PURE AND APPLIED MATHEMATICS A SERIES OF MONOGRAPHS AND TEXTBOOKS

Analytical Methods for Markov Semigroups

Luca Lorenzi Marcello Bertoldi



Analytical Methods for Markov Semigroups

PURE AND APPLIED MATHEMATICS

A Program of Monographs, Textbooks, and Lecture Notes

EXECUTIVE EDITORS

Earl J. Taft
Rutgers University
Piscataway, New Jersey

Zuhair Nashed University of Central Florida Orlando, Florida

EDITORIAL BOARD

M. S. Baouendi Freddy van Oystaeyen

University of California, University of Antwerp,

San Diego Belgium

Jane Cronin Donald Passman

Rutgers University University of Wisconsin,

Jack K. Hale Madison

Georgia Institute of Technology Fred S. Roberts

S. Kobayashi Rutgers University

University of California, David L. Russell

Berkeley Virginia Polytechnic Institute

Marvin Marcus and State University

University of California, Walter Schempp Santa Barbara Universität Siegen

W. S. Massey Mark Teply

Yale University University of Wisconsin,

Anil Nerode Milwaukee

Anii Nerode Milwauke
Cornell University

MONOGRAPHS AND TEXTBOOKS IN PURE AND APPLIED MATHEMATICS

Recent Titles

- J. Haluska, The Mathematical Theory of Tone Systems (2004)
- C. Menini and F. Van Oystaeyen, Abstract Algebra: A Comprehensive Treatment (2004)
- E. Hansen and G. W. Walster, Global Optimization Using Interval Analysis, Second Edition, Revised and Expanded (2004)
- *M. M. Rao*, Measure Theory and Integration, Second Edition, Revised and Expanded (2004)
- W. J. Wickless, A First Graduate Course in Abstract Algebra (2004)
- R. P. Agarwal, M. Bohner, and W-T Li, Nonoscillation and Oscillation Theory for Functional Differential Equations (2004)
- *J. Galambos and I. Simonelli*, Products of Random Variables: Applications to Problems of Physics and to Arithmetical Functions (2004)

Walter Ferrer and Alvaro Rittatore, Actions and Invariants of Algebraic Groups (2005)

Christof Eck, Jiri Jarusek, and Miroslav Krbec, Unilateral Contact Problems: Variational Methods and Existence Theorems (2005)

- M. M. Rao, Conditional Measures and Applications, Second Edition (2005)
- A. B. Kharazishvili, Strange Functions in Real Analysis, Second Edition (2006)

Vincenzo Ancona and Bernard Gaveau, Differential Forms on Singular Varieties: De Rham and Hodge Theory Simplified (2005)

Santiago Alves Tavares, Generation of Multivariate Hermite Interpolating Polynomials (2005)

Sergio Macías, Topics on Continua (2005)

Mircea Sofonea, Weimin Han, and Meir Shillor, Analysis and Approximation of Contact Problems with Adhesion or Damage (2006)

Marwan Moubachir and Jean-Paul Zolésio, Moving Shape Analysis and Control: Applications to Fluid Structure Interactions (2006)

Alfred Geroldinger and Franz Halter-Koch, Non-Unique Factorizations: Algebraic, Combinatorial and Analytic Theory (2006)

Kevin J. Hastings, Introduction to the Mathematics of Operations Research with Mathematica®, Second Edition (2006)

Robert Carlson, A Concrete Introduction to Real Analysis (2006)

John Dauns and Yigiang Zhou, Classes of Modules (2006)

N. K. Govil, H. N. Mhaskar, Ram N. Mohapatra, Zuhair Nashed, and J. Szabados, Frontiers in Interpolation and Approximation (2006)

Luca Lorenzi and Marcello Bertoldi, Analytical Methods for Markov Semigroups (2006)

M. A. Al-Gwaiz and S. A. Elsanousi, Elements of Real Analysis (2006)

Analytical Methods for Markov Semigroups

Luca Lorenzi

University of Parma Parma, Italy

Marcello Bertoldi

Banca Intesa Milano, Italy



Chapman & Hall/CRC Taylor & Francis Group 6000 Broken Sound Parkway NW, Suite 300 Boca Raton, FL 33487-2742

© 2007 by Taylor & Francis Group, LLC Chapman & Hall/CRC is an imprint of Taylor & Francis Group, an Informa business

No claim to original U.S. Government works Printed in the United States of America on acid-free paper 10 9 8 7 6 5 4 3 2 1

International Standard Book Number-10: 1-58488-659-5 (Hardcover) International Standard Book Number-13: 978-1-58488-659-4 (Hardcover)

This book contains information obtained from authentic and highly regarded sources. Reprinted material is quoted with permission, and sources are indicated. A wide variety of references are listed. Reasonable efforts have been made to publish reliable data and information, but the author and the publisher cannot assume responsibility for the validity of all materials or for the consequences of their use.

No part of this book may be reprinted, reproduced, transmitted, or utilized in any form by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying, microfilming, and recording, or in any information storage or retrieval system, without written permission from the publishers.

For permission to photocopy or use material electronically from this work, please access www.copyright.com (http://www.copyright.com/) or contact the Copyright Clearance Center, Inc. (CCC) 222 Rosewood Drive, Danvers, MA 01923, 978-750-8400. CCC is a not-for-profit organization that provides licenses and registration for a variety of users. For organizations that have been granted a photocopy license by the CCC, a separate system of payment has been arranged.

Trademark Notice: Product or corporate names may be trademarks or registered trademarks, and are used only for identification and explanation without intent to infringe.

Visit the Taylor & Francis Web site at http://www.taylorandfrancis.com

and the CRC Press Web site at http://www.crcpress.com

Contents

1	Introduction	xvii
Ι	Markov semigroups in \mathbb{R}^N	1
2	The elliptic equation and the Cauchy problem in $C_b(\mathbb{R}^N)$: the
	uniformly elliptic case	3
	2.0 Introduction	
	2.1 The elliptic equation and the resolvent $R(\lambda)$	7
	2.2 The Cauchy problem and the semigroup	10
	2.3 The weak generator of $T(t)$	21
	2.4 The Markov process	
	2.5 The associated stochastic differential equation \dots	
3	One-dimensional theory	33
	3.0 Introduction	33
	3.1 The homogeneous equation	34
	3.2 The nonhomogeneous equation	
4	Uniqueness results, conservation of probability and n	naxi-
	mum principles	51
	4.0 Introduction	51
	4.1 Conservation of probability and uniqueness	53
	4.1.1 Maximum principles	
	4.1.2 The case when $c \equiv 0 \dots \dots \dots \dots$	62
	4.2 Nonuniqueness	65
5	Properties of $\{T(t)\}$ in spaces of continuous functions	67
	5.0 Introduction	67
	5.1 Compactness of $\{T(t)\}$	68
	5.1.1 The conservative case	69
	5.1.2 The nonconservative case	83
	5.2 On the inclusion $T(t)(C_b(\mathbb{R}^N)) \subset C_0(\mathbb{R}^N)$	85
	5.3 Invariance of $C_0(\mathbb{R}^N)$	91
6	Uniform estimates for the derivatives of $T(t)f$	95
	6.0 Introduction	95
	6.1 Uniform estimates	96

	6.2	Some consequences	113
7		ntwise estimates for the derivatives of $T(t)f$	123
	7.0	Introduction	123
	7.1	The first type of pointwise gradient estimates	124
	7.2	The second type of pointwise gradient estimates	136
	7.3	Further estimates for $A = \Delta + \sum_{j=1}^{N} b_j(x)D_j$	150
8	Inva	riant measures μ and the semigroup in $L^p(\mathbb{R}^N,\mu)$	155
	8.0	Introduction	155
	8.1	Existence, uniqueness and general properties	158
		8.1.1 General properties and uniqueness of the invariant mea-	
		sure of $\{T(t)\}$	158
		8.1.2 Existence by Khas'minskii theorem	168
		8.1.3 Existence by compactness in $C_b(\mathbb{R}^N)$	175
		8.1.4 Existence by symmetry	180
	8.2	Regularity properties of invariant measures	185
		8.2.1 Global L^q -regularity of the density ρ	191
		8.2.2 Global Sobolev regularity	198
	8.3	Some consequences of the estimates of Chapter 6	207
	8.4	The convex case	211
	8.5	Compactness of $T(t)$ and of the embedding $W^{1,p}_{\mu} \subset L^p_{\mu}$	214
	8.6	The Poincaré inequality and the spectral gap \dots	218
	8.7	The logarithmic Sobolev inequality and hypercontractivity .	221
9	The	Ornstein-Uhlenbeck operator	233
	9.0	Introduction	233
	9.1	The formula for $T(t)f$	235
	9.2	Properties of $\{T(t)\}$ in $C_b(\mathbb{R}^N)$	238
	9.3	The invariant measure μ and the semigroup in L^p_{μ}	243
	0.0	9.3.1 The domain of the realization of $\{T(t)\}$ in $L^p_\mu(\mathbb{R}^N)$.	254
		9.3.2 The spectrum of the Ornstein-Uhlenbeck operator in L_{μ}^{r}	
		9.3.3 The sector of analyticity of the Ornstein-Uhlenbeck op-	₁ 20-
		erator in L^p_μ	274
		•	276
	0.4	1 0	
	9.4	The Ornstein-Uhlenbeck operator in $L^p(\mathbb{R}^N)$	279
10		lass of nonanalytic Markov semigroups in $C_b(\mathbb{R}^N)$ and in $\mathbb{R}^N,\mu)$	n 283
	`		283
		Introduction	
	10.1	Nonanalytic semigroups in $C_b(\mathbb{R}^n)$	283
	10.2	Nonanalytic semigroups in $L^p(\mathbb{R}^N, \mu)$	291
TT	M	Parkov semigroups in unhounded open sets	297

11	The	Cauchy-Dirichlet problem	299
	11.0	Introduction	299
	11.1	Two maximum principles	301
		Existence and uniqueness of the classical solution	305
	11.3	Gradient estimates	308
		11.3.1 A priori gradient estimates	309
		11.3.2 An auxiliary problem	315
		11.3.3 Proof of Theorem 11.3.4	324
		11.3.4 A counterexample to the gradient estimates	325
12	The	Cauchy-Neumann problem: the convex case	329
	12.0	Introduction	329
	12.1	Construction of the semigroup and uniform gradient estimates $% \left(1\right) =\left(1\right) \left(1\right)$	332
	12.2	Some consequences of the uniform gradient estimates \dots	344
		Pointwise gradient estimates and their consequences	349
	12.4	The invariant measure of the semigroup	357
13	The	Cauchy-Neumann problem: the nonconvex case	367
	13.0	Introduction	367
	13.1	The case of bounded diffusion coefficients $\dots \dots$	369
		13.1.1 A priori estimate	371
		13.1.2 Existence of the solution to problem (13.0.2) and uni-	
		form gradient estimates	375
		The case when Ω is an exterior domain \dots	388
		Pointwise gradient estimates and their consequences	394
		The invariant measure of the semigroup	402
	13.5	Final remarks	404
II		A class of Markov semigroups in \mathbb{R}^N associated	
	wit.	h degenerate elliptic operators	405
14	The	Cauchy problem in $C_b(\mathbb{R}^N)$	407
	14.0	Introduction	407
	14.1	Some preliminary results on the operator $\mathcal{A}_{\varepsilon}$	409
	14.2	Existence, uniqueness results and uniform estimates	415
		14.2.1 Some preliminary lemmata	416
		proximating semigroups $\{T_{\varepsilon}(t)\}$	419
	140	14.2.3 Construction of the semigroup	431
		Some remarkable properties of the semigroup	439
		The nonhomogeneous elliptic equation	444
IV	7 A	Appendices	465
\mathbf{A}	Basi	ic notions of functional analysis in Banach spaces	467

		Bounded, compact and closed linear operators	467 468 470
	A.4	Some results from interpolation theory	473
В		overview on strongly continuous and analytic semigroups Strongly continuous semigroups B.1.1 On the closure of the sum of generators of strongly con-	477 477
	B.2	tinuous semigroups	480 482
\mathbf{C}	PD	E's and analytic semigroups	487
	C.1	A priori estimates	487
		Classical maximum principles	494
		groups	495
D	Son	ne properties of the distance function	501
E	Fun E.1	ction spaces: definitions and main properties Spaces of functions that are continuous or Hölder continuous	505
		in domains $\Omega \subset \mathbb{R}^N$	505
		E.1.1 Isotropic spaces	505
	.	E.1.2 Anisotropic Hölder spaces in \mathbb{R}^N	506
		Parabolic Hölder spaces	508
	E.3	L^p and Sobolev spaces	509
Re	efere	nces	511
In	\mathbf{dex}		523

Preface

The main objective of this book is to study Markov semigroups, starting from the partial differential equations associated (parabolic and elliptic equations involving a second-order elliptic operator which has unbounded coefficients).

The theory of Markov processes and of related transition semigroups has numerous applications in many fields of science, engineering and economics. Historically, in the mathematical literature the subject is studied using several approaches, with ideas and methods from partial differential equations, Dirichlet forms, stochastic processes, stochastic differential equations, martingale theory.

Somehow the classical semigroup theory is unfit to study this particular class of equations, as well as the classical theory of elliptic differential operators. Indeed, the fact that the coefficients of the operator are unbounded is not merely a technical difficulty, but has significant consequences for the solutions (nonuniqueness of continuous bounded solutions, semigroup not strongly continuous, failure of regularity properties).

These facts lead to looking for specific techniques and results, and here we describe some of them.

The semigroup is studied in spaces of continuous functions and in L^p -spaces of the invariant measure, which is the stationary distribution of the Markov process and it exists under suitable assumptions.

In the first part of the book we study the general properties of the semigroup in spaces of continuous functions: the existence of solutions to the elliptic and to the parabolic equation, the uniqueness properties (and counterexamples to uniqueness), the definition and the properties of the weak generator, which is a specific notion that substitutes the infinitesimal generator of strongly continuous semigroups.

We see also some properties of the associated Markov process and the connection with the uniqueness of the solutions.

Then we focus on the proof of regularity results: global and pointwise estimates of the space derivatives of the semigroup. In particular, first we prove global estimates for the space derivatives and Schauder estimates, similar to those for operators with bounded coefficients. Then, we prove a number of pointwise estimates, which relate the derivatives of the semigroup at a point with the semigroup applied to the derivatives of the function, or to the function itself, at the same point. These estimates are truly characteristic of this class of operators and have interesting consequences (for instance in terms of

Liouville theorems).

We make the same analysis for boundary value problems in unbounded domains, with Dirichlet or Neumann boundary conditions, and for problems involving degenerate operators.

In the part about the invariant measure, we study some different approaches to the problem of the existence of the invariant measure, and we study the properties of the semigroup in L^p -spaces, including the asymptotic behaviour, the Poincaré inequality and the log-Sobolev inequality.

Besides we devote a chapter to the Ornstein–Uhlenbeck semigroup, the most studied example of an operator with unbounded coefficients.

Acknowledgments

The authors acknowledge the financial support of the research projects "Equazioni di evoluzione deterministiche e stocastiche" and "Equazioni di Kolmogorov" of the Italian Ministero dell'Istruzione, dell'Università e della Ricerca (M.I.U.R.), and the financial support of the European Community's Human Potential Programme under contract HPRN-CT-2002-00281 "Evolution Equations". They also wish to thank S. Fornaro, A. Lunardi, G. Metafune, D. Pallara, A. Rhandi, G. Tessitore and G. Tubaro for useful discussions and suggestions during the preparation of this book.

The authors express their gratitude also to K. Payne for many useful suggestions concerning English language.

M. Bertoldi wishes to thank Ph. Clément and the members of the Department of Applied Mathematics of Delft University of Technology for the warm hospitality.

L. Lorenzi wishes to thank the Department of Mathematics of the University of Milan who allowed him usage of its rich library during the preparation of this book.

Finally, the authors thank the staff of Taylor and Francis for their help and useful suggestions they supplied during the preparation of this book.

Marcello Bertoldi Luca Lorenzi

Symbol Descriptions

	Sets		$\mathbb{R}^N \to \mathbb{R}^N$, i.e., div $f =$
\mathbb{N}	set of all positive natural	$ f _{\infty}$	$\sum_{i=1}^{N} D_i f$ the sup-norm of $f: \Omega \subset$
\mathbb{Z}	numbers set of all relative integers	$ J \infty$	$\mathbb{R}^N \to \mathbb{R}$, i.e., $ f _{\infty} :=$
\mathbb{R}	set of all real numbers		$\sup_{\Omega} f $ (whenever is fi-
\mathbb{C}	set of all complex numbers	et (e)	nite)
\mathbb{R}^N	set of all real N-tuples	f' (or f_x)	the first-order derivative of a function f of one
\mathbb{C}^N	set of all complex N-tuples		variable. We use a similar
$A \subset\subset B$	given two subsets $A, B \subset \mathbb{R}^N$ with B open, it means		notation for higher order
	that \overline{A} is contained in B	e(k)	derivatives
B(R)	open disk in \mathbb{R}^N with cen-	$f^{(k)}$	the k -th-order derivative of a function f of one vari-
$\overline{D}(D)$	tre at 0 and radius $R > 0$		able
$\overline{B}(R)$ x + B(R)	the closure of $B(R)$ open disk in \mathbb{R}^N with cen-	$D_{i_1,,i_r}f$	the derivative $\frac{\partial^r f}{\partial x_{i_1},,\partial x_{i_r}}$
w D (10)	tre at x and radius $R > 0$		of the function $f: \mathbb{R}^N \to$
\overline{F}	the closure of F		\mathbb{R} , where $i_1, \ldots, i_r \in$
∂F E + F	the boundary of F given a space X and two	Do t	$\{1,\ldots,N\}$ $\partial^{ \alpha }f$
E + F	subsets of $E, F \subset X$ it de-	$D^{\alpha}f$	the derivative $\frac{\partial^{ \alpha } f}{\partial x^{\alpha_1}, \dots, \partial x^{\alpha_N}}$ of the function f , where
	notes the set of $x \in X$		$\alpha = (\alpha_1, \dots, \alpha_N)$
	such that $x = e + f$ for	$D^k f$	for $k \in \mathbb{N}$, it denotes the
	some $e \in E$ and some $f \in F$		vector consisting of all the
$E \oplus F$	it denotes the set $E +$	$ D^k f _{\infty}$	k -th order derivatives of f for $k \in \mathbb{N}$ it denotes
•	F when E and F are	<i>D</i>	the sup-norm of the
	two subspaces of a vector		vector-valued function
L(X,Y)	space X and $E \cap F = \{0\}$ the set of all the bounded		$D^k f$, namely $ D^k f _{\infty}^2 =$
$L(\Lambda, I)$	linear operators from X to	χ_A	$\sum_{ \alpha =k } \ D^{\alpha} f\ _{\infty}^{2}$ the characteristic function
	Y	ΛA	of the set A , i.e., the func-
L(X)	:= L(X, X)		tion defined by $\chi_A(x) =$
X'	the dual space of the Banach space X , i.e., the sets		1 for any $x \in \mathbb{R}^N$ and
	of bounded linear func-	1	$\chi_A(x) = 0$ for any $x \notin A$ the characteristic function
	tionals from X to \mathbb{R} (\mathbb{C})		of \mathbb{R}^N
	Functions	$\operatorname{Jac} f$	the Jacobian matrix of
S			the vector-valued function $f: \Omega \subset \mathbb{R}^N \to \mathbb{R}^m$
δ_x	the delta function, i.e., $\delta_x(x) = 1$, $\delta_x(y) = 0$ if		$(m, N \in \mathbb{N})$, i.e., the ma-
	$y \neq x$		trix whose j -th line is the
$\operatorname{div} f$	the divergence of $f:\Omega\subset$		vector $(D_1f_j,\ldots,D_Nf_j),$

$\operatorname{supp} f$ $g \circ f$ $\ Q\ _{\infty}$ $\ DQ\ _{\infty}$	where f_j is the j -th component of f the support of the function $f:\Omega\subset\mathbb{R}^N\to\mathbb{R}$ given two functions $f:X\to Y$ and $g:Y\to Z$, it denotes the function defined by $(g\circ f)(x)=g(f(x))$ for any $x\in X$ the Euclidean norm of the matrix $Q=(q_{ij})$ belonging to $L(\mathbb{R}^N)$, i.e., $\ Q\ _\infty^2=\sum_{i,j=1}^Nq_{ij}^2$. If the entries of Q depend on $x\in\Omega$, we use the same notation to denote the norm $\ Q\ _\infty^2=\sum_{i,j=1}^N\ q_{ij}\ _\infty^2$:= $\sup_\Omega\sum_{i,j,h=1}^N\ D_hq_{ij}\ ^2$ for any matrix Q whose entries are continuously differentiable in open set Ω with bounded deriva-	e_j I_r $\lambda_{\min}(A)$ $\lambda_{\max}(A)$ $\mathrm{rank}A$ $\mathrm{span}(E)$ $\mathrm{Tr}B$ $x\otimes x$	nal are $\lambda_1, \ldots, \lambda_N$ the j -th vector of the canonical basis of \mathbb{R}^N the identity matrix with r rows and r columns. When there is no damage of confusion we simply write I the minimum eigenvalue of the matrix A the maximum eigenvalue of the matrix A the maximum number of columns linearly independent of the matrix A given a subset E of a vector space V , it denotes the set of finite linear combinations of elements of E the trace of the $N \times N$ matrix B , i.e., $\text{Tr}(B) = \sum_{i=1}^{N} b_{ii}$ the $N \times N$ -matrix whose
	tives Operators	$\langle x,y \rangle$	entries are $(x \otimes x)_{ij} = x_i x_j$ for any $i, j = 1,, N$ the Euclidean inner prod-
D(A)	the domain of the (linear) operator A		uct between the vectors $x, y \in \mathbb{R}^N$
$\rho(A)$	resolvent set of a linear operator A	$x \cdot y$	$:= \langle x, y \rangle$ Miscellanea
$\sigma(A)$	spectrum of a linear operator A	a^+	the positive part of $a \in$
I $[A, B]$	(in a Banach space X) the identity operator the commutator between the operators A and B ,	a^{-}	\mathbb{R} , i.e., the maximum between a and 0 the negative part of $a \in \mathbb{R}$, i.e., the minimum between
	i.e., $[A, B] = AB - BA$ defined on $D(AB) \cap D(BA)$	$a \vee b$	a and 0 the maximum between a and b
Matr	ix and linear algebra	$a \wedge b$	the minimum between a
$\det B$	the determinant of the matrix B	$ \alpha $	and b the length of the multi- index α , i.e., $ \alpha = \alpha_1 +$
$\operatorname{diag}(\lambda_1,$	$(, \lambda_N)$ the diagonal matrix whose entries on the main diago-	$\alpha!$	$\ldots + \alpha_N$ the factorial of the multi-

$\deg p$	index α , i.e., $\alpha! = \prod_{i=1}^{N} \alpha_i!$ the degree of the polyno-	$\operatorname{dist}(F,G)$) the distance of the set F from the set G , i.e., the
	$\operatorname{mial} p:\mathbb{R}^N o \mathbb{R}$		number $\inf_{x \in F} d(x, G)$
δ_{ij}	the Kronecker delta, i.e.,	dx	the Lebesgue measure in
J	$\delta_{ij} = 1 \text{ if } i = j \text{ and } \delta_{ij} = 0$		$\mathbb{R}^N \ (N \ge 1)$
	otherwise	$\operatorname{Re}\lambda$	the real part of the com-
dist(x, F)) the distance of the point		plex number λ
	x from the set x , i.e.,	$\operatorname{Im} \lambda$	the imaginary part of the
	$\operatorname{dist}(x, F) = \inf_{y \in F} x - y $		complex number λ

Chapter 1

Introduction

Elliptic operators with bounded coefficients have been widely studied in the literature both in \mathbb{R}^N and in open subsets of \mathbb{R}^N , starting from the 1950's, and nowadays they are well understood.

The study of operators with unbounded coefficients is a generalization of this case and recently the interest in this class of operators has grown considerably due to their numerous applications in many fields of science, engineering and economics. The most famous example of an elliptic operator with unbounded coefficients in \mathbb{R}^N $(N \geq 1)$ is the Ornstein-Uhlenbeck operator, defined on smooth functions by

$$\mathcal{A}\varphi(x) = \frac{1}{2} \sum_{i,j=1}^{N} q_{ij} D_{ij} \varphi(x) + \sum_{i,j=1}^{N} b_{ij} x_j D_i \varphi(x), \qquad x \in \mathbb{R}^N,$$

where (q_{ij}) is a constant strictly positive definite matrix and (b_{ij}) is a constant real matrix whose eigenvalues have nonpositive real parts. Such an operator displays all the main peculiarities of the operators with unbounded coefficients. For instance, the associated semigroup $\{T(t)\}$ in $BUC(\mathbb{R}^N)$ is neither strongly continuous nor analytic, whereas the semigroups associated with operators with bounded coefficients are analytic in $C_b(\mathbb{R}^N)$. Nevertheless, it has smoothing effects similar to those enjoyed by analytic semigroups associated with uniformly elliptic operators with bounded coefficients.

Starting from the pioneering papers [10, 76, 90, 91, 92, 93] the literature on elliptic operators with unbounded coefficients has spread out considerably and now we are able to treat uniformly elliptic operators of the type

$$\mathcal{A}\varphi(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}\varphi(x) + \sum_{i=1}^{N} b_i(x)D_i\varphi(x) + c(x)\varphi(x), \quad (1.0.1)$$

under rather weak assumptions on the coefficients, both with analytic and probabilistic methods.

The aim of this book is to present most of the old and recent results on the Markov semigroups associated with elliptic operators with unbounded coefficients, using analytic methods. We mainly consider the case when the coefficients of the operator \mathcal{A} are defined in \mathbb{R}^N , but we also give some very recent results in the case when they are defined in open and unbounded domains of \mathbb{R}^N . The book is divided in three parts.

Part I

In the first part of the book, we are concerned with the following topics:

Existence of the Markov semigroup associated in $C_b(\mathbb{R}^N)$ with the operator \mathcal{A} in (1.0.1). In the case when the coefficients of the operator \mathcal{A} are bounded, the natural way to construct analytically such a semigroup consists in defining, for any $f \in C_b(\mathbb{R}^N)$ (the space of bounded and continuous functions defined in \mathbb{R}^N) and any t > 0, T(t)f as the value at t of the classical solution to the Cauchy problem

$$\begin{cases} D_t u(t,x) - \mathcal{A} u(t,x) = 0, & t > 0, \ x \in \mathbb{R}^N, \\ u(0,x) = f(x), & x \in \mathbb{R}^N. \end{cases}$$
 (1.0.2)

To generalize this procedure to the case when the coefficients are unbounded, we are led to prove existence results for the bounded classical solution of the Cauchy problem (1.0.2), when \mathcal{A} is given by (1.0.1). Here, by bounded classical solution of (1.0.2), we mean a function u which is bounded and continuous in $[0,+\infty)\times\mathbb{R}^N$ and admits first-order time derivative and firstand second-order spatial derivatives, which are continuous in $(0, +\infty) \times \mathbb{R}^N$. This has been done, first, by S. Itô in [76] and, more recently, by R. Azencott in [10], under quite minimal regularity assumptions on the coefficients of the operator \mathcal{A} . The arguments used to prove the existence of a classical solution (which we denote by u_f) are very simple and elegant. They are based both on an approximation argument with Cauchy-Dirichlet problems in bounded and smooth domains, and classical Schauder estimates. In general, u_f is not the unique classical solution to the problem (1.0.2). This is a typical feature of elliptic operators with unbounded coefficients. Nevertheless, u_f enjoys a nice property: when $f \geq 0$, u_f is the minimal positive solution to the problem (1.0.2). This minimality property allows us to define the semigroup $\{T(t)\}$ by setting $T(t)f = u_f(t,\cdot)$ for any $t \geq 0$.

In general, such a semigroup is neither strongly continuous nor analytic in $C_b(\mathbb{R}^N)$. In fact, T(t)f converges to f as t goes to 0, uniformly on compact subsets, but, in general, not uniformly in \mathbb{R}^N , and this happens even if f is uniformly continuous. Some sufficient conditions on the coefficients of the operator \mathcal{A} which imply that the semigroup $\{T(t)\}$ is (resp. is not) analytic in $C_b(\mathbb{R}^N)$ are known in the literature and here we describe some of them.

There is also a natural, purely probabilistic way to introduce the semigroup $\{T(t)\}$. It is known that, under suitable assumptions on the coefficients of the operator \mathcal{A} , there exists a Markov process X in \mathbb{R}^N such that the semigroup $\{T(t)\}$ is the transition semigroup of X, i.e.,

$$(T(t)f)(x) = \mathbb{E}^x \mathbf{1}_{t < \tau} f(X_t), \qquad t > 0, \ x \in \mathbb{R}^N, \ f \in C_b(\mathbb{R}^N),$$
 (1.0.3)

where $\tau \in (0, +\infty]$ is the life time of X (X_t is defined for $t \in [0, \tau)$), and \mathbb{E}^x is the expectation under the probability measure \mathbb{P}_x . Note that $X_0 = x$

xix

 \mathbb{P}_x -almost surely. The process X associated with $\{T(t)\}$ is equivalent under the probability \mathbb{P}_x to the solution ξ_t^x of the stochastic differential equation

$$d\xi_t^x = b(\xi_t^x)dt + \sigma(\xi_t^x)dW_t, \qquad \xi_0^x = x, \tag{1.0.4}$$

where W_t is a N-dimensional Brownian motion, $b: \mathbb{R}^N \to \mathbb{R}^N$ is the coefficient of \mathcal{A} and $\sigma: \mathbb{R}^N \to L(\mathbb{R}^N)$ is such that $Q(x) = \frac{1}{2}\sigma(x)\sigma^*(x)$, where Q(x) is the matrix with elements $q_{ij}(x)$.

Conversely, the stochastic equation (1.0.4) can be the starting point for the construction of $\{T(t)\}$: if ξ_t^x is a solution of the equation (1.0.4) and $\{T(t)\}$ is defined by (1.0.3) with $X_t = \xi_t^x$, then the function $u = T(\cdot)f$ is a solution of the parabolic problem (1.0.2). This follows using results from the stochastic calculus (the Itô formula is the main tool).

The equation (1.0.4) can be considered a random perturbation of the ordinary differential equation

$$\frac{d}{dt}\xi_t^x = b(\xi_t^x). \tag{1.0.5}$$

Of course, here it is not natural to assume that the function b is bounded; in almost all the significant cases b is unbounded. This is an important reason for studying operators with unbounded coefficients.

Note that, according to the formula (1.0.3), the properties of $\{T(t)\}$ may be deduced from the study of the process X. This probabilistic approach has been widely used in the literature (see, e.g., [30, 49, 53, 65, 75, 83, 88, 139]).

Under the same smoothness assumptions on the coefficients as above and using similar approximation arguments by Dirichlet problems in bounded domains, the existence of a solution $v_f \in D_{\text{max}}(\mathcal{A})$ to the elliptic equation

$$\lambda v - \mathcal{A}v = f, \quad \lambda > c_0 := \sup_{x \in \mathbb{R}^N} c(x),$$
 (1.0.6)

can be proved for any $f \in C_b(\mathbb{R}^N)$. Here,

$$D_{\max}(\mathcal{A}) = \left\{ u \in C_b(\mathbb{R}^N) \cap \bigcap_{1 \le p \le +\infty} W_{\text{loc}}^{2,p}(\mathbb{R}^N) : \mathcal{A}u \in C_b(\mathbb{R}^N) \right\}. \quad (1.0.7)$$

As in the parabolic case, the problem (1.0.6) may admit several solutions in $D_{\text{max}}(\mathcal{A})$. In any case, when $f \geq 0$, v_f is the minimal positive solution.

As in the case when the coefficients of the operator \mathcal{A} are bounded, there is a connection between the functions u_f and v_f . In fact, for any $\lambda > c_0$, v_f is the Laplace transform of u_f , in the sense that

$$v_f(x) = \int_0^{+\infty} e^{-\lambda t} u_f(t, x) dt, \qquad x \in \mathbb{R}^N.$$
 (1.0.8)

The main difference with the classical case is that now, in general, the integral term in the right-hand side of (1.0.8) does not converge in $C_b(\mathbb{R}^N)$.

Since, in general, the semigroup $\{T(t)\}$ is neither strongly continuous nor analytic, then the infinitesimal generator does not exist in the classical sense. This gap is filled introducing the concept of a "weak generator". In fact, we can define a family $\{R(\lambda), \ \lambda > c_0\}$ of bounded operators in $C_b(\mathbb{R}^N)$ by setting $R(\lambda)f = v_f$. Such a family is the resolvent family associated with some (closed) operator \widehat{A} . We call \widehat{A} the "weak generator" of the semigroup. The reason for this name is due to the fact that \widehat{A} is a generalization of the classical concept of the infinitesimal generator of a strongly continuous (or analytic) semigroup. Indeed, the weak generator \widehat{A} can be defined, in a equivalent way, using the bounded pointwise convergence: a function f belongs to $D(\widehat{A})$ if and only if

 $\frac{|(T(t)f)(x)-f(x)|}{t} \leq M, \qquad t \in (0,1), \ x \in \mathbb{R}^N,$

for some M > 0, and (T(t)f - f)/t converges pointwise to some bounded and continuous function g as t tends to 0^+ . In such a case $\widehat{A}f = g$.

In general, $D(\widehat{A})$ is properly contained in $D_{\max}(A)$ and \widehat{A} coincides with A on $D(\widehat{A})$. The previous set inclusion is actually a set equality if and only if the elliptic equation (1.0.6) is uniquely solvable in $D_{\max}(A)$.

As in the classical case of bounded coefficients, it is possible to associate a transition family $\{p(t, x; dy) : t > 0, x \in \mathbb{R}^N\}$ (or equivalently a Green's function G) to the semigroup, for which the following representations hold:

$$(T(t)f)(x) = \int_{\mathbb{R}^N} f(y)p(t, x; dy) = \int_{\mathbb{R}^N} G(t, x, y)f(y)dy, \qquad t > 0, \ x \in \mathbb{R}^N.$$
(1.0.9)

The formula (1.0.9) is a keystone for proving some interesting and very useful continuity properties of the semigroup, as well as for showing that the semigroup can be extended to the space $B_b(\mathbb{R}^N)$ of bounded Borel functions. Such a semigroup, still denoted by $\{T(t)\}$, is both irreducible and strong Feller.

All the previous results can be obtained without assuming the uniqueness of the solutions to the problems (1.0.2) and (1.0.6), with the claimed regularity properties. Of course, it is natural to investigate what conditions imply the uniqueness of the classical solution to the Cauchy-Dirichlet problem and of the solution to the elliptic equation in $D_{\text{max}}(\mathcal{A})$. The two problems are not independent of one another. In fact, there exists a unique solution to the elliptic equation in $D_{\text{max}}(\mathcal{A})$ if and only if there exists a unique classical solution to the problem (1.0.2), which is bounded in $[0,T] \times \mathbb{R}^N$ for any T > 0.

In the case when the coefficients are bounded, the uniqueness results are straightforward consequences of the classical maximum principles. If the coefficients of the operator $\mathcal A$ are unbounded, the classical maximum principle may fail. This is the reason why, in general, the elliptic equation and the parabolic Cauchy-Dirichlet problem admit more than one solution. Hence, to prove uniqueness results some additional assumptions on the operator $\mathcal A$ need

to be imposed. The typical assumptions which we assume are those that allow us to prove a generalized maximum principle: the existence of a so-called Lyapunov function φ , i.e., a sufficiently smooth function φ such that

(i)
$$\lim_{|x| \to +\infty} \varphi(x) = +\infty$$
 and (ii) $\sup_{x \in \mathbb{R}^N} \mathcal{A}\varphi(x) - \lambda \varphi(x) < +\infty$, (1.0.10)

for some $\lambda > c_0$. At a first glance, the condition (1.0.10)(ii) may seem a condition related merely to the growth rate of the coefficients at infinity, but actually this is not the case. Indeed, the differential operators

$$\mathcal{A}\varphi(x) = \Delta\varphi(x) - |x|\langle x, D\varphi(x)\rangle, \qquad x \in \mathbb{R}^N$$
 (1.0.11)

and

$$\mathcal{A}\varphi(x) = \Delta\varphi(x) + |x|\langle x, D\varphi(x)\rangle, \qquad x \in \mathbb{R}^N,$$
 (1.0.12)

differ just in the sign of the drift term, but this difference is essential. Indeed, the Cauchy-Dirichlet problem associated with the first operator admits a unique classical solution, while the Cauchy problem associated with the second operator admits several classical solutions.

In the case when $c \equiv 0$, both the functions 1 (i.e., the constant function identically equal to 1) and $T(\cdot)$ 1 solve the Cauchy problem (1.0.2). Hence, T(t)1 \equiv 1 for any t > 0 is a necessary condition for the problem (1.0.2) to admit a unique classical solution. Actually, such a condition is also sufficient. When this condition is satisfied we say that the semigroup is conservative or that the conservation of the probability holds. The reason for this nomenclature is based on the fact that, in this case, the family of measures $\{p(t,x;dy): t>0, x\in\mathbb{R}^N\}$ consists of probability measures.

There are also some conditions under which the problem (1.0.2) admits several classical solutions. For instance, this is the case when there exists a nonnegative bounded and smooth function φ (still called a Lyapunov function) such that $\lambda \varphi - \mathcal{A} \varphi \leq 0$.

The one-dimensional case is easier and has been studied by Feller (see [57]). In such a setting there are necessary and sufficient conditions which give uniqueness of the solution to the elliptic (and, consequently, to the parabolic problem) which can be written in terms of the integrability at infinity of some functions which are strictly related to the coefficients of the operator \mathcal{A} .

The arguments that we have briefly described here are discussed in detail in Chapters 2, 3, 4 and 10.

Study of the main properties of the semigroup $\{T(t)\}$ in the space $C_b(\mathbb{R}^N)$. We discuss two main topics in this book. The first one is the compactness of the semigroup in $C_b(\mathbb{R}^N)$ which has been studied mainly by E.B. Davies in [44] and by G. Metafune, D. Pallara and M. Wacker in [115], in the case when $c \equiv 0$.

In the conservative case, the semigroup is compact if and only if, for any t > 0, the family of measures $\{p(t, x; dy) : x \in \mathbb{R}^N\}$ is tight. This means

that for any t>0 the measures p(t,x;dy) are ε -concentrated in a compact set, uniformly with respect to $x\in\mathbb{R}^N$. This, for instance, is the case when there exist a nonnegative function $\varphi\in C^2(\mathbb{R}^N)$ and a convex function $g\in C^1([0,+\infty))$ such that

$$\lim_{|x|\to\infty}\varphi(x)=+\infty, \quad 1/g \text{ is integrable at } +\infty, \quad \mathcal{A}\varphi(x)\leq -g(\varphi(x)), \quad x\in\mathbb{R}^N.$$

In the case when the family of measures $\{p(t, x; dy), x \in \mathbb{R}^N\}$ is tight for any t > 0, the semigroup, besides being compact, enjoys another important property: it is norm continuous in $(0, +\infty)$. This means that the function $t \mapsto T(t)$ is continuous with respect to the operator topology at any point t > 0. Of course, one cannot expect continuity up to t = 0 since the semigroup is not, in general, strongly continuous.

In the nonconservative case, the compactness of the semigroup can be read, roughly speaking, in terms of some regularity properties of the function $T(t)\mathbf{1}$. Indeed, $T(t)\mathbf{1}$ belongs to $C_0(\mathbb{R}^N)$ (the space of continuous functions defined in \mathbb{R}^N which vanish at infinity), for any t > 0, if and only if the semigroup is compact and maps $C_0(\mathbb{R}^N)$ into itself. In such a case, as in the conservative one, the semigroup is norm continuous and $R(\lambda, \widehat{A})$ is compact as well.

Even though the semigroup $\{T(t)\}$ may fail to be strongly continuous in $C_b(\mathbb{R}^N)$, the function $t \mapsto T(t)f$ is continuous up to t = 0, with respect to the sup-norm, for any $f \in C_0(\mathbb{R}^N)$. Therefore, if the semigroup maps $C_0(\mathbb{R}^N)$ into itself, then its restriction to $C_0(\mathbb{R}^N)$ gives rise to a strongly continuous semigroup. So, it is important to determine suitable conditions on the coefficients of the operator \mathcal{A} which imply that $C_0(\mathbb{R}^N)$ is invariant under the action of the semigroup. We do this in Sections 5.2 and 5.3.

Note that, in general, $\{T(t)\}$ does not map $C_0(\mathbb{R}^N)$ into itself. This is the case, for instance, when the operator \mathcal{A} is defined by

$$\mathcal{A}\varphi(x) = \Delta\varphi(x) - |x|^2 \langle x, D\varphi(x) \rangle, \qquad x \in \mathbb{R}^N,$$

on smooth functions φ .

All the arguments that we have described here are discussed in Chapter 5.

Study of the smoothing effects of the semigroup. We pay attention to the estimates of the space derivatives of T(t)f when $f \in C_b(\mathbb{R}^N)$ or it is smooth. These estimates can be split into two main families: uniform (in the space variables) and pointwise estimates. Starting from the pioneering papers on the Ornstein-Uhlenbeck semigroup it has become clear that under suitable assumptions on the growth rate at infinity of the coefficients of the operator \mathcal{A} , the space derivatives of T(t)f should behave as in the case of the bounded coefficients. Therefore, under reasonable assumptions on the coefficients, one can expect that

$$||D^k T(t)f||_{\infty} \le Ct^{-\frac{k-h}{2}} e^{\omega t} ||f||_{C_t^h(\mathbb{R}^N)}, \qquad t > 0,$$
 (1.0.13)

for any $0 \le h \le k$, any $f \in C_h^h(\mathbb{R}^N)$ and some positive constants C and ω .

We focus our attention on the case when $k \leq 3$. Indeed, in this case, the estimates (1.0.13) are a fundamental tool in order to prove existence results and Schauder estimates for the nonhomogeneous Cauchy problem and the nonhomogeneous elliptic equation associated with the operator \mathcal{A} .

The estimates (1.0.13) were first proved by A. Lunardi in [108] with analytic methods and by S. Cerrai in [29, 30] with probabilistic methods. Recently they have been improved in [18] in the case when $c \equiv 0$. Here, we describe the results of [18]. The choice of focusing attention on the case when $c \equiv 0$ is due to the fact that such an assumption is necessary for the validity of some of the pointwise estimates that we present.

The method used to prove (1.0.13) is very easy and at the same time very elegant. It consists in adapting to our situation the classical Bernstein method. Unfortunately, to make such a method work, one needs to assume stronger assumptions than those needed to construct the semigroup. In particular, some dissipativity conditions are needed. More precisely, some bounds from above on the sum

$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \quad x, \xi \in \mathbb{R}^N$$

are essential to prove (1.0.13). Indeed, without any dissipativity assumption, the estimates (1.0.13) may fail to hold even in the one-dimensional case. Moreover, here we have to assume the uniqueness of the solution to the homogeneous problem (1.0.2).

As it has been claimed, the previous estimates allow us to prove optimal Schauder estimates for both the nonhomogeneous Cauchy problem and the nonhomogeneous elliptic equation associated with the operator \mathcal{A} . This is obtained by an interpolation argument under the same assumptions on f and g as in the case of bounded coefficients. As in this latter case, the solution to the nonhomogeneous Cauchy problem

$$\begin{cases} D_t u(t,x) - \mathcal{A}u(t,x) = g(t,x), & t > 0, \ x \in \mathbb{R}^N, \\ u(0,x) = f(x), & x \in \mathbb{R}^N, \end{cases}$$

can be represented by means of the usual variation-of-constants formula

$$u(t,x) = (T(t)f)(x) + \int_0^t (T(t-s)g(t,\cdot))(x)ds, \qquad t \ge 0, \ x \in \mathbb{R}^N, \ (1.0.14)$$

where now the integral term, in general, does not converge in $C_b(\mathbb{R}^N)$.

The optimal Schauder estimates may also be used to give a partial characterization of $D(\widehat{A})$, showing that $D(\widehat{A}) \subset C_b^{1+\alpha}(\mathbb{R}^N)$ for any $\alpha \in (0,1)$.

The latter type of estimates that we discuss in this book are pointwise estimates. There are two types of pointwise estimates. The first ones are of

the following type:

$$|(D^k T(t)f)(x)|^p \le M_{k,p} \left(T(t)((|f|^2 + \dots + |D^k f|^2)^{\frac{p}{2}}) \right)(x),$$
 (1.0.15)

for any t > 0, any $x \in \mathbb{R}^N$, any $f \in C_b^k(\mathbb{R}^N)$ (k = 1, 2, 3), any $p \in (1, +\infty)$ and some positive constant $M_{k,p}$. They can be proved under the same hypotheses used to get the uniform estimates. Under somewhat heavier assumptions on the coefficients of the operator \mathcal{A} , the estimates (1.0.15) can be improved, eliminating the dependence on f^2 from its right-hand side. In particular, it can be shown that

$$|DT(t)f|^p \le e^{\sigma_p t} T(t)(|Df|^p), \qquad t > 0,$$
 (1.0.16)

for any p > 1. Note that for (1.0.16) to hold it is necessary that $T(t)1 \equiv 1$ for any t > 0 and, hence, $c \equiv 0$. Such more restrictive assumptions cover, for instance, some cases when the coefficients have polynomial growth rate at infinity.

The second type of estimate allows us to give an upper bound of the left-hand side of (1.0.15) for any t > 0 and any $x \in \mathbb{R}^N$ in terms of

$$\psi(t) \left(T(t)((|f|^2 + \ldots + |D^{k-1}f|^2)^{\frac{p}{2}}) \right) (x),$$

 ψ being a positive function which behaves like $t^{-kp/2}$ near t=0 and it is bounded or decreases exponentially at infinity. Then, iterating the arguments and taking the semigroup property into account, we can derive the estimates

$$|(D^k T(t)f)(x)|^p \le C_{k,p} t^{-\frac{pk}{2}} \psi_{k,p}(t) (T(t)(|f|^p))(x), \qquad t > 0, \ x \in \mathbb{R}^N,$$
(1.0.17)

for any k = 1, 2, 3, where the function $\psi_{k,p}$ is bounded at 0 and it behaves like $t^{\frac{pk}{2}}$ or it decreases exponentially to 0 at $+\infty$. This latter estimate improves the uniform estimates (1.0.13), since now $\omega_{k,p}$ may be a negative constant. Consequently, the estimates (1.0.17) allow for a better asymptotic analysis of the semigroup at infinity.

In general, the estimates (1.0.15) cannot be extended to the case p=1, as a well-known counterexample by Wang shows. However, in the particular case when $\mathcal{A} = \Delta + \sum_{i=1}^{N} b_i(\cdot)D_i$ and the b_i 's satisfy suitable growth conditions at infinity, they can be extended also to the case p=1. The importance of obtaining such estimates with p=1 will be clarified later on in this introduction. On the other hand, we stress that the estimates (1.0.17) with p=1 may fail also in the case when the coefficients of the operator $\mathcal A$ are bounded. A very easy counterexample is given by the Gauss-Weierstrass semigroup.

Liouville type theorems are an important consequence of the previous gradient estimates. Indeed, in the case when $\omega_{1,p} < 0$ in (1.0.17), one can show that if $u \in D_{\text{max}}(\mathcal{A})$ satisfies $\mathcal{A}u = 0$, then u is constant. But this is not the only application of these estimates. Other important applications will be seen in the next paragraph.

All the arguments discussed here are contained in Chapters 6 and 7.

Introduction xxv

Study of both the invariant measures associated with the semigroup and the regularity properties of the semigroup in L^p -spaces related to this measure. By definition, an invariant measure is a probability measure μ such that

$$\int_{\mathbb{R}^N} T(t) f \, d\mu = \int_{\mathbb{R}^N} f \, d\mu, \qquad t > 0, \tag{1.0.18}$$

for any $f \in B_b(\mathbb{R}^N)$.

The spaces $L^p(\mathbb{R}^N, \mu)$ are the "right" L^p -spaces for $\{T(t)\}$. Indeed, if an invariant measure exists, then $\{T(t)\}$ maps $L^p(\mathbb{R}^N, \mu)$ into itself. Note that, in general, this property fails for the L^p -spaces related to other measures, even in the case of the Lebesgue measure. In such L^p -spaces, $\{T(t)\}$ is a strongly continuous semigroup for any $p \in [1, +\infty)$. Note that, if $c \equiv 0$ and an invariant measure exists, then the semigroup is conservative. So, from the probabilistic point of view, an invariant measure is a stationary distribution for the Markov process X.

When it exists, the invariant measure of $\{T(t)\}$ is unique and is absolutely continuous with respect to the Lebesgue measure. Moreover, its density is a positive and continuous function (not necessarily bounded). In general, an explicit expression of the density of μ is not available. In any case, in some situations (e.g., in the case when \mathcal{A} is in divergence form), under suitable integrability and smoothness assumptions on the coefficients, one can prove some global L^p and Sobolev regularity properties of the density.

So, the main problem consists in determining suitable conditions ensuring the existence of the invariant measure. The main result in this direction is the Khas'minskii theorem. It states that if there exists a regular function φ such that

$$\varphi \ge 0, \quad \lim_{|x| \to +\infty} \mathcal{A}\varphi(x) = -\infty,$$

then the invariant measure of $\{T(t)\}$ exists.

Let L_p be the infinitesimal generator of $\{T(t)\}$ in $L^p(\mathbb{R}^N, \mu)$. As far as we know, there are only a few situations in which a complete characterization of $D(L_p)$ is available. This is, for instance, the case of the Ornstein-Uhlenbeck semigroup. In any case, one can prove that $D_{\max}(\mathcal{A})$ is always a core of $D(L_p)$. This means that $D_{\max}(\mathcal{A})$ is dense in $D(L_p)$ which is endowed with the graph norm. Moreover, if the pointwise estimate (1.0.17) holds, one can partially characterize the domain of L_p , showing that it is continuously embedded in $W^{1,p}(\mathbb{R}^N,\mu)$ (the set of all the functions whose first-order distributional derivatives are in $L^p(\mathbb{R}^N,\mu)$).

In the case when p=2, a description of the asymptotic behaviour of the semigroup is available. In fact, for any $f \in L^2(\mathbb{R}^N, \mu)$ we have

$$\lim_{t \to +\infty} \int_{\mathbb{R}^N} (T(t)f - \overline{f})^2 d\mu = 0, \tag{1.0.19}$$

where \overline{f} is the projection of f on the subspace of the constant functions. This subspace is the eigenspace corresponding to the eigenvalue $\lambda = 0$ of the operator \mathcal{A} .

If there exists a *spectral gap*, i.e., if

$$\sigma(L_2) \setminus \{0\} \subset \{\text{Re } \lambda \le -\delta\},$$
 (1.0.20)

for some $\delta > 0$, the convergence in (1.0.19) is of exponential type. This is the case, for instance, when the Poincaré inequality

$$\int_{\mathbb{R}^N} |f - \overline{f}|^2 d\mu \le C \int_{\mathbb{R}^N} |Df|^2 d\mu, \qquad f \in W^{1,2}(\mathbb{R}^N, \mu), \tag{1.0.21}$$

is satisfied. A sufficient condition for the validity of the Poincaré inequality is the compactness of the embedding $W^{1,2}(\mathbb{R}^N,\mu) \subset L^2(\mathbb{R}^N,\mu)$. In turn, this is implied by the validity of the log-Sobolev inequality:

$$\int_{\mathbb{R}^N} f^2 \log |f| \, d\mu \le \|f\|_2^2 \log \|f\|_2 + C \int_{\mathbb{R}^N} |Df|^2 d\mu, \quad f \in W^{1,2}(\mathbb{R}^N, \mu).$$
(1.0.22)

For instance, this happens whenever the gradient estimate (1.0.16), with p = 1 and $\sigma_1 = -\alpha < 0$, hold. Here, we see the importance of the pointwise gradient estimates with p = 1 and exponentials of negative type.

We summarize the relations up to now determined in the following scheme.

$$|(DT(t)f)(x)| \leq e^{-\alpha t}(T(t)|Df|)(x), \quad t>0, \ x\in\mathbb{R}^N, \quad \alpha>0,$$

$$\downarrow \\ \text{log-Sobolev inequality } (1.0.22)$$

$$\downarrow \\ W^{1,2}(\mathbb{R}^N,\mu) \subset L^2(\mathbb{R}^N,\mu) \text{ with a compact embedding}$$

$$\downarrow \\ \text{Poincar\'e inequality } (1.0.21)$$

$$\downarrow \\ \text{spectral gap } (1.0.20) \iff \text{exponential convergence to } \overline{f}$$

$$\uparrow \\ T(t) \text{ is compact in } C_b(\mathbb{R}^N) \text{ for any } t>0$$

Introduction xxvii

The log-Sobolev inequality (1.0.22) implies that $f^2 \log |f|$ is integrable whenever $f \in W^{1,2}(\mathbb{R}^N,\mu)$. This Sobolev-type result is very sharp, as an example of L. Gross shows: if μ is the Gaussian measure, there exists a function $f \in W^{1,2}(\mathbb{R}^N,\mu)$ such that $f^2(\log f)(\log \log f)$ is not integrable. In particular, one cannot expect any Sobolev embedding, such as $W^{1,2}(\mathbb{R}^N,\mu) \subset L^p(\mathbb{R}^N,\mu)$, when p > 2.

In the case when

$$\mathcal{A}\varphi(x) = \Delta\varphi(x) - \langle DU(x), D\varphi(x) \rangle, \qquad x \in \mathbb{R}^N, \tag{1.0.23}$$

on smooth functions φ , and U is a smooth function such that e^{-U} is integrable in \mathbb{R}^N , the previous scheme can be improved a bit. Up to a normalization factor, the invariant measure of the associated semigroup is given by

$$d\mu = e^{-U(x)}dx.$$

In this symmetric case, the log-Sobolev inequality (1.0.22) is equivalent to the hypercontractivity of $\{T(t)\}$, that is

$$||T(t)f||_{q(t)} \le ||f||_2, \quad q(t) = 1 + e^{\lambda t}, \quad f \in L^2(\mathbb{R}^N, \mu),$$
 (1.0.24)

where $\|\cdot\|_r$ is the norm in $L^r(\mathbb{R}^N, \mu)$ and $\lambda = 2/C$. The hypercontractivity was first proved by E. Nelson (see [121]) for the Ornstein Uhlenbeck semigroup, while the equivalence with the log-Sobolev inequality was proved by L. Gross (see [69]). On the other hand, D. Bakry and M. Émery (see [12]) showed the hypercontractivity of $\{T(t)\}$ when the operator \mathcal{A} is given by (1.0.23) with U satisfying

$$\langle D^2 U(x)\xi, \xi \rangle \ge \alpha |\xi|^2, \qquad x, \xi \in \mathbb{R}^N,$$

for some $\alpha > 0$. Their proof needs the gradient estimate (1.0.16) with p = 1 and $\sigma_1 < 0$.

Thus, when \mathcal{A} is given by (1.0.23) we can complete the scheme above with the following part:

$$\begin{split} \mathcal{A}\varphi &= \Delta \varphi - \langle DU, D\varphi \rangle, \quad D^2U \geq \alpha > 0 \\ & \downarrow \\ & |(DT(t)f)(x)| \leq e^{-\alpha t} (T(t)|Df|)(x), \quad t > 0, \ x \in \mathbb{R}^N, \ \alpha > 0 \\ & \downarrow \\ & \log\text{-Sobolev inequality } (1.0.22) \\ & \updownarrow \\ & \text{hypercontractivity of } T(t) \end{split}$$

The arguments described here are discussed in detail in Chapter 8.

Finally, due to its importance, we devote Chapter 9 to describe most of the significant results on the Ornstein-Uhlenbeck semigroup both in $C_b(\mathbb{R}^N)$ and in the L^p -space related to its invariant measure. We mainly focus on the nondegenerate Ornstein-Uhlenbeck operator, but some of the results that we describe hold also in the degenerate case.

To conclude this first part of the introduction, we summarize, in the following table, the results that can be proved assuming the existence of a nonidentically vanishing Lyapunov function $\varphi \in C^2(\mathbb{R}^N)$. For simplicity we assume $c \equiv 0$, even though some results hold in the general case. We denote by λ and C positive constants. Moreover, the notation $\varphi \nearrow +\infty$ means $\lim_{|x| \to +\infty} \varphi(x) = +\infty$. Similarly the notation $\varphi \searrow +\infty$ means $\lim_{|x| \to +\infty} \varphi(x) = -\infty$.

1	$\varphi \nearrow +\infty$	$\mathcal{A}\varphi - \lambda\varphi \le C$	uniqueness
2	$0 \le \varphi \le C$	$\mathcal{A}\varphi - \lambda\varphi \ge 0$	nonuniqueness
3	$\varphi \ge 0, \varphi \in C_0(\mathbb{R}^N)$	$\mathcal{A}\varphi - \lambda\varphi \le -C$	$T(t)1 \in C_0(\mathbb{R}^N)$
4	$\varphi \nearrow +\infty,$ $g: \mathbb{R} \to \mathbb{R} \text{ convex},$ $\frac{1}{g} \text{ integrable at } +\infty$	$\mathcal{A}\varphi(x) \le -g(\varphi(x))$	$T(t)$ compact, $C_0(\mathbb{R}^N)$ not invariant
5	$\varphi > 0, \ \varphi \in C_0(\mathbb{R}^N)$	$\mathcal{A}\varphi - \lambda\varphi \le 0$	$C_0(\mathbb{R}^N)$ invariant
6	$\varphi \geq 0$	$\mathcal{A}\varphi \setminus -\infty$	∃ invariant measure

Part II

In this part of the book, we consider the case when \mathbb{R}^N is replaced with an open and unbounded domain of \mathbb{R}^N and we associate homogeneous Dirichlet and Neumann boundary conditions with the operator \mathcal{A} . In these two situations the geometry of the open set Ω plays a crucial role in proving the existence of a semigroup associated with the operator \mathcal{A} . In all the cases when we guarantee the existence of the semigroup, we can also prove some uniform gradient estimates (and pointwise gradient estimates in the case of Neumann boundary conditions).

The case when homogeneous Neumann boundary conditions are associated with the operator \mathcal{A} and Ω is a convex set is the easiest one to handle. The existence of the semigroup is proved by approximating the Cauchy-Neumann problem with a sequence of Cauchy-Neumann problems in suitable bounded and convex domains Ω_n . Using the same arguments as in the case when $\Omega = \mathbb{R}^N$, one can show that the solutions u_n to such approximating problems converge to a smooth function u, with normal derivative vanishing on

Introduction xxix

 $\partial\Omega_n$, satisfying the parabolic equation $D_tu_n - \mathcal{A}u_n = 0$ in Ω_n . Showing the continuity of u up to t = 0 is much more difficult than in the case when $\Omega = \mathbb{R}^N$, since the sequence u_n does not satisfy any monotonicity property. The key point to overcome such a difficulty consists in determining uniform gradient estimates for the solution u_n (similar to those of the case $\Omega = \mathbb{R}^N$), where the constants appearing in the estimates are independent of n. Once such estimates are available, a localization argument shows that u is continuous up to t = 0. The uniform gradient estimates can be proved by adapting the Bernstein method and applying it to the functions u_n . Here, the convexity of Ω_n plays a crucial role in making this machinery work.

Unfortunately, the Bernstein method cannot be adapted to prove the estimates for the higher order derivatives and, to the best of our knowledge, such estimates are available only in the case when Ω is an exterior domain (see Section 13.5).

The semigroup $\{T(t)\}$ is not strongly continuous and, in general, it is not analytic in $C_b(\overline{\Omega})$. In any case, as in Chapter 2, it is possible to define its weak generator and to give a partial characterization of it.

As it has been already remarked, the other two situations considered in the second part of the book are much more involved. This is essentially due to the fact that it is not immediate to adapt the Bernstein method as in the previous situation.

Part III

Finally, in this part of the book, we describe how the approach used in the first part of the book can successfully be applied also to a few cases in which the operator \mathcal{A} is degenerate (and with the coefficients defined on the whole of \mathbb{R}^N). More precisely, we consider degenerate elliptic operators \mathcal{A} of the type

$$\mathcal{A}\varphi(x) = \sum_{i,j=1}^{r} q_{ij}(x)D_{ij}\varphi(x) + \sum_{i,j=1}^{N} b_{ij}x_jD_i\varphi(x), \qquad x \in \mathbb{R}^N, \quad (1.0.25)$$

when $N/2 \leq r < N$ and the matrix (q_{ij}) is strictly positive definite. It is possible to associate a semigroup $\{T(t)\}$ with the operator \mathcal{A} and to prove uniform estimates for the space derivatives of the function T(t)f up to the third-order, when $f \in C_b(\mathbb{R}^N)$. Such estimates are used to prove Schauder estimates for the distributional solutions to both the elliptic equation and the nonhomogeneous Cauchy problem associated with the operator \mathcal{A} .

Unfortunately, the techniques used in the nondegenerate case cannot be easily adapted to this situation. Indeed, such techniques rely essentially on interior Schauder estimates that are not available in the degenerate case.

The construction of the semigroup associated with the problem (1.0.2) and the determination of the uniform estimates are two topics to be treated simultaneously: we need a *priori* estimates on the behaviour of the space derivatives of the function T(t)f in order to guarantee its existence. This forces us to as-

sume stronger assumptions on the coefficients than those in the nondegenerate case.

The underlying idea used to solve the problem is simple: we approximate the operator \mathcal{A} by a sequence of uniformly elliptic operators $\mathcal{A}_{\varepsilon}$ ($\varepsilon > 0$) converging to the operator \mathcal{A} in a sense to be made precise later. With each of such operators we can associate a semigroup of bounded linear operators $\{T_{\varepsilon}(t)\}$. We prove uniform estimates for $\{T_{\varepsilon}(t)\}$, with constants being independent of ε , and, then, letting ε go to 0^+ , we can show that $\{T_{\varepsilon}(t)\}$ converges to a semigroup of bounded operators. Such a semigroup is associated with the operator \mathcal{A} , in the sense that $T(\cdot)f$ turns out to be the (unique) smooth function satisfying $D_t u - \mathcal{A} u = 0$ in $(0, +\infty) \times \mathbb{R}^N$ and the initial condition $u(0, \cdot) = f$, which is bounded in $[0, T] \times \mathbb{R}^N$ for any T > 0.

A fundamental tool for such a machinery work is given by the maximum principle which must hold for both of the operators \mathcal{A} and $\mathcal{A}_{\varepsilon}$.

The behaviour of the derivatives of T(t)f is worse than in the nondegenerate case. For instance, for any $\omega > 0$, one can show that there exists a positive constant C such that

$$||D_i T(t)f||_{\infty} \le \begin{cases} Ct^{-\frac{1}{2}} e^{\omega t}, & i \le r, \\ Ct^{-\frac{3}{2}} e^{\omega t}, & i > r, \end{cases}$$
(1.0.26)

for any t > 0. This situation is not surprising at all, since this is just what happens for the degenerate Ornstein-Uhlenbeck operator defined by (1.0.25) with constant diffusion coefficients. In such a case, the estimates (1.0.26) are known to be optimal near t = 0.

Most of the results holding in the nondegenerate case can be recovered also in this situation. In particular, it is still possible to associate a weak generator with the semigroup and, taking advantage of the uniform estimates, one can (at least partially) characterize its domain. Further, one can deal with the elliptic equation

$$\lambda u - \mathcal{A}u = f \in C_b(\mathbb{R}^N)$$

whose continuous distributional solution is still formally given by the Laplace transform of the semigroup, where the integral, as in the nondegenerate case, is meant in pointwise sense.

Similarly, the continuous distributional solution to the nonhomogeneous Cauchy problem

$$\begin{cases} D_t u(t,x) - \mathcal{A}u(t,x) = g(t,x), & t > 0, x \in \mathbb{R}^N, \\ u(0,x) = f(x), & x \in \mathbb{R}^N, \end{cases}$$

when f and g are sufficiently smooth, is still given by the variation-of-constants formula (1.0.14). Such a formula allows us, also in this situation, to determine Schauder estimates for the solution u to the Cauchy problem via an interpolation argument.

Introduction xxxi

With the aim of making the book as self-contained as possible, we collect in four appendices all the classical results of functional analysis and of the theory of partial differential equations, as well as some properties of the distance function. Finally, in the fifth appendix, we collect all the function spaces that we use in this book.

Part I $\mathbf{Markov} \ \mathbf{semigroups} \ \mathbf{in} \ \mathbb{R}^{N}$

Chapter 2

The elliptic equation and the Cauchy problem in $C_b(\mathbb{R}^N)$: the uniformly elliptic case

2.0 Introduction

In this chapter we consider linear elliptic and parabolic problems in $C_b(\mathbb{R}^N)$ associated with the differential operator \mathcal{A} defined by

$$Au(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}u(x) + \sum_{i=1}^{N} b_{i}(x)D_{i}u(x) + c(x)u(x), \qquad x \in \mathbb{R}^{N},$$

on smooth functions. We assume the following hypotheses on the coefficients of the operator \mathcal{A} .

Hypotheses 2.0.1 (i) $q_{ij} \equiv q_{ji}$ for any i, j = 1, ..., N and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \quad \kappa(x) > 0, \quad \xi, x \in \mathbb{R}^N;$$

- (ii) q_{ij} , b_i (i, j = 1, ..., N) and c belong to $C_{loc}^{\alpha}(\mathbb{R}^N)$ for some $\alpha \in (0, 1)$;
- (iii) there exists $c_0 \in \mathbb{R}$ such that

$$c(x) \le c_0, \quad x \in \mathbb{R}^N.$$

Besides, we introduce the realization A of \mathcal{A} in $C_b(\mathbb{R}^N)$, with domain $D_{\max}(\mathcal{A})$, defined as follows:

$$D_{\max}(\mathcal{A}) = \left\{ u \in C_b(\mathbb{R}^N) \cap \bigcap_{1 \le p < +\infty} W_{\mathrm{loc}}^{2,p}(\mathbb{R}^N) : \mathcal{A}u \in C_b(\mathbb{R}^N) \right\}, \quad Au = \mathcal{A}u.$$
(2.0.1)

In Section 2.1 we prove that, for any $\lambda > c_0$ and any $f \in C_b(\mathbb{R}^N)$, the elliptic equation

$$\lambda u(x) - \mathcal{A}u(x) = f(x), \quad x \in \mathbb{R}^N,$$
 (2.0.2)

admits a solution $u \in D_{\max}(\mathcal{A})$. The idea of the proof is the following. For each $n \in \mathbb{N}$ we consider the Dirichlet problem

$$\begin{cases} \lambda u_n(x) - \mathcal{A}u_n(x) = f(x), & x \in B(n), \\ u_n(x) = 0, & x \in \partial B(n), \end{cases}$$
 (2.0.3)

in the ball $B(n) = \{x \in \mathbb{R}^N : |x| < n\}$. This problem has a unique solution $u_n \in C(\overline{B}(n))$ (in Section C we recall the results about elliptic and parabolic problems in bounded domains that we need throughout this chapter). Using an interior estimate (see Theorem C.1.1), we prove that we can define a function $u : \mathbb{R}^N \to \mathbb{R}$ by setting

$$u(x) := \lim_{n \to +\infty} u_n(x),$$

for any $x \in \mathbb{R}^N$. The function u belongs to $D_{\max}(A)$ is a solution of the equation (2.0.2) and it satisfies the estimate

$$||u||_{\infty} \le \frac{1}{\lambda - c_0} ||f||_{\infty}.$$

We stress that, in general, u is not the unique solution of the equation (2.0.2) in $D_{\text{max}}(\mathcal{A})$. It is the unique solution provided further conditions on the coefficients are satisfied. The problem of the uniqueness of the solution to the equation (2.0.2) will be treated in Chapter 4. At any rate, if $f \geq 0$ then u is the minimal positive solution of (2.0.2) in $D_{\text{max}}(\mathcal{A})$.

Next, we prove that u is given by the formula

$$u(x) = \int_{\mathbb{R}^N} K_{\lambda}(x, y) f(y) dy, \quad x \in \mathbb{R}^N,$$
 (2.0.4)

where K_{λ} is a positive function; K_{λ} is the so-called Green's function associated with the problem (2.0.2). To prove (2.0.4) we recall that the solution u_n of (2.0.3) is given by the formula

$$u_n(x) = \int_{B(n)} K_{\lambda}^n(x, y) f(y) dy, \quad x \in B(n),$$

where K_{λ}^{n} is the Green's function associated with (2.0.3). Using the classical maximum principle we prove that the sequence $\{K_{\lambda}^{n}\}$ is increasing (with respect to $n \in \mathbb{N}$). This leads to the formula (2.0.4) with

$$K_{\lambda}(x,y) := \lim_{n \to +\infty} K_{\lambda}^{n}(x,y), \qquad x, y \in \mathbb{R}^{N}.$$

Thus for any $\lambda > c_0$ we can define the linear operator $R(\lambda)$ in $C_b(\mathbb{R}^N)$ by setting

$$(R(\lambda)f)(x) = \int_{\mathbb{R}^N} K_{\lambda}(x,y)f(y)dy, \quad x \in \mathbb{R}^N.$$

 $R(\lambda)$ is a bounded operator, with $||R(\lambda)||_{L(C_b(\mathbb{R}^N))} \leq (\lambda - c_0)^{-1}$. In Section 2.3 we see that the operators $\{R(\lambda) : \lambda > c_0\}$ are the resolvent operators of a linear operator \widehat{A} in $C_b(\mathbb{R}^N)$. The operator \widehat{A} is called weak generator. This terminology is justified by the theory of parabolic problems and it will be much clearer later on in this chapter.

The parabolic problem that we consider is the following

$$\begin{cases}
D_t u(t,x) - \mathcal{A}u(t,x) = 0, & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N,
\end{cases}$$
(2.0.5)

with $f \in C_b(\mathbb{R}^N)$. In Section 2.2 we prove the existence of a classical solution of such a problem (i.e., a function $u \in C([0, +\infty) \times \mathbb{R}^N) \cap C^{1,2}((0, +\infty) \times \mathbb{R}^N))$, which is bounded in $[0, T] \times \mathbb{R}^N$ for any T > 0 and satisfies $D_t u, D^2 u \in C^{\alpha/2,\alpha}_{loc}((0, +\infty) \times \mathbb{R}^N)$. The idea of the proof is similar to that used in the elliptic case. More precisely, for each $n \in \mathbb{N}$ we consider the Cauchy-Dirichlet problem

$$\begin{cases} D_t u_n(t, x) - \mathcal{A} u_n(t, x) = 0, & t > 0, x \in B(n), \\ u_n(t, x) = 0, & t > 0, x \in \partial B(n), \\ u_n(0, x) = f(x), & x \in B(n), \end{cases}$$
 (2.0.6)

in the ball B(n). By classical results for parabolic Cauchy problems in bounded domains we know that the problem (2.0.6) admits a unique solution $u_n \in C([0,+\infty) \times \overline{B}(n) \setminus (\{0\} \times \partial B(n))) \cap C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty) \times B(n))$. Using Schauder interior estimates (see Theorem C.1.4) and a compactness argument, we prove that we can define a function $u:[0,+\infty) \times \mathbb{R}^N \to \mathbb{R}$ by setting

$$u(t,x) := \lim_{n \to +\infty} u_n(t,x),$$

for any $t \in [0, +\infty)$ and any $x \in \mathbb{R}^N$. Such a function belongs to $C([0, +\infty) \times \mathbb{R}^N) \cap C^{1+\alpha/2, 2+\alpha}_{loc}((0, +\infty) \times \mathbb{R}^N)$, is a solution of the problem (2.0.6) and satisfies the estimate

$$|u(t,x)| \le \exp(c_0 t) ||f||_{\infty}, \qquad t > 0, \ x \in \mathbb{R}^N.$$

In general, as in the elliptic case, u is not the unique classical solution of the problem (2.0.5) which is bounded in $[0,T]\times\mathbb{R}^N$ for any T>0. It turns out to be the unique solution under further assumptions on the coefficients (see Chapter 4). Even though u is not the unique solution to the problem (2.0.5), at any rate it has an intrinsic meaning: in fact, for any $f\geq 0$, u is the minimal solution to the problem (2.0.5) in the sense that, if v is another positive solution to the same Cauchy problem, then $v\geq u$. This is shown in Remark 2.2.3.

Then, we prove that u can be represented by the formula

$$u(t,x) = \int_{\mathbb{R}^N} G(t,x,y) f(y) dy, \quad t > 0, \ x \in \mathbb{R}^N,$$
 (2.0.7)

where G is a positive function, called the fundamental solution. To prove (2.0.7) we argue as in the elliptic case, recalling that the solution u_n of the problem (2.0.6) is given by the formula

$$u_n(t,x) = \int_{B(n)} G_n(t,x,y) f(y) dy, \quad t > 0, \ x \in B(n),$$

where G_n is the fundamental solution of (2.0.6). Using the classical maximum principle we prove that the sequence $\{G_n\}$ is increasing with respect to $n \in \mathbb{N}$. This gives the formula (2.0.7) with

$$G(t, x, y) = \lim_{n \to +\infty} G_n(t, x, y), \quad t > 0, \ x, y \in \mathbb{R}^N,$$

and it allows us to define the linear operator T(t) in $C_b(\mathbb{R}^N)$, for any t > 0, by setting

$$(T(t)f)(x) = \int_{\mathbb{R}^N} G(t, x, y) f(y) dy, \quad t > 0, \ x \in \mathbb{R}^N.$$

We prove that the family $\{T(t)\}$ is a semigroup of linear operators in $C_b(\mathbb{R}^N)$. In general, $\{T(t)\}$ is a strongly continuous semigroup neither in $C_b(\mathbb{R}^N)$ nor in $BUC(\mathbb{R}^N)$ (see Proposition 9.2.6 and [148, Theorem 4.2]). Nevertheless, T(t)f tends to f as t tends to 0, uniformly on compact sets. Then, we show that if f vanishes at infinity, then, actually, T(t)f tends to f as t tends to 0, uniformly in \mathbb{R}^N . But this does not mean that the restriction of $\{T(t)\}$ to $C_0(\mathbb{R}^N)$ is a strongly continuous semigroup, because, in general, $C_0(\mathbb{R}^N)$ is not invariant for $\{T(t)\}$ (see Section 5.3).

In Section 2.3 we study the relation between the semigroup $\{T(t)\}$ and the weak generator \widehat{A} introduced above. In fact, \widehat{A} is the generator of $\{T(t)\}$ with respect to the bounded pointwise convergence. This means that, for any $f \in D(\widehat{A})$, the function $(t,x) \mapsto \{(T(t)f)(x) - f(x)\}/t$ is bounded in $(0,1) \times \mathbb{R}^N$ and

$$\lim_{t\to 0^+}\frac{(T(t)f)(x)-f(x)}{t}=(\widehat{A}f)(x),$$

for any $x \in \mathbb{R}^N$. Further, \widehat{A} is also the generator of the semigroup with respect to the mixed topology, which is the finest locally convex topology which agrees on every norm-bounded subsets of $C_b(\mathbb{R}^N)$ with the topology of the uniform convergence on compact sets. A sequence of functions converges in the mixed topology if and only if it is bounded and it converges locally uniformly. Thus, the convergence in the mixed topology is very similar to the bounded pointwise convergence.

In general, $D(\widehat{A})$ is a proper subset of $D_{\max}(A)$. It turns out to coincide with $D_{\max}(A)$ whenever the elliptic equation (2.0.2) has a unique solution in $D_{\max}(A)$.

It is immediate to see that the semigroup $\{T(t)\}$ can be extended to the space $B_b(\mathbb{R}^N)$ of Borel measurable and bounded functions. We prove that $\{T(t)\}$ is irreducible and it has the strong Feller property in $B_b(\mathbb{R}^N)$ (see Definition 2.2.11). These properties are useful for the theory of the invariant measures treated in Chapter 8.

In the case when $c \equiv 0$, the semigroup $\{T(t)\}$ is associated with a transition function. This leads to the existence of a Markov process associated with $\{T(t)\}$. We briefly deal with the probabilistic approach in Section 2.4. Of course, there is a huge literature on the subject. Here, we just recall the definitions and the main properties of the Markov processes associated with the semigroup. Among them, we see the Dynkin formula and the link with the theory of differential stochastic equations.

2.1 The elliptic equation and the resolvent $R(\lambda)$

In this section we consider the elliptic equation (2.0.2) with $f \in C_b(\mathbb{R}^N)$ and $\lambda > c_0$. We prove that such an equation admits at least a solution u belonging to the domain $D_{\max}(\mathcal{A})$ defined in (2.0.1).

Theorem 2.1.1 For any $f \in C_b(\mathbb{R}^N)$ there exists $u \in D_{\max}(A)$ which solves (2.0.2). Moreover, the following estimate holds:

$$||u||_{\infty} \le \frac{1}{\lambda - c_0} ||f||_{\infty}.$$
 (2.1.1)

Finally, if $f \ge 0$, then $u \ge 0$.

Proof. For any $n \in \mathbb{N}$, we denote by A_n the realization of the operator \mathcal{A} with homogeneous Dirichlet boundary conditions in $C(\overline{B}(n))$. By Proposition C.3.4, the elliptic problem (2.0.3) admits a unique solution $u_n = R(\lambda, A_n)f$ in $\bigcap_{1 \le p \le +\infty} W^{2,p}(B(n))$. Moreover, by (C.2.3), u_n satisfies the estimate

$$||u_n||_{\infty} \le \frac{1}{\lambda - c_0} ||f||_{\infty}.$$
 (2.1.2)

Let us prove that the sequence $\{u_n\}$ converges uniformly on compact sets, and in $W^{2,p}_{loc}(\mathbb{R}^N)$, to a function $u \in D_{\max}(\mathcal{A})$ which satisfies the statement. For this purpose, consider first the case when $f \geq 0$. By the maximum principle, the functions u_n and $u_{n+1} - u_n$ are nonnegative in B(n) for any $n \in \mathbb{N}$, that is the sequence $\{u_n\}$ is nonnegative and increasing. Therefore, it converges pointwise to some nonnegative function $u : \mathbb{R}^N \to \mathbb{R}$. By (2.1.2) it

follows that u satisfies (2.1.1). Moreover, according to Theorem C.1.1 and (2.1.2), the sequence $\{u_n\}$ is bounded in $W^{2,p}(B(R))$ for any $p \in [1, +\infty)$ and any fixed R > 0. Then, by the Sobolev embedding theorems (see [2, Theorem 5.4]), it is bounded in $C^1(\overline{B}(R))$ too, and the Ascoli-Arzelà Theorem implies that it converges to u in $C(\overline{B}(R))$. Besides, applying again Theorem C.1.1 to the function $u_n - u_m$, we deduce that u belongs to $W^{2,p}(B(R))$ and that u_n converges to u in $W^{2,p}(B(R))$, for any $p \in [1, +\infty)$.

Since $Au_n = \lambda u_n - f$ in B(n), it follows that $u \in D_{\max}(A)$ and $Au = \lambda u - f$. This concludes the proof in the case when $f \geq 0$. For an arbitrary $f \in C_b(\mathbb{R}^N)$, it suffices to split $f = f^+ - f^-$ and

$$u_n = R(\lambda, A_n)(f^+) - R(\lambda, A_n)(f^-) := u_{n,1} + u_{n,2},$$

and to apply the previous arguments separately to the sequences $u_{n,1}$ and $u_{n,2}$.

Remark 2.1.2 In general, the equation (2.0.2) admits more than one solution in $D_{\max}(\mathcal{A})$. In Chapters 3 and 5 we show some situations in which the elliptic equation $\lambda u - \mathcal{A}u = 0$ admits a nontrivial solution in $D_{\max}(\mathcal{A})$ (see Examples 3.2.5 and 5.2.5). Nevertheless, in the case when the datum f is nonnegative the solution u provided by Theorem 2.1.1 can be characterized as the minimal positive solution. Indeed, if v is another positive solution, by the maximum principle it follows that $v(x) \geq u_n(x)$ for any $x \in B(n)$ and any $n \in \mathbb{N}$. Letting n go to $+\infty$ gives $v \geq u$.

We now prove that we can associate a positive Green's function with the equation (2.0.2). This will then allow us to define the resolvent operator $R(\lambda)$ for any $\lambda > c_0$.

Theorem 2.1.3 For any $\lambda > c_0$ there exists a linear operator $R(\lambda)$ in $C_b(\mathbb{R}^N)$ such that for any $f \in C_b(\mathbb{R}^N)$ the solution of the equation (2.0.2), provided by Theorem 2.1.1, is represented by

$$u(x) = (R(\lambda)f)(x), \quad x \in \mathbb{R}^N.$$
(2.1.3)

The family of operators $\{R(\lambda): \lambda > c_0\}$ satisfies the estimate

$$||R(\lambda)f||_{\infty} \le \frac{1}{\lambda - c_0} ||f||_{\infty}, \quad f \in C_b(\mathbb{R}^N), \tag{2.1.4}$$

and the resolvent identity

$$R(\lambda)f - R(\mu)f = (\mu - \lambda)R(\mu)R(\lambda)f, \quad c_0 < \lambda < \mu.$$
 (2.1.5)

Moreover, $R(\lambda)$ is injective for any $\lambda > c_0$. Finally, there exists a positive function $K_{\lambda} : \mathbb{R}^N \times \mathbb{R}^N \to \mathbb{R}$ such that

$$(R(\lambda)f)(x) = \int_{\mathbb{R}^N} K_{\lambda}(x, y)f(y)dy, \quad x \in \mathbb{R}^N, \ f \in C_b(\mathbb{R}^N).$$
 (2.1.6)

Proof. Let K_{λ}^n be the Green's function associated with the realization A_n of the operator A in $C(\overline{B}(n))$ with homogeneous Dirichlet conditions (see Proposition C.3.4). With any nonnegative function $f \in C_0(B(n))$, let

$$v_n(x) = \int_{B(n)} (K_{\lambda}^{n+1}(x,y) - K_{\lambda}^n(x,y)) f(y) dy, \quad x \in B(n).$$

Since $f \ge 0$ and $v_n = u_{n+1} - u_n$ by Proposition C.2.3, we have $v_n(x) \ge 0$ for any $x \in B(n)$. The arbitrariness of $f \ge 0$ implies that

$$K_{\lambda}^{n+1}(x,y) - K_{\lambda}^{n}(x,y) \ge 0,$$

for any $x \in B(n)$ and any $y \in E_n(x)$, where $E_n(x)$ is a measurable set such that $B(n) \setminus E(x)$ is negligible. Hence, for any $x \in \mathbb{R}^N$ and any $y \in E_n(x)$, $\{K_{\lambda}^n(x,y)\}$ is an increasing sequence. Therefore, we can define the function $K_{\lambda} : \mathbb{R}^N \times \mathbb{R}^N \to \mathbb{R}$ by setting

$$K_{\lambda}(x,y) := \lim_{n \to +\infty} K_{\lambda}^{n}(x,y), \qquad x \in \mathbb{R}^{N}, \ y \in \bigcap_{n \in \mathbb{N}} E_{n}.$$
 (2.1.7)

This limit is not infinite. Indeed estimate (C.2.3) with $f \equiv 1$ yields

$$\int_{B(n)} K_{\lambda}^{n}(x, y) dy \le \frac{1}{\lambda - c_{0}}, \qquad x \in B(n), \ n \in \mathbb{N},$$

and, then, by monotone convergence

$$\int_{\mathbb{R}^N} K_{\lambda}(x, y) dy \le \frac{1}{\lambda - c_0}, \qquad x \in \mathbb{R}^N,$$

so that, for any $x \in \mathbb{R}^N$, K(x,y) is finite for almost any $y \in \mathbb{R}^N$. Besides, since K_{λ}^n is strictly positive in $B(n) \times B(n)$ for any $n \in \mathbb{N}$, also K_{λ} is.

Now, we observe that the solution of the equation (2.0.2) given by Theorem 2.1.1 can be represented by

$$u(x) = \lim_{n \to +\infty} \left(\int_{\mathbb{R}^N} K_{\lambda}^n(x, y) f^+(y) dy - \int_{\mathbb{R}^N} K_{\lambda}^n(x, y) f^-(y) dy \right),$$

for any $x \in \mathbb{R}^N$. Since both f^+ and f^- are nonnegative, the monotone convergence theorem implies that

$$u(x) = \int_{\mathbb{R}^N} K_{\lambda}(x, y) f(y) dy, \quad x \in \mathbb{R}^N.$$

Thus, the operator $R(\lambda)$ in (2.1.6) is well defined and (2.1.3) holds. Clearly, $R(\lambda)$ is linear, and, by virtue of (2.1.1), it satisfies (2.1.4). Moreover, $R(\lambda)$ is injective. Indeed, if $u = R(\lambda)f = 0$, then f = 0 since $R(\lambda)f$ solves, by construction, the elliptic equation $\lambda u - Au = f$.

To conclude the proof, it remains to prove the resolvent identity (2.1.5). To prove it, we fix $f \in C_b(\mathbb{R}^N)$, $n \in \mathbb{N}$ and we observe that

$$R(\lambda, A_n)f - R(\mu, A_n)f = (\mu - \lambda)R(\lambda, A_n)R(\mu, A_n)f, \quad \mu > \lambda > c_0,$$

(see Sections A.3 and C). By virtue of (C.3.8), this can be rewritten as

$$\begin{split} &\int_{B(n)} K_{\lambda}^{n}(x,y)f(y)dy - \int_{B(n)} K_{\mu}^{n}(x,y)f(y)dy \\ &= (\mu - \lambda) \int_{B(n)} dy \int_{B(n)} K_{\mu}^{n}(x,y)K_{\lambda}^{n}(y,z)f(z)dz. \end{split}$$

Letting n go to $+\infty$, the dominated convergence theorem gives

$$\int_{\mathbb{R}^N} K_{\lambda}(x,y) f(y) dy - \int_{\mathbb{R}^N} K_{\mu}(x,y) f(y) dy$$
$$= (\mu - \lambda) \int_{\mathbb{R}^N} dy \int_{\mathbb{R}^N} K_{\mu}(x,y) K_{\lambda}(y,z) f(z) dz,$$

that is (2.1.5).

2.2 The Cauchy problem and the semigroup

We now prove that for any $f \in C_b(\mathbb{R}^N)$ there exists a solution $u \in C([0, +\infty) \times \mathbb{R}^N) \cap C^{1+\alpha/2, 2+\alpha}((0, +\infty) \times \mathbb{R}^N)$ of the Cauchy problem (2.0.5).

Theorem 2.2.1 For any $f \in C_b(\mathbb{R}^N)$, there exists a solution $u \in C([0, +\infty) \times \mathbb{R}^N)$ of the problem (2.0.5). The function u belongs to $C_{loc}^{1+\alpha/2,2+\alpha}((0,+\infty) \times \mathbb{R}^N)$ and

$$|u(t,x)| \le \exp(c_0 t) ||f||_{\infty}, \qquad t > 0, \ x \in \mathbb{R}^N.$$
 (2.2.1)

Proof. We split the proof into two steps. First, we show that there exists a solution $u \in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\mathbb{R}^N)$ to the differential equation in (2.0.5), and it satisfies (2.2.1). Then, in Step 2, we show that u is continuous up to t=0, and $u(0,\cdot)=f$.

Step 1. For any $n \in \mathbb{N}$, let $u_n \in C([0, +\infty) \times \overline{B}(n) \setminus (\{0\} \times \partial B(n))\} \cap C^{1,2}((0, +\infty) \times B(n))$ be the solution of the Cauchy-Dirichlet problem (2.0.6) (see Proposition C.3.2), which is given by

$$u_n(t,x) = (T_n(t)f)(x), \quad t > 0, \ x \in B(n),$$

where $\{T_n(t)\}$ is the semigroup in $C(\overline{B}(n))$ associated with the Cauchy-Dirichlet problem (2.0.6). From Proposition C.3.2 (see (C.3.4)), for any $n \in \mathbb{N}$ we have

$$|u_n(t,x)| \le \exp(c_0 t) ||f||_{\infty}, \quad t > 0, \ x \in B(n).$$
 (2.2.2)

Now, fix $M \in \mathbb{N}$ and set $D(M) = (0, M) \times B(M)$ and $D'(M) = [1/M, M] \times B(M-1)$. From the interior Schauder estimate (C.1.15) we deduce that

$$||u_n||_{C^{1+\alpha/2,2+\alpha}(D'(M))} \le C_M ||u_n||_{L^{\infty}(D(M))} \le C_M (\exp(c_0 M) \vee 1) ||f||_{\infty},$$
(2.2.3)

for any $n \geq M$, where $C_M > 0$ is a constant independent of $n \in \mathbb{N}$. Fix $\beta \in (0, \alpha)$. By (2.2.3) there exists a subsequence $\{u_n^{(M)}\}$ of $\{u_n\}$ converging in $C^{1+\beta/2,2+\beta}(D'(M))$ to some function $u_{\infty}^{(M)} \in C^{1+\alpha/2,2+\alpha}(D'(M))$. Without loss of generality we can assume that $\{u_n^{(M+1)}\}$ is a subsequence of $\{u_n^{(M)}\}$. Thus, the functions $u_{\infty}^{(M)}$ and $u_{\infty}^{(M+1)}$ coincide in the domain D'(M) and, therefore, we can define the function $u \in C_{\text{loc}}^{1+\alpha/2,2+\alpha}((0,+\infty)\times\mathbb{R}^N)$ by setting

$$u = u_{\infty}^{(M)}$$
 in $D'(M)$.

Moreover, the diagonal subsequence defined by

$$\tilde{u}_n = u_n^{(n)}, \quad n \in \mathbb{N},$$

converges to u in $C^{1+\beta/2,2+\beta}([T_1,T_2]\times K)$ for any compact set $K\subset\mathbb{R}^N$ and any $0< T_1< T_2$. Hence, letting n go to $+\infty$ in the differential equation satisfied by \tilde{u}_n , it follows that u satisfies the equation

$$D_t u(t,x) - \mathcal{A}u(t,x) = 0, \quad t > 0, \ x \in \mathbb{R}^N.$$

Besides, (2.2.1) follows from (2.2.2).

Step 2. To complete the proof we must show that $u \in C([0, +\infty) \times \mathbb{R}^N)$ and u(0,x) = f(x). For this purpose, we take advantage of the semigroup theory. In particular, we will use the representation formula of solutions to Cauchy-Dirichlet problems in bounded domains through semigroups.

Fix $M \in \mathbb{N}$ and let ϑ be any smooth function such that

$$0 \leq \vartheta \leq 1, \qquad \vartheta \equiv 1 \ \ \text{in} \ \ B(M-1), \qquad \vartheta \equiv 0 \ \ \text{outside} \ \ B(M).$$

For any n > M, let $v_n = \vartheta \tilde{u}_n$. As it is easily seen, the function v_n belongs to $C([0, +\infty) \times \overline{B}(M))$ and is the solution of the Cauchy-Dirichlet problem

$$\begin{cases} D_t v_n(t,x) - \mathcal{A}v_n(t,x) = \psi_n(t,x), & t > 0, x \in B(M), \\ v_n(t,x) = 0, & t > 0, x \in \partial B(M), \\ v_n(0,x) = \vartheta(x)f(x), & x \in B(M), \end{cases}$$

where ψ_n is given by

$$\psi_n = -\sum_{i,j=1}^N q_{ij} (2D_i \tilde{u}_n D_j \vartheta + \tilde{u}_n D_{ij} \vartheta) - \tilde{u}_n \sum_{i=1}^N b_i D_i \vartheta.$$

For any t > 0 and any $x \in B(M)$ we have

$$|\psi_n(t,x)| \le K_M \Big(\exp(c_0 t) ||f||_{\infty} + \sum_{i=1}^N ||D_i \tilde{u}_n(t,\cdot)||_{L^{\infty}(B(M))} \Big),$$
 (2.2.4)

where $K_M > 0$ is such that

$$\sum_{i,j=1}^{N} \|q_{ij} D_{ij} \vartheta\|_{L^{\infty}(B(M))} + \sum_{i=1}^{N} \|b_{i} D_{i} \vartheta\|_{L^{\infty}(B(M))} \le K_{M},$$

$$2 \sum_{j=1}^{N} \|q_{ij} D_{j} \vartheta\|_{L^{\infty}(B(M))} \le K_{M}, \quad i = 1, \dots, N.$$

We consider again the interior estimate of Theorem C.1.4. By (C.1.16), the function \tilde{u}_n satisfies the estimate

$$|\sqrt{t}D\tilde{u}_n(t,x)| \le C \|\tilde{u}_n\|_{L^{\infty}(D(M+1))} \le C(\exp(c_0) \lor 1) \|f\|_{\infty},$$

for any $x \in B(M)$, any $t < 1 = \text{dist}(B(M), \partial B(M+1))$ and some positive constant C, independent of n. This yields

$$||D_i \tilde{u}_n(t,\cdot)||_{L^{\infty}(B(M))} \le t^{-1/2} C' ||f||_{\infty}, \quad t \le 1,$$

for any i = 1, ..., N, where $C' = C(\exp(c_0) \vee 1)$. Then, by (2.2.4) it follows that

$$|\psi_n(t,x)| \le K_M'(1+t^{-1/2})||f||_{\infty}, \qquad t \in (0,1], \ x \in B(M),$$
 (2.2.5)

for any n > M, where $K'_M > 0$ is a constant independent of n. Therefore, $\psi_n \in L^1(0, T, C_0(B(M)))$ and we can represent v_n by means of the variation-of-constants formula

$$v_n(t) = T_M(t)(\vartheta f) + \int_0^t T_M(t-s)\psi_n(s)ds, \quad t > 0,$$

where, as usual, $\{T_M(t)\}$ is the semigroup in $C(\overline{B}(M))$ associated with the operator \mathcal{A} with homogeneous Dirichlet conditions on $\partial B(M)$. Since $v_n \equiv \tilde{u}_n$ and $\vartheta \equiv 1$ in B(M-1), by (2.2.2) and (2.2.5) it follows

$$|\tilde{u}_n(t,x) - f(x)| \le ||T_M(t)(\vartheta f) - \vartheta f||_{\infty} + K_M' ||f||_{\infty} \int_0^t e^{c_0(t-s)} (1+s^{-\frac{1}{2}}) ds,$$

for any t > 0 and any $x \in B(M-1)$. Letting n go to $+\infty$ (and taking (C.3.4) into account) we get

$$|u(t,x) - f(x)| \le ||T_M(t)(\vartheta f) - \vartheta f||_{\infty} + K'_M ||f||_{\infty} \int_0^t e^{c_0(t-s)} (1 + s^{-\frac{1}{2}}) ds,$$

which shows that u is continuous at t=0 and $x \in B(M-1)$. Since $M \in \mathbb{N}$ is arbitrary, we have $u \in C([0,+\infty) \times \mathbb{R}^N)$ and $u(0,\cdot) \equiv f$.

Remark 2.2.2 We note that the maximum principle implies that, if $f \geq 0$, the sequence $\{u_n\}$ is positive and increasing. Therefore, u is positive and the whole sequence $\{u_n\}$ converges to u. In the general case where $f \in C_b(\mathbb{R}^N)$, we write $f = f^+ - f^-$, and conclude again that the whole sequence $\{u_n\}$ converges to u in $C^{1+\beta/2,2+\beta}([T_1,T_2]\times K)$ for any $\beta\in(0,\alpha)$, any compact set $K\subset\mathbb{R}^N$ and any $0< T_1< T_2$, applying the previous argument to the functions u_{f^+} and u_{f^-} which are the solutions to the problem (2.0.5) provided by Theorem 2.2.1, corresponding, respectively, to the data f^+ and f^- .

Remark 2.2.3 If $f \geq 0$, then the solution u given by Theorem 2.2.1 is the minimal positive solution of the problem (2.0.5). Indeed, if v is another positive solution, the maximum principle yields $v(t,x) \geq u_n(t,x)$ for any t > 0, any $x \in B(n)$ and any $n \in \mathbb{N}$, and, eventually, $v \geq u$.

We stress that, in general, the problem (2.0.5) is not uniquely solvable in $C_b([0,+\infty)\times\mathbb{R}^N)\cap C^{1+\alpha/2,2+\alpha}_{\mathrm{loc}}((0,+\infty)\times\mathbb{R}^N)$. Indeed, in Example 3.2.5 (see also Proposition 4.1.1) it is proved that the Cauchy problem (2.0.5) associated with the one-dimensional operator $\mathcal{A}u=u''+x^3u'$ admits a nontrivial solution u satisfying $u(0,\cdot)=0$.

We now recall the definition of transition function as it is given in [49]. Here $\mathcal{B}(E)$ denotes the σ -algebra of Borel sets of a topological space E.

Definition 2.2.4 A family of Borel measures $\{p(t, x; \cdot) : t \geq 0, x \in E\}$ is a transition function if the function $p(t, \cdot; B) : E \to \mathbb{R}$ is Borel measurable for any $t \geq 0$ and any $B \in \mathcal{B}(E)$, and

- (i) $p(t, x; E) \le 1$ for any $t \ge 0$ and any $x \in E$;
- (ii) $p(0, x; E \setminus \{x\}) = 0$ for any $x \in E$;
- (iii) $p(t+s,x;B) = \int_E p(s,y;B)p(t,x;dy)$ for any $s,t \ge 0$, any $x \in E$ and any $B \in \mathcal{B}(E)$.

A transition function is normal if $\lim_{t\to 0^+} p(t,x;E) = 1$ for any $x\in E$; it is stochastically continuous if for any open set $U\subset E$ it holds that

$$\lim_{t \to 0^+} p(t, x; U) = 1,$$

whenever $x \in U$.

Theorem 2.2.5 There exists a semigroup of linear operators $\{T(t)\}$ defined in $C_b(\mathbb{R}^N)$ such that, for any $f \in C_b(\mathbb{R}^N)$, the solution of the problem (2.0.5), given by Theorem 2.2.1, is represented by

$$u(t,x) = (T(t)f)(x), \quad t \ge 0, \ x \in \mathbb{R}^N.$$
 (2.2.6)

For any t > 0, T(t) satisfies the estimate

$$||T(t)f||_{\infty} \le \exp(c_0 t) ||f||_{\infty}, \quad f \in C_b(\mathbb{R}^N).$$
 (2.2.7)

Moreover, there exist a family of Borel measures p(t, x; dy) in \mathbb{R}^N such that

$$(T(t)f)(x) = \int_{\mathbb{R}^N} f(y)p(t, x; dy), \quad t \ge 0, \ x \in \mathbb{R}^N,$$
 (2.2.8)

and a function $G:(0,+\infty)\times\mathbb{R}^N\times\mathbb{R}^N\to\mathbb{R}$ such that

$$p(t, x; dy) = G(t, x, y)dy,$$
 $t > 0, x, y \in \mathbb{R}^{N}.$ (2.2.9)

The function G is strictly positive and the functions $G(t,\cdot,\cdot)$ and $G(t,x,\cdot)$ are measurable for any t>0 and any $x\in\mathbb{R}^N$. Further, for almost any fixed $y\in\mathbb{R}^N$, the function $G(\cdot,\cdot,y)$ belongs to the space $C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\mathbb{R}^N)$, and it is a solution of the equation $D_tu-Au=0$.

Finally, if $c_0 \leq 0$ then p(t, x; dy) is a stochastically continuous transition function.

Proof. Step 1: definition and properties of G. For any $k \in \mathbb{N}$ let $G_k \in C((0,+\infty) \times B(k) \times B(k))$ be the fundamental solution of the equation $D_t u - \mathcal{A}u = 0$ in B(k), given by Proposition C.3.2. We extend the function G_k to $(0,+\infty) \times \mathbb{R}^N \times \mathbb{R}^N$ with value zero for $x,y \notin B(k)$ and still denote by G_k the so obtained function. A straightforward computation shows that for any fixed $t \in (0,+\infty)$ and any $x,y \in \mathbb{R}^N$, the sequence $\{G_k(t,x,y)\}$ is increasing. Indeed, for any positive $f \in C(B(k))$, the function

$$w(t,x) = \int_{G(k)} f(y)(G_{k+1}(t,x,y) - G_k(t,x,y))dy, \quad t \ge 0, \ x \in \mathbb{R}^N,$$

is positive as well by virtue of Remark 2.2.2. Recalling that, for any t > 0 and any $x \in B(k)$, the function $G_{k+1}(t,x,\cdot) - G_k(t,x,\cdot)$ is continuous in B(k), we easily deduce that $G_{k+1}(t,x,y) \ge G_k(t,x,y)$ for any t > 0 and any $x,y \in B(k)$, implying that the sequence $\{G_k(t,x,y)\}$ is increasing. Hence, we can define the function $G: (0,+\infty) \times \mathbb{R}^N \times \mathbb{R}^N \to \mathbb{R}$ by setting

$$G(t, x, y) := \lim_{k \to +\infty} G_k(t, x, y), \qquad t > 0, \ x, y \in \mathbb{R}^N.$$

The function G is finite almost everywhere. Indeed, the estimate (C.3.4) with f = 1 yields

$$\int_{B(k)} G_k(t, x, y) dy \le \exp(c_0 t), \qquad t > 0, \ x \in B(k), \ k \in \mathbb{N},$$

and then, by monotone convergence,

$$\int_{\mathbb{R}^N} G(t, x, y) dy \le \exp(c_0 t), \qquad t > 0, \ x \in \mathbb{R}^N, \tag{2.2.10}$$

so that G(t, x, y) is finite for any t > 0, any $x \in \mathbb{R}^N$ and almost any $y \in \mathbb{R}^N$. Since $G_k(t, x, \cdot) > 0$ almost everywhere in B(k) for any t > 0 and any $x \in B(k)$, then G is strictly positive.

Of course, $G(t,\cdot,\cdot)$, $G(t,x,\cdot)$ and $G(t,\cdot,y)$ are measurable functions for any t>0 and any $x,y\in\mathbb{R}^N$ since they are the pointwise limit of measurable functions.

We now prove the regularity properties of G. Fix R, T > 0, $x_0 \in B(R)$ and let $y_0 \in \mathbb{R}^N$ be such that $G(T, x_0, y_0) < +\infty$; actually we have seen that this holds for almost any $y_0 \in \mathbb{R}^N$. If $h, k \in \mathbb{N}$ satisfy R+1 < h < k, the functions $G_h(\cdot, \cdot, y_0)$ and $G_k(\cdot, \cdot, y_0)$ are solutions of the equation $D_t u - \mathcal{A}u = 0$ in $(0, +\infty) \times B(R+1)$ (see Theorem C.1.4), and hence $G_k(\cdot, \cdot, y_0) - G_h(\cdot, \cdot, y_0)$ is as well. Moreover, $G_k(\cdot, \cdot, y_0) - G_h(\cdot, \cdot, y_0)$ is positive and, for any fixed $0 < t_0 < t_1 < T$, it satisfies the following Harnack inequality (see [99]):

$$\sup\{G_k(t, x, y_0) - G_h(t, x, y_0), (t, x) \in [t_0, t_1] \times \overline{B}(R)\}$$

$$\leq C \inf\{G_k(T, x, y_0) - G_h(T, x, y_0), x \in \overline{B}(R)\}$$

$$\leq C(G_k(T, x_0, y_0) - G_h(T, x_0, y_0)),$$

where C>0 is a constant, independent of h and k. Since $G(T,x_0,y_0)<+\infty$, then $\{G_n(\cdot,\cdot,y_0)\}$ turns out to be a Cauchy sequence in $C([t_0,t_1]\times\overline{B}(R))$. Since it converges pointwise to the function $G(t,x,y_0)$, we conclude that $G(\cdot,\cdot,y_0)\in C([t_0,t_1]\times\overline{B}(R))$. Moreover, from Theorem C.1.4 it follows that, for any $t'_0>t_0$, any $t'_1< t_1$ and any R'< R, the sequence $\{G_n(\cdot,\cdot,y_0)\}$ converges also in $C^{1+\beta/2,2+\beta}([t'_0,t'_1]\times\overline{B}(R'))$ for any $\beta\in(0,\alpha)$. Hence, $G(\cdot,\cdot,y_0)\in C^{1+\alpha/2,2+\alpha}([t'_0,t'_1]\times\overline{B}(R'))$. Since $T,R,R',t_0,t'_0,t_1,t'_1>0$ are arbitrary, we get $G(\cdot,\cdot,y_0)\in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\mathbb{R}^N)$.

Finally, since $D_tG_n - \mathcal{A}G_n = 0$, as n goes to $+\infty$, it follows that $D_tG - \mathcal{A}G = 0$.

Step 2: definition and properties of p(t, x; dy) and $\{T(t)\}$. Now, for any t > 0 and any $x \in \mathbb{R}^N$ we define the measure p(t, x; dy) by (2.2.9), while for t = 0 we set $p(t, x; dy) = \delta_x$. Then, for any t > 0, we define the operator T(t) by (2.2.8).

Let us prove that, for any $f \in C_b(\mathbb{R}^N)$, the solution u of the problem (2.0.5), found out in Theorem 2.2.1, is given by (2.2.6). Indeed,

$$u(t,x) = \lim_{k \to +\infty} \int_{\mathbb{R}^N} f(y)G_k(t,x,y)dy, \qquad t > 0, \ x \in \mathbb{R}^N,$$

and we can split it as

$$u(t,x) = \lim_{k \to +\infty} \int_{\mathbb{R}^N} f^+(y) G_k(t,x,y) dy$$
$$-\lim_{k \to +\infty} \int_{\mathbb{R}^N} f^-(y) G_k(t,x,y) dy.$$

By the monotone convergence theorem we immediately deduce that

$$u(t,x) = \int_{\mathbb{R}^N} f^+(y)G(t,x,y)dy - \int_{\mathbb{R}^N} f^-(y)G(t,x,y)dy$$
$$= \int_{\mathbb{R}^N} f(y)G(t,x,y)dy,$$

that is (2.2.6).

To show that $\{T(t)\}$ is a semigroup it suffices to observe that from (C.3.5), using the monotone convergence theorem, it follows that

$$G(t+s,x,y) = \int_{\mathbb{R}^N} G(s,x,z)G(t,z,y)dz, \quad t,s > 0, \ x,y \in \mathbb{R}^N. \quad (2.2.11)$$

Moreover, (2.2.7) is an immediate consequence of (2.2.1) and (2.2.6).

It remains to prove that, if $c_0 \leq 0$, $\{p(t,x;dy): t \geq 0, x \in \mathbb{R}^N\}$ is a stochastically continuous transition function. The condition (i) in Definition 2.2.4 follows immediately from (2.2.10), and the condition (ii) is obvious. The condition (iii) will follow from (2.2.11) once we have proved that the function $p(t,\cdot;B)$ is Borel measurable in \mathbb{R}^N , for any fixed $t\geq 0$ and any $B\in\mathcal{B}(\mathbb{R}^N)$. So, let us prove this property. If t=0 the property is clear. If t>0, consider a sequence of functions $\{f_n\}\subset C_b(\mathbb{R}^N)$ converging almost everywhere to $\chi_B(x)$ and such that $0\leq f_n\leq 1$ for any $n\in\mathbb{N}$. For instance, if B has finite positive Lebesgue measure, one can take, up to a subsequence, $f_n=\rho_n\star\chi_B$ $(n\in\mathbb{N})$, where ρ is a standard mollifier and " \star " denotes convolution. If B has infinite Lebesgue measure, one can consider the sequence of bounded Borel sets $B_M=B\cap B(M)$ and a subsequence $\{f_{n_k}^M\}_{k\in\mathbb{N}}$ of the sequence $f_n^M=\rho_n\star\chi_{B_M}$ $(n\in\mathbb{N})$ pointwise converging a.e. in \mathbb{R}^N to χ_{B_M} as n tends to $+\infty$, and assume that, for any $M\in\mathbb{N}$, $\{f_{n_k}^{M+1}\}_{k\in\mathbb{N}}$ is a subsequence of $\{f_{n_k}^M\}_{k\in\mathbb{N}}$. Then, one can define the wished sequence by setting $f_n=f_n^n$. By the dominated convergence theorem we have

$$p(t, x; B) = \lim_{n \to +\infty} \int_{\mathbb{R}^N} f_n(y) p(t, x; dy) = \lim_{n \to +\infty} (T(t) f_n)(x), \quad x \in \mathbb{R}^N.$$

Hence $p(t, \cdot; B)$ is Borel measurable, being the pointwise limit of a sequence of continuous functions.

Finally we show that p(t, x; dy) is stochastically continuous; clearly this implies that it is also normal. For this purpose, for any $x_0 \in \mathbb{R}^N$ and any

R>0, we introduce the function $f_{x_0,r}:\mathbb{R}^N\to\mathbb{R}$ defined by

$$f_{x_0,r}(x) = \begin{cases} 1 - r^{-1}|x - x_0|, & x \in x_0 + B(r), \\ 0, & x \notin x_0 + B(r). \end{cases}$$

As it is easily seen,

$$f_{x_0,r}(x_0) - (T(t)f_{x_0,r})(x_0) = 1 - \int_{\mathbb{R}^N} f_{x_0,r}(y)p(t,x_0;dy)$$
$$= 1 - \int_{x_0+B(r)} f_{x_0,r}(y)p(t,x_0;dy)$$
$$\geq 1 - p(t,x_0;x_0+B(r)).$$

Hence, letting t go to 0^+ , Theorem 2.2.1 yields

$$\lim_{t \to 0^+} p(t, x_0; x_0 + B(r)) = 1. \tag{2.2.12}$$

Now, we fix an arbitrary open set $U \subset \mathbb{R}^N$. Then, with any $x \in U$, we associate an open ball x + B(r) contained in U. Since

$$1 \ge p(t, x; U) \ge p(t, x; x + B(r)),$$

from (2.2.12) we deduce that p(t, x; U) tends to 1 as t tends to 0^+ and we are done.

Theorem 2.2.6 For any $\lambda > c_0$ we have

$$K_{\lambda}(x,y) = \int_{0}^{+\infty} e^{-\lambda t} G(t,x,y) dt, \qquad x,y \in \mathbb{R}^{N}, \tag{2.2.13}$$

and, for any $f \in C_b(\mathbb{R}^N)$,

$$(R(\lambda)f)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x)dt, \qquad x \in \mathbb{R}^N.$$
 (2.2.14)

Proof. The equality (2.2.13) is a consequence of (C.3.9) with $\Omega = B(n)$, (2.1.7) and the monotone convergence theorem. Then, (2.2.14) follows from (2.2.8), (2.2.13) and the Fubini theorem.

In general, $\{T(t)\}$ is not a strongly continuous semigroup neither in $C_b(\mathbb{R}^N)$ nor in $BUC(\mathbb{R}^N)$ (see [148, Theorem 4.2] and Proposition 9.2.6). Nevertheless, as a straightforward consequence of Theorems 2.2.1 and 2.2.5, we deduce that, for a general $f \in C_b(\mathbb{R}^N)$, T(t)f converges to f as t tends to 0, locally uniformly in \mathbb{R}^N . Actually, as next proposition shows, if f vanishes at infinity, then T(t)f converges to f in $C_b(\mathbb{R}^N)$, as t tends to 0.

Proposition 2.2.7 ([116], Prop. 4.3) For any function $f \in C_0(\mathbb{R}^N)$, T(t)f tends to f in $C_b(\mathbb{R}^N)$, as t tends to 0^+ .

Proof. We prove the statement assuming that $f \in C_c^{\infty}(\mathbb{R}^N)$. The general case then will follow by density. So, let us fix $f \in C_c^{\infty}(\mathbb{R}^N)$ and $x \in \mathbb{R}^N$. Moreover, let $k \in \mathbb{N}$ be such that B(k) contains both x and $\operatorname{supp}(f)$. Then $(A_k f)(x) = (Af)(x)$, where, as usual, A_k denotes the realization of the operator A in $C(\overline{B}(k))$ with homogeneous Dirichlet conditions. Let $u_k(t) = T_k(t)f$, where $\{T_k(t)\}$ is the analytic semigroup generated by A_k . For any t > 0 we have

$$\begin{aligned} u_k(t,x) - f(x) &= \int_0^t \frac{\partial}{\partial s} u_k(s,x) ds \\ &= \int_0^t (A_k T_k(s) f)(x) ds \\ &= \int_0^t (T_k(s) \mathcal{A} f)(x) ds \\ &= \int_0^t ds \int_{\mathbb{R}^N} G_k(s,x,y) \mathcal{A} f(y) dy, \end{aligned}$$

where we have extended $G_k(t, x, \cdot)$ to the whole of \mathbb{R}^N by setting $G_k(t, x, y) = 0$ for any $y \notin B(k)$. Letting k go to $+\infty$ from the dominated convergence theorem it follows that

$$|(T(t)f)(x) - f(x)| = \left| \int_0^t (T(s)\mathcal{A}f)(x)ds \right| \le \|\mathcal{A}f\|_{\infty} \int_0^t \exp(c_0s)ds.$$

Since $x \in \mathbb{R}^N$ is arbitrary, we conclude that

$$||T(t)f - f||_{\infty} \le ||Af||_{\infty} \int_0^t \exp(c_0 s) ds,$$

which proves the proposition.

Remark 2.2.8 The results of the previous proposition do not imply that the restriction of the semigroup to $C_0(\mathbb{R}^N)$ gives rise to a strongly continuous semigroup. Indeed, as it is shown in Section 5.3, in general, $\{T(t)\}$ does not map $C_0(\mathbb{R}^N)$ into itself.

Taking advantage of Theorem 2.2.5 we can prove some interesting properties of the semigroup $\{T(t)\}$.

Proposition 2.2.9 Let $\{f_n\} \subset C_b(\mathbb{R}^N)$ be a bounded sequence of continuous functions converging pointwise to a function $f \in C_b(\mathbb{R}^N)$ as n tends to $+\infty$. Then, $T(\cdot)f_n$ tends to $T(\cdot)f$ locally uniformly in $(0, +\infty) \times \mathbb{R}^N$.

Further, if f_n tends to f uniformly on compact subsets of \mathbb{R}^N , then $T(t)f_n$ converges to T(t)f locally uniformly in $[0, +\infty) \times \mathbb{R}^N$ as n tends to $+\infty$.

Proof. To prove the first part of the proof, we fix $0 < T_1 < T_2$, R > 0, a sequence $\{f_n\} \subset C_b(\mathbb{R}^N)$ converging pointwise to $f \in C_b(\mathbb{R}^N)$ and we prove that $T(\cdot)f_n$ converges to $T(\cdot)f$ in $[T_1,T_2] \times \overline{B}(R)$ as n tends to $+\infty$. As it is immediately seen from (2.2.8) and the dominated convergence theorem, $T(\cdot)f_n$ converges pointwise to $T(\cdot)f$ in $(0,+\infty) \times \mathbb{R}^N$ as n tends to $+\infty$.

Now let K > 0 be such that $\sup_{n \in \mathbb{N}} \|f_n\|_{\infty} \leq K$. Then, taking (2.2.7) into account, we easily deduce that $\sup_{n \in \mathbb{N}} \|T(t)f_n\|_{\infty} \leq K(e^{c_0T} \vee 1)$ for any $t \in [0, T]$. The interior Schauder estimates in Theorem C.1.4 then imply that the sequence $\{T(\cdot)f_n\}$ is bounded in $C^{1+\alpha/2,2+\alpha}([T_1,T_2]\times \overline{B}(R))$. Hence, by the Ascoli-Arzelà Theorem, there exists a subsequence $\{T_{n_k}(\cdot)\}$ converging uniformly in $[T_1,T_2]\times \overline{B}(R)$ to a function $v \in C^{1+\alpha/2,2+\alpha}([T_1,T_2]\times \overline{B}(R))$. Since, $T(\cdot)f_n$ converges pointwise to $T(\cdot)f$ in $(0,+\infty)\times \mathbb{R}^N$, we deduce that $v = T(\cdot)f$ and the whole sequence $\{T(\cdot)f_n\}$ converges to $T(\cdot)f$ uniformly in $[T_1,T_2]\times \overline{B}(R)$.

Now, we suppose that the sequence $\{f_n\} \subset C_b(\mathbb{R}^N)$ converges uniformly to f on compact subsets of \mathbb{R}^N and we show that, for any R, T > 0, $T(\cdot)f_n$ tends to $T(\cdot)f$ uniformly in $[0,T] \times \overline{B}(R)$. Possibly replacing f_n with $f_n - f$, we can suppose that $f \equiv 0$. Moreover, without loss of generality, we can also assume that $\sup_{n \in \mathbb{N}} ||f_n|| \leq 1$.

For any $n \in \mathbb{N}$, let $\varphi_n \in C_0(\mathbb{R}^N)$ be a nonnegative function such that $\chi_{B(n-1)} \leq \varphi_n \leq \chi_{B(n)}$. Moreover, for any $\varepsilon > 0$, let $C_{\varepsilon,R}$ be the set defined by

$$C_{\varepsilon,R} = \left\{ s \geq 0: \ \exists n \in \mathbb{N} \ \text{s.t.} \ \inf_{\substack{(t,x) \in [0,s] \times B(R)}} \left(T(t)(\varphi_n - 1)(x) \geq -\varepsilon \right\}.$$

Let us prove that $C_{\varepsilon,R} = [0, +\infty)$, for any $\varepsilon > 0$. For this purpose, we will show that $C_{\varepsilon,R}$ is both an open and closed interval. Note that $C_{\varepsilon,R}$ is nonempty since it contains 0. To show that $C_{\varepsilon,R}$ is closed, we fix $s \in \overline{C_{\varepsilon,R}}$, $s \neq 0$. Then, there exists a sequence $\{s_n\} \subset C_{\varepsilon,R}$ converging to s as n tends to $+\infty$. Without loss of generality, we can assume that $\{s_n\}$ is either decreasing or increasing. Of course, if $\{s_n\}$ is decreasing, then $s \in C_{\varepsilon,R}$. So, let us consider the case when $\{s_n\}$ is increasing. Since $s_1 \in C_{\varepsilon,R}$, there exists $n_1 \in \mathbb{N}$ such that

$$(T(t)(\varphi_{n_1} - 1))(x) \ge -\varepsilon, \qquad t \in [0, s_1], \quad x \in B(R).$$
 (2.2.15)

Recalling that $\{\varphi_n\}$ is an increasing sequence, it turns out that (2.2.15) is satisfied by any $n \geq n_1$.

By the first part of the proof, we know that $T(\cdot)(\varphi_n - 1)$ converges to 0 uniformly in $[s_1, s] \times B(R)$. Therefore, we can determine $n_0 \in \mathbb{N}$ such that

$$(T(t)(\varphi_n - 1))(x) \ge -\varepsilon, \qquad t \in [s_1, s], \quad \in B(R), \quad n \ge n_0.$$

Now, if we take $\hat{n} = n_0 \vee n_1$, we deduce that

$$(T(t)(\varphi_{\hat{n}} - 1))(x) \ge -\varepsilon, \qquad t \in [0, s], \quad \in B(R).$$

Hence, $s \in C_{\varepsilon,R}$.

To show that C_{ε} is open in $[0, +\infty)$, we fix $s \in C_{\varepsilon,R}$ and prove that, for any $\delta > 0$, $[s, \delta] \subset C_{\varepsilon,R}$. For this purpose, it suffices to argue as above, observing that $T(\cdot)(\varphi_n - 1)$ converges to 0, uniformly in $[s, \delta] \times B(R)$.

Now, since $p(t, x; B(m)) \ge (T(t)\varphi_m)(x)$ for any t > 0, any $x \in \mathbb{R}^N$ and any $m \in \mathbb{N}$, and $C_{\varepsilon,R} = [0, +\infty)$, we easily deduce that, for any arbitrarily fixed T > 0 and any R > 0, there exists $m \in \mathbb{N}$ such that

$$p(t, x; B(m)) \ge (T(t)1)(x) - \varepsilon = p(t, x; \mathbb{R}^N) - \varepsilon, \qquad t \in [0, T], \quad x \in B(R).$$

Therefore,

$$|(T(t)f_n)(x)| \le \int_{B(m)} |f_n(y)| p(t, x; dy) + \int_{\mathbb{R}^N \setminus B(m)} |f_n(y)| p(t, x; dy) dy$$

$$\le \sup_{y \in B(m)} |f_n(y)| + p(t, \mathbb{R}^N \setminus B(m))$$

$$\le \sup_{y \in B(m)} |f_n(y)| + \varepsilon,$$

for any $t \in [0, T]$ and any $x \in B(R)$. Now, the assertion follows.

Remark 2.2.10 Using the formula (2.2.8), the semigroup $\{T(t)\}$ can be extended to a semigroup (which we still denote by $\{T(t)\}$) in the space $B_b(\mathbb{R}^N)$ of all the bounded Borel measurable functions, and, for any $f \in B_b(\mathbb{R}^N)$ and any bounded sequence $\{f_n\} \in C_b(\mathbb{R}^N)$ converging pointwise to f, $(T(t)f_n)(x)$ converges to (T(t)f)(x) for any t > 0 and any $x \in \mathbb{R}^N$. Moreover, by (2.2.9) and (2.2.10) it follows that the estimate (2.2.7) holds also for $f \in B_b(\mathbb{R}^N)$. Similarly, if $\{f_n\} \in B_b(\mathbb{R}^N)$ is a bounded sequence converging pointwise to $f \in B_b(\mathbb{R}^N)$, then $T(\cdot)f_n$ converges to $T(\cdot)f$ pointwise in $[0, +\infty) \times \mathbb{R}^N$. Actually, taking the forthcoming Proposition 2.2.12, into account, we can easily show that $T(\cdot)f_n$ converges locally uniformly in $(0, +\infty) \times \mathbb{R}^N$. Indeed, if $[a, b] \times K$ is a compact set in $(0, +\infty) \times \mathbb{R}^N$, we can split $T(t)f_n = T(t - a/2)T(a/2)f_n$ for any $t \geq a/2$. By the above result and Proposition 2.2.12, the sequence $\{T(a/2)f_n\}$ is contained in $C_b(\mathbb{R}^N)$, is bounded and it converges pointwise to T(a/2)f, as n tends to $+\infty$. Proposition 2.2.9 now implies that $T(\cdot)f_n$ converges to $T(\cdot)f$ in $[a,b] \times K$, as n tends to $+\infty$.

Let us now prove that $\{T(t)\}$ is irreducible and has the strong Feller property. For this purpose, we recall the following definition.

Definition 2.2.11 A semigroup $\{S(t)\}$ in $B_b(\mathbb{R}^N)$ is irreducible if for any nonempty open set $U \subset \mathbb{R}^N$ it holds that

$$(S(t)\chi_U)(x) > 0,$$

for any t > 0 and any $x \in \mathbb{R}^N$. It has the strong Feller property if $S(t)f \in C_b(\mathbb{R}^N)$ for any $f \in B_b(\mathbb{R}^N)$.

Proposition 2.2.12 $\{T(t)\}$ is irreducible and has the strong Feller property.

Proof. Showing that the semigroup is irreducible is easy due to the fact that G is strictly positive (see Theorem 2.2.5).

To prove that $\{T(t)\}$ is strong Feller, fix $f \in B_b(\mathbb{R}^N)$ and let $\{f_n\} \in C_b(\mathbb{R}^N)$ be a bounded sequence converging pointwise to f as n tends to $+\infty$. Applying the interior Schauder estimates in Theorem C.1.4 we deduce that for any compact set $F \subset (0, +\infty) \times \mathbb{R}^N$ there exists a positive constant C = C(F) such that

$$||T(\cdot)f_n||_{C^{1+\alpha/2,2+\alpha}(F)} \le C||f_n||_{\infty}, \qquad n \in \mathbb{N}.$$

Since $\sup_{n\in\mathbb{N}} \|f_n\|_{\infty}$ is finite, and $(T(t)f_n)(x)$ converges to (T(t)f)(x) for any $t\in[0,+\infty)$ and any $x\in\mathbb{R}^N$ (this follows immediately applying the dominated convergence theorem to the formula (2.2.9)), we deduce that T(t)f is continuous in \mathbb{R}^N for any t>0.

Remark 2.2.13 The strictly positiveness of G actually implies that

$$(T(t)\chi_E)(x) > 0,$$
 $t > 0, x \in \mathbb{R}^N,$

for any Borel set $E\subset\mathbb{R}^N$ with positive Lebesgue measure.

2.3 The weak generator of T(t)

Since $\{T(t)\}$ is not strongly continuous in $C_b(\mathbb{R}^N)$ and in general is not strongly continuous in either $C_0(\mathbb{R}^N)$ or $BUC(\mathbb{R}^N)$, we cannot define the infinitesimal generator in the usual sense. Nevertheless, we can still associate a "generator" with $\{T(t)\}$, the so-called weak generator, which has properties similar to those of the infinitesimal generator.

We will provide three equivalent definitions of the weak generator. The first definition that we give was considered in [28] and in [77]: the resolvent operators $\{R(\lambda): \lambda > c_0\}$ given by Theorem 2.1.3 satisfy the resolvent identity (2.1.5) and $R(\lambda)$ is injective in $C_b(\mathbb{R}^N)$ for any $\lambda > c_0$. Hence, by a classical

result of functional analysis (see Proposition A.3.2), we can define the weak generator as the unique linear operator A_1 in $C_b(\mathbb{R}^N)$ such that

$$R(\lambda) = R(\lambda, A_1)$$
 and $Im(R(\lambda)) = D(A_1), \quad \lambda > c_0.$ (2.3.1)

The second definition is based on the bounded pointwise convergence: a sequence $\{f_n\} \subset C_b(\mathbb{R}^N)$ is said to be boundedly and pointwise convergent to $f \in C_b(\mathbb{R}^N)$ if there exists a positive constant C such that $||f_n||_{\infty} \leq C$ for any $n \in \mathbb{N}$ and if $f_n(x)$ converges to f(x) for any $x \in \mathbb{R}^N$ (see [53]). This notion of convergence leads to the following definition of the weak generator which was introduced in [125, 126]:

$$\begin{cases}
D(A_2) = \left\{ f \in C_b(\mathbb{R}^N) : \sup_{t \in (0,1)} \frac{\|T(t)f - f\|_{\infty}}{t} < +\infty \text{ and } \exists g \in C_b(\mathbb{R}^N) : \\
\lim_{t \to 0^+} \frac{(T(t)f)(x) - f(x)}{t} = g(x) \quad \forall x \in \mathbb{R}^N \right\}, \\
(A_2f)(x) = \lim_{t \to 0^+} \frac{(T(t)f)(x) - f(x)}{t}, \quad x \in \mathbb{R}^N, \quad f \in D(A_2).
\end{cases}$$
(2.3.2)

The third definition is based on the notion of mixed topology introduced in [145]. The mixed topology τ^M is the finest locally convex topology which agrees on every norm-bounded subset of $C_b(\mathbb{R}^N)$ with the topology of the uniform convergence on compact sets. Equivalently, it can be defined by the family of seminorms

$$p_{\{a_n\},\{K_n\}}(f) = \sup_{n \in \mathbb{N}} \left\{ a_n \sup_{x \in K_n} |f(x)| \right\}, \quad f \in C_b(\mathbb{R}^N),$$

where $\{a_n\}$ is any sequence of positive numbers converging to zero and $\{K_n\}$ is any sequence of compact subsets of \mathbb{R}^N . Given a sequence $\{f_n\} \subset C_b(\mathbb{R}^N)$ and a function $f \in C_b(\mathbb{R}^N)$ we have

$$\tau^{M} - \lim_{n \to +\infty} f_n = f \Longleftrightarrow \begin{cases} ||f_n||_{\infty} \le C, & n \in \mathbb{N}, \\ f_n \to f & \text{locally uniformly.} \end{cases}$$
 (2.3.3)

For results on transition semigroups and mixed topology we refer the reader to [68].

Thus, we can define the generator of the semigroup in the mixed topology, i.e., the operator $A_3: D(A_3) \to C_b(\mathbb{R}^N)$ defined by

$$\begin{cases}
D(A_3) = \left\{ f \in C_b(\mathbb{R}^N) : \exists g \in C_b(\mathbb{R}^N) : \tau^M - \lim_{t \to 0^+} \frac{T(t)f - f}{t} = g \right\}, \\
A_3 f = \tau^M - \lim_{t \to 0^+} \frac{T(t)f - f}{t}, \quad f \in D(A_3).
\end{cases}$$
(2.3.4)

Proposition 2.3.1 The three operators A_1 , A_2 and A_3 coincide.

To prove Proposition 2.3.1, we need two preliminary lemmata.

Lemma 2.3.2 For any $f \in C_b(\mathbb{R}^N)$, any t > 0, any $x \in \mathbb{R}^N$ and any $\lambda > c_0$ we have

$$(T(t)R(\lambda)f)(x) = \int_0^{+\infty} e^{-\lambda s} (T(t+s)f)(x)ds.$$
 (2.3.5)

Proof. We prove first that for any $\delta > 0$

$$T(t)\left(\int_0^\delta e^{-\lambda s} T(s) f ds\right)(x) = \int_0^\delta e^{-\lambda s} (T(t+s)f)(x) ds, \qquad t > 0 \ x \in \mathbb{R}^N.$$
(2.3.6)

To see it, it suffices to observe that, for any $x \in \mathbb{R}^N$, we have

$$\int_0^\delta e^{-\lambda s} (T(s)f)(x)ds = \lim_{k \to +\infty} \frac{1}{k} \sum_{j=0}^{k-1} e^{-\lambda \delta j/k} (T(\delta j/k)f)(x)$$
$$:= \lim_{k \to +\infty} \sigma_k(f)(x).$$

As it is immediately seen, $(T(t)\sigma_k(f))(x)$ converges to $\int_0^\delta e^{-\lambda s}(T(s+t)f)(x)ds$ as k tends to $+\infty$. On the other hand, since $\{\sigma_k(f)\}_{k\in\mathbb{N}}$ is a bounded sequence in $C_b(\mathbb{R}^N)$, then $(T(t)\sigma_k(f))(x)$ converges to $T(t)(\int_0^\delta e^{-\lambda s}T(s)f)(x)ds$ as k tends to $+\infty$, by Proposition 2.2.9. Thus, (2.3.6) follows.

Now, (2.3.5) follows from (2.3.6) letting δ tend to $+\infty$ and using again Proposition 2.2.9.

Lemma 2.3.3 For any $f \in D(A_2)$, any t > 0 and any $\lambda > c_0$ the functions T(t)f and $R(\lambda)f$ belong to $D(A_2)$. Moreover, $A_2T(t)f = T(t)A_2f$, $A_2R(\lambda)f = R(\lambda)A_2f$.

Proof. Fix $f \in D(A_2)$, t > 0 and K > 0 such that

$$\frac{\|T(h)f - f\|_{\infty}}{h} \le K, \qquad h \in (0, 1). \tag{2.3.7}$$

Then, taking (2.2.7) into account, we get

$$\frac{\|T(h)T(t)f - T(t)f\|_{\infty}}{h} = \left\|T(t)\left(\frac{T(h)f - f}{h}\right)\right\|_{\infty} \le e^{c_0 t}K, \qquad h \in (0, 1).$$

Moreover, since (T(h)f - f)/t converges to A_2f pointwise as h tends to 0^+ , from (2.3.7) and Proposition 2.2.9 we deduce that (T(h)(T(t)f - T(t)f)/h converges pointwise to $T(t)A_2f$ as h tends to 0^+ . That is, $T(t)f \in D(A_2)$ and $A_2T(t)f = T(t)A_2f$.

Next, from (2.3.5) we deduce that

$$\frac{(T(h)R(\lambda)f)(x) - (R(\lambda)f)(x)}{h} = \int_0^{+\infty} e^{-\lambda s} \frac{(T(h+s)f)(x) - (T(s)f)(x)}{h} ds$$
$$= \int_0^{+\infty} e^{-\lambda s} T(s) \left(\frac{T(h)f - f}{h}\right)(x) ds,$$

for any $\lambda > c_0$ and any $x \in \mathbb{R}^N$. Thus, by (2.3.7) we get

$$\frac{\|T(h)R(\lambda)f - R(\lambda)f\|_{\infty}}{h} \le \frac{K}{\lambda - c_0}, \qquad h \in (0, 1).$$
 (2.3.8)

Finally, by Proposition 2.2.9 and the dominated convergence theorem it follows that $(T(h)R(\lambda)f - R(\lambda)f)/h$ converges pointwise to $R(\lambda)A_2f$ as h tends to 0^+ . This implies that $R(\lambda)f \in D(A_2)$ and $A_2R(\lambda)f = R(\lambda)A_2f$.

Proof of Proposition 2.3.1. Let us prove the inclusion $A_1 \subset A_3$. For this purpose, fix $\lambda > c_0$ and $f \in C_b(\mathbb{R}^N)$. Taking (2.3.5) into account, we get

$$\frac{(T(t)R(\lambda)f)(x) - (R(\lambda)f)(x)}{t}$$

$$= \frac{1}{t} \left(\int_0^{+\infty} e^{-\lambda s} (T(s+t)f)(x) ds - \int_0^{+\infty} e^{-\lambda s} (T(s)f)(x) ds \right)$$

$$= \frac{1}{t} \left(\int_t^{+\infty} e^{-\lambda(s-t)} (T(s)f)(x) ds - \int_0^{+\infty} e^{-\lambda s} (T(s)f)(x) ds \right)$$

$$= \frac{e^{\lambda t} - 1}{t} (R(\lambda)f)(x) - \frac{e^{\lambda t}}{t} \int_0^t e^{-\lambda s} (T(s)f)(x) ds, \qquad (2.3.9)$$

for any $t \in (0, +\infty)$ and any $x \in \mathbb{R}^N$. From (2.3.8), we deduce that the function $(t, x) \mapsto t^{-1}(T(t)R(\lambda)f - R(\lambda)f)(x)$ is bounded in $(0, 1] \times \mathbb{R}^N$. Moreover, the right-hand side of (2.3.9) tends to $\lambda R(\lambda)f - f$ locally uniformly in \mathbb{R}^N as t tends to 0^+ . Indeed, for any R > 0 and any $x \in \overline{B}(R)$ it holds that

$$\begin{split} & \left| \frac{e^{\lambda t}}{t} \int_{0}^{t} e^{-\lambda s} (T(s)f)(x) ds - f(x) \right| \\ & \leq \left| \frac{e^{\lambda t}}{t} \int_{0}^{t} e^{-\lambda s} \left((T(s)f)(x) - f(x) \right) ds \right| + \frac{1}{t} \left| \int_{0}^{t} (e^{-\lambda(s-t)} - 1)f(x) ds \right| \\ & \leq e^{(\lambda \vee 0)t} \sup_{s \in [0,t]} \|T(s)f - f\|_{C(\overline{B}(R))} + \|f\|_{\infty} \frac{1}{t} \int_{0}^{t} (e^{-\lambda(s-t)} - 1) ds, \quad (2.3.10) \end{split}$$

and the last side of (2.3.10) converges to 0 as t tends to 0^+ . By the characterization (2.3.3), it follows that $(T(t)R(\lambda)f - R(\lambda)f)/t$ converges to $\lambda R(\lambda)f - f$

in the mixed topology. This implies that $R(\lambda)f \in D(A_3)$ and

$$(A_3 R(\lambda) f)(x) = \lambda(R(\lambda) f)(x) - f(x) = (A_1 R(\lambda) f)(x), \qquad x \in \mathbb{R}^N.$$
(2.3.11)

Therefore, we conclude that $A_1 \subset A_3$.

The inclusion $A_3 \subset A_2$ is clear from the characterization (2.3.3).

Thus, we conclude the proof by proving that $A_2 \subset A_1$. For this purpose, let $f \in D(A_2)$ and fix $\lambda > c_0$. From Lemma 2.3.3 and (2.3.11) we deduce that

$$f = (\lambda - A_1)R(\lambda)f = (\lambda - A_2)R(\lambda)f = R(\lambda)(\lambda f - A_2 f),$$

implying that $f \in D(A_1)$ and $A_1 f = A_2 f$.

Definition 2.3.4 The operator $\widehat{A} := A_1 = A_2 = A_3$, as defined in (2.3.1), (2.3.2) and (2.3.4), is called the weak generator of $\{T(t)\}$.

The weak generator fulfills the following further properties.

Proposition 2.3.5 For any $f \in D(\widehat{A})$ and any fixed $x \in \mathbb{R}^N$, the function $(T(\cdot)f)(x)$ is continuously differentiable in $[0, +\infty)$ and

$$\frac{d}{dt}(T(t)f)(x) = (T(t)\widehat{A}f)(x), \qquad t \ge 0.$$
(2.3.12)

For any sequence $\{f_n\} \subset D(\widehat{A})$ such that f_n and $\widehat{A}f_n$ converge boundedly and pointwise to some functions $f, g \in C_b(\mathbb{R}^N)$, respectively, it holds that $f \in D(\widehat{A})$ and $\widehat{A}f = g$.

Finally, $D(\widehat{A})$ is dense in $C_b(\mathbb{R}^N)$ in the mixed topology.

Proof. Fix $f \in D(\widehat{A})$ and $x \in \mathbb{R}^N$. By Lemma 2.3.3 the right derivative

$$\frac{d^{+}}{dt}(T(t)f)(x) := \lim_{h \to 0^{+}} \frac{(T(t+h)f)(x) - (T(t)f)(x)}{h}$$

exists at any $t \ge 0$ and

$$\frac{d^+}{dt}(T(t)f)(x) = (T(t)\widehat{A}f)(x).$$

Moreover, by Theorem 2.2.5 the function $t \mapsto (T(t)\widehat{A}f)(x)$ is continuous in $[0, +\infty)$. Hence, $(T(\cdot)f)(x)$ is differentiable in $[0, +\infty)$ and (2.3.12) holds.

Next, let $\{f_n\} \subset D(\widehat{A})$ be as in the statement. By the previous step, for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, the function $(T(\cdot)f_n)(x)$ is differentiable in $[0, +\infty)$ and

$$\frac{d}{ds}(T(s)f_n)(x) = (T(s)\widehat{A}f_n)(x), \qquad s \ge 0.$$

Integrating such an equation with respect to $s \in [0, t]$ gives

$$\frac{(T(t)f_n)(x) - f_n(x)}{t} = \frac{1}{t} \int_0^t (T(s)\widehat{A}f_n)(x)ds, \qquad t \ge 0.$$

Letting n go to $+\infty$ and taking Proposition 2.2.9 into account, from the dominated convergence theorem we get

$$\frac{(T(t)f)(x) - f(x)}{t} = \frac{1}{t} \int_0^t (T(s)g)(x)ds.$$

From such an equality we immediately deduce that $f \in D(\widehat{A})$ and $\widehat{A}f = g$.

Finally, to prove that $D(\widehat{A})$ is dense in $C_b(\mathbb{R}^N)$ in the mixed topology it suffices to observe that any bounded and continuous function can be approximated by a sequence of functions $f_n \in C_c^{\infty}(\mathbb{R}^N)$ $(n \in \mathbb{N})$, bounded in the sup-norm and converging locally uniformly to f as n tends to $+\infty$.

Next proposition shows the connections between $D_{\max}(A)$ and $D(\widehat{A})$.

Proposition 2.3.6 We have

$$D_{\max}(\mathcal{A}) \cap C_0(\mathbb{R}^N) \subseteq D(\widehat{A}) \subseteq D_{\max}(\mathcal{A}),$$
 (2.3.13)

where $D_{\max}(A)$ is the set defined in (2.0.1).

Moreover, the following conditions are equivalent:

- (i) $\lambda \in \rho(A)$ for some $\lambda > c_0$, i.e., the equation (2.0.2) has a unique bounded solution $u \in D_{\max}(A)$;
- (ii) $(c_0, +\infty) \subset \rho(A)$;
- (iii) $(A, D_{\max}(\mathcal{A})) = (\widehat{A}, D(\widehat{A})).$

Proof. To prove the inclusion $D_{\max}(\mathcal{A}) \cap C_0(\mathbb{R}^N) \subseteq D(\widehat{A})$ we fix a function $u \in D_{\max}(\mathcal{A}) \cap C_0(\mathbb{R}^N)$, $\lambda > c_0$ and set $f = \lambda u - \mathcal{A}u$. As it is easily seen $f \in C_b(\mathbb{R}^N)$. For any $n \in \mathbb{N}$, let u_n be, as in the proof of Theorem 2.1.1, the solution of the elliptic problem

$$\begin{cases} \lambda u_n - \mathcal{A}u_n = f, & \text{in } B(n), \\ u_n = 0, & \text{on } \partial B(n). \end{cases}$$

The function $u_n - u$ solves the differential equation $\lambda v - Av = 0$ and $u_n - u = -u$ on $\partial B(n)$. By the classical maximum principle, it follows that

$$||u_n - u||_{C(\overline{B}(n))} = \sup_{|x|=n} |u(x)|$$

and, consequently, u_n converges to u as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . On the other hand, from the proof of Theorem 2.1.1 we know that u_n

converges to the function $R(\lambda)f$ as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . It follows that $R(\lambda)f = u$ and, consequently, $u \in D(\widehat{A})$.

The inclusion $D(\widehat{A}) \subseteq D_{\max}(A)$ follows from Theorems 2.1.1 and 2.1.3.

Now, we prove the second part of the proposition. We limit ourselves to showing that "(i) \Rightarrow (iii)" and "(iii) \Rightarrow (ii)", since "(ii) \Rightarrow (i)" is trivial.

"(i) \Rightarrow (iii)". By Theorems 2.1.1 and 2.1.3 it follows immediately that $\widehat{A} \subset A$. Hence, we only need to prove that $A \subset \widehat{A}$. For this purpose, fix $u \in D_{\max}(A)$ and set $f = \lambda u - Au$ and $v = R(\lambda, \widehat{A})f$. Since $\widehat{A} \subset A$, we have $\lambda v - Av = f$. From the property (i), it follows that $u = v \in D(\widehat{A})$, and, therefore, the property (iii) follows.

"(iii) \Rightarrow (ii)". We observe that the property (iii) implies that $\rho(A) = \rho(A)$, which yields the property (ii) by virtue of Theorems 2.1.1 and 2.1.3.

We will give some examples of situations in which $D_{\max}(A) \neq D(\widehat{A})$, in Chapter 4, showing that, in general, the problem (2.0.5) is not uniquely solvable in $C_b(\mathbb{R}^N)$.

2.4 The Markov process

In this section we briefly consider the Markov process associated with the semigroup $\{T(t)\}$ and we show the Dynkin formula. In the whole section we assume that

$$c(x) \le 0, \qquad x \in \mathbb{R}^N. \tag{2.4.1}$$

We introduce a few notations. Let E be a topological space and let \mathcal{B} be the σ -algebra of Borel subsets of E. Moreover, let Ω be an arbitrary set, \mathcal{F} be a σ -algebra on it and $\tau: \Omega \to [0, +\infty]$ be a \mathcal{F} -measurable function. For any $t \geq 0$, we denote by \mathcal{F}_t a σ -algebra on the set $\Omega_t = \{\omega : t < \tau(\omega)\}$, such that $(\mathcal{F}_s)_{|_{\Omega_*}} \subset \mathcal{F}_t \subset \mathcal{F}$ for any 0 < s < t.

Next, we denote by $X = \{X_t : \Omega_t \to E, t \geq 0\}$ a family of functions defined in Ω such that, for any $\omega \in \Omega$, $X_t(\omega) \in E$ is a trajectory defined for $t \in [0, \tau(\omega))$ and such that X_t is \mathcal{F}_t -measurable on Ω_t . Finally, let $\{\mathbb{P}_x : x \in E\}$ be a family of probability measures on (Ω, \mathcal{F}_0) such that the function

$$x \mapsto p(t, x; B) = \mathbb{P}_x(X_t \in B) \tag{2.4.2}$$

is Borel measurable for any fixed t > 0 and $B \in \mathcal{B}$, and such that $\mathbb{P}_x(X_0 = x) = 1$.

The following definition of Markov process is taken from [49].

Definition 2.4.1 X is a Markov process if for any $x \in E$, any $s,t \geq 0$ and

any $B \in \mathcal{B}$ we have

$$\mathbb{P}_x(X_{t+s} \in B | \mathcal{F}_t) = \mathbb{P}_{X_t}(X_s \in B), \quad \mathbb{P}_x \text{-a.s.}, \tag{2.4.3}$$

and for any $\omega \in \Omega_t$ there exists $\omega' \in \Omega_t$ such that $\tau(\omega') = \tau(\omega) - t$ and $X_s(\omega') = X_{t+s}(\omega)$ for any $s \in [0, \tau(\omega'))$.

We say that X is continuous if all the trajectories are continuous. Moreover, we say that two Markov processes are equivalent if they have same transition probabilities $\{p(t, x; dy)\}$.

Definition 2.4.2 A random variable τ' with values in $[0, +\infty]$ is a Markov time of a Markov process X if $\tau' \le \tau$ and $\{t < \tau'\} \in \mathcal{F}_t$ for any t > 0.

The family of measures $\{p(t, x; dy)\}$ defined in (2.4.2) is a transition function (see Definition 2.2.4). Indeed, the conditions (i) and (ii) in Definition 2.2.9 are straightforward, while the condition (iii) follows from (2.4.3). Indeed, for any $B \in \mathcal{B}$ we have

$$p(t+s, x; B) = \mathbb{P}_x(X_{t+s} \in B)$$

$$= \mathbb{E}_x \mathbb{P}_x(X_{t+s} \in B | \mathcal{F}_t)$$

$$= \mathbb{E}_x p(s, X_t; B)$$

$$= \int_E p(s, y; B) p(t, x; dy).$$

Conversely, given a transition function $\{p(t, x; dy)\}$, we say that a Markov process X is associated with it if (2.4.2) holds. It is known that for any normal transition function there exists an associated Markov process; see [49, Theorem 3.2]. In particular, as far as the semigroup $\{T(t)\}$ is concerned, we have the following result.

Theorem 2.4.3 There exists a continuous Markov process X associated with the semigroup $\{T(t)\}$. We have

$$(T(t)f)(x) = \mathbb{E}_x \chi_{t < \tau} f(X_t), \qquad t > 0, \ x \in \mathbb{R}^N, \tag{2.4.4}$$

and

$$(R(\lambda)f)(x) = \mathbb{E}_x \int_0^\tau e^{-\lambda s} f(X_s) ds, \qquad \lambda > 0, \ x \in \mathbb{R}^N,$$
 (2.4.5)

for any $f \in B_b(\mathbb{R}^N)$.

Proof. The existence of X is proved in [49, Theorem 3.2], using the fact that the transition function of $\{T(t)\}$ is normal (see Theorem 2.2.5). See also [65, Theorem 1.6.3]. The continuity of X is proved in [10].

Then (2.4.4) is straightforward: if $f \equiv \chi_B$ is the characteristic function of a Borel set $B \subset \mathbb{R}^N$, then (2.4.4) is just (2.4.2). By linearity, (2.4.4) can be

extended, first, to any simple function f and, then, to any $f \in B_b(\mathbb{R}^N)$, by approximating with simple functions.

Finally, (2.4.5) follows from the resolvent formula (2.2.14) and from (2.4.4), applying the Fubini theorem.

For any set $U \subset \mathbb{R}^N$, we define the first exit time of X from U by

$$\tau_U = \inf\{t : X_t \notin U\}, \quad x \in U, \tag{2.4.6}$$

and we denote by X^U the process induced by X in U, that is

$$X_t^U = \begin{cases} X_t, & t < \tau_U, \\ \infty, & t \ge \tau_U, \end{cases}$$

and we recall the following result (see [10]).

Theorem 2.4.4 Let $U \subset \mathbb{R}^N$ be a regular bounded domain. Then X^U is the Markov process associated with the semigroup $\{T^U(t)\}$.

In the next theorem we state the Dynkin formula, in a slightly different form from the one in [49, Theorem 5.1].

Theorem 2.4.5 Let $U \in \mathbb{R}^N$ be a regular bounded domain, let $\tau' \leq \tau_U$ be a Markov time, and let $\lambda \geq 0$. Let $u \in W^{2,p}_{loc}(\mathbb{R}^N)$ for any $p \in [1, +\infty)$ be such that $Au \in C(\mathbb{R}^N)$; then we have

$$u(x) = \mathbb{E}_x e^{-\lambda \tau'} u(X_{\tau'}) + \mathbb{E}_x \int_0^{\tau'} e^{-\lambda s} (\lambda u - \mathcal{A}u)(X_s) ds, \quad x \in U. \quad (2.4.7)$$

Proof. Let $U_1 \subset \mathbb{R}^N$ be a regular bounded domain such that $\overline{U} \subset U_1$, and let $\vartheta \in C_c^{\infty}(U_1)$ be a function such that $\vartheta \equiv 1$ in U. Define $u' = \vartheta u$ and $f' = \lambda u' - \mathcal{A}u'$. Then we have $f' \in C_c(U_1)$ and $u' = R(\lambda, \mathcal{A}_{U_1})f'$, where $R(\lambda, \mathcal{A}_{U_1})$ is the resolvent operator of $\{T^{U_1}(t)\}$.

In [10, Theorem 1.6] the author proves that X is a strong Markov process; see also [65, Theorem 1.6.3]. Then also X^{U_1} is strong Markov. So we can apply [49, Theorem 5.1] to X^{U_1} , and we have the formula

$$u'(x) = \mathbb{E}_x e^{-\lambda \tau'} u'(X_{\tau'}) + \mathbb{E}_x \int_0^{\tau'} e^{-\lambda s} f'(X_s) ds, \qquad x \in U_1.$$

Since $\tau' \leq \tau_U$ and since u = u' and $f' = \lambda u - \mathcal{A}u$ in U, the restriction of this formula to $x \in U$ gives (2.4.7).

Notice that, taking $\tau' = \tau_U$ in (2.4.7), it follows that the solution of the boundary value problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in U, \\ u(x) = h(x), & x \in \partial U, \end{cases}$$

with $h \in C(\partial U)$, $f \in C(\overline{U})$ and $\lambda \geq 0$, can be represented by the formula

$$u(x) = \mathbb{E}_x e^{-\lambda \tau_U} h(X_{\tau_U}) - \mathbb{E}_x \int_0^{\tau_U} e^{-\lambda s} f(X_s) ds, \quad x \in U.$$
 (2.4.8)

2.5 The associated stochastic differential equation

In this section we consider the stochastic differential equation associated with the differential operator \mathcal{A} .

Let $\{W_t: t \geq 0\}$ be a N-dimensional Wiener process and let $\{\mathcal{F}_t^W: t \geq 0\}$ be the filtration generated by W_t . For any $x \in \mathbb{R}^N$, let $\sigma(x) \in \mathcal{L}(\mathbb{R}^N)$ be the unique positive definite matrix such that $Q(x) = \frac{1}{2}\sigma(x)\sigma(x)^*$. We consider the stochastic differential equation

$$\begin{cases}
d\xi_t^x = b(\xi_t^x)dt + \sigma(\xi_t^x)dW_t, & t > 0, \\
\xi_0^x \equiv x,
\end{cases}$$
(2.5.1)

where $x \in \mathbb{R}^N$ is fixed. This equation is a short writing of the integral equation

$$\xi_t^x = x + \int_0^t b(\xi_s^x) ds + \int_0^t \sigma(\xi_s^x) dW_s, \quad t > 0.$$
 (2.5.2)

We say that ξ_t^x is a solution of the problem (2.5.1) if it is a continuous real process, defined for any $t \geq 0$, adapted to the filtration $\{\mathcal{F}_t^W\}$ and such that (2.5.2) holds almost surely. Moreover, the continuity of ξ_t^x ensures the existence of the stochastic integral in (2.5.2). Note that the continuity of ξ_t^x ensures the existence of the stochastic integral in (2.5.2).

Let us assume the following hypothesis.

Hypothesis 2.5.1 The functions b and σ are continuous and satisfy

$$\|\sigma(x) - \sigma(y)\|_2^2 + 2\langle b(x) - b(y), x - y \rangle \le K_R |x - y|^2, \qquad \quad x, y \in B(R), \ (2.5.3)$$

$$\mathcal{A}(1+|x|^2) = \|\sigma(x)\|_2^2 + 2\langle b(x), x \rangle \le K(1+|x|^2), \qquad x \in \mathbb{R}^N, \quad (2.5.4)$$

where $K, K_R > 0$ are constants and $\|\sigma\|_2^2 = \text{Tr}(\sigma\sigma^*)$.

Theorem 2.5.2 Assume Hypothesis 2.5.1. Then, there exists a unique (up to equivalence) solution ξ_t^x of the problem (2.5.1), which is equivalent to the Markov process X.

Proof. For the proof of the existence and uniqueness of the solution ξ_t^x of the equation (2.5.1) we refer the reader to [53, Theorems 3.7 & 3.11] or [88, Theorem V.1.1]. Then, using the Itô formula we can see that ξ_t^x is equivalent

to X_t . Indeed, fix $t_0 > 0$ and define $v(t, x) = (T(t_0 - t)f)(x)$ for $t \in [0, t_0]$ and $x \in \mathbb{R}^N$. For any $n \in \mathbb{N}$, let $\tau_n(x)$ be the first exit time of ξ_t^x from the ball B(n). Since ξ_t^x is continuous and defined for any t > 0, we have

$$\lim_{n \to +\infty} \tau_n(x) = +\infty, \qquad x \in \mathbb{R}^N, \tag{2.5.5}$$

almost surely. Since $D_t v + \mathcal{A}v = 0$ in $(0, t_0) \times \mathbb{R}^N$, the Itô formula gives

$$v(t \wedge \tau_n(x), \xi_{t \wedge \tau_n(x)}^x) = v(0, x) + \int_0^{t \wedge \tau_n(x)} \sigma(\xi_s^x) Dv(s, \xi_s^x) dW_s, \qquad (2.5.6)$$

for any $t \in [0, t_0]$ and any $x \in B(n)$. If $t < t_0$, then, for any $s \in (0, t \wedge \tau_n(x))$, we have

$$|\sigma(\xi^x_s)Dv(s,\xi^x_s)| \leq \sup_{B(n)} |\sigma| \sup_{[0,t]\times B(n)} |Dv| < +\infty$$

and, therefore,

$$\mathbb{E} \int_0^{t \wedge \tau_n(x)} |\sigma(\xi_s^x) Dv(s, \xi_s^x)|^2 ds < +\infty.$$

Then, the expectation of the stochastic integral in (2.5.6) is zero and we get

$$\mathbb{E} v(t \wedge \tau_n(x), \xi_{t \wedge \tau_n(x)}(x)) = v(0, x) = (T(t_0)f)(x),$$

for any $t < t_0$, $x \in B(n)$, $n \in \mathbb{N}$. Letting n tend to $+\infty$, from (2.5.5) and the dominated convergence theorem it follows that

$$\mathbb{E} v(t, \xi_t^x) = (T(t_0)f)(x), \qquad t < t_0,$$

and finally, letting t tend to t_0 , we get $\mathbb{E} f(\xi^x_{t_0}) = (T(t_0)f)(x)$. This is equivalent to say that $\mathbb{P}(\xi^x_t \in B) = p(t, x; B)$ for any $t \geq 0$ and any $x \in \mathbb{R}^N$ and any Borel set B, where $\{p(t, x; dy)\}$ is the transition function of X. Therefore, ξ and X are equivalent.

Chapter 3

One-dimensional theory

3.0 Introduction

The one-dimensional case is particular and easier. It has been studied by Feller ([57]). Here, we consider the case when the second-order differential operator \mathcal{A} is defined by

$$\mathcal{A}\varphi(x) = q(x)\varphi''(x) + b(x)\varphi'(x), \qquad x \in \mathbb{R}$$

on smooth functions under the following assumptions on the coefficients q and h

Hypothesis 3.0.1 The coefficients q and b are continuous in \mathbb{R} . Moreover

$$q(x) > 0, \qquad x \in \mathbb{R}.$$

Under such a hypothesis, the existence of a solution of the elliptic equation,

$$\lambda u - \mathcal{A}u = f, (3.0.1)$$

in $D_{\max}(\mathcal{A}) = C_b(\mathbb{R}) \cap C_b^2(\mathbb{R})$ can be proved. Note that Hypothesis 3.0.1 is weaker than those in Chapter 2.

A deep analysis of the solutions to the homogeneous equation

$$\lambda u - \mathcal{A}u = 0$$

allows us to prove that, for any $f \in C_b(\mathbb{R})$, there exists at least a solution $u = R_{\lambda}f$ to the equation (3.0.1) which is in $C_b(\mathbb{R}) \cap C^2(\mathbb{R})$. We see that the boundedness at infinity of the solutions to the homogeneous equation is strictly connected to the integrability at infinity of some functions Q and R which depend on the coefficients q and q. This analysis will allow us to determine (integral) conditions on the coefficients which guarantee the uniqueness of the solution $u \in C_b(\mathbb{R}) \cap C^2(\mathbb{R})$ to the equation (3.0.1).

Then, we show that the arguments used in Chapter 2 to find out a solution $u = R(\lambda)f \in D_{\max}(A)$ to (3.0.1) apply also in this situation in which the coefficients are less regular. Moreover, we show that $R_{\lambda}f$ and $R(\lambda)f$ actually coincide.

Finally, under suitable additional hypotheses on q and b, we prove that

$$D_{\max}(\mathcal{A}) = \{ u \in C_b^2(\mathbb{R}) : qu'', bu' \in C_b(\mathbb{R}) \}.$$

3.1 The homogeneous equation

In this section, for any fixed $\lambda > 0$, we study the homogeneous equation

$$\lambda u - (qu'' + bu') = 0. (3.1.1)$$

Remark 3.1.1 Let $u \in C^2(\mathbb{R})$ be a solution to the equation (3.1.1). Then, u can attain neither a positive maximum nor a negative minimum. Indeed, suppose for instance that $x_0 \in \mathbb{R}$ is a positive maximum of the function u. Then, $u'(x_0) = 0$ and $u''(x_0) \leq 0$. From (3.1.1) we get $\lambda u(x_0) = q(x_0)u''(x_0) \leq 0$, which leads us to a contradiction. If u has a negative minimum it suffices to apply the previous argument to the function -u. As a consequence, if u solves (3.1.1) and vanishes at two different points x_0 and x_1 , then u vanishes identically in \mathbb{R} .

The following lemma is a crucial step in order to understand the behaviour of the solutions of (3.1.1) in a neighborhood of $-\infty$ and $+\infty$.

Lemma 3.1.2 There exist a positive decreasing function \overline{u}_1 and a positive increasing function \overline{u}_2 , which solve the equation (3.1.1).

Proof. To prove the existence of a positive decreasing solution \overline{u}_1 to the equation (3.1.1), we introduce the set

$$B = \{b \in \mathbb{R} : \exists u_b \text{ solution to } (3.1.1) \text{ with } u_b(0) = 1, \ u'_b(0) = b$$

and $u_b(x) = 0 \text{ for some } x > 0\}.$ (3.1.2)

Let us prove that B is an interval. First of all, we observe that B is not empty. Indeed, denote by v_1 and v_2 two linearly independent solutions to the equation (3.1.1). Then, the more general solution to (3.1.1) is given by $v = c_1v_1 + c_2v_2$. By the last part of Remark 3.1.1 the matrix whose rows are $(v_1(0), v_2(0))$ and $(v_1(x_0), v_2(x_0))$ is invertible for any $x_0 > 0$, since $v \equiv 0$ is the unique solution to the equation (3.1.1) such that $v(0) = v(x_0) = 0$. This implies that for any $x_0 > 0$ there exists a unique solution v to (3.1.1) such that v(0) = 1, $v(x_0) = 0$. Hence, a = v'(0) belongs to B.

To prove that B is an interval, we show that, if $b \in B$, then $(-\infty, b] \subset B$. For this purpose we observe that, if c < b, then $u_c < u_b$ in $(0, +\infty)$. Indeed, since $u'_c(0) < u'_b(0)$, there exists at least an interval $(0, x_1)$ in which $u_c < u_b$. Suppose that $x_1 < +\infty$. Then, $u_b(x_1) = u_c(x_1)$. This would imply that the function $v = u_b - u_c$, which solves the equation (3.1.1), should have two zeroes. Hence it should be constant in \mathbb{R} , which is a contradiction. Therefore, $u_c < u_b$ in $(0, +\infty)$ and, consequently, u_c vanishes at some point $x \in (0, +\infty)$, so that $c \in B$.

We now show that, if $b \in B$, then u_b is decreasing in \mathbb{R} . As a consequence we deduce that $B \subset (-\infty, 0]$. Let $b \in B$ and let x > 0 be the unique zero of

the function u_b . Then, u_b is decreasing in $(-\infty, x)$, otherwise it should have a positive maximum. Similarly, u_b is decreasing in $[x, +\infty)$. To check it, we observe that $u_b'(x) < 0$. Indeed, if $u_b'(x) = 0$, then u should coincide with the null solution. Therefore, u_b is strictly decreasing in a neighborhood of x. If u_b were not decreasing in $(x, +\infty)$ it should have a negative minimum, which is a contradiction.

We now set $\overline{b}=\sup B$, and $\overline{u}_1=u_{\overline{b}}$. Observe that \overline{u}_1 is decreasing in \mathbb{R} . Indeed, u_b converges to \overline{u}_1 in $C^2([-M,M])$ for any M>0, as b tends to \overline{b} from the left. Since $u_b'\leq 0$ for any $b<\overline{b}$, it follows that $\overline{u}_1'\leq 0$ as well. Let us now prove that \overline{u}_1 is positive in \mathbb{R} . This is equivalent to proving that $\overline{b}\notin B$. By contradiction, suppose that $\overline{b}\in B$. Let x>0 be a positive zero of \overline{u}_1 and let u be the solution to (3.1.1) such that u(0)=1 and u(2x)=0. Arguing as above, we can show that $u\geq \overline{u}_1$. Hence, $u'(0)>\overline{b}$, and this, of course, leads us to a contradiction.

Finally, to prove the existence of a positive increasing solution to (3.1.1) it suffices to set $\overline{u}_2(x) = v(-x)$ for any $x \in \mathbb{R}$, where v is the positive solution to the equation $\lambda v(x) - q(-x)v''(x) + b(-x)v'(x) = 0$, provided by the previous arguments.

The following proposition describes the behaviour of the solutions to the equation (3.1.1). For this purpose, we introduce the functions

$$W(x) = \exp\left(-\int_0^x \frac{b(s)}{q(s)} ds\right), \qquad x \in \mathbb{R}, \tag{3.1.3}$$

$$Q(x) = \frac{1}{q(x)W(x)} \int_0^x W(s)ds, \qquad x \in \mathbb{R}, \tag{3.1.4}$$

$$R(x) = W(x) \int_0^x \frac{1}{q(s)W(s)} ds, \qquad x \in \mathbb{R}.$$
 (3.1.5)

Moreover, we observe that $u \in C^2(\mathbb{R})$ is a solution to the equation (3.1.1) if and only if u solves the differential equation

$$\left(\frac{u'}{W}\right)' = \lambda \frac{u}{qW}.\tag{3.1.6}$$

Therefore, any solution u of (3.1.1) satisfies

$$u'(x) = W(x) \left(u'(0) + \lambda \int_0^x \frac{u(s)}{q(s)W(s)} ds \right), \qquad x \in \mathbb{R}.$$
 (3.1.7)

Remark 3.1.3 Notice that

$$\overline{u}_1(x)\overline{u}_2'(x) - \overline{u}_1'(x)\overline{u}_2(x) = w_0W(x), \qquad x \in \mathbb{R}, \tag{3.1.8}$$

for some positive constant w_0 . The formula (3.1.8) is immediately checked since the function $W^{-1}(\overline{u}_1\overline{u}_2' - \overline{u}_1'\overline{u}_2)$ is positive (by virtue of Lemma 3.1.2) and its first-order derivative identically vanishes in \mathbb{R} .

Proposition 3.1.4 The following properties are met:

- (i) all the solutions to (3.1.1) admit finite limit at $+\infty$ if and only if the function R belongs to $L^1(0,+\infty)$;
- (ii) if $Q \in L^1(0,+\infty)$ and $R \notin L^1(0,+\infty)$ then any positive decreasing solution to (3.1.1) satisfies

$$\lim_{x \to +\infty} \frac{u'(x)}{W(x)} = 0;$$

(iii) if Q, R belong to $L^1(0,+\infty)$ then, for any solution u to (3.1.1), the limits $\lim_{x\to+\infty} u(x)$ and $\lim_{x\to+\infty} u'(x)/W(x)$ exist and are finite. Moreover, there exist two decreasing solutions u and v of (3.1.1) such that

(i)
$$\lim_{x \to +\infty} u(x) = 0, \qquad \lim_{x \to +\infty} \frac{u'(x)}{W(x)} = -1,$$
(ii)
$$\lim_{x \to +\infty} v(x) = 1, \qquad \lim_{x \to +\infty} \frac{v'(x)}{W(x)} = 0;$$
(3.1.9)

(iv) the equation (3.1.1) admits a decreasing solution with $\lim_{x\to+\infty} u(x) > 0$ if and only if $Q \in L^1(0,+\infty)$.

Proof. (i). Since any solution to (3.1.1) is given by a linear combination of the functions \overline{u}_1 and \overline{u}_2 , and \overline{u}_1 is decreasing (see Lemma 3.1.2), it suffices to show that $\lim_{x\to+\infty}\overline{u}_2(x)\in\mathbb{R}$ if and only if $R\in L^1(0,+\infty)$. For this purpose, we observe that, since \overline{u}_2 is increasing and $\overline{u}_2(0)=1$, then

$$R(x) \le W(x) \int_0^x \frac{\overline{u}_2(s)}{q(s)W(s)} ds \le \overline{u}_2(x)R(x), \qquad x > 0.$$
 (3.1.10)

Moreover, by (3.1.7) we can write

$$\overline{u}_2'(x) = W(x) \left(\overline{u}_2'(0) + \lambda \int_0^x \frac{\overline{u}_2(s)}{q(s)W(s)} ds \right), \qquad x > 0.$$
 (3.1.11)

Suppose that \overline{u}_2 is bounded in a neighborhood of $+\infty$. Then, the two terms in the right-hand side of (3.1.11) are in $L^1(0, +\infty)$, since they are both positive. Therefore, from (3.1.10) it follows that $R \in L^1(0, +\infty)$.

Conversely, suppose that $R \in L^1(0, +\infty)$. Then, plugging (3.1.10) into (3.1.11) we see that \overline{u}_2 satisfies the differential inequality

$$\overline{u}_2'(x) \le \overline{u}_2'(0)W(x) + \lambda R(x)\overline{u}_2(x), \qquad x > 0.$$

Therefore, the Gronwall Lemma yields

$$\overline{u}_2(x) \le \exp\left(\lambda \int_0^x R(t)dt\right) \left[1 + \overline{u}_2'(0) \int_0^x W(t) \exp\left(-\lambda \int_0^t R(s)ds\right) dt\right],\tag{3.1.12}$$

for any x > 0. Since R is integrable in $(0, +\infty)$ and

$$W(x) \le \left(\int_0^1 \frac{1}{q(s)W(s)} ds\right)^{-1} R(x), \qquad x \ge 1,$$

we easily deduce that $W \in L^1(0, +\infty)$. Therefore, from (3.1.12) it easily follows that \overline{u}_2 is bounded in $(0, +\infty)$.

(ii). Let u be a positive decreasing solution to (3.1.1). Then, the equation (3.1.6) implies that the function u'/W is negative and increasing in \mathbb{R} . Therefore, there exists $k \leq 0$ such that

$$k = \lim_{x \to +\infty} \frac{u'(x)}{W(x)}.$$

Let us prove that k = 0. For this purpose, we observe that, integrating (3.1.6) from x to c and, then, letting c go to $+\infty$, gives

$$u'(x) = W(x) \left(k - \lambda \int_{x}^{+\infty} \frac{u(s)}{q(s)W(s)} ds \right), \qquad x \in \mathbb{R}.$$
 (3.1.13)

Since $Q \in L^1(0,+\infty)$, the Fubini theorem implies that the function $x \mapsto W(x) \int_x^{+\infty} (q(s)W(s))^{-1}ds$ is integrable in $(0,+\infty)$. The boundedness of u in $(0,+\infty)$ yields the integrability of the function $x \mapsto \int_x^{+\infty} u(s)/(q(s)W(s))ds$ in $(0,+\infty)$. Therefore, from (3.1.13) it follows that, if $k \neq 0$, the function W is integrable in $(0,+\infty)$. Since

$$\frac{1}{q(x)W(x)} \le \left(\int_0^1 W(s)ds\right)^{-1} Q(x), \qquad x \ge 1, \tag{3.1.14}$$

then the function 1/(qW) is integrable in $(0, +\infty)$. But this implies that R is integrable in $(0, +\infty)$ as well. Hence, we get a contradiction.

(iii). Let us prove that, if $Q, R \in L^1(0, +\infty)$, then any solution to (3.1.1) is such that u and u'/W admit finite limits at $+\infty$. By the property (i) we can limit ourselves to showing that the limit $\lim_{x\to +\infty} u'/W$ is finite for any solution u of (3.1.1). For this purpose we observe that, from (3.1.14) and the boundedness of u at $+\infty$, it follows that u/(qW) is integrable in $(0, +\infty)$. Therefore, dividing both the sides of (3.1.7) by W(x) and letting x go to $+\infty$, we obtain that the limit $\lim_{x\to +\infty} u'(x)/W(x)$ is finite.

Let us now set $\overline{u}=\overline{u}_1-c\overline{u}_2$ where c is a constant such that $\lim_{x\to+\infty}\overline{u}(x)=0$. Let us observe that $c\in[0,1)$. Of course, c>0 since \overline{u}_j (j=1,2) is positive in $\mathbb R$. Moreover, $\overline{u}_2\geq\overline{u}_1$ in $[0,+\infty)$ and $\overline{u}_1(0)=\overline{u}_2(0)=1$. If c>1, then \tilde{u} should have a negative minimum and if c=1 it should have a positive maximum or a negative minimum. Of course, these are all contradictions by Remark 3.1.1. Therefore, $c\in[0,1)$. It follows that $\overline{u}(0)>0$ and, since $\lim_{x\to+\infty}\overline{u}(x)=0$, Remark 3.1.1 implies that \overline{u} is decreasing in $\mathbb R$.

Arguing as in the proof of the property (ii) we see that $\overline{u}'(x)/W(x)$ tends to a nonpositive limit k as x tends to $+\infty$. If k=0, from (3.1.13) we would get

$$\frac{\overline{u}'(x)}{\overline{u}(x)} = -\lambda \frac{W(x)}{\overline{u}(x)} \int_{x}^{+\infty} \frac{\overline{u}(s)}{q(s)W(s)} ds \ge -\lambda W(x) \int_{x}^{+\infty} \frac{1}{q(s)W(s)} ds,$$
(3.1.15)

for any $x \in \mathbb{R}$. As it has been shown in the proof of the property (ii) the last side of (3.1.15) is integrable in $(0, +\infty)$ since $Q \in L^1(0, +\infty)$. It follows that the function $\log(\overline{u})$ is bounded from below in $(0, +\infty)$ and, consequently, $\lim_{x \to +\infty} \overline{u}(x) = l > 0$. Since this is a contradiction, k should be strictly negative. Therefore, setting $u = -\tilde{u}/k$, we obtain a solution to (3.1.1) satisfying (3.1.9)(i).

Now, let us show that the problem (3.1.1) admits a positive decreasing solution v satisfying (3.1.9)(ii). For this purpose, let w be the solution to (3.1.1) satisfying w(0) = 0 and w'(0) = 1. According to Remark 3.1.1, w is increasing in \mathbb{R} . Therefore, from (3.1.6) it follows that w'/W is increasing as well and, consequently, $\lim_{x\to +\infty} w'(x)/W(x) \in (1/W(0), +\infty)$.

Let us set $\overline{v} = u + dw$ where u satisfies (3.1.9)(i) and d is a positive constant such that $\lim_{x \to +\infty} \overline{v}'(x)/W(x) = 0$. Since \overline{v} is nonnegative, by (3.1.6) we now easily see that \overline{v} is decreasing in \mathbb{R} . Moreover, since $\lim_{x \to +\infty} \overline{v}'(x)/W(x) = 0$, the previous arguments show that $l := \lim_{x \to +\infty} \overline{v}(x)$ is strictly greater than 0. Thus, the function $v = \overline{v}/l$ is a solution to (3.1.1) satisfying (3.1.9)(ii).

(iv). Let u be a decreasing solution such that $\lim_{x\to+\infty} u(x) = l > 0$. Then, arguing as in the proof of the property (ii), we can show that there exists $k \leq 0$ such that

$$k = \lim_{x \to +\infty} \frac{u'(x)}{W(x)}$$

and u' is given by (3.1.13). It follows that the function u/(qW) is integrable in $(0, +\infty)$. Since all the terms in the right-hand side of (3.1.13) are nonpositive, and the left-hand side is integrable in $(0, +\infty)$, then the function $x \mapsto W(x) \int_x^{+\infty} u(s)/(q(s)W(s))ds$ is integrable in $(0, +\infty)$ as well. Since u is positive and $u \ge l$ in $(0, +\infty)$, it follows that also the function $x \mapsto W(x) \int_x^{+\infty} (q(s)W(s))^{-1}ds$ is integrable in $(0, +\infty)$. Using the Fubini Theorem, we see that this implies that $Q \in L^1(0, +\infty)$.

Conversely, let us suppose that $Q \in L^1(0, +\infty)$ and let us prove that the equation (3.1.1) admits a positive decreasing solution u with $\lim_{x\to +\infty} u(x) = l > 0$. If $R \in L^1(0, +\infty)$, the property (iii) gives us the wished function u. If $R \notin L^1(0, +\infty)$, then any positive and decreasing solution u to (3.1.1) is given by (3.1.13) and k = 0 by the property (ii). Hence u'/u satisfies (3.1.15) and, consequently, arguing as in the proof of the property (iii), we see that $\lim_{x\to +\infty} u(x) = l > 0$.

Concerning the behaviour of the solution to (3.1.1) in a neighborhood of $-\infty$, we have the following result.

Proposition 3.1.5 The following properties are met:

- (i) all the solutions to (3.1.1) admit finite limit at $-\infty$ if and only if the function R belongs to $L^1(-\infty,0)$;
- (ii) if $Q \in L^1(-\infty,0)$ and $R \notin L^1(-\infty,0)$ then any positive increasing solution to (3.1.1) satisfies

$$\lim_{x \to -\infty} \frac{u'(x)}{W(x)} = 0;$$

(iii) if Q, R belong to $L^1(-\infty,0)$ then any solution to (3.1.1) is such that the limits $\lim_{x\to-\infty}u(x)$ and $\lim_{x\to-\infty}u'(x)/W(x)$ exist and are finite. Moreover, there exist two increasing solutions u and v of (3.1.1) such that

(i)
$$\lim_{x \to -\infty} u(x) = 0$$
, $\lim_{x \to -\infty} \frac{u'(x)}{W(x)} = 1$,

(ii)
$$\lim_{x \to -\infty} v(x) = 1$$
, $\lim_{x \to -\infty} \frac{v'(x)}{W(x)} = 0$;

(iv) the equation (3.1.1) admits an increasing solution with $\lim_{x\to-\infty} u(x) > 0$ if and only if $Q \in L^1(-\infty,0)$.

Proof. Let $\tilde{\mathcal{A}}$ be the second order differential operator defined by $\tilde{\mathcal{A}}u = \tilde{q}u'' + \tilde{b}u'$ where $\tilde{q}(x) = q(-x)$, and $\tilde{b}(x) = -b(-x)$ for any $x \in \mathbb{R}$. Let \tilde{W} , \tilde{Q} and \tilde{R} be defined according to (3.1.3)-(3.1.5) with q, b being replaced with \tilde{q} and \tilde{b} . As it is easily seen, $\tilde{W}(x) = W(-x)$, $\tilde{Q}(x) = -Q(-x)$, $\tilde{R}(x) = -R(-x)$. Therefore, \tilde{W} , \tilde{Q} and \tilde{R} are integrable in a neighborhood of $-\infty$ if and only if W, Q and R are integrable in a neighborhood of $+\infty$. Moreover, u solves $\lambda u - \lambda u = 0$ if and only if the function $x \mapsto v(x) := u(-x)$ solves the differential equation $\lambda v - \tilde{\mathcal{A}}v = 0$. Now the assertion follows from Proposition 3.1.4.

We now give the following definitions.

Definition 3.1.6 The point $+\infty$ is said

$$accessible \ if \qquad \begin{cases} \textit{regular, i.e., } Q \in L^1(0,+\infty), \ R \in L^1(0,+\infty), \\ \textit{exit, i.e.,} \qquad Q \not\in L^1(0,+\infty), \ R \in L^1(0,+\infty), \end{cases}$$

unaccessible if
$$\begin{cases} entrance, \ i.e., \ Q \in L^1(0, +\infty), \ R \notin L^1(0, +\infty), \\ natural, \ i.e., \quad Q \notin L^1(0, +\infty), \ R \notin L^1(0, +\infty). \end{cases}$$

Similarly, the point $-\infty$ is said

$$accessible if \qquad \begin{cases} \textit{regular, i.e., } Q \in L^1(-\infty, 0), \ R \in L^1(-\infty, 0), \\ \textit{exit, i.e.,} \qquad Q \notin L^1(-\infty, 0), \ R \in L^1(-\infty, 0), \end{cases}$$

$$unaccessible \ if \ \begin{cases} entrance, \ i.e., \ Q \in L^1(-\infty,0), \ R \notin L^1(-\infty,0), \\ natural, \ i.e., \quad Q \notin L^1(-\infty,0), \ R \notin L^1(-\infty,0). \end{cases}$$

The results in Propositions 3.1.4 and 3.1.5 can be rephrased as follows.

Proposition 3.1.7 The following properties are met:

(i) $+\infty$ (resp. $-\infty$) is regular if and only if the differential equation (3.1.1) admits two positive decreasing (resp. increasing) solutions u_1 and u_2 such that

$$\lim_{x \to +\infty} u_j(x) = j - 1, \qquad \lim_{x \to +\infty} \frac{u'_j(x)}{W(x)} = -2 + j, \qquad j = 1, 2,$$

$$\left(resp. \lim_{x \to -\infty} u_j(x) = j - 1, \qquad \lim_{x \to -\infty} \frac{u'_j(x)}{W(x)} = 2 - j, \quad j = 1, 2\right).$$

In this case all the solutions to (3.1.1) are bounded in $(0, +\infty)$ (resp. in $(-\infty, 0)$);

- (ii) $+\infty$ (resp. $-\infty$) is an exit if and only if all the solutions of (3.1.1) are bounded in $(0, +\infty)$ (resp. in $(-\infty, 0)$) and any positive decreasing (resp. increasing) solution u vanishes at $+\infty$ (resp. at $-\infty$);
- (iii) $+\infty$ (resp. $-\infty$) is an entrance if and only if the differential equation (3.1.1) admits a positive decreasing (resp. increasing) solution u such that

$$\lim_{x \to +\infty} u(x) = 1, \qquad \lim_{x \to +\infty} \frac{u'(x)}{W(x)} = 0,$$

$$\left(resp. \lim_{x \to -\infty} u(x) = 1, \qquad \lim_{x \to -\infty} \frac{u'(x)}{W(x)} = 0 \right),$$

and any other solution of (3.1.1), which is independent of u, is unbounded in $(0, +\infty)$ (resp. in $(-\infty, 0)$);

(iv) $+\infty$ (resp. $-\infty$) is natural if and only the differential equation (3.1.1) admits a positive decreasing (resp. increasing) solution u such that

$$\lim_{x \to +\infty} u(x) = 0, \qquad \lim_{x \to +\infty} \frac{u'(x)}{W(x)} = 0,$$

$$\left(\lim_{x \to -\infty} u(x) = 0, \qquad \lim_{x \to -\infty} \frac{u'(x)}{W(x)} = 0\right),\,$$

and any other solution of (3.1.1), which is independent of u, is unbounded in $(0, +\infty)$ (resp. in $(-\infty, 0)$).

Proof. We just prove the property (iv) since the other properties follow easily from Propositions 3.1.4 and 3.1.5. Moreover, we limit ourselves to dealing with the case when the point $+\infty$ is natural, since the case when $-\infty$ is natural can be then deduced from this one using Proposition 3.1.5 instead of Proposition 3.1.4.

As a first step we observe that, according to Proposition 3.1.4(iv), \overline{u}_1 vanishes at $+\infty$.

Let us now prove that \overline{u}_1'/W vanishes at $+\infty$ as well. For this purpose, we begin by observing that, from (3.1.6) written with \overline{u}_1 instead of u, we easily deduce that \overline{u}_1'/W admits finite (and nonpositive) limit as x tends to $+\infty$ since it is negative and increasing. By contradiction, we assume that the previous limit is negative and we denote it by k. Since any solution of (3.1.1) is a linear combination of \overline{u}_1 and \overline{u}_2 and, according to Proposition 3.1.4(i), the equation (3.1.1) admits solutions which are unbounded in a neighborhood of $+\infty$, it follows that $\overline{u}_2(x)$ tends to $+\infty$ as x tends to $+\infty$. Taking Remark 3.1.3 into account, we can write

$$1 = \frac{W(x)}{W(x)} = \frac{1}{w_0} \left(\frac{\overline{u}_2'(x)}{W(x)} \overline{u}_1(x) - \frac{\overline{u}_1'(x)}{W(x)} \overline{u}_2(x) \right), \qquad x > 0.$$

Recalling that \overline{u}'_2 and \overline{u}_1 are positive in \mathbb{R} and taking the limit as x tends to $+\infty$, we are led to a contradiction.

Conversely, let us assume that there exists a positive decreasing solution \widetilde{u} to the problem (3.1.1) vanishing at $+\infty$ together with the function \widetilde{u}'/W and that any other solution to (3.1.1), independent of \widetilde{u} , is unbounded at $+\infty$. According to Proposition 3.1.4(i), it is clear that $R \notin L^1(0,+\infty)$. To show that $Q \notin L^1(0,+\infty)$, we observe that, denoting by v a solution to (3.1.1) linearly independent of \widetilde{u} , then any solution u to the problem (3.1.1) is given by $u = c_1\widetilde{u} + c_2v$, for some $c_1, c_2 \in \mathbb{R}$. Therefore, u is bounded at $+\infty$ if and only if $c_2 = 0$. But in such a case, u vanishes at $+\infty$. Therefore, Proposition 3.1.4(iv) implies that $Q \notin L^1(0,+\infty)$ and we are done.

3.2 The nonhomogeneous equation

In this section we study the solutions $u \in C^2(\mathbb{R})$ of the nonhomogeneous equation

$$\lambda u - \mathcal{A}u = f, (3.2.1)$$

when $f \in C_b(\mathbb{R})$.

By the classical theory of ordinary differential equations, it is easy to check that the more general solution to the differential equation (3.2.1) is given by

$$u(x) = -\int_0^x \langle \mathcal{W}(x)\mathcal{W}^{-1}(t)e_2, e_1 \rangle \frac{f(t)}{q(t)} dt + c_1 \overline{u}_1(x) + c_2 \overline{u}_2(x), \qquad x \in \mathbb{R},$$
(3.2.2)

where c_1, c_2 are arbitrary real constants, $e_1 = (1, 0), e_2 = (0, 1), \overline{u}_1, \overline{u}_2$ are as in Lemma 3.1.2 and W is the wronskian matrix

$$\mathcal{W}(x) = \begin{pmatrix} \overline{u}_1(x) \ \overline{u}_2(x) \\ \overline{u}_1'(x) \ \overline{u}_2'(x) \end{pmatrix}, \qquad x \in \mathbb{R}.$$

Taking (3.1.8) into account, from (3.2.2) we can write

$$u(x) = \left(c_1 + \frac{1}{w_0} \int_0^x \frac{f(t)}{q(t)W(t)} \overline{u}_2(t)dt\right) \overline{u}_1(x)$$

$$+ \left(c_2 - \frac{1}{w_0} \int_0^x \frac{f(t)}{q(t)W(t)} \overline{u}_1(t)dt\right) \overline{u}_2(x), \tag{3.2.3}$$

for any $x \in \mathbb{R}$.

We can now prove the following result.

Proposition 3.2.1 For any $f \in C_b(\mathbb{R})$ and any $\lambda > 0$, the function $u = R_{\lambda}f$ defined by

$$(R_{\lambda}f)(x) = \int_{-\infty}^{+\infty} G_{\lambda}(x,s)f(s)ds, \qquad x \in \mathbb{R},$$
 (3.2.4)

where $G_{\lambda}: \mathbb{R}^2 \to \mathbb{R}$ is given by

$$G_{\lambda}(x,s) = \begin{cases} \frac{\overline{u}_1(s)\overline{u}_2(x)}{w_0q(s)W(s)}, & x < s, \\ \frac{\overline{u}_1(x)\overline{u}_2(s)}{w_0q(s)W(s)}, & x \geq s, \end{cases}$$

belongs to $C_b(\mathbb{R}) \cap C^2(\mathbb{R})$ and solves the differential equation (3.2.1). Moreover, the operator $R_{\lambda}: C_b(\mathbb{R}) \to C_b(\mathbb{R})$ is bounded and $\|R_{\lambda}\|_{L(C_b(\mathbb{R}^N))} \le 1/\lambda$.

Proof. Observe that a formal computation shows that

$$(R_{\lambda}f)(x) = \frac{\overline{u}_1(x)}{w_0} \int_{-\infty}^x \frac{\overline{u}_2(s)}{q(s)W(s)} f(s)ds + \frac{\overline{u}_2(x)}{w_0} \int_x^{+\infty} \frac{\overline{u}_1(s)}{q(s)W(s)} f(s)ds,$$
(3.2.5)

for any $x \in \mathbb{R}$. So, let us prove that the integral terms in the right-hand side of (3.2.5) are well defined for any $x \in \mathbb{R}$.

Since the function $W^{-1}\overline{u}'_2$ is positive and increasing in \mathbb{R} (as it easily follows from (3.1.6)), then, for any c < x, we have

$$\int_{c}^{x} \frac{\overline{u}_{2}(s)}{q(s)W(s)} |f(s)| ds \leq \|f\|_{\infty} \int_{c}^{x} \frac{\overline{u}_{2}(s)}{q(s)W(s)} ds$$

$$= \frac{1}{\lambda} \|f\|_{\infty} \int_{c}^{x} \left(\frac{\overline{u}'_{2}}{W}\right)'(s) ds$$

$$= \frac{1}{\lambda} \|f\|_{\infty} \left(\frac{\overline{u}'_{2}(x)}{W(x)} - \frac{\overline{u}'_{2}(c)}{W(c)}\right)$$

$$\leq \frac{1}{\lambda} \|f\|_{\infty} \left(\frac{\overline{u}'_{2}(x)}{W(x)} - \lim_{c \to -\infty} \frac{\overline{u}'_{2}(c)}{W(c)}\right)$$

$$\leq \frac{1}{\lambda} \|f\|_{\infty} \frac{\overline{u}'_{2}(x)}{W(x)}.$$
(3.2.6)

Hence, the first integral in the right-hand side of (3.2.5) is well defined. A similar argument shows that, for any c > x,

$$\int_{x}^{c} \frac{\overline{u}_{1}(s)}{q(s)W(s)} |f(s)| ds \le -\frac{1}{\lambda} ||f||_{\infty} \frac{\overline{u}'_{1}(x)}{W(x)}.$$
 (3.2.7)

Hence, also the other integral is well defined. Of course, they both define continuous functions in \mathbb{R} . Therefore, $R_{\lambda}f$ is a continuous function in \mathbb{R} and it is bounded since (3.2.6) gives

$$|(R_{\lambda}f)(x)| \leq ||f||_{\infty} \left(\frac{\overline{u}_1(x)}{w_0 \lambda} \frac{\overline{u}_2'(x)}{W(x)} - \frac{\overline{u}_1'(x)}{w_0 \lambda} \frac{\overline{u}_2(x)}{W(x)} \right) = \frac{1}{\lambda} ||f||_{\infty},$$

for any $x \in \mathbb{R}$. Finally, a straightforward computation shows that $R_{\lambda}f$ solves the differential equation (3.2.1). This finishes the proof.

Thanks to Propositions 3.1.7 and 3.2.1, it is now easy to solve the problem of the uniqueness of the solution $u \in C_b(\mathbb{R}) \cap C^2(\mathbb{R})$ to the equation (3.2.1).

Theorem 3.2.2 The elliptic equation (3.2.1) is uniquely solvable in $C_b(\mathbb{R}) \cap C^2(\mathbb{R})$ for any $\lambda > 0$ and any $f \in C_b(\mathbb{R})$ if and only if $-\infty$ and $+\infty$ are unaccessible. If both $-\infty$ and $+\infty$ are accessible, then any solution $u \in C^2(\mathbb{R})$ to (3.2.1) is bounded in \mathbb{R} .

In Chapter 2 we have seen that if the coefficients q and b belong to $C_{\text{loc}}^{\alpha}(\mathbb{R})$, then a solution $u \in C_{\text{loc}}^{2+\alpha}(\mathbb{R})$ can be obtained by approximating the equation (3.1.1) with the Dirichlet problems

$$\begin{cases} \lambda u(x) - q(x)u''(x) - b(x)u'(x) = f(x), & x \in (-n, n), \\ u(-n) = u(n) = 0, & (3.2.8) \end{cases}$$

and, then, letting n go to 0. Now two questions arise naturally. Does this approximation argument work also in the case when the coefficients are merely continuous in \mathbb{R} and, if this is the case, what are the relations between the solution provided by this method and the solution given by (3.2.4)? The answers to the previous two questions are contained in the following proposition.

Proposition 3.2.3 For any $f \in C_b(\mathbb{R})$, let u_n be the solution to the problem (3.2.8). Then, u_n converges to the function $R_{\lambda}f$ (see (3.2.4)) in $C^2([-a,a])$ for any a > 0, as n tends to $+\infty$.

Proof. Throughout the proof, \overline{u}_1 and \overline{u}_2 will denote the functions in Lemma 3.1.2. Moreover, to simplify the notation we write $u(+\infty)$, respectively $u(-\infty)$, to denote the limit of u at $+\infty$, respectively at $-\infty$, whenever these limits exist. Finally, for any fixed function $f \in C_b(\mathbb{R})$, we denote by g_f and h_f the functions defined by

$$g_f(x) = \frac{1}{w_0} \int_0^x \frac{f(t)}{q(t)W(t)} \overline{u}_1(t)dt, \qquad h_f(x) = \frac{1}{w_0} \int_0^x \frac{f(t)}{q(t)W(t)} \overline{u}_2(t)dt,$$

for any $x \in \mathbb{R}$. First of all, we prove that the sequence $\{u_n\}$ converge in $C^2([-a,a])$, for any a>0, to a function $u \in C^2(\mathbb{R}) \cap C_b(\mathbb{R})$. For this purpose, we observe that a simple computation shows that u_n is given by (3.2.3), with $c=c_1(n)$ and $c_2=c_2(n)$ satisfying

$$\begin{cases}
(c_1(n) + h_f(n))\overline{u}_1(n) + (c_2(n) - g_f(n))\overline{u}_2(n) = 0, \\
(c_1(n) + h_f(-n))\overline{u}_1(-n) + (c_2(n) - g_f(-n))\overline{u}_2(-n) = 0.
\end{cases} (3.2.9)$$

Moreover, by the classical maximum principle (see Theorem C.2.2(i)) it follows that

$$\sup_{x \in [-n,n]} |u_n(x)| \le \frac{1}{\lambda} ||f||_{\infty}, \qquad n \in \mathbb{N}.$$
(3.2.10)

Thus, writing (3.2.3) at $x = \pm 1$, we deduce that $|c_1(n)\overline{u}_1(1) + c_1(n)\overline{u}_2(1)|$ and $|c_1(n)\overline{u}_1(-1) + c_2(n)\overline{u}_2(-1)|$ are bounded by a positive constant, independent of n. Since the matrix A, whose rows are $(\overline{u}_1(-1), \overline{u}_2(-1))$ and $(\overline{u}_1(1), \overline{u}_2(1))$, is invertible, it follows that $|c_1(n)|$ and $|c_2(n)|$ are bounded uniformly with respect to $n \in \mathbb{N}$. Therefore, up to a subsequence, we can assume that $c_1(n)$ and $c_2(n)$ converge to some real numbers $c_1(+\infty)$ and $c_2(+\infty)$. Now, from (3.2.3) is immediate to check that u_n converges in $C^2([-a,a])$ to some function $u \in C^2(\mathbb{R})$ which, of course, satisfies the differential equation (3.2.1). Moreover, the estimate (3.2.10) implies that u is bounded in \mathbb{R} . Therefore, the function $R(\lambda)f$ is well defined for any $f \in C_b(\mathbb{R})$.

Let us now prove that $R(\lambda)f = R_{\lambda}f$, where $R_{\lambda}f$ is given by (3.2.4). For this purpose, it suffices to show that $c_1(n)$ and $c_2(n)$ converge as n tends to $+\infty$ and

$$c_1(+\infty) = -h_f(-\infty), \qquad c_2(+\infty) = g_f(+\infty).$$
 (3.2.11)

Of course, according to Theorem 3.2.2 it suffices to consider the case when at least one between $-\infty$ and $+\infty$ is accessible.

To begin with we observe that, from (3.2.6) and (3.2.7), we easily deduce that the function $\overline{u}_1/(qW)$ is integrable in $(0, +\infty)$, whereas the function $\overline{u}_2/(qW)$ is integrable in $(-\infty, 0)$. Therefore, $g_f(+\infty)$ and $h_f(-\infty)$, as well as $g_1(+\infty)$ and $h_1(-\infty)$, exist and are finite. Moreover, if $\overline{u}_1(+\infty) = 0$ and $\overline{u}_2(+\infty)$ is finite, then $(h_f\overline{u}_1)(+\infty) = 0$. Indeed,

$$\begin{split} |h_f(n)\overline{u}_1(n)| &= \frac{\overline{u}_1(n)}{w_0} \left| \int_0^n \frac{f(t)}{q(t)W(t)} \overline{u}_2(t) dt \right| \\ &\leq \|f\|_{\infty} \frac{\overline{u}_1(n)}{w_0} \int_0^n \frac{\overline{u}_2(t)}{q(t)W(t)} dt \\ &\leq \|f\|_{\infty} \frac{\overline{u}_1(n)\overline{u}_2(n)}{w_0} \left(\int_0^x \frac{1}{q(t)W(t)} dt + \int_x^n \frac{1}{q(t)W(t)} dt \right) \\ &\leq \|f\|_{\infty} \left\{ \frac{\overline{u}_1(n)\overline{u}_2(n)}{w_0} \int_0^x \frac{1}{q(t)W(t)} dt + \overline{u}_2(n) \left(g_1(n) - g_1(x) \right) \right\}, \end{split}$$

for any x > 0. Therefore, letting n go to $+\infty$ we get

$$\limsup_{n \to +\infty} |h_f(n)\overline{u}_1(n)| \le ||f||_{\infty} u_2(+\infty) \left(g_1(+\infty) - g_1(x)\right).$$

Then, letting x go to $+\infty$, we get $\limsup_{n\to+\infty} h_f(n)\overline{u}_1(n) = 0$, namely $(h_f\overline{u}_1)(+\infty) = 0$. Similarly, if $\overline{u}_1(-\infty)$ is real and $\overline{u}_2(-\infty) = 0$, then

$$\lim_{n \to +\infty} \overline{u}_2(-n)g_f(-n) = 0.$$

To complete the proof we need a deeper analysis of the functions \overline{u}_1 and \overline{u}_2 . For this purpose we split the remainder of the proof into several steps.

The case when $+\infty$ and $-\infty$ are both accessible. In such a situation, the functions \overline{u}_1 and \overline{u}_2 are bounded in \mathbb{R} and $\overline{u}_1(+\infty) = \overline{u}_2(-\infty) = 0$. The boundedness of \overline{u}_1 and \overline{u}_2 follows immediately from the properties (i) and (ii) in Proposition 3.1.7. Let us now show that $\overline{u}_1(+\infty) = 0$. This is clear by Proposition 3.1.7(ii) if $+\infty$ is an exit point. So, let us assume that $+\infty$ is regular. A similar argument then can be used to show that $\overline{u}_2(-\infty) = 0$ when $-\infty$ is natural. For this purpose, let u be a decreasing solution of (3.1.1) with $u(+\infty) = 0$, provided by Proposition 3.1.7. Then, we set v = u/u(0). As it is immediately seen, v vanishes at $+\infty$ and v(0) = 1. We claim that $v \geq \overline{u}_1$ in $(0, +\infty)$. Since $\overline{u}_1 > 0$ in \mathbb{R} , it then follows that $\overline{u}_1(+\infty) = 0$. By contradiction suppose that there exists x > 0 such that $\overline{u}_1(x) > v(x)$. Then, $\overline{u}_1 > v$ in $(0, +\infty)$. Indeed if this were not the case, then there should exist a point $x_1 > 0$ such that $\overline{u}_1(x_1) = v(x_1)$. By the last part of Remark 3.1.1 it would follow that $\overline{u}_1 \equiv v$: a contradiction. Therefore, if $\overline{u}_1(x) > v(x)$ for some x > 0, then $\overline{u}_1 > v$ in $(0, +\infty)$. But this implies that $\overline{u}_1'(0) > v'(0)$. Since B is

an interval, this would imply that $\overline{u}'_1(0) \neq \sup B$: a contradiction. Therefore, $\overline{u}_1 \leq v$ in $(0, +\infty)$ and $v'(0) \notin B$ (see (3.1.2)).

Now, from (3.2.9) it easily follows that $c_1(+\infty)$ and $c_2(+\infty)$ exist finite and are given by (3.2.11).

The case when $+\infty$ is accessible and $-\infty$ is unaccessible. According to the properties (i) and (ii) in Proposition 3.1.7, and arguing as in the previous step, it follows that $\overline{u}_1(+\infty)=0$, $\overline{u}_1(-\infty)=+\infty$, $\overline{u}_2(+\infty)\in(0,+\infty)$. Hence, letting n go to $+\infty$ in the first equation in (3.2.9) we deduce that $c_2(+\infty)$ is given by (3.2.11). To prove that $c_1(+\infty)$ is given by (3.2.11), we begin by studying the behaviour of u_2 at $-\infty$. For this purpose, we observe that we can write $\overline{u}_2=d_1\overline{u}_1+d_2u$ where u is an increasing solution to (3.1.1) satisfying $\lim_{x\to-\infty}u'(x)/W(x)=0$ and $u(-\infty)=0$, if $-\infty$ is natural, and $u(-\infty)=1$, if $-\infty$ is an entrance. Since $\overline{u}_2(-\infty)$ is real, we easily see that $d_1=0$ and consequently $\overline{u}_2=u/u(0)$. Therefore, $\overline{u}_2(-\infty)=0$ if $-\infty$ is natural whereas $\overline{u}_2(-\infty)>0$ if $-\infty$ is an entrance. In any case, \overline{u}_2'/W tends to 0 as x tends to $-\infty$. Therefore,

$$|g_{f}(-n)|\overline{u}_{2}(-n) \leq \frac{\|f\|_{\infty}}{w_{0}}\overline{u}_{2}(-n) \int_{-n}^{0} \frac{\overline{u}_{1}(t)}{q(t)W(t)} dt$$

$$= \frac{\|f\|_{\infty}}{w_{0}} \frac{\overline{u}_{2}(-n)}{\lambda} \left(\overline{u}'_{1}(0) - \frac{\overline{u}'_{1}(-n)}{W(-n)}\right)$$

$$= \frac{\|f\|_{\infty}}{w_{0}} \frac{u_{2}(-n)}{\lambda} \overline{u}'_{1}(0) - \frac{\|f\|_{\infty}}{w_{0}} \frac{\overline{u}'_{2}(-n)}{\lambda W(-n)} \overline{u}_{1}(-n) + \frac{1}{\lambda} \|f\|_{\infty}.$$

Therefore,

$$\lim_{n \to +\infty} \frac{g_f(-n)\overline{u}_2(-n)}{\overline{u}_1(-n)} = 0.$$

Thus, letting n go to $+\infty$ in the second equation in (3.2.9), we now easily see that $c_1(+\infty)$ is given by (3.2.11).

The case when $+\infty$ is unaccessible and $-\infty$ is accessible. By changing x to -x, we go back again to the previous case (see the proof of Proposition 3.1.5). Hence, also in this case $c_1(+\infty)$ and $c_2(+\infty)$ are given by (3.2.11). This finishes the proof.

Now, we consider some examples.

Example 3.2.4 Let \mathcal{A} be the operator defined by

$$\mathcal{A}\varphi(x) = \varphi''(x) - x^3\varphi(x), \quad x \in \mathbb{R},$$

on smooth functions φ . Let us show that $+\infty$ and $-\infty$ are both unaccessible. By Theorem 3.2.2 this will imply that the equation $\lambda u - \mathcal{A}u = f$ is uniquely solvable in $C_b(\mathbb{R}) \cap C^2(\mathbb{R})$ for any $f \in C_b(\mathbb{R})$. A straightforward computation shows that

$$W(x) = e^{x^4/4}, \qquad Q(x) = e^{-x^4/4} \int_0^x e^{s^4/4} ds, \qquad R(x) = e^{x^4/4} \int_0^x e^{-s^4/4} ds,$$

for any $x \in \mathbb{R}$. It is readily seen that $R \notin L^1(-\infty,0) \cup L^1(0,+\infty)$. Moreover, using De L'Hôpital rule it can be seen that

$$\lim_{x \to \pm \infty} x^3 Q(x) = 1.$$

Hence $Q \in L^1(-\infty,0) \cap L^1(0,+\infty)$. We conclude that $+\infty$ and $-\infty$ are entrance points.

We will generalize this example to the N-dimensional setting in Chapter 4 (see Example 4.1.13).

Example 3.2.5 Let \mathcal{A} be the one-dimensional differential operator defined by

$$\mathcal{A}\varphi(x) = \varphi''(x) + x^3 \varphi'(x), \qquad x \in \mathbb{R},$$

on smooth functions φ . In this case

$$W(x) = e^{-\frac{x^4}{4}}, \qquad Q(x) = e^{\frac{x^4}{4}} \int_0^x e^{-\frac{s^4}{4}} ds, \qquad R(x) = e^{-\frac{x^4}{4}} \int_0^x e^{\frac{s^4}{4}} ds.$$

Arguing as in the previous example, it follows that $Q \notin L^1(-\infty,0) \cup L^1(0,+\infty)$ and $R \in L^1(-\infty,0) \cap L^1(0,+\infty)$. We conclude that $+\infty$ and $-\infty$ are both exit points. Therefore, according to Theorem 3.2.2, for any $f \in C_b(\mathbb{R})$ the equation $\lambda u - Au = f$ admits more than one solution belonging to $C_b(\mathbb{R}) \cap C^2(\mathbb{R})$.

We will generalize this example to the N-dimensional setting in Chapter 5 (see Example 5.2.5).

Remark 3.2.6 These two examples show us that the uniqueness of the solution $u \in D_{\max}(\mathcal{A})$ to the elliptic equation $\lambda u - \mathcal{A}u = f$ does not depend merely on the growth at infinity of the coefficients of the operator \mathcal{A} . In fact, the operators defined in Examples 3.2.4 and 3.2.5 differ only in the sign of the drift term, but this difference is crucial. Indeed, using the notation of Chapter 2, we have $D(\widehat{A}) = D_{\max}(A)$, if \mathcal{A} is the operator in Example 3.2.4 whereas $D(\widehat{A})$ is properly contained in $D_{\max}(A)$ if \mathcal{A} is the operator in Example 3.2.5.

See also Example 5.2.5 and [88, Section 5.2] for a discussion about these two previous examples in the probabilistic framework.

We study the problem of the uniqueness of the solution $u \in D_{\text{max}}(\mathcal{A})$ to the elliptic equation $\lambda u - \mathcal{A}u = f$, and of the classical solution to the homogeneous Cauchy problem associated with the operator \mathcal{A} (in the N-dimensional setting), in Chapter 4.

To conclude this section we characterize $D_{\text{max}}(\mathcal{A})$ under some additional assumptions of the coefficients q and b. The following theorem has been proved in [58].

Theorem 3.2.7 Suppose that $q \in C(\mathbb{R})$ and $q(x) \geq \kappa_0$ for any $x \in \mathbb{R}$ and some positive constant κ_0 . Further assume that $b \in C^1(\mathbb{R})$ satisfies

$$q(x)b'(x) \le c_1 + c_2(b(x))^2, \qquad x \in \mathbb{R},$$
 (3.2.12)

for some constants $c_1 \in \mathbb{R}$ and $c_2 < 1$. Finally, assume that $+\infty$ and $-\infty$ are both unaccessible. Then,

$$D_{\max}(\mathcal{A}) = \{ u \in C_b^2(\mathbb{R}) : qu'', bu' \in C_b(\mathbb{R}) \}.$$
 (3.2.13)

Proof. According to Theorem 3.2.2 and Proposition 3.2.3, for any $f \in C_b(\mathbb{R})$, the elliptic equation

$$u - \mathcal{A}u = f \tag{3.2.14}$$

admits a unique solution u in $D_{\max}(\mathcal{A})$. Therefore, to prove the assertion it suffices to show that u actually belongs to the space defined in the right-hand side of (3.2.13). For this purpose, for any $n \in \mathbb{N}$, we denote by $u_n \in C^2([-n,n])$ the solution to the Cauchy problem

$$\begin{cases} u(x) - \mathcal{A}u(x) = f(x), & x \in (-n, n), \\ u'(-n) = u'(n) = 0. \end{cases}$$

According to Theorem C.2.2(ii), we have

$$||u_n||_{C([-n,n])} \le ||f||_{C([-n,n])}, \qquad ||\mathcal{A}u_n||_{C([-n,n])} \le 2||f||_{C([-n,n])}, \quad (3.2.15)$$

for any $n \in \mathbb{N}$.

The main step of the proof consists in proving that the C^2 -norm of u_n and the sup-norm of qu''_n and bu'_n are uniformly bounded with respect to n. Then, a compactness argument will allow us to show that u_n converges to a solution of the equation (3.2.14) belonging to the space defined by the right-hand side of (3.2.13).

To begin with, let us prove that there exists a positive constant C, independent of n, such that

$$||qu_n''||_{C([-n,n])} + ||bu_n'||_{C([-n,n])} \le C||f||_{C([-n,n])} \le C||f||_{\infty}, \qquad n \in \mathbb{N}.$$
(3.2.16)

For this purpose, let $x_0 = x_0(n)$ be a point in [-n, n] such that $|b(x_0)u'_n(x_0)| = \|bu'_n\|_{C([-n,n])}$. Up to replacing u_n with $-u_n$ and f with -f, we can assume that x_0 is a maximum of the function bu'_n . Moreover, we can also assume that $x_0 \in (-n, n)$ and $b(x_0) \neq 0$, otherwise $b(x_0)u'_n(x_0) = 0$ and

(3.2.16) would follow immediately. Since $(bu'_n)'(x_0) = 0$, we have $u''_n(x_0) = -b'(x_0)(b(x_0))^{-1}u'_n(x_0)$. Hence, (3.2.12) implies that

$$q(x_0)u_n''(x_0) = -q(x_0)b'(x_0)\frac{u_n'(x_0)}{b(x_0)} \ge -c_1\frac{u_n'(x_0)}{b(x_0)} - c_2b(x_0)u_n'(x_0).$$

Therefore, taking (3.2.15) into account, we deduce that

$$2\|f\|_{C([-n,n])} \ge -f(x_0) + u_n(x_0)$$

$$= q(x_0)u_n''(x_0) + b(x_0)u_n'(x_0)$$

$$\ge (1 - c_2)b(x_0)u_n'(x_0) - c_1\frac{u_n'(x_0)}{b(x_0)}.$$
(3.2.17)

Multiplying the first and the last sides of (3.2.17) by $b(x_0)u'_n(x_0)$ we get

$$2\|bu'_n\|_{C([-n,n])}\|f\|_{C([-n,n])} \ge (1-c_2)\|bu'_n\|_{C([-n,n])}^2 - c_1(u'_n(x_0))^2$$

$$\ge (1-c_2)\|bu'_n\|_{C([-n,n])}^2 - c_1^+\|u'_n\|_{C([-n,n])}^2.$$

If we set

$$x = \|bu_n'\|_{C([-n,n])}, \qquad \alpha = \frac{2}{1-c_2} \|f\|_{C([-n,n])}, \qquad \beta = \frac{c_1^+}{1-c_2} \|u_n'\|_{C([-n,n])}^2,$$

we obtain that x satisfies the inequality $x^2 \leq \alpha x + \beta$, which implies that $x \leq \alpha + \sqrt{\beta}$ or, equivalently,

$$||bu_n'||_{C([-n,n])} \le \frac{2}{1-c_2} ||f||_{C([-n,n])} + \left(\frac{c_1^+}{1-c_2}\right)^{\frac{1}{2}} ||u_n'||_{C([-n,n])}.$$
 (3.2.18)

Now, we observe that there exists a positive constant C, independent of n, such that

$$||v'||_{C([-n,n])} \le C||v||_{C([-n,n])}^{\frac{1}{2}}||v''||_{C([-n,n])}^{\frac{1}{2}}, \tag{3.2.19}$$

for any $v \in C^2([-n,n])$ and any $n \in \mathbb{N}$. Indeed, if n=1 the estimate (3.2.19) follows from the Landau inequality applied to any extension $w \in C_b^2(\mathbb{R})$ of v. To get (3.2.19) for a general n, with a constant being independent of n, it suffices to apply the Landau inequality with n=1 to the function $w:[-1,1] \to \mathbb{R}$ defined by w(x) = v(nx) for any $x \in [-1,1]$.

Now, from (3.2.19) it follows immediately that, for any $\varepsilon > 0$, there exists a positive constant C_{ε} , independent of n, such that

$$||u'_n||_{C([-n,n])} \le \varepsilon ||u''_n||_{C([-n,n])} + C_\varepsilon ||u_n||_{C([-n,n])}$$

$$\le \varepsilon ||u''_n||_{C([-n,n])} + C_\varepsilon ||f_n||_{C([-n,n])},$$
(3.2.20)

for any $n \in \mathbb{N}$. Next we observe that, taking (3.2.15) into account, we can write

$$||u_n''||_{C([-n,n])} \leq \frac{1}{\kappa_0} ||qu_n''||_{C([-n,n])}$$

$$\leq \frac{1}{\kappa_0} \left(||\mathcal{A}u_n||_{C([-n,n])} + ||bu_n'||_{C([-n,n])} \right)$$

$$\leq \frac{1}{\kappa_0} \left(||bu_n'||_{C([-n,n])} + 2||f_n||_{C([-n,n])} \right). \tag{3.2.21}$$

Therefore, from (3.2.18), (3.2.20) (with ε small enough) and (3.2.21), we easily get (3.2.16).

Now we observe that from (3.2.21) we obtain

$$\|u_n''\|_{C([-n,n])} \leq \frac{C}{\kappa_0} \|f\|_{\infty}, \qquad n \in \mathbb{N}.$$

Therefore, according to (3.2.15) and (3.2.19) it follows that $\|u_n\|_{C^2([-n,n])}$ is bounded by a positive constant independent of n. Now, using a compactness argument similar to that used in the proof of Theorem 2.1.1, we can easily show that, up to a subsequence, u_n converges to a function $u \in C^1(\mathbb{R}^N)$ as n tends to $+\infty$, locally uniformly in \mathbb{R} . Since $qu_n'' = u_n - f - bu_n'$ and $q \ge \kappa_0 > 0$, we then deduce that u_n'' converges locally uniformly in \mathbb{R} as well. Therefore, $u \in C^2(\mathbb{R})$ and, since the sequence $\{u_n\}$ is bounded in $C_b^2(\mathbb{R})$, u belongs to $C_b^2(\mathbb{R})$ as well. Moreover, qu'' and bu' are bounded in \mathbb{R} , thanks to (3.2.16). Therefore, u belongs to the space defined by the right-hand side of (3.2.13).

To complete the proof it suffices to observe that since u_n solves, for any $n \in \mathbb{N}$, the differential equation (3.2.14) in [-n, n], then u satisfies such a differential equation in \mathbb{R} . Therefore, $u = R_1 f$ and we are done.

Chapter 4

Uniqueness results, conservation of probability and maximum principles

4.0 Introduction

In this chapter we deal with the problem of the uniqueness of the solution of the elliptic equation

$$\lambda v(x) - \mathcal{A}v(x) = f(x), \qquad x \in \mathbb{R}^N,$$
 (4.0.1)

 $(f \in C_b(\mathbb{R}^N))$ which belongs to

$$D_{\max}(\mathcal{A}) = \left\{ u \in C_b(\mathbb{R}^N) \cap \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2,p}(\mathbb{R}^N) : \ \mathcal{A}u \in C_b(\mathbb{R}^N) \right\}, \quad (4.0.2)$$

and of the solution to the parabolic problem

$$\begin{cases}
D_t u(t,x) - \mathcal{A}u(t,x) = 0, & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N,
\end{cases}$$
(4.0.3)

which belongs to $C([0,+\infty)\times\mathbb{R}^N)\cap C^{1,2}((0,+\infty)\times\mathbb{R}^N)$ and it is bounded in $[0,T]\times\mathbb{R}^N$ for any T>0.

Throughout the chapter, we assume that the hypotheses of Chapter 2 are satisfied. For the reader's convenience, we state them again.

Hypotheses 4.0.1 (i) $q_{ij} \equiv q_{ji}$,

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \quad \kappa(x) > 0, \quad \xi, x \in \mathbb{R}^N;$$

- (ii) q_{ij} , b_i (i, j = 1, ..., N) and c belong to $C_{loc}^{\alpha}(\mathbb{R}^N)$ for some $\alpha \in (0, 1)$;
- (iii) there exists $c_0 \in \mathbb{R}$ such that

$$c(x) \le c_0, \quad x \in \mathbb{R}^N.$$

Under the previous hypotheses, in Chapter 2 we have proved the existence of solutions to the problems (4.0.1) and (4.0.3), with the regularity properties claimed above, for any $f \in C_b(\mathbb{R}^N)$ and any $\lambda > 0$ (see Theorems 2.1.1 and 2.2.1).

The two problems of uniqueness are strictly connected. We exploit this connection in Proposition 4.1.1 (in the general case) and in Proposition 4.1.10 (in the case when $c \equiv 0$).

The following assumption is often considered to get uniqueness results (see, e.g., [83, 104, 139]).

Hypothesis 4.0.2 There exists a positive function $\varphi \in C^2(\mathbb{R}^N)$ satisfying

$$\lim_{|x| \to +\infty} \varphi(x) = +\infty, \qquad \mathcal{A}\varphi(x) - \lambda_0 \varphi(x) \le 0, \quad x \in \mathbb{R}^N, \tag{4.0.4}$$

for some $\lambda_0 > c_0$.

Remark 4.0.3 Observe that one can equivalently assume that there exists $\varphi \in C^2(\mathbb{R}^N)$ such that

$$\lim_{|x| \to +\infty} \varphi(x) = +\infty, \quad \mathcal{A}\varphi(x) - \lambda_0 \varphi(x) \le C, \quad x \in \mathbb{R}^N,$$
 (4.0.5)

for some $C \in \mathbb{R}$. Indeed, if φ satisfies (4.0.5), then the function $\varphi + M$ satisfies Hypothesis 4.0.2 provided the constant M is sufficiently large.

A function $\varphi \in C^2(\mathbb{R}^N)$ satisfying (4.0.4) or (4.0.5) is usually called a Lyapunov function for the operator \mathcal{A} .

The condition (4.0.5) can be made clearer with a particular choice of φ . For instance, if we take $\varphi(x) = \log(m+|x|^2)$, where m is a positive constant, then (4.0.5) reads as follows:

$$(m+|x|^{2})\operatorname{Tr} Q(x) - 2\langle Q(x)x, x \rangle + (m+|x|^{2})\langle b(x), x \rangle$$

$$+ \frac{c(x)}{2}(m+|x|^{2})^{2}\log(m+|x|^{2})$$

$$\leq \frac{\lambda_{0}}{2}(m+|x|^{2})^{2}\log(m+|x|^{2}) + \frac{C}{2}(m+|x|^{2})^{2}, \qquad x \in \mathbb{R}^{N}.$$

$$(4.0.6)$$

Assuming Hypotheses 4.0.1 and 4.0.2 we prove some maximum principles for the elliptic equation and for the parabolic problem. The maximum principles yield the uniqueness of the bounded and continuous solution of the elliptic equation (4.0.1) and the uniqueness of the classical solution of the parabolic problem (4.0.3), which is bounded in $[0,T] \times \mathbb{R}^N$ for any T > 0.

Afterwards, we focus our attention on the case when $c \equiv 0$. In such a case the function $u_1 \equiv \mathbf{1}$ is clearly a bounded solution of the problem (4.0.3) with initial value $f \equiv \mathbf{1}$. Also the function $u_2 = T(\cdot)\mathbf{1}$ is a bounded solution of

the same problem. Therefore, if the problem (4.0.3) admits a unique bounded solution for any $f \in C_b(\mathbb{R}^N)$, then, necessarily,

$$T(t)\mathbf{1} \equiv \mathbf{1}, \qquad t > 0.$$
 (4.0.7)

In general, we only have $0 < T(t)\mathbf{1} \le \mathbf{1}$. In Proposition 4.1.10 we prove that, actually, (4.0.7) is also a sufficient condition guaranteeing the uniqueness of the classical solution to the problem (4.0.3), which is bounded in $[0, T] \times \mathbb{R}^N$ for any T > 0. In this case also the equation (4.0.1) has a unique solution in $D_{\max}(\mathcal{A})$, for any $f \in C_b(\mathbb{R}^N)$.

When (4.0.7) holds, we say that $\{T(t)\}$ is conservative or that conservation of probability holds.

Finally, in Section 4.2 we give a nonuniqueness result. In the case when $c \equiv 0$, some other sufficient conditions implying that the problem (4.0.3) is not uniquely solvable are given in Section 5.2.

4.1 Conservation of probability and uniqueness

The following proposition describes the relation between the uniqueness of the elliptic equation (4.0.1) and the parabolic problem (4.0.3).

Proposition 4.1.1 Consider the following conditions:

- (i) for any $\lambda > c_0$ and any $f \in C_b(\mathbb{R}^N)$, the function $u = R(\lambda)f$ is the unique solution of the elliptic equation (4.0.1) in $D_{\max}(A)$;
- (ii) for any $f \in C_b(\mathbb{R}^N)$, the function $u = T(\cdot)f$ is the unique solution of the parabolic problem (4.0.3) which belongs to $C_b([0,T] \times \mathbb{R}^N) \cap C^{1,2}((0,T) \times \mathbb{R}^N)$ for any T > 0;
- (iii) for any $f \in C_b(\mathbb{R}^N)$, the function $u = T(\cdot)f$ is the unique solution of the parabolic problem (4.0.3) in $C([0, +\infty) \times \mathbb{R}^N) \cap C^{1,2}((0, +\infty) \times \mathbb{R}^N)$ satisfying $|u(t, x)| \leq M \exp(c_0 t)$ for some M > 0, any t > 0 and any $x \in \mathbb{R}^N$.

Then " $(i) \Rightarrow (iii)$ " and " $(ii) \Rightarrow (i)$ ".

Proof. "(i) \Rightarrow (iii)". Let $u \in C([0, +\infty) \times \mathbb{R}^N) \cap C^{1,2}((0, +\infty) \times \mathbb{R}^N)$ be a solution of the parabolic problem (4.0.3) with $u(0, \cdot) = 0$, satisfying

$$|u(t,x)| \le M \exp(c_0 t), \quad t > 0, \ x \in \mathbb{R}^N,$$
 (4.1.1)

for some M > 0. Let us prove that u(t, x) = 0 for any t > 0 and any $x \in \mathbb{R}^N$. For this purpose, fix $\lambda > c_0$, and consider the functions

$$\begin{split} v_{\infty}(x) &= \int_0^{+\infty} e^{-\lambda t} u(t,x) dt, \quad x \in \mathbb{R}^N, \\ v_n(x) &= \int_{1/n}^n e^{-\lambda t} u(t,x) dt, \quad x \in \mathbb{R}^N, \ n \in \mathbb{N}. \end{split}$$

Taking (4.1.1) into account, we easily deduce that $v_n \in C_b(\mathbb{R}^N)$ for any $n \in \mathbb{N} \cup \{0\}$ and

$$||v_n||_{\infty} \le \frac{M}{\lambda - c_0}, \quad n \in \mathbb{N} \cup \{0\}. \tag{4.1.2}$$

We now observe that, for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, we have

$$\mathcal{A}v_n(x) = \int_{1/n}^n e^{-\lambda t} \mathcal{A}u(t, x) dt$$

$$= \int_{1/n}^n e^{-\lambda t} D_t u(t, x) dt$$

$$= e^{-\lambda n} u(n, x) - e^{-\lambda/n} u(1/n, x) + \lambda v_n(x), \qquad (4.1.3)$$

which implies that $\mathcal{A}v_n \in C_b(\mathbb{R}^N)$, so that $v_n \in D_{\max}(\mathcal{A})$. Moreover, by (4.1.1), (4.1.2) and (4.1.3) it follows that there exists a constant C > 0 such that $\|\mathcal{A}v_n\|_{\infty} \leq C$ for any $n \in \mathbb{N}$, and

$$\lim_{n \to +\infty} \mathcal{A}v_n(x) = \lambda v_\infty(x),\tag{4.1.4}$$

for any $x \in \mathbb{R}^N$. We now define, for any $n \in \mathbb{N}$, the function $\varphi_n : \mathbb{R}^N \to \mathbb{R}$ by setting $\varphi_n = \lambda v_n - \mathcal{A}v_n$. By the above results we deduce that $\varphi_n \in C_b(\mathbb{R}^N)$ and, by (i),

$$v_n(x) = (R(\lambda)\varphi_n)(x) = \int_{\mathbb{R}^N} K_\lambda(x, y)\varphi_n(y)dy, \qquad x \in \mathbb{R}^N,$$
 (4.1.5)

where $R(\lambda)$ and K_{λ} are given by Theorem 2.1.3. Moreover, as we can easily see, there exists a constant C'>0 such that $\|\varphi_n\|_{\infty}\leq C'$ for any $n\in\mathbb{N}$. Thus, letting n go to $+\infty$ in (4.1.5), from (4.1.4) and the dominated convergence theorem, we get

$$\lim_{n \to +\infty} v_n(x) = 0, \qquad x \in \mathbb{R}^N,$$

that is $v_{\infty}(x) = 0$ for any $x \in \mathbb{R}^N$. Since $\lambda > c_0$ is arbitrary, by the uniqueness of the Laplace transform, we conclude that u(t,x) = 0 for any t > 0 and any $x \in \mathbb{R}^N$.

"(ii) \Rightarrow (i)". Let $v \in D_{\text{max}}(\mathcal{A})$ be a solution of the equation $\lambda v - \mathcal{A}v = 0$; let us prove that v = 0. By local regularity results for elliptic equations in

bounded domains, we know that $v \in C^{2+\alpha}_{loc}(\mathbb{R}^N)$. Then, the function

$$u(t,x) = e^{\lambda t}v(x), \qquad t > 0, \ x \in \mathbb{R}^N,$$

belongs to $u \in C([0, +\infty) \times \mathbb{R}^N) \cap C^{1,2}((0, +\infty) \times \mathbb{R}^N)$, and it is immediate to check that it is a solution of the problem (4.0.3), with $u(0, \cdot) = v$; besides u is bounded in $[0, T] \times \mathbb{R}^N$ for any T > 0. By (ii), we have u(t, x) = (T(t)v)(x) for any t > 0 and any $x \in \mathbb{R}^N$, and, then, by Theorem 2.2.5,

$$|u(t,x)| \le \exp(c_0 t) ||v||_{\infty}, \quad t > 0, \ x \in \mathbb{R}^N.$$

Since $\sup_{x \in \mathbb{R}^N} |u(t,x)| = \exp(\lambda t) \|v\|_{\infty}$ and $\lambda > c_0$, we conclude that $\|v\|_{\infty} = 0$.

4.1.1 Maximum principles

In this subsection we prove two maximum principles which provide us with uniqueness results for the solution to the equation (4.0.1) and the Cauchy problem (4.0.3). First we prove the following lemma which is a local maximum principle for functions in $W_{\text{loc}}^{2,p}(\mathbb{R}^N)$.

Lemma 4.1.2 Assume Hypotheses 4.0.1 and suppose that $u \in W^{2,p}_{loc}(\mathbb{R}^N)$, for any $p \in [1, +\infty)$, and that $Au \in C(\mathbb{R}^N)$. If x_0 is a local maximum (resp. minimum) of u, then

$$Au(x_0) - c(x_0)u(x_0) \le 0$$
, $(resp. Au(x_0) - c(x_0)u(x_0) \ge 0)$.

Proof. We limit ourselves to considering the case when x_0 is a local maximum point for u, since the case when x_0 is a local minimum follows easily from this one replacing u with -u. Moreover, without loss of generality, we can suppose that c(x) = 0 for any $x \in \mathbb{R}^N$. The general case follows from this one replacing the operator \mathcal{A} with the operator $\mathcal{A}_0 = \mathcal{A} - c$. Possibly replacing u with $u(\cdot - x_0) + C$ for a suitable constant C > 0, we can also assume that $x_0 = 0$ and u(0) > 0. Let r > 0 be such that $u_{|\overline{B}(r)}$ attains its maximum value at 0. Moreover, let $\psi \in C_c^{\infty}(\mathbb{R}^N)$ be such that $\chi_{B(r/2)} \leq \psi \leq \chi_{B(r)}$. Thus, 0

at 0. Moreover, let $\psi \in C_c^{\infty}(\mathbb{R}^N)$ be such that $\chi_{B(r/2)} \leq \psi \leq \chi_{B(r)}$. Thus, 0 is a global maximum point of the function $v = \psi u$. Moreover, the function

$$\mathcal{A}v = \psi \mathcal{A}u + u\mathcal{A}\psi + 2\langle QDu, D\psi \rangle$$

belongs to $C_b(\mathbb{R}^N)$ since, by the Sobolev embedding theorems (see [2, Theorem 5.4]), $u \in C^1(\mathbb{R}^N)$. Therefore, $v \in D_{\max}(A) \cap C_0(\mathbb{R}^N)$ and, consequently, Proposition 2.3.6 implies that $v \in D(\widehat{A})$. Now, since $c \equiv 0$, the family of measures $\{p(t, x; dy) : t > 0, x \in \mathbb{R}^N\}$, introduced in Theorem 2.2.5, defines a transition function and, by (2.2.8), we have

$$(T(t)v)(0)-v(0) \leq \int_{B(r)} (v(y)-v(0))p(t,0;dy) \leq 0.$$

Taking Proposition 2.3.6 into account, we now deduce that

$$\mathcal{A}u(0) = \mathcal{A}v(0) = \widehat{A}v(0) = \lim_{t \to 0^+} \frac{(T(t)v)(0) - v(0)}{t} \le 0.$$

The proof of Lemma 4.1.2 strongly relies on the representation formula (2.2.8) and on the characterization of $D(\widehat{A})$. We now give an alternative proof of Lemma 4.1.2 which has the advantage that it can be easier extended to the case when \mathbb{R}^N is replaced with an open set Ω and $u \in \bigcap_{1 \leq p < +\infty} W^{2,p}_{loc}(\Omega)$ satisfies $\lambda u - \mathcal{A}u \leq 0$ in Ω .

A second proof of Lemma 4.1.2. As in the other proof, we deal only with the case when x_0 is a relative maximum. Moreover, we assume that $c \equiv 0$, $x_0 = 0$ and $u(x) \geq 0$ for any $x \in B(r)$ and some r > 0.

Let us fix a function $\psi \in C_c^{\infty}(\mathbb{R}^N)$ compactly supported in B(r) and such that $\psi(0) = 1$, $D\psi(0) = 0$, $D^2\psi(0) = 0$ and $0 \le \psi(x) < 1$ for any $x \in B(r) \setminus \{0\}$. As it is easily seen, x = 0 is the unique maximum point of the function $v := \psi u$, and $\mathcal{A}v \in C(\mathbb{R}^N)$. Moreover, since $D\psi(0) = 0$ and $D^2\psi(0) = 0$, we deduce that $\mathcal{A}u(0) = \mathcal{A}v(0)$.

Now let $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ be a smooth function such that $0 \leq \varphi(x) \leq 1$ for any $x \in \mathbb{R}^N$, supp $(\varphi) \subset B(1)$, and with $\|\varphi\|_{L^1(\mathbb{R}^N)} = 1$. For any $n \in \mathbb{N}$ we set

$$\varphi_n(x) = n^N \varphi(nx), \qquad v_n = \varphi_n \star v \in C_c^{\infty}(\mathbb{R}^N),$$

where " \star " denotes the convolution operator. Since $v \in C_0(\mathbb{R}^N)$, then v_n converges to v, uniformly in \mathbb{R}^N , as n tends to $+\infty$. Moreover, for any $n \in \mathbb{N}$, v_n has an absolute maximum at some point $x_n \in \overline{B}(r+1/n)$ so that $Av_n(x_n) \leq 0$. Since $D_i v_n = \varphi_n \star D_i v$ and $D_{ij} v_n = \varphi_n \star D_{ij} v$ for any $i, j = 1, \ldots, N$, it follows that

$$\mathcal{A}v_{n}(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}v_{n}(x) + \sum_{j=1}^{N} b_{j}(x)D_{j}v_{n}(x)
= \int_{\mathbb{R}^{N}} \varphi_{n}(y) \sum_{i,j=1}^{N} q_{ij}(x-y)D_{ij}v(x-y)dy
+ \int_{\mathbb{R}^{N}} \varphi_{n}(y) \sum_{j=1}^{N} b_{j}(x-y)D_{j}v(x-y)dy
+ \int_{\mathbb{R}^{N}} \varphi_{n}(y) \sum_{i,j=1}^{N} (q_{ij}(x) - q_{ij}(x-y))D_{ij}v(x-y)dy
+ \int_{\mathbb{R}^{N}} \varphi_{n}(y) \sum_{j=1}^{N} (b_{j}(x) - b_{j}(x-y))D_{j}v(x-y)dy$$

$$= (\varphi_n \star \mathcal{A}v)(x)$$

$$+ \int_{B(1/n)} \varphi_n(y) \sum_{i,j=1}^N (q_{ij}(x) - q_{ij}(x-y)) D_{ij}v(x-y) dy$$

$$+ \int_{B(1/n)} \varphi_n(y) \sum_{i=1}^N (b_j(x) - b_j(x-y)) D_jv(x-y) dy,$$
(4.1.6)

for any $x \in \mathbb{R}^N$. Since $Av \in C_0(\mathbb{R}^N)$, it follows that $\varphi_n \star Av$ tends to Av uniformly in \mathbb{R}^N as n tends to $+\infty$. As far as the remaining terms in the last side of (4.1.6) are concerned, we observe that they vanish as n tends to $+\infty$, uniformly with respect to $x \in \mathbb{R}^N$. Indeed, if $p > N/\alpha$ (α being as in Hypothesis 4.0.1(ii)), then

$$\int_{B(1/n)} \varphi_n(y) \sum_{i,j=1}^N (q_{ij}(x) - q_{ij}(x-y)) D_{ij} v(x-y) dy
+ \int_{B(1/n)} \varphi_n(y) \sum_{j=1}^N (b_j(x) - b_j(x-y)) D_j v(x-y) dy
\leq C_1 n^{-\alpha} \|\varphi_n\|_{L^{p'}(\mathbb{R}^N)} \|D^2 v\|_{L^p(\mathbb{R}^N)} + C_2 n^{-\alpha} \|\varphi_n\|_{L^{p'}(\mathbb{R}^N)} \|Dv\|_{L^p(\mathbb{R}^N)}
= C_1 n^{-\alpha+N/p} \|\varphi\|_{L^{p'}(\mathbb{R}^N)} \|D^2 v\|_{L^p(\mathbb{R}^N)} + C_2 n^{-\alpha+N/p} \|\varphi\|_{L^{p'}(\mathbb{R}^N)} \|Dv\|_{L^p(\mathbb{R}^N)}.$$

Here 1/p + 1/p' = 1,

$$C_1^2 = \sum_{i,j=1}^N [q_{ij}]_{C^{\alpha}(B(r+2))}^2, \qquad C_2^2 = \sum_{j=1}^N [b_j]_{C^{\alpha}(B(r+2))}^2,$$

and $||Du||_p$, $||D^2u||_p$ denote the L^p -norms of the functions

$$x \mapsto \left(\sum_{j=1}^{N} |D_j u(x)|^2\right)^{1/2}, \quad \text{and} \quad x \mapsto \left(\sum_{i,j=1}^{N} |D_{ij} u(x)|^2\right)^{1/2},$$

respectively.

Summing up, we have proved that v_n and Av_n converge, respectively, to v and Av uniformly in \mathbb{R}^N as n tends to $+\infty$.

Since $\{x_n\}$ is a bounded sequence, using a compactness argument we easily see that, up to a subsequence, we can assume that $\{x_n\}$ converges to some point $\hat{x} \in \overline{B}(r)$ as n tends to $+\infty$. By continuity, $v(x_n)$ tends to $v(\hat{x})$ as n tends to $+\infty$. Hence,

$$v(\hat{x}) = \lim_{n \to +\infty} v(x_n) = \lim_{n \to +\infty} v_n(x_n) = \lim_{n \to +\infty} ||v_n||_{\infty} = ||v||_{\infty} = v(0),$$

where the second equality follows from the fact that v_n converges uniformly in \mathbb{R}^N to v. Recalling that 0 is the unique point where v attains its maximum

value, it follows that $\hat{x} = 0$. Since $Av_n \leq 0$ for any $n \in \mathbb{N}$ and it converges uniformly to Av, we immediately deduce that $Av(0) \leq 0$, and the assertion follows.

Theorem 4.1.3 Assume Hypotheses 4.0.1 and 4.0.2. Fix T > 0, $f \in C(\mathbb{R}^N)$, $g: (0,T] \times \mathbb{R}^N \to \mathbb{R}$. Further suppose that the function $u \in C([0,T] \times \mathbb{R}^N)$ is such that $D_t u \in C((0,T] \times \mathbb{R}^N)$, $u(t,\cdot) \in W^{2,p}_{loc}(\mathbb{R}^N)$, $Au(t,\cdot) \in C(\mathbb{R}^N)$ for any $t \in (0,T]$ and any $p \in [1,+\infty)$ and it solves the Cauchy problem

$$\begin{cases} D_t u(t, x) - \mathcal{A} u(t, x) = g(t, x), & t \in (0, T], \ x \in \mathbb{R}^N, \\ u(0, x) = f(x), & x \in \mathbb{R}^N. \end{cases}$$
(4.1.7)

Then, the following properties are met:

(i) If

$$\sup_{x \in \mathbb{R}^N} f(x) < +\infty, \qquad g \le 0 \text{ in } (0, T] \times \mathbb{R}^N,$$

and u satisfies

$$\lim \sup_{|x| \to +\infty} \left(\sup_{t \in [0,T]} \frac{u(t,x)}{\varphi(x)} \right) \le 0, \tag{4.1.8}$$

then

$$u(t,x) \le e^{c_0 t} \max \left\{ 0, \sup_{\mathbb{R}^N} f \right\}, \quad t \in [0,T], \quad x \in \mathbb{R}^N. \quad (4.1.9)$$

(ii) If

$$\inf_{x \in \mathbb{R}^N} f(x) > -\infty, \qquad g \ge 0 \text{ in } (0, T] \times \mathbb{R}^N,$$

u satisfies

$$\liminf_{|x| \to +\infty} \left(\inf_{t \in [0,T]} \frac{u(t,x)}{\varphi(x)} \right) \ge 0,$$

then

$$u(t,x) \ge e^{c_0 t} \min \left\{ 0, \inf_{\mathbb{R}^N} f \right\}, \quad t \in [0,T], \ x \in \mathbb{R}^N.$$
 (4.1.10)

(iii) In particular, for any $f \in C(\mathbb{R}^N)$ such that

$$\lim_{|x| \to +\infty} \frac{f(x)}{\varphi(x)} = 0,$$

there exists at most one solution of the parabolic problem (4.1.7) (with $g \equiv 0$) in $C([0, +\infty) \times \mathbb{R}^N)$ such that $D_t u \in C((0, T] \times \mathbb{R}^N)$, $u(t, \cdot) \in W^{2,p}_{loc}(\mathbb{R}^N)$, $\mathcal{A}u(t, \cdot) \in C(\mathbb{R}^N)$ for any $p \in [1, +\infty)$ and any $t \in (0, T]$, and such that (4.1.8) holds for any T > 0.

Proof. We limit ourselves to proving (4.1.9), since (4.1.10) can be obtained applying (4.1.9) to the function -u, whereas the last part of the statement is a straightforward consequence of (4.1.9) and (4.1.10).

We make some reductions. First, we assume that $c_0 \leq 0$. Indeed, in the case when $c_0 > 0$, it suffices to consider the function $\tilde{u} : [0,T] \times \mathbb{R}^N$ defined by $\tilde{u}(t,x) = e^{-c_0 t} u(t,x)$ for any $t \in [0,T]$ and any $x \in \mathbb{R}^N$, which satisfies (4.1.8) as well as the equation

$$D_t \tilde{u}(t,x) - \mathcal{A}_0 \tilde{u}(t,x) = e^{-c_0 t} g(t,x), \qquad t \in (0,T], \quad x \in \mathbb{R}^N,$$

where $A_0u = Au - c_0u$ has nonpositive zero-order coefficient and it satisfies Hypothesis 4.0.2 with the same Lyapunov function φ and the same λ_0 . Thus, from the case $c_0 = 0$ it follows that

$$\tilde{u}(t,x) \leq \max \left\{ 0, \sup_{\mathbb{R}^N} f \right\}, \qquad \ t \in [0,T], \ x \in \mathbb{R}^N,$$

which yields (4.1.9).

Besides, we can also assume that $\sup_{\mathbb{R}^N} f \leq 0$. Indeed, in the case when $\sup_{\mathbb{R}^N} f > 0$, it suffices to consider the function $\overline{u} = u - \sup_{\mathbb{R}^N} f$ that has a nonpositive initial value and it satisfies the equation

$$D_t \overline{u}(t,x) - A\overline{u}(t,x) = c(x) \sup_{\mathbb{R}^N} f + g(t,x), \qquad t \in (0,T], \ x \in \mathbb{R}^N,$$

where the right-hand side is nonpositive. Then, from the case $\sup_{\mathbb{R}^N} f \leq 0$ we get $\overline{u} \leq 0$, that is (4.1.9).

Taking these reductions into account, we introduce the function

$$v(t,x) = e^{-\lambda_0 t} u(t,x), \qquad t \in [0,T], \quad x \in \mathbb{R}^N,$$

where λ_0 is as in Hypothesis 4.0.2, and, for any $k \in \mathbb{N}$, we introduce the functions $v_k : [0,T] \times \mathbb{R}^N \to \mathbb{R}$ defined by

$$v_k(t,x) = v(t,x) - \frac{1}{k}\varphi(x), \qquad t \in [0,T], \quad x \in \mathbb{R}^N.$$

We prove that

$$v_k(t, x) \le 0,$$
 $t \in [0, T], x \in \mathbb{R}^N.$ (4.1.11)

First we observe that, by Hypothesis 4.0.2,

$$D_t v_k(t, x) - (\mathcal{A} - \lambda_0) v_k(t, x) = e^{-\lambda_0 t} g(t, x) + \frac{1}{k} (\mathcal{A} \varphi(x) - \lambda_0 \varphi(x)) \le 0, (4.1.12)$$

for any $t \in (0, T]$ and any $x \in \mathbb{R}^N$. Moreover, by Hypothesis 4.0.2 and assumption (4.1.8) we deduce that for any $k \in \mathbb{N}$ the function v_k has a maximum point (t_k, x_k) in $[0, T] \times \mathbb{R}^N$. If $t_k = 0$ then (4.1.11) follows, since $f \leq 0$ and $\varphi \geq 0$. If instead $t_k \in (0, T]$, then by Lemma 4.1.2 we have

$$(\mathcal{A} - c(x_k))v_k(t_k, x_k) \le 0, \qquad 0 \le D_t v_k(t_k, x_k),$$

which, combined with (4.1.12), yields

$$(\mathcal{A} - c(x_k))v_k(t_k, x_k) \le (\mathcal{A} - \lambda_0)v_k(t_k, x_k).$$

Since $\lambda_0 > c(x_k)$, we conclude that $v_k(t_k, x_k) \leq 0$, and (4.1.11) follows.

Now, (4.1.11) implies that $v(t,x) \leq k^{-1}\varphi(x)$ for any $t \in [0,T]$, any $x \in \mathbb{R}^N$ and any $k \in \mathbb{N}$. Letting k go to $+\infty$, we get $v \leq 0$, and then also $u \leq 0$, that is (4.1.9).

Remark 4.1.4 The maximum principle in Theorem 4.1.3 implies, in particular, that, under Hypotheses 4.0.1 and 4.0.2, the function $u = T(\cdot)f$ is the unique classical solution of the Cauchy problem (4.0.3).

Next, we prove two maximum principles for the elliptic equation (4.0.1). The former holds for any $\lambda > c_0$ and gives uniqueness of the bounded solution, while the latter holds only in the case when $\lambda > \lambda_0$ (recall that $\lambda_0 > c_0$, see Hypothesis 4.0.2) but it ensures uniqueness also for unbounded solutions.

Theorem 4.1.5 Let Hypotheses 4.0.1, 4.0.2 be satisfied and let $u \in C(\mathbb{R}^N) \cap W^{2,p}_{loc}(\mathbb{R}^N)$, for any $p \in [1, +\infty)$, be such that $Au \in C(\mathbb{R}^N)$. Further, fix $\lambda > c_0$. If u satisfies

$$\sup_{x \in \mathbb{R}^N} u(x) < +\infty \tag{4.1.13}$$

and

$$\lambda u(x) - \mathcal{A}u(x) \le 0, \qquad x \in \mathbb{R}^N,$$
 (4.1.14)

then u > 0.

Similarly, if u satisfies

$$\inf_{x \in \mathbb{R}^N} u(x) > -\infty \tag{4.1.15}$$

and

$$\lambda u(x) - \mathcal{A}u(x) \ge 0, \qquad x \in \mathbb{R}^N,$$
 (4.1.16)

then $u \geq 0$.

In particular, for any $f \in C_b(\mathbb{R}^N)$ and any $\lambda > c_0$, the function $R(\lambda)f$ (see Theorem 2.1.3) is the unique bounded solution to the elliptic equation (4.0.1) in $D_{\max}(A)$.

Proof. We prove the first part of the statement since the second part follows applying the first one to the function -u, while the last part is a straightforward consequence of the two previous parts. Thus, we assume (4.1.13) and (4.1.14).

Consider the function $\tilde{u}:[0,+\infty)\times\mathbb{R}^N\to\mathbb{R}$ defined by

$$\tilde{u}(t,x) = e^{\lambda t} u(x), \qquad t \ge 0, \quad x \in \mathbb{R}^N.$$

As it is easily seen, \tilde{u} satisfies the regularity assumptions in Theorem 4.1.3, the condition (4.1.8) and the equation

$$D_t \tilde{u}(t,x) - \mathcal{A}\tilde{u}(t,x) = e^{\lambda t} (\lambda u(x) - \mathcal{A}u(x)) \le 0, \qquad t > 0, \quad x \in \mathbb{R}^N.$$

Therefore,

$$u(x) \le e^{(c_0 - \lambda)t} \max \left\{ 0, \sup_{\mathbb{R}^N} u \right\}, \qquad t > 0, \quad x \in \mathbb{R}^N.$$

Letting t go to $+\infty$, we conclude that $u \leq 0$.

Theorem 4.1.6 Assume Hypotheses 4.0.1 and 4.0.2. Further, assume that $u \in C(\mathbb{R}^N) \cap W^{2,p}_{loc}(\mathbb{R}^N)$, for any $p \in [1, +\infty)$, is such that $Au \in C(\mathbb{R}^N)$,

$$\limsup_{|x| \to +\infty} \frac{u(x)}{\varphi(x)} \le 0$$
(4.1.17)

and

$$\lambda u(x) - \mathcal{A}u(x) \le 0, \qquad x \in \mathbb{R}^N,$$
 (4.1.18)

for some $\lambda \geq \lambda_0$. Then $u \leq 0$.

If u satisfies

$$\liminf_{|x| \to +\infty} \frac{u(x)}{\varphi(x)} \ge 0$$
(4.1.19)

and

$$\lambda u(x) - \mathcal{A}u(x) \ge 0, \quad x \in \mathbb{R}^N,$$
 (4.1.20)

for some $\lambda \geq \lambda_0$, then $u \geq 0$.

Proof. We prove the first part of the statement; then the second part follows applying the first one to the function -u.

Without loss of generality we can assume that $c_0 \leq 0$. Indeed, in the case when $c_0 > 0$, it suffices to replace the operator \mathcal{A} with the operator $\mathcal{A}_0 = \mathcal{A} - c_0$ which has a nonpositive zero order coefficient. Then, the Lyapunov function φ satisfies

$$\mathcal{A}_0 \varphi(x) - (\lambda_0 - c_0) \varphi(x) \le 0, \quad x \in \mathbb{R}^N,$$

and the function u satisfies

$$(\lambda - c_0)u(x) - \mathcal{A}_0 u(x) < 0, \quad x \in \mathbb{R}^N.$$

Therefore, from the case $c_0 \leq 0$ it follows that $u \leq 0$.

So, let us assume that $c_0 \leq 0$. Then, Hypothesis 4.0.2 implies that, for any $\lambda \geq \lambda_0$,

$$\mathcal{A}\varphi(x) - \lambda\varphi(x) \le 0, \quad x \in \mathbb{R}^N.$$
 (4.1.21)

Now, for any $k \in \mathbb{N}$, we consider the function

$$u_k(x) = u(x) - \frac{1}{k}\varphi(x), \quad x \in \mathbb{R}^N.$$

By (4.1.18) and (4.1.21) we have

$$\lambda u_k(x) - \mathcal{A}u_k(x) \le 0, \quad x \in \mathbb{R}^N.$$
 (4.1.22)

Moreover, by (4.1.17) and Hypothesis 4.0.2, the function u_k has a maximum point x_k in \mathbb{R}^N . By Lemma 4.1.2,

$$(\mathcal{A} - c(x_k))u_k(x_k) \le 0,$$

which, jointly with (4.1.22), yields

$$\lambda u_k(x_k) \le c(x_k)u_k(x_k).$$

Since $\lambda > c(x_k)$, it follows that $u_k(x_k) \leq 0$. This means that $u_k(x) \leq 0$ for any $x \in \mathbb{R}^N$ and any $k \in \mathbb{N}$ and, therefore, $u \leq 0$.

4.1.2 The case when $c \equiv 0$

We now consider the case when $c \equiv 0$. Note that in this case both the functions $u, v : [0, +\infty) \times \mathbb{R}^N \to \mathbb{R}$ defined by $u(t, x) = (T(\cdot) \mathbf{1})(x)$ and v(t, x) = 1 for any $t \in [0, +\infty)$ and any $x \in \mathbb{R}^N$ are bounded solutions of the parabolic problem (4.0.3) with initial value $f \equiv \mathbf{1}$. In general, they can differ at some point $x \in \mathbb{R}^N$. In any case, we have $0 \leq T(t)\mathbf{1} \leq \mathbf{1}$ for any t > 0, as follows from Remark 2.2.3.

With this remark in mind we give the following definition.

Definition 4.1.7 We say that $\{T(t)\}$ is conservative, or that the conservation of probability holds, if $T(t)\mathbf{1} \equiv \mathbf{1}$ for any t > 0.

Let us prove the following preliminary result.

Lemma 4.1.8 Assume Hypotheses 4.0.1 and let $c \equiv 0$. For any fixed $x \in \mathbb{R}^N$ the functions $t \mapsto (T(t)\mathbb{1})(x)$ and $\lambda \mapsto \lambda(R(\lambda)\mathbb{1})(x)$ are, respectively, decreasing and increasing in $(0, +\infty)$.

Proof. To prove the first part of the lemma, we observe that for any t, s > 0 and any $x \in \mathbb{R}^N$, we have $(T(t+s)\mathbb{1})(x) = (T(t)T(s)\mathbb{1})(x) \leq (T(t)\mathbb{1})(x)$. The inequality follows easily observing that, since T(t) maps nonnegative functions into nonnegative functions for any t > 0, then, for any pair of functions f_1 and f_2 with $f_1 \leq f_2$, it holds that $T(t)f_1 \leq T(t)f_2$ for any t > 0.

To prove the last part, we begin by observing that, according to Theorem 2.1.3, $0 \le \lambda R(\lambda) \mathbb{1} \le \mathbb{1}$ for any $\lambda > 0$. Let us now fix $0 < \lambda < \mu$. From the resolvent identity (2.1.5) we have

$$\begin{split} \lambda R(\lambda) \mathbb{1} - \mu R(\mu) \mathbb{1} &= \lambda (R(\mu) \mathbb{1} + (\mu - \lambda) R(\mu) R(\lambda) \mathbb{1}) - \mu R(\mu) \mathbb{1} \\ &= (\lambda - \mu) R(\mu) \mathbb{1} + \lambda (\mu - \lambda) R(\mu) R(\lambda) \mathbb{1} \\ &\leq (\lambda - \mu) R(\mu) \mathbb{1} + (\mu - \lambda) R(\mu) \mathbb{1} = 0. \end{split}$$

Remark 4.1.9 As the proof shows, the lemma holds also in the case when $c \leq 0$. We will use this generalization of Lemma 4.1.8 in Chapter 5.

Now, we can prove the following proposition.

Proposition 4.1.10 Assume Hypotheses 4.0.1 and suppose that $c \equiv 0$. Then, the following conditions are equivalent:

- (i) T(t)1 = 1 for some t > 0;
- (ii) $\{T(t)\}\ is\ conservative;$
- (iii) $R(\lambda)\mathbf{1} = \mathbf{1}/\lambda$ for some $\lambda > 0$;
- (iv) $R(\lambda)\mathbf{1} = \mathbf{1}/\lambda$ for any $\lambda > 0$;
- (v) for any $f \in C_b(\mathbb{R}^N)$, the function $u = T(\cdot)f$ is the unique solution of the parabolic problem (4.0.3) which belongs to $C_b([0,T] \times \mathbb{R}^N) \cap C^{1,2}((0,T) \times \mathbb{R}^N)$ for any T > 0;
- (vi) for any $\lambda > 0$ and any $f \in C_b(\mathbb{R}^N)$, the function $u = R(\lambda)f$ is the unique solution of the elliptic equation (4.0.1) in $D_{\max}(A)$.

Proof. Showing that "(ii) \Rightarrow (i)", "(iv) \Rightarrow (iii)" and "(vi) \Rightarrow (iv)" is trivial, while the implication "(v) \Rightarrow (vi)" has been shown in Proposition 4.1.1. Hence, we can limit ourselves to showing that "(i) \Rightarrow (ii)", "(iii) \Rightarrow (ii)" and "(ii) \Rightarrow (v)".

- "(i) \Rightarrow (ii)". Fix $x \in \mathbb{R}^N$. If $T(t_0)\mathbf{1} \equiv \mathbf{1}$ for some $t_0 > 0$, then, according to the semigroup law, the function $t \mapsto (T(t)\mathbf{1})(x)$ is periodic, with period t_0 . By Lemma 4.1.8 it is also decreasing. Hence, it is constant.
- "(iii) \Rightarrow (ii)" follows from (2.2.14). Indeed, if (T(t)1)(x) were less than 1 for some t > 0 and $x \in \mathbb{R}^N$, from (2.2.14) it would follow that $(R(\lambda)1)(x) < 1/\lambda$.
- "(ii) \Rightarrow (v)". Let u be a solution of the parabolic problem (4.0.3), with $f \equiv 0$, which belongs to $C_b([0,T] \times \mathbb{R}^N) \cap C^{1,2}((0,T) \times \mathbb{R}^N)$ for any T > 0. We prove

that $u \equiv 0$. On the contrary, suppose that $u \not\equiv 0$. By changing sign and multiplying u by a suitable constant if needed, we can assume that

$$\sup_{(0,T)\times\mathbb{R}^N}u>0,\qquad \sup_{(0,T)\times\mathbb{R}^N}|u|<1.$$

Set v = 1 - u. Then v(t, x) > 0 for any t > 0 and any $x \in \mathbb{R}^N$, and

$$\inf_{(0,T)\times\mathbb{R}^N}v<1.$$

Moreover, v is a bounded solution of the problem (4.0.3) with initial value $v(0,\cdot) = 1$. Then, by Remark 2.2.3, we have $T(\cdot)1 \le v$. This implies that $\inf_{\mathbb{R}^N} T(t)1 < 1$ for some $t \in (0,T)$, which is in contradiction with statement (ii).

The following proposition is an immediate consequence of the formula (2.2.8) with $f \equiv 1$.

Proposition 4.1.11 Let $\{p(t,x;dy): t>0, x\in\mathbb{R}^N\}$ be the transition function associated with the semigroup $\{T(t)\}$ (see Theorem 2.2.5). Then, $\{T(t)\}$ is conservative if and only if $p(t,x;\mathbb{R}^N)=1$ for any t>0 and any $x\in\mathbb{R}^N$.

Similarly, taking Theorem 4.1.3 into account, the following sufficient condition, ensuring that $\{T(t)\}$ is conservative, follows.

Proposition 4.1.12 Suppose that $c \equiv 0$ and assume Hypotheses 4.0.1 and 4.0.2. Then, the conservation of probability holds.

Example 4.1.13 Consider the operator A defined on smooth functions by

$$\mathcal{A}f = \Delta f + \langle b(\cdot), Df \rangle,$$

where $b \in C^{\alpha}_{loc}(\mathbb{R}^N)$ for some $\alpha \in (0,1)$. Choosing $\varphi(x) = \log(m+|x|^2)$ for some m > 1, the condition (4.0.5) becomes

$$\langle b(x), x \rangle \le M(m + |x|^2) \log(m + |x|^2), \qquad x \in \mathbb{R}^N,$$
 (4.1.23)

for some M>0 (see (4.0.6)). Therefore, if (4.1.23) holds the corresponding semigroup $\{T(t)\}$ is conservative.

For instance, this is the case when $b(x) = -x|x|^2$ for any $x \in \mathbb{R}^N$. Note that the one-dimensional version of the operator \mathcal{A} , with $b(x) = -x^3$ for any $x \in \mathbb{R}$, has been already considered in Chapter 3 (see Example 3.2.4), where, in fact, we proved, with a different argument, that the associated semigroup is conservative.

Remark 4.1.14 Let us notice that the papers [9, 19, 50] contain some uniqueness results for the classical solution to the problem (4.0.3), with a prescribed growth rate at infinity. In particular such results apply to prove the uniqueness of the bounded solution to the problem (4.0.3). We stress that the results in [9, 19, 50] apply to operators with diffusion coefficients with an at most quadratic growth at infinity.

4.2 Nonuniqueness

By the results in Chapter 3 (see Theorem 3.2.2), in the one-dimensional setting, one can give necessary and sufficient conditions for the elliptic equation (4.0.1) to be not uniquely solvable in $D_{\text{max}}(\mathcal{A})$. Such conditions can be expressed in terms of integrability properties at $\pm \infty$ of the functions Q and R in (3.1.4) and (3.1.5).

In this section we prove a sufficient condition for the nonuniqueness of the solution $v \in D_{\max}(\mathcal{A})$ of the elliptic equation (4.0.1) in the N-dimensional setting (see [83, Section 3.4], or [116, Remark 3.16]). According to Proposition 4.1.1, this result will imply that the parabolic problem (4.0.3) admits several solutions belonging to $C([0, +\infty) \times \mathbb{R}^N) \cap C^{1,2}((0, +\infty) \times \mathbb{R}^N)$ which are bounded in $[0, T] \times \mathbb{R}^N$ for any T > 0. Here, we do not assume that $c \equiv 0$.

Proposition 4.2.1 Suppose that there exists a nonidentically vanishing function $\varphi \in C^2(\mathbb{R}^N)$, such that

$$0 \le \varphi \le C, \qquad \lambda \varphi - \mathcal{A} \varphi \le 0,$$
 (4.2.1)

for some $\lambda > c_0$ and some C > 0. Then, there exists a bounded, continuous and nonidentically vanishing solution of the equation $\lambda u - Au = 0$.

Proof. For any $n \in \mathbb{N}$ consider the solution $u_n \in C(\overline{B}(n))$ of the Dirichlet problem

$$\begin{cases} \lambda u_n(x) - \mathcal{A}u_n(x) = 0, & x \in B(n), \\ u_n(x) = \|\varphi\|_{\infty}, & x \in \partial B(n). \end{cases}$$

From (4.2.1) and the maximum principle it follows that

$$\varphi \le u_n \le \|\varphi\|_{\infty}. \tag{4.2.2}$$

Moreover, $\{u_n\}$ is a decreasing sequence. Indeed, the function $v = u_{n+1} - u_n$ solves the problem

$$\begin{cases} \lambda v(x) - \mathcal{A}v(x) = 0, & x \in B(n), \\ v(x) = u_{n+1}(x) - \|\varphi\|_{\infty} \le 0, & x \in \partial B(n) \end{cases}$$

and, then, it is negative, still by virtue of the maximum principle. Thus, we can define the function $u: \mathbb{R}^N \to \mathbb{R}$ by setting

$$u(x) := \lim_{n \to +\infty} u_n(x), \qquad x \in \mathbb{R}^N.$$

By (4.2.2), u is bounded. Taking the *a priori* interior estimates of Theorem C.1.1 into account and arguing as in the proof of Theorem 2.1.1, it follows that u is a continuous and nonidentically vanishing solution of the equation $\lambda u - \mathcal{A}u = 0$ in \mathbb{R}^N .

Remark 4.2.2 In Section 5.2 we will see other conditions implying nonuniqueness of the bounded solutions. Actually, we will see sufficient conditions ensuring that $T(t)(C_b(\mathbb{R}^N)) \subset C_0(\mathbb{R}^N)$ for any t > 0. It is clear that, in such a situation when $c \equiv 0$, uniqueness does not hold, since we have $T(t)\mathbf{1} \not\equiv \mathbf{1}$.

Chapter 5

Properties of $\{T(t)\}\$ in spaces of continuous functions

5.0 Introduction

In this chapter we study several properties of the semigroup $\{T(t)\}$ in $C_b(\mathbb{R}^N)$. More precisely, we deal with: the compactness of the semigroup in $C_b(\mathbb{R}^N)$, the inclusion $T(t)(C_b(\mathbb{R}^N)) \subset C_0(\mathbb{R}^N)$ and the invariance of $C_0(\mathbb{R}^N)$ under the action of the semigroup.

Compactness. In Section 5.1 we study the compactness of $\{T(t)\}$ in $C_b(\mathbb{R}^N)$. First, we consider the conservative case and, then, the nonconservative one. In the conservative case T(t) is a compact operator, for any t>0, provided that the family $\{p(t,x;dy):x\in\mathbb{R}^N\}$ is tight for any t>0 (see Definition 5.1.2). Here, $\{p(t,x;dy):t>0,\ x\in\mathbb{R}^N\}$ is the transition family defined in Theorem 2.2.5. In such a case the function $t\mapsto T(t)$ is continuous in $(0,+\infty)$ with respect to the operator topology.

We show that the family $\{p(t,x;dy):x\in\mathbb{R}^N\}$ is tight for any t>0 if there exist a strictly positive function $\varphi\in C^2(\mathbb{R}^N)$ and a convex function $g:[0,+\infty)\to\mathbb{R}$ such that

$$\lim_{|x|\to+\infty}\varphi(x)=+\infty, \ \ 1/g \text{ is integrable at } +\infty, \ \ \mathcal{A}\varphi(x)\leq -g(\varphi(x)), \ \ x\in\mathbb{R}^N.$$

It is not difficult to give sufficient conditions on the coefficients of the operator \mathcal{A} guaranteeing that the previous conditions are satisfied. For instance, this is the case of the semigroup generated by the operator

$$\mathcal{A}\varphi(x) = \Delta\varphi(x) - |x|^2 \langle x, D\varphi(x) \rangle, \quad x \in \mathbb{R}^N.$$

In the nonconservative case, we show that there are some connections between the compactness of T(t) (t>0) and the fact that T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$. More precisely, we show that T(t) is compact and maps $C_0(\mathbb{R}^N)$ into itself, for some t>0, if and only if T(t) belongs to $C_0(\mathbb{R}^N)$. In such a case (see Proposition 5.1.12), the resolvent operator $R(\lambda, \widehat{A})$ (see Sections 2.1 and 2.3 for the definitions of \widehat{A} and the resolvent operator $R(\lambda, \widehat{A})$) is compact for any $\lambda > 0$ and the semigroup is norm-continuous.

The inclusion $T(t)(C_b(\mathbb{R}^N)) \subset C_0(\mathbb{R}^N)$. It is not always true that T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ for any t > 0, since it is not always true that

 $C_0(\mathbb{R}^N)$ is invariant under $\{T(t)\}$. For instance, if, for some t > 0, T(t) is conservative and compact, then $C_0(\mathbb{R}^N)$ is not invariant under T(t). In fact, T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$, for some t > 0, if and only if $T(t)\mathbb{1} \in C_0(\mathbb{R}^N)$.

As it is shown in Proposition 5.2.2, T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ for any t > 0, if and only if the resolvent operator $R(\lambda) = R(\lambda, \widehat{A})$ does, for some (and, hence, for any) $\lambda > 0$ or, equivalently, if and only if $C_0(\mathbb{R}^N) \cap D_{\max}(A)$ is the domain of the weak generator \widehat{A} of the semigroup $\{T(t)\}$.

A sufficient condition implying that $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$ is given in Proposition 5.2.4 in the case when there exists a function $\varphi \in C^2(\mathbb{R}^N) \cap C_0(\mathbb{R}^N)$, which is positive outside a compact subset K and satisfies

$$\inf_{x \in \mathbb{R}^N \setminus K} (\lambda \varphi(x) - \mathcal{A}\varphi(x)) = a > 0,$$

for some $\lambda > c_0$, where, as usual, $c_0 = \sup_{x \in \mathbb{R}^N} c(x)$.

Invariance of $C_0(\mathbb{R}^N)$. As it has already been remarked above, it is not always true that $T(t)(C_0(\mathbb{R}^N)) \subset C_0(\mathbb{R}^N)$ for any t > 0. This occurs, for instance, when there exists a strictly positive function $\varphi \in C^2(\mathbb{R}^N) \cap C_0(\mathbb{R}^N)$ such that

$$\lambda \varphi - \mathcal{A} \varphi \ge 0$$
,

for some $\lambda > 0$. In such a case the operator A_0 , defined by $A_0 u = \mathcal{A}u$ for any $u \in D(A_0) = \{u \in D_{\max}(\mathcal{A}) \cap C_0(\mathbb{R}^N) : \mathcal{A}u \in C_0(\mathbb{R}^N)\}$ (i.e., the part of \widehat{A} in $C_0(\mathbb{R}^N)$), is the generator of a strongly continuous semigroup.

5.1 Compactness of $\{T(t)\}$

In this section we study the compactness of the semigroup $\{T(t)\}$ in the space $C_b(\mathbb{R}^N)$ of bounded and continuous functions. The results that we present here are taken from [44] and [115]. In this and in the subsequent sections we assume that the coefficients of the operator \mathcal{A} , defined on smooth functions by

$$Au = \sum_{i,j=1}^{N} q_{ij} D_{ij} u + \sum_{j=1}^{N} b_{j} D_{j} u + cu,$$

satisfy the same hypotheses of Chapter 2. For the reader's convenience we state them again.

Hypotheses 5.1.1 (i) $q_{ij} \equiv q_{ji}$,

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \qquad \kappa(x) > 0, \quad \xi, x \in \mathbb{R}^N;$$

- (ii) q_{ij} , b_i (i, j = 1, ..., N) and c belong to $C_{loc}^{\alpha}(\mathbb{R}^N)$ for some $\alpha \in (0, 1)$;
- (iii) there exists $c_0 \in \mathbb{R}$ such that

$$c(x) \le c_0, \qquad x \in \mathbb{R}^N.$$

We consider both the cases when the semigroup is conservative and nonconservative.

5.1.1 The conservative case

We begin this subsection recalling that the semigroup $\{T(t)\}$ is conservative whenever $T(t)\mathbb{1} = \mathbb{1}$ for any t > 0 or, equivalently, whenever the elliptic equation

$$\lambda u(x) - \mathcal{A}u(x) = f(x), \quad x \in \mathbb{R}^N,$$
 (5.1.1)

and the Cauchy problem

$$\begin{cases}
D_t u(t,x) - \mathcal{A}u(t,x) = 0, & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N,
\end{cases} (5.1.2)$$

are uniquely solvable, respectively, in

$$D_{\max}(\mathcal{A}) = \left\{ u \in C_b(\mathbb{R}^N) \cap \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2,p}(\mathbb{R}^N) : \ \mathcal{A}u \in C_b(\mathbb{R}^N) \right\}$$

and in $C_b([0,T] \times \mathbb{R}^N) \cap C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty) \times \mathbb{R}^N)$, for any T > 0 (see Proposition 4.1.10). Of course, if $\{T(t)\}$ is conservative then, necessarily, $c \equiv 0$.

We also recall that we can associate a transition family $\{p(t, x; dy) : t > 0, x \in \mathbb{R}^N\}$ with the semigroup (see Definition 2.2.4 and Theorem 2.2.5). The connection between the semigroup and the transition family is explained by the formula (2.2.8) which allows us to write T(t)f for any t > 0 and any $f \in C_b(\mathbb{R}^N)$ in terms of the Borel measures p(t, x; dy). For the reader's convenience we rewrite here such a formula:

$$(T(t)f)(x) = \int_{\mathbb{R}^N} f(y)p(t, x; dy), \quad t > 0, \ x \in \mathbb{R}^N, \quad f \in C_b(\mathbb{R}^N).$$
(5.1.3)

With all these remarks in mind we can now show that the compactness of the semigroup $\{T(t)\}$ is equivalent to the tightness of the family of measures $\{p(t,x;\cdot): x\in\mathbb{R}^N\}$ for any t>0, and it implies that the semigroup is norm-continuous in $(0,+\infty)$ (see Definition B.2.4).

Definition 5.1.2 A family of Borel probability measures $\{\mu_{\alpha}\}_{{\alpha}\in\mathcal{F}}$ is said to be tight if for any $\varepsilon > 0$ there exists $\rho > 0$ such that $\mu_{\alpha}(B(\rho)) \geq 1 - \varepsilon$ for any $\alpha \in \mathcal{F}$.

Proposition 5.1.3 Assume that the semigroup $\{T(t)\}$ is conservative. Then, T(t) is compact in $C_b(\mathbb{R}^N)$, for any t > 0, if and only if the family of measures $\{p(t, x; \cdot) : x \in \mathbb{R}^N\}$ is tight for any t > 0. In this case the semigroup is norm-continuous in $(0, +\infty)$, and the resolvent $R(\lambda, \widehat{A})$ is compact for any $\lambda > 0$.

Proof. Fix $t, \varepsilon > 0$ and suppose that T(t) is compact. Let $\{f_n\}$ be a sequence of bounded and continuous functions in \mathbb{R}^N such that $\chi_{B(n-1)} \leq f_n \leq \chi_{B(n)}$ for any $n \in \mathbb{N}$. By Proposition 2.2.9 we deduce that $T(t)f_n$ tends to $T(t)\mathbf{1} = 1$ as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . Since T(t) is compact, then $T(t)f_n$ tends to 1 uniformly in \mathbb{R}^N . Indeed, if this were not the case one could find out a positive number ε_0 and a subsequence $\{f_{n_k}\}$ such that

$$||T(t)f_{n_k} - 1||_{\infty} \ge \varepsilon_0.$$

Hence, no subsequences uniformly converging in \mathbb{R}^N could be extracted from the sequence $\{T(t)f_{n_k}\}$, contradicting the compactness of T(t). It follows that the family of measures $\{p(t,x;\cdot):x\in\mathbb{R}^N\}$ is tight for any t>0, since (see (5.1.3))

$$p(t, x; B(n)) \ge (T(t)f_n)(x), \qquad x \in \mathbb{R}^N.$$

Conversely, assume that, for any t > 0, the family $\{p(t, x; \cdot) : x \in \mathbb{R}^N\}$ is tight. Fix t > 0 and let $\rho_n > 0$ be such that

$$p(t/2, x; B(\rho_n)) \ge 1 - n^{-1}, \quad x \in \mathbb{R}^N.$$
 (5.1.4)

Define the operator $S_{t,n}$ in $C_b(\mathbb{R}^N)$ by

$$(S_{t,n}f)(x) = \int_{B(\rho_n)} (T(t/2)f)(y)p(t/2, x; dy),$$

for any $x \in \mathbb{R}^N$ and any $f \in C_b(\mathbb{R}^N)$. By (5.1.4), we have

$$|(T(t)f)(x) - (S_{t,n}f)(x)| \le \int_{\mathbb{R}^N \setminus B(\rho_n)} |(T(t/2)f)(y)| p(t/2, x; dy) \le n^{-1} ||f||_{\infty},$$
(5.1.5)

for any $f \in C_b(\mathbb{R}^N)$ and any $x \in \mathbb{R}^N$, since $\{T(t)\}$ is a semigroup of contractions (see (2.2.7)). Moreover, $S_{t,n}$ is compact for any $n \in \mathbb{N}$. Indeed, we have

$$S_{t,n} = G_n \circ R_n \circ T(t/2), \qquad n \in \mathbb{N},$$

where $R_n: C_b(\mathbb{R}^N) \to C(\overline{B}(\rho_n))$ is the restriction operator (i.e., $R_n f(x) = f(x)$ for any $x \in \overline{B}(\rho_n)$ and any $f \in C_b(\mathbb{R}^N)$) and $G_n: C(\overline{B}(\rho_n)) \to C_b(\mathbb{R}^N)$ is the bounded operator defined by

$$(G_n h)(x) = \int_{B(\rho_n)} h(y) p(t/2, x; dy) = T(t/2) (h\chi_{B(n)}), \qquad x \in \mathbb{R}^N,$$

for any $h \in C(\overline{B}(\rho_n))$. We observe that $G_n h \in C_b(\mathbb{R}^N)$ for any $h \in C(\overline{B}(\rho_n))$, since $\{T(t)\}$ has the strong Feller property (see Proposition 2.2.12). By the interior Schauder estimates in Theorem C.1.4, we deduce that $R_n \circ T(t/2)$ is bounded from $C_b(\mathbb{R}^N)$ into $C^{2+\alpha}(\overline{B}(\rho_n))$, and the Ascoli-Arzelà Theorem implies that $R_n \circ T(t/2)$ is compact. It follows that $S_{t,n}$ is compact as well. Since $\{S_{t,n}\}$ is a sequence of compact operators converging to T(t) in the operator topology (see (5.1.5)), we deduce that T(t) is, itself, a compact operator. Due to the arbitrariness of t > 0, it follows that T(t) is compact for any t > 0.

Let us now prove that T(t) is norm-continuous for any t>0. We begin by showing that T(t)f is continuous in $(0,+\infty)$ for any $f\in C_b(\mathbb{R}^N)$ with $\|f\|_{\infty}\leq 1$. For this purpose, we fix $\varepsilon>0$, $t_0>0$ and let ρ be sufficiently large such that

$$p(t_0, x; \mathbb{R}^N \setminus B(\rho)) \le \varepsilon, \qquad x \in \mathbb{R}^N.$$

Then, for any h > 0 it holds that

$$(T(t_0 + h)f)(x) - (T(t_0)f)(x)$$

$$= \int_{\mathbb{R}^N} f(y)p(t_0 + h, x; dy) - \int_{\mathbb{R}^N} f(y)p(t_0, x; dy)$$

$$= \int_{B(\rho)} ((T(h)f)(y) - f(y)) p(t_0, x; dy)$$

$$+ \int_{\mathbb{R}^N \setminus B(\rho)} ((T(h)f)(y) - f(y)) p(t_0, x; dy)$$

$$\leq ||T(h)f - f||_{C(\overline{B}(\rho))} + 2\varepsilon$$
(5.1.6)

and the right-hand side of (5.1.6) is less than 3ε provided h is sufficiently small since $T(\cdot)f \in C([0,+\infty)\times\mathbb{R}^N)$ and T(0)f=f (see Theorems 2.2.1, 2.2.5). Hence,

$$\lim_{h \to 0^+} ||T(t_0 + h)f - T(t_0)f||_{\infty} = 0.$$

For negative values of h we fix $\delta \in (0, t_0)$. Then, for any $h \in (\delta - t_0, 0)$ we split

$$T(t_0 + h)f - T(t_0)f = T(t_0 + h - \delta)(T(\delta)f - T(\delta - h)f).$$

Hence,

$$||T(t_0 + h)f - T(t_0)f||_{\infty} = ||T(t_0 + h - \delta)(T(\delta)f - T(\delta - h)f)||_{\infty}$$

$$\leq ||T(\delta)f - T(\delta - h)f||_{\infty}$$

and letting h go to 0^- , by the previous step, we get

$$\lim_{h \to 0^{-}} T(t_0 + h)f = T(t_0)f. \tag{5.1.7}$$

Hence,

$$\lim_{h \to 0} ||T(t_0 + h)f - T(t_0)f||_{\infty} = 0, \qquad t > 0.$$
 (5.1.8)

We can now show that (5.1.8) and the compactness of the semigroup imply that $t \mapsto T(t)$ is norm-continuous in $(0, +\infty)$. Suppose by contradiction that T(t) is not norm-continuous in $(0, +\infty)$. Then, we can find out $t_0, \varepsilon > 0$ two sequences $\{f_n\} \subset C_b(\mathbb{R}^N)$, with $\|f_n\|_{\infty} = 1$, for any $n \in \mathbb{N}$, and $\{t_n\} \subset (0, +\infty)$, tending to t_0 as n tends to $+\infty$, such that

$$\|(T(t_n) - T(t_0))f_n\|_{\infty} \ge \varepsilon, \qquad n \in \mathbb{N}. \tag{5.1.9}$$

Without loss of generality, we can assume that $\{t_n\}$ is a monotone sequence. We first assume that it is decreasing and we set $a_n = t_n - t_0$ for any $n \in \mathbb{N}$. Since $T(t_0)$ is a compact operator, up to a subsequence we can assume that $T(t_0)f_n$ converges uniformly to a function $g \in C_b(\mathbb{R}^N)$ as n tends to $+\infty$. Hence, from (5.1.9), we get

$$||(T(a_n) - I)g||_{\infty} \ge ||(T(a_n) - I)T(t_0)f_n||_{\infty} - ||(T(a_n) - I)(g - T(t_0)f_n)||_{\infty}$$

$$\ge \varepsilon - 2||g - T(t_0)f_n||_{\infty}.$$

Letting n go to $+\infty$, we get

$$\limsup_{n \to +\infty} \|(T(a_n) - I)g\|_{\infty} \ge \varepsilon,$$

which contradicts (5.1.8). Therefore, $\lim_{t\to t_0^+} T(t) = T(t_0)$ in $L(C_b(\mathbb{R}^N))$. Now to prove that T(t) is norm continuous from the left in $(0, +\infty)$, it suffices to argue as in the proof of (5.1.7).

To conclude the proof, we observe that, since T(t) is norm-continuous and bounded in $(0, +\infty)$, the integral

$$\int_0^{+\infty} e^{-\lambda t} T(t) dt$$

converges in the operator topology to $R(\lambda)$, for any $\lambda > 0$. Since T(t) is compact for any t > 0, the operator $R(\lambda)$ is compact as well.

Now, we prove a lemma.

Lemma 5.1.4 Suppose that $\{T(t)\}$ is conservative. Let $\varphi \in C^2(\mathbb{R}^N)$ be any function such that

$$\lim_{|x| \to +\infty} \varphi(x) = +\infty, \qquad \mathcal{A}\varphi(x) \le K, \qquad x \in \mathbb{R}^N,$$

for some K > 0. Then, we have

$$(T(t)\varphi)(x) := \int_{\mathbb{R}^N} \varphi(y)p(t,x;dy) \in \mathbb{R}, \tag{5.1.10}$$

$$(T(t)\mathcal{A}\varphi)(x) := \int_{\mathbb{R}^N} \mathcal{A}\varphi(y)p(t,x;dy) \in \mathbb{R}, \qquad (5.1.11)$$

for any t > 0 and any $x \in \mathbb{R}^N$. Moreover, the function $T(\cdot)\varphi$ belongs to $C^{1+\alpha/2,2+\alpha}_{\mathrm{loc}}((0,+\infty)\times\mathbb{R}^N)\cap C([0,+\infty)\times\mathbb{R}^N)$ and

$$\lim_{t \to 0^+} (T(t)\varphi)(x) = \varphi(x), \quad \left(\frac{\partial}{\partial t} T(t)\varphi\right)(x) \le (T(t)\mathcal{A}\varphi)(x), \quad t > 0, \ x \in \mathbb{R}^N.$$
(5.1.12)

Proof. Without loss of generality, we can suppose that $\varphi \geq 0$. Indeed, if this is not the case, we replace φ with $\varphi - \inf_{x \in \mathbb{R}^N} \varphi(x)$. Observe that $\inf_{x \in \mathbb{R}^N} \varphi(x)$ is a real number, since φ goes to $+\infty$ as |x| goes to $+\infty$. Moreover, $\mathcal{A}\varphi = \mathcal{A}(\varphi - \inf_{x \in \mathbb{R}^N} \varphi(x))$, since the zero-order coefficient of \mathcal{A} identically vanishes in \mathbb{R}^N .

For any $n \in \mathbb{N}$, let $\psi_n \in C_b^2([0,+\infty))$ be any function such that

$$\psi_n(s) = s, \quad s \in [0, n], \quad \psi_n(s) = n + 1, \quad s \ge n + 2,$$

 $0 \le \psi'_n(s) \le 1, \quad \psi''_n(s) \le 0 \quad s \in [0, +\infty).$

Then $\psi_n \circ \varphi$ is bounded and, in particular, it is constant outside some compact set of \mathbb{R}^N . Consequently, the function $\mathcal{A}(\psi_n \circ \varphi)$ has compact support. This implies that $\psi_n \circ \varphi \in D_{\max}(\mathcal{A})$. By Propositions 2.3.6 and 4.1.10, we know that $D_{\max}(\mathcal{A}) = D(\widehat{A})$ (see Section 2.3). Hence, by Lemma 2.3.3 we deduce that

$$T(t)\mathcal{A}(\psi_n \circ \varphi) = T(t)\widehat{A}(\psi_n \circ \varphi) = \widehat{A}T(t)(\psi_n \circ \varphi) = \mathcal{A}T(t)(\psi_n \circ \varphi), \quad t \ge 0.$$
(5.1.13)

Now, let

$$u_n(t,x) = (T(t)(\psi_n \circ \varphi))(x) := \int_{\mathbb{R}^N} \psi_n(\varphi(y)) p(t,x;dy), \qquad t > 0, \ x \in \mathbb{R}^N.$$

Thanks to our assumptions on the functions ψ_n $(n \in \mathbb{N})$, we easily deduce that the sequence $\{\psi_n(s)\}$ is increasing to $\varphi(s)$ for any $s \in \mathbb{R}$. Hence, the monotone convergence theorem yields

$$\lim_{n \to \infty} u_n(t, x) = \lim_{n \to +\infty} \int_{\mathbb{R}^N} \psi_n(\varphi(y)) p(t, x; dy) = u(t, x) := (T(t)\varphi)(x),$$
(5.1.14)

for any t > 0 and any $x \in \mathbb{R}^N$. Now, from (5.1.13) we deduce that, for any s > 0 and any $x \in \mathbb{R}^N$,

$$D_{t}u_{n}(s,x) = \mathcal{A}u_{n}(s,x) = (T(s)\mathcal{A}(\psi_{n} \circ \varphi))(x)$$

$$= \int_{\mathbb{R}^{N}} \left\{ (\psi'_{n} \circ \varphi)\mathcal{A}\varphi + (\psi''_{n} \circ \varphi)\langle QD\varphi, D\varphi\rangle \right\} p(s,x;dy)$$

$$\leq \int_{\mathbb{R}^{N}} (\psi'_{n} \circ \varphi)\mathcal{A}\varphi p(s,x;dy) \leq K, \qquad (5.1.15)$$

where $Q(y) = (q_{ij}(y))$. Integrating (5.1.15) with respect to $s \in [0, t]$, and recalling that the semigroup is positivity preserving, gives

$$0 \le u_n(t,x) \le \psi_n(\varphi(x)) + Kt \le \varphi(x) + Kt, \qquad t > 0, \quad x \in \mathbb{R}^N, \quad (5.1.16)$$

for any $n \in \mathbb{N}$. Then, letting n go to $+\infty$, we get

$$0 \le u(t,x) \le \varphi(x) + Kt < +\infty, \qquad t > 0, \quad x \in \mathbb{R}^N,$$

and (5.1.10) follows.

We now prove (5.1.11) and (5.1.12) and, at the same time, we show that $u \in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\mathbb{R}^N)$. By (5.1.16) and Theorem C.1.4, the sequence $\{u_n\}$ is bounded in $C^{1+\alpha/2,2+\alpha}([\varepsilon,T]\times F)$ for any $\varepsilon,T>0$ and any compact set $F\subset\mathbb{R}^N$. From (5.1.14) and the Ascoli-Arzelà Theorem, we deduce that $u\in C^{1+\alpha/2,2+\alpha}([\varepsilon,T]\times F)$ and u_n converges to u in $C^{1+\beta/2,2+\beta}([\varepsilon,T]\times F)$ for any $\beta<\alpha$. In particular,

$$\lim_{n \to +\infty} D_t u_n(t, x) = D_t u(t, x), \tag{5.1.17}$$

for any t > 0 and any $x \in \mathbb{R}^N$. From (5.1.15) it follows that

$$D_{t}u_{n}(t,x) \leq \int_{\{\mathcal{A}\varphi \geq 0\}} \psi'_{n}(\varphi(y))\mathcal{A}\varphi(y)p(t,x;dy) + \int_{\{\mathcal{A}\varphi < 0\}} \psi'_{n}(\varphi(y))\mathcal{A}\varphi(y)p(t,x;dy).$$
 (5.1.18)

As n tends to $+\infty$, by (5.1.17) and the monotone convergence theorem, we get (5.1.11) and (5.1.12). Note that the first integral in the right-hand side of (5.1.18) converges for any $x \in \mathbb{R}$, since $0 \le \psi_n' \le 1$ for any $n \in \mathbb{N}$, and $\mathcal{A}\varphi$ is bounded from above. Therefore, the second integral in the right-hand side of (5.1.18) converges as well for any $x \in \mathbb{R}^N$ and the second formula in (5.1.12) follows.

To show that u is continuous up to t=0 we argue as in the proof of Theorem 2.2.1. For this purpose, we fix $m\in\mathbb{N}$ and introduce a cut-off function $\vartheta=\vartheta_m\in C_c^\infty(\mathbb{R}^N)$ such that $\chi_{B(m-1)}\leq\vartheta\leq\chi_{B(m)}$. Then, we consider the function $v_n=u_n\vartheta_m$ and we take n sufficiently large such that

$$n \ge \sup_{x \in B(m)} |\varphi(x)|.$$

This choice of n implies that $\vartheta \psi_n(\varphi) = \vartheta \varphi$. As it is easily seen, v_n solves the Cauchy-Dirichlet problem

$$\begin{cases} D_t v_n(t,x) = \mathcal{A}v_n(t,x) + g_n(t,x), & t > 0, x \in B(m), \\ v_n(t,x) = 0, & t > 0, x \in \partial B(m), \\ v_n(0,x) = \vartheta(x)\varphi(x), & x \in \overline{B}(m), \end{cases}$$

for any $n \geq m$, where

$$g_n = -\sum_{i,j=1}^{N} q_{ij} (2D_i u_n D_j \vartheta + u_n D_{ij} \vartheta) - u_n \sum_{i=1}^{N} b_i D_i \vartheta$$

satisfies

$$|g_n(t,x)| \le C_1 \left(\|\varphi\|_{L^{\infty}(B(m))} + \|u_n\|_{L^{\infty}(B(m))} + \sum_{i=1}^N \|D_i u_n(t,\cdot)\|_{L^{\infty}(B(m))} \right),$$

for some positive constant C_1 , independent of n. The estimate (5.1.16) and the interior estimates in Theorem C.1.4 (see (C.1.16)) yield

$$\|\sqrt{t}D_{i}u_{n}(t,\cdot)\|_{L^{\infty}(B(m))} \leq C\|u_{n}\|_{L^{\infty}((0,T)\times B(m+1))}$$

$$\leq C_{2}(\|\varphi\|_{L^{\infty}(B(m+1))} + K),$$

for any $t \in (0, 1)$, any $x \in B(m)$, any i = 1, ..., N and some positive constant C_2 , independent of n. Now, repeating step by step the last part of the proof of Theorem 2.2.1, we deduce that $u(t, \cdot)$ tends to φ uniformly in B(m-1), as t tends to 0. Due to the arbitrariness of m, it follows that $u \in C([0, +\infty) \times \mathbb{R}^N)$.

Thanks to Lemma 5.1.4, we can now prove the following theorem which gives a sufficient condition ensuring the compactness of T(t) for any t > 0.

Theorem 5.1.5 Assume that the semigroup $\{T(t)\}$ is conservative and suppose that there exist a strictly positive function $\varphi \in C^2(\mathbb{R}^N)$ and a convex function $g \in C^1([0,+\infty))$ such that

$$\lim_{|x|\to+\infty}\varphi(x)=+\infty,\quad \frac{1}{g} \text{ is integrable at } +\infty,\quad \mathcal{A}\varphi(x)\leq -g(\varphi(x)),\quad x\in\mathbb{R}^N.$$

$$(5.1.19)$$

Then, T(t) is compact in $C_b(\mathbb{R}^N)$ for any t > 0.

Proof. Without loss of generality, we can suppose that g(0) < 0. Indeed, if this is not the case we replace g with g - C for some large enough positive constant C. Let us set

$$u(t,x) = (T(t)\varphi)(x) = \int_{\mathbb{R}^N} \varphi(y)p(t,x;dy), \quad t > 0, \ x \in \mathbb{R}^N.$$

Observe that, since g is convex and 1/g is integrable in a neighborhood of $+\infty$, then g(x) tends to $+\infty$ as x tends to $+\infty$. In particular, g is bounded from below in $(0, +\infty)$. Hence, by (5.1.19), $\mathcal{A}\varphi$ is bounded from above. Consequently, by Lemma 5.1.4, it follows that $u \in C^{1,2}((0, +\infty) \times \mathbb{R}^N) \cap C([0, +\infty) \times \mathbb{R}^N)$. Moreover.

$$\frac{\partial}{\partial t} u(t,x) \leq \int_{\mathbb{R}^N} \mathcal{A} \varphi(y) p(t,x;dy) \leq -\int_{\mathbb{R}^N} g(\varphi(y)) p(t,x;dy) \leq -g(u(t,x)),$$

for any t>0 and any $x\in\mathbb{R}^N,$ where the last inequality follows from Jensen's inequality.

For any fixed $x \in \mathbb{R}^N$ let $z(\cdot, x)$ be the solution of the Cauchy problem

$$\begin{cases} z'(t,x) = -g(z(t,x)), & t > 0, x \in \mathbb{R}^N, \\ z(0,x) = \varphi(x), & x \in \mathbb{R}^N. \end{cases}$$
 (5.1.20)

Note that $z(\cdot, x)$ is defined in $(0, +\infty)$. Indeed, if $\varphi(x) = x_g$, where x_g denotes the zero of g, then $z(\cdot, x) \equiv x_g$, whereas $z(\cdot, x)$ is increasing and greater than x_g if $\varphi(x) > x_g$. Similarly, $z(\cdot, x)$ is decreasing and less than x_g if $\varphi(x) < x_g$. In any case, $z(\cdot, x)$ exists in $[0, +\infty)$.

Let us now prove that $z(\cdot, x)$ is bounded from above in $[\tau, +\infty)$ for any $\tau > 0$, uniformly with respect to $x \in \mathbb{R}^N$. First, we assume that $\varphi(x) > x_g$. Integrating the differential equation (5.1.20) gives

$$-\int_{\varphi(x)}^{z(t,x)} \frac{dz}{g(z)} = t.$$

Therefore,

$$\int_{z(t,x)}^{+\infty} \frac{dz}{g(z)} = \int_{\varphi(x)}^{+\infty} \frac{dz}{g(z)} - \int_{\varphi(x)}^{z(t,x)} \frac{dz}{g(z)} \ge - \int_{\varphi(x)}^{z(t,x)} \frac{dz}{g(z)} = t > \tau. \quad (5.1.21)$$

Note that all the integral terms are finite since the function 1/g is integrable in a neighborhood of $+\infty$. Since 1/g is not integrable in a neighborhood of x_g , there exists a unique value $\bar{z} = \bar{z}(\tau) > x_g$ such that

$$\int_{\bar{z}}^{+\infty} \frac{dz}{g(z)} = \tau. \tag{5.1.22}$$

The positivity of g in $(x_g, +\infty)$, (5.1.21) and (5.1.22) imply that

$$z(t,x) < \bar{z}, \qquad t > \tau. \tag{5.1.23}$$

Now, suppose that $\varphi(x) \leq x_g$. In such a case, $z(t,x) \leq x_g$ for any $t \in (\tau, +\infty)$. Summing up, we have proved that $z(t,x) \leq \max(\bar{z}, x_g) = \bar{z}$ for any $t > \tau$ and any $x \in \mathbb{R}^N$.

Now, we fix $\varepsilon > 0$ and let $\rho > 0$ be such that $\varphi(x) \geq \bar{z}/\varepsilon$ for any $x \in \mathbb{R}^N \setminus B(\rho)$. Then, we have

$$p(t, x; \mathbb{R}^N \setminus B(\rho)) \le \varepsilon \bar{z}^{-1} \int_{\mathbb{R}^N \setminus B(\rho)} \varphi(y) p(t, x; dy) \le \varepsilon \bar{z}^{-1} u(t, x) \le \varepsilon,$$

for any $x \in \mathbb{R}^N$ and any $t > \tau$. The compactness of T(t), for any t > 0, now follows from Proposition 5.1.3 and the arbitrariness of $\tau > 0$. Note that the proof of the quoted proposition shows that $T(t_0)$ is compact whenever the family $\{p(t_0, x; dy) : x \in \mathbb{R}^N\}$ is tight.

Example 5.1.6 Consider the operator \mathcal{A} defined by

$$Af(x) = \Delta f(x) + \langle b(x), Df(x) \rangle, \quad x \in \mathbb{R},$$

on smooth functions, with the drift b satisfying

$$\langle b(x), x \rangle \le C - M|x|^{2+\varepsilon}, \qquad x \in \mathbb{R}^N,$$

for some $C \in \mathbb{R}$, $M, \varepsilon > 0$. Then, the associated semigroup $\{T(t)\}$ is compact. To see it, it suffices to apply Theorem 5.1.5 with

$$\varphi(x) = |x|^2, \quad x \in \mathbb{R}^N, \quad g(s) = -(2N+C) + Ms^{1+\frac{\varepsilon}{2}}, \quad s > 0.$$
 (5.1.24)

Finally, we give another condition implying that T(t) is a compact operator for any t > 0. For this purpose, we introduce the one-dimensional differential operator C, defined on smooth functions by

$$Cu(r) = q(r) [u''(r) + r^{-1}b(r)u'(r)], \qquad r > 0,$$

where

$$q(r) = \min_{x \in \partial B(r)} \left(\frac{1}{|x|^2} \sum_{i,j=1}^{N} q_{ij}(x) x_i x_j \right) := \min_{x \in \partial B(r)} \mathcal{Q}(x), \tag{5.1.25}$$

$$b(r) = \max_{x \in \partial B(r)} \left(\frac{1}{Q(x)} \sum_{i=1}^{N} (q_{ii}(x) + b_i(x)x_i) - 1 \right) := \max_{x \in \partial B(r)} \mathcal{B}(x). \quad (5.1.26)$$

Proposition 5.1.7 If $+\infty$ is an entrance point for the operator C (see Definition 3.1.6), then T(t) is a compact operator for any t > 0.

To prove the proposition we need two preliminary lemmata.

Lemma 5.1.8 Suppose that $+\infty$ is an entrance point for the operator C. Then, for any R > 0, there exists a positive function $\varphi \in C^2(\mathbb{R}^N \setminus B(R))$ diverging to $+\infty$ as |x| tends to $+\infty$ and such that $\varphi - \mathcal{A}\varphi \geq 0$ in $\mathbb{R}^N \setminus B(R)$. As a consequence, if $z \in C_b([0,T] \times \mathbb{R}^N \setminus B(R)) \cap C^{1,2}((0,T) \times \mathbb{R}^N \setminus \overline{B}(R))$ satisfies

$$\begin{cases} D_t z(t, x) - \mathcal{A} z(t, x) \ge 0, & t \in (0, T), x \in \mathbb{R}^N \setminus \overline{B}(R), \\ z(t, x) \ge 0, & t \in (0, T), x \in \partial B(R), \\ z(0, x) \ge 0, & x \in \mathbb{R}^N \setminus B(R), \end{cases}$$

for some R > 0, then $z \ge 0$.

Proof. We limit ourselves to considering the case when R=1, the general case being completely similar.

Let us determine a function φ satisfying the assertion. For this purpose, we observe that since $+\infty$ is an entrance, according to Proposition 3.1.7(iii), we can determine an increasing positive solution $\psi:[1,+\infty)\to\mathbb{R}$ of the equation $\psi-\mathcal{C}\psi=0$ such that $\lim_{r\to+\infty}\psi(r)=+\infty$. This can be done adapting to this situation the proof of Proposition 3.1.4. The function φ can now be defined by setting $\varphi(x)=\psi(|x|)$ for any $x\in\mathbb{R}^N\setminus B(1)$. Indeed, a straightforward computation shows that

$$\varphi(x) - \mathcal{A}\varphi(x) = \psi(|x|) - \mathcal{Q}(x) \left(\psi''(|x|) + \frac{\mathcal{B}(x)}{|x|} \psi'(|x|) \right), \qquad x \in \mathbb{R}^N \setminus B(1),$$

where Q and B are given by (5.1.25) and (5.1.26). Since ψ and ψ' are both positive, it is now easy to check that

$$\varphi(x) - \mathcal{A}\varphi(x) \ge \mathcal{Q}(x) \left(\frac{1}{q(|x|)} \psi(|x|) - \psi''(|x|) - \frac{b(|x|)}{|x|} \psi'(|x|) \right) = 0,$$

for any $x \in \mathbb{R}^N \setminus B(1)$.

To prove the last part of the lemma, for any $\varepsilon > 0$, we introduce the function z_{ε} defined by $z_{\varepsilon}(t,x) = z(t,x) + \varepsilon e^t \varphi(x) + \varepsilon t$ for any $t \in [0,T]$ and any $x \in \mathbb{R}^N \setminus B(1)$. As it is immediately seen, $D_t z_{\varepsilon} - \mathcal{A} z_{\varepsilon} \geq \varepsilon$ in $(0,T) \times (\mathbb{R}^N \setminus \overline{B}(1))$ and z_{ε} tends to $+\infty$ as |x| tends to $+\infty$, uniformly with respect to $t \in [0,T]$. This implies that z_{ε} has an absolute minimum at some point $(t_0,x_0) \in [0,T] \times \mathbb{R}^N \setminus B(1)$. We cannot have $t_0 > 0$ and $|x_0| > 1$, otherwise we should have $D_t z_{\varepsilon}(t_0,x_0) - \mathcal{A} z(t_0,x_0) \leq 0$ which is a contradiction. Therefore, $t_0 = 0$ or $|x_0| = 1$. In any case we have $z_{\varepsilon}(t_0,x_0) \geq 0$. Letting ε go to 0 the assertion follows.

Lemma 5.1.9 Fix $\delta > 0$. Then, for any $f \in C_b([\delta, +\infty))$ the Cauchy-Dirichlet problem

$$\begin{cases} D_t u(t, x) = \mathcal{C}u(t, x), & t > 0, x > \delta, \\ u(t, \delta) = 0, & t > 0, \\ u(0, x) = f(x), & x > \delta, \end{cases}$$
 (5.1.27)

admits a solution $u \in C_b(([0,+\infty) \times [\delta,+\infty)) \setminus \{(0,\delta)\}) \cap C^{1,2}((0,+\infty) \times (\delta,+\infty))$. Moreover,

$$||u||_{\infty} \le ||f||_{\infty}. \tag{5.1.28}$$

Further, the family of linear operators $\{S(t)\}$ defined by $S(t)f = u(t, \cdot)$ for any t > 0, where u is as above, is a positive semigroup of contractions. Finally, the function $v = S(\cdot)1$ satisfies $D_t v \leq 0$ and $D_r v \geq 0$ in $(0, +\infty) \times (\delta, +\infty)$.

Proof. Let us observe that a straightforward computation shows that the coefficients of the operator \mathcal{C} belong to $C_{\text{loc}}^{\alpha}([\delta, +\infty))$. Moreover, the proof of Lemma 5.1.8 shows that there exists a positive function $\psi \in C^2([\delta, +\infty))$, blowing up as x tends to $+\infty$, such that $\psi - \mathcal{C}\psi = 0$ in $(\delta, +\infty)$. Therefore, to prove the assertion, it suffices to apply the same arguments as in the proof of the forthcoming Theorem 11.2.1.

Let us give just the proof of the last part of the lemma. We set $u = S(\cdot)1$ and we prove that $D_t u$ and $-D_r u$ are nonpositive in $(0, +\infty) \times (\delta, +\infty)$. As a first step we observe that, from the estimate (5.1.28), it follows that $u \leq 1$. Now, using the positivity of the semigroup, we get

$$u(t+s,\cdot) = S(t+s)\mathbf{1} = S(s)S(t)\mathbf{1} \le S(t)\mathbf{1} = u(t,\cdot), \qquad s,t>0,$$

so that the function $u(\cdot, r)$ is nonincreasing in $(0, +\infty)$ for any $r > \delta$ and, consequently, $D_t u \leq 0$ in $(0, +\infty) \times (\delta, +\infty)$.

Now we show that, for any t > 0, the function $u(t, \cdot)$ is nondecreasing in $(0, +\infty)$. Since

$$Cu(t,r) = q(r)W(r)D_r\left(\frac{D_ru(t,\cdot)}{W}\right)(r), \qquad t > 0, \ r > \delta,$$

where W is given by (3.1.3), we deduce that, for any 0 < r < s and any t > 0,

$$\frac{D_r u(t,s)}{W(s)} - \frac{D_r u(t,r)}{W(r)} = \int_r^s \frac{D_t u(t,\xi)}{q(\xi)W(\xi)} d\xi.$$

Since $D_t u$ is nonpositive in $(0, +\infty) \times (\delta, +\infty)$, then the function $w(t, \cdot) := D_r u(t, \cdot)/W$ is nonincreasing in $(\delta, +\infty)$, for any t > 0. Set $l_t = \lim_{r \to +\infty} w(t, r)$. We claim that $l_t \geq 0$ for any t > 0. Of course, once the claim is proved, we will immediately obtain that $w \geq 0$ and, consequently, $D_r u \geq 0$ in $(0, +\infty) \times (\delta, +\infty)$ since W is a strictly positive function. Suppose by contradiction that $l_t < 0$ for some t > 0. Then, there exist t > 0 and a negative constant t > 0 such that

$$D_t u(t,r) \le cW(r), \qquad r \ge r^*.$$

Integrating in $[r^*, r]$ the previous inequality, we get

$$v(t,r) = v(t,r^*) + \int_{r^*}^r D_t u(t,s) ds \le u(t,\delta) + c \int_{r^*}^r W(s) ds.$$

The boundedness of v in $(0, +\infty) \times (\delta, +\infty)$ implies that W is integrable in $(r^*, +\infty)$ and, consequently, in $(\delta, +\infty)$. But this is a contradiction. Indeed, since Q is integrable in $(\delta, +\infty)$, then the function 1/(qW) is integrable in $(\delta, +\infty)$ as well. The integrability of W in $(\delta, +\infty)$ should imply the integrability of R in $(\delta, +\infty)$, which contradicts our assumptions.

Now, we can prove Proposition 5.1.7.

Proof of Proposition 5.1.7. According to Proposition 5.1.3, to prove that T(t) is compact for any t>0, we show that the family of measures $\{p(t,x;dy): x\in\mathbb{R}^N\}$ is tight for any t>0, i.e. we fix $t_0>0$ and we prove that for any $\varepsilon>0$ there exists $\rho>0$ such that

$$p(t_0, x; B(\rho)) \ge 1 - \varepsilon, \qquad x \in \mathbb{R}^N.$$
 (5.1.29)

Let $\{f_n\} \subset C_b(\mathbb{R}^N)$ be a sequence of continuous functions in \mathbb{R}^N such that $\chi_{B(n-1)} \leq f_n \leq \chi_{B(n)}$ for any $n \in \mathbb{N}$. Observe that

$$p(t_0, x; B(n)) = \int_{B(n)} p(t_0, x; dy) \ge \int_{\mathbb{R}^N} f_n(y) p(t_0, x; dy) = (T(t_0) f_n)(x),$$

for any $x \in \mathbb{R}^N$ (see Theorem 2.2.5). Since f_n converges to the function f = 1 as n tends to $+\infty$, locally uniformly in \mathbb{R}^N , by Proposition 2.2.9 it follows that, the function $T(t_0)f_n$ converges locally uniformly in \mathbb{R}^N to the function 1. Hence, it suffices to show that there exist $\rho > 0$ and $n_1 \in \mathbb{N}$ such that

$$(T(t_0)f_n)(x) \ge 1 - \varepsilon, \qquad x \in \mathbb{R}^N \setminus \overline{B}(\rho),$$
 (5.1.30)

for any $n \geq n_1$. To prove (5.1.30) we use a comparison argument with radial functions. For this purpose we split the proof into two steps. First, in Step 1, we study the main properties of the function $v_{\delta} = S(\cdot)1$ that we need in the sequel. Here $\{S(t)\}$ is the semigroup in $C_b([\delta, +\infty))$ defined in Lemma 5.1.9. Then, in Step 2, by means of the function v_{δ} , we construct a radial function w such that $w(t_0, \cdot) \geq 1 - \varepsilon$ and such that, for |x| and n large enough, it satisfies $(T(t_0)f_n)(x) \geq w(t_0, x)$.

Step 1. We claim that, for any δ sufficiently large,

$$0 \le v_{\delta}(t_0, r) \le \eta := 1 - (1 - \varepsilon)^{1/2}, \qquad r \ge \delta.$$
 (5.1.31)

For this purpose, for any $\delta > 0$, we set

$$\psi_{\delta}(r) = 1 - \int_0^{+\infty} e^{-t} v(t, r) dt, \qquad r \ge \delta.$$

The function ψ_{δ} is well defined since v is bounded in $(0, +\infty) \times (\delta, +\infty)$. Moreover, ψ_{δ} satisfies

$$\begin{cases} \psi_{\delta}(r) - \mathcal{C}\psi_{\delta}(r) = 0, & r > \delta, \\ \psi_{\delta}(\delta) = 1. \end{cases}$$
 (5.1.32)

Of course, $\psi_{\delta}(\delta) = 1$ since v(t,0) = 0 for any t > 0. To show that ψ_{δ} satisfies the differential equation in (5.1.32), we introduce, for any $n \in \mathbb{N}$, the function $\psi_{\delta,n} : [\delta, +\infty) \to \mathbb{R}$ defined by

$$\psi_{\delta,n}(r) = \int_0^n e^{-t} v(t,r) dt, \qquad r \ge \delta.$$

It is immediate to check that $\psi_{\delta,n} \in C^2((\delta,+\infty))$ and it solves the differential equation

$$\psi_{\delta,n}(r) - \mathcal{C}\psi_{\delta,n}(r) = 0, \qquad r > \delta. \tag{5.1.33}$$

Moreover, $\psi_{\delta,n}$ converges uniformly to ψ_{δ} in $[\delta, +\infty)$. Let us now show that $\psi_{\delta} \in C^2((\delta, +\infty))$ and it solves the differential equation in (5.1.32) pointwise. For this purpose, we observe that, since the sequence $\{\psi_{\delta,n}\}$ is bounded in $C_b([\delta, +\infty))$, using the same compactness arguments as in the proof of Theorem 2.1.1, one can easily show that, up to a subsequence, $\psi_{\delta,n}$ converges to ψ_{δ} in $W^{2,p}(r_1,r_2)$ for any $\delta < r_1 < r_2$ and any $p \in (1,+\infty)$. According to the Sobolev embedding theorems (see [4, Theorem 5.4]), we deduce that $\psi_{\delta} \in C^1((\delta,+\infty))$ and $\psi'_{\delta,n}$ converges to ψ'_{δ} locally uniformly in $(\delta,+\infty)$. Now, since $q(r)\psi''_{\delta,n}(r) = \psi_{\delta,n}(r) - r^{-1}q(r)b(r)\psi'_{\delta,n}(r)$ for any $r \in (\delta,+\infty)$ and any $n \in \mathbb{N}$, it follows that $q\psi''_{\delta,n}$ (and consequently $\psi''_{\delta,n}$ since q does not vanish in $[\delta,+\infty)$) converges locally uniformly in $(\delta,+\infty)$. Therefore, $\psi_{\delta} \in C^2((\delta,+\infty))$. Now, letting q go to q in q in q in q in q solves the differential equation in q in q

Let us now prove that δ can be chosen sufficiently large so that

$$\psi_{\delta}(r) \ge 1 - \eta t_0 e^{-t_0}, \qquad r \ge \delta. \tag{5.1.34}$$

For this purpose, denote by φ the unique positive decreasing solution of the equation $\varphi - \mathcal{C}\varphi = 0$ in $(1, +\infty)$, such that $\lim_{r \to +\infty} \varphi(r) = 1$ (the existence of such a solution can be established arguing as in the proof of the properties (ii) and (iv) in Proposition 3.1.4). Then, $\psi_{\delta} = \varphi/\varphi(\delta)$. Indeed, the same arguments as in the proof of Proposition 3.1.4 show that, up to a multiplicative constant, ψ_{δ} is the unique bounded solution to the problem (5.1.32). It is now clear that we can fix $\hat{\delta}$ such that $\psi_{\hat{\delta}}$ satisfies (5.1.34).

Now, let $\lambda_m = e^{-t_0} \eta + m^{-1} \ (m \in \mathbb{N})$ and introduce the set

$$E_m = \{ t \ge 0 : e^{-t} \lim_{r \to +\infty} v_{\hat{\delta}}(t, r) \ge \lambda_m \}.$$

Since v is bounded, the set E_m is bounded for any $m \in \mathbb{N}$ and it is not empty since it contains t = 0. Moreover, using the Chebyshev inequality, we get

$$\begin{split} \int_{E_m} dr &= \frac{1}{\lambda_m} \int_{E_m} \lambda_m dr \leq \frac{1}{\lambda_m} \int_{E_m} e^{-t} \lim_{r \to +\infty} v_{\hat{\delta}}(t,r) dt \\ &= \frac{1}{\lambda_m} \lim_{r \to +\infty} \int_{E_m} e^{-t} v_{\hat{\delta}}(t,r) dt \\ &= \frac{1}{\lambda_m} \lim_{r \to +\infty} \int_{E_m} e^{-t} v_{\hat{\delta}}(t,r) dt \\ &\leq \frac{1}{\lambda_m} \lim_{r \to +\infty} \sup(1 - \psi_{\hat{\delta}}(r)) \\ &\leq \frac{\eta}{\lambda_m} t_0 e^{-t_0}. \end{split}$$

82

Therefore, due to the choice of λ_m , the Lebesgue measure of E_m is strictly less than t_0 . From this fact, and recalling that $v_{\hat{\delta}}(t,\cdot)$ is increasing in $[\hat{\delta}, +\infty)$ for any $t > \hat{\delta}$, it follows that we can determine $t_{1,m} < t_0$ such that

$$e^{-t_{1,m}}v_{\hat{\delta}}(t_{1,m},r) < \lambda_m, \qquad r > \hat{\delta}.$$

Since the sequence $\{t_{1,m}\}$ is bounded, up to a subsequence, we can assume that $t_{1,m}$ converges to a point $t_1 \in (0,t_0]$ as m tends to $+\infty$. Moreover, by continuity

$$e^{-t_1}v_{\hat{\delta}}(t_1,r) \le e^{-t_0}\eta, \qquad r \ge \hat{\delta}$$

and, consequently, $v_{\hat{\delta}}(t_1, \cdot) \leq \eta$ in $[\hat{\delta}, +\infty)$. Since $v_{\hat{\delta}}(\cdot, r)$ is nonincreasing in $(0, +\infty)$ for any $r > \hat{\delta}$, the estimate (5.1.31) follows.

Step 2. Now, we prove (5.1.30) with $\rho = \hat{\delta} + 1$. Since, as it has already been pointed out, $T(t)f_n$ converges locally uniformly in $[0, +\infty) \times \mathbb{R}^N$ to 1, we can fix $n_1 \in \mathbb{N}$ such that

$$(T(t)f_n)(x) \ge 1 - \eta,$$
 $t \in (0, t_0), x \in \overline{B}(\hat{\delta} + 1), n \ge n_1.$

Let us now introduce the function $w:[0,+\infty)\times(\mathbb{R}^N\setminus B(\hat{\delta}+1))$ defined by

$$w(t,x) = (1-\eta)(1-v_{\hat{\delta}}(t,|x|)), \qquad t \ge 0, \quad x \in \mathbb{R}^N \setminus B(\hat{\delta}+1).$$

Of course, $w \in C_b([0, +\infty) \times (\mathbb{R}^N \setminus B(\hat{\delta}+1))) \cap C^{1,2}((0, +\infty) \times (\mathbb{R}^N \setminus \overline{B}(\hat{\delta}+1)))$, it satisfies $w \leq 1 - \eta$ in $[0, +\infty) \times (\mathbb{R}^N \setminus B(\hat{\delta}+1))$, $w(0, \cdot) = 0$ and, by (5.1.31), $w(t_0, \cdot) \geq (1 - \eta)^2 = 1 - \varepsilon$ in $\mathbb{R}^N \setminus B(\hat{\delta}+1)$. Moreover, it satisfies the differential equation $D_t w - \mathcal{A}w = g$, where

$$g(t,x) = (\eta - 1)\mathcal{Q}(x) \left[\frac{D_t v(t,|x|)}{\mathcal{Q}(x)} - D_{rr} v(t,|x|) - \frac{\mathcal{B}(x)}{|x|} D_r v(t,|x|) \right],$$

for any t > 0 and any $x \in \mathbb{R}^N \setminus B(\hat{\delta}+1)$. Here, \mathcal{Q} and \mathcal{B} are defined by (5.1.25) and (5.1.26). Since $D_t v$ and $-D_r v$ are nonpositive in $(0, +\infty) \times (\hat{\delta}, +\infty)$, we deduce that

$$g(t,x) \le (\eta - 1) \frac{\mathcal{Q}(x)}{g(|x|)} \left(D_t v(t,|x|) - \mathcal{C}v(t,|x|) \right) = 0.$$

From all the results above, it follows that the function $z = T(\cdot)f_n - w$ satisfies

$$\begin{cases} D_t z(t,x) - \mathcal{A} z(t,x) \geq 0, & t \in (0,t_0), x \in \mathbb{R}^N \setminus \overline{B}(\hat{\delta}+1), \\ z(t,x) \geq 0, & t \in (0,t_0), x \in \mathbb{R}^N \setminus \partial B(\hat{\delta}+1), \\ z(0,x) = f_n(x), & x \in \mathbb{R}^N \setminus B(\hat{\delta}+1), \end{cases}$$

for any $n \ge n_1$. The maximum principle in Lemma 5.1.8 implies that $z \ge 0$ in $(0, t_0) \times (\mathbb{R}^N \setminus B(\hat{\delta} + 1))$ and (5.1.30) follows.

5.1.2 The nonconservative case

We now consider the nonconservative case. In particular, here we do not assume that the coefficient c identically vanishes in \mathbb{R}^N .

The following theorem shows that T(t) is compact and maps $C_0(\mathbb{R}^N)$ into itself, for some t > 0, if and only if it maps the function 1 into a function vanishing at infinity. To prove such a result, we need the following lemma.

Lemma 5.1.10 Suppose that \mathcal{F} is a bounded subset of $C_0(\mathbb{R}^N)$ such that for any $\varepsilon > 0$ there exist $\delta > 0$ with the property

$$\sup_{f \in \mathcal{F}} \sup_{x \in \mathbb{R}^N \setminus B(\delta)} |f(x)| < \varepsilon.$$

Further, suppose that, for any compact set $H \subset \mathbb{R}^N$ and any $\varepsilon > 0$, there exists $\sigma > 0$ such that

$$\sup_{f \in \mathcal{F}} \sup_{\substack{|x-y| \le \sigma \\ x, y \in H}} |f(x) - f(y)| \le \varepsilon.$$

Then, \mathcal{F} is totally bounded in $C_0(\mathbb{R}^N)$.

Proof. Fix $\varepsilon > 0$ and let $\delta > 0$ be such that $|f(x)| \le \varepsilon$ for any $|x| \ge \delta$ and any $f \in \mathcal{F}$. According to our assumptions and the Ascoli-Arzelà Theorem, the set $\mathcal{F}_{\mid B(2\delta)}$, of all the restrictions to $B(2\delta)$ of the functions in \mathcal{F} , is totally bounded. Hence, we can find out $k \in \mathbb{N}$ and $g_1, \ldots, g_k \in C(\overline{B}(2\delta))$ such that

$$\mathcal{F}_{\mid \overline{B}(2\delta)} \subset \bigcup_{j=1}^{k} (g_j + B(\varepsilon)),$$

where $g_j + B(\varepsilon)$ (j = 1, ..., k) denotes the closed ball in $C(\overline{B}(2\delta))$ with centre at g_j and radius ε . Now let φ be any smooth function such that $\chi_{B(\delta)} \leq \varphi \leq \chi_{B(2\delta)}$. Then,

$$\mathcal{F} \subset \bigcup_{j=1}^{k} (\varphi g_j + B(3\varepsilon)), \tag{5.1.35}$$

where the closed balls $\varphi g_j + B(3\varepsilon)$ are now meant in $C_b(\mathbb{R}^N)$. To check (5.1.35), fix $f \in \mathcal{F}$ and let $j \in \{1, \ldots, k\}$ be such that $||f - g_j||_{C(\overline{B}(2\delta))} \leq \varepsilon$. Then, in particular,

$$||f - \varphi g_j||_{C(\overline{B}(\delta))} \le \varepsilon.$$

Moreover,

$$\begin{split} \|f - \varphi g_j\|_{C_b(\mathbb{R}^N \setminus B(\delta))} &\leq \|f\|_{C_b(\mathbb{R}^N \setminus B(\delta))} + \|g_j\|_{C(\overline{B}(2\delta) \setminus B(\delta))} \\ &\leq 2\|f\|_{C_b(\mathbb{R}^N \setminus B(\delta))} + \|f - g_j\|_{C(\overline{B}(2\delta))} \\ &\leq 3\varepsilon. \end{split}$$

84

Summing up, we have

$$||f - \varphi g_i||_{C_k(\mathbb{R}^N)} \le 3\varepsilon,$$

and (5.1.35) follows.

We can now prove the following result.

Theorem 5.1.11 Fix t > 0. Then $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$ if and only if T(t) is compact and $C_0(\mathbb{R}^N)$ is invariant for T(t).

Proof. Suppose that $T(t) \mathbb{1} \in C_0(\mathbb{R}^N)$. Then, using (2.2.8), we get

$$|(T(t)f)(x)| = \left| \int_{\mathbb{R}^N} f(y)p(t,x;dy) \right| \le ||f||_{\infty} \int_{\mathbb{R}^N} p(t,x;dy) = ||f||_{\infty} (T(t)\mathbb{1})(x),$$
(5.1.36)

for any $f \in C_b(\mathbb{R}^N)$ and any $x \in \mathbb{R}^N$. Therefore, T(t) maps $C_b(\mathbb{R}^N)$ in $C_0(\mathbb{R}^N)$. In particular, $C_0(\mathbb{R}^N)$ is invariant.

Let us now prove that T(t) is a compact operator. For this purpose, let

$$B_{\infty}(1) = \{ f \in C_b(\mathbb{R}^N) : ||f||_{\infty} \le 1 \}$$

be the unit ball in $C_b(\mathbb{R}^N)$. According to (5.1.36), $T(t)(B_{\infty}(1))$ is a set of functions vanishing uniformly at infinity. This implies that, for any $\varepsilon > 0$, there exists $\delta > 0$ such that

$$|(T(t)f)(x)| \le \varepsilon, \qquad |x| \ge \delta,$$
 (5.1.37)

for any $f \in B_{\infty}(1)$. Moreover, from the interior Schauder estimates in Theorem C.1.4, it follows that, for any $\rho > 0$, there exists a positive constant $C = C(\rho, t)$ such that

$$||T(t)f||_{C^{2+\alpha}(\overline{B}(\rho))} \le C||f||_{C(\overline{B}(2\rho))} \le C,$$
 (5.1.38)

for any $f \in B_{\infty}(1)$, where α is as in Hypothesis 5.1.1(ii). Therefore, the set of all the restrictions of the functions in $T(t)(B_{\infty}(1))$ to $\overline{B}(\rho)$ is a bounded set of equicontinuous functions. From (5.1.37), (5.1.38) and Lemma 5.1.10 it follows that $T(t)(B_{\infty}(1))$ is compact in $C_b(\mathbb{R}^N)$.

Conversely, fix t > 0 and suppose that T(t) is compact and $C_0(\mathbb{R}^N)$ is invariant for T(t). Let $\{f_n\}$ be a sequence of continuous functions such that $\chi_{B(n-1)} \leq f_n \leq \chi_{B(n)}$ for any $n \in \mathbb{N}$. By Proposition 2.2.9, $(T(t)f_n)(x)$ tends to $(T(t)\mathbf{1})(x)$ for any $x \in \mathbb{R}^N$. Since T(t) is compact, then $T(t)f_n$ tends to $T(t)\mathbf{1}$ uniformly in \mathbb{R}^N . Moreover, since $C_0(\mathbb{R}^N)$ is invariant, then $T(t)f_n \in C_0(\mathbb{R}^N)$ for any $n \in \mathbb{N}$, and, therefore, $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$ as well.

In Section 5.2 we will determine some sufficient conditions implying that $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$ for any t > 0. By virtue of Theorem 5.1.11, such conditions turn out to be sufficient conditions for the compactness of T(t) for any t > 0.

We conclude this section with the following proposition.

Proposition 5.1.12 Suppose that, for any t > 0, T(t) is a compact operator and $C_0(\mathbb{R}^N)$ is invariant under T(t). Then, the semigroup is norm-continuous in $(0, +\infty)$ and the resolvent $R(\lambda, \widehat{A})$ is compact for any $\lambda > c_0$.

Proof. We limit ourselves to proving that the map $t\mapsto T(t)f$ is continuous from the right in $(0,+\infty)$ for any $f\in B_\infty(1)=\{f\in C_b(\mathbb{R}^N):\|f\|_\infty\leq 1\}$. Indeed, once this property is proved, repeating the proof of Proposition 5.1.3, we can easily show that the map $t\mapsto T(t)$ is norm-continuous in $(0,+\infty)$ and $R(\lambda,\widehat{A})$ is compact for any $\lambda>c_0$. Observe that, by Theorem 5.1.11, $T(t)1\in C_0(\mathbb{R}^N)$ for any t>0. Hence, applying (5.1.36), written for $f\in C_b(\mathbb{R}^N)$, we easily see that T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ for any t>0. By Proposition 2.2.7, T(t)g tends to g uniformly in \mathbb{R}^N , as t tends to 0, for any $g\in C_0(\mathbb{R}^N)$. Hence,

$$\lim_{h \to 0^+} \| (T(h) - I)T(t)f \|_{\infty} = 0,$$

for any t>0 and any $f\in C_b(\mathbb{R}^N)$, and this implies that $t\mapsto T(t)f$ is right-continuous in $(0,+\infty)$.

5.2 On the inclusion $T(t)(C_b(\mathbb{R}^N)) \subset C_0(\mathbb{R}^N)$

In this section we determine sufficient conditions implying that T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$. If not explicitly specified, we do not assume that $c \equiv 0$. Let us observe that, without loss of generality, we can limit ourselves to determining sufficient conditions which ensure that $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$. Indeed, as it has been already pointed out in the proof of Proposition 5.1.12, applying (5.1.36) written for $f \in C_b(\mathbb{R}^N)$, we easily deduce that

$$|(T(t)f)(x)| \le ||f||_{\infty}(T(t)\mathbf{1})(x), \qquad t > 0, \quad x \in \mathbb{R}^{N}.$$
 (5.2.1)

Hence, if $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$, then $T(t)f \in C_0(\mathbb{R}^N)$ as well, for any $f \in C_b(\mathbb{R}^N)$. Using (5.2.1), we can easily show that

$$|(R(\lambda, \widehat{A})f)(x)| \le ||f||_{\infty} |(R(\lambda, \widehat{A})\mathbf{1})(x)|, \qquad x \in \mathbb{R}^N, \quad \lambda > c_0, \qquad (5.2.2)$$

which implies that $R(\lambda, \widehat{A})$ maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ if and only if $R(\lambda, \widehat{A})\mathbb{1} \in C_0(\mathbb{R}^N)$. We recall that, for any $f \in C_b(\mathbb{R}^N)$,

$$(R(\lambda, \widehat{A})f)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x)dt, \quad x \in \mathbb{R}^N,$$
 (5.2.3)

where \widehat{A} is the weak generator of the semigroup. See Sections 2.1 and 2.3.

The following proposition shows that T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ if and only if $R(\lambda, \widehat{A})$ does for some (and hence all) $\lambda > 0$. Before stating and proving it, we consider the following lemma.

Lemma 5.2.1 Let, as usual, $c_0 = \sup_{x \in \mathbb{R}^N} c(x)$. Then, for any $x \in \mathbb{R}^N$, the function $t \mapsto (S(t)\mathbf{1})(x) := e^{-c_0t}(T(t)\mathbf{1})(x)$ is not increasing in $[0, +\infty)$.

Proof. It can be obtained applying the proof of Lemma 4.1.8. See Remark 4.1.9.

We can now prove the following proposition.

Proposition 5.2.2 The following properties are equivalent:

- (i) $T(t) \mathbf{1} \in C_0(\mathbb{R}^N)$ for any t > 0;
- (ii) $R(\lambda, \widehat{A}) \mathbb{1} \in C_0(\mathbb{R}^N)$ for any $\lambda > c_0$;
- (iii) $R(\lambda, \widehat{A}) \mathbb{1} \in C_0(\mathbb{R}^N)$ for some $\lambda > c_0$;
- (iv) $D(\widehat{A}) = D_{\max}(A) \cap C_0(\mathbb{R}^N)$, where $D(\widehat{A})$ is defined in Section 2.3.

Proof. We can limit ourselves to proving that "(i) \Rightarrow (ii)" and "(iii) \Rightarrow (i)", since the implication "(ii) \Rightarrow (iii)" is trivial and "(ii) \Leftrightarrow (iv)" follows immediately, from Proposition 2.3.6, observing that (ii) and (5.2.2) imply that $R(\lambda, \widehat{A})f \in C_0(\mathbb{R}^N)$ for any $f \in C_b(\mathbb{R}^N)$.

"(i) \Rightarrow (ii)". Fix $\lambda>0.$ Taking (5.2.3) and Lemma 5.2.1 into account, we deduce that

$$\begin{split} (R(\lambda,\widehat{A})\mathbf{1})(x) &= \int_0^{+\infty} e^{-\lambda t} (T(t)\mathbf{1})(x) dt \\ &= \int_0^{+\infty} e^{-(\lambda-c_0)t} (S(t)\mathbf{1})(x) dt \\ &= \int_0^{\varepsilon} e^{-(\lambda-c_0)t} (S(t)\mathbf{1})(x) dt + \int_{\varepsilon}^{+\infty} e^{-(\lambda-c_0)t} (S(t)\mathbf{1})(x) dt \\ &\leq \varepsilon + (S(\varepsilon)\mathbf{1})(x) \int_{\varepsilon}^{+\infty} e^{-(\lambda-c_0)t} dt \\ &\leq \varepsilon + (S(\varepsilon)\mathbf{1})(x) \int_0^{+\infty} e^{-(\lambda-c_0)t} dt, \end{split}$$

for any $\varepsilon > 0$ and any $x \in \mathbb{R}^N$, where we use the fact that $\{S(t)\}$ is a semigroup of contractions. Since $T(\varepsilon) \mathbb{1} \in C_0(\mathbb{R}^N)$, then $S(\varepsilon) \mathbb{1} \in C_0(\mathbb{R}^N)$ as well. This implies that

$$\limsup_{|x|\to +\infty} (R(\lambda,\widehat{A})1)(x) \leq \varepsilon.$$

From the arbitrariness of $\varepsilon > 0$ and the nonnegativity of the function $R(\lambda, \widehat{A})\mathbf{1}$, we deduce that $R(\lambda, \widehat{A})\mathbf{1} \in C_0(\mathbb{R}^N)$.

"(iii) \Rightarrow (i)". Fix t > 0. Again by the formula (5.2.3) and Lemma 5.2.1 we get

$$(R(\lambda, \widehat{A})\mathbf{1})(x) = \int_0^{+\infty} e^{-(\lambda - c_0)s} (S(s)\mathbf{1})(x) ds$$

$$\geq \int_0^t e^{-(\lambda - c_0)s} (S(s)\mathbf{1})(x) ds$$

$$\geq (S(t)\mathbf{1})(x) \int_0^t e^{-(\lambda - c_0)s} ds, \qquad x \in \mathbb{R}^N,$$

that is

$$0 \leq S(t) 1\!\!1 \leq \left(\frac{\lambda - c_0}{1 - e^{-(\lambda - c_0)t}}\right) R(\lambda, \widehat{A}) 1\!\!1.$$

Thus, we conclude that S(t)1 (and hence T(t)1) belongs to $C_0(\mathbb{R}^N)$.

An interesting consequence of the fact that the semigroup maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ is given by the next proposition.

Proposition 5.2.3 Suppose that $c \equiv 0$ and $T(t_0)$ maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ for some $t_0 > 0$. Then, $||T(t)||_{L(C_b(\mathbb{R}^N))}$ decreases exponentially to 0 as t tends $to +\infty$.

Proof. We begin the proof observing that, since $T(t)\mathbf{1} = T(t_0)T(t-t_0)\mathbf{1}$ for any $t \geq t_0$, then T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ for any $t \geq t_0$. Now, we prove that there exists a positive constant $k_0 \in (0,1)$ such that $(T(t_0)\mathbf{1})(x) \leq k_0$ for any $x \in \mathbb{R}^N$. For this purpose, we notice that, since $T(t_0)\mathbf{1} \in C_0(\mathbb{R}^N)$, there exists $\delta > 0$ such that $(T(t_0)\mathbf{1})(x) \leq 1/2$ for any $|x| \geq \delta$. Moreover, since $T(t)\mathbf{1} \leq \mathbf{1}$ for any t > 0, $(T(t_0)\mathbf{1})(x) < 1$ for any $x \in B(\delta)$ otherwise by the classical maximum principle (see Proposition C.2.3(iii)), $T(t_0)\mathbf{1}$ should be constant in $B(\delta)$ and equal to $\mathbf{1}$, which, of course, cannot be the case.

Now, the semigroup property allows us to show that $||T(t)||_{L(C_b(\mathbb{R}^N))}$ decreases to 0 exponentially as t tends to $+\infty$. For this purpose, we split any $t > t_0$ as $t = nt_0 + r$ with $n \in \mathbb{N}$ and $r \in [0, t_0)$. Recalling that $||T(r)||_{L(C_b(\mathbb{R}^N))} = ||T(r)\mathbf{1}||_{\infty} \le 1$ for any r > 0, we get

$$||T(t)||_{L(C_b(\mathbb{R}^N))} = ||(T(t_0))^n T(r) \mathbf{1}||_{L(C_b(\mathbb{R}^N))}$$

$$\leq ||T(t_0)||_{L(C_b(\mathbb{R}^N))}^n ||T(r) \mathbf{1}||_{\infty}$$

$$\leq \exp(n \log(k_0)),$$

for any $t \ge t_0$. Now observing that $n \ge t/t_0 - 1$, we easily deduce that $\exp(n \log(k_0)) \le k_0^{-1} e^{\omega t}$, where $\omega = t_0^{-1} \log(k_0) < 0$. This finishes the proof.

The following result gives a sufficient condition which guarantees that T(t)1 belongs to $C_0(\mathbb{R}^N)$, in terms of a suitable Lyapunov function φ .

Proposition 5.2.4 Suppose that there exist $\lambda_0 > c_0$, a compact set K and a function $\varphi \in C^2(\mathbb{R}^N \setminus K) \cap C_0(\mathbb{R}^N \setminus K)$ such that

$$\varphi(x) > 0, \quad x \in \mathbb{R}^N \setminus K, \quad \inf_{x \in \mathbb{R}^N \setminus K} (\lambda_0 \varphi(x) - \mathcal{A}\varphi(x)) := a > 0, \quad (5.2.4)$$

Then, $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$ for any t > 0.

Proof. By virtue of Proposition 5.2.2, we can limit ourselves to proving that $R(\lambda_0, \widehat{A})$ belongs to $C_0(\mathbb{R}^N)$.

Let now n_0 be the smallest integer such that $K \subset B(n_0)$. Moreover, let $\gamma = \inf_{x \in \partial B(n_0)} \varphi(x) > 0$ and set $\delta = \min\{a, (\lambda_0 - c_0)\gamma\}$. We will prove that $R(\lambda_0, \widehat{A})\delta \in C_0(\mathbb{R}^N)$. Of course, this will imply that $R(\lambda_0, \widehat{A})\mathbb{1} = \delta^{-1}R(\lambda_0, \widehat{A})\delta$ belongs to $C_0(\mathbb{R}^N)$. For this purpose, we recall that $R(\lambda_0, \widehat{A})\delta$ is the pointwise limit of the sequence $\{u_n\}$, where, for any $n \in \mathbb{N}$, the function u_n is the unique solution in $\bigcap_{1 \le p < +\infty} W^{2,p}(B(n))$ to the problem

$$\begin{cases} \lambda_0 u_n(x) - \mathcal{A}u_n(x) = \delta, & x \in B(n), \\ u_n(x) = 0, & x \in \partial B(n). \end{cases}$$

Moreover, $0 \le u_n \le (\lambda_0 - c_0)^{-1} \delta$; see the proof of Theorem 2.1.1. Now, we observe that, for any $n > n_0$, the function φ satisfies

$$\begin{cases} \lambda_0 \varphi(x) - \mathcal{A}\varphi(x) \ge a, & x \in B(n) \setminus \overline{B}(n_0), \\ \varphi(x) \ge \gamma, & x \in \partial B(n_0), \\ \varphi(x) > 0, & x \in \partial B(n). \end{cases}$$

Therefore, the classical maximum principle (see Theorem C.2.2(i)) implies that

$$\varphi(x) \ge u_n(x), \qquad x \in B(n) \setminus \overline{B}(n_0).$$

Letting n go to $+\infty$, we get

$$0 \le (R(\lambda_0, \widehat{A})\delta)(x) \le \varphi(x), \qquad x \in \mathbb{R}^N \setminus \overline{B}(n_0).$$

Hence, $R(\lambda_0, \widehat{A})\delta \in C_0(\mathbb{R}^N)$.

Example 5.2.5 Consider the operator \mathcal{A} defined on smooth functions by

$$\mathcal{A}f(x) = \Delta f(x) + \langle b(x), Df(x) \rangle, \quad x \in \mathbb{R}^N,$$

where $b \in C^{\alpha}_{loc}(\mathbb{R}^N)$ ($\alpha \in (0,1)$). Moreover, let φ be the function defined by $\varphi(x) = (1+|x|^2)^{-1}$ for any $x \in \mathbb{R}^N$. We have $D\varphi(x) = -2(\varphi(x))^2 x$ and $D^2\varphi(x) = -2(\varphi(x))^2 I_{N\times N} + 8(\varphi(x))^3 x \otimes x$ for any $x \in \mathbb{R}^N$. Therefore,

$$\mathcal{A}\varphi(x) = 8|x|^2(\varphi(x))^3 - 2N(\varphi(x))^2 - 2(\varphi(x))^2\langle b(x), x\rangle, \qquad x \in \mathbb{R}^N. \ (5.2.5)$$

The condition (5.2.4) reads as

$$\langle b(x), x \rangle \ge \frac{a}{2} (1 + |x|^2)^2 - \frac{\lambda_0}{2} (1 + |x|^2) - N + 4 \frac{|x|^2}{1 + |x|^2}, \quad x \in \mathbb{R}^N \setminus K,$$

for some compact set $K \subset \mathbb{R}^N$, some $\lambda_0 > 0$ and some positive constant a. Such a condition is satisfied, for instance, if

$$\langle b(x), x \rangle \ge C + M|x|^4, \tag{5.2.6}$$

for some $C \in \mathbb{R}$ and M > a/2. In such a case, we can take any $\lambda_0 > 0$. According to Proposition 5.2.4, the semigroup $\{T(t)\}$, associated with the operator \mathcal{A} , is such that $T(t)\mathbf{1} \in C_0(\mathbb{R}^N)$ for any t > 0.

The condition (5.2.6) is satisfied, for instance, when the operator \mathcal{A} is given by

$$\mathcal{A}_+ f(x) = \Delta f(x) + |x|^2 \langle x, Df(x) \rangle, \qquad x \in \mathbb{R}^N.$$

This example should be compared with Example 4.1.13, where we considered the operator

$$\mathcal{A}_{-}f(x) = \Delta f(x) - |x|^2 \langle x, Df(x) \rangle, \qquad x \in \mathbb{R}^N,$$

and showed that the associated semigroup is conservative. The operators \mathcal{A}_{+} and \mathcal{A}_{-} differ only in the sign of the drift term. The reason for the difference in the behaviour of the two semigroups can be understood looking at the stochastic equations associated with these two operators. The operator \mathcal{A}_{-} satisfies the Hypotheses 2.5.3 and 2.5.4. Therefore, all the results in Section 2.5 hold. The stochastic equation associated with it is

$$d\xi_t^x = -|\xi_t^x|^2 \xi_t^x dt + dW_t, \qquad t > 0,$$

with the initial condition $\xi_0^x = x \in \mathbb{R}^N$. This equation has a solution defined for any t > 0. Heuristically, one can think of this equation as a perturbation of the ordinary differential equation

$$\frac{d}{dt}\xi_t^x = -|\xi_t^x|^2 \xi_t^x, \qquad t > 0,$$

whose solutions exist for any t > 0.

Repeating the same argument in the case of the operator A_+ leads to the differential equation

$$\frac{d}{dt}\xi_t^x = |\xi_t^x|^2 \xi_t^x, \qquad t > 0,$$

whose not vanishing solutions explode in a finite time. And also the solutions of the stochastic equation

$$d\xi_t^x = |\xi_t^x|^2 \xi_t^x dt + dW_t, \qquad t > 0,$$

with the initial condition $\xi_0^x = x \in \mathbb{R}^N$, do not exist for all t > 0. Indeed, the life time of the Markov process associated with $\{T(t)\}$ is finite since

$$\mathbb{P}_x\{t < \tau\} = (T(t)1)(x) < 1.$$

Example 5.2.6 Let \mathcal{A} be the one-dimensional operator defined by

$$\mathcal{A}\varphi(x) = \varphi''(x) + |x|^3 \varphi'(x), \qquad x \in \mathbb{R}.$$

The associated semigroup $\{T(t)\}$ is not conservative. Indeed, using the notation of Chapter 3 (see the formulas (3.1.3), (3.1.4) and (3.1.5)), it is easy to see that $Q \in L^1(-\infty,0)$, $Q \notin L^1(0,+\infty)$, $R \notin L^1(-\infty,0)$ and $R \in L^1(0,+\infty)$. Therefore, the point $-\infty$ is entrance and, hence, unaccessible while the point $+\infty$ is exit and, hence, accessible. According to Theorem 3.2.2, we conclude that the semigroup $\{T(t)\}$ is not conservative.

We now claim that $R(\lambda, \widehat{A})\mathbb{1}$ and $T(t)\mathbb{1}$ do not belong to $C_0(\mathbb{R}^N)$, for any $\lambda > 0$ and any t > 0, respectively. To show that $u = R(\lambda, \widehat{A})\mathbb{1}$ is not in $C_0(\mathbb{R})$, we observe that u is a solution of the elliptic equation

$$\lambda u(x) - u''(x) - |x|^3 u'(x) = 1, \quad x \in \mathbb{R},$$
 (5.2.7)

and it satisfies $0 \le u \le 1/\lambda$. Moreover, since the coefficients of the operator \mathcal{A} belong to $C^{2+\theta}_{\mathrm{loc}}(\mathbb{R})$ for any $\theta \in (0,1)$, it is readily seen that $u \in C^{4+\theta}_{\mathrm{loc}}(\mathbb{R})$. We claim that $u'(x) \le 0$ for any $x \in \mathbb{R}$. This, of course, will imply that

We claim that $u'(x) \leq 0$ for any $x \in \mathbb{R}$. This, of course, will imply that $u \notin C_0(\mathbb{R})$. By contradiction, suppose that u'(x) > 0 for any $x \in (a, b)$ for some $a, b \in \mathbb{R}$. Let

$$a_0 = \inf\{\alpha : \ u'(x) > 0 \ \forall x \in (\alpha, b)\}.$$

If $a_0 > -\infty$, we have $u'(a_0) = 0$ and u'(x) > 0 for any $x \in (a_0, b)$. Therefore, for such values of x,

$$u''(x) = \lambda u(x) - 1 - |x|^3 u'(x) < 0.$$

But this is a contradiction. Similarly, if $a_0 = -\infty$ we have u'(x) > 0 and $u''(x) \le 0$ for any x < b, so that

$$\lim_{x \to -\infty} u'(x) > 0,$$

which is a contradiction since u is bounded. We conclude that $u' \leq 0$. Since u does not identically vanish in \mathbb{R} , then $u \notin C_0(\mathbb{R})$.

To prove that, for any t > 0, T(t)1 does not belong to $C_0(\mathbb{R}^N)$, we show that $D_xT(t)1 \le 0$. For this purpose, let $u = T(\cdot)1$. According to Lemma 4.1.8 we

know that $D_t u(t,x) \leq 0$ for any $(t,x) \in \mathbb{R}_+ \times \mathbb{R}$. Suppose that $D_x u(t,x) > 0$ at some point $(t,x) \in \mathbb{R}_+ \times \mathbb{R}$. Then, we have

$$D_{xx}u(t,x) = D_tu(t,x) - |x|^3 D_xu(t,x) \le 0.$$

Repeating the same arguments as above, we see that the assumption on the positiveness of $D_x u(t,x)$ leads us to a contradiction. Therefore, $D_x u(t,x) \leq 0$ for any $(t,x) \in \mathbb{R}_+ \times \mathbb{R}$. Since T(t)1 does not identically vanish in \mathbb{R}^N , we conclude that $T(t)1 \notin C_0(\mathbb{R}^N)$.

5.3 Invariance of $C_0(\mathbb{R}^N)$

In this section we determine sufficient conditions implying that $C_0(\mathbb{R}^N)$ is invariant under T(t) for some t > 0. By the results in Section 5.2 we know that this is the case if $T(t)\mathbf{1}$ vanishes at infinity. But this is the case also under weaker assumptions as the next propositions show.

In Proposition 5.2.2 we have seen that T(t) maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ for any t>0 if and only if $R(\lambda, \widehat{A})$ does, for some (and hence all) $\lambda>0$. As we are going to show, the same property holds true as far as the invariance of $C_0(\mathbb{R}^N)$ is concerned. Moreover, in the case when $C_0(\mathbb{R}^N)$ is invariant under the semigroup, then the restriction of the semigroup to $C_0(\mathbb{R}^N)$ defines a strongly continuous semigroup.

Proposition 5.3.1 $C_0(\mathbb{R}^N)$ is invariant under the semigroup $\{T(t)\}$ if and only if it is invariant under $R(\lambda, \widehat{A})$ for any (equivalently, some) $\lambda > 0$. In such a case, the restriction of $\{T(t)\}$ to $C_0(\mathbb{R}^N)$ defines a strongly continuous semigroup. Its infinitesimal generator A_0 is the part of \widehat{A} in $C_0(\mathbb{R}^N)$, i.e.,

$$\begin{cases}
D(A_0) = \{ u \in D_{\max}(\mathcal{A}) \cap C_0(\mathbb{R}^N) : \mathcal{A}u \in C_0(\mathbb{R}^N) \}, \\
A_0u = \mathcal{A}u, \quad \text{for any } u \in D(A_0).
\end{cases}$$
(5.3.1)

Proof. Suppose that T(t) maps $C_0(\mathbb{R}^N)$ into itself for any t > 0. Then, by Proposition 2.2.7 we deduce that the restriction of $\{T(t)\}$ to $C_0(\mathbb{R}^N)$ is a strongly continuous semigroup in $C_0(\mathbb{R}^N)$. Since the resolvent operator associated with the restriction to $C_0(\mathbb{R}^N)$ of $\{T(t)\}$ is the restriction of $R(\lambda, \widehat{A})$ to $C_0(\mathbb{R}^N)$, we immediately deduce that $R(\lambda, \widehat{A})$ maps $C_0(\mathbb{R}^N)$ into itself for any $\lambda > c_0$.

Conversely, let us assume that $R(\lambda, \widehat{A})$ maps $C_0(\mathbb{R}^N)$ into itself for any $\lambda > c_0$. It is easy to show that, for any $\lambda > c_0$, the operator $\lambda I - A_0$ with domain (5.3.1) is bijective from $D(A_0)$ into $C_0(\mathbb{R}^N)$. Moreover,

$$R(\lambda, A_0) = R(\lambda, \widehat{A})_{|C_0(\mathbb{R}^N)}, \quad \lambda > c_0$$
 (5.3.2)

and, consequently,

$$||R(\lambda, A_0)||_{L(C_0(\mathbb{R}^N))} \le \frac{1}{\lambda - c_0}, \quad \lambda > c_0.$$
 (5.3.3)

Indeed, if $u \in D(A_0)$ is such that $\lambda u - A_0 u = 0$, then $u \in D(\widehat{A})$ (see (2.3.13)) and consequently $\lambda u - \widehat{A}u = 0$, implying that u = 0. Now take $f \in C_0(\mathbb{R}^N)$. The function $u = R(\lambda, \widehat{A})f$ is the unique solution in $D(\widehat{A})$ of the equation $\lambda u - Au = f$. Since $R(\lambda, \widehat{A})$ maps $C_0(\mathbb{R}^N)$ into itself, then $u \in D(\widehat{A}) \cap C_0(\mathbb{R}^N)$ and $Au \in C_0(\mathbb{R}^N)$. Still from (2.3.13) we deduce that $u \in D(A_0)$ and it solves the equation $\lambda u - A_0 u = f$. The formula (5.3.2) now easily follows. Similarly, (5.3.3) follows recalling that

$$||R(\lambda, \widehat{A})||_{L(C_b(\mathbb{R}^N))} \le \frac{1}{\lambda - c_0}, \quad \lambda > c_0;$$

see (2.1.4) and Section 2.3.

Now, the Hille-Yosida theorem (see Theorem B.1.5) implies that the operator A_0 with domain (5.3.1) is the generator of a strongly continuous semigroup $\{T_0(t)\}$ in $C_0(\mathbb{R}^N)$. By general results for strongly continuous semigroups we know that

$$R(\lambda, A_0)f = \int_0^{+\infty} e^{-\lambda t} T_0(t) f dt,$$

for any $\lambda > c_0$, where the integral converges in $C_0(\mathbb{R}^N)$. Since $R(\lambda, A_0)$ is the restriction of $R(\lambda, \widehat{A})$ to $C_0(\mathbb{R}^N)$ and

$$(R(\lambda,\widehat{A})g)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)g)(x) dt, \quad x \in \mathbb{R}^N,$$

for any $g \in C_b(\mathbb{R}^N)$ (see (2.2.14), (2.3.1)), we deduce that

$$\int_{0}^{+\infty} e^{-\lambda t} \left(\left(T(t) - T_0(t) \right) f \right) (x) dt = 0,$$

for any $f \in C_0(\mathbb{R}^N)$, any $x \in \mathbb{R}^N$ and any $\lambda > c_0$. The uniqueness of the Laplace transform implies that $T(t)f = T_0(t)f$ for any t > 0. Hence, T(t) maps $C_0(\mathbb{R}^N)$ into itself for any t > 0.

Finally, we show that, if $C_0(\mathbb{R}^N)$ is invariant under $R(\lambda_0, \widehat{A})$ for some $\lambda_0 > c_0$, then it is invariant under $R(\lambda, \widehat{A})$ for any $\lambda > c_0$. For this purpose, we set

$$\mathcal{F} = \{\lambda \in (c_0, +\infty) : C_0(\mathbb{R}^N) \text{ is invariant under } R(\lambda, \widehat{A})\}$$

and prove that \mathcal{F} is open and closed in $(c_0, +\infty)$. This will easily imply that $\mathcal{F} = (c_0, +\infty)$, since, by assumptions, \mathcal{F} is not empty. We begin by proving that \mathcal{F} is an open set. We fix $\lambda_1 \in \mathcal{F}$, $f \in C_0(\mathbb{R}^N)$ and prove that, if λ is

sufficiently close to λ_1 , the equation $\lambda u - \mathcal{A}u = f$ admits a solution $u \in D(A_0)$. As it is immediately seen, $u \in D(A_0)$ is a solution to the previous equation if and only if

 $(\lambda - \lambda_1)R(\lambda_1, \widehat{A})u + u = R(\lambda_1, \widehat{A})f.$ (5.3.4)

Since $C_0(\mathbb{R}^N)$ is invariant under $R(\lambda_1, \widehat{A})$, the operator $G = (\lambda - \lambda_1)R(\lambda_1, \widehat{A}) + I$ is bounded from $C_0(\mathbb{R}^N)$ into itself. Moreover, if $|\lambda - \lambda_1| < |\lambda_1 - c_0|$, then $\|(\lambda - \lambda_1)R(\lambda_1, \widehat{A})\|_{C_0(\mathbb{R}^N)} < 1$ (see again Sections 2.1 and 2.3). Therefore, G is bijective in $C_0(\mathbb{R}^N)$ and the equation (5.3.4) is uniquely solvable in $C_0(\mathbb{R}^N)$. This implies that $R(\lambda, \widehat{A}) f \in C_0(\mathbb{R}^N)$.

Let us now suppose that $\lambda \in (c_0, +\infty)$ is the limit of a sequence $\{\lambda_n\} \subset \mathcal{F}$. Then, for any $n \in \mathbb{N}$, $R(\lambda_n, \widehat{A})$ is a bounded operator in $C_0(\mathbb{R}^N)$. Since $R(\lambda_n, \widehat{A})$ converges to $R(\lambda, \widehat{A})$ in $L(C_b(\mathbb{R}^N))$, it follows that $R(\lambda, \widehat{A})$ belongs to $L(C_0(\mathbb{R}^N))$ as well, and \mathcal{F} is closed.

The next theorem gives a sufficient condition for the invariance of $C_0(\mathbb{R}^N)$. It should be compared with Proposition 5.2.4.

Theorem 5.3.2 Suppose that there exist $\lambda_0 > 0$, a compact set F and a strictly positive function $\varphi \in C^2(\mathbb{R}^N \setminus F) \cap C_0(\mathbb{R}^N \setminus F)$ such that

$$\lambda_0 \varphi(x) - \mathcal{A}\varphi(x) \ge 0, \quad x \in \mathbb{R}^N \setminus F.$$
 (5.3.5)

Then $C_0(\mathbb{R}^N)$ is invariant under T(t) for any t > 0.

Proof. We prove that $C_0(\mathbb{R}^N)$ is invariant under $R(\lambda_0 + 1, \widehat{A})$. Then, the conclusion will follow from Proposition 5.3.1.

We take a compactly supported function $f \in C_0(\mathbb{R}^N)$ and prove that $u = R(\lambda_0, \widehat{A})f$ belongs to $C_0(\mathbb{R}^N)$. The general case will follow by density since any function $f \in C_0(\mathbb{R}^N)$ can be approximated uniformly in \mathbb{R}^N by a sequence of compactly supported functions.

We fix $n_0 \in \mathbb{N}$ such that $B(n_0)$ contains both F and the support of f. Moreover, replacing φ with $\delta \varphi$ for a sufficiently large positive constant δ , we can assume that

$$\inf_{x \in \partial B(n_0)} \varphi \ge (\lambda_0 - c_0)^{-1} ||f||_{\infty}.$$

Finally, for any $n \in \mathbb{N}$, let u_n be the (unique) solution in $\bigcap_{1 \leq p < +\infty} W^{2,p}(B(n))$ of the equation

$$\begin{cases} \lambda_0 u_n(x) - \mathcal{A}u_n(x) = f(x), & x \in B(n), \\ u_n(x) = 0, & x \in \partial B(n). \end{cases}$$

According to the proof of Theorem 2.1.1, u is the pointwise limit of the sequence $\{u_n\}$ and

$$||u_n||_{\infty} \le \frac{1}{\lambda_0 - c_0} ||f||_{\infty},$$

for any $n \in \mathbb{N}$. Since f vanishes outside $B(n_0)$, the function $w_n = \varphi - u_n$ satisfies $\lambda_0 w - \mathcal{A}w \geq 0$ in $B(n) \setminus \overline{B}(n_0)$ $(n > n_0)$, as well as the boundary conditions $w \geq 0$ on $\partial B(n_0) \cup \partial B(n)$. The classical maximum principle (see Theorem C.2.2(i)) now implies that $w \geq 0$ in $B(n) \setminus \overline{B}(n_0)$. Thus, letting n go to $+\infty$, we obtain that

$$0 \le (R(\lambda_0, \widehat{A})f)(x) \le \varphi(x), \qquad x \in \mathbb{R}^N \setminus \overline{B}(n_0).$$

This finishes the proof.

Example 5.3.3 Consider again the operator \mathcal{A} and the function φ defined in Example 5.2.5. Taking (5.2.5) into account, it is immediate to see that the condition (5.3.5) reads as

$$\langle b(x), x \rangle \ge -\frac{\lambda_0}{2} (1+|x|^2) - N + 4 \frac{|x|^2}{1+|x|^2}, \qquad x \in \mathbb{R}^N,$$

for some $\lambda_0 > 0$. If it is satisfied then $C_0(\mathbb{R}^N)$ is invariant. This holds, for instance, if we have

$$\langle b(x), x \rangle \ge C - M|x|^2, \qquad x \in \mathbb{R}^N,$$
 (5.3.6)

for some $C \in \mathbb{R}$ and M > 0. Indeed, it suffices to take $\lambda_0 > 2M$.

The condition (5.3.6) should be compared with the condition (5.2.6), which guarantees that the semigroup associated with the operator \mathcal{A} maps $C_b(\mathbb{R}^N)$ into $C_0(\mathbb{R}^N)$ and hence, in particular, leaves $C_0(\mathbb{R}^N)$ invariant.

We conclude this section with the following proposition which gives a sufficient condition ensuring that $C_0(\mathbb{R}^N)$ is not preserved by the action of the semigroup. It is a immediate corollary of Theorem 5.1.11.

Proposition 5.3.4 Fix t > 0. If $T(t)\mathbf{1} = \mathbf{1}$ and T(t) is compact, then $C_0(\mathbb{R}^N)$ is not invariant for T(t).

Chapter 6

Uniform estimates for the derivatives of T(t)f

6.0 Introduction

In this chapter we prove some uniform estimates for the derivatives of the function T(t)f, when $f \in C_b(\mathbb{R}^N)$ and $\{T(t)\}$ is the semigroup associated with the uniformly elliptic operator

$$A = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij} + \sum_{j=1}^{N} b_{j}(x)D_{j},$$

with unbounded coefficients in \mathbb{R}^N .

The problem of estimating the derivatives of T(t)f has been studied in literature with both analytic ([13, 15, 108]) and probabilistic methods ([29, 30, 143]).

Here, we present the results of [18]. More precisely, we prove uniform estimates for the first-, second- and third-order derivatives of T(t)f. First, we show that, for any $\omega > 0$ and any $k, l \in \mathbb{N}$, with $0 \le k \le l \le 3$, there exists a positive constant $C_{k,l} = C_{k,l}(\omega)$ such that

$$||T(t)f||_{C_h^l(\mathbb{R}^N)} \le C_{k,l} t^{-\frac{l-k}{2}} e^{\omega t} ||f||_{C_h^k(\mathbb{R}^N)}, \qquad f \in C_b^k(\mathbb{R}^N), \ t > 0.$$
 (6.0.1)

Although we limit ourselves to the case when $l \leq 3$, the techniques that we present work as well for l > 3 under suitable additional assumptions on the coefficients.

To prove (6.0.1) we use the Bernstein method (see [14]) and approximate T(t)f by solutions of Cauchy problems in bounded domains. We assume weak dissipativity-type and growth conditions on the coefficients of \mathcal{A} . We notice that some dissipativity condition is necessary, because in general the estimate (6.0.1) fails; see Example 6.1.11.

By interpolation, we can then extend the estimate (6.0.1) to the case when $k, l \in \mathbb{R}_+$, $0 \le k \le l \le 3$. This allows us to prove optimal Schauder estimates for the solution of the nonhomogeneous Cauchy problem

for of the nonhomogeneous Cauchy problem
$$\begin{cases} D_t u(t,x) = \mathcal{A} u(t,x) + g(t,x), & t>0, \ x\in\mathbb{R}^N, \\ u(0,x) = f(x), & x\in\mathbb{R}^N, \end{cases}$$

as well as for the elliptic equation $\lambda u - Au = f$, $(\lambda > 0)$.

6.1 Uniform estimates

Throughout this section we assume the following hypotheses:

Hypotheses 6.1.1 (i) $q_{ij}(x) = q_{ji}(x)$ for any i, j = 1, ..., N and any $x \in \mathbb{R}^N$, and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \qquad \xi, x \in \mathbb{R}^N,$$

for some function $\kappa: \mathbb{R}^N \to \mathbb{R}$ such that

$$\inf_{x \in \mathbb{R}^N} \kappa(x) = \kappa_0 > 0;$$

(ii) there exist a function $\varphi \in C^2(\mathbb{R}^N)$ and two positive constants λ_0 and C such that

$$\lim_{|x| \to +\infty} \varphi(x) = +\infty, \qquad \mathcal{A}\varphi(x) - \lambda_0 \varphi(x) \le C, \quad x \in \mathbb{R}^N;$$

(iii) there exists a constant C > 0 such that

$$\left| \sum_{i,j=1}^{N} q_{ij}(x)x_{j} \right| \leq C(1+|x|^{2})\kappa(x), \qquad \operatorname{Tr}(Q(x)) \leq C(1+|x|^{2})\kappa(x),$$
(6.1.1)

$$\sum_{i=1}^{N} b_i(x)x_i \le C(1+|x|^2)\kappa(x), \tag{6.1.2}$$

for any $x \in \mathbb{R}^N$, i = 1, ..., N.

Moreover, in the next theorems we always assume that one of the following hypotheses holds true:

(iv-1) $q_{ij}, b_j \in C^{1+\delta}_{loc}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any $i,j=1,\ldots,N$, and there exist a positive constant C and a function $d: \mathbb{R}^N \to \mathbb{R}$ with $L_1 := \sup_{x \in \mathbb{R}^N} \{d(x)/\kappa(x)\} < +\infty$, such that $|D_k q_{ij}(x)| \leq C\kappa(x)$ for any $i,j,k=1,\ldots,N$ and

$$\sum_{i,j=1}^{N} D_{i}b_{j}(x)\xi_{i}\xi_{j} \leq d(x)|\xi|^{2}, \qquad x, \xi \in \mathbb{R}^{N};$$
 (6.1.3)

(iv-2) $q_{ij}, b_j \in C^{2+\delta}_{loc}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any $i,j=1,\ldots,N$; Hypothesis 6.1.1(iv-1) holds true and there exist a positive function $r: \mathbb{R}^N \to \mathbb{R}$ and three constants $K_1 \in \mathbb{R}, L_2, L_3 > 0$ such that

$$|D^{\beta}b_j(x)| \le r(x), \qquad x \in \mathbb{R}^N, \ j = 1, \dots, N, \ |\beta| = 2,$$

$$d(x) + L_2 r(x) \le L_3 \kappa(x), \qquad x \in \mathbb{R}^N$$

and

$$\sum_{i,j,h,k=1}^{N} D_{hk} q_{ij}(x) m_{ij} m_{hk} \le K_1 \kappa(x) \sum_{h,k=1}^{N} m_{hk}^2, \qquad x \in \mathbb{R}^N,$$

for any symmetric matrix $M = (m_{hk})$;

(iv-3) $q_{ij}, b_j \in C^{3+\delta}_{loc}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any i, j = 1, ..., N; Hypothesis 6.1.1(iv-2) holds true and there exists a constant C > 0 such that $|D^{\beta}b_j(x)| \leq r(x)$ and $|D^{\beta}q_{ij}(x)| \leq C\kappa(x)$ for any i, j = 1, ..., N, any $|\beta| = 3$ and any $x \in \mathbb{R}^N$.

Remark 6.1.2 Differently from what we did in the previous chapters, here we assume that the function κ is far from zero.

Remark 6.1.3 We remark that in 6.1.1(iv-1) (l = 2, 3) we can take

$$r(x) = L_4(1 + |d(x)|),$$

for any $L_4 > 0$. It is sufficient to fix L_2 and L_3 satisfying

$$L_2 < L_4^{-1}, \qquad L_3 = (1 + L_2 L_4) \left(\sup_{x \in \mathbb{R}^N} \frac{d(x)}{\kappa(x)} \right)^+ + L_2 L_4 \kappa_0^{-1}$$

in (iv-2).

Remark 6.1.4 In some situation, Hypothesis 6.1.1(iii) is easily implied by Hypothesis 6.1.1(iv-1). This is the case, for instance, when there exists a positive constant K such that

$$\int_{0}^{1} \kappa(tx)dt \le K\kappa(x), \qquad x \in \mathbb{R}^{N}. \tag{6.1.4}$$

We limit ourselves to showing that in such a situation, the condition (6.1.2) follows from Hypothesis 6.1.1(iv-1). The same argument can be used to prove that also (6.1.1) is a consequence of Hypothesis 6.1.1(iv-1).

To show (6.1.2) we observe that taking $(x, \xi) = (tz, z)$ in (6.1.3) gives

$$\frac{d}{dt} \sum_{j=1}^{N} b_j(tz) z_j \le L_1 \kappa(tz) |z|^2, \qquad z \in \mathbb{R}^N.$$
 (6.1.5)

Integrating (6.1.5) with respect to $t \in [0,1]$ and taking (6.1.4) into account, we get

$$\sum_{j=1}^{N} b_{j}(z)z_{j} \leq \sum_{j=1}^{N} b_{j}(0)z_{j} + L_{1}|z|^{2} \int_{0}^{1} \kappa(tz)dt$$

$$\leq |z||b(0)| + KL_{1}|z|^{2}\kappa(z)$$

$$\leq C'|z|(1+|z|)\kappa(z), \tag{6.1.6}$$

for any $z \in \mathbb{R}^N$, where $b(0) = (b_1(0), \dots, b_N(0))$, $C' = \max\{|b(0)|\kappa_0^{-1}, KL_1\}$. Let us observe that (6.1.4) is satisfied, for instance, in the case when $\kappa(x) = \overline{\kappa}(|x|)$ for any $x \in \mathbb{R}^N$ and some nondecreasing function $\overline{\kappa} : \mathbb{R}_+ \to \mathbb{R}_+$, and in the case when $\kappa(x) = c + \widetilde{\kappa}(x)$, c and $\widetilde{\kappa}$ being, respectively, a positive constant and a nonnegative homogeneous function of degree $\alpha > 0$.

In the sequel, for any $n \in \mathbb{N}$, we denote by u_n the (unique) classical solution of the Cauchy-Dirichlet problem

$$\begin{cases} D_t u_n(t, x) = \mathcal{A} u_n(t, x), & t > 0, x \in B(n), \\ u_n(t, x) = 0, & t > 0, x \in \partial B(n), \\ u_n(0, x) = f(x), & x \in B(n), \end{cases}$$
(6.1.7)

corresponding to $f \in C_b(\mathbb{R}^N)$ (i.e., the unique solution in $C_b([0, +\infty) \times \overline{B}(n)) \cap C^{1,2}((0, +\infty) \times \overline{B}(n))$). Here, $\vartheta_n : \mathbb{R}^N \to \mathbb{R}$ is defined by

$$\vartheta_n(x) = \varphi(|x|/n), \qquad x \in \mathbb{R}^N, \quad n \in \mathbb{N},$$
 (6.1.8)

 $\varphi\in C_c^\infty(\mathbb{R})$ being any nonincreasing function such that $\chi_{(-1/2,1/2)}\leq \varphi\leq \chi_{(-1,1)}$.

Remark 6.1.5 Repeating the same arguments as in Theorems 2.2.1 and in Remark 2.2.3 and taking Remark 4.1.4 and Theorem 4.1.3 into account, it is easy to check that u_n converges to $u := T(\cdot)f$ locally uniformly in $(0, +\infty) \times \mathbb{R}^N$ as n tends to $+\infty$. From the interior Schauder estimate in Theorem C.1.4(ii), we easily deduce that

$$\lim_{n \to +\infty} \sup_{\varepsilon < t < T} \|u_n(t, \cdot) - u(t, \cdot)\|_{C^3(B(k))} = 0, \tag{6.1.9}$$

for any $0 < \varepsilon < T$ and any $k \in \mathbb{N}$.

To prove the main result of this section we need the following lemma.

Lemma 6.1.6 Let $\Omega \subset \mathbb{R}^N$ be any smooth bounded open set. Assume that Hypothesis 6.1.1(i) is satisfied in $\overline{\Omega}$ and that the coefficients belong to $C^{k+\alpha}(\overline{\Omega})$ for some $\alpha \in (0,1)$ and $k \in \{1,2,3\}$. Then, for any $f \in C_c^j(\overline{\Omega})$ $(j=0,\ldots,k)$, the function $(t,x) \mapsto t^{(k-j)/2}(D^kT_n(t)f)(x)$ is continuous in $[0,T] \times \overline{\Omega}_{\delta}$ for any T > 0 and any $\delta > 0$, where $\Omega_{\delta} = \{x \in \Omega : \delta \leq \inf_{y \in \partial \Omega} |x-y|\}$.

Proof. The proof follows from a density argument. We limit ourselves to proving it in the case when j=k=3, the other cases being completely similar, and even simpler. It is well known that for any $f \in C_c^{\infty}(\Omega)$, the function $D^k T_n(\cdot) f$ (k=1,2,3) is continuous in $[0,T] \times \overline{\Omega}$ (see Proposition C.3.2). Using a method similar to that used in the proof of the following Theorem 6.1.7, we can easily show that there exists a positive constant $C = C_{T,\delta}$, independent of f, such that

$$||T(t)f||_{C^{3}(\overline{\Omega}_{\delta})} \le C||f||_{C^{3}(\overline{\Omega})}, \qquad t \in [0, T].$$
 (6.1.10)

Now with any $f \in C_c^3(\overline{\Omega})$ we associate a sequence of smooth functions $\{f_n\} \subset C_c^{\infty}(\Omega)$ converging to f in $C^k(\overline{\Omega})$. Since, for any $n \in \mathbb{N}$ and $l \in \{1, 2, 3\}$, the function $D^l T(\cdot) f_n$ is continuous in $[0, T] \times \overline{\Omega}$ for any T > 0, taking (6.1.10) into account, we easily get the assertion.

Now, we prove the main result of this section.

Theorem 6.1.7 Let Hypotheses 6.1.1(i)-6.1.1(iii), 6.1.1(iv-l) be satisfied for some $l \in \{1,2,3\}$. Then, for any $\omega > 0$ and any $k = 0,\ldots,l$, there exist constants $C_{k,l} = C_{k,l}(\omega) > 0$ such that

$$||T(t)f||_{C_b^l(\mathbb{R}^N)} \le C_{k,l} t^{-\frac{l-k}{2}} e^{\omega t} ||f||_{C_b^k(\mathbb{R}^N)}, \qquad f \in C_b^k(\mathbb{R}^N), \ t > 0 \quad (6.1.11)$$

holds true. In particular, if k = l we can take $\omega = 0$ in (6.1.11).

Proof. We begin the proof considering the case when k=0 and l=3. For any $n \in \mathbb{N}$, let ϑ_n be the smooth function defined in (6.1.8). We fix $\omega > 0$ and $t_0 > 0$ such that

$$e^{\omega t_0} t^{-\frac{3}{2}} \ge 1, \qquad t \ge t_0,$$
 (6.1.12)

and define the function

$$v_{0,3,n}(t,x) = |u_n(t,x)|^2 + at\vartheta_n^2 |Du_n(t,x)|^2 + a^2 t^2 \vartheta_n^4 |D^2 u_n(t,x)|^2 + a^3 t^3 \vartheta_n^6 |D^3 u_n(t,x)|^2,$$

for any $t \in [0, t_0]$ and any $x \in B(n)$, where $u_n = T_n(\cdot)(\vartheta_n f)$ is the solution of the Cauchy-Dirichlet problem (6.1.7). For notational convenience, throughout the rest of the proof, we drop out the dependence on n, when there is

no damage of confusion. Observing that u_n is in $C([0,+\infty); C(\overline{B}(n)))$ (see Proposition B.2.5(i) and Theorem C.3.6(iv)) and taking Lemma 6.1.6 into account, it follows that $v_{0,3}$ is continuous in $[0,t_0] \times \overline{B}(n)$ and, with some computations, one can see that $v_{0,3}$ solves the Cauchy problem

$$\begin{cases} D_t v_{0,3}(t,x) = \mathcal{A}v_{0,3}(t,x) + g(t,x), & t \in (0,t_0], x \in B(n), \\ v_{0,3}(t,x) = 0, & t \in [0,t_0], x \in \partial B(n), \\ v_{0,3}(0,x) = (\vartheta f)^2(x), & x \in \overline{B}(n), \end{cases}$$
(6.1.13)

where $g = \sum_{j=1}^{8} g_j$ with

$$g_{1} = -2\sum_{i,j=1}^{N} q_{ij} D_{i} u D_{j} u - 2at \vartheta^{2} \sum_{i,j,h=1}^{N} q_{ij} D_{ih} u D_{jh} u$$
$$-2a^{2} t^{2} \vartheta^{4} \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} u D_{jhk} u - 2a^{3} t^{3} \vartheta^{6} \sum_{i,j,h,k,l=1}^{N} q_{ij} D_{ihkl} u D_{jhkl} u,$$

$$(6.1.14)$$

$$g_{2} = -2at|Du|^{2} \sum_{i,j=1}^{N} q_{ij}D_{i}\vartheta D_{j}\vartheta - 12a^{2}t^{2}\vartheta^{2}|D^{2}u|^{2} \sum_{i,j=1}^{N} q_{ij}D_{i}\vartheta D_{j}\vartheta - 30a^{3}t^{3}\vartheta^{4}|D^{3}u|^{2} \sum_{i,j=1}^{N} q_{ij}D_{i}\vartheta D_{j}\vartheta,$$

$$(6.1.15)$$

$$g_{3} = -2at\vartheta \mathcal{A}(\vartheta)|Du|^{2} - 4a^{2}t^{2}\vartheta^{3}\mathcal{A}(\vartheta)|D^{2}u|^{2} - 6a^{3}t^{3}\vartheta^{5}\mathcal{A}(\vartheta)|D^{3}u|^{2}$$

$$-8at\vartheta \sum_{i,j,h=1}^{N} q_{ij}D_{j}\vartheta D_{h}uD_{ih}u - 16a^{2}t^{2}\vartheta^{3} \sum_{i,j,h,k=1}^{N} q_{ij}D_{j}\vartheta D_{hk}uD_{ihk}u$$

$$-24a^{3}t^{3}\vartheta^{5} \sum_{i,j,h,k,l=1}^{N} q_{ij}D_{j}\vartheta D_{hkl}uD_{ihkl}u,$$
(6.1.16)

$$g_{4} = 2at\vartheta^{2} \sum_{j,h=1}^{N} D_{h}b_{j}D_{j}uD_{h}u + 4a^{2}t^{2}\vartheta^{4} \sum_{j,h,k=1}^{N} D_{h}b_{j}D_{jk}uD_{hk}u + 6a^{3}t^{3}\vartheta^{6} \sum_{j,h,k,l=1}^{N} D_{h}b_{j}D_{jkl}uD_{hkl}u,$$

$$(6.1.17)$$

$$g_{5} = 2at\vartheta^{2} \sum_{i,j,h=1}^{N} D_{h}q_{ij}D_{h}uD_{ij}u + 4a^{2}t^{2}\vartheta^{4} \sum_{i,j,h,k=1}^{N} D_{h}q_{ij}D_{hk}uD_{ijk}u + 6a^{3}t^{3}\vartheta^{6} \sum_{i,j,h,k,l=1}^{N} D_{h}q_{ij}D_{hkl}uD_{ijkl}u,$$

$$(6.1.18)$$

$$g_{6} = 2a^{2}t^{2}\vartheta^{4} \left(\sum_{i,j,h,k=1}^{N} D_{hk}q_{ij}D_{ij}uD_{hk}u + \sum_{j,h,k=1}^{N} D_{hk}b_{j}D_{j}uD_{hk}u \right) + 6a^{3}t^{3}\vartheta^{6} \left(\sum_{i,j,h,k,l=1}^{N} D_{hk}q_{ij}D_{ijl}uD_{hkl}u + \sum_{j,h,k,l=1}^{N} D_{hk}b_{j}D_{jl}uD_{hkl}u \right),$$

$$(6.1.19)$$

$$g_7 = 2a^3t^3\vartheta^6 \left(\sum_{i,j,h,k,l=1}^N D_{hkl}q_{ij}D_{ij}uD_{hkl}u + \sum_{j,h,k,l=1}^N D_{hkl}b_jD_juD_{hkl}u \right),$$
(6.1.20)

$$g_8 = a\vartheta^2 |Du|^2 + 2a^2t\vartheta^4 |D^2u|^2 + 3a^3t^2\vartheta^6 |D^3u|^2.$$
 (6.1.21)

Taking the ellipticity condition 6.1.1(i) into account, we easily deduce that

$$g_1 \le -2\kappa |Du|^2 - 2at\vartheta^2 \kappa |D^2u|^2 - 2a^2t^2\vartheta^4 \kappa |D^3u|^2 - 2a^3t^3\vartheta^6 \kappa |D^4u|^2 \quad (6.1.22)$$

and that $g_2 \leq 0$.

To estimate the function g_3 we observe that, by virtue of (6.1.1) and recalling that $\varphi' = 0$ in (-1/2, 1/2), it can be easily shown that

$$\begin{split} &|\operatorname{Tr}(Q(x)D^{2}\vartheta(x))|\\ &\leq |\varphi'(|x|/n)| \left(\frac{1}{|x|n}\operatorname{Tr}(Q(x)) + \frac{1}{|x|^{3}n}\langle Q(x)x,x\rangle\right)\\ &+ |\varphi''(|x|/n)| \frac{1}{|x|^{2}n^{2}}\langle Q(x)x,x\rangle\\ &\leq C\kappa(x) \sup_{\frac{n}{2}\leq |x|\leq n} \left[|\varphi'(|x|/n)| \left(\frac{1+|x|^{2}}{|x|n} + \frac{1+|x|^{2}}{|x|^{3}n} \sum_{i=1}^{N}|x_{i}|\right)\\ &+ C|\varphi''(|x|/n)| \frac{1+|x|^{2}}{|x|^{2}n^{2}} \sum_{i=1}^{N}|x_{i}|\right]\\ &\leq 4C\left(1+2\sqrt{N}\right) \|\varphi'\|_{\infty}\kappa(x) + 4C\sqrt{N}\|\varphi''\|_{\infty}\kappa(x) =: C'\kappa(x), \end{split}$$

for any $x \in B(n)$ and any $n \ge 1$. Similarly,

$$\left| \sum_{i,j=1}^{N} q_{ij}(x) D_{j} \vartheta(x) \right| \leq C \kappa(x) \sup_{\frac{n}{2} \leq |x| \leq n} \left(|\varphi'(|x|/n)| \frac{1+|x|^{2}}{|x|n} \right)$$

$$\leq 4C \|\varphi'\|_{\infty} \kappa(x) =: C'' \kappa(x), \tag{6.1.24}$$

for any $i=1,\ldots,N.$ Using (6.1.2) and recalling that φ' is nonincreasing in \mathbb{R} , we obtain that

$$-\sum_{i=1}^{N} b_i(x) D_i \vartheta(x) = -\varphi'(|x|/n) \frac{1}{|x|n} \sum_{j=1}^{N} b_j(x) x_j$$

$$\leq C\kappa(x) \sup_{\frac{n}{2} \leq |x| \leq n} \left(-\varphi'(|x|/n) \frac{1+|x|^2}{|x|n} \right)$$

$$\leq C'' \kappa(x), \tag{6.1.25}$$

for any $x \in B(n)$.

Taking (6.1.23), (6.1.24) and (6.1.25) into account and recalling that for any $a, b, \varepsilon > 0$ it holds that $ab \leq (4\varepsilon)^{-1}a^2 + \varepsilon^2b^2$, we can now show that

$$\begin{split} g_{3} & \leq 2aC't\kappa\vartheta|Du|^{2} + 4a^{2}C't^{2}\kappa\vartheta^{3}|D^{2}u|^{2} + 6a^{3}C't^{3}\kappa\vartheta^{5}|D^{3}u|^{2} \\ & + 2aC''t\kappa\vartheta|Du|^{2} + 4a^{2}C''t^{2}\kappa\vartheta^{3}|D^{2}u|^{2} + 6a^{3}C''t^{3}\kappa\vartheta^{5}|D^{3}u|^{2} \\ & + 8aC''t\kappa\left(\frac{N}{4\varepsilon}|Du|^{2} + \varepsilon\vartheta^{2}|D^{2}u|^{2}\right) \\ & + 16a^{2}C''t^{2}\kappa\left(\frac{N}{4\varepsilon}\vartheta^{2}|D^{2}u|^{2} + \varepsilon\vartheta^{4}|D^{3}u|^{2}\right) \\ & + 24a^{3}C''t^{3}\kappa\left(\frac{N}{4\varepsilon}\vartheta^{4}|D^{3}u|^{2} + \varepsilon\vartheta^{6}|D^{4}u|^{2}\right) \\ & \leq 2a\left(C' + C'' + C''\frac{N}{\varepsilon}\right)t\kappa|Du|^{2} \\ & + 4a\left(aC't + aC''t + 2C''\varepsilon + aC''t\frac{N}{\varepsilon}\right)t\kappa\vartheta^{2}|D^{2}u|^{2} \\ & + 2a^{2}\left(3aC't + 3aC''t + 8C''\varepsilon + 3aC''t\frac{N}{\varepsilon}\right)t^{2}\kappa\vartheta^{4}|D^{3}u|^{2} \\ & + 24a^{3}C''t^{3}\varepsilon\kappa\vartheta^{6}|D^{4}u|^{2}. \end{split}$$

Taking advantage of Hypothesis 6.1.1(iv-1), we deduce that

$$g_4 \le 2atd \,\vartheta^2 |Du|^2 + 4a^2t^2d \,\vartheta^4 |D^2u|^2 + 6a^3t^3d \,\vartheta^6 |D^3u|^2. \tag{6.1.27}$$

The terms g_5 , g_6 and g_7 can be estimated in a similar way taking Hypothesis 6.1.1(iv-3) into account, and they yield

$$g_{5} \leq atC \frac{N^{2}}{2\varepsilon} \kappa |Du|^{2} + aCN \left(2\varepsilon + at \frac{N}{\varepsilon}\right) t\kappa \vartheta^{2} |D^{2}u|^{2}$$

$$+ a^{2}CN \left(4\varepsilon + 3at \frac{N}{2\varepsilon}\right) t^{2} \kappa \vartheta^{4} |D^{3}u|^{2} + 6a^{3}t^{3}\varepsilon CN\kappa \vartheta^{4} |D^{4}u|^{2},$$

$$(6.1.28)$$

$$g_{6} \leq a^{2}t^{2}\vartheta^{2}r\frac{N^{2}}{2\varepsilon}|Du|^{2} + a^{2}t^{2}\vartheta^{4}\left[2K_{1}\kappa + N\left(2\varepsilon + 3at\frac{N}{2\varepsilon}\right)r\right]|D^{2}u|^{2} + 6a^{3}t^{3}\vartheta^{6}\left(K_{1}\kappa + \varepsilon Nr\right)|D^{3}u|^{2},$$

$$(6.1.29)$$

$$g_{7} \leq a^{3}t^{3}\vartheta^{2}r\frac{N^{3}}{2\varepsilon}|Du|^{2} + a^{3}t^{3}\vartheta^{4}C\kappa\frac{N^{3}}{2\varepsilon}|D^{2}u|^{2} + 2a^{3}t^{3}\vartheta^{6}\varepsilon N(C\kappa N + r)|D^{3}u|^{2}.$$
(6.1.30)

From (6.1.22), (6.1.26)-(6.1.30) we immediately deduce that for any $t \in (0, t_0]$ we have

$$g \leq \left\{ -\kappa_0 + a + \kappa \left[-1 + 2at \left(C' + C'' + C'' \frac{N}{\varepsilon} \right) + atC \frac{N^2}{2\varepsilon} + a^2 t^2 C \frac{N^2}{2\varepsilon} \right] \right.$$

$$\left. + at \left[2d + at(1 + atN) \frac{N^2}{2\varepsilon} r \right] \right\} |Du|^2$$

$$+ a \left\{ -\kappa_0 + 2a + \kappa \left[-1 + 4 \left(aC't + aC''t + 2C''\varepsilon + aC''t \frac{N}{\varepsilon} \right) + CN \left(2\varepsilon + at \frac{N}{\varepsilon} \right) + 2K_1^+ at + a^2 t^2 C \frac{N^3}{2\varepsilon} \right] \right.$$

$$\left. + at \left[4d + \left(2\varepsilon N + 3at \frac{N^2}{2\varepsilon} \right) r \right] \right\} t\vartheta^2 |D^2 u|^2$$

$$+ a^2 \left\{ -\kappa_0 + 3a + \kappa \left[-1 + 2 \left(3aC't + 3aC''t + 8C''\varepsilon + 3aC''t \frac{N}{\varepsilon} \right) + CN \left(4\varepsilon + 3at \frac{N}{2\varepsilon} \right) + 6atK_1^+ + 2at\varepsilon CN^2 \right] \right.$$

$$\left. + 2at \left(3d + 4\varepsilon Nr \right) \right\} t^2 \vartheta^4 |D^3 u|^2$$

$$+ 2a^3 \left(-1 + 12C''\varepsilon + 3\varepsilon CN \right) t^3 \kappa \vartheta^6 |D^4 u|^2.$$

$$(6.1.31)$$

We now choose (a, ε) , sufficiently small, satisfying the following set of inequalities:

$$\begin{cases} 3a - \kappa_0 \leq 0, & 4\varepsilon N \leq 3L_2, & at_0(1 + at_0 N) \frac{N^2}{2\varepsilon} \leq 2L_2, \\ 2\varepsilon N + 3at_0 \frac{N^2}{2\varepsilon} \leq 4L_2, \\ -1 + at_0 \left[4\left(C' + C'' + C'' \frac{N}{\varepsilon} + L_3\right) + C\frac{N^2}{\varepsilon} + 2K_1^+ \right] \\ +2\varepsilon \left(CN + 4C''\right) + a^2 t_0^2 C\frac{N^3}{2\varepsilon} \leq 0, \\ -1 + at_0 \left[6\left(C' + C'' + C'' \frac{N}{\varepsilon} + K_1^+ + L_3\right) + 3C\frac{N^2}{2\varepsilon} + 2\varepsilon CN^2 \right] \\ +4\varepsilon \left(CN + 4C''\right) \leq 0. \end{cases}$$

With such a choice of (a, ε) we get $g(t, x) \leq 0$ for any $t \in (0, t_0]$ and any $x \in B(n)$. From the classical maximum principle we now deduce that

$$|v_{0,3,n}(t,x)| \le \|\vartheta_n f\|_{\infty}^2 \le \|f\|_{\infty}^2, \qquad t \in [0,t_0], \ x \in B(n).$$

Taking the limit as n tends to $+\infty$, by (6.1.9) we deduce that (6.0.1) holds for any $t \in [0, t_0]$, some constant $C_{0,3} = C_{0,3}(t_0) > 0$ and with $\omega = 0$. Using the semigroup property we can then extend the estimate to all the positive t. Indeed, taking (6.1.12) into account, we get, for any $t > t_0$,

$$\begin{split} \|T(t)f\|_{C^3_b(\mathbb{R}^N)} &= \|T(t_0)T(t-t_0)f\|_{C^3_b(\mathbb{R}^N)} \leq C_{0,3}t_0^{-\frac{3}{2}} \|T(t-t_0)f\|_{\infty} \\ &\leq C_{0,3}t_0^{-\frac{3}{2}} \|f\|_{\infty} \leq C_{0,3}t_0^{-\frac{3}{2}} e^{\omega t}t^{-\frac{3}{2}} \|f\|_{\infty}, \end{split}$$

and, so, (6.0.1) follows with $C_{0,3}(\omega) = \max\{C_{0,3}(t_0), C_{0,3}(t_0)t_0^{-3/2}\}.$

In the other cases the proof is very similar. It suffices to apply the quoted arguments to the function

$$v_{k,l,n}(t,x) = \sum_{j=0}^{l} a^{j} t^{(j-k)^{+}} (\vartheta_{n}(x))^{2j} |D^{j} u_{n}(t,x)|^{2}, \qquad t \in (0,t_{0}], \ x \in B(n).$$

Let us just show that, if k=l, we can take $\omega=0$ in (6.0.1). We only consider the case when l=3. A straightforward computation shows that $v_{3,3,n}$ is a classical solution to the Cauchy-Dirichlet problem (6.1.13) with $v_{3,3,n}(0,\cdot)=\sum_{j=0}^{3}|D^{j}(\vartheta_{n}f)|^{2}$ and g_{n} being replaced with $\widetilde{g}_{n}=\sum_{j=1}^{7}\widetilde{g}_{j,n}$, where $\widetilde{g}_{j,n}$ ($j=1,\ldots,7$) are defined by the right-hand sides of (6.1.14)-(6.1.20) after replacing each t, therein appearing as coefficients of u_{n} or of its derivatives, with t=1. Arguing as above we can easily show that \widetilde{g}_{n} can

be estimated, for any t > 0, by the last side of (6.1.31), where we set t = 1and replace the terms $-\kappa_0 + a$, $-\kappa_0 + 2a$ and $-\kappa_0 + 3a$ simply with $-\kappa$. It is now clear that we can take (a, ε) such that $\widetilde{g}_n(t, x) \leq 0$ for any t > 0, any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, and, consequently,

$$|v_{3,3,n}(t,x)| \le \|\vartheta f\|_{\infty}^2 + \|D(\vartheta f)\|_{\infty}^2 + \|D^2(\vartheta f)\|_{\infty}^2 + \|D^3(\vartheta f)\|_{\infty}^2, \quad (6.1.32)$$

for any $t \in [0, t_0]$, any $x \in \overline{B}(n)$ and any $n \in \mathbb{N}$. The estimate (6.1.32) yields (6.0.1) taking the limit as n tends to $+\infty$.

Now, by interpolation, we can extend (6.0.1) to the case when at least one between k and l is not integer.

Theorem 6.1.8 Let Hypotheses 6.1.1(i)–6.1.1(iii), 6.1.1(iv-l) be satisfied for some $l \in \{1,2,3\}$. Then, for any $\omega > 0$, any $k,m = 0,\ldots,l-1$ and any $\alpha, \theta \in [0,1]$ such that $k+\alpha \leq m+\theta$, there exist constants $C_{k+\alpha,m+\theta} =$ $C_{k+\alpha,m+\theta}(\omega) > 0$ such that

$$||T(t)f||_{C_b^{m+\theta}(\mathbb{R}^N)} \le C_{k+\alpha,m+\theta} t^{-\frac{m+\theta-k-\alpha}{2}} e^{\omega t} ||f||_{C_b^{k+\alpha}(\mathbb{R}^N)}, \tag{6.1.33}$$

for any $f \in C_b^{k+\alpha}(\mathbb{R}^N)$ and any $k = 0, \dots, m$. In particular, if $k + \alpha = m + \theta$ we can take $\omega = 0$ in (6.1.33).

Proof. The proof follows from an interpolation argument. We limit ourselves to sketching it in a particular case, since the same techniques can also be applied to all the other cases. So, let us assume that k=m=2 and $0<\alpha\leq$ $\theta < 1$. Moreover, fix $\omega > 0$ and t > 0. Setting $\omega' = \omega/(\theta - \alpha)$, from (6.1.11) with (k, l) = (2, 2) and (k, l) = (2, 3), we deduce that

$$||T(t)||_{L(C_b^2(\mathbb{R}^N), C_b^2(\mathbb{R}^N))} \le C_{2,2}, \qquad ||T(t)||_{L(C_b^2(\mathbb{R}^N), C_b^3(\mathbb{R}^N))} \le C_{2,3} t^{-\frac{1}{2}} e^{\omega' t}, \tag{6.1.34}$$

for any t>0. Recalling that $(C_b^2(\mathbb{R}^N); C_b^3(\mathbb{R}^N))_{\beta,\infty}=C_b^{2+\beta}(\mathbb{R}^N)$, for any $\beta \in (0,1)$ (see Theorem A.4.8), and applying Proposition A.4.2, we easily see from (6.1.34) that T(t) is bounded from $C_b^2(\mathbb{R}^N)$ into $C_b^{2+\beta}(\mathbb{R}^N)$ for any $\beta \in (0,1)$ and

$$||T(t)||_{L(C_b^2(\mathbb{R}^N), C_b^{2+\beta}(\mathbb{R}^N))} \le C_{2,2+\beta} t^{-\frac{\beta}{2}} e^{\beta \omega' t}, \qquad t > 0, \tag{6.1.35}$$

where $C_{2,2+\beta} = C_{2,2}^{1-\beta} C_{2,3}^{\beta}$. Applying the same argument to (6.1.11), with (k,l) = (3,3) and to (6.1.35), we deduce that T(t) is bounded from $C_b^{2+\alpha}(\mathbb{R}^N)$ into $C_b^{2+\beta+(1-\beta)\alpha}(\mathbb{R}^N)$ and

$$\|T(t)\|_{L(C_b^{2+\alpha}(\mathbb{R}^N),C_b^{2+\beta+(1-\beta)\alpha}(\mathbb{R}^N))} \leq C_{2+\alpha,2+\beta+(1-\beta)\alpha} t^{-(1-\alpha)\frac{\beta}{2}} e^{(1-\alpha)\beta\omega' t},$$

where
$$C_{2+\alpha,2+\beta+(1-\beta)\alpha} = C_{2,2+\beta}^{1-\alpha} C_{3,3}^{\alpha}$$
. Now the assertion follows, taking $\beta = (\theta - \alpha)/(1 - \alpha)$.

In some cases, we can extend Theorems 6.1.7 and 6.1.8 to the case when in 6.1.1(iv-j), the j-th-order derivatives of the coefficients are merely continuous in \mathbb{R}^N . As the following theorem shows, this is the case when the condition (6.1.4) is satisfied and there exist m, M > 0 such that

$$\int_0^1 d(tx+y)dt \le M\kappa(x+y), \qquad x \in \mathbb{R}^N, \ y \in B(m). \tag{6.1.36}$$

As it is immediately seen, the previous condition is satisfied, for instance, when d is bounded from above.

Theorem 6.1.9 Suppose that Hypotheses 6.1.1(i)-6.1.1(iii), 6.1.1(iv-j) (j = 1, 2, 3) (with the j-th-order derivatives of the coefficients merely continuous in \mathbb{R}^N) and the conditions (6.1.4) and (6.1.36) are satisfied. Then, (6.1.11) and (6.1.33) hold true for any $k, l \in \mathbb{N}$, $k \leq l \leq j$.

Proof. The proof follows from a density argument. Of course, we can limit ourselves to dealing with (6.1.11), since, as Theorem 6.1.8 shows, (6.1.33) follows easily from (6.0.1).

For any $\varepsilon > 0$ let $\varphi^{\varepsilon}(x) = \varepsilon^{-N} \varphi(x/\varepsilon)$, where $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ is any non-negative even function compactly supported in B(1) and with integral 1. We denote by f^{ε} the convolution between f and φ^{ε} .

Let $\mathcal{A}^{\varepsilon}$ be defined as \mathcal{A} with q_{ij} and b_j being replaced, respectively, with q_{ij}^{ε} and b_j^{ε} for any $i, j = 1, \ldots, N$. As it is immediately seen, q_{ij}^{ε} and b_j^{ε} $(i, j = 1, \ldots, N)$ satisfy Hypotheses 6.1.1(i) and 6.1.1(iv-j), with κ, d, r being replaced with $\kappa^{\varepsilon}, d^{\varepsilon}$ and r^{ε} , and with $C^{\varepsilon} = C$, $L_i^{\varepsilon} = L_i$ for i = 1, 2, 3, $K_1^{\varepsilon} = K_1$ (if j > 1) and $\kappa_0^{\varepsilon} \ge \kappa_0$.

Let us check that q_{ij}^{ε} and b_{j}^{ε} (i, j = 1, ..., N) satisfy Hypothesis 6.1.1(iii) for some positive constant, independent of ε . For this purpose, we observe that (6.1.4) implies that

$$|q_{ij}(x)| \le |q_{ij}(0)| + CK\sqrt{N}|x|\kappa(x), \qquad x \in \mathbb{R}^N, \ i, j = 1, \dots, N,$$

which yields (6.1.1).

A straightforward computation now shows that

$$|q_{ij}^{\varepsilon}(x)| \le |q_{ij}(0)| + CK\sqrt{N}(|x| + \varepsilon)\kappa^{\varepsilon}(x), \qquad x \in \mathbb{R}^N, \ i, j = 1, \dots, N,$$

so that the q_{ij}^{ε} 's satisfy the condition (6.1.1) with a constant being independent of $\varepsilon \in (0,1]$.

Similarly, we can show that the b_j^{ε} 's satisfy (6.1.2) with a positive constant being independent of $\varepsilon \leq m$.

Indeed, combining (6.1.6) and (6.1.36), we deduce that

$$\int_{0}^{1} d^{\varepsilon}(tx)dt = \int_{B(\varepsilon)} dy \int_{0}^{1} d(tx - y)\varphi^{\varepsilon}(y)dt$$

$$\leq M \int_{B(\varepsilon)} dy \int_{0}^{1} \kappa(x - y)\varphi^{\varepsilon}(y)dt$$

$$= M\kappa^{\varepsilon}(x),$$

for any $x \in \mathbb{R}^N$ and any $\varepsilon \leq m$. Now, arguing as in the proof of (6.1.6), we get

$$\sum_{j=1}^{N} b_{j}^{\varepsilon}(x)x_{j} \leq \sum_{j=1}^{N} b_{j}^{\varepsilon}(0)x_{j} + |x|^{2} \int_{0}^{1} d^{\varepsilon}(tx)dt$$
$$\leq \sup_{y \in B(\varepsilon)} |b_{j}(y)||x| + M|x|^{2} \kappa^{\varepsilon}(x),$$

for any $x \in \mathbb{R}^N$, which yields (6.1.2) with a constant being independent of $\varepsilon \leq m$.

Now, applying the same arguments as in the proof of Theorem 6.1.7, we can show that, for any $\omega > 0$ and any $k, l \in \mathbb{N}$, $0 \le k \le l \le j$, there exists a positive constant $C_{k,l} = C_{k,l}(\omega)$, independent of ε and n, such that

$$||T_n^{\varepsilon}(t)f||_{C^1(\overline{B}(n))} \le C_{k,l} t^{-\frac{l-k}{2}} e^{\omega t} ||f||_{C^k(\overline{B}(n))}, \qquad t > 0, \ f \in C_b^k(\overline{B}(n)),$$

where $T^{\varepsilon}(t)$ is defined as T(t) with \mathcal{A} being replaced with $\mathcal{A}^{\varepsilon}$. As n tends to $+\infty$, $T_n^{\varepsilon}(t)f$ tends to a solution $u^{\varepsilon} =: T^{\varepsilon}(t)f$ of the Cauchy problem

$$\begin{cases} D_t u(t, x) = \mathcal{A}^{\varepsilon} u(t, x), & t > 0, x \in \mathbb{R}^N, \\ u(0, x) = f(x), & x \in \mathbb{R}^N, \end{cases}$$

which satisfies

$$||T^{\varepsilon}(t)f||_{C^{l}(\mathbb{R}^{N})} \le C_{k,l}t^{-\frac{l-k}{2}}e^{\omega t}||f||_{C^{k}(\mathbb{R}^{N})}, \qquad t > 0.$$
 (6.1.37)

Theorem C.1.4 and the estimate (6.1.37) imply that there exists an infinitesimal sequence $\{\varepsilon_n\}$ such that $T^{\varepsilon_n}(t)f$ and its space derivatives up to the (j-1)-th-order converge in $C^{1+\delta/2,2+\delta}_{\text{loc}}((0,+\infty)\times\mathbb{R}^N)$ to a function u=:S(t)f satisfying

$$||S(t)f||_{C^{l}(\mathbb{R}^{N})} \le C_{k,l} t^{-\frac{l-k}{2}} e^{\omega t} ||f||_{C^{k}(\mathbb{R}^{N})}, \qquad t > 0.$$
 (6.1.38)

Since the coefficients q_{ij} and b_j (i, j = 1, ..., N) are locally Lipschitz continuous, $q_{ij}^{\varepsilon_n}$ and $b_j^{\varepsilon_n}$ converge locally uniformly in \mathbb{R}^N as n tends to $+\infty$, respectively to q_{ij} and b_j , so that S(t)f satisfies the differential equation

 $D_t w - \mathcal{A} w = 0$ in $(0, +\infty) \times \mathbb{R}^N$. Moreover, for any $f \in C_b^2(\mathbb{R}^N)$, S(t)f converges to f as t tends to 0, locally uniformly in x. This can be seen by a localization argument similar to the one used in the proof of Theorem 2.2.1. For this purpose, we fix $k \in \mathbb{N}$ and let $\vartheta = \vartheta_k$ be as in the proof of Theorem 6.1.7. Moreover, for any $n \in \mathbb{N}$, we set $v_n = \vartheta T^{\varepsilon_n}(\cdot)f$ and observe that v_n is a solution to the Cauchy problem

$$\begin{cases} D_t v_n(t,x) = \mathcal{A}^{\varepsilon_n} v_n(t,x) + \psi_n(t,x), & t \in [0,T], x \in B(k), \\ v_n(t,x) = 0, & t \in [0,T], x \in \partial B(k), \\ v_n(0,x) = \vartheta(x) f(x), & x \in \overline{B}(k), \end{cases}$$

where

$$\psi_n(t,\cdot) = -2T^{\varepsilon_n}(t)f \cdot \mathcal{A}^{\varepsilon_n}(\vartheta) - 2\sum_{i,j=1}^N q_{ij}^{\varepsilon_n} D_i T^{\varepsilon_n}(t)f \cdot D_j \vartheta.$$

Since the coefficients q_{ij}^{ε} and b_{j}^{ε} (i, j = 1, ..., N) converge locally uniformly in \mathbb{R}^{N} , they are equibounded (with respect to $n \in \mathbb{N}$) in B(k) and there exists a positive constant \widetilde{C} such that $\|\mathcal{A}^{\varepsilon_{n}}g\|_{C(\overline{B}(k))} \leq \widetilde{C}\|g\|_{C^{2}(\overline{B}(k))}$ for any $g \in C^{2}(\overline{B}(k))$. Therefore, from (6.1.37) we deduce that there exists a constant \overline{C} , independent of n, such that

$$|\psi_n(t,x)| \le \overline{C}t^{-\frac{1}{2}} ||f||_{\infty}, \qquad t \in (0,T], \quad x \in B(k), \quad n \in \mathbb{N}.$$
 (6.1.39)

The estimate (6.1.39) implies that v_n can be written by the usual variation-of-constants formula as

$$v_n(t,\cdot) = T_{k,n}(t)(\vartheta f) + \int_0^t T_{k,n}(t-s)\psi_n(s,\cdot)ds, \qquad t \in (0,T],$$

where $\{T_{k,n}(t)\}$ is the semigroup associated with the realization $A_{k,n}$ of $\mathcal{A}^{\varepsilon_n}$ in $C(\overline{B}(k))$ with homogeneous Dirichlet boundary conditions (see Theorem C.3.6(iv)).

Since $\vartheta f \in D(A_{k,n}) = \{u \in \bigcap_{1 \leq p < +\infty} W^{2,p}(B(k)) : \mathcal{A}^{\varepsilon_n} u \in C(\overline{B}(k))\}, \text{ the domain of the realization of } \mathcal{A}^{\varepsilon_n} \text{ in } C(\overline{B}(k)) \text{ (see Theorem C.3.6) and } \{T_{k,n}\}$ is a semigroup of contractions in $C(\overline{B}(k))$ for any $n \in \mathbb{N}$, then

$$\begin{split} \|T_{k,n}(t)(\vartheta f) - \vartheta f\|_{\overline{C}(B(k))} &= \left\| \int_0^t T_{k,n}(s) \mathcal{A}^{\varepsilon_n}(\vartheta f) ds \right\|_{C(\overline{B}(k))} \\ &\leq t \|\mathcal{A}^{\varepsilon_n}(\vartheta f)\|_{C(\overline{B}(k))} \leq \overline{C} t \|\vartheta f\|_{C^2(\overline{B}(k))}, \end{split}$$

for any $t \in (0,T)$, which readily yields

$$\lim_{t \to 0^+} \sup_{n \in \mathbb{N}} \|T_{k,n}(t)(\vartheta f) - \vartheta f\|_{C(\overline{B}(k))} = 0. \tag{6.1.40}$$

From (6.1.39) and (6.1.40) we can now easily show that

$$\lim_{t \to 0^+} \sup_{n \in \mathbb{N}} \|v_n(t, \cdot) - \vartheta f\|_{C(\overline{B}(k))} = 0.$$

Recalling that $\vartheta \equiv 1$ in B(k/2), we deduce that

$$|(S(t)f)(x) - f(x)|$$

$$\leq \sup_{n \in \mathbb{N}} |(T^{\varepsilon_n}(t)f)(x) - \vartheta(x)f(x)| + \limsup_{n \to +\infty} |(T^{\varepsilon_n}(t)f)(x) - (S(t)f)(x)|$$

$$\leq \sup_{n \in \mathbb{N}} ||v_k(t, \cdot) - \vartheta f||_{C(\overline{B}(k))}, \tag{6.1.41}$$

for any $t \in (0,T)$ and any $x \in B(k/2)$. Taking the limit as t tends to 0 in both the first and the last side of (6.1.41) gives

$$\lim_{t \to 0^+} \sup_{x \in B(k/2)} |u(t, x) - f(x)| = 0.$$

From the arbitrariness of $k \in \mathbb{N}$, we deduce that S(t)f converges locally uniformly (with respect to x) to f as t tends to 0. Hence S(t)f is a classical solution to the problem

$$\begin{cases} D_t w(t, x) = \mathcal{A} w(t, x), \ t > 0, \ x \in \mathbb{R}^N, \\ w(0, x) = f(x), & x \in \mathbb{R}^N, \end{cases}$$

and Hypothesis 6.1.1(ii) and the maximum principle in Theorem 4.1.3 imply that T(t)f = S(t)f. Since $C_b^2(\mathbb{R}^N)$ is dense in $C_b^1(\mathbb{R}^N)$ (which is endowed with the sup-norm) we can extend the previous equality to all the $f \in C_b^1(\mathbb{R}^N)$, obtaining (6.0.1) in the case when $k \geq 1$.

Finally, with any $f \in C(\mathbb{R}^N)$ and any t > 0 we write T(t) = T(t/2)T(t/2)f and observe that since $T(t/2)f \in C_b^1(\mathbb{R}^N)$, then T(t)f = S(t/2)T(t/2)f. Applying (6.1.38) with (t, f) being replaced with (t/2, T(t/2)f), we easily get (6.0.1) also in the case when k = 0.

Remark 6.1.10 In Theorem 6.1.7 we have shown that the estimate (6.1.11) holds true for some $\omega \geq 0$. In Chapter 7, we will show that, actually, we can take $\omega = 0$ in (6.1.11) (and, consequently, in (6.1.33)) and that, under somewhat heavier assumptions on the coefficients, we can also take $\omega < 0$.

To conclude this section we show that the estimates (6.0.1) may fail to hold without any dissipativity assumption.

Example 6.1.11 Consider the following one-dimensional operator

$$Au(x) = u''(x) + p'(x)u'(x) = e^{-p(x)} \left(e^{p(x)}u'(x)\right)', \quad x \in \mathbb{R},$$

where $p \in C^1(\mathbb{R})$ is a function that will be chosen later on. For any $f \in C_b(\mathbb{R})$, the solutions of the equation Au = f are given by

$$u(x) = C_1 + \int_0^x e^{-p(t)} \left(C_2 + \int_0^t f(s)e^{p(s)} ds \right) dt, \tag{6.1.42}$$

where $C_1, C_2 \in \mathbb{R}$ are arbitrary real constants. Suppose now that $e^p \in L^1(\mathbb{R})$ and that

$$\int_{-\infty}^{+\infty} f(t)e^{p(t)}dt = 0.$$
 (6.1.43)

Choose

$$C_2 = -\int_0^{+\infty} f(t)e^{p(t)}dt = \int_{-\infty}^0 f(t)e^{p(t)}dt.$$

If x > 0 then, from (6.1.42), we formally get

$$u(x) = C_1 - \int_0^x e^{-p(t)} \int_t^{+\infty} f(s)e^{p(s)} ds dt$$
$$= C_1 - \int_0^{+\infty} e^{p(s)} f(s) \int_0^{s \wedge x} e^{-p(t)} dt ds.$$

Suppose now that the function

$$Q(s) = e^{p(s)} \int_0^s e^{-p(t)} dt, \quad s \in \mathbb{R}$$

belongs to $L^1(0,+\infty)$. Then, we get the estimate

$$|u(x)| \le |C_1| + ||f||_{\infty} \int_0^{+\infty} Q(s)ds, \quad x > 0,$$

which implies that $u \in C_b([0, +\infty))$. Similarly, assuming that $Q \in L^1(-\infty, 0)$ we get $u \in C_b((-\infty, 0])$.

Therefore, under the assumption $Q \in L^1(\mathbb{R})$ we conclude that $u \in D_{\max}(\mathcal{A})$. Note that the assumption $Q \in L^1(\mathbb{R})$ implies $e^p \in L^1(\mathbb{R})$. The derivative of u is given by

$$u'(x) = -e^{-p(x)} \int_{x}^{+\infty} f(s)e^{p(s)}ds, \qquad x \in \mathbb{R}.$$

We claim that we can choose the functions p and f so that $Q \in L^1(\mathbb{R})$ and (6.1.43) holds but u' is not bounded. Indeed take

$$p(x) = -x^4 + \log h(x),$$

where h(x) is a regular function satisfying

$$h(x) = \varepsilon_n,$$
 if $x = n - \frac{\delta_n}{2}, n \in \mathbb{N},$
 $\varepsilon_n \le h(x) \le 1,$ if $n - \delta_n < x < n, n \in \mathbb{N},$
 $h(x) = 1,$ otherwise,

where

$$\varepsilon_n = \frac{1}{n} e^{\left(n - \frac{1}{2}\right)^4} - \left(n + \frac{1}{2}\right)^4, \qquad \delta_n = \frac{e^{-n^4}}{n^2} \varepsilon_n$$

Then, we have

$$Q(x) = e^{-x^4} \int_0^x e^{t^4} dt, \quad x < 0, \qquad Q(x) = h(x)e^{-x^4} \int_0^x \frac{e^{t^4}}{h(t)} dt, \quad x > 0.$$

Applying De L'Hôpital rule one sees that $\lim_{x\to-\infty} x^3 Q(x) = 1/4$ and conclude that $Q\in L^1(-\infty,0)$. For any x>0 we have

$$\begin{split} Q(x) & \leq e^{-x^4} \int_0^x \frac{e^{t^4}}{h(t)} dt \\ & \leq e^{-x^4} \int_0^x e^{t^4} dt + e^{-x^4} \sum_{n=1}^{[x]+1} \int_{n-\delta_n}^n \frac{e^{n^4}}{\varepsilon_n} dt \\ & \leq e^{-x^4} \int_0^x e^{t^4} dt + e^{-x^4} \sum_{n=1}^\infty \frac{\delta_n e^{n^4}}{\varepsilon_n} \\ & = e^{-x^4} \int_0^x e^{t^4} dt + e^{-x^4} \sum_{n=1}^\infty \frac{1}{n^2}, \end{split}$$

which shows that $Q \in L^1(0, +\infty)$. Now we choose a function $f \in C_b(\mathbb{R})$ satisfying (6.1.43) and such that f(x) = 1 for any x > 0. Then, for any x > 0 we have

$$u'(x) = -\frac{e^{x^4}}{h(x)} \int_x^{+\infty} h(t)e^{-t^4}dt,$$

for any x > 0, so that at $x = n - \delta_n/2$

$$|u'(n-\delta_n/2)| = \frac{e^{(n-\delta_n/2)^4}}{\varepsilon_n} \int_{n-\frac{\delta_n}{2}}^{+\infty} h(t)e^{-t^4}dt$$

$$\geq \frac{e^{(n-\frac{1}{2})^4}}{\varepsilon_n} \int_{n}^{n+\frac{1}{2}} e^{-t^4}dt$$

$$\geq \frac{e^{(n-\frac{1}{2})^4}}{\varepsilon_n} \frac{1}{2} e^{-(n+\frac{1}{2})^4}$$

$$= \frac{n}{2},$$

which shows that u'(x) is unbounded.

Therefore, the function u belongs to $D_{\max}(A)$ but not to $C_b^1(\mathbb{R})$. This means that (6.0.1) with k = 0, l = 1 fails to hold.

We note that in this case the dissipativity assumption (6.1.3) fails since p'' is unbounded from above. Indeed, let $g: \mathbb{R} \to \mathbb{R}$ be the function defined by $g(x) = \log(h(x))$ for any $x \in \mathbb{R}$. Since $g(n - \delta_n) = g(n) = 0$ and $g(n - \delta_n/2) = \log(\varepsilon_n)$ for any $n \in \mathbb{N}$, then, by the mean value theorem, there exist two points $y_n \in (n - \delta_n, n - \delta_n/2)$ and $z_n \in (n - \delta_n/2, n)$ such that

$$g'(y_n) = -g'(z_n) = \frac{2\log(\varepsilon_n)}{\delta_n}, \quad n \in \mathbb{N}.$$

Applying again the mean value theorem, it follows that there exists $x_n \in (y_n, z_n)$ such that

$$g''(x_n) = -\frac{4\log(\varepsilon_n)}{\delta_n(z_n - y_n)} \ge -\frac{4\log(\varepsilon_n)}{\delta_n^2} = -\frac{4\log(\varepsilon_n)}{\varepsilon_n^2} n^4 e^{2n^4}, \qquad n \in \mathbb{N}.$$

Since ε_n tends to 0 as n tends to $+\infty$, then, for n large enough it holds that

$$p''(x_n) = -12x_n^2 + g''(x_n) \ge -12n^2 + n^4 e^{2n^4},$$

which shows that p'' is unbounded from above.

Remark 6.1.12 As already recalled in the introduction to this chapter, in the paper [29] and in the recent book [30], the author, starting from the stochastic equation

$$\begin{cases} d\xi_t^x = b(\xi_t^x)dt + \sigma(\xi_t^x)dW_t, & t > 0, \\ \xi_0 \equiv x, & \end{cases}$$

and using probabilistic methods, proves uniform estimates, comparable with ours, for the space derivatives of the function T(t)f when $f \in C_b(\mathbb{R}^N)$ and $\{T(t)\}$ is the semigroup associated with the stochastic differential equation.

More precisely, she assumes the following set of assumptions:

(i) $\sigma_{ij} \in C^3(\mathbb{R}^N)$ and there exists $k \geq 0$ such that for any $j = 0, \dots, 3$ it holds

$$\sup_{x \in \mathbb{R}^N} \frac{\|D^{\beta}\sigma(x)\|}{1 + |x|^{k-j}} < +\infty, \qquad |\beta| = j.$$

(ii) $b_i \in C^3(\mathbb{R}^N)$ and there exists $m \geq k$ such that for any $j = 0, \dots, 3$ it holds

$$\sup_{x \in \mathbb{R}^N} \frac{|D^{\beta}b(x)|}{1 + |x|^{2m+1-j}} < +\infty, \qquad |\beta| = j.$$

There exist $a, \gamma > 0$ and $c \in \mathbb{R}$ such that for any $x, h \in \mathbb{R}^N$ it holds

$$\langle b(x+h) - b(x), h \rangle \le -a|h|^{2m+2} + c(1+|x|^{\gamma}).$$

(iii) For any p > 0 there exists $c_p \in \mathbb{R}$ such that

$$\langle b(x) - b(y), x - y \rangle + p \|\sigma(x) - \sigma(y)\|_2^2 \le c_p |x - y|^2.$$

Under the previous set of assumptions, she proves that $T(t)f \in C_b^3(\mathbb{R}^N)$ for any t > 0 and any $f \in B_b(\mathbb{R}^N)$ and that, for any T > 0, there exists a constant $C_T > 0$ such that

$$||D^k T(t)f||_{\infty} \le C_T t^{-\frac{k}{2}} ||f||_{\infty}, \quad t \in (0, T], \ k = 1, 2, 3.$$

Note that also our estimates can be extended to any $f \in B_b(\mathbb{R}^N)$. Indeed, since $\{T(t)\}$ is a semigroup of contractions in $B_b(\mathbb{R}^N)$ and it is also strong Feller (see Remark 2.2.10 and Proposition 2.2.12), then we can split T(t)f = T(t/2)T(t/2)f for any $f \in B_b(\mathbb{R}^N)$ and write, for any $\omega > 0$ and some positive $C = C(\omega)$

$$||D^{k}T(t)f||_{\infty} \leq ||D^{k}T(t/2)||_{L(C_{b}(\mathbb{R}^{N});C_{b}^{k}(\mathbb{R}^{N}))}||T(t/2)f||_{C_{b}(\mathbb{R}^{N})}$$

$$\leq Ce^{\frac{\omega}{2}t}t^{-\frac{k}{2}}||f||_{\infty}, \tag{6.1.44}$$

for any t > 0 and any k = 1, 2, 3. Actually, in view of the results in Section 7.2 (see Remark 7.2.4), we can get rid of the exponential term in (6.1.44).

Remark 6.1.13 As it has been remarked in the introduction to this chapter, uniform estimates similar to those in Theorem 6.1.7 have been obtained in [108] under assumptions on the coefficients of the operator $\mathcal A$ which are comparable with ours. The main differences are in the method used in [108] to prove such uniform estimates. Indeed, A. Lunardi approximates the coefficients of the operator $\mathcal A$ rather than the whole space $\mathbb R^N$ by balls, centered at the origin, as we do here. Moreover, she does not assume that the potential c identically vanishes in $\mathbb R^N$. Here, we assume such an assumption in view of the pointwise estimates of Chapter 7.

6.2 Some consequences

The estimates (6.1.33) can be used to prove optimal Schauder estimates for the elliptic equation

$$\lambda u - \mathcal{A}u = f,\tag{6.2.1}$$

as well as for the Cauchy problem

$$\begin{cases}
D_t u(t,x) = \mathcal{A}u(t,x) + g(t,x), & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = u_0(x), & x \in \mathbb{R}^N.
\end{cases}$$
(6.2.2)

We begin with the elliptic equation and we prove the following lemma.

Lemma 6.2.1 Let $\theta \in (0,3)$ be not an integer, let $I \subset \mathbb{R}$ be an interval, and let $\varphi : I \to C_b^{\theta}(\mathbb{R}^N)$ be such that, for any $x \in \mathbb{R}^N$, the real function $t \mapsto \varphi(t)(x)$ is continuous in I and

$$\|\varphi(t)\|_{C_b^{\theta}(\mathbb{R}^N)} \le c(t), \qquad t \in I,$$

for some function $c \in L^1(I)$. Then, the function

$$f(x) = \int_{I} \varphi(t)(x)dt, \qquad x \in \mathbb{R}^{N}, \tag{6.2.3}$$

belongs to $C_b^{\theta}(\mathbb{R}^N)$ and

$$||f||_{C_{\iota}^{\theta}(\mathbb{R}^{N})} \leq K||c||_{L^{1}(I)},$$

for some positive constant K.

Proof. We begin the proof observing that, according to [141, Section 2.7.2], for any $\theta \in (0,3)$, $C_b^{\theta}(\mathbb{R}^N)$ can be characterized as the space of functions $g \in C_b(\mathbb{R}^N)$ such that

$$[[g]]_{\theta} = \sup_{\substack{x,h \in \mathbb{R}^N \\ h \neq 0}} |h|^{-\theta} \left| \sum_{l=0}^{3} (-1)^l f(x+lh) \right| < +\infty$$

and the norm

$$g \mapsto \|g\|_{\infty} + [[g]]_{\theta}$$

is equivalent to the classical norm of $C_b^{\theta}(\mathbb{R}^N)$. Hence, if φ is as in the statement of the lemma, then for any $x,h\in\mathbb{R}^N$, with $h\neq 0$, we have

$$\left|\sum_{l=0}^{3} (-1)^l \int_I \varphi(t)(x+lh) dt\right| \leq \int_I \left|\sum_{l=0}^{3} (-1)^l \varphi(t)(x+lh)\right| dt \leq K|h|^{\theta} \int_I c(t) dt, \tag{6.2.4}$$

for some positive constant K. The estimate (6.2.4) implies that the function f in (6.2.3) belongs to $C_b^{\theta}(\mathbb{R}^N)$, and the statement follows.

Now, we observe that by virtue of Hypothesis 6.1.1(ii), Theorem 4.1.5 implies that, for any $f \in C_b(\mathbb{R}^N)$ and any $\lambda > c_0$, the function $R(\lambda)f$ defined by

$$(R(\lambda)f)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x)dt, \qquad x \in \mathbb{R}^N,$$
 (6.2.5)

is the unique solution to the elliptic equation (6.2.1) in

$$D_{\max}(\mathcal{A}) = \left\{ u \in C_b(\mathbb{R}^N) \cap \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2,p}(\mathbb{R}^N) : \ \mathcal{A}u \in C_b(\mathbb{R}^N) \right\}, \ Au = \mathcal{A}u.$$

As a consequence, Proposition 2.3.6 implies that $(A, D_{\text{max}}(A)) = (\widehat{A}, D(\widehat{A}))$ with equivalence of the corresponding norms.

With this remark in mind, we can now prove the following theorem, which provides us a partial characterization of $D_{\text{max}}(\mathcal{A})$ and gives optimal Schauder estimates for the solution of the elliptic equation (6.2.1).

Theorem 6.2.2 Suppose that Hypotheses 6.1.1(i)-6.1.1(iii), 6.1.1(iv-l) are satisfied (for some l=1,2). Then, $D_{\max}(\mathcal{A})$ is contained in $C_b^{\theta}(\mathbb{R}^N)$, for any $\theta \in (0,1]$, if l=1, and, for any $\theta \in (0,2)$, if l=2. Moreover, for any $\omega > 0$ and any $\theta \in (0,l)$, and also for $\theta = 1$ if l=1, there exists a positive constant $C = C(\theta,\omega)$ such that

$$||u||_{C_h^{\theta}(\mathbb{R}^N)} \le C||u||_{\infty}^{1-\frac{\theta}{2}}||(\omega - \mathcal{A})u||_{\infty}^{\frac{\theta}{2}}, \quad u \in D_{\max}(\mathcal{A}).$$
 (6.2.6)

Finally, let the previous assumptions be satisfied with l=3. Then, for any $f \in C_b^{\theta}(\mathbb{R}^N)$ ($\theta \in (0,1)$) and any $\lambda > 0$, there exist a unique solution $u \in C_b^{2+\theta}(\mathbb{R}^N)$ of the elliptic equation $\lambda u - \mathcal{A}u = f$ and a positive constant $C = C(\theta, \lambda)$ such that

$$||u||_{C_b^{2+\theta}(\mathbb{R}^N)} \le C||f||_{C_b^{\theta}(\mathbb{R}^N)}.$$
 (6.2.7)

Proof. To prove the first part we assume that l=3, the other case being completely similar and even simpler. Let $u \in D_{\max}(\mathcal{A})$ be a nonidentically vanishing function and fix $\omega > 0$. For any $\lambda > \omega$, set $\varphi = \lambda u - \mathcal{A}u$. Then,

$$u(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)\varphi)(x) dt, \qquad x \in \mathbb{R}^N,$$

so that, according to Lemma 6.2.1 and the estimate (6.1.33), $f \in C_b^{\theta}(\mathbb{R}^N)$ and, if $\theta \neq 1$, then

$$\begin{split} \|u\|_{C_b^{\theta}(\mathbb{R}^N)} &\leq C \frac{\Gamma(1 - \theta/2)}{(\lambda - \omega)^{1 - \theta/2}} \|\varphi\|_{\infty} \\ &\leq C \frac{\Gamma(1 - \theta/2)}{(\lambda - \omega)^{1 - \theta/2}} \left((\lambda - \omega) \|u\|_{\infty} + \|(\omega - \mathcal{A})u\|_{\infty} \right), \quad (6.2.8) \end{split}$$

for some positive constant C, independent of λ , where Γ denotes the Gamma function. Taking the minimum for $\lambda > \omega$ in (6.2.8), the estimate (6.2.6) follows.

Now, we prove that for any $f \in C_b^{\theta}(\mathbb{R}^N)$ $(\theta \in (0,1))$ and any $\lambda > 0$, there exists a (unique) solution $u \in C_b^{2+\theta}(\mathbb{R}^N)$ of the equation (6.2.1). So we assume that Hypothesis 6.1.1(iv-3) is satisfied. By virtue of Theorem 4.1.5, to prove the assertion, it suffices to show that the function $u = R(\lambda)f$ has the claimed regularity properties.

We set

$$u(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)\varphi)(x) dt, \qquad x \in \mathbb{R}^N,$$

and prove that, for any $\alpha \in (\alpha, 1)$,

$$u \in (C_b^{\alpha}(\mathbb{R}^N), C_b^{2+\alpha}(\mathbb{R}^N))_{1-(\alpha-\theta)/2, +\infty}$$

This is enough for our aims. Indeed, by a well known result in interpolation theory (see Theorem A.4.8) it holds that

$$(C_b^{\alpha}(\mathbb{R}^N), C_b^{2+\alpha}(\mathbb{R}^N))_{1-(\alpha-\theta)/2, +\infty} = C_b^{2+\theta}(\mathbb{R}^N).$$

In order to show that $u \in (C_b^{\alpha}(\mathbb{R}^N), C_b^{2+\alpha}(\mathbb{R}^N))_{1-(\alpha-\theta)/2,+\infty}$, we apply the same arguments as in [105]. For this purpose, we split, for any $\xi > 0$, $u = a_{\xi} + b_{\xi}$, where

$$a_{\xi}(x) = \int_0^{\xi} e^{-\lambda t} (T(t)f)(x)dt, \qquad b_{\xi}(x) = \int_{\xi}^{+\infty} e^{-\lambda t} (T(t)f)(x)dt,$$

for any $x \in \mathbb{R}^N$. From Lemma 6.2.1 and the estimate (6.1.33), where we take $\omega = \lambda/2$, we deduce that

$$||a_{\xi}||_{C_b^{\alpha}(\mathbb{R}^N)} \le C||f||_{C_b^{\theta}(\mathbb{R}^N)} \int_0^{\xi} t^{-\frac{\alpha-\theta}{2}} dt = C' \xi^{1-\frac{\alpha-\theta}{2}} ||f||_{C_b^{\theta}(\mathbb{R}^N)},$$

$$\|b_{\xi}\|_{C^{2+\alpha}_b(\mathbb{R}^N)} \leq C \|f\|_{C^{\theta}_b(\mathbb{R}^N)} \int_{\xi}^{+\infty} t^{-1-\frac{\alpha-\theta}{2}} dt = C'' \xi^{-\frac{\alpha-\theta}{2}} \|f\|_{C^{\theta}_b(\mathbb{R}^N)},$$

for some positive constants C, C', C'', independent of ξ . Hence,

$$\xi^{-1+\frac{\alpha-\theta}{2}}\left(\|a_\xi\|_{C^{\alpha}_b(\mathbb{R}^N)} + \xi\|b_\xi\|_{C^{2+\alpha}_b(\mathbb{R}^N)}\right) \leq C'\|f\|_{C^{\theta}_b(\mathbb{R}^N)} + C''\|f\|_{C^{\theta}_b(\mathbb{R}^N)},$$

for any $\xi > 0$. This implies that (see Definition A.4.1)

$$\sup_{\xi \in (0,1)} \xi^{-1 + \frac{\alpha - \theta}{2}} K(\xi, u) \le (C' + C'') \|\varphi\|_{C_b^{\theta}(\mathbb{R}^N)},$$

implying that $u \in (C_b^{\alpha}(\mathbb{R}^N), C_b^{2+\alpha}(\mathbb{R}^N))_{1-(\alpha-\theta)/2,\infty}$ and, consequently, that (6.2.7) holds true.

Remark 6.2.3 Actually, the results in Chapter 7 will show that, in some situations, (6.1.11) and, consequently, (6.1.33) hold with an exponential term of negative type (see Remark 7.2.4). In such a situation, the same arguments as in the proof of Theorem 6.2.2 show that we can take $\omega = 0$ in (6.2.6).

As far as the Cauchy problem (6.2.2) is concerned, we can prove two results which provide us a (unique) solution u to such a problem and give sharp estimates for its space derivatives. As in the classical case, we give the following definition.

Definition 6.2.4 The function $u:[0,+\infty)\times\mathbb{R}^N\to\mathbb{R}$ given by

$$u(t,x) = (T(t)u_0)(x) + \int_0^t (T(t-s)g(s,\cdot))(x)ds, \qquad t > 0, \ x \in \mathbb{R}^N, \ (6.2.9)$$

is called mild solution to the problem (6.2.2).

Throughout the rest of this section we always assume that Hypotheses 6.1.1(i)–6.1.1(iii) and 6.1.1(iv-3) are satisfied.

Proposition 6.2.5 Fix $\alpha, \theta \in (0,1)$, $\theta < \beta \leq 2 + \theta$. Moreover, let $u_0 \in C_b(\mathbb{R}^N)$ and let g be a continuous function in $(0,T] \times \mathbb{R}^N$ such that $g(t,\cdot) \in C_b^{\beta}(\mathbb{R}^N)$ for any $t \in (0,T]$ and

$$\sup_{0 < t \le T} t^{\alpha} \|g(t, \cdot)\|_{C_b^{\beta}(\mathbb{R}^N)} < +\infty.$$

Then, the function u defined in (6.2.9) belongs to $C([0,T] \times \mathbb{R}^N) \cap C^{1,2}((0,T] \times \mathbb{R}^N)$. Moreover, u is the unique bounded classical solution to the problem (6.2.2). Finally, $u(t,\cdot) \in C_b^{2+\theta}(\mathbb{R}^N)$ for any $t \in (0,T]$ and there exists a positive constant C > 0, independent of u, such that

$$||u(t,\cdot)||_{\infty} + t^{1+\frac{\theta}{2}} ||u(t,\cdot)||_{C_b^{2+\theta}(\mathbb{R}^N)} \le C \left(||u_0||_{\infty} + \sup_{0 < t \le T} t^{\alpha} ||g(t,\cdot)||_{C_b^{\beta}(\mathbb{R}^N)} \right),$$

for any $t \in [0, T]$.

Proof. The uniqueness of the bounded classical solution to the problem (6.2.9) is a straightforward consequence of Hypothesis 6.1.1(ii) and Theorem 4.1.3. So, let us prove that the function u in (6.2.9) is actually a bounded classical solution to the problem (6.2.2).

Throughout the proof, we denote by C positive constants, independent of s and t, which may vary from line to line.

To begin with, we will deal with the function defined by the integral term in (6.2.9). We will denote it by v.

As a first step, we show that v is well defined and it is continuous in $[0,T] \times \mathbb{R}^N$. For this purpose, we observe that the function $(r,s,x) \mapsto (T(r)g(s,\cdot))(x)$ is continuous in $[0,T] \times (0,T] \times \mathbb{R}^N$. To see it, it suffices to observe that, for any $s \in [0,T]$, the function $(r,x) \mapsto (T(r)g(s,\cdot))(x)$ is continuous in $[0,T] \times \mathbb{R}^N$, and, by virtue of Proposition 2.2.9, for any $r \in [0,T]$ and any $x \in \mathbb{R}^N$, the function $s \mapsto (T(r)g(s,\cdot))(x)$ is continuous in [0,T], uniformly with respect to r and x on compact subsets of \mathbb{R}^N . Moreover,

$$|T(r)(g(s,\cdot))(x)| \le ||g(s,\cdot)||_{\infty} \le Cs^{-\alpha}, \qquad s \in (0,T], \quad x \in \mathbb{R}^N$$

for some C > 0. This is enough to conclude that v is continuous in $[0, T] \times \mathbb{R}^N$.

Let us now prove that $v(t,\cdot) \in C_b^{2+\theta}(\mathbb{R}^N)$ for any $t \in (0,T]$. By the estimate (6.1.33) we have

$$||T(t-s)||_{L(C_h^{\beta}(\mathbb{R}^N), C_h^{2+\theta}(\mathbb{R}^N))} \le C(t-s)^{1-\frac{\beta-\theta}{2}},$$

for any $0 \le s < t \le T$. Therefore,

$$||T(t-s)g(s)||_{C_b^{2+\theta}(\mathbb{R}^N)} \le \frac{C}{s^{\alpha}(t-s)^{1-(\beta-\theta)/2}} \sup_{0 < t < T} t^{\alpha} ||g(t,\cdot)||_{C_b^{\beta}(\mathbb{R}^N)}.$$

Since $s\mapsto (s^{\alpha}(t-s)^{1-(\beta-\alpha)/2})^{-1}$ is in $L^1(0,t)$, by Lemma 6.2.1, we have $v(t,\cdot)\in C_b^{2+\theta}(\mathbb{R}^N)$ for any $t\in(0,T]$, and

$$||v(t,\cdot)||_{C_b^{2+\theta}(\mathbb{R}^N)} \le C \left(\int_0^t \frac{1}{s^{\alpha}(t-s)^{1-(\beta-\theta)/2}} ds \right) \sup_{0 < t \le T} t^{\alpha} ||g(t,\cdot)||_{C_b^{\beta}(\mathbb{R}^N)}$$

$$= C' t^{-\alpha + \frac{\beta-\theta}{2}} \sup_{0 < t \le T} t^{\alpha} ||g(t,\cdot)||_{C_b^{\beta}(\mathbb{R}^N)}. \tag{6.2.10}$$

Therefore, the function $t \mapsto v(t, \cdot)$ is bounded in $[\varepsilon, T]$ with values in $C^{2+\theta}(K)$ for any compact set $K \subset \mathbb{R}^N$ and any $\varepsilon \in (0, T)$. Since v is continuous, then it belongs to C([0, T]; C(K)). By Propositions A.4.4 and A.4.6(i), the map v belongs to $C([\varepsilon, T] \times C^2(K))$. Therefore, the first- and second-order derivatives of v are continuous in $[\varepsilon, T] \times K$ and, hence, in $(0, T] \times \mathbb{R}^N$.

Let us now consider the regularity of v with respect to t. For this purpose we observe that, for any t>0, any $x\in\mathbb{R}^N$ and any h>0, sufficiently close to 0, one has

$$\frac{v(t+h,x) - v(t,x)}{h} = \frac{1}{h} \int_{t}^{t+h} (T(t+h-s)g(s,\cdot))(x)ds + \frac{1}{h} \int_{0}^{t} ((T(t+h-s) - T(t-s))g(s,\cdot))(x)ds.$$
(6.2.11)

As it is easily seen, the first integral term in the right-hand side of (6.2.11) tends to g(t,x) as h tends to 0^+ . As far as the second term is concerned, we observe that, taking Theorem 2.2.1 into account, we can show that, for any $s \in (0,T]$ and any $x \in \mathbb{R}^N$, the function $t \mapsto (T(t-s)g(s,\cdot))(x)$ is continuously differentiable in (s,T] and

$$D_t(T(t-s)g(s,\cdot))(x) = (\mathcal{A}T(t-s)g(s,\cdot))(x).$$

Therefore,

$$((T(t+h-s)-T(t-s))g(s,\cdot))(x) = \int_0^h (D_t T(t-s+r)g(s,\cdot))(x)dr$$
$$= \int_0^h (\mathcal{A}T(t-s+r)g(s,\cdot))(x)dr.$$

Now, using (6.1.33), we can show that

$$\begin{split} |(\mathcal{A}T(t+r-s)g(s,\cdot))(x)| &\leq \frac{C}{s^{\alpha}}(t+r-s)^{\min\left(\frac{\beta}{2}-1,0\right)}\sup_{0 < t \leq T}t^{\alpha}\|g(t,\cdot)\|_{C^{\beta}_{b}(\mathbb{R}^{N})}, \\ &\leq \frac{C}{s^{\alpha}}(t-s)^{\min\left(\frac{\beta}{2}-1,0\right)}\sup_{0 < t \leq T}t^{\alpha}\|g(t,\cdot)\|_{C^{\beta}_{b}(\mathbb{R}^{N})}, \end{split}$$

for any $t \in (s,T]$, any $x \in B(n)$, any $r \in (0,h)$ and some positive constant C = C(n,T). Hence,

$$|((T(t+h-s)-T(t-s))g(s,\cdot))(x)| \le \frac{C}{s^{\alpha}}(t-s)^{\min(\frac{\beta}{2}-1,0)}|h| \sup_{0 < t \le T} t^{\alpha}||g(t,\cdot)||_{C_b^{\beta}(\mathbb{R}^N)}$$
(6.2.12)

for any s,t,h as above. From (6.2.12) and the dominated convergence theorem, we obtain that the second term in the right-hand side of (6.2.11) tends to $\mathcal{A}v(t,x)$ as h tends to 0^+ , for any t>0 and any $x\in\mathbb{R}^N$. Therefore, $v(\cdot,x)$ is differentiable from the right in (0,T) for any $x\in\mathbb{R}^N$, and

$$D_t^+ v(t,x) = \int_0^t (\mathcal{A}T(t-s)g(s,\cdot))(x)ds + g(t,x), \qquad t \in (0,T) \times \mathbb{R}^N.$$

Let us now prove that

$$\int_0^t (\mathcal{A}T(t-s)g(s,\cdot))(x)ds = \mathcal{A}\int_0^t (T(t-s)g(s,\cdot))(x)ds, \tag{6.2.13}$$

for any $t \in (0,T]$ and any $x \in \mathbb{R}^N$. This will imply that $D_t v^+ = \mathcal{A}v + g$. But, since we have already shown that the functions $D^{\gamma}v$ ($|\gamma| = 1,2$) are continuous in $(0,T] \times \mathbb{R}^N$, it follows that $\mathcal{A}v$ is continuous in $(0,T] \times \mathbb{R}^N$ as well. Therefore, $D_t v$ exists in $(0,T] \times \mathbb{R}^N$ and it is therein continuous. Hence, v is differentiable with respect to the time variable in $(0,T] \times \mathbb{R}^N$ and $D_t v = \mathcal{A}v + g$.

We begin by proving (6.2.13). For this purpose, we observe that the function $s\mapsto D_x^\gamma(T(t-s)g(s,\cdot))_{\mid K}$ belongs to $C((0,T);C(K))\cap L^1((0,T);C(K))$ for any compact set $K\subset\mathbb{R}^N$, any multi-index γ with $|\gamma|=1,2$ and any $t\in(0,T]$. Since the realization of the derivative D_x^γ in C(K) is a closed operator, it follows that

$$D_x^{\gamma} \int_0^t (T(t-s)g(s,\cdot))(x)ds = \int_0^t (D_x^{\gamma}T(t-s)g(s,\cdot))(x)ds,$$

for any $t \in (0,T]$ and any $x \in K$. Hence, the formula (6.2.13) immediately follows.

To conclude the proof, let us deal with the function $T(\cdot)u_0$. According to Theorem 2.2.1 we already know that such a function belongs to $C^{1,2}((0,+\infty)\times$

 \mathbb{R}^N) $\cap C_b([0,+\infty) \times \mathbb{R}^N)$ and $D_tT(t)u_0 = \mathcal{A}T(t)u_0$ for any $t \in (0,+\infty)$. We now observe that, by virtue of (6.1.33), the function $t \mapsto t^{1+\theta/2}T(t)u_0$ is bounded with values in $C_b^{2+\theta}(\mathbb{R}^N)$. Hence, the function

$$u(t,x) = (T(t)u_0)(x) + v(t,x), t \in [0,T], x \in \mathbb{R}^N,$$

belongs to $C^{1,2}((0,T]\times\mathbb{R}^N)$, it satisfies (6.2.2), and

$$\sup_{0 < t \le T} t^{1 + \frac{\theta}{2}} \|u(t, \cdot)\|_{C_b^{2 + \theta}(\mathbb{R}^N)} \le C \left(\|u_0\|_{\infty} + \sup_{0 < t \le T} t^{\alpha} \|g(t, \cdot)\|_{C_b^{\beta}(\mathbb{R}^N)} \right).$$

If, in addition, g is bounded with values in $C_b^{2+\theta}(\mathbb{R}^N)$, then $v(t,\cdot)$ is bounded up to t=0 with values in $C_b^{2+\theta}(\mathbb{R}^N)$ since, by (6.1.33),

$$||T(t)||_{L(C_b^{\beta}(\mathbb{R}^N); C_b^{2+\theta}(\mathbb{R}^N))} \le C(t-s)^{-1+\frac{\beta-\theta}{2}},$$

and the right-hand side of the previous inequality defines a function in $L^1(s,t)$. Taking once more advantage of (6.1.33), we obtain that also the function $t \mapsto T(t)u_0$ is bounded with values in $C_b^{2+\theta}(\mathbb{R}^N)$. Then, arguing as above, we easily see that $u \in C^{1,2}([0,T] \times \mathbb{R}^N)$.

Proposition 6.2.6 Let T > 0, $\theta \in (0,1)$ and let $g : [0,T] \times \mathbb{R}^N \to \mathbb{R}$ be a bounded and continuous function such that $g(t,\cdot) \in C_b^{\theta}(\mathbb{R}^N)$ for any t and

$$\sup_{t\in[0,T]}\|g(t,\cdot)\|_{C^\theta_b(\mathbb{R}^N)}<+\infty.$$

Moreover, let $u_0 \in C_b^{2+\theta}(\mathbb{R}^N)$. Then, the function u in (6.2.9) is the unique strict solution to the problem (6.2.2) (i.e., $D_t u, D_x u, D_{xx} u \in C_b([0,T] \times \mathbb{R}^N)$ and u solves (6.2.2)). Moreover, there exists a positive constant $C = C_T$ such that

$$\sup_{t \in [0,T]} \|u(t,\cdot)\|_{C_b^{2+\theta}(\mathbb{R}^N)} \le C \bigg(\|u_0\|_{C_b^{2+\theta}(\mathbb{R}^N)} + \sup_{t \in [0,T]} \|g(t,\cdot)\|_{C_b^{\theta}(\mathbb{R}^N)} \bigg). \quad (6.2.14)$$

Proof. From Proposition 6.2.5, we know that, under our assumptions, the function u defined by (6.2.9) is the unique classical solution to the problem (6.2.2) belonging to $C^{1,2}([0,T]\times\mathbb{R}^N)$. For notational convenience, throughout the rest of the proof, we denote by v the function defined by the integral term in (6.2.9), so that we can write $u(t,\cdot) = T(t)u_0 + v(t,\cdot)$ for any $t \in [0,T]$. Let us prove that u is bounded with values in $C_b^{2+\theta}(\mathbb{R}^N)$ and it satisfies

Let us prove that u is bounded with values in $C_b^{2+\theta}(\mathbb{R}^N)$ and it satisfies (6.2.14). For this purpose, we begin by observing that, due to (6.1.33), we already know that $T(t)u_0 \in C_b^{2+\theta}(\mathbb{R}^N)$ for any $t \geq 0$ and

$$||T(t)||_{C_{t}^{2+\theta}(\mathbb{R}^{N})} \le C_{1}||u_{0}||_{C_{t}^{2+\theta}(\mathbb{R}^{N})},$$
 (6.2.15)

for some positive constant C_1 and any $t \in [0, T]$. To show that v is bounded with values in $C_b^{2+\theta}(\mathbb{R}^N)$ we adapt the techniques used in the proof of Theorem 6.2.2, namely, we show that, for any $t \in [0, T]$ and any $\alpha \in (\theta, 1)$,

$$v(t,\cdot)\in (C^{\alpha}_b(\mathbb{R}^N),C^{2+\alpha}_b(\mathbb{R}^N))_{1-(\alpha-\theta)/2,\infty},$$

recalling that $(C_b^{\alpha}(\mathbb{R}^N), C_b^{2+\alpha}(\mathbb{R}^N))_{1-(\alpha-\theta)/2,+\infty} = C_b^{2+\theta}(\mathbb{R}^N)$ with equivalence of the corresponding norms (see Theorem A.4.8). For this purpose, we split $v(t,\cdot) = a_{\xi}(t,\cdot) + b_{\xi}(t,\cdot)$, where, for any $\xi \in (0,1)$, we set

$$a_{\xi}(t,x) = \begin{cases} \int_{t-\xi}^{t} (T(t-s)g(s,\cdot))(x)ds, & \text{if } \xi \leq t, \\ \int_{0}^{t} (T(t-s)g(s,\cdot))(x)ds, & \text{if } \xi > t, \end{cases}$$

$$b_{t,\xi}(x) = \begin{cases} \int_{0}^{t-\xi} (T(t-s)g(s,\cdot))(x)ds, & \text{if } \xi \leq t, \\ 0, & \text{if } \xi > t, \end{cases}$$

for any $x \in \mathbb{R}^N$. Taking Lemma 6.2.1 and the estimate (6.1.33) into account, we easily deduce that, for any $\alpha \in (\theta, 1)$ and any $t \in [0, T]$, $a_{\xi}(t, \cdot) \in C_b^{\alpha}(\mathbb{R}^N)$ and $b_{\xi}(t, \cdot) \in C_b^{2+\alpha}(\mathbb{R}^N)$. Moreover,

$$||a_{\xi}(t,\cdot)||_{C_{b}^{\alpha}(\mathbb{R}^{N})} \leq C_{2} \sup_{0 \leq s \leq t} \int_{\max(t-\xi,0)}^{t} (t-s)^{-\frac{\alpha-\theta}{2}} ds,$$
$$\leq C_{3} \xi^{1-\frac{\alpha-\theta}{2}} \sup_{0 \leq s \leq t} ||g(s,\cdot)||_{C_{b}^{\theta}(\mathbb{R}^{N})}$$

and

$$||b_{\xi}(t,\cdot)||_{C_{b}^{2+\alpha}(\mathbb{R}^{N})} \leq C_{4} \sup_{0 \leq s \leq t} \int_{0}^{\max(t-\xi,0)} (t-s)^{-1-\frac{\alpha-\theta}{2}} ds,$$
$$\leq C_{5} \xi^{-\frac{\alpha-\theta}{2}} \sup_{0 \leq s \leq t} ||g(s,\cdot)||_{C_{b}^{\theta}(\mathbb{R}^{N})},$$

for some positive constants C_2, \ldots, C_5 , independent of ε , so that

$$\xi^{-1+\frac{\alpha-\theta}{2}}K(\xi,v(t,\cdot)) \leq \xi^{-1+\frac{\alpha-\theta}{2}} \|a_{\xi}(t,\cdot)\|_{C_b^{\alpha}(\mathbb{R}^N)} + \xi^{\frac{\alpha-\theta}{2}} \|b_{\xi}(t,\cdot)\|_{C_b^{\alpha}(\mathbb{R}^N)}$$

$$\leq (C_4 + C_5) \sup_{0 \leq s \leq T} \|g(s,\cdot)\|_{C_b^{\theta}(\mathbb{R}^N)},$$

for any $\xi \in (0,1)$, implying that $v(t,\cdot) \in (C_b^{\alpha}(\mathbb{R}^N), C_b^{2+\alpha}(\mathbb{R}^N))_{1-(\alpha-\theta)/2,\infty}$ for any $t \in (0,T)$ and that

$$\sup_{0 < t < T} \|v(t, \cdot)\|_{C_b^{2+\theta}(\mathbb{R}^N)} \le C_6 \sup_{0 < t < T} \|g(t, \cdot)\|_{C_b^{\theta}(\mathbb{R}^N)}, \tag{6.2.16}$$

for some positive constant C_6 . Summing up, we have proved that $u(t,\cdot) \in C_b^{2+\theta}(\mathbb{R}^N)$ for any $t \in [0,T]$. Moreover, combining (6.2.15) and (6.2.16), the estimate (6.2.14) now follows.

Chapter 7

Pointwise estimates for the derivatives of T(t)f

7.0 Introduction

This chapter is devoted to prove some pointwise estimates for the first-, second- and third-order derivatives of T(t)f, when $f \in C_b^k(\mathbb{R}^N)$ (k = 0, 1, 2, 3). Under the same assumptions on the coefficients as in Chapter 6, we prove that, for any k = 1, 2, 3 and any $p \in (1, +\infty)$, there exists a constant $M_{k,p} > 0$ such that

$$\left(\sum_{i=0}^{k} |D^{i}T(t)f(x)|^{2}\right)^{\frac{\nu}{2}} \leq M_{k,p} \left(T(t) \left(\sum_{i=0}^{k} |D^{i}f|^{2}\right)\right)^{\frac{\nu}{2}}\right)(x), \tag{7.0.1}$$

for any t > 0, any $x \in \mathbb{R}^N$ and any $f \in C_b^k(\mathbb{R}^N)$. Under somewhat heavier assumptions, we show that

$$\left(\sum_{i=1}^{k} |D^{i}T(t)f(x)|^{2}\right)^{\frac{\nu}{2}} \leq \hat{M}_{k,p}e^{\sigma_{k,p}t} \left(T(t)\left(\sum_{i=1}^{k} |D^{i}f|^{2}\right)\right)^{\frac{\nu}{2}}\right)(x), \quad (7.0.2)$$

for any t > 0 and any $x \in \mathbb{R}^N$, where $\hat{M}_{k,p}$ and $\sigma_{k,p}$ are, respectively, a positive and a negative constant.

The estimates (7.0.1) and (7.0.2) are then used to prove the sharper estimate

$$|(D^{k}T(t)f)(x)|^{p} \leq \left(\frac{\widetilde{\sigma}_{k,\min\{p,2\}}}{1 - e^{-\widetilde{\sigma}_{k,\min\{p,2\}}t}} \varphi_{k,\min\{p,2\}}(t)\right)^{\max\{1,\frac{p}{2}\}} \times \left(T(t)((|f|^{2} + \ldots + |D^{k-1}f|^{2})^{\frac{p}{2}})\right)(x), \quad (7.0.3)$$

holding for any $(t,x) \in (0,+\infty) \times \mathbb{R}^N$, any $f \in C_b^{k-1}(\mathbb{R}^N)$ and any p > 1, where $\widetilde{\sigma}_{k,p}$ is a real constant and $\varphi_{k,r} \in C([0,+\infty))$ is a suitable function which behaves as $t^{1-r/2}$ near 0 and it is such that the term in the first brackets in the right-hand side of (7.0.3) stays bounded at infinity, or it decreases to 0 exponentially. Taking the semigroup property into account, from (7.0.3) we get

$$|(D^k T(t)f)(x)|^p \le C_{k,p} \frac{e^{\omega_{k,p}t}}{t^{pk/2}} (T(t)(|f|^p))(x), \qquad t > 0, \ x \in \mathbb{R}^N,$$
 (7.0.4)

for any $f \in C_b(\mathbb{R}^N)$, any p > 1 and some constants $C_{k,p} > 0$, blowing up as p tends to 1, and $\omega_{k,p} \in \mathbb{R}$.

In the particular case when $q_{ij}(x) = \delta_{ij}$, i.e., when $\mathcal{A} = \Delta + \sum b_i(x)D_i$, we prove the estimate (7.0.1) also for p = 1. Such pointwise estimates are typical for transition semigroups of Markov processes, and they have been already studied for the first-order derivatives (k = 1); see [11, 13, 15]. Here, we present the results of [18].

On the contrary, the estimate (7.0.4) cannot be extended, in general, to the case when p = 1. Counterexamples are easily obtained in the simple case $\mathcal{A} = \Delta$ (see Example 7.3.3).

In the case when $\omega_{1,p} \leq 0$, the estimate (7.0.4) with k=1 allows us to obtain a Liouville type theorem, namely, in such a situation we can show that if $\mathcal{A}u=0$, then u is constant. If $\omega_{1,p}>0$, in general, such a result fails. Counterexamples are given in [128] also in the one-dimensional case.

Sometimes in what follows, when there is no damage of confusion, we write u instead of $T(\cdot)f$.

7.1 The first type of pointwise gradient estimates

We begin by proving the following lemma which will be essential to prove the first type of pointwise estimates.

Proposition 7.1.1 Let $k \in \{1, 2, 3\}$ and let Hypotheses 6.1.1(i)-6.1.1(iii) and 6.1.1(iv-k) be satisfied. Then, for any $f \in C_b^k(\mathbb{R}^N)$, the function $(t, x) \mapsto (D^kT(t)f)(x)$ is continuous in $[0, +\infty) \times \mathbb{R}^N$.

Proof. Fix $f \in C_b^k(\mathbb{R}^N)$ and set $u = T(\cdot)f$. The regularity of u for t > 0 is a classical result, recalled in Theorem C.1.4. Thus we have only to prove the regularity at t = 0. We will do it using a localization argument.

Fix $x_0 \in \mathbb{R}^N$ and let Ω be a smooth bounded neighborhood of x_0 . Moreover, let $\vartheta \in C_c^{\infty}(\Omega)$ be such that $\vartheta \equiv 1$ in a smaller neighborhood $\Omega_0 \subset \Omega$ of x_0 . Set $v(t,x) = \vartheta(x)u(t,x)$ for any t > 0 and any $x \in \Omega$. Then, the function v satisfies the equation $D_t v - \mathcal{A}v = \psi$ in $(0, +\infty) \times \Omega$, where

$$\psi(t,x) = -u(t,x)\mathcal{A}\vartheta(x) - 2\sum_{i,j=1}^{N} q_{ij}(x)D_iu(t,x)D_j\vartheta(x),$$

for any t>0 and any $x\in\Omega$, and the boundary condition v(t,x)=0 for any t>0 and any $x\in\partial\Omega$. Moreover, it is readily seen that there exists a constant C>0 such that

$$\|\psi(t,\cdot)\|_{\infty} \leq C\|u(t,\cdot)\|_{C^1(\overline{\Omega})} \leq C\frac{C_{0,1}e^T}{\sqrt{t}}\|f\|_{\infty}, \qquad t \in (0,T],$$

where we have used the estimate (6.0.1) with (k,l) = (0,1). In particular the function $t \mapsto \|\psi(t,\cdot)\|_{\infty}$ belongs to $L^1(0,T)$ for any T > 0, and, therefore, we can write

$$v(t,x) = (T_{\Omega}(t)(\vartheta f))(x) + \int_0^t (T_{\Omega}(t-s)\psi(s,\cdot))(x)ds, \qquad t > 0, \ x \in \Omega,$$

where $\{T_{\Omega}(t)\}$ is the semigroup associated with the realization of the operator \mathcal{A} with homogeneous Dirichlet boundary condition in $C(\overline{\Omega})$. Now, let k = 1, 2, 3 and $f \in C_b^k(\mathbb{R}^N)$; using the classical gradient estimates for $\{T_{\Omega}(t)\}$ (see Proposition C.3.2) and the estimate (6.0.1), we get

$$\begin{split} \|D^k T_{\Omega}(t-s)\psi(s,\cdot)\|_{\infty} &\leq \frac{C_T}{\sqrt{t-s}} \|\psi(s,\cdot)\|_{C^{k-1}(\overline{\Omega})} \\ &\leq C \frac{C_T}{\sqrt{t-s}} \|u(s,\cdot)\|_{C^k(\overline{\Omega})} \\ &\leq C_{k,k} C \frac{C_T}{\sqrt{t-s}} \|f\|_{C_b^k(\mathbb{R}^N)}, \end{split}$$

for any $0 < s < t \le T$, where $C, C_T > 0$ are constants. This means that the function $s \mapsto \|D^k T_{\Omega}(t-s)\psi(s,\cdot)\|_{\infty}$ belongs to $L^1(0,t)$ for any $t \in (0,T)$ and, therefore, we can write

$$|D^{k}v(t,x) - (D^{k}T_{\Omega}(t)(\vartheta f))(x)| = \left| \int_{0}^{t} (D^{k}T_{\Omega}(t-s)\psi(s,\cdot))(x)ds \right|$$

$$\leq CC_{kk}C_{T} \int_{0}^{t} \frac{1}{\sqrt{t-s}} \|f\|_{C_{b}^{k}(\mathbb{R}^{N})} ds$$

$$= 2CC_{kk}C_{T}t \|f\|_{C_{b}^{k}(\mathbb{R}^{N})},$$

for any $t \in (0,T]$ and any $x \in \Omega$. This implies that the function $D^k v$ is continuous in $[0,T] \times \Omega_0$ since, by virtue of Lemma 6.1.6, $D^k(T_{\Omega}(t)(\vartheta f))$ tends to $D^k(\vartheta f)$ uniformly in $\Omega_\delta := \{x \in \Omega : \operatorname{dist}(x,\partial\Omega) \geq \delta\}$, for any $\delta > 0$, as t tends to 0. Since $v \equiv u$ in a neighborhood of x_0 , it follows that $D^k u$ is continuous at $(0,x_0)$.

Now, we can prove the following theorem.

Theorem 7.1.2 Fix $k \in \{1,2,3\}$ and let Hypotheses 6.1.1(i)–6.1.1(iii) and 6.1.1(iv-k) be satisfied. Then, for any $f \in C_b^k(\mathbb{R}^N)$ and any $p \in (1,+\infty)$, there exists a positive constant $M_{k,p}$ such that

$$\left(\sum_{j=0}^{k} |(D^{j}T(t)f)(x)|^{2}\right)^{\frac{p}{2}} \leq M_{k,p} \left(T(t) \left(\sum_{j=0}^{k} |D^{j}f|^{2}\right)^{\frac{p}{2}}\right)(x), \tag{7.1.1}$$

for any t > 0 and any $x \in \mathbb{R}^N$.

Proof. We begin with the case when k=3 and $p \in (1,2]$. For any $\delta > 0$ we introduce the function $w_{\delta} : [0,+\infty) \times \mathbb{R}^N \to \mathbb{R}$ defined by

$$w_{\delta} = (\alpha |u|^2 + \beta |Du|^2 + \gamma |D^2u|^2 + |D^3u|^2 + \delta)^{\frac{p}{2}}, \qquad (7.1.2)$$

where α, β, γ are positive constants to be fixed later on, and $u = T(\cdot)f$. By Theorems 2.2.1, C.1.4(ii) and Proposition 7.1.1, $w_{\delta} \in C_b([0, +\infty) \times \mathbb{R}^N) \cap C^{1,2}((0, +\infty) \times \mathbb{R}^N)$ and a straightforward computation shows that w_{δ} solves the Cauchy problem

$$\begin{cases}
D_t w_{\delta}(t, \cdot) = \mathcal{A}w_{\delta}(t, \cdot) + g_{\delta}(t, \cdot), & t > 0, \\
w_{\delta}(0, \cdot) = \left(\alpha |f|^2 + \beta |Df|^2 + \gamma |D^2 f|^2 + |D^3 f|^2 + \delta\right)^{\frac{p}{2}}, & (7.1.3)
\end{cases}$$

where

$$\begin{split} g_{\delta} &= p \left(\alpha |u|^2 + \beta |Du|^2 + \gamma |D^2u|^2 + |D^3u|^2 + \delta \right)^{\frac{p}{2} - 1} \\ &\times \left(-\alpha \sum_{i,j=1}^{N} q_{ij} D_i u D_j u - \beta \sum_{i,j,h=1}^{N} q_{ij} D_{ih} u D_{jh} u \right. \\ &- \gamma \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} u D_{jhk} u - \sum_{i,j,h,k,l=1}^{N} q_{ij} D_{ihkl} u D_{jhkl} u \\ &+ \beta \sum_{i,j,h=1}^{N} D_h q_{ij} D_h u D_{ij} u + 2 \gamma \sum_{i,j,h,k=1}^{N} D_h q_{ij} D_{hk} u D_{ijk} u \\ &+ 3 \sum_{i,j,h,k,l=1}^{N} D_h q_{ij} D_{hkl} u D_{ijkl} u + \beta \sum_{j,h=1}^{N} D_h b_j D_j u D_h u \\ &+ 2 \gamma \sum_{j,h,k=1}^{N} D_h b_j D_{jk} u D_{hk} u + 3 \sum_{j,h,k,l=1}^{N} D_h b_j D_{jkl} u D_{hkl} u \\ &+ \gamma \sum_{i,j,h,k=1}^{N} D_{hk} q_{ij} D_{ij} u D_{hk} u + 3 \sum_{i,j,h,k,l=1}^{N} D_{hk} q_{ij} D_{ijl} u D_{hkl} u \\ &+ \gamma \sum_{j,h,k=1}^{N} D_{hk} b_j D_j u D_{hk} u + 3 \sum_{j,h,k,l=1}^{N} D_{hk} b_j D_{jl} u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_j u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{hkl} q_{ij} D_{ij} u D_{hkl} u + \sum_{i,j,h,k,l=1}^{N} D_{hkl} b_j D_{jl} u D_{hkl} u \\ &+ \sum_{i,j,h,k=1}^{N} D_{ij} u D$$

$$\begin{split} &+ p(2-p) \left(\alpha |u|^{2} + \beta |Du|^{2} + \gamma |D^{2}u|^{2} + |D^{3}u|^{2} + \delta\right)^{\frac{p}{2}-2} \\ &\times \sum_{i,j=1}^{N} q_{ij} \left(\alpha u D_{i} u + \beta \sum_{h=1}^{N} D_{h} u D_{ih} u + \gamma \sum_{h,k=1}^{N} D_{hk} u D_{ihk} u \right. \\ &+ \sum_{h,k,l=1}^{N} D_{hkl} u D_{ihkl} u \right) \\ &\times \left(\alpha u D_{j} u + \beta \sum_{h=1}^{N} D_{h} u D_{jh} u + \gamma \sum_{h,k=1}^{N} D_{hk} u D_{jhk} u \right. \\ &+ \sum_{h,k,l=1}^{N} D_{hkl} u D_{jhkl} u \right). \end{split}$$

Now, let $h, k \in \{0, 1, 2, 3\}$ be fixed. Applying the Cauchy-Schwarz inequality twice (first to the inner product $(\xi, \eta) \mapsto \langle Q(x)\xi, \eta \rangle$ and then to the Euclidean one) we deduce that

$$\sum_{i,j=1}^{N} q_{ij} \sum_{|\alpha|=h} D^{\alpha} u D_{i} D^{\alpha} u \sum_{|\beta|=k} D^{\beta} u D_{j} D^{\beta} u$$

$$= \sum_{|\alpha|=h} \sum_{|\beta|=k} D^{\alpha} u D^{\beta} u \sum_{i,j=1}^{N} q_{ij} D_{i} D^{\alpha} u D_{j} D^{\beta} u$$

$$\leq \sum_{|\alpha|=h} |D^{\alpha} u| \left(\sum_{i,j=1}^{N} q_{ij} D_{i} D^{\alpha} u D_{j} D^{\alpha} u\right)^{\frac{1}{2}}$$

$$\times \sum_{|\beta|=k} |D^{\beta} u| \left(\sum_{i,j=1}^{N} q_{ij} D_{i} D^{\beta} u D_{j} D^{\beta} u\right)^{\frac{1}{2}}$$

$$\leq |D^{h} u| |D^{k} u| \left(\sum_{|\alpha|=h} \sum_{i,j=1}^{N} q_{ij} D_{i} D^{\alpha} u D_{j} D^{\alpha} u\right)^{\frac{1}{2}}$$

$$\times \left(\sum_{|\beta|=k} \sum_{i,j=1}^{N} q_{ij} D_{i} D^{\beta} u D_{j} D^{\beta} u\right)^{\frac{1}{2}}.$$

$$(7.1.4)$$

This estimate can be used in order to get

$$\sum_{i,j=1}^{N} q_{ij} \left(\alpha u D_{i} u + \beta \sum_{h=1}^{N} D_{h} u D_{ih} u + \gamma \sum_{h,k=1}^{N} D_{hk} u D_{ihk} u \right)$$

$$+ \sum_{h,k,l=1}^{N} D_{hkl} u D_{ihkl} u$$

$$\times \left(\alpha u D_{j} u + \beta \sum_{h=1}^{N} D_{h} u D_{jh} u + \gamma \sum_{h,k=1}^{N} D_{hk} u D_{jhk} u \right)$$

$$+ \sum_{h,k,l=1}^{N} D_{hkl} u D_{jhkl} u$$

$$+ \sum_{h,k,l=1}^{N} D_{hkl} u D_{jhkl} u \right)$$

$$\leq \left[\alpha |u| \left(\sum_{i,j=1}^{N} q_{ij} D_{i} u D_{j} u \right)^{\frac{1}{2}} + \beta |D u| \left(\sum_{i,j,h=1}^{N} q_{ij} D_{ih} u D_{jh} u \right)^{\frac{1}{2}} \right)$$

$$+ \gamma |D^{2} u| \left(\sum_{i,j,h,k,l=1}^{N} q_{ij} D_{ihkl} u D_{jhkl} u \right)$$

$$+ |D^{3} u| \left(\sum_{i,j,h,k,l=1}^{N} q_{ij} D_{ihkl} u D_{jhkl} u \right)$$

$$\leq (\alpha |u|^{2} + \beta |D u|^{2} + \gamma |D^{2} u|^{2} + |D^{3} u|^{2})$$

$$\times \left(\alpha \sum_{i,j=1}^{N} q_{ij} D_{i} u D_{j} u + \beta \sum_{i,j,h=1}^{N} q_{ij} D_{ih} u D_{jh} u \right)$$

$$+ \gamma \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} u D_{jhk} u + \sum_{i,j,h,k,l=1}^{N} q_{ij} D_{ihkl} u D_{jhkl} u \right).$$

$$(7.1.5)$$

Taking (7.1.5) into account, it is immediate to check that

$$g_{\delta} \leq p \left\{ (1-p) \left(\alpha \sum_{i,j=1}^{N} q_{ij} D_{i} u D_{j} u + \beta \sum_{i,j,h=1}^{N} q_{ij} D_{ih} u D_{jh} u \right) + (1-p) \left(\gamma \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} u D_{jhk} u + \sum_{i,j,h,k,l=1}^{N} q_{ij} D_{ihkl} u D_{jhkl} u \right) + \beta \sum_{i,j,h=1}^{N} D_{h} q_{ij} D_{h} u D_{ij} u + 2\gamma \sum_{i,j,h,k=1}^{N} D_{h} q_{ij} D_{hk} u D_{ijk} u + 3 \sum_{i,j,h,k,l=1}^{N} D_{h} q_{ij} D_{hkl} u D_{ijkl} u + \beta \sum_{j,h=1}^{N} D_{h} b_{j} D_{j} u D_{h} u \right\}$$

$$\begin{split} &+2\gamma\sum_{j,h,k=1}^{N}D_{h}b_{j}D_{jk}uD_{hk}u+3\sum_{j,h,k,l=1}^{N}D_{h}b_{j}D_{jkl}uD_{hkl}u\\ &+\gamma\sum_{i,j,h,k=1}^{N}D_{hk}q_{ij}D_{ij}uD_{hk}u+3\sum_{i,j,h,k,l=1}^{N}D_{hk}q_{ij}D_{ijl}uD_{hkl}u\\ &+\gamma\sum_{j,h,k=1}^{N}D_{hk}b_{j}D_{j}uD_{hk}u+3\sum_{j,h,k,l=1}^{N}D_{hk}b_{j}D_{jl}uD_{hkl}u\\ &+\sum_{i,j,h,k,l=1}^{N}D_{hkl}q_{ij}D_{ij}uD_{hkl}u+\sum_{j,h,k,l=1}^{N}D_{hkl}b_{j}D_{j}uD_{hkl}u\\ &\times(\alpha|u|^{2}+\beta|Du|^{2}+\gamma|D^{2}u|^{2}+|D^{3}u|^{2}+\delta)^{\frac{p}{2}-1}. \end{split}$$

Hence, using Hypotheses 6.1.1(i) and 6.1.1(iv-3), we get, for any $\varepsilon, \varepsilon_1 > 0$,

$$g_{\delta} \leq p \left\{ \left((1-p)\alpha\kappa + C\beta \frac{N^{2}}{4\varepsilon_{1}}\kappa + \beta d + r \frac{N^{3}}{4\varepsilon_{2}} + r\gamma \frac{N^{2}}{4\varepsilon_{3}} \right) |Du|^{2} \right.$$

$$\left. + \left[(1-p+C\varepsilon_{1}N)\beta\kappa + \left(C\gamma \frac{N^{2}}{2\varepsilon_{1}} + C \frac{N^{3}}{4\varepsilon_{1}} + \gamma K_{1} \right) \kappa + 2d\gamma \right.$$

$$\left. + rN \left(\gamma\varepsilon_{3} + \frac{3N}{4\varepsilon_{2}} \right) \right] |D^{2}u|^{2}$$

$$\left. + \left[\left((1-p+2C\varepsilon_{1}N)\gamma + C \frac{3N^{2}}{4\varepsilon_{1}} + \varepsilon_{1}CN^{2} + 3K_{1} \right) \kappa \right.$$

$$\left. + 3d + 4r\varepsilon_{2}N \right] |D^{3}u|^{2} + (1-p+3C\varepsilon_{1}N) \kappa |D^{4}u|^{2} \right\}$$

$$\times \left(\alpha|u|^{2} + \beta|Du|^{2} + \gamma|D^{2}u|^{2} + |D^{3}u|^{2} + \delta \right)^{\frac{p}{2}-1}.$$

Now, we choose $\varepsilon_1 = (p-1)/(3CN)$ to make the coefficient of $|D^4u|^2$ vanish, $\varepsilon_2 = 3L_2/(4N)$ and $\varepsilon_3 = \gamma^{-1}$, where L_2 is defined in Hypothesis 6.1.1(iv-3). Moreover, we choose β and γ satisfying

$$\begin{cases} \frac{N^4}{3L_2} + \frac{N^2 \gamma^2}{4} \le L_2 \beta, \\ N\left(1 + \frac{N^2}{L_2}\right) \le 2L_2 \gamma. \end{cases}$$

With this choice of β and γ , from (7.1.6) we get

$$\begin{split} g_{\delta} & \leq p \left\{ \left((1-p)\alpha\kappa + \frac{3C^2N^3}{4(p-1)}\beta\kappa + \beta(d+L_2r) \right) |Du|^2 \right. \\ & \left. + \left[\left(\frac{2-2p}{3}\beta + \frac{3C^2N^3}{4(p-1)}(2\gamma+N) + \gamma K_1 \right) \kappa + 2\gamma(d+L_2r) \right] |D^2u|^2 \right. \end{split}$$

$$+ \left[\left(\frac{1-p}{3} \gamma + \frac{9C^2N^3}{4(p-1)} + \frac{(p-1)N}{3} + 3K_1 \right) \kappa \right] \\
+ 3(d+L_2r) \left[|D^3u|^2 \right] \\
\times \left(\alpha |u|^2 + \beta |Du|^2 + \gamma |D^2u|^2 + |D^3u|^2 + \delta \right)^{\frac{p}{2}-1} \\
\leq p \left\{ \left((1-p)\alpha + \frac{3C^2N^3}{4(p-1)}\beta + \beta L_3 \right) \kappa |Du|^2 \right. (7.1.7) \\
+ \left(\frac{2-2p}{3}\beta + \frac{3C^2N^3}{4(p-1)}(2\gamma + N) + \gamma K_1 + 2\gamma L_3 \right) \kappa |D^2u|^2 \\
+ \left(\frac{1-p}{3}\gamma + \frac{9C^2N^3}{4(p-1)} + \frac{(p-1)N}{3} + 3(K_1 + L_3) \kappa |D^3u|^2 \right) \\
\times \left(\alpha |u|^2 + \beta |Du|^2 + \gamma |D^2u|^2 + |D^3u|^2 + \delta \right)^{\frac{p}{2}-1} \right]$$

Hence, up to taking larger β and γ and fixing α suitably large, we can make the right-hand side of (7.1.7) nonpositive in $(0, +\infty) \times \mathbb{R}^N$. The maximum principle in Theorem 4.1.3 now implies that

$$w_{\delta} \leq T(\cdot) \left(\alpha |f|^2 + \beta |Df|^2 + \gamma |D^2 f|^2 + |D^3 f|^2 + \delta \right),$$

in $(0, +\infty) \times \mathbb{R}^N$, for any $\delta > 0$. Taking the limit as δ tends to 0, from Proposition 2.2.9 we get

$$(\alpha |u(t,x)|^2 + \beta |Du(t,x)|^2 + \gamma |D^2u(t,x)|^2 + |D^3u(t,x)|^2)^{\frac{p}{2}}$$

$$\leq (T(t) (\alpha |f|^2 + \beta |Df|^2 + \gamma |D^2f|^2 + |D^3f|^2)) (x),$$

for any t > 0 and any $x \in \mathbb{R}^N$. Then (7.1.1) follows.

Finally we consider the case when p > 2. Since

$$\left(\sum_{j=0}^{3} |(D^{j}T(t)f)(x)|^{2}\right)^{\frac{p}{2}}$$

$$\leq \left(M_{3,2}\left(T(t)\left(|f|^{2} + |Df|^{2} + |D^{2}f|^{2} + |D^{3}f|^{2}\right)\right)(x)\right)^{\frac{p}{2}},$$

we get (7.1.1) observing that, since $T(t)\psi$ is given by (2.2.8) and the measures p(t, x; dy) in (2.2.9) are probability measures for any t > 0 and any $x \in \mathbb{R}^N$ (since $c \equiv 0$), then the Jensen inequality implies that

$$(T(t)\psi)^{p/2} \le T(t)(\psi^{p/2}),$$

for any t > 0 and any nonnegative $\psi \in C_b(\mathbb{R}^N)$.

To get (7.1.1) in the case when $p \in (1,2)$ and k = 1,2, one can apply the previous arguments to the function

$$w_{\delta} = \left(\alpha |T(\cdot)f|^2 + |DT(\cdot)f|^2 + \delta\right)^{\frac{p}{2}}, \qquad \delta > 0.$$

if k = 1 and to the function

$$w_{\delta} = \left(\alpha |T(\cdot)f|^2 + \beta |DT(\cdot)f|^2 + |D^2T(\cdot)f|^2 + \delta\right)^{\frac{p}{2}},$$

for any $\delta > 0$, if k = 2, where α and β are positive constants to be properly fixed. Actually one gets

$$\left(\sum_{j=0}^{k} |(D^{j}T(t)f)(x)|^{2} + \delta\right)^{\frac{p}{2}} \le M_{k,p} \left(T(t) \left(\sum_{j=0}^{k} |D^{j}f|^{2} + \delta\right)^{\frac{p}{2}}\right)(x), (7.1.8)$$

for any t > 0, any $x \in \mathbb{R}^N$ and k = 1, 2.

In the case when p > 2 (7.1.1) then follows from the case p = 2, applying the Jensen inequality. This finishes the proof.

We now show that under more restrictive assumptions on the coefficients of the operator \mathcal{A} we can improve the estimates in Theorem 7.1.2.

Hypotheses 7.1.3 (i) Hypotheses 6.1.1(i)–6.1.1(iii) are satisfied. Moreover, we always assume that one of the following hypotheses holds true:

(ii-1) $q_{ij}, b_j \in C^{1+\delta}_{loc}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any $i,j=1,\ldots,N$, and there exist $\sigma \in (0,1)$ and a constant C>0 such that $|D^\beta q_{ij}(x)| \leq C(\kappa(x))^\sigma$ for any $x \in \mathbb{R}^N$, any $|\beta|=1$ and any $i,j=1,\ldots,N$. Moreover, there exist two constants $L>0, \ p_0 \in (1,2]$ and a function $d:\mathbb{R}^N \to \mathbb{R}$ such that

$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \le d(x) |\xi|^2, \qquad x, \xi \in \mathbb{R}^N$$
 (7.1.9)

and

$$0 \ge C_1(p_0) := \sup_{\mathbb{R}^N} \left(\frac{C^2 N^3}{4(p_0 - 1)\kappa_0^{1 - \sigma}} \kappa^{\sigma} + d \right); \tag{7.1.10}$$

(ii-2) $q_{ij}, b_j \in C^{2+\delta}_{loc}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any $i,j=1,\ldots,N$, and there exist $\sigma \in (0,1)$, a constant C>0 and a positive function $r:\mathbb{R}^N\to\mathbb{R}$ such that $|D^2b_i(x)| \leq r(x)$ and $|D^\beta q_{ij}(x)| \leq C(\kappa(x))^\sigma$ for any $x\in\mathbb{R}^N$, any $|\beta|=1,2$ and any $i,j=1,\ldots,N$. Moreover, the condition (7.1.9) is satisfied and there exist two constants L>0 and $p_0\in (1,2]$ such that

$$0 \ge C_2(p_0) := \sup_{\mathbb{R}^N} \left(d + Lr + \frac{C^2 N^3}{4(p_0 - 1)} \kappa_0^{\sigma - 1} \kappa_0^{\sigma} \right). \tag{7.1.11}$$

Finally, there exists a constant $K_1 \in \mathbb{R}$ such that

$$\sum_{i,j,h,k=1}^{N} D_{hk} q_{ij}(x) m_{ij} m_{hk} \le K_1(\kappa(x))^{\sigma} \sum_{h,k=1}^{N} m_{hk}^2,$$

for any symmetric matrix $M = (m_{hk})_{h,k=1}^N$ and any $x \in \mathbb{R}^N$;

(ii-3) $q_{ij}, b_j \in C^{3+\delta}_{loc}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any $i, j = 1, \ldots, N$; Hypothesis 7.1.3(ii-2) holds true. Moreover, $|D^{\beta}b_i(x)| \leq r(x)$ and $|D^{\beta}q_{ij}(x)| \leq C(\kappa(x))^{\sigma}$ for any $i, j = 1, \ldots, N$, any $|\beta| = 3$ and any $x \in \mathbb{R}^N$.

Remark 7.1.4 Hypotheses 7.1.3(ii-k) (k = 1, 2, 3) allow us to consider the case when the coefficients are of polynomial type.

We are now able to improve the result in Theorem 7.1.2.

Theorem 7.1.5 Let k = 1, 2, 3 and let Hypotheses 7.1.3(i), 7.1.3(ii-k) be satisfied. Then, there exists a positive constant $\hat{M}_{k,p}$ such that

$$\left(\sum_{j=1}^{k} |(D^{j}T(t)f)(x)|^{2}\right)^{\frac{p}{2}} \leq \hat{M}_{k,p}e^{\omega_{k,p}t} \left(T(t)\left(\sum_{j=1}^{k} |D^{j}f|^{2}\right)^{\frac{p}{2}}\right)(x), \quad (7.1.12)$$

for any t > 0, any $x \in \mathbb{R}^N$, any $f \in C_b^k(\mathbb{R}^N)$ and any $p \in (p_0, +\infty)$, where $\omega_{1,p} = pC_1(p)$, with $C_1(p)$ being defined as in (7.1.10) with p instead of p_0 , and $\omega_{k,p}$ (k = 2,3) is any real number greater than $p\kappa_0^{\sigma}C_2(p)$, where $C_2(p)$ is defined according to (7.1.11) with p instead of p_0 . In particular, for k = 1, 2, 3, we can take $\omega_{k,p} < 0$.

Proof. The proof is close to that of Theorem 7.1.2. Hence, we just point out the main differences. To get (7.1.12), in the case when $p \in (p_0, 2]$ and k = 3, one has to deal with the function w_{δ} defined in (7.1.2) with $\alpha = 0$ and β, γ to be determined. As it is immediately seen, the function w_{δ} so defined solves the Cauchy problem (7.1.3) with g_{δ} satisfying (7.1.6) (with $\alpha = 0$ everywhere).

Taking Hypothesis 7.1.3(ii-3) into account and using the same techniques as in the proof of Theorem 7.1.2 to estimate the right-hand side of (7.1.6), we get

$$g_{\delta} \leq p \left\{ \left(\left(C \frac{N^2}{4\varepsilon_2} \kappa^{\sigma} + d \right) \beta + r \frac{N^2}{4\varepsilon_1} (N + \gamma^2) \right) |Du|^2 \right.$$

$$+ \left[\left((1 - p)\kappa_0^{1 - \sigma} + C\varepsilon_2 N \right) \beta \kappa^{\sigma} + \left(C \gamma \frac{N^2}{2\varepsilon_3} + C \frac{N^3}{4\varepsilon_3} + K_1 \gamma \right) \kappa^{\sigma} \right.$$

$$\left. + 2d\gamma + r N \left(\varepsilon_1 + \frac{3N}{4\varepsilon_1} \right) \right] |D^2 u|^2$$

$$+\left[\left((1-p)\kappa_0^{1-\sigma} + 2C\varepsilon_3N\right)\gamma\kappa^{\sigma} + \left(C\frac{3N^2}{4\varepsilon_3} + \varepsilon_3CN^2 + 3K_1\right)\kappa^{\sigma} + 3d + 4r\varepsilon_1N\right]|D^3u|^2 + \left((1-p)\kappa_0^{1-\sigma} + 3C\varepsilon_3N\right)\kappa^{\sigma}|D^4u|^2\right\} \times \left(\beta|Du|^2 + \gamma|D^2u|^2 + |D^3u|^2 + \delta\right)^{\frac{p}{2}-1}.$$

$$(7.1.13)$$

Due to the estimate (7.1.11), we can find out $s_1 \in (0,1)$ such that

$$0 > C_2(p, s_1) = \sup_{x \in \mathbb{R}^N} \left(\frac{d(x) + Lr(x)}{(\kappa(x))^{\sigma}} + \frac{C^2 N^3}{4(p-1)(1-s_1)} \kappa_0^{\sigma-1} \right).$$
 (7.1.14)

We now choose

$$\varepsilon_2 = (p-1)(1-s_1)\frac{\kappa_0^{1-\sigma}}{CN}, \qquad \varepsilon_3 = (p-1)\frac{\kappa_0^{1-\sigma}}{3CN}$$

and ε_1 in order to minimize the function $g:(0,+\infty)\to\mathbb{R}$ defined by $g(x)=N\max\{4x/3,(4x^2+3N)/(8\gamma x)\}$. We obtain $\varepsilon_1=3\sqrt{N}/(2\sqrt{8\gamma-3})$, if $\gamma\geq 3/8$. With these choices of ε_1 , ε_2 and ε_3 , we get

$$g_{\delta} \leq p \left\{ \beta \left(\frac{C^{2}N^{3}}{4(p-1)(1-s_{1})} \kappa_{0}^{\sigma-1} \kappa^{\sigma} + d + r \frac{\sqrt{(8\gamma-3)N^{3}}}{6\beta} (N+\gamma^{2}) \right) |Du|^{2} \right.$$

$$+ \left[s_{1}(1-p)\kappa_{0}^{1-\sigma}\beta\kappa^{\sigma} + \left((2\gamma+N) \frac{3C^{2}N^{3}}{4(p-1)\kappa_{0}^{1-\sigma}} + K_{1}\gamma \right) \kappa^{\sigma} \right.$$

$$+ 2\gamma \left(d + \frac{2N^{\frac{3}{2}}}{\sqrt{8\gamma-3}} r \right) \right] |D^{2}u|^{2}$$

$$+ \left[\frac{1}{3} (1-p)\kappa_{0}^{1-\sigma}\gamma\kappa^{\sigma} + \left(\frac{9C^{2}N^{3}}{4(p-1)\kappa_{0}^{1-\sigma}} + \frac{1}{3}(p-1)N\kappa_{0}^{1-\sigma} + 3K_{1} \right) \kappa^{\sigma} \right.$$

$$+ 3 \left(d + \frac{2N^{\frac{3}{2}}}{\sqrt{8\gamma-3}} r \right) \right] |D^{3}u|^{2} \right\}$$

$$\times \left(\beta |Du|^{2} + \gamma |D^{2}u|^{2} + |D^{3}u|^{2} + \delta \right)^{\frac{p}{2}-1}.$$

Now we choose $\beta, \gamma > 0$ such that

$$\begin{cases} A_1(\beta,\gamma) := \frac{\sqrt{(8\gamma - 3)N^3}}{6\beta}(N + \gamma^2) - L \le 0, \\ A_2(\gamma) := \frac{2N^{\frac{3}{2}}}{\sqrt{8\gamma - 3}} - L \le 0, \\ C_3(\beta,\gamma,p) := s_1(1 - p)\kappa_0^{1-\sigma}\beta + (4\gamma + 3N)\frac{C^2N^3}{4(p - 1)\kappa_0^{1-\sigma}} + K_1\gamma < 0, \\ C_4(\gamma,p) := \frac{1}{3}(1 - p)\kappa_0^{1-\sigma}\gamma + \frac{3C^2N^3}{2(p - 1)\kappa_0^{1-\sigma}} + \frac{1}{3}(p - 1)N\kappa_0^{1-\sigma} + 3K_1 < 0. \end{cases}$$

and we get

$$g_{\delta}(t,x) \leq \omega_{3,p} \left(\beta |Du|^{2} + \gamma |D^{2}u|^{2} + |D^{3}u|^{2} \right)$$

$$\times \left(\beta |Du|^{2} + \gamma |D^{2}u|^{2} + |D^{3}u|^{2} + \delta \right)^{\frac{p}{2} - 1}$$

$$\leq \omega_{3,p} w(t,x) - \omega_{3,p} \delta^{\frac{p}{2}},$$

$$(7.1.15)$$

for any t > 0 and any $x \in \mathbb{R}^N$, where

$$\omega_{3,p} = p \kappa_0^{\sigma} \inf_{\beta > 0, \gamma > 3/8, A_1(\beta,\gamma), A_2(\gamma) \le 0 \atop C_3(\beta,\gamma,p), C_4(\gamma,p) < 0, j = 2,3} \max\{C_2(p), C_3(\beta,\gamma,p)/\gamma, C_4(\gamma,p)\}.$$

Note that the previous infimum is $C_2(p)$. Of course, $\omega_{3,p} \geq C_2(p)$. Hence, we just need to show that $\omega_{3,p} \leq C_2(p)$. For this purpose, we observe that, since $\lim_{\gamma \to +\infty} C_4(\gamma,p) = -\infty$, we can determine γ_0 such that $C_3(\gamma_0,p) < C_2(p)$. Similarly, since $\lim_{\beta \to +\infty} C_3(\beta,\gamma_0,p) = -\infty$, we can determine $\beta_0 > 0$ such that $C_3(\beta_0,\gamma_0,p)/\gamma_0 < C_2(p)$. Of course, without loss of generality we can assume that $\gamma_0 > 3/8$ and $\max\{A_1(\beta_0,\gamma_0),A_2(\gamma_0) \leq 0\}$. Therefore, $\omega_{3,p} \leq C_2(p)$. So, let us fix $\beta,\gamma \geq 1$ such that $\omega_{3,p} = p\kappa_0^\sigma C_2(p)$.

Now, from (7.1.15) we easily deduce that

$$D_t w_{\delta}(t, x) \leq \mathcal{A}w_{\delta}(t, x) + \omega_{3,p} w_{\delta}(t, x) - \omega_{3,p} \delta^{\frac{p}{2}}.$$

Set now $z_{\delta} = e^{-\omega_{3,p}t}(w_{\delta} - \delta^{p/2})$ and observe that z_{δ} satisfies

$$\begin{cases}
D_t z_{\delta}(t, \cdot) \leq \mathcal{A} z_{\delta}(t, \cdot), & t > 0, \\
z_{\delta}(0, \cdot) \leq (\beta |Df|^2 + \gamma |D^2 f|^2 + |D^3 f|^2 + \delta)^{\frac{p}{2}}.
\end{cases}$$
(7.1.16)

The maximum principle in Theorem 4.1.3 implies that

$$z_{\delta}(t,x) \le \left(T(t)\left(\beta|Df|^2 + \gamma|D^2f|^2 + |D^3f|^2 + \delta\right)^{\frac{p}{2}}\right)(x), \quad t > 0, \ x \in \mathbb{R}^N,$$

for any $\delta > 0$. Taking the limit as δ tends to 0, from Proposition 2.2.9 we get

$$\left(\beta |Du(t,x)|^2 + \gamma |D^2u(t,x)|^2 + |D^3u(t,x)|^2 \right)^{\frac{p}{2}}$$

$$\leq e^{\omega_{3,p}t} \left(T(t) \left(\beta |Df|^2 + \gamma |D^2f|^2 + |D^3f|^2 \right)^{\frac{p}{2}} \right) (x), \qquad t > 0, \ x \in \mathbb{R}^N,$$

and (7.1.12) follows with $\hat{M}_{3,p} = \max\{\beta^{p/2}, \gamma^{p/2}\}.$

Next, to get (7.1.12) in the case when p > 2, it suffices to repeat the same arguments as in the proof of Theorem 7.1.2.

To get (7.1.1) in the case when $p \in (1,2)$ and k=2, one can apply the above arguments to the function

$$w_{\delta} = \left(\beta |DT(\cdot)f|^2 + |D^2T(\cdot)f|^2 + \delta\right)^{\frac{p}{2}}, \qquad \delta > 0,$$

if k=2, where β is a positive constant to be determined. From (7.1.13), where we disregard the terms whose coefficients do not depend on β, γ , we take $\gamma=1$, $\varepsilon_1=2L/N$, $\varepsilon_3=(p-1)\kappa_0^{1-\sigma}/(2CN)$ and we make the same choices of ε_2 as above; we get

$$\begin{split} g_{\delta} &\leq p \bigg\{ \beta \left(\frac{C^2 N^3}{4(p-1)(1-s_1)} \kappa_0^{\sigma-1} \kappa^{\sigma} + d + \frac{N^3}{8\beta L} r \right) |Du|^2 \\ &\quad + \left(s_1 (1-p) \beta \kappa_0^{1-\sigma} + \frac{C^2 N^3}{(p-1)\kappa_0^{1-\sigma}} + K_1 \right) \kappa^{\sigma} |D^2 u|^2 \bigg\} \\ &\quad \times \left(\beta |Du|^2 + |D^2 u|^2 + \delta \right)^{\frac{p}{2}-1}. \end{split}$$

Choosing β satisfying

$$\begin{cases} N^3 \le 8\beta L^2, \\ C_5(\beta, p) := s_1(1-p)\beta \kappa_0^{1-\sigma} + \frac{C^2 N^3}{(p-1)\kappa_0^{1-\sigma}} + K_1 < 0, \end{cases}$$

we get

$$(\beta |Du|^2 + |D^2u|^2 + \delta)^{\frac{p}{2}} \le e^{\omega_{2,p}t} \left(T(\cdot)(\beta |Df|^2 + |D^2u|^2 + \delta)^{\frac{p}{2}} \right) + \delta^{\frac{p}{2}}, (7.1.17)$$

with

$$\omega_{2,p} = p\kappa_0^{\sigma} \min_{\substack{\beta \ge N^4/(8L^2) \\ C_{\kappa}(\beta,p) < 0}} \max\{C_2(p,s_1), C_5(\beta,p)\} = C_2(p,s_1).$$

Finally, in the case when k = 1, taking $\beta = 1$ and $\varepsilon_2 = (p-1)\kappa_0^{1-\sigma}/(CN)$ in (7.1.13) and disregarding the terms whose coefficients are independent of β , we get

$$g_{\delta} \le p \left(\frac{C^2 N^3}{4(p-1)\kappa_0^{1-\sigma}} \kappa^{\sigma} + d \right) |Du|^2$$

and, consequently, we get the assertion with $\omega_{1,p}$ as in the assertion of the theorem. This finishes the proof.

Remark 7.1.6 The conditions (7.1.10) and (7.1.11) can be a bit relaxed and, under these new assumptions, the estimate (7.1.12) can be proved with positive constants $\omega_{k,p}$ (k = 1, 2, 3). More precisely, it suffices to assume that

$$\sup_{x \in \mathbb{R}^N} \frac{d(x)}{(\kappa(x))^{\sigma}} < +\infty,$$

to have (7.1.12) with k = 1,

$$\sup_{x \in \mathbb{R}^N} \frac{d(x) + L_2 r(x)}{(\kappa(x))^{\sigma}} < +\infty,$$

(with $L_2 = 2/\sqrt{5}$, if N = 1, and $L_2 = N^{3/2}/\sqrt{8}$, if N > 1) to have (7.1.12) with k = 2,

$$\sup_{x \in \mathbb{R}^N} \frac{3d(x) + \sqrt{N^3(N+1)}r(x)}{3(\kappa(x))^{\sigma}} < +\infty,$$

to have (7.1.12) with k = 3.

7.2 The second type of pointwise gradient estimates

In this section we prove a second type of pointwise estimates. For this purpose, we first prove a lemma.

Lemma 7.2.1 If $g_n, g : [0,T] \times \mathbb{R}^N$ $(n \in \mathbb{N})$, are continuous functions such that $||g_n||_{\infty} \leq M$ for any $n \in \mathbb{N}$ and $\lim_{n \to +\infty} g_n = g$, uniformly in $[\varepsilon, T - \varepsilon] \times B(k)$ for any k > 0 and any $\varepsilon \in (0,T)$, then

$$\lim_{n \to +\infty} T_n(t)g_n(t) = T(t)g(t),$$

uniformly in $[\varepsilon, T - \varepsilon] \times B(k)$ for any k > 0 and any $\varepsilon \in (0, T)$.

If $f_{\varepsilon}, f_0 \in C_b(\mathbb{R}^N)$ ($\varepsilon > 0$) are such that $||f_{\varepsilon}||_{\infty} \leq M$ for any $\varepsilon > 0$ and $\lim_{\varepsilon \to 0^+} f_{\varepsilon} = f_0$ uniformly in B(k) for any k > 0, then, for any t > 0, we have

$$\lim_{\varepsilon \to 0^+} T(t - \varepsilon) f_{\varepsilon} = T(t) f_0,$$

uniformly in B(k) for any k > 0.

Proof. Throughout the proof, in order to simplify the notation, sometimes we do not explicitly write the dependence on x of the functions that we will consider. All the estimates that we write are meant pointwise in x.

Let g_n, g be as above and observe that

$$|T(t)f - T_n(t)f| \le (T(t) - T_n(t))|f|, \qquad t > 0.$$
 (7.2.1)

To check (7.2.1), we recall that (see Theorem 2.2.5 and Proposition C.3.2)

$$(T_n(t)f)(x) = \int_{B(n)} G_n(t,x,y)f(y)dy, \qquad (T(t)f)(x) = \int_{\mathbb{R}^N} G(t,x,y)f(y)dy$$

for any t > 0 and any $x \in B(n)$, where G_n and G are positive, respectively, in $(0, +\infty) \times B(n) \times B(n)$ and in $(0, +\infty) \times \mathbb{R}^N \times \mathbb{R}^N$. Moreover $G_n \leq G$ in

$$(0,+\infty)\times B(n)\times B(n)$$
. Hence,

$$|(T(t)f)(x) - (T_n(t)f)(x)| = \left| \int_{\mathbb{R}^N} \left(G(t, x, y) - G_n(t, x, y) \chi_{B(n)} \right) f(y) dy \right|$$

$$\leq \int_{\mathbb{R}^N} (G(t, x, y) - G_n(t, x, y) \chi_{B(n)}) |f(y)| dy$$

$$= ((T(t) - T_n(t))|f|) (x),$$

for any t > 0 and any $x \in B(n)$. Hence, taking (7.2.1) into account, we get

$$|T_n(t)g_n(t) - T(t)g(t)| \le |T_n(t)(g_n(t) - g(t))| + |(T(t) - T_n(t))g(t)|$$

$$\le T_n(t)(|g_n(t) - g(t)|) + |(T(t) - T_n(t))|g(t)|$$

$$\le T(t)(|g_n(t) - g(t)|) + M(T(t) - T_n(t))\mathbf{1}, \quad (7.2.2)$$

for any $n \in \mathbb{N}$.

Let us observe that the second term in the last side of (7.2.2) converges to zero uniformly in $[\varepsilon, T] \times B(k)$ for any $\varepsilon, k > 0$, by Theorem 2.2.1. To estimate the first term we observe that, for any $t \in [\varepsilon, T - \varepsilon]$ and any $x \in \mathbb{R}^N$, we have

$$\left(T(t)\left(|g_n(t) - g(t)|\right)\right)(x) \le \left(T(t)\left(\sup_{t \in [\varepsilon, T - \varepsilon]} |g_n(t) - g(t)|\right)\right)(x).$$
(7.2.3)

Since $\sup_{t\in[\varepsilon,T-\varepsilon]}|g_n(t)-g(t)|$ is a bounded and continuous function in \mathbb{R}^N , converging to 0, locally uniformly in \mathbb{R}^N , from Proposition 2.2.9 we deduce that the right-hand side of (7.2.3) converges to 0, locally uniformly, as n tends to $+\infty$, and the conclusion follows.

Now, we prove the second part of the assertion. Let f_{ε} , f_0 be as in the statement of the lemma; we have

$$|T(t-\varepsilon)f_{\varepsilon}-T(t)f_{0}| \leq |T(t-\varepsilon)f_{\varepsilon}-T(t-\varepsilon)f_{0}| + |T(t-\varepsilon)f_{0}-T(t)f_{0}|, (7.2.4)$$

for any $\varepsilon > 0$. By Proposition 2.2.9, for any fixed t > 0, the first term in the right-hand side of (7.2.4) converges to 0 as ε tends to 0, uniformly in $[0,T] \times B(k)$ for any T,k > 0. The second term in the right-hand side of (7.2.4) converges to zero uniformly in $[0,T] \times B(k)$, for any k > 0, as well, since the function $T(t)f_0$ is continuous in $[0,T] \times B(k)$, for any T,k > 0.

We can now prove our estimates. For notational convenience, we set

$$\hat{\omega}_k = \hat{\omega}_{k,p} = \begin{cases} 0, & \text{if Hypotheses 6.1.1 hold} \\ \omega_{k,p}, & \text{if Hypotheses 7.1.3 hold} \end{cases}, \quad k = 1, 2, 3,$$

where $\omega_{k,p}$ are defined in the proof of Theorem 7.1.5. Moreover, we denote by $M_k = M_{k,p}$ both the constants in (7.1.1) and in (7.1.12). Finally, we set

 $\tilde{L}=L_2$ if Hypotheses 6.1.1 are satisfied, and $\tilde{L}=L$ if Hypotheses 7.1.3 are satisfied.

Theorem 7.2.2 Let $k \in \{1, 2, 3\}$ and suppose that either Hypotheses 6.1.1(i) – 6.1.1(iii) and 6.1.1(iv-k) or Hypotheses 7.1.3(i) and 7.1.3(ii-k) hold. Then, for any $f \in C_b^{k-1}(\mathbb{R}^N)$ we have

$$| (D^{k}T(t)f)(x)|^{p}$$

$$\leq \left(\frac{\hat{\omega}_{k,p\wedge2}}{1 - e^{-\hat{\omega}_{k,p\wedge2}t}} \psi_{k,p\wedge2}(t)\right)^{\max\{1,\frac{p}{2}\}} \left(T(t)\left(\sum_{j=0}^{k-1} |D^{j}f|^{2}\right)^{\frac{p}{2}}\right)(x),$$
 (7.2.5)

for any t > 0, any $x \in \mathbb{R}^N$ and any $p > \hat{p}$, where $\hat{p} = 1$, under Hypotheses 6.1.1, and $\hat{p} = p_0$, under Hypothesis 7.1.3, whereas

$$\psi_{1,r}(t) = b_{r,1}t^{1-\frac{p}{2}}(1+t)^{\frac{p}{2}}, \qquad t > 0,$$

for any $r \in (1, 2]$, and

$$\psi_{k,r}(t) = b_{r,k} \left(1 + \frac{e^{\hat{\omega}_{k-1}t} - 1}{\hat{\omega}_{k-1}} \right)^{\frac{r}{2}} \left(t + \frac{e^{\hat{\omega}_{k-1}t} - 1}{\hat{\omega}_{k-1}} \right)^{1 - \frac{r}{2}},$$

for any t > 0, any $r \in (1,2]$ and any k = 1,2,3, $b_{k,r}$ ($r \in (1,2]$, k = 2,3) being positive constants that can be explicitly determined from the data (see (7.2.23) and (7.2.25)). When $\hat{\omega}_{k-1} = 0$ we agree that $(1 - e^{-\hat{\omega}_{k-1}t})/\hat{\omega}_{k-1} = t$. Finally, a = 1, under Hypotheses 6.1.1, and $a = \alpha$ under Hypotheses 7.1.3.

Proof. We first consider the case when k=3 and $p\in (\hat{p},2)$. We fix $\delta,t,n\in\mathbb{N}$ and let $\theta_n:\mathbb{R}^N\to\mathbb{R}$ be the cut-off function defined by $\vartheta_n(x)=\varrho(|x|/n)$, where $\varrho\in C^\infty([0,+\infty))$ is any nonincreasing function such that $\chi_{(0,1/2)}\leq \varphi\leq \chi_{(0,1)}$. For any $\alpha,\beta>0$ and any $f\in C^2_b(\mathbb{R}^N)$ we define the function $g_\delta:[0,t]\to C(B(n))$ by

$$= T_n(s) \left\{ \left(\alpha |T_n(t-s)f|^2 + \beta \vartheta_n^2 |DT_n(t-s)f|^2 + \vartheta_n^4 |D^2 T_n(t-s)f|^2 + \delta \right)^{\frac{p}{2}} - \delta^{\frac{p}{2}} \right\},$$

$$(7.3)$$

(7.2.6)

for any $0 \le s \le t$, where $\{T_n(\cdot)\}$ is the semigroup generated in $C_b(\overline{B}(n))$ by the realization A_n of the operator \mathcal{A} with homogeneous Dirichlet boundary conditions (see Section C). To simplify the notation, throughout the remainder of the proof, when there is no damage of confusion, we drop out the dependence of the functions considered on n. Moreover, we set $\varphi(r) = \varphi_n(r) := T_n(t-r)f$ for any $r \in [0, t]$.

As it is easily seen the function

$$\left(\alpha|\varphi(r)|^2 + \beta\vartheta^2|D\varphi(r)|^2 + \vartheta^4|D^2T\varphi(r)|^2 + \delta\right)^{\frac{p}{2}} - \delta^{\frac{p}{2}}$$

belongs to $D(A_n)$ for any $r \in [0, t)$. Recalling that $\mathcal{A}(\delta^{p/2}) = 0$, we get

$$\begin{split} g_{\delta}'(s) &= pT_{n}(s) \left[\left(\alpha \sum_{i,j=1}^{N} q_{ij} D_{i} \varphi D_{j} \varphi + \beta \vartheta^{2} \sum_{i,j,h=1}^{N} q_{ij} D_{ih} \varphi D_{jh} \varphi \right. \\ &+ \vartheta^{4} \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} \varphi D_{jhk} \varphi + \frac{\beta}{2} \mathcal{A}(\vartheta^{2}) |D\varphi|^{2} + \frac{1}{2} \mathcal{A}(\vartheta^{4}) |D^{2}\varphi|^{2} \right. \\ &+ 4\beta \vartheta \sum_{i,j,h=1}^{N} q_{ij} D_{i} \vartheta D_{h} \varphi D_{jh} \varphi + 8 \vartheta^{3} \sum_{i,j,h,k=1}^{N} q_{ij} D_{j} \vartheta D_{hk} \varphi D_{ihk} \varphi \\ &- \beta \vartheta^{2} \sum_{i,j,h=1}^{N} D_{h} q_{ij} D_{h} \varphi D_{ij} \varphi - \beta \vartheta^{2} \sum_{j,h=1}^{N} D_{h} b_{j} D_{j} \varphi D_{h} \varphi \\ &- 2 \vartheta^{4} \sum_{i,j,h,k=1}^{N} D_{h} q_{ij} D_{hk} \varphi D_{ijk} \varphi - 2 \vartheta^{4} \sum_{j,h,k=1}^{N} D_{h} b_{j} D_{jk} \varphi D_{hk} \varphi \\ &- \vartheta^{4} \sum_{i,j,h,k=1}^{N} D_{hk} q_{ij} D_{ij} \varphi D_{hk} \varphi - \vartheta^{4} \sum_{j,h,k=1}^{N} D_{hk} b_{j} D_{j} \varphi D_{hk} \varphi \right) \\ &\times \left(\alpha |\varphi|^{2} + \beta \vartheta^{2} |D\varphi|^{2} + \vartheta^{4} |D^{2}\varphi|^{2} + \delta \right)^{\frac{p}{2} - 1} \right] \\ &- \frac{p(2-p)}{4} T_{n}(s) \left[\sum_{i,j=1}^{N} q_{ij} D_{i} \left(\alpha \varphi^{2} + \beta \vartheta^{2} |D\varphi|^{2} + \vartheta^{4} |D^{2}\varphi|^{2} \right) \\ &\times D_{j} \left(\alpha \varphi^{2} + \beta \vartheta^{2} |D\varphi|^{2} + \vartheta^{4} |D^{2}\varphi|^{2} + \delta \right)^{\frac{p}{2} - 2} \right]. \end{split}$$

Arguing as in the proof of (7.1.4) and observing that

$$(a+b)^2 \le (1+\varepsilon)a^2 + (1+\varepsilon^{-1})b^2$$

for any $a, b, \varepsilon > 0$, we easily deduce that

$$\frac{1}{4} \sum_{i,j=1}^{N} q_{ij} D_i \left(\alpha \varphi^2 + \beta \vartheta^2 |D\varphi|^2 + \vartheta^4 |D^2 \varphi|^2 \right) \times D_j \left(\alpha \varphi^2 + \beta \vartheta^2 |D\varphi|^2 + \vartheta^4 |D^2 \varphi|^2 \right)$$

$$\leq (1+\varepsilon) \left[\alpha |\varphi| \left(\sum_{i,j=1}^{N} q_{ij} D_{i} \varphi D_{j} \varphi \right)^{\frac{1}{2}} \right.$$

$$\left. + \beta \vartheta^{2} |D\varphi| \left(\sum_{i,j,h=1}^{N} q_{ij} D_{ih} \varphi D_{jh} \varphi \right)^{\frac{1}{2}} \right.$$

$$\left. + \vartheta^{4} |D^{2}\varphi| \left(\sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} \varphi D_{jhk} \varphi \right)^{\frac{1}{2}} \right]^{2}$$

$$+ \left(1 + \frac{1}{\varepsilon} \right) \left(\beta \vartheta |D\varphi|^{2} + 2\vartheta^{3} |D^{2}\varphi|^{2} \right)^{2} \sum_{i,j=1}^{N} q_{ij} D_{i} \vartheta D_{j} \vartheta$$

$$\leq (1+\varepsilon) \left[\alpha \sum_{i,j=1}^{N} q_{ij} D_{i} \varphi D_{j} \varphi + \beta \vartheta^{2} \sum_{i,j,h=1}^{N} q_{ij} D_{ih} \varphi D_{jh} \varphi$$

$$+ \vartheta^{4} \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} \varphi D_{jhk} \varphi \right]$$

$$\times \left(\alpha |\varphi|^{2} + \beta \vartheta^{2} |D\varphi|^{2} + \vartheta^{4} |D^{2}\varphi|^{2} \right)$$

$$+ \left(1 + \frac{1}{\varepsilon} \right) \left(\beta |D\varphi|^{2} + 4\vartheta^{2} |D^{2}\varphi|^{2} \right)$$

$$\times \left(\beta \vartheta^{2} |D\varphi|^{2} + \vartheta^{4} |D^{2}\varphi|^{2} \right) \sum_{i,j=1}^{N} q_{ij} D_{i} \vartheta D_{j} \vartheta.$$

Hence, taking $\varepsilon = (p-1)/(4-2p)$ we get

$$g_{\delta}'(s) \ge pT_{n}(s) \left\{ \left[\frac{p-1}{2} \alpha \sum_{i,j=1}^{N} q_{ij} D_{i} \varphi D_{j} \varphi + \frac{p-1}{2} \beta \vartheta^{2} \sum_{i,j,h=1}^{N} q_{ij} D_{ih} \varphi D_{jh} \varphi \right. \right. \\ \left. + \frac{p-1}{2} \vartheta^{4} \sum_{i,j,h,k=1}^{N} q_{ij} D_{ihk} \varphi D_{jhk} \varphi + \frac{\beta}{2} \mathcal{A}(\vartheta^{2}) |D\varphi|^{2} \right. \\ \left. + \frac{1}{2} \mathcal{A}(\vartheta^{4}) |D^{2}\varphi|^{2} + 4\beta \vartheta \sum_{i,j,h=1}^{N} q_{ij} D_{i} \vartheta D_{h} \varphi D_{jh} \varphi \right. \\ \left. + 8\vartheta^{3} \sum_{i,j,h,k=1}^{N} q_{ij} D_{j} \vartheta D_{hk} \varphi D_{ihk} \varphi - \beta \vartheta^{2} \sum_{i,j,h=1}^{N} D_{h} q_{ij} D_{h} \varphi D_{ij} \varphi \right. \\ \left. - \beta \vartheta^{2} \sum_{i,h=1}^{N} D_{h} b_{j} D_{j} \varphi D_{h} \varphi - 2 \vartheta^{4} \sum_{i,j,h,k=1}^{N} D_{h} q_{ij} D_{hk} \varphi D_{ijk} \varphi \right.$$

$$-2\vartheta^{4} \sum_{j,h,k=1}^{N} D_{h}b_{j}D_{jk}\varphi D_{hk}\varphi - \vartheta^{4} \sum_{i,j,h,k=1}^{N} D_{hk}q_{ij}D_{ij}\varphi D_{hk}\varphi$$

$$-\vartheta^{4} \sum_{j,h,k=1}^{N} D_{hk}b_{j}D_{j}\varphi D_{hk}\varphi$$

$$-\frac{3-p}{p-1}(2-p)\left(\beta|D\varphi|^{2} + 4\vartheta^{2}|D^{2}\varphi|^{2}\right) \sum_{i,j=1}^{N} q_{ij}D_{i}\vartheta D_{j}\vartheta$$

$$\times \left(\alpha|\varphi|^{2} + \beta\vartheta^{2}|D\varphi|^{2} + \vartheta^{4}|D^{2}\varphi|^{2} + \delta\right)^{\frac{p}{2}-1}$$

$$\left. \left(7.2.7\right) \right.$$

Taking Hypothesis 6.1.1(iii) and (6.1.24) into account, one can easily check that

$$\sum_{i,j=1}^{N} q_{ij}(x)D_i\vartheta(x)D_j\vartheta(x) \le C_1\kappa(x), \tag{7.2.8}$$

for any $x \in B(n)$ and some positive constant C_1 , independent of n. Moreover, in view of (6.1.23)-(6.1.25) and (7.2.8) we can write

$$\mathcal{A}(\vartheta^2) \ge -C_2 \kappa,\tag{7.2.9}$$

$$\mathcal{A}(\vartheta^4) = 2\vartheta^2 \mathcal{A}(\vartheta^2) + 2\sum_{i,j=1}^N q_{ij} D_i \vartheta^2 D_j \vartheta^2 \ge 2\vartheta^2 \mathcal{A}(\vartheta^2) \ge -2C_2 \vartheta^2 \kappa, \quad (7.2.10)$$

$$\left|\vartheta \sum_{i,j,h=1}^{N} q_{ij} D_i \vartheta D_h \varphi D_{jh} \varphi\right| \le \frac{C_3 N}{4\varepsilon_1} \kappa |D\varphi|^2 + \varepsilon_1 C_3 \vartheta^2 \kappa |D^2 \varphi|^2, \qquad (7.2.11)$$

$$\left| \vartheta^{3} \sum_{i,j,h,k=1}^{N} q_{ij} D_{j} \vartheta D_{hk} \varphi D_{ihk} \varphi \right| \leq \frac{C_{3} N}{4\varepsilon_{2}} \vartheta^{2} \kappa |D^{2} \varphi|^{2} + \varepsilon_{2} C_{3} \vartheta^{4} \kappa |D^{3} \varphi|^{2},$$

$$(7.2.12)$$

for any $\varepsilon > 0$, where C_2, C_3 are two positive constants independent of n. Now, we observe that from Hypotheses 6.1.1, or Hypotheses 7.1.3, and (7.2.8), (7.2.9)-(7.2.12), we get

$$\begin{split} g_{\delta}'(s) \\ &\geq pT_n(s) \left\{ \left[\left(\left(\frac{p-1}{2} \alpha - \frac{C_2}{2} \beta - \beta \frac{C_3 N}{\varepsilon_1} + \beta (p-2) C_1 \frac{3-p}{p-1} \right) \kappa - \beta^2 \frac{C N^2}{4\varepsilon} \kappa^a \right. \right. \\ &\left. - \vartheta^2 \left(\beta d + r \frac{N^2}{4\varepsilon} \right) \right) |D\varphi|^2 \\ &\left. + \left(\left(\beta \left(\frac{p-1}{2} - 4\varepsilon_1 C_3 \right) - C_2 - \frac{2C_3 N}{\varepsilon_2} + 4 \frac{3-p}{p-1} (p-2) C_1 \right) \kappa \right. \\ &\left. - \left(C\varepsilon N + \frac{C N^2}{2\varepsilon_3} + K_1^+ \right) \kappa^a - \vartheta^2 \left(2d + r\varepsilon N \right) \right) \vartheta^2 |D^2 \varphi|^2 \\ &\left. + \left[\left(\frac{p-1}{2} - 8\varepsilon_2 C_3 \right) \kappa - 2C\varepsilon_3 N \kappa^a \right] \vartheta^4 |D^3 \varphi|^2 \right] \\ &\times \left(\alpha |\varphi|^2 + \beta \vartheta^2 |D\varphi|^2 + \vartheta^4 |D^2 \varphi|^2 + \delta \right)^{\frac{p}{2}-1} \right\}, \end{split}$$

where a=1, under Hypotheses 6.1.1, and $a=\sigma$, under Hypotheses 7.1.3. We now choose $\varepsilon=2\tilde{L}/N$, $\varepsilon_1=2\varepsilon_2=(p-1)/(16C_3)$, $\varepsilon_3=(p-1)\kappa_0^{1-a}/(16CN)$ and $\beta\geq N^3/(8\tilde{L}^2)$. With this choice of $\varepsilon,\varepsilon_1,\varepsilon_2,\varepsilon_3$ and β we get

$$g_{\delta}'(s) \ge pT_n(s) \left\{ \kappa \left(H_1(\beta, p) |D\varphi|^2 + H_2(\beta, p) \vartheta^2 |D^2 \varphi|^2 + \frac{p-1}{8} \kappa_0^{1-a} \vartheta^4 |D^3 \varphi|^2 \right) \right.$$
$$\left. \times (\alpha |\varphi|^2 + \beta \vartheta^2 |D\varphi|^2 + \vartheta^4 |D^2 \varphi|^2 + \delta)^{\frac{p}{2} - 1} \right\},$$

with

$$H_1(\alpha, \beta, p) := \frac{p-1}{2}\alpha - \frac{C_2}{2}\beta - 16\frac{C_3^2N}{p-1}\beta + \frac{3-p}{p-1}(p-2)C_1\beta - \frac{CN^3}{8\tilde{L}}\kappa_0^{a-1}\beta^2 - A\beta,$$
 (7.2.13)

$$H_2(\beta, p) := \frac{p-1}{4}\beta - C_2 - \frac{64C_3^2N}{p-1} + 4\frac{3-p}{p-1}(p-2)C_1 - \left(\frac{8C^2N^3}{p-1}\kappa_0^{a-1} + K_1^+ + 2C\tilde{L}\right)\kappa_0^{a-1} - 2A, \quad (7.2.14)$$

where $A = (L_3)^+$, under Hypotheses 6.1.1, and A = 0 otherwise.

It is immediate to notice that, both in the case when a=1 and $a=\sigma$, we can fix α, β sufficiently large so that $H_1(\alpha, \beta, p)$ and $H_2(\beta, p)$ are both positive and $\beta \geq N^3/(8\tilde{L}^2)$. Therefore, for such α 's and β 's we get

$$g_{\delta}'(s) \ge c_p T_n(s) \left(\vartheta^4 \left(|D\varphi|^2 + |D^2 \varphi|^2 + |D^3 \varphi|^2 \right) \right) \times \left(\alpha \varphi^2 + \beta \vartheta^2 |D\varphi|^2 + \vartheta^4 |D^2 \varphi|^2 + \delta \right)^{\frac{p}{2} - 1}, \quad (7.2.15)$$

where

$$c_p = c_p(\alpha, \beta) := p\kappa_0 \min\left\{H_1(\alpha, \beta, p), H_2(\beta, p), \frac{p-1}{8}\kappa_0^{1-a}\right\}.$$
 (7.2.16)

Then, integrating (7.2.15) in $[\varepsilon, t - \varepsilon]$ ($\varepsilon > 0$), and recalling that $\{T_n(t)\}$ is a positive semigroup, we get

$$c_{p} \int_{\varepsilon}^{t-\varepsilon} \left(T_{n}(s) \left(\vartheta^{4}(|D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2}) \right) \times (\alpha \varphi^{2} + \beta \vartheta^{2}|D\varphi|^{2} + \vartheta^{4}|D^{2}\varphi|^{2} + \delta)^{\frac{p}{2}-1} \right) (x) ds$$

$$\leq \left(T_{n}(t-\varepsilon) \left((\alpha|T_{n}(\varepsilon)f|^{2} + \beta \vartheta^{2}|DT_{n}(\varepsilon)f|^{2} + \vartheta^{4}|D^{2}T_{n}(\varepsilon)f|^{2} + \delta)^{\frac{p}{2}} - \delta^{\frac{p}{2}} \right) (x).$$

$$(7.2.17)$$

Now, we observe that the same arguments as in Remark 6.1.5 show that $D^{\gamma}\varphi_n$ converges to $D^{\gamma}T(t-\cdot)f$ uniformly in $[\varepsilon,t-\varepsilon]\times B(k)$, as n tends to $+\infty$, for any $\varepsilon\in(0,t/2)$, any $k\in\mathbb{N}$ and any $|\gamma|\leq 3$. Therefore, applying Lemma 7.2.1 we get

$$\lim_{n \to \infty} \left(T_n(s) \left(\vartheta_n^4 (|D\varphi_n(s)|^2 + |D^2 \varphi_n(s)|^2 + |D^3 \varphi_n(s)|^2 \right) \right.$$

$$\times \left(\alpha |D\varphi_n(s)|^2 + \beta \vartheta_n^2 |D\varphi_n(s)|^2 + \vartheta_n^4 |D^2 \varphi_n(s)|^2 + \delta \right)^{\frac{p}{2} - 1} \right) \right) (x)$$

$$= \left(T(s) \left((|DT(t - s)f|^2 + |D^2 T(t - s)f|^2 + |D^3 T(t - s)f|^2 \right) \right.$$

$$\times \left(\alpha |DT(t - s)f|^2 + \beta |DT(t - s)f|^2 \right.$$

$$\left. + |D^2 T(t - s)f|^2 + \delta \right)^{\frac{p}{2} - 1} \right) \right) (x),$$

the convergence being uniform in $[\varepsilon, t - \varepsilon] \times B(k)$ for any k > 0, and

$$\lim_{n \to +\infty} (T_n(t-\varepsilon)((\alpha|T_n(\varepsilon)f|^2 + \beta\vartheta_n^2|DT_n(\varepsilon)f|^2 + \vartheta_n^4|D^2T_n(\varepsilon)f|^2)^{\frac{p}{2}}))(x)$$

$$= (T(t-\varepsilon)((\alpha|T(\varepsilon)f|^2 + \beta|DT(\varepsilon)f|^2 + |D^2T(\varepsilon)f|^2)^{\frac{p}{2}}))(x),$$

uniformly for $s \in [\varepsilon, t - \varepsilon]$. Then, from (7.2.17) we get

$$c_{p} \int_{\varepsilon}^{t-\varepsilon} \left(T(s) \left((|DT(t-s)f|^{2} + |D^{2}T(t-s)f|^{2} + |D^{3}T(t-s)f|^{2} \right) \right) \times \left(\alpha |T(t-s)f|^{2} + \beta |DT(t-s)f|^{2} + |D^{2}T(t-s)f|^{2} + \delta \right)^{\frac{p}{2}-1} \right) (x) ds$$

$$\leq \left(T(t-\varepsilon) \left(\alpha |T(\varepsilon)f|^{2} + \beta |DT(\varepsilon)f|^{2} + |D^{2}T(\varepsilon)f|^{2} \right)^{\frac{p}{2}} \right) (x),$$

$$(7.2.18)$$

for any $x \in \mathbb{R}^N$. Now, from Proposition 7.1.1 it follows that

$$\lim_{\varepsilon \to 0^+} D^j T(\varepsilon) f = D^j f, \qquad j = 0, 1, 2,$$

uniformly in B(k) for any k > 0. Therefore, from Lemma 7.2.1 we have

$$\lim_{\varepsilon \to 0^+} \left(T(t - \varepsilon) \left((\alpha |T(\varepsilon)f|^2 + \beta |DT(\varepsilon)f|^2 + |D^2T(\varepsilon)f|^2)^{\frac{p}{2}} \right) \right) (x)$$

$$= \left(T(t) \left((\alpha |f|^2 + \beta |Df|^2 + |D^2f|^2)^{\frac{p}{2}} \right) \right) (x),$$
(7.2.19)

for any $x \in \mathbb{R}^N$.

As far as the first integral term in (7.2.18) is concerned, we observe that, by virtue of Theorem 7.1.2, for any $x \in \mathbb{R}^N$ the function

$$s \mapsto \left(T(s) \Big(\left(|DT(t-s)f|^2 + |D^2T(t-s)f|^2 + |D^3T(t-s)f|^2 \right)^{\frac{p}{2}} \right) \Big) \left(x\right)$$

behaves as $\widehat{C}(t-s)^{-p/2}$ for some positive constant \widehat{C} , independent of x. Therefore, the dominated convergence theorem and (7.2.19) imply that we can take the limit as ε tends to 0^+ in (7.2.18), getting

$$c_{p} \int_{0}^{t} \left(T(s) \left(\left(|DT(t-s)f|^{2} + |D^{2}T(t-s)f|^{2} + |D^{3}T(t-s)f|^{2} \right) \right) \times \left(\alpha |T(t-s)f|^{2} + \beta |DT(t-s)f|^{2} + |D^{2}T(t-s)f|^{2} + \delta \right)^{\frac{p}{2}-1} \right) (x) ds$$

$$\leq \left(T(t) \left((\alpha f^{2} + \beta |Df|^{2} + |D^{2}f|^{2})^{\frac{p}{2}} \right) \right) (x),$$
(7.2.20)

for any $x \in \mathbb{R}^N$.

For notational convenience we now set $\varphi = T(t-s)f$. Let us observe that, from (7.1.1) or (7.1.12) (with k=3), from the Young inequality and the inequality

$$T(t)(fg) \le (T(t)(f^q))^{\frac{1}{q}} (T(t)(g^r))^{\frac{1}{r}},$$

holding for any positive $f, g \in C_b(\mathbb{R}^N)$ and any r, q > 0 such that 1/r + 1/q = 1 (see Theorem 2.2.5), we deduce that, for any $\gamma \in \mathbb{R}$,

$$(|DT(t)f|^{2} + |D^{2}T(t)f|^{2} + |D^{3}T(t)f|^{2})^{\frac{p}{2}}$$

$$= (|DT(s)(T(t-s)f)|^{2} + |D^{2}T(s)(T(t-s)f)|^{2} + |D^{3}T(s)(T(t-s)f)|^{2})^{\frac{p}{2}}$$

$$\leq M_{3}e^{\hat{\omega}_{3}s}T(s) \left[\left(\frac{a-\sigma}{1-\sigma}|\varphi|^{2} + |D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2} \right)^{\frac{p}{2}} \right]$$

$$\leq M_{3}e^{\hat{\omega}_{3}s}T(s) \left[\left(\frac{a-\sigma}{1-\sigma}|\varphi|^{2} + |D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2} \right)^{\frac{p}{2}}$$

$$\times \left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta \right)^{-\gamma}$$

$$\times \left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta \right)^{\gamma} \right]$$

$$\leq M_3 e^{\hat{\omega}_3 s} \left\{ T(s) \left(\left(\alpha |\varphi|^2 + \beta |D\varphi|^2 + |D^2 \varphi|^2 + \delta \right)^{\frac{2\gamma}{2-p}} \right) \right\}^{1-\frac{p}{2}}$$

$$\times \left\{ T(s) \left[\left(\frac{a-\sigma}{1-\sigma} |\varphi|^2 + |D\varphi|^2 + |D^2 \varphi|^2 + |D^3 \varphi|^2 \right) \right.$$

$$\left. \times \left(\alpha |\varphi|^2 + \beta |D\varphi|^2 + |D^2 \varphi|^2 + \delta \right)^{-\frac{2\gamma}{p}} \right] \right\}^{\frac{p}{2}}.$$

Choosing $\gamma = p(2-p)/4$, using the Jensen and Young inequalities, we get

$$(|DT(t)f|^{2} + |D^{2}T(t)f|^{2} + |D^{3}T(t)f|^{2})^{\frac{p}{2}}$$

$$\leq M_{3}e^{\hat{\omega}_{3}s} \left\{ T(s) \left((\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta)^{\frac{p}{2}} \right) \right\}^{1-\frac{p}{2}}$$

$$\times \left\{ T(s) \left[\left(\frac{a - \sigma}{1 - \sigma} |\varphi|^{2} + |D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2} \right) \right.$$

$$\times \left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta)^{\frac{p}{2} - 1} \right] \right\}^{\frac{p}{2}}$$

$$\leq M_{3}e^{\hat{\omega}_{3}s} \left\{ \frac{p}{2} \varepsilon^{\frac{2}{p}} T(s) \left[\left(\frac{a - \sigma}{1 - \sigma} |\varphi|^{2} + |D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2} \right) \right.$$

$$\times \left(\alpha \varphi^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta)^{\frac{p}{2} - 1} \right]$$

$$+ \left(1 - \frac{p}{2} \right) \varepsilon^{\frac{2}{p - 2}} T(s) \left[\left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta \right)^{\frac{p}{2}} \right] \right\}$$

$$\leq M_{3}e^{\hat{\omega}_{3}s} \left\{ \frac{p}{2} \varepsilon^{\frac{p}{p}} T(s) \left[\left(|D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2} \right) \right.$$

$$\times \left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta \right)^{\frac{p}{2} - 1} \right]$$

$$+ \left(1 - \frac{p}{2} \right) \varepsilon^{\frac{2}{p - 2}} T(s) \left[\left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta \right)^{\frac{p}{2}} \right]$$

$$+ \frac{p}{2} \varepsilon^{\frac{p}{p}} \frac{a - \sigma}{1 - \sigma} \frac{1}{\alpha} T(s) \left[\left(\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{3}\varphi|^{2} + \delta \right)^{\frac{p}{2}} \right] \right\}.$$

$$(7.2.21)$$

Hence, recalling that

$$(|\varphi|^{2} + |D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta)^{\frac{p}{2}}$$

$$\leq M_{2}e^{\hat{\omega}_{2}(t-s)}T(t-s)\left((f^{2} + |Df|^{2} + |D^{2}f|^{2} + \delta)^{\frac{p}{2}}\right) + \delta^{\frac{p}{2}}$$

$$+ \frac{1-a}{1-\sigma}T(t-s)(|f|^{p}),$$

(see (7.1.8) and (7.1.17)), from (7.2.21) we get, for any $\varepsilon > 0$,

$$(|DT(t)f|^{2} + |D^{2}T(t)f|^{2} + |D^{3}T(t)f|^{2})^{\frac{p}{2}}$$

$$\leq M_{3}e^{\hat{\omega}_{3}s} \left\{ \frac{p}{2} \varepsilon^{\frac{2}{p}} T(s) \left[(|D\varphi|^{2} + |D^{2}\varphi|^{2} + |D^{3}\varphi|^{2}) \times (\alpha|\varphi|^{2} + \beta|D\varphi|^{2} + |D^{2}\varphi|^{2} + \delta)^{\frac{p}{2}-1} \right]$$

$$+ \left(1 - \frac{p}{2}\right) \max(1, \alpha, \beta)^{\frac{p}{2}} M_{2} \varepsilon^{\frac{2}{p-2}} e^{\hat{\omega}_{2}(t-s)}$$

$$\times T(t) \left[(|f|^{2} + |Df|^{2} + |D^{2}f|^{2} + \delta)^{\frac{p}{2}} \right]$$

$$+ \left(1 - \frac{p}{2}\right) \max(1, \alpha, \beta)^{\frac{p}{2}} \varepsilon^{\frac{2}{p-2}} \delta^{\frac{p}{2}}$$

$$+ \left(1 - \frac{p}{2}\right) \max(1, \alpha, \beta)^{\frac{p}{2}} \frac{1 - a}{1 - \sigma} \varepsilon^{\frac{2}{p-2}} T(t) (|f|^{p})$$

$$+ \frac{p}{2} \varepsilon^{\frac{p}{p}} \frac{a - \sigma}{1 - \sigma} \frac{1}{\alpha} \max(1, \alpha, \beta)^{\frac{p}{2}} M_{2} e^{\hat{\omega}_{2}(t-s)}$$

$$\times T(t) \left[(f^{2} + |Df|^{2} + |D^{2}f|^{2} + \delta)^{\frac{p}{2}} \right]$$

$$+ \frac{p}{2} \varepsilon^{\frac{p}{p}} \frac{a - \sigma}{1 - \sigma} \frac{1}{\alpha} \delta^{\frac{p}{2}} \max(1, \alpha, \beta)^{\frac{p}{2}} \right\}$$

$$(7.2.22)$$

Now, we multiply the first and the last side of (7.2.22) by $e^{-\hat{\omega}_3 s}$, integrate in (0,t), pointwise with respect to $x \in \mathbb{R}^N$, taking (7.2.20) into account and taking $\alpha \geq 1$. Finally, we let δ tend to 0^+ . We get

$$\begin{split} &\frac{1 - e^{-\hat{\omega}_3 t}}{\hat{\omega}_3} \left(|DT(t)f|^2 + |D^2T(t)f|^2 + |D^3T(t)f|^2 \right)^{\frac{p}{2}} \\ & \leq M_3 \max(\alpha, \beta)^{\frac{p}{2}} \left\{ \frac{p}{2} \left(\frac{1}{c_p} + \frac{a - \sigma}{1 - \sigma} M_2 \frac{e^{\hat{\omega}_2 t} - 1}{\hat{\omega}_2} \right) \varepsilon^{\frac{2}{p}} \right. \\ & \qquad \qquad + \left(1 - \frac{p}{2} \right) \left(M_2 \frac{e^{\hat{\omega}_2 t} - 1}{\hat{\omega}_2} + \frac{1 - a}{1 - \sigma} t \right) \varepsilon^{\frac{2}{p - 2}} \right\} \\ & \times T(t) \left((f^2 + |Df|^2 + |D^2f|^2)^{\frac{p}{2}} \right) \end{split}$$

Minimizing with respect to $\varepsilon > 0$ gives

$$\frac{1 - e^{-\hat{\omega}_3 t}}{\hat{\omega}_3} \left(|DT(t)f|^2 + |D^2T(t)f|^2 + |D^3T(t)f|^2 \right)^{\frac{p}{2}} \\
\leq M_3 \max\{\alpha, \beta\}^{\frac{p}{2}} \left(\frac{1}{c_p} + \frac{a - \sigma}{1 - \sigma} M_2 \frac{e^{\hat{\omega}_2 t} - 1}{\hat{\omega}_2} \right)^{\frac{p}{2}} \left(M_2 \frac{e^{\hat{\omega}_2 t} - 1}{\hat{\omega}_2} + \frac{1 - a}{1 - \sigma} t \right)^{1 - \frac{p}{2}} \\
\times T(t) \left((f^2 + |Df|^2 + |D^2f|^2)^{\frac{p}{2}} \right).$$

Now, (7.2.5) with $p \in (1,2)$ follows taking

$$b_{3,p} = M_3 \max \left\{ M_2, \frac{1-a}{1-\sigma} \right\}^{1-\frac{p}{2}}$$

$$\times \min \left\{ \max(\alpha, \beta)^{\frac{p}{2}} \max \left(\frac{1}{c_p(\alpha, \beta)}, \frac{a-\sigma}{1-\sigma} M_2 \right)^{\frac{p}{2}} :$$

$$\alpha \ge 1, \ H_1(\alpha, \beta, p), H_2(\beta, p) > 0 \right\}, \tag{7.2.23}$$

where $H_1(\alpha, \beta, p)$ and $H_2(\beta, p)$ are given by (7.2.13) and (7.2.14). Note that the previous minimum exists since the function

$$(\alpha, \beta) \mapsto \max\{\alpha, \beta\} (c_p(\alpha, \beta))^{-\frac{p}{2}}$$

with domain $D = \{(\alpha, \beta) \in \mathbb{R}^2 : H_j(\alpha, \beta) > 0, \ j = 1, 2\}$ tends to $+\infty$ both as $\|(\alpha, \beta)\|$ tends to $+\infty$ and as (α, β) tends to any point on ∂D .

The case when p=2 is similar and even simpler. It suffices to apply the previous arguments to the function g_0 defined by (7.2.6) (with $\delta=0$ and p=2).

In the case when p > 2, (7.2.5) (with k = 3) can be obtained from the case when p = 2 observing that

$$|(D^{3}T(t)f)(x)|^{p} = (|(D^{3}T(t)f)(x)|^{2})^{\frac{p}{2}}$$

$$\leq \left(\frac{\hat{\omega}_{3,2}}{1 - e^{-\hat{\omega}_{3,2}t}}\psi_{3,2}(t)\right)^{\frac{p}{2}}$$

$$\times \left(\left(T(t)\left(f^{2} + |Df|^{2} + |D^{2}f|^{2}\right)\right)(x)\right)^{\frac{p}{2}}$$

and, then, applying the Hölder inequality.

The proof of (7.2.5) in the case when k=1,2 is completely similar. In the case when k=2 one can apply the previous arguments to the function g_{δ} in (7.2.6), obtained disregarding the terms whose coefficients do not depend on α and β and setting $\beta=1$. After some computations one finds that g'_{δ} is still given by (7.2.7) with the obvious changes induced by the changes in the definition of the function g_{δ} . Therefore, taking $\varepsilon=(p-1)\kappa_0^{1-a}/(8CN)$ and $\varepsilon_1=(p-1)/(16C_3)$, one gets

$$g_{\delta}'(s) \ge pT_n(s) \left\{ \left(H_3(\alpha, p) \kappa |D\varphi|^2 + \frac{p-1}{8} \kappa_0^{1-a} \kappa^a \vartheta^2 |D^2 \varphi|^2 \right) \times (\alpha |\varphi|^2 + \vartheta^2 |D\varphi|^2 + \delta)^{\frac{p}{2}-1} \right\},$$

where

$$H_3(\alpha, p) = \frac{p-1}{2}\alpha - \frac{C_2}{2} - \frac{16C_3^2N}{p-1} + (p-2)C_1\frac{3-p}{p-1} - \frac{2C^2N^3}{p-1}\kappa_0^{2a-2} - A,$$

where, again, $A = (L_3)^+$, under Hypotheses 6.1.1, and A = 0 otherwise. Repeating the same arguments as above, one can now prove that

$$\left(\sum_{j=1}^{2} |(D^{j}T(t)f)(x)|^{2}\right)^{\frac{p}{2}} \leq \left(\frac{\omega_{2,p\wedge2}}{1 - e^{-\omega_{2,p\wedge2}t}} \psi_{2,p\wedge2}(t)\right)^{\max\{1,\frac{p}{2}\}} \left(T(t)\left(\sum_{j=0}^{1} |D^{j}f|^{2}\right)^{\frac{p}{2}}\right)(x), \tag{7.2.24}$$

for any t > 0, any $x \in \mathbb{R}^N$ and any $p \in (1, +\infty)$, where $\psi_{2,p \wedge 2}$ is as in the statement of the theorem with

$$b_{2,p} = M_2 \max \left(M_1, \frac{1-a}{1-\sigma} \right)^{1-\frac{p}{2}} \times \min \left\{ \alpha^{p/2}, \max \left(\frac{1}{c_p'(\alpha)}, \frac{a-\sigma}{1-\sigma} M_1 \right)^{\frac{p}{2}} : \alpha \ge 1, \ H_3(\alpha, p) > 0 \right\}.$$
(7.2.25)

Here,

$$c'_p(\alpha) = c'_p(\alpha, p) := p\kappa_0 \min\left\{H_3(\alpha, p), \frac{p-1}{8}\right\}.$$

In the case when k = 1, the previous arguments actually show that

$$b_{1,p} = M_1 \max\left(\frac{1}{p(p-1)}, \frac{a-\sigma}{1-\sigma}\right)^{\frac{p}{2}}$$

and

$$|(DT(t)f)(x)|^p \le \left(\frac{\omega_{1,p\wedge 2}}{1 - e^{-\omega_{1,p\wedge 2}t}}\psi_{1,p\wedge 2}(t)\right)^{\max\{1,\frac{p}{2}\}} (T(t)(|f|^p))(x), \quad (7.2.26)$$

for any t > 0 and any $x \in \mathbb{R}^N$.

Corollary 7.2.3 Let k = 2, 3. Moreover, let Hypotheses 6.1.1(i)-6.1.1(iii) and 6.1.1(iv-k), or Hypotheses 7.1.3(i) and 7.1.3(ii-k) hold. Then, for any $f \in C_b(\mathbb{R}^N)$ and any $p \in (\hat{p}, +\infty)$ (see Theorem 7.2.2), we have

$$|(D^{k}T(t)f)(x)|^{p} \leq \left(\sum_{j=0}^{k-1} k^{j} \prod_{h=0}^{j} \frac{\widetilde{\omega}_{k-h,p\wedge 2}}{1 - e^{-\widetilde{\omega}_{k-h,p\wedge 2}t}} \psi_{k-h,p\wedge 2}(t/k)\right)^{\max\{1,\frac{p}{2}\}} \times (T(t)(|f|^{p}))(x), \tag{7.2.27}$$

for any t > 0 and any $x \in \mathbb{R}^N$, where $\widetilde{\omega}_{k,r} = \frac{1}{k} \hat{\omega}_{j,r}$, and $\psi_{j,r}$ and $\hat{\omega}_{j,r}$ are as in Theorem 7.2.2.

Proof. We begin by proving (7.2.27) in the case when k = 3 and $p \le 2$. We fix $f \in C_b(\mathbb{R}^N)$ and t > 0. Applying (7.2.5) to the function T(2t/3)f we get

$$\leq \frac{|(D^3T(t)f)(x)|^p}{1 - e^{-\hat{\omega}_{3,p}t/3}} \psi_{3,p}(t/3) \times T(t/3) \left\{ \left(|T(2t/3)f|^2 + |DT(2t/3)f|^2 + |D^2T(2t/3)f|^2 \right)^{\frac{p}{2}} \right\}(x),$$

$$(7.2.28)$$

for any $x \in \mathbb{R}^N$. Now, taking (7.2.24) into account, with T(t)f replaced with T(2t/3)f, we get

$$(|T(2t/3)f|^{2} + |DT(2t/3)f|^{2} + |D^{2}T(2t/3)f|^{2})^{\frac{p}{2}}$$

$$\leq \frac{\hat{\omega}_{2,p}}{1 - e^{-\hat{\omega}_{2,p}t/3}} \psi_{2,p}(t/3)T(t/3) \left\{ (|T(t/3)f|^{2} + |DT(t/3)f|^{2})^{\frac{p}{2}} \right\}$$

$$+ T(2t/3)(|f|^{p})$$

$$\leq \frac{\hat{\omega}_{2,p}}{1 - e^{-\hat{\omega}_{2,p}t/3}} \psi_{2,p}(t/3) \left[\frac{\hat{\omega}_{1,p}}{1 - e^{-\hat{\omega}_{1,p}t/3}} \psi_{1,p}(t/3) + 1 \right] T(2t/3)(|f|^{p})$$

$$+ T(2t/3)(|f|^{p}).$$

$$(7.2.29)$$

Combining (7.2.28) and (7.2.29) gives (7.2.27) for $p \le 2$. If p > 2 and k = 3, (7.2.27) follows from the case when p = 2 applying the Hölder inequality. The proof of (7.2.27) (with k = 2) is similar: we write

$$|(D^2T(t)f)(x)|^2 = |(D^2T(t/2)T(t/2)f)(x)|^2$$

and apply (7.2.5) to T(t/2)f with, respectively, k = 1 and k = 2. We omit the details.

Remark 7.2.4 The estimates in (7.2.5) (with k=1) and in Corollary 7.2.3 allow us to improve the uniform estimates for the derivatives of T(t) in Theorem 6.1.7. Indeed, they show that under Hypotheses 6.1.1(i)-6.1.1(iii) and 6.1.1(ii-k) (k=1,2,3), the sup-norm of $D^kT(t)f$ stays bounded as t tends to $+\infty$, whereas it decreases exponentially to 0 as t tends to $+\infty$ if Hypotheses 7.1.3(i) and 7.1.3(ii-k) are satisfied.

The estimate (7.2.5) (with k = 1) can also be used to prove a Liouville type theorem. Namely, under Hypotheses 7.1.3(i) and 7.1.3(ii-1) such an estimate allows us to prove that, if $u \in C_b(\mathbb{R}^N)$ satisfies the equation Au = 0, then it is constant. As it has already been stressed in the introduction, such a result fails in general. We refer the reader to [128] for further details.

Theorem 7.2.5 Suppose that Hypotheses 7.1.3(i) and 7.1.3(ii-1) hold. If $f \in C_b(\mathbb{R}^N)$ satisfies $\mathcal{A}f = 0$, then u is constant.

Proof. We observe that if $f \in C_b(\mathbb{R}^N)$ is such that $\mathcal{A}f = 0$, then, by local elliptic regularity, $u \in D_{\max}(\mathcal{A})$ (see (2.0.1)). Moreover, for any $g \in D_{\max}(\mathcal{A})$ and any $x \in \mathbb{R}^N$, the function $t \mapsto (T(t)g)(x)$ is continuously differentiable in $[0, +\infty)$ and $(D_tT(t)g)(x) = (\mathcal{A}T(t)g)(x) = (T(t)\mathcal{A}g)(x)$ (see Lemma 2.3.3, Propositions 2.3.5, 2.3.6 and 4.1.10). Hence $D_tT(t)f = 0$ for any t > 0, so that $T(t)f \equiv f$ for any t > 0. From (7.2.5) we deduce that

$$|(DT(t)f)(x)| \le \frac{\widetilde{C}e^{-\omega t}}{\sqrt{t}} \left((T(t)f^2)(x) \right)^{\frac{1}{2}} \le \frac{\widetilde{C}e^{-\omega t}}{\sqrt{t}} ||f||_{\infty}, \qquad t > 0, \ x \in \mathbb{R}^N,$$

$$(7.2.30)$$

for some \widetilde{C} , $\omega > 0$. Letting t go to $+\infty$ in (7.2.30), we get

$$Df(x) = \lim_{t \to +\infty} (DT(t)f)(x) = 0, \qquad x \in \mathbb{R}^N,$$

so that f is constant.

7.3 Further estimates for $\mathcal{A} = \Delta + \sum_{j=1}^{N} b_j(x) D_j$

Now, we consider the particular case when $q_{ij} \equiv \delta_{ij}$ (i, j = 1, ..., N). In this case we show that, under Hypotheses 7.1.3, the gradient estimate (7.1.1) can be proved also for p = 1.

Theorem 7.3.1 Fix k = 1, 2, 3 and suppose that $q_{ij} \equiv \delta_{ij}$ (i, j = 1, ..., N). Moreover, let Hypotheses 7.1.3(i), 7.1.3(ii-k) hold with the conditions (7.1.10) and (7.1.11) being replaced with the following one:

$$d(x) + L_k r(x) \le K, \qquad x \in \mathbb{R}^N, \tag{7.3.1}$$

to be satisfied for some constants $L_k > 0$ and $K \in \mathbb{R}$. Then,

$$|(D^k T(t)f)(x)| \le M_k e^{\omega_k t} \left(T(t) \left(\sum_{j=1}^k |D^j f|^2 \right)^{\frac{1}{2}} \right) (x), \qquad t > 0, \ x \in \mathbb{R}^N,$$
(7.3.2)

where $M_k > 0$ and ω_k are constants which can be explicitly determined (see the proof). In particular, $\omega_k < 0$ (k = 2, 3) if K < 0.

Finally, if k = 1, assume that the coefficient b_j belongs to $C_{loc}^{1+\delta}(\mathbb{R}^N)$ for some $\delta \in (0,1)$ and any $j = 1, \ldots, N$, and (7.1.9) is satisfied with the function d being replaced with a real constant d_0 . Then,

$$|(DT(t)f)(x)| \le e^{d_0t}(T(t)|Df|)(x), \qquad t > 0, \ x \in \mathbb{R}^N.$$
 (7.3.3)

Proof. We limit ourselves to sketching the proof of (7.3.2) in the case when k=3, since the proof of (7.3.2) (with k=2) and (7.3.3) are similar and even simpler. The proof that we provide is similar to that of Theorem 7.1.2. For the reader's convenience we go into details.

Let $\delta > 0$, $f \in C_b^1(\mathbb{R}^N)$ and consider the function $w_\delta : (0, +\infty) \times \mathbb{R}^N \to \mathbb{R}$ defined by

$$w_{\delta}(t,x) = (\alpha |Du(t,x)|^2 + \beta |D^2u(t,x)|^2 + |D^3u(t,x)|^2 + \delta)^{\frac{1}{2}},$$

where, as usual, $u = T(\cdot)f$, and α, β are positive constants to be fixed later on. Arguing as in the proof of Theorem 7.1.2 and taking the Cauchy-Schwarz inequality into account, we can easily show that w_{δ} turns out to solve the Cauchy problem

$$\begin{cases} D_t w_{\delta}(t, x) = \mathcal{A}w_{\delta}(t, x) + g_{\delta}(t, x), & t > 0, x \in \mathbb{R}^N, \\ w_{\delta}(0, x) = \left(\alpha |Df(x)|^2 + \beta |D^2 f(x)|^2 + |D^3 f(x)|^2 + \delta\right)^{\frac{1}{2}}, & x \in \mathbb{R}^N, \end{cases}$$

where (see (7.1.6))

$$g_{\delta} \leq w_{\delta}^{-1} \left(\alpha \sum_{j,h=1}^{N} D_{h} b_{j} D_{j} u D_{h} u + 2\beta \sum_{j,h,k=1}^{N} D_{h} b_{j} D_{jk} u D_{hk} u \right)$$

$$+3 \sum_{j,h,k,l=1}^{N} D_{h} b_{j} D_{jkl} u D_{hkl} u + \beta \sum_{j,h,k=1}^{N} D_{hk} b_{j} D_{j} u D_{hk} u$$

$$+3 \sum_{j,h,k,l=1}^{N} D_{hk} b_{j} D_{jl} u D_{hkl} u + \sum_{j,h,k=1}^{N} D_{hkl} b_{j} D_{j} u D_{hkl} u \right).$$

From Hypotheses 7.1.3(ii-3) and (7.3.1), and arguing as in the proof of (7.1.13), we get for any $\varepsilon > 0$

$$g_{\delta} \leq w_{\delta}^{-1} \left\{ \left(\alpha d + \frac{N^2}{4\varepsilon} \beta r + \frac{N^3}{4\varepsilon_1} r \right) |Du|^2 + \left(\beta (2d + \varepsilon Nr) + \frac{3N^2}{4\varepsilon_1} r \right) |D^2 u|^2 + (3d + 4\varepsilon_1 rN) |D^3 u|^2 \right\}.$$

We choose $\varepsilon = 4\varepsilon_1/3 = L_3/N$ and, then, α and β such that

$$\beta > \frac{N^3}{L_3^2}, \qquad \alpha > \frac{N^3}{24L_3^2}(3\beta + 8N).$$

With these choices of ε , ε_1 , α , β we get

$$g_{\delta} \leq \left(\alpha |Du|^{2} + \beta |D^{2}u|^{2} + |D^{3}u|^{2} + \delta\right)^{-\frac{1}{2}} \times (\alpha K |Du|^{2} + 2K |D^{2}u|^{2} + K |D^{3}u|^{2})$$

$$\leq \max\{K, 2K/\beta\}w_{\delta} - K^{-}\delta^{\frac{1}{2}}.$$

Repeating the same arguments in the last part of the proofs of Theorems 7.1.2 and 7.1.5, we can easily get (7.3.2) with $\omega_3 = \max\{K, 2K/\beta\}$.

In the case when k = 2, arguing likewise, we get $\omega_2 = \max\{K, 2K\}$.

Remark 7.3.2 Using the Jensen inequality as in the proof of Theorem 7.1.2, from (7.3.3) we get

$$|(DT(t)f)(x)|^p \le e^{d_0pt}(T(t)(|Df|^p))(x), \qquad t > 0, \ x \in \mathbb{R}^N,$$
 (7.3.4)

for any $p \in [1, +\infty)$.

Moreover, adapting the proof of Theorem 7.2.2 to this case, we can show that, for any $f \in C_b(\mathbb{R}^N)$, any t > 0 and any $x \in \mathbb{R}^N$, it holds:

$$|(DT(t)f)(x)|^{p} \le \frac{2^{p}p^{1-p/2}d_{0}}{p-1)^{p/2}(1-e^{-pd_{0}t})}t^{1-\frac{p}{2}}(T(t)(|f|^{p}))(x), \tag{7.3.5}$$

for any $p \in (1, 2]$ and

$$|(DT(t)f)(x)|^p \le \left(\frac{d_0}{1 - e^{-2d_0t}}\right)^{\frac{p}{2}} (T(t)(|f|^p))(x),$$
 (7.3.6)

for any $p \in (2, +\infty)$.

We will use this estimate in Section 8.3.

In general, the estimates (7.2.5) and (7.2.27) fail for p=1 also in the case when the coefficients of the operator \mathcal{A} are bounded. Here, we provide a simple situation in which this happens, taking $\mathcal{A} = \Delta$.

Example 7.3.3 Let $\{T(t)\}$ be the heat semigroup in \mathbb{R} , i.e.,

$$(T(t)f)(x) = \frac{1}{\sqrt{4\pi t}} \int_{\mathbb{R}} e^{-\frac{(x-y)^2}{4t}} f(y) dy, \qquad t > 0, \ x \in \mathbb{R}.$$

The space derivative of T(t)f is given by

$$(DT(t)f)(x) = \frac{1}{2t\sqrt{4\pi t}} \int_{\mathbb{R}} (y-x)e^{-\frac{(x-y)^2}{4t}} f(y)dy, \qquad t > 0, \ x \in \mathbb{R}.$$

Fix R > 0 and let $f \in C_b(\mathbb{R})$ be such that $0 \le f \le 1$, f(x) = 0 for any $x < R - R^{-1}$ and f(x) = 1 for any x > R. Then,

$$(T(t)f)(0) \le \frac{1}{\sqrt{4\pi t}} \int_{R-R^{-1}}^{+\infty} e^{-\frac{|y|^2}{4t}} dy,$$

$$(DT(t)f)(0) \ge \frac{1}{2t\sqrt{4\pi t}} \int_{R}^{+\infty} ye^{-\frac{|y|^2}{4t}} dy.$$

7.3. Further estimates for
$$A = \Delta + \sum_{j=1}^{N} b_j(x)D_j$$
 153

Therefore, $(DT(t)f)(0) \ge c_R(T(t)f)(0)$, where

$$c_R = \frac{1}{2t} \int_R^{+\infty} y e^{-\frac{|y|^2}{4t}} dy \left(\int_{R-R^{-1}}^{+\infty} e^{-\frac{|y|^2}{4t}} dy \right)^{-1}.$$

Using De L'Hôspital rule, it is readily seen that c_R tends to $+\infty$ as R goes to $+\infty$. This means that no pointwise estimates similar to (7.2.5) can hold for p=1.

Chapter 8

Invariant measures μ and the semigroup in $L^p(\mathbb{R}^N, \mu)$

8.0 Introduction

In this chapter we deal with the invariant measures of the semigroup $\{T(t)\}$ associated with the operator \mathcal{A} defined on smooth functions by

$$\mathcal{A}u(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}u(x) + \sum_{j=1}^{N} b_{j}(x)D_{j}u(x), \qquad x \in \mathbb{R}^{N}.$$
 (8.0.1)

We recall that, according to Remark 2.2.10, $\{T(t)\}$ is a semigroup of contractions in $B_b(\mathbb{R}^N)$.

Throughout the chapter, for any measure μ , we write L^p_{μ} for $L^p(\mathbb{R}^N, \mu)$ and we denote by $\|\cdot\|_p$ the norm of L^p_{μ} . Moreover, we write $W^{k,p}_{\mu}$ for $W^{k,p}(\mathbb{R}^N, \mu)$ for any $k \in \mathbb{N}$ and any $p \in [1, +\infty]$.

By definition, a probability measure μ is an invariant measure for $\{T(t)\}$ if

$$\int_{\mathbb{R}^N} T(t)f \, d\mu = \int_{\mathbb{R}^N} f \, d\mu, \tag{8.0.2}$$

for any $f \in B_b(\mathbb{R}^N)$ or, equivalently, for any $f \in C_c^\infty(\mathbb{R}^N)$ (see Lemma 8.1.3). In Section 8.1 we show that if there exists an invariant measure of $\{T(t)\}$, then the semigroup can be extended to a semigroup of bounded operators in the space L^p_μ for any $p \in [1, +\infty)$. Such an extension, which we still denote by $\{T(t)\}$, enjoys many interesting properties. First of all, it defines a strongly continuous semigroup for any $p \in [1, +\infty)$. Moreover, it is possible to describe the behaviour of the function T(t)f in L^p_μ when t approaches $+\infty$. More precisely, for any $f \in L^p_\mu$ we have

$$\lim_{t \to +\infty} \|T(t)f - \overline{f}\|_p = 0, \quad \text{ where } \ \overline{f} = \int_{\mathbb{R}^N} f \, d\mu.$$

Such a result implies a Liouville type theorem, i.e., it implies that if $u \in D(L_p)$ satisfies $L_p u = 0$ for some $p \in (1, +\infty)$, then u is constant. Here, and in all the chapter, by L_p we denote the infinitesimal generator of the semigroup in L^p_μ .

The invariant measure μ of $\{T(t)\}$, whenever existing, is unique and it is equivalent to the Lebesgue measure on the σ -algebra of the Borel sets of \mathbb{R}^N , in the sense that a Borel set A is negligible with respect to μ if and only if it is negligible with respect to the Lebesgue measure. Moreover, the density $\rho = d\mu/dx$ satisfies

$$\operatorname{ess inf}_{x \in B(r)} \rho(x) > 0, \qquad r > 0.$$

We will prove the existence of an invariant measure of $\{T(t)\}$ in three different ways and situations. First we show a classical result by Khas'minskii, which guarantees the existence of the invariant measure in terms of a Lyapunov function related to the operator \mathcal{A} . Next, we show that an invariant measure exists whenever T(t) is compact in $C_b(\mathbb{R}^N)$ for any t > 0. Finally, we consider the case when the operator \mathcal{A} is given, on smooth functions, by

$$Au(x) = \Delta u(x) - \langle DU(x) + G(x), Du(x) \rangle, \quad x \in \mathbb{R}^{N}$$
(8.0.3)

and the functions U and G belong, respectively, to $C^{1+\alpha}_{loc}(\mathbb{R}^N)$ and $C^1(\mathbb{R}^N, \mathbb{R}^N)$ for some $\alpha \in (0,1)$. We prove that if $e^{-U} \in L^1(\mathbb{R}^N)$ and $\operatorname{div} G = \langle G, DU \rangle$, then the measure

$$\mu(dx) = K^{-1}e^{-U(x)}dx, \qquad K = \int_{\mathbb{R}^N} e^{-U(x)}dx,$$

is the invariant measure of $\{T(t)\}$. Moreover, the semigroup $\{T(t)\}$ and its infinitesimal generator L_2 are symmetric in L^2_{μ} and the following formula holds:

$$\int_{\mathbb{R}^N} \langle Df, Dg \rangle d\mu = -\int_{\mathbb{R}^N} gL_2 f \, d\mu, \qquad f \in D(L_2), \quad g \in W^{1,2}_{\mu}.$$

Notice that, when $f \in C_c^{\infty}(\mathbb{R}^N)$, such a formula follows immediately from an integration by parts. The proof that μ is an invariant measure of $\{T(t)\}$ is obtained using variational methods in L_{μ}^p .

In Section 8.2 we show some regularity properties of the invariant measure μ . We first show that if the diffusion coefficients q_{ij} $(i,j=1,\ldots,N)$ are continuously differentiable in \mathbb{R}^N , then μ has a density ρ which belongs to $W^{1,p}_{\text{loc}}(\mathbb{R}^N)$. In particular, according to the Sobolev embedding theorems, ρ is a continuous function in \mathbb{R}^N . Next, we specialize to the case when

$$Au = \sum_{i,j=1}^{N} D_i(q_{ij}D_ju) + \sum_{i=1}^{N} b_iD_iu,$$

on smooth functions. Allowing both the diffusion and the drift coefficients to be unbounded in \mathbb{R}^N , we show some global L^p - and $W^{1,p}$ -regularity properties of ρ . In particular, we show that if $b_i \in L^p_\mu$ for any $i = 1, \ldots, N$, then $\rho \in C_b(\mathbb{R}^N)$. Further, in the particular case when the diffusion coefficients belong

to $C_b^1(\mathbb{R}^N)$, we prove some global $W^{1,q}$ - and $W^{2,q}$ -regularity properties of ρ . Finally, under some additional assumptions on the coefficients of the operator \mathcal{A} , we show some pointwise bounds for the function ρ .

In Section 8.3 we prove some estimates for the derivatives of T(t)f (up to the third-order) and for the gradient of the resolvent operator in the L^p_{μ} -norm. Such estimates follow integrating the pointwise estimates proved in Chapter 7.

In Section 8.4 we consider the operator \mathcal{A} in (8.0.3) when U is a convex function. In this case we can characterize the domain $D(L_2)$ and we obtain further estimates for the derivatives of the resolvent $R(\lambda, L_2)f$.

In Section 8.5 we study the compactness of the embedding $W_{\mu}^{1,p} \subset L_{\mu}^{p}$ in the symmetric case. We show that the embedding is compact whenever

$$\lim_{|x|\to +\infty} |DU(x)| = +\infty$$

and

$$|\Delta U(x)| \le \delta |DU(x)|^2 + M, \qquad x \in \mathbb{R}^N,$$

in the case when $p \geq 2$, or

$$\langle D^2U(x)DU(x),DU(x)\rangle \geq (\delta |DU(x)|^2 + M)|DU(x)|^2, \qquad x \in \mathbb{R}^N,$$

if $p \in (1,2)$. Here $\delta \in (0,1)$ and M > 0 are suitable constants.

A consequence of the compactness of the embedding of $W_{\mu}^{1,p}$ in L_{μ}^{p} is the Poincaré inequality in L_{μ}^{p} , that is the inequality

$$\int_{\mathbb{R}^N} |f - \overline{f}|^p d\mu \le C \int_{\mathbb{R}^N} |Df|^p d\mu, \qquad f \in W^{1,p}_\mu.$$

In the case when p=2, the Poincaré inequality can be proved even in the non-symmetric case, using the pointwise estimates of Chapter 7. Such an inequality has the following consequences: first it allows us to show that T(t)f converges exponentially to \overline{f} in L^2_{μ} as t tends to $+\infty$; secondly, it implies that L_2 has the spectral gap property in L^2_{μ} , that is

$$\sigma(L_2) \setminus \{0\} \subset \{\lambda \in \mathbb{C}, \text{Re } \lambda \leq -1/C\}.$$

The Poincaré inequality and the spectral gap property are discussed in Section 8.6.

Finally, in Section 8.7 we consider the logarithmic Sobolev inequality, i.e., the inequality

$$\int_{\mathbb{R}^{N}} f^{p} \log f d\mu \le ||f||_{p}^{p} \log ||f||_{p} + \frac{p}{\lambda} \int_{\mathbb{R}^{N}} f^{p-2} |Df|^{2} d\mu,$$

for positive and regular functions f. Such an inequality was first studied by Leonard Gross in [69]. For p = 2, it implies that $f^2 \log f$ is integrable for any

positive $f \in W^{1,2}_{\mu}$. This is a very sharp result, as it can be seen in the case when the invariant measure μ is a Gaussian measure (see Example 8.7.6). In particular, such an example shows that the Sobolev embedding theorems, in general, fail to hold when μ is an invariant measure.

We prove the logarithmic Sobolev inequality using the pointwise estimates of Chapter 7. Then, adapting a proof by Gross, we see that in the symmetric case the logarithmic Sobolev inequality implies the hypercontractivity of $\{T(t)\}$. This means that the operator T(t) is a contraction from L^2_μ to $L^{q(t)}_\mu$ for any t>0, where $q(t)=1+e^{\lambda t}$. This is a sharp result too, since in general T(t) is not bounded from L^2_μ to L^q_μ for any q>q(t). The hypercontractivity was first proved by Edward Nelson in [121] for the Ornstein-Uhlenbeck semigroup, and, in fact, it was the reason for which Gross studied the logarithmic Sobolev inequalities.

8.1 Existence, uniqueness and general properties

In this section, we deal with the problem of the existence and uniqueness of the invariant measure of $\{T(t)\}$ and we prove some general properties of the invariant measures. If not otherwise specified, we always assume that the coefficients of the operator \mathcal{A} satisfy the same assumptions as in Chapter 2 that we rewrite here for the reader's convenience.

Hypotheses 8.1.1 (i) $q_{ij} \equiv q_{ji}$ for any i, j = 1, ..., N and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \quad \kappa(x) > 0, \quad \xi, x \in \mathbb{R}^N;$$

(ii) q_{ij} and b_i (i, j = 1, ..., N) belong to $C_{loc}^{\alpha}(\mathbb{R}^N)$ for some $\alpha \in (0, 1)$.

8.1.1 General properties and uniqueness of the invariant measure of $\{T(t)\}$

To begin with we give a characterization of the invariant measures of $\{T(t)\}$ in terms of the weak generator \widehat{A} of the semigroup (see Section 2.3).

Proposition 8.1.2 A Borel probability measure μ is an invariant measure of $\{T(t)\}$ if and only if

$$\int_{\mathbb{R}^N} \widehat{A} f d\mu = 0, \qquad f \in D(\widehat{A}). \tag{8.1.1}$$

To prove the proposition, we first need the following lemma.

Lemma 8.1.3 A Borel probability measure μ such that

$$\int_{\mathbb{R}^N} T(t) f d\mu = \int_{\mathbb{R}^N} f d\mu, \qquad t > 0,$$
(8.1.2)

for any $f \in C_c^{\infty}(\mathbb{R}^N)$ is an invariant measure of $\{T(t)\}$.

Proof. Fix $f \in B_b(\mathbb{R}^N)$. We claim that we can approximate f with a sequence $\{f_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ which is bounded with respect to the sup-norm and such that f_n tends to f almost everywhere in \mathbb{R}^N as n tends to $+\infty$. Observe that a convolution argument with standard mollifiers proves the claim when $f \in C_b(\mathbb{R}^N)$. For a general $f \in B_b(\mathbb{R}^N)$ it suffices to observe that, according to [53, Proposition 3.4.2]), we can determine a sequence $\{f_n\} \subset C_b(\mathbb{R}^N)$ such that $\|f_n\|_{\infty} \leq C$ for any $n \in \mathbb{N}$ and $\lim_{n \to +\infty} f_n(x) = f(x)$ for almost any $x \in \mathbb{R}^N$.

Now, the assertion easily follows from the dominated convergence theorem, writing (8.1.2) with f_n instead of f and letting n go to $+\infty$. Indeed, by Remark 2.2.10, $(T(t)f_n)(x)$ converges to (T(t)f)(x) as n tends to $+\infty$, for any t > 0 and any $x \in \mathbb{R}^N$, and $\sup_{n \in \mathbb{N}} ||T(t)f_n||_{\infty} < +\infty$.

Proof of Proposition 8.1.2. First we recall that, according to (2.3.12), for any $f \in D(\widehat{A})$, any t > 0 and any $x \in \mathbb{R}^N$, we have

$$(T(t)f)(x) - f(x) = \int_0^t (T(s)\widehat{A}f)(x)ds.$$
 (8.1.3)

Now, let μ be an invariant measure of $\{T(t)\}$ and fix $f \in D(\widehat{A})$. Then

$$\int_{\mathbb{R}^N} \frac{T(t)f - f}{t} \, d\mu = 0, \qquad t > 0.$$
 (8.1.4)

By (8.1.3), recalling that $\{T(t)\}$ is a semigroup of contractions since $c \equiv 0$ (see Theorem 2.2.5), it follows that $t^{-1}|(T(t)f)(x) - f(x)| \leq \|\widehat{A}f\|_{\infty}$ for any t > 0 and any $x \in \mathbb{R}^N$. Thus (8.1.1) follows letting t tend to 0 in (8.1.4), using the dominated convergence theorem.

Conversely, assume that μ is a Borel probability measure satisfying (8.1.1), and let $f \in C_c^{\infty}(\mathbb{R}^N)$. By Lemma 2.3.3 and Proposition 2.3.6, f and T(s)f belong to $D(\widehat{A})$ and $T(s)\widehat{A}f = \widehat{A}T(s)f$ for any s > 0. Integrating (8.1.3) in \mathbb{R}^N and using the Fubini theorem, we get

$$\int_{\mathbb{R}^N} (T(t)f - f)d\mu = \int_0^t ds \int_{\mathbb{R}^N} \widehat{A}T(s)fd\mu = 0, \qquad t > 0$$

and, then, the conclusion follows from Lemma 8.1.3.

Remark 8.1.4 A probability measure μ which solves the equation (8.1.1) for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ is usually called *infinitesimally invariant*. In general an infinitesimally invariant measure μ is not an invariant measure of the semigroup $\{T(t)\}$. Indeed, in [136, Example 1.12], W. Stannat shows that the measure $d\mu = e^{-x^2}dx$ is infinitesimally invariant for the one-dimensional elliptic operator defined by

$$Au(x) = u''(x) - (2x + 6e^{x^2})u'(x), \qquad x \in \mathbb{R},$$

on smooth functions, but it is not invariant for the semigroup associated with the operator A.

Checking that the measure μ is infinitesimally invariant for the operator $\mathcal A$ is an easy task. According to the forthcoming Proposition 8.1.10, to prove that it is not invariant for the associated semigroup $\{T(t)\}$ it suffices to show that $\{T(t)\}$ is not conservative. According to Theorem 3.2.2, this is the case if $+\infty$ or $-\infty$ are accessible. In fact, we are going to prove that $+\infty$ is an entrance and $-\infty$ is an exit point. For this purpose, we show that the functions $Q, R: \mathbb R \to \mathbb R$ defined by

$$Q(x) = \frac{1}{W(x)} \int_0^x W(t) dt, \qquad R(x) = W(x) \int_0^x \frac{1}{W(t)} dt, \qquad x \in \mathbb{R},$$

where

$$W(x) = \exp\left(x^2 + 6\int_0^x e^{t^2} dt\right), \qquad x \in \mathbb{R}$$

(see (3.1.3)-(3.1.5)), are such that $Q \in L^1(0,+\infty)$, $Q \notin L^1(-\infty,0)$, $R \in L^1(-\infty,0)$, $R \notin L^1(0,+\infty)$. But this can be easily shown, observing that

$$\lim_{x\to +\infty} x^4 Q(x) = \lim_{x\to -\infty} x^4 R(x) = 0, \qquad \lim_{x\to -\infty} Q(x) = \lim_{x\to +\infty} R(x) = +\infty.$$

Let us now exploit some properties of the invariant measures.

Proposition 8.1.5 Let μ be an invariant measure of $\{T(t)\}$. Then, μ is equivalent to the Lebesgue measure m on the σ -algebra of the Borel sets of \mathbb{R}^N (in the sense that μ and the Lebesgue measure have the same sets of zero measure). Besides, for any fixed r > 0 the density ρ of μ with respect to the Lebesgue measure satisfies

$$\operatorname*{ess\,inf}_{x\in B(r)}\rho(x)>0.$$

Proof. For any Borel set $B \subset \mathbb{R}^N$, any t > 0 and any $x \in \mathbb{R}^N$ we have

$$(T(t)\chi_B)(x) = \int_B G(t, x, y) dy$$

(see Theorem 2.2.5), and, by (8.0.2),

$$\mu(B) = \int_{\mathbb{R}^N} \chi_B d\mu = \int_{\mathbb{R}^N} T(t) \chi_B d\mu.$$

Therefore, if m(B) = 0, then $(T(t)\chi_B)(x) = 0$ for any t > 0 and any $x \in \mathbb{R}^N$, since G is strictly positive in $(0, +\infty) \times \mathbb{R}^N \times \mathbb{R}^N$. Thus, $\mu(B) = 0$. Conversely, if m(B) > 0, then $(T(t)\chi_B)(x) > 0$ for any $x \in \mathbb{R}^N$ and, therefore, $\mu(B) > 0$. Thus μ and m are equivalent.

Now, let r > 0 and fix R > r. Moreover, let G_R be the Green's function in B(R) associated with the realization of the operator \mathcal{A} with homogeneous Dirichlet boundary conditions in $C(\overline{B}(R))$ (see Proposition C.3.2). Since $G_R(t,x,y) \leq G(t,x,y)$ for any R > 0, any t > 0 and any $x,y \in B(R)$ (see the proof of Theorem 2.2.5), then, for any positive function $f \in B_b(\mathbb{R}^N)$, we have

$$\int_{\mathbb{R}^N} f(x)\rho(x)dx = \int_{\mathbb{R}^N} (T(1)f)(x)\rho(x)dx$$
$$= \int_{\mathbb{R}^N} \int_{\mathbb{R}^N} G(1,x,y)f(y)\rho(x)dy dx$$
$$\geq \int_{\mathbb{R}^N} f(y) \int_{B(R)} G_R(1,x,y)\rho(x)dx dy.$$

Since f is arbitrary, we have

$$\rho(y) \ge F(y) = \int_{B(R)} G_R(1, x, y) \rho(x) dx,$$

for almost any $y \in B(R)$. Now, since $G_R(1,\cdot,\cdot)$ is positive, bounded and continuous in $B(R) \times B(R)$, it follows that the function F is positive, bounded and continuous in B(R). Therefore, $\rho(y) \ge \inf_{B(r)} F > 0$ for almost any $y \in B(r)$.

As an immediate consequence of the results in Proposition 8.1.5, we get the following corollary.

Corollary 8.1.6 Any invariant measure μ of $\{T(t)\}$ can be extended to a complete probability measure defined on the σ -algebra of all the Lebesgue measurable sets.

Moreover, since the Lebesgue measure is regular, Proposition 8.1.5 immediately implies that μ is regular as well. Hence, applying [135, Theorem 3.14], we get the following useful result.

Corollary 8.1.7 If μ is an invariant measure of $\{T(t)\}$, then $C_c^{\infty}(\mathbb{R}^N)$ is dense in L_{μ}^p for any $p \in [1, +\infty)$.

We can now prove the following result.

Proposition 8.1.8 Let μ be an invariant measure of $\{T(t)\}$. For any $p \in [1, +\infty)$, $\{T(t)\}$ extends to a strongly continuous semigroup of contractions in L^p_{μ} .

Proof. We begin the proof recalling that

$$(T(t)f)(x) = \int_{\mathbb{R}^N} f(y)p(t, x; dy), \qquad t > 0, \ x \in \mathbb{R}^N, \ f \in C_b(\mathbb{R}^N),$$

where p(t, x; dy) are positive measures satisfying $p(t, x; \mathbb{R}^N) \leq 1$ for any t > 0 and any $x \in \mathbb{R}^N$.

Using the Hölder inequality, we easily deduce that

$$|(T(t)f)(x)|^p = \left| \int_{\mathbb{R}^N} f(y)p(t,x;dy) \right|^p$$

$$\leq \int_{\mathbb{R}^N} |f(y)|^p p(t,x;dy) = (T(t)(|f|^p))(x),$$

for any $f \in C_b(\mathbb{R}^N)$, any t > 0 and any $x \in \mathbb{R}^N$. Therefore,

$$\int_{\mathbb{R}^N} |T(t)f|^p d\mu \le \int_{\mathbb{R}^N} T(t)(|f|^p) d\mu = \int_{\mathbb{R}^N} |f|^p d\mu, \qquad t > 0.$$

Thus, since $C_b(\mathbb{R}^N)$ is dense in L^p_μ (due to Corollary 8.1.7), then, for any t>0, the operator T(t) can be uniquely extended to a contraction in L^p_μ . The strong continuity of $\{T(t)\}$ in L^p_μ now is an immediate consequence of Propositions 2.2.7 and A.1.2.

As it has been pointed out in the introduction to this chapter, for any $p \in [1, +\infty)$, we denote by L_p the generator of $\{T(t)\}$ in L^p_μ . We simply write L for L_2 . From (8.0.2) it readily follows that

$$\int_{\mathbb{D}^{N}} L_{p} f d\mu = 0, \qquad f \in D(L_{p}), \quad p \in [1, +\infty).$$
 (8.1.5)

Besides, for any $p \in [1, +\infty)$ the resolvent $R(\lambda, L_p)$ is an extension of $R(\lambda)$ (see Theorem 2.1.3) to L^p_μ and

$$||R(\lambda, L_p)f||_p \le \lambda^{-1}||f||_p, \qquad f \in D(L_p), \ \lambda > 0, \ p \in [1, +\infty).$$
 (8.1.6)

As the following proposition shows, $D(\widehat{A})$ is a core of L_p .

Proposition 8.1.9 Let $\{T(t)\}$ admit an invariant measure μ . Then, $D(\widehat{A})$ is a core of L_p , for any $p \in [1, +\infty)$.

Proof. Let us begin by observing that $D(\widehat{A}) \subset D(L_p)$. This follows easily recalling that $R(\lambda, L_p)_{|C_b(\mathbb{R}^N)} = R(\lambda)$ and $R(\lambda)(C_b(\mathbb{R}^N)) = D(\widehat{A})$.

Now, since $C_c^{\infty}(\mathbb{R}^N) \subset D(\widehat{A})$ and $C_c^{\infty}(\mathbb{R}^N)$ is dense in L_{μ}^p (see Proposition 2.3.6 and Corollary 8.1.7), then $D(\widehat{A})$ is dense in L_{μ}^p as well. Besides, $T(t)(D(\widehat{A})) \subset D(\widehat{A})$ for any t > 0 (see Lemma 2.3.3). Thus, the conclusion follows from Proposition B.1.10.

As the following proposition shows, if $\{T(t)\}$ admits an invariant measure, then the semigroup is conservative.

Proposition 8.1.10 Assume Hypotheses 8.1.1. If there exists an invariant measure μ of $\{T(t)\}$, then $\{T(t)\}$ is conservative.

Proof. Fix t > 0. According to Theorem 2.2.5, $0 < T(t) \mathbb{1} \le \mathbb{1}$ for any t > 0. Besides, by (8.0.2) with $f \equiv \mathbb{1}$, we have

$$\int_{\mathbb{R}^N} T(t) \mathbf{1} d\mu = \int_{\mathbb{R}^N} \mathbf{1} d\mu = 1, \qquad t > 0,$$

and, therefore, $T(t)1 \equiv 1 \mu$ -almost everywhere in \mathbb{R}^N . Since T(t)1 is a continuous function and μ is equivalent to the Lebesgue measure, then $T(t)1 \equiv 1$ everywhere in \mathbb{R}^N , for any t > 0.

Now, we show that an invariant measure of $\{T(t)\}$ is always ergodic, i.e., that

$$\lim_{t \to +\infty} \frac{1}{t} \int_0^t T(s) f ds = \int_{\mathbb{R}^N} f d\mu, \qquad f \in L^2_\mu,$$

where the limit is meant in L^2_{μ} .

Proposition 8.1.11 Let μ be an invariant measure of $\{T(t)\}$. Then μ is ergodic.

To prove the proposition, we need some preliminary results.

Lemma 8.1.12 Let H be a Hilbert space and let $T \in L(H)$ be a bounded operator such that $||T^k||_{L(H)} \leq M$ for any $k \in \mathbb{N}$, some constant M > 0 and any $k \in \mathbb{N}$. Then, the sequence of bounded operators $\{T_n\}$, defined by

$$P_n x = \frac{1}{n} \sum_{k=0}^{n-1} T^k x, \qquad x \in H,$$

converges pointwise to a bounded operator P_{∞} which is a projection on the kernel of the operator I-T.

Proof. We begin the proof by observing that a straightforward computation shows that $P_n x$ converges to x in H, as n tends to $+\infty$, for any $x \in \text{Ker}(I-T)$. Similarly, $P_n x$ converges in H for any $x \in (I-T)(H)$. Since the norms of the operators T_n are equibounded by M, by Proposition A.1.2, $P_n x$ converges in H for any $x \in \overline{(I-T)(H)}$.

We now prove that the sequence $\{P_nx\}$ converges for any $x \in H$. Since the previous sequence is bounded, up to a subsequence, we can assume that there exists $y \in H$ such that P_nx converges weakly to y as n tends to $+\infty$. As a consequence, $T(P_n)x$ converges weakly to Ty in H. Since

$$T(P_n x) = P_n x + \frac{1}{n} (T^n x - x), \qquad n \in \mathbb{N},$$
 (8.1.7)

we immediately deduce that $y \in \text{Ker}(I-T)$. We now split x = y + (x-y) and we claim that $x - y \in \overline{(I-T)(X)}$. Once the claim is proved, we will conclude that $P_n x$ converges in the strong topology of H. To prove the claim, we observe that x-y is the weak limit of the sequence $x-P_n x$, and $x-P_n x \in (I-T)(X)$ for any $n \in \mathbb{N}$. Indeed,

$$x - P_n x = \frac{1}{n} \sum_{k=0}^{n-1} (I - T^k)(x) = (I - T) \left(\frac{1}{n} \sum_{k=0}^{n-1} \sum_{j=0}^{k-1} T^j \right) (x).$$

Since (I-T)(H) is convex, its weak and strong closures coincide. The claim follows.

Now, let us set $P_{\infty}x = \lim_{n \to +\infty} P_n x$ for any $x \in H$. To conclude the proof, we show that P_{∞} is a projection on Ker(I-T). Letting n go to $+\infty$ in (8.1.7) we get $TP_{\infty}x = P_{\infty}x$. This implies that

$$T^k P_{\infty} x = P_{\infty} x, \qquad x \in H, \quad k \in \mathbb{N},$$
 (8.1.8)

and, consequently, $P_{\infty}^2 = P_{\infty}$, so that P_{∞} is a projection. Since we already know that $\operatorname{Ker}(I-T) \subset P_{\infty}(H)$, we just need to show that if $P_{\infty}x = x$, then Tx = x. But this is a straightforward consequence of (8.1.8).

Proposition 8.1.13 (Von Neumann) For any $f \in L^2_{\mu}$ the limit

$$\lim_{t \to +\infty} \frac{1}{t} \int_0^t T(s) f ds$$

is well defined. Here, the integral is meant in the L^2_{μ} -norm. Moreover, if we denote by $P_{\infty}f$ the previous limit, the operator P_{∞} is a projection on the subspace $C:=\{f\in L^2_{\mu}: T(t)f=f\ \mu-a.e.\ for\ any\ t>0\}.$

The set C enjoys the following properties:

(i) if
$$f \in C$$
, then $|f| \in C$;

- (ii) if $f, g \in C$, then $f \wedge g$ and $f \vee g$ belong to C. In particular, f^+ and f^- belong to C;
- (iii) if $f \in C$, then, for any $\lambda \in \mathbb{R}$, the function $\chi_{\{x: f(x) > \lambda\}}$ belongs to C.

Finally,

$$\int_{\mathbb{R}^N} P_{\infty} f d\mu = \int_{\mathbb{R}^N} f d\mu, \qquad f \in L^2_{\mu}. \tag{8.1.9}$$

Proof. Let us introduce the operator $P_t \in L(L^2_\mu)$ defined by

$$P_t f = \frac{1}{t} \int_0^t T(s) f ds, \qquad f \in L^2_\mu, \quad t > 0.$$

Denoting, respectively, by [t] and $\{t\}$ the integer and the fractional part of t > 0, we can write

$$P_{t}f = \frac{1}{t} \sum_{k=0}^{[t]-1} \int_{0}^{1} T(s+k)fds + \frac{1}{t} \int_{0}^{\{t\}} T(s+[t])fds$$
$$= \frac{[t]}{t} \frac{1}{[t]} \sum_{k=0}^{[t]-1} T^{k}(1)(P_{1}f) + \frac{\{t\}}{t} T^{[t]}(1)(P_{\{t\}}f). \tag{8.1.10}$$

Since the L^2_{μ} -norm of $T^k(1)$ is bounded by 1 for any $k \in \mathbb{N}$, Lemma 8.1.12 implies that there exists a projection P_* such that $\frac{1}{n} \sum_{k=0}^{n-1} T^k(1) f$ converges to P_*f , as n tends to $+\infty$, for any fixed $f \in L^2_{\mu}$. Therefore, letting t go to $+\infty$ in (8.1.10), we easily deduce that P_t converges pointwise to $P_{\infty} = P_* \circ P_1$ as t tends to $+\infty$.

To prove that P_{∞} is a projection, it suffices to show that

$$T(r) \circ P_{\infty} = P_{\infty}, \qquad r > 0. \tag{8.1.11}$$

Indeed, this will imply that $P_1 \circ P_{\infty} = P_{\infty}$ and $P_* \circ P_{\infty} = P_{\infty}$, which, of course, gives $P_{\infty}^2 = P_{\infty}$. To prove (8.1.11), we observe that

$$T(r)P_{t}f = \frac{1}{t}T(r)\int_{0}^{t}T(s)fds = \frac{1}{t}\int_{0}^{t}T(r+s)fds = \frac{1}{t}\int_{r}^{r+t}T(s)fds$$
$$= P_{t}f + \frac{1}{t}\int_{0}^{r}(T(t)-1)T(s)fds. \tag{8.1.12}$$

Hence, letting t go to $+\infty$ in (8.1.12) gives (8.1.11).

Now, showing that P_{∞} is a projection on C is immediate. The inclusion $C \subset P_{\infty}(L_{\mu}^2)$ is straightforward, whereas the other inclusion follows from (8.1.11). Indeed, since $P_{\infty}f = f$ for any $f \in C$, we have $C \subset P_{\infty}(L_{\mu}^2)$. Hence, the first part of the assertion follows.

To prove the second part of the proposition, we begin by observing that since $|T(t)g| \leq T(t)|g|$ for any t>0 and any function $g\in C_b(\mathbb{R}^N)$, then, by density, such an inequality can be extended to any $g\in L^2_\mu$. Fix now $f\in C$. Then, $|f|=|T(t)f|\leq T(t)|f|$ for any t>0. Since μ is an invariant measure of $\{T(t)\}$ then $\|T(t)|f|-|f|\|_1=0$, so that T(t)|f|=|f| μ -almost everywhere for any t>0. Therefore, $|f|\in C$. Now, since $f^+=(f+|f|)/2$ and $f^-=(f-|f|)/2$, we easily deduce that f^+ and f^- belong to C if f does. Finally, since $f\vee g=g+(f-g)^+$ and $f\wedge g=g+(f-g)^-$, we easily deduce that $f\vee g$ and $f\wedge g$ belong to C if f and g do.

To prove the property (iii), we fix $\lambda \in \mathbb{R}$, $f \in C$ and we set

$$f_n = (n(f - \lambda)^+) \wedge 1, \qquad n \in \mathbb{N}.$$

As it is immediately seen, f_n converges pointwise to $\chi_{\{x:\ f(x)>\lambda\}}$ as n tends to $+\infty$. By Remark 2.2.10 it follows that $T(\cdot)f_n$ tends to $T(\cdot)\chi_{\{x:\ f(x)>\lambda\}}$ pointwise in $(0,+\infty)\times\mathbb{R}^N$. Since each function f_n belongs to C, the function $\chi_{\{x:\ f(x)>\lambda\}}$ is in C as well.

Finally, since

$$\int_{\mathbb{R}^N} P_t f d\mu = \frac{1}{t} \int_{\mathbb{R}^N} d\mu \int_0^t T(s) f ds = \frac{1}{t} \int_0^t ds \int_{\mathbb{R}^N} T(s) f d\mu = \int_{\mathbb{R}^N} f d\mu,$$

for any t > 0, letting t go to $+\infty$, the dominated convergence theorem yields (8.1.9).

Proof of Proposition 8.1.11. We begin the proof observing that, if A is a Borel set, then χ_A belongs to C if and only if $\mu(A)=0$ or $\mu(A)=1$. Indeed, suppose that $\chi_A \in C$ and $\mu(A)>0$. Then, since $\{T(t)\}$ is irreducible and strong Feller (see Proposition 2.2.12) the function $T(t)\chi_A$ is continuous in \mathbb{R}^N and it is strictly positive for any t>0 (see Remark 2.2.13). Therefore, $T(t)\chi_A \neq \chi_A$ on $\mathbb{R}^N \setminus A$. But since $\chi_A \in C$, then $T(t)\chi_A = \chi_A$ μ -almost everywhere in \mathbb{R}^N , which implies that $\mu(\mathbb{R}^N \setminus A) = 0$ or, equivalently, $\mu(A) = 1$.

We can now show that C contains only the constant functions. For this purpose, let $f \in C$ be a nonconstant function. Then, by Proposition 8.1.13(iii) the characteristic function of the set $A_{\lambda} = \{x: f(x) > \lambda\}$ is in C, for any $\lambda \in \mathbb{R}$. Therefore, $\mu(A_{\lambda}) = 0$ or $\mu(A_{\lambda}) = 1$. Observe that there should exist $\lambda \in \mathbb{R}$ such that $\mu(A_{\lambda}) = 1$. Otherwise, since

$$\mathbb{R}^{N} = \{x \in \mathbb{R}^{N} : f(x) = \pm \infty\} \cup \bigcup_{n \in \mathbb{N} \cup \{+\infty\}} A_{-n},$$

it should follow that $\mu(\mathbb{R}^N) = 0$. Similarly, if $\mu(A_{\lambda}) = 1$ for any $\lambda \in \mathbb{R}$, then $f(x) = +\infty$ for μ -almost any $x \in \mathbb{R}^N$, which cannot be the case. Set $\lambda_0 = \sup\{\lambda : \mu(A_{\lambda}) = 1\}$. We infer that $f = \lambda_0$ μ -almost everywhere in \mathbb{R}^N . Indeed, by definition of λ_0 , $f \geq \lambda_0$ μ -almost everywhere in \mathbb{R}^N . Moreover,

since for any $\lambda > \lambda_0$, $\mu(A_{\lambda}) = 0$ it follows that $\mu(A_{\lambda_0}) = 0$, so that f is μ -almost everywhere constant in \mathbb{R}^N .

Since C contains only the constant functions, its dimension is 1. Therefore, we can find out a linear operator $S \in (L^2_\mu)'$ such that $P_\infty f = S(f) \mathbb{1}$ for any $f \in L^2_\mu$. According to the Riesz-Fisher representation theorem, there exists $g \in L^2_\mu$ such that

$$Sf = \int_{\mathbb{R}^N} fg d\mu, \qquad f \in L^2_\mu.$$

To complete the proof we just need to show that g=1 μ -almost everywhere. For this purpose we observe that from (8.1.9) we get

$$\int_{\mathbb{R}^N} f d\mu = \int_{\mathbb{R}^N} P_{\infty} f d\mu = \int_{\mathbb{R}^N} f g d\mu, \qquad f \in L^2_{\mu}.$$

By the arbitrariness of f, it follows that g=1 μ -almost everywhere, and we are done.

Remark 8.1.14 The results of Proposition 8.1.13 imply, in particular, that there exists a subsequence $\{t_n\}$ diverging to $+\infty$ such that $P_{t_n}f$ converges pointwise, as n tends to $+\infty$, to \overline{f} . As the Chacon-Ornstein theorem show (see [124, Chapter 3, Section 8]), the pointwise limit is \overline{f} , for any $f \in L^1_\mu$, and we can take $t_n = n$ for any $n \in \mathbb{N}$.

Now we can prove that $\{T(t)\}$ admits at most one invariant measure. The uniqueness of the invariant measure has been proved in a more general context by Doob. We refer the reader to [42] for further details.

Theorem 8.1.15 There exists at most one invariant measure of $\{T(t)\}$.

Proof. Let μ_1 and μ_2 be two invariant measures of $\{T(t)\}$. Let us prove that $\mu_1 = \mu_2$. According to Proposition 8.1.11, μ_1 and μ_2 are ergodic measures. Therefore, for any Borel set A, we can find out two Borel sets M_1 and M_2 such that $\mu_j(M_j) = 1$ for j = 1, 2 and

$$\lim_{n \to +\infty} \frac{1}{n} \int_0^n (T(s)\chi_A)(x)ds = \int_{\mathbb{R}^N} \chi_A d\mu_j = \mu_j(A), \qquad x \in M_j, \quad j = 1, 2$$

(see Remark 8.1.14). Since the measures μ_1 and μ_2 are equivalent (see Proposition 8.1.5), then $M_1 \cap M_2 \neq \emptyset$. Therefore, $\mu_1(A) = \mu_2(A)$ and we are done.

To conclude this subsection we deal with the behaviour of T(t) as t tends to $+\infty$ and we prove a Liouville type theorem. The following result is due to Doob. For a proof, see, e.g., [42, Theorems 3.4.2 & 4.2.1].

Theorem 8.1.16 For any $f \in L^p_u$ we have

$$\lim_{t \to +\infty} ||T(t)f - \overline{f}||_p = 0, \tag{8.1.13}$$

where $\overline{f} = \int_{\mathbb{R}^N} f d\mu$.

As a straightforward consequence of Theorem 8.1.16 we get the following Liouville type theorem.

Theorem 8.1.17 Suppose that the semigroup $\{T(t)\}$ admits an invariant measure μ . Moreover, let $p \in [1, +\infty)$ and let $u \in D(L_p)$ be such that $L_p u = 0$. Then, u is constant.

Proof. Let u be as in the statement of the theorem. Since $\{T(t)\}$ is a strongly continuous semigroup, then the function $t\mapsto T(t)u$ is continuously Fréchet differentiable in $[0,+\infty)$ with values in L^p_μ and $D_tT(t)u=T(t)L_pu=0$ for any t>0 and any $p\in[1,+\infty)$ (see Section B.1). This implies that the function $t\mapsto T(t)u$ is constant and, consequently, T(t)u=u for any t>0. Letting t tend to $+\infty$, from (8.1.13), we immediately deduce that $u=\int_{\mathbb{R}^N}ud\mu$, so that u is constant.

8.1.2 Existence by Khas'minskii theorem

The main result concerning the existence of an invariant measure of a Markov semigroup is the Khas'minskii theorem. To prove it, we need some preliminaries.

Theorem 8.1.18 (Prokhorov) A family \mathcal{F} of Borel probability measures on \mathbb{R}^N is tight (see Definition 5.1.2) if and only if, for any sequence $\{\mu_n\} \subset \mathcal{F}$, there exists a subsequence weakly* convergent in the dual space $C_b(\mathbb{R}^N)'$ to a probability measure.

Proof. To begin the proof, we assume that the family \mathcal{F} is tight and we fix a sequence $\{\mu_n\} \subset \mathcal{F}$. For any $n \in \mathbb{N}$, let us consider the restriction $\mu_{n,1}$ of the measure μ_k to $\overline{B}(1)$. Since $C(\overline{B}(1))$ is separable the weak* topology of $(C(\overline{B}(1))')$ is metrizable. Therefore, up to a subsequence, $\mu_{n,1}$ weakly* converges to some measure μ_1 . Applying the same argument to the restrictions of the sequence $\{\mu_n\}$ to the ball B(k) $(k \in \mathbb{N})$ and using a diagonal procedure, we can determine a subsequence $\{\mu_{k_n}\} \subset \{\mu_n\}$ such that for any $m \in \mathbb{N}$, μ_{k_n} converges weakly* to a Borel measure $\mu^{(m)}$ on $C(\overline{B}(m))$.

Let us now observe that, for any positive function $f \in C_b(\mathbb{R}^N)$ and any $m \in \mathbb{N}$, one has

$$\int_{\overline{B}(m)} f d\mu^{(m)} = \lim_{n \to +\infty} \int_{\overline{B}(m)} f d\mu_{k_n} \le \lim_{n \to +\infty} \int_{\overline{B}(m+1)} f d\mu_{k_n}.$$
 (8.1.14)

Now, fix a Borel set B and a bounded sequence $\{f_j\} \subset C_b(\mathbb{R}^N)$ of positive functions converging pointwise to χ_B as j tends to $+\infty$. Taking (8.1.14) into account, we can write

$$\mu^{(m)}(B \cap \overline{B}(m)) = \int_{\overline{B}(m)} \chi_B d\mu^{(m)} = \lim_{j \to +\infty} \int_{\overline{B}(m)} f_j d\mu^{(m)}$$

$$\leq \lim_{j \to +\infty} \int_{\overline{B}(m+1)} f_j d\mu^{(m+1)} = \int_{\overline{B}(m+1)} \chi_B d\mu^{(m+1)}$$

$$= \mu^{(m+1)}(B \cap \overline{B}(m)).$$

Since the sequence $\{\mu^{(m)}(B \cap \overline{B}(m))\}$ is positive, increasing and bounded by 1, we can define a function μ on the σ algebra of the Borel sets of \mathbb{R}^N by setting

$$\mu(B) := \lim_{m \to +\infty} \mu^{(m)}(B \cap \overline{B}(m)),$$

for any Borel set B. As it is immediately seen, μ is a Borel measure.

Let us now prove that μ is a probability measure and μ_{k_n} converges weakly* to μ . For this purpose, we fix $\varepsilon > 0$ and take $h \in \mathbb{N}$ such that

$$\mu_{k_n}(\mathbb{R}^N \setminus B(h)) \le \varepsilon, \qquad n \in \mathbb{N}.$$

Moreover, for any m > h, we denote by g_m any continuous function such that $\chi_{\overline{B}(m)\backslash B(h+1)} \leq g_m \leq \chi_{\overline{B}(m+1)\backslash B(h)}$. Then,

$$\mu(\overline{B}(m) \setminus B(h+1)) \leq \int_{\overline{B}(m)} g_m d\mu$$

$$= \lim_{n \to +\infty} \int_{\overline{B}(m)} g_m d\mu_{k_n}$$

$$\leq \limsup_{n \to +\infty} \mu_{k_n}(\overline{\mathbb{R}}^N \setminus B(h))$$

$$\leq \varepsilon.$$

Letting m tend to $+\infty$ gives $\mu(\mathbb{R}^N \setminus B(h+1)) \leq \varepsilon$ and we are almost done. Indeed, if $f \in C_b(\mathbb{R}^N)$,

$$\left| \int_{\mathbb{R}^{N}} f d\mu - \int_{\mathbb{R}^{N}} f d\mu_{k_{n}} \right| \leq \left| \int_{\overline{B}(h+1)} f d\mu - \int_{\overline{B}(h+1)} f d\mu_{k_{n}} \right|$$

$$+ \left| \int_{\mathbb{R}^{N} \setminus \overline{B}(h+1)} f d\mu - \int_{\mathbb{R}^{N} \setminus \overline{B}(h+1)} f d\mu_{k_{n}} \right|$$

$$\leq \left| \int_{\overline{B}(h+1)} f d\mu - \int_{\overline{B}(h+1)} f d\mu_{k_{n}} \right| + 2\varepsilon \|f\|_{\infty}.$$

$$(8.1.15)$$

Now, observing that $\mu(\overline{B}(h+1)) = \mu^{(m)}(B(h+1))$ for any m > h+1 and letting n go to $+\infty$, (8.1.14) and (8.1.15) give

$$\lim_{n \to +\infty} \left| \int_{\mathbb{R}^N} f d\mu - \int_{\mathbb{R}^N} f d\mu_{k_n} \right| \le 2\varepsilon ||f||_{\infty}, \tag{8.1.16}$$

and the arbitrariness of $\varepsilon > 0$ and $f \in C_b(\mathbb{R}^N)$ allow us to conclude that μ_{k_n} weakly* converges to μ . Moreover, taking $f \equiv 1$ in (8.1.16), we obtain that μ is a probability measure.

Conversely, let us assume that \mathcal{F} is relatively weak* compact and let us prove that it is also tight. By contradiction, assume that there exist $\varepsilon > 0$ and a sequence $\{\mu_n\} \subset \mathcal{F}$ such that

$$\mu_n(B(n)) < 1 - \varepsilon, \qquad n \in \mathbb{N}.$$

Since \mathcal{F} is relatively weak* compact we can assume, up to a subsequence, that μ_n converges to some probability measure μ . Let us prove that

$$\mu(B(k)) \le \liminf_{n \to +\infty} \mu_n(B(k)) \le \liminf_{n \to +\infty} \mu_n(B(n)) \le 1 - \varepsilon, \tag{8.1.17}$$

for any k sufficiently large. Of course, (8.1.17) will lead us to a contradiction, since $\mu(B(k))$ converges to $\mu(\mathbb{R}^N) = 1$ as k tends to $+\infty$.

So, let us prove the claim. As it is easily seen, it suffices to show that

$$\limsup_{n \to +\infty} \mu_n(\mathbb{R}^N \setminus B(k)) \le \mu(\mathbb{R}^N \setminus B(k)). \tag{8.1.18}$$

For this purpose, we fix $\varepsilon > 0$ and $m \in \mathbb{N}$ sufficiently large such that

$$\mu(\mathbb{R}^N \setminus B(k-1/m)) \le \mu(\mathbb{R}^N \setminus B(k)) + \varepsilon.$$

Further, we introduce the function $f: \mathbb{R}^N \to \mathbb{R}$ defined by

$$f(x) = \begin{cases} 0, & |x| \le k - \frac{1}{m}, \\ m|x| + 1 - mk, & k - \frac{1}{m} < |x| < k, \\ 1, & |x| \ge k, \end{cases}$$

and we observe that

$$\mu_n(\mathbb{R}^N \setminus B(k)) = \int_{\mathbb{R}^N \setminus B(k)} f d\mu_n \le \int_{\mathbb{R}^N \setminus B(k-1/m)} f d\mu_n.$$

Letting n go to $+\infty$, we deduce that

$$\limsup_{n \to +\infty} \mu_n(\mathbb{R}^N \setminus B(k)) \le \int_{\mathbb{R}^N \setminus B(k-1/m)} f d\mu \le \mu(\mathbb{R}^N \setminus B(k-1/m))$$
$$\le \mu(\mathbb{R}^N \setminus B(k)) + \varepsilon.$$

The arbitrariness of $\varepsilon > 0$ yields the inequality (8.1.18), and we are done.

Let $\{p(t,x;dy): t>0, x\in\mathbb{R}^N\}$ be the same transition family as in Theorem 2.2.5, and define the family of probability measures $\{r(t,x;dy), t>0, x\in\mathbb{R}^N\}$ by setting

$$r(t, x; B) = \frac{1}{t} \int_0^t p(s, x; B) ds, \qquad B \in \mathcal{B}(\mathbb{R}^N).$$

We recall that $p(s, x; B) = (T(s)\chi_B)(x)$.

Let us prove the following preliminary theorem ([83, Theorem 3.2.1]). In the sequel we will use only the first part of it.

Theorem 8.1.19 (Krylov-Bogoliubov) If for some $t_0 > 0$ and some $x_0 \in \mathbb{R}^N$ the family of measures $\{r(t, x_0; dy), t > t_0\}$ is tight, then the semigroup $\{T(t)\}$ admits an invariant measure μ .

Conversely, if there exists an invariant measure μ , then

$$\lim_{R \to +\infty} \liminf_{t \to +\infty} r(t, x; \mathbb{R}^N \setminus B(R)) = 0,$$

for any $x \in \mathbb{R}^N \setminus E$, where E is a negligible set with respect to the Lebesgue measure.

Proof. Suppose that the family of measures $\{r(t, x_0; dy), t > t_0\}$ is tight. By Prokhorov theorem, there exists a sequence $\{t_n\}$ diverging to $+\infty$ as n tends to $+\infty$ such that $r(t_n, x_0; dy)$ converges weakly* to a probability measure μ . We claim that μ is an invariant measure of $\{T(t)\}$. Indeed, for any $f \in B_b(\mathbb{R}^N)$ and any t > 0, we have

$$\int_{\mathbb{R}^{N}} (T(t)f)(y)r(t_{n}, x_{0}; dy) = \frac{1}{t_{n}} \int_{0}^{t_{n}} (T(t+s)f)(x_{0})ds$$

$$= \frac{1}{t_{n}} \int_{t}^{t+t_{n}} (T(s)f)(x_{0})ds$$

$$= \int_{\mathbb{R}^{N}} f(y)r(t_{n}, x_{0}; dy) + \frac{1}{t_{n}} \int_{t_{n}}^{t_{n}+t} (T(s)f)(x_{0})ds$$

$$- \frac{1}{t_{n}} \int_{0}^{t} (T(s)f)(x_{0})ds.$$

Since $||T(t)||_{L(B_b(\mathbb{R}^N))} \leq 1$ for any $t \geq 0$, letting n tend to $+\infty$ we get

$$\int_{\mathbb{R}^N} T(t) f \, d\mu = \int_{\mathbb{R}^N} f \, d\mu,$$

that is μ is an invariant measure of $\{T(t)\}$.

Conversely, assume that μ is an invariant measure of $\{T(t)\}$. For any fixed $x \in \mathbb{R}^N$ the function

$$R \mapsto \liminf_{t \to +\infty} r(t, x; \mathbb{R}^N \setminus B(R))$$

is decreasing. Therefore, we can define

$$r(x) = \lim_{R \to +\infty} \liminf_{t \to +\infty} r(t, x; \mathbb{R}^N \setminus B(R)), \qquad x \in \mathbb{R}^N.$$

Now, we observe that

$$\mu(\mathbb{R}^N \setminus B(R)) = \frac{1}{t} \int_0^t \mu(\mathbb{R}^N \setminus B(R)) ds$$

$$= \frac{1}{t} \int_0^t ds \int_{\mathbb{R}^N} T(s) \chi_{\mathbb{R}^N \setminus B(R)} d\mu$$

$$= \frac{1}{t} \int_0^t ds \int_{\mathbb{R}^N} p(s, x; \mathbb{R}^N \setminus B(R)) d\mu$$

$$= \int_{\mathbb{R}^N} r(t, x; \mathbb{R}^N \setminus B(R)) d\mu,$$

for any t > 0, and then, by Fatou's lemma, we get

$$\mu(\mathbb{R}^N \setminus B(R)) = \liminf_{t \to +\infty} \int_{\mathbb{R}^N} r(t, x; \mathbb{R}^N \setminus B(R)) d\mu$$
$$\geq \int_{\mathbb{R}^N} \liminf_{t \to +\infty} r(t, x; \mathbb{R}^N \setminus B(R)) d\mu.$$

Letting R tend to $+\infty$, by the monotone convergence theorem we get

$$0 \ge \lim_{R \to +\infty} \int_{\mathbb{R}^N} \liminf_{t \to +\infty} r(t, x; \mathbb{R}^N \setminus B(R)) d\mu = \int_{\mathbb{R}^N} r d\mu,$$

which implies that r = 0, μ -almost everywhere. Since μ is equivalent to the Lebesgue measure, the conclusion follows.

We can now prove the main result of this section, that is the Khas'minskii theorem, which gives a sufficient condition for the tightness of the measures $\{r(t, x_0; dy), t > 1\}$ and, hence, for the existence of an invariant measure for the semigroup $\{T(t)\}$.

Theorem 8.1.20 (Khas'minskii) Suppose that there exists a function $\varphi \in C^2(\mathbb{R}^N)$ such that

$$\varphi \ge 0,$$
 $\lim_{|x| \to +\infty} \mathcal{A}\varphi(x) = -\infty.$

Then, for any fixed $x_0 \in \mathbb{R}^N$, the measures $\{r(t, x_0; dy), t > 1\}$ are tight. As a consequence, the semigroup $\{T(t)\}$ admits an invariant measure.

Proof. Let X be the Markov process associated with $\{T(t)\}$ (see Section 2.4), and let $\tau_n = \tau_{B(n)}$ be the first exit time of X from the ball B(n), as defined in (2.4.6). Fix t > 0 and $x \in \mathbb{R}^N$; from the formula (2.4.7) with $\tau' = t \wedge \tau_n$ we have

$$\mathbb{E}_{x}\varphi(X_{t\wedge\tau_{n}}) = \varphi(x) + \mathbb{E}_{x} \int_{0}^{t\wedge\tau_{n}} \mathcal{A}\varphi(X_{s})ds, \tag{8.1.19}$$

for any $n \in \mathbb{N}$. Now, for any R > 0, let

$$-M_R = \sup_{|x| > R} \mathcal{A}\varphi(x),$$

and let $K \geq 0$ be such that $\mathcal{A}\varphi(x) \leq K$ for any $x \in \mathbb{R}^N$. Then, clearly

$$\mathcal{A}\varphi(x) \le K - \chi_{\{|x| > R\}} M_R, \qquad x \in \mathbb{R}^N,$$

and, by (8.1.19), we get

$$\mathbb{E}_{x}\varphi(X_{t\wedge\tau_{n}}) - \varphi(x) \leq Kt - M_{R}\mathbb{E}_{x} \int_{0}^{t\wedge\tau_{n}} \chi_{\{|X_{s}|>R\}} ds, \qquad t > 0, \ x \in \mathbb{R}^{N}.$$

Hence,

$$M_R \mathbb{E}_x \int_0^{t \wedge \tau_n} \chi_{\{|X_s| > R\}} ds \le Kt + \varphi(x) - \mathbb{E}_x \varphi(X_{t \wedge \tau_n}) \le Kt + \varphi(x), \quad (8.1.20)$$

for any t > 0 and any $x \in \mathbb{R}^N$. Now, since $\{T(t)\}$ is conservative, τ_n tends to $+\infty$ almost surely, and then

$$\lim_{n \to +\infty} \mathbb{E}_x \int_0^{t \wedge \tau_n} \chi_{\{|X_s| > R\}} ds = \mathbb{E}_x \int_0^t \chi_{\{|X_s| > R\}} ds$$
$$= \int_0^t \mathbb{E}_x \chi_{\{|X_s| > R\}} ds$$
$$= t r(t, x; \mathbb{R}^N \setminus B(R)).$$

Thus, letting n tend to $+\infty$ in (8.1.20), we obtain

$$r(t, x; \mathbb{R}^N \setminus B(R)) \le \frac{K + \varphi(x)}{M_R},$$

for any t>1. Since $\lim_{R\to+\infty}M_R=+\infty$, it follows that the measures $\{r(t,x;dy),\ t>1\}$ are tight, for any $x\in\mathbb{R}^N$.

In [116] Khas'minskii theorem is proved in a similar way, taking advantage of the Krylov-Bogoliubov theorem, but without using the formula (2.4.7). For the sake of completeness we give this proof. In Chapters 12 and 13 we will

adapt such a proof, to prove the existence of an invariant measure for the semigroups therein considered.

A second proof of Theorem 8.1.20. We are going to prove that for any $t_0 > 0$ and any $x_0 \in \mathbb{R}^N$ the family of measures $\{r(t, x_0; dy) : t > t_0\}$ is tight.

Let $\{\psi_n\} \in C^{\infty}([0, +\infty))$ be a sequence of smooth functions with the following properties:

- (i) $\psi_n(t) = t$ for any $t \in [0, n]$;
- (ii) $\psi_n(t) = n + \frac{1}{2}$ for any $t \ge n + 1$;
- (iii) $\psi'_n(t) \in [0,1]$ and $\psi''_n(t) \leq 0$ for any $t \geq 0$.

As it is immediately seen the function $\varphi_n = \psi_n \circ \varphi$ belongs to $D_{\max}(\mathcal{A})$ for any $n \in \mathbb{N}$ (see (2.0.1)). Since the semigroup $\{T(t)\}$ is conservative (see Remark 4.0.3 and Theorem 4.1.3), Propositions 2.3.5 and 2.3.6 imply that the function $(t,x) \mapsto u_n(t,x) := (T(t)\varphi_n)(x)$ is differentiable with respect to t in $[0,+\infty) \times \mathbb{R}^N$ and

$$D_{t}u_{n}(t,x) = (T(t)\mathcal{A}\varphi_{n})(x)$$

$$= \int_{\mathbb{R}^{N}} (\mathcal{A}\varphi_{n})p(t,x;dy)$$

$$= \int_{\mathbb{R}^{N}} \left(\psi'_{n}(\varphi)\mathcal{A}\varphi + \psi''_{n}(\varphi)\sum_{i,j=1}^{N} q_{ij}D_{i}\varphi D_{j}\varphi\right)p(t,x;dy), \quad (8.1.21)$$

for any $t \in [0, +\infty)$, any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$. We now fix t > 0 and $x \in \mathbb{R}^N$, and integrate (8.1.21) with respect to $s \in [0, t]$. Recalling that $u_n \geq 0$ in $[0, +\infty) \times \mathbb{R}^N$ and $\psi_n'' \leq 0$ in $[0, +\infty)$, we get

$$-\varphi_{n}(x) \leq \int_{0}^{t} ds \int_{\mathbb{R}^{N}} \psi'_{n}(\varphi)(\mathcal{A}\varphi)p(s, x; dy)$$

$$= \int_{0}^{t} ds \int_{E} \psi'_{n}(\varphi)(\mathcal{A}\varphi)p(s, x; dy)$$

$$+ \int_{0}^{t} ds \int_{\mathbb{R}^{N} \setminus E} \psi'_{n}(\varphi)(\mathcal{A}\varphi)p(s, x; dy), \qquad (8.1.22)$$

where $E = \{y \in \mathbb{R}^N : \mathcal{A}\varphi(y) \geq 0\}$. Since E is a bounded set, the dominated convergence theorem implies that the first integral term in the last side of (8.1.22) converges to

$$\int_0^t ds \int_E (\mathcal{A}\varphi) p(s,x;dy),$$

as n tends to $+\infty$. Further, since the sequence $\{\psi'_n(s)\}$ is increasing to 1 for any $s \in [0, +\infty)$, and $\mathcal{A}\varphi \leq 0$ in $\mathbb{R}^N \setminus E$, using the monotone convergence

theorem, it is immediate to check that the second integral term converges to

$$\int_0^t ds \int_{\mathbb{R}^N \setminus E} (\mathcal{A}\varphi) p(s, x; dy),$$

which is finite due to (8.1.22). Therefore, again from (8.1.22), we get

$$-\varphi(x) \le \int_0^t ds \int_{\mathbb{R}^N} (\mathcal{A}\varphi)p(s, x; dy). \tag{8.1.23}$$

Now, fix ε and let $\rho > 0$ be such that $\mathcal{A}\varphi < -\varepsilon^{-1}$ outside the ball $B(\rho)$. From (8.1.23) we obtain that

$$-\varphi(x) \leq \int_{0}^{t} ds \int_{B(\rho)} (\mathcal{A}\varphi)p(s, x; dy) + \int_{0}^{t} ds \int_{\mathbb{R}^{N} \setminus B(\rho)} (\mathcal{A}\varphi)p(s, x; dy)$$

$$\leq Mt - \frac{1}{\varepsilon} \int_{0}^{t} p(s, x; \mathbb{R}^{N} \setminus B(\rho))ds$$

$$= Mt - \frac{t}{\varepsilon} r(t, x; \mathbb{R}^{N} \setminus B(\rho)),$$

for any $x \in \mathbb{R}^N$, where $M = \sup_{\mathbb{R}^N} \mathcal{A}\varphi$. Consequently,

$$r(t, x, \mathbb{R}^N \setminus B(\rho)) \le \varepsilon \frac{Mt + \varphi(x)}{t},$$

which, of course, implies that for any $t_0 > 0$ and any $x \in \mathbb{R}$ the family of measures $\{r(t, x; dy) : t > t_0\}$ is tight.

8.1.3 Existence by compactness in $C_b(\mathbb{R}^N)$

The compactness of $\{T(t)\}$ in $C_b(\mathbb{R}^N)$, which we studied in Section 5.1, implies the existence of an invariant measure as well as other remarkable properties. To begin with, we consider the following lemma.

Lemma 8.1.21 If for some $t_0 > 0$ and some $x_0 \in \mathbb{R}^N$ the family of measures $\{p(t, x_0; dy), t > t_0\}$ is tight, then the family of measures $\{r(t, x_0; dy), t > t_0\}$ is tight, as well.

Proof. For any $t > t_0$ we have

$$r(t, x_0; B) = \frac{1}{t} \int_0^{t_0} p(s, x_0; B) ds + \frac{1}{t} \int_{t_0}^t p(s, x_0; B) ds$$
$$\leq \sup_{0 < s < t_0} p(s, x_0; B) + \sup_{s > t_0} p(s, x_0; B).$$

Therefore, it is sufficient to prove that the family of measures $\{p(t, x_0; dy), t \in [0, t_0]\}$ is tight. This follows from the Prokhorov theorem 8.1.18. Indeed, consider a sequence $\{p(t_n, x_0; dy)\}$ where $t_n \in [0, t_0]$ for any $n \in \mathbb{N}$. Up to a subsequence, $\{t_n\}$ converges to some $t^* \in [0, t_0]$. Taking Theorem 2.2.5 into account, we get

$$\lim_{n \to +\infty} \int_{\mathbb{R}^N} f(y) p(t_n, x_0; dy) = \lim_{n \to +\infty} (T(t_n) f)(x_0)$$

$$= (T(t^*) f)(x_0)$$

$$= \int_{\mathbb{R}^N} f(y) p(t^*, x_0; dy),$$

for any $f \in C_b(\mathbb{R}^N)$. Therefore, the sequence $\{p(t_n, x_0; dy)\}$ is weakly* convergent to the measure $p(t^*, x_0; dy)$ in the dual space $C_b(\mathbb{R}^N)'$.

Theorem 8.1.22 If T(t) is compact in $C_b(\mathbb{R}^N)$ for any t > 0, then there exists an invariant measure of $\{T(t)\}$. Moreover, for any $p \in (1, +\infty)$, T(t) and $R(\lambda, L_p)$ are compact operators in L_p^p for any t > 0 and any $\lambda > 0$, and the spectrum of L_p consists of isolated eigenvalues.

Proof. Fix $t_0 > 0$, $x_0 \in \mathbb{R}^N$ and $\varepsilon > 0$. Since $\{T(t)\}$ is conservative, by Proposition 5.1.3 there exists R > 0 such that

$$p(t_0, y; \mathbb{R}^N \setminus B(R)) \le \varepsilon, \qquad y \in \mathbb{R}^N.$$

Recalling that p(t, x; dy) is a probability measure for any t > 0 and any $x \in \mathbb{R}^N$, it follows that, for any $t > t_0$,

$$p(t, x_0; \mathbb{R}^N \setminus B(R)) = \int_{\mathbb{R}^N} p(t_0, y; \mathbb{R}^N \setminus B(R)) p(t - t_0, x_0; dy) \le \varepsilon,$$

namely, the family of measures $\{p(t, x_0; dy), t > t_0\}$ is tight. Then, the existence of an invariant measure of $\{T(t)\}$ follows from Theorem 8.1.19 and Lemma 8.1.21.

We now fix an arbitrary t>0 and we prove that T(t) is compact in L^p_μ . For this purpose, we observe that, since μ is equivalent to the Lebesgue measure on the σ -algebra of the Borel sets of \mathbb{R}^N (see Proposition 8.1.5), then $L^\infty_\mu \subset L^\infty(\mathbb{R}^N, dx)$. Moreover, since $\{T(t)\}$ is strong Feller (see Proposition 2.2.12), then it maps L^∞_μ into $C_b(\mathbb{R}^N)$. Therefore, writing $T(t) = T(t/2) \circ T(t/2)$ and taking Proposition A.1.1(ii) into account, it is immediate to check that T(t) is compact from L^∞_μ into $C_b(\mathbb{R}^N)$ and, since $C_b(\mathbb{R}^N)$ is continuously embedded in L^∞_μ , T(t) is compact from L^∞_μ into itself.

Let now $\overline{B}(1)$ be the closed ball in L^{∞}_{μ} with centre at 0 and radius 1 and fix $t \in (0, +\infty)$. Since $T(t)(\overline{B}(1))$ is compact in L^{∞}_{μ} and the set of all the

bounded simple functions is dense in L^{∞}_{μ} then, for any $\varepsilon > 0$, there exists a finite number of simple functions $f_1, \ldots, f_m \in L^{\infty}_{\mu}$ $(m \in \mathbb{N})$ such that

$$T(t)(\overline{B}(1)) \subset \bigcup_{j=1}^{m} \{ f \in L_{\mu}^{\infty} : ||f - f_{j}||_{L_{\mu}^{\infty}} \leq \varepsilon \}.$$

Without loss of generality, we can assume that $f_j = \sum_{i=1}^k \alpha_i^{(j)} \chi_{F_i}$ where F_i are disjoint measurable sets with positive measures such that $\sum_{i=1}^k F_i = \mathbb{R}^N$ and $\alpha_i^{(j)} \in \mathbb{R}$ for any $i = 1, \ldots, k$ and any $j = 1, \ldots, M$. We now consider the operator P_{ε} defined on measurable functions by

$$P_{\varepsilon}f = \sum_{i=1}^{k} \left(\frac{1}{\mu(F_j)} \int_{F_j} f d\mu\right) \chi_{F_j}.$$

As it is easily seen,

$$||P_{\varepsilon}||_{L(L_{u}^{\infty})} = ||P_{\varepsilon}||_{L(L_{u}^{1})} = 1,$$

and $P_{\varepsilon}f_{j} \equiv f_{j}$ for any $j=1,\ldots,N$. Fix $f \in L_{\mu}^{\infty}$ such that $\|f\|_{L_{\mu}^{\infty}}=1$ and let $j \in \{1,\ldots,m\}$ be such that $\|T(t)f-f_{j}\|_{L_{\mu}^{\infty}} \leq \varepsilon$. Then,

$$\|P_{\varepsilon}T(t)f - T(t)f\|_{L^{\infty}_{\mu}} \le \|P_{\varepsilon}(T(t)f - f_j)\|_{L^{\infty}_{\mu}} + \|f_j - T(t)f\|_{L^{\infty}_{\mu}} \le 2\varepsilon.$$

Therefore,

$$||P_{\varepsilon}T(t) - T(t)||_{L(L_{u}^{\infty})} \le 2\varepsilon.$$
(8.1.24)

Similarly,

$$\|P_{\varepsilon}T(t) - T(t)\|_{L(L_{\mu}^{1})} \le \|P_{\varepsilon}T(t)\|_{L(L_{\mu}^{1})} + \|T(t)\|_{L(L_{\mu}^{1})} \le 2\|T(t)\|_{L(L_{\mu}^{1})}. \quad (8.1.25)$$

Hence, by (8.1.24), (8.1.25) and the Riesz-Thorin interpolation theorem (see Theorem A.4.9), we get

$$||P_{\varepsilon}T(t) - T(t)||_{L(L_{\mu}^{p})} \leq ||P_{\varepsilon}T(t) - T(t)||_{L(L_{\mu}^{\infty})}^{1 - \frac{1}{p}} ||P_{\varepsilon}T(t) - T(t)||_{L(L_{\mu}^{1})}^{\frac{1}{p}}$$

$$\leq 2||T(t)||_{L(L_{\mu}^{1})}^{\frac{1}{p}} \varepsilon^{1 - \frac{1}{p}},$$

for any $p \in (1, +\infty)$. Since the operator $P_{\varepsilon}T(t)$ has finite rank for any $\varepsilon > 0$, then it is compact in L^p_{μ} and, consequently, T(t) is compact in L^p_{μ} as well.

Now, we show that $R(\lambda, L_p)$ is compact for any $\lambda \in \rho(L_p)$. For this purpose, we prove that the semigroup $\{T(t)\}$ is norm continuous in $(0, +\infty)$. Indeed, once this property is checked, taking (B.1.3) into account, we will get

$$R(\lambda, L_p) = \int_0^{+\infty} e^{-\lambda t} T(t) dt,$$

for any λ with positive real part, where the integral is meant in L^p_{μ} . Proposition A.1.1(i) then will allow us to conclude that $R(\lambda, L_p)$ is compact for any λ as above. Then the resolvent identity (see the formula (A.3.2)), together with Proposition A.1.1(ii), will yield the compactness of $R(\lambda, L_p)$ for any $\lambda \in \rho(L_p)$.

So, let us prove that $\{T(t)\}$ is norm-continuous in $(0, +\infty)$. Let us fix $t_0 > 0$ and let us set $K = \overline{T(t_0)(B(1))}$, where now B(1) denotes the open unit ball in L^p_μ with centre at 0. Since $T(t_0)$ is a compact operator, then K is compact in L^p_μ . According to Remark B.1.2, T(t)f tends to f as t tends to f, uniformly with respect to f on compact subsets of L^p_μ . Therefore,

$$\lim_{t \to 0^+} T(t+t_0)f = T(t_0)f,$$

uniformly with respect to $f \in B(1)$, and this implies that T(t) tends to $T(t_0)$ in $L(L^p_u)$ from the right. Similarly, if t < 0, we can fix $\delta \in (0, t_0)$ and write

$$||T(t_0+t)f - T(t_0)f||_{L^p_{\mu}} \le ||T(t_0+t-\delta)(T(\delta)f - T(\delta-t)f)||_{L^p_{\mu}} \le ||T(\delta)f - T(\delta-t)f||_{L^p_{\mu}}.$$
(8.1.26)

Letting t tend to 0^- , we deduce that the last side of (8.1.26) vanishes. Hence, $T(\cdot)f$ is norm continuous in $(0, +\infty)$.

Finally, we observe that, since $R(1, L_p)$ is compact (and invertible), then its spectrum consists of isolated eigenvalues. Therefore, the spectrum of L_p consists of isolated eigenvalues as well.

Remark 8.1.23 Applying [7, Proposition 2.6], one can also show that, under the assumptions of Theorem 8.1.22, the spectrum of the operator L_p is independent of p and it coincides with the spectrum of $(A, D_{\max}(A))$ in $C_b(\mathbb{R}^N)$ (see (2.0.1)).

To conclude this subsection, we show that, under the assumptions of Theorem 8.1.22, the spectral gap for the operator L_p holds for any $p \in (1, +\infty)$, i.e., there exists $\delta > 0$ such that $\sigma(L_p) \setminus \{0\} \subset \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda < -\delta\}$.

Theorem 8.1.24 Under the assumptions of Theorem 8.1.22, the spectral gap holds for the operator L_p , for any $p \in (1, +\infty)$. Moreover, T(t)f tends exponentially to \overline{f} in L^2_μ , as t tends to $+\infty$.

Proof. Fix $p \in (1, +\infty)$ and denote, as in Subsection 8.1.1, by P_{∞} the operator defined in L^p_u by

$$P_{\infty}f = \int_{\mathbb{R}^N} f d\mu, \qquad f \in L^p_{\mu}.$$

Since P_{∞} is a projection, then $L^p_{\mu} = H \oplus K$, where $H = \operatorname{Ker}(P_{\infty})$ and $K = \operatorname{Ker}(I - P_{\infty})$. As it is immediately seen, T(t) commutes with P_{∞} for

any t>0. Therefore, H and K are both invariant under the action of the semigroup and the restrictions of the semigroup to H and K give rise to two strongly continuous semigroups, whose infinitesimal generators are the restrictions of L_p to H and K, respectively. In particular, for any t>0, T(t) coincides with the identity operator on K. As a consequence, for any $f\in L^p_\mu$ such that $f=f_1+f_2$ with $f_1\in H$ and $f_2\in K$, we have

$$T(t)f = T(t)_{\mid H} f_1 + f_2.$$

Now, we are going to prove that $T(t)_{\mid H}$ decreases exponentially as t tends to $+\infty$. As a first step, we claim that $\sigma(T(1)_{\mid H}) \subset B(1)$. Since $\{T(t)\}$ is a semigroup of contractions, then $\sigma(T(1)) \subset \overline{B}(1)$. Let us show that, if $\lambda \in \partial B(1)$, then $\lambda \in \rho(T(1)_{\mid H})$. On the contrary, we assume that there exists $\lambda \in \sigma(T(1)_{\mid H}) \cap \partial B(1)$. Since T(1) is a compact operator, λ belongs to the point spectrum of T(1). Therefore, there exists $f \in H$, with $f \neq 0$, such that $T(1)f = \lambda f$. Using the semigroup rule, we obtain that $T(n)f = \lambda^n f$ for any $n \in \mathbb{N}$. From Theorem 8.1.16 it follows that T(n)f tends to $P_{\infty}f = 0$ in L^p_{μ} . Therefore, $\lambda^n f$ should tend to 0 as n tends to $+\infty$, but, of course, this cannot be the case. The claim follows and, in particular, we deduce that there exists $\delta < 1$ such that

$$\lim_{n \to +\infty} \|T(1)^n\|_{L(H)}^{\frac{1}{n}} = \lim_{n \to +\infty} \|T(n)\|_{L(H)}^{\frac{1}{n}} = \delta.$$

Using the semigroup law it is now immediate to check that there exists $\omega < 0$ such that

$$||T(t)||_{L(H)} \le Ce^{\omega t}, \qquad t > 0.$$

Since $\{T(t)_{|H}\}$ is a strongly continuous semigroup in H, then $\inf\{\operatorname{Re}\lambda:\lambda\in\sigma((L_p)_{|H})\}\leq\omega$ (see Section B.1).

To conclude the proof of the first part of the theorem, it suffices to show that

$$\sigma(L_p) = \sigma((L_p)_{|H}) \cup \sigma((L_p)_{|K}) = \sigma((L_p)_{|H}) \cup \{0\}.$$
 (8.1.27)

Note that the second equality in (8.1.27) is immediately checked since $T(t)_{|K} = I$ for any t > 0.

As a first step to prove (8.1.27), we observe that $0 \in \sigma(L)$ and $0 \notin \sigma((L_p)_{|H})$. Indeed, Theorem 8.1.17 implies that Lu = 0 if and only if u is constant. Therefore, $0 \in \sigma(L_p)$. But, since 0 is the only constant in H, then $0 \notin \sigma((L_p)_{|H})$.

Now, let $\lambda \in \rho(L)$, $\lambda \neq 0$, and let $f \in H$. Set $u = R(\lambda, L_p)f$, so that $\lambda u - L_p u = f$. Since $\overline{f} = 0$ and $\overline{L_p u} = 0$, by (8.1.5), then $\overline{u} = 0$ as well, so

that $u \in H$. This implies that $\lambda \in \rho((L_p)_{\mid H})$ and $R(\lambda, (L_p)_{\mid H}) = R(\lambda, L_p)_{\mid H}$. Therefore, $\sigma((L_p)_{\mid H}) \cup \{0\} \subset \sigma(L_p)$.

To prove the other inclusion in (8.1.27), we fix $\lambda \in \rho((L_p)_{\mid H})$, $\lambda \neq 0$, and we show that $\lambda \in \rho(L_p)$. For this purpose, we observe that, for any $f \in L_{\mu}^p$, the function $u = R(\lambda, (L_p)_{\mid H})(f - \overline{f}) + \lambda^{-1}\overline{f}$ is a solution to the equation $\lambda u - L_p u = f$. Note that $\overline{f} \in D(L_p)$, since the constants belong to $D_{\max}(A)$, which is a core of L_p (see Propositions 2.3.6, 4.1.10 and 8.1.9). Hence, the operator $\lambda I - L_p$ is surjective in L_{μ}^p .

To prove that $\lambda I - L_p$ is also one to one, we observe that, since P_{∞} commutes with T(t) for any t > 0, then it commutes with L_p on $D(L_p)$. Therefore, if $\lambda u = L_p u$ for some $u \in D(L_p)$, then both $u_1 = P_{\infty} u$ and $u_2 = (I - P_{\infty})u$ solve the equation $\lambda v = L_p v$. Since $\lambda \notin \sigma((L_p)_{\mid H}) \cup \sigma((L_p)_{\mid K})$, then $u_1 \equiv u_2 \equiv 0$ and, consequently, $u \equiv 0$.

To conclude the proof of the theorem, we observe that the function $f - \overline{f}$ belongs to H for any $f \in L^2_{\mu}$. Therefore,

$$||T(t)f - \overline{f}||_2 = ||T(t)f - T(t)\overline{f}||_2 \le 2Ce^{\omega t}||f||_2, \qquad t > 0,$$

and we are done.

8.1.4 Existence by symmetry

In this subsection we study the case when the operator A is given by

$$\mathcal{A}u(x) = \Delta u(x) - \langle DU(x) + G(x), Du(x) \rangle, \qquad x \in \mathbb{R}^N,$$
(8.1.28)

on smooth functions, under the following hypotheses on U and G.

Hypotheses 8.1.25 (i) $U: \mathbb{R}^N \to \mathbb{R}$ belongs to $C^{1+\alpha}_{loc}(\mathbb{R}^N)$, for some $\alpha \in (0,1)$, and $e^{-U} \in L^1(\mathbb{R}^N)$;

(ii) $G \in C^1(\mathbb{R}^N, \mathbb{R}^N)$, $\operatorname{div} G = \langle G, DU \rangle$ and

$$\int_{\mathbb{R}^N} |G(x)| e^{-U(x)} dx < +\infty.$$

Under Hypotheses 8.1.25 we can define the probability measure

$$\mu(dx) = K^{-1}e^{-U(x)}dx, \qquad K = \int_{\mathbb{R}^N} e^{-U(x)}dx.$$

By means of an integration by parts, we get

$$\int_{\mathbb{R}^N} \mathcal{A}f \, g \, d\mu = -\int_{\mathbb{R}^N} \langle Df, Dg \rangle d\mu, \qquad f \in C_c^{\infty}(\mathbb{R}^N), \quad g \in W_{\text{loc}}^{1,p}(\mathbb{R}^N),$$
(8.1.29)

for any $p \in [1, +\infty)$. In particular,

$$\int_{\mathbb{R}^N} \mathcal{A}f d\mu = 0, \qquad f \in C_c^{\infty}(\mathbb{R}^N). \tag{8.1.30}$$

The main result of the section is the following theorem.

Theorem 8.1.26 Assume Hypotheses 8.1.25. Then, the semigroup $\{T(t)\}$ associated with the operator \mathcal{A} in $C_b(\mathbb{R}^N)$ is conservative, μ is the invariant measure of $\{T(t)\}$, and $C_c^{\infty}(\mathbb{R}^N)$ is a core of the infinitesimal generator L_p in L_p^p for any $p \in [1, +\infty)$.

Proof. For any $p \in [1, +\infty)$ consider the operator $A_p : C_c^{\infty}(\mathbb{R}^N) \to L^p_{\mu}$ defined by $A_p u = \mathcal{A}u$ for any $u \in C_c^{\infty}(\mathbb{R}^N)$. As a first step we prove that A_p is dissipative, that is

$$\lambda \|u\|_{L^p_\mu} \le \|\lambda u - A_p u\|_{L^p_\mu}, \qquad \lambda > 0, \quad u \in D(A_p).$$
 (8.1.31)

For this purpose, let $u \in D(A_p)$ and set $f = \lambda u - A_p u$. Multiplying both the sides of the previous equality by $\operatorname{sign}(u)|u|^{p-1}$, and integrating in \mathbb{R}^N , we get

$$\lambda \int_{\mathbb{R}^N} |u|^p d\mu - \int_{\mathbb{R}^N} A_p u \operatorname{sign}(u) |u|^{p-1} d\mu = \int_{\mathbb{R}^N} f \operatorname{sign}(u) |u|^{p-1} d\mu$$

$$\leq \int_{\mathbb{R}^N} |f| |u|^{p-1} d\mu. \tag{8.1.32}$$

Now, if $p \ge 2$ integrating by parts we get

$$\int_{\mathbb{R}^{N}} A_{p} u \operatorname{sign}(u) |u|^{p-1} d\mu = -\int_{\mathbb{R}^{N}} \langle Du, D(\operatorname{sign}(u)|u|^{p-1}) \rangle d\mu$$
$$= -(p-1) \int_{\mathbb{R}^{N}} |Du|^{2} |u|^{p-2} d\mu \le 0.$$
 (8.1.33)

Hence, from (8.1.32) and (8.1.33) we easily get (8.1.31).

In the case when $p \in [1, 2)$ we consider the sequence $\{\varphi_n\} \subset C^{\infty}(\mathbb{R})$ defined by $\varphi_n(x) = x(x^2 + n^{-1})^{(p-2)/2}$ for any $x \in \mathbb{R}^N$. Since any φ_n is an increasing function in \mathbb{R}^N , converging pointwise to $\operatorname{sign}(x)|x|^{p-1}$ as n tends to $+\infty$, by dominated convergence and by (8.1.29) we have

$$\begin{split} \int_{\mathbb{R}^N} A_p u \operatorname{sign}(u) |u|^{p-1} d\mu &= \lim_{n \to +\infty} \int_{\mathbb{R}^N} A_p u \, \varphi_n(u) d\mu \\ &= - \lim_{n \to +\infty} \int_{\mathbb{R}^N} \varphi_n'(u) |Du|^2 d\mu \leq 0, \end{split}$$

and, then, (8.1.31) follows from (8.1.32).

As a second step we show that the range of $\lambda I - A_p$ is dense in L^p_μ for some $\lambda > 0$. To prove this, let $g \in L^q_\mu(\mathbb{R}^N)$ (where 1/p + 1/q = 1, or $q = +\infty$ if p = 1) be such that

$$\int_{\mathbb{R}^N} (\lambda u - A_p u) g d\mu = 0,$$

for any $u \in D(A_p)$. First we assume that $p \in (1, +\infty)$. By classical regularity results for weak solutions of elliptic equations (see Theorem C.1.3(ii)), g belongs to $W_{\text{loc}}^{1,q}(\mathbb{R}^N)$. If p=1, then $g \in L_{\text{loc}}^2(\mathbb{R}^N)$. Therefore, still from Theorem C.1.3(ii), we deduce that $g \in W_{\text{loc}}^{1,2}(\mathbb{R}^N)$. Now, from the formula (8.1.29) we get

$$\lambda \int_{\mathbb{R}^N} ug \, d\mu + \int_{\mathbb{R}^N} \langle Du, Dg \rangle d\mu = 0, \qquad u \in C_c^{\infty}(\mathbb{R}^N).$$
 (8.1.34)

By density, (8.1.34) can be extended to any compactly supported function $u \in W^{1,p}(\mathbb{R}^N)$, if $p \in (1, +\infty)$ and to any compactly supported function $u \in W^{1,2}(\mathbb{R}^N)$, if p = 1. Now, for any $n \in \mathbb{N}$, let $\eta_n \in C_c^{\infty}(\mathbb{R}^N)$ be a smooth function such that $\eta_n \equiv 1$ in B(n), $0 \le \eta_n \le \eta_{n+1}$, and $|D\eta_n| \le 1/n$. If $1 , taking <math>u = \eta_p^2 \text{sign}(g)|g|^{q-1}$ in (8.1.34), we get

$$\begin{split} &\lambda \int_{\mathbb{R}^N} \eta_n^2 |g|^q d\mu + (q-1) \int_{\mathbb{R}^N} \eta_n^2 |g|^{q-2} |Dg|^2 d\mu \\ &= -2 \int_{\mathbb{R}^N} \eta_n \mathrm{sign}(g) |g|^{q-1} \langle D\eta_n, Dg \rangle d\mu \\ &\leq \frac{1}{n} \int_{\mathbb{R}^N} \eta_n^2 |Dg|^2 |g|^{q-2} d\mu + \frac{1}{n} \|g\|_q^q. \end{split}$$

Hence,

$$\lambda \int_{\mathbb{R}^N} \eta_n^2 |g|^q d\mu \le \frac{1}{n} \|g\|_{L^q_\mu}^q,$$

for any $n \geq 2$. Letting n tend to $+\infty$, by monotone convergence we get g = 0. The case when p = 1 can be treated as the case when p = 2. Suppose now that p > 2 and, for any $n, m \in \mathbb{N}$, let $u_{n,m} = \eta_n^2 \varphi_m(g)$, where the sequence $\{\varphi_n\}$ is as above with p being replaced by q. Then, $u_{n,m} \in W^{1,p}(\mathbb{R}^N)$ and has compact support for any $n, m \in \mathbb{N}$. Writing (8.1.34) with $u_{n,m}$ instead of u, we get

$$\lambda \int_{\mathbb{R}^{N}} \eta_{n}^{2} g^{2} \left(g^{2} + \frac{1}{m}\right)^{\frac{q-2}{2}} d\mu$$

$$+ (q-1) \int_{\mathbb{R}^{N}} \eta_{n}^{2} g^{2} \left(g^{2} + \frac{1}{m}\right)^{\frac{q-4}{2}} |Dg|^{2} d\mu$$

$$\leq \frac{2}{n} \int_{\mathbb{R}^{N}} \eta_{n} |g| \left(g^{2} + \frac{1}{m}\right)^{\frac{q-2}{2}} |Dg| d\mu. \tag{8.1.35}$$

Letting m tend to $+\infty$ in (8.1.35), we get

$$\lambda \int_{\mathbb{R}^N} \eta_n^2 |g|^q d\mu + (q-1) \int_{\mathbb{R}^N} \eta_n^2 |g|^{q-2} |Dg|^2 d\mu$$

$$\leq \frac{2}{n} \int_{\mathbb{R}^N} \eta_n |g|^{q-1} |Dg| d\mu. \tag{8.1.36}$$

Since the right-hand side of (8.1.36) is finite, the second integral in the left-hand side is finite as well. Therefore, applying the Cauchy-Schwarz inequality gives

$$\begin{split} &\lambda \int_{\mathbb{R}^N} \eta_n^2 |g|^q d\mu + (q-1) \int_{\mathbb{R}^N} \eta_n^2 |g|^{q-2} |Dg|^2 d\mu \\ &\leq \frac{1}{n} \|g\|_q^q + \frac{1}{n} \int_{\mathbb{R}^N} \eta_n^2 |g|^{q-2} |Dg|^2 d\mu. \end{split}$$

Hence, for $n(q-1) \ge 1$, we get

$$\lambda \int_{\mathbb{R}^N} \eta_n^2 |g|^q d\mu \le \frac{1}{n} \|g\|_q^q$$

and, letting n go to $+\infty$, we obtain g=0 also in this case.

Now, according to Proposition B.1.8, the closure $\overline{A_p}$ of the operator A_p in L^p_μ is the generator of a strongly continuous semigroup $\{S(t)\}$. By density, (8.1.30) implies that

$$\int_{\mathbb{D}^N} \overline{A_p} f d\mu = 0, \qquad f \in D(\overline{A_p}).$$

Thus, for any $f \in D(\overline{A_p})$ we have

$$\frac{d}{dt} \int_{\mathbb{D}^N} S(t) f d\mu = \int_{\mathbb{D}^N} \overline{A_p} S(t) f d\mu = 0, \qquad t > 0,$$

so that

$$\int_{\mathbb{R}^N} S(t) f d\mu = \int_{\mathbb{R}^N} f d\mu, \qquad t > 0.$$
 (8.1.37)

Since $D(\overline{A_p})$ is dense in L^p_{μ} , the equality (8.1.37) holds for any $f \in L^p_{\mu}$.

Next, we prove that the semigroup $\{T(t)\}$ can be extended to L^p_μ with a strongly continuous semigroup of contractions and, then, that the so extended semigroup coincides with $\{S(t)\}$. Note that $\{T(t)\}$ is well defined since the coefficients of the operator \mathcal{A} satisfy Hypotheses 8.1.1. For this purpose, for any R>0 we denote by $\{T_R(t)\}$ the semigroup associated with the realization of the operator \mathcal{A} with homogeneous Dirichlet boundary conditions in $C(\overline{\mathcal{B}}(R))$. Let us observe that the measure μ is a subinvariant measure of $\{T_R(t)\}$ for any R>0, that is

$$\int_{B(R)} T_R(t) f d\mu \le \int_{B(R)} f d\mu, \qquad t > 0,$$
 (8.1.38)

for any positive function $f \in C_0(B(R))$. Indeed, differentiating the left-hand side of (8.1.38) with respect to t and integrating by parts we get

$$\begin{split} \int_{B(R)} \frac{d}{dt} T_R(t) f d\mu &= \int_{B(R)} \mathcal{A} T_R(t) f d\mu \\ &= K^{-1} \int_{\partial B(R)} \left(\frac{\partial}{\partial \nu} T_R(t) f \right) e^{-U(y)} \, \sigma(dy), \end{split}$$

where ν is the unit outward normal vector to $\partial B(R)$ and $\sigma(dy)$ is the surface measure on $\partial B(R)$. Since $T_R(t)f$ is nonnegative in B(R) and it vanishes on $\partial B(R)$ (see Proposition C.3.2), we have $\partial T_R(t)f/\partial\nu \leq 0$, and then (8.1.38) follows.

Since $T_R(t)f$ converges to T(t)f in a dominated way, then, letting R tend to infinity, the inequality (8.1.38) holds also when $T_R(t)f$ is replaced with T(t)f, for any nonnegative function $f \in C_c(\mathbb{R}^N)$ and B(R) is replaced by \mathbb{R}^N . Using an approximation argument we can show that (8.1.38) holds for any nonnegative function $f \in C_b(\mathbb{R}^N)$. Arguing as in the proof of Proposition 8.1.8, we can extend $\{T(t)\}$ to a strongly continuous semigroup in L^p_μ for any $p \in [1, +\infty)$.

Now, let L_p be the generator of $\{T(t)\}$ in L_μ^p . Since $\overline{A_p}$ and L_p coincide with \mathcal{A} on $C_c^\infty(\mathbb{R}^N)$, which is a core of $\overline{A_p}$, then L_p is an extension of $\overline{A_p}$. But since $\overline{A_p}$ and L_p are both generators of strongly continuous semigroups of contractions, then for any $\lambda > 0$ we have

$$L^p_{\mu}=(\lambda-\overline{A_p})(D(\overline{A_p}))=(\lambda-L_p)(D(\overline{A_p}))\subset (\lambda-L_p)(D(L_p))=L^p_{\mu},$$

which implies that $(\lambda - L_p)(D(\overline{A_p})) = (\lambda - L_p)(D(L_p))$ or, equivalently, that $D(\overline{A_p}) = D(L_p)$, so that $\overline{A_p}$ and L_p coincide. Therefore, $S(t) \equiv T(t)$ in L_μ^p for any t > 0 (see Proposition B.1.4). Thus, by (8.1.37), μ is the invariant measure of $\{T(t)\}$. Hence, by Proposition 8.1.10, it follows that $\{T(t)\}$ is conservative.

Proposition 8.1.27 D(L) is continuously and densely embedded in $W^{1,2}_{\mu}$. Further,

$$\int_{\mathbb{R}^N} \langle Dg, Dh \rangle d\mu = -\int_{\mathbb{R}^N} Lg \, h \, d\mu, \qquad g \in D(L), \quad h \in W^{1,2}_{\mu}. \tag{8.1.39}$$

Finally, for any $\lambda > 0$,

$$||DR(\lambda, L)f||_2 \le \sqrt{\frac{2}{\lambda}} ||f||_2, \qquad f \in L^2_\mu.$$
 (8.1.40)

Proof. The proof of the first part of the assertion is a straightforward consequence of the formula (8.1.29) and Theorem 8.1.26. Indeed, fix $\psi \in D(L)$. Since $C_c^{\infty}(\mathbb{R}^N)$ is a core of D(L), there exists a sequence $\{\psi_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ such that ψ_n and $L\psi_n$ tend, respectively, to ψ and $L\psi$ as n tends to $+\infty$. Writing (8.1.29) with $f = g = \psi_n - \psi_m$ $(n, m \in \mathbb{N})$ we get

$$||D\psi_n - D\psi_m||_2^2 \le ||\psi_n - \psi_m||_2 ||L\psi_n - L\psi_m||_2.$$

Therefore, $\{\psi_n\}$ is a Cauchy sequence in $W^{1,2}_{\mu}$. Since it converges to ψ in L^2_{μ} , we deduce that $\psi \in W^{1,2}_{\mu}$ and, then, writing (8.1.29) with $f = g = \psi$, we obtain that D(L) is continuously embedded in $W^{1,2}_{\mu}$.

To conclude the proof of the first part of the proposition, we have to show that D(L) is dense in $W^{1,2}_{\mu}$. But this follows immediately from the forthcoming Proposition 8.7.3.

Now, the formula (8.1.39) can be easily proved by a density argument, recalling that $L\varphi = \mathcal{A}\varphi$ for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$.

Finally, to prove the formula (8.1.40) it suffices to apply the formula (8.1.39) with $g = h = R(\lambda, L)f$, recalling that

$$||R(\lambda, L)f||_2 \le |\lambda|^{-1}||f||_2$$
 and $||LR(\lambda, L)||_2 \le 2||f||_2$.

Remark 8.1.28 In Section 8.4, we will improve the results in Proposition 8.1.27 in the case when the drift term is the gradient of a convex function.

To conclude this section, let us recall the following result.

Proposition 8.1.29 Under Hypotheses 8.1.25, T(t)f belongs to $W^{1,2}_{\mu}$ for any $f \in L^2_{\mu}$ and any t > 0. Moreover,

$$||DT(t)f||_2^2 \le \frac{1}{2t} \left(||f||_2^2 - ||T(t)f||_2^2 \right), \qquad t > 0.$$

Proof. See [62, Lemma 1.3.3].

8.2 Regularity properties of invariant measures

In this section we give some conditions on the coefficients of the operator \mathcal{A} implying global regularity properties of the invariant measure μ .

In Proposition 8.1.5 we have seen that, if the semigroup $\{T(t)\}$ admits an invariant measure μ , then μ is absolutely continuous with respect to the

Lebesgue measure and its density is positive. Now, we specialize to the case when \mathcal{A} is given on smooth functions u by

$$Au = \sum_{i,j=1}^{N} D_i(q_{ij}D_j)u + \sum_{i=1}^{N} b_iD_iu := A_0u + \sum_{i=1}^{N} b_iD_iu.$$
 (8.2.1)

Let us state the main hypotheses on the coefficients q_{ij} and b_i (i, j = 1, ..., N) which will be assumed throughout this section.

Hypotheses 8.2.1 (i) $q_{ij} = q_{ji} \in C^{1+\alpha}_{loc}(\mathbb{R}^N)$ for any i, j = 1, ..., N $(N \ge 2)$ and some $\alpha \in (0, 1)$ and there exists a positive constant κ_0 such that

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_{i}\xi_{j} \ge \kappa_{0}|\xi|^{2}, \qquad x, \xi \in \mathbb{R}^{N};$$
 (8.2.2)

- (ii) the functions $x \mapsto (1+|x|^2)^{-1}q_{ij}(x)$ and $x \mapsto (1+|x|)^{-1}D_iq_{ij}(x)$ belong to L^1_μ for any $i, j = 1, \dots, N$;
- (iii) $b_i \in C^{\alpha}_{loc}(\mathbb{R}^N)$ for any i = 1, ..., N.

Remark 8.2.2 Hypothesis 8.2.1(ii) is satisfied for instance in the case when the function $x \mapsto (1+|x|^2)^{-1}q_{ij}(x)$ and $x \mapsto (1+|x|)^{-1}D_iq_{ij}(x)$ $(i,j=1,\ldots,N)$ are bounded in \mathbb{R}^N . Another sufficient condition for the integrability of the previous functions (with respect to the measure μ) can be easily given under the assumptions of the forthcoming Proposition 8.2.14, which guarantees the integrability of the function $x \mapsto \exp(\delta|x|^{\beta})$ for some positive constants β and δ .

In order to present the main results of this section (proved very recently by G. Metafune, D. Pallara and A. Rhandi in [114]) we recall the following theorem, which has been proved in [20, 21] and gives a local regularity property of the function ρ .

Theorem 8.2.3 Under Hypotheses 8.2.1(i) and 8.2.1(iii), the function ρ belongs to $W_{\text{loc}}^{1,p}(\mathbb{R}^N)$ for any $p \in [1, +\infty)$. As a consequence, ρ is continuous in \mathbb{R}^N .

Proof. We split the proof into three steps.

Step 1. Here, we show that $\rho \in L^p_{loc}(\mathbb{R}^N)$ for any p < N/(N-1). For this purpose, we begin by observing that, according to Proposition 8.1.2,

$$\int_{\mathbb{R}^N} \sum_{i,j=1}^N q_{ij} D_{ij} \varphi d\mu = -\int_{\mathbb{R}^N} \sum_{i=1}^N \overline{b_i} D_i \varphi d\mu, \qquad \varphi \in C_c^{\infty}(\mathbb{R}^N), \tag{8.2.3}$$

where

$$\overline{b}_i := b_i + \sum_{j=1}^{N} D_i q_{ij}, \qquad i = 1, \dots, N.$$

By density, it is immediate to extend (8.2.3) to any function $\varphi \in C_c^2(\mathbb{R}^N)$. Now, we fix R > 0 and a function $\vartheta \in C_c^{\infty}(B(R))$ such that $\chi_{B(R/2)} \leq \vartheta \leq \chi_{B(R)}$. For any $\psi \in C^2(\overline{B}(R))$ we write (8.2.3) with $\varphi = \psi \vartheta$. We get

$$\left| \int_{B(R)} \sum_{i,j=1}^{N} q_{ij} D_{ij} \psi \, \vartheta d\mu \right| \leq \left| 2 \int_{B(R)} \sum_{i,j=1}^{N} q_{ij} D_{i} \vartheta D_{j} \psi d\mu \right|$$

$$+ \int_{B(R)} \psi \sum_{i,j=1}^{N} q_{ij} D_{ij} \vartheta d\mu + \int_{\mathbb{R}^{N}} \vartheta \sum_{i=1}^{N} \overline{b}_{i} D_{i} \psi d\mu$$

$$+ \int_{\mathbb{R}^{N}} \psi \sum_{i=1}^{N} \overline{b}_{i} D_{i} \vartheta d\mu \right|$$

$$\leq C_{1} \|\psi\|_{C^{1}(B(R))}, \tag{8.2.4}$$

where C_1 is a positive constant, depending on R, but being independent of ψ . Let $f \in C_c^{\infty}(B(R))$ be a smooth function. By [66, Theorem 6.14 and Lemma 9.17], the equation

$$\begin{cases}
\sum_{i,j=1}^{N} q_{ij}(x)D_{ij}u(x) = f(x), & x \in B(R), \\
u(x) = 0, & x \in \partial B(R),
\end{cases}$$
(8.2.5)

admits a unique solution $u \in C^2(\overline{B}(R))$ satisfying

$$||u||_{W^{2,q}(B(R))} \le C_2 ||f||_{L^q(B(R))},$$

for any q > N and some positive constant $C_2 = C_2(q, R)$, independent of f. Using the Sobolev embedding theorems (see [2, Theorem 5.4]) we deduce that

$$||u||_{C^1(\overline{B}(R))} \le C_3 ||f||_{L^q(B(R))},$$

with C_3 being independent of f. Therefore, plugging $\psi = u$ into (8.2.4), we deduce that

$$\left| \int_{B(R)} f \, \vartheta \rho dx \right| \le C_1 C_3 ||f||_{L^q(B(R))}.$$

By the arbitrariness of $f \in C_c^{\infty}(B(R))$ and q > N the function $\rho \vartheta$ belongs to $L^p(B(R))$ for any $p \in [1, N/(N-1))$. Since $\rho \vartheta = \rho$ in B(R/2) and R is arbitrary, the assertion easily follows.

Step 2. We now show that $\rho \in W^{1,p}_{\mathrm{loc}}(\mathbb{R}^N)$ for any $p \in [1,N/(N-1))$. So, let us fix $p \in (1,N/(N-1))$ and $M \in \mathbb{N}$. For any $x_0 \in \overline{B}(M)$ and any R>0 we fix two functions $\eta, \psi \in C^2(x_0 + \overline{B}(R))$ such that $\chi_{x_0 + B(R/2)} \leq \eta \leq \chi_{x_0 + B(R)}$ and ψ vanishes on $x_0 + \partial B(R)$. We are going to prove that we can fix R sufficiently small (depending only on M) such that $\rho \in W^{1,p}(x_0 + B(R/2))$ for any $x_0 \in B(M)$. The arbitrariness of $x_0 \in B(M)$ and M>0, then, will imply that $\rho \in W^{1,p}_{\mathrm{loc}}(\mathbb{R}^N)$. Without loss of generality, we can assume that R<1.

Writing (8.2.3) with $\varphi = \psi \eta$ and using the Poincaré inequality, we get

$$\left| \int_{x_{0}+B(R)} \eta \sum_{i,j=1}^{N} q_{ij} D_{ij} \psi d\mu \right|$$

$$\leq 2 \left| \int_{x_{0}+B(R)} \sum_{i,j=1}^{N} q_{ij} D_{i} \eta D_{j} \psi d\mu \right| + \left| \int_{x_{0}+B(R)} \psi \sum_{i,j=1}^{N} q_{ij} D_{ij} \eta d\mu \right|$$

$$+ \left| \int_{x_{0}+B(R)} \eta \sum_{i=1}^{N} \overline{b}_{i} D_{i} \psi d\mu \right| + \left| \int_{x_{0}+B(R_{0})} \psi \sum_{i=1}^{N} \overline{b}_{i} D_{i} \eta d\mu \right|$$

$$\leq C_{4} \int_{x_{0}+B(R)} (|\psi| + |D\psi|) d\mu$$

$$\leq C_{5} \|D\psi\|_{L^{p/(p-1)}(x_{0}+B(R))}, \tag{8.2.6}$$

for some positive constants C_4 and C_5 , depending on R, M and the sup-norm of the coefficients of A in B(M+1), but independent of ψ and x_0 .

For any set of functions $f_i \in C_c^{\infty}(x_0 + B(R))$ (i = 1, ..., N), let us denote by $u \in C^2(x_0 + \overline{B}(R))$ the solution to the elliptic boundary problem (8.2.5) with $\sum_{i=1}^N D_i f_i$ instead of f. Plugging $\psi = u$ into (8.2.6), we get

$$\left| \int_{x_0 + B(R)} \eta \sum_{i=1}^N D_i f_i d\mu \right| \le C_5 \|Du\|_{L^{p/(p-1)}(x_0 + B(R))}. \tag{8.2.7}$$

Let us show that

$$||Du||_{L^{p/(p-1)}(x_0+B(R))} \le C_6 \sum_{i=1}^N ||D_i f_i||_{W^{-1,p/(p-1)}(x_0+B(R))}, \tag{8.2.8}$$

for some positive constant C_6 , independent of R, x_0 and f, provided that R is sufficiently small. Here, $W^{-1,p/(p-1)}(x_0+B(R))$ denotes the dual of the space $W_0^{1,p}(x_0+B(R))$. From the estimates (8.2.7) and (8.2.8) we will then immediately deduce that $\eta\rho\in W_0^{1,p}(x_0+B(R))$ and, since $\eta=1$ in $x_0+B(R/2)$, that $\rho\in W^{1,p}(x_0+B(R/2))$.

From now on, we denote by C_j positive constants, depending on M, but independent of x_0 , R and f.

To prove (8.2.8) we begin by observing that, since

$$\sum_{i,j=1}^{N} D_i(q_{ij}D_ju) = \sum_{i=1}^{N} D_if_i + \sum_{i,j=1}^{N} D_iq_{ij}D_ju := g_1 + g_2,$$

then

$$||Du||_{L^{p/(p-1)}(x_0+B(R))}$$

$$\leq C_7 \left(||g_1||_{W^{-1,p/(p-1)}(x_0+B(R))} + ||g_2||_{W^{-1,p/(p-1)}(x_0+B(R))} \right);$$
 (8.2.9)

see, e.g., [64, Section 4.3] or [80, Section 5.5].

To estimate the norm of the function g_2 we observe that, since p < N, then $L^{pN/(N(p-1)+p)}(x_0 + B(R)) \subset W^{-1,p/(p-1)}(x_0 + B(R))$ and

$$||k||_{W^{-1,p/(p-1)}(x_0+B(R))} \le C_8 ||k||_{L^{pN/(N(p-1)+p)}(x_0+B(R))}.$$

for any $k \in W^{-1,p/(p-1)}(x_0 + B(R))$. Therefore,

$$||g_{2}||_{W^{-1,p/(p-1)}(x_{0}+B(R))} \leq C_{9}||g_{2}||_{L^{pN/(N(p-1)+p)}(x_{0}+B(R))}$$

$$\leq C_{10}||h||_{L^{N}(x_{0}+B(R))}||Du||_{L^{p/(p-1)}(x_{0}+B(R))}$$

$$\leq C_{11}R||h||_{L^{\infty}(B(M+1))}||Du||_{L^{p/(p-1)}(x_{0}+B(R))}.$$
(8.2.10)

Here, $h^2 = \sum_{i,j=1}^N |D_i q_{ij}|^2$. From (8.2.9) and (8.2.10) we deduce that $\|Du\|_{L^{p/(p-1)}(x_0+B(R))}$

$$\leq C_{12} \left(R \|Du\|_{L^{p/(p-1)}(x_0 + B(R))} + \|g_1\|_{W^{-1,p/(p-1)}(x_0 + B(R))} \right). \tag{8.2.11}$$

Taking R sufficiently small, we get the estimate (8.2.8).

Step 3. In this step we conclude the proof, using a bootstrap argument. Since $\rho \in W^{1,p}_{\mathrm{loc}}(\mathbb{R}^N)$ for any $p \in [1,N/(N-1))$, the Sobolev embedding theorems imply that $\rho \in L^p_{\mathrm{loc}}(\mathbb{R}^N)$ for any $p \in [1,N/(N-2))$. Now, repeating the argument in Step 2, we deduce that $\rho \in W^{1,p}_{\mathrm{loc}}(\mathbb{R}^N)$ for any $p \in [1,N/(N-2))$. Iterating this argument, we can show that $\rho \in L^p_{\mathrm{loc}}(\mathbb{R}^N)$ for any $p \in [1,+\infty)$.

Now, we observe that we can adapt the arguments in Step 2 to prove that $\rho \in W^{1,p}_{loc}(\mathbb{R}^N)$ also in the case when p > N. As it is immediately seen, we just need to modify the estimate (8.2.10). For this purpose, we observe that, if p > N, then $L^1(x_0 + B(R))$ is continuously embedded in $W^{-1,p/(p-1)}(x_0 + B(R))$ and

$$||k||_{W^{-1,p/(p-1)}(x_0+B(R))} \le C_{13}R^{\frac{p-N}{p}}||k||_{L^1(x_0+B(R))},$$

for any $k \in W^{-1,p/(p-1)}(x_0 + B(R))$. Hence, in this case we have

$$||g_2||_{W^{-1,p/(p-1)}(x_0+B(R))} \le C_{14} R^{\frac{p-N}{p}} ||g_2||_{L^1(x_0+B(R))}$$

$$\le C_{15} R^{\frac{p-N}{p}} ||h||_{L^p(x_0+B(R))} ||Du||_{L^{p/(p-1)}(x_0+B(R))}$$

$$\le C_{16} R ||h||_{L^{\infty}(B(M+1))} ||Du||_{L^{p/(p-1)}(x_0+B(R))}.$$

Taking R sufficiently small, we get (8.2.8) and, consequently, we obtain that $u \in W_{loc}^{1,p}(\mathbb{R}^N)$ also in this case.

- **Remark 8.2.4** (i) The previous theorem applies also to the case when μ is just an infinitesimally invariant measure. See Remark 8.1.4 for the definition of infinitesimally invariant measures.
 - (ii) The result in Theorem 8.2.3 holds also in the case when N=1. To see it, we denote by x the variables in \mathbb{R} and we introduce the operator \mathcal{B} defined on smooth functions of two variables (x,y) by

$$\mathcal{B}u(x,y) = \mathcal{A}u(x,y) + D_{yy}u(x,y) - yD_yu(x,y), \qquad (x,y) \in \mathbb{R}^2.$$

Moreover, we denote by $\widetilde{\mu}$ the measure in \mathbb{R}^2 whose density with respect to the Lebesgue measure is given by

$$\widetilde{\rho}(x,y) = \frac{1}{\sqrt{2\pi}}\rho(x)e^{-\frac{1}{2}y^2}, \qquad (x,y) \in \mathbb{R}^2.$$

Of course, the coefficients of the operator \mathcal{B} satisfy the assumptions of Theorem 8.2.3. Moreover, for any $\varphi \in C_c^{\infty}(\mathbb{R}^2)$, we have

$$\int_{\mathbb{R}^{2}} \mathcal{B}\varphi \, d\widetilde{\mu} = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-\frac{1}{2}y^{2}} dy$$

$$\times \int_{\mathbb{R}} (\mathcal{A}\varphi(x,y) + \varphi_{yy}(x,y) - y\varphi_{y}(x,y))\rho(x) dx$$

$$= \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-\frac{1}{2}y^{2}} \left(\int_{\mathbb{R}} (\varphi_{yy}(x,y) - y\varphi_{y}(x,y))\rho(x) dx \right) dy$$

$$= \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} e^{-\frac{1}{2}y^{2}} (D_{yy} - yD_{y}) \left(\int_{\mathbb{R}} \varphi(x,y)\rho(x) dx \right) dy$$

$$= \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} (D_{yy}e^{-\frac{1}{2}y^{2}} + D_{y}(ye^{-\frac{1}{2}y^{2}}) \left(\int_{\mathbb{R}^{2}} \varphi(x,y)\rho(x) dx \right) dy$$

$$= 0.$$

Therefore, from the results in Theorem 8.2.3, the function $\widetilde{\rho}$ belongs to $W^{1,p}_{\mathrm{loc}}(\mathbb{R}^2)$ and, hence, $\rho \in W^{1,p}_{\mathrm{loc}}(\mathbb{R})$ for any $p \in [1,+\infty)$.

(iii) The reason why, throughout this section, we decide to deal only with the case when $N \geq 2$, is based on the fact that, when N=1, an explicit representation formula for the density of the infinitesimally invariant measure μ is available. Indeed, in such a case, a straightforward integration by parts shows that

$$\int_{\mathbb{R}} \{q(x)\rho'(x) - b(x)\rho(x)\}\varphi'(x)dx = 0,$$

for any $\varphi \in C_c^{\infty}(\mathbb{R})$. Hence,

$$q(x)\rho'(x) - b(x)\rho(x) = C, \qquad x \in \mathbb{R},$$

for some real constant C.

8.2.1 Global L^q -regularity of the density ρ

In this subsection we are devoted to prove some L^q -regularity results for the function ρ . By virtue of Remark 8.2.4(iii), we limit ourselves to dealing with the case when N > 2.

Theorem 8.2.5 Assume Hypothesis 8.2.1. Further, suppose that $b_i \in L^p_\mu$ for any i = 1, ..., N and some $p \in [2, +\infty)$. Then, the following properties are met:

- (i) if $N \geq 3$ and $p \in [2, N)$, then $\rho \in L^q(\mathbb{R}^N)$ for any $q \leq \frac{N}{N-p}$;
- (ii) if p = N, then $\rho \in L^q(\mathbb{R}^N)$ for any $q \in [1, +\infty)$;
- (iii) if p > N, then $\rho \in C_b(\mathbb{R}^N)$.

Remark 8.2.6 As it has been already noticed in Remark 8.2.2, one can take advantage of the forthcoming Proposition 8.2.14 for a sufficient condition ensuring that the integrability assumptions on the coefficients b_i (i = 1, ..., N) are satisfied.

Proof of Theorem 8.2.5. To prove the assertion we use a bootstrap argument. Since the proof is rather long we divide it into three steps. First, in Step 1 we assume that p=2 and we prove that ρ belongs to $L^{N/(N-2)}(\mathbb{R}^N)$, if N>2, and to $L^q(\mathbb{R}^N)$ for any $q\in [1,+\infty)$, if N=2. For this purpose, according to the Sobolev embedding theorems (see [2, Theorem 5.4]), it suffices to show that $\sqrt{\rho} \in W^{1,2}(\mathbb{R}^N)$. Next in Step 2, where we assume that N>2, we prove that, if $\rho \in L^r(\mathbb{R}^N)$ for some $r\in (1,+\infty)$, then actually it belongs to $L^q(\mathbb{R}^N)$, where q=N(r(p-2)+2)/(p(N-2)). Finally, in Step 3 we conclude the proof. Throughout the proof, for notational convenience, we set

$$q(\xi, \eta) = \sum_{i,j=1}^{N} q_{ij}\xi_i \eta_j, \qquad \xi, \eta \in \mathbb{R}^N.$$

We also stress that, throughout the proof, actually we do not require μ to be the invariant measure of $\{T(t)\}$. We just need that μ is infinitesimally invariant, i.e.,

$$\int_{\mathbb{R}^N} \mathcal{A}\varphi \, d\mu = 0, \qquad \varphi \in C_c^{\infty}(\mathbb{R}^N). \tag{8.2.12}$$

Of course, according to Proposition 8.1.2, if μ is the invariant measure of $\{T(t)\}$, then the previous condition is always satisfied.

Step 1. Integrating by parts (taking Theorem 8.2.3 into account), from the formula (8.2.12) we easily see that

$$\int_{\mathbb{R}^N} q(D\rho, D\varphi) dx = \int_{\mathbb{R}^N} \sum_{j=1}^N b_j D_j \varphi d\mu, \qquad (8.2.13)$$

for any $\varphi \in C_c^\infty(\mathbb{R}^N)$. Since ρ is continuous (see again Theorem 8.2.3), then $b\rho \in L^2_{\mathrm{loc}}(\mathbb{R}^N)$. Therefore, by density, we can extend (8.2.13) to any $\varphi \in W^{1,2}(\mathbb{R}^N)$ with compact support.

Now, for any ε, k such that $0 < \varepsilon < 1 < k$, we introduce the function $\rho_{\varepsilon,k} : \mathbb{R}^N \to \mathbb{R}$ defined by

$$\rho_{\varepsilon,k}(x) = (\rho(x) \vee \varepsilon) \wedge k, \qquad x \in \mathbb{R}^N.$$
 (8.2.14)

Moreover, let $\{\vartheta_n\}$ be the sequence of smooth functions defined by $\vartheta_n(x) = \vartheta(x/n)$ for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, where $\vartheta \in C_c^{\infty}(\mathbb{R}^N)$ is such that $\chi_{B(1/2)} \leq \vartheta \leq \chi_{B(1)}$. A straightforward computation shows that there exists a positive constant C such that

$$|D\vartheta_n(x)| \le \frac{C}{1+|x|}, \qquad |D^2\vartheta_n(x)| \le \frac{C}{1+|x|^2}, \quad x \in \mathbb{R}^N.$$
 (8.2.15)

As it is easily seen, the function $\log(\rho_{\varepsilon,k})$ belongs to $W^{1,p}_{\mathrm{loc}}(\mathbb{R}^N) \cap L^{\infty}(\mathbb{R}^N)$ for any $n \in \mathbb{N}$. Plugging $\varphi = \vartheta_n^2 \log(\rho_{\varepsilon,k})$ in (8.2.13), we get

$$\int_{\mathbb{R}^N} \vartheta_n^2 \chi_{\{\varepsilon < \rho < k\}} \frac{q(D\rho, D\rho)}{\rho} dx = -2 \int_{\mathbb{R}^N} \vartheta_n \log(\rho_{\varepsilon, k}) q(D\rho, D\vartheta_n) dx
+ \int_{\mathbb{R}^N} \vartheta_n^2 \sum_{j=1}^N b_j D_j \rho \chi_{\{\varepsilon < \rho < k\}} dx
+ 2 \int_{\mathbb{R}^N} \vartheta_n \rho \log(\rho_{\varepsilon, k}) \sum_{j=1}^N b_j D_j \vartheta_n dx
:= I_n + J_n + K_n.$$

Let us estimate the term I_n . An integration by parts shows that

$$\begin{split} I_n &= 2 \int_{\mathbb{R}^N} \log(\rho_{\varepsilon,k}) q(D\vartheta_n, D\vartheta_n) d\mu + 2 \int_{\mathbb{R}^N} \vartheta_n \log(\rho_{\varepsilon,k}) \sum_{i,j=1}^N q_{ij} D_{ij} \vartheta_n d\mu \\ &+ 2 \int_{\mathbb{R}^N} \vartheta_n \log(\rho_{\varepsilon,k}) \sum_{i,j=1}^N D_i q_{ij} D_j \vartheta_n d\mu \\ &+ 2 \int_{\mathbb{R}^N} \vartheta_n \chi_{\{\varepsilon < \rho < k\}} \frac{q(D\rho, D\vartheta_n)}{\rho} d\mu. \end{split}$$

Using the Young inequality, we get

$$|I_{n}| \leq C_{\varepsilon,k} \int_{\mathbb{R}^{N}} \left(q(D\vartheta_{n}, D\vartheta_{n}) + \operatorname{Tr}(QD^{2}\vartheta_{n}) + \sum_{i,j=1}^{N} |D_{i}q_{ij}D_{j}\vartheta_{n}| \right) d\mu$$

$$+ 2 \left(\int_{\mathbb{R}^{N}} \vartheta_{n}^{2} \chi_{\{\varepsilon < \rho < k\}} \frac{q(D\rho, D\rho)}{\rho} dx \right)^{\frac{1}{2}} \left(\int_{\mathbb{R}^{N}} q(D\vartheta_{n}, D\vartheta_{n}) \rho dx \right)^{\frac{1}{2}}$$

$$\leq C_{\varepsilon,k} \int_{\mathbb{R}^{N}} \left(q(D\vartheta_{n}, D\vartheta_{n}) + \operatorname{Tr}(QD^{2}\vartheta_{n}) + \sum_{i,j=1}^{N} |D_{i}q_{ij}D_{j}\vartheta_{n}| \right) d\mu$$

$$+ \frac{1}{\delta^{2}} \int_{\mathbb{R}^{N}} q(D\vartheta_{n}, D\vartheta_{n}) d\mu + \delta \int_{\mathbb{R}^{N}} \vartheta_{n}^{2} \chi_{\{\varepsilon < \rho < k\}} \frac{q(D\rho, D\rho)}{\rho} dx, \quad (8.2.16)$$

where $C_{\varepsilon,k} = |\log(\varepsilon)| \vee \log(k)$. Similarly, we have

$$J_n + K_n \le \delta \int_{\mathbb{R}^N} \vartheta_n^2 \chi_{\{\varepsilon < \rho < k\}} \frac{|D\rho|^2}{\rho} dx + \frac{1}{4\delta^2} \int_{\mathbb{R}^N} |b|^2 \rho dx + \frac{CC_{\varepsilon,k}}{n} \int_{\mathbb{R}^N} |b| d\mu,$$
(8.2.17)

for any $\delta>0.$ From (8.2.16) and (8.2.17) and Hypothesis 8.2.1(i) it follows that

$$\begin{split} &\left(1-\delta-\frac{\delta}{\kappa_0}\right)\int_{\mathbb{R}^N}\vartheta_n^2\chi_{\{\varepsilon<\rho< k\}}\frac{q(D\rho,D\rho)}{\rho}dx\\ &\leq C_{\varepsilon,k}\int_{\mathbb{R}^N}\left(q(D\vartheta_n,D\vartheta_n)+\mathrm{Tr}(QD^2\vartheta_n)+\sum_{i,j=1}^N|D_iq_{ij}D_j\vartheta_n|\right)d\mu\\ &+\frac{1}{\delta^2}\int_{\mathbb{R}^N}q(D\vartheta_n,D\vartheta_n)d\mu+\frac{CC_{\varepsilon,k}}{n}\int_{\mathbb{R}^N}|b|d\mu+\frac{1}{4\delta^2}\int_{\mathbb{R}^N}|b|^2d\mu. \end{split}$$

Letting n go to $+\infty$ and taking Hypothesis 8.2.1(ii) and the condition (8.2.15) into account, from the dominated convergence theorem we deduce that

$$\left(1 - \delta - \frac{\delta}{\kappa_0}\right) \int_{\mathbb{R}^N} \chi_{\{\varepsilon < \rho < k\}} \frac{q(D\rho, D\rho)}{\rho} dx \le \frac{1}{4\delta^2} \int_{\mathbb{R}^N} |b|^2 \rho dx,$$

for any $\delta > 0$ and any $0 < \varepsilon < 1 < k$. Now, taking δ small enough and letting ε go to 0^+ and k go to $+\infty$, we deduce that $\rho^{-1}q(D\rho, D\rho)$ (and, hence, $\rho^{-1}|D\rho|^2$) belongs to $L^1(\mathbb{R}^N)$ and

$$\kappa_0 \int_{\mathbb{R}^N} \frac{|D\rho|^2}{\rho} dx \le \int_{\mathbb{R}^N} \frac{q(D\rho, D\rho)}{\rho} dx \le \widetilde{C} \int_{\mathbb{R}^N} |b|^2 d\mu, \tag{8.2.18}$$

for a suitable positive constant \widetilde{C} . Hence, $\sqrt{\rho} \in W^{1,2}(\mathbb{R}^N)$. Note that $\rho \in L^1(\mathbb{R}^N)$ according to Proposition 8.1.5.

Step 2. Now, we assume that N>2 and that $b_i\in L^p_\mu$ for any $i=1,\ldots,N$, and we prove that, if $\rho\in L^r(\mathbb{R}^N)$ for some $r\in (1,+\infty)$, then, actually, it belongs to $L^q(\mathbb{R}^N)$, where q=N(r(p-2)+2)/(p(N-2)). For this purpose, we prove that $\rho^{(\beta-1)/2}|D\rho|\in L^2(\mathbb{R}^N)$, where $\beta=(r-1)(1-2/p)$, and

$$\kappa_0^2 \int_{\mathbb{R}^N} \rho^{\beta - 1} |D\rho|^2 dx \le \int_{\mathbb{R}^N} |b|^2 \rho^{\beta + 1} dx. \tag{8.2.19}$$

The estimate (8.2.19) will imply that the function $\rho^{(\beta+1)/2}$ is an element of $W^{1,2}(\mathbb{R}^N)$ (note that, since $\beta+1<2r$ and $\rho\in L^1(\mathbb{R})$, then $\rho\in L^s(\mathbb{R}^N)$ for any $s\in [1,r]$). Again, the Sobolev embedding theorems will yield $\rho\in L^q(\mathbb{R}^N)$.

So, let us prove (8.2.19). We observe that the integral in the right-hand side of (8.2.19) converges since, by the Hölder inequality, we have

$$\int_{\mathbb{R}^{N}} |b|^{2} \rho^{\beta+1} dx \le \left(\int_{\mathbb{R}^{N}} |b|^{p} d\mu \right)^{\frac{2}{p}} \left(\int_{\mathbb{R}^{N}} \rho^{r} dx \right)^{1-\frac{2}{p}}, \tag{8.2.20}$$

and the right-hand side of (8.2.20) is finite. The estimate (8.2.19) can be proved by applying the same arguments as in Step 1. Indeed, plugging $\varphi_n = \vartheta_n^2 \rho_{\varepsilon,k}^{\beta}$ into (8.2.13) we get

$$\beta \int_{\mathbb{R}^{N}} \vartheta_{n}^{2} \rho^{\beta-1} \chi_{\{\varepsilon < \rho < k\}} q(D\rho, D\rho) dx = -2 \int_{\mathbb{R}^{N}} \vartheta_{n} \rho_{\varepsilon, k}^{\beta} q(D\rho, D\vartheta_{n}) dx$$

$$+\beta \int_{\mathbb{R}^{N}} \vartheta_{n}^{2} \rho^{\beta} \chi_{\{\varepsilon < \rho < k\}} \sum_{i=1}^{N} b_{i} D_{i} \rho dx$$

$$+2 \int_{\mathbb{R}^{N}} \vartheta_{n} \rho \rho_{\varepsilon, k}^{\beta} \sum_{i=1}^{N} b_{i} D_{i} \theta_{n} dx$$

$$= I_{n} + J_{n} + K_{n}. \tag{8.2.21}$$

With the same computations as in Step 1, we can easily show that

$$\left(\beta \left(1 - \frac{\delta}{\kappa_0}\right) - \delta\right) \int_{\mathbb{R}^N} \eta_n^2 \rho^{\beta - 1} \chi_{\{\varepsilon < \rho < k\}} q(D\rho, D\rho) dx$$

$$\leq \frac{1}{4\delta^2} \int_{\mathbb{R}^N} |b|^2 \rho^{\beta + 1} dx + C(\varepsilon, k, \delta, n), \tag{8.2.22}$$

where $C(\varepsilon, k, \delta, n)$ is a positive constant which tends to 0 as n tends to $+\infty$, for any fixed ε, k, δ . Therefore, letting first n go to $+\infty$ and, then, ε and k, respectively, to 0 and $+\infty$, we deduce that the function $\rho^{\beta-1}q(D\rho, D\rho)$ (and, consequently, the function $\rho^{\beta-1}|D\rho|^2$) belongs to $L^1(\mathbb{R}^N)$. Now, from (8.2.21) the formula (8.2.19) easily follows.

Step 3. Here we make the bootstrap argument work to prove the assertions (i)-(iii). We first assume that N > 2. In such a situation the results of Steps 1

and 2 allow us to show that $\rho \in L^{q_n}(\mathbb{R}^N)$ for any $n \in \mathbb{N}$, where q_n is defined recursively by the following relation

$$\begin{cases} q_{n+1} = \frac{N(p-2)}{p(N-2)}q_n + \frac{2N}{p(N-2)}, & n \ge 1, \\ q_1 = \frac{N}{N-2}. \end{cases}$$

Therefore,

$$q_n = \frac{N^n(p-2)^n}{p^{n-1}(N-2)^n(p-N)} + \frac{N}{N-p}, \qquad n \ge 2,$$

if $p \neq N$, whereas

$$q_n = \frac{2n+N-2}{N-2}, \qquad n \ge 2,$$

if p = N. In the case when $p \in (2, N)$ it is immediate to check that q_n converges to N/(N-p) as n tends to $+\infty$. Hence, to complete the proof of the assertion (i) it suffices to show that the sequence $\{\|\rho\|_{L^{q_n}(\mathbb{R}^N)}\}$ is bounded. Indeed, once this is proved, we will easily deduce that

$$\|\rho\|_{L^{N/(N-p)}(B(k))} = \lim_{n \to +\infty} \|\rho\|_{L^{q_n}(B(k))} \leq \limsup_{n \to +\infty} \|\rho\|_{L^{q_n}(\mathbb{R}^N)} \leq \sup_{n \in \mathbb{N}} \|\rho\|_{L^{q_n}(\mathbb{R}^N)},$$

for any $k \in \mathbb{N}$, and the Fatou lemma will imply that $\|\rho\|_{L^{N/(N-p)}(\mathbb{R}^N)}$ is finite. Let us observe that according to the formulas (8.2.19) and (8.2.20) and the Sobolev embedding theorems, we have

$$\|\rho\|_{L^{q_{n+1}}(\mathbb{R}^N)} \leq C_1^{\frac{2N}{q_{n+1}(N-2)}} \|\rho\|_{L^{q_n}(\mathbb{R}^N)}^{\frac{N}{N-2}\frac{p-2}{p}\frac{q_n}{q_{n+1}}},$$

where $C_1 = C||b||_p$, C being the constant appearing in the Sobolev embedding theorems. Equivalently, setting $\alpha_n = \log(\|\rho\|_{L^{q_n}(\mathbb{R}^N)})$, we have

$$\alpha_{n+1} \le \frac{2N}{q_{n+1}(N-2)} |\log(C_1)| + \frac{N}{N-2} \frac{p-2}{p} \frac{q_n}{q_{n+1}} \alpha_n.$$
 (8.2.23)

Since N(p-2)/(p(N-2)) < 1 and $\{q_n\}$ is increasing, from (8.2.23) it follows that there sequence $\{\alpha_n\}$ satisfies

$$\alpha_{n+1} \le \beta + \gamma \alpha_n, \qquad n \in \mathbb{N},$$

for some $\beta > 0$ and some $\gamma \in (0,1)$, and this implies that $\{\alpha_n\}$ and, consequently, $\{\|\rho\|_{L^{q_n}(\mathbb{R}^N)}\}$ are bounded. On the other hand, in the case when $p \geq N$ the sequence $\{q_n\}$ is increasing and it diverges to $+\infty$. Hence, the assertion (ii) easily follows. To prove that ρ belongs to $L^{\infty}(\mathbb{R}^N)$ when p > N,

we will show that the L^{q_n} -norms of ρ are bounded by a constant C, independent of n. This will be enough for our purposes. Indeed, since ρ is continuous then, for any $k \in \mathbb{N}$,

$$\|\rho\|_{C(B(k))} = \lim_{n \to +\infty} \|\rho\|_{L^{q_n}(B(k))} \le \limsup_{n \to +\infty} \|\rho\|_{L^{q_n}(\mathbb{R}^N)} < C.$$

The arbitrariness of k then will imply that $\rho \in L^{\infty}(\mathbb{R}^N)$.

To prove the boundedness of $\|\rho\|_{L^{q_n}(\mathbb{R}^N)}$ we argue as in the case when p < N. From (8.2.23) we easily deduce that

$$\alpha_n \le \beta_{n-1} + \sum_{k=1}^{n-2} \beta_{n-1-k} \prod_{l=n-k}^{n-1} \gamma_l + \alpha_1 \prod_{k=1}^{n-1} \gamma_k, \qquad n \ge 2,$$

where

$$\beta_j = \frac{2N}{q_{j+1}(N-2)} |\log(C_1)|, \quad \gamma_j = \frac{N}{N-2} \frac{p-2}{p} \frac{q_j}{q_{j+1}} = 1 - \frac{2N}{p(N-2)} \frac{1}{q_{j+1}},$$
(8.2.24)

for any j. Observing that $\gamma_l \leq 1$ for any l, we get

$$\alpha_n \le \beta_{n-1} + \sum_{k=1}^{n-2} \beta_k + \alpha_1.$$

Since, $\sum_{k=1}^{+\infty} \beta_k < +\infty$, we immediately deduce that

$$\alpha_n \le \alpha_1 + \sum_{k=1}^{+\infty} \beta_k,$$

for any $n \in \mathbb{N}$ and we are done.

To conclude the proof, let us consider the case when N=2. The property (ii) easily follows from the Sobolev embedding theorems since, as we have shown in Step 1, the function $\sqrt{\rho}$ belongs to $W^{1,2}(\mathbb{R}^N)$.

To prove the property (iii) in the case when N=2 we observe that, since $L^p_{\mu}(\mathbb{R}^N)$ is continuously embedded in $L^2_{\mu}(\mathbb{R}^N)$, then $\rho \in L^q(\mathbb{R}^N)$ for any $q \in [1,+\infty)$. Therefore, it belongs to $L^{r_0/(r_0-2)}(\mathbb{R}^N)$, where r_0 is such that $2 < r_0 < p$.

Adapting the proof of Step 2 to this situation, we can easily show that, if $\rho \in L^r(\mathbb{R}^N)$ for some r > 1, then, actually, it belongs to $L^q(\mathbb{R}^N)$, where $q = r_0(r(p-2)+2)/(p(r_0-2))$.

Now, applying the same bootstrap argument used in the case when $N \geq 3$, we can show that $\rho \in L^{q_n}(\mathbb{R}^N)$ for any $n \in \mathbb{N}$, where now the sequence $\{q_n\}$ is defined recursively by

$$\begin{cases} q_{n+1} = \frac{r_0(p-2)}{p(r_0-2)}q_n + \frac{2r_0}{p(r_0-2)}, & n \ge 2, \\ q_1 = \frac{r_0}{r_0-2}. & \end{cases}$$

Since $r_0(p-2)/(p(r_0-2)) > 1$, we can conclude that $\rho \in C_b(\mathbb{R}^N)$ repeating step by step the same arguments used in the case when N > 2.

Remark 8.2.7 In fact, the proof of Theorem 8.2.5 shows some differentiability properties of the density ρ . In particular, it shows that, if $b_i \in L^p_\mu$ for any $i=1,\ldots,N$ and some $p \geq (N+2)/2$, p>2, then $\rho \in W^{1,2}(\mathbb{R}^N)$. To see it, first suppose that $N\geq 3$. By Theorem 8.2.5(i), $\rho \in L^q(\mathbb{R}^N)$ for any $q\in [1,N/(N-p)]$. Hence, in particular, it belongs to $L^{2N/(N-2)}(\mathbb{R}^N)$. Therefore, we can apply Step 2 in the proof of Theorem 8.2.5, taking $\beta=1$. This shows that $D\rho \in L^2(\mathbb{R}^N)$, and we are done.

In the case when N=2, Theorem 8.2.5 implies that $\rho \in L^q(\mathbb{R}^N)$ for any $q \in [1, +\infty]$. In particular, it belongs to $L^{2(p-1)/(p-2)}(\mathbb{R}^N)$ and, again, we can apply Step 2 in the proof of Theorem 8.2.5 to get the assertion.

A result similar to whose in Theorem 8.2.5(iii) can be proved, without assuming p > N, in the case when the function b is much more regular.

We refer the reader to the forthcoming Proposition 8.2.14 for a sufficient condition ensuring that the integrability assumptions on the coefficients b_i (i = 1, ..., N) in the following theorem are satisfied.

Theorem 8.2.8 Suppose that $b_i \in W^{1,\infty}_{loc}(\mathbb{R}^N)$ and b_i , div $b \in L^p_\mu$ for some p > N/2, $p \geq 2$ and any i = 1, ..., N. Then $\rho \in C_b(\mathbb{R}^N)$.

Proof. The proof is similar to that of Theorem 8.2.5. To make the same arguments as in the proof of the quoted theorem work, we need to show that if $\rho \in L^r(\mathbb{R}^N)$ for some r > 1, then the function $\rho^{(\beta+1)/2}$ belongs to $W^{1,2}(\mathbb{R}^N)$, where $\beta = (r-1)(1-1/p)$. For this purpose it suffices to show that

$$\kappa_0(\beta+1) \int_{\mathbb{R}^N} \rho^{\beta-1} |D\rho|^2 dx \le -\int_{\mathbb{R}^N} \rho^{\beta+1} \operatorname{div} b \, dx, \tag{8.2.25}$$

since integral in the right-hand side of (8.2.25) converges (compare with (8.2.20)). Multiplying the (distributional) identity $\mathcal{A}_0\rho = \operatorname{div}(\rho b)$ by $\vartheta_n^2 \rho_{\varepsilon,k}^\beta$ (where $\rho_{\varepsilon,k}$ is given by (8.2.14) and $\{\vartheta_n\}$ is the sequence of smooth functions defined by $\vartheta_n(x) = \vartheta(x/n)$ for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, $\vartheta \in C_c^\infty(\mathbb{R}^N)$ being any smooth function such that $\chi_{B(1/2)} \leq \vartheta \leq \chi_{B(1)}$), we obtain again (8.2.21). Here, \mathcal{A}_0 is given by (8.2.1). The terms I_n and K_n can be estimated as in Step 1 in the proof of Theorem 8.2.5. As far as the term J_n is concerned, we observe that $\rho^\beta \chi_{\{\varepsilon < \rho < k\}} D\rho = (\beta+1)^{-1} D\rho_{\varepsilon,k}^{\beta+1}$. Therefore, since $\beta+1 < r$, both the functions $\rho_{\varepsilon,k}^{\beta+1}$ and $\rho^{\beta+1}$ belong to $L^1(\mathbb{R}^N)$. Hence, integrating by parts, we can write

$$J_n = -\frac{\beta}{\beta + 1} \int_{\mathbb{R}^N} \rho_{\varepsilon,k}^{\beta + 1} \sum_{i=1}^N b_i D_i(\vartheta_n^2) dx - \frac{\beta}{\beta + 1} \int_{\mathbb{R}^N} \rho_{\varepsilon,k}^{\beta + 1} \vartheta_n^2 \operatorname{div} b \, dx.$$

Now, letting n go to $+\infty$, yields

$$\lim_{n \to +\infty} J_n = -\frac{\beta}{\beta + 1} \int_{\mathbb{R}^N} \rho_{\varepsilon,k}^{\beta + 1} \operatorname{div} b \, dx.$$

Therefore, arguing as in the proof of (8.2.22), we get

$$\beta(1-\delta) \int_{\mathbb{R}^N} \rho^{\beta-1} \chi_{\{\varepsilon < \rho < k\}} \sum_{i,j=1}^N q_{ij} D_i \rho D_j \rho dx \le -\frac{\beta}{\beta+1} \int_{\mathbb{R}^N} \rho_{\varepsilon,k}^{\beta+1} \operatorname{div} b \, dx.$$

Finally, letting ε and δ go to 0^+ and k go to $+\infty$, (8.2.25) follows.

8.2.2 Global Sobolev regularity

In this subsection we prove some Sobolev regularity results for ρ under the following more restrictive assumptions on the diffusion coefficients q_{ij} (i, j = 1, ..., N).

Hypotheses 8.2.9 (i) Hypotheses 8.2.1(i) and 8.2.1(iii) are satisfied;

(ii) the functions q_{ij} belong to $C_b^1(\mathbb{R}^N)$ for any i, j = 1, ..., N.

Theorem 8.2.10 Under Hypotheses 8.2.9 the following properties are met:

(i) if
$$b_i \in L^1_{\mu}$$
 $(i = 1, ..., N)$, then $\rho \in L^q(\mathbb{R}^N)$ for any $q \in \left[1, \frac{N}{N-1}\right]$;

- (ii) if $b_i \in L^p_\mu$ (i = 1, ..., N) for some $p \in (1, 2)$, then $\rho \in W^{1,q}(\mathbb{R}^N)$ for any $q \in \left(1, \frac{N}{N-p+1}\right)$;
- (iii) if $b_i \in L^p_\mu$ (i = 1, ..., N) for some $p \in [2, N)$, then $\rho \in W^{1,q}(\mathbb{R}^N)$ for any $q \in \left[1, \frac{N}{N-p+1}\right]$;
- (iv) if $b_i \in L^N_\mu$ (i = 1, ..., N), then $\rho \in W^{1,q}(\mathbb{R}^N)$ for any $q \in [1, N)$;
- (v) if $b_i \in L^p_\mu$ (i = 1, ..., N) for some p > N, then $\rho \in W^{1,q}(\mathbb{R}^N)$ for any $q \in [1, p]$.

Proof. We begin the proof checking the property (i). For this purpose, we observe that, according to (8.2.12), we can write

$$\int_{\mathbb{R}^N} (\varphi - \mathcal{A}_0 \varphi) \, d\mu = \int_{\mathbb{R}^N} \left(\varphi + \sum_{i=1}^N b_i D_i \varphi \right) d\mu,$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$. Here \mathcal{A}_0 is defined by (8.2.1). Then, using the Hölder inequality we get

 $\left| \int_{\mathbb{R}^N} (\varphi - \mathcal{A}_0 \varphi) \, d\mu \right| \le C \|\varphi\|_{1,\infty}, \tag{8.2.26}$

for some positive constant C, independent of φ . Let us now show that (8.2.26) implies that

$$\left| \int_{\mathbb{R}^N} f d\mu \right| \le C \|f\|_{L^q(\mathbb{R}^N)}, \tag{8.2.27}$$

for any $f \in L^q(\mathbb{R}^N)$ and any q > N. This, of course, will lead us to the assertion thanks to the Riesz-Fisher representation theorem. Without loss of generality we assume that $f \in C_c^\infty(\mathbb{R}^N)$. According to [66, Theorems 9.15], the equation $w - \mathcal{A}_0 w = f$ admits a unique solution $w \in W^{2,q}(\mathbb{R}^N)$. Moreover, there exists a positive constant C, independent of f, such that $\|w\|_{W^{2,q}(\mathbb{R}^N)} \leq C\|f\|_{L^q(\mathbb{R}^N)}$ (see [66, Lemma 9.17]). Since q > N, the Sobolev embedding theorems imply that $w \in C_b^1(\mathbb{R}^N)$ and $\|w\|_{C_b^1(\mathbb{R}^N)} \leq \widetilde{C}\|f\|_{L^q(\mathbb{R}^N)}$ for some constant \widetilde{C} , independent of w.

To get (8.2.27) we just need to show that we can plug w into (8.2.26). For this purpose, we use an approximation argument. Let $\{\vartheta_n\}$ be the sequence of smooth functions defined by $\vartheta_n(x) = \vartheta(x/n)$ for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, where $\vartheta \in C_c^{\infty}(\mathbb{R}^N)$ is such that $\chi_{B(1/2)} \leq \vartheta \leq \chi_{B(1)}$. As it is easily seen, the function $\vartheta_n w$ tends to w in $W^{2,q}(\mathbb{R}^N)$ and, consequently, $\mathcal{A}_0(\vartheta_n w)$ tends to $\mathcal{A}_0 w$ in $C_b(\mathbb{R}^N)$. We now regularize the sequence $\{\vartheta_n w\}$ by convolution in a standard way, introducing the sequence $\{w_{\varepsilon,n}\}$ defined by

$$w_{\varepsilon,n} = \int_{\mathbb{R}^N} \vartheta_n(y) w(y) \xi_{\varepsilon}(x-y) dy, \qquad x \in \mathbb{R}^N,$$

for any $\varepsilon > 0$ and any $n \in \mathbb{N}$, where $\xi_{\varepsilon}(x) = \varepsilon^{-N} \xi(\varepsilon^{-1} x)$ and $\xi \in C_c^{\infty}(\mathbb{R}^N)$ is a smooth function with $\|\xi\|_{L^1(\mathbb{R}^N)} = 1$. Of course, $w_{\varepsilon,n} \in C_c^{\infty}(\mathbb{R}^N)$ and it tends to $\vartheta_n w$ in $W^{2,q}(\mathbb{R}^N)$ as ε tends to 0. Moreover,

$$(\mathcal{A}_{0}w_{\varepsilon,n})(x) = \int_{\mathbb{R}^{N}} (\mathcal{A}_{0}(\vartheta_{n}w))(y)\xi_{\varepsilon}(x-y)dy$$

$$+ \sum_{i,j=1}^{N} \int_{\mathbb{R}^{N}} (q_{ij}(x) - q_{ij}(x-y))(D_{ij}(\vartheta_{n}w))(x-y)\xi_{\varepsilon}(y)dy$$

$$+ \sum_{i=1}^{N} \int_{\mathbb{R}^{N}} (c_{i}(x) - c_{i}(x-y))(D_{i}(\vartheta_{n}w))(x-y)\xi_{\varepsilon}(y)dy,$$

$$(8.2.28)$$

for any $x \in \mathbb{R}^N$, where $c_i = \sum_{j=1}^N D_j q_{ij}$ for any i = 1, ..., N. Since $\mathcal{A}_0 w \in BUC(\mathbb{R}^N)$, then $\mathcal{A}_0(\vartheta_n w) \in BUC(\mathbb{R}^N)$ as well. Therefore, the first term in the right-hand side of (8.2.28) tends to $\mathcal{A}(\vartheta_n w)$ uniformly in \mathbb{R}^N , as ε tends to

0. Since $q_{ij} \in \operatorname{Lip}(\mathbb{R}^N)$ for any $i,j=1,\ldots,N$, a similar argument shows that the second term in the right-hand side of (8.2.28) converges to zero uniformly in \mathbb{R}^N as ε tends to 0. Finally, the third term converges to 0 locally uniformly in \mathbb{R}^N . It follows that $\mathcal{A}_0 w_{\varepsilon,n}$ tends to $\mathcal{A}_0(\vartheta_n w)$ locally uniformly in \mathbb{R}^N . Moreover, there exists a positive constant \overline{C} , independent of ε and n, such that $\|\mathcal{A}_0 w_{\varepsilon,n}\|_{\infty} \leq \overline{C}$ for any $\varepsilon > 0$ and any $n \in \mathbb{N}$. We have so proved that there exists a sequence $\{w_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ converging to w in $W^{2,q}(\mathbb{R}^N)$ and such that $\mathcal{A}_0 w_n$ tends to $\mathcal{A}_0 w$ locally uniformly in \mathbb{R}^N and the sequence $\{\mathcal{A}_0 w_n\}$ is bounded in $C_b(\mathbb{R}^N)$. As a consequence, writing (8.2.26) with w_n instead of φ and letting n go to $+\infty$ easily yields (8.2.27).

Now we can prove the property (ii). We argue as in Step 2 of the proof of Theorem 8.2.5. Suppose that $\rho \in L^q(\mathbb{R}^N)$ for some q > 1. Then, using (8.2.12) and the Hölder inequality, we can show that

$$\left| \int_{\mathbb{R}^N} \mathcal{A}_0 \varphi d\mu \right| \le C \|\varphi\|_{1,r'},$$

where 1/r + 1/r' = 1 and

$$\frac{1}{r} = \frac{1}{p} + \left(1 - \frac{1}{p}\right) \frac{1}{q}.\tag{8.2.29}$$

According to Theorem C.1.3(i), ρ belongs to $W^{1,r}(\mathbb{R}^N)$ and since r < N (recall that $N \geq 2$) the Sobolev embedding theorems then imply that $\rho \in L^{\widetilde{q}}(\mathbb{R}^N)$ where

$$\frac{1}{\widetilde{q}} = \frac{1}{p} - \frac{1}{N} + \left(1 - \frac{1}{p}\right) \frac{1}{q}.$$

Therefore, using a bootstrap argument, we can now easily show that $\rho \in L^{q_n}(\mathbb{R}^N)$ for any $n \in \mathbb{N}$, where $s_n := 1/q_n \ (n \in \mathbb{N})$ is defined recursively by

$$\begin{cases} s_{n+1} = \frac{N-p}{pN} + \frac{p-1}{p}s_n, \\ s_1 = \alpha, \end{cases}$$

 α being any real number belonging to the interval ((N-1)/N, 1). Since the sequence $\{s_n\}$ converges to (N-p)/N, then q_n tends to N/(N-p) as n tends to $+\infty$. Using (8.2.29) the assertion follows.

Let us now prove the property (iii). The arguments that we are going to use can be also adapted to prove the properties (iv) and (v). According to Theorem 8.2.5(i) we know that $\rho \in L^{N/(N-p)}(\mathbb{R}^N)$. Therefore, by the Hölder inequality and our assumptions, it is immediate to check that the functions $b_i \rho$ belong to $L^{(N/(N-p+1)}(\mathbb{R}^N)$ for any $i = 1, \ldots, N$. Therefore, from (8.2.12) we deduce that

$$\left| \int_{\mathbb{R}^N} \mathcal{A}_0 \varphi d\mu \right| \le C \|\varphi\|_{W^{1,\frac{N}{p-1}}(\mathbb{R}^N)}, \tag{8.2.30}$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$. Theorem C.1.3(i) now implies that the function ρ belongs to $W^{1,N/(N-p+1)}(\mathbb{R}^N)$. Since $\rho^{-1}|D\rho|^2$ belongs to $L^1(\mathbb{R}^N)$ (see Step 1 in the proof of Theorem 8.2.5), using the Hölder inequality we deduce that $|D\rho| \in L^1(\mathbb{R}^N)$. Therefore, $\rho \in W^{1,1}(\mathbb{R}^N)$, and, hence, it belongs to $W^{1,q}(\mathbb{R}^N)$ for any $q \in [1, N/(N-p+1)]$.

Remark 8.2.11 For a sufficient condition ensuring that the integrability assumptions on the coefficients b_i (i = 1, ..., N) are satisfied, we refer the reader to the forthcoming Proposition 8.2.14.

We now deal with the regularity of the second-order derivatives of ρ .

Theorem 8.2.12 In addition to Hypotheses 8.2.9, let $b_i \in W^{1,\infty}_{loc}(\mathbb{R}^N) \cap L^p_{\mu}$ for any i = 1, ..., N and div $b \in L^p_{\mu}$, for some $p \geq 2$. Then, the following properties are met:

(i) if
$$p \in (N/2, N)$$
, then $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in \left(1, \frac{pN}{pN - p^2 + 3p - 2}\right]$;

(ii) if
$$p = N$$
, then $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in \left(1, \frac{N^2}{3N-2}\right)$;

(iii) if
$$p \in (N, 2N)$$
, then $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in \left(1, \frac{p^2}{3p-2}\right]$;

(iv) if
$$p = 2N$$
, then $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in (1, \frac{p}{2})$;

(v) if
$$p > 2N$$
, then $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in \left(1, \frac{p}{2}\right]$.

Proof. To prove the first three properties it suffices to show that, if p > N/2 and $p \geq 2$, and $D\rho \in L^r(\mathbb{R}^N)$ for some $r \in (1, +\infty)$, then $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in (1, s]$, where

$$\frac{1}{s} = \left(1 - \frac{2}{p}\right)\frac{1}{r} + \frac{2}{p}.\tag{8.2.31}$$

Indeed, once this property is checked, the conditions (i)-(iii) will immediately follow from Theorem 8.2.10.

Let us observe that, using (8.2.12) and integrating by parts, we can write

$$\int_{\mathbb{R}^N} (\mathcal{A}_0 \varphi) \rho dx = \int_{\mathbb{R}^N} \left(\rho \operatorname{div} b + \sum_{i=1}^N b_i D_i \rho \right) \varphi dx,$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$, where \mathcal{A}_0 is defined by (8.2.1). Since the function $f := \rho \operatorname{div} b + \sum_{i=1}^N b_i D_i \rho$ belongs to $L_{\operatorname{loc}}^m(\mathbb{R}^N)$ for any $m \in [1, +\infty)$ (see Theorem 8.2.3), according to Theorem C.1.3(iv), $\rho \in W_{\operatorname{loc}}^{2,m}(\mathbb{R}^N)$ for any $m \in \mathbb{R}^n$

 $[1, +\infty)$. Moreover, $\rho \operatorname{div} b \in L^m(\mathbb{R}^N)$ for any $m \in [1, p]$, since $\rho \in C_b(\mathbb{R}^N)$ (see Theorem 8.2.8). Now, using the Hölder inequality, we can write

$$\int_{\mathbb{R}^{N}} |b|^{\alpha} |D\rho|^{\alpha} dx = \int_{\mathbb{R}^{N}} |b|^{\alpha} |D\rho|^{\alpha - \frac{2}{\beta}} |D\rho|^{\frac{2}{\beta}} \rho^{-\frac{1}{\beta}} \rho^{\frac{1}{\beta}} dx$$

$$\leq \left(\int_{\mathbb{R}^{N}} \frac{|D\rho|^{2}}{\rho} dx \right)^{\frac{1}{\beta}} \left(\int_{\mathbb{R}^{N}} |b|^{\alpha\beta} d\mu \right)^{\frac{1}{\beta}}$$

$$\times \left(\int_{\mathbb{R}^{N}} |D\rho|^{\frac{\alpha\beta - 2}{\beta - 2}} dx \right)^{\frac{\beta - 2}{\beta}}, \tag{8.2.32}$$

for any $\alpha \geq 1$ and any $\beta \geq 2$ and, taking (8.2.18) into account, we easily see that $\alpha = s$ is the largest exponent such that the right-hand side of (8.2.32) is finite. Summing up, we have proved that $g := \sum_{i=1}^N b_i D_i \rho + \rho \text{div } b \in L^s(\mathbb{R}^N)$. Since $\mathcal{A}_0 \rho = g$, the Calderón-Zygmund estimates imply that $\rho \in W^{2,s}(\mathbb{R}^N)$. Hence, the statement follows with g = s.

To prove that $\rho \in W^{2,q}(\mathbb{R}^N)$ for any $q \in (1,s)$, we now observe that since $\rho \in W^{2,s}(\mathbb{R}^N)$, then $D\rho \in L^s(\mathbb{R}^N)$. Moreover, it belongs to $L^1(\mathbb{R}^N)$ by virtue of Theorem 8.2.10(iii) (recall that $p \geq 2$). Therefore, ρ belongs to $L^{r_*}(\mathbb{R}^N)$, where $r_* \in (1,s)$ is chosen so that $q^{-1} = (1-2/p)r_*^{-1} + 2/p$. Now, from the first part of the proof, we get $\rho \in W^{2,q}(\mathbb{R}^N)$.

To conclude the proof, let us check the properties (iv) and (v). To begin with, we show that $D\rho \in L^q(\mathbb{R}^N)$ for any $q < +\infty$ if p = 2N and that $D\rho \in L^\infty(\mathbb{R}^N)$ if p > 2N. Iterating the results in the first part of the proof, it is easy to check that $\rho \in W^{2,q_{n+1}}(\mathbb{R}^N)$ for any $n \in \mathbb{N}$ such that $q_n < N$. Here, $\tau_n := 1/q_n$ is defined recursively by

$$\tau_{n+1} = \left(1 - \frac{2}{p}\right)\tau_n + \frac{2N - p + 2}{pN}, \quad n \ge 2.$$

We claim that, if p > 2N, there exists $n_0 \in \mathbb{N}$ such that $q_{n_0} > N$. The Sobolev embedding theorems then will imply that $D\rho \in C_b(\mathbb{R}^N)$. To prove the claim, we observe that if $q_n < N$ for any $n \le m$ and some $m \in \mathbb{N}$, then

$$\tau_n = \left(1 - \frac{2}{p}\right)^n \left(\tau_1 - 1 + \frac{p-2}{2N}\right) + 1 + \frac{2-p}{2N},$$

for any $n \leq m+1$. Since the sequence $\{\tau_n\}$ converges to $1+(2-p)(2N)^{-1} < N^{-1}$ as n tends to $+\infty$, there should exist $n \in \mathbb{N}$ such that $q_n > N$. In the case when p = 2N, the sequence τ_n converges to N^{-1} . Therefore, $\rho \in W^{2,\alpha}(\mathbb{R}^N)$ for any $\alpha < N$, and, consequently, $D\rho \in L^r(\mathbb{R}^N)$ for any $r \in (1, +\infty)$.

Now we are almost done. Indeed, to prove the properties (iv) and (v) it suffices to prove that the function $g = \sum_{i=1}^{N} b_i D_i \rho + \rho \text{div } b$ belongs to $L^q(\mathbb{R}^N)$ for any $q \in (1, p/2)$, if p = 2N, and it belongs also to $L^{p/2}(\mathbb{R}^N)$, if p > 2N. Once these properties are checked, the Calderón-Zygmund estimates will

imply that $u \in W^{2,q}(\mathbb{R}^N)$. Of course, without loss of generality we can limit ourselves to dealing with the function $|bD\rho|$. Indeed, since $\rho \in C_b(\mathbb{R}^N)$, it is immediate to check that all the other terms in the definition of g belong to $L^q(\mathbb{R}^N)$ for any $q \in (1, p/2)$, if p = 2N, and for any $q \in (1, p/2]$, if p > 2N. From the Hölder inequality we get

$$\int_{\mathbb{R}^{N}} |b|^{q} |D\rho|^{q} dx = \int_{\mathbb{R}^{N}} |b|^{q} |D\rho|^{q - \frac{2q}{p}} |D\rho|^{\frac{2q}{p}} \rho^{-\frac{q}{p}} \rho^{\frac{q}{p}} dx
\leq \left(\int_{\mathbb{R}^{N}} |D\rho|^{\frac{q(p-2)}{p-2q}} dx \right)^{\frac{p-2q}{p}} \left(\int_{\mathbb{R}^{N}} \frac{|D\rho|^{2}}{\rho} dx \right)^{\frac{q}{p}}
\times \left(\int_{\mathbb{R}^{N}} |b|^{p} d\mu \right)^{\frac{q}{p}},$$
(8.2.33)

for any $q \in (1, p/2)$, if p = N/2, and

$$\int_{\mathbb{R}^{N}} |b|^{\frac{p}{2}} |D\rho|^{\frac{p}{2}} dx = \int_{\mathbb{R}^{N}} |b|^{\frac{p}{2}} |D\rho|^{\frac{p}{2}-1} |D\rho|^{\frac{p}{2}-1} |D\rho|^{\frac{1}{2}} \rho^{\frac{1}{2}} dx
\leq ||D\rho||_{\infty}^{\frac{p}{2}-1} \left(\int_{\mathbb{R}^{N}} \frac{|D\rho|^{2}}{\rho} dx \right)^{\frac{1}{2}} \left(\int_{\mathbb{R}^{N}} |b|^{p} d\mu \right)^{\frac{1}{2}}, \quad (8.2.34)$$

if p > 2N. Since the last sides of (8.2.33) and (8.2.34) are finite, the assertion follows.

To conclude this subsection we prove some pointwise bounds for the function ρ . For this purpose we need to assume some additional assumptions on the measure μ and on the drift coefficients b_i $(i=1,\ldots,N)$. For notational convenience, for any $\delta, \beta > 0$ we denote by $V_{\delta,\beta}$ the function defined by $V_{\delta,\beta}(x) = \exp(\delta|x|^{\beta})$ for any $x \in \mathbb{R}^N$.

Theorem 8.2.13 In addition to Hypotheses 8.2.9 assume that $V_{\delta,\beta} \in L^1_{\mu}$ for some $\beta, \delta > 0$. Then, the following properties are met:

(i) if $|b(x)| \leq C \exp(|x|^{\gamma})$ for some C > 0, $\gamma < \beta$ and any $x \in \mathbb{R}^N$, then there exist $c_1, c_2 > 0$ such that

$$\rho(x) \le c_1 \exp(-c_2|x|^{\beta}), \qquad x \in \mathbb{R}^N;$$

(ii) if in addition to the previous set of hypotheses, $b_i \in W^{1,\infty}_{loc}(\mathbb{R}^N)$ (i = 1, ..., N) and $|b(x)| + |\operatorname{div} b(x)| \leq C \exp(|x|^{\gamma})$ for some C > 0, some $\gamma < \beta$ and any $x \in \mathbb{R}^N$, then there exist $c_1, c_2 > 0$ such that

$$|D\rho(x)| \le c_1 \exp(-c_2|x|^{\beta}), \qquad x \in \mathbb{R}^N.$$

Proof. We begin the proof by checking the property (i). Let $w : \mathbb{R}^N \to \mathbb{R}$ be the function defined by $w(x) = \exp(c_2|x|^{\beta})$ for any $x \in \mathbb{R}^N$, where the positive

constant c_2 will be chosen in the sequel. Moreover, set $\varphi = w\psi$, where ψ is any function belonging to $C_c^{\infty}(\mathbb{R}^N)$. Since μ is the invariant measure of $\{T(t)\}$, then

$$\int_{\mathbb{R}^N} \mathcal{A}_0 \varphi d\mu = -\int_{\mathbb{R}^N} \sum_{i=1}^N b_i D_i \varphi d\mu, \qquad (8.2.35)$$

or, equivalently, by differentiation,

$$\int_{\mathbb{R}^{N}} w \rho \mathcal{A}_{0} \psi dx
= -\int_{\mathbb{R}^{N}} \left(\psi \mathcal{A}_{0} w + 2 \sum_{i,j=1}^{N} q_{ij} D_{i} \psi D_{j} w + w \sum_{i=1}^{N} b_{i} D_{i} \psi + \psi \sum_{i=1}^{N} b_{i} D_{i} w \right) d\mu.$$
(8.2.36)

Here \mathcal{A}_0 is defined by (8.2.1). Now, we fix p > N and we prove that the right-hand side of (8.2.36) can be estimated by $C\|\psi\|_{W^{1,p'}(\mathbb{R}^N)}$ for a suitable positive constant C, independent of ψ . Here 1/p + 1/p' = 1. Theorem C.1.3(i) then will imply that $\rho w \in W^{1,p}(\mathbb{R}^N)$ and the Sobolev embedding theorems will yield $\rho w \in C_b(\mathbb{R}^N)$. The assertion will follow.

So, let us fix q > p and choose $c_2 < \delta/q$. According to our assumptions, it is immediate to check that the functions w, Dw and A_0w belong to L^q_μ and $\rho \in C_b(\mathbb{R}^N)$ (see Theorem 8.2.5(iii)). Therefore, the first two terms in the right-hand side of (8.2.36) can be estimated as wished. Indeed, let us consider, for instance, the first one. We have

$$\left| \int_{\mathbb{R}^{N}} \psi \mathcal{A}_{0} w d\mu \right| = \left| \int_{\mathbb{R}^{N}} \psi (\mathcal{A}_{0} w) \rho dx \right|$$

$$\leq \|\rho\|_{\infty}^{1 - \frac{1}{p}} \left| \int_{\mathbb{R}^{N}} \psi (\mathcal{A}_{0} w) \rho^{\frac{1}{p}} dx \right|$$

$$\leq \|\rho\|_{\infty}^{1 - \frac{1}{p}} \left(\int_{\mathbb{R}^{N}} |\mathcal{A}_{0} w|^{p} d\mu \right)^{\frac{1}{p}} \left(\int_{\mathbb{R}^{N}} |\psi|^{p'} dx \right)^{\frac{1}{p'}}. \quad (8.2.37)$$

As far as the last two terms in the right-hand side of (8.2.36) are concerned, we observe that the functions wb_i and b_iD_iw belong to $L^p_\mu(\mathbb{R}^N)$ for any $i=1,\ldots,N$. Indeed, according to our assumptions, the functions b_i $(i=1,\ldots,N)$ belong to $L^p_\mu(\mathbb{R}^N)$ for any $r \in [1,+\infty)$. Hence, in particular, they belong to $L^{pq/(q-p)}_\mu(\mathbb{R}^N)$. The Hölder inequality now implies that wb_i and b_iD_iw belong to $L^p_\mu(\mathbb{R}^N)$ for any $i=1,\ldots,N$, and, repeating the same arguments as in the proof of (8.2.37), we are done.

To conclude the proof, let us check the property (ii). For this purpose, we begin by observing that, according to Theorem 8.2.12(v), $\rho \in W^{2,p}(\mathbb{R}^N)$ for any $p \in (1, +\infty)$. Therefore, integrating by parts the identity (8.2.35), we easily see that

$$\int_{\mathbb{R}^N} (\mathcal{A}_0 \rho) \varphi dx = \int_{\mathbb{R}^N} \{ \rho(\operatorname{div} b) + \langle b, D\rho \rangle \} \varphi dx,$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ or, equivalently, $A_0 \rho = \rho(\operatorname{div} b) + \langle b, D\rho \rangle$. It follows that

$$\mathcal{A}_0(\rho w) = w\rho \operatorname{div} b + w \sum_{i=1}^N b_i D_i \rho + \rho(A_0 w) + 2 \sum_{i,j=1}^N q_{ij} D_i w D_j \rho. \quad (8.2.38)$$

Now, we replace the constant c_2 with a smaller constant to make ρDw bounded in \mathbb{R}^N . Our purpose consists in showing that the right-hand side of (8.2.38) belongs to $L^p(\mathbb{R}^N)$ for some p > N. This will imply that ρw belongs to $W^{2,p}(\mathbb{R}^N)$ and the Sobolev embedding theorems then will yield $D(\rho w) \in C_b(\mathbb{R}^N)$ and, consequently, $wD\rho \in C_b(\mathbb{R}^N)$.

As in the proof of the property (i), one sees that $w\rho(\operatorname{div} b), \rho(A_0w) \in L^p(\mathbb{R}^N)$. As far as the terms containing $D\rho$ are concerned, we observe that

$$\int_{\mathbb{R}^{N}} \left| w \sum_{i=1}^{N} b_{i} D_{i} \rho \right|^{p} dx \leq \int_{\mathbb{R}^{N}} w^{p} |b|^{p} |D\rho|^{p-1} |D\rho|^{p-\frac{1}{2}} \rho^{\frac{1}{2}} dx \\
\leq \|D\rho\|_{\infty}^{p-1} \left(\int_{\mathbb{R}^{N}} \frac{|D\rho|^{2}}{\rho} dx \right)^{\frac{1}{2}} \left(\int_{\mathbb{R}^{N}} w^{2p} |b|^{2p} d\mu \right)^{\frac{1}{2}},$$

and the last side of the previous chain of inequalities is finite due to our assumptions and to (8.2.18). In a similar way one can estimate the term $|Dw||D\rho|$.

A situation in which $V_{\delta,\beta} \in L^1_{\mu}$, for suitable values of β and δ , is provided by the next proposition. We stress that it holds under Hypotheses 8.2.1.

Proposition 8.2.14 Assume that

$$\lim_{|x| \to +\infty} \sup \left\{ c\lambda_{\max}(x) + |x|^{-\beta} \left(\sum_{i=1}^{N} b_i(x)x_i + \sum_{i,j=1}^{N} D_j q_{ij}(x)x_i \right) \right\} < 0, \quad (8.2.39)$$

for some $c, \beta > 0$, where, for any $x \in \mathbb{R}^N$, $\lambda_{\max}(x)$ denotes the maximum eigenvalue of the matrix $(q_{ij}(x))$. Then, the operator A admits a (unique) invariant measure μ . Moreover, the function $V_{\delta,\beta}$ is integrable with respect to μ for any $\delta < \beta^{-1}c$.

Proof. As it is easily seen,

$$\mathcal{A}V_{\delta,\beta}(x) = \delta\beta |x|^{\beta-1} e^{\delta|x|^{\beta}}
\times \left(\frac{\text{Tr}(Q(x))}{|x|} + \frac{\beta-2}{|x|^3} \sum_{i,j=1}^N q_{ij} x_i x_j + \delta\beta |x|^{\beta-3} \sum_{i,j=1}^N q_{ij}(x) x_i x_j \right)
+ \frac{1}{|x|} \sum_{i=1}^N b_i x_i + \frac{1}{|x|} \sum_{i,j=1}^N D_j q_{ij}(x) x_i \right),$$

$$\leq \delta \beta |x|^{\beta - 2} e^{\delta |x|^{\beta}} \left((N + |\beta - 2| + \delta \beta |x|^{\beta}) \lambda_{\max}(x) + \sum_{i=1}^{N} b_i x_i + \sum_{i,j=1}^{N} D_j q_{ij} x_i \right), \tag{8.2.40}$$

for any $x \in \mathbb{R}^N$, and due to (8.2.39) and the choice of δ and β , the last side of (8.2.40) is negative for large |x|, and this implies that $\mathcal{A}V_{\delta,\beta}(x)$ tends to $-\infty$ as |x| tends to $+\infty$. Therefore, according to Khas'minskii theorem 8.1.20, the semigroup associated with the operator \mathcal{A} admits a (unique) invariant measure μ .

Let us now prove that the function $V_{\delta,\beta} \in L^1_{\mu}$. From (8.2.40) we easily deduce that, if $\beta \geq 1$, then $V_{\delta,\beta}(x) \leq |\mathcal{A}V_{\delta,\beta}(x)|$ for |x| sufficiently large, whereas, if $\beta \in (0,1)$, then, for any $\varepsilon > 0$, $|V_{(1-\varepsilon)\delta,\beta}(x)| \leq |\mathcal{A}V_{\delta,\beta}(x)|$ still for sufficiently large |x|. Therefore, in both the cases it suffices to show that $\mathcal{A}V_{\delta,\beta} \in L^1_{\mu}$ for any $\delta < \beta^{-1}c$ to deduce that $V_{\delta,\beta} \in L^1_{\mu}$.

To prove that $\mathcal{A}V_{\delta,\beta} \in L^1_{\mu}$ we observe that, by a simple approximation argument, we can show that

$$\int_{\mathbb{R}^N} \mathcal{A}\varphi d\mu = 0, \tag{8.2.41}$$

for any $\varphi \in C_b^2(\mathbb{R}^N)$ with compact support. Since $\mathcal{A}1 = 0$, we can extend (8.2.41) to any $\varphi \in C_b^2(\mathbb{R}^N)$ which is constant outside a ball. Now, let $\{\psi_n\} \in C_b^2([0,+\infty))$ be a sequence of increasing functions such that $\psi_n(t) = t$ for any $t \in [0,n], \psi(t) = n+1/2$ for any $t \geq n+1$, and $\psi_n''(t) \leq 0$ for any $t \geq 0$. Then, the function $V_n = \psi_n \circ V_{\delta,\beta}$ belongs to $C_b^2(\mathbb{R}^N)$ and it is constant outside a ball. Therefore,

$$\int_{\mathbb{R}^N} \mathcal{A}V_n d\mu = 0, \qquad n \in \mathbb{N}. \tag{8.2.42}$$

Fix now M sufficiently large such that $\mathcal{A}V_{\delta,\beta}(x) < 0$ for any $x \notin B(M)$ and, then, fix a large enough n so that $V_n = V_{\delta,\beta}$ in B(M). Taking (8.2.42) into account, we can write

$$\int_{\mathbb{R}^N \backslash B(M)} |\mathcal{A}V_n| d\mu = -\int_{\mathbb{R}^N \backslash B(M)} |\mathcal{A}V_n| d\mu = \int_{B(M)} \mathcal{A}V_n d\mu = \int_{B(M)} \mathcal{A}V_{\delta,\beta} d\mu.$$

Now, according to Fatou's lemma, we deduce that the function $\mathcal{A}V_{\delta,\beta}$ is integrable in $\mathbb{R}^N \setminus B(M)$ (and, hence, in \mathbb{R}^N) with respect to the measure μ .

Assuming some more regularity on the diffusion coefficients and some bounds on the growth at infinity of the drift term, one can show also lower bounds for ρ .

Theorem 8.2.15 ([114], Theorem 6.3) Assume that $q_{ij} \in C_b^3(\mathbb{R}^N)$ and that $b_i \in C^2(\mathbb{R}^N)$ (i, j = 1, ..., N) satisfy

$$|b_i(x)| + |Db_i(x)| + |D^2b_i(x)| \le C(1 + |x|^{\beta - 1}), \qquad x \in \mathbb{R}^N, \quad i = 1, \dots, N,$$

for some $\beta > 0$ and C > 0. Then

$$\rho(x) \ge \exp(-M(1+|x|^{\beta})), \qquad x \in \mathbb{R}^N,$$

where M > 0 is a positive constant depending only on C, κ_0 and $||q_{ij}||_{C_b^3(\mathbb{R}^N)}$ (i, j = 1, ..., N).

8.3 Some consequences of the estimates of Chapter 7

In this section we show some interesting consequences of the estimates proved in Chapter 7, when the semigroup $\{T(t)\}$ admits an invariant measure.

Remark 8.3.1 Note that if there exists a function φ such that

$$\lim_{|x| \to +\infty} \varphi(x) = +\infty, \qquad \lim_{|x| \to +\infty} \mathcal{A}\varphi(x) = -\infty,$$

then both the assumptions of Khas'minskii theorem and Hypothesis 6.1.1(ii) are satisfied.

The following proposition is a consequence of the results in Theorem 7.2.2 and Corollary 7.2.3.

Proposition 8.3.2 Suppose that Hypotheses 6.1.1(i)-6.1.1(iii), and Hypothesis 6.1.1(iv-k) or 7.1.3(ii-k) (k = 1, 2, 3) are satisfied, and let $p \in (1, +\infty)$. Then, for any t > 0, T(t) maps L^p_μ into $W^{k,p}_\mu$ and

$$||D^{k}T(t)f||_{p} \leq \left(k^{2} \prod_{j=1}^{k} \frac{\widetilde{\omega}_{k,p \wedge 2}}{1 - e^{-\widetilde{\omega}_{k,p \wedge 2} t}} \psi_{j,r}(t/k)\right)^{\max\{\frac{1}{p},\frac{1}{2}\}} ||f||_{p}, \qquad t > 0,$$
(8.3.1)

for any $f \in L^p_{\mu}$. Here, $\widetilde{\omega}_{k,p} = 0$ under Hypothesis 6.1.1(iv-k) and $\widetilde{\omega}_{k,p} < 0$ is given by Theorem 7.1.5 if Hypothesis 7.1.3(ii-k) is satisfied. Moreover, $\psi_{k,p}$ is given in Theorem 7.2.2. In particular, the L^p_{μ} -norm of $D^kT(t)f$ behaves as $t^{-1/2}$ as t tends to 0, whereas, at infinity, it stays bounded, if Hypothesis 6.1.1(iv-k) is satisfied, and it decreases exponentially if Hypothesis 7.1.3(ii-k) is satisfied.

Proof. The proof is a straightforward consequence of Theorem 7.2.2 and Corollary 7.2.3. Indeed, let first $f \in C_b(\mathbb{R}^N)$; integrating (7.2.5) (if k=1) or (7.2.27) (if k=2,3) in \mathbb{R}^N , we deduce that T(t)f belongs to $W_{\mu}^{l,p}$ and it satisfies (8.3.1) for any t>0. In the general case when $f \in L_{\mu}^p$, consider a sequence $\{f_n\} \subset C_b(\mathbb{R}^N)$ converging to f in L_{μ}^p . Writing (8.3.1) for $f_n - f_m$ it follows that $\{T(t)f_n\}$ is a Cauchy sequence in $W_{\mu}^{l,p}$. Therefore, T(t)f belongs to $W_{\mu}^{k,p}$ and satisfies (8.3.1).

Proposition 8.3.3 Let Hypotheses 6.1.1(i)–6.1.1(iii) and 6.1.1(iv-1) be satisfied and let p > 1. Then, $D(L_p) \subset W^{1,p}_{\mu}$ and, for any $\omega > 0$, there exists a positive constant M_{ω} such that

$$||Df||_p \le M_{\omega,p} ||f||_p^{\frac{1}{2}} ||(L_p - \omega)f||_p^{\frac{1}{2}}, \qquad f \in D(L_p).$$
 (8.3.2)

Under Hypotheses 6.1.1(i)-6.1.1(iii) and 7.1.3(i), the estimate (8.3.2) holds true with $\omega = 0$.

Proof. The proof is similar to that of Theorem 6.2.2. Indeed, fix $f \in D(L_p)$, $\lambda > 0$ and let ω be as in the statement of the proposition. Moreover, let us set $u = (\lambda + \omega)f - L_p f$. Then, we have

$$f = \int_0^{+\infty} e^{-(\lambda + \omega)t} T(t) u \, dt. \tag{8.3.3}$$

Now, from Proposition 8.3.2 we deduce that in correspondence of ω we can find out a positive constant $C = C_{\omega}$ such that

$$||DT(t)f||_p \le \frac{Ce^{\omega t}}{\sqrt{t}}||f||_p, \qquad t > 0.$$
 (8.3.4)

Using (8.3.3) and (8.3.4) it is immediate to check that $Df \in L^p_\mu$ and

$$||Df||_{p} \leq \int_{0}^{+\infty} e^{-(\lambda+\omega)t} ||DT(t)u||_{p} dt$$

$$\leq C_{\omega,p} \left(\int_{0}^{+\infty} t^{-\frac{1}{2}} e^{-\lambda t} dt \right) ||\lambda f - (L_{p} - \omega)f||_{p}$$

$$\leq \sqrt{\pi} C_{\omega,p} \left(\sqrt{\lambda} ||f||_{p} + \frac{1}{\sqrt{\lambda}} ||(L_{p} - \omega)f||_{p} \right). \tag{8.3.5}$$

Minimizing with respect to $\lambda \in (0, +\infty)$ the last side of (8.3.5), we get the assertion with $M_{\omega,p} = 2\sqrt{2}C_{\omega,p}$.

In the remainder of this section, we consider the particular case when

$$Au(x) = \Delta u(x) + \langle b(x), Du(x) \rangle, \qquad x \in \mathbb{R}^N,$$

under the following assumptions on the function b.

Hypotheses 8.3.4 (i) the drift coefficient b belongs to $C^{1+\delta}_{loc}(\mathbb{R}^N, \mathbb{R}^N)$ for some $\delta > 0$. Moreover, there exists $d_0 \in \mathbb{R}$ such that

$$\langle b(x) - b(y), x - y \rangle \le d_0 |x - y|^2, \qquad x, \xi \in \mathbb{R}^N;$$
 (8.3.6)

- (ii) Hypothesis 6.1.1(ii) is satisfied and the semigroup associated with the operator \mathcal{A} admits an invariant measure.
- **Remark 8.3.5** (i) If the drift term b satisfies Hypotheses 8.3.4, then it satisfies the condition (7.1.9). To check this latter condition, it suffices to write (8.3.6) with $y=x+t\xi$, dividing, then, both the members by t^2 and letting t go to 0. Therefore, the assumptions of Theorem 7.3.1 are satisfied. Therefore, the estimates (7.3.4) and (7.3.5) apply. In particular, these latter two estimates imply that for any $f \in L^p_\mu(\mathbb{R}^N)$ and any t > 0, it holds that

$$||DT(t)f||_p \le \frac{2p^{1/p-1/2}d_0^{1/p}}{(p-1)^{1/2}(1-e^{-pd_0t})^{1/p}}t^{\frac{1}{p}-\frac{1}{2}}||f||_p,$$

for any $p \in (1, 2]$ and

$$||DT(t)f||_p \le \left(\frac{d_0}{1 - e^{-2d_0t}}\right)^{\frac{1}{2}} ||f||_p,$$

for any $p \in (2, +\infty)$.

Finally, Proposition 8.3.3 holds true.

- (ii) Hypothesis 8.3.4(ii) is satisfied, for instance, under the assumptions of Remark 8.3.1.
- (iii) In the particular case when $d_0 < 0$, Hypothesis 8.3.4(ii) is a consequence of (8.3.4)(i) and the previous point (ii). Indeed, it suffices to take $\varphi(x) = |x|^2$ for any $x \in \mathbb{R}^N$.

Further, in this situation we can also prove the following formulas.

Proposition 8.3.6 Assume that Hypotheses 8.3.4 hold and let p, q > 1 be such that $p^{-1} + q^{-1} = 1$. For any $f \in D(L_p)$ and any $g \in D(L_q)$, the function fg belongs to $D(L_1)$ and

$$L_1(fg) = fL_q g + gL_p f + 2\langle Df, Dg \rangle. \tag{8.3.7}$$

In particular,

$$\int_{\mathbb{R}^N} g L_p f d\mu = -\int_{\mathbb{R}^N} f L_q g d\mu - 2 \int_{\mathbb{R}^N} \langle Df, Dg \rangle d\mu, \tag{8.3.8}$$

and, for any $f \in D(L)$, we have

$$\int_{\mathbb{R}^N} f L f \, d\mu = -\int_{\mathbb{R}^N} |Df|^2 d\mu. \tag{8.3.9}$$

Proof. Assume first $f, g \in D(\widehat{A})$. Since $D(\widehat{A}) = D_{\max}(A)$ (see Propositions 2.3.6 and 4.1.10), then $fg \in D(\widehat{A})$ and

$$A(fg) = f\widehat{A}g + g\widehat{A}f + 2\langle Df, Dg\rangle. \tag{8.3.10}$$

Now, let $f \in D(L_p)$ and $g \in D(L_q)$. Since $D(\widehat{A})$ is a core of L_p and L_q (see Proposition 8.1.9), then there exist two sequences $\{f_n\}, \{g_n\} \subset D(\widehat{A})$ such that $f_n, L_p f_n$ converge, respectively, to f and $L_p f$ in L_p^p , whereas $g_n, L_q g_n$ converge, respectively, to g and $L_q g$ in L_μ^q , as n tends to $+\infty$. Then, $f_n g_n$ converges to fg in L_μ^1 ; besides by (8.3.2), for any f in f in f converges to f in f and f in f

Formula (8.3.8) now easily follows integrating (8.3.7) in \mathbb{R}^N and taking (8.1.5) into account. Formula (8.3.9) then follows from (8.3.8) taking f = g.

In the symmetric case we can prove the following improvement of the formula (8.3.8), which generalizes the formula 8.1.39.

Proposition 8.3.7 Assume that Hypotheses 8.1.25 hold with $U \in C^{2+\alpha}_{loc}(\mathbb{R}^N)$ and $G \in C^{1+\alpha}_{loc}(\mathbb{R}^N)$. Further, assume that the function U therein defined satisfies

$$\langle DU(x) + G(x) - DU(y) - G(y), y - x \rangle \le d_0 |x - y|^2, \qquad x, y \in \mathbb{R}^N,$$

for some constant $d_0 \in \mathbb{R}$. Finally, let p, q > 1 be such that $p^{-1} + q^{-1} = 1$. Then, for any $f \in D(L_p)$ and any $g \in W_u^{1,q}$ we have

$$\int_{\mathbb{R}^N} g L_p f d\mu = -\int_{\mathbb{R}^N} \langle Df, Dg \rangle d\mu. \tag{8.3.11}$$

Proof. If $f, g \in D(\widehat{A})$, then (8.3.11) follows from (8.1.39). Indeed, it suffices to observe that since $D(\widehat{A}) \subset D(L_2)$, then

$$\int_{\mathbb{R}^N} g L_p f d\mu = \int_{\mathbb{R}^N} g L_2 f d\mu = -\int_{\mathbb{R}^N} \langle Df, Dg \rangle d\mu.$$

Consider now the general case when $f \in D(L_p)$ and $g \in W_{\mu}^{1,q}$. Let $\{f_n\} \subset D(\widehat{A})$ be a sequence such that f_n and $\widehat{A}f_n$ converge in L_{μ}^p , respectively, to f and $L_p f$, and let $\{g_n\} \subset D(\widehat{A})$ converge to g in $W_{\mu}^{1,q}$. Then, by the estimate (8.3.2), f_n converges to f in $W_{\mu}^{1,q}$. Thus (8.3.11) follows letting n tend to $+\infty$.

0

8.4 The convex case

In this section we consider the particular case when the operator $\mathcal A$ is given by

$$\mathcal{A}u(x) = \Delta u(x) - \langle DU(x), Du(x) \rangle, \qquad x \in \mathbb{R}^N, \tag{8.4.1}$$

under the following hypothesis on the function U.

Hypothesis 8.4.1 U belongs to $C^2(\mathbb{R}^N)$, $e^{-U} \in L^1(\mathbb{R}^N)$ and

$$\sum_{i,j=1}^{N} D_{ij} U(x) \xi_i \xi_j \ge 0, \qquad x, \xi \in \mathbb{R}^N.$$
 (8.4.2)

Note that since Hypothesis 8.4.1 is stronger than Hypotheses 8.1.25, then all the results of Section 8.1.4 hold and, in particular, the measure

$$\mu(dx) = K^{-1}e^{-U(x)}dx, \qquad K = \int_{\mathbb{R}^N} e^{-U(x)}dx,$$
 (8.4.3)

is the invariant measure of $\{T(t)\}$.

The results of this section have been proved in [40], in a more general context.

Theorem 8.4.2 Assume Hypothesis 8.4.1. Then

$$D(L) = \{ u \in W_u^{2,2} : \langle DU, Du \rangle \in L_u^2 \}$$
 (8.4.4)

and

$$||D^2u||_2 \le ||Lu||_2, \qquad u \in D(L),$$
 (8.4.5)

$$||u||_{W^{2,2}_{\mu}} \le \frac{3}{2} (||u||_2 + ||Lu||_2), \qquad u \in D(L).$$
 (8.4.6)

Proof. We consider first the case when $u \in C_c^{\infty}(\mathbb{R}^N)$. Fix $i \in \{1, ..., N\}$; by (8.1.39) we have

$$\int_{\mathbb{R}^N} |D(D_i u)|^2 d\mu = -\int_{\mathbb{R}^N} L(D_i u) D_i u d\mu.$$

Writing $L(D_i u) = D_i L u + \langle D(D_i U), Du \rangle$ and summing as i runs from 1 to N we obtain

$$\int_{\mathbb{R}^N} \sum_{i,j=1}^N |D_{ij}u|^2 d\mu = -\int_{\mathbb{R}^N} \left\{ \sum_{i=1}^N (D_i L u) D_i u + \sum_{i,j=1}^N D_{ij} U D_i u D_j u \right\} d\mu.$$

Now, by (8.1.39) and (8.4.2) it follows that

$$\int_{\mathbb{R}^{N}} \sum_{i,j=1}^{N} |D_{ij}u|^{2} d\mu \le -\int_{\mathbb{R}^{N}} \sum_{i=1}^{N} (D_{i}Lu) D_{i}u d\mu = \int_{\mathbb{R}^{N}} |Lu|^{2} d\mu,$$

that is (8.4.5).

The estimate (8.4.6) with $u \in C_c^{\infty}(\mathbb{R}^N)$ now follows from (8.1.39) and (8.4.5).

Let us now prove the set equality (8.4.4). For this purpose, we fix $u \in D(L)$. By Theorem 8.1.26, $C_c^{\infty}(\mathbb{R}^N)$ is a core of L; therefore, there exists a sequence $\{u_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ such that u_n and Lu_n converge in L_{μ}^2 , respectively, to u and Lu. By (8.4.6) applied to $u_n - u_m$ it follows that $u \in W_{\mu}^{2,2}$ and that u_n converges to u in $W_{\mu}^{2,2}$. As a straightforward consequence, u satisfies (8.4.5) and (8.4.6). Moreover, since $\langle DU, Du_n \rangle = Lu_n - \Delta u_n$, letting n go to $+\infty$ we get $\langle DU, Du \rangle \in L_{\mu}^2$. This proves the inclusion " \subset " in (8.4.4).

Conversely, let $u \in W_{\mu}^{2,2}$ be such that $\langle DU, Du \rangle \in L_{\mu}^{2}$, and let $f \in W_{\mu}^{1,2}$; we prove that

$$\int_{\mathbb{R}^N} \langle Du, Df \rangle d\mu = -\int_{\mathbb{R}^N} (\mathcal{A}u) f d\mu, \qquad (8.4.7)$$

from which it follows that $u \in D(L)$. Indeed, if \widetilde{L} denotes the realization of the operator \mathcal{A} with domain given by the right-hand side of (8.4.4), the formula (8.4.7) implies that the operator $I - \widetilde{L}$ is injective. Since $(\widetilde{L}, D(\widetilde{L}))$ is an extension of (L, D(L)), then $I - \widetilde{L}$ is also surjective. Therefore, $1 \in \rho(\widetilde{L}) \cap \rho(L)$ from which we get

$$L^2_{\mu}=(I-\widetilde{L})(D(\widetilde{L}))\supset (I-\widetilde{L})(D(L))=(I-L)(D(L))=L^2_{\mu}$$

and, consequently, $D(L) = D(\widetilde{L})$.

So, let us prove the formula (8.4.7). For this purpose, we consider a sequence $\{f_n\} \in C_c^{\infty}(\mathbb{R}^N)$ converging to f in $W_{\mu}^{1,2}$ (see the last part of the proof of Proposition 8.1.27). An integration by parts shows that

$$\int_{\mathbb{R}^N} \langle Du, Df_n \rangle d\mu = -\int_{\mathbb{R}^N} f_n (\Delta u - \langle DU, Du \rangle) d\mu, \qquad n \in \mathbb{N}$$

Letting n tend to $+\infty$, we get (8.4.7).

As a consequence of the results in Theorem 8.4.2, we get the following corollary.

Corollary 8.4.3 Assume Hypothesis 8.4.1. Then, for any $\lambda > 0$

$$||D^2R(\lambda, L)f||_2 \le 2||f||_2,$$
 (8.4.8)

for any $f \in L^2_{\mu}$.

Proof. By (8.4.4), the function $R(\lambda, L)f$ belongs to $W_{\mu}^{2,2}$ for any $f \in L_{\mu}^{2}$. Setting $u = R(\lambda, L)f$, by (8.4.5) we deduce that

$$\int_{\mathbb{R}^N} |D^2 u|^2 d\mu \le ||Lu||_2^2 = ||\lambda u - f||_2^2 \le 4||f||_2^2,$$

that is (8.4.8).

Remark 8.4.4 In [40] the authors prove some more refined results. Indeed, they consider the operator \mathcal{A} in (8.4.1) assuming that U is a convex function which goes to $+\infty$ as |x| tends to $+\infty$. No regularity assumptions on U are made. When U is not differentiable at $x \in \mathbb{R}^N$, DU(x) is meant as the element with minimal norm in the subdifferential $\partial U(x)$ of U at x, where $\partial U(x) = \{y \in \mathbb{R}^N : U(\xi) \geq U(x) + \langle y, \xi - x \rangle, \ \forall \xi \in \mathbb{R}^N \}$. Under these rather weak assumptions on U, they prove that the realization A in L^2_μ of the operator \mathcal{A} with domain $D(A) = \{u \in W^{2,2}_\mu : \langle DU, Du \rangle \in L^2_\mu \}$ is a dissipative self-adjoint operator. Therefore, it generates a strongly continuous analytic semigroup in L^2_μ . Moreover, they prove that all the results in Theorem 8.4.2 and Corollary 8.4.3 hold also in this situation.

To prove their results, the authors use an approximation argument. In the case when U is not smooth, they approximate U with a sequence of smooth functions by means of the Moreau-Yosida approximations of U (say $\{U_{\alpha} : \alpha > 0\}$), which are defined as follows:

$$U_{\alpha}(x) = \inf_{y \in \mathbb{R}^N} \left(U(y) + \frac{1}{2\alpha} |x - y|^2 \right), \qquad x \in \mathbb{R}^N, \ \alpha > 0.$$

Each function U_{α} is convex differentiable and

$$U_{\alpha}(x) \le U(x), \qquad |DU_{\alpha}(x)| \le |DU(x)|,$$

$$\lim_{\alpha \to 0^+} U_{\alpha}(x) = U(x), \qquad \lim_{\alpha \to 0^+} DU_{\alpha}(x) = DU(x),$$

for any $x \in \mathbb{R}^N$. Moreover, any U_{α} is Lipschitz continuous in \mathbb{R}^N and its Lipschitz constant is $1/\alpha$.

To the approximated operators $\mathcal{A}_{\alpha} := \Delta u - \langle DU_{\alpha}, Du \rangle$ ($\alpha > 0$), the results in Theorem 8.4.2 and Corollary 8.4.3 apply with the same constants appearing in (8.4.4)-(8.4.6) and (8.4.8). Taking advantage of this fact, the authors show that, if $f \in C_c^{\infty}(\mathbb{R}^N)$, then (up to a subsequence) $R(\lambda, A_{\alpha})f$ converges weakly in $W_{\mu}^{2,2}(\mathbb{R}^N)$ to a function $u \in \{u \in W_{\mu}^{2,2} : \langle DU, Du \rangle \in L_{\mu}^2 \}$, which turns out to be the (unique) solution of the equation $\lambda u - \mathcal{A}u = f$ in such a space, and it satisfies the estimates in Theorem 8.4.2 and Corollary 8.4.3. Here A_{α} is the realization of the operator \mathcal{A}_{α} in $L_{\mu_{\alpha}}^2$, with domain given by (8.4.4) with μ_{α} instead of μ , where μ_{α} is defined accordingly to (8.4.3) with U being replaced by U_{α} .

Finally, using a density argument, they conclude that the operator A, with domain D(A), generates a strongly continuous analytic semigroup in L^2_{μ} and it satisfies the estimates in Theorem 8.4.2 and Corollary 8.4.3.

8.5 Compactness of T(t) and of the embedding $W^{1,p}_{\mu}\subset L^p_{\mu}$

In this section we prove some compactness result in the symmetric case, i.e., in the case when the operator \mathcal{A} is given by (8.4.1) under the following hypothesis on U.

Hypothesis 8.5.1 The function U belongs to $C^2(\mathbb{R}^N)$ and the function e^{-U} is integrable in \mathbb{R}^N .

We recall that in this case the invariant measure of the associated semigroup is given by $d\mu = K^{-1}e^{-U(x)}dx$ (see Theorem 8.1.26). The results of this section are due to G. Metafune.

Lemma 8.5.2 Assume that the function U satisfies Hypothesis 8.5.1 as well as the following condition

$$\Delta U(x) \le \delta_1 |DU(x)|^2 + M_1, \qquad x \in \mathbb{R}^N, \tag{8.5.1}$$

for some $\delta_1 \in (0,1)$ and $M_1 > 0$. Then, for any $p \in [2,+\infty)$ there exists a positive constant C such that

$$\int_{\mathbb{R}^N} |f|^p |DU|^2 d\mu \le C \|f\|_{W^{1,p}_{\mu}}^p, \qquad f \in W^{1,p}_{\mu}. \tag{8.5.2}$$

If, moreover,

$$\langle D^2 U(x) D U(x), D U(x) \rangle \ge (\delta_2 |D U(x)|^2 + M_2) |D U(x)|^2, \qquad x \in \mathbb{R}^N,$$
(8.5.3)

for some $\delta_2 > \delta_1 - 1$ and some $M_2 \in \mathbb{R}$, then, for any $p \in [1, 2)$, there exists a positive constant C such that

$$\int_{\mathbb{R}^N} |f|^p |DU|^p d\mu \le C \|f\|_{W^{1,p}_{\mu}}^p, \qquad f \in W^{1,p}_{\mu}.$$
 (8.5.4)

Proof. First we prove the assertion in the case when f has compact support. Fix $g \in W^{1,p}_{\mu}$ with compact support and $\alpha \in (1,p]$. Since $L^q_{\mu} \subset L^r_{\mu}$ for any

q > r, and g is compactly supported, the function $|g|^{\alpha}(1+|DU|^2)$ belongs to L^1_{μ} . Now, integrating by parts we get

$$\int_{\mathbb{R}^N} |g|^{\alpha} (1 + |DU|^2) d\mu = \int_{\mathbb{R}^N} |g|^{\alpha} d\mu - K^{-1} \int_{\mathbb{R}^N} |g|^{\alpha} \langle DU, De^{-U} \rangle dx$$

$$= \int_{\mathbb{R}^N} |g|^{\alpha} (1 + \Delta U) d\mu + \alpha \int_{\mathbb{R}^N} |g|^{\alpha - 2} g \langle Dg, DU \rangle d\mu.$$
(8.5.5)

Assume now that $p \in [1, 2)$. Letting α tend to 1 in (8.5.5), we get

$$\int_{\mathbb{R}^{N}} |g|(1+|DU|^{2})d\mu = \int_{\mathbb{R}^{N}} \{|g|(1+\Delta U) + \operatorname{sign}(g)\langle Dg, DU \rangle\} d\mu, \quad (8.5.6)$$

where we have set sign(0) = 0. Choosing $g = |f|^p (1 + |DU|^2)^{(p-2)/2}$ and observing that

$$Dg = p|f|^{p-2}fDf(1+|DU|^2)^{\frac{p-2}{2}} + (p-2)|f|^p(1+|DU|^2)^{\frac{p-4}{2}}D^2UDU,$$

from (8.5.6) we obtain that

$$\int_{\mathbb{R}^{N}} |f|^{p} (1 + |DU|^{2})^{\frac{p}{2}} d\mu = \int_{\mathbb{R}^{N}} |f|^{p} (1 + |DU|^{2})^{\frac{p-2}{2}} (1 + \Delta U) d\mu
+ p \int_{\mathbb{R}^{N}} |f|^{p-2} f (1 + |DU|^{2})^{\frac{p-2}{2}} \langle Df, DU \rangle d\mu
+ (p-2) \int_{\mathbb{R}^{N}} |f|^{p} (1 + |DU|^{2})^{\frac{p-4}{2}} \langle D^{2}UDU, DU \rangle d\mu
:= I_{1} + I_{2} + I_{3}.$$

Using (8.5.1), the Hölder inequality and (8.5.3), we get

$$I_{1} \leq \delta_{1} \int_{\mathbb{R}^{N}} |f|^{p} (1 + |DU|^{2})^{\frac{p}{2}} d\mu + (M_{1} + 1 - \delta_{1}) \int_{\mathbb{R}^{N}} |f|^{p} (1 + |DU|^{2})^{\frac{p-2}{2}} d\mu$$

$$\leq \delta_{1} \int_{\mathbb{R}^{N}} |f|^{p} (1 + |DU|^{2})^{\frac{p}{2}} d\mu + (M_{1} + 1 - \delta_{1}) \int_{\mathbb{R}^{N}} |f|^{p} d\mu,$$

$$I_{2} \leq p \left(\int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2})^{\frac{p}{2}} d\mu \right)^{\frac{p-1}{p}} \left(\int_{\mathbb{R}^{N}} |Df|^{p} d\mu \right)^{\frac{1}{p}}$$
$$\leq \varepsilon \int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2})^{\frac{p}{2}} d\mu + C_{\varepsilon,p} \int_{\mathbb{R}^{N}} |Df|^{p} d\mu,$$

for any $\varepsilon > 0$ and some positive constant $C_{\varepsilon,p}$. Finally,

$$\begin{split} I_{3} &\leq (p-2) \int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2})^{\frac{p-4}{2}} (\delta_{2}^{-}|DU|^{2} + M_{2}^{-}) |DU|^{2} d\mu \\ &\leq (p-2) \int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2})^{\frac{p-2}{2}} (\delta_{2}^{-}|DU|^{2} + M_{2}^{-}) d\mu \\ &\leq (p-2) \delta_{2}^{-} \int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2})^{\frac{p}{2}} d\mu + |(p-2)(M_{2}^{-} - \delta_{2}^{-})| \int_{\mathbb{R}^{N}} |f|^{p} d\mu. \end{split}$$

Since $\delta_1 - \delta_2^- < 1$, choosing $\varepsilon > 0$ small enough, (8.5.4) follows.

In the case when $p \geq 2$, the estimate (8.5.2) follows for any compactly supported function $f \in W^{1,p}_{\mu}$ by (8.5.5) with $\alpha = p$ and g = f, using (8.5.1) to obtain the inequalities

$$\int_{\mathbb{R}^N} |f|^p (1 + \Delta U) d\mu \le \delta_1 \int_{\mathbb{R}^N} |f|^p (1 + |DU|^2) d\mu + (M_1 + 1 - \delta_1) \int_{\mathbb{R}^N} |f|^p d\mu,$$

$$\int_{\mathbb{R}^{N}} |f|^{p-1} |Df| (1+|DU|^{2})^{\frac{1}{2}} d\mu$$

$$\leq \left(\int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2})^{\frac{p}{2p-2}} d\mu \right)^{\frac{p-1}{p}} \left(\int_{\mathbb{R}^{N}} |Df|^{p} d\mu \right)^{\frac{1}{p}}$$

$$\leq \left(\int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2}) d\mu \right)^{\frac{p-1}{p}} \left(\int_{\mathbb{R}^{N}} |Df|^{p} d\mu \right)^{\frac{1}{p}}$$

$$\leq \varepsilon \int_{\mathbb{R}^{N}} |f|^{p} (1+|DU|^{2}) d\mu + C_{\varepsilon} \int_{\mathbb{R}^{N}} |Df|^{p} d\mu, \qquad \varepsilon > 0,$$

and then choosing $\varepsilon > 0$ small enough.

The general case when $f \in W^{1,p}_{\mu}$ follows by approximation: consider a sequence of cut-off functions $\{\vartheta_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ such that $\vartheta_n \equiv 1$ in B(n) and $\vartheta_n \equiv 0$ outside B(n+1); the function $\eta_n f$ satisfies (8.5.2), if $p \in [2, +\infty)$, or (8.5.4) if $p \in [1, 2)$. Then, the conclusion follows from the monotone convergence theorem.

Theorem 8.5.3 Suppose that the assumptions in Lemma 8.5.2 hold and assume that

$$\lim_{|x| \to +\infty} |DU(x)| = +\infty. \tag{8.5.7}$$

Then, the embedding $W^{1,p}_{\mu} \subset L^p_{\mu}$ is compact for any $p \in [1, +\infty)$. In particular T(t) is compact in L^2_{μ} for any t > 0.

Proof. We have to prove that the set

$$E = \left\{ f \in W_{\mu}^{1,p} : \|f\|_{W_{\mu}^{1,p}} \le 1 \right\}$$

is totally bounded in L^p_u . For this purpose, fix r > 0 and let

$$E_r = \bigg\{ f_{\big|B(r)}: \ f \in E \bigg\}, \qquad r > 0.$$

Since U is continuous, the norms of $L^p(B(r), dx)$ and $L^p(B(r), \mu)$ are equivalent on the σ -algebra of the Borel sets of \mathbb{R}^N (in the sense that there exists a positive constant C such that $C^{-1}\mu(A) \leq m(A) \leq C\mu(A)$ for any Borel set A, where by m we denote the Lebegue measure). Therefore, E_r is bounded in $W^{1,p}(B(r), dx)$ and then, by Rellich-Kondrachov theorem (see [2, Theorem 6.2]), it is totally bounded in $L^p(B(r), dx)$ and in $L^p(B(r), \mu)$ as well. Thus, for any $\varepsilon > 0$ there exists a finite number of functions g_1, \ldots, g_n in $L^p(B(r), \mu)$ such that

$$E_r \subset \bigcup_{i=1}^n \{f: \|f - g_i\|_{L^p(B(r),\mu)} < \varepsilon\}.$$

Moreover, by (8.5.2) and (8.5.4) it follows that

$$\int_{\mathbb{R}^N \backslash B(r)} |f|^p d\mu \leq \frac{1}{\inf\limits_{\mathbb{R}^N \backslash B(r)} |DU|^{p \wedge 2}} \int_{\mathbb{R}^N} |f|^p |DU|^{p \wedge 2} d\mu \leq \frac{C}{\inf\limits_{\mathbb{R}^N \backslash B(r)} |DU|^{p \wedge 2}},$$

for any $f \in E$. Choosing r large enough, from (8.5.7) we get

$$\left(\int_{\mathbb{R}^N \setminus B(r)} |f|^p d\mu\right)^{\frac{1}{p}} \le \varepsilon, \qquad f \in E.$$

Therefore, extending the functions g_j (j = 1, ..., n) by zero to the whole \mathbb{R}^N we obtain

$$E \subset \bigcup_{i=1}^{n} \{ f : \|f - g_i\|_p < 2\varepsilon \},$$

that is, E is totally bounded in L_{μ}^{p} .

The compactness of T(t) in L^2_{μ} , for any t>0, now follows from Proposition 8.1.29.

The following corollary is an immediate consequence of Proposition 8.3.2 and Theorem 8.5.3.

Corollary 8.5.4 Assume that the operator A in (8.4.1) satisfies both Hypotheses 8.3.4 and the assumptions of Theorem 8.5.3 for some p > 1. Then, T(t) is compact in L^p_μ for any t > 0.

However, in general the embedding $W^{1,p}_{\mu} \subset L^p_{\mu}$ is not compact. Here, we see a counterexample.

Example 8.5.5 Let N=1 and let $U\in C^2(\mathbb{R})$ be any function such that U(x)=x for any x>0 and U(x)=0 for any $x\leq -1$. Moreover, let $\mu(dx)=e^{-U(x)}dx$. Consider the sequence of functions

$$f_n(x) = \chi_{(0,+\infty)} x^n, \qquad x \in \mathbb{R}, \quad n \in \mathbb{N}.$$

Then, we have

$$||f_n||_1 = \int_0^{+\infty} x^n e^{-x} dx = n!$$

and

$$||Df_n||_1 = n \int_0^{+\infty} x^{n-1} e^{-x} dx = n!.$$

Thus the sequence of functions $g_n = (n!)^{-1} f_n$ $(n \in \mathbb{N})$ is bounded in $W^{1,1}_{\mu}$, but it is not relatively compact in L^1_{μ} because $\lim_{n \to +\infty} g_n(x) = 0$ for any $x \in \mathbb{R}$, whereas $\|g_n\|_{L^1_{\mu}} = 1$ for any $n \in \mathbb{N}$.

In this example Hypotheses 8.1.25 and the conditions (8.5.1) and (8.5.3) are satisfied, so that the estimates (8.5.2) and (8.5.4) hold, but the condition (8.5.7) fails.

To conclude this section, we observe that, taking Lemma 8.5.2 into account, we can improve the results in Theorem 8.4.2.

Proposition 8.5.6 Let the function U satisfy Hypothesis 8.4.1 and the condition (8.5.1). Then, $D(L) = W_{\mu}^{2,2}$ and the graph-norm of D(L) is equivalent to the $W_{\mu}^{2,2}$ -norm.

8.6 The Poincaré inequality and the spectral gap

In this section we study the Poincaré inequality in L^p_μ . We say that the Poincaré inequality holds if there exists C>0 such that

$$\int_{\mathbb{R}^N} |f - \overline{f}|^p d\mu \le C \int_{\mathbb{R}^N} |Df|^p d\mu, \qquad f \in W^{1,p}_{\mu}, \tag{8.6.1}$$

where $\overline{f} = \int_{\mathbb{R}^N} f d\mu$. The following proposition gives a first sufficient condition for the Poincaré inequality to hold.

Proposition 8.6.1 Let $p \in [1, +\infty)$. If the embedding $W^{1,p}_{\mu} \subset L^p_{\mu}$ is compact, then the Poincaré inequality (8.6.1) holds.

Proof. Fix $p \in [1, +\infty)$ and suppose, by contradiction, that (8.6.1) does not hold. Then, there exists a sequence $\{f_n\} \subset W^{1,p}_{\mu}$ such that

$$\overline{f_n} = 0, ||f_n||_p = 1, ||Df_n||_p \le \frac{1}{n}, (8.6.2)$$

for any $n \in \mathbb{N}$. Since $W^{1,p}_{\mu}$ is compactly embedded in L^p_{μ} , then there exists a subsequence $\{f_{n_k}\}$ converging in L^p_{μ} to a function f^* , which satisfies

$$\overline{f^*} = 0, \qquad ||f^*||_p = 1.$$
 (8.6.3)

Now fix r > 0; according to Proposition 8.1.5, $L^p(B(r), \mu) \subset L^p(B(r)) := L^p(B(r), dx)$. Therefore,

$$\lim_{k \to +\infty} ||f_{n_k} - f^*||_{L^1(B(r))} = 0 \tag{8.6.4}$$

and, moreover, from (8.6.2) it follows that

$$||Df_{n_k}||_{L^1(B(r))} \le \frac{C}{n_k}, \qquad k \in \mathbb{N},$$
 (8.6.5)

for some C > 0. Now, from the Poincaré inequality in $W^{1,1}(B(r))$ (see e.g., [54, Theorem 4.5.2]) we have

$$||f_{n_k} - (f_{n_k})_r||_{L^1(B(r))} \le C' ||Df_{n_k}||_{L^1(B(r))},$$
 (8.6.6)

where

$$(f_{n_k})_r = \frac{1}{\omega_N r^N} \int_{B(r)} f_{n_k} dx,$$

and ω_N is the Lebesgue measure of the ball $B(1) \subset \mathbb{R}^N$. Letting k tend to $+\infty$ in (8.6.6), taking (8.6.4) and (8.6.5) into account, it follows that

$$||f^* - (f^*)_r||_{L^1(B(r))} = 0.$$

Therefore, f^* is almost everywhere constant in B(r). Since r > 0 is arbitrary, f^* is almost everywhere constant in \mathbb{R}^N . But this is in contradiction with (8.6.3).

As an immediate consequence of the previous proposition and the results of Section 8.5, we get the following result.

Theorem 8.6.2 Under the assumptions of Theorem 8.5.3 the Poincaré inequality (8.6.1) holds for any $p \in [1, +\infty)$.

Now, we provide another proof of the Poincaré inequality, based on the pointwise gradient estimate (7.3.4).

220

Theorem 8.6.3 Assume that Hypotheses 8.3.4 hold for some $d_0 < 0$. Then, the Poincaré inequality (8.6.1) holds for p = 2 and with $C = -1/d_0$.

Proof. Observe that (8.3.2) implies that D(L) is dense in $W^{1,2}_{\mu}$. Indeed, it contains $C_c^{\infty}(\mathbb{R}^N)$ which is dense in $W^{1,p}_{\mu}$; see the last part of the proof of Proposition 8.1.27. Hence, it is sufficient to consider the case when $f \in D(L)$. Then, by (8.3.9) we have

$$\frac{d}{dt}\int_{\mathbb{R}^N}|T(t)f|^2d\mu=2\int_{\mathbb{R}^N}T(t)f\,LT(t)f\,d\mu=-2\int_{\mathbb{R}^N}|DT(t)f|^2d\mu,$$

for any t > 0. Using the pointwise estimate (7.3.4), with p = 2, we obtain

$$\frac{d}{dt} \int_{\mathbb{R}^N} |T(t)f|^2 d\mu \geq -2e^{2d_0t} \int_{\mathbb{R}^N} T(t) \left(|Df|^2 \right) d\mu = -2e^{2d_0t} \int_{\mathbb{R}^N} |Df|^2 d\mu.$$

Now, integrating with respect to t in $(0, +\infty)$ and using Theorem 8.1.16, we get

$$\overline{f}^2 - \int_{\mathbb{R}^N} f^2 d\mu \geq \frac{1}{d_0} \int_{\mathbb{R}^N} |Df|^2 d\mu,$$

that is (8.6.1) for p = 2 and with $C = -1/d_0$.

We now show that the Poincaré inequality implies that T(t)f converges exponentially to \overline{f} in L^2_{μ} , as t tends to $+\infty$. This improves Theorem 8.1.16 and should be compared with Theorem 8.1.24. In the proof of Proposition 8.6.4 we will need that $D(L) \subset W^{1,2}_{\mu}$ and that

$$\int_{\mathbb{R}^{N}} fLf \, d\mu = -\int_{\mathbb{R}^{N}} |Df|^{2} d\mu, \qquad f \in D(L).$$
 (8.6.7)

Both the two previous properties hold. Actually we have already proved them twice: first in the symmetric case, that is under Hypotheses 8.1.25, see Proposition 8.1.27. Secondly we have proved them in the nonsymmetric case in Propositions 8.3.3 and 8.3.6, under Hypotheses 8.3.4.

Proposition 8.6.4 Assume Hypotheses 8.1.25 or Hypotheses 8.3.4. Moreover, assume that the Poincaré inequality (8.6.1) holds with p = 2. Then,

$$||T(t)f - \overline{f}||_2 \le e^{-\frac{t}{C}}||f - \overline{f}||_2, \qquad t > 0,$$
 (8.6.8)

for any $f \in L^2_{\mu}$, where C > 0 is the same constant as in (8.6.1). Further, we have a spectral gap for L, namely

$$\sigma(L)\setminus\{0\}\subset\{\lambda\in\mathbb{C}:\ \mathrm{Re}\ \lambda\leq-1/C\}\,.$$

Proof. First we prove (8.6.8). By density, it is sufficient to consider the case when $f \in D(L)$. Moreover, we can assume that $\overline{f} = 0$, since the general case then follows by considering the function $f - \overline{f}$. Note that $\overline{f} = 0$ implies that $\overline{T}(t)f = 0$ for any t > 0.

Let us observe that the function $t \mapsto ||T(t)f||_2^2$ is differentiable in $[0, +\infty)$, and, by the relation (8.6.7), we have

$$\frac{d}{dt} \|T(t)f\|_2^2 = 2 \int_{\mathbb{R}^N} T(t) f \, LT(t) f \, d\mu = -2 \int_{\mathbb{R}^N} |DT(t)f|^2 d\mu.$$

Now, the Poincaré inequality (8.6.1) yields

$$\frac{d}{dt}\|T(t)f\|_2^2 \leq -\frac{2}{C}\int_{\mathbb{R}^N}|T(t)f-\overline{T(t)f}|^2d\mu = -\frac{2}{C}\int_{\mathbb{R}^N}|T(t)f|^2d\mu.$$

Then, the estimate (8.6.8) follows from the Gronwall Lemma.

To prove the spectral gap property, it suffices to argue as in the proof of Theorem 8.1.24, observing that the estimate (8.6.8) implies that

$$||T(t)||_{L(H)} \le e^{-\frac{t}{C}}, \qquad t > 0,$$
 (8.6.9)

where, as in the proof of the quoted theorem, we have set $H = \{ f \in L^2_{\mu} : \overline{f} = 0 \}$. In particular, (8.6.9) implies that

$$\sigma(L_{\big|H})\subset\left\{\lambda\in\mathbb{C}:\operatorname{Re}\lambda\leq-\frac{1}{2}\right\}.$$

See the beginning of Section B.1.

8.7 The logarithmic Sobolev inequality and hypercontractivity

In this section we deal with the logarithmic Sobolev inequality (in short LSI). Fix $p \in [1, +\infty)$; the logarithmic Sobolev inequality reads as follows:

$$\int_{\mathbb{R}^{N}} f^{p} \log f d\mu \le \|f\|_{p}^{p} \log \|f\|_{p} + \frac{p}{\lambda} \int_{\mathbb{R}^{N}} f^{p-2} |Df|^{2} d\mu, \tag{8.7.1}$$

where $\lambda > 0$ and $f \in L^p_\mu$ is a regular positive function. In this subsection we see some conditions under which the LSI holds and its main consequences.

Roughly speaking the LSI is, for invariant measures, the counterpart of the Sobolev embedding theorems which hold when the underlining measure is the Lebesgue one. Indeed, it is well known that, if both f and its gradient

222

belong to some L^p -space related to the Lebesgue measure, then $|f|^q$ belongs to $L^1(\mathbb{R}^N)$ for some q>p. In general, this is no more the case for a general measure (hence, in particular, for invariant measure). We can see it by a simple counterexample.

Example 8.7.1 Let μ be the Gaussian measure, i.e., $\mu(dx) = \pi^{-N/2}e^{-|x|^2}$. Moreover, for any p > 1 and any $\varepsilon > 0$, let $u : \mathbb{R}^N \to \mathbb{R}$ be defined by

$$u(x) = \exp\left(\frac{2}{2p+\varepsilon}|x|^2\right), \qquad x \in \mathbb{R}^N.$$

Then, u belongs to L^p_μ , is continuously differentiable in \mathbb{R}^N and

$$D^{\alpha}u(x) = P_{|\alpha|}(x)u(x), \qquad x \in \mathbb{R}^N,$$

where $P_{|\alpha|}$ is a real polynomial with degree $|\alpha|$. Since

$$\int_{\mathbb{R}^N} |D^{\alpha}u(x)|^p d\mu = \frac{1}{\pi^{N/2}} \int_{\mathbb{R}^N} |P_{|\alpha|}(x)|^p \exp\bigg(-\frac{\varepsilon}{2p+\varepsilon}|x|^2\bigg) dx < +\infty,$$

then $u \in W^{k,p}_{\mu}$ for any $k \in \mathbb{N}$.

However, u does not belong to $L_u^{p+\varepsilon}$. Indeed,

$$\int_{\mathbb{R}^N} |u(x)|^{p+\varepsilon} d\mu = \frac{1}{\pi^{N/2}} \int_{\mathbb{R}^N} \exp\bigg(\frac{\varepsilon}{2p+\varepsilon} |x|^2\bigg) dx = +\infty.$$

In the case when $p \geq 2$, what one can only expect is that $f^p \log(f) \in L^1_\mu$, if f is positive. Indeed, by the Hölder inequality we have

$$\int_{\mathbb{R}^N} |f|^{p-2} |Df|^2 d\mu \le \left(\int_{\mathbb{R}^N} |f|^p d\mu \right)^{\frac{p-2}{p}} \left(\int_{\mathbb{R}^N} |Df|^p d\mu \right)^{\frac{2}{p}}.$$

This result is very sharp, as Example 8.7.6 shows.

Theorem 8.7.2 Assume that Hypotheses 8.3.4 hold with $d_0 < 0$. Then, the LSI (8.7.1) holds, with $\lambda = 2d_0$, for any $p \in [1, +\infty)$ and any nonnegative $f \in D(\widehat{A})$.

Proof. We split the proof into two steps. First in Step 1 we prove the assertion for the functions $f \in D(\widehat{A})$ (see Section 2.3) such that $f \geq \delta$ for some positive constant δ . Then, in Step 2, we extend the result to all the nonnegative functions $f \in D(\widehat{A})$,

Step 1. Let $f \in D(\widehat{A})$ be strictly greater than δ for some $\delta > 0$. We first prove the LSI with p = 1. By Proposition 8.3.3, $T(t)f \in W^{1,2}_{\mu}$ for any t > 0.

Moreover, by Remark 2.2.3 and Theorem 2.2.5, $T(t)f \geq T(t)\delta \equiv \delta$ for any $t \geq 0$. Therefore, we can define the function

$$\psi(t) = e^{-2d_0t} \int_{\mathbb{R}^N} \frac{1}{T(t)f} |DT(t)f|^2 d\mu, \qquad t \ge 0.$$

The pointwise estimate (7.3.4) with p = 2 implies that

$$\psi(t) \le \int_{\mathbb{R}^N} \frac{1}{T(t)f} |T(t)(|Df|)|^2 d\mu, \qquad t > 0.$$

Moreover, applying the Hölder inequality to the representation formula (2.2.8) we get

$$|T(t)(|Df|)|^2 = \left(T(t)\left(\sqrt{f}\frac{1}{\sqrt{f}}|Df|\right)\right)^2 \leq (T(t)f)T(t)\left(\frac{1}{f}|Df|^2\right);$$

therefore,

$$\psi(t) \le \int_{\mathbb{R}^N} T(t) \left(\frac{1}{f} |Df|^2\right) d\mu = \int_{\mathbb{R}^N} \frac{1}{f} |Df|^2 d\mu = \psi(0). \tag{8.7.2}$$

Consider now the function $F:[0,+\infty)\to\mathbb{R}$ defined by

$$F(t) = \int_{\mathbb{R}^N} T(t) f \log(T(t)f) d\mu, \qquad t \ge 0.$$

By Theorem 8.1.16, T(t)f tends to \overline{f} in L^2_{μ} as t tends to $+\infty$; besides, since $T(t)f \geq \delta > 0$ for any t > 0, and since the function $x \mapsto \log x$ is Lipschitz continuous in $[\delta, +\infty)$, it follows that $\log T(t)f$ tends to $\log \overline{f}$ in L^2_{μ} as t tends to $+\infty$. This implies that

$$\lim_{t \to +\infty} F(t) = \overline{f} \log \overline{f}. \tag{8.7.3}$$

Moreover, $F \in C^1([0, +\infty))$ and a straightforward computation yields

$$F'(t) = \int_{\mathbb{R}^N} (\mathcal{A}T(t)f) \log T(t) f \, d\mu, \qquad t \ge 0, \tag{8.7.4}$$

since, by (8.3.8), $\int_{\mathbb{R}^N} \mathcal{A}T(t)fd\mu = \int_{\mathbb{R}^N} LT(t)fd\mu = 0$ for any t > 0. Now, since

$$\mathcal{A}(\log u) = \frac{1}{u}\mathcal{A}u - \frac{1}{u^2}|Du|^2, \qquad u \in D(\widehat{A}),$$

for any positive function $u \in C^2(\mathbb{R}^N)$, it follows that $\log T(t) f \in D_{\max}(\mathcal{A})$ for any t > 0 (see (2.0.1)). Since $\{T(t)\}$ is conservative (see Proposition 8.1.10),

224

from Propositions 2.3.6, 4.1.10 and 8.1.9 we deduce that $\log T(t) f \in D(L)$ for any t > 0. Now, (8.3.8) and (8.7.4) yield

$$F'(t) = -\int_{\mathbb{R}^N} T(t) f L(\log T(t) f) d\mu - \int_{\mathbb{R}^N} 2\langle DT(t) f, D \log T(t) f \rangle d\mu$$
$$= -\int_{\mathbb{R}^N} \frac{1}{T(t) f} |DT(t) f|^2 d\mu = -e^{2d_0 t} \psi(t), \qquad t \ge 0.$$

Integrating with respect to t in $(0, +\infty)$ and using (8.7.2) and (8.7.3), we get

$$\overline{f}\log \overline{f} - F(0) = -\int_0^{+\infty} e^{2d_0 t} \psi(t) dt \ge -\frac{1}{2d_0} \psi(0),$$

that is the LSI (8.7.1) with p = 1.

Finally, we consider the case when $p \in (1, +\infty)$. We have

$$\mathcal{A}(f^p) = pf^{p-1}\mathcal{A}f + p(p-1)f^{p-2}|Df|^2,$$

for any $f \in D(\widehat{A})$. According to Theorem 6.2.2, $D(\widehat{A}) = D_{\max}(\mathcal{A}) \subset C_b^1(\mathbb{R}^N)$, and thus $f^p \in D_{\max}(\mathcal{A})$. Moreover $f^p \geq \delta^p > 0$, and thus, by the first part of the proof, the function f^p satisfies (8.7.1) with p = 1. This yields the conclusion.

Step 2. We now consider the general case, first extending the validity of the LSI to all the nonnegative functions $f \in D(\widehat{A})$. For this purpose, we fix a nonnegative $f \in D(\widehat{A})$ and, for any $\delta > 0$, we define the function $f_{\delta} = f + \delta$. Since $\mathcal{A}f_{\delta} = \mathcal{A}f$ and $D_{\max}(\mathcal{A}) = D(\widehat{A})$, it is immediate to check that $f_{\delta} \in D(\widehat{A})$. Therefore, we can apply (8.7.1), with f_{δ} instead of f, getting

$$\int_{\mathbb{D}^{N}} f_{\delta}^{p} \log f_{\delta} d\mu \le \|f_{\delta}\|_{p}^{p} \log \|f_{\delta}\|_{p} + \frac{p}{2d_{0}} \int_{\mathbb{D}^{N}} f_{\delta}^{p-2} |Df_{\delta}|^{2} d\mu.$$
 (8.7.5)

We are going to show that we can let δ go to 0^+ in the previous inequality, obtaining (8.7.1). As a first step we apply the Fatou lemma. From (8.7.5), we deduce that

$$\int_{\mathbb{R}^N} f^p \log f d\mu \le \liminf_{\delta \to 0^+} \left(\|f_\delta\|_p^p \log \|f_\delta\|_p + \frac{p}{2d_0} \int_{\mathbb{R}^N} f_\delta^{p-2} |Df|^2 d\mu \right). \tag{8.7.6}$$

Now, since $f_{\delta} \leq f_1 \in C_b(\mathbb{R}^N) \subset L^p_{\mu}$ for any $\delta \in (0,1)$, from the dominated convergence theorem we immediately deduce that f_{δ} tends to f in L^p_{μ} as δ tends to 0. Therefore, $\|f_{\delta}\|_p^p \log \|f_{\delta}\|_p$ tends to $\|f\|_p^p \log \|f\|_p$ as δ vanishes. So, we just need to consider the last term in the right-hand side of (8.7.6). We observe that, if $p \leq 2$, then $f_{\delta}^{p-2} \leq f^{p-2}$ and, therefore, we are done. On the other hand, in the case when $p \geq 2$, by the Hölder inequality, it follows that the function $f_{\delta}^{p-2}|Df|^2$ is in L^1_{μ} and, as above, $f_{\delta}^{p-2}|Df|^2 \leq f_1^{p-1}|Df|^2$. Hence, the dominated convergence theorem implies that the last integral in (8.7.5)

converges to the corresponding one, where we replace f_{δ} with f. Therefore, the estimate (8.7.1) follows also in this case.

In order to extend the LSI (8.7.1) to all the positive functions $f \in W_{\mu}^{1,p}$, we prove the following proposition.

Proposition 8.7.3 Let μ be the invariant measure of $\{T(t)\}$ and denote by ρ its density with respect to the Lebesgue measure. Suppose that ρ is locally bounded in \mathbb{R}^N . Then, $C_c^{\infty}(\mathbb{R}^N)$ is dense in $W_u^{1,p}$.

Proof. The proof is similar to the classical proof of the density of $C_c^{\infty}(\mathbb{R}^N)$ in the Sobolev space $W^{1,p}(\mathbb{R}^N)$ associated with the Lebesgue measure. For the reader's convenience, we go into details. Let $f \in W_{\mu}^{1,p}(\mathbb{R}^N)$. We approximate f by a sequence of compactly supported functions $f_n \in W_{\mu}^{1,p}$, by setting $f_n = f\xi_n$, where $\xi_n(x) = \xi(|x|/n)$, for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, $\xi \in C^{\infty}([0, +\infty))$ being any smooth function such that $\xi = 1$ in [0, 1/2] and $\xi = 0$ in $[1, +\infty)$. Each function f_n belongs to $W_{\mu}^{1,p}$ and the dominated convergence theorem implies that f_n converges to f in $W_{\mu}^{1,p}$. Therefore, we can limit ourselves to proving the assertion in the case when f is compactly supported. Let $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ be a smooth function, compactly supported in B(1) such that $\varphi \equiv 1$ in B(1/2), $0 \le \varphi \le 1$ in \mathbb{R}^N and $\|\varphi\|_{L^1(\mathbb{R}^N)} = 1$. We set $\varphi_n(x) = n^N \varphi(x/n)$ for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, and, then, for any f compactly supported in \mathbb{R}^N , we set

$$f_n(x) = (f \star \varphi_n)(x) = \int_{\mathbb{R}^N} \varphi_n(x - y) f(y) dy, \qquad x \in \mathbb{R}^N, \ n \in \mathbb{N}.$$

It is immediate to check that all the functions f_n belong to $C_c^{\infty}(\mathbb{R}^N)$ and Supp $f_n \subset B(1) + \text{supp } f$. Therefore,

$$\int_{\mathbb{R}^N} |f_n - f|^p d\mu = \int_{B(K)} |f_n - f|^p d\mu$$

$$\leq \sup_{x \in B(K)} \rho(x) \int_{B(K)} |f_n - f|^p dx$$

$$\leq \sup_{x \in B(K)} \rho(x) \int_{\mathbb{R}^N} |f_n - f|^p dx,$$

where K is such that $B(1) + \operatorname{supp}(f) \subset B(K)$. Since, it is well known that f_n tends to f if $L^p(\mathbb{R}^N)$, from the previous chain of inequalities we deduce that f_n tends to f in L^p_μ as well. The same argument can be applied to check the convergence of Df_n to Df. Indeed, $Df_n = Df \star \varphi_n$ for any $n \in \mathbb{N}$.

Remark 8.7.4 The assumptions of Proposition 8.7.3 are satisfied, for instance, under Hypothesis 8.2.1(i).

Theorem 8.7.5 Under the assumptions of Proposition 8.7.3, the logarithmic Sobolev inequality (8.7.1) holds for any nonnegative function $f \in W^{1,p}_{\mu}$ and any $p \geq 2$.

Proof. Fix a nonnegative function $f \in W^{1,p}_{\mu}$ and let $\{f_n\} \in C^{\infty}_{c}(\mathbb{R}^{N})$ be a sequence of smooth functions converging to f in $W^{1,p}_{\mu}$ as n tends to $+\infty$. As the proof of Proposition 8.7.3 shows, we can assume that f_n is nonnegative for any $n \in \mathbb{N}$. Since $C^{\infty}_{c}(\mathbb{R}^{N}) \subset D(\widehat{A})$ (see Proposition 2.3.6), we can write the estimate (8.7.1) with f replaced with f_n $(n \in \mathbb{N})$ obtaining

$$\int_{\mathbb{R}^N} f_n^p \log f_n d\mu \le \|f_n\|_p^p \log \|f_n\|_p + \frac{p}{2d_0} \int_{\mathbb{R}^N} f_n^{p-2} |Df_n|^2 d\mu.$$

Without loss of generality, we can also assume that f_n converges to f pointwise μ -almost everywhere in \mathbb{R}^N . Therefore, the Fatou lemma implies that

$$\int_{\mathbb{R}^{N}} f^{p} \log f d\mu \leq \|f\|_{p}^{p} \log \|f\|_{p} + \frac{p}{2d_{0}} \liminf_{n \to +\infty} \int_{\mathbb{R}^{N}} f_{n}^{p-2} |Df_{n}|^{2} d\mu.$$

So, to conclude the proof, we just need to show that

$$\liminf_{n \to +\infty} \int_{\mathbb{R}^N} f_n^{p-2} |Df_n|^2 d\mu = \int_{\mathbb{R}^N} f^{p-2} |Df|^2 d\mu.$$
(8.7.7)

For this purpose, we observe that

$$\left| \int_{\mathbb{R}^N} f_n^{p-2} |Df_n|^2 d\mu - \int_{\mathbb{R}^N} f^{p-2} |Df|^2 d\mu \right|$$

$$\leq \int_{\mathbb{R}^N} f_n^{p-2} ||Df_n|^2 - |Df|^2 |d\mu + \int_{\mathbb{R}^N} |f_n^{p-2} - f^{p-2}||Df|^2 d\mu$$

$$:= I_1 + I_2.$$

As far as I_1 is concerned, we observe that the Hölder inequality implies that

$$I_1 \le \|f_n\|_p^{p-2} \left(\int_{\mathbb{R}^N} ||Df_n|^2 - |Df|^2|^{\frac{p}{2}} \right)^{\frac{2}{p}}$$

$$\le \|f_n\|_p^{p-2} (\|Df_n\|_p + \|Df\|_p) \|Df_n - Df\|_p,$$

so that it vanishes as n tends to $+\infty$. As far as I_2 is concerned, we observe that if $p \in [2,3]$, then $|f_n^{p-2} - f^{p-2}| \le |f_n - f|^{p-2}$ for any $n \in \mathbb{N}$. Therefore, the Hölder inequality yields

$$I_2 \le \int_{\mathbb{R}^N} |f_n - f|^{p-2} |Df|^2 d\mu \le ||f_n - f||_p^{p-2} ||Df||_p^2.$$
 (8.7.8)

On the other hand, if p > 3, then

$$|f_n^{p-2} - f_n^{p-2}| \le (p-2)|f_n^{p-3} + f^{p-3}||f_n - f|$$

$$\le (p-2)(2^{4-p} \lor 1)|f_n + f|^{p-3}|f_n - f|,$$

for any $n \in \mathbb{N}$. Therefore,

$$I_{2} \leq (2^{4-p} \vee 1)(p-2) \int_{\mathbb{R}^{N}} |f_{n} + f|^{p-3} |f_{n} - f| |Df|^{2} d\mu$$

$$\leq (2^{4-p} \vee 1)(p-2) \left(\int_{\mathbb{R}^{N}} |f_{n} - f|^{\frac{p}{p-2}} |f_{n} + f|^{\frac{p(p-3)}{p-2}} d\mu \right)^{\frac{p-2}{p}} \|Df\|_{p}^{2}$$

$$\leq (2^{4-p} \vee 1)(p-2) \|f_{n} - f\|_{p} (\|f_{n}\|_{p} + \|f\|_{p})^{\frac{p-3}{p}} \|Df\|_{p}^{2}. \tag{8.7.9}$$

From (8.7.8) and (8.7.9) we deduce that, both in the cases when $p \in [2,3]$ and p > 3, I_2 tends to 0 as n tends to $+\infty$. Summing up, (8.7.7) follows and this completes the proof.

Example 8.7.6 Let $\mu(dx) = (2\pi)^{-1/2} \exp(-x^2/2) dx$ be the one-dimensional Gaussian measure. Further, let $f : \mathbb{R} \to \mathbb{R}$ be the function defined by

$$f(x) = \frac{e^{\frac{x^2}{4}}}{\{(x^2 + 2)\log(x^2 + 2)\}^{\frac{3}{4}}} := e^{\frac{x^2}{4}}h(x),$$

for any $x \geq M$, where M is such that $f(x) \geq 1$ for any $x \in (-\infty, M) \cup (M, +\infty)$. As it is immediately seen, f belongs to L^2_{μ} , since at $\pm \infty$ h^2 is asymptotic to the function $x \mapsto |x|^{-3/2} (\log(|x|)^{-3/4})$ which is integrable in a neighborhood of $\pm \infty$. Moreover, f is continuously differentiable in \mathbb{R} and

$$f'(x) = \frac{x}{2} \frac{(x^2 - 1)\log(x^2 + 2) - 3}{((x^2 + 2)\log(x^2 + 2))^{\frac{7}{4}}} e^{\frac{x^2}{4}},$$

for any $x \in (-\infty, -M) \cup (M, +\infty)$. Therefore, $x \mapsto (f'(x))^2 \exp(-x^2/2)$ is asymptotic to the function $x \mapsto |x|^{-1} (\log(|x|)^{-3/2})$ at $\pm \infty$, and this latter function is integrable in a neighborhood of $\pm \infty$.

On the other hand, the function

$$x \mapsto (f(x))^2 \log(f(x)) \log(\log(f(x))) \exp(-x^2/2)$$

is asymptotic to the function $x \mapsto (2|x|^2 \log(|x|))^{-1/2}$ at $\pm \infty$, and this latter function is integrable neither in a neighborhood of $+\infty$ nor in a neighborhood of $-\infty$.

This example, due to L. Gross (see [69, p. 1074]), shows that the LSI (8.7.1) is sharp.

Remark 8.7.7 As a consequence of Proposition 8.7.3, Remark 8.7.4 and Theorem 8.7.5, we immediately deduce that, if $Au = \Delta u - \langle DU, Du \rangle$, with U satisfying Hypotheses 8.1.25 as well as the dissipative type condition

$$\langle DU(x) - DU(y), y - x \rangle \le d_0 |x - y|^2, \qquad x, y \in \mathbb{R}^N, \tag{8.7.10}$$

228

for some $d_0 \in \mathbb{R}$, then the LSI may be written for any nonnegative $f \in W^{1,p}_{\mu}(\mathbb{R}^N)$ $(p \geq 2)$. Since $D(L_p) \subset W^{1,p}_{\mu}$ (see Proposition 8.3.3), if $p \geq 2$, we can write the LSI for any $f \in D(L_p)$.

The following proposition shows that actually the LSI can be written for any nonnegative $f \in D(L_p)$ also in the case when $p \in (1, 2)$. For this purpose, we show that integrating by parts we can write the second integral term in the right-hand side of (8.7.1) in a more suitable form.

Proposition 8.7.8 Assume Hypotheses 8.1.25 with the function U wherein defined satisfying (8.7.10). Then, for any $p \in (1, +\infty)$ and any positive function $f \in D(L_p)$, we have

$$\int_{\mathbb{R}^N} f^{p-2} |Df|^2 d\mu = -\frac{1}{(p-1)} \int_{\mathbb{R}^N} f^{p-1} L_p f \, d\mu. \tag{8.7.11}$$

In particular the LSI (8.7.1) holds true for any nonnegative $f \in D(L_p)$ $(p \in (1, +\infty))$.

Proof. Let us prove (8.7.11). First, we suppose that $p \geq 2$ and fix $f \in D(L_p)$. Then, by Proposition 8.3.3, $f \in W_{\mu}^{1,p}$ and, therefore, $f^{p-1} \in W_{\mu}^{1,p/(p-1)}$; besides $D(f^{p-1}) = (p-1)f^{p-2}Df$. Thus, (8.7.11) follows from the formula (8.3.11).

Now let $p \in (1,2)$ and let us fix $f \in D(L_p)$. For any $\varepsilon > 0$ we consider the function $g_{\varepsilon} = (f^2 + \varepsilon)^{(p-1)/2}$. As it is easily seen, g_{ε} belongs to $W_{\mu}^{1,p/(p-1)}$. Applying the previous argument with g_{ε} instead of f, we get

$$\int_{\mathbb{R}^{N}} f(f^{2} + \varepsilon)^{\frac{p-3}{2}} |Df|^{2} d\mu = -\frac{1}{p-1} \int_{\mathbb{R}^{N}} (f^{2} + \varepsilon)^{\frac{p-1}{2}} L_{p} f d\mu.$$

Letting ε go to 0^+ , by monotone and dominated convergence, we get (8.7.11). To conclude the proof, let us show that the LSI (8.7.1) can be extended to all the positive functions $f \in D(L_p)$ for any $p \in (1, +\infty)$. Thanks to Remark 8.7.7, we can limit ourselves to considering the case when $p \in (1, 2)$. For this purpose, we fix $f \in D(L_p)$ for some $p \in (1, 2)$. Since $D(\widehat{A})$ is a core of L_p (see Proposition 8.1.9), we can determine a sequence $\{f_n\} \subset D(\widehat{A})$ converging to f, in L^p_μ , as f tends to f and such that f tends to f to f as well. Arguing as in the proof of Theorem 8.7.5, we can easily show that

$$\lim_{n \to +\infty} \int_{\mathbb{R}^N} f_n^{p-1} L_p f_n d\mu = \int_{\mathbb{R}^N} f^{p-1} L_p f d\mu.$$

Now, using twice (8.7.11) gives

$$\lim_{n \to +\infty} \int_{\mathbb{R}^N} f_n^{p-2} |Df_n|^2 d\mu = -\frac{1}{p-1} \lim_{n \to +\infty} \int_{\mathbb{R}^N} f_n^{p-1} L_p f_n d\mu$$

$$= -\frac{1}{p-1} \int_{\mathbb{R}^N} f^{p-1} L_p f d\mu$$

$$= \int_{\mathbb{R}^N} f^{p-2} |Df|^2 d\mu.$$

Therefore, writing the LSI with f_n instead of f and, then, letting n go to $+\infty$, we can show the assertion. We refer the reader to the proof of Theorem 8.7.2 for further details.

The following result, due to L. Gross (see [69, 70]), states that the LSI implies that the semigroup $\{T(t)\}$ is hypercontractive, that is for any t>0 the operator T(t) is a contraction from L^2_μ to $L^{q(t)}_\mu$, where $q(t)=1+e^{\lambda t}$. Actually, the hypercontractivity is equivalent to the logarithmic Sobolev inequality (see again [69, 70]). As the LSI, also the hypercontractivity is a very sharp result: in general T(t) is not bounded from L^2_μ to L^p_μ when p>q(t).

Theorem 8.7.9 Assume Hypotheses 8.1.25 and (8.7.10) with $d_0 < 0$. Then,

$$||T(t)f||_{q(t)} \le ||f||_2, \qquad q(t) = 1 + e^{\lambda t},$$
 (8.7.12)

for any t > 0 and any $f \in L^2_{\mu}$, where λ is the constant appearing in (8.7.1).

Proof. First, we assume that $f \in D(\widehat{A})$ and $f \geq \delta > 0$. Consider the function

$$g(t,x) = ((T(t)f)(x))^{q(t)}, t \ge 0, x \in \mathbb{R}^N.$$

Since $f \in D(\widehat{A})$, then, for any fixed $x \in \mathbb{R}^N$, $g(\cdot,x) \in C^1([0,+\infty))$ (see Proposition 2.3.5) and

$$\begin{split} \frac{\partial g}{\partial t}(t,x) &= q(t)((T(t)f)(x))^{q(t)-1}(\widehat{A}T(t)f)(x) \\ &+ ((T(t)f)(x))^{q(t)}q'(t)\log((T(t)f)(x)), \end{split}$$

for any $t \geq 0$. Since $\widehat{A}T(t)f = T(t)\widehat{A}f$ and $0 < \delta \leq T(t)f \leq ||f||_{\infty}$ for any t > 0 (see Lemma 2.3.3), then, for any fixed T > 0, there exists a constant $C_T > 0$ such that

$$\left| \frac{\partial g}{\partial t}(t, x) \right| \le C_T, \qquad t \ge 0, \quad x \in \mathbb{R}^N.$$

It follows that the function $G:[0,+\infty)\to\mathbb{R}$ defined by

$$G(t) = \int_{\mathbb{R}^N} g(t, \cdot) d\mu, \qquad t \ge 0,$$

230

belongs to $C^1([0,+\infty))$ and

$$G'(t) = \int_{\mathbb{R}^N} \frac{\partial g}{\partial t}(t, \cdot) d\mu, \qquad t \ge 0.$$

Now, consider the function $z:[0+\infty)\to\mathbb{R}$ defined by

$$z(t) = \log ||T(t)f||_{q(t)} = \frac{1}{q(t)} \log G(t), \qquad t \ge 0.$$

It belongs to $C^1([0,+\infty))$ and

$$z'(t) = -\frac{q'(t)}{(q(t))^2} \log G(t) + \frac{G'(t)}{q(t)G(t)}$$

$$= -\frac{q'(t)}{q(t)} \log ||T(t)f||_{q(t)} + \frac{1}{||T(t)f||_{q(t)}^{q(t)}} \int_{\mathbb{R}^N} (T(t)f)^{q(t)-1} \widehat{A}T(t)f d\mu$$

$$+ \frac{q'(t)}{q(t)||T(t)f||_{q(t)}^{q(t)}} \int_{\mathbb{R}^N} (T(t)f)^{q(t)} \log T(t)f d\mu, \tag{8.7.13}$$

for any $t \geq 0$. Observing that $q'(t) = \lambda(q(t) - 1)$ and that $\widehat{A} = L$ on $D(\widehat{A})$ (see Proposition 8.1.9), and using the formula (8.7.11), we get

$$\int_{\mathbb{R}^N} (T(t)f)^{q(t)-1} \widehat{A} T(t) f d\mu = -\frac{q'(t)}{\lambda} \int_{\mathbb{R}^N} (T(t)f)^{q(t)-2} |DT(t)f|^2 d\mu.$$
 (8.7.14)

Replacing (8.7.14) into (8.7.13) we get

$$\begin{split} \frac{q(t)}{q'(t)} \|T(t)f\|_{q(t)}^{q(t)} z'(t) &= -\|T(t)f\|_{q(t)}^{q(t)} \log \|T(t)f\|_{q(t)} \\ &- \frac{q(t)}{\lambda} \int_{\mathbb{R}^N} (T(t)f)^{q(t)-2} |DT(t)f|^2 d\mu \\ &+ \int_{\mathbb{R}^N} (T(t)f)^{q(t)} \log T(t) f d\mu. \end{split} \tag{8.7.15}$$

Now, according to Theorem 8.7.2, the function T(t)f satisfies the LSI (8.7.1) for any $p \in [1, +\infty)$ and hence, in particular, for p = q(t). This implies that the right-hand side of (8.7.15) is negative. Consequently, $z(t) \leq z(0)$, which yields (8.7.12).

Next, we consider the case when $f \in D(\widehat{A})$ is nonnegative. In this situation (8.7.12) follows from the previous step, by approximating f with $f + \varepsilon$ ($\varepsilon > 0$). In particular, (8.7.12) holds for any nonnegative $f \in C_c^{\infty}(\mathbb{R}^N)$.

Finally, let $f \in L^2_{\mu}$. Consider a sequence of functions $\{f_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ converging to |f| in L^2_{μ} . Then $T(t)f_n$ converges to T(t)|f| in L^2_{μ} as n tends to $+\infty$. By (8.7.12), applied to $T(t)(f_n - f_m)$, we see that $T(t)f_n$ converges to

T(t)|f| in $L^{q(t)}_{\mu}$ as well, and that |f| satisfies (8.7.12). Since $|T(t)f| \leq T(t)|f|$ (recall that $\{T(t)\}$ is a positive semigroup, see Remark 2.2.3 and Theorem 2.2.5), f satisfies (8.7.12) as well.

The assumptions under which we proved the LSI in Theorem 8.7.2 imply also the Poincaré inequality (by Theorem 8.6.3). But one can also prove directly that the LSI implies the Poincaré inequality. We show here the proof in [134].

Proposition 8.7.10 If the LSI (8.7.1) holds for any $f \in D(\widehat{A})$ such that $f \geq \delta > 0$, then the Poincaré inequality (8.6.1) holds with p = 2 and $C = 2/\lambda$.

Proof. Without loss of generality, we can assume that $\overline{f} = 0$, since the general case then follows by considering the function $f - \overline{f}$. Moreover, to fix ideas, we assume that $\delta = 1/2$. Let $\varepsilon > 0$ be such that

$$\varepsilon \|f\|_{\infty} < \frac{1}{2} \tag{8.7.16}$$

and define the function

$$g = 1 + \varepsilon f \in D(\widehat{A}).$$

Then $g \geq 1/2$, and from the LSI (8.7.1) with p = 2, it follows that

$$\int_{\mathbb{R}^N} (1+\varepsilon f)^2 \log(1+\varepsilon f) d\mu \le \|g\|_2^2 \log \|g\|_2 + \frac{2\varepsilon^2}{\lambda} \int_{\mathbb{R}^N} |Df|^2 d\mu. \tag{8.7.17}$$

Now, we use the Taylor expansion of the function

$$\varphi(s) = (1+s)^2 \log(1+s), \qquad s > -1,$$

at s = 0. We have

$$\varphi(s) = s + \frac{3}{2}s^2 + r(s), \qquad r(s) = \frac{\varphi'''(\xi_s)}{6}s^3, \quad s > -1,$$
 (8.7.18)

where ξ_s is a suitable point in the interval joining 0 and s. Substituting the expansion (8.7.18) in (8.7.17), and dividing by ε^2 , we get

$$\frac{3}{2} \|f\|_{2}^{2} + \frac{1}{\varepsilon^{2}} \int_{\mathbb{R}^{N}} r(\varepsilon f) d\mu \leq \frac{1}{2\varepsilon^{2}} (1 + \varepsilon^{2} \|f\|_{2}^{2}) \log(1 + \varepsilon^{2} \|f\|_{2}^{2}) + \frac{2}{\lambda} \int_{\mathbb{R}^{N}} |Df|^{2} d\mu. \tag{8.7.19}$$

Since $\varphi'''(s) = 2/(1+s)$, we have

$$|r(s)| \le \frac{2}{3}|s|^3, \qquad s > -\frac{1}{2}.$$
 (8.7.20)

Now, from (8.7.16) and (8.7.20) it follows that

$$\lim_{\varepsilon \to 0^+} \frac{1}{\varepsilon^2} \int_{\mathbb{R}^N} r(\varepsilon f) d\mu = 0.$$

232

Moreover,

$$\lim_{\varepsilon \to 0^+} \frac{1}{2\varepsilon^2} (1 + \varepsilon^2 ||f||_2^2) \log(1 + \varepsilon^2 ||f||_2^2) = \frac{1}{2} ||f||_2^2.$$

Therefore, letting ε tend to 0^+ in (8.7.19), we obtain the Poincaré inequality (8.6.1) with $C = 2/\lambda$.

Remark 8.7.11 For higher order logarithmic Sobolev inequalities we refer the reader to the papers [55, 56], where the author proves that, if μ is a Gaussian measure and $f \in W^{2,2}_{\mu}$, then $f^2(\log(f))^2$ is in L^1_{μ} .

Chapter 9

The Ornstein-Uhlenbeck operator

9.0 Introduction

In this chapter we present some results concerned with the Ornstein-Uhlenbeck operator, which is the prototype of an elliptic operator with unbounded coefficients. Such an operator is defined on smooth functions φ by

$$(\mathcal{A}\varphi)(x) = \frac{1}{2} \operatorname{Tr}(QD^2 \varphi(x)) + \langle Bx, D\varphi(x) \rangle, \qquad x \in \mathbb{R}^N,$$
 (9.0.1)

where Q and B are $N \times N$ constant matrices, with Q (strictly) positive definite and $B \neq 0$. Here, if not otherwise specified, we consider only the case when Q is strictly positive. In any case, some of the results that we present hold also when Q is a degenerate positive definite matrix and the operator \mathcal{A} in (9.0.1) is hypoelliptic, i.e., when the matrix

$$\int_0^t e^{sB} Q e^{sB^*} ds$$

is strictly positive definite for any t > 0. Such a condition can be expressed also by saying that the kernel of Q does not contain any invariant subspace of B^* (see, e.g., [98]).

Firstly, in Section 9.1, we show that an explicit formula for the Ornstein-Uhlenbeck semigroup is available both in the nondegenerate and in the degenerate case. Having such a formula simplifies the study of the main properties of the semigroup. For instance, one can prove uniform estimates for the space derivatives of any order of the function T(t)f when $f \in C_b(\mathbb{R}^N)$ just differentiating under the integral sign. We do this in Section 9.2 in the nondegenerate case. The case of the degenerate Ornstein-Uhlenbeck operator is much more involved. It has been studied by A. Lunardi in [107]. Here, we limit ourselves to state the main results of [107].

As it has been claimed several times, the Ornstein-Uhlenbeck semigroup is neither analytic nor strong continuous in $C_b(\mathbb{R}^N)$. In particular, T(t)f tends to f in $C_b(\mathbb{R}^N)$ as t tends to 0^+ , if and only if $f \in BUC(\mathbb{R}^N)$ and $f(e^{tB}\cdot)$ tends to f uniformly in \mathbb{R}^N .

In Section 9.3, we deal with the invariant measure of $\{T(t)\}$. We show that when the spectrum of the matrix B is contained in the left halfplane, the

Ornstein-Uhlenbeck semigroup admits the Gaussian measure

$$\mu(dx) = \frac{1}{\sqrt{(2\pi)^N \det Q_{\infty}}} e^{-\frac{1}{2}\langle Q_{\infty}^{-1} x, x \rangle} dx$$

as the (unique) invariant measure, both in the nondegenerate and in the degenerate case. Here,

$$Q_{\infty} = \int_0^{+\infty} e^{sB} Q e^{sB^*} ds. \tag{9.0.2}$$

The assumptions on the location of the spectrum of the matrix B turns out to be also necessary to guarantee the existence of the invariant measure of $\{T(t)\}$.

From the results in Chapter 8, we know that the extension of the Ornstein-Uhlenbeck semigroup to the L^p -spaces associated with the invariant measure μ (in short L^p_{μ}) gives rise to a strongly continuous semigroup for any $p \in [1, +\infty)$. Moreover, for any $f \in L^p_{\mu}$, T(t)f is still given by the same formula as in the case when $f \in C_b(\mathbb{R}^N)$.

Actually, in the nondegenerate case, $\{T(t)\}$ is also analytic for any $p \in (1, +\infty)$. Also in this situation, having an explicit representation formula for T(t)f is of much help. Indeed, we can quite easily show that T(t) maps L^p_μ into $W^{k,p}_\mu$ (for any $k \in \mathbb{N}$) and we can also give precise estimates on the behaviour of the space derivatives of T(t)f in L^p_μ when t approaches 0^+ .

An important feature of the Ornstein-Uhlenbeck semigroup in L^p_μ is that a complete characterization of its infinitesimal generator L_p is available. More precisely, we show that

$$D(L_p) = W_{\mu}^{2,p}, \qquad p \in (1, +\infty)$$

and that the graph norm is equivalent to the Euclidean norm of $W^{2,p}_{\mu}$. Such a result was firstly proved by A. Lunardi and G. Da Prato in the Hilbert case, and then it has been proved by G. Metafune, D. Pallara, A. Rhandi and R. Schnaubelt for a general p.

Since the Ornstein-Uhlenbeck semigroup in L^p_{μ} is compact for any $p \in (1, +\infty)$, the spectrum of L_p is a discrete set. It has been completely characterized by G. Metafune, D. Pallara and E. Priola in terms of the eigenvalues $\lambda_1, \ldots, \lambda_N$ of the matrix B. More precisely, they show that

$$\sigma(L_p) = \left\{ \lambda = \sum_{i=1}^r n_i \lambda_i : n_i \in \mathbb{N} \cup \{0\}, \ i = 1, \dots, r \right\}.$$

In particular, $\sigma(L_p)$ is independent of p and the eigenfunctions of L_p are polynomials with degree at most $(\text{Re}\lambda)/s(B)$. Moreover, the eigenvalues of L_p are all semisimple if and only if B is diagonalizable in \mathbb{C} .

The picture changes drastically in the case when p=1. Indeed, in such a situation $\sigma(L_1)$ is the set of all the $\lambda \in \mathbb{C}$ with nonpositive real part. In particular, any λ with negative real part is an eigenvalue of L_1 .

Since the Ornstein-Uhlenbeck semigroup is analytic in L^p_μ , the function $z \mapsto T(z)$ is analytic in a sector centered at the origin. A recent result by R. Chill, E. Fasangova, G. Metafune and D. Pallara explicitly characterizes the amplitude of such a sector.

Next, we exploit some relations between the Ornstein-Uhlenbeck semigroup and the Hermite polynomials.

Finally, in Section 9.4, we consider the Ornstein-Uhlenbeck semigroup in the L^p -spaces associated with the Lebesgue measure, for any $p \in [1, +\infty)$. In such a setting $\{T(t)\}$ is strongly continuous but it is not analytic. The domain of its infinitesimal generator A_p is the set of all the functions $u \in W^{2,p}(\mathbb{R}^N)$ such that $\mathcal{A}u \in L^p(\mathbb{R}^N)$. Moreover, the spectrum of A_p consists of all the complex numbers with real parts not greater than -(Tr(B))/p. In particular, differently from what happens when the underlining measure is the invariant measure, the spectrum depends explicitly on p.

9.1 The formula for T(t)f

One of the best features of the Ornstein-Uhlenbeck operator is that an explicit formula for the associated semigroup is available both in the nondegenerate and in the degenerate case. Such a formula is due to Kolmogorov (see [85]).

Let us introduce some notation. For any t > 0, we denote by $Q_t \in L(\mathbb{R}^N)$ the matrix defined by

$$Q_t = \int_0^t e^{sB} Q e^{sB^*} ds. (9.1.1)$$

It is readily seen that Q_t (t > 0) is a positive definite matrix, and it is strictly positive definite whenever Q is.

Moreover, for any strictly positive definite matrix K and any $a \in \mathbb{R}^N$, we denote by $\mathcal{N}(a, K)$ the Gaussian measure defined by

$$\mathcal{N}(a,K)(dx) = \frac{1}{\sqrt{(2\pi)^N \det K}} e^{-\frac{1}{2}\langle K^{-1}(x-a), x-a \rangle} dx. \tag{9.1.2}$$

We can now prove the following theorem.

Theorem 9.1.1 For any $f \in C_b(\mathbb{R}^N)$, the Cauchy problem

$$\begin{cases}
D_t u(t,x) = \mathcal{A}u(t,x), & t > 0, \ x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N,
\end{cases}$$
(9.1.3)

admits a unique classical solution u which is bounded in $[0,T] \times \mathbb{R}^N$ for any T > 0. In fact, such a solution is bounded in $[0,+\infty) \times \mathbb{R}^N$ and

$$\sup_{t>0} \|u(t,\cdot)\|_{C_b(\mathbb{R}^N)} \le \|f\|_{\infty}. \tag{9.1.4}$$

Moreover, u is given by

$$u(t,x) = (T(t)f)(x) := \frac{1}{\sqrt{(2\pi)^N \det Q_t}} \int_{\mathbb{R}^N} e^{-\frac{1}{2}\langle Q_t^{-1}(e^{tB}x - y), e^{tB}x - y\rangle} f(y) dy,$$
(9.1.5)

for any t > 0 and any $x \in \mathbb{R}^N$. The semigroup $\{T(t)\}$ is called the Ornstein-Uhlenbeck semigroup associated with the operator \mathcal{A} in $C_b(\mathbb{R}^N)$.

Proof. First of all, we observe that the existence of the classical solution to the problem (9.1.3) as well as the estimate (9.1.4) are guaranteed by Theorem 2.2.1, whereas its uniqueness is an immediate consequence of Theorem 4.1.3, where we can choose as a Lyapunov function, the function $\varphi : \mathbb{R}^N \to \mathbb{R}$ defined by $\varphi(x) = |x|^2$ for any $x \in \mathbb{R}^N$. Indeed, an explicit computation shows that

$$(\mathcal{A}\varphi)(x) = \operatorname{Tr} Q + 2\langle Bx, x \rangle \le \operatorname{Tr} Q + 2\|B\|_{\infty}|x|^2, \qquad x \in \mathbb{R}^N.$$

Hence, it suffices to take $\lambda = 2\|B\|_{\infty}$ to get $\sup_{\mathbb{R}^N} (\mathcal{A}\varphi - \lambda\varphi) < +\infty$ and, by Remark 4.0.3, this is enough to apply Theorem 4.1.3.

So, to complete the proof we just need to check the formula (9.1.5). For this purpose, we use stochastic calculus. Indeed, the stochastic differential equation associated with the operator \mathcal{A} is the following one:

$$d\xi_t^x = B\xi_t^x dt + \sigma dW_t, \qquad \xi_0^x \equiv x,$$

where $\sigma = Q^{1/2}$, W_t is a standard N-dimensional Brownian motion and $x \in \mathbb{R}^N$. It is easily verified that the solution of this equation is the process

$$\xi_t^x = e^{tB}x + \int_0^t \sigma e^{(t-s)B} dW_s, \qquad t > 0.$$

This is a Gaussian process. For any fixed t > 0 and any $x \in \mathbb{R}^N$ the mean value of ξ_t^x is $e^{tB}x$ and the covariance matrix is the matrix Q_t defined in (9.1.1). Therefore, the distribution of ξ_t^x is the normal distribution $\mathcal{N}(e^{tB}x, Q_t)(dx)$. This implies that the semigroup $\{T(t)\}$ is given by the formula

$$(T(t)f)(x) = \mathbb{E} f(\xi_t^x) = \int_{\mathbb{R}^N} f(y) \mathcal{N}(e^{tB}x, Q_t)(dy), \quad x \in \mathbb{R}^N,$$

for any $f \in B_b(\mathbb{R}^N)$, and (9.1.5) follows.

Let us give also an analytical proof of the formula (9.1.5). For this purpose, we look for a solution u of (9.1.3) in the form

$$u(t,x) = v(t, M(t)x), t > 0, x \in \mathbb{R}^N,$$
 (9.1.6)

where the matrix-valued function $M:(0,+\infty)\to L(\mathbb{R}^N)$, defined by $t\mapsto M(t)$, is to be properly chosen.

A straightforward computation shows that the function v solves the Cauchy problem

$$\begin{cases}
D_t v(t, x) = \mathcal{A}(t) v(t, x), & t > 0, \ x \in \mathbb{R}^N, \\
v(0, x) = f(M(0)x), & x \in \mathbb{R}^N,
\end{cases}$$
(9.1.7)

where $\mathcal{A}(t)$ is given by

$$\mathcal{A}(t) = \frac{1}{2} \text{Tr}(\Lambda(t) D^2 v(t, x)) + \langle (M(t) B - M'(t)) (M(t))^{-1} x, D v(t, x) \rangle, (9.1.8)$$

with $\Lambda(t) = M(t)Q(M(t))^*$, for any t > 0.

Now, we choose M(t) in order to eliminate the drift term in (9.1.8). This is possible if we choose $M(\cdot)$ satisfying the differential equation $M'(\cdot) = M(\cdot)B$. For instance, we can take $M(t) = e^{tB}$ for any t > 0. With this choice of $M(\cdot)$ the elliptic equation in (9.1.7) reduces to

$$\begin{cases}
D_t v(t, x) = \frac{1}{2} \text{Tr}(e^{tB} Q e^{tB^*} D^2 v(t, x)), & t > 0, \ x \in \mathbb{R}^N, \\
u(0, x) = f(x), & x \in \mathbb{R}^N.
\end{cases}$$
(9.1.9)

Taking the Fourier transform of both the sides of the differential equation in (9.1.9) gives the ordinary differential equation

$$D_t \hat{v}(t,\xi) = -\frac{1}{2} \langle \Lambda(t)\xi, \xi \rangle \, \hat{v}(t,\xi), \qquad t > 0, \ \xi \in \mathbb{R}^N.$$

Hence,

$$\hat{v}(t,\xi) = \exp\left(-\frac{1}{2}\langle Q_t\xi,\xi\rangle\right)\hat{f}(\xi), \qquad t > 0, \ \xi \in \mathbb{R}^N, \qquad (9.1.10)$$

where Q_t is given by (9.1.1). Taking the inverse Fourier transform gives

$$v(t,x) = \frac{1}{\sqrt{(2\pi)^N \det Q_t}} \int_{\mathbb{R}^N} e^{-\frac{1}{2}\langle Q_t^{-1}(x-y), x-y \rangle} f(y) dy.$$
 (9.1.11)

Now, the formulas (9.1.6) and (9.1.11) lead us to (9.1.5).

Remark 9.1.2 The same results as in Theorem 9.1.1 hold true also in the case when Q is singular (but still positive definite) provided that the matrix Q_t in (9.1.1) is nonsingular for any t > 0. The semigroup defined by the formula (9.1.5) is still called the (degenerate) Ornstein-Uhlenbeck semigroup associated with the operator \mathcal{A} in (9.0.1). As it has been stressed in the introduction to this chapter, the condition $\det Q_t > 0$, for any t > 0, is equivalent to the hypoellipticity of the operator \mathcal{A} in the sense of Hörmander (see [73]).

9.2 Properties of $\{T(t)\}\$ in $C_b(\mathbb{R}^N)$

In this section we study some properties of $\{T(t)\}$ in the space $C_b(\mathbb{R}^N)$. To begin with, let us consider the following proposition.

Proposition 9.2.1 The following properties are met:

- (i) $\{T(t)\}\ is\ conservative\ in\ C_b(\mathbb{R}^N);$
- (ii) $C_0(\mathbb{R}^N)$ is invariant for $\{T(t)\}$;
- (iii) for any t > 0, T(t) is not compact in $C_b(\mathbb{R}^N)$.

Proof. The property (i) follows immediately from Proposition 4.1.10 since T(t)1 = 1 for any t > 0.

The property (ii) follows immediately from Example 5.3.3.

Finally, the property (iii) is now a straightforward consequence of the properties (i) and (ii) and Proposition 5.3.4.

The estimates for the derivatives of $\{T(t)f\}$ can be easily obtained as a consequence of the results in Chapters 6 and 7. They can also be deduced differentiating under the integral sign the representation formula (9.1.5). This is the approach used in [39] that allows the authors to prove uniform (and pointwise) estimates for the space derivatives of T(t)f of any order.

Proposition 9.2.2 For any t > 0 and any $f \in C_b(\mathbb{R}^N)$, the function T(t)f belongs to $C_b^k(\mathbb{R}^N)$ for any $k \in \mathbb{N}$. In particular, if $f \in C_b^1(\mathbb{R}^N)$,

$$D_i T(t) f = \sum_{j=1}^{N} (e^{tB^*})_{ij} T(t) D_j f, \qquad t > 0, \ i, j = 1, \dots, N.$$
 (9.2.1)

Moreover, for any $\varepsilon > 0$ and any $h \in \mathbb{N}$ with $h \leq k$, there exists a positive constant $C = C_{\varepsilon}$ such that

$$||T(t)f||_{C_b^k(\mathbb{R}^N)} \le C \frac{e^{k(s(B)+\varepsilon)t}}{1 \wedge t^{\frac{k-h}{2}}} ||f||_{C_b^h(\mathbb{R}^N)}, \qquad t > 0, \tag{9.2.2}$$

for any $f \in C_b^h(\mathbb{R}^N)$, where s(B) is the spectral bound of the matrix B.

Proof. The estimate (9.2.2) can be proved by induction on k. Let us begin by proving it in the case when h=0. For k=1 the estimate (9.2.2), as well as the formula (9.2.1), follows easily differentiating the formula (9.1.5) under the integral sign and observing that, for any $\delta > 0$, there exists a constant $\widetilde{C}_1 = \widetilde{C}_1(\delta)$ such that

$$||e^{tB}||_{\infty} \le \widetilde{C}_1 e^{(s(B)+\delta)t}, \qquad t > 0.$$
 (9.2.3)

Let us suppose that T(t)f belongs to $C_b^{m-1}(\mathbb{R}^N)$ for any t>0 and (9.2.2) holds for all the (m-1)-th order derivatives of T(t)f. We are going to prove that the m-th order derivative $D_{i_1,\ldots,i_m}T(t)f$ exists in $C_b(\mathbb{R}^N)$ and it satisfies (9.2.2). For this purpose, we observe that, using (9.2.1), we get

$$D_{i_{2},...,i_{m}}T(t)f$$

$$= D_{i_{2},...,i_{m}}T(t/n) \cdot ... \cdot T(t/n)f$$

$$= D_{i_{2},...,i_{m-1}} \left\{ T((1-1/n)t) \sum_{j_{m}=1}^{N} (e^{\frac{n-1}{n}B^{*}t})_{i_{m}j_{m}} D_{j_{m}}T(t/n)f \right\}$$

$$= T((1-(m-1)/n)t) \sum_{j_{2},...,j_{m}=1}^{N} (e^{\frac{n-m+1}{n}tB^{*}})_{i_{2}j_{2}} \cdot ... \cdot (e^{\frac{n-1}{n}tB^{*}})_{i_{m}j_{m}} \times D_{j_{2}}T(t/n) \circ ... \circ D_{j_{m}}T(t/n)f,$$

$$(9.2.4)$$

for any n > m and any t > 0. Similarly,

$$D_{j_2}T(t/n) \circ \dots \circ D_{j_m}T(t/n)f$$

$$= T((m/n - 2/n)t) \sum_{l_2,\dots,l_{m-1}=1}^{N} (e^{\frac{1}{n}tB^*})_{j_{m-1}l_{m-1}} \cdot \dots \cdot (e^{\frac{m-2}{n}tB^*})_{j_2l_2}$$

$$\times D_{l_2,\dots,l_{m-1},j_m}T(t/n)f, \qquad (9.2.5)$$

for any t > 0. From (9.2.4), (9.2.5) and (9.2.2) (with k = 1) we easily deduce that $D_{i_2,...,i_m}T(t) \in C_b^1(\mathbb{R}^N)$. Moreover,

$$D_{i_{1},...,i_{m}}T(t)f$$

$$= \sum_{j_{2},...,j_{m}=1}^{N} \left(e^{\frac{n-m+1}{n}tB^{*}}\right)_{i_{2}j_{2}} \cdot ... \cdot \left(e^{\frac{n-1}{n}tB^{*}}\right)_{i_{m}j_{m}}$$

$$\times \sum_{l_{1},...,l_{m-1}=1}^{N} \left(e^{\frac{1}{n}tB^{*}}\right)_{j_{m-1}l_{m-1}} \cdot ... \cdot \left(e^{\frac{m-2}{n}tB^{*}}\right)_{j_{2}l_{2}} \left(e^{\frac{n-2}{n}tB^{*}}\right)_{i_{1}l_{1}}$$

$$\times T((1-2/n)t)D_{l_{1}}T(t/n)D_{l_{2},...,l_{m-1},j_{m}}T(t/n)f.$$

$$(9.2.6)$$

Therefore, taking $\delta = 1/n$ in (9.2.3) and using (9.2.2) with k = 1, k = m-1 and $\varepsilon = 1/n$, we get

$$||D_{i_1,...,i_m}T(t)f||_{\infty} \le C_n \exp\left\{\left(m - \frac{1}{n}\right)\left(s(B) + \frac{1}{n}\right)\right\} \frac{1}{1 \wedge t^{m/2}} ||f||_{\infty},$$

for any t > 0 and some positive constant C_n . The estimate (9.2.2) with k = m follows, taking n sufficiently large.

The estimate (9.2.2) with h = k follows immediately from (9.2.3). Indeed, a straightforward computation shows that

$$D_{i_1,\dots,i_k}T(t)f = \sum_{j_1,\dots,j_k=1}^{N} (e^{tB^*})_{i_1j_1} \cdot \dots \cdot (e^{tB^*})_{i_kj_k}T(t)D_{j_1,\dots,j_k}f, \quad (9.2.7)$$

for any $i_1, \ldots, i_k \in \{1, \ldots, N\}$, any $k \in \mathbb{N}$ and any t > 0.

Finally, in the case when 0 < h < k, the estimate (9.2.2) follows from (9.2.6) (with m = k) observing that, thanks to (9.2.7), we have

$$D_{l_1}T(t/n)D_{l_2,\dots,l_{k-1},j_k}T(t/n)f$$

$$= \sum_{r_1,\dots,r_{h+1}=1}^{N} (e^{\frac{1}{n}tB^*})_{l_1r_1}(e^{\frac{1}{n}tB^*})_{l_{k-h+1}r_2}\cdot\dots\cdot(e^{\frac{1}{n}tB^*})_{l_{k-1}r_h}(e^{\frac{1}{n}tB^*})_{j_kr_{h+1}}$$

$$\times T(t/n)(D_{r_1,l_2,\dots,l_{k-h}}T(t/n)D_{r_2,\dots,r_{h+1}}f),$$

for any t > 0.

Remark 9.2.3 (i) From the proof of Proposition 9.2.2, it is clear that if (9.2.3) holds with $\delta = 0$, then the estimate (9.2.2) can be written also with $\varepsilon = 0$.

(ii) In view of Remark 7.2.4, when k = 1, 2, 3, the uniform estimate (9.2.2) is useful only in the particular case when s(B) < 0.

Remark 9.2.4 In the case when the matrix Q is singular and the operator \mathcal{A} is hypoelliptic, uniform estimates of the Ornstein-Uhlenbeck semigroup are still available, but they differ from those in Proposition 9.2.2 since, as we will see in Chapter 14, the behaviour of the space derivatives of T(t)f near t=0 is anisotropic. Such estimates, which have been proved in [107, Proposition 3.2], can be written in an elegant way if the Ornstein-Uhlenbeck is written not in the usual Euclidean basis of \mathbb{R}^N but in a suitable orthonormal basis $\{e_i': i=1,\ldots,N\}$ associated with the matrix Q.

To determine such a basis, we begin by observing that the operator \mathcal{A} is hypoelliptic if and only if there exists $m \leq N-1$ such that $\operatorname{rank} H_m = N$, where the matrix $H_m \in L(\mathbb{R}^{mN}, \mathbb{R}^N)$ is given by $H_m = [Q^{1/2}, BQ^{1/2}, \dots, B^mQ^{1/2}]$. Such a condition is usually called the Kalman rank condition and its equivalence with the hypoellipticity condition is proved in [149, Chapter 1]. We denote by r the smallest integer such that $\operatorname{rank} H_r = N$, and by V_i ($i = 0, \dots, r-1$) the image of \mathbb{R}^N through the matrix H_i ($0 = 1, \dots, r$). It is now easy to check that $\mathbb{R}^N = \bigoplus_{k=0}^r E_k(\mathbb{R}^N)$ where E_0 is the orthogonal projection of \mathbb{R}^N in V_0 and E_i ($j = i, \dots, r$) is the orthogonal projection of \mathbb{R}^N on V_{i-1}^{\perp} in V_i , $i = 1, \dots, r$. Then, we define the orthonormal basis $\{e'_i : i = 1, \dots, N\}$ of \mathbb{R}^N choosing e'_j in $E_i(\mathbb{R}^N)$ for any $j \in \{a_i + 1, \dots, a_{i+1}\} := I_i$ ($i = 0, \dots, r$),

where $a_0 = 0$, $a_{r+1} = N$. In such a basis it can be shown that, for any $h \in \mathbb{N}$, any multi-index α with $|\alpha| = h$, and any $\varepsilon > 0$, there exists a constant $C_h = C_h(\varepsilon)$ such that

$$||D^{\alpha}T(t)f||_{\infty} \le C_h e^{(s(B)+\varepsilon)t} t^{-\frac{h}{2}-m_{\alpha}} ||f||_{\infty}, \qquad t > 0,$$
 (9.2.8)

where, $m_{\alpha} = \sum_{i=0}^{r} \sum_{j=a_i+1}^{a_{i+1}} \tau(\alpha_j)$. Here, $\tau(\alpha_j) = k$, where k is such that $j \in I_k$.

As we have already several times recalled, both the Ornstein-Uhlenbeck semigroup and the degenerate Ornstein-Uhlenbeck semigroup are neither analytic nor strongly continuous in $C_b(\mathbb{R}^N)$ and in $BUC(\mathbb{R}^N)$. Now we provide a proof of these properties. To begin with we show that $\{T(t)\}$ is not analytic, adapting the proof of [39, Lemma 3.3]. This result holds also in the case when Q is singular.

Proposition 9.2.5 The Ornstein-Uhlenbeck semigroup is analytic neither in $C_b(\mathbb{R}^N)$ nor in $BUC(\mathbb{R}^N)$.

Proof. We begin the proof considering the case when Q is not singular. Of course, we can limit ourselves to proving that $\{T(t)\}$ is not analytic in $BUC(\mathbb{R}^N)$. For this purpose, we prove that there exists $h \in \mathbb{R}^N$ such that

$$\sup_{h \in \mathbb{N}} \|\mathcal{A}T(1)f_h\|_{\infty} = +\infty, \tag{9.2.9}$$

where $f_h(x) = \sin(\langle h, x \rangle)$ for any $x \in \mathbb{R}^N$. Indeed, we have

$$(T(1)f_h)(x) = e^{-\frac{1}{2}\langle Q_1 h, h \rangle} \sin(\langle e^B x, h \rangle), \qquad x, h \in \mathbb{R}^N$$
 (9.2.10)

(see also (9.3.6)). From (9.2.10) it follows that the function $\operatorname{Tr}(QD^2T(1)f_h)$ is bounded in \mathbb{R}^N for any $h \in \mathbb{R}^N$. Therefore, to prove (9.2.9) we need to show that the function $x \mapsto \langle Bx, (DT(1)f_h)(x) \rangle$ is unbounded in \mathbb{R}^N for some $h \in \mathbb{R}^N$. As it is easily seen

$$(B^*(DT(1)f_h)(x))_i = (B^*e^{B^*}h)_i e^{-\frac{1}{2}\langle Q_1h,h\rangle}\cos(\langle e^Bx,h\rangle),$$

for any $x \in \mathbb{R}^N$ and any i = 1, ..., N. Since $B \neq 0$, there exist $h_0 \in \mathbb{R}^N$ and $j \in \{1, ..., N\}$ such that $(B^*e^{B^*}h_0)_j \neq 0$. As a consequence, the function

$$r \mapsto \langle rBe_j, (DT(1)f_{h_0})(re_j) \rangle = r(B^*e^{B^*}h_0)_j e^{-\frac{1}{2}\langle Q_1h_0, h_0 \rangle} \cos(r(e^{B^*}h_0)_j)$$

is unbounded in \mathbb{R} and, therefore, $\langle B \cdot, DT(1) f_{h_0} \rangle$ is unbounded in \mathbb{R}^N . This completes the proof in this case.

Now we assume that Q is nonnegative definite and singular, and we denote by $\tilde{\mathcal{A}}$ the operator \mathcal{A} in the orthonormal basis $\{e'_i: i=1,\ldots,N\}$ defined

in Remark 9.2.4. Observe that $\mathcal{A}=M\tilde{\mathcal{A}}M^*$ for some orthogonal matrix M, where

$$\tilde{\mathcal{A}}\varphi = \sum_{i,j=1}^{a_1} q'_{ij} D_{ij} \varphi + \langle B \cdot, D\varphi \rangle,$$

for any smooth function φ , $Q'=(q'_{ij})$ and B' being suitable matrices, with Q' strictly positive definite, and a_1 being as in Remark 9.2.4. Then, $T(t)=M\tilde{T}(t)M^*$ for any t>0, where $\{\tilde{T}(t)\}$ denotes the Ornstein-Uhlenbeck semigroup associated with the operator $\tilde{\mathcal{A}}$. Therefore, we can limit ourselves to showing that $\{\tilde{T}(t)\}$ is not analytic in $BUC(\mathbb{R}^N)$. For this purpose, we can repeat step by step the proof given in the nondegenerate case.

The next proposition describes the set of strong continuity of $\{T(t)\}$. It holds for both nondegenerate and degenerate Ornstein-Uhlenbeck semigroups.

Proposition 9.2.6 Let $f \in C_b(\mathbb{R}^N)$. Then, $||T(t)f - f||_{\infty}$ tends to 0 as t tends to 0 if and only if $f \in BUC(\mathbb{R}^N)$ and

$$\lim_{t \to 0^+} (f(e^{tB}x) - f(x)) = 0,$$

uniformly with respect to $x \in \mathbb{R}^N$.

Proof. First of all let us notice that, if T(t)f tends to f uniformly in \mathbb{R}^N , then $f \in BUC(\mathbb{R}^N)$. Indeed, Proposition 9.2.2 ensures that, for any $f \in C_b(\mathbb{R}^N)$, the function T(t)f belongs to $BUC(\mathbb{R}^N)$. Therefore, in the rest of the proof we assume that $f \in BUC(\mathbb{R}^N)$.

For any t > 0, let G_t be the linear operator in $BUC(\mathbb{R}^N)$ defined by

$$(G_t f)(x) = \frac{1}{\sqrt{(2\pi)^N \det Q_t}} \int_{\mathbb{R}^N} f(x - y) e^{-\frac{1}{2} \langle Q_t^{-1} y, y \rangle} dy, \qquad x \in \mathbb{R}^N.$$

Then, we have $(T(t)f)(x) = (G_t f)(e^{tB}x)$ for any t > 0 and any $x \in \mathbb{R}^N$. Let us observe that $G_t f$ tends to f uniformly in \mathbb{R}^N as n tends to $+\infty$. Indeed, for any $n \in \mathbb{N}$,

$$|(G_{t}f)(x) - f(x)| = \left| \frac{1}{\sqrt{(2\pi)^{N}}} \int_{\mathbb{R}^{N}} e^{-\frac{1}{2}|y|^{2}} \left(f(x + \sqrt{Q_{t}}y) - f(x) \right) dy \right|$$

$$\leq \frac{1}{\sqrt{(2\pi)^{N}}} \int_{B(n)} e^{-\frac{1}{2}|y|^{2}} |f(x + \sqrt{Q_{t}}y) - f(x)| dy$$

$$+ \frac{2}{\sqrt{(2\pi)^{N}}} ||f||_{\infty} \int_{\mathbb{R}^{N} \setminus B(n)} e^{-\frac{1}{2}|y|^{2}} dy, \qquad t > 0, \ x \in \mathbb{R}^{N}.$$

$$(9.2.11)$$

From (9.2.11) it is now clear that for any $\varepsilon > 0$ we can find first n sufficiently large and, then, t sufficiently close to 0 such that both the two last integrals are less then $\varepsilon/2$, uniformly with respect to $x \in \mathbb{R}^N$.

Now, the conclusion follows writing

$$(T(t)f)(x) - f(x) = (G_t f)(e^{tB}x) - f(e^{tB}x) + f(e^{tB}x) - f(x),$$

for any t > 0 and any $x \in \mathbb{R}^N$.

To conclude this section we characterize the spectrum of the weak generator \widehat{A} of the Ornstein-Uhlenbeck operator in $C_b(\mathbb{R}^N;\mathbb{C})$ when Q is not singular. For the proof we refer the reader to [112, Corollary 6.3], where the characterization of the spectrum of the restriction of \widehat{A} to $BUC(\mathbb{R}^N;\mathbb{C})$ is given. But the spectra of \widehat{A} and its restriction to $BUC(\mathbb{R}^N;\mathbb{C})$ actually coincide. Indeed, of course, $\sigma(\widehat{A}_{|BUC(\mathbb{R}^N;\mathbb{C})}) \subset \sigma(\widehat{A})$. Moreover, since $D(\widehat{A}) \subset BUC(\mathbb{R}^N;\mathbb{C})$ the point spectra of \widehat{A} and $\widehat{A}_{|BUC(\mathbb{R}^N;\mathbb{C})}$ coincide. Further, if $\operatorname{Re} \lambda > 0$, then $\lambda \in \rho(\widehat{A})$. Indeed, fix $f \in C_b(\mathbb{R}^N;\mathbb{C})$ and consider a bounded sequence $\{f_n\} \subset C_c^{\infty}(\mathbb{R}^N;\mathbb{C})$ converging locally uniformly in \mathbb{R}^N to f. Taking the estimates in Proposition 9.2.2 into account it is easy to check that

$$(R(\lambda, \widehat{A}_{\mid BUC(\mathbb{R}^N; \mathbb{C})}) f_n)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t) f_n)(x) dt, \qquad (9.2.12)$$

for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, where we still denote by $\{T(t)\}$ the extension of the Ornstein-Uhlenbeck semigroup to complex-valued functions. From formula (9.2.12) and Proposition 2.2.9, it is immediate to check that $u_n = R(\lambda, \widehat{A}_{\mid BUC(\mathbb{R}^N;\mathbb{C})}) f_n$ converges locally uniformly to a function $u \in C_b(\mathbb{R}^N;\mathbb{C})$.

Since $\hat{A}u_n = \lambda u_n - f_n$ and

$$||R(\lambda, \widehat{A}_{\mid BUC(\mathbb{R}^N; \mathbb{C})}) f_n||_{\infty} \le K ||f_n||_{\infty}, \qquad n \in \mathbb{N},$$
 (9.2.13)

for some positive constant K, independent of n, it follows that $\{\widehat{A}u_n\}$ is bounded in $C_b(\mathbb{R}^N;\mathbb{C})$ and it converges locally uniformly in \mathbb{R}^N . Therefore, according to Proposition 2.3.5, $u \in D(\widehat{A})$ and $\lambda u - \widehat{A}u = f$. Moreover, by (9.2.13) we get $\|u\|_{\infty} \leq K\|f\|_{\infty}$ since we can assume that $\|f_n\|_{\infty} \leq \|f\|_{\infty}$ for any $n \in \mathbb{N}$.

Theorem 9.2.7 If $\sigma(B) \cap i\mathbb{R} = \emptyset$ then $\sigma(\widehat{A}) = \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \leq 0\}.$

9.3 The invariant measure μ and the semigroup in L^p_μ

In this section we assume that

$$\sigma(B) \subset \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda < 0\}.$$
 (9.3.1)

As a straightforward consequence, the matrix Q_t (see (9.1.1)) can be defined also for $t = +\infty$. Q_{∞} is a positive definite matrix and it is strictly positive definite if Q_t is strictly positive definite for some t > 0.

We set

$$\mu(dx) := \mathcal{N}(0, Q_{\infty})(dx) = \frac{1}{\sqrt{(2\pi)^N \det Q_{\infty}}} e^{-\frac{1}{2}\langle Q_{\infty}^{-1} x, x \rangle} dx := \rho(x) dx. \quad (9.3.2)$$

As the following theorem shows, μ is the invariant measure of both the nondegenerate and the degenerate Ornstein-Uhlenbeck semigroups.

Proposition 9.3.1 If (9.3.1) holds, then the Gaussian measure μ is the unique invariant measure of the semigroup $\{T(t)\}$.

Proof. In the case of the nondegenerate Ornstein-Uhlenbeck semigroup, the uniqueness of the invariant measure of $\{T(t)\}$ follows immediately from Theorem 8.1.15.

Actually, Theorem 8.1.15 holds also in the case when the diffusion matrix Q is singular. Indeed, to make the proof of that proposition work, one needs to be able to extend the semigroup with continuity to a strong Feller and irreducible semigroup in to $B_b(\mathbb{R}^N)$ and to prove the convergence of the sequence $(T(t)f_n)(x)$ to (T(t)f)(x), as n tends to $+\infty$, for any t>0 and any $x\in\mathbb{R}^N$, whenever the sequence $\{f_n\}\subset B_b(\mathbb{R}^N)$ converges in a dominated way to f. Of course, these two properties are satisfied also by the degenerate Ornstein-Uhlenbeck semigroup, as it can be easily checked using the representation formula (9.1.5).

So, let us prove that, in both the nondegenerate and the degenerate case $\{T(t)\}$, the Gaussian measure μ is the invariant measure of $\{T(t)\}$. Throughout the proof, we extend $\{T(t)\}$ to the space of bounded and continuous complex-valued functions in the natural way, and we still denote by $\{T(t)\}$ the so extended semigroup.

Denote by \mathcal{E} the linear span of the set $\{f_h : h \in \mathbb{R}^N\}$ where $f_h : \mathbb{R}^N \to \mathbb{C}$ is defined by $f_h(x) = e^{i\langle x,h\rangle}$ for any $x,h \in \mathbb{R}^N$. We observe that \mathcal{E} is dense in L^1_μ . To check this property, it suffices to show that, if $T \in (L^1_\mu)'$ vanishes on \mathcal{E} , then T = 0. For this purpose, we recall that $(L^1_\mu)' = L^\infty_\mu \subset L^\infty(\mathbb{R}^N, dx)$ (see [135, Theorem 6.16]). Fix $T \in (L^1_\mu)'$ vanishing on \mathcal{E} and let $g \in L^\infty(\mathbb{R}^N, dx)$ be such that $Tf = \int_{\mathbb{R}^N} fgd\mu$ for any $f \in L^1_\mu$. Then

$$\int_{\mathbb{R}^N} g(x) e^{i\langle x,h\rangle} e^{-\frac{1}{2}\langle Q_\infty x,x\rangle} dx = 0,$$

for any $h \in \mathbb{R}^N$. It follows that the Fourier transform of the function $x \mapsto g(x)e^{-\langle Q_{\infty}x,x\rangle/2}$ identically vanishes in \mathbb{R}^N . By the uniqueness of the Fourier transform, we deduce that $q \equiv 0$.

Hence, to prove that μ is the invariant measure of the Ornstein-Uhlenbeck semigroup, it suffices to show that

$$\int_{\mathbb{R}^N} T(t) f_h d\mu = \int_{\mathbb{R}^N} f_h d\mu, \qquad t > 0, \ h \in \mathbb{R}^N.$$
 (9.3.3)

For this purpose, fix $a \in \mathbb{R}^N$ and a strictly positive definite matrix K, and let us compute the Fourier transform of the probability measure $\mathcal{N}(a, K)(dx)$ (see (9.1.2)).

Let M be an orthogonal matrix such that $MKM^* = \operatorname{diag}(\lambda_1, \ldots, \lambda_N)$. Then

$$\int_{\mathbb{R}^{N}} f_{h} \mathcal{N}(a, K)(dy) = \frac{e^{i\langle a, h \rangle}}{\sqrt{(2\pi)^{N} \det K}} \int_{\mathbb{R}^{N}} e^{i\langle M^{*}y, h \rangle} \exp\left(-\sum_{i=1}^{N} \frac{y_{i}^{2}}{2\lambda_{i}}\right) dy$$

$$= \frac{e^{i\langle a, h \rangle}}{\sqrt{(2\pi)^{N} \lambda_{1} \cdot \ldots \cdot \lambda_{N}}} \prod_{k=1}^{N} \int_{\mathbb{R}} e^{iy_{k}(Mh)_{k}} e^{-\frac{y_{k}^{2}}{2\lambda_{k}}} dy_{k}$$

$$= e^{i\langle a, h \rangle} \prod_{k=1}^{N} e^{-\frac{1}{2}\lambda_{k}(Mh)_{k}^{2}}$$

$$= e^{i\langle a, h \rangle - \frac{1}{2}\langle Kh, h \rangle}, \tag{9.3.4}$$

for any $h \in \mathbb{R}^N$. Therefore,

$$\int_{\mathbb{R}^N} f_h d\mu = \int_{\mathbb{R}^N} e^{i\langle h, y \rangle} \mathcal{N}(0, Q_\infty)(dy) = e^{-\frac{1}{2}\langle Q_\infty h, h \rangle}, \qquad h \in \mathbb{R}^N. \quad (9.3.5)$$

On the other hand, we have

$$(T(t)f_h)(x) = \int_{\mathbb{R}^N} e^{i\langle h, y \rangle} \mathcal{N}(e^{tB}x, Q_t)(dy) = e^{i\langle e^{tB}x, h \rangle - \frac{1}{2}\langle Q_t h, h \rangle}, \qquad (9.3.6)$$

for any t > 0, any $x, h \in \mathbb{R}^N$, and then

$$\int_{\mathbb{R}^N} T(t) f_h d\mu = e^{-\frac{1}{2} \langle Q_t h, h \rangle} \int_{\mathbb{R}^N} e^{i \langle e^{tB^*} h, x \rangle} \mathcal{N}(0, Q_\infty)(dx)$$
$$= e^{-\frac{1}{2} \langle (Q_t + e^{tB} Q_\infty e^{tB^*}) h, h \rangle}. \tag{9.3.7}$$

Now, observing that

$$Q_{t} + e^{tB}Q_{\infty}e^{tB^{*}} = \int_{0}^{t} e^{sB}Qe^{sB^{*}}ds + \int_{0}^{+\infty} e^{(t+s)B}Qe^{(t+s)B^{*}}ds$$
$$= \int_{0}^{t} e^{sB}Qe^{sB^{*}}ds + \int_{t}^{+\infty} e^{sB}Qe^{sB^{*}}ds = Q_{\infty}, \quad (9.3.8)$$

for any t > 0, from (9.3.5) and (9.3.7) we get (9.3.3).

Remark 9.3.2 As it is shown in [43, Section 11.2.3], (9.3.1) is also a necessary condition to guarantee the existence of an invariant measure of $\{T(t)\}$.

In view of Remark 9.3.2, throughout the rest of this section we assume that the condition (9.3.1) is satisfied.

According to Proposition 8.1.8, the semigroup $\{T(t)\}$ can be extended to a strongly continuous semigroup in $L^p_{\mu}(\mathbb{R}^N)$. As in Chapter 8, we simply write L^p_{μ} for $L^p(\mathbb{R}^N,\mu)$ and denote by $\|\cdot\|_p$ its norm. Moreover, we denote $W^{k,p}_{\mu}$ the Sobolev space of the functions in L^p_{μ} with weak derivatives up to k-th-order in L^p_{μ} . Finally we denote by $L_p: D(L_p) \subset L^p_{\mu} \to L^p_{\mu}$ $(p \in [1,+\infty))$ the infinitesimal generator of $\{T(t)\}$ in L^p_{μ} and we simply write L for L_2 .

To prove the main result of this first part of the section (i.e., the analyticity of $\{T(t)\}$ in L^p_μ for any $p \in (1, +\infty)$), we need some preliminary results.

Lemma 9.3.3 For any $p \in [1, +\infty)$, $C_c^{\infty}(\mathbb{R}^N)$ is dense in L_{μ}^p and in $W_{\mu}^{k,p}$ for any $k \in \mathbb{N}$.

Proof. The proof can be obtained using the same truncation argument as in the proof of Proposition 8.7.3.

Lemma 9.3.4 Let $p \in [1, +\infty)$. Then, for any $u \in W^{1,p}_{\mu}$ and any $j = 1, \ldots, N$, the function $x \mapsto x_j u(x)$ belongs to L^p_{μ} . Moreover, there exists a positive constant C_j such that

$$\left(\int_{\mathbb{R}^N} |x_j|^p |u(x)|^p d\mu\right)^{\frac{1}{p}} \le C_j \left(\int_{\mathbb{R}^N} (|u(x)|^p + |Du(x)|^p) d\mu\right)^{\frac{1}{p}}.$$

Proof. Without loss of generality we can limit ourselves to proving the assertion in the case when j=1. Moreover, we can assume that $Q_{\infty}=\operatorname{diag}(\lambda_1,\ldots,\lambda_n)$. Indeed, we can always reduce to this situation by means of a suitable linear change of variables as in the proof of Proposition 9.3.1. From the formula (9.1.5) we immediately deduce that

$$\rho(x) = \frac{1}{\sqrt{(2\pi)^N \lambda_1 \cdot \ldots \cdot \lambda_n}} \exp\left(-\sum_{i=1}^N \frac{x_j^2}{2\lambda_j}\right), \qquad x \in \mathbb{R}^N.$$

Therefore,

$$x_1 \rho(x) = -\lambda_1 D_1 \rho(x), \qquad x \in \mathbb{R}^N. \tag{9.3.9}$$

Let us now fix $u \in C_c^{\infty}(\mathbb{R}^N)$ and assume that p > 1 (the case p = 1 is similar and much easier). Taking (9.3.9) into account and integrating by parts, we deduce the following chain of inequalities

$$\int_{\mathbb{R}^N} |x_1 u(x)|^p d\mu$$

$$\leq -\lambda_1 \int_{\mathbb{R}^N} |x_1 |x_1|^{p-2} |u(x)|^p D_1 \rho(x) dx$$

$$= \lambda_1 p \int_{\mathbb{R}^N} x_1 u(x) |x_1 u(x)|^{p-2} D_1 u(x) d\mu + \lambda_1 (p-1) \int_{\mathbb{R}^N} |x_1|^{p-2} |u(x)|^p d\mu$$

= $I_1 + I_2$. (9.3.10)

Using the Hölder and Young inequalities we immediately see that

$$I_{1} \leq \lambda_{1} p \left(\int_{\mathbb{R}^{N}} |x_{1} u(x)|^{p} d\mu \right)^{\frac{p-1}{p}} \left(\int_{\mathbb{R}^{N}} |D_{1} u|^{p} d\mu \right)^{\frac{1}{p}}$$

$$\leq \frac{1}{4} \int_{\mathbb{R}^{N}} |x_{1} u(x)|^{p} d\mu + K_{1} \int_{\mathbb{R}^{N}} |D_{1} u|^{p} d\mu, \tag{9.3.11}$$

for any $\varepsilon > 0$ and some positive constant $K_1 = K_1(\varepsilon)$.

As far as I_2 is concerned, we observe that, if $p \geq 2$, by the Young inequality, we can determine a positive constant K_2 such that

$$|x_1|^{p-2} \le \frac{1}{4\lambda_1(p-1)}|x_1|^p + K_2, \qquad x_1 \in \mathbb{R}.$$

Therefore,

$$I_2 \le \frac{1}{4} \int_{\mathbb{R}^N} |x_1|^p |u(x)|^p d\mu + K_3 \int_{\mathbb{R}^N} |u(x)|^p d\mu, \tag{9.3.12}$$

for some positive constant K_3 .

If p < 2 we have to apply a different argument to estimate the term I_2 . For notational convenience we split $x \in \mathbb{R}^N$ as $x = (x_1, z)$. Moreover, we denote by $d\mu_1$ and $d\mu_2$ the measures in \mathbb{R} and \mathbb{R}^{N-1} whose densities with respect to the Lebesgue measure are

$$\rho_1(x_1) = \frac{1}{\sqrt{(2\pi)\lambda_1}} \exp\left(-\frac{x_1^2}{2\lambda_1}\right), \qquad x_1 \in \mathbb{R}$$

and

$$\rho_2(z) = \frac{1}{\sqrt{(2\pi)^{N-1}\lambda_2 \cdot \ldots \cdot \lambda_N}} \exp\left(-\sum_{j=1}^{N-1} \frac{z_j^2}{2\lambda_{j+1}}\right), \qquad z \in \mathbb{R}^{N-1},$$

respectively. Applying the Fubini-Tonelli theorem we deduce that

$$\begin{split} I_2 &\leq 2\lambda_1 \int_{\mathbb{R}^{N-1}} d\mu_2 \int_{\mathbb{R}} |x_1|^{p-2} |u(x_1, z)|^p d\mu_1 \\ &= 2\lambda_1 \int_{\mathbb{R}^{N-1}} d\mu_2 \int_{|x_1| \geq \sqrt{8\lambda_1}} |x_1|^{p-2} |u(x_1, z)|^p d\mu_1 \\ &+ 2\lambda_1 \int_{\mathbb{R}^{N-1}} d\mu_2 \int_{-\sqrt{8\lambda_1}}^{\sqrt{8\lambda_1}} |x_1|^{p-2} |u(x_1, z)|^p d\mu_1 \end{split}$$

$$\leq \frac{1}{4} \int_{\mathbb{R}^{N-1}} d\mu_2 \int_{\mathbb{R}} |x_1|^p |u(x_1, z)|^p d\mu_1
+ \int_{\mathbb{R}^{N-1}} d\mu_2 \int_{-\sqrt{8\lambda_1}}^{\sqrt{8\lambda_1}} |x_1|^{p-2} |u(x_1, z)|^p d\mu_1.$$
(9.3.13)

Let us estimate the second integral in the last side of (9.3.13). For this purpose we observe that since $W^{1,p}(-\sqrt{8\lambda_1},\sqrt{8\lambda_1})$ is continuously embedded in $L^{\infty}(-\sqrt{8\lambda_1},\sqrt{8\lambda_1})$ (see [2, Theorem 5.4]), then

$$\begin{split} & \int_{-\sqrt{8\lambda_{1}}}^{\sqrt{8\lambda_{1}}} |x_{1}|^{p-2} |u(x_{1},z)|^{p} d\mu_{1} \\ & \leq \left(\sup_{|x_{1}| \leq \sqrt{8\lambda_{1}}} |u(x_{1},z)| \right)^{p} \int_{-\sqrt{8\lambda_{1}}}^{\sqrt{8\lambda_{1}}} |x_{1}|^{p-2} dx \\ & \leq K_{4} \int_{-\sqrt{8\lambda_{1}}}^{\sqrt{8\lambda_{1}}} (|u(x_{1},z)|^{p} + |D_{1}u(x_{1},z)|^{p}) dx_{1} \\ & \leq K_{4} \sup_{|x_{1}| \leq \sqrt{8\lambda_{1}}} |\rho_{1}(x_{1})|^{-1} \int_{-\sqrt{8\lambda_{1}}}^{\sqrt{8\lambda_{1}}} (|u(x_{1},z)|^{p} + |D_{1}u(x_{1},z)|^{p}) d\mu_{1}, \end{split}$$

for some positive constant K_4 . Integrating in \mathbb{R}^{N-1} with respect to the measure μ_2 gives

$$\int_{\mathbb{R}^{N-1}} d\mu_2 \int_{-\sqrt{8\lambda_1}}^{\sqrt{8\lambda_1}} |x_1|^{p-2} |u(x_1, z)|^p d\mu_1 \le K_5 ||u||_{1, p}^p,$$

for some constant $K_5 > 0$. Therefore,

$$I_2 \le \frac{1}{4} \int_{\mathbb{R}^{N-1}} d\mu_2 \int_{\mathbb{R}} |x_1|^p |u(x_1, z)|^p d\mu_1 + K_5 ||u||_{1, p}^p. \tag{9.3.14}$$

Summing up, from (9.3.10)-(9.3.12) and (9.3.14), we deduce that there exists a positive constant K_6 , independent of u, such that

$$\int_{\mathbb{R}^N} |x_1 u(x)|^p d\mu \le K_6 ||u||_{1,p}^p. \tag{9.3.15}$$

Since by Lemma 9.3.3 $C_c^{\infty}(\mathbb{R}^N)$ is dense in $W_{\mu}^{1,p}$, the estimate (9.3.15) yields the assertion.

The following corollary is now a straightforward consequence of Lemma 9.3.4.

Corollary 9.3.5 The realization of the Ornstein-Uhlenbeck operator A in L^p_μ with domain $W^{2,p}_\mu$ is a bounded linear operator.

Lemma 9.3.6 For any $f \in L^p_{\mu}$ $(p \in [1, +\infty))$ and any t > 0, T(t)f is given by the formula (9.1.5).

Proof. In order to avoid misunderstanding, throughout the proof we denote by $\{T(t)\}$ the Ornstein-Uhlenbeck semigroup in $C_b(\mathbb{R}^N)$ and by $\{T_p(t)\}$ its extension to L_p^p .

To begin with, let us observe that since, for any t>0, Q_t and Q_{∞} differ in a (strictly) positive definite matrix, then $Q_t<Q_{\infty}$ in the sense of the positive matrices (i.e., $\langle Q_t h, h \rangle < \langle Q_{\infty} h, h \rangle$ for any $h \in \mathbb{R}^N$). Hence, $Q_t Q_{\infty}^{-1} < Q_{\infty} Q_{\infty}^{-1} = I$ or, equivalently, $Q_{\infty}^{-1} = Q_t^{-1} Q_t Q_{\infty}^{-1} < Q_t^{-1}$. As a consequence,

$$\rho_t(x) := \frac{1}{\sqrt{(2\pi)^N \det Q_t}} e^{-\frac{1}{2}\langle Q_t^{-1} x, x \rangle} \le C_t \rho(x), \qquad x \in \mathbb{R}^N,$$

for some positive constant C_t , and this implies that $L^1_{\mu} \subset L^1_{\mu_t}$ where $\mu_t(dx) = \rho_t(x)dx$. Therefore, if $\{f_n\}$ is a sequence of smooth functions converging to f in L^p_{μ} as n tends to $+\infty$, then, for any t > 0 and any $x \in \mathbb{R}^N$, $f_n(e^{tB}x + \cdot)$ converges to $f(e^{tB}x + \cdot)$ in L^1_{μ} and, hence,

$$\lim_{n \to +\infty} (T(t)f_n)(x) = \lim_{n \to +\infty} \int_{\mathbb{R}^N} f_n(e^{tB}x + y)\mu_t(dy) = \int_{\mathbb{R}^N} f(e^{tB}x + y)\mu_t(dy).$$

Now, the assertion follows if we recall that, by definition, $T(t)f_n$ converges to $T_p(t)f$ in L^p_u (see Proposition 8.1.8).

Remark 9.3.7 Note that the results in Corollary 9.3.5 and Lemma 9.3.6 hold true also in the case when the matrix Q is singular.

Arguing as in the proof of Proposition 9.2.2, we can prove the following estimates for the derivatives of the function T(t)f in the L^p_μ -norm.

Proposition 9.3.8 For any $p \in (1, +\infty)$, any $k \in \mathbb{N}$, any t > 0 and any $f \in L^p_{\mu}$ the function T(t)f belongs to $W^{k,p}_{\mu}$. Moreover, for any multi-index α and any $\varepsilon > 0$, there exists a positive constant $C = C(\varepsilon, |\alpha|, p)$ such that

$$||D^{\alpha}T(t)f||_{p} \le Ce^{|\alpha|(s(B)+\varepsilon)t}t^{-\frac{|\alpha|}{2}}||f||_{p}, \qquad t > 0,$$
 (9.3.16)

for any $p \in (1, +\infty)$, and

$$||D^{\alpha}T(t)f||_{p} \le Ce^{|\alpha|(s(B)+\varepsilon)t} \sum_{|\beta|=|\alpha|} ||D^{\beta}f||_{p}, \qquad t > 0,$$
 (9.3.17)

for any $p \in [1, +\infty)$. Further, the function $t \mapsto D^{\alpha}T(t)f$ is continuous in $(0, +\infty)$ with values in L^p_{μ} for any $f \in L^p_{\mu}$ $(p \in (1, +\infty))$.

Proof. Since $C_c^{\infty}(\mathbb{R}^N)$ is dense in $W_{\mu}^{k,p}$ (see Lemma 9.3.3), it suffices to prove the assertion in the case when $f \in C_c^{\infty}(\mathbb{R}^N)$.

To prove the estimate (9.3.16) it suffices to argue by induction on the length of $|\alpha|$ as in the proof of Proposition 9.2.2. So, we can limit ourselves to checking (9.3.16) in the case when $|\alpha| = 1$. For this purpose, we fix $f \in C_c^{\infty}(\mathbb{R}^N)$, $i \in \{1, ..., N\}$, $p \in (1, +\infty)$ and observe that

$$(D_i T(t) f)(x) = -\frac{1}{\sqrt{(2\pi)^N \det Q_t}} \int_{\mathbb{R}^N} (e^{tB^*} Q_t^{-1} y)_i e^{-\frac{1}{2} \langle Q_t^{-1} y, y \rangle} f(e^{tB} x - y) dy,$$

for any t > 0 and any $x \in \mathbb{R}^N$. Since

$$|(e^{tB^*}Q_t^{-1}y)_i| = |\langle Q_t^{-1/2}y, Q_t^{-1/2}e^{tB}e_i\rangle| \leq |Q_t^{-1/2}y||(Q_t^{-1/2}e^{tB})_i|,$$

for any t > 0, any $y \in \mathbb{R}^N$ and any i = 1, ..., N, using the Hölder inequality and taking (9.2.3) into account, we easily see that

$$|(D_{i}T(t)f)(x)|^{p} \leq \left(\int_{\mathbb{R}^{N}} |(e^{tB^{*}}Q_{t}^{-1}y)_{i}|^{\frac{p}{p-1}} e^{-\frac{1}{2}\langle Q_{t}^{-1}y,y\rangle} dy\right)^{p-1} (T(t)(|f|^{p}))(x)$$

$$\leq \|(Q_{t}^{-1/2}e^{tB})_{i}\|^{\frac{p}{p-1}} (T(t)(|f|^{p}))(x)$$

$$\times \left(\frac{1}{\sqrt{(2\pi)^{N}\det Q_{t}}} \int_{\mathbb{R}^{N}} |Q_{t}^{-1/2}y|^{\frac{p}{p-1}} e^{-\frac{1}{2}\langle Q_{t}^{-1}y,y\rangle} dy\right)^{p-1}$$

$$\leq C_{p} t^{-\frac{1}{2}} e^{(s(B)+\varepsilon)t} (T(t)(|f|^{p}))(x), \tag{9.3.18}$$

for any t > 0, any $x \in \mathbb{R}^N$ and some positive constant C_p . Hence integrating the first and the last term in (9.3.18) in \mathbb{R}^N , with respect to the measure μ , we get (9.3.16).

The estimate (9.3.17) is easier to be proved: indeed, it suffices to iterate the formula (9.2.1).

Finally, let us prove that for any multi-index α , the function $D^{\alpha}T(\cdot)f$ is continuous in $(0, +\infty)$ with values in L^p_{μ} . To see it, we observe that the estimate (9.3.16) implies that

$$\|D^{\alpha}T(t)f - D^{\alpha}T(s)f\|_{p} \le Ce^{|\alpha|(s(B)+\varepsilon)t}s^{-\frac{|\alpha|}{2}}\|T(t-s)f - f\|_{p},$$

for any 0 < s < t and any $p \in (1, +\infty)$. Now, the strong continuity of $\{T(t)\}$ in L^p_μ yields the assertion.

Now, we can show that the Ornstein-Uhlenbeck semigroup is analytic in L^p_μ for any $p \in (1, +\infty)$. Such a result has been proved in [39] and in [61, 67] also in infinite dimensional settings. Here, we present a simplified proof taken from [106].

Theorem 9.3.9 For any $p \in (1, +\infty)$, the Ornstein-Uhlenbeck semigroup is analytic in L^p_μ .

Proof. Fix $p \in (1, +\infty)$. According to Theorem B.2.8 to prove the assertion we can limit ourselves to showing that the function $T(\cdot)f$ is differentiable in $(0, +\infty)$, for any $f \in L^p_\mu$, and there exists a positive constant C, independent of f, such that

$$||tD_tT(t)f||_p \le C||f||_p, \qquad t \in (0,1).$$
 (9.3.19)

Suppose first that $f \in C_c^{\infty}(\mathbb{R}^N)$. In such a case, taking Propositions 2.3.5, 2.3.6 and 4.1.10 into account, it is easy to check that the function $T(\cdot)f$ is differentiable in $[0, +\infty)$ with values in L^p_{μ} and $D_t T(t) f = \mathcal{A}T(t) f$ for any $t \geq 0$. Moreover, taking Corollary 9.3.5 and the estimate (9.3.16) into account, it is immediate to check that

$$t\|\mathcal{A}T(t)f\|_p \le C_1\|f\|_p, \qquad t > 0,$$
 (9.3.20)

for some positive constant C_1 , independent of f, and (9.3.19) follows.

Now, suppose that $f \in L^p_\mu$ and let $\{f_n\} \in C^\infty_c(\mathbb{R}^N)$ be a sequence of smooth functions converging to f in L^p_μ (see Lemma 9.3.3). Since

$$\frac{T(t)f_n - T(t_0)f_n}{t - t_0} - \mathcal{A}T(t_0)f_n = \frac{1}{t - t_0} \int_{t_0}^t (\mathcal{A}T(s)f_n - \mathcal{A}T(t_0)f_n)ds, \ (9.3.21)$$

for any $t_0, t \in (0, +\infty)$ and any $n \in \mathbb{N}$ (where the integral is meant in L^p_μ), letting n go to $+\infty$, and observing that (9.3.20) is satisfied by f_n ($n \in \mathbb{N}$) and f, with the same constant C_1 , we easily see that (9.3.21) holds true also with f instead of f_n . Now using the continuity of the function $t \mapsto \mathcal{A}T(t)f$ in $(0, +\infty)$ with values in L^p_μ (see Corollary 9.3.5 and Proposition 9.3.8), we can conclude that $T(\cdot)f$ is differentiable in $(0, +\infty)$ with values in L^p_μ and $D_tT(t)f = \mathcal{A}T(t)f$ for any t > 0. Hence, the estimate (9.3.19) follows and we are done.

The next proposition characterizes the symmetric Ornstein-Uhlenbeck semigroups, i.e., the case when T(t) coincides with its adjoint operator for any t>0. The characterization of the symmetric Ornstein-Uhlenbeck semigroups has been studied in [32] and [33] also in the infinite dimensional setting.

Proposition 9.3.10 The Ornstein-Uhlenbeck semigroup is symmetric in L^2_{μ} if and only if $BQ = QB^*$. In such a case L is self-adjoint. Further, if $B = -Q^{-1}_{\infty}$, then

$$\int_{\mathbb{R}^N} Lf \, g d\mu = -\int_{\mathbb{R}^N} \langle QDf, Dg \rangle d\mu, \qquad (9.3.22)$$

for any $f, g \in D(L)$.

Proof. As a first step we show that, for any t > 0, the adjoint operator of T(t) is given by

$$(T(t)^*f)(x) = \int_{\mathbb{R}^N} f(Q_{\infty}^{1/2} e^{tB_{\infty}^*} Q_{\infty}^{-1/2} x + Q_{\infty}^{1/2} (I - e^{tB_{\infty}^*} e^{tB_{\infty}})^{1/2} Q_{\infty}^{-1/2} y) d\mu,$$
(9.3.23)

for any t > 0 and any $f \in \mathcal{D} := \{g_h^{(j)} : h \in \mathbb{R}^N, j = 1, 2\}$, where $g_h^{(1)}(x) = \cos(\langle h, x \rangle)$ and $g_h^{(2)}(x) = \sin(\langle h, x \rangle)$ for any x and any h in \mathbb{R}^N . Here, $B_{\infty} = Q_{\infty}^{-1/2}BQ_{\infty}^{1/2}$. Note that the matrix $(I - e^{tB_{\infty}^*}e^{tB_{\infty}})^{1/2}$ is well defined. Indeed, using the formula (9.3.8), it is easy to check that

$$Q_t = Q_{\infty}^{1/2} (I - e^{tB_{\infty}} e^{tB_{\infty}^*}) Q_{\infty}^{1/2}, \qquad t > 0.$$
 (9.3.24)

As a consequence, the matrix $I - e^{tB_{\infty}} e^{tB_{\infty}^*}$ is strictly definite positive and this implies that $\|e^{tB_{\infty}}\|_{L(L_{\mu}^2)} = \|e^{tB_{\infty}^*}\|_{L(L_{\mu}^2)} < 1$ for any t > 0. Therefore, $I - e^{tB_{\infty}^*} e^{tB_{\infty}}$ is a strictly positive definite matrix.

Let us fix t > 0 and $g \in \mathcal{D}$, and denote by R(t) the operator defined by the right-hand side of (9.3.23). Since \mathcal{D} is dense in L^2_{μ} (see the proof of Proposition (9.3.1)), to prove that $T(t)^*g = R(t)g$ it suffices to show that

$$\int_{\mathbb{R}^N} gT(t)fd\mu = \int_{\mathbb{R}^N} fR(t)gd\mu, \qquad (9.3.25)$$

for any $f \in \mathcal{D}$. For this purpose, we begin by observing that, from the formula (9.3.6), we get

$$T(t)g_h^{(j)} = e^{-\frac{1}{2}\langle Q_t h, h \rangle} g_{e^{tB^*}h}^{(j)}, \qquad j = 1, 2.$$
 (9.3.26)

Similarly, setting $P_{1,t} = Q_{\infty}^{1/2} (I - e^{tB_{\infty}^*} e^{tB_{\infty}})^{1/2} Q_{\infty}^{-1/2}$, $P_{2,t} = Q_{\infty}^{1/2} e^{tB_{\infty}^*} Q_{\infty}^{-1/2}$, we obtain that

$$(R(t)g_h^{(j)})(x) = \int_{\mathbb{R}^N} g_h^{(j)}(P_{2,t}x + P_{1,t}y)\mathcal{N}(0, Q_\infty)(dy)$$

$$= \int_{\mathbb{R}^N} g_h^{(j)}(P_{1,t}y)\mathcal{N}(P_{1,t}^{-1}P_{2,t}x, Q_\infty)(dy)$$

$$= \int_{\mathbb{R}^N} g_h^{(j)}(y)\mathcal{N}(P_{2,t}x, P_{1,t}Q_\infty P_{1,t}^*)(dy)$$

$$= e^{-\frac{1}{2}\langle Q_\infty^{1/2}(I - e^{tB_\infty^*}e^{tB_\infty})Q_\infty^{1/2}h,h\rangle} g_{Q_\infty^{-1/2}e^{tB_\infty}Q_\infty^{1/2}h}^{(j)}, \qquad (9.3.27)$$

for any $x \in \mathbb{R}^N$ and j = 1, 2, where we took advantage of (9.3.4) in the last side of (9.3.27), and $\mathcal{N}(a, K)$ is defined in (9.1.2) for any $a \in \mathbb{R}^N$ and any positive definite matrix K.

Since the density of μ (with respect to the Lebesgue measure) is even with respect to each variable, from (9.3.26) and (9.3.27) we easily deduce that we can limit ourselves to checking (9.3.25) in the case when $(f,g)=(g_h^{(j)},g_k^{(j)})$, for j=1,2. This can easily be done, observing that $2g_h^{(j)}g_k^{(j)}=g_{h-k}^{(j)}+(-1)^{j-1}g_{h+k}^{(j)}$ for any $h,k\in\mathbb{R}^N$ and j=1,2. Indeed, we get

$$\begin{split} \int_{\mathbb{R}^{N}} g_{k}^{(j)} T(t) g_{h}^{(j)} d\mu &= \frac{1}{2} e^{-\frac{1}{2} \langle Q_{t} h, h \rangle} \left(e^{-\frac{1}{2} \langle Q_{\infty}(e^{tB^{*}}h - k), e^{tB^{*}}h - k \rangle} \right. \\ &+ (-1)^{j-1} e^{-\frac{1}{2} \langle Q_{\infty}(e^{tB^{*}}h + k), e^{tB^{*}}h + k \rangle} \right), \end{split}$$

whereas

$$\begin{split} \int_{\mathbb{R}^N} g_h^{(j)} R(t) g_k^{(j)} d\mu \\ &= \frac{1}{2} e^{-\frac{1}{2} \langle Q_\infty^{1/2} (I - e^{tB_\infty^*} e^{tB_\infty}) Q_\infty^{1/2} k, k \rangle} \\ &\quad \times \left(e^{\frac{1}{2} \langle Q_\infty (h - Q_\infty^{-1/2} e^{tB_\infty} Q_\infty^{1/2} k), h - Q_\infty^{-1/2} e^{tB_\infty} Q_\infty^{1/2} k \rangle} \right. \\ &\quad \left. + (-1)^{j-1} e^{\frac{1}{2} \langle Q_\infty (h + Q_\infty^{-1/2} e^{tB_\infty} Q_\infty^{1/2} k), h + Q_\infty^{-1/2} e^{tB_\infty} Q_\infty^{1/2} k \rangle} \right), \end{split}$$

from which the equality (9.3.26) immediately follows observing that

$$e^{tB}Q_{\infty}e^{tB^*} + Q_t = Q_{\infty}, \qquad t > 0.$$

Now, the formulas (9.3.26) and (9.3.27) imply that T(t) is symmetric for any t > 0 if and only

(i)
$$Q_{\infty}^{-1/2} e^{tB_{\infty}} Q_{\infty}^{1/2} = e^{tB^*},$$
 (ii) $Q_{\infty}^{1/2} (I - e^{tB_{\infty}^*} e^{tB_{\infty}}) Q_{\infty}^{1/2} = Q_t.$ (9.3.28)

Differentiating both the members of (9.3.28)(i) at t=0, we see that such a condition holds if and only if $B_{\infty}Q_{\infty}^{1/2}=Q_{\infty}^{1/2}B^*$ or, equivalently, if and only if B_{∞} is symmetric. Taking (9.3.24) into account, the condition (9.3.28)(i) is now an immediate consequence of (9.3.28)(i). Summing up, we have shown that T(t) is symmetric in L_{μ}^2 for any t>0 if and only if the matrix B_{∞} is symmetric or, equivalently, if and only if $BQ_{\infty}=Q_{\infty}B^*$. In such a case, according to [123, Chapter 1, Corollary 10.6], the operator L is self-adjoint in L_{μ}^2 .

To conclude the first part of the proof, let us show that $BQ_{\infty} = Q_{\infty}B^*$ if and only if $BQ = QB^*$. Showing that $BQ = QB^*$ implies that $BQ_{\infty} = Q_{\infty}B^*$ follows immediately from (9.0.2). Conversely, let us assume that $BQ_{\infty} = Q_{\infty}B^*$. Using the formula (9.3.8) it is immediate to check that

$$Q_t B^* = BQ_t, t > 0.$$
 (9.3.29)

Differentiating (9.3.29) at t = 0 gives immediately $BQ = QB^*$.

We now assume that $B=-Q_{\infty}^{-1}$ and we prove the formula (9.3.22). For this purpose, since

$$D_i \rho(x) = -\frac{1}{2} \rho(x) D_i(\langle Q_{\infty}^{-1} x, x \rangle) = -\rho(x) (Q_{\infty}^{-1} x)_i, \qquad i = 1, \dots, N,$$

an integration by parts gives

$$\int_{\mathbb{R}^N} \langle QDf(x), Dg(x) \rangle d\mu$$

$$= \sum_{i,j=1}^N \int_{\mathbb{R}^N} q_{ij} D_i f(x) D_j g(x) \rho(x) dx$$

$$= -\sum_{i,j=1}^N \int_{\mathbb{R}^N} q_{ij} g(x) (D_{ij} f(x) \rho(x) + D_i f(x) D_j \rho(x)) dx$$

$$= -\sum_{i,j=1}^{N} \int_{\mathbb{R}^{N}} q_{ij}g(x) \left(D_{ij}f(x) - D_{i}f(x)(Q_{\infty}^{-1}x)_{j} \right) \rho(x)dx$$
$$= -\int_{\mathbb{R}^{N}} g(x) \left(\operatorname{Tr}\left(QD^{2}f(x) \right) - \langle QDf(x), Q_{\infty}^{-1}x \rangle \right) d\mu,$$

for any $f, g \in C_c^{\infty}(\mathbb{R}^N)$, that is (9.3.22). Since $C_c^{\infty}(\mathbb{R}^N)$ is a core of L (see the forthcoming Lemma 9.3.13), the formula (9.3.22) follows for any $f, g \in D(L)$.

To conclude the first part of this section, let us show that the Ornstein-Uhlenbeck semigroup satisfies the Poincaré inequality.

Theorem 9.3.11 The Ornstein-Uhlenbeck semigroup satisfies the Poincaré inequality (8.6.1) with p=2 and $C_{\varepsilon}=-(s(B)+\varepsilon)^{-1}$ for any $\varepsilon<-s(B)$.

Proof. It can be obtained repeating step by step the proof of Theorem 8.6.3, replacing everywhere the constant d_0 defined in (8.3.6) with $s(B) + \varepsilon$, where ε is as in the assertion of the theorem.

Remark 9.3.12 Note that our assumptions on the matrix B do not imply, in general, that the constant d_0 in (8.3.6) is negative. This is the reason why we cannot apply immediately Theorem 8.6.3.

Suppose, for instance, that

$$B = \begin{pmatrix} -1 & 1 \\ -6 & -4 \end{pmatrix}.$$

Then, B has two eigenvalues with negative real parts. Moreover,

$$\langle B\xi, \xi \rangle = -\xi_1^2 - 5\xi_1\xi_2 - 4\xi_2^2, \qquad \xi = (\xi_1, \xi_2) \in \mathbb{R}^2,$$

which is not strictly less than 0 in \mathbb{R}^2 .

9.3.1 The domain of the realization of $\{T(t)\}$ in $L^p_\mu(\mathbb{R}^N)$

In this subsection we give a complete characterization of the domain of L_p for any $p \in [1, +\infty)$. The results that we present are taken from [118]. We notice that in the case when p=2 the characterization of the domain of L_2 has been proved in [106] and in [36] (also in the infinite dimensional case) and, then, it has been generalized to a more general context first in [41] and, then, in [40]. Here, we consider the case when Q is not singular.

The following lemma will be very useful to characterize the domain of L_p .

Lemma 9.3.13 For any $p \in [1, +\infty)$, $C_c^{\infty}(\mathbb{R}^N)$ is a core of L_p .

Proof. We first prove that the Schwarz space \mathcal{S} , of all the smooth functions $f: \mathbb{R}^N \to \mathbb{R}$ such that $\lim_{|x| \to +\infty} |x|^k D^{\alpha} f(x) = 0$ for any $k \in \mathbb{N}$ and any $\alpha \in \mathbb{N}^N$, is a core of L_p . For this purpose, let us notice that the same arguments as in the first part of the proof of Theorem 9.3.9 show that \mathcal{S} is contained in $D(L_p)$ for any $p \in [1, +\infty)$. Moreover, a straightforward computation shows that T(t) maps \mathcal{S} into itself for any t > 0. Therefore, according to Lemma 9.3.3 and Proposition B.1.10, \mathcal{S} is a core of $\{T(t)\}$.

Now to prove that $C_c^{\infty}(\mathbb{R}^N)$ is a core of L_p , it suffices to show that, for any $f \in \mathcal{S}$, there exists a sequence $\{f_n\} \subset C_c^{\infty}(\mathbb{R}^N)$ such that f_n and $\mathcal{A}f_n$ converge, respectively, to f and $\mathcal{A}f$ in L_{μ}^p , as n tends to $+\infty$. So, let $f \in \mathcal{S}$ and, for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$, let $\vartheta_n(x) = \vartheta(x/n)$ where $\vartheta \in C_c^{\infty}(\mathbb{R}^N)$ satisfies $\chi_{B(1/2)} \leq \vartheta \leq \chi_{B(1)}$. Then, the function $f_n = f\vartheta_n$ belongs to $C_c^{\infty}(\mathbb{R}^N)$ for any $n \in \mathbb{N}$ and it converges to f in $W_{\mu}^{2,p}$ as n tends to $+\infty$. Moreover, since

$$\langle x, Df_n(x) \rangle = \vartheta_n(x)\langle x, Df(x) \rangle + \frac{1}{n}f(x)\langle x, D\vartheta(n^{-1}x) \rangle, \qquad x \in \mathbb{R}^N,$$

 $\mathcal{A}f_n$ converges to $\mathcal{A}f$ in L^p_μ .

Now, we characterize the domain of L_p for any $p \in (1, +\infty)$, when the Ornstein-Uhlenbeck operator \mathcal{A} is given by

$$(\mathcal{A}u)(x) = \frac{1}{2}\Delta u(x) - \sum_{i=1}^{N} \frac{x_i}{2\lambda_i} D_i u(x), \qquad x \in \mathbb{R}^N,$$
(9.3.30)

where λ_i $(i=1,\ldots,N)$ are suitable positive constants. This a crucial step in order to characterize the domain of the more general Ornstein-Uhlenbeck operator. In the case when \mathcal{A} is given by (9.3.30), $\{T(t)\}$ is symmetric (see Proposition 9.3.10) and it is given by

$$(T(t)f)(x) = \frac{1}{\sqrt{(2\pi)^N \prod_{i=1}^N \lambda_i (1 - e^{-\frac{t}{\lambda_i}})}} \times \int_{\mathbb{R}^N} \exp\left(-\sum_{i=1}^N \frac{(e^{-\frac{t}{2\lambda_i}} x_i - y_i)^2}{2\lambda_i (1 - e^{-\frac{t}{\lambda_i}})}\right) f(y) dy, \quad (9.3.31)$$

for any t>0, any $x\in\mathbb{R}^N$ and any $f\in L^p_\mu$. Moreover, the density ρ of the invariant measure of the semigroup $\{T(t)\}$, with respect to the Lebesgue measure, is given by

$$\rho(x) = \frac{1}{\sqrt{(2\pi)^N \lambda_1 \cdot \ldots \cdot \lambda_N}} \exp\left(-\sum_{i=1}^N \frac{x_i^2}{2\lambda_i}\right), \qquad x \in \mathbb{R}^N.$$

Proposition 9.3.14 Suppose that the operator A is given by (9.3.30). Then, for any $p \in (1, +\infty)$, $D(L_p) = W_{\mu}^{2,p}$.

Proof. Since the proof is rather long, we divide it into three steps.

Step 1. Let us introduce the semigroup $\{T_i(t)\}\ (i=1,\ldots,N)$ defined in $L^p_\mu(\mathbb{R}^N)\ (p\in(1,+\infty)$ by

$$(T_i(t)f)(x) = \frac{1}{\sqrt{2\pi\lambda_i(1 - e^{-\frac{t}{\lambda_i}})}} \int_{\mathbb{R}} \exp\left(-\frac{y^2}{2\lambda_i(1 - e^{-\frac{t}{\lambda_i}})}\right) f(E_i(t, x, y)) dy,$$
(9.3.32)

for any t>0 and any $x\in\mathbb{R}^N$, where $E_i(t,x,y)\in\mathbb{R}^N$ is defined by

$$\langle E_i(t,x,y),e_j\rangle = \begin{cases} x_j, & \text{if } i \neq j; \\ e^{-t/(2\lambda_i)}x_i - y, & \text{if } i = j. \end{cases}$$

Arguing as in the proof of Proposition 8.1.8 and Theorem 9.3.9, it can be shown that $\{T_i(t)\}$ is an analytic strongly continuous semigroup of positive contractions in L^p_μ . We denote by $L_{p,i}$ its infinitesimal generator and we prove that it coincides with the operator $A_{p,i}$ defined by $A_{p,i}u(x) = \frac{1}{2}D_{ii}u(x) - (2\lambda_i)^{-1}x_iD_iu(x)$ for any $u \in D(A_{p,i}) = \{u \in L^p_\mu : D_iu, D_{ii}u \in L^p_\mu\}$. For this purpose, we observe that, arguing as in the proof of Lemma 9.3.13, we can easily show that $C_c^\infty(\mathbb{R}^N)$ is a core of $L_{p,i}$ and that $L_{p,i} = A_{p,i}$ on $C_c^\infty(\mathbb{R}^N)$. Since, as the proof of Lemma 9.3.4 shows, $A_{p,i}$ is a bounded operator in $D(A_{p,i})$ endowed with the norm $\|u\|_{i,p} = \|u\|_p + \|D_iu\|_p + \|D_{ii}u\|_p$, then $L_{p,i}$ is the closure of the operator $A_{p,i}$. Therefore, to prove the assertion it suffices to show that $A_{p,i}$ is closed in L^p_μ . This will follow immediately if we prove that there exists a positive constant C_1 such that

$$||D_i u||_p + ||D_{ii} u||_p \le C_1 (||u||_p + ||A_{i,p} u||_p), \qquad u \in C_c^{\infty}(\mathbb{R}^N),$$
 (9.3.33)

for any i = 1, ..., N. Equivalently, we can prove that

$$\int_{\mathbb{R}} |u'|^p d\mu_i + \int_{\mathbb{R}} |u''|^p d\mu_i \le C_2 \int_{\mathbb{R}} (|u|^p + |\tilde{L}_i u|^p) d\mu_i, \qquad i = 1, \dots, N,$$
(9.3.34)

for any $u \in C_c^{\infty}(\mathbb{R})$ and some positive constant C_2 , where

$$(\tilde{L}_i u)(x) = \frac{1}{2} u''(x) - \frac{1}{2\lambda_i} x u'(x), \qquad x \in \mathbb{R},$$

and $d\mu_i(x) = (\sqrt{2\pi\lambda_i})^{-1} \exp(-x^2/(2\lambda_i)) dx$.

Let us observe that we can limit ourselves to showing that there exists $C_3 > 0$ such that

$$\int_{\mathbb{R}} |xu'(x)|^p d\mu_i \le C_3 \int_{\mathbb{R}} (|u|^p + |\tilde{L}_i u|^p) d\mu_i, \qquad u \in C_c^{\infty}(\mathbb{R}).$$
 (9.3.35)

Indeed, once (9.3.35) is checked, it will be immediate to show that the $L_{\mu_i}^p$ -norm of u'' satisfies (9.3.34). Concerning the $L_{\mu_i}^p$ -norm of u', we observe that

$$\frac{1}{\sqrt{2\pi\lambda_{i}}}|u'(x)|^{p}e^{-\frac{x^{2}}{2\lambda_{i}}} = \frac{1}{\sqrt{2\pi\lambda_{i}}}\int_{-\infty}^{x} \frac{d}{dt}\left(|u'(t)|^{p}e^{-\frac{t^{2}}{2\lambda_{i}}}\right)dt$$

$$\leq p\int_{\mathbb{R}}|u'(x)|^{p-1}|u''(x)|d\mu_{i} + \frac{1}{\lambda_{i}}\int_{\mathbb{R}}|x||u'(x)|^{p}d\mu_{i}$$

$$\leq p\left(\int_{\mathbb{R}}|u'(x)|^{p}d\mu_{i}\right)^{\frac{p-1}{p}}\left(\int_{\mathbb{R}}|u''(x)|^{p}d\mu_{i}\right)^{\frac{1}{p}}$$

$$+\frac{1}{4}\int_{\mathbb{R}}|u'(x)|^{p}d\mu_{i} + C_{4}\int_{\mathbb{R}}|x|^{p}|u'(x)|^{p}d\mu_{i}$$

$$\leq \frac{1}{2}\int_{\mathbb{R}}|u'(x)|^{p}d\mu_{i} + C_{4}\int_{\mathbb{R}}|x|^{p}|u'(x)|^{p}d\mu_{i}$$

$$+C_{5}\int_{\mathbb{R}}|u''(x)|^{p}d\mu_{i}, \qquad (9.3.36)$$

for any $x \in \mathbb{R}$ and some positive constants C_4 and C_5 . Here, in the last two inequalities we took advantage of the Young inequality. Integrating (9.3.36) in \mathbb{R} with respect to the Lebesgue measure gives

$$\int_{\mathbb{R}} |u'(x)|^p d\mu_i \le 2C_4 \int_{\mathbb{R}} |x|^p |u'(x)|^p d\mu_i + 2C_5 \int_{\mathbb{R}} |u''(x)|^p d\mu_i$$

and the estimate for u' in (9.3.34) follows.

So, let us prove (9.3.35). For this purpose, we fix $u \in C_c^{\infty}(\mathbb{R})$, we multiply both the sides of the equation $u - \tilde{L}_1 u =: f$ by $e^{-x^2/2\lambda_i}$ and, then, we integrate in $(-\infty, x)$ and in $(x, +\infty)$, with respect to the measure $d\mu_i$. We get

$$u'(x) = 2 \int_{x}^{\pm \infty} (f(y) - u(y)) e^{\frac{x^2 - y^2}{2\lambda_i}} dy, \qquad x \in \mathbb{R}^N.$$

Therefore,

$$u'(x)e^{-\frac{x^2}{2p\lambda_i}} = 2\int_x^{\pm\infty} h(y)e^{\frac{x^2-y^2}{2p'\lambda_i}}dy, \qquad x \in \mathbb{R}^N,$$

where 1/p + 1/p' = 1 and we have set $h(y) = e^{-y^2/(2p\lambda_i)}(f(y) - u(y))$ for any $y \in \mathbb{R}$. Consequently,

$$\int_{\mathbb{R}} |xu'(x)|^p d\mu_i
= 2 \int_0^{+\infty} \left| x \int_x^{+\infty} h(y) e^{\frac{x^2 - y^2}{2p'\lambda_i}} dy \right|^p d\mu_i + 2 \int_{-\infty}^0 \left| x \int_{-\infty}^x h(y) e^{\frac{x^2 - y^2}{2p'\lambda_i}} dy \right|^p d\mu_i
\le 4 \int_0^{+\infty} \left| x \int_x^{+\infty} (|h(y)| + |h(-y)|) e^{\frac{x^2 - y^2}{2p'\lambda_i}} dy \right|^p d\mu_i.$$

Now, observing that $e^{(x^2-y^2)/(2p'\lambda_i)} \le e^{x(x-y)/(p'\lambda_i)}$ for any $0 \le x \le y$, and using the Hölder inequality and then the Fubini-Tonelli theorem, we deduce that

$$\int_{\mathbb{R}} |xu'(x)|^p d\mu_i \le 4 \int_0^{+\infty} dx \left(x \int_x^{+\infty} (|h(y)| + |h(-y)|)^p e^{\frac{x(x-y)}{p'\lambda_i}} dy \right)
\times \left(\int_x^{+\infty} x e^{\frac{x(x-y)}{p'\lambda_i}} dy \right)^{p-1}
= 4(p'\lambda_i)^{p-1} \int_0^{+\infty} (|h(y)| + |h(-y)|)^p dy \int_0^y x e^{\frac{x(x-y)}{p'\lambda_i}} dx.$$
(9.3.37)

Since

$$\begin{split} \int_0^y x e^{\frac{x(x-y)}{p'\lambda_i}} dx &\leq y^2 \int_0^1 e^{-\frac{y^2 t(1-t)}{p'\lambda_i}} dt \\ &= y^2 \int_0^{\frac{1}{2}} e^{-\frac{y^2 t(1-t)}{p'\lambda_i}} dt + y^2 \int_{\frac{1}{2}}^1 e^{-\frac{y^2 t(1-t)}{p'\lambda_i}} dt \\ &= 2y^2 \int_0^{\frac{1}{2}} e^{-\frac{y^2 t(1-t)}{p'\lambda_i}} dt \\ &\leq 2y^2 \int_0^{\frac{1}{2}} e^{-\frac{y^2 t}{2p'\lambda_i}} dt \leq 4p'\lambda_i, \end{split}$$

from (9.3.37) we easily deduce that

$$\int_{\mathbb{R}} |xu'(x)|^p d\mu_i \le 2^{p+3} (p'\lambda_i)^p ||h||_{L^p(\mathbb{R})}$$

$$\le 2^{p+3} (p'\lambda_i)^p (||u||_p + ||\tilde{L}_i u||_p),$$

and (9.3.35) follows.

Step 2. We now prove that

$$D(L_p) = \{ u \in L^p_\mu : D_i u, \ D_{ii} u \in L^p_\mu, \ i = 1, \dots, N \}.$$
 (9.3.38)

By virtue of Step 1, we can assume that N>1. A straightforward computation shows that

$$(T(t)f = (T_1(t) \circ \ldots \circ T_N(t)) f, \qquad t > 0, \ f \in L^p_u.$$

Since for any $i, j \in \{1, ..., N\}$ the operators $L_{p,i}$ and $L_{p,j}$ commute in the resolvent sense, then a general theorem for commuting analytic semigroups implies that L_p is the closure of the operator $\sum_{i=1}^{N} L_{p,i}$ with domain given by

the right-hand side of (9.3.38) (see [38, Theorem 3.3]). Hence, to prove (9.3.38) it suffices to show that the operator $\sum_{i=1}^{N} L_{p,i}$ is closed. For this purpose, we use Theorem B.1.15. According to Proposition B.1.13(iii), the operators $I - L_{i,p}$ (i = 1, ..., N) admit bounded imaginary powers in L_{μ}^{p} with power angles $\theta_{i,p} := \theta_{I-L_{i,p}}$ less or equal to $\pi/2$. Now, since $I - L_{i,2}$ is self-adjoint in L_{μ}^{2} (see Proposition 9.3.10), from Proposition B.1.13(v) we deduce that $\theta_{i,2} = 0$, and the Riesz-Thorin interpolation theorem (see Theorem A.4.9) implies that $\theta_{i,p} < \pi/2$ for any $p \in (1, +\infty)$. Hence, Theorem B.1.15 applies and yields the closedness of the sum $\sum_{i=1}^{N} (I - L_{i,p})$ (equivalently, the closedness of the operator $\sum_{i=1}^{N} L_{i,p}$) in $\{u \in L_{\mu}^{p} : D_{i}u, D_{ii}u \in L_{\mu}^{p}, i = 1, ..., N\}$.

Step 3. To conclude the proof, we need to show that $W^{2,p}_{\mu} = \{u \in L^p_{\mu} : D_i u, \ D_{ii} u \in L^p_{\mu}, \ i = 1, \dots, N\}$. For this purpose, we fix $u \in L^p_{\mu}$ such that $D_i u, D_{ii} u \in L^p_{\mu}$, for any $i = 1, \dots, N$, and introduce the function $v \in L^p(\mathbb{R}^N)$ defined by

$$v(x) = u(x) \exp\left(-\sum_{i=1}^{N} \frac{x_i^2}{2p\lambda_i}\right), \qquad x \in \mathbb{R}^N,$$

and we prove that $\Delta v \in L^p(\mathbb{R}^N)$. Since

$$\begin{split} D_{ii}v(x) &= \left(D_{ii}u(x) - \frac{x_i}{p\lambda_i}D_ju(x) - \frac{1}{p\lambda_i}u(x) + \frac{x_i^2}{(p\lambda_i)^2}u(x)\right) \\ &\times \exp\left(-\sum_{i=1}^N \frac{x_i^2}{2p\lambda_i}\right), \end{split}$$

for any $x \in \mathbb{R}^N$ and any i = 1, ..., N, according to the proof of Lemma 9.3.4, the function $x \mapsto x_i D_i u$ belongs to $L^p_{\mu}(\mathbb{R}^N)$. Repeating the same arguments as in the proof of the quoted lemma, we can easily show that the function $x \mapsto x_i^2 u(x)$ belongs to L^p_{μ} as well. Therefore, $D_{ii}v \in L^p(\mathbb{R}^N)$ and, consequently, $\Delta v \in L^p(\mathbb{R}^N)$. By classical results for elliptic equations (see Theorem C.1.3(iii)), $v \in W^{2,p}(\mathbb{R}^N)$.

As a consequence of the previous results, we also deduce that the functions $x \mapsto x_i x_j v(x)$ and $x \mapsto x_i D_i v(x)$ belong to $L^p(\mathbb{R}^N)$ for any $i, j \in \{1, \dots, N\}$. Since

$$D_{ij}u(x) = \left(D_{ij}v(x) + \frac{x_i}{p\lambda_i}D_jv(x) + \frac{x_j}{p\lambda_j}D_iv(x) + \frac{x_ix_j}{p^2\lambda_i\lambda_j}v(x)\right) \times \exp\left(\sum_{i=1}^N \frac{x_i^2}{2p\lambda_i}\right),$$

for any $x \in \mathbb{R}^N$ and any $i, j \in \{1, ..., N\}$, it follows that $D_{ij}u \in L^p_\mu$.

In order to characterize the domain of the more general Ornstein-Uhlenbeck operator \mathcal{A} , let us prove the following proposition which allows us to transform

the operator \mathcal{A} in a perturbation of the operator defined in (9.3.30). For this purpose, with any nonsingular matrix M we associate the operator Φ_M defined by $(\Phi_M u)(x) = u(Mx)$, for any $x \in \mathbb{R}^N$ and any $u \mathbb{R}^N \to \mathbb{R}$. It is clearly seen that Φ_M is an isometry between $L^p_{\bar{u}}$ and $L^p_{\bar{u}}$ $(p \in [1, +\infty))$, where

$$\tilde{\mu}(dx) = \frac{1}{|\det M| \sqrt{(2\pi)^N \det Q_{\infty}}} e^{-\frac{1}{2}\langle (M^*)^{-1} Q_{\infty}^{-1} M^{-1} x, x \rangle} dx := \tilde{\rho}(x) dx.$$
(9.3.39)

Lemma 9.3.15 There exists a nonsingular matrix $M \in L(\mathbb{R}^N)$ such that the operator $\tilde{\mathcal{A}} = \Phi_{M^{-1}} \mathcal{A} \Phi_M$ is the Ornstein-Uhlenbeck operator defined, on smooth functions, by

$$(\tilde{\mathcal{A}}u)(x) = \mathcal{A}^0 u(x) + \langle B^0 x, Du(x) \rangle, \qquad x \in \mathbb{R}^N,$$
 (9.3.40)

where

$$(\mathcal{A}^0 u)(x) = \frac{1}{2} \Delta u(x) - \sum_{i=1}^N \frac{x_i}{2\lambda_i} D_i u(x), \qquad x \in \mathbb{R}^N$$
 (9.3.41)

and

$$B^{0} = MBM^{-1} + \frac{1}{2}\operatorname{diag}(\lambda_{1}^{-1}, \dots, \lambda_{N}^{-1}). \tag{9.3.42}$$

The measure

$$\tilde{\mu}(dx) = \frac{1}{\sqrt{(2\pi)^N \lambda_1 \cdot \ldots \cdot \lambda_N}} \exp\left(-\sum_{i=1}^N \frac{x_i^2}{2\lambda_i}\right) dx$$
 (9.3.43)

is the invariant measure of both the semigroup $\{\tilde{T}(t)\}$ associated with the operator $\tilde{\mathcal{A}}$ and the semigroup $\{T^0(t)\}$ associated with the operator \mathcal{A}^0 . Finally, denoting by \tilde{L}_p the generator of $\{\tilde{T}(t)\}$ in $L^p_{\tilde{\mu}}$ $(p \in (1, +\infty))$, it holds that $D(L_p) = \Phi_M(D(\tilde{L}_p))$ and $\tilde{L}_p := \Phi_{M^{-1}} L_p \Phi_M$.

Proof. Since Q is strictly positive definite, then the matrix $Q^{-1/2}$ is well defined. Now, let M_1 be an orthogonal matrix such that

$$M_1 Q^{-1/2} Q_{\infty}(Q^{-1/2}) M_1^* = \operatorname{diag}(\lambda_1, \dots, \lambda_N).$$

A straightforward computation shows that, setting $M = M_1 Q^{-1/2}$, we have

$$(\tilde{\mathcal{A}}u)(x) = \frac{1}{2}\Delta u(x) + \langle MBM^{-1}x, Du(x)\rangle, \qquad x \in \mathbb{R}^N.$$
 (9.3.44)

Therefore, $\tilde{\mathcal{A}}$ is still an Ornstein-Uhlenbeck operator. The decomposition in (9.3.40)-(9.3.42) now easily follows.

The associated Ornstein-Uhlenbeck semigroup is given by (9.1.5) where now the matrix Q_t is replaced with the matrix

$$\tilde{Q}_{t} = \int_{0}^{t} e^{sMBM^{-1}} e^{s(M^{-1})^{*}B^{*}M^{*}} ds$$

$$= \int_{0}^{t} M e^{sB} M^{-1} (M^{-1})^{*} e^{sB^{*}} M^{*} ds$$

$$= \int_{0}^{t} M e^{sB} Q e^{sB^{*}} M^{*} ds = MQ_{t}M^{*}, \qquad t > 0. \qquad (9.3.45)$$

In particular, $\tilde{Q}_{\infty} = MQ_{\infty}M^* = \operatorname{diag}(\lambda_1, \dots, \lambda_N)$. Therefore, according to Proposition 9.3.1, the measure $\tilde{\mu}$ in (9.3.43) is the invariant measure of both the semigroups $\{\tilde{T}(t)\}$ and $\{T^0(t)\}$.

Let us now denote by \tilde{L}_p the infinitesimal generator of $\{\tilde{T}(t)\}$ in $L^p_{\tilde{\mu}}$ $(p \in (1, +\infty))$. Since by Lemma 9.3.13, $C_c^{\infty}(\mathbb{R}^N)$ is a core of both L_p and \tilde{L}_p , and $\tilde{L}_p = \tilde{\mathcal{A}} = \Phi_{M^{-1}} \mathcal{A} \Phi_M = \Phi_{M^{-1}} L_p \Phi_M$ in $C_c^{\infty}(\mathbb{R}^N)$, we easily deduce that $D(\tilde{L}_p) = \Phi_{M^{-1}}(D(L_p))$ and $\tilde{L}_p = \Phi_{M^{-1}} L_p \Phi_M$. This completes the proof.

Lemma 9.3.16 Let $\tilde{\mu}$ and B^0 be the measure and the matrix defined, respectively, in (9.3.39) and (9.3.42). Then, for any $p \in [1, +\infty)$, the family of bounded operators $\{S(t): t \geq 0\}$ in $L^p_{\tilde{\mu}}$, defined by $(S(t)f)(x) = f(e^{tB^0}x)$ for any $t \in [0, +\infty)$, any $x \in \mathbb{R}^N$ and any $f \in L^p_{\mu}$, is a semigroup of isometries. Moreover, $C^\infty_c(\mathbb{R}^N)$ is a core of the infinitesimal generator B^0_p and $(B^0_p u)(x) = \langle B^0 x, Du(x) \rangle$ for any $u \in C^\infty_c(\mathbb{R}^N)$ and any $x \in \mathbb{R}^N$.

 ${\it Proof.}$ To begin with, let us observe that a straightforward computation shows that

$$\int_{\mathbb{R}^N} |S(t)f|^p d\tilde{\mu}$$

$$= \frac{e^{-t\operatorname{Tr} B^0}}{\sqrt{(2\pi)^N(\lambda_1 \cdot \ldots \cdot \lambda_N)}} \int_{\mathbb{R}^N} |f(y)|^p \exp\left(-\frac{1}{2} \langle \tilde{Q}_{\infty}^{-1} e^{-tB^0} y, e^{-tB^0} y \rangle\right) dy,$$

for any t > 0, where $\tilde{Q}_{\infty} = \text{diag}(\lambda_1, \dots, \lambda_N)$. To prove that $||S(t)f||_p = ||f||_p$, it suffices to show that $\text{Tr } B^0 = 0$ and

$$\langle \tilde{Q}_{\infty}^{-1} e^{-tB^0} h, e^{-tB^0} h \rangle = \langle \tilde{Q}_{\infty}^{-1} h, h \rangle, \qquad t > 0, \ h \in \mathbb{R}^N.$$
 (9.3.46)

Applying (9.3.8) to the Ornstein-Uhlenbeck operator $\tilde{\mathcal{A}}$ (see (9.3.44)), we get

$$\tilde{Q}_t e^{-t(MBM^{-1})^*} + e^{tMBM^{-1}} \tilde{Q}_{\infty} = \tilde{Q}_{\infty} e^{-t(MBM^{-1})^*},$$
 (9.3.47)

for any t > 0, where \tilde{Q}_t is given by (9.3.45). Differentiating both the sides of (9.3.47) at t = 0 gives

$$I + MBM^{-1}\tilde{Q}_{\infty} = -\tilde{Q}_{\infty}(MBM^{-1})^*$$

or, equivalently,

$$\tilde{Q}_{\infty}^{-1} B^0 \tilde{Q}_{\infty} = -(B^0)^*. \tag{9.3.48}$$

From (9.3.48) we get $\operatorname{Tr} B^0 = 0$ and

$$\tilde{Q}_{\infty}^{-1}e^{-tB^{0}} = e^{t(B^{0})^{*}}\tilde{Q}_{\infty}^{-1}, \qquad t > 0,$$

which implies (9.3.46).

Let us now observe that $\{S(t)\}$ is a strongly continuous group. Indeed, for any $f \in C_c^{\infty}(\mathbb{R}^N)$, $f(e^{tB^0}\cdot)$ converges uniformly to f as t tends to 0. Therefore, S(t)f tends to f as t tends to 0 in $L^p_{\tilde{\mu}}$. Since $C_c^{\infty}(\mathbb{R}^N)$ is dense in $L^p_{\tilde{\mu}}$ (see Lemma 9.3.3) and $\{S(t)\}$ is a semigroup of isometries in $L^p_{\tilde{\mu}}$, using Proposition A.1.2, it is immediate to check that, for any $f \in L^p_{\tilde{\mu}}$, S(t)f tends to f as t tends to 0.

Denote now by B_p^0 the infinitesimal generator of the semigroup $\{S(t)\}$ in L_μ^p . To prove that $C_c^\infty(\mathbb{R}^N)$ is a core of B_p^0 , one can argue as in the proof of Lemma 9.3.13. Indeed, it is easy to verify that the Schwarz space $\mathcal S$ is contained in $D(B_p^0)$ and $(B_p^0f)(x) = \langle B^0x, Df(x) \rangle$ for any $x \in \mathbb{R}^N$ and any $f \in \mathcal S$. Moreover, S(t) maps $\mathcal S$ into itself for any t > 0. Therefore, by Proposition B.1.10, $\mathcal S$ is a core of $\mathcal B_p^0$. Then, the same approximation argument as in the proof of Lemma 9.3.13 shows that $C_c^\infty(\mathbb{R}^N)$ is a core of B_p^0 and $(B_p^0f)(x) = \langle Bx, Du(x) \rangle$ for any $x \in \mathbb{R}^N$ and any $u \in W_{\tilde{\mu}}^{2,p}$.

Theorem 9.3.17 For any $p \in (1, +\infty)$, $D(L_p) = W_{\mu}^{2,p}$ and $L_p u = \mathcal{A}u$ for any $u \in W_{\mu}^{2,p}$.

Proof. By virtue of the Lemma 9.3.15, to characterize $D(L_p)$, it suffices to characterize $D(\tilde{L}_p)$. In particular, since Φ_M is an isometry between $W_{\tilde{\mu}}^{2,p}$ and $W_{\mu}^{2,p}$, if we show that $D(\tilde{L}_p) = W_{\tilde{\mu}}^{2,p}$ $(p \in (1, +\infty))$, we will immediately deduce that $D(L_p) = W_{\mu}^{2,p}$.

We fix now, and for the rest of the proof, an arbitrary $p \in (1, +\infty)$. Moreover, we denote by L_0^p the infinitesimal generator of the Ornstein-Uhlenbeck semigroup $\{T^0(t)\}$ in L_μ^p , associated with the operator \mathcal{A}^0 in (9.3.41), and, by B_p^0 , the operator in Lemma 9.3.16. By Proposition 9.3.14, $D(L_p^0) = W_{\tilde{\mu}}^{2,p}$ and $L_p^0 u = \mathcal{A}^0 u$ for any $u \in W_{\tilde{\mu}}^{2,p}$. Since $C_c^\infty(\mathbb{R}^N)$ is a core of \tilde{L}_p (see Lemma 9.3.13) and the operator $L_p^0 + B_p^0$ with domain $W_{\tilde{\mu}}^{2,p}$ coincides with the bounded operators $\tilde{\mathcal{A}}$ (see Corollary 9.3.5), then \tilde{L}_p is the closure of the sum of the operators L_p^0 and B_p^0 . To complete the proof it suffices to show that the operator $L_p^0 + B_p^0$, with domain $W_\mu^{2,p}$, is closed. For this purpose, we prove that all the assumptions of Theorem B.1.16 are satisfied.

As a first step we observe that, repeating the same arguments as in the proof of Proposition 9.3.14 and observing that the operator L_2^0 is self-adjoint in $L_{\tilde{\mu}}^2$ (see Proposition 9.3.10), one can easily check that, for any $\lambda > 0$, both the

operators $I-L_p^0$ and $\lambda I-B_p^0$ admit bounded imaginary powers. Moreover, by Proposition B.1.13(iii) it follows that the power angle $\theta_{\lambda I-B_p^0} \leq \pi/2$ of $\lambda I-B_p^0$ is not greater than $\pi/2$. On the other hand, since L_2^0 is self-adjoint, then, by Proposition B.1.13(v), $\theta_{I-L_2^0}=0$. Therefore, $\theta_{I-L_2^0}<\pi/2$ for any $p\in(1,+\infty)$.

We now show that if λ is sufficiently large, then the operators $I-L_p^0$ and $\lambda I-B_p^0$ satisfy the condition (B.1.6). For this purpose, for any $\lambda>0$, any $\mu\in\rho(B_p^0)$ and any $\nu\in L_p^0$, we define the bounded operator $E(\lambda,\mu,\nu):C_b^4(\mathbb{R}^N)\to L_{\tilde{\mu}}^p$ by setting

$$E(\lambda,\mu,\nu)u = (I-L_p^0)R(\nu+1,L_p^0)[R(1,L_p^0),R(\lambda+\mu,B_p^0)]u, \qquad u \in W_{\tilde{\mu}}^{2,p}.$$

An explicit computation and Proposition 9.2.2 show that both $\{S(t)\}$ and $\{T^0(t)\}$ map $C_b^4(\mathbb{R}^N)$ into itself for any t>0 and there exist two constants $M\geq 1$ and $\omega_0\geq 0$ such that

$$e^{-\omega_0 t} \|S(t)\|_{L(C_h^4(\mathbb{R}^N))} + \|T^0(t)\|_{L(C_h^4(\mathbb{R}^N))} \le M, \qquad t > 0.$$

This implies that both the resolvent operators $R(\xi + \omega_0, B_p^0)$ and $R(\xi, L_p^0)$ map $C_b^4(\mathbb{R}^N)$ into itself for any $\xi \in \mathbb{C}$ with $\text{Re } \xi > 0$. Therefore, if $\lambda > \omega_0$,

$$\begin{split} &[R(1,L_p^0),R(\lambda+\mu,B_p^0)]\\ &=-R(\lambda+\mu,B_p^0)R(1,L_p^0)[B_p^0,L_p^0]R(1,L_p^0)R(\lambda+\mu,B_p^0). \end{split}$$

Observing that

$$\begin{split} &(I-L_p^0)R(\nu+1,L_p^0)R(\lambda+\mu,B_p^0)R(1,L_p^0)u\\ &=-(I-L_p^0)R(\nu+1,L_p^0)[R(1,L_p^0),R(\lambda+\mu,B_p^0)]u\\ &+(I-L_p^0)R(\nu+1,L_p^0)R(1,L_p^0)R(\lambda+\mu,B_p^0)u\\ &=-E(\lambda,\mu,\nu)u+(I-L_p^0)R(1,L_p^0)R(\nu+1,L_p^0)R(\lambda+\mu,B_p^0)u\\ &=-E(\lambda,\mu,\nu)u+R(\nu+1,L_p^0)R(\lambda+\mu,B_p^0)u, \end{split}$$

for any $u \in C_b^4(\mathbb{R}^N)$, we finally get

$$E(\lambda, \mu, \nu)u = \{E(\lambda, \mu, \nu) - R(\nu + 1, L_p^0)R(\lambda + \mu, B_p^0)\}[B_p, L_p^0]R(1, L_p^0)R(\lambda + \mu, B_p^0)u,$$
(9.3.49)

for any $u \in C_b^4(\mathbb{R}^N)$ and any λ sufficiently large.

Observe that the equality in (9.3.49) can be extended to all the functions $u \in L^p_\mu$. To see it, it suffices to show that the operator $[B^0_p, L^0_p]R(1, L_p)$: $C^4_b(\mathbb{R}^N) \to L^p_\mu$ extends to a bounded operators in L^p_μ , but this is immediate since an explicit computation and (9.3.48) show that

$$([B_p, L_p^0]u)(x) = -\operatorname{Tr}(B^0 D^2 u(x)) + \frac{1}{2} \langle (B^0 + (B^0)^*) \tilde{Q}_{\infty}^{-1} x, Du(x) \rangle,$$

for any $x \in \mathbb{R}^N$ and any $u \in C_b^4(\mathbb{R}^N)$. Lemma 9.3.4 and Proposition 9.3.14 then allow us to conclude.

Now, we observe that, according to Proposition B.1.13(ii), we can determine two angles φ_1 and φ_2 with $\varphi_1 > \max\{\theta_{B_p^0}, \pi/2\}, \varphi_2 \in (\theta_{L_p^0}, \pi/2)$ and $\varphi_1 + \varphi_2 < \pi$, such that

$$||R(\mu+\lambda, B_p^0)||_{L(L_{\tilde{\mu}}^p)} \le \frac{C}{|\mu+\lambda|}, \qquad ||R(\nu+1, L_p^0)||_{L(L_{\tilde{\mu}}^p)} \le \frac{C}{1+|\nu|}, (9.3.50)$$

for any $|\arg \mu| < \pi - \varphi_1$, any $|\arg \nu| < \pi - \varphi_2$ and some positive constant C, independent of μ, ν . Note that we can take $|\nu| + 1$ instead of $|\nu|$ in the second estimate in (9.3.50) since the operator $\nu \mapsto R(\nu+1, L_p^0)$ is bounded in a neighborhood of 0.

Now, from (9.3.49) and (9.3.50) we get

$$||E(\lambda,\mu,\nu)||_{L(L^p_{\tilde{\mu}})} \le \frac{C_1}{|\mu+\lambda|} ||E(\lambda,\mu,\nu)||_{L(L^p_{\tilde{\mu}})} + \frac{C_2}{(1+|\nu|)|\mu+\lambda|^2}, \quad (9.3.51)$$

for some positive constants C_1 and C_2 , independent of λ, μ, ν . Taking $\lambda > \max\{\omega_0, 2C_1\}$, we easily deduce that

$$||E(\lambda,\mu,\nu)||_{L(L^p_{\check{\mu}})} \le \frac{2C_2}{(1+|\nu|)|\mu+\lambda|^2} \le \frac{2C_2}{(1+|\nu|)|\mu|^2},$$

for any $\mu, \nu \in \mathbb{C} \setminus \{0\}$ such that $|\arg \mu| < \pi - \varphi_1$ and $|\arg \nu| < \pi - \varphi_2$. Theorem B.1.16 allows us to conclude that the operator $(\lambda - B_p^0) + (I - L_p^0)$ (or, equivalently, the operator $L_p^0 + B_p^0$) is closed in $W_{\tilde{\mu}}^{2,p}$. This concludes the proof.

Remark 9.3.18 In the case when p=1 the characterization of $D(L_1)$ is still an open problem. From Lemma 9.3.13, it follows easily that $D(L_1)$ is the closure of $C_c^{\infty}(\mathbb{R}^N)$ with respect to the graph norm of the operator L_1 .

On the other hand, the set equality $D(L_1) = W_{\mu}^{2,1}$ cannot hold, since, in particular, $D(L_1)$ is not continuously embedded in $W_{\mu}^{1,1}$. Indeed, suppose on the contrary that $D(L_1)$ is continuously embedded in $W_{\mu}^{1,1}$. Then, the operator $R(1,L_1):W_{\mu}^{1,1}\to W_{\mu}^{1,1}$ should be compact. Therefore the spectrum of the part of L_1 in $W_{\mu}^{1,1}$ (i.e., the restriction of the operator L_1 to the set $\{u \in D(L_1): L_1u \in W_{\mu}^{1,1}\}$) should be a discrete set. But this cannot be the case, since, as it is easily seen, it coincides with $\sigma(L_1)$, which, according to the forthcoming Theorem 9.3.24, is a halfplane.

9.3.2 The spectrum of the Ornstein-Uhlenbeck operator in L^p_μ

In this subsection we explicitly characterize the spectrum of the operator L_p for any $p \in [1, +\infty)$. The results that we present here are taken from [113] and they hold also for degenerate Ornstein-Uhlenbeck operators.

First of all, we show that the operator $R(\lambda, L_p)$ is compact in L^p_μ for any $p \in (1, +\infty)$.

Theorem 9.3.19 For any $p \in (1, +\infty)$, $W_{\mu}^{1,p}$ is compactly embedded in L_{μ}^{p} . As a consequence, for any p as above, T(t) and $R(\lambda, L_{p})$ are compact operators in L_{μ}^{p} , respectively, for any t > 0 and any $\lambda \in \rho(L_{p})$.

Proof. Of course, we can limit ourselves to showing that $W_{\mu}^{1,p}$ is compactly embedded in L_{μ}^{p} for any $p \in (1, +\infty)$. Indeed, since T(t) maps L_{μ}^{p} in $W_{\mu}^{1,p}$ for any t > 0 (see Proposition 9.3.8), the compactness of T(t) then easily follows. Finally, since $R(\lambda, L_{p})$ is the Laplace transform of the semigroup, for any λ with positive real part (see the formula (B.1.3)), then, taking Proposition A.1.1 and the resolvent formula (A.3.2) into account, we will deduce that $R(\lambda, L_{p})$ is compact as well.

To prove that $W_{\mu}^{1,p}$ is compactly embedded in L_{μ}^{p} , we adapt the technique in the proof of Theorem 8.5.3. For this purpose, it suffices to show that

$$\lim_{r \to +\infty} \int_{\mathbb{R}^N \setminus B(r)} |f|^p d\mu = 0, \tag{9.3.52}$$

uniformly with respect to $f \in W^{1,p}_{\mu}$ with $||f||_{W^{1,p}_{\mu}} \leq 1$. Taking Lemma 9.3.4 into account, we can write

$$\int_{\mathbb{R}^N \backslash B(r)} |f|^p d\mu \leq \frac{1}{r^p} \int_{\mathbb{R}^N} |xf(x)|^p d\mu \leq \frac{C}{r^p} \|f\|_{W^{1,p}_\mu},$$

for any $f \in W^{1,p}_{\mu}$ and some positive constant C, independent of f. Now, (9.3.52) easily follows.

As a consequence of Theorem 9.3.19, the spectrum of L_p consists of at most a sequence of eigenvalues for any $p \in (1, +\infty)$. In the remainder of this subsection, we characterize explicitly the spectrum of L_p . To begin with, we observe that 0 is an eigenvalue of the operator L_p for any $p \in (1, +\infty)$ and, according to the Liouville type theorem 8.1.17 and Proposition 9.3.8, the corresponding eigenspace is one-dimensional and consists of all the constant functions. As the following proposition shows, if u is a generalized eigenfunction corresponding to the eigenvalue $\lambda \in \mathbb{C}$, then u is a polynomial. Of course, if u is a polynomial it belongs to $W_{\mu}^{2,p} = D(L_p)$ for any $p \in (1, +\infty)$ (see Theorem 9.3.17).

Proposition 9.3.20 Fix $p \in (1, +\infty)$ and let $u \in D(L_p^r)$ satisfy $(\lambda - L_p)^r u = 0$ for some $\lambda \in \mathbb{C}$ and some $r \in \mathbb{N}$. Then, $\operatorname{Re} \lambda \leq 0$ and u is a polynomial with degree at most equal to $\operatorname{Re} \lambda/s(B)$.

Proof. We prove the assertion by induction on r. Suppose that r = 1 and let $\lambda \in \mathbb{C}$ and $u \in D(L_p)$ be such that $\lambda u - L_p u = 0$. Then, as it is easily checked,

 $T(t)u=e^{\lambda t}u$ for any $t\geq 0$ and, from Proposition 9.3.8, we get $u\in W^{k,p}_{\mu}$ for any $k\in\mathbb{N}$. From the formula (9.3.17), we easily deduce that, for any $\varepsilon>0$ and any multi-index α , there exists a positive constant $C=C(\varepsilon,|\alpha|)$ such that

$$e^{\operatorname{Re}\lambda t} \|D^{\alpha}u\|_{p} \le Ce^{|\alpha|(s(B)+\varepsilon)t} \sum_{|\beta|=|\alpha|} \|D^{\beta}u\|_{p}, \qquad t > 0.$$

From such an inequality we obtain that if $\operatorname{Re} \lambda > |\alpha| s(B)$, then C should be zero. Therefore, the derivatives of u of order $k > \operatorname{Re} \lambda/s(B)$ should vanish and the assertion follows in this particular case.

Now, suppose that the assertion is true with r = n > 1 and let us prove it with r = n + 1. For this purpose we observe that if $u \in D(L_p^{n+1})$ satisfies $(\lambda u - L_p u)^{n+1} = 0$, then the function $v = \lambda u - L_p u$ belongs to $D(L_p^n)$ and it satisfies the equation $(\lambda I - L_p)^r v = 0$. Therefore, v is a polynomial of degree at most $\text{Re } \lambda/s(B)$. Moreover, since $(\lambda I - L_p)^s v = 0$ for any $s \geq n + 1$, we have

$$T(t)u = e^{\lambda t} \sum_{i=0}^{n} \frac{t^{i}}{i!} (\lambda I - L_{p})^{i} u = e^{\lambda t} u + e^{\lambda t} \sum_{i=0}^{n-1} \frac{t^{i+1}}{(i+1)!} (\lambda I - L_{p})^{i} v, \quad (9.3.53)$$

for any t > 0. From the formula (9.3.53), we obtain that

$$D^{\alpha}T(t)u = e^{\lambda t}D^{\alpha}u + e^{\alpha t}\sum_{i=0}^{n-1} \frac{t^{i+1}}{(i+1)!}D^{\alpha}(\lambda v - L_{p}v)^{i}, \qquad t > 0,$$

for any multi-index α . Since v is a polynomial with degree at most $\text{Re}\lambda/s(B)$, then

$$D^{\alpha}T(t)u = e^{\lambda t}D^{\alpha}u, \qquad t > 0,$$

for any multi-index α with $|\alpha| > \text{Re } \lambda/s(B)$. Now the proof goes on as in the case when r = 1.

We now show that the problem of determining the eigenvalues of the operator L_p can be stated in an equivalent (and simpler as far as computations are concerned) way. For this purpose, we introduce the operator $\mathcal{B}_p:W^{2,p}_{\mu}\to L^p_{\mu}$ defined by

$$\mathcal{B}_p u(x) = \langle Bx, Du(x) \rangle, \qquad x \in \mathbb{R}^N, \quad u \in W^{2,p}_{\mu}.$$
 (9.3.54)

According to Lemma 9.3.4, \mathcal{B}_p is well defined and it is bounded in $W_{\mu}^{2,p}$.

Proposition 9.3.21 *Let* $p \in (1, +\infty)$ *. The following statements are equivalent:*

(i) $\lambda \in \mathbb{C}$ is an eigenvalue of the operator L_p ;

(ii) λ is an eigenvalue of the operator \mathcal{B}_p and the corresponding eigenspace contains a nonzero homogeneous polynomial.

Proof. Let $\lambda \in \mathbb{C}$ be an eigenvalue of the operator L_p . Further, let u be a corresponding eigenfunction so that $\lambda u - \mathcal{B}_p u = \sum_{i,j=1}^N q_{ij} D_{ij} u$. By Proposition 9.3.20, u is a polynomial. We denote by m its degree. If m=0,1 then it is immediate to check that λ is an eigenvalue of \mathcal{B}_p . So, let us assume that $m \geq 2$. Note that for any $n \in \mathbb{N}$, both \mathcal{B}_p and L_p map the set \mathcal{P}_n of all the polynomials with degree at most n into itself. Suppose that $\lambda I - \mathcal{B}_p$ is invertible in \mathcal{P}_{m-2} . Then, there exists a polynomial v with degree m-2 such that

$$\lambda v - \mathcal{B}_p v = \sum_{i,j=1}^N q_{ij} D_{ij} u.$$

The function w = u - v, which is a nontrivial polynomial with degree m, turns out to be an eigenfunction of the operator L_p corresponding to the eigenvalue λ . If the operator $\lambda I - \mathcal{B}_p$ is not invertible in \mathcal{P}_{m-2} , of course λ is an eigenvalue of \mathcal{B}_p and we can take as a corresponding eigenfunction a polynomial w with degree m-2.

To complete the proof of the property (ii) we have to check that there exists a homogeneous polynomial among the eigenfunctions of the operator \mathcal{B}_p corresponding to the eigenvalue λ . For this purpose, it suffices to notice that, for any $k \in \mathbb{N}$, \mathcal{B}_p maps the space of homogeneous polynomials with degree k into itself. Therefore, the homogeneous part with maximum degree of the eigenfunction w is itself an eigenfunction.

Let us now assume that the property (ii) holds and let us prove that the property (i) holds as well. For this purpose, we assume that $\lambda \in \mathbb{C}$ is an eigenvalue of the operator \mathcal{B}_p and that the homogeneous polynomial u is a corresponding eigenfunction. Of course,

$$\lambda u - L_p u = -\sum_{i,j=1}^{N} q_{ij} D_{ij} u.$$

Repeating the same arguments used above, now applied to the operator L_p , we can easily show that λ is an eigenvalue of L_p . This concludes the proof.

By virtue of Proposition 9.3.21, to characterize the spectrum of the operator A_p we can limit ourselves to characterize the spectrum of the operator \mathcal{B}_p , which is easier to handle.

Theorem 9.3.22 Denote by $\lambda_1, \ldots, \lambda_r$ the r distinct eigenvalues of the matrix B. Then, for any $p \in (1, +\infty)$,

$$\sigma(L_p) = \left\{ \lambda = \sum_{i=1}^r n_i \lambda_i : n_i \in \mathbb{N} \cup \{0\}, \ i = 1, \dots, r \right\}.$$
 (9.3.55)

In particular, $\sigma(L_p)$ is independent of p.

Proof. To begin with, we observe that a straightforward computation shows that $v: \mathbb{R}^N \to \mathbb{C}$ is an eigenfunction of the operator \mathcal{B}_p , corresponding to the eigenvalue $\lambda \in \mathbb{C}$, if and only if

$$v(e^{tB}x) = e^{\lambda t}v(x), t > 0, x \in \mathbb{R}^{N}.$$
 (9.3.56)

According to Proposition 9.3.21, we can assume that v is a homogeneous polynomial. Therefore, the equality (9.3.56) can be extended to all $z \in \mathbb{C}$.

Denote by $\lambda_1, \ldots, \lambda_r$ the distinct eigenvalues of the matrix B. By a general result of linear algebra (see [100, Chapter 11, Section 6]), there exists a nonsingular $N \times N$ complex valued matrix M such that $B = M^{-1}CM$, with $C = \operatorname{diag}(C_1, \ldots, C_r)$, where each block C_i is a Jordan block, which can be split into the sum $C_i = \lambda_i I_{n_i} + N_{n_i}$, n_i being the algebraic multiplicity of the eigenvalue λ_i and N_{λ_i} being a nihilpotent matrix. It follows that $e^{tB} = M^{-1}e^{tC}M$ and, consequently, setting $w(z) = v(M^{-1}z)$ for any $z \in \mathbb{C}^N$, we easily see that v satisfies (9.3.56) if and only if

$$w(e^{tC}z) = e^{\lambda t}w(z), \qquad t > 0, \ z \in \mathbb{C}^N. \tag{9.3.57}$$

We can now prove the inclusion " \subset " in the formula (9.3.55). For this purpose, let $\lambda \in \sigma(L_p)$ and let v be a corresponding eigenfunction. Since we have assumed that v is a homogeneous polynomial, w is a homogeneous polynomial as well and we can write

$$w(z) = \sum_{|\alpha_1| + \dots + |\alpha_n| = n} a_{\alpha_1, \dots, \alpha_r} \prod_{j=1}^r z_j^{\alpha_j}, \qquad z \in \mathbb{C}^N,$$

for some $n \in \mathbb{N}$ and some $a_{\alpha_1,\dots,\alpha_r} \in \mathbb{C}$ $(|\alpha_1| + \dots + |\alpha_n| = n)$, where $z = (z_1,\dots,z_r)$ and $z_i \in \mathbb{C}^{n_i}$. Since $e^{tC}z = (e^{tC_1}z_1,\dots,e^{tC_r}z_r)$ and N_i is nihilpotent, we have

$$e^{tC_i}y = e^{\lambda t}(P_{1,i}(t,y), \dots, P_{r_i,i}(t,y)), \qquad t > 0, \ y \in \mathbb{C}^{n_i}, \qquad i = 1, \dots, r,$$

where $P_{k,i}$ (i = 1, ..., r) is a polynomial in both its entries. It follows that

$$e^{\lambda t}w(z) = w(e^{tC}z)$$

$$= \sum_{|\alpha_1|+\dots+|\alpha_n|=n} a_{\alpha_1,\dots,\alpha_r}e^{(\lambda_1|\alpha_1|+\dots+\lambda_n|\alpha_n|)t}P_{\alpha_1,\dots,\alpha_r}(t,z),$$
(9.3.58)

for any t>0 and any $z\in\mathbb{C}^N$. Here, $P_{\alpha_1,\ldots,\alpha_r}$ is a polynomial. Fixing $z^*\in\mathbb{C}^N$ such that $w(z^*)\neq 0$ and letting t tend first to $+\infty$ and then to $-\infty$ in (9.3.58), we easily see that $\lambda=\lambda_1|\alpha_1|+\ldots+\lambda_n|\alpha_n|$ for some α_1,\ldots,α_n , and the inclusion " \subset " in (9.3.55) follows.

We now show the other inclusion in (9.3.55). For this purpose, let $\lambda = \sum_{i=1}^{r} n_i \lambda_i$ for some $n_i \in \mathbb{N}$ (i = 1, ..., r) and let v be the polynomial defined by

$$w(z) = \hat{z}_1^{n_1} \cdot \hat{z}_2^{n_2} \cdot \ldots \cdot \hat{z}_r^{n_r}, \qquad z \in \mathbb{C}^N,$$

where \hat{z}_i is the last component of the vector z_i $(i=1,\ldots,r)$ defined above. Since the last row of the matrix N_i is 0 for any $i=1,\ldots,r$, then, for any $z\in\mathbb{C}^{n_i}$, any $k\in\mathbb{N}$ and any $i=1,\ldots,r$, the last entry of the vector N_i^kh is 0. It follows that the last entry of the vector $e^{tC_i}z_i$ is $e^{\lambda t}\hat{z}_i$ for any $i=1,\ldots,r$ and, consequently,

$$w(e^{tC}z) = e^{\lambda t}w(z), \qquad t > 0, \ z \in \mathbb{C}^N.$$

Hence, λ is an eigenvalue of \mathcal{B}_p and the (homogeneous) polynomial v defined by v(x) = w(Mx) for any $x \in \mathbb{R}^N$ is a corresponding eigenfunction. Proposition 9.3.21 allows us to conclude that $\lambda \in \sigma(L_p)$. This completes the proof.

Some more information is available on the eigenspaces corresponding to the eigenvalues of the operator L_p . In the following theorem we give a necessary and sufficient condition which guarantees that the eigenvalues of L_p are semisimple. For notational convenience, for any linear operator S with compact resolvent operator, we denote by $i_{\lambda}(S)$ the index of the eigenvalue λ (i.e., the smallest integer k such that $\operatorname{Ker}(\lambda I - S)^k = \operatorname{Ker}(\lambda I - S)^{k+1}$). According to Proposition A.3.7, $i_{\lambda}(S)$ is well defined for any $\lambda \in \sigma(S)$ and $R(\cdot, S)$ has a pole of order $i_{\lambda}(S)$ at λ .

Theorem 9.3.23 The index $i_{\lambda}(L_p)$ is one for any eigenvalue λ of L_p (equivalently, all the eigenvalues of L_p are semisimple) if and only if the matrix B is diagonalizable in \mathbb{C} .

Proof. To show the assertion we prove that for any eigenvalue λ of the operator A_p , it holds that

$$1 + \max \left\{ \sum_{i=1}^{r} n_i (i_{\lambda_i}(B) - 1) : \sum_{i=1}^{r} n_i \lambda_i = \lambda \right\} \le i_{\lambda}(L_p) \le 1 + \frac{\operatorname{Re}\lambda}{s(B)} (i(B) - 1),$$
(9.3.59)

where i(B) denotes the maximum of the indices $i_{\lambda_i}(B)$ (i = 1, ..., r).

To prove (9.3.59) we proceed into some steps. As in the proof of Theorem 9.3.22, we split any $z \in \mathbb{C}^N$ into blocks as $z = (z_1, \ldots, z_r)$, where $z_i \in \mathbb{C}^{n_i}$ and n_i is the algebraic multiplicity of the eigenvalue λ_i of the matrix B. Moreover, we recall (see again the proof of Theorem 9.3.22) that we can determine a nonsingular matrix M such that $B = M^{-1}CM$, where $C = \text{diag}(C_1, \ldots, C_r)$ and each block C_i is a Jordan block, which can be split into the sum $C_i = \lambda_i I_{n_i} + N_{n_i}$ for some nihilpotent matrix N_{λ_i} . By well known results of linear algebra, $i_{\lambda_i}(B)$ coincides with the algebraic multiplicity of the eigenvalue λ , i.e., it coincides with n_i . As a consequence

$$e^{tC_i} = e^{t\lambda_i} p_i(t), \qquad t \in \mathbb{R}, \quad i = 1, \dots, N,$$
 (9.3.60)

where p_i are suitable matrices whose entries are polynomials with degree at most $i_{\lambda_i}(B) - 1$. Moreover, there exists $h \in \mathbb{R}^{n_i}$ such that at least one component of $p_i(t)x$ is a polynomial with degree $i_{\lambda_i}(B) - 1$.

Step 1. Here, we prove that $i_{\lambda}(\mathcal{B}_p) \leq i_{\lambda}(L_p)$. For this purpose, we show that, for any $\mu \in \rho(L_p)$, it holds that

$$R(\mu, \mathcal{B}_p)u = \sum_{i=1}^{+\infty} P_i R(\mu, L_p) P_i u,$$
 (9.3.61)

for any polynomial u. Here, P_i $(i \in \mathbb{N} \cup \{0\})$ is the canonical projection of the space of polynomials into the space \mathcal{H}_i of the homogeneous polynomials with degree i.

Once the formula (9.3.61) is checked, we can quite easily show that $R(\cdot, L_p)$ has a pole at λ , whose order is greater or equal to the order of the pole of $R(\cdot, \mathcal{B}_p)$ at λ . Indeed, using such a formula we deduce that for any polynomial u the function $\mu \mapsto (\mu - \lambda)^{i_{\lambda}(L_p)}R(\mu, \mathcal{B}_p)u$ can be analytically extended to $\lambda + B(r) \subset \mathbb{C}$ for some r > 0, independent of u. By density, $\mu \mapsto (\mu - \lambda)^{i_{\lambda}(L_p)}R(\mu, \mathcal{B}_p)u$ is analytic in $\lambda + B(r)$ for any $u \in L_{\mu}^p$, and the Banach-Steinhaus theorem (see [147, Chapter 2, Section 4]) implies that $\|(\mu - \lambda)^{i_{\lambda}(L_p)}R(\mu, \mathcal{B}_p)\|_{L(L_{\mu}^p)} \leq C$ for any $\mu \in \lambda + B(r)$ and some positive constant C, independent of λ . Of course, this implies that $R(\mu, \mathcal{B}_p)u$ has a pole of order $m \leq i_{\lambda}(L_p)$ at λ . Therefore, $\lambda \in \sigma(\mathcal{B}_p)$. According to the formula (A.3.5), $(\lambda I - \mathcal{B}_p)^m P = 0$ and $(\lambda I - \mathcal{B}_p)^{m-1} P \neq 0$, where P is the spectral projection with λ , i.e.,

$$P = \frac{1}{2\pi i} \int_{\gamma} R(\xi, \mathcal{B}_p) d\xi,$$

 γ being the boundary of a suitable ball centered at λ and oriented counterclockwise (see Proposition A.3.5). Therefore, $P(L_{\mu}^{p}) \subset \mathcal{E}_{\lambda}$, where by \mathcal{E}_{λ} we have denoted the space of generalized eigenfunction of \mathcal{B}_{p} corresponding to λ . But since the restriction of $\lambda I - \mathcal{B}_{p}$ to $(I - P)(L_{\mu}^{p})$ is an invertible operator (see again Proposition A.3.5), then $P(L_{\mu}^{p}) = \mathcal{E}_{\lambda}$. Now, it is immediate to check that $i_{\lambda}(\mathcal{B}_{p})$ coincides with m.

By linearity we can limit ourselves to checking (9.3.61) in the case when $u \in \mathcal{H}_i$ ($i \in \mathbb{N}$). Set $v = R(\mu, L_p)u$. Since \mathcal{B}_p maps \mathcal{H}_n into itself, we easily get

$$u = P_i u = P_i (\mu v - L_p v) = P_i (\mu v - \mathcal{B}_p v) = (\mu I - \mathcal{B}_p) P_i v.$$

The formula (9.3.61) now follows.

Step 2. We now prove that

$$i_{\lambda}(\mathcal{B}_p) = 1 + \max\left\{\sum_{i=1}^{r} n_i(i_{\lambda_i}(B) - 1) : \sum_{i=1}^{r} n_i \lambda_i = \lambda\right\}.$$
 (9.3.62)

Let v be a generalized eigenfunction of \mathcal{B}_p corresponding to the eigenvalue $\lambda = m_1\lambda_1 + \ldots + m_r\lambda_r$ such that $(\lambda I - \mathcal{B}_p)^{i_\lambda(\mathcal{B}_p)-1}v \neq 0$ and $(\lambda I - \mathcal{B}_p)^{i_\lambda(\mathcal{B}_p)}v = 0$. Let us show that v is a linear combination of polynomials $p_i \in \mathcal{H}_{m_i}$, $i = 1, \ldots, r$, such that each p_i depends only on the block of variables z_i . For this purpose, we observe that $v(e^{tB}x) = (S(t)v)(x)$ for any $t \geq 0$ and any $x \in \mathbb{R}^N$, where $\{S(t)\}$ is the strongly continuous group generated by the closure of the operator \mathcal{B}_p in L^p_μ (the strong continuity of $\{S(t)\}$ can be proved by arguing as in the proof of Lemma 9.3.16). Since v is a generalized eigenfunction of \mathcal{B}_p , then v is a polynomial. This can be easily checked arguing as in the proof of Proposition 9.3.20, observing that

$$||D^{\alpha}S(t)f||_{p} \le Ce^{|\alpha|(s(B)+\varepsilon)t} \sum_{|\beta|=|\alpha|} ||D^{\alpha}f||_{p}, \qquad t > 0,$$

for any multi-index α , any $\varepsilon > 0$, any $f \in W^{k,p}_{\mu}$ and some positive constant $C = C(|\alpha|, \varepsilon)$. Moreover, arguing as in the proof of (9.3.53), we can show that

$$(S(t)v)(x) = v(e^{tB}x) = e^{t\lambda}p(t,x), t > 0, x \in \mathbb{R}^N,$$
 (9.3.63)

where p is a polynomial in (t, x) whose degree in t is less or equal to $i_B(\lambda) - 1$. Since v is a polynomial, we can extend (9.3.63) to any $z \in \mathbb{C}^N$. Moreover, we can determine some constants $c_{\alpha_1,...,\alpha_r}$ such that

$$v(z) = \sum_{\substack{|\alpha_1|+\ldots+|\alpha_r| \le n}} c_{\alpha_1,\ldots,\alpha_r} \prod_{i=1}^r z_i^{\alpha_i}, \qquad z \in \mathbb{C}^N,$$

and, hence,

$$v(e^{tB}z) = \sum_{|\alpha_1|+\ldots+|\alpha_r| \le n} c_{\alpha_1,\ldots,\alpha_r} e^{t(\lambda_1|\alpha_1|+\ldots+\lambda_r|\alpha_r|)} p_{\alpha_1,\ldots,\alpha_r}(t,z), \quad (9.3.64)$$

for any t>0 and any $z\in\mathbb{C}^N$, where $p_{\alpha_1,\ldots,\alpha_r}$ are polynomials in (t,x) with degree at most $\sum_{i=1}^r |\alpha_i| (i_{\lambda_i}(B)-1)$ in t. Comparing (9.3.63) and (9.3.64) for large t, we easily deduce that $c_{\alpha_1,\ldots,\alpha_r}=0$, if $|\alpha_1|\lambda_1+\ldots+|\alpha_r|\lambda_r\neq\lambda$. It also follows that each polynomial $p_{\alpha_1,\ldots,\alpha_r}$ depends only on the block of variables x_i $(i=1,\ldots,r)$ for which $\alpha_i\neq0$.

From (9.3.63) and (9.3.64) we now get that

$$p(t,z) = q(t,z) := \sum_{|\alpha_1|\lambda_1 + \ldots + |\alpha_r|\lambda_r = \lambda} c_{\alpha_1,\ldots,\alpha_r} p_{\alpha_1,\ldots,\alpha_r}(t,z), \quad t > 0, \ z \in \mathbb{C}^N.$$

Moreover, from the arguments before Step 1, we deduce that there exists $z \in \mathbb{C}^N$ such that the degree of $q(\cdot, z)$ is $\max\{\sum_{i=1}^r n_i(i_{\lambda_i}(B)-1): \sum_{i=1}^r n_i\lambda_i = \lambda\}$ and the degree of $p(\cdot, z)$ is $i_{\lambda}(\mathcal{B}_p) - 1$. Now, (9.3.62) easily follows.

Step 3. We now show that

$$i_{\lambda}(L_p) \le 1 + \frac{\text{Re}\lambda}{s(B)}(i(B) - 1).$$
 (9.3.65)

For this purpose, let $u = \sum_{|\alpha| \leq m} c_{\alpha} x^{\alpha}$ be a polynomial (with degree $m \leq \text{Re } \lambda/s(B)$, see Proposition 9.3.20) such that $(\lambda I - L_p)^{i_{\lambda}(L_p)-1}u \neq 0$ and $(\lambda I - L_p)^{i_{\lambda}(L_p)}u = 0$. From the formula (9.3.53) with $n = i_{\lambda}(L_p) - 1$, we deduce that

$$(T(t)u)(x) = e^{\lambda t}p(t,x), \qquad t > 0, \ x \in \mathbb{R}^N,$$
 (9.3.66)

and there exists $x_0 \in \mathbb{R}^N$ such that the degree of $p(\cdot, x)$ is $i_{\lambda}(L_p) - 1$. A long but straightforward computation shows that

$$\begin{split} (T(t)u)(x) &= \sum_{|\alpha| \leq m} c_{\alpha} \sum_{h \leq \alpha} \binom{\alpha}{h} \left(e^{tB}x\right)^{h} \\ &\qquad \times \frac{1}{\sqrt{(2\pi)^{N} \det Q_{t}}} \int_{\mathbb{R}^{N}} y^{\alpha - h} e^{-\frac{1}{2}\langle Q_{t}^{-1}y, y \rangle} dy \\ &= \sum_{|\alpha| \leq m} c_{\alpha} \sum_{h \leq \alpha \atop |\alpha - h| \text{ even}} d_{\alpha, h} \binom{\alpha}{h} \left(e^{tB}x\right)^{h} D_{w}^{\alpha - h} \left((Q_{t}w, w)^{\frac{|\alpha - h|}{2}}\right)_{|w = 0}, \end{split}$$

for any t > 0 and any $x \in \mathbb{R}^N$, where $d_{\alpha,h}^{-1} = \left(\frac{2}{|\alpha - h|}\right)!$. Taking (9.3.60) into account, it easily follows that

$$(Q_t)_{ij} = \sum_{h=1}^r e^{\lambda_h t} p_{ij}^{(h)}(t) + c_{ij}^{(h)}, \qquad t > 0, \ i, j = 1, \dots, N,$$

where $p_{ij}^{(h)}$ are suitable polynomials with degree at most 2i(B)-2, and $c_{ij}^{(h)}$ are complex constants. Therefore, $D_w^{\alpha-h}((Q_tw,w)^{|\alpha-h|/2})_{|w=0} = \sum_{i=1}^{l_1} e^{\tau_i t} q_i(t)$ for any t>0 and some $l_1 \geq m$, where τ_i are integer multiples of $\lambda_1,\ldots,\lambda_r$ and q_i are polynomials with degree at most $(i(B)-1)(|\alpha|-|h|)$. Summing up,

$$(T(t)u)(x) = \sum_{i=1}^{l_2} e^{\xi_i t} r_i(t, x), \qquad t > 0, \ x \in \mathbb{R}^N,$$

for some $l_2 \in \mathbb{N}$, where ξ_i (i = 1, ..., r) are again integer multiples of $\lambda_1, ..., \lambda_r$, and r_i , as functions of t, are polynomials with degree at most m(i(B) - 1). Now, the estimate (9.3.65) easily follows. This completes the proof.

Now, we consider the case when p=1. In such a situation the picture drastically changes. Indeed, as the following theorem shows, the spectrum of L_1 is no more a discrete set but it is a halfplane. The proof of this property is based on the results of the forthcoming Section 9.4, where the spectrum of the Ornstein-Uhlenbeck operator in L^p -spaces associated with the Lebesgue measure is characterized.

Theorem 9.3.24 The spectrum of the Ornstein-Uhlenbeck operator in L^1_{μ} is the halfplane $\{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \leq 0\}$. Moreover, any $\lambda \in \mathbb{C}$ with negative real part is an eigenvalue of L_1 .

Proof. Let $\psi = 1/\rho$, where ρ is the density of the invariant measure μ with respect to the Lebesgue measure (see (9.3.2)). Denote by $I: L^1(\mathbb{R}^N) \to L^1_\mu$ the isometry defined by $If = f\psi$ for any $f \in L^1(\mathbb{R})$, and by \tilde{L}_1 the operator defined on $I^{-1}(D(L_1))$ by $\tilde{L}_1 = I^{-1}L_1I$. As it is immediately seen \tilde{L}_1 is the generator of a strongly continuous semigroup of contractions in $L^1(\mathbb{R}^N)$ and $\sigma(L_1) = \sigma(\tilde{L}_1)$. Therefore, we can limit ourselves to characterize the spectrum of the operator \tilde{L}_1 in $L^1(\mathbb{R}^N)$. For this purpose, we observe that since $C_c^{\infty}(\mathbb{R}^N) \subset D(L_1)$ and I maps $C_c^{\infty}(\mathbb{R}^N)$ into itself, then $C_c^{\infty}(\mathbb{R}^N)$ is contained in $D(\tilde{L}_1)$. A straightforward computation shows that $\tilde{L}_1 = \tilde{A}$ in $C_c^{\infty}(\mathbb{R}^N)$, where

$$(\tilde{\mathcal{A}}u)(x) = (\mathcal{A}u)(x) + \rho(x) \sum_{i,j=1}^{N} q_{ij} D_i \psi(x) D_j u(x) + \rho(x) u(x) (\mathcal{A}\psi)(x)$$
$$= \frac{1}{2} \operatorname{Tr} (QD^2 u(x)) + \langle (B + QQ_{\infty}^{-1})x, Du \rangle + (\rho \mathcal{A}\psi)(x) u(x),$$

for any $x \in \mathbb{R}^N$ and any $u \in C_c^{\infty}(\mathbb{R}^N)$. Now, we observe that differentiating the formula (9.3.8) at t = 0 gives

$$Q + BQ_{\infty} + Q_{\infty}B^* = 0. (9.3.67)$$

Using (9.3.67) we immediately deduce that

$$2\langle Q_{\infty}^{-1}Bx,x\rangle = 2\langle BQ_{\infty}Q_{\infty}^{-1}x,Q_{\infty}^{-1}x\rangle = -\langle QQ_{\infty}^{-1}x,Q_{\infty}^{-1}x\rangle, \qquad x\in\mathbb{R}^N$$

and, therefore,

$$\rho(x)(\mathcal{A}\psi)(x) = \frac{1}{2} \text{Tr} (QQ_{\infty}^{-1}) + \frac{1}{2} \langle QQ_{\infty}^{-1}x, Q_{\infty}^{-1}x \rangle + \langle B^*Q_{\infty}^{-1}x, x \rangle$$
$$= \frac{1}{2} \text{Tr} (QQ_{\infty}^{-1}) := k.$$

Using again (9.3.67) to obtain

$$B + QQ_{\infty}^{-1} = -Q_{\infty}B^*Q_{\infty}^{-1}, \tag{9.3.68}$$

we finally get

$$(\tilde{\mathcal{A}}u)(x) = \frac{1}{2} \text{Tr} \left(Q D^2 u(x) \right) - \langle Q_{\infty} B^* Q_{\infty}^{-1} x, D u(x) \rangle + k u(x), \qquad x \in \mathbb{R}^N.$$

Since $\hat{\mathcal{A}} := \tilde{\mathcal{A}} - kI$ is an Ornstein-Uhlenbeck, according to the forthcoming Propositions 9.4.1 and 9.4.2, the realization \hat{L}_1 of $\hat{\mathcal{A}}$ in $L^1(\mathbb{R}^N)$ is the generator

of a strongly continuous semigroup of contractions and $C_c^{\infty}(\mathbb{R}^N)$ is a core of \hat{L}_1 . We claim that $D(\tilde{L}_1) = D(\hat{L}_1)$. Indeed, since $(\tilde{L}_1, D(\tilde{L}_1))$ is a closed operator and $C_c^{\infty}(\mathbb{R}^N)$ is a core of \hat{L}_1 , $D(\hat{L}_1) \subset D(\tilde{L}_1)$ and $\hat{L} = \tilde{L} - k$ in $D(\hat{L}_1)$. To prove the other inclusion, we observe that $\lambda - \tilde{L}_1$ and $\lambda I - \hat{L}_1$ are both invertible if λ is sufficiently large. Therefore,

$$L^1(\mathbb{R}^N) = (I - \tilde{L}_1)(D(\tilde{L}_1)) \supset (I - \tilde{L}_1)(D(\hat{L}_1)) = (\lambda I - \hat{L}_1)(D(\hat{L}_1)) = L^1(\mathbb{R}^N),$$

which implies that $D(\tilde{L}_1) = D(\hat{L}_1)$, so that $(\tilde{L}_1, D(\tilde{L}_1))$ and $(\hat{L}_1 - kI, D(\hat{L}_1))$ do coincide. It follows that $\sigma(\tilde{L}_1) = \{\lambda \in \mathbb{C} : \lambda + k \in \sigma(\hat{L}_1)\}.$

Now, since the spectrum of the matrix $-Q_{\infty}B^*Q_{\infty}^{-1}$ is contained in the halfplane $\{\lambda \in \mathbb{C} : \text{Re}\lambda < 0\}$, according to the forthcoming Theorem 9.4.3, the spectrum of \hat{L}_1 is the halfplane $\{\lambda \in \mathbb{C} : \text{Re}\lambda \leq \text{Tr}\left(Q_{\infty}B^*Q_{\infty}^{-1}\right)\}$. Moreover, any λ with $\text{Re}\lambda < \text{Tr}\left(Q_{\infty}B^*Q_{\infty}^{-1}\right)$ is an eigenvalue of \hat{L}_1 .

To complete the proof it suffices to show that $k = -\text{Tr}(Q_{\infty}B^*Q_{\infty}^{-1})$ and this can be easily done taking the traces of both the sides of (9.3.68).

9.3.3 The sector of analyticity of the Ornstein-Uhlenbeck operator in L^p_μ

In this subsection we present a recent result due to R. Chill, E. Fasangova, G. Metafune and D. Pallara which gives a precise characterization of the sector of analyticity of the Ornstein-Uhlenbeck operator in L^p_{μ} $(p \in (1, +\infty))$, that is of the largest sector $\Sigma_{\theta} = \{\lambda \in \mathbb{C} : |\arg \lambda| < \theta\}$ where the function $z \mapsto T(z)$ is analytic with values in $L(L^p_{\mu})$.

Theorem 9.3.25 For any $p \in (1, +\infty)$ let $\theta_p \in (0, \pi/2]$ be defined by

$$\cot \theta_p = \frac{\sqrt{(p-2)^2 + p^2 \|I + 2Q^{-\frac{1}{2}}Q_{\infty}B^*Q^{-\frac{1}{2}}\|_{\infty}^2}}{2\sqrt{p-1}}.$$
 (9.3.69)

Then, Σ_{θ_p} is the biggest sector, with centre at the origin, where the Ornstein-Uhlenbeck semigroup can be analytically extended. Moreover, the so extended semigroup is contractive in L^p_μ . Finally, for p=2 the set of analyticity of the function $z\mapsto T(z)$ is the set

$$\{z \in \mathbb{C} : \|Q_{\infty}^{\frac{1}{2}} e^{zB} Q_{\infty}^{-\frac{1}{2}}\|_{\infty} \le 1\},$$
 (9.3.70)

and $\{T(z)\}$ is contractive in it.

Proof. See [31, Theorems 1, 2 and Remark 6].

Remark 9.3.26 (i) The formula (9.3.69) gives a relation between the sector of analyticity of the semigroup $\{T(t)\}$ and the sector (with center at 0) Σ_{θ_0} of

analyticity of the resolvent operator of the matrix $R := Q^{-1/2}Q_{\infty}B^*Q^{-1/2}$. Indeed,

$$\pi - \theta_0 = \arctan(\|I + 2Q^{-\frac{1}{2}}Q_{\infty}B^*Q^{-\frac{1}{2}}\|_{\infty}). \tag{9.3.71}$$

To check (9.3.71) we begin by observing that, due to the formula (9.3.67), $\frac{1}{2}I + Q^{-1/2}Q_{\infty}B^*Q^{-1/2}$ is a skew-symmetric matrix and, consequently, R is normal. Therefore $|I + 2R||_{\infty}$ coincides with the spectral radius r(I + 2R) of I + 2R, i.e., the radius of the smallest ball centered at the origin which contains all the eigenvalues of the matrix I + 2R. Still using (9.3.67) it is easy to check all the eigenvalues of the matrix R lie on the straight lines $\{\lambda \in \mathbb{C} : \operatorname{Re} \lambda = -1/2\}$, and, therefore, $\operatorname{arctan}(r(I + 2R)) = \pi - \theta_0$.

(ii) The formula (9.3.69) also shows that when the Ornstein-Uhlenbeck semi-group is symmetric (i.e., when $BQ = QB^*$ or, equivalently, when $Q_{\infty}B^* = BQ_{\infty}$, see the proof of Proposition 9.3.10), the sector of analyticity of $\{T(t)\}$ depends only on p and it equals $\pi/2$ when p=2. Indeed, in such a case, the formula (9.3.67) gives $Q^{-1/2}Q_{\infty}B^*Q^{-1/2}=-\frac{1}{2}I$.

Note that the symmetry of the Ornstein-Ühlenbeck semigroup in L^2_{μ} is also a necessary condition for its sector of analyticity to be $\Sigma_{\pi/2}$. Indeed, if $\{T(t)\}$ is analytic in $\Sigma_{\pi/2}$, then $\cot \theta_p = 0$. But this implies that $I + 2Q^{-1/2}Q_{\infty}B^*Q^{-1/2} = 0$ and, by virtue of (9.3.67), it follows that $Q_{\infty}B^* = B^*Q_{\infty}$, so that $\{T(t)\}$ is symmetric in L^2_{μ} , by Proposition 9.3.10.

(iii) In general the domain of analyticity of the functions $z \mapsto T(z)$ is larger than the sector Σ_{θ_p} . For instance this is the case of the one-dimensional Ornstein-Uhlenbeck semigroup and $p \neq 2$ (see [52]). But in general also in the multidimensional case, the domain of analyticity of $\{T(z)\}$ is larger than Σ_{θ_p} . We can easily see it when p=2. Indeed, consider the two dimensional Ornstein-Uhlenbeck operator

$$(\mathcal{A}u)(x,y) = \frac{1}{2}\Delta u(x,y) - (ax - by)D_x u(x,y) - ayD_y u(x,y), \qquad (x,y) \in \mathbb{R}^2,$$

where a and b are real constants, with a > 0 and $b \neq 0$. A straightforward computation shows that the associated Ornstein-Uhlenbeck semigroup is not symmetric in L^2_{μ} . Therefore, θ_2 is strictly less than $\pi/2$. Moreover, since

$$e^{zB} = \begin{pmatrix} e^{-az} \ bz e^{-az} \\ 0 \ e^{-az} \end{pmatrix}, \qquad z \in \mathbb{C},$$

then

$$\lim_{\substack{|z|\to +\infty\\ \|Q_{\infty}^{1/2}e^{zB^*}Q_{\infty}^{-1/2}\|_{\infty}\leq 1}}=0,$$

for any $\theta < \pi/2$, so that there exist some points $z \in \mathbb{C}$ which satisfy (9.3.70) and do not belong to Σ_{θ_2} .

9.3.4 Hermite polynomials

In this subsection we show some relations connecting the Hermite polynomials and the Ornstein-Uhlenbeck operators. For this purpose, we partially follow the approach in [109].

For any multi-index α , the N-dimensional Hermite polynomial $H_{\alpha}^{(N)}$ is defined by

$$H_{\alpha}^{(N)}(x) = (-1)^{|\alpha|} e^{\frac{1}{2}|x|^2} D^{\alpha} e^{-\frac{1}{2}|x|^2}, \qquad x \in \mathbb{R}^N.$$

Equivalently,

$$H_{\alpha}^{(N)}(x) = \prod_{k=1}^{N} H_{\alpha_k}^{(1)}(x_k), \qquad x \in \mathbb{R}^N,$$
 (9.3.72)

where $\alpha = (\alpha_1, \dots, \alpha_N)$.

Throughout this first part of the subsection we denote by \mathcal{A} the Ornstein-Uhlenbeck operator \mathcal{A} defined in $W^{2,2}_{\mu}$ by

$$\mathcal{A}u(x) = \Delta u(x) - \langle x, Du(x) \rangle, \qquad x \in \mathbb{R}^N, \quad u \in W^{2,2}_{\mu}. \tag{9.3.73}$$

Here,

$$\mu(dx) = \mathcal{N}(0, I)(dx) = \frac{1}{\sqrt{(2\pi)^N}} e^{-\frac{1}{2}|x|^2} dx.$$

Let us first consider the one-dimensional case and we introduce the operators in $W^{1,2}_\mu$ defined by

$$\partial f(x) = f'(x), \qquad \partial^* f(x) = -f'(x) + x f(x), \qquad x \in \mathbb{R}, \qquad f \in W^{1,2}_\mu.$$

The notation is justified by the identity

$$\int_{\mathbb{R}} \partial f \, g d\mu = \int_{\mathbb{R}} f \partial^* g d\mu, \qquad f, g \in W^{1,2}_{\mu},$$

which follows integrating by parts, if $f, g \in C_c^{\infty}(\mathbb{R})$, and by approximation with compactly supported function for a general pair of functions $f, g \in W_{\mu}^{1,2}$ (see Lemmas 9.3.3 and 9.3.4).

With the notation now introduced, we can rewrite the Ornstein-Uhlenbeck operator \mathcal{A} in the more compact form

$$\mathcal{A}u = -\partial^* \partial u, \qquad x \in \mathbb{R}^N, \qquad u \in D(L).$$
 (9.3.74)

We summarize some basic properties of the Hermite polynomials. In particular we show that they are an orthonormal basis of L^2_{μ} consisting of eigenfunctions of L.

Proposition 9.3.27 For any $n \in \mathbb{N} \cup \{0\}$, the following properties are met:

(i) the Hermite polynomials $\{H_n^{(1)}\}$ can be equivalently defined by recurrence by

$$H_0^{(1)} = 1, H_n^{(1)} = \partial^* H_{n-1}^{(1)}, n \ge 1;$$

- (ii) $H_n^{(1)}$ is a polynomial with degree n with leading term x^n ;
- (iii) $\partial H_{n+1}^{(1)} = (n+1)H_n^{(1)};$
- (iv) $\mathcal{A}H_n^{(1)} = -nH_n^{(1)};$
- (v) $\{H_n^{(1)}/\sqrt{n!}\}$ is an orthonormal basis of L_μ^2 . Here, we set 0!=1.

Proof. The properties (i), (ii) and (iii) can be proved by induction. In particular, to prove the property (iii) one can take advantage of the fact that $[\partial, \partial^*] = I$ in $W^{2,2}_{\mu}$.

The property (iv) is now a straightforward consequence of (i), (iii) and (9.3.74). Indeed, we have

$$\mathcal{A}H_{n}^{(1)} = -\partial^{*}\partial H_{n}^{(1)} = -\partial^{*}(nH_{n-1}^{(1)}) = -nH_{n}^{(1)}. \tag{9.3.75}$$

Let us now prove the property (v). For this purpose we observe that each polynomial of order $m \in \mathbb{N}$ belongs to $\mathrm{span}\,(H_0^{(1)},\ldots,H_m^{(1)})$. Therefore, to prove that $\{H_n^{(1)}/\sqrt{n!}\}$ is a basis of L_μ^2 , it suffices to show that the set of all the polynomials are a dense subspace of L_μ^2 . So, let us fix a function $f \in L_\mu^2$ such that

$$0 = \int_{\mathbb{R}} f(x)x^n d\mu = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} f(x)x^n e^{-\frac{1}{2}x^2} dx, \qquad n \in \mathbb{N} \cup \{0\}.$$

Then, the Fourier transform of the function $x \mapsto f(x)e^{-x^2/2}$ is an entire function, whose derivatives at 0 vanish. Therefore, $f(x)e^{-x^2/2} = 0$ for any $x \in \mathbb{R}$ and, consequently, f = 0.

To conclude the proof, let us prove that $\{H_n^{(1)}/\sqrt{n!}\}_{n\in\mathbb{N}}$ is an orthonormal system in L^2_{μ} . Fix $m,n\in\mathbb{N}$ with $m\leq n$ and denote by $(\cdot,\cdot)_{2,\mu}$ the Euclidean inner product in L^2_{μ} . Then, we have

$$(H_m^{(1)}, H_n^{(1)})_{2,\mu} = (H_m^{(1)}, \partial^* H_{n-1}^{(1)})_{2,\mu}$$

= $(\partial H_m^{(1)}, H_{n-1}^{(1)})_{2,\mu} = -n(H_{m-1}^{(1)}, H_{n-1}^{(1)})_{2,\mu}$

and, iterating the argument, we finally get

$$(H_m^{(1)}, H_n^{(1)})_{2,\mu} = \frac{n!}{(n-m)!} (H_0^{(1)}, H_{n-m}^{(1)})_{2,\mu}.$$

Hence, if n = m, we get $||H_n^{(1)}||_2 = \sqrt{n!}$, while, if m < n, we have

$$(H_0^{(1)},H_{n-m}^{(1)})_{2,\mu}=(\partial H_0^{(1)},H_{n-m-1}^{(1)})_{2,\mu}=0$$

and, consequently,
$$(H_m^{(1)}, H_n^{(1)})_{2,\mu} = 0$$
. The property (v) now follows.

We now consider the multidimensional case N > 1. According to the onedimensional case, for any k = 1, ..., N, we define the operators ∂_k and ∂_k^* in $W_{\mu}^{1,2}$ by setting

$$\partial_k f(x) = \frac{\partial f}{\partial x_k}(x), \qquad \partial_k^* f(x) = -\frac{\partial f}{\partial x_k}(x) + x_k f(x), \qquad x \in \mathbb{R}^N.$$

With this notation the Ornstein-Uhlenbeck operator in (9.3.73) can be rewritten in the compact form $\mathcal{A} = -\sum_{k=1}^{N} \partial_k^* \partial_k$. We can now prove the N-dimensional version of Proposition 9.3.27.

Proposition 9.3.28 The following properties are met:

- (i) $H_{\alpha+e_k}^{(N)} = \partial_k^* H_{\alpha}^{(N)}$, for any multi-index α and any $k \in \{1, \dots, N\}$;
- (ii) for any multi-index α , $H_{\alpha}^{(N)}$ is a polynomial with degree $|\alpha|$, whose leading term is given by $x^{\alpha} = \prod_{i=1}^{N} x^{\alpha_i}$, and it depends only on the variables x_i such that $\alpha_i \neq 0$;
- (iii) $\partial_k H_{\alpha}^{(N)} = \alpha_k H_{\alpha e_{\iota}}^{(N)};$
- (iv) $\mathcal{A}H_{\alpha}^{(N)} = -|\alpha|H_{\alpha}^{(N)}$:
- (v) $\{H_{\alpha}^{(N)}/\sqrt{\alpha!}\}_{\alpha\in\Lambda}$ is an orthonormal basis of L^2_{μ} .

Proof. The properties (i)-(iii) are immediate consequences of the results in Proposition 9.3.27 and formula (9.3.72), whereas the property (iv) can be checked adapting the proof of the property (v) in the quoted proposition.

Propositions 9.3.27 and 9.3.28 can be easily extended to the more general Ornstein-Uhlenbeck operator

$$\mathcal{A}u(x) = \frac{1}{2} \text{Tr}\left(QD^2 u(x)\right) + \langle Bx, Du(x)\rangle, \qquad x \in \mathbb{R}^N, \tag{9.3.76}$$

with $\sigma(B) \subset \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda < 0\}.$

Corollary 9.3.29 Let μ be the invariant measure of the Ornstein-Uhlenbeck semigroup associated with the operator A in (9.3.76) and let R be a nonsingular matrix such that $RQ_{\infty}R^* = \operatorname{diag}(\lambda_1, \ldots, \lambda_N)$. Then, the family of polynomials $\{K_{\alpha}^{(N)}\}$ defined by $K_{\alpha}^{N}(x) = \tilde{H}_{\alpha}^{(N)}(Rx)$, where

$$\tilde{H}_{\alpha}^{(N)}(x) = \prod_{k=1}^{N} H_{\alpha_k}^{(1)} \left(\frac{x_k}{\sqrt{\lambda_k}}\right), \qquad x \in \mathbb{R}^N,$$

is a complete orthonormal set in L^2_{μ} .

Proof. Let $D=\operatorname{diag}\left(\sqrt{\lambda_1},\ldots,\sqrt{\lambda_N}\right)$ and set $M=D^{-1}R$. From the remarks before Lemma 9.3.15, it follows that the operator $\Phi_{M^{-1}}$, defined by $(\Phi_{M^{-1}}f)(x)=f(M^{-1}x)$ for any $x\in\mathbb{R}^N$ and any function $f:\mathbb{R}^N\to\mathbb{R}$, is a linear isometry between L^p_μ and $L^p_{\overline{\mu}}$, where

$$\tilde{\mu}(dx) = \frac{1}{\sqrt{(2\pi)^N}} \exp\left(-\sum_{i=1}^N \frac{x_i^2}{2}\right) dx.$$

The assertion now follows from the property (iv) in Proposition 9.3.28.

9.4 The Ornstein-Uhlenbeck operator in $L^p(\mathbb{R}^N)$

To conclude the analysis of the Ornstein-Uhlenbeck operator, in this section we recall some results concerning the realization of the Ornstein-Uhlenbeck operator \mathcal{A} in $L^p(\mathbb{R}^N)$. Here, B is any nonzero matrix and Q is strictly positive definite. The results that we present are taken from [112, 118].

Proposition 9.4.1 For any $p \in [1, +\infty)$ the Ornstein-Uhlenbeck semigroup $\{T(t)\}$ can be extended to a strongly continuous semigroup (which we still denote by $\{T(t)\}$) to $L^p(\mathbb{R}^N)$. For any $f \in L^p(\mathbb{R}^N)$ and any t > 0, T(t)f is given by the right-hand side of (9.1.5). Finally,

$$||T(t)||_{L(L^p(\mathbb{R}^N))} \le e^{-\frac{t}{p}\text{Tr }B}, \qquad t > 0.$$
 (9.4.1)

Proof. Since $C_c^{\infty}(\mathbb{R}^N)$ is dense in $L^p(\mathbb{R}^N)$ for any $p \in [1, +\infty)$, it suffices to prove that T(t)f tends to f in $L^p(\mathbb{R}^N)$ as t tends to 0^+ , and that

$$||T(t)f||_{L^p(\mathbb{R}^N)} \le e^{-\frac{t}{p}\operatorname{Tr} B} ||f||_{L^p(\mathbb{R}^N)}, \qquad t > 0,$$
 (9.4.2)

for any $f \in C_c^{\infty}(\mathbb{R}^N)$. Denote by $\{S(t)\}$ the semigroup in $L^p(\mathbb{R}^N)$ defined by $(S(t)f)(x) = f(e^{tB}x)$ for any t > 0, any $x \in \mathbb{R}^N$ and any $f \in L^p(\mathbb{R}^N)$. As it is easily seen, for any $f \in C_c^{\infty}(\mathbb{R}^N)$ and any t > 0, we can write $T(t)f = S(t) \circ G_t$, where

$$(G_t f)(x) = \frac{1}{\sqrt{(2\pi)^N \det Q_t}} \int_{\mathbb{R}^N} e^{-\frac{1}{2}\langle Q_t^{-1} y, y \rangle} f(x - y) dy, \qquad t > 0, \ x \in \mathbb{R}^N.$$

Let us show that $\{S(t)\}$ is a strongly continuous (semi)group in $L^p(\mathbb{R}^N)$ and that

$$||S(t)||_{L(L^p(\mathbb{R}^N))} \le e^{-\frac{t}{p}\operatorname{Tr} B}, \qquad t > 0.$$
 (9.4.3)

The estimate (9.4.3) easily follows by a change of variable in the integral defining $||S(t)f||_p$, recalling that $\det(e^{-tB}) = e^{-t\operatorname{Tr} B}$ for any $t \in \mathbb{R}$.

To show that $\{S(t)\}$ is a strongly continuous semigroup, we observe that if $f \in C_c^{\infty}(\mathbb{R}^N)$, then S(t)f converges to f uniformly in \mathbb{R}^N as t tends to 0^+ . Since the support of S(t)f is compact in \mathbb{R}^N and there exists R > 0 such that $\operatorname{supp}(S(t)f) \subset B(R)$ for any $t \in [0,1]$, then S(t)f converges to f also in $L^p(\mathbb{R}^N)$ as t tends to 0^+ . Now, using the density of $C_c^{\infty}(\mathbb{R}^N)$ in $L^p(\mathbb{R}^N)$ and taking (9.4.3) into account, from Proposition A.1.2 we deduce that S(t)f converges to f as t tends to 0^+ for any $f \in L^p(\mathbb{R}^N)$.

As far as $\{G_t\}$ is concerned, we observe that the Young inequality implies that, for any t > 0, G_t is a contraction in $L^p(\mathbb{R}^N)$. Moreover, standard arguments show that $G_t f$ converges to f in $L^p(\mathbb{R}^N)$, as t tends to 0^+ , for any $f \in L^p(\mathbb{R}^N)$. Hence, the semigroup $\{T(t)\}$ is strongly continuous in $L^p(\mathbb{R}^N)$. Finally, (9.4.2) easily follows from (9.4.3) and the contractiveness of $\{G_t\}$.

In the rest of the section we denote by $A_p: D(A_p) \subset L^p(\mathbb{R}^N) \to L^p(\mathbb{R}^N)$ the infinitesimal generator of $\{T(t)\}$ in $L^p(\mathbb{R}^N)$. The following proposition gives a complete characterization of the operator A_p and of its domain.

Proposition 9.4.2 For any $p \in (1, +\infty)$,

$$D(A_p) = \{ u \in W^{2,p}(\mathbb{R}^N) : Au \in L^p(\mathbb{R}^N) \},$$
 (9.4.4)

and $A_p u = \mathcal{A}u$ for any $u \in D(A_p)$. Moreover, there exist two positive constants C_1 and C_2 such that

$$C_1(\|u\|_p + \|\mathcal{A}u\|_p) \le \|u\|_{W^{2,p}(\mathbb{R}^N)} + \|x \mapsto \langle Bx, Du(x) \rangle\|_p \le C_2(\|u\|_p + \|\mathcal{A}u\|_p),$$

(9.4.5)
for any $u \in D(A_p)$. Finally, for any $p \in [1, +\infty)$, $C_c^{\infty}(\mathbb{R}^N)$ is a core of A_p .

Proof. First of all we observe that, using the same arguments as in Lemma 9.3.13, it can be easily checked that $C_c^{\infty}(\mathbb{R}^N)$ is a core of A_p for any $p \in [1, +\infty)$. In particular, this implies that A_p is the closure of $\mathcal{A}: C_c^{\infty}(\mathbb{R}^N) \to L^p(\mathbb{R}^N)$ for any $p \in [1, +\infty)$.

Now, let us prove the formulas (9.4.4) and (9.4.5) for any $p \in (1, +\infty)$. For this purpose, we adapt to the present situation the proof of Theorem 9.3.17. Let us introduce the operators $A_p^0: W^{2,p}(\mathbb{R}^N) \to L^p(\mathbb{R}^N)$, defined by $A_p^0 u = \frac{1}{2} \text{Tr}(QD^2 u)$ for any $u \in W^{2,p}(\mathbb{R}^N)$, and $B_p^0: \{u \in L^p(\mathbb{R}^N): x \mapsto \langle Bx, Du(x) \rangle \in L^p(\mathbb{R}^N) \} \to L^p(\mathbb{R}^N)$, defined by $B_p^0 u = \langle B \cdot, Du \rangle$ for any $u \in D(B_p^0)$, where Du is meant in the sense of distributions. Note that B_p^0 is the infinitesimal generator of the strongly continuous group $\{S(t)\}$ defined in the proof of Proposition 9.4.1.

Let us observe that since $C_c^{\infty}(\mathbb{R}^N)$ is a core of $D(A_p)$, and $A_p = A_p^0 + B_p^0$ in $C_c^{\infty}(\mathbb{R}^N)$, then A_p is the closure of the operator $A_p^0 + B_p^0 : D(A_p^0) \cap D(B_p^0) = \{u \in W^{2,p}(\mathbb{R}^N) : Au \in L^p(\mathbb{R}^N)\} \to L^p(\mathbb{R}^N)$. Therefore, to complete the proof, it suffices to show that the operator $A_p^0 + B_p^0 : D(A_p^0) \cap D(B_p^0) \to L^p(\mathbb{R}^N)$ is closed.

According to Proposition B.1.13(iv), the operator $\omega I - A_p^0$ admits bounded imaginary powers with power angle $\theta_{\omega I - A_p^0} = 0$ for any $\omega > 0$. Similarly, taking the estimate (9.4.3) into account and applying Proposition B.1.13(iii), we deduce that the operator $\omega I - B_p^0$ admits bounded imaginary powers for any $\omega > -(\text{Tr }B)/p$, with power angle $\theta_{\omega I - B_p^0} \leq \pi/2$.

To apply Theorem B.1.16, we need to show that the condition (B.1.6) is satisfied. For this purpose, following the arguments in the proof of Theorem 9.3.17, we replace the space $C_b^4(\mathbb{R}^N)$ with the space \mathcal{G}_k defined by

$$\mathcal{G}_k = \left\{ u \in W^{k,2}(\mathbb{R}^N) : x \mapsto (1 + |x|^2)^{k/2} D^\alpha u \in L^2(\mathbb{R}^N) \text{ for any } |\alpha| \le k \right\},$$

where k = k(p) is to be properly chosen in order that \mathcal{G}_k is contained in the domain of the commutator $[A_p^0, B_p^0]$. It is then easy to check that

$$[A_p^0, B_p^0]u = \operatorname{Tr}(QB^*D^2u), \qquad u \in \mathcal{G}_k.$$

Therefore, the operator $[A_p^0, B_p^0]R(1, A_p^0)$ extends to a bounded linear operator in $L^p(\mathbb{R}^N)$.

To make the arguments in the proof of Theorem 9.3.17 work, it suffices to show that, for any $k \in \mathbb{N}$, $R(\lambda, A_p^0)$ and $R(\lambda, B_p^0)$ leave \mathcal{G}_k invariant, if $\operatorname{Re}\lambda$ is sufficiently large. Showing that \mathcal{G}_k is invariant for $R(\lambda, A_p^0)$ is an easy task, taking advantage of the Fourier transform. As far as $R(\lambda, B_p^0)$ is concerned, we begin by showing that the map

$$x \mapsto (\mathcal{E}_{\alpha}(S(t)u))(x) := (1 + |x|^2)^{k/2} (D^{\alpha}S(t)u)(x)$$

belongs to $L^2(\mathbb{R}^N)$ for any multi-index with length at most k. Since,

$$DS(t)f = e^{tB^*}S(t)Df, t > 0, f \in W^{1,p}(\mathbb{R}^N), (9.4.6)$$

we can limit ourselves to proving that $\mathcal{E}_0 u \in L^2(\mathbb{R}^N)$. For this purpose, we observe that

$$|x|^k |(S(t)u)(x)| \le ||e^{-tB}||_{\infty}^k (S(t)(\mathcal{E}_0 u))(x), \qquad t > 0, \ x \in \mathbb{R}^N.$$
 (9.4.7)

From (9.4.7) we easily deduce that $\mathcal{E}_0(S(t)f) \in L^2(\mathbb{R}^N)$. Now, iterating (9.4.6) and taking (9.4.7) into account, we can easily show that, up to replacing C_k and ω_k with larger constants,

$$\|\mathcal{E}_{\alpha}(S(t)u)\|_{L^{2}(\mathbb{R}^{N})} \le C_{k}e^{\omega_{k}t}\|\mathcal{E}_{0}(|D^{|\alpha|}u|)\|_{L^{2}(\mathbb{R}^{N})}, \qquad t > 0, \tag{9.4.8}$$

for any multi-index α with length k. Since $R(\lambda, B_p^0)$ is the Laplace transform of the semigroup $\{S(t)\}$ (see the formula (B.1.3)), from (9.4.8) we deduce that if Re λ is sufficiently large, then $R(\lambda, B_p^0)$ maps \mathcal{G}_k into itself. This completes the proof.

To conclude this section we characterize the spectrum of $(A_p, D(A_p))$. Since the proof is rather technical we prefer not to go into details referring the reader to the original paper [112] for the proof. **Theorem 9.4.3** Suppose that either the eigenvalues of the matrix B or the eigenvalues of the matrix -B have all negative real parts. Then, for any $p \in [1, +\infty)$, it holds that

$$\sigma(A_p) = \{ \lambda \in \mathbb{C} : \operatorname{Re}\lambda \le -(\operatorname{Tr}B)/p \}. \tag{9.4.9}$$

Proof. See [112, Theorems 4.4, 4.7, 4.11 & 4.12].

The following proposition is now a straightforward consequence of the formula (9.4.9).

Proposition 9.4.4 For any $p \in [1, +\infty)$ the Ornstein-Uhlenbeck semigroup is not analytic in $L^p(\mathbb{R}^N)$.

Remark 9.4.5 The results in Theorem 9.4.3 should be compared with the corresponding results in Theorem 9.3.22. In particular, it should be noticed that the spectrum of the Ornstein-Uhlenbeck operator in $L^p(\mathbb{R}^N)$ strictly depends on p (unless in the case when $\operatorname{Tr} B = 0$), whereas the spectrum in L^p_μ is p-invariant.

Chapter 10

A class of nonanalytic Markov semigroups in $C_b(\mathbb{R}^N)$ and in $L^p(\mathbb{R}^N, \mu)$

10.0 Introduction

In Chapter 9 we have seen that the Ornstein-Uhlenbeck semigroup is neither analytic nor strongly continuous in $BUC(\mathbb{R}^N)$. In this chapter we deal with Markov semigroups associated with the operator \mathcal{A} defined on smooth functions by

$$Au(x) = \Delta u(x) + \sum_{i,j=1}^{N} b_j(x)D_j u(x), \qquad x \in \mathbb{R}^N,$$
 (10.0.1)

where b_j (j = 1, ..., N) are unbounded Lipschitz continuous functions in \mathbb{R}^N . We provide conditions on $b = (b_1, ..., b_N)$ implying that $\{T(t)\}$ is not analytic in $C_b(\mathbb{R}^N)$. Next, we consider the semigroup $\{T(t)\}$ in $L^p(\mathbb{R}^N, \mu)$ (μ being the invariant measure associated with the operator \mathcal{A}) and we still provide suitable growth conditions on b implying that $\{T(t)\}$ is not analytic in $L^p(\mathbb{R}^N, \mu)$. The results of this chapter are taken from [117].

10.1 Nonanalytic semigroups in $C_b(\mathbb{R}^N)$

The main result of this section is Theorem 10.1.1. We recall that $(\widehat{A}, D(\widehat{A}))$ denotes the weak generator of the semigroup $\{T(t)\}$. See Section 2.3.

Theorem 10.1.1 Assume that the coefficients b_j (j = 1,...,N) in (10.0.1) are locally Lipschitz continuous. Further, assume that there exist three sequences $\{r_n\}$, $\{\lambda_n\} \subset (0,+\infty)$ and $\{\sigma_n\} \subset \mathbb{R}^N$ such that

$$r_n \le M, \qquad n \in \mathbb{N},$$
 (10.1.1)

for some positive constant M;

$$\lim_{n \to +\infty} \frac{r_n}{\lambda_n^2} = 0; \tag{10.1.2}$$

$$\lim_{n \to +\infty} \frac{r_n}{\lambda_n} b(\lambda_n x + \sigma_n) = h \in \mathbb{R}^N, \quad h \neq 0,$$
 (10.1.3)

uniformly with respect to x on compact subsets of \mathbb{R}^N . Then, $(\widehat{A}, D(\widehat{A}))$ is not sectorial in $C_b(\mathbb{R}^N)$. In particular, if $r_n = 1$ for any $n \in \mathbb{N}$, then the spectrum of $(\widehat{A}, D(\widehat{A}))$ contains the imaginary axis.

Proof. To prove the assertion we argue by contradiction. We show that, if $(\widehat{A}, D(\widehat{A}))$ were sectorial, then the semigroup $\{S(t)\} \subset L(C_0(\mathbb{R}^N))$, defined by (S(t)f)(x) = f(x+th) for any t > 0, any $x \in \mathbb{R}^N$ and any $f \in C_0(\mathbb{R}^N)$, should be analytic. But this cannot be the case since the infinitesimal generator of $\{S(t)\}$, is the operator $B: \{u \in C_0(\mathbb{R}^N): \frac{\partial u}{\partial h} \in C_0(\mathbb{R}^N)\} \to C_0(\mathbb{R}^N)$ defined by $Bu = \langle h, Du \rangle$ for any $u \in D(B)$, whose spectrum contains the imaginary axis. Here, h is as in (10.1.3).

So, let us assume that $(\widehat{A}, D(\widehat{A}))$ is sectorial and let us introduce, for any $n \in \mathbb{N}$ the isometries $I_n : C_b(\mathbb{R}^N) \to C_b(\mathbb{R}^N)$ defined by

$$(I_n u)(x) = u\left(\frac{x - \sigma_n}{\lambda_n}\right), \qquad x \in \mathbb{R}^N, \quad u \in C_b(\mathbb{R}^N), \quad n \ge 1.$$

For any $n \in \mathbb{N}$ we denote by \mathcal{A}_n the second-order differential operator defined on smooth functions by

$$(\mathcal{A}_n u)(x) = \frac{r_n}{\lambda_n^2} \Delta u(x) + \frac{r_n}{\lambda_n} \sum_{j=1}^N b_j (\lambda_n x + \sigma_n) D_j u(x), \qquad x \in \mathbb{R}^N, \quad (10.1.4)$$

and we denote by $\{T_n(t)\}$ the positive semigroup of contractions in $C_b(\mathbb{R}^N)$ associated with the operator \mathcal{A}_n . Since $\mathcal{A}_n = r_n I_n^{-1} \mathcal{A} I_n$, it is easy to check that, for any $n \in \mathbb{N}$, the weak generator \widehat{A}_n of the semigroup $\{T_n(t)\}$ is the operator $\widehat{A}_n : D(\widehat{A}_n) := I_n^{-1}(D(\widehat{A})) \to C_b(\mathbb{R}^N)$ defined by $\widehat{A}_n = r_n(I_n^{-1}\widehat{A}I_n)$. Therefore, $\rho(\widehat{A}_n) = r_n\rho(\widehat{A})$ and

$$R(\lambda, \widehat{A}_n) = \frac{1}{r_n} I_n^{-1} R(r_n^{-1} \lambda, \widehat{A}) I_n, \qquad (10.1.5)$$

for any $n \in \mathbb{N}$.

Since we are assuming that $(\widehat{A}, D(\widehat{A}))$ is sectorial, then there exist two positive constants C and K such that

$$||R(\lambda, \widehat{A})||_{L(C_b(\mathbb{R}^N))} \le \frac{C}{|\lambda|}, \qquad \text{Re } \lambda > K.$$
 (10.1.6)

Using (10.1.5), (10.1.6) and the assumptions on the sequence $\{r_n\}$, it is now immediate to check that

$$||R(\lambda, \widehat{A}_n)||_{L(C_b(\mathbb{R}^N))} \le \frac{C}{|\lambda|}, \qquad \text{Re } \lambda > MK.$$
 (10.1.7)

Now, we fix $f \in C_c^{\infty}(\mathbb{R}^N)$ and check that, for any $\lambda \in \mathbb{C}$, with $\operatorname{Re} \lambda > MK$, and any $x \in \mathbb{R}^N$, it holds that

$$\lim_{n \to +\infty} (R(\lambda, \widehat{A}_n)f)(x) = (R(\lambda, B)f)(x). \tag{10.1.8}$$

Of course, once (10.1.8) is proved, letting n tend to $+\infty$ in (10.1.7) we will get

 $||R(\lambda, B)f||_{\infty} \le \frac{C}{|\lambda|} ||f||_{\infty},$ Re $\lambda > MK$,

first for any $f \in C_c^{\infty}(\mathbb{R}^N)$ and then, by density, for any $f \in C_0(\mathbb{R}^N)$. Theorem B.2.7 then will imply that B is sectorial: a contradiction.

To show (10.1.8) we modify the proof of the classical Trotter-Kato theorem (see Theorem B.1.11).

As a first step we prove that, for any $f \in C_c^{\infty}(\mathbb{R}^N)$, the function $T_n(\cdot)f$ converges to $S(\cdot)f$ in $[0,T] \times \mathbb{R}^N$ as n tends to $+\infty$, for any T > 0. For this purpose, we begin by observing that, since S(t) maps $C_c^{\infty}(\mathbb{R}^N)$ into itself for any $t \geq 0$, and $C_c^{\infty}(\mathbb{R}^N)$ is contained in $D(\widehat{A}_n)$ (see Proposition 2.3.6), by Proposition 2.3.5,

$$\frac{d}{dt}(T_n(t)S(s)f)(x) = (T_n(t)\widehat{A}_nS(s)f)(x), \qquad s, t > 0, \quad x \in \mathbb{R}^N,$$

for any $f \in C_c^{\infty}(\mathbb{R}^N)$. Moreover, since $\{S(t)\}$ is a strongly continuous semigroup of contractions and the domain of its infinitesimal generator contains $C_c^{\infty}(\mathbb{R}^N)$, then the map $t \mapsto S(t)f$ is differentiable in $[0, +\infty)$ with values in $C_0(\mathbb{R}^N)$ and its derivative at t is BS(t)f. Therefore, we have

$$(T_n(t)f)(x) - (S(t)f)(x) = -\int_0^t \frac{d}{ds} \{ (T_n(t-s)S(s)f)(x) \} ds$$
$$= \int_0^t (T_n(t-s)(\widehat{A}_n - B)S(s)f)(x) ds,$$

for any t > 0 and any $x \in \mathbb{R}^N$, which implies that

$$||T_n(t)f - S(t)f||_{\infty} \le T \sup_{s \in [0,T]} ||(\widehat{A}_n - B)S(s)f||_{\infty},$$
 (10.1.9)

for any $t \in [0, T]$ and any T > 0. Let us observe that the right-hand side of (10.1.9) converges to 0 as n tends to $+\infty$. Indeed, recalling that $\widehat{A}_n S(s) f = \mathcal{A}_n S(s) f$ (see again Chapter 2), we have

$$|(\widehat{A}_n S(s)f)(x) - (BS(s)f)(x)|$$

$$\leq \frac{r_n}{\lambda_n^2} ||\Delta S(s)f||_{\infty} + \left| \frac{r_n}{\lambda_n} b(\lambda_n x + \sigma_n) - h \right| ||DS(s)f||_{\infty} \chi_{\text{supp}(S(s)f)}(x),$$

for any $s \in [0, T]$ and any $x \in \mathbb{R}^N$. Since $\operatorname{supp}(S(s)f) \subset \overline{B}(|h|T) + \operatorname{supp}(f)$ for any $s \in [0, T]$, using the conditions (10.1.2) and (10.1.3), we easily deduce

that $\|(\widehat{A}_n - B)S(s)f\|_{\infty}$ tends to 0 as n tends to $+\infty$. Therefore, from (10.1.9) it follows that $T_n(\cdot)$ tends to $S(\cdot)f$ uniformly in $[0,T] \times \mathbb{R}^N$.

Now, by (2.2.14) and (B.1.3) we can write

$$(R(\lambda, \widehat{A}_n)f)(x) - (R(\lambda, B)f)(x) = \int_0^{+\infty} e^{-\lambda t} (T_n(t) - S(t)f)(x) dt,$$

for any $x \in \mathbb{R}^N$. Consequently,

$$|(R(\lambda, \widehat{A}_n)f)(x) - (R(\lambda_0, B)f)(x)|$$

$$\leq \int_0^{+\infty} e^{-\operatorname{Re}\lambda t} |(T_n(t)f)(x) - (S(t)f)(x)| dt, \qquad (10.1.10)$$

for any $x \in \mathbb{R}^N$, and the right-hand side of (10.1.10) tends to 0 as n tends to $+\infty$, by the dominated convergence theorem. Therefore, (10.1.8) follows.

To conclude the proof, let us prove that if $r_n = 1$ for any $n \in \mathbb{N}$, then the spectrum of $(\widehat{A}, D(\widehat{A}))$ contains the imaginary axis. For this purpose, we observe that from (10.1.5) we get

$$\|R(\lambda,\widehat{A})\|_{L(C_b(\mathbb{R}^N))} = \|R(\lambda,\widehat{A}_n)\|_{L(C_b(\mathbb{R}^N))},$$

for any $\lambda \in \rho(\widehat{A}_n) = \rho(\widehat{A})$. Therefore, since $\rho(\widehat{A})$ contains the halfplane $\{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > 0\}$, by (10.1.8) we have

$$\|R(\lambda,B)\|_{L(C_b(\mathbb{R}^N))} \leq \limsup_{n \to +\infty} \|R(\lambda,\widehat{A}_n)\|_{L(C_b(\mathbb{R}^N))} = \|R(\lambda,\widehat{A})\|_{L(C_b(\mathbb{R}^N))}.$$

But, since $i\mathbb{R} \subset \sigma(B)$, it follows that

$$\lim_{\substack{\lambda \to is \\ \text{Re}\lambda > 0}} ||R(\lambda, A)||_{L(C_b(\mathbb{R}^N))} = +\infty,$$

for any s > 0, so that $i\mathbb{R} \subset \sigma(\widehat{A})$ (see Proposition A.3.4).

Now we look for sufficient conditions under which (10.1.1)-(10.1.3) are satisfied.

Corollary 10.1.2 Assume that there exist two sequences $\{\tau_n\} \subset \mathbb{R}^N$ and $\{\lambda_n\} \subset (0,+\infty)$ such that $b(\tau_n) \neq 0$ for any $n \in \mathbb{N}$, $\{\lambda_n\}$ is bounded, $\lambda_n|b(\tau_n)|$ tends to $+\infty$ as n tends to $+\infty$ and

$$\lim_{n \to +\infty} \frac{|b(\tau_n + \lambda_n x) - b(\tau_n)|}{|b(\tau_n)|} = 0,$$
(10.1.11)

uniformly on compact subsets of \mathbb{R}^N . Then, $(\widehat{A}, D(\widehat{A}))$ is not sectorial.

Proof. Let $\{\sigma_n\}$ be a subsequence of $\{\tau_n\}$ such that $b(\sigma_n)/|b(\sigma_n)|$ converges to some $h \in \mathbb{R}^N$ as n tends to $+\infty$ Then, if we set $r_n = \lambda_n |b(\sigma_n)|^{-1}$ for any $n \in \mathbb{N}$, it is immediate to check that the three sequences $\{r_n\}$, $\{\lambda_n\}$ and $\{\sigma_n\}$ satisfy the conditions (10.1.1)-(10.1.3).

Corollary 10.1.3 Let $b_j \in C^1(\mathbb{R}^N)$ for any j = 1, ..., N. Further, assume that there exists a sequence $\{\tau_n\} \subset \mathbb{R}^N$, diverging to $+\infty$, such that $|b(\tau_n)| \geq K|\tau_n|^{\alpha}$ for some $K, \alpha > 0$ and any $n \in \mathbb{N}$. If

$$\sum_{i,j=1}^{N} |D_i b_j(x)|^2 \le K(1+|x|^{\beta})^2, \qquad x \in \mathbb{R}^N,$$
(10.1.12)

for some $\beta < \alpha$, then the assumptions of Corollary 10.1.2 are satisfied.

Proof. Without loss of generality, we can assume that $\lambda_n = 1$ for any $n \in \mathbb{N}$. The mean value theorem yields

$$|b(\tau_n + \lambda_n x) - b(\tau_n)| \le R \sup_{0 \le \theta \le 1} \left(\sum_{i,j=1}^N |D_i b_j (\tau_n + \theta \lambda_n x)|^2 \right)^{\frac{1}{2}}$$

$$\le KR(1 + (|\tau_n| + MR)^{\beta}), \tag{10.1.13}$$

for any $|x| \leq R$. Dividing both the sides of (10.1.13) by $|b(\tau_n)|$, we easily see that the condition (10.1.11) is satisfied since, by assumption, $|b(\tau_n)|$ tend to $+\infty$ faster than $|\tau_n|^{\beta}$.

Remark 10.1.4 The conditions in Corollary 10.1.3 are always satisfied in the case when b_j are polynomials for any $j=1,\ldots,N$. Indeed, let b_i be the polynomial of maximal degree and define $\tau_n=nx_0$ for any $n\in\mathbb{N}$, where $x_0\in\mathbb{R}^N$ is such that the homogeneous part of b_i of maximum degree does not vanish at x_0 . It is immediate to check that

$$|b(\tau_n)| \ge |b_i(\tau_n)| \ge K n^{\deg(b_i)},$$

for some positive constant K > 0 and any n sufficiently large. Moreover, since $\sum_{i,j=1}^{N} |D_i b_j|^2$ is a polynomial of degree $2(\deg(b_i) - 1)$, the condition (10.1.12) is satisfied with $\beta = \deg(b_i) - 1 < \alpha := \deg(b_i)$.

Here we see, with techniques different from those in Chapter 9, that the nondegenerate Ornstein-Uhlenbeck semigroup is not analytic in $C_b(\mathbb{R}^N)$.

Corollary 10.1.5 Let $b_j \in C^1(\mathbb{R}^N)$ (j = 1, ..., N). Assume that there exist: (i) a sequence $\{\tau_n\} \subset \mathbb{R}^N$, such that $|b(\tau_n)|$ diverges to $+\infty$ as n tends to $+\infty$, (ii) positive numbers γ, δ, C such that

$$\left(\sum_{i,j=1}^{N} |D_i b_j(x)|^2\right)^{\frac{1}{2}} \le \gamma |b(x)|^{\frac{3}{2}} + C,$$

for any $x \in \mathbb{R}^N$ with $|x - \tau_n| \le \delta |b(\tau_n)|^{-1/2}$. Then $(\widehat{A}, D(\widehat{A}))$ is not sectorial in $C_b(\mathbb{R}^N)$.

Proof. We begin the proof by observing that, without loss of generality, we can assume that $\delta < (4\gamma)^{-1}$ and that $b(\tau_n) \neq 0$ for any $n \in \mathbb{N}$. Next we fix an arbitrary $n \in \mathbb{N}$ and observe that, for any $s \in (0, \delta)$ and any $|x| \leq s|b(\tau_n)|^{-1/2}$, we have

$$\frac{|b(\tau_n + x) - b(\tau_n)|}{|b(\tau_n)|} \le \frac{|x|}{|b(\tau_n)|} \sup_{0 \le \theta \le 1} \left(\sum_{i,j=1}^N |D_i b_j(\tau_n + \theta x)|^2 \right)^{\frac{1}{2}} \\
\le \gamma s \sup_{0 \le \theta \le 1} \frac{|b(\tau_n + \theta x)|^{\frac{3}{2}}}{|b(\tau_n)|^{\frac{3}{2}}} + Cs|b(\tau_n)|^{-\frac{3}{2}}.$$
(10.1.14)

We set

$$F_n(s) = \sup_{|x| \le s|b(\tau_n)|^{-\frac{1}{2}}} \frac{|b(\tau_n + x)|}{|b(\tau_n)|}, \qquad s \in [0, \delta).$$

From (10.1.14) we get

$$F_n(s) \le 1 + \gamma s(F_n(s))^{\frac{3}{2}} + Cs|b(\tau_n)|^{-\frac{3}{2}}, \qquad s \in [0, \delta).$$

We claim that $F_n(s) < 4$ for any $s \in [0, \delta)$ and any n sufficiently large. To prove the claim, we begin by noting that, for n large enough, we have

$$1 + Cs|b(\tau_n)|^{-\frac{3}{2}} \le 1 + C\delta|b(\tau_n)|^{-\frac{3}{2}} \le 2.$$

It follows that

$$F_n(s) \le 2 + \gamma s(F_n(s))^{\frac{3}{2}}, \qquad s \in [0, \delta).$$
 (10.1.15)

Recalling that $4\delta\gamma < 1$, we get

$$F_n(s) < 2 + \frac{1}{4}(F_n(S))^{\frac{3}{2}}, \qquad s \in [0, \delta).$$

Since F_n is continuous in $[0, +\infty)$ with $F_n(0) = 1$ for any $n \in \mathbb{N}$, and [0, 4) is the biggest interval containing 1 in which the inequality $x < 2 + x^{3/2}/4$ holds, we easily deduce that $F_n(s) \in [0, 4)$ for any $s \in [0, \delta)$, and the claim follows. Thus, for n large enough, the equation (10.1.14) yields, for $|x| \leq s|b(\tau_n)|^{-1/2}$,

$$\frac{|b(\tau_n + x) - b(\tau_n)|}{|b(\tau_n)|} \le 8\gamma s + Cs|b(\tau_n)|^{-\frac{3}{2}}.$$
 (10.1.16)

Now, fix R > 0 and let $\lambda_n = |b(\tau_n)|^{-\frac{3}{4}}$. For any $y \in B(R)$ and n sufficiently large such that $R|b(\tau_n)|^{-1/4} < \delta$, we have $|\lambda_n y| \leq s|b(\tau_n)|^{-1/2}$ with $s = R|b(\tau_n)|^{-1/4}$. Applying (10.1.16), we get

$$\frac{|b(\tau_n + \lambda_n x) - b(\tau_n)|}{|b(\tau_n)|} \le 8R\gamma |b(\tau_n)|^{-\frac{1}{4}} + CR|b(\tau_n)|^{-\frac{7}{4}}.$$

Corollary 10.1.2 now allows us to conclude the proof.

Next corollary gives suitable conditions on the drift term of the operator \mathcal{A} implying that the spectrum of \widehat{A} contains the imaginary axis.

Corollary 10.1.6 Assume that $b_j \in C^1(\mathbb{R})$ for any j = 1, ..., N and there exists a sequence $\{\tau_n\} \subset \mathbb{R}^N$, diverging to infinity, such that $|b(\tau_n)|$ tends to $+\infty$ as n tends to $+\infty$. Further, assume that $\sum_{i,j=1}^N |D_i b_j(x)|^2$ and |b(x)|/|x| tend to 0 as |x| tends to $+\infty$. Then, the spectrum of $(\widehat{A}, D(\widehat{A}))$ contains the imaginary axis. As a consequence, $(\widehat{A}, D(\widehat{A}))$ is not sectorial.

Proof. Fix R > 0. By the mean value theorem we have

$$|b(\tau_n + \lambda_n x) - b(\tau_n)| \le R\lambda_n \sup_{0 \le \theta \le 1} \left(\sum_{i,j=1}^N |D_i b_j(\tau_n + \theta \lambda_n x)|^2 \right)^{\frac{1}{2}},$$

for any $|x| \leq R$ and any sequence $\{\lambda_n\} \subset (0, +\infty)$.

We now take $\lambda_n = |b(\tau_n)|$ and $r_n = 1$ for any $n \in \mathbb{N}$, and we show that

$$\lim_{n \to +\infty} \sup_{0 \le \theta \le 1} \left(\sum_{i,j=1}^{N} |D_i b_j (\tau_n + \theta \lambda_n x)|^2 \right)^{\frac{1}{2}} = 0,$$

uniformly with respect to $x \in \overline{B}(R)$. For this purpose it suffices to show that $|\tau_n + \theta \lambda_n x|$ tends to $+\infty$ as n tends to $+\infty$, uniformly with respect to $\theta \in [0, 1]$ and $x \in \overline{B}(R)$. Since b is sublinear, we have

$$|\tau_n + \theta \lambda_n x| \ge |\tau_n| \left(1 - \frac{R|b(\tau_n)|}{|\tau_n|}\right) \ge \frac{1}{2} |\tau_n|,$$

for any $\theta \in [0, 1]$, any $x \in \overline{B}(R)$ and any n sufficiently large. Therefore,

$$\lim_{n \to +\infty} \inf_{\theta \in [0,1] \atop x \in \overline{B}(R)} |\tau_n + \theta \lambda_n x| = +\infty.$$

Theorem 10.1.1 and Corollary 10.1.2 allow us to conclude.

Up to now we have shown sufficient conditions implying that $(\widehat{A}, D(\widehat{A}))$ is not sectorial. In the next theorem we provide suitable conditions on the drift coefficient of the operator \mathcal{A} which guarantee that $(\widehat{A}, D(\widehat{A}))$ is sectorial in $C_b(\mathbb{R}^N)$.

Theorem 10.1.7 Let $b_j \in C^1(\mathbb{R}^N) \cap L^p(\mathbb{R}^N)$, for some $p \in (N, +\infty)$ and any j = 1, ..., N, be such that div b is bounded from below. Then, $(\widehat{A}, D(\widehat{A}))$ is sectorial in $C_b(\mathbb{R}^N)$.

Proof. As a first step we introduce the formal adjoint of A, i.e., the operator

$$\mathcal{A}^* = \Delta - \sum_{i=1}^{N} b_i D_i - \operatorname{div} b,$$

and we prove that its realization A^* in $L^1(\mathbb{R}^N)$ with domain $D(A^*) = \{u \in L^1(\mathbb{R}^N) : \Delta u, (\operatorname{div} b)u \in L^1(\mathbb{R}^N)\}$ (Δu being meant in the sense of distributions) generates an analytic semigroup in $L^1(\mathbb{R}^N)$. Then, we use this result to show that $(\widehat{A}, D(\widehat{A}))$ is sectorial.

Let us observe that, since div b is bounded from below and it is locally bounded, according to [82, Theorem I, Lemmas 6 & 8] the operator

$$B_0 = \Delta - \operatorname{div} b + \inf_{\mathbb{R}^N} (\operatorname{div} b) := B + \inf_{\mathbb{R}^N} (\operatorname{div} b),$$

with domain $D(B_0) = D(A^*)$, generates an analytic semigroup in $L^1(\mathbb{R}^N)$. Moreover,

$$\|\Delta u\|_{L^1(\mathbb{R}^N)} + \|(\operatorname{div} b)u\|_{L^1(\mathbb{R})} \le C(\|u\|_{L^1(\mathbb{R}^N)} + \|Bu\|_{L^1(\mathbb{R}^N)}), \qquad (10.1.17)$$

for any $u \in D(A^*)$.

We now recall that if $u \in L^1(\mathbb{R}^N)$ is such that $\Delta u \in L^1(\mathbb{R}^N)$, then $u \in W^{1,q}(\mathbb{R}^N)$ for any $q \in [1, N/(N-1))$ and there exists a positive constant C = C(N,q), independent of u, such that

$$||Du||_{L^{q}(\mathbb{R}^{N})} \le C(N, q) \left(||u||_{L^{1}(\mathbb{R}^{N})} + ||\Delta u||_{L^{1}(\mathbb{R}^{N})} \right), \tag{10.1.18}$$

see [140, Theorem 5.8]. Applying the estimate (10.1.18), with q = p/(p-1), to the function $v_{\lambda} : \mathbb{R}^{N} \to \mathbb{R}$ ($\lambda > 0$), defined by $v(x) = u(\lambda x)$ for any $x \in \mathbb{R}^{N}$, and minimizing with respect to $\lambda > 0$, we get the following inequality:

$$\|Du\|_{L^{\frac{p}{p-1}}(\mathbb{R}^N)} \leq 2C(N,p)\|u\|_{L^1(\mathbb{R}^N)}^{\frac{p-N}{2p}}\|\Delta u\|_{L^1(\mathbb{R}^N)}^{\frac{N+p}{2p}}.$$

Hölder's inequality then gives

$$\begin{split} \left\| \sum_{i=1}^{N} b_{i} D_{i} u \right\|_{L^{1}(\mathbb{R}^{N})} &\leq \|b\|_{L^{p}(\mathbb{R}^{N})} \|D u\|_{L^{\frac{p}{p-1}}(\mathbb{R}^{N})} \\ &\leq 2C(N, p) \|b\|_{L^{p}(\mathbb{R}^{N})} \|u\|_{L^{1}(\mathbb{R}^{N})}^{\frac{p-N}{2p}} \|\Delta u\|_{L^{1}(\mathbb{R}^{N})}^{\frac{p+N}{2p}}, \quad (10.1.19) \end{split}$$

for any $u \in L^1(\mathbb{R}^N)$ such that $\Delta u \in L^1(\mathbb{R}^N)$.

Now, from (10.1.17) and (10.1.19) it follows that the term $u \mapsto \langle b, Du \rangle$ is a small perturbation of (B, D(B)). Therefore, according to Theorem B.2.10, $(A^*, D(A^*))$ generates an analytic semigroup in $L^1(\mathbb{R}^N)$.

We are now in a position to prove that $(\widehat{A}, D(\widehat{A}))$ is sectorial in $C_b(\mathbb{R}^N)$. Possibly replacing A^* with $A^* - \omega$, with ω sufficiently large, we can assume that

$$||R(\lambda, A^*)||_{L(L^1(\mathbb{R}^N))} \le \frac{M}{|\lambda|},$$

for any $\lambda \in \mathbb{C}$ with positive real part. Let us set $D_{\lambda} = (\lambda - A^*)(C_c^{\infty}(\mathbb{R}^N))$ for such λ 's. Since $C_c^{\infty}(\mathbb{R}^N)$ is a core of A^* (see [82, Theorem IV]), then D_{λ} is dense in $L^1(\mathbb{R}^N)$ for any $\text{Re } \lambda > 0$. Therefore, if $u \in D(\widehat{A}) \subset C_b(\mathbb{R}^N)$ and λ is as above, then

$$\begin{aligned} \|u\|_{\infty} &= \sup \left\{ \int_{\mathbb{R}^N} u \varphi \, dx : \varphi \in D_{\lambda}, \ \|\varphi\|_{L^1(\mathbb{R}^N)} \le 1 \right\} \\ &\leq \sup \left\{ \int_{\mathbb{R}^N} u (\lambda - A^*) v \, dx : v \in C_c^{\infty}(\mathbb{R}^N), \ \|v\|_{L^1(\mathbb{R}^N)} \le \frac{M}{|\lambda|} \right\} \\ &= \sup \left\{ \int_{\mathbb{R}^N} v (\lambda - \widehat{A}) u \, dx : v \in C_c^{\infty}(\mathbb{R}^N), \ \|v\|_{L^1(\mathbb{R}^N)} \le \frac{M}{|\lambda|} \right\} \\ &\leq \frac{M}{|\lambda|} \|(\lambda - \widehat{A}) u\|_{\infty}. \end{aligned}$$

Therefore, $||R(\lambda, \widehat{A})||_{L(C_b(\mathbb{R}^N))} \leq M/|\lambda|$ and the assertion follows from Theorem B.2.7.

Example 10.1.8 Let \mathcal{A} be the operator defined on smooth functions of two variables by

$$Au(x,y) = D_x^2 u(x,y) + D_y^2 u(x,y) + b_0(\sqrt{x^2 + y^2})(xu_y - yu_x), \qquad (x,y) \in \mathbb{R}^2,$$

where $b_0 \in C^1(\mathbb{R})$. Note that the drift $b(x,y) = b_0(\sqrt{x^2 + y^2})(-y,x)$ has divergence 0. The assumptions of Theorem 10.1.7 hold if

$$\int_0^{+\infty} r|b(r)|^q dr < +\infty,$$

for some q > 2. In such a case, the semigroup associated with the operator \mathcal{A} is analytic in $C_b(\mathbb{R}^N)$.

10.2 Nonanalytic semigroups in $L^p(\mathbb{R}^N, \mu)$

In this section we introduce a class of Markov semigroups $\{T(t)\}$, which admit an invariant measure μ , but which are not analytic in $L^p_{\mu} := L^p(\mathbb{R}^N, \mu)$. Let \mathcal{A} be the second-order elliptic operator defined by

$$\mathcal{A}\varphi = \Delta\varphi - \sum_{i=1}^{N} (D_i U + G_i) D_i \varphi, \qquad (10.2.1)$$

on smooth functions, where U and G satisfy the following hypotheses:

Hypotheses 10.2.1 (i) $U \in C^2(\mathbb{R}^N)$ and $e^{-U} \in L^1(\mathbb{R}^N)$;

(ii) $G \in C^1(\mathbb{R}^N, \mathbb{R}^N)$ and satisfies

$$\int_{\mathbb{R}^N} |G| e^{-U} dx < +\infty$$

and

$$\operatorname{div} G - \sum_{i=1}^{N} G_i D_i U = 0.$$
 (10.2.2)

These conditions will be kept throughout this section and ensure that the semigroup $\{T(t)\}$ associated with \mathcal{A} admits the measure

$$d\mu = Me^{-U(x)}dx, \qquad M = \left(\int_{\mathbb{R}^N} e^{-U(y)}dy\right)^{-1},$$

as its (unique) invariant measure (see Subsection 8.1.4). As it has been shown in Proposition 8.1.8, the semigroup $\{T(t)\}$ extends to a strongly continuous semigroup of contractions in L^p_μ for any $p \in [1, +\infty)$. Using the same notations as in Chapter 8, we still denote by $\{T(t)\}$ the so extended semigroup, and by $(L_p, D(L_p))$ its infinitesimal generator. Moreover, we write L instead of L_2 . Let us state the main result of this section.

Theorem 10.2.2 Suppose that there exist some constants k > 0, $0 < \beta < \alpha$ and a sequence $\{\sigma_n\} \subset \mathbb{R}^N$, diverging to infinity, such that

(i)
$$|G(\sigma_n)| \ge k |\sigma_n|^{\alpha}$$
, $n \ge 1$;
(ii) $|\Delta U(x)| + |DU(x)| + |DG(x)| < k(1 + |x|^{\beta})$, $x \in \mathbb{R}^N$. (10.2.3)

Then, the semigroup $\{T(t)\}$ is not analytic in L^p_μ for any $p \in [1, +\infty)$.

Proof. We first prove the assertion in the case when p=2. For this purpose, let us consider the isometry $R:L^2(\mathbb{R}^N)\to L^2_\mu$ defined by $Rf=f\psi$ for any $f\in L^2(\mathbb{R}^N)$, where

$$\psi(x) = M^{-\frac{1}{2}} e^{\frac{1}{2}U(x)}, \qquad x \in \mathbb{R}^N.$$

Further, let us define the operator $\widetilde{L}_2: D(\widetilde{L}_2) \subset L^2(\mathbb{R}^N) \to L^2(\mathbb{R}^N)$, with domain $D(\widetilde{L}_2) = \{g \in L^2(\mathbb{R}^N) : \psi g \in D(L_2)\}$, by setting

$$\widetilde{L}_2 u = R^{-1} L_2 R u, \qquad u \in D(\widetilde{L}_2).$$

Since, as it has already been recalled, the semigroup $\{T(t)\}$ is strongly continuous in L^2_{μ} , then it is immediate to check that \widetilde{L}_2 is the generator of a

strongly continuous semigroup of contractions in $L^2(\mathbb{R}^n)$ (say $\{\widetilde{T}(t)\}$) defined by

$$\widetilde{T}(t)f = R^{-1}T(t)(Rf), \qquad t \ge 0, \quad f \in L^2(\mathbb{R}^N).$$

According to Propositions 2.3.6 and 8.1.9, $C_c^{\infty}(\mathbb{R}^N) \subset D(L_2)$. Therefore, $C_c^{\infty}(\mathbb{R}^N)$ is contained in $D(\widetilde{L}_2)$ and, since $L_2u = \mathcal{A}u$ for any $u \in C_c^{\infty}(\mathbb{R}^N)$, then

$$(\widetilde{L}_2 u)(x) = \Delta u(x) - \sum_{j=1}^{N} G_j(x) D_j u(x) - c(x) u(x), \qquad x \in \mathbb{R}^N, \quad (10.2.4)$$

for any $u \in C_c^{\infty}(\mathbb{R}^N)$, where the potential c is given by

$$c = \frac{1}{2} \left(\frac{1}{2} |DU|^2 - \Delta U + \sum_{i=1}^{N} G_i D_i U \right).$$

We claim that the operator \widetilde{L}_2 is not sectorial in $L^2(\mathbb{R}^N)$. Of course, this will imply that L_2 is not sectorial in L^2_μ as well.

To prove the claim, we adapt the technique in the proof of Theorem 10.1.1 to this situation. For this purpose, let us set $\gamma = \frac{1}{2}(\alpha + \beta)$ and introduce the sequences $\{\lambda_n\}$ and $\{r_n\}$ defined by

$$\lambda_n = \frac{1}{|\sigma_n|^{\gamma} + 1}, \qquad r_n = \frac{1}{|G(\sigma_n)|(|\sigma_n|^{\gamma} + 1)}, \qquad n \ge 1.$$

(Up to replacing $\{\sigma_n\}$ with a suitable subsequence, we can assume that $G(\sigma_n) \neq 0$ for any $n \in \mathbb{N}$, so that the sequence $\{r_n\}$ is well defined.) Moreover, for any $n \in \mathbb{N}$, we introduce the operator I_n defined by

$$(I_n u)(x) = u\left(\frac{x - \sigma_n}{\lambda_n}\right), \qquad x \in \mathbb{R}^N, \quad u \in L^2(\mathbb{R}^N), \quad n \ge 1$$

and $\widetilde{L}_{2,n} = r_n I_n^{-1} \widetilde{L}_2 I_n$ with domain $D(\widetilde{L}_{2,n}) = I_n^{-1}(D(\widetilde{L}_2))$. As it is immediately seen, $\widetilde{L}_{2,n}$ is the infinitesimal generator of the strongly continuous semigroup of contractions $\{\widetilde{T}_n(t)\}$ in $L^2(\mathbb{R}^N)$, defined by $\widetilde{T}_n(t) = I_n^{-1} \widetilde{T}(r_n t) I_n$, for any $t \geq 0$.

Let us show that there exists $h \in \mathbb{R}^N$ such that, for any $u \in C_c^{\infty}(\mathbb{R}^N)$, the function $\widetilde{L}_{2,n}u$ converges to $\mathcal{B}_{-h}u := \sum_{i=1}^N h_i D_i u$ as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . For this purpose, we observe that, by (10.2.4), we have

$$(\widetilde{L}_{2,n}u)(x) = \frac{r_n}{\lambda_n^2} \Delta u(x) - \frac{r_n}{\lambda_n} \sum_{i=1}^N G_i(\lambda_n x + \sigma_n) D_i u(x) - r_n c(\lambda_n x + \sigma_n) u(x),$$

for any $x \in \mathbb{R}^N$, any $n \in \mathbb{N}$ and any $u \in C_c^{\infty}(\mathbb{R}^N)$.

Due to the choice of the sequences $\{\sigma_n\}$ and $\{r_n\}$, it is immediate to check that r_n/λ_n^2 tends to 0 as n tends to $+\infty$. Moreover, by the mean value theorem and (10.2.3)(ii) we have

$$|G(\lambda_n x + \sigma_n)| \le |G(\sigma_n)| + kR(1 + (R + |\sigma_n|)^{\beta}),$$
 (10.2.5)

for any $x \in \overline{B}(R)$ and any R > 0. Hence,

$$\lim_{n \to +\infty} \frac{|G(\sigma_n + \lambda_n x) - G(\sigma_n)|}{|G(\sigma_n)|} = 0,$$

locally uniformly in \mathbb{R}^N . Up to replacing $\{\sigma_n\}$ with a suitable subsequence, we can assume that $G(\sigma_n)/|G(\sigma_n)|$ converges to some vector $h \in \mathbb{R}^N$. Therefore,

$$\frac{r_n}{\lambda_n^2} \Delta u - \frac{r_n}{\lambda_n} \sum_{i=1}^N G_i(\lambda_n \cdot + \sigma_n) D_i u$$

converges to $\mathcal{B}_h u$ uniformly in \mathbb{R}^N as n tends to $+\infty$ and, hence, in $L^2(\mathbb{R}^N)$, since the support of u is compact.

Let us now show that $c(\lambda_n \cdot + \sigma_n)u$ converges to 0 in $L^2(\mathbb{R}^N)$. For this purpose, we note that, thanks to (10.2.3)(ii),

$$|c(x)| \le C \left((1+|x|^{\beta})^2 + (1+|x|^{\beta})|G(x)| \right), \qquad x \in \mathbb{R}^N, \quad (10.2.6)$$

for some positive constant C. Estimates (10.2.6) and (10.2.5) imply that the sequence $\{r_n|c(\lambda_n\cdot+\sigma_n)|\}$ converges to 0, locally uniformly in \mathbb{R}^N , and we are done.

Now the proof follows immediately, arguing by contradiction. Indeed, if \widetilde{L}_2 were sectorial in $L^2(\mathbb{R}^N)$, then the operators $\widetilde{L}_{2,n}$ should be sectorial as well, and there should exist K, C > 0, independent of n (observe that $R(\lambda, \widetilde{L}_{2,n}) = r_n^{-1} I_n^{-1} R(r_n^{-1} \lambda, \widetilde{L}_2) I_n$), such that

$$\|R(\lambda,\widetilde{L}_{2,n})u\|_{L^2(\mathbb{R}^N)} \leq \frac{C}{|\lambda|} \|u\|_{L^2(\mathbb{R}^N)},$$

for any $\lambda \in \mathbb{C}$, with Re $\lambda > K$ and any $u \in C_c^{\infty}(\mathbb{R}^N)$. Since the realization B_h of the operator \mathcal{B}_h generates a strongly continuous semigroup in $L^2(\mathbb{R}^N)$ and $C_c^{\infty}(\mathbb{R}^N)$ is a core of B_h , using Trotter-Kato theorem (see Theorem B.1.11), it should follow that

$$||R(\lambda, B_h)u||_{L^2(\mathbb{R}^N)} \le \frac{C}{|\lambda|} ||u||_{L^2(\mathbb{R}^N)},$$

for any $u \in L^2(\mathbb{R}^N)$, which, of course, is a contradiction.

To prove the assertion with a general $p \in [1, +\infty)$ it now suffices to observe that, if T(t) were analytic in L^p_μ for some p as above $(p \neq 2)$, then, by Stein interpolation theorem, it should be analytic in L^2_μ as well. For more details, we refer the reader to [45, Theorem 1.4.2].

Example 10.2.3 We now provide two concrete examples of differential operators in \mathbb{R}^2 , such that the associated semigroups $\{T(t)\}$ are not analytic in L^2_{μ} . For this purpose, let \mathcal{A}_1 and \mathcal{A}_2 be the operators defined on smooth functions by

$$A_1 u(x,y) = \Delta u(x,y) - \left(x + y(x^2 + y^2)^{r/2}\right) D_x u(x,y)$$
$$- \left(y - x(x^2 + y^2)^{r/2}\right) D_y u(x,y)$$

and

$$\mathcal{A}_2 u(x,y) = \Delta u(x,y) - (x^2 + y^2) \left(x - y(x^2 + y^2)^{r/2} \right) D_x u(x,y)$$
$$-(x^2 + y^2) \left(y - x(x^2 + y^2)^{r/2} \right) D_y u(x,y),$$

for some r>1 and any $(x,y)\in\mathbb{R}^2$. We can rewrite the operators \mathcal{A}_1 and \mathcal{A}_2 in the compact form (10.2.1) taking $U_j(x,y)=(2j)^{-1}(x^2+y^2)^j$, $G_j(x,y)=(x^2+y^2)^{r/2+(j-1)}(y,-x)$ for any $(x,y)\in\mathbb{R}^2$ and j=1,2. It is immediate to check that the condition (10.2.2) is satisfied. Moreover, if we take $\sigma_n=(n,0)$ for any $n\in\mathbb{N}$, also the assumptions of Theorem 10.2.2 are satisfied with $\alpha=r-1+2j$ and $\beta=r-2+2j$, j=1,2. Therefore, the semigroup $\{T_j(t)\}$ associated with the operator \mathcal{A}_j is not analytic in $L^p_{\mu_j}$ $(p\in[1,+\infty))$ either for j=1 or for j=2, where

$$d\mu_1 = \frac{1}{2\pi} e^{-\frac{1}{2}(x^2 + y^2)} dx dy,$$

$$d\mu_2 = \left(\int_{\mathbb{R}^2} e^{-\frac{1}{4}(x^2 + y^2)^2} dx dy \right)^{-1} e^{-\frac{1}{4}(x^2 + y^2)^2} dx dy.$$

Part II

Markov semigroups in unbounded open sets

Chapter 11

The Cauchy-Dirichlet problem

11.0 Introduction

In this chapter we consider the parabolic Cauchy-Dirichlet problem

$$\begin{cases} D_t u(t,x) - \mathcal{A} u(t,x) = 0, & t > 0, \ x \in \Omega, \\ u(t,x) = 0, & t > 0, \ x \in \partial \Omega, \\ u(0,x) = f(x), & x \in \Omega, \end{cases}$$
(11.0.1)

where $\Omega \subset \mathbb{R}^N$ is an unbounded connected smooth open set; f is a bounded and continuous function on Ω ; \mathcal{A} is the usual second-order uniformly elliptic operator

$$\mathcal{A} = \sum_{i,j=1}^{N} q_{ij} D_{ij} + \sum_{i=1}^{N} b_i D_i + c, \tag{11.0.2}$$

with unbounded coefficients

We show that, under rather weak assumptions of the smoothness of the coefficients q_{ij} , b_j (i, j = 1, ..., N) and c and assuming the existence of a Lyapunov function associated with the operator \mathcal{A} , the Cauchy-Dirichlet problem (11.0.1) admits a unique classical solution (i.e., a function $u \in C^{1,2}((0, +\infty) \times \Omega)$, bounded in $(0, +\infty) \times \Omega$ and continuous in $[0, +\infty) \times \overline{\Omega} \setminus (\{0\} \times \partial \Omega)$ which solves (11.0.1)). This, as usually, will allow us to define a semigroup $\{T(t)\}$ of bounded linear operators on $C_b(\Omega)$.

Next, assuming somewhat heavier assumptions on the growth rate of the coefficients at infinity, we show that such a semigroup satisfies an uniform gradient estimate similar to that proved in Chapter 6. The results of this chapter have been recently proved in [59].

The chapter is split into sections as follows. First, in Section 11.1, we show two maximum principles for the solution to the problem (11.0.1). In the first one, which is rather similar to that proved in Chapter 4, we deal with classical solutions to the problem (11.0.1) which are continuous in $[0, +\infty) \times \overline{\Omega}$. In the second one, we weaken a bit the assumptions on the continuity of u. Indeed, we just assume that u is continuous in $[0, +\infty) \times \overline{\Omega} \setminus (\{0\} \times \partial \Omega)$. This latter maximum principle is the keystone to prove the uniform gradient estimates.

In Section 11.2, by virtue of the previous two maximum principles and some classical Schauder estimates, we prove the existence and uniqueness of the

classical bounded solution u to (11.0.1), and, then, we define the semigroup $\{T(t)\}.$

Next, in Section 11.3, under somewhat heavier assumptions on the growth rate of the coefficients of the operator \mathcal{A} , we prove that the gradient estimate

$$||DT(t)f||_{\infty} \le C \frac{e^{\omega t}}{\sqrt{t}} ||f||_{\infty}, \qquad t > 0,$$
 (11.0.3)

holds for any $\omega > 0$, some positive constant $C = C(\omega)$ and any $f \in C_b(\Omega)$. First, in Subsection 11.3.1 we prove (11.0.3) for classical solutions to the problem (11.0.1) enjoying some more regularity properties. This is done applying the Bernstein method as we did in Chapter 6. Unfortunately, it is not immediate to adapt the procedure in the proof of Theorem 6.1.7 to our situation, due to the fact that no a priori information of the sign of Du on $(0, +\infty) \times \partial \Omega$ is available. To overcome such a difficulty, we first prove the gradient estimate (11.0.3) at the boundary of Ω by a comparison argument with a suitable one-dimensional parabolic problem. This and the maximum principle in Proposition 11.1.3 then allow us to prove the uniform estimate (11.0.3).

Finally, in Subsections 11.3.2 and 11.3.3, adapting to our situation the approach introduced by P. Cannarsa and V. Vespri in [26, 27], we prove (11.0.3) for any classical solution to the problem (11.0.1). This is done in the following way. The operator \mathcal{A} is approximated by a family of uniformly elliptic operators $\mathcal{A}_{\varepsilon}$ such that, for any $p \geq 2$, their realizations in $L^p(\Omega)$ generate analytic semigroups $\{S_{\varepsilon}(t)\}$. Then, using the Sobolev embedding theorems, it is shown that, for any $f \in C_c^{\infty}(\Omega)$, the function $u_{\varepsilon} = S_{\varepsilon}(\cdot)f$ is the classical solution to the problem (11.0.1), with \mathcal{A} being replaced with the operator $\mathcal{A}_{\varepsilon}$. Then, letting ε go to 0^+ and using a compactness argument based on classical Schauder estimates, it follows that u_{ε} converges to T(t)f, which satisfies all the additional assumptions needed, in Proposition 11.3.7, to prove the gradient estimate (11.0.3). Hence, (11.0.3) follows in this particular case. Finally, in the general case when $f \in C_b(\Omega)$, the estimate (11.0.3) is obtained by approximating f with a suitable sequence of $C_c^{\infty}(\Omega)$ -smooth functions converging, locally uniformly and in a dominated way, to f.

Unfortunately, to the best of our knowledge, estimates for the higher-order derivatives of the function T(t)f when $f \in C_b(\Omega)$ seem not to be available for a general domain Ω . As it is readily seen, differently from what happens when $\Omega = \mathbb{R}^N$ (see Chapters 6 and 7), the Bernstein method cannot be easily extended to estimate second- and third-order derivatives of the function T(t)f and this is essentially due to the boundary conditions.

This fact prevents us to make the machinery of Chapter 6 work to study the nonhomogeneous elliptic problem and the nonhomogeneous Cauchy-Dirichlet problem in Ω and to prove optimal Schauder estimates. To the best of our knowledge there are only a few results in this direction (see Remark 11.3.16)

We conclude this introduction with the following remark.

Remark 11.0.1 Differently from what happens in the whole of \mathbb{R}^N , there never exists an invariant measure μ associated with the semigroup $\{T(t)\}$ considered in this chapter. Indeed, if this were the case, then

$$\int_{\Omega} T(t)d\mu = \int_{\Omega} f d\mu, \qquad (11.0.4)$$

for any $f \in C_b(\Omega)$. But, according to Corollary 11.2.2, it holds that $0 \le T(t) \mathbf{1} \le \mathbf{1}$ for any t > 0. Hence, from (11.0.4), it should follow that $T(t) \mathbf{1} \equiv \mathbf{1}$ which is a contradiction.

11.1 Two maximum principles

Here, we prove two maximum principles for the classical solutions to the problem (11.0.1). Throughout this section we assume that the following conditions are satisfied:

Hypotheses 11.1.1 (i) Ω is an unbounded connected open set with uniformly $C^{2+\alpha}$ -boundary for some $\alpha \in (0,1)$;

(ii) $q_{ij} \equiv q_{ji} \in C_b^1(\Omega)$ for any i, j = 1, ..., N and there exists $\kappa_0 > 0$ such that

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_{i}\xi_{j} \ge \kappa_{0}|\xi|^{2}, \qquad x \in \Omega, \ \xi \in \mathbb{R}^{N};$$
 (11.1.1)

- (iii) $c(x) \leq 0$ for any $x \in \Omega$;
- (iv) $q_{ij}, b_i, c \in C^{1+\alpha}(\Omega \cap B(R))$ for any i, j = 1, ..., N and any R > 0;
- (v) there exist a positive function $\varphi \in C^2(\overline{\Omega})$ and a positive constant λ_0 such that

$$\lim_{\substack{|x| \to +\infty \\ x \in \overline{\Omega}}} \varphi(x) = +\infty$$

and

$$\sup_{x \in \Omega} (\mathcal{A}\varphi(x) - \lambda_0 \varphi(x)) < +\infty.$$

The following proposition provides us a maximum principle for classical solutions to the Cauchy-Dirichlet problem (11.0.1), which are continuous in $[0,T] \times \overline{\Omega}$.

Proposition 11.1.2 Assume that Hypotheses 11.1.1 are satisfied. Fix T > 0 and let $u \in C([0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$ be a bounded function satisfying

$$\begin{cases}
D_t u(t,x) - \mathcal{A}u(t,x) \le 0, & t \in (0,T], x \in \Omega, \\
u(t,x) \le 0, & t \in (0,T], x \in \partial\Omega, \\
u(0,x) \le 0, & x \in \Omega.
\end{cases}$$
(11.1.2)

Then, $u \leq 0$. In particular, if $u \in C([0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$ is a bounded solution to the Cauchy problem

$$\begin{cases} D_t u(t,x) - \mathcal{A} u(t,x) = 0, & 0 < t \le T, x \in \Omega, \\ u(t,x) = 0, & 0 < t \le T, x \in \partial \Omega, \\ u(0,x) = 0, & x \in \Omega, \end{cases}$$

then, u = 0.

Proof. The proof is close to that of Theorem 4.1.3. Of course we can limit ourselves to proving the first part since the second part can then be deduced applying the first one to the functions u and -u.

We set $v = e^{-\lambda_0 t}u$ and we prove that $v \leq 0$ in $[0, T] \times \overline{\Omega}$. For this purpose, for any $n \in \mathbb{N}$, we introduce the function $v_n : [0, T] \times \overline{\Omega} \to \mathbb{R}$ defined by

$$v_n(t,x) = v(t,x) - \frac{\varphi(x)}{r}, \qquad t \in [0,T], \ x \in \overline{\Omega}.$$

Replacing φ with $\varphi + C$ for some suitable constant C > 0, we can assume that $\mathcal{A}\varphi - \lambda_0 \varphi \leq 0$. Now, a straightforward computation shows that v_n satisfies (11.1.2) with the operator \mathcal{A} being replaced with $\mathcal{A} - \lambda_0$. Due to our assumption on φ , the function v_n attains its maximum at some point (t_n, x_n) . Repeating the same arguments as in the proof of Theorem 4.1.3, we can see that $v_n \leq 0$ in $(0, T) \times \Omega$. Thus, letting n go to $+\infty$ gives $v \leq 0$ in $[0, T] \times \overline{\Omega}$.

The next proposition provides us with a maximum principle for classical solutions to the problem (11.0.1) which are not continuous in $\{0\} \times \partial \Omega$.

We introduce a few notation which will be used in the proof of the next proposition and, more generally, throughout this chapter. By $d: \overline{\Omega} \to \mathbb{R}$ and Ω_{δ} ($\delta > 0$) we denote, respectively, the distance function

$$d(x) = \operatorname{dist}(x, \partial\Omega), \qquad x \in \overline{\Omega}$$

and the set

$$\Omega_{\delta} = \{ x \in \overline{\Omega} : d(x) < \delta \}. \tag{11.1.3}$$

In view of the assumptions on Ω , the function d is C^2 -smooth, with bounded second-order derivatives in Ω_{δ} , if $\delta > 0$ is sufficiently small. See [66, Lemma 14.16] and also Appendix D.

Proposition 11.1.3 Suppose that Hypotheses 11.1.1 are satisfied and let $u \in C^{1,2}((0,T)\times\Omega)\cap C([0,T)\times\overline{\Omega}\setminus (\{0\}\times\partial\Omega))$ be bounded in $(0,T)\times\Omega$. If $D_tu\leq \mathcal{A}u$ in $(0,T)\times\Omega$ and $u\leq 0$ in $(0,T)\times\partial\Omega\cup (\{0\}\times\Omega)$, then $u\leq 0$ in $(0,T)\times\Omega$.

Finally, if $D_t u \leq Au$ and u is bounded in $(0,T) \times \partial \Omega \cup (\{0\} \times \Omega)$ (T > 0), then

$$\sup_{(t,x)\in(0,T)\times\Omega}u(t,x)\leq \|u(0,\cdot)\|_{\infty}+\sup_{(t,\xi)\in(0,T)\times\partial\Omega}|u(t,\xi)|.$$

Proof. We limit ourselves to proving the first part of the proposition, since the last assertion then easily follows applying this part to the functions u-K, where $K=\sup_{(t,\xi)\in(0,T)\times\partial\Omega}|u(t,\xi)|+\|u(0,\cdot)\|_{\infty}$.

To begin with, we observe that there exists a function $g \in C^2(\overline{\Omega})$ such that

$$g \ge 0$$
 in Ω , $|Dg| = 1$ on $\partial \Omega$, $\partial \Omega = \{x \in \overline{\Omega} : g(x) = 0\}$.

Indeed, let η be a smooth function such that $0 \le \eta \le 1$ in Ω , $\eta \equiv 1$ in $\Omega_{\delta/2}$ and $\eta \equiv 0$ outside $\Omega_{3\delta/4}$. It is easy to check that the function $g = \eta d + 1 - \eta$ satisfies the claim.

Now, we introduce the function $v = e^{-\lambda_0 t}u$, where λ_0 is given by Hypothesis 11.1.1(v), and we prove that $v \leq 0$ in $(0,T) \times \Omega$. For this purpose, we fix R > 1 and consider the set

$$\Omega_R = \Omega \cap B(R) = \{g > 0\} \cap \{R^2 - |x|^2 > 0\}.$$

Moreover, we define the function $v_R:(0,T)\times\overline{\Omega}_R\to\mathbb{R}$ by setting

$$v_R(t,x) = v(t,x) - \frac{\|v\|_{\infty}}{C_R} \varphi(x), \qquad t \in (0,T), \quad x \in \overline{\Omega}_R,$$

where $C_R = \inf_{[0,T] \times (\partial B(R) \cap \Omega)} \varphi$. As in the proof of Proposition 11.1.2,we can assume that $\mathcal{A}\pi - \lambda_0 \varphi \leq 0$. By Hypothesis 11.1.1(v), C_R tends to $+\infty$ as R tends to $+\infty$. Further, it is easy to see that v_R satisfies

$$\begin{cases} D_t v_R - \mathcal{A} v_R + \lambda_0 v_R \leq 0, & \text{in } (0, T) \times \Omega_R, \\ v_R \leq 0, & \text{in } (0, T) \times \partial \Omega_R, \\ v_R \leq 0, & \text{in } \{0\} \times \Omega_R. \end{cases}$$

We claim that $v_R \leq 0$ in $(0,T) \times \Omega_R$. Once the claim is proved, letting R tend to $+\infty$, we will obtain that $v \leq 0$ in $(0,T) \times \Omega$ and we will be done.

To prove that $v_R \leq 0$ in $(0,T) \times \overline{\Omega}_R$, for any $\varepsilon, \lambda > 0$ we introduce the function $\psi : (0,T) \times \Omega_R \to \mathbb{R}$ defined by

$$\psi(t,x) = \frac{1}{t^{\varepsilon\kappa_0}} \left[\exp\left(\lambda t - \frac{\varepsilon(g(x))^2}{t}\right) + \exp\left(\lambda t - \frac{\varepsilon(h(x))^2}{t}\right) \right],$$

for any t>0 and any $x\in\Omega_R$, where κ_0 is given by (11.1.1) and $h(x)=R^2-|x|^2$ for any $x\in\mathbb{R}^N$. A straightforward computation shows that

$$D_{t}\psi(t,\cdot) - \mathcal{A}\psi(t,\cdot) = \frac{1}{t^{2}} \left\{ (\lambda - c)t^{2} - \varepsilon \left[\kappa_{0} - 2\langle QDg, Dg \rangle - 2g(\mathcal{A} - c)g \right] t + \varepsilon g^{2} \left[1 - 4\varepsilon \langle QDg, Dg \rangle \right] \right\} \psi(t,\cdot)$$

$$+ \frac{1}{t^{2}} \left\{ (\lambda - c)t^{2} - \varepsilon \left[\kappa_{0} - 2\langle QDh, Dh \rangle - 2h(\mathcal{A} - c)h \right] t + \varepsilon h^{2} \left[1 - 4\varepsilon \langle QDh, Dh \rangle \right] \right\} \psi(t,\cdot)$$

$$:= \frac{1}{t^{2}} \left(\Lambda_{1}(t,\cdot) + \Lambda_{2}(t,\cdot) \right), \qquad (11.1.4)$$

for any $t \in (0, +\infty)$. We now claim that we can fix $\varepsilon > 0$ sufficiently small and λ sufficiently large such that $(D_t - \mathcal{A})\psi \geq 0$ in $(0, +\infty) \times \Omega_R$. To prove the claim, we begin by considering the first term in the last side of (11.1.4). Since $|Dg| \geq 1$ on $\partial\Omega$, we can fix $\delta > 0$ such that $|Dg(x)| \geq 3/4$ if $x \in \Omega_R$ and $g(x) \leq \delta$. Then, we split $\Omega_R = \{x \in \Omega_R : g(x) > \delta\} \cup \{x \in \Omega_R : g(x) \leq \delta\} := \Omega_{R,1} \cup \Omega_{R,2}$. If $x \in \Omega_{R,1}$, we have

$$\Lambda_1(t,x) \ge \lambda t^2 - \varepsilon \left(\kappa_0 + 2 \sup_{\Omega_R} |g(\mathcal{A} - c)g| \right) t + \varepsilon \delta^2 \left(1 - 4\varepsilon \|Q\|_{\infty} \sup_{\Omega_R} |Dg|^2 \right). \tag{11.1.5}$$

It is now clear that, if we fix $\varepsilon = (8\|Q\|_{\infty} \sup_{\Omega_R} |Dg|^2)^{-1}$ and, then, $\lambda = \lambda(\varepsilon, \delta)$ large enough, we can make the right-hand side of (11.1.5) nonnegative.

Now, let $x \in \Omega_{R,2}$. Then,

$$\kappa_0 - 2\langle Q(x)Dg(x), Dg(x)\rangle - 2g(x)\left((\mathcal{A} - c)g\right)(x)$$

$$\leq \kappa_0 - 2\kappa_0|Dg(x)|^2 + 2g(x)\sup_{\Omega_R}|(\mathcal{A} - c)g|$$

$$\leq -\frac{1}{2}\kappa_0 + \frac{1}{2t}(g(x))^2 + \frac{t}{2}\sup_{\Omega_R}|(\mathcal{A} - c)g|^2.$$

Therefore,

$$\Lambda_1(t,x) \ge \left(\lambda - \frac{1}{2} \sup_{\Omega_R} |(\mathcal{A} - c)g|^2\right) t^2 + \frac{1}{2} \varepsilon \kappa_0 t + \varepsilon g^2 \left(\frac{1}{2} - 4\varepsilon \|Q\|_{\infty} \sup_{\Omega_R} |Dg|^2\right). \tag{11.1.6}$$

Again, we can fix $\varepsilon = (16\|Q\|_{\infty} \sup_{\Omega_R} |Dg|^2)^{-1}$ and, then, $\lambda = \lambda(\varepsilon)$ large enough such that the right-hand side of (11.1.6) is nonnegative. Similarly, we can argue with the function Λ_2 . Indeed, the keystone to prove that Λ_1 is nonnegative in $(0, +\infty) \times \Omega_R$ was the fact that $|Dg| \ge 1$ when g = 0. Now, if we consider the function h it is easy to check that, since R > 1, then $|Dh| \ge 1$

when h=0. Summing up, we have shown that we can fix the parameters λ and ε such that $D_t\psi - \mathcal{A}\psi \geq 0$ in $(0, +\infty) \times \Omega_R$.

Let us now suppose by contradiction that $0 < M = \sup_{(0,T) \times \Omega_R} v_R$ for some T > 0. For any a > 0 we introduce the function $v_{a,R}$ defined by

$$v_{a,R}(t,x) = v_R(t,x) - Ma^{\varepsilon \kappa_0} \psi(t+a,x),$$

for any $(t,x) \in [0,T] \times \overline{\Omega}_R \setminus (\{0\} \times \Omega \cap \partial B(R))$. Clearly $(D_t - \mathcal{A} + \lambda_0)v_{a,R} \leq 0$ in $(0,T) \times \Omega_R$. Take $\eta > 0$ such that $\lambda a - \varepsilon \eta^2 a^{-1} > 0$ and introduce the set

$$I_{\eta,R} = \{ x \in \overline{\Omega}_R : \min\{g(x), h(x)\} \le \eta \}.$$

For any $x \in I_{\eta,R}$ we have

$$a^{\varepsilon\kappa_0}\psi(a,x)\geq 2\exp\left(\lambda a-\frac{\varepsilon\eta^2}{a}\right)>1.$$

By continuity, there exists $\delta > 0$ such that, for any $t \in [0, \delta]$ and any $x \in I_{\eta,R}$,

$$a^{\varepsilon\kappa_0}\psi(t+a,x) > 1.$$

It follows that $v_{a,R} < M - M = 0$ in $[0, \delta] \times I_{\eta,R}$.

Since $v_R(0,\cdot) \leq 0$ in $\Omega_R \setminus I_{\eta,R}$, we have $v_{a,R}(0,\cdot) < 0$ in $\Omega_R \setminus I_{\eta,R}$. The boundedness of Ω_R and the continuity of u in $[0,T) \times \Omega_R \setminus I_{\eta,R}$ imply that $v_{a,R} \leq 0$ in $[0,\delta] \times (\Omega_R \setminus I_{\eta,R})$ for some $\delta > 0$. Hence, $v_{a,R}$ is nonpositive in $[0,\delta] \times \overline{\Omega_R} \setminus (\{0\} \times \partial\Omega \cap B(R))$.

Applying the classical maximum principle in $[\delta, T) \times \overline{\Omega}_R$, we get that $v_{a,R} \leq 0$ in $[\delta, T) \times \overline{\Omega}_R$ and, consequently, we obtain that $v_{a,R}$ is nonpositive in $[0, T) \times \overline{\Omega}_R \setminus (\{0\} \times \partial \Omega_R)$.

Finally, letting a go to 0^+ , we find that $v_R \leq 0$ in $[0,T) \times \overline{\Omega}_R \setminus (\{0\} \times \partial \Omega_R)$, and we are done, letting R tend to $+\infty$.

Remark 11.1.4 Observe that the above theorem covers also the case of certain nonsmooth domains, whose boundaries can be described by a finite number of functions g_i as in the statement. For further details, we refer the reader to [59, Theorem A.2].

11.2 Existence and uniqueness of the classical solution

In this section we prove that, under Hypotheses 11.1.1, the Cauchy-Dirichlet problem (11.0.1) has a unique bounded classical solution u (i.e., a function $u \in C^{1,2}((0,+\infty)\times\Omega)$ which is continuous in $[0,+\infty)\times\overline{\Omega}\setminus(\{0\}\times\partial\Omega)$, bounded in $(0,+\infty)\times\Omega$ and solves the Cauchy problem (11.0.1)).

Theorem 11.2.1 Suppose that Hypotheses 11.1.1 are satisfied. Then, for any $f \in C_b(\Omega)$, the Cauchy-Dirichlet problem (11.0.1) admits a unique bounded classical solution u. Moreover,

$$||u||_{\infty} \le ||f||_{\infty}$$

and $u \ge 0$ if $f \ge 0$.

Further, if $f \in C_c^{2+\alpha}(\Omega)$, then u belongs to $C^{1+\alpha/2,2+\alpha}((0,T)\times(\Omega\cap B(R)))$ for any R,T>0. Moreover, Du belongs to $C^{1+\alpha/2,2+\alpha}((\varepsilon,T)\times\Omega')$ for any $0<\varepsilon< T$ and any bounded open set $\Omega'\subset \Omega$. In particular, $Du\in C^{1,2}((0,+\infty)\times\Omega)$.

Proof. Uniqueness follows from Proposition 11.1.3. So, let us prove that the Cauchy-Dirichlet problem (11.0.1) actually admits a solution with the claimed regularity properties. We split the proof into two steps. First in Step 1, we consider the case when $f \in C_c^{2+\alpha}(\Omega)$ and then, in Step 2, we deal with the general case.

Step 1. Let $f \in C_c^{2+\alpha}(\Omega)$. To prove that the problem (11.0.1) admits a classical solution, we use an approximation argument different from that used in the proof of Theorem 2.2.1. Instead of approximating the domain, we rather approximate the coefficients of the operator \mathcal{A} . For this purpose, we consider the sequence of uniformly elliptic operators $\mathcal{A}^{(n)}$ with bounded coefficients in $C^{\alpha}(\Omega)$ defined by

$$\mathcal{A}^{(n)} = \sum_{i,j=1}^{N} q_{ij} D_{ij} + \sum_{i=1}^{N} b_i^{(n)} D_i + c^{(n)} u, \qquad n \in \mathbb{N},$$

where the coefficients $b_i^{(n)}$ $(i=1,\ldots,N)$ and $c^{(n)}$ coincide, respectively, with b_i and c in $\Omega \cap B(n)$, and $c^{(n)} \leq 0$ in Ω . Let $u_n \in C^{1+\alpha/2,2+\alpha}((0,T) \times \Omega) \cap C([0,T] \times \overline{\Omega})$, for any T>0, be the classical solution of (11.0.1), with $\mathcal{A}^{(n)}$ instead of \mathcal{A} (see Proposition C.3.2). The maximum principle in Proposition 11.1.2 yields $\|u_n\|_{\infty} \leq \|f\|_{\infty}$ for any $n \in \mathbb{N}$. Let us fix R>0 and observe that, since Ω is unbounded and connected, $\operatorname{dist}(\Omega \setminus B(R+1), \Omega \cap B(R)) > 0$. Since $\mathcal{A}^{(n)} = \mathcal{A}^{(m)} = \mathcal{A}$ in $\Omega \cap B(R+1)$ for $n, m \geq R+1$, from the local Schauder estimates in Theorem C.1.5 (see (C.1.19)), for any T>0, there exists a positive constant C, independent of n and m, such that

$$||u_n - u_m||_{C^{1+\alpha/2,2+\alpha}((0,T)\times(\Omega\cap B(R)))} \le C||u_n - u_m||_{C((0,T)\times(\Omega\cap B(R+1)))}$$

$$\le 2C||f||_{\infty}.$$

Now, using a compactness argument as in the proof of Theorem 2.2.1, we can determine a subsequence $\{u_{n_k}\}$ converging in $C^{1,2}(F)$, for any compact set $F \subset [0,T] \times \overline{\Omega}$, to a function $u \in C^{1+\alpha/2,2+\alpha}_{loc}([0,T] \times \overline{\Omega})$, which solves the Cauchy-Dirichlet problem (11.0.1) and satisfies $||u||_{\infty} \leq ||f||_{\infty}$. Moreover, by

the maximum principle in Proposition 11.1.2, we deduce that u is nonnegative, whenever f is. Note that, applying the same arguments as above to any subsequence of $\{u_n\}$, one can actually show that the whole sequence $\{u_n\}$ converges to u in $C^{1,2}(F)$, for any compact set $F \subset (0,T) \times \overline{\Omega}$.

The further regularity properties of the function u follow from Theorem C.1.4(ii).

Step 2. Now, we consider the general case when $f \in C_b(\Omega)$. Let $\{f_n\} \in C_c^{2+\alpha}(\Omega)$ be a sequence of smooth functions converging to f uniformly on compact subsets of Ω and such that $||f_n||_{\infty} \leq ||f||_{\infty}$ for any $n \in \mathbb{N}$. Moreover, let $u_n \in C^{1+\alpha/2,2+\alpha}((0,T)\times(\Omega\cap B(R)))$, for any R,T>0, be the solution of (11.0.1) given by Step 1, with f_n instead of f. Let us fix $\varepsilon > 0$. According to Theorem C.1.5, for any $\varepsilon \in (0,T)$ we can determine a positive constant C, independent of n and m, such that

$$||u_n - u_m||_{C^{1+\alpha/2,2+\alpha}((\varepsilon,T)\times(\Omega\cap B(R)))} \le 2C||f||_{\infty}, \qquad n,m \in \mathbb{N}.$$

Now, as in Step 1, we can find out a subsequence $\{u_{n_k}\}$ converging to a function $u \in C^{1+\alpha/2,2+\alpha}((\varepsilon,T)\times(\Omega\cap B(R)))$ for any $\varepsilon,R>0$, $\varepsilon< T$, which solves the equation $D_tu-Au=0$ in $(0,T)\times\Omega$ and vanishes in $(0,T)\times\partial\Omega$.

To conclude the proof, it remains to show that $u(t,\cdot)$ converges to f as t tends to 0^+ , uniformly on compact sets of Ω .

For notational convenience, in the rest of the proof we denote by u_g the classical solution to (11.0.1) corresponding to $g \in C_c^{2+\alpha}(\Omega)$.

We fix a compact set $K \subset \Omega$ and a function $\vartheta \in C_c^{2+\alpha}(\Omega)$ such that $0 \le \vartheta \le 1$ in Ω and $\vartheta \equiv 1$ in K. Moreover, we split

$$u_{n_k} = u_{\vartheta f_{n_k}} + u_{(1-\vartheta)f_{n_k}}, \qquad k \in \mathbb{N}, \tag{11.2.1}$$

and we claim that there exists a positive constant C, independent of k such that

$$|u_{(1-\vartheta)f_{n_k}}(t,\cdot)| \le C(\mathbf{1} - u_{\vartheta}(t,\cdot)), \qquad t > 0.$$
 (11.2.2)

To prove (11.2.2), it suffices to observe that, for any $m \in \mathbb{N}$ and any t > 0,

$$|T_m(t)((1-\vartheta)f_{n_k})| \leq ||f_{n_k}||_{\infty} (1-T_m(t)\vartheta) \leq C (1-T_m(t)\vartheta),$$

where $\{T_m\}$ is the semigroup associated with the operator $\mathcal{A}^{(m)}$, and let m go to $+\infty$, using the fact that $(1-\vartheta)f_{n_k}$ and ϑ are compactly supported in $\overline{\Omega}$. Now, subtracting ϑf from both the sides of (11.2.1) and letting k go to $+\infty$ give

$$|u(t,\cdot) - \vartheta f| \le |u_{\vartheta f} - \vartheta f| + C(1 - u_{\vartheta}),$$

for any t > 0. Since $\vartheta f, \vartheta \in C_c(\overline{\Omega})$, then $u_{\vartheta f}$ and u_{ϑ} tends, respectively, to ϑf and ϑ as t tends to 0, uniformly in $\overline{\Omega}$. Since $\vartheta \equiv 1$ on K, it follows that $u(t,\cdot)$ tends to f uniformly on K as t tends to 0^+ . This shows that actually u is the classical solution to (11.0.1). The proof is now complete.

Corollary 11.2.2 The family $\{T(t)\}$ defined by $T(t)f = u(t, \cdot)$ for any t > 0 and any $f \in C_b(\Omega)$, where u is the classical solution to the problem (11.0.1), is a semigroup of positive contractions in $C_b(\Omega)$.

Proof. The semigroup law (i.e., the equality T(t+s) = T(t)T(s) for any s, t > 0) and the positivity of the semigroup (i.e., $T(t)f \ge 0$ whenever $f \ge 0$) follow immediately from the maximum principle in Proposition 11.1.3 and Theorem 11.2.1.

11.3 Gradient estimates

This section is devoted to prove that, under Hypotheses 11.1.1 and suitable additional growth assumptions on the coefficients of the operator \mathcal{A} , for any $\omega > 0$ there exists a positive constant $C = C(\omega)$ such that

$$||DT(t)f||_{\infty} \le C \frac{e^{\omega t}}{\sqrt{t}}, \qquad t > 0, \tag{11.3.1}$$

for any $f \in C_b(\Omega)$. Of course, it suffices to prove (11.3.1) in some time-domain (0,T), and then to use the semigroup rule to extend it to all the positive times (see the proof of Theorem 6.1.7). As mentioned at the very beginning of this chapter the arguments that we need to prove (11.3.1) differ from those used in Chapter 6 in the case when $\Omega = \mathbb{R}^N$.

Throughout this section we assume the following additional assumptions on the coefficients of the operator \mathcal{A} . We recall that Ω_{δ} is defined in (11.1.3).

Hypothesis 11.3.1 There exist some constants $k, M, \beta, d_1, d_2 \in \mathbb{R}$, s < 1/2 and $\delta > 0$ such that

$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \le (-s c(x) + k) |\xi|^2, \qquad x \in \Omega, \ \xi \in \mathbb{R}^N, \quad (11.3.2)$$

$$\sum_{j=1}^{N} b_j(x) D_j d(x) \le M, \qquad x \in \Omega_{\delta}, \tag{11.3.3}$$

$$|Dc(x)| \le \beta(1 - c(x)), \qquad x \in \Omega, \tag{11.3.4}$$

$$|b(x)| \le d_1 e^{d_2|x|}, \qquad x \in \Omega.$$
 (11.3.5)

Remark 11.3.2 Observe that if s = 0 in (11.3.2) and Ω is star-shaped with respect to 0, then Hypothesis 11.1.1(v) is a straightforward consequence of the positivity of -c and the boundedness of q_{ij} (i, j = 1, ..., N). Indeed, since

$$\langle b(x), x \rangle = \langle b(0), x \rangle + \int_0^1 \frac{d}{dt} \langle b(tx), x \rangle dt$$
$$= \langle b(0), x \rangle + \int_0^1 \sum_{i,j=1}^N D_i b_j(tx) x_i x_j dt,$$

for any $x \in \Omega$, from (11.3.2) we get

$$\langle b(x), x \rangle \le \langle b(0), x \rangle + k|x|^2, \qquad x \in \Omega.$$

Hence, if we take $\varphi(x) = 1 + |x|^2$ for any $x \in \Omega$, we get

$$\mathcal{A}\varphi(x) = 2\text{Tr}(Q(x)) + 2\langle b(x), x \rangle + c(x)(1+|x|^2)$$

$$\leq 2\text{Tr}(Q(x)) + 2|b(0)||x| + k|x|^2 + c(x)(1+|x|^2).$$

Since $c \leq 0$ and the q_{ij} 's are bounded for any i, j = 1, ..., N, we can determine λ_0 sufficiently large such that $\mathcal{A}\varphi - \lambda_0\varphi \leq 0$ and Hypothesis 11.1.1(v) is satisfied taking as a Lyapunov function the function $\varphi : \mathbb{R}^N \to \mathbb{R}$ defined by $\varphi(x) = 1 + |x|^2$ for any $x \in \mathbb{R}^N$.

Remark 11.3.3 As we will see in Subsection 11.3.4, the gradient estimate (11.3.1) may fail if we do not assume Hypothesis 11.3.1.

Let us state the main result of this section. Its proof will be given in Subsection 11.3.3.

Theorem 11.3.4 Suppose that the coefficients of the operator A satisfy Hypotheses 11.1.1 and 11.3.1. Then, the classical solution u to the Cauchy-Dirichlet problem (11.0.1) satisfies the gradient estimates (11.3.1).

11.3.1 A priori gradient estimates

In this subsection we prove the gradient estimate (11.3.1) assuming some more regularity on the classical solution u to the problem (11.0.1). To begin with, let us prove the following lemma which plays a crucial role in obtaining the a priori gradient estimates.

Lemma 11.3.5 Let a, b, δ be positive constants and let $g : [0, +\infty) \times [0, \delta] \to \mathbb{R}$ be the classical solution to the Cauchy-Dirichlet problem

$$\begin{cases}
g_t(t,r) = ag_{rr}(t,r) + bg_r(t,r), & t > 0, \quad r \in (0,\delta), \\
g(t,0) = 0, \quad g(t,\delta) = 1, & t > 0, \\
g(0,r) = 1, & r \in (0,\delta),
\end{cases}$$
(11.3.6)

i.e., $g \in C([0,T] \times [0,\delta] \setminus (\{0\} \times \{0,\delta\})) \cap C^{1,2}((0,T) \times [0,\delta])$. Then $g_r \geq 0$, $g_{rr} \leq 0$ in $(0,+\infty) \times (0,\delta)$ and there exists a positive constant C such that

$$0 \le g(t,r) \le \frac{C}{\sqrt{t}}r, \qquad t > 0, \ r \in (0,\delta).$$
 (11.3.7)

Proof. The existence and the uniqueness of the classical solution g to the problem (11.3.6) follows from the results in Section 11.2, where we replace Ω with $(0, \delta)$.

To prove that g satisfies the claimed properties, we find out an explicit representation formula for g. For this purpose, let us consider the differential operator (B, D(B)) defined by

$$Bu = au'' + bu'$$
 $D(B) = \{u \in C^2([0, \delta]) : u(0) = u(\delta) = 0\}.$

According to Theorem C.3.6(iv), (B, D(B)) generates an analytic semigroup $\{R(t)\}$ of positive contractions in $C([0, \delta])$. Let us introduce the function $\psi : [0, \delta] \to \mathbb{R}$ defined by

$$\psi(r) = q \int_0^r e^{-b s/a} ds, \qquad r \in [0, \delta],$$

where $q = a(1 - e^{-b\delta/a})/b$. As it is immediately seen, $B\psi = 0$, $\psi(0) = 0$ and $\psi(\delta) = 1$. A straightforward computation shows that $g(t, \cdot) = R(t)(1 - \psi) + \psi$ for any t > 0. Since the function $t \mapsto R(t)(1 - \psi)$ is analytic in $(0, +\infty)$ with values in $D(B^n)$ for any $n \in \mathbb{N}$ (see Theorem B.2.2(iv)), we immediately deduce that g is a C^{∞} -smooth function in $(0, +\infty) \times [0, \delta]$. To show that g_r is nonpositive in $(0, +\infty) \times (0, \delta)$ we begin by observing that, according to the maximum principle, we have

$$0 \le g(t, \cdot) \le 1,$$
 $t > 0.$ (11.3.8)

Hence $R(t)(\mathbf{1} - \psi) \leq \mathbf{1} - \psi$ for any t > 0. Moreover, from (11.3.8) we also deduce that

$$g(t+s,\cdot) = R(t+s)(\mathbb{1} - \psi) + \psi$$
$$= R(t)R(s)(\mathbb{1} - \psi) + \psi$$
$$\leq R(t)(\mathbb{1} - \psi) + \psi = g(t,\cdot),$$

for any s, t > 0. Therefore, the function $g(\cdot, r)$ is decreasing for any $r \in (0, \delta)$. This implies that $g_t(t, r) \leq 0$ for any $t \in (0, +\infty)$ and any $r \in (0, \delta)$. From the differential equation solved by g, we now deduce that

$$ag_{rr}(t,r) + bg_r(t,r) \le 0, \qquad t > 0, \ r \in (0,\delta),$$

or, equivalently, that

$$\frac{d}{dr}\left(e^{br/a}g_r\right) \le 0.$$

Hence, the function $r \mapsto e^{br/a}g_r(t,r)$ is decreasing in $(0,\delta)$ for any t>0. Since $g(t,\delta)=1$ and $0 \le g \le 1$, we deduce that $g_r(t,\delta) \ge 0$ and, consequently, $g_r(t,\cdot) \ge 0$ for any t>0.

Now, since

$$g_{rr} = \frac{1}{a}(g_t - bg_r),$$

by difference, we immediately deduce that $g_{rr} \leq 0$ in $(0, +\infty) \times (0, \delta)$.

To conclude the proof, let us check (11.3.7). For this purpose, we observe that since $\{R(t)\}$ is an analytic semigroup, for any T>0 there exists a positive constant $\widetilde{C}=\widetilde{C}(T)$ such that

$$||ag_{rr}(t,\cdot) + bg_r(t,\cdot)||_{\infty} \le \frac{\widetilde{C}}{t}, \qquad t \in (0,T).$$
 (11.3.9)

Now, since $C^1([0,\delta])$ belongs to the class $J_{1/2}$ between $C([0,\delta])$ and $C^2([0,\delta])$ (see Proposition A.4.4), then there exists a positive constant C such that

$$||v||_{C^{1}([0,\delta])} \leq C||v||_{\infty}^{\frac{1}{2}} \left(||v||_{C^{1}([0,\delta])}^{\frac{1}{2}} + ||v''||_{\infty}^{\frac{1}{2}}\right)$$

$$\leq \frac{C}{2\varepsilon} ||v||_{\infty} + C\varepsilon ||v||_{C^{1}([0,\delta])} + C\varepsilon ||v''||_{\infty}, \qquad (11.3.10)$$

for any $v \in C^2([0,\delta])$ and any $\varepsilon > 0$. Now, choosing $C\varepsilon < 1$ we deduce that

$$||v||_{C^1([0,\delta])} \le C(\varepsilon)||v||_{\infty} + \frac{C\varepsilon}{1 - C\varepsilon}||v''||_{\infty}, \tag{11.3.11}$$

for some positive constant $C(\varepsilon)$, blowing up as ε tends to 0^+ . Hence, from (11.3.9) and (11.3.11), we deduce that

$$\|g_{rr}(t,\cdot)\|_{\infty} \le C_1 \left(\frac{1}{t} + \|g_r(t,\cdot)\|_{\infty}\right) \le \frac{C_2(\varepsilon)}{t} + \frac{C\varepsilon}{1 - C\varepsilon} \|g_{rr}(t,\cdot)\|_{\infty},$$

for any $t \in (0,T)$, any T > 0 and some positive constants C_1 and C_2 , independent of $t \in (0,T)$. Hence, up to replacing ε with a smaller constant, we finally obtain that, for any T > 0, there exists a positive constant $\widetilde{C} = \widetilde{C}(T)$ such that

$$||g_{rr}(t,\cdot)||_{\infty} \le \frac{C_2}{t}, \qquad t \in (0,T).$$

Now, from the first part of (11.3.10) and (11.3.11), we get

$$||g_r(t,\cdot)||_{\infty} \le \overline{C} ||g(t,\cdot)||_{\infty}^{\frac{1}{2}} ||g_{rr}(t,\cdot)||_{\infty}^{\frac{1}{2}} \le \overline{C} ||g_{rr}(t,\cdot)||_{\infty}^{\frac{1}{2}}, \qquad t > 0.$$

where \overline{C} is independent of t > 0, and we immediately get (11.3.7). This finishes the proof.

Taking advantage of Lemma 11.3.5, we can prove the following a priori gradient estimates on the boundary of Ω .

Proposition 11.3.6 Assume that Hypotheses 11.1.1 and the estimate (11.3.3) are satisfied. Then, there exists a positive constant C_0 depending on κ_0, M, δ, T and the sup-norm of the diffusion coefficients q_{ij} (i, j = 1, ..., N) such that any bounded classical solution u of (11.0.1), differentiable with respect to the space variables in $(0,T) \times \overline{\Omega}$, satisfies the estimate

$$|Du(t,\xi)| \le \frac{C_0}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0,T), \ \xi \in \partial\Omega. \tag{11.3.12}$$

Proof. For any $x \in \Omega_{\delta}$ (see (11.1.3)) let $\xi(x)$ be the unique point on $\partial\Omega$ satisfying $|x - \xi(x)| = d(x)$. Then, we can write

$$x = \xi(x) - d(x)\nu(\xi(x)),$$

where $\nu(\xi)$ is the outer unit normal to $\partial\Omega$ at ξ . Moreover, $Dd(x) = -\nu(\xi(x))$ for any $x \in \Omega_{\delta}$. To proceed, we observe that, since u = 0 on $\partial\Omega$,

$$|Du(t,\xi)| = \left| \frac{\partial u}{\partial \nu}(t,\xi) \right|, \qquad t > 0, \ \xi \in \partial \Omega.$$

Hence, in order to prove the claim it is enough to show that

$$(T(t)\mathbf{1})(x) \le \frac{C_0}{\sqrt{t}}d(x), \qquad t \in (0,T), \ x \in \Omega_{\delta},$$
 (11.3.13)

for some C_0 as in the statement of the proposition, where $\{T(t)\}$ is the semigroup defined in Corollary 11.2.2. Indeed, once (11.3.13) is proved, using the fact that the semigroup is order preserving, for any $f \in C_b(\Omega)$, any $\xi \in \partial \Omega$ and $x = \xi + d(x)\nu(\xi)$, we can write

$$|(T(t)f)(x) - (T(t)f)(\xi)| = |(T(t)f)(x)| \le ||f||_{\infty} (T(t)\mathbf{1})(x) \le \frac{C_0}{\sqrt{t}} d(x) ||f||_{\infty},$$
(11.3.14)

and (11.3.12) easily follows dividing both the first and the last side of (11.3.14) by d and letting x go to ξ .

To prove (11.3.13) we use a comparison argument. For this purpose, let $z:(0,T)\times\Omega_\delta\to\mathbb{R}$ be the function defined by

$$z(t,x) = g(t,d(x)), \qquad t > 0, \ x \in \Omega_{\delta},$$

 $g:[0,+\infty)\times[0,\delta]\to\mathbb{R}$ being the solution to the problem (11.3.6), where we take $a=\kappa_0$ and $b=M_0:=\sup_{x\in\Omega}(\mathcal{A}-c)d$ (which is finite, since the diffusion coefficients are bounded and the drift b satisfies (11.3.3)). By (11.3.7) we know that

$$|z(t,x)| = g(t,d(x)) \le \frac{C}{\sqrt{t}}d(x), \qquad t > 0, \ x \in \Omega_{\delta},$$

for some constant C depending on the same quantities as in the statement of the proposition. Thus, to prove (11.3.13) we have only to show that

$$(T(t)\mathbf{1})(x) \le z(t,x), \qquad t > 0, \ x \in \Omega_{\delta}.$$
 (11.3.15)

Let us consider the function $v = z - T(\cdot)\mathbf{1}$ in $(0, +\infty) \times \Omega_{\delta}$. In view of the assumptions on Ω , the function d is in $C^2(\Omega_{\delta})$ and, consequently, $v \in C^{1,2}((0,+\infty) \times \Omega_{\delta})$, it is continuous in $[0,+\infty) \times \overline{\Omega}_{\delta} \setminus (\{0\} \times \partial \Omega_{\delta})$, bounded in $(0,+\infty) \times \Omega_{\delta}$, and nonnegative in $(0,+\infty) \times \partial \Omega_{\delta} \cup (\{0\} \times \Omega_{\delta})$. Moreover,

$$D_t v - \mathcal{A}v = \left(\kappa_0 - \sum_{i,j=1}^N q_{ij} D_i dD_j d\right) g_{rr}(\cdot, d(\cdot))$$

$$+ (M_0 - \mathcal{A}d + c d) g_r(\cdot, d(\cdot)) - c g(\cdot, d(\cdot)) \ge 0,$$

since $g, g_r, -g_{rr}$ are nonnegative. The maximum principle in Proposition 11.1.3 now yields (11.3.15) and it concludes the proof.

The following proposition provides us an a priori estimate of Du.

Proposition 11.3.7 Let Hypotheses 11.1.1 and 11.3.1, but estimate (11.3.5), be satisfied. Then, for any T > 0, there exists a positive constant C, depending on κ_0, k, s, β, T and $||DQ||_{\infty}$, such that

$$||Du(t,\cdot)||_{\infty} \le \frac{C}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0,T),$$
 (11.3.16)

for any bounded classical solution u of (11.0.1) such that

- (i) Du belongs to $C^{1,2}((0,+\infty)\times\Omega)$;
- (ii) $\sqrt{t} |Du|$ is continuous in $[0, +\infty) \times \overline{\Omega} \setminus (\{0\} \times \partial \Omega)$, bounded in $(0, T) \times \Omega$ for any T > 0 and it satisfies

$$\lim_{t \to 0^+} \sqrt{t} |Du(t, x)| = 0, \qquad x \in \Omega,$$

as well as the gradient estimate (11.3.12).

Proof. Changing c to c-1 (hence u to $e^{-t}u$) we may assume that $|Dc| \le -\beta c$ (see (11.3.4)). To get (11.3.16) we apply the Bernstein method. For this purpose, we define the function

$$v(t,x) = |u(t,x)|^2 + at |Du(t,x)|^2, \qquad t > 0, \ x \in \Omega,$$

where the constant a>0 will be chosen later. Due to assumptions (i) and (ii), v belongs to $C^{1,2}((0,+\infty)\times\Omega)$, it is continuous in $[0,+\infty)\times\overline{\Omega}\setminus(\{0\}\times\partial\Omega)$, bounded in $(0,+\infty)\times\Omega$ and $v(0,\cdot)=f^2$. We claim that we can fix a>0,

depending only on $\kappa_0, h, k, s, \beta, T$ and $||DQ||_{\infty}$ (which is finite, since $q_{ij} \in C_b^1(\Omega)$ for any $i, j = 1, \ldots, N$), such that

$$D_t v(t, x) - A v(t, x) \le 0,$$
 $t \in (0, T), x \in \Omega.$ (11.3.17)

Proposition 11.1.3 and estimate (11.3.12) will imply that

$$v(t,x) \leq \|f\|_{\infty} + \sup_{(t,\xi) \in (0,T) \times \partial \Omega} at \, |Du(t,\xi)|^2 \leq (1 + aC_0^2) \|f\|_{\infty}^2,$$

for any $t \in (0,T)$ and any $x \in \Omega$. Therefore, the estimate (11.3.16) will follow with $C = (C_0^2 + a^{-1})^{1/2}$.

To prove (11.3.17) we observe that v satisfies the equation

$$D_t v - \mathcal{A}v = a|Du|^2 - 2\sum_{i,j=1}^N q_{ij} D_i u D_j u + g_1 + g_2,$$

where

$$g_1 = at \left(2\sum_{i,j=1}^N D_i b_j D_i u D_j u + 2u \sum_{j=1}^N D_j c D_j u + c |Du|^2 \right) + cu^2,$$

$$g_2 = 2at \left(\sum_{i,j,h=1}^N D_h q_{ij} D_h u D_{ij} u - \sum_{i,j,h=1}^N q_{ij} D_{ih} u D_{jh} u \right).$$

Using Hypothesis 11.3.1 and the inequality $2ab \le \varepsilon a^2 + \varepsilon^{-1}b^2$, which holds for any $a,b,\varepsilon>0$, and recalling that $c\le 0$, we can easily show that, for any $\varepsilon>0$,

$$D_{t}v - \mathcal{A}v \leq c \left(1 - \frac{a\beta}{\varepsilon}t\right)u^{2}$$

$$+ \left\{a - 2\kappa_{0} + at\left[2k - (2s - 1)c + \frac{\|DQ\|_{\infty}}{\varepsilon} - \beta\varepsilon c\right]\right\} |Du|^{2}$$

$$+ (\varepsilon \|DQ\|_{\infty} - 2\kappa_{0})at |D^{2}u|^{2}.$$

$$(11.3.18)$$

Recalling that 2s < 1 and choosing $\varepsilon < \min\{2\kappa_0 \|DQ\|_{\infty}, (1-2s)\beta^{-1}\}$, from (11.3.18) we get

$$D_t v - \mathcal{A}v \le c \left(1 - \frac{a\beta}{\varepsilon}T\right)u^2 + \left\{a - 2\kappa_0 + aT\left[2k + \frac{\|DQ\|_{\infty}}{\varepsilon}\right]\right\} |Du|^2.$$

Now, taking a small enough, we get immediately (11.3.17).

11.3.2 An auxiliary problem

As a second step, in order to prove Theorem 11.3.4, in this subsection we introduce an auxiliary problem set in L^p -spaces. We still consider an elliptic operator \mathcal{A} of the type (11.0.2), but we assume that its coefficients satisfy the following set of hypotheses.

Hypotheses 11.3.8 (i) the coefficients q_{ij} , b_j (i, j = 1, ..., N) and c satisfy Hypotheses 11.1.1(ii) and 11.1.1(iv);

(ii) there exist $0 < \sigma < \sqrt{2\kappa_0}$ and $\beta > 0$ such that

$$c(x) \le -1, \quad |Dc(x)| \le \beta |c(x)|, \quad |b(x)| \le \sigma |c(x)|^{\frac{1}{2}}, \quad x \in \Omega$$

We are going to show that, under the previous hypotheses, the operator \mathcal{A} generates an analytic semigroup in $L^p(\Omega)$, for any $p \geq 2$. For this purpose we adapt the ideas of [26, 27, 119], where the case $\Omega = \mathbb{R}^N$ is considered.

We introduce the space D_p defined by

$$D_p = \{ u \in W^{2,p}(\Omega) \cap W_0^{1,p}(\Omega) : c \, u \in L^p(\Omega) \},$$

and we endow it with the norm

$$||u||_{D_p} = ||u||_{W^{2,p}(\Omega)} + ||cu||_{L^p(\Omega)}, \qquad u \in D_p.$$
 (11.3.19)

As it is immediately seen, D_p is a Banach space when it is endowed with the norm in (11.3.19). Moreover, we consider the dense subspace D of D_p defined as follows:

$$D=\{u\in C^{\infty}(\overline{\Omega}): u_{\big|\partial\Omega}=0, \; \operatorname{supp} u\subset\subset\overline{\Omega}\}.$$

Finally, we denote by A_0 the operator defined on smooth functions by

$$\mathcal{A}_0 \varphi = \sum_{i,j=1}^N q_{ij} D_{ij} \varphi,$$

and by A_p the realization of the operator \mathcal{A} in D, defined by $A_p u = \mathcal{A}u$ for any $u \in D$.

In what follows to shorten the notation we denote by $\|\cdot\|_p$ and $\|\cdot\|_{j,p}$ (j=1,2) the usual norms of $L^p(\Omega)$ and $W^{j,p}(\Omega)$, respectively.

Lemma 11.3.9 For any $p \ge 2$ there exists a positive constant C, depending on N, p, β, R and the coefficients q_{ij} (i, j = 1, ..., N), such that

$$||c|^{1/2}Du||_p \le \varepsilon ||\mathcal{A}_0 u||_p + C\varepsilon^{-1}(||u||_p + ||cu||_p),$$
 (11.3.20)

for any $0 < \varepsilon < R$ and any $u \in D$.

Proof. Fix $u \in D$. Integrating by parts and using the fact that u = 0 on $\partial\Omega$ and $p \geq 2$ we obtain

$$\begin{split} \int_{\Omega} |c|^{\frac{p}{2}} |Du|^p dx &= \int_{\Omega} |c|^{\frac{p}{2}} |Du|^{p-2} |Du|^2 dx \\ &= \frac{p}{2} \int_{\Omega} |c|^{\frac{p}{2}-1} \langle Dc, Du \rangle |Du|^{p-2} u dx \\ &- (p-2) \int_{\Omega} |c|^{\frac{p}{2}} u |Du|^{p-4} \sum_{i,j=1}^{N} D_{ij}^2 u D_i u D_j u dx \\ &- \int_{\Omega} |c|^{\frac{p}{2}} u |Du|^{p-2} \Delta u \, dx. \end{split}$$

Using the Hölder inequality, and observing that, by Hypothesis (11.3.8)(ii), $|Dc| \leq \beta |c| \leq \beta |c|^{\frac{3}{2}}$, we get

$$\begin{split} & \int_{\Omega} |c|^{\frac{p}{2}} |Du|^p dx \\ & \leq \frac{\beta p}{2} \int_{\Omega} |u| |Du|^{p-1} |c|^{\frac{p-1}{2}} |c| dx \\ & + (p-2+\sqrt{N}) \int_{\Omega} |c|^{\frac{p}{2}-1} |Du|^{p-2} |c| |u| |D^2u| dx \\ & \leq \frac{\beta p}{2} \bigg(\int_{\Omega} |c|^{\frac{p}{2}} |Du|^p dx \bigg)^{1-\frac{1}{p}} \bigg(\int_{\Omega} |c|^p |u|^p dx \bigg)^{\frac{1}{p}} \\ & + (p-2+\sqrt{N}) \bigg(\int_{\Omega} |c|^{\frac{p}{2}} |Du|^p dx \bigg)^{1-\frac{2}{p}} \bigg(\int_{\Omega} |c|^p |u|^p dx \bigg)^{\frac{1}{p}} \bigg(\int_{\Omega} |D^2u|^p dx \bigg)^{\frac{1}{p}}, \end{split}$$

or, equivalently,

$$\||c|^{\frac{1}{2}}Du\|_p^2 \leq \frac{\beta p}{2}\|c\,u\|_p\||c|^{\frac{1}{2}}Du\|_p + (p-2+\sqrt{N})\|c\,u\|_p\|D^2u\|_p.$$

It follows that

$$\begin{split} \||c|^{\frac{1}{2}}Du\|_{p} &\leq \frac{\beta p}{2} \|c\,u\|_{p} + \sqrt{p-2+\sqrt{N}} \, \|c\,u\|_{p}^{\frac{1}{2}} \|D^{2}u\|_{p}^{\frac{1}{2}} \\ &\leq C\varepsilon^{-1} \|c\,u\|_{p} + \varepsilon \|D^{2}u\|_{p}, \end{split}$$

for any $\varepsilon \in (0,1)$, with C depending on β, p , and the statement follows with $\|D^2u\|_p$ instead of $\|\mathcal{A}_0u\|_p$. To complete the proof it suffices to observe that

$$||D^2u||_p \le C(||u||_p + ||\mathcal{A}_0u||_p), \tag{11.3.21}$$

for some positive constant C, independent of u (see [66, Lemma 9.17]).

In what follows to simplify the notation we set

$$q(\xi,\eta) = \sum_{i,j=1}^{N} q_{ij}\xi_i\eta_j, \qquad \xi,\eta \in \mathbb{R}^N.$$
 (11.3.22)

Moreover, we introduce the function $b': \mathbb{R}^N \to \mathbb{R}$ defined by

$$b'_i(x) = b_i(x) - \sum_{j=1}^N D_j q_{ij}(x), \qquad x \in \overline{\Omega}.$$

From Hypotheses 11.3.8 we obtain that

$$|b'(x)| \le \sigma |c(x)|^{\frac{1}{2}} + K,$$
 (11.3.23)

for a positive constant K depending only on the sup-norm of the gradient of the diffusion coefficients.

Proposition 11.3.10 For any $p \geq 2$ there exist two constants λ_0 and λ_1 such that

$$(\lambda - \lambda_0) \|u\|_p \le \|\lambda u - A_p u\|_p, \qquad \lambda > \lambda_0, \tag{11.3.24}$$

$$C_1 \|u\|_{D_p} \le \|\lambda u - A_p u\|_p \le C_2 \|u\|_{D_p}, \qquad \lambda > \lambda_1,$$
 (11.3.25)

for any $u \in D$ and some positive constants C_1, C_2 depending only on λ , N, p, β , σ , κ_0 and the sup-norm of the coefficients q_{ij} (i, j = 1, ..., N).

Proof. To shorten the notation, we denote by C any positive constant depending at most on $\lambda, N, p, \beta, \sigma$ and the coefficients q_{ij} (i, j = 1, ..., N), which may vary from line to line.

To prove (11.3.24) we multiply the identity $f = \lambda u - A_p u$ by $u|u|^{p-2}$ and integrate by parts over Ω . We get

$$\int_{\Omega} (\lambda - c)|u|^p dx + (p - 1) \int_{\Omega} |u|^{p-2} q(Du, Du) dx$$

$$= \int_{\Omega} u|u|^{p-2} f dx + \int_{\Omega} u|u|^{p-2} \langle b', Du \rangle dx.$$
(11.3.26)

Using (11.3.23), the Cauchy-Schwarz and Young inequalities we deduce that

$$\int_{\Omega} u|u|^{p-2} \langle b', Du \rangle dx \le \sigma \int_{\Omega} |c|^{\frac{1}{2}} |u|^{p-1} |Du| dx + K \int_{\Omega} |u|^{p-1} |Du| dx$$
$$\le \sigma \left(\int_{\Omega} |c| |u|^{p} dx \right)^{\frac{1}{2}} \left(\int_{\Omega} |u|^{p-2} |Du|^{2} dx \right)^{\frac{1}{2}}$$

$$+K\left(\int_{\Omega}|u|^{p-2}|Du|^{2}dx\right)^{\frac{1}{2}}\left(\int_{\Omega}|u|^{p}dx\right)^{\frac{1}{2}}$$

$$\leq \frac{1}{2}\int_{\Omega}|c||u|^{p}dx + \frac{\sigma^{2}}{2}\int_{\Omega}|u|^{p-2}|Du|^{2}dx$$

$$+\varepsilon\int_{\Omega}|u|^{p-2}|Du|^{2}dx + \frac{K^{2}}{4\varepsilon}\int_{\Omega}|u|^{p}dx. \tag{11.3.27}$$

Combining (11.3.26) and (11.3.27) gives

$$\left(\lambda - \frac{K^2}{4\varepsilon}\right) \int_{\Omega} |u|^p dx + \frac{1}{2} \int_{\Omega} |c| |u|^p dx + \left((p-1)\kappa_0 - \frac{\sigma^2}{2} - \varepsilon\right) \int_{\Omega} |u|^{p-2} |Du|^2 dx \leq \left(\int_{\Omega} |f|^p dx\right)^{\frac{1}{p}} \left(\int_{\Omega} |u|^p dx\right)^{1-\frac{1}{p}}.$$
(11.3.28)

Since $0 < \sigma < \sqrt{2\kappa_0}$, $p \ge 2$ and $c \le -1$, we can fix ε sufficiently small and, then, take $\lambda > \lambda_0 := K^2/(4\varepsilon)$ to make all the terms in the left-hand side of (11.3.28) nonnegative. The estimate (11.3.24) now easily follows.

Let us now show the first inequality in (11.3.25). We begin by estimating cu. For this purpose, we multiply the identity $\lambda u - A_p u = f$ by $|c|^{p-1}u|u|^{p-2}$ and integrate by parts over Ω . Taking (11.1.1) into account, we obtain

$$\int_{\Omega} (\lambda |c|^{p-1} + |c|^{p}) |u|^{p} dx + \kappa_{0}(p-1) \int_{\Omega} |c|^{p-1} |u|^{p-2} |Du|^{2} dx$$

$$\leq \int_{\Omega} (\lambda |c|^{p-1} + |c|^{p}) |u|^{p} dx + (p-1) \int_{\Omega} |c|^{p-1} |u|^{p-2} q(Du, Du) dx$$

$$= (p-1) \int_{\Omega} |c|^{p-2} u |u|^{p-2} q(Du, Dc) dx + \int_{\Omega} |c|^{p-1} u |u|^{p-2} \langle b', Du \rangle dx$$

$$+ \int_{\Omega} |c|^{p-1} u |u|^{p-2} f dx. \tag{11.3.29}$$

Let us estimate the last side of (11.3.29). Arguing as in the proof of (11.3.27), we get

$$\left| \int_{\Omega} |c|^{p-1} u |u|^{p-2} \langle b', Du \rangle dx \right|$$

$$\leq \left(\frac{\sigma^2}{2} + \varepsilon \right) \int_{\Omega} |c|^{p-1} |u|^{p-2} |Du|^2 dx + \frac{1}{2} \int_{\Omega} |c|^p |u|^p dx + \frac{K^2}{4\varepsilon} \int_{\Omega} |c|^{p-1} |u|^p dx,$$
(11.3.30)

for any $\varepsilon > 0$. Similarly, using Hypothesis 11.3.8(ii), we get

$$\int_{\Omega} |c|^{p-2} |u|^{p-1} |q(Du, Dc)| dx$$

$$\leq \|Q\|_{\infty} \int_{\Omega} |c|^{p-2} |Dc| |u|^{p-1} |Du| dx$$

$$\leq \|Q\|_{\infty} \beta \left(\int_{\Omega} |c|^{p-1} |u|^{p-2} |Du|^{2} dx \right)^{\frac{1}{2}} \left(\int_{\Omega} |c|^{p-1} |u|^{p} dx \right)^{\frac{1}{2}}$$

$$\leq \varepsilon \int_{\Omega} |c|^{p-1} |u|^{p-2} |Du|^{2} dx + \frac{\|Q\|_{\infty}^{2} \beta^{2}}{4\varepsilon} \int_{\Omega} |c|^{p-1} |u|^{p} dx. \tag{11.3.31}$$

Summing up, from (11.3.29)-(11.3.31) we get

$$\left(\lambda - \frac{K^2}{4\varepsilon} - (p-1)\frac{\|Q\|_{\infty}^2 \beta^2}{4\varepsilon}\right) \int_{\Omega} |c|^{p-1} |u|^p dx + \frac{1}{2} \int_{\Omega} |c|^p |u|^p dx$$

$$+ \left((p-1)(\kappa_0 - \varepsilon) - \frac{\sigma^2}{2} - \varepsilon\right) \int_{\Omega} |c|^{p-1} |u|^{p-2} |Du|^2 dx$$

$$\leq \left(\int_{\Omega} |f|^p dx\right)^{\frac{1}{p}} \left(\int_{\Omega} |c|^p |u|^p dx\right)^{1-\frac{1}{p}}.$$

Hence, choosing ε sufficiently small and, then,

$$\lambda > \lambda_1 := \frac{K^2 + (p-1)\|Q\|_{\infty}^2 \beta^2}{4\varepsilon},$$

we get

$$||c u||_p \le 2||f||_p. \tag{11.3.32}$$

We now use Lemma 11.3.9 with $\varepsilon=(2\sigma)^{-1}$ to estimate the second-order derivatives of u. We have

$$\begin{split} \|\langle b, Du \rangle\|_{p} &\leq \sigma \||c|^{\frac{1}{2}} Du\|_{p} \\ &\leq \frac{1}{2} \|\mathcal{A}_{0}u\|_{p} + 2\sigma^{2} C \|u\|_{p} + 2\sigma^{2} C \|cu\|_{p} \\ &\leq \frac{1}{2} \|f\|_{p} + \frac{1}{2} \|\langle b, Du \rangle\|_{p} + \left(\frac{1}{2} + 2\sigma^{2} C\right) \|cu\|_{p} + \left(\frac{\lambda}{2} + 2\sigma^{2} C\right) \|u\|_{p}. \end{split}$$

Hence, taking (11.3.24) and (11.3.32) into account, we get $\|\langle b, Du \rangle\|_p \leq C \|f\|_p$ and, then, by difference, $\|\mathcal{A}_0 u\|_p \leq C \|f\|_p$. From (11.3.21) we get $\|D^2 u\|_p \leq C \|f\|_p$. Moreover, since $c \leq -1$, from (11.3.20) we deduce that

$$||Du||_p \le C||f||_p.$$

Now, the first inequality in (11.3.25) follows.

To prove the other estimate in (11.3.25) it suffices to show that

$$\|\langle b, Du \rangle\|_p \le C \|u\|_{D_p},$$
 (11.3.33)

for any $u \in D$, and this can easily be done, taking (11.3.20), with $\varepsilon = 1$, and Hypothesis (11.3.8)(ii) into account.

Remark 11.3.11 The estimate (11.3.33) and the density of D in D_p show that the operator A_p is well defined in D_p . We still denote by A_p the operator so extended to D_p . Of course, such an operator satisfies all the results in Proposition 11.3.10. In particular, by (11.3.25), the graph norm of A_p is equivalent to the norm of D_p .

Proposition 11.3.12 For any $p \in [2, +\infty)$, (A_p, D_p) generates a strongly continuous analytic semigroup $\{S(t)\}$ in $L^p(\Omega)$.

Proof. Without loss of generality, we replace \mathcal{A} with the operator $B_p = A_p - kI$, defined in $D(B_p) = D_p$, where $k \geq 1$ is a suitable positive constant to be fixed.

We split the proof into two steps. First we prove that B_p generates a strongly continuous semigroup in $L^p(\Omega)$ and then, in Step 2, we show that such a semigroup is analytic.

Step 1. To begin with, we fix $\sigma' \in (\sigma, \sqrt{2\kappa_0})$ and k > 0 sufficiently large such that

$$|b'(x)| \le \sigma' |c'(x)|^{\frac{1}{2}}, \qquad x \in \Omega,$$
 (11.3.34)

where c' = c - k. Since now we can take K = 0 in (11.3.23), the estimate (11.3.28) implies that B_p is dissipative in $L^p(\Omega)$.

Let us show $\lambda - B_p$ is surjective for some $\lambda > 0$. The Lumer-Phillips theorem (see Theorem B.1.7), then, will imply that B_p is the generator of a strongly continuous semigroup of contractions. For this purpose, for any $\varepsilon > 0$, we set

$$b_{\varepsilon} = \frac{b}{\sqrt{1 - \varepsilon c}}, \qquad c'_{\varepsilon} = \frac{c}{1 - \varepsilon c} - k.$$
 (11.3.35)

As it is immediately seen, b_{ε} and c_{ε} satisfy Hypothesis 11.3.8, uniformly with respect to $\varepsilon > 0$ with the same constants σ and β , provided that k > 1. Since b_{ε} and c_{ε} are bounded, the realization $B_{\varepsilon,p}$ of the operator $\mathcal{B}_{\varepsilon} = \mathcal{A}_0 + \langle b_{\varepsilon}, D \rangle + c'_{\varepsilon}$ in $L^p(\Omega)$, with domain $D(B_{\varepsilon,p}) = W^{2,p}(\Omega) \cap W_0^{1,p}(\Omega)$, generates an analytic semigroup (see Theorem C.3.6(ii)). Moreover, (11.3.24) implies that there exists λ_0 such that

$$(\lambda - \lambda_0) \|u\|_p \le \|\lambda u - B_{\varepsilon,p} u\|_p, \qquad u \in D(B_{\varepsilon,p}), \quad \lambda > \lambda_0,$$

for any $\varepsilon > 0$. Since $D(B_{\varepsilon,p})$ is dense in $L^p(\Omega)$, applying the Lumer-Phillips theorem to the operator $B_{\varepsilon,p} - \lambda_0 I$, we easily deduce that the resolvent of $B_{\varepsilon,p}$ contains the half-line $(\lambda_0, +\infty)$ for any $\varepsilon \in (0,1)$.

Now, given $f \in L^p(\Omega)$, let $u_{\varepsilon} \in W^{2,p}(\Omega) \cap W_0^{1,p}(\Omega)$ be such that $(2\lambda_0 - B_{\varepsilon,p})u_{\varepsilon} = f$. According to Proposition 11.3.10 we know that

$$||u_{\varepsilon}||_{2,p} + ||c_{\varepsilon}u_{\varepsilon}||_{p} \le C||f||_{p},$$

for some positive constant C, independent of ε . By a weak compactness argument, we can determine a sequence $\{\varepsilon_n\}$ converging to 0 such that u_{ε_n} tends weakly to a function u in $W^{2,p}(\Omega) \cap W_0^{1,p}(\Omega)$. Moreover, we can assume that u_{ε_n} tends to u both strongly in $W_{\text{loc}}^{1,p}(\Omega)$ and a.e. in Ω . Applying Fatou's lemma, we easily see that $\|cu\|_p \leq C\|f\|_p$. Hence, $u \in D(B_p)$. Similarly, it is easy to check that u solves the differential equation $\lambda u - B_p u = f$. Therefore, the operator $2\lambda_0 - B_p$ is surjective, and the Lumer-Phillips theorem applies.

Step 2. We now prove that the semigroup $\{R(t)\}$ generated by B_p is analytic. For this purpose, we fix $u \in D$ and set $u^* := \overline{u}|u|^{p-2}$. An integration by parts shows that

$$-\operatorname{Re} \int_{\Omega} u^* B_p u \, dx = \operatorname{Re} \int_{\Omega} \left(\frac{p}{2} |u|^{p-2} q(Du, D\overline{u}) + \frac{p-2}{2} \overline{u}^2 |u|^{p-4} q(Du, Du) \right) dx$$
$$-\operatorname{Re} \int_{\Omega} \overline{u} |u|^{p-2} \langle b', Du \rangle \, dx - \int_{\Omega} c' |u|^p dx, \tag{11.3.36}$$

where q is given by (11.3.22). Since

$$\operatorname{Re}(|u|^2 q(Du, D\overline{u})) = q(\operatorname{Re}(\overline{u}Du), \operatorname{Re}(\overline{u}Du)) + q(\operatorname{Im}(\overline{u}Du), \operatorname{Im}(\overline{u}Du))$$
(11.3.37)

and

$$\operatorname{Re}\left(\overline{u}^{2}q(Du,Du)\right) = q(\operatorname{Re}(\overline{u}Du),\operatorname{Re}(\overline{u}Du)) - q(\operatorname{Im}(\overline{u}Du),\operatorname{Im}(\overline{u}Du)),$$
(11.3.38)

from (11.3.36)-(11.3.38) we easily deduce that

$$-\operatorname{Re} \int_{\Omega} u^* B_p u dx = (p-1) \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Re}(\overline{u}Du)) dx$$
$$+ \int_{\Omega} |u|^{p-4} q(\operatorname{Im}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx$$
$$- \int_{\Omega} |u|^{p-2} \langle b', \operatorname{Re}(\overline{u}Du) \rangle dx - \int_{\Omega} c' |u|^p dx. \quad (11.3.39)$$

The condition (11.3.34) and the Hölder inequality imply that

$$\left| \int_{\Omega} |u|^{p-2} \langle b', \operatorname{Re}(\overline{u}Du) \rangle dx \right|$$

$$\leq \sigma' \int_{\Omega} |c'|^{\frac{1}{2}} |\operatorname{Re}(\overline{u}Du)| |u|^{p-2} dx$$

$$\leq \sigma' \left(\int_{\Omega} |c'| |u|^p dx \right)^{\frac{1}{2}} \left(\int_{\Omega} |u|^{p-4} |\operatorname{Re}(\overline{u}Du)|^2 dx \right)^{\frac{1}{2}} \\
\leq \frac{\sigma'}{\sqrt{\kappa_0}} \left(\int_{\Omega} |c'| |u|^p dx \right)^{\frac{1}{2}} \left(\int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Re}(\overline{u}Du)) dx \right)^{\frac{1}{2}}. (11.3.40)$$

If we set

$$\begin{split} F^2 &:= \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Re}(\overline{u}Du)) dx, \\ G^2 &:= \int_{\Omega} |u|^{p-4} q(\operatorname{Im}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx, \\ H^2 &:= \int_{\Omega} |c'| |u|^p dx, \end{split}$$

from (11.3.39) and (11.3.40) we get

$$-\operatorname{Re} \int_{\Omega} u^* B_p u \, dx \ge \left(p - 1 - \frac{(\sigma')^2}{2\kappa_0} \right) F^2 + G^2 + \frac{1}{2} H^2. \tag{11.3.41}$$

Now, we consider the imaginary part of $\int_{\Omega} u^* B_p u dx$. Integrating by parts, we can easily show that

$$\operatorname{Im} \int_{\Omega} u^* B_p u \, dx = -\operatorname{Im} \int_{\Omega} \left(\frac{p}{2} |u|^{p-2} q(Du, D\overline{u}) + \frac{p-2}{2} \overline{u}^2 |u|^{p-4} q(Du, Du) \right) dx + \operatorname{Im} \int_{\Omega} \overline{u} |u|^{p-2} \left\langle b', Du \right\rangle dx. \tag{11.3.42}$$

Since $\operatorname{Im}(q(Du, D\overline{u})) = 0$ and $\operatorname{Im}(q(\overline{u}Du, \overline{u}Du)) = 2q(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du))$, we get

$$\operatorname{Im} \int_{\Omega} u^* B_p u \, dx = (p-2) \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx + \operatorname{Im} \int_{\Omega} \overline{u} |u|^{p-2} \langle b', Du \rangle \, dx.$$
 (11.3.43)

Using, first, the Cauchy-Schwarz inequality for the inner product induced by the matrix Q and, then, the Hölder inequality, we can show that

$$\left| \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx \right| \le F G,$$

whereas the other term in the right-hand side of (11.3.43) can be estimated as in (11.3.40), replacing everywhere the real part of $\overline{u}Du$ with its imaginary part. Summing up, we get

$$\left| \operatorname{Im} \int_{\Omega} u^* B_p u \, dx \right| \le (p-2)FG + \frac{\sigma'}{\sqrt{\kappa_0}} GH. \tag{11.3.44}$$

Combining (11.3.41) and (11.3.44), it follows that we can determine a positive constant C such that

$$\left| \operatorname{Im} \int_{\Omega} u^* B_p u \, dx \right| \le C \left(-\operatorname{Re} \int_{\Omega} u^* B_p u \, dx \right), \tag{11.3.45}$$

for any $u \in D$. By density, (11.3.45) can be extended to any $u \in D(B_p)$.

Formula (11.3.45) immediately implies that the numerical range $r(B_p)$ of B_p is contained in the sector $\{\lambda \in \mathbb{C} : |\operatorname{Im} \lambda| \leq -C \operatorname{Re} \lambda\}$ so that, applying Theorem A.3.6, we can complete the proof.

To conclude this subsection we show some regularity properties of the function $S(\cdot)f$.

Proposition 11.3.13 Let p > N+1. Then, for any $f \in C_c^{\infty}(\Omega)$ the function $u = S(\cdot)f$ is the bounded classical solution of the Cauchy-Dirichlet problem (11.0.1). Moreover, $u \in C^{1+\alpha/2,2+\alpha}((0,T) \times (\Omega \cap B(R)))$ for any R > 0. Further, $Du \in C_b([0,T] \times \overline{\Omega}) \cap C^{1+\alpha/2,2+\alpha}((\varepsilon,T) \times \Omega')$ for any $0 < \varepsilon < T$, any $\Omega' \subset\subset \Omega$ and any R > 0 and it belongs to $C^{1+\alpha/2,2+\alpha}((0,T) \times (\Omega \cap B(R)))$.

Proof. Fix $f \in C_c^{\infty}(\Omega) \subset D(A_p)$. Since $C_c^{\infty}(\Omega)$ is dense in $L^p(\Omega)$, according to Proposition 11.3.10 and Remark B.2.6, the function $t \mapsto S(t)f$ is continuous in $[0, +\infty)$ with values in $W^{2,p}(\Omega)$. Since p > N+1, the Sobolev embedding theorems (see [2, Theorem 5.4]) and Remark B.2.6 imply that both u and Du are bounded and continuous in $[0, T] \times \overline{\Omega}$ for any T > 0. To conclude that u is a classical solution to (11.0.1), we have to show that $u \in C^{1,2}((0, +\infty) \times \Omega)$. Since $\{S(t)\}$ is analytic, the function u is continuously differentiable in $[0, +\infty)$ with values in $W^{2,p}(\Omega)$ (see again Remark B.2.6). The Sobolev embedding theorems yield $D_t u \in C([0, +\infty); C_b(\overline{\Omega}))$.

Now, fix $\varepsilon \in (0,T)$ and set

$$\tau = \sup_{\varepsilon \le t \le T} \left(\|u(t, \cdot)\|_{W^{2,p}(\Omega)} + \|D_t u(t, \cdot)\|_{W^{2,p}(\Omega)} \right). \tag{11.3.46}$$

Since $u(t,\cdot) \in D(A_p^n)$ for any $t \in [\varepsilon,T]$ and any $n \in \mathbb{N}$, the function $A_p u(t,\cdot)$ belongs to $W^{2,p}(\Omega)$. Observing that

$$Tr(QD^{2}u(t,\cdot)) = A_{p}u(t,\cdot) - \langle b, Du(t,\cdot) \rangle - cu(t,\cdot)$$
$$= D_{t}u(t,\cdot) - \langle b, Du(t,\cdot) \rangle - cu(t,\cdot), \qquad t \in (\varepsilon, T],$$

and $u(t,\cdot) \in W^{2,p}(\Omega)$, by difference, we deduce that $\operatorname{Tr}(QD^2u(t,\cdot)) \in W^{1,p}_{\operatorname{loc}}(\Omega)$. By local regularity results for elliptic equations in L^p -spaces, we deduce that $u(t,\cdot) \in W^{3,p}_{\operatorname{loc}}(\Omega)$ for any $t \in [\varepsilon,T]$ (see Theorem C.1.1). Moreover, since the coefficients of the operator $\mathcal A$ are locally bounded in Ω , from (11.3.46) we deduce that, for any pair of bounded open sets $\Omega_1 \subset\subset \Omega_2 \subset\subset \Omega$, it holds that

$$||u(t,\cdot)||_{W^{3,p}(\Omega_1)} \le C_1 \left(||A_p u(t,\cdot) - \langle b, D u(t,\cdot) \rangle - c u(t,\cdot)||_{W^{1,p}(\Omega_2)} + ||u||_{L^p(\Omega_2)} \right)$$

$$\le C_2 \tau, \tag{11.3.47}$$

for some positive constants $C_1 = C_1(\Omega_1, \Omega_2)$ and $C_2 = C_2(\Omega_1, \Omega_2)$ and any $t \in [\varepsilon, T]$. We have thus proved that $D^2u(t, \cdot) \in W^{1,p}(\Omega_1)$ for any $t \in [\varepsilon, T]$. Since p > N + 1, the Sobolev embedding theorems imply that $D^2u(t, \cdot) \in C^{\theta}(\Omega_1)$ for some $\theta \in (0, 1)$ and, by virtue of (11.3.47),

$$\sup_{\varepsilon < t < T} \|D^2 u(t, \cdot)\|_{C^{\theta}(\Omega_1)} < +\infty. \tag{11.3.48}$$

Now, without loss of generality, we assume that Ω_1 has a smooth boundary. Therefore, applying Proposition A.4.6 with $X=C(\overline{\Omega}_1), Y=C^{2+\theta}(\overline{\Omega}_1), Z=C^2(\overline{\Omega}_1), I=[\varepsilon,T]$ and $\theta=0$ (and taking Proposition A.4.4 into account), we immediately deduce that $D^2u\in C([\varepsilon,T]\times\overline{\Omega}_1)$ and the arbitrariness of ε,T and Ω_1 implies that $D^2u\in C((0,+\infty)\times\Omega)$. We have so proved that u is a classical solution to the problem (11.0.1).

The last part of the assertion follows from Proposition 11.1.3 and Theorem 11.2.1. This finishes the proof.

11.3.3 Proof of Theorem 11.3.4

To prove that the bounded classical solution to the Cauchy-Dirichlet problem (11.0.1) satisfies the gradient estimate (11.0.3) we use an approximation argument. First we prove the assertion in the case when $f \in C_c^{\infty}(\Omega)$ and, then, in the general case.

Step 1. Let $f \in C_c^{\infty}(\Omega)$. For any $\varepsilon \in (0,1)$, let $c_{\varepsilon} : \mathbb{R}^N \to \mathbb{R}$ be defined by

$$c_{\varepsilon}(x) = -\varepsilon \exp\left(4d_2\sqrt{1+|x|^2}\right), \qquad x \in \mathbb{R}^N.$$

Taking (11.3.5) into account, it can be easily shown that for any $\sigma > 0$ there exists $C_{\sigma,\varepsilon} > 0$ such that

$$|b| \le \sigma |c_{\varepsilon} + c|^{1/2} + C_{\sigma,\varepsilon}. \tag{11.3.49}$$

Moreover, using (11.3.4), we get

$$|D(c_{\varepsilon} + c)| \le \beta_0 (1 - c_{\varepsilon} - c), \tag{11.3.50}$$

where $\beta_0 := \max(\beta, 4d_2)$, β and d_2 being given by (11.3.4) and (11.3.5).

Let $\mathcal{A}_{\varepsilon} = \mathcal{A} + c_{\varepsilon} - k$. Taking (11.3.49) and (11.3.50) into account, one can check that, if k is sufficiently large, the operator $\mathcal{A}_{\varepsilon}$ satisfies Hypotheses 11.3.8 with β, σ being replaced by β_0 and some $\sigma' \in (\sigma_0, \sqrt{2\kappa_0})$. Hence, it generates an analytic semigroup $\{S_{\varepsilon}(t)\}$ in $L^p(\Omega)$ for any $p \geq 2$. By Proposition 11.3.13 we know that if p > N + 1, the function $u_{\varepsilon} = S_{\varepsilon}(\cdot)f$ is the bounded classical solution to the problem (11.0.1), where we replace \mathcal{A} with the operator $\mathcal{A}_{\varepsilon}$. Moreover, Du_{ε} is bounded and continuous in $[0, T] \times \overline{\Omega}$, for any T > 0, and it belongs to $C^{1,2}((0, +\infty) \times \Omega)$. Hence, u_{ε} satisfies the assumptions of Proposition 11.3.7. Since the coefficients of the operator $\mathcal{A}_{\varepsilon}$ satisfy Hypotheses 11.3.1

with constants being independent of $\varepsilon \in (0,1)$ (see (11.3.49) and (11.3.50)), we can find out a positive constant C = C(T), independent of ε , such that

$$||Du_{\varepsilon}(t,\cdot)||_{\infty} \le \frac{C}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0,T).$$
(11.3.51)

Since, for any $\alpha \in (0,1)$, the C^{α} -norms of the coefficients of the operator $\mathcal{A}_{\varepsilon}$ are bounded, in bounded subsets of Ω , uniformly with respect to ε , arguing as in Step 1 of the proof of Theorem 11.2.1, we can show that, up to a subsequence, u_{ε} converges to $T(\cdot)f$ in $C^{1,2}([0,T]\times(\Omega\cap B(R)))$, for any R>0, as ε tends to 0^+ . Moreover, letting ε go to 0^+ in (11.3.51), we get

$$||DT(t)f||_{\infty} \le \frac{C}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0, T),$$
 (11.3.52)

and the assertion follows in this case.

Step 2. Given a function $f \in C_b(\Omega)$, we approximate it by a sequence $\{f_n\} \in C_c^{\infty}(\Omega)$ of smooth functions converging to f uniformly on compact subsets of Ω and such that $\|f_n\|_{\infty} \leq \|f\|_{\infty}$ for any $n \in \mathbb{N}$. Arguing as in Step 2 of the proof of Theorem 11.2.1, we easily see that, up to a subsequence, $T(\cdot)f_n$ converges to $T(\cdot)f$ in $C^{1,2}(F)$ for any compact set $F \subset (0,T) \times \Omega$ and any T > 0. By (11.3.52), we know that

$$||DT(t)f_n||_{\infty} \le \frac{C}{\sqrt{t}} ||f_n||_{\infty} \le \frac{C}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0, T).$$

Letting n go to $+\infty$, we get the assertion.

11.3.4 A counterexample to the gradient estimates

In this subsection we show that the gradient estimate (11.0.3) fails, in general, if Hypothesis 11.3.1 is not satisfied. The following result generalizes an example in [144].

Example 11.3.14 Let us consider the following Cauchy-Dirichlet problem in $\mathbb{R}^2_+ = \{(x,y) \in \mathbb{R}^2, \ x > 0\}$

$$\begin{cases}
D_t u(t, x, y) \in \mathbb{R}, & x > 0 \\
D_t u(t, x, y) = \Delta u(t, x, y) + g(y) D_x u(t, x, y), & t > 0, (x, y) \in \mathbb{R}_+^2, \\
u(t, 0, y) = 0, & t > 0, y \in \mathbb{R}, \\
u(0, x, y) = 1, & (x, y) \in \mathbb{R}_+^2, \\
(11.3.53)
\end{cases}$$

where $g(y) = \sqrt{1 + y^2}$ for any $y \in \mathbb{R}$.

Observe that (11.3.3) fails. Indeed, in this situation d(x, y) = x and, consequently, the condition (11.3.3) reads as

$$\sqrt{1+y^2} \le M,$$
 $(x,y) \in \Omega_{\delta} = \{(x,y) \in \mathbb{R}^2 : x \in (0,\delta)\},\$

which, of course, leads us to a contradiction, letting y go to ∞ . However, Theorem 11.2.1 guarantees that the problem (11.3.53) admits a unique bounded classical solution u. Since Ω is convex, to prove that the gradient estimate does not hold, it suffices to show that, for any t > 0, $u(t, \cdot)$ is not uniformly continuous in Ω . This can be done showing that, for any t, x > 0,

$$\sup_{y>0} u(t, x, y) = 1. \tag{11.3.54}$$

Indeed, if $u(t,\cdot)$ were uniformly continuous in \mathbb{R}^2_+ , then, for any t>0, the function $x\mapsto w_t(x)=\sup_{y>0}u(t,x,y)$ would be continuous in $[0,+\infty)$. Since $w_t(0)=0$, we would get $w_t(x)<1$ for any x in a sufficiently small neighborhood of x=0, contradicting (11.3.54).

To prove (11.3.54), we fix n > 0 and take $r_n = \sqrt{n^2 - 1}$. Moreover, we denote by $v = v_n$ the classical solution to the problem

$$\begin{cases} D_t v(t, x, y) = \Delta v(t, x, y) + n D_x v(t, x, y), & t > 0, & (x, y) \in R_n, \\ v(t, x, y) = 0, & t > 0, & (x, y) \in \partial R_n, \\ v(0, x, y) = 1, & (x, y) \in R_n, \end{cases}$$

where $R_n = (0, +\infty) \times (r_n, +\infty)$. We are going to show that for any t, x > 0

(i)
$$\lim_{n \to +\infty} \sup_{y > r_n} v_n(t, x, y) = 1;$$
 (ii) $u(t, x, y) \ge v_n(t, x, y), \quad n \in \mathbb{N}.$ (11.3.55)

Clearly (i) and (ii) give (11.3.54). Indeed, they give $\sup_{y>0} u(t,x,y) \geq 1$ and the maximum principle in Proposition 11.1.3 implies that $u \leq 1$.

Let us prove (11.3.55)(i). We look for the solution v_n in the form $v_n(t, x, y) = a_n(t, x)b_n(t, y)$, with $a = a_n$ and $b = b_n$ solving, respectively, the Cauchy problems

$$\begin{cases} D_t a(t,x) = D_{xx} a(t,x) + n D_x a(t,x), & t > 0, \ x > 0, \\ a(t,0) = 0, & t > 0, \ x > 0, \\ a(0,x) = 1, & x > 0 \end{cases}$$

and

$$\begin{cases} D_t b(t, y) = D_{yy} b(t, y), & t > 0, \ y > r_n, \\ b(t, r_n) = 0, & t > 0, \\ b(0, y) = 1, & y > r_n. \end{cases}$$
(11.3.56)

To obtain an explicit formula for a_n , we begin by observing that $a_n(t,x) = a_1(n^2t, nx)$. Next, setting $z(t,x) = e^{x/2}e^{t/4}a_1(t,x)$, we see that z solves the

Cauchy problem

$$\begin{cases} D_t z(t,x) = D_{xx} z(t,x), & t > 0, \ x > 0, \\ z(t,0) = 0, & t > 0, \\ z(0,x) = e^{\frac{x}{2}}, & x > 0. \end{cases}$$
 (11.3.57)

Due to the well known representation formulas for the solutions to (11.3.56) and (11.3.57), we can write

$$a_n(t,x) = \frac{e^{-\frac{n^2t}{4}}}{n\sqrt{4\pi t}} \int_0^{+\infty} \left(e^{-\frac{|nx-z|^2}{4n^2t}} - e^{-\frac{|nx+z|^2}{4n^2t}} \right) e^{\frac{z-nx}{2}} dz$$

$$= \frac{1}{\sqrt{\pi}} \int_{-\frac{x+nt}{2\sqrt{t}}}^{+\infty} e^{-s^2} ds - \frac{e^{-nx}}{\sqrt{\pi}} \int_{\frac{x-nt}{2\sqrt{t}}}^{+\infty} e^{-s^2} ds, \qquad (11.3.58)$$

$$b_n(t,y) = \frac{1}{\sqrt{\pi t}} \int_0^{y-r_n} e^{-\frac{z^2}{4t}} dz,$$

for any t, x > 0, any $y > r_n$ and any $n \in \mathbb{N}$. Hence,

$$\sup_{y>r_n} a_n(t, x, y) = a_n(t, x) \sup_{y>r_n} b_n(t, y) = a_n(t, x).$$

Letting n go to $+\infty$, from (11.3.58) we immediately get (11.3.55).

To prove (11.3.55)(ii) we use the maximum principle in Proposition 11.1.3. Set $w = u - v_n$ in $(0, +\infty) \times R_n$. We have w = 0 in $\{0\} \times R_n$. Moreover, $w \ge 0$ in $(0, +\infty) \times \partial R_n$. Indeed, the quoted maximum principle implies that $u \ge 0$ in $(0, +\infty) \times \Omega$. To conclude that $u \ge v_n$ in $(0, +\infty) \times \mathbb{R}^2_+$, it suffices to show that

$$D_t w(t, x, y) \ge \Delta w(t, x, y) + g(y) D_x w(t, x, y), \qquad t > 0, \quad (x, y) \in R_n.$$
(11.3.59)

For this purpose we observe that

$$D_t w(t, x, y) = \Delta w(t, x, y) + g(y) D_x w(t, x, y) + [g(y) - n] D_x a_n(t, x) b_n(t, y),$$

for any t > 0 and any $(x, y) \in R_n$. From (11.3.58) we easily deduce that $a_n(t, \cdot)$ is increasing in $(0, +\infty)$. Therefore, since $\sqrt{1+y^2} \ge n$ for any $y \in (r_n, +\infty)$, it follows that $[g(\cdot) - n]D_x a_n b_n \ge 0$ in $(0, +\infty) \times R_n$ and (11.3.59) follows.

Remark 11.3.15 We remark that, if we take $g(y) = -\sqrt{1+y^2}$ in (11.3.53), then Hypotheses 11.1.1 and 11.3.1 are satisfied. Therefore, the gradient estimate (11.0.3) holds. This shows that (11.3.2)-(11.3.5) are not merely conditions on the growth rate at infinity of the coefficients.

Remark 11.3.16 As it has been claimed in the introduction of this chapter, as far as we know, uniform estimates for higher order derivatives of the function T(t)f seem to be not available in a general domain Ω . This prevents us to perform the same techniques as in Chapter 6 to prove optimal Schauder estimates for both the nonhomogeneous elliptic problem with Dirichlet boundary conditions and the nonhomogeneous Cauchy-Dirichlet problem associated with the operator \mathcal{A} .

The situation is different when Ω is an exterior domain. Indeed, very recently, estimates for the higher-order derivatives of the function T(t)f have been proved in [71].

We remark that in the particular case when Ω is the halfspace $\Omega = \{x \in \mathbb{R}^N : \langle x, v \rangle > 0\}$ (v being an unitary vector in \mathbb{R}^N) and \mathcal{A} is the operator defined by

$$Au(x) = \Delta u(x) + \sum_{i,j=1}^{N} b_{ij} x_j D_i u(x), \qquad x \in \Omega,$$

where $B = (b_{ij})$ is a matrix such that v is an eigenvector of both B and B^* , optimal Schauder estimates for the elliptic problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ u(x) = 0, & x \in \partial\Omega, \end{cases}$$
 (11.3.60)

 $(\lambda \geq 0)$, are available and they have been proved by E. Priola in [127]. To state them, we need to introduce some notation. We denote by Λ any orthogonal matrix such that $\Lambda(\Omega) := \mathbb{R}^N_+ = \{x = (x_1, \dots, x_N) \in \mathbb{R}^N : x_1 > 0\}$. Moreover we set $\widetilde{B} = \Lambda B \Lambda^*$ and we denote by $\widetilde{B}_1 \in L(\mathbb{R}^{N-1})$ the matrix obtained from \widetilde{B} by erasing the first line and the first row.

Assuming that $f \in C_b^{\alpha}(\mathbb{R}^N_+)$ for some $\alpha \in (0,1)$, E. Priola shows that, for any $\lambda \geq 0$, the problem (11.3.60) admits a unique solution $u \in C_b^{2+\alpha}(\mathbb{R}^N_+)$ if and only if

$$[[f]] := \sup_{\substack{t>0\\y\in\mathbb{R}^{N-1}}} t^{-\frac{\alpha}{2}} |f(\Lambda^{-1}(0,e^{t\widetilde{B}_1y})) - f(\Lambda^{-1}(0,y))| < +\infty.$$

In such a case, there exists a positive constant C, independent of f, such that

$$||u||_{C_b^{2+\alpha}(\Omega)} \le C([[f]] + ||f||_{C_b^{\alpha}(\mathbb{R}^N)}).$$

This result shows that, differently from what happens when $\Omega = \mathbb{R}^N$, when we have Dirichlet boundary conditions on $\partial\Omega$ (and Ω is not an exterior domain), we cannot expect, in general, to prove optimal Schauder estimates for the solution to the elliptic equation in Ω , only assuming that the data are Hölder continuous in Ω : we need to assume some additional conditions on them.

Chapter 12

The Cauchy-Neumann problem: the convex case

12.0 Introduction

Let Ω be a smooth convex unbounded domain in \mathbb{R}^N . In this chapter we consider both the parabolic problem with homogeneous Neumann boundary conditions

$$\begin{cases} D_t u(t,x) - \mathcal{A} u(t,x) = 0, & t > 0, \quad x \in \Omega, \\ \frac{\partial u}{\partial \nu}(t,x) = 0, & t > 0, \quad x \in \partial \Omega, \\ u(0,x) = f(x), & x \in \Omega \end{cases}$$
 (12.0.1)

and the elliptic problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ \frac{\partial u}{\partial \nu}(x) = 0, & x \in \partial\Omega, \end{cases}$$
 (12.0.2)

when $f \in C_b(\overline{\Omega})$. Here, \mathcal{A} is given, as usual, by

$$\mathcal{A}\varphi(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}\varphi(x) + \sum_{j=1}^{N} b_{j}(x)D_{i}\varphi(x) + c(x)\varphi(x), \qquad x \in \Omega,$$
(12.0.3)

on smooth functions and $\nu(x)$ denotes the outer unit normal to $\partial\Omega$ at $x\in\partial\Omega$. Under suitable assumptions on the coefficients of the operator \mathcal{A} , we show that the Cauchy-Neumann problem (12.0.1) admits a unique classical solution u (i.e., a function $u\in C([0,+\infty)\times\overline\Omega)\cap C^{0,1}((0,+\infty)\times\overline\Omega)\cap C^{1,2}((0,+\infty)\times\Omega)$ solving the Cauchy-Neumann problem (12.0.1) pointwise) which is bounded in $(0,T)\times\Omega$ for any T>0). This will allow us to associate a semigroup $\{T(t)\}$ of bounded operators with the Cauchy-Neumann problem (12.0.1) by setting $T(t)f=u(t,\cdot)$ for any t>0.

As in Chapter 2, the solution to the problem (12.0.1) is obtained by approximating our problem with a sequence of Cauchy-Neumann problems in (convex) bounded domains Ω_n $(n \in \mathbb{N})$. The Neumann boundary condition

gives some problems. Indeed, differently to what happens in the case considered in Chapter 2, it is not immediate to show that the solutions u_n to the Cauchy-Neumann problems in the bounded sets Ω_n converge to a solution to the problem (12.0.1). To overcome such a difficulty, we prove an a priori gradient estimate for the functions u_n , with constants being independent of n. This forces us to assume stronger hypotheses on the coefficients than those in Chapter 2. In particular, we have to assume some dissipativity and growth conditions on the coefficients of the operator \mathcal{A} .

Once this gradient estimate and a suitable maximum principle for bounded classical solutions to the Cauchy-Neumann problem (12.0.1) is proved, one can show that the sequence $\{u_n\}$ converges to a bounded classical solution u to the problem (12.0.1) and that T(t)f satisfies the following estimate:

$$||DT(t)f||_{\infty} \le \frac{C_T}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0, T),$$
 (12.0.4)

for any $f \in C_b(\overline{\Omega})$, any T > 0 and some positive constant C_T , independent of f. Moreover, one can also show that

$$||DT(t)f||_{\infty} \le C||f||_{C_{h}^{1}(\overline{\Omega})}, \qquad t > 0,$$
 (12.0.5)

for any $f \in C^1_{\nu}(\overline{\Omega})$ and C, as above, is a positive constant independent of f. The gradient estimates

$$||Du_n(t,\cdot)||_{\infty} \le \frac{C_T}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0,T), \ f \in C_b(\overline{\Omega}_n)$$

and

$$||Du_n(t,\cdot)||_{\infty} \le C||f||_{C^1_{\nu}(\overline{\Omega}_n)}, \qquad t \in (0,T), \ f \in C^1_{\nu}(\overline{\Omega}_n)$$

are proved by using the Bernstein method as we did in Chapter 6. Here $c_0 = \sup_{\Omega} c$ and C_T and C are two positive constants independent, respectively, of $t \in (0,T)$ and of t > 0. Moreover, they are independent of n as well.

Concerning the elliptic problem (12.0.2), we show that, for any $\lambda > c_0 := \sup_{\Omega} c(x)$ and any $f \in C_b(\overline{\Omega})$, it admits a unique solution $u \in D(A)$, where

$$D(\mathcal{A}) = \left\{ u \in C_b(\overline{\Omega}) \cap \bigcap_{1 \le p < +\infty} W^{2,p}(\Omega \cap B(R)) \text{ for any } R > 0 :$$

$$\mathcal{A}u \in C_b(\overline{\Omega}), \frac{\partial u}{\partial \nu}(x) = 0 \text{ for any } x \in \partial \Omega \right\}.$$

$$(12.0.6)$$

Moreover, using (12.0.4) and (12.0.5), we then show that $D(A) \subset C_b^1(\mathbb{R}^N)$ and that $C_b^1(\overline{\Omega})$ is of class $J_{1/2}$ between $C_b(\overline{\Omega})$ and D(A), that is

$$||Du||_{\infty} \le M_{\omega} ||u||_{\infty}^{\frac{1}{2}} ||(A - \omega)u||_{\infty}^{\frac{1}{2}}, \qquad f \in D(A),$$

for any $\omega > 0$.

In Section 12.3 we come back to the parabolic Cauchy-Neumann problem and, under somewhat heavier assumptions on the growth of the coefficients at infinity, we prove the pointwise estimate

$$|(DT(t)f)(x)|^p \le e^{\overline{\sigma}_p t} (T(t)|Df|^p)(x), \qquad t > 0, \ x \in \overline{\Omega}, \ f \in C^1_{\nu}(\overline{\Omega}), \tag{12.0.7}$$

where $\overline{\sigma}_p$ are suitable real constants. But, here, it must be $T(t)\mathbf{1} \equiv \mathbf{1}$, otherwise (12.0.7) fails for $f \equiv \mathbf{1}$. Thus, we suppose that $c \equiv 0$. Next, using (12.0.7) we prove the pointwise estimates

$$|(DT(t)f)(x)|^{p} \le \left(\frac{\overline{\sigma}_{2}}{2\kappa_{0}(1 - e^{-\overline{\sigma}_{2}t})}\right)^{\frac{p}{2}} (T(t)(|f|^{p}))(x), \qquad p \in [2, +\infty),$$
(12.0.8)

$$|(DT(t)f)(x)|^p \le \frac{c_p \overline{\sigma}_p t^{1-\frac{p}{2}}}{\kappa_0 (1 - e^{-\overline{\sigma}_p t})} (T(t)(|f|^p))(x), \qquad p \in (1, 2), \quad (12.0.9)$$

for any t > 0 and any $x \in \overline{\Omega}$, where $\overline{\sigma}_p/(1 - e^{-\overline{\sigma}_p t})$ is replaced with 1/t when $\overline{\sigma}_p = 0$.

As in Chapter 7, such estimates lead to some interesting results. First, they allow to prove that

$$||DT(t)f||_{\infty} \le \left(\frac{\overline{\sigma}_2}{2\kappa_0(1 - e^{-\overline{\sigma}_2 t})}\right)^{\frac{1}{2}} ||f||_{\infty}, \qquad t > 0, \ f \in C_b(\overline{\Omega}),$$

which improves (12.0.4), since it shows that the constant can be taken independent of T and, when $k_0 < 0$, it shows that the sup-norm of DT(t)fdecreases exponentially to 0 as t tends to $+\infty$.

Next, when $\overline{\sigma}_2 \leq 0$, (12.0.8) and (12.0.9) provide us with a Liouville type theorem. Namely, they imply that the constants are the only distributional solutions to Au = 0 which belong to D(A).

We also show that, under the same assumptions which allow us to prove the uniform gradient estimates,

$$|(DT(t)f)(x)|^p \le \left(\frac{2-p}{p}\right)^{1-\frac{p}{2}} \left(\frac{1}{p(p-1)\kappa_0 t} + 1\right)^{\frac{p}{2}} (T(t)(|f|^p))(x),$$

for any t > 0, any $x \in \Omega$, any $f \in C_b(\overline{\Omega})$ and any p > 1. Again, we can use this estimate to improve the gradient estimates proved in Section 12.1, when t approaches $+\infty$.

Finally, in Section 12.4, we briefly generalize some results of Chapter 8 to the case of the invariant measure associated with the semigroup $\{T(t)\}$ considered in this chapter. As in the case when $\Omega = \mathbb{R}^N$, whenever an invariant measure exists, the semigroup can be extended to a strongly continuous semigroup defined in the L^p space related to the measure μ , for any $p \in [1, +\infty)$.

Using the pointwise estimates of Section 12.3 we prove gradient estimates for the semigroup in $L^p(\Omega, \mu)$. Such estimates allow us to give a partial characterization of the domain $D(L_p)$ of the infinitesimal generator of $\{T(t)\}$ in $L^p(\Omega, \mu)$.

In the last part of the section we deal with the case when

$$\mathcal{A}\varphi = \frac{1}{2}\Delta\varphi - \langle DU, D\varphi \rangle,$$

on smooth functions φ , and U is a convex function. In such a situation a complete description of $D(L_2)$ is available. Moreover, the Poincaré inequality can be proved, provided a suitable dissipative condition is satisfied.

12.1 Construction of the semigroup and uniform gradient estimates

In this section we prove that for any $f \in C_b(\overline{\Omega})$, the Cauchy-Neumann problem (12.0.1) admits a unique classical solution (i.e., a function $u \in C([0, +\infty) \times \overline{\Omega}) \cap C^{1,2}((0, +\infty) \times \overline{\Omega})$ that solves pointwise the problem (12.0.1)), which is bounded in $[0,T] \times \overline{\Omega}$ for any T > 0. This will allow us to associate a semigroup of bounded linear operators with the problem (12.0.1), as explained in the introduction. To carry out our programme we assume the following set of hypotheses.

Hypotheses 12.1.1 (i) Ω is a convex unbounded open set of class $C^{2+\alpha}$ for some $\alpha \in (0,1)$;

(ii) the coefficients q_{ij} and b_j (i, j = 1, ..., N) and c belong to $C^{1+\alpha}_{loc}(\overline{\Omega})$ and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \qquad x \in \Omega, \quad \xi \in \mathbb{R}^N,$$
 (12.1.1)

for some function $\kappa : \overline{\Omega} \to \mathbb{R}$ such that $0 < \kappa_0 := \inf_{x \in \mathbb{R}^N} \kappa(x)$;

(iii) there exist some constants $q_0, \gamma > 0, c_0, k_0, \beta \in \mathbb{R}, \beta < 1/2$ such that

$$|Dq_{ij}(x)| \le q_0 \kappa(x), \quad x \in \Omega, \tag{12.1.2}$$

$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \le (-\beta c(x) + k_0) |\xi|^2, \quad x \in \Omega, \ \xi \in \mathbb{R}^N, \quad (12.1.3)$$

$$\sup_{x \in \Omega} c(x) = c_0, \tag{12.1.4}$$

$$|Dc(x)| \le \gamma (1 + |c(x)|), \quad x \in \Omega; \tag{12.1.5}$$

(iv) there exist a function $\varphi \in C^2(\overline{\Omega})$ and a constant $\lambda_0 > c_0$ such that

$$\lim_{|x| \to +\infty} \varphi(x) = +\infty, \quad \sup_{\Omega} (\mathcal{A}\varphi - \lambda_0 \varphi) < +\infty, \quad \frac{\partial \varphi}{\partial \nu}(x) \ge 0, \quad x \in \partial \Omega.$$
(12.1.6)

Remark 12.1.2 Suppose that $0 \in \Omega$. Then $\langle x, \nu(x) \rangle \geq 0$ for any $x \in \partial \Omega$. Therefore, one might look for a function φ satisfying (12.1.6) in the form $\varphi(x) = g(|x|^2)$, for any $x \in \Omega$, where $g: [0, +\infty) \to \mathbb{R}$ is a smooth and increasing function such that $\lim_{t \to +\infty} g(t) = +\infty$.

We now introduce a nested sequence $\{\Omega_n\}$ of bounded convex domains, with $C^{2+\alpha}$ -smooth boundaries, such that

$$\bigcup_{n>0} \Omega_n = \Omega, \qquad B(n) \cap \partial\Omega \subset \partial\Omega_n, \quad n>0.$$
 (12.1.7)

Moreover, we denote by $\{T_n(t)\}$ the strongly continuous analytic semigroup generated by the realization A_n of the operator \mathcal{A} with homogeneous Neumann boundary conditions in Ω_n , i.e., by the operator $A_n: D(A_n) \subset C_b(\overline{\Omega}) \to C_b(\overline{\Omega})$ defined by

$$A_n u = \mathcal{A}u, \qquad u \in D(A_n), \tag{12.1.8}$$

where

$$D(A_n) = \left\{ u \in \bigcap_{1 \le p < +\infty} W^{2,p}(\Omega_n) : \ \mathcal{A}u \in C(\overline{\Omega}_n), \ \frac{\partial u}{\partial \nu}(x) = 0, \ x \in \partial \Omega_n \right\},\tag{12.1.9}$$

(see Theorem C.3.6(v)). According to Proposition C.3.3, for any $f \in C(\overline{\Omega}_n)$, the function $u_n = T_n(\cdot)f$ belongs to $C([0, +\infty) \times \overline{\Omega}_n) \cap C^{1+\alpha/2, 2+\alpha}([\varepsilon, T] \times \overline{\Omega}_n)$ for any $0 < \varepsilon < T$ and it is the unique classical solution of the Cauchy problem

$$\begin{cases} D_t u_n(t,x) = \mathcal{A}u_n(t,x), & t > 0, x \in \Omega_n, \\ \frac{\partial u_n}{\partial \nu}(t,x) = 0, & t > 0, x \in \partial\Omega_n, \\ u_n(0,x) = f(x), & x \in \Omega_n. \end{cases}$$
(12.1.10)

Moreover,

$$||T_n(t)||_{L(C(\overline{\Omega}_n))} \le e^{c_0 t}, \qquad t > 0,$$
 (12.1.11)

and $T_n(t)f \geq 0$ for any $t \geq 0$ and any nonnegative function $f \in C(\overline{\Omega}_n)$. We can now state the main result of this section.

Theorem 12.1.3 Under Hypotheses 12.1.1 there exists a unique classical solution $u \in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\overline{\Omega})$ to the Cauchy-Neumann problem (12.0.1), which is bounded in $[0,T]\times\overline{\Omega}$ for any T>0. Moreover, $u_n=T_n(\cdot)f$ converges to u uniformly on compact sets of $(0,+\infty)\times\overline{\Omega}$. Further, there exists a semigroup $\{T(t)\}$ in $C_b(\overline{\Omega})$ such that $u(t,\cdot)=T(t)f$ for any $t\geq 0$ and

$$||T(t)f||_{\infty} \le e^{c_0 t} ||f||_{\infty}, \qquad f \in C_b(\overline{\Omega}).$$

Finally, for any $\omega > 0$, there exists a positive constant C_{ω} such that

$$||DT(t)f||_{\infty} \le C_{\omega} \frac{e^{\omega t}}{\sqrt{t}} ||f||_{\infty}, \qquad t > 0,$$
 (12.1.12)

for any $f \in C_b(\overline{\Omega})$.

Proof. Fix two open sets Ω' and Ω'' such that $\Omega' \subset \Omega''$ and $\operatorname{dist}(\Omega', \Omega \setminus \Omega'') > 0$, and let $n_0 \in \mathbb{N}$ be such that $\Omega'' \subset \Omega_n$ for any $n \geq n_0$. According to Theorem C.1.5, there exists a positive constant C, independent of n, such that

$$||u_n||_{C^{1+\alpha/2,2+\alpha}([\varepsilon,T]\times\Omega')} \le C||f||_{\infty},$$

for any $n \geq n_0$. This estimate implies that there exist a subsequence $T_{n_k}(\cdot)f$ and a continuous function $u:(0,+\infty)\times\overline{\Omega}\to\mathbb{R}$ such that

$$\lim_{k \to +\infty} (T_{n_k}(t)f)(x) = u(t,x), \quad t > 0, \ x \in \overline{\Omega}.$$

Moreover, $T_{n_k}(\cdot)f$ converges to u in $C^{1+\beta/2,2+\beta}([\varepsilon,T]\times\Omega')$ for any $\beta<\alpha$, any bounded set $\Omega'\subset\Omega$ and any $0<\varepsilon< T$. As it is immediately seen, u is a solution of the equation $D_tu-\mathcal{A}u=0$ in $(0,+\infty)\times\Omega$ and its normal derivative vanishes on $(0,+\infty)\times\partial\Omega$. Moreover, $u\in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\overline{\Omega})$. By (12.1.11) we get

$$|u(t,x)| \le e^{c_0 t} ||f||_{\infty}, \qquad t > 0, \ x \in \overline{\Omega}.$$
 (12.1.13)

Hence, u is bounded in $(0,T] \times \overline{\Omega}$ for any T > 0.

In contrast to what we showed in Section 2.2, when we dealt with the Cauchy problem in the whole of \mathbb{R}^N , in general, the sequence $\{T_n(t)f\}$ is not increasing for any fixed t > 0 and any nonnegative function $f \in C_b(\overline{\Omega})$. This lack of monotonicity is strictly related to the choice of Neumann boundary conditions.

To prove the continuity of u on $\{0\} \times \overline{\Omega}$, we observe that the arguments used in Step 2 in the proof of Theorem 2.2.1, based on a localization argument, cannot be immediately adapted to our situation, since they only allow us to prove the continuity of $T(\cdot)f$ on $\{0\} \times \Omega$. To overcome this difficulty, we apply a different technique which requires us to prove some gradient estimates for the function $T_n(t)f$, with constants being independent of n, and a maximum principle for bounded solutions to the problem (12.0.1). To prove such a gradient estimate we strongly take advantage of the convexity of Ω .

Proposition 12.1.4 Let Hypotheses 12.1.1 be satisfied and fix T > 0. Then, there exists a constant $C_T > 0$, independent of $n \in \mathbb{N}$, such that

$$||DT_n(t)f||_{\infty} \le \frac{C_T}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0, T],$$
 (12.1.14)

for any $f \in C_b(\overline{\Omega})$. Moreover

$$||DT_n(t)f||_{\infty} \le C_1 e^{c_0 \frac{t}{2}} ||f||_{C_b^1(\overline{\Omega})}, \qquad t \ge 0, \ x \in \Omega_n,$$
 (12.1.15)

for any $f \in C_b^1(\Omega_n)$ such that $\partial f/\partial \nu = 0$ on $\partial \Omega_n$ and some positive constant C_1 , independent of n. Here, c_0 is given by (12.1.4).

Proof. Let us begin the proof by checking the estimate (12.1.14). Without loss of generality, we can suppose that $c_0 \leq -1$. Indeed, once the assertion is proved in this situation, the general case will follow by replacing \mathcal{A} with the operator $\mathcal{A}' = \mathcal{A} - (c_0 + 1)I$. Moreover, we can suppose that $f \in D(A_n)$. Indeed, for a general $f \in C_b(\overline{\Omega})$ it suffices to split $T_n(t)f = T_n(t/2)T_n(t/2)f$ and to observe that $T_n(t/2)f \in D(A_n)$.

Let n > 0, set $u_n = T_n(\cdot)f$ and define

$$v_n(t,x) = |u_n(t,x)|^2 + at|Du_n(t,x)|^2, t \in (0,T], x \in \overline{\Omega}_n, (12.1.16)$$

where a>0 will be chosen later. Since $u_n\in C([0,+\infty);D(A_n))\cap C^1((0,+\infty);D_{A_n}(1+\alpha/2,\infty))$ and $D(A_n)\subset C_b^1(\overline{\Omega}_n)$ with a continuous embedding (see Theorems B.2.2 and C.3.6(v)), we easily deduce that $Du_n\in C([0,+\infty)\times\overline{\Omega}_n)\cap C^{1,1}((0,+\infty)\times\overline{\Omega}_n)$. Moreover, according to Theorem C.1.4(ii), Du_n is twice continuously differentiable with respect to the space variables in $(0,+\infty)\times\Omega_n$. It follows that $v_n\in C([0,+\infty)\times\overline{\Omega}_n)\cap C^{1,1}((0,+\infty)\times\overline{\Omega}_n)\cap C^{0,2}((0,+\infty)\times\Omega_n)$ and $v_n(0,\cdot)=0$.

We claim that, for any T > 0 arbitrarily fixed, a can be properly chosen such that the function v_n satisfies

$$\begin{cases} D_t v_n(t,x) - \mathcal{A}v_n(t,x) \le 0, & t \in (0,T], x \in \Omega_n, \\ \frac{\partial v_n}{\partial \nu}(t,x) \le 0, & t \in (0,T], x \in \partial \Omega_n. \end{cases}$$

Once the claim is proved, the classical maximum principle will yield $v_n \leq |f|^2$ in $(0,T) \times \Omega_n$, and (12.1.14) will easily follow, with $C_T = a^{-1/2}$. In the sequel, to simplify the notation, we drop out the dependence on n, when there is no damage of confusion.

As a first step, we prove that

$$\frac{\partial v}{\partial \nu}(t,x) \le 0, \qquad t \in (0,T], \ x \in \partial \Omega_n.$$
 (12.1.17)

This is the crucial point of the proof and it is the only part in which the convexity of Ω plays a crucial role. Since, by assumptions, $\partial u_n/\partial \nu$ identically vanishes on $\partial \Omega_n$ we have

$$\frac{\partial v}{\partial \nu}(t,x) = 2u \frac{\partial u}{\partial \nu}(t,x) + 2at \sum_{i,j=1}^{N} D_{ij}^{2} u(t,x) D_{i} u(t,x) \nu_{j}(x)$$

$$= 2at \sum_{i,j=1}^{N} D_{ij}^{2} u(t,x) D_{i} u(t,x) \nu_{j}(x), \qquad (12.1.18)$$

for any $t \in (0,T]$ and any $x \in \partial \Omega_n$. Now, we differentiate the relation

$$\frac{\partial u}{\partial \nu} = Du \cdot \nu = 0,$$

with respect to x_i , obtaining

$$\sum_{i=1}^{N} D_{ij}u(t,x)\nu_i(x) + \sum_{i=1}^{N} D_iu(t,x)D_j\nu_i(x) = 0.$$
 (12.1.19)

Multiplying by $D_j u(t, x)$ both the sides of (12.1.19) and summing with respect to j = 1, ..., N, we get

$$\sum_{i,j=1}^{N} D_{ij}u(t,x)D_{j}u(t,x)\nu_{i}(x) + \sum_{i=1}^{N} D_{i}u(t,x)D_{j}u(t,x)D_{j}\nu_{i}(x) = 0. \quad (12.1.20)$$

From (12.1.18) and (12.1.20), we get

$$\frac{\partial v}{\partial \nu}(t,x) = -2at \sum_{i,j=1}^{N} D_i \nu_j(x) D_i u(t,x) D_j u(t,x), \qquad t \in (0,T], \ x \in \partial \Omega_n.$$

Since Ω_n is convex, $D\nu(x)$ is a positive definite matrix for any $x \in \Omega_n$ and (12.1.17) follows.

Let us now show that if a is suitably small, then $D_t v - Av \leq 0$ in $(0, T] \times \Omega_n$. With some computations one can see that v satisfies the equation

$$D_t v - A v = a|Du|^2 - 2\sum_{i,j=1}^N q_{ij} D_i u D_j u + g_1 + g_2,$$

for any $t \in (0,T]$ and any $x \in \Omega_n$, where

$$g_1(t,x) = 2at \sum_{i,j=1}^{N} D_i b_j(x) D_i u(t,x) D_j u(t,x) + atc(x) |Du(t,x)|^2$$
$$+2atu(t,x) \sum_{i=1}^{N} D_i c(x) D_i u(t,x) + c(x) |u(t,x)|^2,$$

$$g_{2}(t,x) = -2at \sum_{i,j,h=1}^{N} q_{ij}(x) D_{ih} u(t,x) D_{jh} u(t,x)$$
$$+2at \sum_{i,j,h=1}^{N} D_{h} q_{ij}(x) D_{h} u(t,x) D_{ij} u(t,x).$$

Using (12.1.5) we get

$$2atu \sum_{i=1}^{N} D_i c D_i u \le 2at|u||Du||Dc|$$

$$\le 2a\gamma t (1-c)|u||Du|$$

$$\le a\frac{\gamma}{2c}t(1-c)u^2 + 2a\gamma \varepsilon t (1-c)|Du|^2,$$

for any $\varepsilon > 0$. Recalling that $c \leq -1$ and using (12.1.3), it follows that

$$g_1 \le \left(1 - a\frac{\gamma}{\varepsilon}t\right)cu^2 - at(2\beta - 1 + 2\gamma\varepsilon)c|Du|^2 + 2at(k_0 + \gamma\varepsilon)|Du|^2.$$

Since $\beta < 1/2$ we can choose $\varepsilon > 0$ such that $2\beta - 1 + 2\gamma \varepsilon < 0$. Therefore,

$$g_1 \le \left(1 - \frac{a\gamma}{\varepsilon}t\right)cu^2 + 2at(k_0 + \gamma\varepsilon)|Du|^2,$$
 (12.1.21)

in $(0,T] \times \Omega_n$.

Now, we consider the function g_2 . By (12.1.2) we have

$$\sum_{i,j,h=1}^{N} D_h q_{ij} D_h u D_{ij} u \leq q_0 \kappa |Du| \sum_{i,j=1}^{N} |D_{ij} u|
\leq q_0 \kappa N |Du| |D^2 u|
\leq \frac{q_0^2 N^2}{4\varepsilon'} \kappa |Du|^2 + \varepsilon' \kappa |D^2 u|^2, \qquad (12.1.22)$$

for any $\varepsilon' > 0$. Then, using (12.1.1), we get

$$g_2 \le a \frac{q_0^2 N^2}{2\varepsilon'} t\kappa |Du|^2 + 2at(\varepsilon' - 1)\kappa |D^2 u|^2,$$

in $(0,T] \times \Omega_n$. Choosing $\varepsilon' < 1$ we obtain

$$g_2 \le a \frac{q_0^2 N^2}{2\varepsilon'} t\kappa |Du|^2, \tag{12.1.23}$$

in $(0,T] \times \Omega_n$. The estimates (12.1.21) and (12.1.23) imply that

$$D_t v - \mathcal{A}v \le \left(1 - a\frac{\gamma}{\varepsilon}T\right)c u^2 + \left\{a + 2aT(k_0 + \gamma\varepsilon)^+ + \left(a\frac{q_0^2 N^2}{2\varepsilon'}T - 2\right)\kappa\right\}|Du|^2,$$

in $(0,T] \times \Omega_n$. It is now clear that there exists a sufficiently small value of a = a(T) > 0, independent of n, such that $D_t v - \mathcal{A}v \leq 0$ in $(0,T] \times \Omega_n$.

Now, we briefly prove the estimate (12.1.15). It suffices to show that it holds for any $f \in D(A_n)$ (see (12.1.9)). Indeed, $D(A_n)$ is dense in $C^1_{\nu}(\overline{\Omega}_n)$. This is a classical result, and it can be checked adapting the proof of the forthcoming Lemma 13.1.10.

So, let us check (12.1.15) for any $f \in D(A_n)$. For this purpose, we introduce the function $w_n : [0, +\infty) \times \overline{\Omega}$ defined by

$$w_n(t,x) = |u_n(t,x)|^2 + a |Du_n(t,x)|^2, t \ge 0, x \in \overline{\Omega}_n.$$

The above arguments show that $w_n \in C([0,T] \times \overline{\Omega}_n)$ for any T > 0. Moreover, it belongs to $C^{1,2}((0,T] \times \Omega)$. Straightforward computations show that the function $w = w_n$ satisfies

$$D_t w(t, x) - \mathcal{A}w(t, x) = -2 \sum_{i,j=1}^{N} q_{ij}(x) D_i u(t, x) D_j u(t, x) + h_1(t, x) + h_2(t, x),$$

for any $t \in (0,T]$ and any $x \in \Omega_n$, where $h = h_1 + h_2$ and h_1 and h_2 are defined as g_1 and g_2 with the coefficient $a \cdot t$ being everywhere replaced with a. Moreover, it satisfies $\partial w/\partial \nu = 0$ on $\partial \Omega_n$ and $w(0,\cdot) = f^2 + a|Df|^2$. Arguing as above, we can show that for a suitably small value of a > 0, independent of n and T, we have $h \leq 0$. Thus, the classical maximum principle yields

$$w(t,x) \le e^{c_0 t} \sup_{x \in \Omega_n} w(0,x) \le e^{c_0 t} (\|f\|_{\infty}^2 + a \|Df\|_{\infty}^2), \quad t > 0, \ x \in \Omega_n,$$

and (12.1.15) follows with
$$C_1 = a^{-1/2} \vee 1$$
.

Now, we prove a maximum principle for bounded classical solutions to the Neumann parabolic problem in Ω .

Theorem 12.1.5 If $z \in C([0,T] \times \overline{\Omega}) \cap C^{0,1}((0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$ is bounded and satisfies

$$\begin{cases} D_t z(t,x) - \mathcal{A} z(t,x) \le 0, & t \in (0,T], \ x \in \Omega, \\ \frac{\partial z}{\partial \nu}(t,x) \le 0, & t \in (0,T], \ x \in \partial \Omega, \\ z(0,x) \le 0, & x \in \Omega, \end{cases}$$

then

$$z(t,x) \leq 0, \hspace{1cm} t \in [0,T], \hspace{0.2cm} x \in \overline{\Omega}. \hspace{1cm} (12.1.24)$$

In particular, the Cauchy-Neumann problem (12.0.1) admits at most one classical solution which is bounded and continuous in $[0,T] \times \overline{\Omega}$. Moreover, it satisfies $||u(t,\cdot)||_{\infty} \leq e^{c_0t}||f||_{\infty}$, where c_0 is given by (12.1.4).

Proof. Let φ be as in (12.1.6). We can suppose that $\varphi \geq 0$ and $(\mathcal{A} - \lambda)\varphi \leq 0$; otherwise we replace φ with $\varphi + C$, for a suitable constant C > 0, and take a larger λ if needed.

Denote by $v:[0,T]\times\overline{\Omega}\to\mathbb{R}$ the function given by $v(t,x)=e^{-\lambda t}z(t,x)$ for any $t\in[0,T]$ and any $x\in\overline{\Omega}$, and consider the sequence $\{v_n\}$ defined by

$$v_n(t,x) = v(t,x) - \frac{1}{n}\varphi(x), \qquad t \in [0,T], \quad x \in \Omega.$$

We have

$$\begin{cases} D_t v_n(t,x) - (\mathcal{A} - \lambda) v_n(t,x) \leq 0, & t \in (0,T], \ x \in \Omega, \\ \frac{\partial v_n}{\partial \nu}(t,x) \leq 0, & t \in (0,T], \ x \in \partial \Omega, \\ v_n(0,x) \leq -\frac{1}{n} \inf_{\Omega} \varphi, & x \in \Omega. \end{cases}$$

For any $n \in \mathbb{N}$ the function v_n attains its maximum in $[0,T] \times \overline{\Omega}$ at some point (t_n, x_n) . Moreover, we have

$$\lim_{n \to +\infty} v_n(t_n, x_n) = \lim_{n \to +\infty} \sup_{[0,T] \times \overline{\Omega}} v_n = \sup_{[0,T] \times \overline{\Omega}} v.$$
 (12.1.25)

According to the strong maximum principle (see [60, Theorem 2.14]), we cannot have $t_n > 0$ and $x_n \in \partial \Omega$. If $(t_n, x_n) \in (0, T] \times \Omega$ we have

$$D_t v_n(t_n, x_n) \ge 0$$
, $\mathcal{A}v_n(t_n, x_n) - c(x_n)v_n(t_n, x_n) \le 0$,

and, therefore,

$$(\lambda - c(x_n)) \sup_{[0,T] \times \overline{\Omega}} v_n = (\lambda - c(x_n))v_n(t_n, x_n) \le (\lambda + D_t - \mathcal{A})v_n(t_n, x_n) \le 0,$$

which yields

$$\sup_{[0,T]\times\overline{\Omega}} v_n \le 0.$$

If instead $t_n = 0$ we have

$$v_n(t_n, x_n) \le -\frac{1}{n} \inf_{\Omega} \varphi.$$

Thus letting n go to $+\infty$ and taking (12.1.25) into account, we obtain (12.1.24).

Remark 12.1.6 It is worth noticing that in the proof of the quoted theorem, we never used the assumption that Ω is convex.

We can now conclude the proof of Theorem 12.1.3.

Proof of Theorem 12.1.3 (continued). To prove that u is a classical solution to the problem (12.0.1), it remains to show that u is continuous at t = 0.

The proof is similar to that of Theorem 2.2.1 and it strongly relies on the gradient estimate (12.1.14). Hence we limit ourselves to sketching it.

For any arbitrarily fixed $x_0 \in \overline{\Omega}$, we consider two open neighborhoods $U_1 \subset U_0$ of x_0 such that $\Omega_0 := U_0 \cap \Omega$ is sufficiently smooth. Then, we introduce a smooth function $\vartheta \in C^{\infty}(\overline{\Omega}_0)$ such that $\vartheta \equiv 0$ in a neighborhood of $\Omega \cap \partial U_0$, $\vartheta \equiv 1$ in $U_1 \cap \Omega$ and $\partial \vartheta / \partial \nu = 0$ in $U_0 \cap \partial \Omega$. Finally, for any $k \in \mathbb{N}$ such that

 $\Omega_0 \subset \Omega_{n_k}$, we introduce the function $v_{n_k} = \vartheta T_{n_k}(\cdot) f$. Such a function solves the equation

$$D_t v_{n_k}(t, x) - \mathcal{A}v_{n_k}(t, x) = \psi_{n_k}(t, x), \quad t > 0, \ x \in \Omega_0,$$

where

$$\psi_{n_k}(t,x) = -(T_{n_k}(t)f)(x)((\mathcal{A} - c)\vartheta)(x) - 2\sum_{i,j=1}^{N} q_{ij}(x)(D_i T_{n_k}(t)f)(x)D_j\vartheta(x),$$

for any t > 0 and any $x \in \Omega_0$, and, due to the choice of the function ϑ , its normal derivative vanishes on $\partial \Omega_0$.

From (12.1.14), we immediately see that

$$\|\psi_{n_k}(t,\cdot)\|_{\infty} \le \frac{C}{\sqrt{t}}, \quad t \in (0,T),$$
 (12.1.26)

for any $k \in \mathbb{N}$, and some positive constant C = C(T), independent of k. We now represent v_{n_k} through the variation-of-constants formula

$$v_{n_k}(t,\cdot) = T_0(t)(\vartheta f) + \int_0^t T_0(t-s)\psi_{n_k}(s,\cdot)ds, \qquad t > 0,$$

where $\{T_0(t)\}$ is the strongly continuous analytic semigroup generated by the realization of the operator \mathcal{A} , with homogeneous Neumann boundary conditions, in $C(\overline{\Omega}_0)$ (see Theorem C.3.6(v)). Using the estimates (12.1.11) and (12.1.26) it is easy to show that

$$\begin{aligned} |(T_{n_k}(t))f(x) - f(x_0)| &= |v_{n_k}(t, x) - f(x_0)| \\ &\leq |(T_0(t)(\vartheta f))(x) - f(x_0)| + C_1 \int_0^t s^{-\frac{1}{2}} e^{c_0(t-s)} ds, \end{aligned}$$

for any $x \in \Omega_1$ and some positive constant C_1 . Letting k go to $+\infty$, we easily see that u is continuous at $(0, x_0)$.

In conclusion, we have proved that $u \in C([0, +\infty) \times \overline{\Omega}) \cap C^{1+\alpha/2, 2+\alpha}_{loc}((0, +\infty) \times \overline{\Omega})$ and it is a solution of the Cauchy-Neumann problem (12.0.1). Moreover, since u is bounded in $[0, T] \times \Omega$ for any T > 0 (see (12.1.13)), by Theorem 12.1.5, u is the unique bounded classical solution to the problem (12.0.1).

Now, we prove that the whole sequence $T_n(\cdot)f$ converges to u in $C^{1,2}(F)$ for any compact set $F \subset (0, +\infty) \times \overline{\Omega}$. For this purpose, we observe that, applying all the above arguments to any subsequence $\{T_{n_h}(\cdot)f\}$, we can show that there exists a subsequence $\{T_{n_h_k}(\cdot)f\}$ converging to u in $C^{1,2}(F)$ for any compact set F. This, of course, implies that the whole sequence $\{T_n(\cdot)f\}$ converges to u in $C^{1,2}(F)$, and we are done.

Now, if we set (T(t)f)(x) := u(t,x) for any $t \in (0,+\infty)$, any $x \in \overline{\Omega}$ and any $f \in C_b(\overline{\Omega})$, where u is the solution to the problem (12.0.1) satisfying the initial

condition $u(0,\cdot) = f$, it is immediate to check that $\{T(t)\}$ is a semigroup of the linear operators in $C_b(\overline{\Omega})$.

Finally, letting n go to $+\infty$ in (12.1.14), we deduce that T(t)f satisfies (12.1.12) in (0,T). Then, we can use the semigroup law to extend it to all the positive t (we refer the reader to the proof of Theorem 6.1.7 for the details).

Our purpose now consists in showing that the more f is smooth, the more the behaviour near t=0 of the function DT(t)f is good. More precisely, we want to show that

$$||DT(t)f||_{\infty} \le Ce^{c_0\frac{t}{2}}||f||_{C_h^1(\overline{\Omega})}, \qquad t > 0,$$
 (12.1.27)

for any $f \in C^1_{\nu}(\overline{\Omega})$. Unfortunately, it is not immediate to deduce (12.1.27) from (12.1.15), since this latter estimate has been proved for the functions f whose normal derivative vanish on $\partial\Omega_n$. To overcome such a difficulty, we use an approximation argument: given $f \in C^1_{\nu}(\overline{\Omega})$ we approximate it by a sequence $\{f_n\} \subset C^1_b(\overline{\Omega})$ for which (12.1.15) holds for any $n \in \mathbb{N}$. Then, to conclude that (12.1.27) holds, we need to show that $DT(t)f_n$ converges to DT(t)f as n tends to $+\infty$ for any t > 0, at least pointwise. This convergence result is proved in the next proposition which shows us that $\{T(t)\}$ satisfies the same continuity properties as the semigroup defined in Chapter 2.

Proposition 12.1.7 If $\{f_n\} \subset C_b(\overline{\Omega})$ is a bounded sequence converging pointwise in Ω to a function $f \in C_b(\overline{\Omega})$, then $T(\cdot)f_n$ converges, as n tends to $+\infty$, to $T(\cdot)f$ in $C^{1,2}([\varepsilon,T]\times\overline{\Omega}')$ for any $0<\varepsilon< T$ and any bounded set $\Omega'\subset\Omega$.

Further, if $\{f_n\}$ converges to f uniformly on compact subsets of $\overline{\Omega}$, then $T(\cdot)f_n$ converges to $T(\cdot)f$ uniformly in $[0,T] \times \overline{\Omega}'$ for any T > 0 and any bounded set $\Omega' \subset \Omega$.

Finally T(t)f can be represented in the form

$$(T(t)f)(x) = \int_{\Omega} f(y)p(t, x; dy), \quad t > 0, \ x \in \overline{\Omega}, \tag{12.1.28}$$

where p(t, x; dy) is a positive finite Borel measure on Ω . In the case when $c \equiv 0$, p(t, x; dy) is a probability measure for any t > 0 and any $x \in \overline{\Omega}$.

Proof. Throughout the proof, without loss of generality, we assume that $f \equiv 0$. To prove the first part of the proposition it suffices to adapt the arguments in the proof of Theorem 12.1.3. So, we limit ourselves to sketching the proof, pointing out the main differences.

Let $\{f_n\}$ be a bounded sequence in $C_b(\overline{\Omega})$ converging pointwise to zero in Ω , and set $u_n = T(\cdot)f_n$. Applying the Schauder estimate (C.1.19) and the maximum principle to the sequence $\{u_n\}$, we can extract a subsequence $\{u_{n_k}\}$ converging in $C^{1,2}([\varepsilon,T]\times\overline{\Omega}')$ to a function $u\in C^{1,2}((0,+\infty)\times\overline{\Omega})$,

for any $0 < \varepsilon < T$ and any bounded set $\Omega' \subset \Omega$. Moreover, the function u is a bounded solution of the equation $D_t u - \mathcal{A} u = 0$ in $(0, +\infty) \times \Omega$, and its normal derivative vanishes on the boundary of Ω . To conclude that $u \equiv 0$, by Theorem 12.1.5 it suffices to show that u is continuous up to t = 0 and that $u(0,\cdot) = 0$. This can be obtained by a localization argument. So, we fix $x_0 \in \overline{\Omega}$. Moreover, we consider (as in the proof of Theorem 12.1.3) two open neighborhoods $U_1 \subset U_0$ of x_0 such that $\Omega_0 := U_0 \cap \Omega$ is sufficiently smooth. Further, we introduce a smooth function $\vartheta \in C^{\infty}(\overline{\Omega}_0)$ such that $\vartheta \equiv 0$ in a neighborhood of $\Omega \cap \partial U_0$, $\vartheta \equiv 1$ in $U_1 \cap \Omega$ and $\partial \vartheta / \partial \nu = 0$ in $U_0 \cap \partial \Omega$. Finally, we set $v_{n_k} = \vartheta u_{n_k}$. Then, we can write

$$v_{n_k}(t,\cdot) = T_0(t)(\vartheta f_{n_k}) + \int_0^t T_0(t-s)\psi_{n_k}(s,\cdot)ds, \qquad t > 0,$$

where $\{T_0(t)\}$ is the semigroup generated by the realization of \mathcal{A} in $C(\overline{\Omega}_0)$ with homogeneous Neumann boundary conditions (see Theorem C.3.6(v)) and

$$\psi_{n_k} = -u_{n_k} (\mathcal{A} - c)\vartheta - 2 \sum_{i,j=1}^{N} q_{ij} D_i u_{n_k} D_j \vartheta.$$

Using the gradient estimate (12.1.12) and the boundedness of $\{f_{n_k}\}$ it follows that

$$|v_{n_k}(t,x)| \le |(T_0(t)\vartheta f_{n_k})(x)| + C\sqrt{t}, \quad t \in [0,T], \ x \in \overline{\Omega}_0, \quad k \in \mathbb{N}, \ (12.1.29)$$

where C > 0 is a constant independent of $k \in \mathbb{N}$. We now claim that $T_0(t)(\vartheta f_{n_k})$ vanishes as k tends to 0 for any $t \in [0,T]$. Once the claim is proved, letting k go to $+\infty$ in (12.1.29), we will get $|u(t,x)| \leq C\sqrt{t}$ and, consequently, u will turn out to be continuous at $(0, x_0)$, where it will vanish. To prove the claim, we take advantage of the L^p -theory. We observe that, for any $p \in (N, +\infty)$, the semigroup $\{T_0(t)\}$ extends to an analytic semigroup in $L^p(\Omega_0)$ (see Theorem C.3.6(iii)), and, by the Sobolev embedding theorem (see [2, Theorem 5.4]), the domain of its generator A_p is continuously embedded in $C(\overline{\Omega}_0)$. Since ϑf_{n_k} converges to zero in $L^p(\Omega_0)$ and $||T_0(t)(\vartheta f_{n_k})||_{D(A_p)} \leq Ct^{-1}||f_{n_k}||_{L^p}$ for some positive constant C, independent of k, we deduce that $T_0(t)(\vartheta f_{n_k})$ converges to 0 in $D(A_p)$ and, consequently, uniformly in $\overline{\Omega}$, as k tends to $+\infty$. So far, we have proved that the subsequence u_{n_k} converges to zero in $C^{1,2}([\varepsilon,T]\times\overline{\Omega}')$ for any $0<\varepsilon< T$ and any bounded $\Omega' \subset \Omega$. As in the proof of Theorem 12.1.3 we can now prove that the whole sequence $\{u_n\}$ converges to zero in $C^{1,2}([\varepsilon,T]\times\overline{\Omega}')$ for any $0<\varepsilon< T$ and any bounded set $\Omega' \subset \Omega$, as stated.

Suppose now that $\{f_n\}$ converges to zero uniformly on compact subsets of $\overline{\Omega}$. By (12.1.29), with v_n instead of v_{n_k} , we get

$$|u_n(t,x)| \le ||T_0(t)(\vartheta f_n)||_{\infty} + C\sqrt{t} \le e^{c_0 T} ||\vartheta f_n||_{\infty} + C\sqrt{t}, \quad t \in [0,T], \ x \in \overline{\Omega}_1,$$

for any $n \in \mathbb{N}$, where C > 0 does not depend on $n \in \mathbb{N}$. Therefore, for any $\varepsilon > 0$, we have

$$||u_n||_{C([0,T]\times\overline{\Omega}_1)} \le e^{\varepsilon c_0} ||\vartheta f_n||_{\infty} + C\sqrt{\varepsilon} + ||u_n||_{C([\varepsilon,T]\times\overline{\Omega}_1)}.$$

Since u_n and ϑf_n vanish as n tends to $+\infty$, respectively in $[\varepsilon, T] \times \overline{\Omega}$ and in $\overline{\Omega}$, we get

$$\limsup_{n \to \infty} \|u_n\|_{C([0,T] \times \overline{\Omega}_1)} \le C\sqrt{\varepsilon},$$

which, of course, implies that u_n converges to zero uniformly in $[0, T] \times \overline{\Omega}_1$. Since Ω_1 is arbitrary, the conclusion follows.

We can now prove (12.1.28). By the Riesz representation theorem, for any $x \in \overline{\Omega}$, there exists a positive finite Borel measure p(t, x; dy) in Ω such that (12.1.28) holds for any $f \in C_0(\Omega)$. Now, if $f \in C_b(\overline{\Omega})$, we consider a bounded sequence $\{f_n\} \subset C_0(\Omega)$ which converges to f uniformly on the compact sets of Ω . Writing (12.1.28) with f_n instead of f and letting n go to $+\infty$, we obtain the desired formula, by dominated convergence.

To conclude the proof, we observe that, if $c \equiv 0$, then $T(t) \mathbb{1} \equiv \mathbb{1}$ and, consequently, $\int_{\Omega} p(t, x; dy) = 1$ for any t > 0 and any $x \in \overline{\Omega}$. This implies that p(t, x; dy) are all probability measures.

We can now prove (12.1.27).

Proposition 12.1.8 Under Hypotheses 12.1.1, there exists a positive constant C such that (12.1.27) holds for any $f \in C^1_{\nu}(\overline{\Omega})$.

Proof. To prove (12.1.27) we fix $f \in C^1_{\nu}(\overline{\Omega})$ and, for any $k \in \mathbb{N}$, we consider a function $\vartheta_k \in C^1_{\nu}(\overline{\Omega})$ such that

$$0 \le \vartheta_k \le 1$$
, $\|D\vartheta_k\|_{\infty} \le L$, $\vartheta_k \equiv 1$ in Ω_k , $\frac{\partial \vartheta_k}{\partial \nu} = 0$ on $\partial \Omega$,

where L>0 is a constant independent of $k\in\mathbb{N}$ and Ω_k is as in (12.1.7). If n_0 is such that $\operatorname{supp}(\vartheta_k)\subset\overline{\Omega}_{n_0}$, then, for any $n>n_0$, the function $f_k=\vartheta_k f$ belongs to $C^1_{\nu}(\overline{\Omega}_n)$. Thus, $T_n(t)f_k$ satisfies the estimate (12.1.15). Since, according to Theorem 12.1.3, $T_n(t)f_k$ tends to $T(t)f_k$ in $C^1(\overline{\Omega}_n)$ for any $n\in\mathbb{N}$, letting n go to $+\infty$, we get

$$|(DT(t)f_k)(x)| \le C_1 e^{\frac{c_0}{2}t} ||f_k||_{C_b^1(\overline{\Omega})} \le C_1 e^{\frac{c_0}{2}t} (1+L) ||f||_{C_b^1(\overline{\Omega})},$$

for any t > 0 and $x \in \Omega$. Now, using Proposition 12.1.7, we obtain that $DT(t)f_n(x)$ converges to (DT(t)f)(x) for any t > 0 and any $x \in \Omega$. Therefore, (12.1.27) follows with $C = C_1(1 + L)$.

12.2 Some consequences of the uniform gradient estimates

In this section we show some useful consequences of the uniform gradient estimates proved in Section 12.1. To begin with, we observe that, using (12.1.28), it is easy to see that the semigroup can be extended to the space $B_b(\Omega)$ of all the bounded Borel functions $f: \Omega \to \mathbb{R}$. We still denote by $\{T(t)\}$ such a semigroup. Moreover, as the following proposition shows, the so obtained semigroup is irreducible (i.e., $(T(t)\chi_U)(x) > 0$ for any t > 0, any $x \in \Omega$ and any open set $U \subset \Omega$) and it is strong Feller (i.e., it maps $B_b(\Omega)$ into $C_b(\overline{\Omega})$).

Proposition 12.2.1 The semigroup $\{T(t)\}$ is irreducible and strong Feller.

Proof. To begin with, let us prove that $\{T(t)\}$ is irreducible. Fix an open set $U \subset \Omega$. We are going to prove that $(T(t)\chi_U)(x) > 0$ for any t > 0 and any $x \in \overline{\Omega}$. For this purpose, we approximate χ_U by a sequence of nonnegative continuous functions f_n converging monotonically to χ_U and such that f_1 does not identically vanish in Ω . Then, according to the representation formula (12.1.28), $(T(t)f_n)(x)$ converges monotonically to $(T(t)\chi_U)(x)$ as n tends to $+\infty$, for any t>0 and any $x\in\overline{\Omega}$. To conclude that $T(\cdot)\chi_U>0$, it suffices to show that $T(t)f_1>0$ for any t>0. By contradiction, let us suppose that there exist $t_0>0$ and $x_0\in\Omega$ such that $(T(t_0)f_1)(x_0)=0$. The nonnegativity of the semigroup $\{T(t)\}$ immediately implies that (t_0,x_0) is a minimum point of the function $T(\cdot)f_1$. Therefore, according to Proposition C.2.3(iii), $T(\cdot)f_1\equiv 0$ for any $t\in(0,t_0]$ and any $x\in\Omega'$, where Ω' is any bounded open subset of Ω , which contains the point x_0 . In particular, taking t=0, we get $f_1\equiv 0$ in Ω' and hence $f_1\equiv 0$ in Ω , which, of course, cannot be the case.

Similarly, if $(T(t_0)f_1)(x_0) = 0$ at some point $(t_0, x_0) \in (0, +\infty) \times \partial\Omega$, we can find out a bounded open set Ω' of class $C^{2+\alpha}$ such that $x_0 \in \partial\Omega$ and $\Omega' \cap (x_0 + B(r)) \subset \Omega \cap (x_0 + B(r))$ for some r > 0. Then, according to Proposition D.0.5, Ω' satisfies the interior sphere condition, so that $\partial T(t_0)f/\partial\nu > 0$ at $x = x_0$ (see Proposition C.2.3(iii)), which, of course, cannot be the case.

To prove that the semigroup $\{T(t)\}$ is strong Feller, we adapt the technique of [37, Theorem 3.2], which strongly relies on the gradient estimate (12.0.4). For this purpose, we fix $f \in B_b(\Omega)$ and introduce a bounded sequence $\{f_n\} \in C_b(\overline{\Omega})$ converging pointwise to f. By (12.0.4), for any t > 0 and any $x_0 \in \overline{\Omega}$, we can determine a constant C = C(t), such that

$$||DT(t)f_n||_{\infty} \le C, \qquad n \in \mathbb{N},$$

that, due to the convexity of Ω , implies that $T(t)f_n$ is a Lipschitz continuous function, uniformly with respect to $n \in \mathbb{N}$. Letting n go to $+\infty$, we deduce that T(t)f is Lipschitz continuous as well. This completes the proof.

Remark 12.2.2 In fact, in the proof of Proposition 12.2.1 we have shown that if $f \in C_b(\overline{\Omega})$ is a nonnegative function which does not identically vanish in Ω , then T(t)f > 0 in $\overline{\Omega}$ for any t > 0.

We now show that, if $f \in C^1_{\nu}(\overline{\Omega})$, then the gradient of $T(\cdot)f$ is continuous at t = 0.

Proposition 12.2.3 If $f \in C^1_{\nu}(\overline{\Omega})$ then

$$\lim_{t\to 0^+} \|T(t)f - f\|_{C^1_b(\overline{\Omega}')} = 0, \tag{12.2.1}$$

for any open bounded set $\Omega' \subset \Omega$, i.e., $DT(\cdot)f$ is continuous up to $\{0\} \times \overline{\Omega}$.

Proof. The proof is similar to that of Theorem 12.1.3. We fix $x_0 \in \overline{\Omega}$ and introduce two open neighborhoods U_0 , U_1 of x_0 with $U_1 \subset U_0$. Moreover, we set $\Omega_0 = U_0 \cap \Omega$, $\Omega_1 = U_1 \cap \Omega$ and we assume that Ω_0 is sufficiently smooth. Finally, we denote by $\vartheta \in C^{\infty}(\overline{\Omega}_0)$ any function vanishing in a neighborhood of $\Omega \cap \partial U_0$ and such that $\vartheta \equiv 1$ in Ω_1 , $\partial \vartheta / \partial \nu \equiv 0$ on $\partial \Omega \cap U_0$.

The same arguments as in the proof of Theorem 12.1.3 show that

$$v_n(t,\cdot) := \vartheta T(\cdot)f = T_0(t)(\vartheta f) + \int_0^t T_0(t-s)\psi(s,\cdot)ds,$$

where $\{T_0(t)\}\$ is the semigroup generated by the realization of the operator \mathcal{A} in $C(\overline{\Omega}_0)$ with homogeneous Neumann boundary conditions, and

$$\psi = -(T(\cdot)f)(\mathcal{A} - c)\vartheta - 2\sum_{i,j=1}^{N} q_{ij}(D_iT(\cdot)f)D_j\vartheta.$$

According to Proposition 12.1.8, for any T>0 there exists a constant $C_1>0$ such that

$$\|\psi(t)\|_{\infty} \le C_1 \|f\|_{C_{-1}^1(\overline{\Omega})}, \qquad t \in (0, T].$$
 (12.2.2)

Moreover, since $\{u \in C^1(\overline{\Omega}_0) : \partial u/\partial \nu = 0 \text{ on } \partial \Omega_0\}$ belongs to the class $J_{1/2}$ between $C_b(\overline{\Omega}_0)$ and the domain of $\{T_0(t)\}$ (see Theorem C.3.6(v)), we easily deduce that

$$||DT_0(t)g||_{\infty} \le \frac{C_2}{\sqrt{t}} ||g||_{C(\overline{\Omega}_0)} \le \frac{C_2}{\sqrt{t}} ||g||_{\infty}, \qquad t \in (0, T],$$
 (12.2.3)

for any $g \in C_b(\overline{\Omega})$, where $C_2 > 0$ is a suitable constant. Now, from (12.2.2) and (12.2.3) we get

$$|Dv(t,x) - Df(x_0)| = \left| (DT_0(t)(\vartheta f))(x) + \int_0^t (DT_0(t-s)\psi(s,\cdot))(x)ds - Df(x_0) \right|$$

$$\leq |DT_0(t)(\vartheta f)(x) - Df(x_0)| + C_1 ||f||_{C_b^1(\overline{\Omega})} \int_0^t \frac{C_2}{\sqrt{t-s}} ds,$$

for any $t \in (0,T)$ and any $x, x_0 \in U_0$. Since $DT_0(\cdot)(\vartheta f)$ is continuous at $(0,x_0)$, then Dv is continuous at $(0,x_0)$ as well. Finally, since $v = T(\cdot)f$ in Ω_1 , we conclude that DT(t)f is continuous at $(0,x_0)$, and (12.2.1) easily follows.

We now use the previous gradient estimates to characterize the weak generator \widehat{A} of the semigroup $\{T(t)\}$ and to solve the elliptic problem

$$\begin{cases} \lambda u - \mathcal{A}u, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial\Omega, \end{cases}$$
 (12.2.4)

when $f \in C_b(\overline{\Omega})$.

As in the case of the Dirichlet problem in the whole \mathbb{R}^N , the family of bounded operators $\{R(\lambda)\}$ defined by

$$(R(\lambda)f)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x)dt, \qquad x \in \overline{\Omega},$$

for any $f \in C_b(\overline{\Omega})$ and any $\lambda > c_0$, is the resolvent family associated with some closed operator $\widehat{A}: D(\widehat{A}) \subset C_b(\overline{\Omega}) \to C_b(\overline{\Omega})$. So from now on, we write $R(\lambda, \widehat{A})$ instead of $R(\lambda)$. Adapting the proofs in Section 2.3, it can be shown that the weak generator $(\widehat{A}, D(\widehat{A}))$ is given by

$$\begin{split} D(\widehat{A}) &= \bigg\{ f \in C_b(\overline{\Omega}) : \sup_{t \in (0,1)} \frac{\|T(t)f - f\|_{\infty}}{t} < +\infty \ \text{ and } \ \exists g \in C_b(\overline{\Omega}) \ \text{ s.t.} \\ \lim_{t \to 0^+} \frac{(T(t)f)(x) - f(x)}{t} &= g(x), \ \text{ for any } x \in \overline{\Omega} \bigg\}, \end{split}$$

$$\widehat{A}f(x) = \lim_{t \to 0^+} \frac{(T(t)f)(x) - f(x)}{t}, \quad x \in \overline{\Omega}, \qquad f \in D(\widehat{A}).$$

Our aim now is to show that $D(\widehat{A}) = D(\mathcal{A})$ and $\widehat{A} = \mathcal{A}$, where

$$D(\mathcal{A}) = \left\{ u \in C_b(\overline{\Omega}) \cap \bigcap_{1 \le p < +\infty} W^{2,p}(\Omega \cap B(R)) \text{ for any } R > 0 : \\ \mathcal{A}u \in C_b(\overline{\Omega}), \ \frac{\partial u}{\partial \nu}(x) = 0, \text{ for any } x \in \partial \Omega \right\}.$$
 (12.2.5)

For this purpose, we first prove the following maximum principle for the elliptic problem (12.2.4).

Proposition 12.2.4 Let $u \in C_b(\overline{\Omega}) \cap W^{2,p}(\Omega \cap B(R))$, for any R > 0 and any $p \in [1, +\infty)$, be such that $Au \in C_b(\overline{\Omega})$ and

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) \leq 0, & x \in \Omega, \\ \frac{\partial u}{\partial \nu}(x) \leq 0, & x \in \partial \Omega, \end{cases}$$

for some $\lambda \geq \lambda_0$, where λ_0 is as in (12.1.6). Then $u \leq 0$. In particular, for any $\lambda \geq \lambda_0$ there exists at most one solution $u \in D(A)$ to the problem (12.2.4).

Proof. As in the proof of Theorem 12.1.5, we may assume that $\varphi \geq 0$ and $\mathcal{A}\varphi - \lambda_0 \varphi \leq 0$. We introduce the sequence

$$u_n(x) = u(x) - \frac{1}{n}\varphi(x), \quad x \in \Omega,$$

and we note that

$$\begin{cases} \lambda u_n(x) - \mathcal{A}u_n(x) \le 0, & x \in \Omega, \\ \frac{\partial u_n}{\partial \nu}(x) \le 0, & x \in \partial \Omega. \end{cases}$$
 (12.2.6)

If we prove that $u_n \leq 0$ for any $n \in \mathbb{N}$, the conclusion will follow letting n go to $+\infty$. By contradiction, suppose that u_n has a maximum point at $x_n \in \overline{\Omega}$. Adapting the second proof of Lemma 4.1.2 to this situation, we can show that if $x_n \in \Omega$, then $((\mathcal{A} - c)u_n)(x_n) \leq 0$. Thus, by (12.2.6) we have $(\lambda - c(x_n))u_n(x_n) \leq 0$ and, hence, $u_n(x_n) \leq 0$.

Suppose now that $x_n \in \partial\Omega$ is such that $u_n(x_n) > 0$ and $u_n(x) < u_n(x_n)$ for any $x \in \Omega$. These assumptions will lead us to a contradiction since they will imply that $(\partial u_n/\partial\nu)(x_n) > 0$.

Let $y + B(r) \subset \Omega$ be such that $(y + \overline{B}(r)) \cap \partial\Omega = \{x_n\}$ and assume that $u_n > 0$ in y + B(r). Moreover, fix $\alpha > 0$ such that the function $z : \mathbb{R}^N \to \mathbb{R}$, defined by $z(x) = e^{-\alpha|x-y|^2} - e^{-\alpha r^2}$ for any $x \in \mathbb{R}^N$, satisfies Az > 0 in $D = \{x \in \mathbb{R}^N : r/2 < |x-y| < r\}$. Then, set $w_n = u_n + \varepsilon z$, where $\varepsilon > 0$ is such that

$$w_n(x) < u_n(x_n), \qquad x \in y + \partial B(r/2). \tag{12.2.7}$$

A straightforward computation shows that

$$Aw_n(x) = Au_n(x) + \epsilon Az(x) > \lambda u_n(x) > 0, \qquad x \in D.$$
 (12.2.8)

Let \hat{x}_n be the maximum point of w_n in \overline{D} . \hat{x}_n belongs to ∂D (this is a classical result that can be obtained arguing as in the proof of Lemma 4.1.2) and, due to (12.2.7), it belongs to $y + \partial B(r)$. Since $z \equiv 0$ on $y + \partial B(r)$ and x_n is a maximum point for the function u_n in Ω , it follows that $w_n(\hat{x}_n) = w_n(x_n)$. Therefore, x_n is a maximum point of w_n in \overline{D} as well. As a consequence,

$$\frac{\partial w_n}{\partial \nu}(x_n) = \frac{\partial u_n}{\partial \nu}(x_n) + \varepsilon \frac{\partial z}{\partial \nu}(x_n) \ge 0.$$

Since $\partial z/\partial \nu(x_n) < 0$, it follows that $(\partial u_n/\partial \nu)(x_n) > 0$.

Proposition 12.2.5 For any $f \in C_b(\overline{\Omega})$ and any $\lambda > c_0$ (see (12.1.4)), the function $u = R(\lambda, \widehat{A})f$ belongs to D(A) and solves the problem (12.2.4). Moreover, $D(\widehat{A}) = D(A)$ and $\widehat{A}v = Av$, for any $v \in D(A)$. Finally, $D(A) \subset C_{\nu}^1(\overline{\Omega})$

with a continuous embedding and, for any $\omega > c_0$, there exists a positive constant M_{ω} such that

$$||Du||_{\infty} \le M_{\omega} ||u||_{\infty}^{\frac{1}{2}} ||(\mathcal{A} - \omega) u||_{\infty}^{\frac{1}{2}},$$
 (12.2.9)

for any $u \in D(A)$.

Proof. Let $f \in C_b(\overline{\Omega})$ and, for any $\lambda > c_0$, set $u = R(\lambda, \widehat{A})f$ and $u_n = R(\lambda, A_n)f$ $(n \in \mathbb{N})$, where A_n is the generator of the semigroup $\{T_n(t)\}$ (see (12.1.8), (12.1.9)). Of course,

$$u_n(x) = \int_0^{+\infty} e^{-\lambda t} (T_n(t)f)(x)dt, \quad x \in \overline{\Omega}_n.$$
 (12.2.10)

Therefore, for any $\lambda > c_0$ and any $n \in \mathbb{N}$, it holds that

$$||u_n||_{\infty} \le \frac{1}{\lambda - c_0} ||f||_{\infty}, \qquad ||\mathcal{A}u_n||_{\infty} \le \left(1 + \frac{\lambda}{\lambda - c_0}\right) ||f||_{\infty}.$$

Since $T_n(\cdot)f$ tends to $T(\cdot)f$ locally uniformly in $(0, +\infty) \times \Omega$ (see Theorem 12.1.3), from (12.2.10) taking (12.1.11) into account, we immediately see that

$$\lim_{n \to +\infty} u_n = u,$$

pointwise in $\overline{\Omega}$ and in $L^p(\Omega_k)$, for any $k \in \mathbb{N}$. Furthermore, by the L^p -estimates in Theorem C.1.2, we have

$$||u_m - u_n||_{W^{2,p}(\Omega_k)} \le c(p,k)||u_m - u_n||_{L^p(\Omega_{k+1})}, \quad m, n > k,$$

for any $p \in (1, +\infty)$ and any $k \in \mathbb{N}$, where c(p, k) > 0 is a constant independent of m, n. Therefore, u_n converges to u in $W^{2,p}(\Omega_k)$, for any $k \in \mathbb{N}$ and any $p \in [1, +\infty)$. Hence, $u \in W^{2,p}(\Omega \cap B(R))$, for any R > 0 and any $p \in [1, +\infty)$. The Sobolev embedding theorems (see [2, Theorem 5.4]) imply that u_n converges to u also in $C^1(\overline{\Omega}_k)$ for any $k \in \mathbb{N}$, and, therefore, $\partial u/\partial \nu = 0$ on $\partial \Omega$. Finally, since $\lambda u_n - \mathcal{A} u_n = f$ for any $n \in \mathbb{N}$, letting n go to $+\infty$, it follows that $\lambda u - \mathcal{A} u = f$ in Ω . Therefore, u belongs to $D(\mathcal{A})$ and solves the problem (12.2.4). Since $R(\lambda, \widehat{A})$ is surjective from $C_b(\overline{\Omega})$ onto $D(\widehat{A})$, we deduce that $D(\widehat{A}) \subset D(\mathcal{A})$ and $\widehat{A} = \mathcal{A}$ in $D(\widehat{A})$.

Conversely, let $u \in D(A)$ and set $f = \lambda u - Au \in C_b(\overline{\Omega})$, where $\lambda = 2 \max(\lambda_0, c_0)$, and c_0 , λ_0 are given, respectively, by (12.1.4) and (12.1.6). By the above results, the function $v = R(\lambda, \widehat{A})f$ is a bounded solution of the problem (12.2.4) as well. Proposition 12.2.4 yields that $u \equiv v$, implying, in particular, that $u \in D(\widehat{A})$.

To complete the proof, let us prove (12.2.9). For this purpose, we fix $u \in D(\mathcal{A})$, $\omega > c_0$ and $\lambda > 0$. Then, we set

$$u(x) = (R(\lambda + \omega, \widehat{A})f)(x) = \int_0^{+\infty} e^{-(\lambda + \omega)t} (T(t)f)(x) dt, \qquad x \in \overline{\Omega},$$
(12.2.11)

where $f = (\lambda + \omega) u - Au$. By virtue of the estimate (12.1.12), we may differentiate under the integral sign in (12.2.11) obtaining

$$Du(x) = \int_0^{+\infty} e^{-(\lambda + \omega)t} (DT(t)f)(x) dt, \quad x \in \overline{\Omega}$$

and, then,

$$|Du(x)| \le C_{\omega} ||f||_{\infty} \int_{0}^{+\infty} \frac{e^{-\lambda t}}{\sqrt{t}} dt = \frac{C_{\omega} \sqrt{\pi}}{\sqrt{\lambda}} ||f||_{\infty}, \qquad x \in \Omega,$$

where $C_{\omega} > 0$ is a constant. Therefore,

$$||Du||_{\infty} \le C_{\omega} \sqrt{\pi} \left(\sqrt{\lambda} ||u||_{\infty} + \frac{||(\mathcal{A} - \omega) u||_{\infty}}{\sqrt{\lambda}} \right),$$

and, taking the minimum over $\lambda > 0$, we get (12.2.9).

12.3 Pointwise gradient estimates and their consequences

In this section we prove some pointwise estimates for the gradient of T(t)f, similar to those proved in Chapter 7. We assume that $c \equiv 0$. Indeed, as it is immediately seen, some of such estimates fail if $T(t)\mathbb{1} \not\equiv \mathbb{1}$ (take $f \equiv \mathbb{1}$ in (12.3.2)) which is the case if $c \not\equiv 0$.

Proposition 12.3.1 Assume that the coefficient c of the operator A identically vanishes in Ω and that Hypotheses 12.1.1 are satisfied with the condition (12.1.2) being replaced by the following one:

$$\sum_{i,j=1}^{N} \left(\sum_{h=1}^{N} D_h q_{ij}(x) \xi_h \right)^2 \le q_0 \kappa(x) |\xi|^2, \qquad x \in \Omega, \ \xi \in \mathbb{R}^N.$$
 (12.3.1)

Then, for any p > 1 and any $f \in C^1_{\nu}(\overline{\Omega})$, we have

$$|(DT(t)f)(x)|^p \le e^{\sigma_p t} (T(t)(|Df|^p))(x), \qquad t \ge 0, \ x \in \overline{\Omega}, \tag{12.3.2}$$

where $\sigma_p = pk_0 + \frac{p}{4}q_0$, if $p \ge 2$ and $\sigma_p = pk_0 + \frac{p}{4(p-1)}q_0$, if 1 .

Further, if $q_{ij} \equiv \delta_{ij}$ for any i, j = 1, ..., N, then (12.3.2) can be extended also to the case when p = 1, setting $\sigma_p = k_0$ for any $p \in [1, +\infty)$.

Proof. The proof is similar to and even simpler than those of Theorems 7.1.2 and 7.3.1. Hence, we just sketch it in the case when $p \in (1,2]$ and the diffusion coefficients q_{ij} depend on x. For this purpose, we fix $\delta > 0$ and introduce the function $w : [0, +\infty) \times \overline{\Omega} \to \mathbb{R}$ defined by

$$w(t,x) = (|Du(t,x)|^2 + \delta)^{\frac{p}{2}}, \qquad t > 0, \quad x \in \overline{\Omega},$$

where $u = T(\cdot)f$. By Propositions 12.1.8, 12.2.3 and Theorems 12.1.3, C.1.4, $w \in C_b([0,+\infty) \times \overline{\Omega}) \cap C^{0,1}((0,+\infty) \times \overline{\Omega}) \cap C^{1,2}((0,+\infty) \times \Omega)$ and it satisfies the equation

$$D_t w(t, x) - \mathcal{A}w(t, x) = f_1(t, x) + f_2(t, x), \quad t > 0, \ x \in \Omega,$$

where

$$f_{1} = p \left(|Du|^{2} + \delta \right)^{\frac{p}{2} - 1} \times \left(\sum_{i,j,h=1}^{N} D_{h}q_{ij}D_{h}uD_{ij}u + \sum_{i,j=1}^{N} D_{i}b_{j}D_{i}uD_{j}u - \text{Tr}(QD^{2}uD^{2}u) \right),$$
(12.3.3)

$$f_2 = p(2-p) (|Du|^2 + \delta)^{\frac{p}{2}-2} \langle QD^2 uDu, D^2 uDu \rangle.$$

Taking (12.1.3) (with $c \equiv 0$) and (12.3.1) into account, we obtain that

$$f_1 \le p \left(|Du|^2 + \delta \right)^{\frac{p-2}{2}} \times \left(\varepsilon \kappa |D^2 u|^2 + \frac{q_0}{4\varepsilon} |Du|^2 + k_0 |Du|^2 - \text{Tr}(QD^2 uD^2 u) \right),$$

for any $\varepsilon > 0$. On the other hand, applying (7.1.4) with $|\alpha| = |\beta| = 1$, we obtain

$$\langle QD^2uDu, D^2uDu \rangle \leq |Du|^2 \text{Tr}(QD^2uD^2u).$$

Then, choosing $\varepsilon = p - 1$, we get

$$f_1 + f_2 \le \sigma_p (|Du|^2 + \delta)^{\frac{p-2}{2}} |Du|^2 = \sigma_p w - \delta \sigma_p (|Du|^2 + \delta)^{\frac{p-2}{2}},$$

which yields

$$D_t w - \mathcal{A}w \le \sigma_p w - (\sigma_p \wedge 0)\delta^{\frac{p}{2}}.$$

Therefore, by a comparison argument we see that

$$w(t,\cdot) \le e^{\sigma_p t} T(t) \left((|Df|^2 + \delta)^{\frac{p}{2}} \right), \qquad t > 0,$$

if $\sigma_p \geq 0$, and

$$w(t,\cdot) \le e^{\sigma_p t} T(t) \left((|Df|^2 + \delta)^{\frac{p}{2}} - \delta^{\frac{p}{2}} \right) + \delta^{\frac{p}{2}}, \qquad t > 0,$$

if $\sigma_p < 0$. Then, letting δ go to 0^+ , we get the assertion from Proposition 12.1.7.

As it is immediately seen, the assumption (12.3.1) is rather restrictive, since it does not allow us to consider diffusion coefficients of polynomial type. As the following proposition shows, this trouble may be overcome assuming a dissipativity condition of negative type on the drift term, in order to balance the growth at infinity of the derivatives of the q_{ij} 's (i, j = 1, ..., N).

Proposition 12.3.2 Assume that the coefficient c of the operator A identically vanishes in Ω . Moreover, assume that Hypotheses 12.1.1 but (12.1.2) are satisfied with k_0 being replaced by a negative function $k: \Omega \to \mathbb{R}$, and that there exist two constants C > 0 and $\alpha \in (0,1]$ such that

$$|Dq_{ij}(x)| \le C(\kappa(x))^{\alpha}, \qquad x \in \Omega, \ i, j = 1, \dots, N$$

and

$$\widetilde{\sigma}_p := \sup_{x \in \Omega} \left(\frac{C^2 N^2}{4p_0(p_0 - 1)\kappa_0^{1-\alpha}} (\kappa(x))^{\alpha} + k(x) \right) < +\infty.$$
 (12.3.4)

for some $p_0 \in (1,2)$. Then, for any $p \geq p_0$,

$$|(DT(t)f)(x)|^p \le e^{\widetilde{\sigma}_p t}(T(t)(|Df|^p))(x), \qquad t \ge 0, \ x \in \overline{\Omega}. \tag{12.3.5}$$

Proof. It is similar to that of Proposition 12.3.1. The only difference is in the estimate of term $\sum_{i,j,h=1}^{N} D_h q_{ij} D_i u D_{jh} u$ which now should be estimated as follows:

$$\left| \sum_{i,j,h=1}^{N} D_h q_{ij} D_h u D_{ij} u \right| \leq C \kappa^{\alpha} \left(\frac{N^2}{4\varepsilon} |Du|^2 + \varepsilon |D^2 u|^2 \right),$$

for any $\varepsilon > 0$ and, then, suitably choosing ε in order to make the coefficient in front of $|D^2u|^2$ in the expression of $f_1 + f_2$ vanish.

We can now use the gradient estimates in Propositions 12.3.1 and 12.3.2 to prove a second type of pointwise gradient estimates.

Proposition 12.3.3 Under the assumptions of Proposition 12.3.1, for any $f \in C_b(\overline{\Omega})$ we have

$$|(DT(t)f)(x)|^p \le \left(\frac{\sigma_2}{2\kappa_0(1 - e^{-\sigma_2 t})}\right)^{\frac{p}{2}} (T(t)(|f|^p))(x), \qquad t > 0, \ x \in \overline{\Omega},$$
(12.3.6)

for any $p \geq 2$, and

$$|(DT(t)f)(x)|^p \le \frac{c_p \sigma_p t^{1-\frac{p}{2}}}{\kappa_0 (1 - e^{-\sigma_p t})} (T(t)(|f|^p))(x), \qquad t > 0, \ x \in \overline{\Omega}, \ (12.3.7)$$

for any $1 , where <math>c_p = (p(p-1))^{-p/2}$ and σ_p is given by Proposition 12.3.1. When $\sigma_p = 0$ in (12.3.6) and (12.3.7), we replace $\sigma_p/(1 - e^{-\sigma_p t})$ with 1/t.

The same results hold true under the assumptions of Proposition 12.3.2, provided we take $p \geq p_0$ and replace everywhere σ_p with $\widetilde{\sigma}_p$.

Proof. Without loss of generality, we limit ourselves to proving the assertion under the assumptions of Proposition 12.3.1.

We first prove (12.3.6). For any $\delta > 0$ and any $n \in \mathbb{N}$, we introduce the function $\Phi_n : (0,t) \to C_b(\overline{\Omega})$ defined by

$$\Phi_n(s) = T_n(s) \Big((|T_n(t-s)f|^2 + \delta)^{\frac{p}{2}} \Big),$$

where $\{T_n(t)\}$ is, as usual, the semigroup introduced before Theorem 12.1.3. Moreover, we set $g_n(s) = (|T_n(t-s)f|^2 + \delta)^{p/2}$ for any $s \in (0,t)$. As it is immediately seen, $g_n(s) \in D(A_n)$ for any $s \in (0,t)$ (see (12.1.9)). This implies that $A_n\Phi_n = T_n(\cdot)A_ng_n$. Hence, taking the positivity of the semigroup $\{T_n(t)\}$ into account, we get

$$\Phi'_{n}(s) = T_{n}(s) \left(A_{n}g_{n}(s) - p(g(s))^{1-\frac{2}{p}}T_{n}(t-s)f A_{n}T_{n}(t-s)f \right)
= T_{n}(s) \left(p(g_{n}(s))^{1-\frac{4}{p}} \left((p-1)|T_{n}(t-s)f|^{2} + \delta \right) \right)
\times \langle QDT_{n}(t-s)f, DT_{n}(t-s)f \rangle
\geq p(p-1)\kappa_{0}
\times T_{n}(s) \left((g_{n}(s))^{1-\frac{4}{p}}|DT_{n}(t-s)f|^{2} \left(|T_{n}(t-s)f|^{2} + \frac{\delta}{p-1} \right) \right)
\geq p(p-1)\kappa_{0}T_{n}(s) \left((g_{n}(s))^{1-\frac{2}{p}}|DT_{n}(t-s)f|^{2} \right).$$
(12.3.8)

Now, we fix $x \in \overline{\Omega}$ and $\varepsilon \in (0, t/2)$. Integrating (12.3.8) with respect to s from ε to $t - \varepsilon$, we get

$$\left(T_n(t-\varepsilon)\left((|T_n(\varepsilon)f|^2+\delta)^{\frac{p}{2}}\right)\right)(x)$$

$$\geq p(p-1)\kappa_0 \int_{\varepsilon}^{t-\varepsilon} \left(T_n(s)\left((g_n(s))^{1-\frac{2}{p}}|DT_n(t-s)f|^2\right)\right)(x)ds.$$

Since $\{T_n(t)\}$ is a strongly continuous semigroup, then, for any $x \in \overline{\Omega}$, the function $(T_n(t-\varepsilon)\{(|T_n(\varepsilon)f|^2+\delta)^{\frac{p}{2}}\})(x)$ tends to $(T_n(t)\{(|f|^2+\delta)^{\frac{p}{2}}\})(x)$

as ε goes to 0. Therefore, from the monotone convergence theorem, we get

$$\left(T_n(t)\left((|f|^2 + \delta)^{\frac{p}{2}}\right)\right)(x)
\geq p(p-1)\kappa_0 \int_0^t \left(T_n(s)\left((g_n(s))^{1-\frac{2}{p}}|DT_n(t-s)f|^2\right)\right)(x)ds, \quad (12.3.9)$$

for any $x \in \Omega$. Now, arguing as in the proof of Proposition 12.3.1, it easily follows that for any $n \in \mathbb{N}$ the semigroup $\{T_n(t)\}$ satisfies

$$|(DT_n(t)f)(x)|^p \le e^{\sigma_p t} (T_n(t)(|Df|^p))(x), \qquad t \ge 0, \ x \in \overline{\Omega},$$

where σ_p is the same as in (12.3.2). Hence, using the Hölder inequality, we can write

$$|DT_n(t)f|^p \le e^{\sigma_p s} T_n(s) \left(|DT_n(t-s)f|^p \right)$$

$$= e^{\sigma_p s} T_n(s) \left((g_n(s))^{\frac{p}{2} - 1} |DT_n(t-s)f|^p \left(g_n(s) \right)^{1 - \frac{p}{2}} \right)$$

$$\le e^{\sigma_p s} \left\{ T_n(s) \left((g_n(s))^{1 - \frac{2}{p}} |DT_n(t-s)f|^2 \right) \right\}^{\frac{p}{2}} \left\{ T_n(s) g_n(s) \right\}^{1 - \frac{p}{2}}.$$

Then, using the Young and Jensen inequalities and recalling that $(a+b)^{p/2} \le a^{p/2} + b^{p/2}$ for any a, b > 0, we get

$$|DT_n(t)f|^p \le e^{\sigma_p s} \left\{ \frac{p}{2} \varepsilon^{\frac{2}{p}} T_n(s) \left((g_n(s))^{1-\frac{2}{p}} |DT_n(t-s)f|^2 \right) + \left(1 - \frac{p}{2} \right) \varepsilon^{\frac{2}{p-2}} \left(T_n(t) (|f|^p) + \delta^{\frac{p}{2}} \right) \right\}.$$

Integrating from 0 to t and taking (12.3.9) into account gives

$$\frac{1 - e^{-\sigma_p t}}{\sigma_p} |DT_n(t)f|^p \le \frac{\varepsilon^{\frac{2}{p}}}{2(p-1)\kappa_0} T_n(t) \left((|f|^2 + \delta)^{\frac{p}{2}} \right) + t \left(1 - \frac{p}{2} \right) \varepsilon^{\frac{2}{p-2}} \left(T_n(t) (|f|^p) + \delta^{\frac{p}{2}} \right),$$

where, if $\sigma_p = 0$, we replace the term $(1 - e^{-\sigma_p t})/\sigma_p$ with t. Letting δ go to 0 and, then, minimizing with respect to $\varepsilon \in (0, +\infty)$, we deduce (12.3.6) with $\{T(t)\}$ being replaced with the semigroup $\{T_n(t)\}$. To get (12.3.6) it is now sufficient to take the limit as n goes to $+\infty$, recalling that $T_n(\cdot)f$ converges to $T(\cdot)f$ in $C^{1,2}(D)$ for any compact set $D \subset (0, +\infty) \times \overline{\Omega}$ (see the last part of the proof of Theorem 12.1.3).

To prove (12.3.7) it suffices to apply the Jensen inequality and (12.3.6) with p=2. Indeed, since p(t,x;dy) are probability measures for any $t>0, x\in\overline{\Omega}$,

it follows that $(T(t)(|f|^2))^{p/2} \leq T(t)(|f|^p)$. Therefore,

$$\begin{split} |(DT(t)f)(x)|^p &= \left(|(DT(t)f)(x)|^2 \right)^{\frac{p}{2}} \\ &\leq \left(\frac{\sigma_2}{2\kappa_0 (1 - e^{-\sigma_2 t})} (T(t)(f^2))(x) \right)^{\frac{p}{2}} \\ &\leq \left(\frac{\sigma_2}{2\kappa_0 (1 - e^{-\sigma_2 t})} \right)^{\frac{p}{2}} (T(t)(|f|^p))(x) \end{split}$$

and (12.3.7) follows.

Corollary 12.3.4 For any $f \in C_b(\overline{\Omega})$ it holds that

$$||DT(t)f||_{\infty} \le \left(\frac{\overline{\sigma}_2}{2\kappa_0(1 - e^{-\overline{\sigma}_2 t})}\right)^{\frac{1}{2}} ||f||_{\infty}, \qquad t > 0,$$

if $\overline{\sigma}_2 \neq 0$, and

$$||DT(t)f||_{\infty} \le \frac{1}{\sqrt{2\kappa_0 t}} ||f||_{\infty}, \qquad t > 0,$$

if $\overline{\sigma}_2 = 0$. Here, $\overline{\sigma}_2 = \sigma_2$ under the assumptions of Proposition 12.3.1, whereas $\overline{\sigma}_2 = \widetilde{\sigma}_2$ under the assumptions of Proposition 12.3.2.

Corollary 12.3.4 gives uniform gradient estimates sharper than those of Theorem 12.1.3. Indeed, it shows that we can take $\omega=0$ in (12.1.12). Moreover, under the assumptions of Proposition 12.3.1 it gives an explicit description of the constant C, in terms of q_0 and k_0 (see (12.1.3) and (12.3.1)). Finally, in the case when $\sigma_2 < 0$, (12.3.4) shows that the sup-norm of the gradient of T(t)f vanishes exponentially as t tends to $+\infty$.

Another interesting consequence of the gradient estimate in Proposition 12.3.3 is shown in the following proposition which provides us a Liouville type theorem. We omit the proof since it can be obtained arguing as in the proof of Theorem 7.2.5.

Proposition 12.3.5 Suppose that Proposition 12.3.3 holds with $\overline{\sigma}_2 \leq 0$ (where $\overline{\sigma}_2$ is as in Corollary 12.3.4). If $f \in D(A)$ is such that Af = 0, then f is constant.

To conclude this section we show that under the same assumptions as in Section 12.1, we can prove estimates similar to (12.3.2) even if a bit weaker. In any case, they can allow to improve the uniform gradient estimate (12.1.12). In fact, they show that as t tends to $+\infty$ the sup-norm of DT(t)f stays bounded.

Proposition 12.3.6 Under Hypotheses 12.1.1, there exists a constant $C_p > 0$ such that

$$|(DT(t)f)(x)|^{\frac{p}{2}} \le C_p T(t) \left((f^2 + |Df|^2)^{\frac{p}{2}} \right) (x), \qquad t > 0, \quad x \in \overline{\Omega},$$
(12.3.10)

for any $f \in C^1_{\nu}(\overline{\Omega})$.

Proof. Since the proof is similar to that of Proposition 12.3.1, we limit ourselves to sketching it, pointing out the main differences. In the case when $p \in (1,2]$ and $f \in C_c^{2+\alpha}(\overline{\Omega})$, we introduce, for any $\delta > 0$, the function w defined by

$$w(t,x) = (a|u(t,x)|^2 + |Du(t,x)|^2 + \delta)^{\frac{p}{2}}, \qquad t > 0, \quad x \in \Omega.$$

Here a>0 is a real parameter to be fixed later on. By virtue of Propositions 12.1.8 and 12.2.3, the function w is bounded and continuous in $[0, +\infty) \times \overline{\Omega}$. Moreover, by Theorem C.1.4, w is differentiable once with respect to the time variable and twice with respect to the space variables in $(0, +\infty) \times \Omega$, and it solves the differential equation $D_t w - \mathcal{A}w = f_1 + f_2$, where now

$$f_{1} = p w^{1-\frac{2}{p}} \left(\sum_{i,j,h=1}^{N} D_{h} q_{ij} D_{h} u D_{ij} u + \sum_{i,j=1}^{N} D_{i} b_{j} D_{i} u D_{j} u - a \langle Q D u, D u \rangle - \text{Tr}(Q D^{2} u D^{2} u) \right),$$
(12.3.11)

$$f_2 = p(2-p)w^{1-\frac{4}{p}}\langle Q(auDu + D^2uDu), auDu + D^2uDu\rangle.$$

Moreover, $\partial w/\partial \nu \leq 0$ in $(0, +\infty) \times \partial \Omega$. Using the Cauchy-Schwarz inequality twice, first for the inner product induced by the matrix Q and, then, for the Euclidean one, we get

$$\langle Q(auDu + D^2uDu), auDu + D^2uDu \rangle$$

$$\leq \left[\left(\text{Tr}(QD^2uD^2u) \right)^{\frac{1}{2}} |Du| + a|u| \left(\langle QDu, Du \rangle \right)^{\frac{1}{2}} \right]^2$$

$$\leq (a|u|^2 + |Du|^2) \left\{ \text{Tr}(QD^2uD^2u) + a\langle QDu, Du \rangle \right\}.$$

Hence,

$$f_2 \le p(2-p)(au^2 + |Du|^2)^{\frac{p}{2}-1} \left\{ \text{Tr}(QD^2uD^2u) + a\langle QDu, Du \rangle \right\}.$$

Therefore, we get

$$D_{t}w - \mathcal{A}w = p w^{1-\frac{2}{p}} \left\{ \sum_{i,j,h=1}^{N} D_{h}q_{ij}D_{h}uD_{ij}u + \sum_{i,j=1}^{N} D_{i}b_{j}D_{i}uD_{j}u - (p-1)\left[a\langle QDu, Du\rangle + \text{Tr}(QD^{2}uD^{2}u)\right] \right\}.$$
(12.3.12)

The first term in the right-hand side of (12.3.12) is immediately estimated by (12.1.22) (where we take $\varepsilon' = (p-1)$), whereas, to estimate the remaining ones, it suffices to use the dissipativity condition (12.1.3) and the ellipticity condition (12.1.1) as in the proof of Proposition 12.1.4. So, finally, we get

$$D_t w - \mathcal{A}w \le p w^{1-\frac{2}{p}} \left(K_p + \frac{k_0}{\kappa_0} - a(p-1) \right) \kappa |Du|^2,$$

where K_p is a suitable positive constant. It is now clear that we can choose a sufficiently large such that $D_t w - \mathcal{A}w \leq 0$ in $(0, +\infty) \times \Omega$. Hence, from the maximum principle in Theorem 12.1.5 we get

$$(a|u(t,x)|^2 + |Du(t,x)|^2 + \delta)^{\frac{p}{2}} \le (T(t)((af^2 + |Df|^2 + \delta)^{\frac{p}{2}}))(x),$$

for any t > 0 and any $x \in \Omega$. Letting δ go to 0, the assertion follows.

Remark 12.3.7 Repeating verbatim the proof of Proposition 12.3.3, it can be easily checked that, for any $n \in \mathbb{N}$, the semigroup $\{T_n(t)\}$, introduced in the proof of Theorem 12.1.3, satisfies (12.3.10) with the same constant C_p .

Theorem 12.3.8 Under Hypotheses 12.1.1, for any $p \in (1, +\infty)$ and any $f \in C_b(\overline{\Omega})$, we have

$$|(DT(t)f)(x)|^p \le \left(\frac{1}{p(p-1)\kappa_0 t} + 1\right)^{\frac{p}{2}} (T(t)(|f|^p))(x),$$
 (12.3.13)

for any t > 0. In particular, the function $DT(\cdot)f$ is bounded in $(a, +\infty) \times \Omega$ for any a > 0.

Proof. It can be obtained arguing as in the proof of Proposition 12.3.3. The only difference is that now, using (12.3.10) with T(t) replaced with $T_n(t)$, we obtain

$$|DT_n(t)f|^p \le C_p \left\{ \frac{p}{2} \varepsilon^{\frac{2}{p}} T_n(s) \left[(g_n(s))^{1-\frac{2}{p}} (|T_n(t-s)f|^2 + |DT_n(t-s)f|^2) \right] + \left(1 - \frac{p}{2} \right) \varepsilon^{\frac{2}{p-2}} \left(T_n(t)|f|^p + \delta^{\frac{p}{2}} \right) \right\}.$$

Using the Jensen inequality as in the last part of the proof of Proposition 12.3.3, we obtain

$$T_n(s) \left((g_n(s))^{1-\frac{2}{p}} |T_n(t-s)f|^2 \right) \le T_n(s) g_n(s)$$

$$\le T_n(s) \left(T_n(t-s)(|f|^p) + \delta^{\frac{p}{2}} \right)$$

$$= T_n(t)(|f|^p) + \delta^{\frac{p}{2}}.$$

Hence,

$$|DT_n(t)f|^p \le C_p \left\{ \frac{p}{2} \varepsilon^{\frac{2}{p}} T_n(s) \left[(g_n(s))^{1-\frac{2}{p}} |DT_n(t-s)f|^2 \right] + \left(1 - \frac{p}{2} \right) \varepsilon^{\frac{2}{p-2}} \left(T_n(t) (|f|^p) + \delta^{\frac{p}{2}} \right) + \frac{p}{2} \varepsilon^{\frac{2}{p}} (T_n(t) (|f|^p) + \delta^{\frac{p}{2}}) \right\},$$

and now (12.3.13) follows arguing as in the proof of Proposition 12.3.3.

12.4 The invariant measure of the semigroup

In this section we briefly generalize some results of Chapter 8 to the case of the invariant measure associated with the semigroup $\{T(t)\}$ considered in this chapter. To simplify the notation, throughout this section we simply write L^p_μ and $W^{k,p}_\mu$ $(k \in \mathbb{N}, p \in [1, +\infty))$ instead of $L^p(\Omega, \mu)$ and $W^{k,p}(\Omega, \mu)$. Moreover, we denote by $\|\cdot\|_p$ the usual norm of L^p_μ .

In view of the extension of the gradient estimates in Section 12.3 to this setting, we assume that c identically vanishes in $\overline{\Omega}$.

To begin with let us prove the following theorem.

Theorem 12.4.1 There exists at most one invariant measure of $\{T(t)\}$. Such an invariant measure is absolutely continuous with respect to the Lebesgue measure and its density ϱ is a positive function which belongs to $W_{\text{loc}}^{1,p}(\Omega)$ for any $p \in (1, +\infty)$. In particular, ϱ is continuous in Ω (not necessarily bounded).

Proof. We split the proof into three steps.

Step 1. Here, we assume that μ is an invariant measure of $\{T(t)\}$. Repeating with slight changes the same arguments as in the proof of Theorem 8.2.2, we can easily show that μ admits a density $\varrho \in W^{1,p}_{loc}(\Omega)$ (see also [21, Section 2] for more details). Moreover, using the Harnack inequality (see [142, Corollary 5.3]), we can show that

$$\sup_{x \in K} \varrho(x) \le C \inf_{x \in K} \varrho(x), \tag{12.4.1}$$

for any compact set K and some positive constant C=C(K). Of course, the estimate (12.4.1) and the continuity of ϱ in Ω imply that ϱ is everywhere positive.

To conclude this step of the proof, let us show that μ is absolutely continuous with respect to the Lebesgue measure. For this purpose, let $\{\Omega_n\}$ be a sequence of bounded sets such that $\Omega_n \subset \Omega_{n+1}$ for any $n \in \mathbb{N}$ and $\bigcup_{n \in \mathbb{N}} \Omega_n = \Omega$. By the above results,

$$\mu(\Omega_n) = \int_{\Omega_n} \varrho dx, \qquad n \in \mathbb{N}.$$

The Fatou lemma now implies that

$$1 = \mu(\Omega) = \int_{\Omega} \varrho dx.$$

Therefore, ϱ is everywhere positive in Ω and it belongs to $L^1(\Omega)$, i.e., μ is absolutely continuous with respect to the Lebesgue measure on the σ -algebra of the Borel set of Ω .

Step 2. We now prove that there exists at most a unique invariant measure of $\{T(t)\}$. For this purpose, we begin by observing that the semigroup $\{T(t)\}$ is ergodic, i.e.,

$$\lim_{t \to +\infty} \frac{1}{t} \int_0^t T(s) f ds = \int_{\Omega} f d\mu, \qquad f \in L^2_{\mu}. \tag{12.4.2}$$

To prove the formula (12.4.2), one can argue as in the proof of Proposition 8.1.11 with minor changes. Indeed, Proposition 8.1.13 holds also in this situation as it can be easily seen repeating step by step the same proof (note in particular that, using the formula (12.1.28), one can show that, if $\{f_n\}$ is a bounded sequence of bounded and continuous functions converging to a function $f \in B_b(\mathbb{R}^N)$ pointwise, then $T(\cdot)f_n$ converges to $T(\cdot)f$ pointwise in $[0, +\infty) \times \overline{\Omega}$ as n tends to $+\infty$). We need to modify the proof of Proposition 8.1.11 only to show that $\chi_A \in C = \{f \in L^2_\mu : T(t)f = f \mu$ -a.e. for any $t > 0\}$ if and only if $\mu(A) = 0$ or $\mu(A) = 1$. For this purpose, suppose that $\chi_A \in C$ and $\mu(A) > 0$. Then, $T(1)\chi_A = \chi_A$ μ -almost everywhere in Ω . Therefore, $T(1)\chi_A > 0$ μ -almost everywhere in A. Since the semigroup is strong Feller (see Proposition 12.2.1), then $T(1)\chi_A \in C_b(\overline{\Omega})$. Moreover, according to Remark 12.2.2, the function $T(2)\chi_A$ is everywhere positive in Ω . Therefore, the equality $T(2)\chi_A = \chi_A$ can hold μ -almost everywhere in Ω if and only if $\mu(\Omega \setminus A) = 0$ that is if and only if $\mu(A) = 1$.

Step 3. We can now conclude the proof. Suppose that μ_1 and μ_2 are two different invariant measures of $\{T(t)\}$. Then, according to the Chacon-Ornstein theorem (see [124, Chapter 3, Section 8]), for any Borel set A we can determine two sets M_1 and M_2 , with $\mu_j(M_j) = 1$ (j = 1, 2) such that

$$\lim_{n \to +\infty} \frac{1}{n} \int_0^n (T(s)\chi_A)(x)ds = \int_{\Omega} \chi_A d\mu_j = \mu_j(A), \qquad x \in M_j, \quad j = 1, 2.$$
(12.4.3)

Let us observe that $M_1 \cap M_2 \neq \emptyset$. Indeed, since the densities ϱ_1 and ϱ_2 of μ_1 and μ_2 (with respect to the Lebesgue measure) are positive in Ω , then, for any compact set $K \subset \Omega$, the restriction μ_j^K of the measure μ_j to K (j=1,2) is equivalent to the Lebesgue measure (restricted to the Borel sets of K). Fix a compact set K such that $\mu_j(M_j \cap K) > 0$ (j=1,2) and suppose, by contradiction, that $M_1 \cap M_2 = \emptyset$. Since $\mu_1(\Omega \setminus M_1) = 0$, then $\mu_1^K(M_1 \cap K) = \mu_1^K(K)$. Moreover,

$$\mu_1^K((M_1 \cup M_2) \cap K) = \mu_1^K(M_1 \cap K) + \mu_1^K(M_2 \cap K) \le \mu_1^K(K) = \mu_1^K(M_1 \cap K).$$

Hence, $\mu_1^K(M_2 \cap K) = 0$. Since, μ_1^K and μ_2^K are equivalent to the Lebesgue measure, then $\mu_2^K(M_2 \cap K) = 0$, which, of course, cannot be the case.

Now, taking $x \in M_1 \cap M_2$, from (12.4.3), we easily deduce that $\mu_1(A) = \mu_2(A)$ and the arbitrariness of A allows us to conclude that $\mu_1 = \mu_2$.

The following theorem gives a precise description of the behaviour of the function T(t)f as t tends to $+\infty$, for any $f \in L^2_{\mu}$. For the proof, we refer the reader to [42, Theorems 3.4.2 & 4.2.1].

Theorem 12.4.2 For any $f \in L^p_\mu$ we have

$$\lim_{t \to +\infty} ||T(t)f - \overline{f}||_p = 0,$$

where $\overline{f} = \int_{\Omega} f d\mu$.

As in the case when $\Omega = \mathbb{R}^N$ the main existence result of an invariant measure of $\{T(t)\}$ is the Khas'minskii theorem.

Theorem 12.4.3 (Khas'minskii) Suppose that there exists a function $\varphi \in C^2(\overline{\Omega})$ such that

$$\varphi \geq 0,$$
 $\lim_{|x| \to +\infty} \mathcal{A}\varphi(x) = -\infty,$ $\frac{\partial \varphi}{\partial \nu} = 0$ on $\partial \Omega$.

Then, the semigroup $\{T(t)\}$ admits an invariant measure μ .

Proof. It can be obtained arguing as in the proof of Theorem 8.1.20, adapting to our situation the proof of Theorems 8.1.18 and 8.1.19.

From now on, we always assume that the semigroup $\{T(t)\}$ admits an invariant measure. In such a situation, as the following proposition shows, the semigroup can be extended to a strongly continuous semigroup of contractions defined in $L^p(\Omega)$ for any $p \in [1, +\infty)$. We will still denote by $\{T(t)\}$ such a semigroup.

Proposition 12.4.4 Let μ be the invariant measure of $\{T(t)\}$. Then, the semigroup can be extended to a strongly continuous semigroup of contractions in L^p_{μ} for any $p \in [1, +\infty)$. Moreover, the set $D(\mathcal{A})$ (see (12.2.5)) is a core of the infinitesimal generator L_p of the semigroup in L^p_{μ} .

Proof. It can be obtained arguing as in the proofs of Propositions 8.1.8 and 8.1.9.

We now use the gradient estimates in Section 12.3 to derive some gradient estimates for functions in L^p_u .

Proposition 12.4.5 Under the hypotheses of Proposition 12.3.3, T(t) maps L^p_μ into $W^{1,p}_\mu$, for any t > 0. Moreover, under the assumptions of Proposition 12.3.1,

$$||DT(t)f||_{p} \leq \begin{cases} \left(\frac{c_{p}\sigma_{p}t^{1-\frac{p}{2}}}{\kappa_{0}(1-e^{-\sigma_{p}t})}\right)^{\frac{1}{p}}||f||_{p}, & p \in (1,2), \\ \left(\frac{\sigma_{2}}{2\kappa_{0}(1-e^{-\sigma_{2}t})}\right)^{\frac{1}{2}}||f||_{p}, & p \geq 2, \end{cases}$$

$$(12.4.4)$$

for any t > 0, where c_p and σ_p are the same as in the quoted proposition. When $\sigma_p = 0$ in (12.3.6) and (12.3.7), we replace $\sigma_p/(1 - e^{-\sigma_p t})$ with 1/t.

Similarly, under the assumptions of Proposition 12.3.2 the same estimate holds for any $p \geq p_0$, provided we replace everywhere σ_p with $\tilde{\sigma}_p$, which is given by (12.3.4).

Proof. It follows arguing as in the proof of Proposition 8.3.2, taking the pointwise gradient estimates in Proposition 12.3.3 into account.

As the following proposition shows, the estimates (12.4.4) can be used to give a partial characterization of the domain of the infinitesimal generator of the semigroup $\{T(t)\}$ in L^p_μ .

Proposition 12.4.6 Let p > 1, under the assumptions of Proposition 12.3.1 and $p \ge p_0$, under the assumptions of Proposition 12.3.2. Then, $D(L_p) \subset W_{\mu}^{1,p}$ and, for any $\omega > 0$, there exists a positive constant $M_{\omega,p}$ such that

$$||Df||_p \le M_{\omega,p} ||f||_p^{\frac{1}{2}} ||(L_p - \omega)f||_p^{\frac{1}{2}}, \qquad f \in D(L_p).$$
(12.4.5)

Moreover, under the assumptions of Proposition 12.3.1, the estimate (12.4.5) holds true also with $\omega = 0$, if $\sigma_p < 0$. Similarly, under the assumptions of Proposition 12.3.2, the estimate (12.4.5) holds true also with $\omega = 0$, if $\tilde{\sigma}_p < 0$.

Proof. The proof is similar to that of Proposition 8.3.3. So we skip the details.

To conclude this section, we consider the case when the operator $\mathcal A$ is given by

$$\mathcal{A}\varphi = \Delta\varphi - \langle DU, D\varphi \rangle, \tag{12.4.6}$$

on smooth functions, under the following hypothesis on U:

Hypothesis 12.4.7 The function U belongs to $C^2(\mathbb{R}^N)$,

$$\lim_{|x| \to +\infty} U(x) = +\infty$$

and

$$\sum_{i,j=1}^{N} D_{ij} U(x) \xi_i \xi_j \ge 0, \qquad x \in \mathbb{R}^N, \ \xi \in \mathbb{R}^N.$$

Let us denote by μ the measure defined by

$$\mu(dx) = K^{-1}e^{-U(x)}dx, \qquad K = \int_{\Omega} e^{-U(x)}dx.$$
 (12.4.7)

Lemma 12.4.8 The measure μ in (12.4.7) is the invariant measure of $\{T(t)\}$.

Proof. Let us fix $f \in C_b(\overline{\Omega})$ and denote by $\{T_n(t)\}$ $(n \in \mathbb{N})$ the approximating semigroup defined in Section 12.1. We know that the function $T_n(t)f$ belongs to $C^2(\overline{\Omega}_n)$ for any t > 0. Moreover, for any t > 0, $\partial T_n(t)f/\partial \nu = 0$ on $\partial \Omega_n$ and $T_n(t)f$ converges pointwise to T(t)f as n tends to $+\infty$. An integration by parts shows that

$$\int_{\Omega_n} \mathcal{A}T_n(t) f d\mu = 0,$$

for any t > 0. Therefore,

$$\frac{d}{dt} \int_{\Omega_n} T_n(t) f d\mu = \int_{\Omega} \mathcal{A} T_n(t) f d\mu = 0, \qquad t > 0,$$

which implies that

$$\int_{\Omega_n} T_n(t) f d\mu = \int_{\Omega_n} f d\mu, \qquad t > 0, \ n \in \mathbb{N}.$$

Letting n go to $+\infty$, from the dominated convergence theorem we deduce that

$$\int_{\Omega} T(t)fd\mu = \int_{\Omega} fd\mu, \qquad t > 0,$$

which, of course, implies that μ is the invariant measure of $\{T(t)\}.$

As it has been proved by G. Da Prato and A. Lunardi in [40], a precise characterization of the domain of the infinitesimal generator of $\{T(t)\}$ in L^2_{μ} is available, when \mathcal{A} is given by (12.4.6). As in Chapter 8, we simply write L instead of L_2 .

Theorem 12.4.9 Under Hypothesis 12.4.7, the resolvent set $\rho(L)$ contains the halfline $(0, +\infty)$ and

$$D(L) = \left\{ u \in W_{\mu}^{2,2} : \langle DU, Du \rangle \in L_{\mu}^{2}, \ \frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega \right\}.$$
 (12.4.8)

Moreover,

$$||D^k R(\lambda, L)||_{L(L^2_u)} \le 2^k \lambda^{\frac{k}{2} - 1}, \qquad k = 0, 1, 2.$$
 (12.4.9)

Proof. We split the proof into two steps.

Step 1. Here, we prove that, for any $f \in L^2_{\mu}$ and any $\lambda > 0$, the Neumann problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ \frac{\partial u}{\partial \nu}(x) = 0, & x \in \partial\Omega, \end{cases}$$
 (12.4.10)

admits a unique solution u belonging to the space defined by the right-hand side of (12.4.8) and

$$||D^k u||_2 \le 2^k \lambda^{\frac{k}{2} - 1} ||f||_2. \tag{12.4.11}$$

Let us first prove the uniqueness part. For this purpose, let u be a solution to the problem (12.4.10) (corresponding to $f \equiv 0$) with the claimed regularity properties. An integration by parts shows that

$$\lambda \int_{\Omega} u^2 d\mu = \lambda \int_{\Omega} (\mathcal{A}u) u d\mu = -\frac{1}{2} \int_{\Omega} |Du|^2 d\mu,$$

and, of course, this implies that u = 0.

To prove the existence part, we approximate the operator \mathcal{A} with the sequence of operators $\{\mathcal{A}_n\}$ defined as the operator \mathcal{A} , with U being replaced by the function U_n given by

$$U_n(x) = U(x) + \frac{n}{2}(\operatorname{dist}(x,\Omega))^2, \qquad x \in \mathbb{R}^N, \ n \in \mathbb{N}.$$

As it is immediately seen, the function U_n tends to $+\infty$ as |x| tends to $+\infty$. Moreover, since Ω is convex, U_n is convex in \mathbb{R}^N . Therefore, according to Remark 8.4.4, the realization $L^{(n)}$ in $L^2_{\mu_n}$ of the operator \mathcal{A}_n with domain $D(L^{(n)}) = \{u \in W^{2,2}(\mathbb{R}^N, \mu_n) : \langle DU_n, Du \rangle \in L^2(\mathbb{R}^N, \mu_n) \}$ generates a strongly continuous semigroup in $L^2(\mathbb{R}^N, \mu_n)$, where

$$\mu_n(dx) = K_n^{-1} e^{-U_n(x)} dx, \qquad K_n = \int_{\mathbb{R}^N} e^{-U_n(x)} dx.$$

Let us extend the function f to the whole of \mathbb{R}^N by zero and let us still denote by f such an extended function. For any $\lambda > 0$, the equation

$$\lambda u - \mathcal{A}_n u = f$$

admits a unique solution $u \in D(L^{(n)})$ which satisfies

$$||D^k u_n||_{L^2(\mathbb{R}^N, \mu_n)} \le 2^k \lambda^{\frac{k}{2} - 1} ||f||_{L^2(\mathbb{R}^N, \mu_n)}, \qquad k = 0, 1, 2;$$
(12.4.12)

see Theorem 8.4.2 and Corollary 8.4.3.

We now use a compactness argument to prove that, up to a subsequence, u_n converges to a solution of the problem (12.4.10) which belongs to $W^{2,2}_{\mu}$ and satisfies $\langle DU, Du \rangle \in L^2_{\mu}$. For this purpose, we begin by observing that

$$\|f\|_{L^2(\mathbb{R}^N,\mu_n)} = \left(\frac{1}{K_n} \int_{\Omega} f^2 e^{-U} dx\right)^{\frac{1}{2}} = \left(\frac{\int_{\Omega} e^{-U} dx}{\int_{\mathbb{R}^N} e^{-U_n} dx}\right)^{\frac{1}{2}} \|f\|_2.$$

Therefore, the sequence $\{\|f\|_{L^2(\mathbb{R}^N,\mu_n)}\}$ is bounded. From (12.4.12) we now deduce that the restriction of the functions u_n to Ω gives rise to a bounded sequence in $W^{2,2}_{\mu}$. Indeed, $\|D^k u\|_{L^2(\Omega,\mu)} \leq \|D^k u\|_{L^2(\Omega,\mu_n)}$. Hence, there exists a subsequence $\{u_{n_k}\}\subset \{u_n\}$ which converges weakly in $W^{2,2}_{\mu}$ and strongly in $W^{3/2,2}(\Omega\cap B(R))$, to a function $u\in W^{2,2}_{\mu}$, for any R>0. Moreover, we can assume that u_n converges to u pointwise a.e. in Ω . This implies that the function u solves the differential equation in (12.4.10). Since u and $\Delta u\in L^2_{\mu}$, then the function $\langle DU, Du \rangle$ belongs to L^2_{μ} . Further, (12.4.9) follows by virtue of (12.4.12).

Finally, let us prove that the normal derivative of u identically vanishes on $\partial\Omega$. Using the formula (8.1.29), we can show that

$$\int_{\mathbb{R}^{N}} \langle Du_{n}, D\psi \rangle d\mu = -\int_{\mathbb{R}^{N}} \mathcal{A}_{n} u_{n} \psi d\mu_{n}$$

$$= -\int_{\mathbb{R}^{N} \setminus \Omega} (\mathcal{A}_{n} u_{n}) \psi d\mu_{n} - \int_{\Omega} (\mathcal{A}_{n} u_{n}) \psi d\mu_{n}$$

$$= -\lambda \int_{\mathbb{R}^{N} \setminus \Omega} u_{n} \psi d\mu_{n} - \int_{\Omega} (\mathcal{A}_{n} u_{n}) \psi d\mu_{n}$$

$$= -\lambda \int_{\mathbb{R}^{N} \setminus \Omega} u_{n} \psi d\mu_{n} + \int_{\Omega} \langle Du_{n}, D\psi \rangle d\mu_{n}$$

$$-\frac{1}{K_{n}} \int_{\partial \Omega} \frac{\partial u_{n}}{\partial \nu} \psi e^{-U_{n}} d\sigma,$$

for any $\psi \in C_c^{\infty}(\mathbb{R}^N)$, where $d\sigma$ is the surface measure related to $\partial\Omega$. Therefore,

$$\frac{1}{K_n} \int_{\partial \Omega} \frac{\partial u_n}{\partial \nu} \psi e^{-U_n} d\sigma = -\int_{\mathbb{R}^N \setminus \Omega} \langle Du_n, D\psi \rangle d\mu_n - \lambda \int_{\mathbb{R}^N \setminus \Omega} u_n \psi d\mu_n,$$
(12.4.13)

for any $\psi \in C_c^{\infty}(\mathbb{R}^N)$. Let us observe that the right-hand side of (12.4.13) vanishes as n tends to $+\infty$. Indeed, the sequence $\{u_n\}$ is bounded in $W_{\mu}^{2,2}$

and, as it is easily seen, the $W^{1,2}(\mathbb{R}^N \setminus \Omega, \mu_n)$ -norm of ψ goes to 0 as n tends to $+\infty$. As far as the left-hand side in (12.4.13) is concerned, we observe that, since u_n converges to u in $W^{3/2,2}(\Omega \cap \operatorname{supp} \psi)$, then $\partial u_n/\partial \nu$ tends to $\partial u/\partial \nu$ in $L^2(\partial \Omega \cap \operatorname{supp} \psi)$ as n tends to $+\infty$. So, we have

$$\int_{\partial\Omega} \frac{\partial u}{\partial \nu} \psi e^{-U} d\sigma = \lim_{n \to +\infty} \int_{\partial\Omega} \frac{\partial u_n}{\partial \nu} \psi e^{-U_n} d\sigma$$

from which we deduce that $\partial u/\partial \nu = 0$ on $\partial \Omega$.

Finally, the formula (12.4.9) follows from (12.4.12), letting n go to $+\infty$.

Step 2. Denote by \widehat{L} the realization of the operator \mathcal{A} in L^2_{μ} with domain given by the right-hand side of (12.4.8).

Since the operator \tilde{L} is dissipative and self-adjoint in L^2_{μ} , according to Theorem B.2.9, it generates a strongly continuous analytic semigroup $\{S(t)\}$ in L^2_{μ} . To conclude the proof, we need to show that $T(t) \equiv S(t)$ for any t > 0. For this purpose, we begin by observing that \tilde{L} and L coincide in $C_c^{\infty}(\Omega)$. To see it, it suffices to observe that $C_c^{\infty}(\Omega)$ is properly contained in the space defined by the right-hand side of (12.4.8). Moreover, it is also contained in D(L). Indeed, according to Proposition 12.4.4, the space defined by (12.2.5) is a core of L.

Now, fix $f \in C_c^{\infty}(\Omega)$ and t > 0. Moreover, let $\psi : [0,t] \to L_{\mu}^2$ be the function defined by $\psi(s) = S(t-s)T(s)f$ for any $s \in [0,t]$. Since $\{S(t)\}$ and $\{T(t)\}$ are both strongly continuous semigroups and $\{T(t)\}$ is also analytic, then the function ψ is continuously differentiable in [0,t) with values in L_{μ}^2 and

$$\psi'(s) = -S(t-s)T(s)Lf + S(t-s)\widetilde{L}T(s)f$$

= -S(t-s)T(s)Lf + S(t-s)T(s)\widetilde{L}f
= 0.

Therefore, $\psi(s) = \psi(0)$ for any $s \in (0, t)$. Since ψ is continuous up to s = t, we deduce that $\psi(t) = \psi(0)$ or, equivalently, that $T(t)f \equiv S(t)f$. This completes the proof.

To conclude this section we prove the Poincaré inequality.

Theorem 12.4.10 Suppose that A is given by (12.4.6) with $U \in C^2(\mathbb{R}^N)$ satisfying

$$\sum_{i,j=1}^{N} D_{ij} U(x) \xi_i \xi_j \ge d_0 |\xi|^2, \qquad x \in \Omega, \ \xi \in \mathbb{R}^N,$$

for some negative constant d_0 . Then,

$$\int_{\mathbb{R}^N} |f - \overline{f}|^2 d\mu \le -\frac{1}{d_0} \int_{\mathbb{R}^N} |Df|^2 d\mu, \qquad f \in W_{\mu}^{1,2}.$$
 (12.4.14)

Moreover,

$$||T(t)f - \overline{f}||_2 \le e^{\frac{t}{d_0}} ||f - \overline{f}||_2, \qquad t > 0,$$

for any $f \in L^2_{\mu}$. Finally, we have a spectral gap for L, namely

$$\sigma(L) \setminus \{0\} \subset \{\lambda \in \mathbb{C} : \text{ Re } \lambda \le 1/d_0\}.$$

Proof. The Poincaré inequality can be proved arguing as in the proof of Theorem 8.6.3, taking Theorem 12.4.2 into account and observing that according to Proposition 12.3.1,

$$|(DT(t)f)(x)|^2 \le e^{2d_0t}(T(t)(|Df|^2))(x), \qquad t > 0, \ x \in \Omega.$$

The second and the last part of the theorem can be proved repeating the same arguments as in the proof of Proposition 8.6.4. Hence, we omit the details.

Chapter 13

The Cauchy-Neumann problem: the nonconvex case

13.0 Introduction

In this chapter we consider the case when Ω is a nonconvex unbounded open subset of \mathbb{R}^N , uniformly of class $C^{2+\alpha}$, for some $\alpha \in (0,1)$, and we show that, under somewhat heavier assumptions on the coefficients of the operator

$$\mathcal{A}\varphi(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}\varphi(x) + \sum_{j=1}^{N} b_j(x)D_i\varphi(x) + c(x)\varphi(x), \qquad x \in \Omega,$$
(13.0.1)

we can recover the results proved in Chapter 12. In the convex case the keystone to prove the existence of a classical solution to the problem

$$\begin{cases}
D_t u(t,x) = \mathcal{A}u(t,x), & t > 0, x \in \Omega, \\
\frac{\partial u}{\partial \nu}(t,x) = 0, & t > 0, x \in \partial\Omega, \\
u(0,x) = f(x), & x \in \overline{\Omega}
\end{cases}$$
(13.0.2)

(i.e., a function $u \in C([0,+\infty) \times \overline{\Omega}) \cap C^{0,1}((0,+\infty) \times \overline{\Omega}) \cap C^{1,2}((0,+\infty) \times \Omega)$ which solves the problem (13.0.2) pointwise) was the a priori estimate (12.0.4)for the classical solution u_n to the Cauchy problem (12.1.10) in bounded domains. Such an estimate was proved by using the Bernstein method. The convexity of Ω played a crucial role since it allowed us to approximate Ω by convex bounded domains Ω_n , and the convexity of Ω_n was essential to apply the Bernstein method. Indeed, if F is a smooth convex open set, then any smooth function w defined in F, with normal derivative identically vanishing on ∂F , satisfies $\partial |Dw|^2/\partial \nu \leq 0$ on ∂F (see Proposition 12.1.4). In the case when Ω is not convex, there is no a priori information on the sign of the normal derivative of $|Du_n|^2$ on $\partial\Omega$ and so the technique of Section 12.1 cannot be applied. To overcome such a difficulty, here we do not approximate the domain but we rather deal directly with the problem in the whole of Ω . Taking the geometry of Ω into account, we introduce a suitable positive function m and we prove that, if $\partial u/\partial \nu = 0$ on $\partial \Omega$, then $\partial (m|Du|^2)/\partial \nu < 0$ on $\partial \Omega$. Once we have this preliminary result, we apply the Bernstein method to the function

 $v := u^2 + atm|Du|^2$, u being the classical solution to the problem (13.0.2), and we get (12.0.4) for the function u. To apply such a technique we need to assume that u is sufficiently smooth in order to be able to differentiate v once with respect to time and twice with respect to the time variables in $(0, +\infty) \times \Omega$.

Next, we focus our attention on showing that the Cauchy-Neumann problem (13.0.2) actually admits a (unique) classical solution which satisfies the smoothness assumptions that we needed to apply the Bernstein method.

To guarantee the uniqueness of the bounded classical solution to the problem (13.0.2), we assume as in the previous Chapters, the existence of a Lyapunov function for the operator \mathcal{A} , which allows us to prove a maximum principle.

To prove the existence of a solution to the problem (13.0.2) with the claimed regularity, we adapt to this situation the technique of Chapter 11. For this purpose, we consider auxiliary problems in $L^p(\Omega)$, which involve suitable new operators $\mathcal{A}_{\varepsilon}$ ($\varepsilon > 0$) and we show that, for p large enough and f smooth enough, there exists a unique bounded classical solution u_{ε} to the problem (13.0.2) with \mathcal{A} replaced with $\mathcal{A}_{\varepsilon}$. The regularity of the functions u_{ε} and the suitable choice of the operators $\mathcal{A}_{\varepsilon}$ allow us to apply the a priori gradient estimate proved before to the functions u_{ε} ($\varepsilon > 0$) obtaining that

$$||Du_{\varepsilon}(t,\cdot)||_{\infty} \le \frac{C_T}{\sqrt{t}} ||f||_{\infty}, \qquad t \in (0,T],$$
(13.0.3)

for any T > 0 and some positive constant C_T , depending on T, but being independent of ε . At this point, we show that there exists a sequence $\{u_{\varepsilon_n}\}$ which converges to a solution u of the problem (13.0.2). Letting n go to $+\infty$ in (13.0.3), we find that u satisfies (13.0.3) as well.

A standard approximation argument yields the gradient estimate in the general case when f is bounded and continuous. Finally, the semigroup rule can be used to show that, for any $\omega > 0$, there exists a positive constant $C = C(\omega)$ such that

$$||Du(t,\cdot)||_{\infty} \le \frac{C}{\sqrt{t}} e^{\omega t} ||f||_{\infty}, \qquad t > 0.$$
 (13.0.4)

This procedure works if we assume, among other assumptions, that the diffusion coefficients are bounded in Ω with their first-order derivatives. In particular, this assumption is essential when we deal with the auxiliary problem in $L^p(\Omega)$. However, if Ω is an exterior domain, we can allow the diffusion coefficients to be unbounded in Ω . In such a situation, we approximate the operator \mathcal{A} by a sequence of elliptic operators $\{\mathcal{A}_n\}$ with bounded diffusion coefficients to which we can apply the previous results, getting the existence of a unique bounded classical solution u_n satisfying the gradient estimate (13.0.4). Then, letting n go to $+\infty$, we finally see that u_n converges to the (unique) solution to the problem (13.0.2), which, of course, satisfies (12.0.4).

As a consequence of the results so far proved we can define, as usual, a semi-group $\{T(t)\}$ of linear bounded operators in the space $C_b(\overline{\Omega})$. We show that the results in Proposition 12.1.7 as well as all the results in Subsection 12.2 can be extended to the case when Ω is not convex. Concerning the pointwise estimate, we confine ourselves to the case when $c \equiv 0$. In the case when the diffusion coefficients are bounded we show that the semigroup $\{T(t)\}$ satisfies (12.0.7) with e^{k_0pt} being replaced with $C_pe^{M_pt}$ for some constants $C_p > 0$ and $M_p \in \mathbb{R}$. In the case when Ω is an exterior domain, we are able to recover the estimate (12.0.7), by assuming new conditions on the coefficients, which are slightly more restrictive than those used to prove the uniform gradient estimate. But we still allow the q_{ij} 's to be unbounded. On the other hand, still when Ω is an exterior domain, we can prove an estimate similar to (12.3.10), under the same set of hypotheses used to get (12.0.4).

Next, we derive another type of pointwise gradient estimates that, in some cases, allow us to improve the estimates on the behaviour of the sup-norm of DT(t)f when t approaches infinity and $f \in C_b(\overline{\Omega})$, as well as to prove a Liouville type theorem for the operator \mathcal{A} .

Finally, in Section 13.4, we briefly generalize to this situation some results of Chapter 12.

The results of this chapter have been proved in [16].

13.1 The case of bounded diffusion coefficients

We begin this section by recalling that, if Ω is uniformly of class $C^{2+\alpha}$ for some $\alpha \in (0,1)$, then the distance function

$$x \mapsto d(x) = \operatorname{dist}(x, \partial\Omega),$$

belongs to $C_b^2(\Gamma(\rho))$ for some $\rho > 0$, where

$$\Gamma(\rho) := \{ x \in \overline{\Omega} : d(x) < \rho \}. \tag{13.1.1}$$

Moreover,

$$Dd(x) = -\nu(\xi(x)), \tag{13.1.2}$$

where $\xi(x)$ is the unique point on $\partial\Omega$ such that $d(x,\partial\Omega) = |x - \xi(x)|$. In particular, formula (13.1.2) implies that |Dd(x)| = 1 for any $x \in \Gamma(\rho)$. We refer to Appendix D for the proof of these properties concerning the distance function.

Finally, according to Remark D.0.3,

$$M_0 := \inf_{x \in \partial \Omega} \left\{ \frac{\partial \nu}{\partial \tau}(x) \cdot \tau, \ |\tau| = 1, \ \tau \cdot \nu(x) = 0 \right\} > -\infty.$$

370

Therefore, we can define the constant ω_0 which will be used throughout this chapter, by setting

$$\omega_0 = \max\{0, -M_0\}. \tag{13.1.3}$$

We can now state the hypotheses that we always assume to be satisfied in this section.

Hypotheses 13.1.1 (i) Ω is an open nonconvex unbounded set uniformly of class $C^{2+\alpha}$ for some $\alpha \in (0,1)$;

(ii) the coefficients q_{ij} and b_j (i, j = 1, ..., N) and c belong to $C^{1+\alpha}_{loc}(\overline{\Omega})$. Moreover, $q_{ij} \in C^1_b(\overline{\Omega})$, for any i, j = 1, ..., N, and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa_0|\xi|^2, \qquad x \in \Omega, \quad \xi \in \mathbb{R}^N,$$
 (13.1.4)

for some positive constant κ_0 ;

(iii) $c(x) \leq 0$ for any $x \in \overline{\Omega}$ and there exist two constants c_0 and γ such that $-c_0 := \sup_{x \in \overline{\Omega}} c(x), \qquad |Dc(x)| \leq \gamma (1 - c(x)), \quad x \in \Omega;$

(iv) there exist $k_1, k_2 \in \mathbb{R}$ and $s_1, s_2 \geq 0$ such that

(a)
$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \le (k_1 - s_1 c(x)) |\xi|^2, \quad x \in \Omega, \quad \xi \in \mathbb{R}^N;$$

(b)
$$-\sum_{j=1}^{N} b_j(x) D_j d(x) \le k_2 - s_2 c(x), \quad x \in \Gamma(\rho),$$

with

$$s_1 + \frac{\omega_0}{2}s_2 < \frac{1}{2}; \tag{13.1.5}$$

(v) there exist two constants $d_1, d_2 \ge 0$ such that

$$|b(x)| \le d_1 \exp(d_2|x|), \quad x \in \Omega;$$

(vi) there exist a function $\varphi \in C^2(\overline{\Omega})$ and a constant $\lambda_0 > c_0$ such that

$$\lim_{|x|\to+\infty}\varphi(x)=+\infty,\quad \sup_{x\in\Omega}(\mathcal{A}-\lambda_0)\varphi(x)<+\infty,\quad \frac{\partial\varphi}{\partial\nu}(x)\geq 0,\ x\in\partial\Omega.$$

The requirement that $q_{ij} \in C_b^1(\overline{\Omega})$ (i, j = 1, ..., N) as well as Hypothesis 13.1.1(v) are essential in order to prove the existence of a bounded classical solution to the problem (13.0.2). The geometry of Ω is taken into account by Hypothesis 13.1.1(iv)(b) which requires that the growth of the component of the drift along the outward normal direction is balanced by the potential in a neighborhood of $\partial\Omega$.

13.1.1 A priori estimate

In this subsection we are devoted to prove the following a priori estimate

$$|Du(t,x)| \le C_T t^{-\frac{1-k}{2}} \|f\|_{C_b^k(\overline{\Omega})}, \qquad t \in (0,T], \ x \in \overline{\Omega}, \ k = 0,1, \quad (13.1.6)$$

for any T > 0 and any $f \in C_b^k(\overline{\Omega})$ (satisfying $\partial f/\partial \nu = 0$ if k = 1), when u is a bounded classical solution to the problem (13.0.2) with the further property that $Du \in C_b([0,T] \times \overline{\Omega}) \cap C^{0,1}((0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$. We will deal with the existence of such a solution in the next subsection. In order to prove (13.1.6) we adapt to our situation the classical Bernstein method. For this purpose, we introduce the function $m: \overline{\Omega} \to \mathbb{R}$ defined by

$$m(x) = 1 + \omega_0 \psi(d(x)), \qquad x \in \overline{\Omega},$$
 (13.1.7)

where ω_0 is given by (13.1.3) and $\psi \in C_b^2([0,+\infty))$ is any function such that

$$0 \le \psi \le \frac{\varepsilon_0}{2}, \quad \psi(0) = 0, \quad \psi(s) = \frac{\varepsilon_0}{2}, \quad s > \varepsilon_0,$$
 (13.1.8)

$$\psi'(0) = 1, \quad 0 \le \psi' \le 1, \quad -\frac{2}{\varepsilon_0} \le \psi'' \le 0,$$
 (13.1.9)

for some $0 < \varepsilon_0 < \rho$, being ρ given by (13.1.1). As it is readily seen,

$$1 \le m \le 1 + \frac{\omega_0 \varepsilon_0}{2}$$

$$m(x) = 1 + \frac{\omega_0 \varepsilon_0}{2}, \quad x \in \Omega \setminus \Gamma(\varepsilon_0), \quad m(x) = 1, \quad x \in \partial\Omega.$$

Let us prove the following fundamental lemma.

Lemma 13.1.2 Let $u \in C^2(\overline{\Omega})$ be such that $\partial u/\partial \nu = 0$ on $\partial \Omega$. Then, the function $v := m|Du|^2$ satisfies

$$\frac{\partial v}{\partial \nu}(x) \le 0, \quad x \in \partial \Omega.$$

Proof. Since $Du(x) \cdot \nu(x) = 0$ for any $x \in \partial \Omega$, differentiating along a unit tangent vector τ , we obtain

$$\frac{\partial}{\partial \tau}(Du(x) \cdot \nu(x)) = D^2 u(x) \tau \cdot \nu(x) + \frac{\partial \nu}{\partial \tau}(x) \cdot Du(x) = 0, \quad x \in \partial\Omega.$$
 (13.1.10)

We now observe that, from (13.1.2), (13.1.3) and (13.1.9), it follows that, if $x \in \partial \Omega$ and $\tau \cdot \nu(x) = 0, |\tau| = 1$, then

$$\frac{\partial m}{\partial \nu}(x) - 2\frac{\partial \nu}{\partial \tau}(x) \cdot \tau = \omega_0 \,\psi'(0) Dd(x) \cdot \nu(x) - 2\frac{\partial \nu}{\partial \tau}(x) \cdot \tau$$

$$= -\omega_0 - 2\frac{\partial \nu}{\partial \tau}(x) \cdot \tau \le 0. \tag{13.1.11}$$

Therefore, if we choose, for any $x \in \partial \Omega$,

$$\tau = \tau(x) = m(x)|Du(x)|^{-1}Du(x) = |Du(x)|^{-1}Du(x),$$

from (13.1.10) and (13.1.11) we get

$$\begin{split} \frac{\partial v}{\partial \nu}(x) &= \frac{\partial m}{\partial \nu}(x)|Du(x)|^2 + 2m(x)D^2u(x)Du(x) \cdot \nu(x) \\ &= |Du(x)|^2 \left\{ \frac{\partial m}{\partial \nu}(x) - 2\frac{\partial \nu}{\partial \tau}(x) \cdot \tau \right\} \leq 0, \quad x \in \partial\Omega. \end{split}$$

We can now prove the *a priori* estimate (13.1.6) requiring, for the time being, a further regularity property to the function u, that will be removed later.

Proposition 13.1.3 Assume that Hypotheses 13.1.1 are satisfied and let $u \in C([0,T] \times \overline{\Omega}) \cap C^{0,1}((0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$ be a bounded solution of the Cauchy-Neumann problem (13.0.2). Further, assume that the function Du is bounded in $[0,T] \times \overline{\Omega}$ and it belongs to $C^{0,1}((0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$, for any T > 0. Then, there exists a constant C > 0 such that (13.1.6) holds. In particular, if $c \equiv 0$, then (13.1.6), with k = 1, holds with a constant being independent of T.

Proof. We begin by proving (13.1.6) with k=0. For this purpose, let us define the function

$$v(t,x) = |u(t,x)|^2 + a t m(x) |Du(t,x)|^2, \quad t \ge 0, \ x \in \overline{\Omega},$$

where a > 0 is a parameter that will be chosen later. By the assumptions, $v \in C_b([0,T] \times \overline{\Omega}) \cap C^{0,1}((0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$ and $v(0,\cdot) = f^2$.

We claim that there exists a suitable value of a>0, independent of f,t,x, such that

$$\begin{cases}
D_t v(t, x) - \mathcal{A}v(t, x) \le M v, & t \in (0, T], \ x \in \Omega, \\
\frac{\partial v}{\partial \nu}(t, x) \le 0 & t \in (0, T], \ x \in \partial \Omega,
\end{cases}$$
(13.1.12)

for some constant $M \geq 0$, having the same dependence as a. Then, by the maximum principle in Theorem 12.1.5, applied to the function $e^{-Mt}v$, we get

$$v(t,x) \leq e^{MT} \sup_{x \in \overline{\Omega}} v(0,x) = e^{MT} \|f\|_{\infty}^2, \qquad t \in [0,T], \ x \in \overline{\Omega},$$

which yields (13.1.6), with k = 0 and $C_T = e^{MT/2}a^{-1/2}$, since $m(x) \ge 1$ for any $x \in \overline{\Omega}$.

According to Lemma 13.1.2, $\partial v/\partial \nu \leq 0$ on $\partial \Omega$. Thus we have only to prove that v satisfies the differential inequality in (13.1.12). A straightforward computation shows that v solves the following equation:

$$D_{t}v - \mathcal{A}v = am|Du|^{2} + cu^{2} - 2\langle QDu, Du \rangle$$

$$+2at \left\{ -m \operatorname{Tr} (QD^{2}uD^{2}u) + \frac{1}{2}mc|Du|^{2} \right.$$

$$+ m \sum_{i,j,h=1}^{N} D_{h}q_{ij}D_{h}uD_{ij}u + m \sum_{i,j=1}^{N} D_{i}b_{j}D_{i}uD_{j}u$$

$$+ mu\langle Dc, Du \rangle - \frac{\omega_{0}}{2}\psi''(d)\langle QDd, Dd \rangle |Du|^{2}$$

$$- \frac{\omega_{0}}{2}\psi'(d)(\mathcal{A}d - cd)|Du|^{2} - 2\omega_{0}\psi'(d)\langle QDd, D^{2}uDu \rangle \right\}.$$

$$(13.1.13)$$

Let us estimate each term in the right-hand side of (13.1.13). By the ellipticity condition (13.1.4) we have

$$-m \operatorname{Tr} (QD^2 u D^2 u) \le -m \kappa_0 |D^2 u|^2. \tag{13.1.14}$$

Using the boundedness of Dq_{ij} (i, j = 1, ..., N) and the Hölder and Young inequalities, we can write

$$\begin{split} m \sum_{i,j,h=1}^{N} D_{h} q_{ij} D_{h} u D_{ij} u &\leq m \bigg(\sum_{i,j=1}^{N} |\langle D q_{ij}, D u \rangle|^{2} \bigg)^{\frac{1}{2}} \bigg(\sum_{i,j=1}^{N} |D_{ij} u|^{2} \bigg)^{\frac{1}{2}} \\ &\leq \frac{1}{4\delta} \|DQ\|_{\infty} m |Du|^{2} + \delta \|DQ\|_{\infty} m |D^{2} u|^{2}, \ (13.1.15) \end{split}$$

for any $\delta > 0$. From Hypothesis 13.1.1(iv)(a) we get

$$m\sum_{i,j=1}^{N} D_i b_j D_i u D_j u \le k_1 m |Du|^2 - s_1 m c |Du|^2$$
(13.1.16)

and, from Hypothesis 13.1.1(iii), it follows that, for arbitrary $\delta > 0$,

$$mu\langle Dc, Du \rangle \le \gamma m(1-c)|u||Du| \le \gamma \delta m(1-c)|Du|^2 + \frac{\gamma}{4\delta} m(1-c)u^2$$
. (13.1.17)

Using (13.1.9) and the Hölder inequality yields

$$-\frac{\omega_0}{2}\psi''(d)\langle QDd, Dd\rangle |Du|^2 \le \frac{\omega_0}{\varepsilon_0} ||Q||_{\infty} |Du|^2.$$
 (13.1.18)

Analogously, by Hypothesis 13.1.1(iv)(b) and (13.1.8) we have

$$-\frac{\omega_0}{2}\psi'(d)(\mathcal{A}d - cd)|Du|^2 = -\frac{\omega_0}{2}\psi'(d)\left(\text{Tr}(QD^2d) + \langle b, Dd \rangle\right)|Du|^2$$

$$\leq \frac{\omega_0}{2}\left(\|Q\|_{\infty}\|D^2d\|_{\infty} + k_2\right)|Du|^2 - \frac{\omega_0}{2}s_2c|Du|^2.$$
(13.1.19)

Finally

$$-2\omega_0 \psi'(d) \langle QDd, D^2 u Du \rangle \le 2\omega_0 \|Q\|_{\infty} |D^2 u| |Du|$$

$$\le \frac{\omega_0}{2\delta} \|Q\|_{\infty} |Du|^2 + 2\delta\omega_0 \|Q\|_{\infty} |D^2 u|^2. \quad (13.1.20)$$

Now, combining the estimates (13.1.14)-(13.1.20) and using the fact that $m \ge 1$, whenever it is necessary, we get

$$\begin{split} D_{t}v - \mathcal{A}v &\leq (a \, m - 2\kappa_{0})|Du|^{2} + cu^{2} \\ &+ 2at \bigg\{ \left(\delta \|DQ\|_{\infty} + 2\delta\omega_{0} \|Q\|_{\infty} - \kappa_{0} \right) m|D^{2}u|^{2} \\ &- \left(s_{1} + \gamma\delta + \frac{\omega_{0}}{2} s_{2} - \frac{1}{2} \right) mc|Du|^{2} \\ &+ \bigg[\frac{\|DQ\|_{\infty}}{4\delta} + k_{1} + \gamma\delta + \omega_{0} \|Q\|_{\infty} \left(\frac{1}{\varepsilon_{0}} + \|D^{2}d\|_{\infty} \right) \\ &+ \frac{k_{2}\omega_{0}}{2} + \frac{\omega_{0}}{2\delta} \|Q\|_{\infty} \bigg] m \, |Du|^{2} + \frac{\gamma}{4\delta} m \, (1 - c)u^{2} \bigg\}. \end{split}$$

By (13.1.5) we can choose δ small enough to have

$$||DQ||_{\infty}\delta + 2\omega_0\delta||Q||_{\infty} - \kappa_0 \le 0, \quad s_1 + \gamma\delta + \frac{\omega_0}{2}s_2 - \frac{1}{2} \le 0.$$

Therefore, we obtain

$$D_t v - Av \le (a m - 2\kappa_0)|Du|^2 + cu^2 + 2aT \left\{ Km |Du|^2 + \frac{\gamma}{4\delta} (1 - c)m u^2 \right\},$$

where we have set

$$K := \frac{\|DQ\|_{\infty}}{4\delta} + k_1 + \gamma\delta + \omega_0 \|Q\|_{\infty} \left(\frac{1}{\varepsilon_0} + \|D^2 d\|_{\infty}\right) + \frac{k_2 \omega_0}{2} + \frac{\omega_0}{2\delta} \|Q\|_{\infty}.$$

Since $m \leq 1 + \omega_0 \varepsilon_0/2$, it follows that

$$D_t v - \mathcal{A}v \le \left[\left(1 + \frac{\omega_0 \varepsilon_0}{2} \right) a (1 + 2TK) - 2\kappa_0 \right] |Du|^2$$
$$- \left[-1 + aT \left(1 + \frac{\omega_0 \varepsilon_0}{2} \right) \frac{\gamma}{2\delta} \right] c u^2 + aT \left(1 + \frac{\omega_0 \varepsilon_0}{2} \right) \frac{\gamma}{2\delta} u^2.$$

Now, it is clear that there exists a sufficiently small value a > 0, which is independent of f, t, x, such that v satisfies the differential inequality in (13.1.12) with $M = a T \gamma (2 + \omega_0 \varepsilon_0)/(4\delta)$.

In order to prove (13.1.6) with k = 1, it suffices to replace the function v with $u^2 + am|Du|^2$ and to argue as above.

13.1.2 Existence of the solution to problem (13.0.2) and uniform gradient estimates

The aim of this subsection is to prove the existence of the bounded classical solution u to the problem (13.0.2) and to show that u satisfies the estimate (13.1.6). The crucial point consists in proving the statement in the case when f is smooth. Indeed, the general case then will follow by approximating $f \in C_b(\overline{\Omega})$ with a sequence of smooth functions. For this purpose, by adapting the technique of Subsection 11.3.2, we approximate the operator \mathcal{A} by new operators $\mathcal{A}_{\varepsilon} = \mathcal{A} + c_{\varepsilon}$, where the auxiliary potential c_{ε} is chosen in such a way that $\mathcal{A}_{\varepsilon}$ generates a strongly continuous analytic semigroup $\{T_{\varepsilon}(t)\}$ in $L^p(\Omega)$ for any $p \in [2, +\infty)$. We prove that, for any $\varepsilon > 0$, the function $u_{\varepsilon} = T_{\varepsilon}(\cdot)f$ solves the problem (13.0.2), with $\mathcal{A}_{\varepsilon}$ replacing \mathcal{A} and, for p large enough, it enjoys the regularity properties of Proposition 13.1.3. Hence, it verifies (13.1.6). Taking the limit as ε goes to 0^+ allows us to prove the existence of a classical solution of (13.0.2) as well as of (13.1.6) for such a solution.

Generation of analytic semigroups in L^p -spaces

For notational convenience, throughout this subsection, we denote by $\|\cdot\|_p$ and $\|\cdot\|_{k,p}$ the usual norms of the spaces $L^p(\Omega)$ and $W^{k,p}(\Omega)$ $(k \in \mathbb{N})$, respectively.

We assume more restrictive conditions on the coefficients of the operator \mathcal{A} in (13.0.1). More precisely, in this subsection we always assume the following hypotheses.

Hypotheses 13.1.4 (i) the coefficients q_{ij} and b_j (i, j = 1, ..., N) and c belong to $C_{\text{loc}}^{1+\alpha}(\overline{\Omega})$ for some $\alpha \in (0,1)$. Moreover $q_{ij} \in C_b^1(\overline{\Omega})$, for any i, j = 1, ..., N, and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa_0|\xi|^2, \qquad x \in \Omega, \quad \xi \in \mathbb{R}^N,$$

for some positive constant κ_0 ;

(ii) there exist $\beta > 0$ and $\sigma \in (0, \sqrt{2\kappa_0})$ such that

(i)
$$c(x) \le -1$$
, (ii) $|Dc(x)| \le \beta |c(x)|$, (iii) $|b(x)| \le \sigma |c(x)|^{\frac{1}{2}}$, (13.1.21)

376

for any $x \in \Omega$.

For any $p \in [2, +\infty)$, we introduce the Banach space

$$D_p = \left\{ u \in W^{2,p}(\Omega) : \frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega, \quad cu \in L^p(\Omega) \right\},$$
 (13.1.22)

which is endowed with the norm

$$||u||_{D_p} = ||u||_{2,p} + ||c u||_p, \qquad u \in D_p,$$

and the subspace

$$D = \left\{ u \in C_c^2(\overline{\Omega}) \ : \ \frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega \right\}.$$

As the following lemma shows D is dense in D_p for any $p \in [2, +\infty)$.

Lemma 13.1.5 For any $p \in [2, +\infty)$, D is dense in D_p .

Proof. Let U_n and g_n be as in Definition D.0.1 and let $\{x_i + B(R)\}_{i \in \mathbb{N}}$ be a countable family of balls with the following properties:

- (i) the closure of $x_i + B(2R)$ is contained in Ω , for any $i \in \mathbb{N}$;
- (ii) $\cup_{i\in\mathbb{N}}(x_i+B(R))$ covers $\Omega\setminus\Omega_{\varepsilon}$, where $\Omega_{\varepsilon}=\{x\in\mathbb{R}^N:d(x)<\varepsilon\}$ and $\varepsilon>0$ is given by the condition (ii) of Definition D.0.1;
- (iii) there exists $s \in \mathbb{N}$ such that at most s among the balls $\{x_i + B(2R)\}$ overlap.

We now introduce a covering $\{\Omega_k\}_{k\in\mathbb{N}}$ of $\overline{\Omega}$ defined as follows:

$$\Omega_{2k} = x_k + B(R), \qquad \Omega_{2k+1} = V_k = g_k^{-1}(B(1/2)),$$

and with this covering we associate a partition of the unity $\{\vartheta_k\}$ such that

- (i) the C^2 -norms of the functions ϑ_k are bounded by a constant independent of k;
- (ii) ϑ_{2k+1} satisfies the Neumann condition $\frac{\partial \vartheta_{2k+1}}{\partial \nu} = 0$ on $\Omega_{2k+1} \cap \partial \Omega$.

Now, let $u \in W^{2,p}(\Omega)$ be such that $\partial u/\partial \nu = 0$ on $\partial \Omega$ and $cu \in L^p(\Omega)$, and fix $\varepsilon > 0$. We claim that there exists a function $\varphi \in C^2(\overline{\Omega})$ with compact support in $\overline{\Omega}$, such that $\partial \varphi/\partial \nu = 0$ on $\partial \Omega$ and

$$||u - \varphi||_{W^{2,p}(\Omega)} + ||cu - c\varphi||_{L^p(\Omega)} < \varepsilon.$$

$$(13.1.23)$$

To prove the claim, for any $k \in \mathbb{N}$ we set $u_k = \vartheta_k u$, $a_{2k} = \|c\|_{L^{\infty}(x_k + B(2R))}$, $a_{2k+1} = \|c\|_{L^{\infty}(V_k \cap \Omega)}$. We observe that

$$u(x) = \sum_{k=1}^{+\infty} u_k(x), \qquad x \in \Omega,$$

and that $a_k \geq 1$ since, by assumptions, $c \leq -1$. At this point, the idea is to approximate each function u_k by regular functions. Consider first the case when k is even. Then, $u_k \in W^{2,p}(\mathbb{R}^N)$ with supp $u_k \subseteq x_{k/2} + B(R)$. By classical results, there exists a function $v_k \in C^2(\mathbb{R}^N)$ with supp $v_k \subseteq x_{k/2} + B(2R)$ such that

$$||u_k - v_k||_{W^{2,p}(\mathbb{R}^N)} = ||u_k - v_k||_{W^{2,p}(x_{k/2} + B(2R))} < \frac{\varepsilon}{4^k a \iota}.$$
 (13.1.24)

It follows that

$$||cu_k - cv_k||_{L^p(x_{k/2} + B(2R))} \le a_k ||u_k - v_k||_{L^p(x_{k/2} + B(2R))} < \frac{\varepsilon}{4^k}.$$
 (13.1.25)

Now, assume that k is odd and consider the function \tilde{u}_k defined by $\tilde{u}_k(y) = u_k(g_n^{-1}(y))$ for any $y \in B^+(1/2) = \{z \in B(1/2) \mid z_N > 0\}$. Since the change of variables is of class C^2 , $\tilde{u}_k \in W^{2,p}(B^+(1/2))$ and, if $y \in B(1/2)$ with $y_N = 0$, then, taking Remark D.0.2 into account, we get

$$\frac{\partial \tilde{u}_k}{\partial y_N}(y) = \langle D\tilde{u}_k(y), e_N \rangle = \langle (Du_k)(g_n^{-1}(y)), \operatorname{Jac}(g_n^{-1})(y)e_N \rangle
= -\frac{1}{\alpha_n(g_n^{-1}(y))} \langle (Du_k)(g_n^{-1}(y)), \nu(g_n^{-1}(y)) \rangle = 0,$$

since $\partial u_k/\partial \nu = u\partial \vartheta_k/\partial \nu + \vartheta_k\partial u/\partial \nu = 0$, thanks to the choice of ϑ_k . It follows that the extension of \tilde{u}_k with value zero in $\mathbb{R}^N_+ \setminus B^+(1/2)$ belongs to $W^{2,p}(\mathbb{R}^N_+)$ and its normal derivative identically vanishes on $\{y_N=0\}$. A standard technique based on a truncation argument and convolution with functions which are even with respect to the last variable allows us to construct, for any $\varepsilon'>0$, a new function $\tilde{v}_k \in C^2(\mathbb{R}^N_+)$ with supp $\tilde{v}_k \subseteq B(1)$, $\partial \tilde{v}_k/\partial y_N=0$, if $y\in B(1)$ and $y_N=0$, such that

$$\|\tilde{v}_k - \tilde{u}_k\|_{W^{2,p}(B^+(1))} \le \frac{\varepsilon'}{4^k a_k}.$$

It is immediate to check that the function $v_k = \tilde{v}_k(g_n(\cdot))$ belongs to $C^2(U_n \cap \overline{\Omega})$, it has support contained in U_n and its normal derivative vanishes on $U_n \cap \partial \Omega$. Since the C^2 -norm of g_n can be taken independent of n (by the condition (iii) in Definition D.0.1), we can choose ε' sufficiently small such that

$$||v_k - u_k||_{W^{2,p}(U_n \cap \Omega)} < \frac{\varepsilon}{4k} a_k, \qquad ||cv_k - cu_k||_{L^p(U_n \cap \Omega)} < \frac{\varepsilon}{4k}.$$
 (13.1.26)

At this point, for any $n \in \mathbb{N}$, we set

$$v_n(x) = \sum_{k=1}^n v_k(x), \quad x \in \overline{\Omega}.$$

Of course, $v_n \in C_c^2(\overline{\Omega})$ and $\partial v_n/\partial \nu = 0$ on $\partial \Omega$. Moreover, using (13.1.24), (13.1.25) and (13.1.26), we obtain

$$\|u - v_n\|_{D_p} \le \frac{2}{3}\varepsilon + \left\| \left(1 - \sum_{k=1}^n \vartheta_k \right) u \right\|_{W^{2,p}(\Omega)} + \left\| \left(1 - \sum_{k=1}^n \vartheta_k \right) V u \right\|_{L^p(\Omega)}.$$
(13.1.27)

As n goes to $+\infty$, the two last terms in the right-hand side of (13.1.27) vanish by dominated convergence. Therefore, choosing n sufficiently large, we get (13.1.23) with $\varphi = v_n$.

Now, our purpose consists in showing that, under Hypotheses 13.1.4, the operator \mathcal{A} with domain D_p is well defined and generates an analytic semigroup $\{T_p(t)\}$, and that, for any $f \in C_b(\overline{\Omega})$ and any p > N, the function $T_p(t)f$ is a classical solution to the problem (13.0.2) satisfying the assumptions of Proposition 13.1.3.

For this purpose, we are going to adapt the techniques in Subsection 11.3.2, which are based on integration by parts. We note that since the conormal vector associated with the operator Q needs not to coincide with the normal derivative on $\partial\Omega$, integrating by parts, we are led to some surface integrals which we would avoid. For this purpose, we show that we can replace the matrix Q with a new matrix Q' such that $\text{Tr}(QD^2u) = \text{Tr}(Q'D^2u)$ for any $u \in D_p$ and the conormal vector associated with Q' coincides with the normal vector on $\partial\Omega$. This allows us to eliminate the surface integrals. For this purpose, for any $x \in \Omega$ we introduce the matrix S(x) defined by

$$S(x)y = \vartheta(x)\{\langle y, Q(x)Dd(x)\rangle Dd(x) - \langle y, Dd(x)\rangle Q(x)Dd(x)\}, \quad y \in \mathbb{R}^N,$$

where $\vartheta \in C_b^2(\mathbb{R}^N)$ is such that $\vartheta = 1$ in $\Gamma(\rho/2)$ and it vanishes outside $\Gamma(\rho)$ (see (13.1.1)). Then, we set Q'(x) = Q(x) + S(x). As the following lemma shows, Q' has all the claimed properties.

Lemma 13.1.6 We have:

- (i) $q'_{ij} \in C_b^1(\overline{\Omega})$ (i, j = 1, ..., N) and $\sum_{i,j=1}^N q'_{ij}(x)\xi_i\xi_j \geq \kappa_0|\xi|^2$ for any $x \in \Omega$ and any $\xi \in \mathbb{R}^N$;
- (ii) $\operatorname{Tr}(QA)(x) = \operatorname{Tr}(Q'A)(x)$, for any symmetric matrix A and any $x \in \overline{\Omega}$;
- (iii) $Q'(x)\nu(x) = \langle \nu(x), Q(x)\nu(x)\rangle \nu(x)$, for any $x \in \partial\Omega$.

Moreover, for any $u \in D_p$ and any $v \in W^{1,q}(\Omega)$, with $\frac{1}{p} + \frac{1}{q} = 1$, we have

$$\int_{\Omega} (A_0 u) v \, dx = -\sum_{i,j=1}^{N} \int_{\Omega} v D_j q'_{ij} \, D_i u \, dx - \sum_{i,j=1}^{N} \int_{\Omega} q'_{ij} D_i u \, D_j v \, dx. \quad (13.1.28)$$

Proof. Since S is skew-symmetric, the properties (i) and (ii) follow immediately. Similarly, the property (iii) follows easily taking (13.1.2) into account. In order to prove (13.1.28), it suffices to write $A_0u = \text{Tr}(Q'D^2u)$, and to integrate by parts using the property (iii).

For notational convenience, we now introduce the vector-valued function $b': \overline{\Omega} \to \mathbb{R}^N$ defined by

$$b'_{i}(x) = \sum_{j=1}^{N} D_{j} q'_{ij}(x) - b_{i}(x), \qquad x \in \overline{\Omega}, \quad i = 1, \dots, N.$$

Taking the assumption (13.1.21) into account, it is immediate to see that

$$|b'| \le |b| + \sqrt{N} \|DQ'\|_{\infty} \le \sigma |c|^{\frac{1}{2}} + \sqrt{N} \|DQ'\|_{\infty}.$$
 (13.1.29)

Proposition 13.1.7 For any $p \geq 2$, the operator A_p is well defined in D_p . Moreover, there exists a positive constant λ_0 , depending on $\kappa_0, \sigma, p, N, \|Q\|_{\infty}, \|DQ\|_{\infty}, \gamma$, such that

$$\|\lambda u - A_p\|_p \ge (\lambda - \lambda_0) \|u\|_p, \qquad \lambda > \lambda_0, \quad u \in D_p.$$
 (13.1.30)

Moreover, for any $\lambda > \lambda_0$, there exist two constants $C_1, C_2 > 0$ (depending also on λ), such that

$$C_1 \|u\|_{D_p} \le \|\lambda u - Au\|_p \le C_2 \|u\|_{D_p}, \quad u \in D_p.$$
 (13.1.31)

Proof. It can be obtained arguing as in the proof of Lemma 11.3.9 and Proposition 11.3.10, writing $Au = \text{Tr}(Q'D^2u) + \langle b, Du \rangle + cu$. Indeed, observe that, according to Lemma 13.1.6, the conormal derivative relevant to the operator $u \mapsto \text{Tr}(Q'D^2u)$ coincides, up to a multiplication factor, with the normal derivative. Therefore, integrating by parts gives rise to the same integrals as in the proof of Proposition 11.3.10.

Next step consists in proving that, for any $p \in [2, +\infty)$, the operator A with domain D_p generates an analytic semigroup of linear operators in $L^p(\Omega)$. For this purpose, we approximate the operator \mathcal{A} by operators with bounded coefficients.

Proposition 13.1.8 For any $p \geq 2$, (A_p, D_p) generates a strongly continuous analytic semigroup $\{T_p(t)\}$ in $L^p(\Omega)$.

380

Proof. The proof is close to that of Proposition 11.3.12, so that we limit ourselves to sketching it. To prove that A_p generates a strongly continuous analytic semigroup, we fix $\sigma' \in (\sigma, \sqrt{2\kappa_0})$ and replace the operator A_p with the operator $B_p = A_p - k$ defined in $D(A_p)$, choosing k sufficiently large such that

$$|b'(x)| \le \sigma' |c'(x)|^{\frac{1}{2}}, \qquad x \in \overline{\Omega},$$

where c' = c - k. This is possible since b' satisfies (13.1.29). Then, we approximate B_p by the operators $B_{p,\varepsilon} = \text{Tr}(Q'D^2 \cdot) + \langle b_{\varepsilon}, D \cdot \rangle + c'_{\varepsilon}$ where b_{ε} and c'_{ε} are the same as in (11.3.35), i.e.,

$$b_{\varepsilon} = \frac{b}{\sqrt{1 - \varepsilon c}}, \qquad c'_{\varepsilon} = \frac{c}{1 - \varepsilon c} + k.$$

Arguing as in the proof of Proposition 11.3.12 and taking Proposition 13.1.7 into account, we can show that, for any $\lambda > \lambda_0$ (where λ_0 is given by Proposition 13.1.7) and any $f \in L^p(\Omega)$, there exists a sequence of functions $u_{\varepsilon_n} \in W^{2,p}(\Omega)$, solutions to the equations $\lambda u_{\varepsilon_n} - B_{p,\varepsilon_n} u_{\varepsilon_n} = f$, such that $\partial u_{\varepsilon_n}/\partial \nu = 0$ on $\partial \Omega$ and $c_{\varepsilon_n} u_{\varepsilon_n} \in L^p(\Omega)$. Moreover, u_{ε_n} converges weakly in $W^{2,p}(\Omega)$ to a solution $u \in W^{2,p}(\Omega)$ of the equation $\lambda u - B_p u = f$ such that $cu \in L^p(\Omega)$. To conclude that $u \in D_p$ it suffices to show that $\partial u/\partial \nu = 0$ on $\partial \Omega$. By the classical interior L^p -estimates applied to the operator $\lambda - A_0$ (see Theorem C.1.1), for any pair of bounded and smooth open sets $\Omega' \subset \Omega'' \subset \Omega$, we have

$$\|u_{\varepsilon_n} - u_{\varepsilon_m}\|_{W^{2,p}(\Omega')} \le C(\|u_{\varepsilon_n} - u_{\varepsilon_m}\|_{L^p(\Omega'')} + \|f_{\varepsilon_n} - f_{\varepsilon_m}\|_{L^p(\Omega'')}),$$

where $f_{\varepsilon_n}=f+\langle b_{\varepsilon_n},Du_{\varepsilon_n}\rangle+c_{\varepsilon_n}'u_{\varepsilon_n}$ and C is a suitable positive constant independent of n,m. Since u_{ε_n} and f_{ε_n} converge to u and $f+\langle b,Du\rangle+cu$, respectively, in $L^p(\Omega\cap B(R+1))$, u_{ε_n} converges to u in $W^{2,p}(\Omega')$. Therefore, by continuity $\partial u_{\varepsilon_n}/\partial \nu$ converges to $\partial u/\partial \nu$ on $\partial \Omega'$ and, consequently, $\partial u/\partial \nu=0$ on $\partial \Omega'$ as well. From the arbitrariness of Ω' , it follows that $\partial u/\partial \nu=0$ on $\partial \Omega$. Hence, the operator $\lambda I-B_p$ is surjective for any $\lambda>\lambda_0$. Since it is quasi-dissipative, due to (13.1.30), and since it has domain dense in $L^p(\Omega)$, according to the Lumer-Phillips theorem (see Theorem B.1.7), the operator B_p generates a strongly continuous semigroup $\{R(t)\}$ in $L^p(\Omega)$.

To show that $\{R(t)\}$ is analytic, it suffices to show that there exists a positive constant ς such that, for any $u \in D_p$,

$$\left| \operatorname{Im} \int_{\Omega} u^* B_p u \, dx \right| \le \varsigma \left(-\operatorname{Re} \int_{\Omega} u^* B_p u dx \right), \tag{13.1.32}$$

where $u^* = |u|^{p-2}\bar{u}$. Indeed, (13.1.32) implies that the numerical range of B_p is contained in the sector $\{\lambda \in \mathbb{C} : |\operatorname{Im} \lambda| \leq -\varsigma \operatorname{Re} \lambda\}$, and Theorem A.3.6 yields the analyticity of $\{R(t)\}$.

Integrating by parts, it is immediate to check that the real and imaginary parts of $\int_{\Omega} u^* B_p u \, dx$ are given, respectively, by (11.3.36) and (11.3.42), with q' instead of q, where q' is the bilinear form induced by the matrix Q'. Since Q and Q' differ only in a skew symmetric matrix S, it follows that

$$\operatorname{Re}(\overline{|u|^2}q'(Du,Du)) = \operatorname{Re}(|u|^2q(Du,Du))$$

and

$$\operatorname{Re}(\overline{u}^2 q'(Du, Du)) = \operatorname{Re}(\overline{u}^2 q(Du, Du)).$$

Hence, the real part of $\int_{\Omega} u^* B_p u \, dx$ can be estimated by

$$-\operatorname{Re} \int_{\Omega} u^* B_p u \, dx \ge \left(p - 1 - \frac{(\sigma')^2}{2\kappa_0} \right) F^2 + G^2 + \frac{1}{2} H^2, \tag{13.1.33}$$

where

$$F^{2} = \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Re}(\overline{u}Du)) dx,$$

$$G^{2} = \int_{\Omega} |u|^{p-4} q(\operatorname{Im}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx,$$

$$H^{2} = \int_{\Omega} |c'| |u|^{p} d\mu$$

(see the proof of Proposition 11.3.12 for further details).

As far as the imaginary part of the integral is concerned, we observe that

$$\operatorname{Im}(|u|^2 q'(Du, D\overline{u})) = q'(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) + q'(\operatorname{Im}(\overline{u}Du), \operatorname{Re}(\overline{u}Du))$$
$$= -2s(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)),$$

where s denotes the bilinear form induced by the matrix S. Moreover,

$$\operatorname{Im}(q'(\overline{u}Du, \overline{u}Du)) = -q'(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) + q'(\operatorname{Im}(\overline{u}Du), \operatorname{Re}(\overline{u}Du))$$
$$= 2q(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)).$$

Therefore,

$$\operatorname{Im} \int_{\Omega} u^* B_p u \, dx = -p \int_{\Omega} |u|^{p-4} s(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx$$
$$+ (p-2) \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx$$
$$- \operatorname{Im} \int_{\Omega} |u|^{p-2} \overline{u} \langle b', Du \rangle \, dx,$$

that differs from (11.3.43) only in the integral containing the bilinear form s, that can be estimated by

$$\left| \int_{\Omega} |u|^{p-4} s(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx \right|$$

$$\leq \|S\|_{\infty} \int_{\Omega} |u|^{p-4} |\operatorname{Re}(\overline{u}Du)| |\operatorname{Im}(\overline{u}Du)| dx$$

$$\leq \|S\|_{\infty} \left(\int_{\Omega} |u|^{p-4} |\operatorname{Re}(\overline{u}Du)|^{2} dx \right)^{\frac{1}{2}} \left(\int_{\Omega} |u|^{p-4} |\operatorname{Im}(\overline{u}Du)|^{2} dx \right)^{\frac{1}{2}}$$

$$\leq \frac{1}{\kappa_{0}} \|S\|_{\infty} F G. \tag{13.1.34}$$

Using (11.3.44), we can write

$$\left| (p-2) \int_{\Omega} |u|^{p-4} q(\operatorname{Re}(\overline{u}Du), \operatorname{Im}(\overline{u}Du)) dx - \operatorname{Im} \int_{\Omega} |u|^{p-2} \overline{u} \langle b', Du \rangle dx \right|$$

$$\leq (p-2)FG + \frac{\sigma'}{\sqrt{\kappa_0}} GH.$$
(13.1.35)

Combining (13.1.34) and (13.1.35), we get

$$\left| \operatorname{Im} \int_{\Omega} u^* B_p u \, dx \right| \le \left(p - 2 + \frac{p}{\kappa_0} \|S\|_{\infty} \right) FG + \frac{\sigma'}{\sqrt{\kappa_0}} GH. \tag{13.1.36}$$

From (13.1.33) and (13.1.36) we obtain (13.1.32) provided we choose $\varsigma \ge 1$ large enough. The proof is now complete.

We now study the relation between the semigroup $\{T_p(t)\}\$ and the Cauchy-Neumann problem (13.0.2).

Proposition 13.1.9 For any p > N and $f \in C_c^{2+\alpha}(\overline{\Omega})$ such that $\partial f/\partial \nu = 0$ on $\partial \Omega$, the function $u = T_p(\cdot)f$ is the unique bounded classical solution to the problem (13.0.2). Moreover, $u \in C_{\text{loc}}^{1+\alpha/2,2+\alpha}([0,+\infty) \times \overline{\Omega})$ and Du belongs to $C_b([0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$ for any T > 0.

Proof. Let f be as in the statement of the proposition. As it is easily seen, $f \in D_p$ (see (13.1.22)). Therefore, the density of D_p in $L^p(\Omega)$ implies that the function u is continuous from $[0, +\infty)$ in D_p (see Lemma 13.1.5, Theorem B.2.2 and Proposition B.2.5) and, hence, in $W^{2,p}(\Omega)$. Since p > N, according to the Sobolev embedding theorems (see [2, Theorem 5.4]), $u \in C([0, +\infty); C_b^1(\overline{\Omega}))$. In particular, u and Du are continuous functions in $[0, +\infty) \times \overline{\Omega}$ and they are bounded in $[0, T] \times \overline{\Omega}$ for any T > 0.

Let us now prove that $u \in C^{1,2}((0,T] \times \Omega)$ for any T > 0. For this purpose, we observe that, since $\{T_p(t)\}$ is an analytic semigroup, $u \in C^1([\varepsilon,T];D_p)$ for

any $0 < \varepsilon < T$. Using again the inclusion $D_p \subseteq W^{2,p}(\Omega)$ and the Sobolev embedding theorems, we find that $D_t u \in C([\varepsilon, T] \times \overline{\Omega})$ and, by the arbitrariness of ε , we have $D_t u \in C((0, T] \times \overline{\Omega})$.

In order to show that the second-order space derivatives of u are continuous in $(0,T]\times\Omega$, we consider two bounded open sets Ω_1 and Ω_2 , contained in Ω such that $\mathrm{dist}(\Omega_1,\Omega\setminus\Omega_2)>0$. For any fixed $t,t_0\in[\varepsilon,T]$, the function $v=u(t,\cdot)-u(t_0,\cdot)$ solves the equation

$$\sum_{i,j=1}^{N} q_{ij} D_{ij} v = T_p(t) \mathcal{A}f - T_p(t_0) \mathcal{A}f - \langle b, Dv \rangle - cv := g(t, t_0, \cdot),$$

where the right-hand side belongs to $W^{1,p}(\Omega_2)$. By the classical L^p -regularity theory (see Theorem C.1.2), v belongs to $W^{3,p}(\Omega_1)$ and

$$||u(t,\cdot) - u(t_0,\cdot)||_{W^{3,p}(\Omega_1)}$$

$$\leq C_1 (||g(t,t_0,\cdot)||_{W^{1,p}(\Omega_2)} + ||u(t,\cdot) - u(t_0,\cdot)||_{W^{1,p}(\Omega_2)})$$

$$\leq C_2 (||T(t)\mathcal{A}f - T(t_0)\mathcal{A}f||_{W^{1,p}(\Omega)} + ||u(t,\cdot) - u(t_0,\cdot)||_{W^{2,p}(\Omega)}),$$

for some positive constants C_1 and C_2 , independent of t, t_0 . Since the map $t \mapsto T(t)\mathcal{A}f$ is continuous in $(0, +\infty)$ with values in D_p , then, in particular, it is continuous in $(0, +\infty)$ with values in $W^{1,p}(\Omega)$. Since u is continuous in $(0, +\infty)$ with values in $W^{2,p}(\Omega)$, $u \in C((0, +\infty); W^{3,p}(\Omega_1))$. Therefore, using once more the Sobolev embedding theorems, we deduce that $u \in C([\varepsilon, T]; C^2(\overline{\Omega}))$, which implies that $Du, D^2u \in C((0, +\infty) \times \overline{\Omega})$, by the arbitrariness of ε, T and Ω . Thus, we have established that $u \in C^{1,2}((0, +\infty) \times \overline{\Omega})$, and, consequently, that u is a bounded classical solution to the Cauchy-Neumann problem (13.0.2).

To show that $u \in C^{1+\alpha/2,2+\alpha}_{loc}([0,+\infty)\times\overline{\Omega})$ and $Du \in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty)\times\overline{\Omega})$, let $\{\Omega_n\}$ be a sequence of smooth and bounded open sets such that

(i)
$$\Omega_n \subset \Omega_{n+1}$$
, $n \in \mathbb{N}$, (ii) $\partial \Omega \subset \bigcup_{n \in \mathbb{N}} \partial \Omega_n$, (iii) $\Omega = \bigcup_{n \in \mathbb{N}} \Omega_n$.

Denote by $\{T_n(t)\}$ the analytic semigroup generated by the realization A_n of the operator \mathcal{A} with homogeneous Neumann boundary conditions in $C(\overline{\Omega}_n)$ (see Theorem C.3.6(v)). As it is easily seen, if n is sufficiently large, then $f \in D_A(\alpha+1,\infty) = \{u \in C^{2+\alpha}(\overline{\Omega}_n) : \partial u/\partial \nu = 0 \text{ on } \partial \Omega_n\}$ (see Theorem C.3.6). This implies that for such values of n, the function $u_n = T_n(\cdot)f$ belongs to $C_{\text{loc}}^{1+\alpha/2,2+\alpha}([0,+\infty)\times\overline{\Omega}_n)$ and $Du_n \in C_{\text{loc}}^{1+\alpha/2,2+\alpha}((0,+\infty)\times\Omega_n)$. Moreover, by the classical maximum principle, it follows that

$$||u_n||_{\infty} \le ||f||_{\infty}, \qquad n \in \mathbb{N}. \tag{13.1.37}$$

Now, we fix $k \in \mathbb{N}$ such that supp $f \subset \overline{\Omega}_k$, and two bounded smooth open subsets $\Omega' \subset \Omega'' \subset C$, such that $\operatorname{dist}(\Omega', \Omega \setminus \Omega'') > 0$. By the classical

Schauder estimates in Theorems C.1.4, C.1.5 and by (13.1.37), we deduce that there exist two constants $C_1 = C_1(k)$ and $C_2 = C_2(\varepsilon, \Omega', \Omega'')$ such that

$$||u_n - u_m||_{C^{1+\alpha/2, 2+\alpha}([0,T] \times \overline{\Omega}_k)} \le C_1 ||u_n - u_m||_{C([0,T] \times \overline{\Omega}_{k+1})} \le 2C_1 ||f||_{\infty},$$

$$||Du_n - Du_m||_{C^{1+\alpha/2, 2+\alpha}([\varepsilon,T] \times \overline{\Omega}')} \le C_2 ||u_n - u_m||_{C([0,T] \times \overline{\Omega}'')} \le 2C_2 ||f||_{\infty},$$

for any n,m>k. By a compactness argument, we can determine a subsequence $\{u_{n_k}\}$ converging to a function $v\in C^{1+\alpha/2,2+\alpha}_{\mathrm{loc}}([0,+\infty)\times\overline{\Omega})$ in $C^{1,2}(F)$, for any compact set $F\subset [0,+\infty)\times\overline{\Omega}$, such that Du_{n_k} converges to $Dv\in C^{1+\alpha/2,2+\alpha}_{\mathrm{loc}}((0,+\infty)\times\Omega)$ in $C^{1,2}([\varepsilon,T]\times\Omega')$ for any $\varepsilon\in (0,T)$ and any $\Omega'\subset\subset\Omega$. As it is immediately seen, v is a classical solution to the problem (13.0.2) and, consequently, it coincides with u, by Theorem 12.1.5. We have so proved that u admits continuous third-order space derivatives in $(0,+\infty)\times\Omega$. This finishes the proof.

Proof of the existence of the classical bounded solution to (13.0.2)

We are now in position to prove the existence of the solution to the problem (13.0.2) together with the gradient estimate (13.1.6). For this purpose, as it has been already pointed out at the beginning of the section, we approximate the operator \mathcal{A} by operators satisfying the assumptions of Subsection 13.1.2, including the extra condition (13.1.4). To begin with, we prove the following approximation result which will be useful in the sequel.

Lemma 13.1.10 Let $f \in C_b(\overline{\Omega})$. Then, there exists a sequence of functions $\{f_n\} \subset C_c^{2+\alpha}(\overline{\Omega})$ with the following properties:

- (i) $||f_n||_{\infty} \leq C$, for some positive constant C and any $n \in \mathbb{N}$;
- (ii) $\frac{\partial f_n}{\partial \nu} = 0$ on $\partial \Omega$ for any $n \in \mathbb{N}$;
- (iii) f_n converges to f as n tends to $+\infty$, uniformly on compact sets of $\overline{\Omega}$.

Moreover, let $f \in C^1_{\nu}(\overline{\Omega})$. Then, there exists a sequence of functions $\{f_n\} \subset C^{2+\alpha}_c(\overline{\Omega})$ with the following properties:

- (i) $\partial f_n/\partial \nu = 0$ on $\partial \Omega$ for any $n \in \mathbb{N}$;
- (ii) $||f_n||_{C_b^k(\overline{\Omega})} \leq C||f||_{C_b^k(\overline{\Omega})}$ (k = 0, 1), for any $n \in \mathbb{N}$ and some constant C > 0, independent of n;
- (iii) f_n and Df_n converge to f and Df, respectively, as n tends to $+\infty$, uniformly on compact sets of $\overline{\Omega}$.

Proof. We first prove the second part of the statement, since we need it to show the first one. In order to prove the second part of the statement, it is sufficient to repeat, step by step, the same construction as in the proof of Lemma 13.1.5 and to observe that $\sum_{k=1}^{n} \vartheta_k$ and $\sum_{k=1}^{n} D\vartheta_k$ converge as r tends to $+\infty$, respectively to $\mathbf{1}$ and $\mathbf{0}$, uniformly on any compact set Ω' of $\overline{\Omega}$. Moreover, we can also assume that the $C^{2+\alpha}$ -norm of ϑ_k can be estimated by a constant independent of k.

We can now prove the first part of the lemma. For this purpose, we observe that if $\{T(t)\}$ denotes the semigroup associated with the Laplacian with homogeneous Neumann boundary condition in Ω , then, for any $n \in \mathbb{N}$, the function $g_n = T(1/n)f$ belongs to $C^2(\overline{\Omega})$, $\partial g_n/\partial \nu = 0$ on $\partial \Omega$ and $\|g_n\|_{\infty} \leq \|f\|_{\infty}$. Moreover, g_n converges to f as n tends to $+\infty$, uniformly on compact sets of $\overline{\Omega}$. Now, applying the second part of the lemma to any function g_n , we can conclude the proof.

Theorem 13.1.11 Under Hypotheses 13.1.1, for any $f \in C_b(\overline{\Omega})$, the Cauchy-Neumann problem (13.0.2) admits a unique bounded classical solution u. The function u belongs to $C_{\text{loc}}^{1+\alpha/2,2+\alpha}((0,+\infty)\times\overline{\Omega})$ and it satisfies the gradient estimate (13.1.6) with k=0. If, in addition, $f \in C_{\nu}^{1}(\overline{\Omega})$, then the estimate (13.1.6) holds with k=1.

Proof. The uniqueness of the bounded classical solution to the problem (13.0.2) follows immediately from Theorem 12.1.5. So, let us prove the existence part.

In the case when $f \in C_c^{2+\alpha}(\overline{\Omega})$ the proof follows repeating the same arguments as in Step 1 of the proof of Theorem 11.3.4. Hence, it is omitted.

Let us now consider the case when $f \in C_b(\overline{\Omega})$. Let $\{f_n\}$ be a sequence of functions in $C_c^{2+\alpha}(\overline{\Omega})$ with $\partial f_n/\partial \nu = 0$ on $\partial \Omega$ and assume that f_n converges to f uniformly on compact sets of $\overline{\Omega}$ and it satisfies $||f_n||_{\infty} \leq C||f||_{\infty}$ with C being independent of n (see Lemma 13.1.10). Denote by u_n the solution to the problem (13.0.2), with initial datum f_n . Arguing as in Step 2 of the proof of Theorem 11.3.4, we can easily show that, up to a subsequence, u_n converges in $C^{1,2}(F)$, for any compact set $F \subset (0, +\infty) \times \overline{\Omega}$, to a function $u \in C_{\text{loc}}^{1+\alpha/2,2+\alpha}((0,+\infty) \times \overline{\Omega})$ which solves the differential equation $D_t u - \mathcal{A}u = 0$ and such that $\partial u/\partial \nu = 0$ on $\partial \Omega$.

It remains to show that u is continuous at $(0, x_0)$ with value $f(x_0)$ for any $x_0 \in \overline{\Omega}$, in order to conclude that u is the bounded classical solution to (13.0.2). To this aim it suffices to argue as in the proof of Proposition 12.1.7, taking the gradient estimate (13.1.6) into account.

In the case when $f \in C^1_{\nu}(\overline{\Omega})$, we can assume that the sequence $\{f_n\}$ satisfies also $\|f_n\|_{C^1_b(\overline{\Omega})} \leq C\|f\|_{C^1_b(\overline{\Omega})}$. Then, letting n go to $+\infty$ in the following estimate

$$|Du_n(t,x)| \le C_T ||f_n||_{C_h^1(\overline{\Omega})} \le C_T' ||f||_{C_h^1(\overline{\Omega})}, \qquad 0 < t \le T, \ x \in \overline{\Omega},$$

we get (13.1.6) with k = 1.

Remark 13.1.12 In the case when $f \in C_c^{2+\alpha}(\overline{\Omega})$, the Schauder estimates in Proposition 13.1.9, that we used to prove the convergence of the sequence u_n , show that Du_n is bounded in $[0,T] \times \overline{\Omega}$ for any T>0 and it converges to Du in $[0,T] \times \overline{\Omega}'$ for any bounded open set $\Omega' \subset \Omega$. Therefore, the function Du is continuous in $[0,+\infty) \times \overline{\Omega}$ and it is bounded in $[0,T] \times \overline{\Omega}$ for any T>0. If $c \equiv 0$, then Du is bounded in $[0,+\infty) \times \overline{\Omega}$, since the estimate (13.1.6) with k=1 holds with $T=+\infty$ (see the end of the proof of Proposition 13.1.3).

Finally, by classical regularity results (see Theorem C.1.4), Du belongs to $C^{1,2}((0,+\infty)\times\Omega)$.

For any $f \in C_b(\overline{\Omega})$, any t > 0 and any $x \in \overline{\Omega}$, we set (T(t)f)(x) = u(t,x), where, according to Theorem 13.1.11, u is the unique bounded classical solution to the problem (13.0.2). Then, $\{T(t)\}$ is a semigroup of linear operators in $C_b(\overline{\Omega})$, such that $\|T(t)\|_{L(C_b(\overline{\Omega}))} \leq 1$ for any t > 0, as a consequence of Theorem 12.1.5. To conclude this section we state some remarkable properties of the semigroup $\{T(t)\}$, which we need in Subsection 13.3.

Theorem 13.1.13 Let $\{f_n\} \subset C_b(\overline{\Omega})$ be a bounded sequence converging pointwise to some function $f \in C_b(\overline{\Omega})$ as n tends to $+\infty$. Then, for any $0 < \varepsilon < T < +\infty$ and any bounded set $\Omega' \subset \overline{\Omega}$, $T(\cdot)f_n$ converges to $T(\cdot)f$ in $C^{1,2}([\varepsilon,T]\times\Omega')$. Further, if f_n converges to f locally uniformly in $\overline{\Omega}$, then $T(\cdot)f_n$ converges to $T(\cdot)f$ uniformly in $[0,T]\times\Omega'$.

As a consequence, for any t > 0 and $x \in \overline{\Omega}$ there exists a positive, Borel measure $p(t, x; \cdot)$ such that $p(t, x; \Omega) \leq 1$ and

$$(T(t)f)(x) = \int_{\Omega} f(y)p(t, x; dy), \qquad f \in C_b(\overline{\Omega}).$$
 (13.1.38)

Proof. It suffices to argue as in the proof of Proposition 12.1.7.

Let \widehat{A} be the weak generator of the semigroup $\{T(t)\}$ defined in Section 12.2. The following proposition characterizes the operator \widehat{A} .

Proposition 13.1.14 It holds that

$$\begin{cases}
D(\widehat{A}) = \left\{ u \in C_b(\overline{\Omega}) \cap \bigcap_{1 \le p < +\infty} W^{2,p}(\Omega \cap B(R)) & \text{for any } R > 0 : \\
Au \in C_b(\overline{\Omega}), & \frac{\partial u}{\partial \nu}(x) = 0 & \text{for any } x \in \partial \Omega \right\}, \\
\widehat{A}u = Au, & u \in D(\widehat{A}).
\end{cases}$$
(13.1.39)

In particular, A and T(t) commute on $D(\widehat{A})$ for any t > 0. Finally, $D(\widehat{A}) \subset C^1_{\nu}(\overline{\Omega})$ with a continuous embedding and, for any $\omega > c_0$, there exists a positive constant M_{ω} such that

$$||Du||_{\infty} \le M_{\omega} ||u||_{\infty}^{\frac{1}{2}} ||(\mathcal{A} - \omega) u||_{\infty}^{\frac{1}{2}},$$
 (13.1.40)

for any $u \in D(\widehat{A})$.

Proof. The proof is similar to that of Proposition 12.2.5. So we limit ourselves to sketching it, pointing out the main differences. To prove that, for any $f \in C_b(\overline{\Omega})$, the function

$$u(x) = R(\lambda, \widehat{A})f(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x)dt, \qquad x \in \overline{\Omega}$$

belongs to the space defined in the right-hand side of (13.1.39) and it solves the elliptic equation $\lambda u - \mathcal{A}u = f$, we use an approximation argument which slightly differs from that used in the proof of the quoted proposition. We first consider the case when $f \in C_c^{2+\alpha}(\overline{\Omega})$ is such that $\partial f/\partial \nu = 0$ on $\partial \Omega$. For any $n \in \mathbb{N}$, we introduce the operator $A^{(n)}$ with bounded coefficients belonging to $C^{\alpha}(\Omega \cap B(R))$ for any R > 0, defined by

$$A^{(n)}u = \sum_{i,j=1}^{N} q_{ij}D_{ij}u + \sum_{i=1}^{N} b_i^{(n)}D_iu + c^{(n)}u,$$

for any $u \in D(A^{(n)})$ where

$$D(A^{(n)}) = \left\{ u \in \bigcap_{p \ge 1} W^{2,p}(\Omega \cap B(R)) \text{ for any } R > 0 : \\ u, A^{(n)}u \in C_b(\overline{\Omega}), \ \partial u/\partial \nu = 0 \text{ on } \partial \Omega \right\}.$$

We assume that $b_i^{(n)} \equiv b_i$ and $c^{(n)} \equiv c$ in $\Omega \cap B(n)$ and $c^{(n)} \leq 0$ in $\overline{\Omega}$. According to Proposition C.3.3 (with $c_0 = 0$) and Theorem C.3.6(v), we deduce that the operator $A^{(n)}$ generates an analytic semigroup of contractions $\{T_n(t)\}$ in $C_b(\overline{\Omega})$. Arguing as in Step 1 of the proof of Theorem 11.3.4, it turns out that $T_n(t)f$ converges to T(t)f pointwise in $\overline{\Omega}$. Therefore, by dominated convergence, the sequence $R(\lambda, A^{(n)})f = \int_0^{+\infty} e^{-\lambda t} T_n(t) f dt$ converges to $u = R(\lambda, \widehat{A})f$ pointwise in $\overline{\Omega}$ and in $L^p(\Omega')$ for any bounded subset Ω' of Ω and any $p \in [1, +\infty)$. Now arguing as in the proof of Proposition 12.2.5, we can show that $u \in D(A)$.

In the general case, when $f \in C_b(\overline{\Omega})$, we consider a sequence of functions $\{f_n\} \subset C_c^{2+\alpha}(\overline{\Omega})$, bounded in $C_b(\overline{\Omega})$, with $\partial f_n/\partial \nu = 0$ on $\partial \Omega$, and converging to f pointwise in $\overline{\Omega}$ (see Lemma 13.1.10). Using the representation formula

(13.1.38), we can show that $(T(t)f_n)(x)$ converges to (T(t)f)(x) for any t > 0 and any $x \in \overline{\Omega}$. Since $||T(t)f_n||_{\infty} \leq ||f_n||_{\infty} \leq C$, C being a positive constant independent of n, the dominated convergence theorem implies that $R(\lambda, \widehat{A})f_n$ tends to $R(\lambda, \widehat{A})f$ in $L^p(\Omega \cap B(R))$ for any R > 0 and any $p \in [1, +\infty)$. Now, arguing as in the previous step, we can show that $R(\lambda, \widehat{A})f$ belongs to the space defined by the right-hand side of (13.1.39) and solves the differential equation $\lambda u - Au = f$. This completes the first part of the proof which shows that the realization A of the operator A with domain given by the right-hand side of (13.1.39) is an extension of \widehat{A} .

To show that $D(A) \subset D(\widehat{A})$ and to prove (13.1.40) it suffices to argue as in the proof of Proposition 12.2.5. For this purpose, we remark that Proposition 12.2.4, which is the main ingredient to prove the inclusion $D(A) \subset D(\widehat{A})$, holds also when Ω is nonconvex.

13.2 The case when Ω is an exterior domain

In this section we deal with the case when $\Omega \subset \mathbb{R}^N$ is an open subset of \mathbb{R}^N such that $\mathbb{R}^N \setminus \Omega$ is bounded and has uniformly $C^{2+\alpha}$ -smooth boundary. Our aim consists in extending the gradient estimate (13.1.6) to some cases in which the diffusion coefficients q_{ij} $(i,j=1,\ldots,N)$ are unbounded in Ω . Let us state the hypotheses that we always assume in this section.

Hypotheses 13.2.1 (i) Hypotheses 13.1.1(i), 13.1.1(v), 13.1.1(vi) are satisfied;

- (ii) the functions $q_{ij} \equiv q_{ji}$, b_j and c (i, = 1, ..., N) enjoy the same regularity assumptions as in Hypothesis 13.1.1(ii);
- (iii) there exist constants $\beta_0 > 0$ and $r \in [0, \sqrt{2}/N)$ and a function $\kappa : \mathbb{R}^N \to \mathbb{R}$ such that

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \qquad x \in \Omega, \quad \xi \in \mathbb{R}^N, \quad 0 < \kappa_0 := \inf_{x \in \overline{\Omega}} \kappa(x);$$

$$|Dq_{ij}(x)| \le \beta_0 \kappa(x) + r|c(x)|^{\frac{1}{2}} (\kappa(x))^{\frac{1}{2}}, \qquad x \in \Omega, \quad i, j = 1, \dots, N;$$

(iv) there exists a positive constant k_1 such that

$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \le k_1 |\xi|^2, \quad x \in \Omega, \quad \xi \in \mathbb{R}^N.$$

Set $\Omega_{\rho} = \Omega \setminus \Gamma(\rho)$, where $\Gamma(\rho)$ has been defined in (13.1.1) and fix, now and for the rest of this section, a real number $\overline{R} > 0$ such that

$$\mathbb{R}^N \setminus \Omega_\rho \subseteq B(\overline{R}).$$

For any $n \in \mathbb{N}$ such that $n > \overline{R}$, let us introduce a nondecreasing odd function $\psi_n \in C_b^2(\mathbb{R})$ such that

- (i) $\psi_n(x) = x$ for any $0 \le x \le n$;
- (ii) $\psi_n(x) = n + \frac{1}{2} \text{ for any } x \ge n + 1;$
- (iii) $0 \le \psi'_n(x) \le 1$ for any $x \in \mathbb{R}$.

Then, we set $\Psi_n(x) = (\psi_n(x_1), \dots, \psi_n(x_N))$ for any $x = (x_1, \dots, x_N) \in \mathbb{R}^N$, and we define the functions $q_{ij}^{(n)}$, $c^{(n)}$ by

$$q_{ij}^{(n)}(x) = q_{ij}(\Psi_n(x)), \qquad c^{(n)}(x) = c(\Psi_n(x)), \qquad x \in \Omega.$$

Finally, we introduce the operator $\mathcal{A}^{(n)}$ defined on smooth functions by

$$\mathcal{A}^{(n)}u(x) = \sum_{i,j=1}^{N} q_{ij}^{(n)}(x)D_{ij}u(x) + \sum_{i=1}^{N} b_i(x)D_iu(x) + c^{(n)}(x)u(x), \qquad x \in \Omega.$$
(13.2.1)

We observe that, by construction, $q_{ij}^{(n)}, c^{(n)}$ are bounded in $\overline{\Omega}$ together with their first-order derivatives. Moreover, $q_{ij}^{(n)}, c^{(n)} \in C^{1+\alpha}(\Omega \cap B(R))$, for any R > 0 and they satisfy Hypotheses 13.1.1(iii) and 13.2.1(ii), with $\kappa(x)$ replaced with $\kappa^{(n)}(x) = \nu(\Psi_n(x))$ and with the same constants κ_0, β_0, r . We note that the functions $b_i^{(n)}(x) = b_i(\Psi_n(x))$, $i = 1, \ldots, N$, do not preserve Hypothesis 13.2.1(ii), in general. This is the reason why we do not approximate the drift.

We assume that $\mathcal{A}^{(n)}$, $n > \overline{R}$, satisfies the following additional hypothesis.

Hypothesis 13.2.2 There exist a positive function $\varphi_n \in C^2(\overline{\Omega})$ and a positive constant $\lambda_0^{(n)} > 0$ such that

$$\lim_{|x|\to+\infty}\varphi_n(x)=+\infty, \qquad \sup_{\Omega}\left(\mathcal{A}^{(n)}\varphi_n-\lambda_0^{(n)}\varphi_n\right)<+\infty,$$

$$\frac{\partial \varphi_n}{\partial \nu}(x) \ge 0, \quad x \in \partial \Omega.$$

Remark 13.2.3 A case in which both the Hypotheses 13.1.1(vi) and 13.2.2 are satisfied is when there exist two constants $\lambda_0 > 0$ and $C \in \mathbb{R}$ such that

$$2\operatorname{Tr}(Q(x)) + 2\langle F(x), x \rangle + (c(x) - \lambda_0)(1 + |x|^2) \le C,$$

$$2\langle F(x), x \rangle - \lambda_0 (1 + |x|^2) \le C,$$

in a neighborhood of ∞ . Indeed, in such a case the function φ defined by

$$\varphi(x) = \psi(x) + (1 - \psi(x))(1 + |x|^2), \qquad x \in \overline{\Omega},$$

where $\psi \in C_c^2(B(\overline{R}+1))$ is such that $\psi = 1$ in $B(\overline{R})$, is a Lyapunov function for both the operators \mathcal{A}^n $(n \in \mathbb{N})$ and \mathcal{A} .

At this point, by applying to the operator $\mathcal{A}^{(n)}$, $n > \overline{R}$, the results of the previous section and, in particular, Theorem 13.1.11, we deduce that, for any $f \in C_b(\overline{\Omega})$, the parabolic problem

$$\begin{cases} D_t u(t,x) - \mathcal{A}^{(n)} u(t,x) = 0, & t > 0, x \in \Omega, \\ \frac{\partial u}{\partial \nu}(t,x) = 0, & t > 0, x \in \partial\Omega, \\ u(0,x) = f(x), & x \in \Omega, \end{cases}$$
(13.2.2)

admits a unique bounded classical solution $u_n \in C^{1+\alpha/2,2+\alpha}_{loc}((0,+\infty) \times \overline{\Omega})$ and for any $T \in (0,+\infty)$ there exists $C^{(n)} > 0$ such that

$$|Du_n(t,x)| \le C^{(n)} t^{-\frac{1-k}{2}} ||f||_{C_b^k(\overline{\Omega})}, \qquad t \in (0,T], \ x \in \overline{\Omega}, \ k = 0,1,$$

for any $f \in C_b^k(\overline{\Omega})$, satisfying $\partial f/\partial \nu = 0$ on $\partial \Omega$ if k=1. We observe that the dependence of $C^{(n)}$ on n relies on the fact that, according to Proposition 13.1.3, such a constant depends on the C^1 -norms of the diffusion coefficients $q_{ij}^{(n)}$ $(i,j=1,\ldots,N)$, which in fact depend on n and blow up as n tends to $+\infty$. Now, we want to show that, in the framework of this section, it is possible to modify a little bit the computations in the proof of Proposition 13.1.3 in order to make the constant C in (13.1.6), independent of n. Once this step is done, a standard approximation argument will prove that the sequence $\{u_n\}$ converges to the unique bounded classical solution to the problem (13.0.2) which satisfies (13.1.6).

In order to carry out our program, we observe that, by construction, $q_{ij}^{(n)}(x) = q_{ij}(x)$ (i, j = 1, ..., N) and $c^{(n)}(x) = V(x)$ if $n > \overline{R}$ and $x \in \Gamma(\rho)$. Hence, we have

$$(i) - \left((\mathcal{A}^{(n)} - c^{(n)})d \right)(x) = -\left((\mathcal{A} - c)d \right)(x) \le M_1, \quad x \in \Gamma(\rho), \ n > \overline{R},$$

(ii)
$$|Q^{(n)}(x)Dd(x)| = |Q(x)Dd(x)| \le M_2, \quad x \in \Gamma(\rho), \ n > \overline{R},$$
 (13.2.3)

where $M_1 \in \mathbb{R}$ and $M_2 \geq 0$ depend on the sup-norm of q_{ij}, b_i, c (i, j = 1, ..., N) on the compact set $\overline{\Gamma(\rho)}$. We use such conditions in the next proposition.

Proposition 13.2.4 Let u be the bounded classical solution to the parabolic problem (13.2.2), where $\mathcal{A}^{(n)}$ is defined by (13.2.1). Then, u verifies (13.1.6), with a constant C_T independent of n.

Proof. We limit ourselves to proving the statement in the case when u satisfies the regularity assumptions of Proposition 13.1.3. The general case then will follow arguing as in Theorem 13.1.11. Moreover, since the proof we provide is similar to that of Proposition 13.1.3, we just sketch it, pointing out the main differences in the case when k = 0 in (13.1.6).

Fix $n \in \mathbb{N}$ and introduce the function $v: [0, +\infty) \times \overline{\Omega} \to \mathbb{R}$ defined by

$$v(t,x) = |u(t,x)|^2 + a t m(x) |Du(t,x)|^2, t \ge 0, x \in \overline{\Omega}.$$

Since v satisfies the boundary condition $\partial v/\partial \nu \leq 0$ on $\partial \Omega$, we have only to show that $D_t v - \mathcal{A}^{(n)} v \leq M v$, for some a, M > 0 independent of n. Recalling that $D_t v - \mathcal{A}^{(n)} v$ is given by (13.1.13) with \mathcal{A} replaced with $\mathcal{A}^{(n)}$ and observing that $q_{ij}^{(n)}$, $c^{(n)}$ satisfy Hypothesis 13.2.1(iii), with κ being replaced with $\kappa^{(n)}$, we get

$$-m\operatorname{Tr}(Q^{(n)}D^2uD^2u) \le -m\kappa^{(n)}|D^2u|^2, \tag{13.2.4}$$

$$\begin{split} m \sum_{i,j,k=1}^{N} D_{k} q_{ij}^{(n)} D_{k} u D_{ij} u &\leq N m \left(\beta_{0} \kappa^{(n)} + r |c^{(n)}|^{\frac{1}{2}} \left(\kappa^{(n)} \right)^{\frac{1}{2}} \right) |D u| |D^{2} u| \\ &\leq \frac{N^{2} \beta_{0}^{2}}{4 \varepsilon} m \kappa^{(n)} |D u|^{2} + \varepsilon m \kappa^{(n)} |D^{2} u|^{2} \\ &- \tilde{\varepsilon} N^{2} r^{2} m c^{(n)} |D u|^{2} + \frac{1}{4 \tilde{\varepsilon}} m \kappa^{(n)} |D^{2} u|^{2}, (13.2.5) \end{split}$$

for any $\varepsilon, \tilde{\varepsilon} \in (0,1)$. From Hypotheses 13.2.1(iv) it follows that

$$m \sum_{i,j=1}^{N} D_i b_j D_i u D_j u \le k_1 m |Du|^2, \tag{13.2.6}$$

and from Hypothesis 13.1.1(iii), which is satisfied also by $c^{(n)}$ with the same constant γ , we get, for arbitrary $\delta > 0$,

$$m u \langle Dc^{(n)}, Du \rangle \le \gamma \delta m (1 - c^{(n)}) |Du|^2 + \frac{\gamma}{4\delta} m (1 - c^{(n)}) u^2.$$

Using (13.1.9) and (13.2.3)(ii) it follows that

$$-\frac{\omega_0}{2}\psi''(d(\cdot))\sum_{i,j=1}^{N} q_{ij}^{(n)} D_i d(\cdot) D_j d(\cdot) |Du|^2 \le \frac{\omega_0}{\varepsilon_0} M_2 |Du|^2.$$
 (13.2.7)

Analogously, by (13.1.8) and (13.2.3)(i) we obtain

$$-\frac{\omega_0}{2}\psi'(d(\cdot))(\mathcal{A}^{(n)}d(\cdot) - c^{(n)}d(\cdot))|Du|^2 \le \frac{\omega_0}{2}M_1|Du|^2.$$
 (13.2.8)

Finally, for any $\varepsilon > 0$,

$$-2\omega_0 \psi'(d(\cdot)) \langle Q^{(n)} Dd, D^2 u Du \rangle \le 2\omega_0 M_2 |D^2 u| |D u|$$

$$\le \omega_0^2 M_2^2 \varepsilon^{-1} |D u|^2 + \varepsilon |D^2 u|^2. \quad (13.2.9)$$

From the estimates (13.2.4)-(13.2.9) and recalling that m satisfies (13.1.7), we get

$$D_{t}v_{n} - \mathcal{A}^{(n)}v_{n} \leq \left\{ a \left(1 + \frac{\omega_{0}\varepsilon_{0}}{2} \right) \right.$$

$$\times \left[1 + 2T \left(k_{1} + \gamma\delta + \frac{\omega_{0}M_{2}}{\varepsilon_{0}} + \frac{\omega_{0}M_{1}}{2} + \frac{\omega_{0}^{2}M_{2}^{2}}{\varepsilon} \right) \right]$$

$$+ \kappa^{(n)} \left[-2 + \left(1 + \frac{\omega_{0}\varepsilon_{0}}{2} \right) \frac{aTN^{2}}{2\varepsilon} \beta_{0}^{2} \right] \right\} |Du_{n}|^{2}$$

$$- at \left(2\tilde{\varepsilon}N^{2}r^{2} + 2\gamma\delta - 1 \right) m c^{(n)} |Du_{n}|^{2}$$

$$+ \left(1 + \frac{\omega_{0}\varepsilon_{0}}{2} \right) \frac{aT\gamma}{2\delta} u_{n}^{2} + \left[1 - \left(1 + \frac{\omega_{0}\varepsilon_{0}}{2} \right) \frac{aT\gamma}{2\delta} \right] c^{(n)} u_{n}^{2}$$

$$+ at \left(-2 + 2\varepsilon + \frac{1}{2\tilde{\varepsilon}} + 2\frac{\varepsilon}{\kappa_{0}} \right) m \kappa^{(n)} |D^{2}u_{n}|^{2}, \quad (13.2.10)$$

in $(0,T] \times \Omega$ for any arbitrarily fixed T>0. Since $r\in [0,\sqrt{2}/N)$ we can choose $\tilde{\varepsilon}\in (1/4,1)$ such that $-1+2\tilde{\varepsilon}N^2r^2>0$. Then, we choose a sufficiently small value of $\varepsilon>0$ in order to make the last term in the right-hand side of (13.2.10) vanish. At this point, we fix $\delta>0$ such that also the third term vanishes and, then, a>0 such that

$$-1 + aTN^2 \left(1 + \frac{\omega_0 \varepsilon_0}{2}\right) \frac{\beta_0^2}{4\varepsilon} < 0$$

and

$$-1 + aT\left(1 + \frac{\omega_0 \varepsilon_0}{2}\right) \frac{\gamma}{2\delta} < 0.$$

With these choices of the parameters we obtain

$$\begin{split} D_t v_n - \mathcal{A}^{(n)} v_n \\ &\leq \left[a \left(1 + \frac{\omega_0 \varepsilon_0}{2} \right) + a T (2 + \omega_0 \varepsilon_0) \left(k_1 + \gamma \delta + \frac{\omega_0}{\varepsilon_0} M_2 + \frac{\omega_0}{2} M_1 + \frac{\omega_0^2 M_2^2}{\varepsilon} \right) \right. \\ &+ \kappa_0 \left(1 + \frac{\omega_0 \varepsilon_0}{2} \right) \left(-2 + a T \frac{N^2 \beta_0^2}{2\varepsilon} \right) \left. \right] |Du_n|^2 + \left(1 + \frac{\omega_0 \varepsilon_0}{2} \right) \frac{a T \gamma}{2\delta} u_n^2. \end{split}$$

Choosing a smaller value of a, if necessary, we finally obtain $D_t v_n - \mathcal{A}^{(n)} v_n \leq M v_n$ in $(0,T] \times \Omega$, where $M = aT(2+\omega_0 \varepsilon_0)\gamma/(4\delta)$. Note that all the constants

involved are independent of n. Hence, the conclusion follows as in Proposition 13.1.3.

We are now in position to prove the main theorem of the section.

Theorem 13.2.5 For any $f \in C_b(\overline{\Omega})$ the Cauchy problem (13.0.2) admits a unique bounded classical solution u, which also belongs to $C_{\text{loc}}^{1+\alpha/2,2+\alpha}((0,+\infty)\times\overline{\Omega})$. Moreover, the estimate (13.1.6) holds true with k=0. Finally, if $f \in C_{\nu}^{1}(\overline{\Omega})$, then the estimate (13.1.6) holds also with k=1.

Proof. We limit ourselves to proving the assertion when $f \in C_b(\overline{\Omega})$, since the other case follows straightforward from this one.

For any fixed $f \in C_b(\overline{\Omega})$ and any $n > \overline{R}$, let u_n be the unique bounded classical solution to the problem (13.2.2). As it has been already pointed out, $u_n \in C_{\text{loc}}^{1+\alpha/2,2+\alpha}((0,+\infty) \times \overline{\Omega})$. Moreover, Theorem 12.1.5 yields $||u_n||_{\infty} \le ||f||_{\infty}$ and u_n satisfies the gradient estimate

$$|Du_n(t,x)| \le C_T t^{-\frac{1-k}{2}} ||f||_{C_b(\overline{\Omega})}, \qquad t \in (0,T], \quad x \in \overline{\Omega}, \quad (13.2.11)$$

for any T > 0 and some positive constant C_T , independent of n, as a consequence of Proposition 13.2.4.

In order to prove that there exists a subsequence of $\{u_n\}$, which converges to the bounded classical solution to the problem (13.0.2) in $C^{1,2}([\varepsilon,T]\times\Omega')$ for any $\varepsilon\in(0,T)$ and any $\Omega'\subset\overline{\Omega}$, we argue as in Theorem 13.1.11. Hence, we may skip in this context some details. First, we show that a subsequence $\{u_{n_k}\}$ converges to a solution u to the differential equation in (13.0.2), which satisfies the boundary condition $\partial u/\partial\nu=0$ on $\partial\Omega$. For this purpose, we apply the Schauder estimates in Theorem C.1.5 to the function u_n , observing that the constants which appear can be taken independent of n, since for any bounded set $\widetilde{\Omega}\subset\Omega$, one has $\mathcal{A}^{(n)}=\mathcal{A}$ in $\widetilde{\Omega}$, if n is sufficiently large. Then, to prove the continuity of u at t=0, we use a localization argument and the gradient estimate (13.2.11). For the same reason as above, also in this case all the constants involved can be taken independent of n.

Remark 13.2.6 We remark that all the results in Remark 13.1.12, Theorem 13.1.13 and Proposition 13.1.14 can be extended to the case when Ω is an exterior domain. The proof of Remark 13.1.12 follows from the arguments in the proof of Theorem 13.2.5, whereas the proof of Theorem 13.1.13 can be obtained just as in the case when the q_{ij} 's are bounded (i, j = 1, ..., N). In the proof of Proposition 13.1.14 only a slight modification is needed. Since now the coefficients q_{ij} need not to be bounded, one has to change the definition of the approximating operators $A^{(n)}$ replacing the q_{ij} 's with a sequence of bounded coefficients $q_{ij}^{(n)}$ such that the operator $A^{(n)}$ is still uniformly elliptic in Ω and $q_{ij}^{(n)} = q_{ij}$ in $\Omega \cap B(n)$. Then, the proof can be obtained just repeating the same arguments as in the quoted proposition.

13.3 Pointwise gradient estimates and their consequences

In this section we are devoted to proving some pointwise gradient estimates for the function T(t)f. We start by proving the following estimate

$$|(DT(t)f)(x)|^p \le C_p e^{M_p t} (T(t)(|Df|^p))(x), \qquad t > 0, \quad x \in \Omega, \quad (13.3.1)$$

in the case when the diffusion coefficients are bounded. Here, M_p is a real constant that can be explicitly determined (see (13.3.3)). As it is immediately seen, $T(\cdot)\mathbf{1} \equiv \mathbf{1}$ is a necessary condition in order to have (13.3.1), hence we need to assume $c \equiv 0$.

Theorem 13.3.1 Let Hypotheses 13.1.1(i), 13.1.1(ii), 13.1.1(iv)-13.1.1(vi) be satisfied. Further, assume that $c \equiv 0$. Then, for any $p \in (1, +\infty)$, there exist two constants $C_p > 0$ and $M_p \in \mathbb{R}$, such that (13.3.1) holds for any $f \in C^1_{\nu}(\overline{\Omega})$.

Proof. The techniques used are similar to those in the proof of Proposition 12.3.1. The only differences are in the case when $p \in (1,2]$. Hence, we sketch the proof only in this case, pointing out the main differences. We assume that $f \in C_c^{2+\alpha}(\overline{\Omega})$ and $\partial f/\partial \nu = 0$ on $\partial \Omega$. The general case then follows by density by virtue of Lemma 13.1.10 and Theorem 13.1.13. We replace the function w defined in the proof of Proposition 12.3.1 with the new function w defined by

$$w(t,x) = (m(x)|Du(t,x)|^2 + \delta)^{\frac{p}{2}}, \qquad t > 0, \quad x \in \Omega,$$

where $u = T(\cdot)f$. We sketch the proof in the case when $p \in (1, 2]$, pointing out the main differences.

According to Remark 13.1.12, w is differentiable once with respect to the time variable and twice with respect to the space variables in $(0, +\infty) \times \Omega$ and it is bounded and continuous in $[0, +\infty) \times \overline{\Omega}$. Moreover,

$$D_t w - \mathcal{A}w = mf_1 + f_2 + f_3,$$

where f_1 is given by (12.3.3) (with $m|Du|^2 + \delta$ instead of $|Du|^2 + \delta$),

$$f_2 = p(m|Du|^2 + \delta)^{\frac{p}{2} - 1} \left(-\frac{1}{2}|Du|^2 \mathcal{A}m - 2\sum_{i,j,k=1}^{N} q_{ij}D_j m D_k u D_{ik} u \right),$$

$$f_3 = p(2-p)(m|Du|^2 + \delta)^{\frac{p}{2}-2} \sum_{i,j=1}^{N} q_{ij} \left(\frac{1}{2} |Du|^2 D_i m + m(D^2 u D u)_i \right) \times \left(\frac{1}{2} |Du|^2 D_j m + m(D^2 u D u)_j \right).$$

Using (13.1.14), (13.1.15) (with $||DQ||_{\infty}^{-1}\varepsilon$ instead of δ) and (13.1.16) we get

$$f_1 \le p \left(m|Du|^2 + \delta \right)^{\frac{p}{2} - 1}$$

$$\times \left\{ \left(\frac{1}{4\varepsilon} \|DQ\|_{\infty}^2 + k_1 \right) |Du|^2 + \varepsilon |D^2u|^2 - \text{Tr}(QD^2uD^2u) \right\}.$$

Since $Am = \omega_0 \psi'(d(\cdot)) Ad + \omega_0 \psi''(d(\cdot)) \langle Qd, d \rangle$, from (13.1.18), (13.1.19) and (13.1.20), with ε instead of δ , we get

$$f_{2} \leq p\omega_{0} \left(m|Du|^{2} + \delta \right)^{\frac{p}{2} - 1} \times \left\{ \|Q\|_{\infty} \left(\frac{1}{\varepsilon_{0}} + \frac{1}{2} \|D^{2}d\|_{\infty} + \frac{1}{2\varepsilon} + \frac{k_{2}}{2} \right) |Du|^{2} + 2\varepsilon \|Q\|_{\infty} |D^{2}u|^{2} \right\},\,$$

for any $\varepsilon > 0$. Finally, using the Cauchy-Schwarz inequality twice, first for the inner product induced by the matrix Q and, then, for the Euclidean inner product, and recalling that

$$(\alpha + \beta)^2 \le (1 + \varepsilon)\alpha^2 + \frac{1 + \varepsilon}{\varepsilon}\beta^2,$$

for any $\alpha, \beta, \varepsilon > 0$, we get

$$\sum_{i,j=1}^{N} q_{ij} \left(\frac{1}{2} |Du|^{2} D_{i} m + m(D^{2} u D u)_{i} \right) \left(\frac{1}{2} |Du|^{2} D_{j} m + m(D^{2} u D u)_{j} \right)$$

$$\leq \left[\frac{1}{2} \left(\langle QDm, Dm \rangle \right)^{\frac{1}{2}} |Du|^{2} + m \left(\text{Tr}(QD^{2} u D^{2} u) \right)^{\frac{1}{2}} |Du| \right]^{2}$$

$$\leq m |Du|^{2} \left[\frac{1}{2} \left(\langle QDm, Dm \rangle \right)^{\frac{1}{2}} |Du| + \left(m \text{Tr}(QD^{2} u D^{2} u) \right)^{\frac{1}{2}} \right]^{2}$$

$$\leq (m |Du|^{2} + \delta)$$

$$\times \left(m(1 + \varepsilon) \text{Tr}(QD^{2} u D^{2} u) + m \frac{1 + \varepsilon}{4\varepsilon} |Du|^{2} \langle QDm, Dm \rangle \right). \quad (13.3.2)$$

Therefore,

$$f_3 \le p(2-p)(m|Du|^2 + \delta)^{\frac{p}{2}-1} \times \left(m(1+\varepsilon)\operatorname{Tr}(QD^2uD^2u) + m\omega_0^2 \frac{1+\varepsilon}{4\varepsilon} \|Q\|_{\infty} |Du|^2 \right).$$

Now, choosing $\varepsilon > 0$ small enough to have $1 - (1 + \varepsilon)(2 - p) > 0$ from the previous estimate and (13.1.14) we get

$$\begin{split} &D_t w - \mathcal{A} w \\ &\leq p \left(m |Du|^2 + \delta \right)^{\frac{p}{2} - 1} \\ &\qquad \times \left\{ \left[\frac{\|DQ\|_{\infty}^2}{4\varepsilon} + k_1 + \frac{\omega_0}{\varepsilon_0} \|Q\|_{\infty} + \frac{\omega_0}{2} \|D^2 d\|_{\infty} \|Q\|_{\infty} + \frac{\omega_0}{2\varepsilon} \|Q\|_{\infty} \right. \\ &\qquad \qquad \left. + \frac{\omega_0}{2} k_2 - \frac{\omega_0^2 \varepsilon_0}{4 + 2\omega_0 \varepsilon_0} k_2^- + \frac{(1 + \varepsilon)(2 - p)}{4\varepsilon} \omega_0^2 \|Q\|_{\infty} \right] m |Du|^2 \\ &\qquad \qquad \left. + \left[\varepsilon + \frac{\varepsilon}{2} \omega_0 \varepsilon_0 + 2\omega_0 \varepsilon \|Q\|_{\infty} - (1 - (1 + \varepsilon)(2 - p))\kappa_0 \right] |D^2 u|^2 \right\}, \end{split}$$

where $k_2^- = \min\{0, k_2\}$. Let us now choose $\varepsilon > 0$ such that

$$\varepsilon + \frac{1}{2}\omega_0\varepsilon_0\varepsilon - \kappa_0 + (1+\varepsilon)(2-p)\kappa_0 + 2\varepsilon\omega_0\|Q\|_{\infty} = 0$$

and set

$$M_{p} = p \left\{ \frac{\|DQ\|_{\infty}^{2}}{4\varepsilon} + k_{1} + \frac{\omega_{0}}{\varepsilon_{0}} \|Q\|_{\infty} + \frac{\omega_{0}}{2} \|D^{2}d\|_{\infty} \|Q\|_{\infty} + \frac{\omega_{0}}{2\varepsilon} \|Q\|_{\infty} + \frac{\omega_{0}}{2} k_{2} - \frac{\omega_{0}^{2}\varepsilon_{0}}{4 + 2\omega_{0}\varepsilon_{0}} k_{2}^{-} + \frac{(1 + \varepsilon)(2 - p)}{4\varepsilon} \omega_{0}^{2} \|Q\|_{\infty} \right\}.$$

$$(13.3.3)$$

Therefore, we get $D_t w - \mathcal{A}w \leq M_p w - (M_p \wedge 0)\delta^{\frac{p}{2}}$. Arguing as in the proof of Theorem 7.1.5, taking the maximum principle in Theorem 12.1.5 into account, we deduce that

$$w(t,x) \le e^{M_p t} \left(T(t) \left((f^2 + m|Df|^2)^{\frac{p}{2}} \right) \right) (x),$$

for any t > 0 and any $x \in \overline{\Omega}$, which gives us (13.3.1) with $C_p = (1 + \omega_0 \varepsilon_0/2)^{p/2}$.

Now, we prove the second type of pointwise estimates.

Theorem 13.3.2 Let Hypotheses 13.1.1(i), 13.1.1(ii), 13.1.1(iv)-13.1.1(vi) be satisfied. Further, assume that $c \equiv 0$. Then, for any $p \in (1, +\infty)$ and any $f \in C_b(\overline{\Omega})$

$$|(DT(t)f)(x)|^p \le K_{p,t} t^{-\frac{p}{2}} (T(t)(|f|^p))(x), \qquad t \in (0, +\infty), \quad x \in \overline{\Omega},$$
(13.3.4)

where

$$K_{p,t} = \begin{cases} \frac{(2 + \omega_0 \varepsilon_0)^{\frac{p}{2}}}{[2p(p-1)\kappa_0]^{\frac{p}{2}}} \frac{M_p t}{1 - e^{-M_p t}}, & \text{if } p \in (1,2], \\ K_{2,t}^{\frac{p}{2}}, & \text{if } p > 2, \end{cases}$$

with M_p being given by (13.3.3). In the case when $M_p = 0$, the term $M_p t (1 - e^{-M_p t})^{-1}$ should be replaced with 1.

Proof. We limit ourselves to sketching the proof (13.3.4) in the case when $f \in C_c^{2+\alpha}(\overline{\Omega})$ is such that $\partial f/\partial \nu = 0$ on $\partial \Omega$, and $p \in (1,2]$, since it is similar to that of Proposition 12.3.3. The general case then will follow from an approximation argument (see Lemma 13.1.10) and the Jensen inequality.

For any $\delta > 0$, we introduce the function $\Phi : (0,t) \to C_b(\overline{\Omega})$ defined by

$$\Phi(s) = T(s) \Big((|T(t-s)f|^2 + \delta)^{\frac{p}{2}} \Big),$$

and we set, for any $s \in (0,t)$, $g(s) = (|T(t-s)f|^2 + \delta)^{p/2}$. The first step consists in showing that

$$\Phi'(s) \ge p(p-1)\kappa_0 T(s) \left((g(s))^{1-\frac{2}{p}} |DT(t-s)f|^2 \right),$$

for any $s \in (0,t)$. This can be done as in the proof of Proposition 12.3.3, after proving that $g(s) \in D(\widehat{A})$ for any $s \in (0,t)$, where $D(\widehat{A})$ is defined in Proposition 13.1.14. This is not immediate since now the operator \mathcal{A} has unbounded coefficients. By Remark 13.1.12, $g(s) \in C_b^1(\overline{\Omega}) \cap C^2(\Omega)$ and it satisfies $\partial g(s)/\partial \nu = 0$ on $\partial \Omega$. To conclude that $g(s) \in D(A)$, we just need to show that $\mathcal{A}g(s) \in C_b(\overline{\Omega})$. For this purpose, we observe that

$$\mathcal{A}g(s) = p(g(s))^{1-\frac{2}{p}} \left\{ T(t-s)f \, \mathcal{A}T(t-s)f + \langle QDT(t-s)f, DT(t-s)f \rangle \right\}$$
$$+ p(p-2)(g(s))^{1-\frac{4}{p}} |T(t-s)f|^2 \langle QDT(t-s)f, DT(t-s)f \rangle.$$

Since $f \in D(A)$ then, by Proposition 13.1.14, $\mathcal{A}T(t-s)f = T(t-s)\mathcal{A}f$ and, therefore, $\mathcal{A}g(s)$ is bounded and continuous in $\overline{\Omega}$. This implies that $\mathcal{A}\Phi(s) = T(s)\mathcal{A}g(s)$ for any $s \in (0,t)$.

The second step consists in showing that

$$\left(T(t)\left((|f|^2+\delta)^{\frac{p}{2}}\right)\right)(x)$$

$$\geq p(p-1)\kappa_0 \int_0^t \left(T(s)\left((g(s))^{1-\frac{2}{p}}|DT(t-s)f|^2\right)\right)(x)ds,$$

for any $x \in \overline{\Omega}$. So, arguing as in the proof of Proposition 12.3.3, we are led to showing that $\left(T(t-\varepsilon)\left((|T(\varepsilon)f|^2+\delta)^{\frac{p}{2}}\right)\right)(x)$ tends to $\left(T(t)\left((f^2+\delta)^{\frac{p}{2}}\right)\right)(x)$

as ε tends to 0, for any $x \in \overline{\Omega}$. This was immediate in the proof of the quoted proposition, since $\{T_n(t)\}$ was a strongly continuous semigroup, whereas, now, it is guaranteed by Theorem 13.1.13.

From now on, the same arguments in the proof of Proposition 12.3.3 yield to (13.3.4). Hence, we skip the details.

Now, we show that the pointwise estimate (13.3.4) can be extended to the case when Ω is an exterior domain. For this purpose, we apply the same technique as in Chapter 7, first proving that, for any $p \in (1, +\infty)$, there exists a constant $C_p > 0$ such that

$$|(DT(t)f)(x)|^p \le C_p(T(t)(f^2 + |Df|^2)^{\frac{p}{2}})(x), \qquad t > 0, \ x \in \overline{\Omega}, \quad (13.3.5)$$

for any $f \in C_b^1(\overline{\Omega})$ such that $\partial f/\partial \nu = 0$ on $\partial \Omega$. As above we still assume that $c \equiv 0$.

Theorem 13.3.3 Under Hypotheses 13.2.1 and 13.2.2 (where we take $c \equiv 0$) there exists a constant $C_p > 0$ such that (13.3.5) holds true.

Proof. Since the proof is similar to that of Theorem 13.3.1, we limit ourselves to sketching it, pointing out the main differences. In the case when $p \in (1,2]$ and $f \in C_c^{2+\alpha}(\overline{\Omega})$, we introduce, for any $\delta > 0$, the function w defined by

$$w(t,x) = \left(a|u(t,x)|^2 + m(x)|Du(t,x)|^2 + \delta\right)^{\frac{p}{2}}, \qquad t > 0, \quad x \in \Omega.$$
 (13.3.6)

Here a>0 is a real parameter to be fixed later on. By virtue of Remark 13.2.6, the function w is differentiable once with respect to the time variable and twice with respect to the space variables in $(0, +\infty) \times \Omega$ and it is continuous in $[0, +\infty) \times \overline{\Omega}$, due to Remark 13.2.6. Moreover, it satisfies the equation $D_t w - \mathcal{A} w = m f_1 + f_2 + f_3$, where f_1 is given by (12.3.11) (with w as in (13.3.6) and a being replaced by a/m),

$$f_2 = -p w^{1-\frac{2}{p}} \left(\frac{1}{2} |Du|^2 \mathcal{A}m + 2\langle QDm, D^2 uDu \rangle \right)$$

and

$$f_3 = \frac{p}{2} \left(1 - \frac{p}{2} \right) \langle Q(D(au^2 + m|Du|^2), D(au^2 + m|Du|^2) \rangle.$$

The function f_1 can be immediately estimated as in (13.2.5) and (13.2.6), where we take $c^{(n)} = 0$ and replace $\kappa^{(n)}$ with κ . Similarly, since

$$\mathcal{A}m = \omega_0 \psi'(d(\cdot)) \mathcal{A}d + \omega''(d(\cdot)) \langle Qd, d \rangle,$$

 f_2 can be estimated by (13.2.7) and (13.2.9). To estimate f_3 we observe that, arguing as in the proof of (13.3.2), we get

$$\langle Q(D(au^{2} + m|Du|^{2}), D(au^{2} + m|Du|^{2}) \rangle$$

$$\leq \left[(\langle QDm, Dm \rangle)^{\frac{1}{2}} |Du| + 2m \left(\text{Tr}(QD^{2}uD^{2}u) \right)^{\frac{1}{2}} |Du| \right]$$

$$+2a|u| (\langle QDu, Du \rangle)^{\frac{1}{2}}$$

$$\leq (a|u|^{2} + m|Du|^{2})$$

$$\times \left\{ m \left[(\langle QDm, Dm \rangle)^{\frac{1}{2}} |Du| + 2 \left(\text{Tr}(QD^{2}uD^{2}u) \right)^{\frac{1}{2}} \right]^{2} \right.$$

$$+4a\langle QDu, Du \rangle \right\}$$

$$\leq (a|u|^{2} + m|Du|^{2}) \left(4m(1 + \varepsilon) \text{Tr}(QD^{2}uD^{2}u) \right.$$

$$+ m \frac{1 + \varepsilon}{\varepsilon} |Du|^{2} \langle QDm, Dm \rangle + a \langle QDu, Du \rangle \right).$$

Hence, we obtain for $1 - (1 + \varepsilon)(2 - p) > 0$,

$$D_{t}w - \mathcal{A}w$$

$$\leq p \left(u^{2} + m|Du|^{2} + \delta\right)^{\frac{p}{2} - 1}$$

$$\times \left\{ \left[\left(\frac{N^{2}\beta_{0}^{2}}{4\varepsilon} + \frac{2a(1-p)}{2 + \omega_{0}\varepsilon_{0}} \right) \kappa + k_{1} + \frac{\omega_{0}}{2}M_{1} - \frac{\omega_{0}^{2}\varepsilon_{0}}{4 + 2\omega_{0}\varepsilon_{0}}M_{1}^{-} + \left(\frac{1}{\varepsilon_{0}} + \frac{1}{2\varepsilon} + \frac{(1+\varepsilon)(2-p)}{4\varepsilon}\omega_{0} \right) \omega_{0}M_{2} \right] m|Du|^{2} + \left[\left(1 + \frac{1}{2}\omega_{0}\varepsilon_{0} \right) \varepsilon + 2\varepsilon\omega_{0}M_{2}\kappa_{0}^{-1} - 1 + (1+\varepsilon)(2-p) \right] \kappa|D^{2}u|^{2} \right\}.$$

Now we can choose $\varepsilon = \varepsilon_1 > 0$ such that

$$\left(1 + \frac{1}{2}\omega_0\varepsilon_0\right)\varepsilon_1 - 1 + (1 + \varepsilon_1)(2 - p) + 2\varepsilon_1\omega_0M_2\kappa_0^{-1} = 0. \quad (13.3.7)$$

Then, we fix a sufficiently large such that

$$L_a := \frac{N^2 \beta_0^2}{4\varepsilon_1} + \frac{2a(1-p)}{2 + \omega_0 \varepsilon_0} < 0$$

and

$$L_a \kappa_0 + k_1 + \frac{\omega_0}{2} M_1 - \frac{\omega_0^2 \varepsilon_0}{4 + 2\omega_0 \varepsilon_0} M_1^- + \left(\frac{1}{\varepsilon_0} + \frac{1}{2\varepsilon_1} + \frac{(1 + \varepsilon_1)(2 - p)}{4\varepsilon_1} \omega_0\right) \omega_0 M_2 \le 0.$$

With this choice of ε_1 and a, we easily see that

$$D_t w - \mathcal{A} w \leq 0.$$

Therefore, using the same technique as in the proof of Theorem 13.3.1, we finally get (13.3.5) in the case when $p \in (1, 2]$, with

$$C_p = \max \left\{ a^{p/2}, \left(1 + \frac{\omega_0 \varepsilon_0}{2}\right)^{\frac{p}{2}} \right\}.$$

The other cases follow just as in the proof of the quoted theorem.

The estimates (13.3.5) can be improved in the case when the dissipative condition is satisfied with the constant k_1 replaced with a function (still denoted by k_1) which, roughly speaking, balance the growth of the coefficients at infinity. In such a situation, we can recover the results in Theorem 13.3.1 and, in some particular cases, we can allow the exponential term appearing in the estimate of DT(t)f to be of negative type. We reformulate Hypotheses 13.2.1 and 13.2.2 as follows:

Hypotheses 13.3.4 (i) the coefficients $q_{ij} = q_{ji}$ (i, j = 1, ..., N) satisfy

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \qquad \xi \in \mathbb{R}^N, \quad x \in \overline{\Omega},$$

for some function κ such that $0 < \kappa_0 = \inf_{x \in \overline{\Omega}} \kappa(x)$. Moreover, there exist $\sigma \in (0,1]$ and $\beta_0 > 0$ such that

$$|Dq_{ij}(x)| \le \beta_0(\kappa(x))^{\sigma}, \qquad x \in \overline{\Omega};$$

- (ii) Hypotheses 13.1.1(i), 13.1.1(v), 13.1.1(vi) and 13.2.2 are satisfied;
- (iii) there exist a function $k_1: \mathbb{R}^N \to \mathbb{R}$ and $p_0 \in (1,2)$ such that

$$\sum_{i,j=1}^{N} D_i b_j(x) \xi_i \xi_j \le k_1(x) |\xi|^2, \qquad \xi \in \mathbb{R}^N, \quad x \in \overline{\Omega}.$$

Moreover,

$$L_1 := \sup_{x \in \mathbb{R}^N} \left\{ k_1(x) + \frac{N^2 \beta_0^2}{4\varepsilon_1} (\kappa(x))^{\sigma} \right\} < +\infty,$$

where

$$\varepsilon_1 = \frac{2(p_0 - 1)\kappa_0}{(6 + \omega_0 \varepsilon_0 - 2p_0)\kappa_0 + 4\omega_0 M_2}.$$

Remark 13.3.5 In the case when $\lim_{|x|\to+\infty} k_1(x)(\kappa(x))^{-\sigma} = -\infty$, Hypothesis 13.3.4(iii) is satisfied by any $p_0 \in (1,2]$.

Theorem 13.3.6 Under Hypotheses 13.3.4 the estimate (13.3.1) holds true for any $p \in [p_0, +\infty)$, with M_p being replaced with ω_p , where

$$\omega_{p} = L_{1} + \frac{\omega_{0}}{2} M_{1} - \frac{\omega^{2} \varepsilon_{0}}{4 + 2\omega_{0} \varepsilon_{0}} M_{1}^{-} + \left(\frac{1}{2\varepsilon_{1}} + \frac{1}{\varepsilon_{0}} + \frac{(1 + \varepsilon_{1})(2 - p)}{4\varepsilon_{1}} \omega_{0}\right) \omega_{0} M_{2},$$

$$(13.3.8)$$

$$if \ p \in (1, 2] \ and \ \omega_{p} = \omega_{2}^{p/2} \ if \ p \in (2, +\infty).$$

Proof. The proof can be obtained arguing as in the proof of Theorem 13.3.1, estimating the terms containing Q and its gradient as in the proof of Theorem 13.3.3.

We are now in a position to extend the results in Theorem 13.3.2 to the case when Ω is an exterior domain.

Theorem 13.3.7 Under the same assumptions as in Theorem 13.3.3, the estimate (13.3.4) holds true with $K_{p,t}t^{-p/2}$ replaced with $\widetilde{K}_{p,t}$, where

$$\widetilde{K}_{p,t} = \left(\frac{1}{[p(p-1)t\kappa_0]^{p/2}} + 1\right) \left(1 + \frac{\omega_0 \varepsilon_0}{2}\right)^{\frac{p}{2}},$$

if $p \in (1, 2]$, and $\widetilde{K}_{p,t} = \widetilde{K}_{2,t}^{p/2}$ if $p \in (2, +\infty)$.

Similarly, under the assumptions of Theorem 13.3.6, $\{T(t)\}$ satisfies (13.3.4) for any $p \ge p_0$, with M_p being replaced with ω_p given by (13.3.8).

Proof. The proof of the second part of the theorem can be obtained arguing as in the proof of Theorem 13.3.2, whereas the proof of the first part can be obtained applying the technique in the proof of Theorem 12.3.8 to the semigroup $\{T_n(t)\}$ associated with the operator $\mathcal{A}^{(n)}$ defined in (13.2.1). This yields (13.3.4) with $T_n(t)$ instead of T(t). Recalling that a suitable subsequence of $T_n(t)f$ converges to T(t)f in $C^{1,2}(F)$ for any compact set $F \subset (0, +\infty) \times \overline{\Omega}$ (see the proof of Theorem 13.2.5), we deduce the assertion.

To conclude this section, we briefly see some consequence of all the previous pointwise gradient estimates. First of all, as the following proposition shows, they allow us to improve the uniform gradient estimate (13.1.6) with k=0, in the case when $c\equiv 0$. We skip the proof since it is immediate.

Proposition 13.3.8 Under the assumptions of Theorem 13.3.2, for any $f \in C_b(\overline{\Omega})$ it holds that

$$||DT(t)f||_{\infty} \le \left(\frac{M_2(2+\omega_0\varepsilon_0)}{4\kappa_0(1-e^{-M_2t})}\right)^{\frac{1}{2}}||f||_{\infty}, \qquad t > 0,$$
 (13.3.9)

if $M_2 \neq 0$ and

$$||DT(t)f||_{\infty} \le \left(\frac{2 + \omega_0 \varepsilon_0}{4\kappa_0 t}\right)^{\frac{1}{2}} ||f||_{\infty}, \qquad t > 0,$$
 (13.3.10)

if $M_2 = 0$.

Similarly, under the assumptions of Theorem 13.3.3, for any $f \in C_b(\overline{\Omega})$, it holds that

$$||DT(t)f||_{\infty} \le \left(\frac{2+\omega_0\varepsilon_0}{4t\kappa_0}(1+2\kappa_0t)\right)^{\frac{1}{2}}||f||_{\infty}, \qquad t > 0.$$

Finally, under the assumptions of Theorem 13.3.6, $\{T(t)\}$ satisfies (13.3.9) (with M_2 replaced with ω_2 given by (13.3.8)) and (13.3.10).

Proposition 13.3.9 Let the assumptions of Theorem 13.3.2 be satisfied and suppose that $M_p \leq 0$ for some $p \in (1, +\infty)$. If $f \in D(\widehat{A})$ (see (13.1.39)) is such that Af = 0, then f is constant. The same result holds in the case when Ω is an exterior domain, under the assumptions of Theorem 13.3.6, provided that $\omega_p \leq 0$ for some $p \in [p_0, +\infty)$.

Proof. The proof is similar to that of Theorem 7.2.5. Hence it is omitted.

13.4 The invariant measure of the semigroup

In this section we deal with the invariant measure μ of the semigroup whenever it exists. To simplify the notation, we simply write L^p_{μ} and $W^{1,p}_{\mu}$ $(p \in [1, +\infty))$ instead of $L^p(\Omega, \mu)$ and $W^{1,p}(\Omega, \mu)$. Moreover, we denote by $\|\cdot\|_p$ the usual norm of L^p_{μ} .

In view of the extension of the gradient estimates in Section 13.3 to this setting, we assume that c identically vanishes in $\overline{\Omega}$.

First of all we observe that, repeating just the same proofs as in Section 12.4, we can show that, when the invariant measure exists, it is unique and is absolutely continuous with respect to the restriction of the Lebesgue measure to the σ -algebra of Borel sets of Ω . Moreover, its density ϱ is everywhere

positive in Ω and it belongs to $W_{\text{loc}}^{1,p}(\Omega)$ for any $p \in [1, +\infty)$. In particular, ϱ is a continuous function in Ω . The Khas'minskii theorem, in the version of Theorem 12.4.3, is still the main existence theorem of an invariant measure.

When $\{T(t)\}$ admits an invariant measure it can be extended to a strongly continuous semigroup of contractions in L^p_μ for any $p \in [1, +\infty)$, and the set $D(\widehat{A})$ (see (13.1.39)) is a core of the infinitesimal generator L_p of the semigroup.

The following proposition is the main result of this section: under the assumptions of Section 13.3 it allows us to prove that the semigroup maps L^p_μ into $W^{1,p}_\mu$. Moreover, it provides us some useful estimates of the behaviour of the gradient of T(t)f near t=0, when $f\in L^p_\mu$.

Proposition 13.4.1 Assume that the semigroup $\{T(t)\}$ admits an invariant measure and let Hypotheses 13.1.1(i), 13.1.1(ii), 13.1.1(iv)–13.1.1(vi) be satisfied. Further, suppose that $c \equiv 0$. Then, T(t) maps L^p_μ into $W^{l,p}_\mu$, for any t > 0 and any $p \in (1, +\infty)$. Moreover,

$$||DT(t)f||_{p} \leq \begin{cases} \frac{(2+\omega_{0}\varepsilon_{0})^{\frac{1}{2}}}{[2p(p-1)\kappa_{0}]^{\frac{1}{2}}} \left(\frac{M_{p}t}{1-e^{-M_{p}t}}\right)^{\frac{1}{p}} t^{-\frac{1}{2}} ||f||_{p}, \ p \in (1,2), \\ \left(\frac{2+\omega_{0}\varepsilon_{0}}{4\kappa_{0}} \frac{M_{2}t}{1-e^{-M_{2}t}}\right)^{\frac{1}{2}} t^{-\frac{1}{2}} ||f||_{p}, \qquad p \geq 2, \end{cases}$$

$$(13.4.1)$$

for any t>0 and any $p\in(1,+\infty)$, where M_p is given by (13.3.3). In the case when $M_p=0$, the term $M_pt(1-e^{-M_pt})^{-1}$ should be replaced with 1.

The same results hold in the particular case when Ω is an exterior domain, and Hypotheses 13.3.4 are satisfied. In such a case, the estimate (13.4.1) is satisfied for any $p \geq p_0$, where p_0 is given by Hypothesis 13.3.4(iii), with the constant M_p being replaced with the constant ω_p defined by (13.3.8).

Finally, if Hypotheses 13.2.1 and 13.2.2 (where we take $c \equiv 0$) are satisfied, then T(t) maps L^p_μ into $W^{1,p}_\mu$ for any t > 0 and any $p \in (1, +\infty)$. Moreover,

$$||DT(t)f||_{p} \leq \begin{cases} \left(\frac{1}{[p(p-1)t\kappa_{0}]^{p/2}} + 1\right)^{\frac{1}{p}} \left(1 + \frac{\omega_{0}\varepsilon_{0}}{2}\right)^{\frac{1}{2}} ||f||_{p}, & p \in [p_{0}, 2), \\ \left(\frac{1}{2t\kappa_{0}} + 1\right)^{\frac{1}{2}} \left(1 + \frac{\omega_{0}\varepsilon_{0}}{2}\right)^{\frac{1}{2}} t^{-\frac{1}{2}} ||f||_{p}, & p \geq 2. \end{cases}$$

$$(13.4.2)$$

A partial characterization of the domain of the infinitesimal generator of the semigroup $\{T(t)\}$ in L^p_μ now follows easily from the estimates in Proposition 13.4.1. We skip the proof since it can be obtained arguing as in the proof of Proposition 8.3.3.

Proposition 13.4.2 Suppose that Hypotheses 13.1.1(i), 13.1.1(ii), 13.1.1(iv)–13.1.1(vi), or Hypotheses 13.3.4 are satisfied. Then, $D(L_p)$ is continuously

embedded in $W^{1,p}_{\mu}$ for any $p \in (1,+\infty)$ (under Hypothesis 13.1.1) and for any $p \geq p_0$ (under Hypothesis 13.3.4). Moreover, for any $\omega > 0$, there exists a positive constant M_{ω} such that

$$||Df||_p \le M_{\omega,p} ||f||_p^{\frac{1}{2}} ||(L_p - \omega)f||_p^{\frac{1}{2}}, \qquad f \in D(L_p).$$
 (13.4.3)

In the case when the estimates in Proposition 13.4.1 are satisfied with an exponential term of negative type, we can take $\omega = 0$ in (13.4.3).

The same results hold under Hypotheses 13.2.1 and 13.2.2 (where we take $c \equiv 0$), for any $p \in (1, +\infty)$.

13.5 Final remarks

To conclude the chapter, we observe that the results of these chapter have been very recently generalized in [17, 71] in the case when Ω is an exterior domain Ω . To be more precise, under assumptions comparable with those in Hypothesis 6.1.1, uniform estimates up to the third-order have been proved in [71], for the semigroup associated with the operator

$$\mathcal{A}\varphi = \sum_{i,j=1}^{N} q_{ij} D_{ij} u + \sum_{j=1}^{N} b_j D_j u,$$

with boundary conditions $\mathcal{B}u = 0$ on $\partial\Omega$, where \mathcal{B} is a rather general class of differential boundary operators, including the Dirichlet and Neumann boundary operators.

Such estimates have been the keystone to prove optimal Schauder estimates for both the elliptic equation

$$\begin{cases} \lambda u - \mathcal{A}u = f, & \text{in } \Omega, \\ \mathcal{B}u = 0, & \text{on } \partial\Omega, \end{cases}$$

(for λ positive) and the parabolic problem

$$\begin{cases} D_t u - \mathcal{A} u = f, & t > 0, x \in \Omega, \\ \mathcal{B} u = 0, & t > 0, x \in \partial \Omega, \\ u(0, \cdot) = u_0, & x \in \overline{\Omega}, \end{cases}$$

adapting the arguments in Chapter 6.

Finally, in [17], the pointwise estimates of Section 13.3 have been obtained removing the Hypothesis 13.1.1 and 13.2.2.

Part III

A class of Markov semigroups in \mathbb{R}^N associated with degenerate elliptic operators

Chapter 14

The Cauchy problem in $C_b(\mathbb{R}^N)$

14.0 Introduction

In this chapter we generalize the results in Chapters 2 and 6 to a class of linear degenerate elliptic operators in $C_b(\mathbb{R}^N)$ of the type

$$(\mathcal{A}u)(x) = \sum_{i,j=1}^{r} q_{ij}(x)D_{ij}u(x) + \sum_{i,j=1}^{N} b_{ij}x_{j}D_{i}u(x), \qquad x \in \mathbb{R}^{N}, \quad (14.0.1)$$

under the following assumptions on the coefficients and on r.

Hypotheses 14.0.1 (i) $N/2 \le r < N$ and

$$\sum_{i,j=1}^{r} q_{ij}(x)\xi_i\xi_j \ge \kappa(x)|\xi|^2, \qquad \xi \in \mathbb{R}^r, \ x \in \mathbb{R}^N,$$

for some function $\kappa : \mathbb{R}^N \to (0, +\infty)$ such that $\kappa_0 := \inf_{x \in \mathbb{R}^N} \kappa(x) > 0$;

(ii) there exists a positive constant C such that

$$|D^{\alpha}q_{ij}(x)| \le C|x|^{(1-|\alpha|)^{+}} \sqrt{\kappa(x)},$$
 (14.0.2)

for any $x \in \mathbb{R}^N$, any i, j = 1, ..., r and any $|\alpha| \leq 3$;

(iii) the matrix B can be split as

$$B = \begin{pmatrix} B_1 & B_2 \\ B_3 & B_4 \end{pmatrix},$$

with $B_1 \in L(\mathbb{R}^r)$, $B_2, B_3^* \in L(\mathbb{R}^{N-r}, \mathbb{R}^r)$, $B_4 \in L(\mathbb{R}^{N-r})$. Moreover, $\operatorname{rank}(B_3) = N - r$.

The prototype of this class of elliptic operators is the degenerate Ornstein-Uhlenbeck operator

$$\mathcal{A}u(x) = \frac{1}{2} \sum_{i,j=1}^{N} q_{ij} D_{ij} u(x) + \sum_{i,j=1}^{N} b_{ij} x_j D_i u(x), \qquad x \in \mathbb{R}^N, \qquad (14.0.3)$$

where the matrix $Q = (q_{ij})$ is singular and positive definite, while the matrix

$$Q_t = \int_0^t e^{sB} Q e^{sB^*} ds, \qquad t > 0$$
 (14.0.4)

is strictly definite positive for any t>0 (see Chapter 9). In such a case some results, similar to those that we present here, have been proved by A. Lunardi in [107], without assuming that $r\geq N/2$ (see also Remark 9.2.4). Note that the condition (14.0.4) is equivalent to the hypoellipticity of the operator \mathcal{A} . Elliptic operators similar to those considered in this chapter have been also investigated, generally with different techniques, by other authors. Among them, we quote the papers [63, 84, 95, 96, 97, 98, 110, 111, 122].

Here, we show that, if A is given by (14.0.1) with the coefficients satisfying Hypotheses 14.0.1, then the Cauchy problem

$$\begin{cases}
D_t u(t,x) = \mathcal{A}u(t,x), & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N,
\end{cases}$$
(14.0.5)

admits, for any $f \in C_b(\mathbb{R}^N)$, a unique classical solution u (i.e., a unique function u continuous in $[0, +\infty) \times \mathbb{R}^N$ and continuously differentiable in $(0, +\infty) \times \mathbb{R}^N$, once with respect to time and twice with respect to the space variables, which solves (14.0.5)), which is bounded in $[0, +\infty) \times \mathbb{R}^N$. Therefore, as in the nondegenerate case, we can then associate a semigroup $\{T(t)\}$ of bounded linear operators with the operator \mathcal{A} by setting $T(\cdot)f = u$. The results that we present are taken from [102, 103].

We stress that the coefficients q_{ij} may be unbounded in \mathbb{R}^N , but Hypotheses 14.0.1 give some bounds on their growth at infinity (see (14.1.3)). Notice that, due to our assumptions on Q and B, the hypoellipticity condition (14.0.4) is satisfied for any t > 0 and any $x \in \mathbb{R}^N$.

We construct the semigroup $\{T(t)\}$ as the "limit" (in a sense to be made precise later on) as ε tends to 0^+ of a family of semigroups $\{T_{\varepsilon}(t)\}$ associated with the uniformly elliptic operators

$$A_{\varepsilon} = A + \varepsilon \sum_{j=r+1}^{N} D_{jj}, \qquad \varepsilon > 0,$$
 (14.0.6)

and to which all the results of Chapter 2 can be applied.

To prove the convergence of the function $T_{\varepsilon}(\cdot)f$ to $T(\cdot)f$, we first determine suitable estimates for the sup-norm of the space derivatives of the function $T_{\varepsilon}(t)f$, for any t>0, with the constants therein appearing, being independent of ε (see (14.2.1)-(14.2.4)). Such estimates replace the interior estimate (C.1.15) and make the compactness argument used in the proof of Theorem 2.2.1 work also in this situation.

Once the convergence of $T_{\varepsilon}(\cdot)f$ to a solution to the Cauchy problem (14.0.5) is proved, sharp estimates on the behaviour of the sup-norm of the space

derivatives of $T_{\varepsilon}(t)f$ as t approaches 0^+ can be obtained, just letting ε go to 0^+ in (14.2.1)-(14.2.4).

We remark also that some properties that we proved for Markov semigroups associated with uniformly elliptic operators in $C_b(\mathbb{R}^N)$ can be generalized also to this situation (see Section 14.3). In particular, the semigroup $\{T(t)\}$ is strong Feller and irreducible.

The uniform estimates are used in Section 14.4 to prove some regularity properties for the distributional solutions to both the elliptic equation

$$\lambda u(x) - \mathcal{A}u(x) = f(x), \qquad x \in \mathbb{R}^N, \qquad f \in C_b(\mathbb{R}^N), \quad \lambda > 0,$$

and to the nonhomogeneous Cauchy problem

$$\begin{cases} D_t u(t,x) = \mathcal{A}u(t,x) + g(t,x), & t > 0, x \in \mathbb{R}^N, \\ u(0,x) = f(x), & x \in \mathbb{R}^N. \end{cases}$$

when f and g are sufficiently smooth function.

To conclude this section, let us introduce a few notations.

Notations. Throughout this chapter, for any $x \in \mathbb{R}^N$, we denote by $Q(x) \in L(\mathbb{R}^r)$ the (strictly) positive definite matrix defined by $(Q(x))_{ij} = q_{ij}(x)$ for any $1 \le i, j \le r$. Similarly, for any $\varepsilon \ge 0$ and any $x \in \mathbb{R}^N$, we denote by $Q^{(\varepsilon)}(x) = (q_{ij}^{(\varepsilon)}(x)) \in L(\mathbb{R}^N)$ the matrix defined as follows: $q_{ij}^{(\varepsilon)}(x) = q_{ij}(x)$ if $i, j \le r$, $q_{ij}^{(\varepsilon)}(x) = \delta_{ij}\varepsilon$ if $i \lor j > r$.

14.1 Some preliminary results on the operator $\mathcal{A}_{\varepsilon}$

In this section we recall and prove some remarkable properties of the semi-group $\{T_{\varepsilon}(t)\}$ in $C_b(\mathbb{R}^N)$, associated with the operator $\mathcal{A}_{\varepsilon}$ ($\varepsilon > 0$) (see (14.0.6)). Such results will be used in following sections.

To begin with, we recall that, according to the results in Section 2.1, for any $\varepsilon > 0$, $u = T_{\varepsilon}(\cdot)f$ belongs to $C_b([0, +\infty) \times \mathbb{R}^N) \cap C_{\text{loc}}^{1+\alpha/2, 2+\alpha}((0, +\infty) \times \mathbb{R}^N)$ and it solves the Cauchy problem

$$\begin{cases}
D_t u(t,x) = \mathcal{A}_{\varepsilon} u(t,x), & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N,
\end{cases}$$
(14.1.1)

for any $f \in C_b(\mathbb{R}^N)$. Moreover,

$$||T_{\varepsilon}(t)f||_{C_{h}(\mathbb{R}^{N})} \le ||f||_{\infty}, \qquad t > 0, \quad f \in C_{b}(\mathbb{R}^{N}).$$
 (14.1.2)

Actually, $T_{\varepsilon}(\cdot)f$ is the unique classical solution to the Cauchy problem (14.1.1), which is bounded in $[0,T] \times \mathbb{R}^N$ for any T > 0. Indeed, the assumptions on the growth of the diffusion coefficients q_{ij} $(i,j=1,\ldots,N)$ in Hypotheses 14.0.1 imply that there exists a positive constant \hat{C} such that

$$\kappa(x) \le \hat{C}|x|^2, \qquad |q_{ij}(x)| \le \hat{C}|x|^2, \quad i, j = 1, \dots, N,$$
(14.1.3)

for any $x \in \mathbb{R}^N$. Therefore, the function $\varphi : \mathbb{R}^N \to \mathbb{R}$, defined by $\varphi(x) = 1 + |x|^2$ for any $x \in \mathbb{R}^N$, is a Lyapunov function for the operator $\mathcal{A}_{\varepsilon}$ ($\varepsilon > 0$), (i.e., it satisfies Hypothesis 4.0.2). The uniqueness of the bounded classical solution to the problem (14.1.1) now follows immediately from Theorem 4.1.3.

Now, we prove some boundedness and continuity properties up to t = 0 of the functions $(t, x) \mapsto t^{(j-k)/2}(T_{\varepsilon}(t)f)(x)$ when $f \in C_b^k(\mathbb{R}^N)$ $(j, k \in \mathbb{N}, 1 \le j \le k \le 3)$, similar to those in Theorem 6.1.7.

Proposition 14.1.1 Suppose that Hypotheses 14.0.1 are satisfied. Then, for any T > 0, any $\varepsilon > 0$ and any k = 0, 1, 2, 3, there exists a positive constant $\widetilde{C} = \widetilde{C}_{T,\varepsilon}$ such that

$$\sum_{j=0}^{3} t^{\frac{(j-k)^{+}}{2}} \|D^{j} T_{\varepsilon}(t) f\|_{\infty} \leq \widetilde{C} \|f\|_{C^{k}(\mathbb{R}^{N})}, \qquad t \in (0, T].$$
 (14.1.4)

Proof. To begin with, we note that the estimate (14.1.4) does not follow immediately from Theorem 6.1.7. The techniques therein used need to be adapted to our situation. Indeed, the quoted theorem to be applied requires that the modulus of derivatives of the diffusion coefficients of the operator $\mathcal{A}_{\varepsilon}$ at any $x \in \mathbb{R}^N$ could be estimated by $C\overline{\kappa}(x)$, where $\overline{\kappa}(x)$ is the minimum eigenvalue of the matrix $Q^{(\varepsilon)}(x)$. Therefore, it requires that the derivatives of the coefficients are bounded in \mathbb{R} , which may be not the case in our situation.

We limit ourselves to proving (14.1.4) in the case when (k,l)=(0,3), the other cases being similar and even easier. For any $n \in \mathbb{N}$, let $v_{n,\varepsilon}:(0,+\infty)\times\mathbb{R}^N\to\mathbb{R}$ be the function defined by

$$\begin{split} v_{n,\varepsilon}(t,x) &= |u_{n,\varepsilon}(t,x)|^2 + at\vartheta_n^2 |Du_{n,\varepsilon}(t,x)|^2 + a^2t^2\vartheta_n^4 |D^2u_{n,\varepsilon}(t,x)|^2 \\ &+ a^3t^3\vartheta_n^6 |D^3u_{n,\varepsilon}(t,x)|^2, \end{split}$$

for any t > 0 and any $x \in \overline{B}(n)$, where $u_{n,\varepsilon}$ denotes the classical solution to the problem

$$\begin{cases} D_t u_{n,\varepsilon}(t,x) = \mathcal{A}_{\varepsilon} u_{n,\varepsilon}(t,x), & t > 0, x \in B(n), \\ u_{n,\varepsilon}(t,x) = 0, & t > 0, x \in \partial B(n), \\ u_{n,\varepsilon}(0,x) = \vartheta_n(x) f(x), & x \in B(n), \end{cases}$$

and $\vartheta_n \in C_c^{\infty}(\mathbb{R}^N)$ is such that $\chi_{B(n/2)} \leq \vartheta_n \leq \chi_{B(n)}$. From now on, to simplify the notation, when there is no damage of confusion, we do not write

explicitly the dependence on n and ε of the functions that we consider. As it is immediately seen, the function v solves the Cauchy problem

$$\begin{cases} D_t v(t, x) = \mathcal{A}_{\varepsilon} v(t, x) + g(t, x), & t \in [0, T], x \in B(n), \\ v(t, x) = 0, & t \in [0, T], x \in \partial B(n), \\ v(0, x) = (f(x))^2, & x \in B(n), \end{cases}$$

where $g = \sum_{j=1}^{5} g_j$ is given by

$$g_{1} = -2 \sum_{i,j=1}^{N} q_{ij}^{(\varepsilon)} D_{i} u D_{j} u - 2at \vartheta^{2} \sum_{i,j,h=1}^{N} q_{ij}^{(\varepsilon)} D_{ih} u D_{jh} u$$
$$-2a^{2} t^{2} \vartheta^{4} \sum_{i,j,h,k=1}^{N} q_{ij}^{(\varepsilon)} D_{ihk} u D_{jhk} u - 2a^{3} t^{3} \vartheta^{6} \sum_{i,j,h,k,l=1}^{N} q_{ij}^{(\varepsilon)} D_{ihkl} u D_{jhkl} u,$$

$$g_{2} = -2at\vartheta(\mathcal{A}_{\varepsilon}\vartheta)|Du|^{2} - 4a^{2}t^{2}\vartheta^{3}(\mathcal{A}_{\varepsilon}\vartheta)|D^{2}u|^{2} - 6a^{3}t^{3}\vartheta^{5}(\mathcal{A}_{\varepsilon}\vartheta)|D^{3}u|^{2}$$

$$-8at\vartheta \sum_{i,j,h=1}^{N} q_{ij}^{(\varepsilon)} D_{j}\vartheta D_{h}u D_{ih}u - 16a^{2}t^{2}\vartheta^{3} \sum_{i,j,h,k=1}^{N} q_{ij}^{(\varepsilon)} D_{j}\vartheta D_{hk}u D_{ihk}u$$

$$-24a^{3}t^{3}\vartheta^{5} \sum_{i,j,h,k,l=1}^{N} q_{ij}^{(\varepsilon)} D_{j}\vartheta D_{hkl}u D_{ihkl}u + 2at\vartheta^{2} \sum_{i,j=1}^{N} b_{ij} D_{il}u D_{j}u$$

$$+4a^{2}t^{2}\vartheta^{4} \sum_{i,j,h=1}^{N} b_{ij} D_{ih}u D_{jh}u + 6a^{3}t^{3}\vartheta^{6} \sum_{i,j,h,k=1}^{N} b_{ij} D_{ihk}u D_{jhk}u,$$

$$\begin{split} g_3 &= 2at\vartheta^2 \sum_{i,j=1}^r \sum_{h=1}^N D_h q_{ij} D_h u D_{ij} u + 4a^2 t^2 \vartheta^4 \sum_{i,j=1}^r \sum_{h,k=1}^N D_h q_{ij} D_{hk} u D_{ijk} u \\ &+ 6a^3 t^3 \vartheta^6 \sum_{i,j=1}^r \sum_{h,k,l=1}^N D_h q_{ij} D_{hkl} u D_{ijkl} u \\ &+ 6a^3 t^3 \vartheta^6 \sum_{i,j=1}^r \sum_{h,k,l=1}^N D_{hk} q_{ij} D_{ijl} u D_{hkl} u \\ &+ 2a^2 t^2 \vartheta^4 \sum_{i,j=1}^r \sum_{h,k=1}^N D_{hk} q_{ij} D_{ij} u D_{hk} u \\ &+ 2a^3 t^3 \vartheta^6 \sum_{i,j=1}^r \sum_{h,k,l=1}^N D_{hkl} q_{ij} D_{ij} u D_{hkl} u, \end{split}$$

$$g_4 = -2at|Du|^2 \sum_{i,j=1}^N q_{ij}^{(\varepsilon)} D_i \vartheta D_j \vartheta - 12a^2 t^2 \vartheta^2 |D^2 u|^2 \sum_{i,j=1}^N q_{ij}^{(\varepsilon)} D_i \vartheta D_j \vartheta$$
$$-30a^3 t^3 \vartheta^4 |D^3 u|^2 \sum_{i,j=1}^N q_{ij}^{(\varepsilon)} D_i \vartheta D_j \vartheta,$$

$$g_5 = a\vartheta^2 |Du|^2 + 2a^2t\vartheta^4 |D^2u|^2 + 3a^3t^2\vartheta^6 |D^3u|^2.$$

In the rest of the proof, to simplify the notation, we denote by $D_{\#,1}^k u$ (resp. $D_{\#,2}^k u$) (k=1,2,3,4) the vectors whose entries are the k-th-order derivatives of u corresponding, respectively, to multi-indices α with at least a component less than r+1, and multi-indices α with all the components greater than r.

Taking Hypothesis 14.0.1(i) into account, we easily deduce that

$$g_{1} \leq -2(\varepsilon \wedge \kappa_{0})|Du|^{2} - 2at\vartheta^{2}\kappa|D_{\#,1}^{2}u|^{2}$$

$$-2at\vartheta^{2}\varepsilon|D_{\#,2}^{2}u|^{2} - 2a^{2}t^{2}\vartheta^{4}\kappa|D_{\#,1}^{3}u|^{2}$$

$$-2a^{2}t^{2}\vartheta^{4}\varepsilon|D_{\#,2}^{3}u|^{2} - 2a^{3}t^{3}\vartheta^{6}\kappa|D_{\#,1}^{4}u|^{2} - 2a^{3}t^{3}\vartheta^{6}\varepsilon|D_{\#,2}^{4}u|^{2}.$$

$$(14.1.5)$$

To estimate the function g_2 we observe that, arguing as in the proof of (6.1.23)-(6.1.25) and taking (14.1.3) into account, we get

$$|(\mathcal{A}\vartheta)(x)| \le C_1, \qquad |(Q^{(\varepsilon)}(x)D\vartheta(x))_i| \le C_1 \begin{cases} \sqrt{\kappa(x)}, & \text{if } i \le r, \\ 1, & \text{if } i > r, \end{cases}$$
 (14.1.6)

for any $x \in \mathbb{R}^N$ and some positive constant C_1 , independent of ε and x. Now, using (14.1.6), we obtain

$$\begin{split} & \left| 8at\vartheta \sum_{i,j,h=1}^{N} q_{ij}^{(\varepsilon)} D_{j}\vartheta D_{h}u D_{ih}u \right| \\ & \leq 8at\vartheta \left| \sum_{i=1}^{r} \sum_{j,h=1}^{N} q_{ij}^{(\varepsilon)} D_{j}\vartheta D_{h}u D_{ih}u \right| + 8at\vartheta \left| \sum_{i=r+1}^{N} \sum_{j,h=1}^{N} q_{ij}^{(\varepsilon)} D_{j}\vartheta D_{h}u D_{ih}u \right| \\ & \leq 8at\vartheta C_{1} \sqrt{\kappa} \sum_{i=1}^{r} \sum_{h=1}^{N} |D_{h}u| |D_{ih}u| + 8at\vartheta C_{1} \sum_{i=r+1}^{N} \sum_{h=1}^{N} |D_{h}u| |D_{ih}u| \\ & \leq 8atC_{1} \left(\frac{r}{4\delta} |Du|^{2} + \delta\vartheta^{2}\kappa |D_{\#,1}^{2}u|^{2} \right) \\ & + 8atC_{1} \left(\frac{N-r}{4\delta} |Du|^{2} + \frac{\delta}{\kappa_{0}} \vartheta^{2}\kappa |D_{\#,1}^{2}u|^{2} + \delta\vartheta^{2} |D_{\#,2}^{2}u|^{2} \right) \\ & \leq 8atC_{1} \frac{N}{4\delta} |Du|^{2} + 8atC_{1} \delta \left(1 + \frac{1}{\kappa_{0}} \right) \vartheta^{2}\kappa |D_{\#,1}^{2}u|^{2} + 8atC_{1} \delta\vartheta^{2} |D_{\#,2}^{2}u|^{2}. \end{split}$$

Similarly, we can estimate all the other terms in the definition of g_2 and g_3 , so that, finally, we get

$$g_{2} \leq 2at \left\{ C_{1} + C_{1} \frac{N}{\delta} + \|B\| \right\} |Du|^{2}$$

$$+4at \left\{ 2C_{1}\delta \left(1 + \frac{1}{\kappa_{0}} \right) + at \left(\frac{C_{1}}{\kappa_{0}} + \frac{NC_{1}}{\delta \kappa_{0}} + \frac{\|B\|}{\kappa_{0}} \right) \right\} \vartheta^{2} \kappa |D_{\#,1}^{2} u|^{2}$$

$$+4at \left\{ 2C_{1}\delta + at \left(C_{1} + C_{1} \frac{N}{\delta} + \|B\| \right) \right\} \vartheta^{2} |D_{\#,2}^{2} u|^{2}$$

$$+2a^{2}t^{2} \left\{ 8C_{1}\delta \left(1 + \frac{1}{\kappa_{0}} \right) + 3at \left(\frac{C_{1}N}{\delta \kappa_{0}} + \frac{C_{1}}{\kappa_{0}} + \frac{\|B\|}{\kappa_{0}} \right) \right\} \vartheta^{4} \kappa |D_{\#,1}^{3} u|^{2}$$

$$+2a^{2}t^{2} \left\{ 8C_{1}\delta + 3at \left(\frac{C_{1}N}{\delta} + C_{1} + \|B\| \right) \right\} \vartheta^{4} |D_{\#,2}^{3} u|^{2}$$

$$+24a^{3}t^{3}C_{1}\delta \left(1 + \frac{1}{\kappa_{0}} \right) \vartheta^{6} \kappa |D_{\#,1}^{4} u|^{2} + 24a^{3}t^{3}C_{1}\delta \vartheta^{6} |D_{\#,2}^{4} u|^{2}$$

$$(14.1.7)$$

and

$$g_{3} \leq atC \frac{N^{2}}{2\delta} |Du|^{2} + atCN \left\{ 2\delta + at \left(\frac{N}{\delta} + N + \frac{N}{\kappa_{0}} \right) + a^{2}t^{2} \frac{N}{2\delta} \right\} \vartheta^{2} \kappa |D_{\#,1}^{2}u|^{2}$$

$$+ a^{2}t^{2}CN^{2} \left(\frac{1}{\delta} + 1 + atC \frac{N}{2\delta} \right) \vartheta^{2} |D_{\#,2}^{2}u|^{2}$$

$$+ a^{2}t^{2}CN \left\{ 4\delta + atN \left(\frac{3}{2\delta\kappa_{0}} + 3 + \frac{3}{\kappa_{0}} + 2\delta \right) \right\} \vartheta^{4} \kappa |D_{\#,1}^{3}u|^{2}$$

$$+ a^{3}t^{3}CN^{2} \left(\frac{3}{2\delta} + 3 \right) \vartheta^{4} |D_{\#,2}^{3}|^{2} + 6a^{3}t^{3}C\delta N\vartheta^{6} \kappa |D_{\#,1}^{4}u|^{2},$$

$$(14.1.8)$$

for any $\delta > 0$, where C is given by (14.0.2). For further details, see the proof of the estimate (6.1.26).

Now, let us fix T > 0. Observing that $g_4 \leq 0$ in $(0, +\infty) \times \mathbb{R}^N$, from (14.1.5), (14.1.7) and (14.1.8) we deduce that

$$g \leq \left\{ -2(\varepsilon \wedge \kappa_0) + aT \left(2C_1 + 2C_1 \frac{N}{\delta} + 2\|B\| + \frac{CN^2}{2\delta} \right) \right\} |Du|^2$$

$$+ at \left\{ -2 + 8C_1 \delta \left(1 + \frac{1}{\kappa_0} \right) + 2CN\delta + 4aT \left(\frac{C_1}{\kappa_0} + \frac{NC_1}{\delta \kappa_0} + \frac{\|B\|}{\kappa_0} \right) \right.$$

$$+ aTCN \left(\frac{N}{\delta} + N + \frac{N}{\kappa_0} + aT \frac{N}{2\delta} \right) \right\} |D_{\#,1}^2 u|^2$$

$$\begin{split} +at \bigg\{ -2\varepsilon + 8C_1\delta + 4aT \left(C_1 + \frac{C_1N}{\delta} + \|B\| \right) \\ +aTCN^2 \left(\frac{1}{\delta} + 1 + aT\frac{CN}{2\delta} \right) \bigg\} \vartheta^2 |D_{\#,2}^2 u|^2 \\ +a^2t^2 \bigg\{ -2 + 16C_1\delta \left(1 + \frac{1}{\kappa_0} \right) + 4CN\delta + 6aT \left(\frac{C_1N}{\delta\kappa_0} + \frac{C_1}{\kappa_0} + \frac{\|B\|}{\kappa_0} \right) \\ +aTCN^2 \left(\frac{3}{2\delta\kappa_0} + 3 + \frac{3}{\kappa_0} + 2\delta \right) \bigg\} \vartheta^4 \kappa |D_{\#,1}^3 u|^2 \\ +a^2t^2 \bigg\{ -2\varepsilon + 16C_1\delta + 6aT \left(\frac{C_1N}{\delta} + C_1 + \|B\| \right) \\ +aTCN^2 \left(\frac{3}{2\delta} + 3 \right) \bigg\} \vartheta^4 |D_{\#,2}^3 u|^2 \\ +2a^3t^3 \bigg\{ -1 + 3aT\delta \left[3CN + 4C_1 \left(1 + \frac{1}{\kappa_0} \right) \right] \bigg\} \vartheta^6 |D_{\#,1}^4 u|^2 \\ +2a^3t^3 \bigg\{ -\varepsilon + 12TC_1\delta \bigg\} \vartheta^6 |D_{\#,2}^4 u|^2, \end{split}$$

for any $t \in (0, T]$. A straightforward computation shows that we can choose (a, δ) such that $g \leq 0$ in $(0, T] \times B(n)$. Theorem 4.1.3 now implies that

$$|v_{n,\varepsilon}(t,x)| \le \widetilde{C} ||f||_{\infty}, \qquad t \in [0,T], \quad x \in \overline{B}(n),$$

for some positive constant \widetilde{C} , independent of u and n. Since, by Proposition 2.2.1, $u_{n,\varepsilon}$ converges to u_{ε} on compact sets of $(0,+\infty)\times\mathbb{R}^N$, as n tends to $+\infty$, letting n go to $+\infty$ we get (14.1.4).

The following theorem guarantees the continuity of the functions $(t,x) \mapsto t^{j/2}(D^jT_{\varepsilon}(t)f)(x)$ at t=0, for any j=1,2,3 and any $f\in C_b(\mathbb{R}^N)$. Its proof is essentially based upon Proposition 2.2.9 and the interior estimates in Theorem C.1.4.

Proposition 14.1.2 Under Hypotheses 14.0.1, for any $f \in C_b^k(\mathbb{R}^N)$ (k = 0, ..., 3) and any $\varepsilon > 0$, the function $(t, x) \mapsto t^{(j-k)^+/2}(D^jT_{\varepsilon}(t)f)(x)$ is continuous in $[0, +\infty) \times \mathbb{R}^N$ for any j = 0, ..., 3. In particular,

(i)
$$\lim_{t \to 0^+} t^{\frac{j-k}{2}} (D^j T_{\varepsilon}(t) f)(x) = 0, \quad \text{if } j > k,$$

(ii) $\lim_{t \to 0^+} (D^j T_{\varepsilon}(t) f)(x) = D^j f(x), \quad \text{if } j \leq k,$ (14.1.9)

for any $x \in \mathbb{R}^N$ and j, k = 0, 1, 2, 3.

Proof. The continuity of the functions $(t,x) \mapsto t^{(j-k)^+/2}(D^jT_{\varepsilon}(t)f)(x)$ in $(0,+\infty) \times \mathbb{R}^N$ is a classical result recalled in Theorem C.1.4. Moreover, the formula (14.1.9)(ii) can be proved by arguing as in the proof of Proposition

7.1.1. So, we just need to show that the formula (14.1.9)(i) holds. We limit ourselves to dealing with the case when k=0, since the other cases are similar and even simpler. We use a localization argument. For any $x \in \mathbb{R}^N$ we fix r, R > 0 with R-r > 1. Applying (C.1.16) and (C.1.18) with $\Omega' = x + B(r)$ and $\Omega = x + B(R)$ yields

$$\sup_{\substack{(t,y)\in(0,1)\times(x+B(r))}} t^{\frac{1}{2}} |Dv(t,y)| + \sup_{\substack{(t,y)\in(0,1)\times(x+B(r))}} t |D^{2}v(t,y)|$$

$$+ \sup_{\substack{(t,y)\in(0,1)\times(x+B(r))}} t^{\frac{3}{2}} |D^{3}v(t,y)|$$

$$\leq C \sup_{\substack{(t,y)\in(0,1)\times(x+B(R))}} |v(t,y)|,$$
(14.1.10)

for any solution to the parabolic equation $D_t v - \mathcal{A}_{\varepsilon} v = 0$ and some positive constant $C = C(r, R, \varepsilon)$. Now, we fix a sequence $\{f_n\} \in C_b^3(\mathbb{R}^N)$ of smooth functions, bounded in the sup-norm and converging locally uniformly in \mathbb{R}^N to f as n tends to $+\infty$. Applying (14.1.10) with $v = T_{\varepsilon}(\cdot)(f - f_n)$ and taking Proposition 2.2.9 into account, we deduce that the function $(t,x) \mapsto \psi_{j,n}(t,x) := t^{j/2}(D^jT_{\varepsilon}(t)f_n)(x)$ (j=1,2,3) converges uniformly in $(0,T) \times (x+B(\overline{R}))$ to the function $(t,x) \mapsto \psi_j(t,x) := t^{j/2}(D^jT_{\varepsilon}(t)f)(x)$. By (14.1.9)(ii), each function $\psi_{n,j}$ is continuous in $[0,1] \times (x+\overline{B}(r))$ and it vanishes at t=0. Hence, each ψ_j (j=1,2,3) is continuous in $[0,1] \times (x+\overline{B}(r))$ and it vanishes at t=0, as well. The estimate (14.1.9)(i) follows.

14.2 Existence, uniqueness results and uniform estimates

In this section we show that, for any $f \in C_b(\mathbb{R}^N)$, the problem (14.0.5) admits a unique classical solution u bounded in $[0, +\infty) \times \mathbb{R}^N$, where we recall that by classical solution we mean, as usual, a function $u : [0, +\infty) \times \mathbb{R}^N \to \mathbb{R}$ continuous in $[0, +\infty) \times \mathbb{R}^N$, continuously differentiable, once with respect to time and twice with respect to the space variables in $(0, +\infty) \times \mathbb{R}^N$, which satisfies (14.0.5). Moreover, we show that the family of linear operators $\{T(t)\} \in L(C_b(\mathbb{R}^N))$ defined by $T(t)f = u(t, \cdot)$, for any t > 0 and any $f \in C_b(\mathbb{R}^N)$, gives rise to a semigroup of linear operators satisfying

$$||D_i T(t) f||_{\infty} \le C e^{\omega t} t^{-\frac{1}{2} - H(i - r)} ||f||_{\infty}, \qquad t > 0,$$
 (14.2.1)

$$||D_{ij}T(t)f||_{\infty} \le Ce^{\omega t}t^{-(1+H(i-r)+H(j-r))}||f||_{\infty}, \quad t > 0,$$
 (14.2.2)

$$\|D_{ijh}T(t)f\|_{\infty} \leq Ce^{\omega t}t^{-\frac{3}{2}-H(i-r)-H(j-r)-H(h-r)}\|f\|_{\infty}, \qquad t>0, \quad (14.2.3)$$

for any i, j, h = 1, ..., N and some positive constants C and ω , independent of f and t. Here, H(s) = 0 if $s \le 0$ and H(s) = 1 if s > 0. Furthermore, we

prove that the more f is regular, the more the estimates (14.2.1)-(14.2.3) can be improved. More precisely, we show that

$$||T(t)f||_{C_b^k(\mathbb{R}^N)} \le Ce^{\omega t} ||f||_{C_b^k(\mathbb{R}^N)}, \qquad t > 0, \quad k = 1, 2, 3,$$
 (14.2.4)

$$||D_{ij}T(t)f||_{\infty} \le Ce^{\omega t}t^{-\frac{1}{2}-H(i-r)}||f||_{C_h^1(\mathbb{R}^N)}, \qquad t > 0, \quad i \le j, \quad (14.2.5)$$

$$||D_{ijh}T(t)f||_{\infty} \le Ce^{\omega t}t^{c_{ijk}^k}||f||_{C_b^k(\mathbb{R}^N)}, \qquad t > 0, \quad 1 \le j \le h, \quad k = 0, 1, 2,$$
(14.2.6)

where

$$c_{ijh}^k = \frac{3-k}{2} + H(i-r) + ((2-k) - (1-k)^+)H(j-r) + (1-k)^+H(h-r),$$

and C, ω are as above.

As it has already been claimed in the introduction, we will construct the semigroup $\{T(t)\}$ as the limit as ε tends to 0^+ (in a sense to be made clear) of the semigroup $\{T_{\varepsilon}(t)\}$. As a first step in this direction, in subsection 14.2.2 we show that each semigroup $\{T_{\varepsilon}(t)\}$ satisfies (14.2.1)-(14.2.6) with constants being independent of ε . This will allow us to apply a compactness argument to prove the convergence of the function $T_{\varepsilon}(\cdot)f$ to what we will denote by $T(\cdot)f$, for any $f \in C_b(\mathbb{R}^N)$. Moreover, letting ε go to 0^+ in the previous estimates written for the approximating semigroups, we will see that $\{T(t)\}$ satisfies (14.2.1)-(14.2.6) as well.

14.2.1 Some preliminary lemmata

In this section we collect some very easy results of linear algebra that we need in this chapter. For the reader's convenience we give the proofs.

Lemma 14.2.1 Suppose that $A \in L(\mathbb{R}^n, \mathbb{R}^m)$ is a given matrix. Then, there exists a matrix $C \in L(\mathbb{R}^m, \mathbb{R}^n)$ such that

$$AC + C^*A^*$$

is strictly positive (resp. negative) definite if and only if $n \ge m$ and $\operatorname{rank}(A) = m$.

Let $B \in L(\mathbb{R}^n, \mathbb{R}^m)$. Then, there exists a matrix $C \in L(\mathbb{R}^m, \mathbb{R}^n)$ such that

$$CB + B^*C^*$$

is uniformly positive (resp. negative) definite if and only if $m \ge n$ and $\operatorname{rank}(B) = n$.

Proof. As far as the first statement is concerned, we limit ourselves to proving that we can choose C such that the matrix $AC + C^*A^*$ (resp. $CB + B^*C^*$) is strictly positive definite. Replacing C by -C we get the assertion also in the other case.

Suppose that the matrix $AC + C^*A^*$ is strictly positive definite. Then, for any $\xi \in \mathbb{R}^m \setminus \{0\}$, we have

$$0 < \langle (AC + C^*A^*)\xi, \xi \rangle = 2\langle AC\xi, \xi \rangle.$$

Hence, the matrix AC is not singular, i.e., its rank equals m. Suppose that n < m. Then the ranks of A and C equal at most n and, consequently, the rank of AC is at most n. This implies that AC is singular. Hence, $n \ge m$. Moreover, rank(A) = m, otherwise rank(AC) would be strictly less than m and AC would be singular.

Vice versa, let us assume that $n \geq m$ and $\operatorname{rank}(A) = n$. Moreover, let $D \in L(\mathbb{R}^n)$ be an invertible matrix such that

$$AD = (A_1 A_2),$$

where $A_1 \in L(\mathbb{R}^m)$ is invertible and $A_2 \in L(\mathbb{R}^{n-m}, \mathbb{R}^m)$. Let $\widetilde{C} \in L(\mathbb{R}^m, \mathbb{R}^n)$ be the matrix defined by

$$\widetilde{C} = \begin{pmatrix} A_1^{-1} K \\ 0 \end{pmatrix},$$

 $K\in L(\mathbb{R}^m)$ being any strictly positive definite matrix. We set $C=D\widetilde{C}$ and observe that

$$AC = AD\widetilde{C} = A_1 A_1^{-1} K = K.$$

Hence, $AC + C^*A^* = 2K$ so that the matrix $AC + C^*A^*$ is strictly positive definite.

Let us prove the last part of the lemma. The same arguments as above show that if $CB + B^*C^*$ is strictly positive definite, then $m \ge n$. Vice versa, suppose that $m \ge n$, rank(B) = n, and let $C = B^*$. Then,

$$\langle (CB + B^*C^*)\xi, \xi \rangle = 2\langle B^*B\xi, \xi \rangle = 2\|B\xi\|^2,$$

for any $\xi \in \mathbb{R}^n$. Since $\operatorname{rank}(B) = n$, then the mapping $T : \mathbb{R}^n \to \mathbb{R}^m$, defined by $T(\xi) = B\xi$ for any $\xi \in \mathbb{R}^N$, is injective. It follows that the matrix B^*B is strictly positive definite and this finishes the proof.

Lemma 14.2.2 Fix $k, n \in \mathbb{N}$ and let $A(t) \in L(\mathbb{R}^N)$ be the matrix defined by

$$\begin{pmatrix} A_{11}t^{k} & A_{12}t^{k+1} & \cdots & A_{1n}t^{k+n-1} \\ A_{12}^{*}t^{k+1} & A_{22}t^{k+2} & \cdots & A_{2n}t^{k+n} \\ \vdots & \ddots & \ddots & \vdots \\ A_{1n}^{*}t^{k+n-1} & A_{2n}^{*}t^{k+n} & \cdots & A_{nn}t^{k+2n-2} \end{pmatrix}, \qquad t > 0,$$

where $A_{ij} \in L(\mathbb{R}^{N_j}, \mathbb{R}^{N_i})$ $(N_1 + \ldots + N_n = N)$ and $A_{ii} = A_{ii}^*$ for any $i = 1, \ldots, n$. Then, A(t) is positive (resp. negative) definite, for any t > 0, if and only if it is positive (resp. negative) definite at t = 1. In such a case if, for any $\xi \in \mathbb{R}^N$, we split $\xi = (\xi_1, \ldots, \xi_n)$ with $\xi_i \in \mathbb{R}^{N_i}$, we have

$$\langle A(t)\xi, \xi \rangle \ge \lambda_{\min}(A(1)) \sum_{j=0}^{n-1} t^{k+2j} |\xi_{j+1}|^2, \qquad t > 0,$$
 (14.2.7)

if A(1) is positive definite, and

$$\langle A(t)\xi,\xi\rangle \le \lambda_{\max}(A(1)) \sum_{j=0}^{n-1} t^{k+2j} |\xi_{j+1}|^2, \qquad t>0,$$
 (14.2.8)

if A(1) is negative definite.

Proof. Of course, if A(t) is strictly positive (resp. negative) definite for any t > 0, then, in particular, it is strictly positive (resp. negative) definite at t = 1.

Vice versa, assume that A(1) is strictly positive definite. For any t > 0 and any $\xi \in \mathbb{R}^N$, split as in the statement of the lemma, let

$$\overline{\xi}^T = (t^{k/2}\xi_1, t^{k/2+1}\xi_2, \dots, t^{k/2+n-1}\xi_n)^T.$$

As it is immediately seen,

$$\langle A(t)\xi, \xi \rangle = \langle A(1)\overline{\xi}, \overline{\xi} \rangle.$$

Therefore, A(t) is strictly positive (resp. negative) definite and the inequalities (14.2.7), (14.2.8) follow.

Lemma 14.2.3 Suppose that $Q, A \in L(\mathbb{R}^N)$ are two positive definite matrices. Further, assume that the submatrix $Q_0 = (q_{ij})_{i,j=1}^r$ is strictly positive definite and $q_{ij} = 0$ if $i \lor j > r$. Then,

$$\operatorname{Tr}(QA) \ge \lambda_{\min}(Q_0)\operatorname{Tr}(A_1),$$

where A_1 is the submatrix obtained from A by erasing the last N-r rows and lines.

Proof. Since Q_0 is strictly positive definite, then there exist an orthogonal matrix $B = (b_{ij})_{i,j=1}^N \in L(\mathbb{R}^r)$ and a diagonal matrix $\Lambda = \operatorname{diag}(\lambda_1, \ldots, \lambda_r)$ such that $Q_0 = B^*\Lambda B$. This implies that the matrix $\widetilde{B} \in L(\mathbb{R}^N)$, defined by $\widetilde{b}_{ij} = b_{ij}$ if $1 \leq i, j \leq r$, $\widetilde{b}_{ij} = \delta_{ij}$ if $i \vee j > r$, is orthogonal and $\widetilde{B}Q\widetilde{B}^* = \widetilde{\Lambda} = \operatorname{diag}(\lambda_1, \ldots, \lambda_r, 0, \ldots, 0)$. Hence,

$$\operatorname{Tr}\left(QA\right)=\operatorname{Tr}\left(\widetilde{B}^{-1}\widetilde{\Lambda}\widetilde{B}A\right)=\operatorname{Tr}\left(\widetilde{\Lambda}\widetilde{B}A\widetilde{B}^{-1}\right)=\operatorname{Tr}\left(\widetilde{\Lambda}\widetilde{B}A\widetilde{B}^{*}\right).$$

We now observe that since A is positive definite, then $\widetilde{B}A\widetilde{B}^* =: (c_{ij})_{i,j=1}^N$ is. This, in particular, implies that $c_{ii} \geq 0$ for any i = 1, ..., N. It follows that

$$\operatorname{Tr}\left(\Lambda \widetilde{B} A \widetilde{B}^*\right) = \sum_{j=1}^r \lambda_j \, c_{jj} \ge \lambda \operatorname{Tr}\left(B A_1 B^*\right) = \lambda_{\min}(Q_0) \operatorname{Tr}\left(A_1\right),$$

and the assertion follows.

14.2.2 Uniform estimates for the space derivatives of the approximating semigroups $\{T_{\varepsilon}(t)\}$

To begin with, we introduce a few notations. We set $D_{*,1}w = (D_1w, \ldots, D_rw)$ and $D_{*,2}w = (D_{r+1}w, \ldots, D_Nw)$. Then, we introduce, instead of the corresponding tensors, the vectors $D_*^k w$ (k=2,3,4) consisting of all the derivatives $D_{i_1,\ldots,i_k}w$ ordered as follows: $D_{i_1,\ldots,i_k}w$ precedes $D_{j_1,\ldots,j_k}w$ if $i_l \leq j_l$ for any $l=1,\ldots,k$ and there exists $l_0 \in \{1,\ldots,k\}$ such that $i_{l_0} < j_{l_0}$, or $\{j_1,\ldots,j_k\}$ contains more indexes $j_l \geq r+1$ than the set $\{i_1,\ldots,i_k\}$. Finally, we set $D_*^k w^T = ((D_{*,1}^k w)^T,\ldots,(D_{*,k+1}^k w)^T)$, where the vector $D_{*,j}^k w$ contains all the derivatives $D_{i_1,\ldots,i_k}^k w$ with $i_{k+1-j} \leq r < i_{k+2-j}$ (when such inequalities are meaningful). For instance, if N=4, k=3 and r=2, then

$$(D_{*,1}^3 w)^T = (D_{111}w, D_{112}w, D_{122}w, D_{222}w),$$

$$(D_{*,2}^3 w)^T = (D_{113}w, D_{114}w, D_{123}w, D_{124}w, D_{223}w, D_{224}w),$$

$$(D_{*,3}^3 w)^T = (D_{133}w, D_{134}w, D_{144}w, D_{233}w, D_{234}w, D_{244}w),$$

$$(D_{*,4}^3 w)^T = (D_{333}w, D_{334}w, D_{344}w, D_{444}w).$$

Theorem 14.2.4 Let $\varepsilon > 0$ and assume that Hypotheses 14.0.1 are satisfied. Then, for any k = 1, 2, 3 and any $\omega > 0$, there exists a positive constant $C = C(\omega)$, independent of ε , such that

$$||D_{*,j}^k T_{\varepsilon}(t)f||_{\infty} \le Ce^{\omega t} t^{-\frac{k+2j-2}{2}} ||f||_{\infty}, \quad t > 0, \quad j = 1, \dots, k+1, \quad (14.2.9)$$

for any $f \in C_b(\mathbb{R}^N)$ and any $\varepsilon > 0$.

Proof. In order to simplify the notation, throughout the proof, we simply write u instead of $T_{\varepsilon}(\cdot)f$. Moreover, for any function v depending on t and x we write v(t) when we want to point out only the dependence on t. Finally, we set

$$n_m^1 = \frac{1}{2}m(m+1), \qquad n_m^2 = \frac{1}{6}m(m+1)(m+2), \quad m \in \mathbb{N}.$$
 (14.2.10)

Let us introduce the function $\xi:[0,+\infty)\times\mathbb{R}^N\to\mathbb{R}$ defined by

$$\xi(t,x) = \frac{1}{2}\alpha^{3}(u(t,x))^{2} + \langle F(t)Du(t,x), Du(t,x) \rangle + \langle G(t)D_{*}^{2}u(t,x), D_{*}^{2}u(t,x) \rangle + \langle H(t)D_{*}^{3}u(t,x), D_{*}^{3}u(t,x) \rangle,$$

for any t > 0 and any $x \in \mathbb{R}^N$, where

$$F(t) = \begin{pmatrix} \alpha t I_r & 4t^2 F_1 \\ 4t^2 F_1^* & t t^3 I_{N-r} \end{pmatrix}, \tag{14.2.11}$$

$$G(t) = \begin{pmatrix} t^2 I_{n_r^1} & 0 & 0\\ 0 & \alpha^{-\frac{7}{16}} t^4 I_{r(N-r)} & \alpha^{-\frac{4}{5}} t^5 G_1\\ 0 & \alpha^{-\frac{4}{5}} t^5 G_1^* & \alpha^{-\frac{7}{8}} t^6 I_{n_{N-r}^1} \end{pmatrix},$$
(14.2.12)

$$H(t) = \begin{pmatrix} \alpha^{-\frac{7}{16}} t^3 I_{n_r^2} & 0 & 0 & 0\\ 0 & \alpha^{-\frac{7}{8}} t^5 I_{(N-r)n_r^1} & 0 & 0\\ 0 & 0 & \alpha^{-1} t^7 I_{rn_{N-r}^1} & \alpha^{-\frac{13}{12}} t^8 H_1\\ 0 & 0 & \alpha^{-\frac{13}{12}} t^8 H_1^* & \alpha^{-\frac{9}{8}} t^9 I_{n_{N-r}^2} \end{pmatrix}, (14.2.13)$$

for any t > 0. Here,

- (i) F_1 is any matrix such that $B_3F_1 + F_1^*B_3^*$ is strictly negative definite (such a matrix exists by virtue of Lemma 14.2.1);
- (ii) $-\iota = \lambda_{\max}(B_3F_1 + F_1^*B_3^*);$
- (iii) $G_1 \in L(\mathbb{R}^{n_{N-r}^1}, \mathbb{R}^{r(N-r)})$ and $H_1 \in L(\mathbb{R}^{n_{N-r-2}^2}, \mathbb{R}^{rn_{N-r-1}^1})$ are suitable matrices (with entries being independent of α) to be determined later on, as well as the positive constant α .

We require that the matrices F(t), G(t) and H(t) are strictly positive definite for any t > 0. According to Lemma 14.2.2 it suffices to assume that F(1), G(1) and H(1) are strictly positive definite and, as a straightforward computation shows, this is the case if we assume that

$$\begin{cases}
\iota \alpha - 4 \|F_1\|^2 > 0, \\
\alpha^{-\frac{7}{16}} - \alpha^{-\frac{4}{5}} \|G_1\|^2 > 0, \\
\alpha^{-\frac{17}{8}} - \alpha^{-\frac{13}{12}} \|H_1\|^2 > 0.
\end{cases}$$
(14.2.14)

From Propositions 14.1.1 and 14.1.2 it follows that the function ξ is a classical solution of the Cauchy problem

$$\begin{cases} D_t \xi(t, \cdot) = \mathcal{A}_{\varepsilon} \xi(t, \cdot) + g_{\varepsilon}(t, \cdot), \ t > 0, \\ \xi(0, \cdot) = \frac{1}{2} \alpha^3 f^2, \end{cases}$$

where $g^{(\varepsilon)} = g_1^{(\varepsilon)} + g_2 + g_3$ and

$$g_1^{(\varepsilon)}(t,x) = -\alpha^3 \langle Q^{(\varepsilon)}(x)Du(t,x), Du(t,x) \rangle$$

$$-2\text{Tr}(Q^{(\varepsilon)}(x)D^2u(t,x)F(t)D^2u(t,x))$$

$$-2\sum_{i,j=1}^N q_{ij}^{(\varepsilon)}(x)\langle G(t)D_*^2D_iu(t,x), D_*^2D_ju(t,x) \rangle$$

$$-2\sum_{i,j=1}^N q_{ij}^{(\varepsilon)}(x)\langle H(t)D_*^3D_iu(t,x), D_*^3D_ju(t,x) \rangle; \qquad (14.2.15)$$

$$g_{2}(t,x) = \langle F'(t)Du(t,x), Du(t,x) \rangle + \langle G'(t)D_{*}^{2}u(t,x), D_{*}^{2}u(t,x) \rangle$$

$$+ \langle H'(t)D_{*}^{3}u(t,x), D_{*}^{3}u(t,x) \rangle$$

$$+ \langle (BF(t) + F(t)B^{*})Du(t,x), Du(t,x) \rangle$$

$$+ 2\langle G(t)[D_{*}^{2}, \langle Bx, D \rangle]u(t,x), D_{*}^{2}u(t,x) \rangle$$

$$+ 2\langle H(t)[D_{*}^{3}, \langle Bx, D \rangle]u(t,x), D_{*}^{3}u(t,x) \rangle; \qquad (14.2.16)$$

$$g_{3}(t,x) = 2 \sum_{i,j=1}^{r} D_{ij}u(t,x) \langle F(t)Dq_{ij}(x), Du(t,x) \rangle$$

$$+2 \sum_{i,j=1}^{r} \langle G(t)[D_{*}^{2}, q_{ij}(x)D_{ij}]u(t,x), D_{*}^{2}u(t,x) \rangle$$

$$+2 \sum_{i,j=1}^{r} \langle H(t)[D_{*}^{3}, q_{ij}(x)D_{ij}]u(t,x), D_{*}^{3}u(t,x) \rangle, \quad (14.2.17)$$

for any t > 0 and any $x \in \mathbb{R}^N$.

We claim that we can fix α and the matrices G_1 and H_1 such that both $g_{\varepsilon} \leq 0$ in $(0, T_0] \times \mathbb{R}^N$ for some $T_0 > 0$, independent of ε , and the conditions (14.2.14) are satisfied. Theorem 4.1.3 then will imply that

$$\xi(t,x) \le \frac{1}{2}\alpha^3 (f(t,x))^2, \qquad t \in (0,T_0], \quad x \in \mathbb{R}^N.$$
 (14.2.18)

Since we are assuming that the matrices F(t), G(t) and H(t) are strictly positive definite for any t > 0, then, by (14.2.18), all the terms in the definition of ξ will turn out to be bounded from above by $\frac{1}{2}\alpha^3f^2$ in $(0,T_0] \times \mathbb{R}^N$ and, consequently, Lemma 14.2.2 will lead us to the estimate (14.2.9) in $(0,T_0]$ with $\omega = 0$ and C replaced with a new constant C_1 . The semigroup rule, then, will allow us to extend the previous estimate to all the positive t, as in the proof of Theorem 6.1.7. So, let us prove the claim. For this purpose, we begin by observing that, since the matrices F(t), $(\langle G(t)D_*^2D_iu(t), D_*^2D_ju(t)\rangle)_{ij}$ and

 $(\langle H(t)D_*^3D_iu(t), D_*^3D_ju(t)\rangle)_{ij}$ are strictly positive definite, then $g_1^{(\varepsilon)} \leq g_1$ in $(0, +\infty) \times \mathbb{R}^N$, where the function g_1 is obtained from $g_1^{(\varepsilon)}$ replacing, at any $x \in \mathbb{R}^N$, the matrix $Q^{(\varepsilon)}(x)$ with $Q^{(0)}(x)$. Now, Lemma 14.2.3 implies that

$$g_{1}(t) \leq -\alpha^{3} \kappa |D_{*,1} u(t)|^{2} - 2\kappa \sum_{i=1}^{r} (D^{2} u(t) F(t) D^{2} u(t))_{ii}$$
$$-2\kappa \sum_{i=1}^{r} \langle G(t) D_{*}^{2} D_{i} u(t), D_{*}^{2} D_{i} u(t) \rangle$$
$$-2\kappa \sum_{i=1}^{r} \langle H(t) D_{*}^{3} D_{i} u(t), D_{*}^{3} D_{i} u(t) \rangle.$$

Hence, using properly the inequality

$$2\alpha^{\gamma}t^{\beta}ab \le \alpha^{\gamma_1}t^{\beta_1}a^2 + \alpha^{\gamma_2}t^{\beta_2}b^2 \qquad (2\gamma = \gamma_1 + \gamma_2, \ 2\beta = \beta_1 + \beta_2), \ (14.2.19)$$

holding for any $\alpha, a, b, t > 0$, we obtain

$$\begin{split} g_{1}(t) &\leq -\alpha^{3}\kappa |D_{*,1}u(t)|^{2} - 2\alpha t\kappa \langle K_{1}D_{*,1}^{2}u(t), D_{*,1}^{2}u(t)\rangle - 2\iota\kappa t^{3}|D_{*,2}^{2}u(t)|^{2} \\ &+ \|K_{2}\|\kappa(\alpha^{\frac{1}{2}}t|D_{*,1}^{2}u(t)|^{2} + \alpha^{-\frac{1}{2}}t^{3}|D_{*,2}^{2}u(t)|^{2}) \\ &- 2t^{2}\kappa \langle K_{3}D_{*,1}^{3}u(t), D_{*,1}^{3}u(t)\rangle - 2\alpha^{-\frac{7}{16}}t^{4}\kappa \langle K_{4}D_{*,2}^{3}u(t), D_{*,2}^{3}u(t)\rangle \\ &- 2\alpha^{-\frac{7}{8}}t^{6}\kappa |D_{*,3}^{3}u(t)|^{2} + \alpha^{-\frac{3}{5}}t^{4}\|K_{5}\|\kappa |D_{*,2}^{3}u(t)|^{2} \\ &+ \alpha^{-1}t^{6}\|K_{5}\|\kappa |D_{*,3}^{3}u(t)|^{2} - 2\alpha^{-\frac{7}{16}}t^{3}\kappa \langle K_{6}D_{*,1}^{4}u(t), D_{*,1}^{4}u(t)\rangle \\ &- 2\alpha^{-\frac{7}{8}}t^{5}\kappa \langle K_{7}D_{*,2}^{4}u(t), D_{*,2}^{4}u(t)\rangle \\ &- 2\alpha^{-1}t^{7}\kappa \langle K_{8}D_{*,3}^{4}u(t), D_{*,3}^{4}u(t)\rangle - 2\alpha^{-\frac{9}{8}}t^{9}\kappa |D_{*,4}^{4}u(t)|^{2} \\ &+ \|K_{9}\|\kappa(\alpha^{-\frac{49}{48}}t^{7}|D_{*,3}^{4}u(t)|^{2} + \alpha^{-\frac{55}{48}}t^{9}|D_{*,4}^{4}u(t)|^{2}), \end{split}$$

for any t > 0, where $K_1, K_3, K_4, K_6, K_7, K_8$ are suitable diagonal matrices whose minimum eigenvalue is 1, whereas the entries of the matrices K_2 , K_5 and K_9 depend linearly only on the entries of F_1 , G_1 and H_1 , respectively. In particular, all the previous matrices are independent of α . Therefore,

$$g_{1}(t) \leq -\alpha^{3} \kappa |D_{*,1} u(t)|^{2} + (-2\alpha + \alpha^{\frac{1}{2}} ||K_{2}||) t \kappa |D_{*,1}^{2} u(t)|^{2}$$

$$+ (-2\iota + \alpha^{-\frac{1}{2}} ||K_{2}||) \kappa t^{3} |D_{*,2}^{2} u(t)|^{2}$$

$$-2t^{2} \kappa |D_{*,1}^{3} u(t)|^{2} + (-2\alpha^{-\frac{7}{16}} + \alpha^{-\frac{3}{5}} ||K_{5}||) \kappa t^{4} |D_{*,2}^{3} u(t)|^{2}$$

$$+ (-2\alpha^{-\frac{7}{8}} + \alpha^{-1} ||K_{5}||) \kappa t^{6} |D_{*,3}^{3} u(t)|^{2} - 2\alpha^{-\frac{7}{16}} t^{3} \kappa |D_{*,1}^{4} u(t)|^{2}$$

$$-2\alpha^{-\frac{7}{8}} t^{5} \kappa |D_{*,2}^{4} u(t)|^{2} + (-2\alpha^{-1} + \alpha^{-\frac{49}{48}} ||K_{9}||) \kappa t^{7} |D_{*,3}^{4} u(t)|^{2}$$

$$+ (-2\alpha^{-\frac{9}{8}} + \alpha^{-\frac{55}{48}} ||K_{9}||) \kappa t^{9} |D_{*,4}^{4} u(t)|^{2},$$

$$(14.2.20)$$

for any t > 0.

Throughout the rest of the proof, to simplify the notation, we will denote by $o_t(t^k)$ $(k \geq 0)$ any function h, depending on t, possibly on α , $\|G_1\|$ and $\|H_1\|$, but being independent of x, such that $\lim_{t\to 0} t^{-k}h(t)=0$. Moreover, in the estimates for g_2 and g_3 we will write explicitly only the terms which are not negligible, as t tends to 0^+ , with respect to the terms in the right-hand side of (14.2.20), and we use the notation now introduced to denote all the other ones. For instance, if a term is negligible with respect to $-2\iota\kappa t^3|D_{*,2}^2u(t)|^2$ we simply denote it by $o_t(t^3)\kappa|D_{*,2}^2u(t)|^2$, or by $o_t(t^3)|D_{*,2}^2u(t)|^2$, if it is independent of κ .

In order to estimate the function $g_{2,\varepsilon}$ we observe that

$$\begin{cases}
[D_*^2, \langle Bx, D \rangle] u(t, x) = \mathcal{L} D_*^2 u(t, x), \\
[D_*^3, \langle Bx, D \rangle] u(t, x) = \mathcal{M} D_*^3 u(t, x),
\end{cases} t > 0, x \in \mathbb{R}^N, (14.2.21)$$

where

$$\mathcal{L} = \begin{pmatrix} L_1 \ L_2 \ 0 \\ L_3 \ L_4 \ L_5 \\ 0 \ L_6 \ L_7 \end{pmatrix}, \qquad \mathcal{M} = \begin{pmatrix} M_1 \ M_2 \ 0 & 0 \\ M_3 \ M_4 \ M_5 & 0 \\ 0 \ M_6 \ M_7 \ M_8 \\ 0 \ 0 \ M_9 \ M_{10} \end{pmatrix}, \tag{14.2.22}$$

 L_j (j = 1, ..., 7) and M_j (j = 1, ..., 10) being suitable matrices whose entries linearly depend on the entries of B, but are independent of α, G_1, H_1 . Using properly the inequality (14.2.19), we get

$$\begin{split} &2\langle G(t)[D_{*}^{2},\langle Bx,D\rangle]u(t),D_{*}^{2}u(t)\rangle\\ &\leq 2t^{2}\|L_{1}\||D_{*,1}^{2}u(t)|^{2}+\|L_{2}\|(\alpha^{\frac{1}{2}}t|D_{*,1}^{2}u(t)|^{2}+\alpha^{-\frac{1}{2}}t^{3}|D_{*,2}^{2}u(t)|^{2})\\ &+\alpha^{-\frac{4}{5}}\|G_{1}^{*}\|\|L_{3}\|\left(t^{2}|D_{*,1}^{2}u(t)|^{2}+t^{8}|D_{*,3}^{2}u(t)|^{2}\right)\\ &+\alpha^{-\frac{7}{16}}\|L_{3}\|\left(t^{2}|D_{*,1}^{2}u(t)|^{2}+t^{6}|D_{*,2}^{2}u(t)|^{2}\right)+2\alpha^{-\frac{7}{16}}t^{4}\|L_{4}\||D_{*,2}^{2}u(t)|^{2}\\ &+2\alpha^{-\frac{4}{5}}t^{5}\|G_{1}\|\|L_{6}\||D_{*,2}^{2}u(t)|^{2}\\ &+\|L_{5}\|(\alpha^{-\frac{1}{24}}t^{3}|D_{*,2}^{2}u(t)|^{2}+\alpha^{-\frac{5}{6}}t^{5}|D_{*,3}^{2}u(t)|^{2})\\ &+\alpha^{-\frac{4}{5}}\|G_{1}\|\left(\|L_{4}\|+\|L_{7}\|\right)\left(t^{4}|D_{*,2}^{2}u(t)|^{2}+t^{6}|D_{*,3}^{2}u(t)|^{2}\right)\\ &+\alpha^{-\frac{7}{8}}t^{6}\|L_{6}\|\left(|D_{*,2}^{2}u(t)|^{2}+|D_{*,3}^{2}u(t)|^{2}\right)\\ &+\alpha^{-\frac{4}{5}}\lambda_{\max}(G_{1}^{*}L_{5}+L_{5}^{*}G_{1})t^{5}|D_{*,3}^{2}u(t)|^{2}+2\alpha^{-\frac{7}{8}}t^{6}\|L_{7}\||D_{*,3}^{2}u(t)|^{2}, \end{split}$$

that we can rewrite, with the notation introduced above, as

$$2\langle G(t)[D_*^2, \langle Bx, D\rangle]u(t), D_*^2u(t)\rangle$$

$$= \{\alpha^{\frac{1}{2}} \|L_2\| + o_t(1)\}t |D_{*,1}^2u(t)|^2$$

$$+ \{\alpha^{-\frac{1}{2}} \|L_2\| + \alpha^{-\frac{1}{24}} \|L_5\| + o_t(1)\}t^3 |D_{*,2}^2u(t)|^2$$

$$+ \{\alpha^{-\frac{4}{5}} \lambda_{\max}(G_1^* L_5 + L_5^* G_1) + \alpha^{-\frac{5}{6}} \|L_5\| + o_t(1)\}t^5 |D_{*,3}^2u(t)|^2. \quad (14.2.23)$$

Similarly,

$$\begin{split} & 2\langle H(t)[D_{*}^{3},\langle Bx,D\rangle]u(t),D_{*}^{3}u(t)\rangle \\ & \leq \{\alpha^{-\frac{1}{8}}\|M_{2}\|+o_{t}(1)\}|D_{*,1}^{3}u(t)|^{2} \\ & + \{\alpha^{-\frac{3}{4}}\|M_{2}\|+\alpha^{-\frac{1}{2}}\|M_{5}\|+o_{t}(1)\}t^{2}|D_{*,2}^{3}u(t)|^{2} \\ & + \{\alpha^{-\frac{43}{48}}\|M_{8}\|+\alpha^{-\frac{5}{4}}\|M_{5}\|+o_{t}(1)\}t^{6}|D_{*,3}^{3}u(t)|^{2} \\ & + \{\alpha^{-\frac{53}{48}}\|M_{8}\|+\alpha^{-\frac{13}{12}}\lambda_{\max}(H_{1}^{*}M_{8}+M_{8}^{*}H_{1})+o_{t}(1)\}t^{8}|D_{*,4}^{3}u(t)|^{2}. \end{split}$$

$$(14.2.24)$$

Now, observing that

$$\langle (4(B_3F_1 + F_1^*B_3^*)D_{*,2}u(t), D_{*,2}u(t)) \rangle \le -4\iota |D_{*,2}u(t)|^2, \qquad t > 0,$$

from (14.2.23) and (14.2.24) we get

$$g_{2}(t) \leq a_{1}(t)|D_{*,1}u(t)|^{2} + a_{2}(t)|D_{*,2}u(t)|^{2} + a_{3}(t)|D_{*,1}^{2}u(t)|^{2}$$

$$+a_{4}(t)|D_{*,2}^{2}u(t)|^{2} + a_{5}(t)|D_{*,3}^{2}u(t)|^{2} + a_{6}(t)|D_{*,1}^{3}u(t)|^{2}$$

$$+a_{7}(t)|D_{*,2}^{3}u(t)|^{2} + a_{8}(t)|D_{*,3}^{3}u(t)|^{2} + a_{9}(t)|D_{*,4}^{3}u(t)|^{2}, \quad (14.2.25)$$

where

$$\begin{split} a_1(t) &= \alpha + 8\alpha^{\frac{1}{2}} \|F_1\| + \alpha^{\frac{5}{2}} \|B_3\| + o_t(1), \\ a_2(t) &= \{-\iota + \alpha^{-\frac{1}{2}} (8\|F_1\| + \|B_3\|) + o_t(1)\} t^2, \\ a_3(t) &= \{4 + \alpha^{\frac{1}{2}} \|L_2\| + o_t(1)\} t, \\ a_4(t) &= \{4\alpha^{-\frac{7}{16}} + \alpha^{-\frac{1}{2}} \|L_2\| \|L_5\| + \alpha^{-\frac{1}{24}} + o_t(1)\} t^3, \\ a_5(t) &= \{6\alpha^{-\frac{7}{8}} + \alpha^{-\frac{4}{5}} \lambda_{\max}(G_1^*L_5 + L_5^*G_1) + \alpha^{-\frac{5}{6}} \|L_5\| + o_t(1)\} t^5, \\ a_6(t) &= \{15\alpha^{-\frac{7}{16}} + \alpha^{-\frac{1}{8}} \|M_2\| + o_t(1)\} t^2, \\ a_7(t) &= \{10\alpha^{-\frac{7}{8}} + \alpha^{-\frac{3}{4}} \|M_2\| + \alpha^{-\frac{1}{2}} \|M_5\| + o_t(1)\} t^4, \\ a_8(t) &= \{\alpha^{-\frac{43}{48}} \|M_8\| + 7\alpha^{-1} + \alpha^{-\frac{5}{4}} \|M_5\| + o_t(1)\} t^6, \\ a_9(t) &= \{9\alpha^{-\frac{9}{8}} + \alpha^{-\frac{53}{48}} \|M_8\| + \alpha^{-\frac{13}{12}} \lambda_{\max}(H_1^*M_8 + M_8^*H_1) + o_t(1)\} t^8. \end{split}$$

Finally, we consider the function $g_{3,\varepsilon}$ and we observe that, for any $i, j = 1, \ldots, N$,

$$\begin{cases}
[D_*^2, q_{ij}(x)D_{ij}]u(t, x) = D_{ij}u(t, x)D_*^2q_{ij}(x) + \mathcal{N}D_*^3u(t, x), \\
[D_*^3, q_{ij}(x)D_{ij}]u(t, x) = D_{ij}u(t, x)D_*^3q_{ij}(x) + \mathcal{R}(x)D_*^3u(t, x) \\
+ \mathcal{S}(x)D_*^4u(t, x),
\end{cases} (14.2.26)$$

where, for any $x \in \mathbb{R}^N$, the matrices $\mathcal{N}(x) \in L(\mathbb{R}^{n_N^2}, \mathbb{R}^{n_N^1})$, $\mathcal{P}(x) \in L(\mathbb{R}^{n_N^2})$ and $\mathcal{R}(x) \in L(\mathbb{R}^{n_N^3}, \mathbb{R}^{n_N^2})$ $(n_N^3 := N(N+1)(N+2)(N+3)/24)$, split according to the splitting of the vectors $D_*^k u$ (k=2,3,4), are given by

$$\mathcal{N}(x) = \begin{pmatrix} N_1(x) & 0 & 0 & 0 \\ N_2(x) & N_3(x) & 0 & 0 \\ 0 & N_4(x) & 0 & 0 \end{pmatrix}, \tag{14.2.27}$$

$$\mathcal{P}(x) = \begin{pmatrix} P_1(x) & 0 & 0 & 0 \\ P_2(x) & P_3(x) & 0 & 0 \\ P_4(x) & P_5(x) & 0 & 0 \\ 0 & P_6(x) & 0 & 0 \end{pmatrix}, \tag{14.2.28}$$

$$\mathcal{R}(x) = \begin{pmatrix} R_1(x) & 0 & 0 & 0 & 0 \\ R_2(x) & R_3(x) & 0 & 0 & 0 \\ 0 & R_4(x) & R_5(x) & 0 & 0 \\ 0 & 0 & R_6(x) & 0 & 0 \end{pmatrix},$$
(14.2.29)

the entries of the matrices $N_j(x)$ (j = 1, ..., 4), $P_j(x)$, $R_j(x)$ (j = 1, ..., 6) being linear combinations of the entries of the derivatives of the diffusion coefficients $q_{ij}(x)$. In particular, \mathcal{N} , \mathcal{P} and \mathcal{R} are independent of α , G_1 , H_1 and there exists a positive constant G_2 , independent of x, such that

$$||N_i(x)|| + ||P_j(x)|| + ||R_j(x)|| \le C_2 \sqrt{\kappa(x)}, \qquad x \in \mathbb{R}^N,$$
 (14.2.30)

for any $i=1,\ldots,4$ and any $j=1,\ldots,6$. Hence, using properly the inequality (14.2.19) (where now a and b are given, respectively, by $\sqrt{\kappa}|D_{*,i}^ku|$ and $|D_{*,j}^mu|$ for suitable i,j,k,m), and taking Hypothesis 14.0.1(ii) into account, it is easy

to check that

$$g_{3}(t) \leq o_{t}(1)|D_{*,1}u(t)|^{2} + o_{t}(t^{2})|D_{*,2}u(t)|^{2} + o_{t}(t)\kappa|D_{*,1}^{2}u(t)|^{2}$$

$$+o_{t}(t^{3})\kappa|D_{*,2}^{2}u(t)|^{2} + o_{t}(t^{5})|D_{*,3}^{2}u(t)|^{2} + o_{t}(t^{2})\kappa|D_{*,1}^{3}u(t)|^{2}$$

$$+o_{t}(t^{4})\kappa|D_{*,2}^{3}u(t)|^{2} + o_{t}(t^{6})\kappa|D_{*,3}^{3}u(t)|^{2} + o_{t}(t^{8})|D_{*,4}^{3}u(t)|^{2}$$

$$+o_{t}(t^{3})\kappa|D_{*,1}^{4}u(t)|^{2} + o_{t}(t^{5})\kappa|D_{*,2}^{4}u(t)|^{2} + o_{t}(t^{7})\kappa|D_{*,3}^{4}u(t)|^{2},$$

$$(14.2.31)$$

for any t > 0. Summing up, from (14.2.20), (14.2.25) and (14.2.31) we deduce that

$$\begin{split} g^{(\varepsilon)}(t) &\leq \{-\kappa\alpha^3 + o_{\alpha}(\alpha^3) + o_t(1)\}|D_{*,1}u(t)|^2 \\ &+ \{-\iota + o_{\alpha}(1) + o_t(1)\}t^2|D_{*,2}u(t)|^2 \\ &+ \{-2\alpha + o_{\alpha}(\alpha) + o_t(1)\}\kappa t|D_{*,1}^2u(t)|^2 \\ &+ \{-2\iota + o_{\alpha}(1) + o_t(1)\}\kappa t^3|D_{*,2}^2u(t)|^2 \\ &+ \{\lambda_{\max}(G_1^*L_5 + L_5^*G_1)\alpha^{-\frac{4}{5}} + o_{\alpha}(\alpha^{-\frac{4}{5}}) + o_t(1)\}t^5|D_{*,3}^2u(t)|^2 \\ &+ \{-2\kappa + o_{\alpha}(1) + o_t(1)\kappa\}t^2|D_{*,1}^3u(t)|^2 \\ &+ \{-2\alpha^{-\frac{7}{16}} + o_{\alpha}(\alpha^{-\frac{7}{16}}) + o_t(1)\}\kappa t^4|D_{*,2}^3u(t)|^2 \\ &+ \{-2\alpha^{-\frac{7}{8}} + o_{\alpha}(\alpha^{-\frac{7}{8}}) + o_t(1)\}\kappa t^6|D_{*,3}^3u(t)|^2 \\ &+ \{\alpha^{-\frac{13}{12}}\lambda_{\max}(H_1^*M_8 + M_8^*H_1) + o_{\alpha}(\alpha^{-\frac{13}{12}}) + o_t(1)\}t^8|D_{*,4}^3u(t)|^2 \\ &+ \{-2\alpha^{-\frac{7}{8}} + o_t(1)\kappa\}t^3|D_{*,1}^4u(t)|^2 \\ &+ \{-2\alpha^{-\frac{7}{8}} + o_t(1)\}\kappa t^5|D_{*,2}^4u(t)|^2 \\ &+ \{-2\alpha^{-1} + o_{\alpha}(\alpha^{-1}) + o_t(1)\}\kappa t^7|D_{*,3}^4u(t)|^2 \\ &+ \{-2\alpha^{-\frac{9}{8}} + o_{\alpha}(\alpha^{-\frac{9}{8}})\}\kappa t^9|D_{*,4}^4u(t)|^2, \end{split}$$

where by $o_{\alpha}(\alpha^k)$ $(k \in \mathbb{R})$ we have denoted any function $h:(0,+\infty) \to \mathbb{R}$, depending on α , and possibly on G_1 and H_1 , but being independent of t, such that $\lim_{\alpha \to +\infty} \alpha^{-k} h(\alpha) = 0$.

To prove that $g^{(\varepsilon)}(t,x) \leq 0$ for any t in a right neighborhood of 0 (independent of ε) and any $x \in \mathbb{R}^N$, we show that we can fix $\alpha, T_0 > 0$ and the matrices G_1 and H_1 so that all the terms in the right-hand side of (14.2.32) are negative in $(0,T_0] \times \mathbb{R}^N$. As a first step we prove that we can fix $\alpha > 0$ and the matrices G_1 and H_1 so that

$$\sup_{x \in \mathbb{R}^N} \hat{a}_j(x) < 0, \qquad j = 1, \dots, 9, \tag{14.2.33}$$

where $\hat{a}_j(x)$ are obtained from the terms in curly brackets in the right-hand side of (14.2.32), disregarding the terms of type $o_t(1)$. Once (14.2.33) is proved, it will be an easy task to check that we can fix $T_0 > 0$ such that the right-hand side of (14.2.32) is negative in $(0, T_0] \times \mathbb{R}^N$.

An easy asymptotic analysis shows that all the coefficients \hat{a}_j $(j=1,\ldots,9)$ satisfy (14.2.33) for $\alpha>0$ sufficiently large, provided that G_1 and H_1 can be chosen so that the matrices $G_1^*L_5+L_5^*G_1$ and $H_1^*M_8+M_8^*H_1$ are strictly negative definite. By virtue of Lemma 14.2.1 and Hypothesis 14.0.1(i), this is the case if the ranks of the matrices $L_5\in L(\mathbb{R}^{n_{N-r}^1},\mathbb{R}^{r(N-r)})$ and $M_8\in L(\mathbb{R}^{n_{N-r}^2},\mathbb{R}^{rn_{N-r}^1})$ are, respectively, n_{N-r}^1 and n_{N-r}^2 . Straightforward computations show that, up to rearranging the rows, we can split L_5 and M_8 into blocks (according to the splitting of the vectors D_3^*u , k=3,4) as follows:

$$L_5 = S_0, \qquad M_8 = \begin{pmatrix} S_0 & 0 & \cdots & \cdots & 0 \\ \star & S_1 & 0 & \cdots & 0 \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & 0 \\ \star & \cdots & \cdots & \star & S_{N-r-1} \end{pmatrix},$$

where $S_j \in L(\mathbb{R}^{n_{N-r-j-1}^1}, \mathbb{R}^{r(N-r-j)})$ is given by

$$S_{j} = \begin{pmatrix} B_{3}^{j} & 0 & \cdots & 0 \\ \star & B_{3}^{j+1} & 0 & \cdots & 0 \\ \vdots & \ddots & \ddots & \vdots & \vdots \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & 0 \\ \star & \cdots & \cdots & \star & B_{3}^{N-r-1} \end{pmatrix}, \qquad j = 0, \dots, N-r-1,$$

and the matrix B_3^j is obtained from B_3^* by removing the first j columns. Here and above by " \star " we denote suitable matrices whose entries depend linearly only on the entries of B_3 (recall that the entries of the matrices L_5 and M_8 are independent of α). Since, by Hypothesis 14.0.1(iii), $\operatorname{rank}(B_3) = N - r$, then $\operatorname{rank}(L_5) = n_{N-r}^1$ and $\operatorname{rank}(M_8) = n_{N-r}^2$. Hence, from Lemma 14.2.1, we deduce that, if we set $G_1 = -L_5$, and $H_1 = -M_8$, then the matrices $G_1^*L_5 + L_5^*G_1$ and $H_1^*M_8 + M_8^*H_1$ are strictly negative definite. Therefore, if $\alpha > 0$ is large enough, the estimate (14.2.33) holds. Up to choosing a larger α , we can also assume that the conditions (14.2.14) are satisfied. Then, fixing T_0 sufficiently small, but independent of ε , we obtain that $g^{(\varepsilon)} \leq 0$ in $(0, T_0] \times \mathbb{R}^N$ and F(t), G(t) and H(t) strictly positive definite for any t > 0. By the above remarks, this concludes the proof.

As in the nondegenerate case, the more the initial datum f is regular, the more we can improve the estimates of the derivatives of $T_{\varepsilon}(t)f$ near t=0. We state this fact in the following theorem.

Theorem 14.2.5 Under the same assumptions as in Theorem 14.2.4, for any k = 1, 2, 3, $h \in \mathbb{N}$ with $h \leq k$, there exist two positive constants ω and C, independent of ε , such that

$$||D_{*,j}^k T_{\varepsilon}(t)f||_{\infty} \le Ce^{\omega t} t^{-(j-h-1)^+ - \frac{k-h}{2}} ||f||_{C_b^h(\mathbb{R}^N)}, \quad t > 0, \quad j \le k+1,$$
(14.2.34)

for any $f \in C_b^h(\mathbb{R}^N)$ and any $\varepsilon > 0$. In particular, if h = k, (14.2.34) holds true for any $\omega > 0$ and some positive constant $C = C(\omega)$.

Proof. Since the proof is close to that of Theorem 14.2.4, we just sketch it, pointing out the main differences. We confine ourselves to proving (14.2.34) when h=k=3, the other cases being similar and even simpler. As in the previous proof we write u for u_{ε} , and u(t) when we want to stress the dependence on t of u but we are not interested in the dependence on the space variables.

Let us introduce the function $\xi_{3,\varepsilon}:[0,+\infty)\times\mathbb{R}^N\to\mathbb{R}$ defined by

$$\xi_3 = \frac{1}{2}\alpha^3 u^2 + \langle F D u, D u \rangle + \langle G D_*^2 u, D_*^2 u \rangle + \langle H D_*^3 u, D_*^3 u \rangle, \qquad (14.2.35)$$

where F = F(1) (with F_1 being replaced with $(\|B_4\| + 1)F_1$), G = G(1) (with $G_1 = -L_5$) and H = H(1) (with $H_1 = -M_8$) (see (14.2.11)-(14.2.13) and (14.2.22)) and α is a positive constant to be determined later on. Using Proposition 14.1.2 we can show that ξ_3 is continuous up to t = 0. Moreover, straightforward computations show that it satisfies

$$\begin{cases} D_t \xi_3(t,\cdot) = \mathcal{A}_{\varepsilon} \xi_3(t,\cdot) + \hat{g}_{\varepsilon}(t,\cdot), & t > 0, \\ \xi_3(0,\cdot) = \frac{1}{2} \alpha^3 f^2 + \langle F Df, Df \rangle + \langle G D_*^2 f, D_*^2 f \rangle + \langle H D_*^3 f, D_*^3 f \rangle, \end{cases}$$

where $\hat{g}^{(\varepsilon)} = \hat{g}_1^{(\varepsilon)} + \hat{g}_2 + \hat{g}_3$, the functions $g_1^{(\varepsilon)}$ and \hat{g}_j (j = 1, 2) being obtained from the corresponding functions $g_1^{(\varepsilon)}$ and g_j (j = 1, 2) in the proof of Theorem 14.2.4, by replacing everywhere F(t), G(t) and H(t) with F, G and H, and disregarding the terms containing the matrices F', G' and H'.

The proof now follows the same ideas as in the proof of Theorem 14.2.4. The function $\hat{g}_{1,\varepsilon}$ can be estimated, as the right-hand side of (14.2.20), by

$$\begin{split} g_1(t) &\leq -\alpha^3 \kappa |D_{*,1} u(t)|^2 + (-2\alpha + \alpha^{\frac{1}{2}} \|K_2\|) \kappa |D_{*,1}^2 u(t)|^2 \\ &+ (-2\iota + \alpha^{-\frac{1}{2}} \|K_2\|) \kappa |D_{*,2}^2 u(t)|^2 \\ &- 2\kappa |D_{*,1}^3 u(t)|^2 + (-2\alpha^{-\frac{7}{16}} + \alpha^{-\frac{3}{5}} \|K_5\|) \kappa |D_{*,2}^3 u(t)|^2 \\ &+ (-2\alpha^{-\frac{7}{8}} + \alpha^{-1} \|K_5\|) \kappa |D_{*,3}^3 u(t)|^2 - 2\alpha^{-\frac{7}{16}} \kappa |D_{*,1}^4 u(t)|^2 \end{split}$$

$$-2\alpha^{-\frac{7}{8}}\kappa|D_{*,2}^{4}u(t)|^{2} + (-2\alpha^{-1} + \alpha^{-\frac{49}{48}}||K_{9}||)\kappa|D_{*,3}^{4}u(t)|^{2} + (-2\alpha^{-\frac{9}{8}} + \alpha^{-\frac{55}{48}}||K_{9}||)\kappa t^{9}|D_{*,4}^{4}u(t)|^{2}.$$

$$(14.2.36)$$

Here, the matrices K_1, \ldots, K_8 are defined as the corresponding ones in the formula (14.2.20), with the only difference that now $G_1 = -L_5$, $H_1 = -M_8$. As far as the functions \hat{g}_2 and \hat{g}_3 are concerned, using properly the inequality (14.2.19) (with $\beta = 0$) and arguing as in the proof of (14.2.25) and (14.2.31), we get

$$\hat{g}_{2}(t) + \hat{g}_{3}(t) \leq -\iota |D_{*,2}u(t)|^{2} - 2\alpha^{-\frac{4}{5}}\lambda_{\min}(L_{5}^{*}L_{5})|D_{*,3}^{2}u(t)|^{2}
-2\lambda_{\min}(M_{8}^{*}M_{8})\alpha^{-\frac{13}{12}}|D_{*,4}^{3}u(t)|^{2} + o_{\alpha}(\alpha^{3})|D_{*,1}u(t)|^{2}
+o_{\alpha}(1)|D_{*,2}u(t)|^{2} + o_{\alpha}(\alpha)|D_{*,1}^{2}u(t)|^{2} + o_{\alpha}(1)|D_{*,2}^{2}u(t)|^{2}
+o_{\alpha}(1)|D_{*,3}^{2}u(t)|^{2} + o_{\alpha}(1)|D_{*,1}^{3}u(t)|^{2} + o_{\alpha}(\alpha^{-\frac{7}{16}})|D_{*,2}^{3}u(t)|^{2}
+o_{\alpha}(\alpha^{-\frac{7}{8}})|D_{*,3}^{3}u(t)|^{2} + o_{\alpha}(\alpha^{-\frac{13}{12}})|D_{*,4}^{3}u(t)|^{2}
+o_{\alpha}(\alpha^{-\frac{7}{16}})\kappa|D_{*,1}^{4}u(t)|^{2} + o_{\alpha}(\alpha^{-\frac{7}{8}})\kappa|D_{*,2}^{4}u(t)|^{2}
+o_{\alpha}(\alpha^{-1})\kappa|D_{*,2}^{4}u(t)|^{2},$$
(14.2.37)

for any t > 0, where, as in the proof of Theorem 14.2.4, $o_{\alpha}(\alpha^k)$ $(k \in \mathbb{R})$ denote real functions such that $\lim_{\alpha \to +\infty} \alpha^{-k} o_{\alpha}(\alpha^k) = 0$. Therefore, from (14.2.36) and (14.2.37), we obtain

$$\begin{split} \hat{g}^{(\varepsilon)}(t) &\leq -\{\alpha^{3}\kappa + o_{\alpha}(\alpha^{3})\}|D_{*,1}u(t)|^{2} - \{\iota + o_{\alpha}(1)\}|D_{*,2}u(t)|^{2} \\ &- \{2\alpha + o_{\alpha}(\alpha)\}\kappa|D_{*,1}^{2}u(t)|^{2} - \{2\iota + o_{\alpha}(1)\}\kappa|D_{*,2}^{2}u(t)|^{2} \\ &- \{2\alpha^{-\frac{4}{5}}\lambda_{\min}(L_{5}^{*}L_{5}) + o_{\alpha}(\alpha^{-\frac{4}{5}})\}|D_{*,3}^{2}u(t)|^{2} \\ &- \{2\kappa + o_{\alpha}(1)\}|D_{*,1}^{3}u(t)|^{2} - \{2\alpha^{-\frac{7}{16}} + o_{\alpha}(\alpha^{-\frac{7}{16}})\}\kappa|D_{*,2}^{3}u(t)|^{2} \\ &- \{2\alpha^{-\frac{7}{8}} + o_{\alpha}(\alpha^{-\frac{7}{8}})\}\kappa|D_{*,3}^{3}u(t)|^{2} \\ &- \{2\alpha^{-\frac{13}{12}}\lambda_{\min}(M_{8}^{*}M_{8}) + o_{\alpha}(\alpha^{-\frac{13}{12}})\}|D_{*,4}^{3}u(t)|^{2} \\ &- 2\alpha^{-\frac{7}{16}}\kappa|D_{*,1}^{4}u(t)|^{2} - 2\alpha^{-\frac{7}{8}}\kappa t^{5}|D_{*,2}^{4}u(t)|^{2} \\ &- \{2\alpha^{-1} + o_{\alpha}(\alpha^{-1})\}\kappa|D_{*,3}^{4}u(t)|^{2} - \{2\alpha^{-\frac{9}{8}} + o_{\alpha}(\alpha^{-\frac{9}{8}})\}\kappa|D_{*,4}^{4}u(t)|^{2}. \end{split}$$

$$(14.2.38)$$

Now, from (14.2.38) and the condition (14.2.14) (where we replace F_1 with $(\|B_4\|+1)F_1$), it follows that we can fix $\alpha>0$ such that F,G,H are strictly positive definite and $\hat{g}^{(\varepsilon)} \leq 0$ in $(0,+\infty) \times \mathbb{R}^N$. This implies that

$$||T_{\varepsilon}(t)f||_{C_{\varepsilon}^{3}(\mathbb{R}^{N})} \le C_{0}||f||_{C_{\varepsilon}^{3}(\mathbb{R}^{N})}, \qquad t \in (0, T_{0}], \quad f \in C_{b}^{3}(\mathbb{R}^{N}), \quad (14.2.39)$$

for some positive T_0 , and some positive constant $C = C(T_0)$. The semigroup rule then allows us to extend (14.2.39) to all the positive t obtaining (14.2.34).

To prove (14.2.34) with h, k = 1, 2, it suffices to apply the previous arguments to the functions ξ_k defined by

$$\xi_k = \frac{1}{2}\alpha^3 u^2 + \langle FDu, Du \rangle + (k-1)\langle GD_*^2 u, D_*^2 u \rangle, \qquad k = 1, 2,$$

where F and G are as in (14.2.35).

Finally, to prove (14.2.34) with (h,k) = (1,2) and with h = 1,2, k = 3, it suffices to apply the previous arguments, respectively, to the functions $\xi_{1,2}$, $\xi_{1,3}$ and $\xi_{2,3}$ defined by

$$\xi_{1,k}(t,x) = \xi_1(t,x) + \langle G(t)D_*^2 u(t,x), D_*^2 u(t,x) \rangle + (k-1)\langle H_1(t)D_*^3 u(t,x), D_*^3 u(t,x) \rangle, \quad k = 2, 3,$$

for any t > 0, $x \in \mathbb{R}^N$, and

$$\xi_{2,3}(t,x) = \xi_2(t,x) + \langle H_2(t)D_*^3 u(t,x), D_*^3 u(t,x) \rangle, \qquad t > 0, \quad x \in \mathbb{R}^N,$$

where

$$G(t) = \begin{pmatrix} tI_{n_r^1} & 0 & 0 \\ 0 & \alpha^{-\frac{7}{16}}tI_{(N-r)n_r^1} & -\alpha^{-\frac{4}{5}}t^2L_5 \\ 0 & -\alpha^{-\frac{4}{5}}t^2L_5^* & \alpha^{-\frac{7}{8}}t^3I_{n_N^1} \end{pmatrix},$$

$$H_1(t) = \begin{pmatrix} \alpha^{-\frac{7}{16}} t^2 I_{n_r^2} & 0 & 0 & 0 \\ 0 & \alpha^{-\frac{7}{8}} t^2 I_{(N-r)n_r^1} & 0 & 0 \\ 0 & 0 & \alpha^{-1} t^4 I_{rn_{N-r}^1} - \alpha^{-\frac{13}{12}} t^5 M_8 \\ 0 & 0 & -\alpha^{-\frac{13}{12}} t^5 M_8^* & \alpha^{-\frac{9}{8}} t^6 I_{n_{N-r}^2} \end{pmatrix},$$

$$H_2(t) = \begin{pmatrix} \alpha^{-\frac{7}{16}}tI_{n_r^2} & 0 & 0 & 0 \\ 0 & \alpha^{-\frac{7}{8}}tI_{(N-r)n_r^1} & 0 & 0 \\ 0 & 0 & \alpha^{-1}tI_{rn_{N-r}^1} & -\alpha^{-\frac{13}{12}}t^2M_8 \\ 0 & 0 & -\alpha^{-\frac{13}{12}}t^2M_8^* & \alpha^{-\frac{9}{8}}t^3I_{n_{N-r}^2} \end{pmatrix},$$

for any t > 0. Here, as in the proof of Theorem 14.2.4, n_m^1 and n_m^2 $(m \in \mathbb{N})$ are given by (14.2.10).

14.2.3 Construction of the semigroup

In this section we prove that, for any $f \in C_b(\mathbb{R}^N)$, the Cauchy problem (14.0.5), associated with the degenerate elliptic operator \mathcal{A} , admits a unique classical solution u_f . This will allow us to define a semigroup of bounded linear operators in $C_b(\mathbb{R}^N)$ by setting $T(t)f = u_f(t, \cdot)$ for any t > 0 and any $f \in C_b(\mathbb{R}^N)$. At the same time we will also show that the semigroup $\{T(t)\}$ satisfies the uniform estimates (14.2.1)-(14.2.6).

The following theorem will be fundamental in order to prove our results, since it provides us a useful maximum principle for the classical solution to the Cauchy problem (14.0.5).

Theorem 14.2.6 Assume Hypotheses 4.0.1 and 4.0.2. Fix T > 0, $f \in C(\mathbb{R}^N)$, $g: (0,T] \times \mathbb{R}^N \to \mathbb{R}$. Further suppose that the function $u \in C([0,T] \times \mathbb{R}^N)$ is such that $D_t u \in C((0,T] \times \mathbb{R}^N)$, $u(t,\cdot) \in W^{2,p}_{loc}(\mathbb{R}^N)$, $Au(t,\cdot) \in C(\mathbb{R}^N)$ for any $t \in (0,T]$ and any $p \in [1,+\infty)$ and it solves the Cauchy problem

$$\begin{cases}
D_t u(t,x) - \mathcal{A}u(t,x) = g(t,x), & t \in (0,T], x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N.
\end{cases}$$
(14.2.40)

Then, the following properties are met:

(i) if

$$\sup_{x \in \mathbb{R}^N} f(x) < +\infty, \qquad g \le 0 \text{ in } (0, T] \times \mathbb{R}^N,$$

and u satisfies

$$\lim_{|x| \to +\infty} \sup_{t \in [0,T]} \frac{u(t,x)}{\varphi(x)} \le 0, \tag{14.2.41}$$

then

$$u(t,x) \leq \max\left\{0, \sup_{\mathbb{R}^N} f\right\}, \quad t \in [0,T], \ x \in \mathbb{R}^N.$$

(ii) If

$$\inf_{x \in \mathbb{R}^N} f(x) > -\infty, \qquad g \ge 0 \text{ in } (0, T] \times \mathbb{R}^N,$$

u satisfies

$$\lim_{|x| \to +\infty} \inf \left(\inf_{t \in [0,T]} \frac{u(t,x)}{\varphi(x)} \right) \ge 0,$$

then

$$u(t,x) \geq \min\left\{0, \inf_{\mathbb{R}^N} f\right\}, \quad \ t \in [0,T], \ \ x \in \mathbb{R}^N.$$

(iii) In particular, for any $f \in C(\mathbb{R}^N)$ such that

$$\lim_{|x| \to +\infty} \frac{f(x)}{\varphi(x)} = 0,$$

there exists at most one solution of the parabolic problem (4.1.7) (with $g \equiv 0$) in $C([0, +\infty) \times \mathbb{R}^N)$ such that $D_t u \in C((0, T] \times \mathbb{R}^N)$, $u(t, \cdot) \in W^{2,p}_{loc}(\mathbb{R}^N)$, $\mathcal{A}u(t, \cdot) \in C(\mathbb{R}^N)$ for any $p \in [1, +\infty)$ and any $t \in (0, T]$, and such that (14.2.41) holds for any T > 0.

Proof. The proof can be obtained arguing just as in the proof of Theorem 4.1.3. Indeed, such a proof can be repeated with no changes also in the case when the operator \mathcal{A} is elliptic but not uniformly elliptic. The essential tool needed in the proof of Theorem 4.1.3 is the existence of a Lyapunov function $\varphi \in C^2(\mathbb{R}^N)$ such that $\sup_{\mathbb{R}^N} (\mathcal{A}\varphi - \lambda\varphi) \leq 0$ for some $\lambda > 0$. In our situation, using (14.1.3) it is easy to check that the function φ defined by $\varphi(x) = 1 + |x|^2$ for any $x \in \mathbb{R}^N$ is a Lyapunov function for the operator \mathcal{A} corresponding to $\lambda = 2(\sqrt{N}\hat{C} + \|B\|)$, where \hat{C} is given by (14.1.3).

We can now prove the main result of this section.

Theorem 14.2.7 Under Hypotheses 14.0.1, for any $f \in C_b(\mathbb{R}^N)$ there exists a unique classical solution u to the problem (14.0.5). Moreover, if we set $u(t,\cdot) = T(t)f$ for any t > 0, the family $\{T(t)\}$ is an order preserving semigroup of contractions in $C_b(\mathbb{R}^N)$ satisfying the estimates (14.2.1)-(14.2.6). Finally, for any compact set $D \subset (0, +\infty) \times \mathbb{R}^N$ and any $f \in C_b(\mathbb{R}^N)$, $T_{\varepsilon}(\cdot)f$ converges to $T(\cdot)f$ in $C^{1,2}(D)$ as ε goes to 0^+ .

Proof. We split the proof into several steps. First in Steps 1 to 3 we prove that, for any $f \in C_b(\mathbb{R}^N)$, the problem (14.0.5) admits a (unique) classical solution u_f ; we define the operator T(t) (t > 0) and we show that $T_{\varepsilon}(\cdot)f$ converges to $T(\cdot)f$ in $C^{1,2}(D)$ as ε tends to 0^+ for any compact set $D \subset (0,+\infty)\times\mathbb{R}^N$. Moreover, we show that $\{T(t)\}$ is an order preserving semigroup of contractions in $C_b(\mathbb{R}^N)$ and it satisfies (14.2.1), (14.2.2), (14.2.4) (with k = 1,2) and (14.2.5). Then, in Steps 4 and 5, we show that u_f is thrice-continuously differentiable with respect to the space variables in $(0,+\infty)\times\mathbb{R}^N$ and T(t) satisfies the estimates (14.2.3), (14.2.4) (with k = 3) and (14.2.6) (with k = 0,1,2), for any t > 0.

Step 1. For any $f \in C_b(\mathbb{R}^N)$ and any $\varepsilon \in (0,1]$, we set $u_{\varepsilon} = T_{\varepsilon}(\cdot)f$. Using the estimates (14.2.1) and (14.2.2), we deduce that the family of functions $\{u_{\varepsilon} : \varepsilon \in (0,1)\}$ is contained in $B([T_0,T];C_b^3(\mathbb{R}^N))$, for any $0 < T_0 < T$, with norm being independent of ε . Moreover, since $D_t u_{\varepsilon} = \mathcal{A}_{\varepsilon} u_{\varepsilon}$, then $\{D_t u_{\varepsilon} : \varepsilon \in (0,1)\}$ is equibounded in $[T_0,T] \times \overline{B}(R)$ for any R > 0. It follows that $u_{\varepsilon} \in \text{Lip}([T_0,T];C(\overline{B}(R))) \cap B([T_0,T] \times C^3(\overline{B}(R)))$. From Propositions A.4.4 and A.4.6, we deduce that $u_{\varepsilon} \in C^{(1-\alpha)/3}([T_0,T];C^{2+\alpha}(\overline{B}(R)))$ and $D_t u_{\varepsilon} \in C^{(1-\alpha)/3}([T_0,T];C^{2+\alpha}(\overline{B}(R)))$

 $C^{(1-\alpha)/3}([T_0,T];C(\overline{B}(R)))$ with norms being independent of ε . It follows that the families of functions $\{D_t^{\alpha}D_x^{\beta}u_{\varepsilon}:\varepsilon\in(0,1)\}\ (2\alpha+|\beta|\leq 2)$ are equibounded and equicontinuous in $[T_0,T]\times\overline{B}(R)$ for any $0< T_0< T$ and any R>0. Hence, there exists an infinitesimal sequence $\{\varepsilon_n\}$ such that, for any compact set $D\subset(0,+\infty)\times\mathbb{R}^N$, u_{ε_n} converges in $C^{1,2}(D)$ to some function u_f , which, of course, solves the differential equation $D_tu_f-\mathcal{A}u_f=0$.

We now assume that $f \in C_c^2(\mathbb{R}^N)$ and prove that u_f is continuous up to t = 0. For this purpose we observe that, since $C_c^2(\mathbb{R}^N)$ is contained in the domain of the weak generator of the operator $\mathcal{A}_{\varepsilon}$ (see Propositions 2.3.6 and 4.1.10), by Lemma 2.3.3 and Proposition 2.3.5 we can write

$$(T_{\varepsilon_n}(t)f)(x) - f(x) = \int_0^t (D_t T_{\varepsilon_n}(s)f)(x)ds$$
$$= \int_0^t (\mathcal{A}_{\varepsilon_n} T_{\varepsilon_n}(s)f)(x)ds$$
$$= \int_0^t (T_{\varepsilon_n}(s)\mathcal{A}_{\varepsilon_n}f)(x)ds,$$

for any t > 0 and any $x \in \mathbb{R}^N$. Using (14.1.2) we get

$$||T_{\varepsilon_n}(t)f - f||_{\infty} \leq \sup_{x \in \mathbb{R}^N} \int_0^t |(T_{\varepsilon_n}(s)\mathcal{A}_{\varepsilon_n}f)(x)| ds \leq ||\mathcal{A}_{\varepsilon_n}f||_{\infty}t \leq Ct||f||_{C_b^2(\mathbb{R}^N)},$$

for any t > 0 and any $n \in \mathbb{N}$, where C is a positive constant, independent of n. Letting n go to $+\infty$, we deduce that $u_f(t,\cdot)$ tends to f uniformly as t tends to 0^+ . Hence, the function u_f is a classical solution to the problem (14.0.5).

Repeating the same arguments as above and taking Theorem 14.2.6 into account, we can show that any sequence $\{u_{\varepsilon'_n}\}$, with ε'_n vanishing as n go to $+\infty$, admits a subsequence converging to u_f in $C^{1,2}(D)$ for any compact set $D \subset (0, +\infty) \times \mathbb{R}^N$. This implies that $T_{\varepsilon}(\cdot)f$ converges to u_f in $C^{1,2}(D)$ as ε tends to 0^+ for any D as above.

Step 2. We now assume that $f \in C_0(\mathbb{R}^N)$ and let $u_{\varepsilon_n} = T_{\varepsilon_n}(\cdot)f$ be as in Step 1. Moreover, we denote by $\{f_m\} \in C_c^2(\mathbb{R}^N)$ any sequence converging to f uniformly in \mathbb{R}^N . We fix $m \in \mathbb{N}$ and observe that, by Step 1, $T_{\varepsilon_n}(t)f_m$ converges to $T(t)f_m$ as n tends to $+\infty$, locally uniformly in \mathbb{R}^N and for any t > 0. According to (14.1.2),

$$||T_{\varepsilon_n}(t)f - T_{\varepsilon_n}(t)f_m||_{\infty} \le ||f - f_m||_{\infty}, \quad t > 0, \ n, m \in \mathbb{N}.$$

So, letting n go to $+\infty$, we get

$$||u_f(t,\cdot) - T(t)f_m||_{\infty} \le ||f - f_m||_{\infty}, \quad t > 0, \quad m \in \mathbb{N}.$$
 (14.2.42)

Hence, from (14.2.42) it follows that

$$||u_f(t,\cdot) - f||_{\infty} \le ||u_f(t,\cdot) - T(t)f_m||_{\infty} + ||T(t)f_m - f_m||_{\infty} + ||f_m - f||_{\infty}$$

$$\le 2||f - f_m||_{\infty} + ||T(t)f_m - f_m||_{\infty},$$

for any t > 0 and any $m \in \mathbb{N}$, and, from Step 1, we deduce that $u_f(t, \cdot)$ tends to f uniformly in \mathbb{R}^N as t tends to 0^+ . Hence, the function u_f is a classical solution to the problem (14.0.5). Then, with the same arguments as in Step 1 we can easily show that $T_{\varepsilon}(\cdot)f$ converges to $T(\cdot)f$ in $C^{1,2}(D)$ as ε goes to 0^+ , for any compact set $D \subset (0, +\infty) \times \mathbb{R}^N$.

Step 3. We now assume that $f \in C_b(\mathbb{R}^N)$ and adapt to our situation the technique in the proof of Theorem 11.2.1. We preliminarily observe that by Theorem 4.1.3, $T_{\varepsilon}(t)\mathbb{1} = \mathbb{1}$ for any t > 0. Hence, $T(t)\mathbb{1}$ is well defined for any t > 0 and $T(\cdot)\mathbb{1} = \mathbb{1}$.

We now fix a compact set $K \subset \mathbb{R}^N$ and a smooth function η , compactly supported in \mathbb{R}^N , such that $\eta = 1$ in K and $0 \le \eta \le 1$. By linearity and Step 2, we easily see that for any t > 0, the function $T(t)(1-\eta)$ is well defined. Moreover, since $\{T_{\varepsilon}(t)\}$ is an order preserving semigroup, $T_{\varepsilon}(t)(1-\eta) \ge 0$ for any t > 0 and any $\varepsilon \in (0,1)$. Thus,

$$0 \le T(t)(1-\eta) = 1 - T(t)\eta, \qquad t > 0.$$

Since, by Step 2, $T(t)\eta$ tends to η , uniformly in \mathbb{R}^N , then $T(t)(1-\eta)$ tends to 0 as t tends to 0^+ , uniformly in K.

Now, let $u_{\varepsilon_n} = T_{\varepsilon_n}(\cdot)f$ and u_f be as in Step 1. Splitting $T_{\varepsilon_n}(t)(1-\eta)f) = T_{\varepsilon_n}(t)f - T_{\varepsilon_n}(\eta f)$, it is immediate to check that $T_{\varepsilon_n}(\cdot)((1-\eta)f)$ converges in $C^{1,2}(D)$ to $u_f(t,\cdot) - T(t)(\eta f)$, for any compact set $D \subset (0,+\infty) \times \mathbb{R}^N$. Since

$$|(T_{\varepsilon_n}(t)((\mathbb{1}-\eta)f))(x)| \le ||f||_{\infty} (T_{\varepsilon_n}(t)(\mathbb{1}-\eta))(x), \qquad t > 0, \ x \in \mathbb{R}^N, \ n \in \mathbb{N},$$

for any $n \in \mathbb{N}$, then

$$|u_f(t,x) - (T(t)(\eta f))(x)| \le ||f||_{\infty} (T(t)(1-\eta))(x), \qquad t > 0, \quad x \in \mathbb{R}^N.$$

It follows that $u_f(t,\cdot) - T(t)(\eta f)$ vanishes uniformly in K, as t tends to 0^+ . Recalling that $T(t)(\eta f)$ tends to f, uniformly, in K as t tends to 0^+ , we easily deduce that $u_f(t,\cdot)$ converges to f, as t tends to 0^+ , uniformly in K, as well. By the arbitrariness of K, u_f is continuous up to t=0 and it is a classical solution to the problem (14.0.5). Arguing again as in Step 1, we can then show that $T_{\varepsilon}(\cdot)f$ converges to $T(\cdot)f$ in $C^{1,2}(D)$ as ε goes to 0^+ , for any compact set $D \subset (0, +\infty) \times \mathbb{R}^N$.

Now, the estimates (14.2.1), (14.2.2), (14.2.4) (with k=1,2) and (14.2.5) easily follow letting ε go to 0^+ in (14.2.9) and (14.2.34), recalling that the constant occurring in these latter estimates are independent of ε .

To conclude the proof of this step, we observe that the family of contractive operators $\{T(t)\}$ (T(0) = I) is an order preserving semigroup of linear operators in $C_b(\mathbb{R}^N)$. The semigroup rule follows from Theorem 14.2.6 since, for any $f \in C_b(\mathbb{R}^N)$ and any s > 0, both the functions $u(t, \cdot) = T(t)T(s)f$ and $v(t, \cdot) = T(t+s)f$ are both classical solutions to the Cauchy problem (14.0.5) with f replaced with T(s)f. The maximum principle also implies that $\{T(t)\}$

is an order preserving semigroup, and, since $T_{\varepsilon}(t)$ are all contractions, then T(t) is a contraction as well, for any $t \geq 0$.

Step 4. We now prove (14.2.3), (14.2.4) (with k=3) and (14.2.6) (in the case when $i \leq r$), using a localization argument. Without loss of generality, we can assume that $f \in C_b^3(\mathbb{R}^N)$. Indeed, once (14.2.3) and (14.2.6) are established for any $f \in C_b^3(\mathbb{R}^N)$, then they can be extended, respectively, to the cases when $f \in BUC(\mathbb{R}^N)$ and $f \in BUC^k(\mathbb{R}^N)$ (k=0,1,2) by a density argument, by approximating $f \in BUC^k(\mathbb{R}^N)$ with a sequence of functions in $C_b^3(\mathbb{R}^N)$, converging uniformly to f in $BUC^k(\mathbb{R}^N)$. Finally, for a general $f \in C_b^k(\mathbb{R}^N)$, it suffices to split, for any t > 0, T(t)f = T(t/2)T(t/2)f and apply the above results with f replaced with $T(t/2)f \in BUC^k(\mathbb{R}^N)$.

Observe that, to prove the previous estimates, it suffices to show that, for any $i \in \{1, ..., r\}$ and any $j, h \in \{1, ..., N\}$, the function $D_{jh}T(t)f$ is continuously differentiable with respect to the i-th space variable. Indeed, suppose, for instance, that $i = 1, j \le r, h > r$ and $f \in C_b(\mathbb{R}^N)$. The estimate (14.2.9) (with k = 3) implies that, for any $x_2, ..., x_N$, the function $x \mapsto (D_{jh}T_{\varepsilon}(t)f)(x, x_2, ..., x_N)$ is Lipschitz continuous in \mathbb{R} and

$$[(D_{jh}T_{\varepsilon}(t)f)(\cdot, x_2, \dots, x_N)]_{\operatorname{Lip}(\mathbb{R})} \le Ce^{\omega t}t^{-\frac{5}{2}} ||f||_{C_b(\mathbb{R}^N)}, \qquad t > 0,$$

for suitable $C, \omega > 0$, independent of x_2, \ldots, x_N . Since $D_{jh}T_{\varepsilon}(t)f$ converges to $D_{jh}T(t)f$ locally uniformly, then the function $(D_{jh}T(t)f)(\cdot, x_2, \ldots, x_N)$ is Lipschitz continuous in \mathbb{R} as well, and

$$[(D_{jh}T(t)f)(\cdot, x_2, \dots, x_N)]_{\text{Lip}(\mathbb{R})} \le Ce^{\omega t}t^{-\frac{5}{2}}||f||_{C_b(\mathbb{R}^N)}, \qquad t > 0.$$

Therefore, if $D_{jh}T(t)f$ is continuously differentiable with respect to the direction e_1 , then the function $D_{1jh}T(t)f$ satisfies (14.2.6).

So, we fix $i \leq r$ and $j, h \leq N$, and prove that the function $D_{jh}u = D_{jh}T(\cdot)f$ is continuously differentiable in $(0, +\infty) \times \mathbb{R}^N$ with respect to the *i*-th variable. For this purpose, let $\eta_R : \mathbb{R}^N \to \mathbb{R}$ be a smooth function compactly supported in B(R) (R > 0) such that $\eta_R \equiv 1$ in B(R/2). For any $k \in \mathbb{R}$, with $|k| \leq 1$ we introduce the operator τ_k^h defined on $C_b(\mathbb{R}^N)$ by

$$\tau_k^h \psi(x) = \frac{\psi(x + ke_h) - \psi(x)}{k}, \qquad x \in \mathbb{R}^N, \qquad \psi \in C_b(\mathbb{R}^N).$$

Moreover, we set $v_{\varepsilon,k,R}^h = \tau_k^h v_{\varepsilon,R}$ where $v_{\varepsilon,R} = u_{\varepsilon} \eta_R$ and $u_{\varepsilon} = T_{\varepsilon}(\cdot)f$. As it is easily seen, the function $v_{\varepsilon,k,R}^h$ is the classical solution to the Cauchy problem

$$\begin{cases} D_t v_{\varepsilon,k,R}^h(t,x) = \mathcal{A}_{\varepsilon} v_{\varepsilon,k,R}^h(t,x) + g_{\varepsilon,k,R}^h(t,x), & t > 0, x \in \mathbb{R}^N, \\ v_{\varepsilon,h,R}^k(0,x) = \tau_k^h(\eta_R f)(x), & x \in \mathbb{R}^N, \end{cases}$$

where, for any t > 0,

$$g_{\varepsilon,k,R}^{h}(t,\cdot) = -\tau_{k}^{h}(u_{\varepsilon}(t,\cdot)\mathcal{A}_{\varepsilon}\eta_{R}) - 2\sum_{l,m=1}^{r} q_{lm}D_{l}u_{\varepsilon}(t,\cdot+ke_{h})\tau_{k}^{h}(D_{m}\eta_{R})$$

$$-2\sum_{l,m=1}^{r} q_{lm}(\tau_{k}^{h}D_{l}u_{\varepsilon}(t,\cdot))D_{m}\eta_{R}$$

$$-2\sum_{l,m=1}^{r} (\tau_{k}^{h}q_{lm})D_{l}u_{\varepsilon}(t,\cdot+ke_{h})D_{m}\eta_{R}(\cdot+ke_{h})$$

$$-2\varepsilon\sum_{m=r+1}^{N} D_{m}u_{\varepsilon}(t,\cdot+ke_{h})(\tau_{k}^{h}D_{m}\eta_{R})$$

$$-2\varepsilon\sum_{m=r+1}^{N} (\tau_{k}^{h}D_{m}u_{\varepsilon}(t,\cdot))D_{m}\eta_{R}$$

$$+\sum_{l=1}^{r} (\tau_{k}^{h}q_{lm})D_{lm}v_{\varepsilon,R}(t,\cdot) + \sum_{l=1}^{N} b_{lh}D_{l}v_{\varepsilon,R}(t,\cdot). \quad (14.2.43)$$

By the results of Chapter 6, $v^h_{\varepsilon,k,R}$ can be represented by the variation-of-constants formula

$$v_{\varepsilon,k,R}^h(t,x) = (T_{\varepsilon}(t)(\tau_k^h(\eta_R f)))(x) + \int_0^t (T_{\varepsilon}(t-s)g_{\varepsilon,k,R}^h(s,\cdot))(x)ds, \quad (14.2.44)$$

for any t > 0 and any $x \in \mathbb{R}^N$. We are going to show that we can take the limit as ε tends to 0^+ in (14.2.44) and write

$$v_{k,R}^{h}(t,x) := (\tau_{k}^{h}(u\eta_{R})(t,\cdot))(x)$$
$$= (T(t)(\tau_{k}^{h}(\eta_{R}f)))(x) + \int_{0}^{t} (T(t-s)g_{k,R}^{h}(s,\cdot))(x)ds, \quad (14.2.45)$$

for any t>0 and any $x\in\mathbb{R}^N$, where $g_{k,R}^h$ is obtained from $g_{\varepsilon,k,R}^h$ by replacing u_ε with u and letting $\varepsilon=0$ in (14.2.43). Of course, thanks to the previous steps, it suffices to deal with the convolution term in (14.2.44). Since u_ε converges to u in $C^{1,2}(D)$ for any compact set $D\subset (0,+\infty)\times\mathbb{R}^N$ and η_R is compactly supported in \mathbb{R}^N , then the continuous function $g_{\varepsilon,k,R}^h$ converges uniformly in \mathbb{R}^N to the function $g_{k,R}^h$ as ε tends to 0^+ . This implies that, for any s,t>0, $T_\varepsilon(t)g_{\varepsilon,k,R}^h(s,\cdot)$ converges to $T(t)g_{k,R}^h(s,\cdot)$ locally uniformly in \mathbb{R}^N as ε tends to 0^+ . Indeed, for any compact set $K\subset\mathbb{R}^N$, we have

$$\sup_{x \in K} |(T_{\varepsilon}(t)g_{\varepsilon,k,R}^{h}(s,\cdot))(x) - (T(t)g_{k,R}^{h}(s,\cdot))(x)|$$

$$\leq \sup_{x \in K} |(T_{\varepsilon}(t)g_{\varepsilon,k,R}^{h}(s,\cdot))(x) - (T_{\varepsilon}(t)g_{k,R}^{h}(s,\cdot))(x)|$$

$$+ \sup_{x \in K} |(T_{\varepsilon}(t)g_{k,R}^{h}(s,\cdot))(x) - (T(t)g_{k,R}^{h}(s,\cdot))(x)|$$

$$\leq \|g_{\varepsilon,k,R}^{h} - g_{k,R}^{h}\|_{\infty} + \sup_{x \in K} |(T_{\varepsilon}(t)g_{k,R}^{h}(s,\cdot))(x) - (T(t)g_{k,R}^{h}(s,\cdot))(x)|,$$

and, by virtue of Step 1, the last side of the previous chain of inequalities vanishes as ε tends to 0^+ . Moreover, since $f \in C_b^3(\mathbb{R}^N)$ and $\{T_{\varepsilon}(t)\}$ is a semigroup of contractions for any $\varepsilon > 0$, then $T_{\varepsilon}(\cdot)g_{\varepsilon,k,R}^h$ is bounded in $[0,T] \times \mathbb{R}^N$ for any T > 0, uniformly with respect to $\varepsilon \in (0,1)$. Therefore, letting ε go to 0^+ in (14.2.44), by the dominated convergence theorem we get (14.2.45).

Next step consists in showing that we can let k go to 0 in (14.2.45) getting the fundamental representation formula

$$D_h v_R(t,x) = (T(t)(D_h(\eta_R f)))(x) + \int_0^t (T(t-s)g_R^h(s,\cdot))(x)ds, \quad (14.2.46)$$

for any t > 0 and any $x \in \mathbb{R}^N$, where

$$g_{R}^{h} = -D_{h}(u\mathcal{A}\eta_{R}) - 2\sum_{l,m=1}^{r} q_{lm}D_{l}uD_{hm}\eta_{R} - 2\sum_{l,m=1}^{r} q_{lm}D_{hl}uD_{m}\eta_{R}$$
$$-2\sum_{l,m=1}^{r} (D_{h}q_{lm})D_{l}uD_{m}\eta_{R} + \sum_{l,m=1}^{r} (D_{h}q_{lm})D_{lm}v_{R} + \sum_{l=1}^{N} b_{hl}D_{l}v_{R},$$

$$(14.2.47)$$

and $v_R = u\eta_R$. The convergence of the integral term in (14.2.45) follows from the dominated convergence theorem, since $g_{k,R}^h$ converges uniformly in \mathbb{R}^N , to the function g_R^h , as k tends to 0. To show this, it suffices to observe that, for any $\psi \in C(\mathbb{R}^N)$ such that $D_h \psi \in C(\mathbb{R}^N)$, the function $\tau_k^h \psi$ converges to $D_h \psi$, locally uniformly in \mathbb{R}^N , as k tends to 0. Similarly, $\tau_k^h (\eta_R f)$ and $v_{k,R}^h (t,\cdot)$ (t>0) converge uniformly, respectively, to $D_h (\eta_R f)$ and $D_h v_R(t,\cdot)$ as k tends to 0.

Now, taking advantage of the representation formula (14.2.46), we can show that the function $D_h v_R$ is twice continuously differentiable in $(0,T) \times \mathbb{R}^N$ with respect to the *i*-th and *j*-th space variable, for any T > 0. To simplify the notation, in the rest of the proof, we denote by C_j positive constants which may depend on R, but are independent of t, ε and k.

Using a simple interpolation argument, from (14.2.34) (with h = k = 2, 3) we deduce that $T_{\varepsilon}(\cdot)f \in B((0,T); C_b^{2+\alpha}(\mathbb{R}^N))$ with norm being independent of ε . Therefore, letting ε go to 0^+ , we get $T(\cdot)f \in B((0,T); C_b^{2+\alpha}(\mathbb{R}^N))$ as well and, consequently, $g_R^h \in B((0,T); C_b^{\alpha}(\mathbb{R}^N))$ for any $\alpha \in (0,1)$.

Next, interpolating the estimates (14.2.2) and (14.2.5) we get

$$||D_{ij}T(t)\psi||_{\infty} \le C_1 t^{-\frac{3}{4}} ||\psi||_{C_{k}^{5/6}(\mathbb{R}^{N})}, \tag{14.2.48}$$

for any $\psi \in C_b^{\alpha}(\mathbb{R}^N)$. The estimate (14.2.48) implies that the function $s \mapsto \|D_{ij}T(t-s)g_R^{h}(s,\cdot)\|_{\infty}$ is in $L^1(0,t)$. Hence, (14.2.46) and Step 3 imply that

the function $D_h v_R$ is twice continuously differentiable with respect to the *i*-th and *j*-th space variables in $(0,T) \times \mathbb{R}^N$. Since $v_R \equiv u$ in B(R/2) and R is arbitrarily fixed, we deduce that $D_h u$ is twice continuously differentiable with respect to the *i*-th and *j*-th space variables as well. The estimates (14.2.3) and (14.2.4) (with k = 3) follow.

Step 5. We now prove the estimate (14.2.6) in the case when i, j, h > r. The same arguments as in Step 4 show that it is not restrictive to consider the case when $f \in C_b^4(\mathbb{R}^N)$. We are going to show that the right-hand side of (14.2.46) defines a function which is twice continuously differentiable in $(0,+\infty)\times\mathbb{R}^N$, with respect to space variables x_i and x_j . As a first step, we show that the function g_R^h in (14.2.47) belongs to $B((0,T); C_b^{3/2}(\mathbb{R}^N))$ for any T>0. Of course, this is the case if $D_{lm}u\in B((0,T); C^{3/2}(\overline{B}(R)))$ for any $l,m=1,\ldots,r$. To show that $D_{lm}u\in B((0,T); C^{3/2}(\overline{B}(R)))$ we use a bootstrap argument, first showing that it belongs to $B((0,T); C_b^{1+\theta/3}(B(2R)))$ for any $\theta\in(0,1)$. For this purpose, we replace the function η_R defined in Step 4, with the function η_{4R} . Interpolating the estimates (14.2.5) and (14.2.6) (with k=1) we get

$$||D_{lm}T(t)\psi||_{C_b^{\theta/3}(\mathbb{R}^N)} \le C_2 t^{-\frac{\theta+1}{2}} ||\psi||_{C_b^1(\mathbb{R}^N)}, \qquad t \in (0,T), \quad 1 \le l \le r,$$

for any $\psi \in C_b^1(\mathbb{R}^N)$. Hence, the function $t \mapsto \|D_{lm}T(t)\psi\|_{C_b^{\theta/3}(\mathbb{R}^N)}$ is integrable in (0,T). Therefore, since by Step 4, $T(\cdot)\psi$ and $D_{lm}T(\cdot)\psi$ both belong to $B((0,T);C_b^1(\mathbb{R}^N))$ for any $l=1,\ldots,r$, it follows, first, that $g_{4R}^h \in B((0,T);C_b^1(\mathbb{R}^N))$ and, then, using the formula (14.2.46), that $D_{lm}u$ belongs to $B((0,T);C^{1+\theta/3}(\overline{B}(R)))$. As a straightforward consequence, from (14.2.47) (with R replaced with 2R) we deduce that $g_{2R}^h \in B((0,T);C_b^{1+\theta/3}(\mathbb{R}^N))$ for any $\theta \in (0,1)$. Now, we interpolate first (14.2.6), respectively, with k=1 and k=2, and then (14.2.2) and (14.2.5), obtaining

$$||D_{lm}T(t)\psi||_{C_b^1(\mathbb{R}^N)} \le C_3 t^{\frac{\theta}{2}-2} ||\psi||_{C_b^{1+\theta/3}(\mathbb{R}^N)}, \tag{14.2.49}$$

$$||D_{lm}T(t)\psi||_{\infty} \le C_4 t^{\frac{\theta-3}{6}} ||\psi||_{C_h^{1+\theta/3}(\mathbb{R}^N)}, \tag{14.2.50}$$

for any $t \in (0,T)$ and any $1 \leq l, m \leq r$. Again, interpolating (14.2.49) and (14.2.50) yields

$$||D_{lm}T(t)\psi||_{C_b^{1/2}(\mathbb{R}^N)} \le C_5 t^{\frac{4\theta-15}{12}} ||\psi||_{C_b^{1+\theta/3}(\mathbb{R}^N)}, \qquad t \in (0,T), \quad 1 \le l \le r.$$

$$(14.2.51)$$

Taking $\theta = 4/5$ in (14.2.51), we easily see that the function

$$s\mapsto \|D_{lm}T(t-s)g_{2R}^h(s)\|_{C_b^{1/2}(\mathbb{R}^N)}$$

is integrable in (0,T). Hence, from the formula (14.2.46) we get $D_{lm}v_{2R} \in B((0,T); C_b^{3/2}(\mathbb{R}^N))$ and, consequently, $D_{lm}u \in B((0,T); C^{3/2}(\overline{B}(R)))$, since $v_{2R} \equiv u$ in $(0,+\infty) \times B(R)$.

Now, we are almost done. Indeed, interpolating (14.2.4) and (14.2.5), we get

$$||D_{ij}T(t)\psi||_{\infty} \le C_6 t^{-\frac{3}{4}} ||\psi||_{C_b^{3/2}(\mathbb{R}^N)}, \qquad t \in (0,T), \quad i,j > r,$$

that, due to the above results, implies that the map $s \mapsto \|D_{ij}T(t-s)g_R^h(s,\cdot)\|_{\infty}$ is integrable in (0,t). Therefore, from (14.2.46) we easily deduce that $D_{ijk}u(t,\cdot)$ exists for any t>0.

Remark 14.2.8 In fact, in Step 2 of the proof of Theorem 14.2.7 we have shown that, for any $f \in C_0(\mathbb{R}^N)$, T(t)f converges to f uniformly in \mathbb{R}^N as t tends to 0^+ .

Remark 14.2.9 Uniform gradient estimates similar to those in Theorem 14.2.4 have been proved in [37] in the case when the operator \mathcal{A} is obtained from the degenerate Ornstein-Uhlenbeck operator in (14.0.3) replacing the drift term Bx, with the term Bx + F(x). The smooth function F is assumed to satisfy

$$\langle DF(x)G_t\xi,\xi\rangle \le C\langle G_t\xi,\xi\rangle, \qquad t>0, \ x,\xi\in\mathbb{R}^N,$$

for some constant $C \in \mathbb{R}$, and the matrix G_t is defined as in (14.0.4) with B being replaced with the matrix -B. Moreover, F should either belong to $C_b^2(\mathbb{R}^N, \mathbb{R}^N)$ or be Lipschitz continuous and satisfy

$$\langle F(x) - F(y), x - y \rangle \le \eta |x - y|^2, \qquad x, y \in \mathbb{R}^N,$$

for some constant $\eta \in \mathbb{R}$.

14.3 Some remarkable properties of the semigroup

In this section we first prove a continuity property of the semigroup $\{T(t)\}$ that will play a fundamental role in order to prove the Schauder estimates of Section 14.4. Then, we prove that $\{T(t)\}$ is strong Feller and, finally, we characterize the domain of its weak generator.

Proposition 14.3.1 Let $\{f_n\} \subset C_b(\mathbb{R}^N)$ be a bounded sequence of continuous functions converging to $f \in C_b(\mathbb{R}^N)$ locally uniformly in \mathbb{R}^N . Then, $T(\cdot)f_n$ converges to $T(\cdot)f$ locally uniformly in $[0, +\infty) \times \mathbb{R}^N$ and in $C^{1,2}(D)$ for any compact set $D \subset (0, +\infty) \times \mathbb{R}^N$.

Proof. Without loss of generality, throughout the proof we assume that $f \equiv 0$ and $\sup_{n \in \mathbb{N}} \|f_n\|_{\infty} \leq 1$. As a first step, we show that $T(\cdot)f_n$ tends to 0 in $C^{1,2}(D)$, for any compact set $D \subset (0,+\infty) \times \mathbb{R}^N$. Since the function $T(\cdot)f_n$ solves the differential equation $D_t u - \mathcal{A} u = 0$ in $(0,+\infty) \times \mathbb{R}^N$, by the estimate (14.2.1)-(14.2.3) it follows that the sequence $\{T(\cdot)f_n\}$ is bounded in $C^{1+\alpha/2,2+\alpha}(K)$ for any compact set $K \subset (0,+\infty) \times \mathbb{R}^N$ and any $\alpha \in (0,1)$. Therefore, the Ascoli-Arzelà theorem implies that there exists a subsequence $\{T(\cdot)f_{n_k}\}$ converging in $C^{1,2}(D)$ to some function g, for any compact set $D \subset (0,+\infty) \times \mathbb{R}^N$. As it is easily seen, the function g solves the differential equation $D_t g - \mathcal{A} g = 0$. Let us prove that $g \equiv 0$. For this purpose, according to Theorem 14.2.6, it suffices to show that g can be extended by continuity at g = 0 by setting g(g, g) = 0. So, let g = 0 be a compact set and take any smooth function g = 0 compactly supported in g = 0 in g = 0. Recalling that g = 0 is a semigroup of contractions (see Theorem 14.2.7), we can write

$$|(T(t)f_{n_k})(x) - (T(t)(\eta f_{n_k}))(x)| \le ||f_{n_k}||_{\infty} ||\mathbf{1} - (T(t)\eta)(x)|| \le ||\mathbf{1} - (T(t)\eta)(x)||,$$
 (14.3.1)

for any t > 0 and any $x \in \mathbb{R}^N$. Now, since f_{n_k} converges locally uniformly in \mathbb{R}^N as k tends to $+\infty$, the function $T(t)(\eta f_{n_k})$ converges uniformly in \mathbb{R}^N to 0. Hence, letting k go to $+\infty$ in (14.3.1) gives

$$|g(t,x) - (T(t)(\eta f))(x)| \le |\mathbf{1} - (T(t)\eta)(x)|, \quad t > 0, \ x \in \mathbb{R}^N.$$

Since both η and ηf are compactly supported in \mathbb{R}^N , then, by Remark 14.2.8, $T(t)\eta$ and $T(t)(\eta f)$ tend, respectively, to η and 0 uniformly in \mathbb{R}^N , as t tends to 0^+ . Therefore, from (14.3.1) we deduce that g tends to 0 as t tends to 0^+ , uniformly in F. The arbitrariness of F implies that g can be extended by continuity at t = 0 setting $g(0, \cdot) = 0$, and we are done.

Repeating the same arguments as above we can show that any sequence $\{T(\cdot)f_{n_k}\}$ admits a subsequence which converges to 0 in $C^{1,2}(D)$ for any set D as above. This implies that, actually, all the sequence $\{T(\cdot)f_n\}$ converges in $C^{1,2}(D)$ to 0, for any D as above.

Now, to check that $T(\cdot)f_n$ converges to 0 locally uniformly in $[0, +\infty) \times \mathbb{R}^N$ it suffices to argue as in the proof of Proposition 2.2.9.

Corollary 14.3.2 There exists a family of probability Borel measures

$$\{p(t, x; dy): t > 0, x \in \mathbb{R}^N\},\$$

such that

$$(T(t)f)(x) = \int_{\mathbb{R}^N} f(y)p(t, x; dy), \qquad t > 0, \ x \in \mathbb{R}^N,$$
 (14.3.2)

for any $f \in C_b(\mathbb{R}^N)$. It follows that, if $\{f_n\} \in C_b(\mathbb{R}^N)$ is a bounded sequence converging pointwise to $f \in C_b(\mathbb{R}^N)$, then $T(\cdot)f_n$ tends to $T(\cdot)f$ pointwise.

Proof. It can be obtained arguing as in the proof of Proposition 12.1.7, taking Proposition 14.3.1 into account. Therefore, we omit the details.

Next, arguing as in the proof of Proposition 12.2.1 and taking the gradient estimate in (14.2.9) into account, the following proposition can be proved.

Proposition 14.3.3 The semigroup $\{T(t)\}$ is strong Feller.

We now observe that since $\{T(t)\}$ is a semigroup of contractions, then for any $\lambda > 0$ we can define the linear operator $R(\lambda) \in L(C_b(\mathbb{R}^N))$ by setting

$$(R(\lambda)f)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x) dt, \qquad x \in \mathbb{R}^N.$$

As in Chapter 2, it is immediate to check that $\{R(\lambda) : \lambda > 0\}$ is the resolvent family associated with some closed operator $\widehat{A} : D(\widehat{A}) \subset C_b(\mathbb{R}^N) \to C_b(\mathbb{R}^N)$: the weak generator of the semigroup $\{T(t)\}$. From now on, we write $R(\lambda, \widehat{A})$ instead of $R(\lambda)$.

In the next proposition we characterize $D(\widehat{A})$ and we show that $\widehat{A}u = \mathcal{A}u$ for any $u \in D(\widehat{A})$, where $\mathcal{A}u$ is meant in the sense of distributions. The following lemma is the keystone in order to prove the forthcoming Proposition 14.3.5.

Lemma 14.3.4 For any $f \in C_b^2(\mathbb{R}^N)$ such that $Af \in C_b(\mathbb{R}^N)$ it holds that

$$T(t)\mathcal{A}f = \mathcal{A}T(t)f, \qquad t > 0. \tag{14.3.3}$$

Proof. To prove (14.3.3) we first show that

$$T_{\varepsilon}(t)\mathcal{A}_{\varepsilon}f = \mathcal{A}_{\varepsilon}T_{\varepsilon}(t)f, \qquad t > 0,$$
 (14.3.4)

for any $\varepsilon > 0$ and any $f \in C_b^2(\mathbb{R}^N)$ such that $\mathcal{A}f \in C_b(\mathbb{R}^N)$. Here, $\mathcal{A}_{\varepsilon}$ is given by (14.0.6). For this purpose, we recall that, according to Lemma 2.3.3 and Propositions 2.3.6, 4.1.1(i), $T(t)\mathcal{A}_{\varepsilon}u = \mathcal{A}_{\varepsilon}T(t)u$ for any $u \in D_{\max}(\mathcal{A}_{\varepsilon})$, where

$$D_{\max}(\mathcal{A}_{\varepsilon}) := \left\{ g \in C_b(\mathbb{R}^N) \cap \bigcap_{1 \le p < +\infty} W_{\mathrm{loc}}^{2,p}(\mathbb{R}^N) : \mathcal{A}_{\varepsilon}g \in C_b(\mathbb{R}^N) \right\},\,$$

and $A_{\varepsilon}g = \mathcal{A}_{\varepsilon}g$ for any $g \in D_{\max}(\mathcal{A}_{\varepsilon})$. Since $f \in D_{\max}(\mathcal{A}_{\varepsilon})$ for any $\varepsilon > 0$, then (14.3.4) follows.

Now, we are almost done. Indeed, a straightforward computation shows that $\mathcal{A}_{\varepsilon}f$ converges uniformly in \mathbb{R}^N to $\mathcal{A}f$ as ε tends to 0^+ . Moreover, from Theorem 14.2.7 we know that $T_{\varepsilon}(t)f$ converges to T(t)f in $C^2(K)$, as ε tends to 0^+ , for any t > 0. Therefore, from (14.1.2) we get

$$||T_{\varepsilon}(t)\mathcal{A}_{\varepsilon}f - T(t)\mathcal{A}f||_{C(K)}$$

$$\leq ||T_{\varepsilon}(t)(\mathcal{A}_{\varepsilon}f - \mathcal{A}f)||_{C(K)} + ||(T_{\varepsilon}(t) - T(t))\mathcal{A}f||_{C(K)}$$

$$\leq ||\mathcal{A}_{\varepsilon}f - \mathcal{A}f||_{\infty} + ||(T_{\varepsilon}(t) - T(t))\mathcal{A}f||_{C(K)},$$

and the last side of the previous chain of inequalities vanishes as ε tends to 0^+ . This implies that $T_{\varepsilon}(t)\mathcal{A}_{\varepsilon}f$ tends to $T(t)\mathcal{A}f$, locally uniformly in \mathbb{R}^N , for any t>0. Similarly, for any t>0, $\mathcal{A}_{\varepsilon}T_{\varepsilon}(t)f$ tends to $\mathcal{A}T(t)f$ locally uniformly in \mathbb{R}^N as ε tends to 0^+ . Hence, taking the limit as ε tends to 0^+ in (14.3.4), we get (14.3.3).

We are now in a position to prove the following proposition.

Proposition 14.3.5 The following characterization holds true:

$$D(\widehat{A}) = \Big\{ f \in C_b(\mathbb{R}^N) : \exists \{f_n\} \subset C_b^2(\mathbb{R}^N), \exists g \in C_b(\mathbb{R}^N) : \\ f_n \to f, \ \mathcal{A}f_n \to g \text{ locally uniformly in } \mathbb{R}^N \\ \text{and} \ \sup_{n \in \mathbb{N}} (\|f_n\|_{\infty} + \|\mathcal{A}f_n\|_{\infty}) < +\infty \Big\}.$$
 (14.3.5)

Moreover, Af = Af for any $f \in D(A)$. Here and above, Af is meant in the sense of distributions.

Proof. We adapt to our situation the technique of [107, Theorem 6.2]. For this purpose, let us denote by A_0 the realization of the operator \mathcal{A} in $C_b(\mathbb{R}^N)$ with domain $D(A_0) = \{ f \in C_b^2(\mathbb{R}^N) : \mathcal{A}f \in C_b(\mathbb{R}^N) \}$. Taking Lemma 14.3.4 into account, it is easy to check that, for any $f \in D(A_0)$,

$$(R(\lambda, \widehat{A})\mathcal{A}f)(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)\mathcal{A}f)(x)dt$$

$$= \int_0^{+\infty} e^{-\lambda t} (\mathcal{A}T(t)f)(x)dt$$

$$= \int_0^{+\infty} e^{-\lambda t} \left(\frac{\partial}{\partial t}T(t)f\right)(x)dt$$

$$= -f(x) + \lambda (R(\lambda, \widehat{A})f)(x), \tag{14.3.6}$$

for any $x \in \mathbb{R}^N$ and any $\lambda > 0$. Therefore, $R(\lambda, \widehat{A})(\lambda f - \mathcal{A}f) = f$ and, consequently, $f \in D(\widehat{A})$ and $\widehat{A}f = \mathcal{A}f$.

Now let $f \in D_0$ (the function space defined by the right-hand side of (14.3.5)). By the above results we know that

$$f_n = R(\lambda, \widehat{A})(\lambda f_n - \mathcal{A}f_n), \qquad n \in \mathbb{N}, \quad \lambda > 0.$$
 (14.3.7)

Let us now show that, for any bounded sequence $\{h_n\} \subset C_b(\mathbb{R}^N)$, converging to some function $h \in C_b(\mathbb{R}^N)$ locally uniformly in \mathbb{R}^N , the function $R(\lambda, \widehat{A})h_n$ converges to $R(\lambda, \widehat{A})h$, locally uniformly in \mathbb{R}^N , as well. For this purpose, we observe that, for any compact set $K \subset \mathbb{R}^N$,

$$||R(\lambda, \widehat{A})(h_n - h)||_{C(K)} \le \int_0^{+\infty} e^{-\lambda t} ||T(t)(h_n - h)||_{C(K)} dt.$$

By Proposition 14.3.1, $||T(t)(h_n - h)||_{C(K)}$ converges to 0 as n tends to $+\infty$, for any $t \ge 0$. Moreover,

$$||T(t)(h_n - h)||_{C(K)} \le ||T(t)(h_n - h)||_{\infty} \le ||h_n - h||_{\infty} \le ||h_n|| + ||h|| \le C,$$

for any $n \in \mathbb{N}$ and some positive constant C, independent of n. A straightforward application of the dominated convergence theorem shows that $R(\lambda, \widehat{A})h_n$ converges to $R(\lambda, \widehat{A})h$ as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . Hence, letting n go to $+\infty$ in (14.3.7) gives $f = R(\lambda, \widehat{A})(\lambda f - g)$ and, consequently, $f \in D(\widehat{A})$ and $g = \widehat{A}f$. Moreover, for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ we have

$$\int_{\mathbb{R}^N} \mathcal{A} f_n \, \varphi dx = \int_{\mathbb{R}^N} f_n \mathcal{A}^* \varphi dx, \qquad n \in \mathbb{N}, \tag{14.3.8}$$

where by \mathcal{A}^* we denote the adjoint of the operator \mathcal{A} , i.e., the operator defined by

$$(\mathcal{A}^*\psi)(x) = \sum_{i,j=1}^r D_{ij}(q_{ij}\psi)(x) - \sum_{i,j=1}^N b_{ij}x_j D_i\psi(x) - \psi(x)\text{Tr}(B), \qquad x \in \mathbb{R}^N,$$
(14.3.9)

for any smooth function ψ . Hence, letting n go to $+\infty$ in (14.3.8), we obtain that $g = \mathcal{A}f$ and, consequently, $\widehat{A}f = \mathcal{A}f$.

Now, we prove that $D(\widehat{A}) \subset D_0$. For this purpose, fix $u \in D(\widehat{A})$ and $\lambda > 0$. Further, let $f \in C_b(\mathbb{R}^N)$ be such that $u = R(\lambda, \widehat{A})f$. By convolution, we can regularize f obtaining a sequence $\{f_n\} \subset C_b^2(\mathbb{R}^N)$ bounded in $C_b(\mathbb{R}^N)$ and converging locally uniformly to f as n tends to $+\infty$. Taking Theorem 14.2.7 into account, we deduce that $R(\lambda, \widehat{A})f_n$ belongs to $C_b^2(\mathbb{R}^N)$ for any $n \in \mathbb{N}$. Moreover,

$$(\mathcal{A}R(\lambda,\widehat{A})f_n)(x) = \int_0^{+\infty} e^{-\lambda t} (\mathcal{A}T(t)f_n)(x)dt$$
$$= -f_n(x) + \lambda(R(\lambda,\widehat{A})f_n)(x), \qquad (14.3.10)$$

for any $x \in \mathbb{R}^N$ and any $n \in \mathbb{N}$. Since, by the above results, $R(\lambda, \widehat{A})f_n$ converges to u, locally uniformly in \mathbb{R}^N , letting n go to $+\infty$ in (14.3.10) we deduce that $\mathcal{A}R(\lambda, \widehat{A})f_n$ converges to $\lambda u - f$, locally uniformly in \mathbb{R}^N . Moreover, as we can easily see, the sequences $\{R(\lambda, \widehat{A})f_n\}$ and $\{\mathcal{A}R(\lambda, \widehat{A})f_n\}$ are bounded in $C_b(\mathbb{R}^N)$. Therefore, $u \in D_0$.

14.4 The nonhomogeneous elliptic equation and the nonhomogeneous Cauchy problem

In this section we are devoted to prove existence, uniqueness and regularity properties of the bounded and continuous distributional solution to the elliptic equation

$$\lambda u(x) - \mathcal{A}u(x) = f(x), \qquad x \in \mathbb{R}^N,$$
 (14.4.1)

and to the parabolic Cauchy problem

$$\begin{cases}
D_t u(t,x) = \mathcal{A}u(t,x) + g(t,x), & t > 0, x \in \mathbb{R}^N, \\
u(0,x) = f(x), & x \in \mathbb{R}^N.
\end{cases}$$
(14.4.2)

The main results that we prove are contained in the following two theorems.

Theorem 14.4.1 Assume that Hypotheses 14.0.1 hold and let $\theta \in (0,1)$, $\lambda > 0$. Then, for any $f \in \mathcal{C}_b^{\theta,\theta/3}(\mathbb{R}^N)$, there exists a function $u \in \mathcal{C}_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)$ solving the equation (14.4.1) in the sense of distributions. Moreover, there exists a positive constant C, independent of u and f, such that

$$||u||_{C_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)} \le C||f||_{C_b^{\theta,\theta/3}(\mathbb{R}^N)}.$$
(14.4.3)

Such a function u is the unique distributional solution to the equation (14.4.1) which is bounded and continuous in \mathbb{R}^N and it is twice continuously differentiable in \mathbb{R}^N with respect to the first r variables, with bounded derivatives.

Theorem 14.4.2 Assume that Hypotheses 14.0.1 are satisfied and let $\theta \in (0,1)$, T > 0. Further, let $f \in C_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)$ and let $g \in C_b([0,T] \times \mathbb{R}^N)$ be such that $g(t,\cdot) \in C_b^{\theta,\theta/3}(\mathbb{R}^N)$ for any $t \in [0,T]$ and

$$\sup_{t \in [0,T]} \|g(t,\cdot)\|_{\mathcal{C}^{\theta,\theta/3}_b(\mathbb{R}^N)} < +\infty.$$

Then, there exists a function $u \in C_b([0,T] \times \mathbb{R}^N)$, solution to the problem (14.4.2) in the sense of distributions, such that $u(t,\cdot) \in \mathcal{C}_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)$ for any $t \in [0,T]$ and

$$\sup_{t \in [0,T]} \|u(t,\cdot)\|_{\mathcal{C}_{b}^{2+\theta,(2+\theta)/3}(\mathbb{R}^{N})} \le C (\|f\|_{\mathcal{C}_{b}^{2+\theta,(2+\theta)/3}(\mathbb{R}^{N})} + \sup_{t \in [0,T]} \|g(t,\cdot)\|_{\mathcal{C}_{b}^{\theta,\theta/3}(\mathbb{R}^{N})}), \tag{14.4.4}$$

for some positive constant C, independent of u, f and g. Moreover, u is the unique distributional solution to the problem (14.4.2) which is bounded and continuous in $[0,T] \times \mathbb{R}^N$ and it is twice continuously differentiable with respect to the first r space variables in $[0,T] \times \mathbb{R}^N$, with bounded derivatives.

To prove Theorems 14.4.1 and 14.4.2, in the spirit of Chapter 6, we first characterize the behaviour of the semigroup in the anisotropic space $C_b^{3\theta,\theta}(\mathbb{R}^N)$ $(\theta \in [0,1])$. More precisely, we show that for any $0 < \alpha \le \beta \le 1$ and any $\omega > 0$, there exists a positive constant $C = C(\alpha, \beta, \omega)$ such that

$$\|T(t)\|_{L(\mathcal{C}_{b}^{3\alpha,\alpha}(\mathbb{R}^{N}),\mathcal{C}_{b}^{3\beta,\beta}(\mathbb{R}^{N}))} \leq Ce^{\omega t}t^{-\frac{3(\beta-\alpha)}{2}}, \qquad t>0. \tag{14.4.5}$$

Two steps are the main ingredients to prove (14.4.5):

Step (i): the interpolation set equality

$$(C_b(\mathbb{R}^N), C_b^{3,1}(\mathbb{R}^N))_{\theta,\infty} = C_b^{3\theta,\theta}(\mathbb{R}^N), \tag{14.4.6}$$

with equivalence of the corresponding norms, for any $\theta \in (0,1)$;

Step (ii): the estimate

$$\|T(t)\|_{L(C_b^{3k,k}(\mathbb{R}^N),C_b^{3,1}(\mathbb{R}^N))} \le Ce^{\omega t}t^{-\frac{3(1-k)}{2}}, \qquad t>0, \quad k=0,1. \quad (14.4.7)$$

Once the Steps (i) and (ii) are established, an interpolation argument allows us to prove (14.4.5). We begin by proving Step (i).

Proposition 14.4.3 For any $\theta \in (0,1)$, the formula (14.4.6) holds true with equivalence of the corresponding norms.

Proof. Throughout the proof we identify \mathbb{R}^N with $\mathbb{R}^r \times \mathbb{R}^{N-r}$. Moreover, we denote by C_k $(k \in \mathbb{N})$ positive constants which are independent of s, t and $f \in C_b(\mathbb{R}^N)$. Finally, since the proof is rather long, we split it into three steps.

Step 1. Let us denote by A_1 and A_2 , respectively, the realization in $C_b(\mathbb{R}^N)$ of the operators $A_1 = \sum_{i=1}^r D_{ii}$ and $A_2 = \sum_{i=r+1}^N D_{ii}$ with domains

$$D(A_1) = \left\{ u \in C_b(\mathbb{R}^N) : u(\cdot, y) \in \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2, p}(\mathbb{R}^r), \quad \forall y \in \mathbb{R}^{N-r}, \right.$$
$$\mathcal{A}_1 u \in C_b(\mathbb{R}^N) \right\}$$

and

$$D(A_2) = \left\{ u \in C_b(\mathbb{R}^N) : u(x, \cdot) \in \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2,p}(\mathbb{R}^{N-r}), \quad \forall x \in \mathbb{R}^r, \right.$$
$$\mathcal{A}_2 u \in C_b(\mathbb{R}^N) \left. \right\}.$$

Further, let us set $A_3 = -A_1^2$. In this step of the proof we show that

$$(X, D(A_j))_{\beta,\infty} = \mathcal{C}_b^{2(2-j)\beta, 2(j-1)\beta}(\mathbb{R}^N), \quad j = 1, 2,$$
 (14.4.8)

$$(X, D(A_3))_{\beta,\infty} = \mathcal{C}_b^{4\beta,0}(\mathbb{R}^N),$$
 (14.4.9)

for any $\beta \in (0,1)$, with equivalence of the corresponding norms. For this purpose, we will take advantage of the characterization of the interpolation spaces $(X, D(A))_{\beta,\infty}$ given in Proposition B.2.12, when A is the generator of an analytic semigroup in a Banach space X. Indeed, taking Theorem C.3.6(iv) into account, it is easy to check that A_1 and A_2 are sectorial operators. Moreover, since the realization of the Laplacian in $C_b(\mathbb{R}^N)$ satisfies (B.2.1) for any $\varphi \in (\pi/2, \pi)$, then the operator A_1 satisfies (B.2.1) for any $\varphi \in (\pi/2, \pi)$ as well. Therefore, according to Theorem B.2.10, also the operator A_3 is sectorial in $C_b(\mathbb{R}^N)$.

In the sequel, we simply write $D_{A_j}(\beta, \infty)$ and $D_{A_3}(\beta, \infty)$ to denote the interpolation spaces $(X, D(A_j))_{\beta,\infty}$ and $(X, D(A_3))_{\infty}$, (j = 1, 2).

We begin by proving (14.4.8) in the case when j = 1. For this purpose, we preliminarily observe that the analytic semigroup $\{T_1(t)\}$, associated with the operator A_1 , is given by

$$(T_1(t)f)(x,y) = \frac{1}{(4\pi t)^{r/2}} \int_{\mathbb{R}^r} e^{-\frac{|x-z|^2}{4t}} f(z,y) dz, \qquad t > 0, \ (x,y) \in \mathbb{R}^N,$$

for any $f \in C_b(\mathbb{R}^N)$.

To prove the inclusion $\mathcal{C}_b^{2\beta,0}(\mathbb{R}^N) \subset D_{A_1}(\beta,\infty)$, we fix $f \in \mathcal{C}_b^{2\beta,0}(\mathbb{R}^N)$ and observe that

$$(T_1(t)f)(x,y) - f(x,y)$$

$$= \frac{1}{2(4\pi)^{r/2}} \int_{\mathbb{R}^r} e^{-\frac{|z|^2}{4}} (f(x - \sqrt{t}z, y) - 2f(x, y) + f(x + \sqrt{t}z, y)) dz,$$

for any t > 0 and any $(x, y) \in \mathbb{R}^N$. Therefore,

$$||T_1(t)f - f||_{\infty} \le \frac{t^{\beta}}{(4\pi)^{r/2}} [f]_{\mathcal{C}_b^{2\beta,0}(\mathbb{R}^N)} \int_{\mathbb{R}^r} e^{-\frac{|z|^2}{4}} |z|^{2\beta} dz, \tag{14.4.10}$$

and Proposition B.2.12 implies that $C_b^{2\beta,0}(\mathbb{R}^N)$ is continuously embedded in $D_{A_1}(\beta,\infty)$.

Let us now prove the other inclusion in (14.4.8). We begin by observing that

(i)
$$||D_x T_1(t)f||_{\infty} \le C_1 t^{-\frac{1}{2}} ||f||_{C_b(\mathbb{R}^N)},$$
 (ii) $||A_1 T_1(t)f||_{\infty} \le C_1 t^{-1} ||f||_{\infty},$ (14.4.11)

for any $f \in C_b(\mathbb{R}^N)$, where D_x denotes the gradient with respect to the first r variables. Let us fix $f \in D_{A_1}(\beta, \infty)$ and assume that $\beta \in (0, 1/2)$. Then, for

any t > 0, any $x_1, x_2 \in \mathbb{R}^r$ and any $y \in \mathbb{R}^{N-r}$, it holds that

$$|f(x_{2},y) - f(x_{1},y)|$$

$$\leq |(T_{1}(t)f)(x_{2},y) - f(x_{2},y)| + |(T_{1}(t)f)(x_{2},y) - (T_{1}(t)f)(x_{1},y)|$$

$$+|(T_{1}(t)f)(x_{1},y) - f(x_{1},y)|$$

$$\leq 2[[f]]_{D_{A_{1}}(\beta,\infty)}t^{\beta} + ||D_{x}T_{1}(t)f||_{\infty}|x_{2} - x_{1}|, \qquad (14.4.12)$$

where $[[f]]_{D_{A_1}(\beta,\infty)}$ is given by Proposition B.2.12. So, we need to estimate the sup-norm of $D_xT_1(t)f$. As it is easily seen,

$$D_i T_1(n) f - D_i T_1(t) f = \int_t^n D_i A_1 T_1(s) f \, ds, \qquad t \in (0, n), \qquad (14.4.13)$$

for any $i=1,\ldots,r$. Combining (14.4.11)(ii) and Proposition B.2.12, we deduce that

$$||A_1T_1(t)f||_{\infty} \le C_2 t^{\beta-1} ||f||_{D_A(\beta,\infty)}, \qquad t \in (0,+\infty).$$
 (14.4.14)

Now, from (14.4.11)(i) and (14.4.14), we get

$$||D_{i}A_{1}T_{1}(s)f||_{\infty} = ||D_{i}T_{1}(s/2)A_{1}T_{1}(s/2)f||_{\infty}$$

$$\leq ||D_{i}T_{1}(s/2)||_{L(C_{b}(\mathbb{R}^{N}))}||A_{1}T_{1}(s/2)f||_{\infty}$$

$$\leq C_{3}s^{\beta-\frac{3}{2}}||f||_{D_{A_{1}}(\beta,\infty)}, \qquad (14.4.15)$$

for any s > 0, so that we can let n go to $+\infty$ in (14.4.13) obtaining

$$D_i T_1(t) f = -\int_t^{+\infty} D_i A_1 T_1(s) f ds, \qquad t > 0,$$

and

$$||D_x T_1(t)f||_{\infty} \le C_4 t^{\beta - \frac{1}{2}} ||f||_{D_{A_1}(\beta,\infty)}.$$
(14.4.16)

Now, taking $t = |x_2 - x_1|^2$ in (14.4.12) and (14.4.16), we get

$$|f(x_2, y) - f(x_1, y)| \le C_5 ||f||_{D_{A_1}(\beta, \infty)} |x_2 - x_1|^{2\beta},$$

for any $|x_2 - x_1| \le 1$. Moreover, if $|x_2 - x_1| > 1$ we have

$$|f(x_2, y) - f(x_1, y)| \le 2||f||_{\infty} \le 2||f||_{D_{A_1}(\beta, \infty)}|x_2 - x_1|^{2\beta},$$

so that $D_{A_1}(\beta, \infty) \subset C_b^{2\beta,0}(\mathbb{R}^N)$ with a continuous embedding. Suppose now that $\beta > 1/2$. According to (14.4.13) and (14.4.15) we have

$$||D_i T_1(t_2) f - D_i T(t_1) f||_{\infty} \le C_6 |t_2 - t_1|^{\beta - 1/2} ||f||_{D_{A_1}(\beta, \infty)}, \qquad t_1, t_2 > 0,$$
(14.4.17)

for any i = 1, ..., r. This implies that $DT_1(t)f$ converges uniformly in \mathbb{R}^N as t tends to 0. Since $T_1(\cdot)f$ is continuous in $[0, +\infty) \times \mathbb{R}^N$ and $T_1(0)f = f$, we obtain that $f \in C_b^1(\mathbb{R}^N)$.

Now, taking the estimates (14.4.11)(i) and (14.4.16) (which, of course, holds true also in the case when $\beta \in (1/2, 1)$) into account, we get

$$||D_x D_i T_1(t) f||_{\infty} = ||D_x T(t/2) D_i T(t) f||_{\infty}$$

$$\leq ||D_x T(t/2)||_{\infty} ||D_i T(t) f||_{\infty}$$

$$\leq C_1 C_4 t^{\beta - 1} ||f||_{D_{A_1}(\beta, \infty)}.$$
(14.4.18)

Hence, from (14.4.17) (where we take $t_1 = 0$) and (14.4.18), it follows that

$$|D_{i}f(x_{2}, y) - D_{i}f(x_{1}, y)|$$

$$\leq |(D_{i}T_{1}(t)f)(x_{2}, y) - D_{i}f(x_{2}, y)| + |(D_{i}T_{1}(t)f)(x_{2}, y) - (D_{i}T_{1}(t)f)(x_{1}, y)|$$

$$+|(D_{i}T_{1}(t)f)(x_{1}, y) - D_{i}f(x_{1}, y)|$$

$$\leq 2||D_{i}T_{1}(t)f - f||_{\infty} + ||D_{x}D_{i}T_{1}(t)f||_{\infty}|x - y|$$

$$\leq 2C_{6}t^{\beta - 1/2}||u||_{D_{A_{1}}(\beta, \infty)} + C_{1}C_{4}t^{\beta - 1}|x_{2} - x_{1}|||f||_{D_{A_{1}}(\beta, \infty)},$$

for any t > 0, any $x_1, x_2 \in \mathbb{R}^r$, any $y \in \mathbb{R}^{N-r}$ and any $1 = 1 \dots, r$. The same arguments as in the case when $\beta < 1/2$ allow us to conclude that $D_{A_1}(\beta, \infty) \subset C_b^{2\beta,0}(\mathbb{R}^N)$ with a continuous embedding.

So far, we have proved (14.4.8) (with j=1) in the case when $\beta \neq 1/2$. Let us now consider the case when $\beta = 1/2$. For this purpose, we observe that

$$D_{A_1}(1/2,\infty) = (D_{A_1}(1/4,\infty), D_{A_1}(3/4,\infty))_{1/2,\infty}$$
$$= (C_h^{1/2,0}(\mathbb{R}^N), C_h^{3/2,0}(\mathbb{R}^N))_{1/2,\infty}.$$

Since $(C_b^{1/2}(\mathbb{R}^r), C_b^{3/2}(\mathbb{R}^r))_{1/2,\infty} = C_b^1(\mathbb{R}^r)$ (see Theorem A.4.8) and the operator $T_y: C_b(\mathbb{R}^N) \to C_b(\mathbb{R}^r)$, defined by $T_y f = f(\cdot, y)$ for any $f \in C_b(\mathbb{R}^N)$, belongs to $L(C_b^{(2k+1)/2,0}(\mathbb{R}^N), C_b^{(2k+1)/2}(\mathbb{R}^r))$ (k = 0, 1) with norms being independent of y, then Proposition A.4.2 allows us to conclude that $D_{A_1}(1/2, \infty)$ is continuously embedded in $C_b^{1,0}(\mathbb{R}^N)$. This concludes the proof of (14.4.8), in the case when j = 1.

The proof of (14.4.8), with j = 2, is completely similar, so we skip it.

Let us prove (14.4.9). From the general interpolation theory, we know that, for any $\beta \in (0,1)$,

$$D_{A_3}(\beta/2,\infty) = (C_b(\mathbb{R}^N), D(A_1^2))_{\beta/2,\infty} = (D(A_1), D(A_1^2))_{\beta,\infty} = D_{A_1}(1+\beta,\infty),$$
(14.4.19)

with equivalence of the corresponding norms. Hence, (14.4.9) follows immediately if we show that

$$D_{A_1}(1+\beta,\infty) = C_b^{2+2\beta,0}(\mathbb{R}^N),$$
 (14.4.20)

with equivalence of the corresponding norms. The embedding $C_b^{2+2\beta,0}(\mathbb{R}^N) \subset D_{A_1}(1+\beta,\infty)$ is an obvious consequence of (14.4.8) and the characterization of $D(A_1)$. To prove the other embedding, we fix $f \in D_{A_1}(1+\beta,\infty)$ and prove that f admits second-order derivatives, with respect to the first r variables, which belong to $C_b^{2\beta,0}(\mathbb{R}^N)$. For this purpose, we observe that, by (B.2.6), we can write

$$f = \int_0^{+\infty} e^{-t} T_1(t) g dt, \qquad (14.4.21)$$

where $g := f - \mathcal{A}_1 f$ belongs to $D_{A_1}(\beta, \infty) = \mathcal{C}_b^{2\beta,0}(\mathbb{R}^N)$. Taking (14.4.16) into account, it is immediate to check that f is once-continuously differentiable with respect to the first r variables and

$$||D_x f||_{\infty} \le C_7 ||g||_{D_{A_1}(\beta,\infty)}. \tag{14.4.22}$$

Splitting $D_{ij}T_1(t)g = D_jT_1(t/2)D_iT_1(t/2)g$ for any t > 0 and any $i, j = 1, \ldots, r$, and using (14.4.11)(i) and again (14.4.16), we can show that

$$||D_{ij}T_1(t)g||_{\infty} \le C_8 t^{\beta-1} ||g||_{D_{A_1}(\beta,\infty)}, \qquad i,j=1,\ldots,r. \quad (14.4.23)$$

Hence, f is twice continuously differentiable in \mathbb{R}^N with respect to the first r variables and

$$||D_{ij}f||_{\infty} \le C_9 ||g||_{D_{A_1}(\beta,\infty)} \le C_{10} ||f||_{D_{A_1}(1+\beta,\infty)}, \qquad t > 0, \quad i,j = 1,\dots,r.$$
(14.4.24)

This implies that $f \in C_b^{2,0}(\mathbb{R}^N)$ with norm bounded by $C_{11} ||f||_{D_{A_1}(1+\beta,\infty)}$. Since we already know that $D_{A_1}(\beta,\infty) = C_b^{2\beta,0}(\mathbb{R}^N)$, it is sufficient to prove that $D_{ij}f \in D_{A_1}(\beta,\infty)$. Using (14.4.11)(ii) and (14.4.23), we get

$$\|\xi^{1-\beta}A_1T_1(\xi)D_{ij}f\|_{\infty} \le \left\| \int_0^{+\infty} \xi^{1-\beta}e^{-t}A_1T_1(\xi+t/2)D_{ij}T_1(t/2)gdt \right\|$$

$$\le C_{12}\|g\|_{D_{A_1}(\beta,\infty)} \int_0^{+\infty} \frac{1}{(1+t)t^{1-\beta}}dt,$$

for any $\xi \in (0,1)$. Therefore, all the second order derivatives of f (with respect to the directions e_i , e_j , with $i, j \leq r$) are in $D_{A_1}(\beta, \infty) = \mathcal{C}_b^{2\beta,0}(\mathbb{R}^N)$ (see again Proposition B.2.12) and

$$||D_{ij}f||_{\mathcal{C}_b^{2\beta,0}(\mathbb{R}^N)} \le C_{13}||g||_{D_{A_1}(\beta,\infty)} \le C_{14}||f||_{D_{A_1}(1+\beta,\infty)}.$$

The formula (14.4.20) now follows.

Step 2. Here, taking advantage of the results in Step 1, we show that

$$(C_b(\mathbb{R}^N), C_b^{3\alpha,\alpha}(\mathbb{R}^N))_{\gamma,\infty} = C_b^{3\alpha\gamma,\alpha\gamma}(\mathbb{R}^N), \tag{14.4.25}$$

for any $\alpha \in (0, 4/3)$ and any $\gamma \in (0, 1)$. This can be done adapting the proof given in [107, Theorem 2.2]. We begin by proving (14.4.25) in the case when $\alpha \in (0, 2/3)$. By (14.4.8) we can write

$$C_b^{3\alpha,\alpha}(\mathbb{R}^N) = D_{A_1}(3\alpha/2,\infty) \cap D_{A_2}(\alpha/2,\infty).$$

The characterization (14.4.25) will follow immediately if we show that

$$(C_b(\mathbb{R}^N), D_{A_1}(3\alpha/2, \infty) \cap D_{A_2}(\alpha/2, \infty))_{\gamma, \infty}$$

$$= D_{A_1}(3\gamma\alpha/2, \infty) \cap D_{A_2}(\gamma\alpha/2, \infty), \qquad (14.4.26)$$

for any $\gamma \in (0,1)$, with equivalence of the corresponding norms.

The inclusion " \subset " in (14.4.26) is immediate if we observe that, for any triplet of Banach spaces X,Y,Z such that $Z \subset Y \subset X$ with continuous embeddings, it holds that $(X,Z)_{\theta,\infty} \subset (X,Y)_{\theta,\infty}$, for any $\theta \in (0,1)$, with continuous embedding.

So, let us prove the other inclusion in (14.4.26). For this purpose, we fix $f \in D_{A_1}(3\gamma\alpha/2,\infty) \cap D_{A_2}(\gamma\alpha/2,\infty)$ and define the function $u:[0,+\infty) \to C_b(\mathbb{R}^N)$ by setting

$$u(t) = T_1(t^{2/(3\alpha)})T_2(t^{2/\alpha})f, \qquad t \in (0,1].$$

Using Proposition B.2.12 yields

$$||u(t)||_{D_{A_{1}}(3\alpha/2,\infty)} \le C_{15} \Big(||T_{1}(t^{2/(3\alpha)})||_{L(C_{b}(\mathbb{R}^{N}))} ||T_{2}(t^{2/\alpha})||_{L(C_{b}(\mathbb{R}^{N}))} ||f||_{C_{b}(\mathbb{R}^{N})}$$

$$+ ||T_{2}(t^{2/(3\alpha)})||_{L(C_{b}(\mathbb{R}^{N}))} \sup_{0 < \xi \le 1} ||\xi^{1-2/(3\alpha)} A_{1}T_{1}(\xi + t^{2/(3\alpha)})f||_{C_{b}(\mathbb{R}^{N})} \Big)$$

$$\le C_{16} \Big(||f||_{C_{b}(\mathbb{R}^{N})} + \sup_{0 < \xi \le 1} \xi^{1-2/(3\alpha)} (\xi + t^{2/(3\alpha)})^{(3\alpha-2)/2} ||f||_{D_{A_{1}}(3\alpha/2,\infty)} \Big)$$

$$\le C_{17} t^{(3\alpha-2)/2} ||f||_{D_{A_{1}}(3\alpha/2,\infty)},$$

$$(14.4.27)$$

for any $t \in (0,1]$. Similarly,

$$||u(t)||_{D_{A_2}(\alpha/2,\infty)} \le C_{18} t^{(\alpha-2)/2} ||f||_{D_{A_2}(\alpha/2,\infty)}$$
(14.4.28)

and

$$||u(t) - f||_{C_b(\mathbb{R}^N)} \le ||T_1(t^{2/(3\alpha)})||_{L(C_b(\mathbb{R}^N))} ||(T_2(t^{2/\alpha}) - 1)f||_{C_b(\mathbb{R}^N)}$$

$$\le C_{19} t^{\gamma} \Big(||f||_{D_{A_1}(3/\alpha/2,\infty)} + ||f||_{D_{A_2}(\alpha/2,\infty)} \Big), \quad (14.4.29)$$

for any $t \in (0, 1]$. Splitting f = u(t) + (f - u(t)) and using (14.4.27)-(14.4.29), we get

$$t^{-\gamma}K(t,f) \le C_{20}\Big(\|f\|_{D_{A_1}(3\alpha/2,\infty)} + \|f\|_{D_{A_2}(\alpha/2,\infty)}\Big), \qquad t \in (0,1],$$
(14.4.30)

which implies that $f \in (C_b(\mathbb{R}^N), D_{A_1}(3\alpha/2, \infty) \cap D_{A_2}(\alpha/2, \infty))_{\gamma,\infty}$. We have so proved both the set equality in (14.4.26) and the equivalence of the norms of the spaces in the left- and right-hand sides of (14.4.26).

Now, suppose that $\alpha \in [2/3, 4/3)$. Then, according to (14.4.9), $C_b^{3\alpha,0}(\mathbb{R}^N) = D_{A_3}(3\alpha/4, \infty)$ with equivalence of the corresponding norms. Therefore, repeating the above arguments, with the operator A_1 being now replaced with A_3 , we can prove (14.4.25).

Step 3. We can now conclude the proof proving (14.4.6). From (14.4.25), with $\alpha \in (1, 4/3)$ and $\gamma = 1/\alpha$, we deduce that

$$C_b^{3,1}(\mathbb{R}^N) \subset C_b^{3,1}(\mathbb{R}^N) \in K_{1/\alpha}(C_b(\mathbb{R}^N), C_b^{3\alpha,\alpha}(\mathbb{R}^N)).$$

Since $C_b^3(\mathbb{R}^r)$ and $C_b^1(\mathbb{R}^{N-r})$ belong, respectively, to $J_{1/\alpha}(C_b(\mathbb{R}^r), C_b^{3\alpha}(\mathbb{R}^r))$ and $J_{1/\alpha}(C_b(\mathbb{R}^{N-r}), C_b^{\alpha}(\mathbb{R}^{N-r}))$ (see Proposition A.4.4), then $C_b^{3,1}(\mathbb{R}^N)$ belongs to $J_{1/\alpha}(C_b(\mathbb{R}^N), C_b^{3\alpha,\alpha}(\mathbb{R}^N))$. The reiteration theorem (see Theorem A.4.7) and (14.4.25) imply that

$$(C_b(\mathbb{R}^N), C_b^{3,1}(\mathbb{R}^N))_{\theta,\infty} = (C_b(\mathbb{R}^N), C_b^{3\alpha,\alpha}(\mathbb{R}^N))_{\theta/\alpha,\infty} = C_b^{3\theta,\theta}(\mathbb{R}^N),$$

for any $\theta \in (0,1)$, with equivalence of the corresponding norms. The set equality (14.4.6) now follows.

Applying the reiteration Theorem A.4.7, we get the following corollary.

Corollary 14.4.4 For any $\alpha, \theta \in (0,1)$ and any $\alpha \leq \beta < 1$,

$$(\mathcal{C}_b^{3\alpha,\alpha}(\mathbb{R}^N),\mathcal{C}_b^{3,1}(\mathbb{R}^N))_{\theta,\infty} = \mathcal{C}_b^{3(\alpha+(1-\alpha)\theta),\alpha+(1-\alpha)\theta}(\mathbb{R}^N), \qquad (14.4.31)$$

with equivalence of the corresponding norms.

Remark 14.4.5 Adapting the techniques in the proof of Proposition 14.4.3 one can show that (14.4.25) holds true for any $\alpha > 1$ and, hence, by reiteration, that for any $\alpha, \beta > 0$ and any $\theta \in (0,1)$,

$$(\mathcal{C}_{b}^{\alpha,\alpha/3}(\mathbb{R}^{N}),\mathcal{C}_{b}^{\beta,\beta/3}(\mathbb{R}^{N}))_{\theta,\infty} = \mathcal{C}_{b}^{\alpha+\theta(\beta-\alpha),(\alpha+\theta(\beta-\alpha))/3}(\mathbb{R}^{N}),$$

with equivalence of the corresponding norms.

As far as Step (ii) is concerned, we observe that, when k = 0, the estimate (14.4.7) follows easily from (14.2.1)-(14.2.3). Hence, we just need to show (14.4.7) when k = 1. For this purpose, we adapt the Bernstein method to these anisotropic spaces.

In the following theorem we use the same notation introduced before Theorem 14.2.4.

Theorem 14.4.6 Let $\varepsilon > 0$ and assume that Hypotheses 14.0.1 are satisfied. Then, there exist two positive constants C and ω such that (14.4.7) holds true, with k = 1.

Proof. As in the proof of Theorem 14.2.4, we limit ourselves to showing that for any $\omega > 0$, there exists a positive constant C, independent of ε such that

$$||T_{\varepsilon}(t)||_{L(C_{h}^{3,1}(\mathbb{R}^{N}), C_{h}^{3,1}(\mathbb{R}^{N}))} \le Ce^{\omega t}, \qquad t > 0.$$
 (14.4.32)

Here, $\{T_{\varepsilon}(t)\}$ is the semigroup defined in Section 14.2. Once (14.4.32) is proved we will be almost done. Indeed, suppose that (14.4.32) holds true. Since $T_{\varepsilon}(\cdot)f$ tends to $T(\cdot)f$ in $C^{1,2}(D)$ for any compact set $D \subset (0, +\infty) \times \mathbb{R}^N$ and any $f \in C_b(\mathbb{R}^N)$ (see Theorem 14.2.7), then, for any t > 0 the sup-norms of both DT(t)f and $D_{*,1}^2T(t)f$ are bounded by the right-hand side of (14.4.32). As far as the vector $D_{*,1}^3T(t)f$ is concerned, we observe that (14.4.32) implies that

$$[(D_{*,1}^2 T_{\varepsilon}(t)f)(\cdot, x_{r+1}, \dots, x_N)]_{\text{Lip}(\mathbb{R}^r)} \le Ce^{\omega t} ||f||_{C_{k}^{3,1}(\mathbb{R}^N)}, \qquad t > 0$$

for any t > 0 and any $(x_{r+1}, \ldots, x_N) \in \mathbb{R}^{N-r}$, and, letting ε go to 0^+ , yields

$$[(D_{*,1}^2 T(t)f)(\cdot, x_{r+1}, \dots, x_N)]_{\operatorname{Lip}(\mathbb{R}^r)} \le C e^{\omega t} ||f||_{C_h^{3,1}(\mathbb{R}^N)}, \quad (14.4.33)$$

and we are done. Indeed, since $T(t)f\in C^3_b(\mathbb{R}^N)$ for any t>0 (see Theorem 14.2.7), from (14.4.33) we deduce that

$$||D_{*,1}^3 T(t)f||_{\infty} \le Ce^{\omega t} ||f||_{C_b^{3,1}(\mathbb{R}^N)}, \qquad t > 0,$$

and (14.4.7) follows.

So, let us prove (14.4.32). For notational convenience we simply write u instead of $T_{\varepsilon}(\cdot)f$. Let us introduce the function $\xi:[0,+\infty)\times\mathbb{R}^N\to\mathbb{R}$ defined by

$$\xi(t,x) = \frac{1}{2}\alpha^{3}(u(t,x))^{2} + \langle F Du(t,x), Du(t,x) \rangle + \langle G(t)D_{*}^{2}u(t,x), D_{*}^{2}u(t,x) \rangle + \langle H(t)D_{*}^{3}u(t,x), D_{*}^{3}u(t,x) \rangle,$$

for any t > 0 and any $x \in \mathbb{R}^N$, where the matrices $F \in L(\mathbb{R}^N)$, $G(t) \in L(\mathbb{R}^{n_N^1})$ and $H(t) \in L(\mathbb{R}^{n_N^2})$ $(t \ge 0)$ are defined by

$$F = \begin{pmatrix} \alpha I_r & \beta F_1 \\ \beta F_1^* & -\iota I_{N-r} \end{pmatrix}, \qquad G(t) = \begin{pmatrix} I_{n_r^1} & 0 & 0 \\ 0 & \alpha^{-\frac{7}{16}} t I_{r(N-r)} & -\alpha^{-\frac{4}{5}} t^2 L_5 \\ 0 & -\alpha^{-\frac{4}{5}} t^2 L_5^* & \alpha^{-\frac{7}{8}} t^3 I_{n_{N-r}^1} \end{pmatrix}, \tag{14.4.34}$$

$$H(t) = \begin{pmatrix} \alpha^{-\frac{4}{3}} I_{n_r^2} & -\alpha^{-\frac{4}{3}} t M_2 & \frac{1}{2} t^2 \alpha^{-\frac{4}{3}} H_1 & -\frac{1}{6} \alpha^{-\frac{4}{3}} t^3 H_2 \\ -\alpha^{\frac{4}{3}} t M_2^* & \alpha^{-\frac{8}{9}} t^2 I_{(N-r)n_r^1} & 0 & 0 \\ \frac{1}{2} \alpha^{-\frac{4}{3}} t^2 H_1^* & 0 & \alpha^{-1} t^4 I_{rn_{N-r}^1} - \alpha^{-\frac{13}{12}} t^5 M_8 \\ -\frac{1}{6} \alpha^{-\frac{4}{3}} t^3 H_2^* & 0 & -\alpha^{-\frac{13}{12}} t^5 M_8^* & \alpha^{-\frac{9}{8}} t^6 I_{n_{N-r}^2} \end{pmatrix}.$$

$$(14.4.35)$$

Here, n_m^1 and n_m^2 $(m \in \mathbb{N})$ are given by (14.2.10), $\beta = 2||B_4|| + 1$, $F_1 \in L(\mathbb{R}^{N-r}, \mathbb{R}^r)$ is any matrix such that $B_3F_1 + F_1^*B_3^*$ is strictly negative definite and $-\iota$ denotes its maximum (negative) eigenvalue. Such a matrix exists since we have assumed that $r \geq N/2$. Moreover, $H_1 = M_2M_5$ and $H_2 = M_2M_5M_8$, where the matrices M_2 , M_5 , M_8 , as well as L_5 , are given by (14.2.22). Finally α is a positive constant to be determined later on.

Taking Propositions 14.1.1 and 14.1.2 into account, it is easy to check that the function ξ is a classical solution of the Cauchy problem

$$\begin{cases} D_t \xi(t,\cdot) = \mathcal{A}_{\varepsilon} \xi(t,\cdot) + g_{\varepsilon}(t,\cdot), & t > 0, \\ \xi(0,\cdot) = \frac{1}{2} \alpha^3 f^2 + \langle F \, Df, Df \rangle + \langle E \, D_{*,1}^2 f, D_{*,1}^2 f \rangle + \alpha^{-\frac{4}{3}} \langle K_1 D_{*,1}^3 f, D_{*,1}^3 f \rangle, \end{cases}$$

where $g^{(\varepsilon)}=g_1^{(\varepsilon)}+g_2+g_3$, and $g_1^{(\varepsilon)}$, g_2 and g_3 are given by (14.2.15)-(14.2.17) with F, G(t) and H(t) as in (14.4.34) and (14.4.35). To get the assertion, as in the proof of Theorem 14.2.4, it suffices to show that there exists $\alpha>0$ such that F, G(t) and H(t) are strictly positive definite for any t>0 and $g^{(\varepsilon)}\leq 0$ in $(0,T_0]\times\mathbb{R}^N$ for some T_0 independent of ε . Indeed, then Theorem 14.2.6 and the semigroup rule will lead us to (14.4.32) We refer the reader to the proof of Theorem 14.2.4 for more details.

Due to the particular structure of the matrices F, G(t) and H(t), it is easy to check that the previous matrices are strictly positive definite for any t>0, if α is sufficiently large. Hence, we can limit ourselves to showing that $g_{\varepsilon} \leq 0$ in $(0, T_0] \times \mathbb{R}^N$ for a suitable choice of T_0 and the parameter α . We begin by estimating the function $g_1^{(\varepsilon)}$. As it is immediately seen, $g_1^{(\varepsilon)} \leq g_1$, where g_1 is obtained from $g_1^{(\varepsilon)}$ by replacing everywhere the matrix $Q^{(\varepsilon)}$ with the matrix $Q^{(0)}$. Arguing as in the proof of (14.2.20) we can prove that

$$\begin{split} g_1(t) & \leq -\alpha^3 \kappa |D_{*,1} u(t)|^2 - \kappa \{2\alpha - (2\|B_4\| + 1)\alpha^{\frac{1}{2}} \|K_2\|\} |D_{*,1}^2 u(t)|^2 \\ & - \kappa \{2\iota - (2\|B_4\| + 1)\alpha^{-\frac{1}{2}} \|K_2\|\} |D_{*,2}^2 u(t)|^2 - 2\kappa |D_{*,1}^3 u(t)|^2 \\ & - \kappa t (2\alpha^{-\frac{7}{16}} - \alpha^{-\frac{3}{5}} \|K_5\|) |D_{*,2}^3 u(t)|^2 - \kappa t^3 (2\alpha^{-\frac{7}{8}} - \alpha^{-1} \|K_5\|) |D_{*,3}^3 u(t)|^2 \\ & - \kappa \{2\alpha^{-\frac{4}{3}} - \alpha^{-\frac{3}{2}} (\|K_9\| + \|K_{10}\| + \|K_{11}\|)\} |D_{*,1}^4 u(t)|^2 \\ & - \kappa t^2 (2\alpha^{-\frac{8}{9}} - \alpha^{-\frac{7}{6}} \|K_9\|) |D_{*,2}^4 u(t)|^2 \end{split}$$

$$-\kappa t^{4} (2\alpha^{-1} - \alpha^{-\frac{7}{6}} \|K_{10}\| - \alpha^{-\frac{49}{48}} \|K_{12}\|) |D_{*,3}^{4} u(t)|^{2} -\kappa t^{6} (2\alpha^{-\frac{9}{8}} - \alpha^{-\frac{7}{6}} \|K_{11}\| - \alpha^{-\frac{55}{48}} \|K_{12}\|) |D_{*,4}^{4} u(t)|^{2}.$$
 (14.4.36)

The function g_2 can be estimated using properly the inequality (14.2.19) and taking (14.2.21) and (14.2.22) into account. We get

$$\begin{split} g_2(t) &\leq \{2\alpha \|B_1\| + o_{\alpha}(\alpha)\} |D_{*,1}u(t)|^2 + \{-\iota + o_{\alpha}(1)\} |D_{*,2}u(t)|^2 \\ &+ \{2\|L_1\| + o_{\alpha}(1) + o_t(1)\} |D_{*,1}^2u(t)|^2 \\ &+ \{\alpha^{-\frac{1}{16}} \|L_5\| + o_{\alpha}(\alpha^{-\frac{1}{16}}) + o_t(1)\} |D_{*,2}^2u(t)|^2 \\ &+ t^2 \{-2\alpha^{-\frac{4}{5}} \lambda_{\min}(L_5^*L_5) + o_{\alpha}(\alpha^{-\frac{4}{5}}) + o_t(1)\} |D_{*,3}^2u(t)|^2 \\ &+ \{2\alpha^{-\frac{4}{3}} \|M_1\| + o_t(1)\} |D_{*,1}^3u(t)|^2 \\ &+ t \{\alpha^{-\frac{8}{9}} (\|M_5\| + 2) + o_{\alpha}(\alpha^{-\frac{8}{9}}) + o_t(1)\} |D_{*,2}^3u(t)|^2 \\ &+ t^3 \{\alpha^{-\frac{8}{9}} \|M_5\| + o_{\alpha}(\alpha^{-\frac{8}{9}}) + o_t(1)\} |D_{*,3}^3u(t)|^2 \\ &+ t^5 \{-2\alpha^{-\frac{13}{12}} \lambda_{\min}(M_8^*M_8) + o_{\alpha}(\alpha^{-\frac{13}{12}}) + o_t(1)\} |D_{*,4}^3u(t)|^2, (14.4.37) \end{split}$$

for any t>0, where, to simplify the notation, we denote by $o_t(t^k)$ $(k\geq 0)$ any function of t (possibly depending also on α) such that $\lim_{t\to 0} t^{-k} o_t(t^k) = 0$, and by $o_{\alpha}(\alpha^k)$ $(k\in\mathbb{R})$ any function depending only on α and such that $\lim_{\alpha\to +\infty} \alpha^{-k} o_{\alpha}(\alpha^k) = 0$. Note that, according to the proof of Theorem 14.2.4, $\lambda_{\min}(L_5L_5^*)$ and $\lambda_{\min}(M_8^*M_8)$ are both positive.

Let us now consider the function g_3 . Taking (14.2.26)-(14.2.30) and Hypothesis 14.0.1(ii) into account, and applying (14.2.19) with $a = \sqrt{\kappa} |D_{*,i}^h u|$ and $b = |D_{*,j}^h u|$ ($i \le k, j = 1, ..., k + 1$), it is easy to check that all the terms in the definition of the function g_3 are negligible (as t tends to 0^+) with respect to the terms in (14.4.36) and (14.4.37). Now, combining these two estimates, we get

$$\begin{split} g^{(\varepsilon)}(t) &\leq \{-\alpha^3 + o_{\alpha}(\alpha^3)\} |D_{*,1}u(t)|^2 + \{-\iota + o_{\alpha}(1)\} |D_{*,2}u(t)|^2 \\ &\quad + \{-2\alpha + o_{\alpha}(\alpha) + o_t(1)\} \kappa |D_{*,1}^2 u(t)|^2 \\ &\quad + \{-2\iota + o_{\alpha}(1) + o_t(1)\} \kappa |D_{*,2}^2 u(t)|^2 \\ &\quad + t^2 \{-2\lambda_{\min}(L_5^*L_5)\alpha^{-\frac{4}{5}} + o_{\alpha}(\alpha^{-\frac{4}{5}}) + o_t(1)\} |D_{*,3}^2 u(t)|^2 \\ &\quad + \{-2\kappa + o_{\alpha}(1) + o_t(1)\} |D_{*,1}^3 u(t)|^2 \\ &\quad + t \{-2\alpha^{-\frac{7}{16}} \kappa + o_{\alpha}(\alpha^{-\frac{7}{16}}) + o_t(1)\} |D_{*,2}^3 u(t)|^2 \\ &\quad + t^3 \{-2\alpha^{-\frac{7}{8}} \kappa + o_{\alpha}(\alpha^{-\frac{7}{8}}) + o_t(1)\} |D_{*,3}^3 u(t)|^2 \\ &\quad + t^5 \{-2\alpha^{-\frac{13}{12}} \lambda_{\min}(M_8^*M_8) + o_{\alpha}(\alpha^{-\frac{13}{12}}) + o_t(1)\} |D_{*,4}^3 u(t)|^2 \\ &\quad - \{2\alpha^{-\frac{4}{3}} + o_{\alpha}(\alpha^{-\frac{4}{3}}) + o_t(1)\} \kappa |D_{*,1}^4 u(t)|^2 \end{split}$$

$$-t^{2} \{ 2\alpha^{-\frac{8}{9}} + o_{\alpha}(\alpha^{-\frac{8}{9}}) + o_{t}(1) \} \kappa |D_{*,2}^{4}u(t)|^{2}$$

$$-t^{4} \{ 2\alpha^{-1} + o_{\alpha}(\alpha^{-1}) + o_{t}(1) \} \kappa |D_{*,3}^{4}u(t)|^{2}$$

$$-t^{6} \{ 2\alpha^{-\frac{9}{8}} + o_{\alpha}(\alpha^{-\frac{9}{8}}) \} \kappa |D_{*,4}^{4}u(t)|^{2}.$$

$$(14.4.38)$$

It is now clear that we can fix T_0 sufficiently close to 0 and α large enough, so as to make all the terms in (14.4.38) negative in $(0, T_0] \times \mathbb{R}^N$.

We are now able to prove (14.4.5).

Proposition 14.4.7 For any $0 \le \alpha \le \beta \le 3$ there exist two positive constants $C = C(\alpha, \beta)$ and $\omega = \omega(\alpha, \beta)$ such that (14.4.5) holds true.

Proof. As it has been already mentioned, the proof is based on an interpolation argument and it is similar to that of Theorem 6.1.8. For the reader's convenience, we go into details.

As a first step, we apply (A.4.3) with $X_1 = X_2 = C_b(\mathbb{R}^N)$, $Y_1 = Y_2 = C_b^{3,1}(\mathbb{R}^N)$ and $\theta = \alpha$. Taking (14.4.6) into account, we get

$$||T(t)||_{L(\mathcal{C}_{b}^{3\alpha,\alpha}(\mathbb{R}^{N}))} \le C_1 e^{\omega_1 t}, \qquad t > 0,$$
 (14.4.39)

for some positive constants C_1 and ω_1 , independent of t. Next, applying (A.4.3) with $X_1 = C_b(\mathbb{R}^N)$, $X_2 = Y_1 = Y_2 = C_b^{3,1}(\mathbb{R}^N)$, $\theta = \alpha$ and, still taking (14.4.6) into account, we deduce that

$$||T(t)||_{L(\mathcal{C}_{h}^{3\alpha,\alpha}(\mathbb{R}^{N}),C_{h}^{3,1}(\mathbb{R}^{N}))} \le C_{2}e^{\omega_{2}t}t^{-\frac{3(1-\alpha)}{2}}, \qquad t > 0,$$
 (14.4.40)

for some positive constants C_2 and ω_2 , independent of t.

Finally, interpolating (14.4.39) and (14.4.40), with $\theta = (\beta - \alpha)/(1 - \alpha)$, and taking (14.4.31) into account, we deduce (14.4.5) in its generality.

In order to prove Theorem 14.4.1, we show that for any $f \in C_b(\mathbb{R}^N)$ and any $\lambda > 0$ the function

$$u(x) = \int_0^{+\infty} e^{-\lambda t} (T(t)f)(x)dt, \qquad x \in \mathbb{R}^N, \tag{14.4.41}$$

is a distributional solution to the elliptic equation

$$\lambda u - \mathcal{A}u = f. \tag{14.4.42}$$

Moreover, we prove that if $f \in \mathcal{C}_b^{\theta,\theta/3}(\mathbb{R}^N)$ for some $\theta \in (0,1)$, then $u \in \mathcal{C}_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)$ and it satisfies (14.4.3). Finally, we show that the function u in (14.4.41) is the unique solution of the equation (14.4.42) among all the distributional solutions which are twice continuously differentiable with respect to the first r variables, and are bounded with their classical derivatives.

The following proposition provides some regularity properties of the functions belonging to $D(\widehat{A})$.

Proposition 14.4.8 Let $u \in D(\widehat{A})$ be such that $\widehat{A}u \in C_b^{\theta,\theta/3}(\mathbb{R}^N)$ for some $\theta \in [0,1)$. Then, $u \in C_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)$ and there exists a positive constant C_1 , independent of u, such that

$$||u||_{\mathcal{C}_{b}^{2+\theta,(2+\theta)/3}(\mathbb{R}^{N})} \le C_{1}(||u||_{\infty} + ||\widehat{A}u||_{\mathcal{C}_{b}^{\theta,\theta/3}(\mathbb{R}^{N})}).$$
(14.4.43)

In particular, $D(\widehat{A})$ is continuously embedded in $C_b^{2,2/3}(\mathbb{R}^N)$. Finally, if $f \in C_b^{\theta,\theta/3}(\mathbb{R}^N)$ and u solves the equation $\lambda u - \widehat{A}u = f$, then

$$||u||_{\mathcal{C}_{b}^{2+\theta,(2+\theta)/3}(\mathbb{R}^{N})} \le C_{2}||f||_{\mathcal{C}_{b}^{\theta,\theta/3}(\mathbb{R}^{N})},$$
 (14.4.44)

where C_2 is independent of f and u.

Proof. In view of Corollary 14.4.4, to prove that $D(\widehat{A}) \subset \mathcal{C}_b^{2,2/3}(\mathbb{R}^N)$ with a continuous embedding, one can fix a suitable large λ_0 (i.e., λ_0 larger than the constant ω appearing in Proposition 14.4.7) and show that the function $R(\lambda_0, \widehat{A})f$ belongs to $(C_b^{\alpha,\alpha/3}(\mathbb{R}^N), C_b^{2+\alpha,(2+\alpha)/3}(\mathbb{R}^N))_{1-\alpha/2}$ for any $f \in C_b(\mathbb{R}^N)$. This can be done using the same techniques as in the proof of Theorem 6.2.2, which rely on the estimates (14.4.7). Similarly, writing $u = R(\lambda, \widehat{A})(\lambda u - \widehat{A}u)$, one can then easily show (14.4.43) and (14.4.44) (with $\lambda > \omega$, where ω is still given by Proposition 14.4.7). Hence, we skip the details.

To prove (14.4.44) (with $\lambda \in (0, \omega]$), it suffices to observe that, using the resolvent identity (A.3.2), one can easily check that if u satisfies the equation $\lambda u - \widehat{A}u = f$, then $\lambda_0 u - \widehat{A}u = g := f + (\lambda_0 - \lambda)u$ and $u = R(\lambda_0, A)g$. The arguments needed to prove the first part of the proposition (see also the proof of (6.2.8)) show that

$$||u||_{C_b^{\theta,\theta/3}(\mathbb{R}^N)} \le K_1 ||g||_{C_b(\mathbb{R}^N)} \le C \left(||f||_{\infty} + |\lambda - \lambda_0|||u||_{\infty} \right) \le K_2 ||f||_{\infty},$$

for some positive constants K_1 and K_2 , independent of f. It follows that

$$\|g\|_{C_b^{\theta,\theta/3}(\mathbb{R}^N)} \le (K_2+1)\|f\|_{C_b^{\theta,\theta/3}(\mathbb{R}^N)}.$$
 (14.4.45)

Hence, (14.4.44) (with $\lambda = \lambda_0$) and (14.4.45) imply that

$$\|u\|_{C_b^{2+\theta,(2+\theta)/3}(\mathbb{R}^N)} \leq C_2 \|g\|_{C_b^{\theta,\theta/3}(\mathbb{R}^N)} \leq C_2 (K_2+1) \|f\|_{C_b^{\theta,\theta/3}(\mathbb{R}^N)}$$

and we are done.

In order to prove that the problem (14.4.42) admits actually a unique distributional solution which is twice continuously differentiable with respect to the first r variables, we need the following lemma.

Lemma 14.4.9 Let \mathcal{B} the first-order differential operator formally defined by

$$\mathcal{B}u(x) = \langle Bx, Du(x) \rangle, \qquad x \in \mathbb{R}^N.$$

Then, for any $u \in C_b(\mathbb{R}^N)$ such that $\mathcal{B}u \in C_b(\mathbb{R}^N)$, there exists a sequence of smooth functions u_n such that $\mathcal{B}u_n \in C_b(\mathbb{R}^N)$ for any $n \in \mathbb{N}$ and it converges to $\mathcal{B}u$ locally uniformly.

Proof. Let φ be a smooth function compactly supported in B(1), such that $0 \leq \varphi(x) \leq 1$ for any $x \in \mathbb{R}^N$ and $\|\varphi\|_{L^1(\mathbb{R}^N)} = 1$. For any $n \in \mathbb{N}$ we define the function $u_n : \mathbb{R}^N \to \mathbb{R}$ by setting

$$u_n(x) = (u \star \varphi_n)(x) := \int_{\mathbb{R}^N} u(x - y)\varphi_n(y)dy, \qquad x \in \mathbb{R}^N, \qquad (14.4.46)$$

where $\varphi_n(x) = n^N \varphi(nx)$ for any $x \in \mathbb{R}^N$,

As it is easily seen, u_n converges to u as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . Let us show that $\mathcal{B}u_n$ converges to $\mathcal{B}u$ locally uniformly in \mathbb{R}^N , as well. To compute $\mathcal{B}u_n$ we first assume that $u \in C_b^1(\mathbb{R}^N)$ and observe that, for any $i, j = 1, \ldots, N$, we have

$$\begin{split} x_j D_i u_n(x) &= x_j \int_{\mathbb{R}^N} (D_i u)(x-y) \varphi_n(y) dy \\ &= \int_{\mathbb{R}^N} (x_j - y_j)(D_i u)(x-y) \varphi_n(y) dy + \int_{\mathbb{R}^N} (D_i u)(x-y) y_j \varphi_n(y) dy \\ &= \int_{\mathbb{R}^N} (x_j - y_j)(D_i u)(x-y) \varphi_n(y) dy + \int_{\mathbb{R}^N} u(x-y) D_i(y_j \varphi_n(y)) dy \\ &= \int_{\mathbb{R}^N} (x_j - y_j)(D_i u)(x-y) \varphi_n(y) dy + \delta_{ij} \int_{\mathbb{R}^N} u(x-y) \varphi_n(y) dy \\ &+ \int_{\mathbb{R}^N} u(x-y) y_j D_i \varphi_n(y) dy. \end{split}$$

Therefore,

$$\mathcal{B}u_n = \mathcal{B}u \star \varphi_n + \text{Tr}(B)u_n + u \star \mathcal{B}\varphi_n. \tag{14.4.47}$$

Now, fix $u \in C_b(\mathbb{R}^N)$ and let $\{u_m\} \subset C_b^1(\mathbb{R}^N)$ be a sequence of smooth functions converging to u locally uniformly. Moreover, let $u_n^m = u_m \star \varphi_n$. As we can easily see, u_n^m and $\mathcal{B}u_n^m$ converge, respectively, to u_n and $\mathcal{B}u_n$, locally uniformly in \mathbb{R}^N , as m tends to $+\infty$. Moreover, $\mathcal{B}u^m$ converges to $\mathcal{B}u$ in the sense of distributions. This implies that $\mathcal{B}u^m \star \varphi_n$ converges to $\mathcal{B}u \star \varphi_n$ pointwise in \mathbb{R}^N as m tends to $+\infty$. Therefore, writing (14.4.47) with u_n replaced with u_n^m and letting m go to $+\infty$, we see that (14.4.47) holds true also in the case when $u \in C_b(\mathbb{R}^N)$.

Now, we are almost done. Indeed, since $\mathcal{B}u \in C_b(\mathbb{R}^N)$, then $\mathcal{B}u \star \varphi_n$ converges locally uniformly to $\mathcal{B}u$ as n tends to $+\infty$. Finally,

$$(u \star \mathcal{B}\varphi_n)(x) = \sum_{i,j=1}^N b_{ij} \int_{\mathbb{R}^N} u(x-y)y_j D_i \varphi_n(y) dy$$
$$= \sum_{i,j=1}^N b_{ij} \int_{\mathbb{R}^N} u(x-z/n)z_j D_i \varphi(z) dz$$

and the last side of the previous chain of equalities tends to

$$u(x)\sum_{i,j=1}^{N}b_{ij}\int_{\mathbb{R}^{N}}z_{i}D_{j}\varphi(z)dz = -u(x)\operatorname{Tr}(B),$$

as n tends to $+\infty$, locally uniformly in \mathbb{R}^N . Therefore, $\mathcal{B}u_n$ converges locally uniformly to $\mathcal{B}u$, as n tends to $+\infty$.

We are now able to prove Theorem 14.4.1.

Proof of Theorem 14.4.1. (Existence). The existence part follows immediately from Propositions 14.3.5 and 14.4.8. Indeed, Proposition 14.3.5 shows that the function $u = R(\lambda, \widehat{A})f$ is a distributional solution to the equation (14.4.42), whereas Proposition 14.4.8 ensures that u enjoys all the regularity properties claimed in the statement of the theorem.

(Uniqueness). Here, we prove that $v \equiv 0$ is the unique distributional solution to the equation $\lambda v - \mathcal{A}v = 0$ which is bounded and continuous in \mathbb{R}^N and admits classical derivatives $D_{ij}v \in C_b(\mathbb{R}^N)$ for any $i,j=1,\ldots,r$. This is equivalent to prove that any solution v to the previous equation, with the claimed regularity, belongs to $D(\widehat{A})$. For this purpose, let $\{\hat{v}_m\}$ be the sequence defined by $\hat{v}_m = T(1/m)v$ for any $m \in \mathbb{N}$. By Theorem 14.2.7, any function \hat{v}_m belongs to $C_b^2(\mathbb{R}^N)$ and \hat{v}_m converges locally uniformly to v as m tends to $+\infty$. Moreover, $\|\hat{v}_m\|_{\infty} \leq \|v\|_{\infty}$ for any $m \in \mathbb{N}$. We now show that

$$\mathcal{A}\hat{v}_m = T(1/m)\mathcal{A}v, \qquad m \in \mathbb{N}. \tag{14.4.48}$$

This will imply that $\mathcal{A}\hat{v}_m$ converges locally uniformly to $\mathcal{A}v$ and $\|\mathcal{A}\hat{v}_m\|_{\infty} \leq \|\mathcal{A}v\|_{\infty}$ for any $m \in \mathbb{N}$. Hence, by Proposition 14.3.5, $v \in D(A)$ and, consequently, $v \equiv 0$.

So, let us prove (14.4.48). For this purpose, we approximate v by convolution, by the sequence of smooth functions $\{v_n\} \in C_b^2(\mathbb{R}^N)$ defined as in (14.4.46). As we can easily see, v_n converges locally uniformly to v as n tends to $+\infty$ and D_iv_n , $D_{ij}v_n$ converge locally uniformly, respectively, to D_iv and $D_{ij}v$ for any $1 \leq i, j \leq r$. Moreover, by Lemma 14.4.9, $\mathcal{B}v_n$ converges to $\mathcal{B}v$ locally uniformly in \mathbb{R}^N . Therefore, $\mathcal{A}v_n$ converges to $\mathcal{A}v$ locally uniformly in \mathbb{R}^N , as well. Now, the formula (14.4.48) follows observing that

 $T(1/m)\mathcal{A}v_n = \mathcal{A}T(1/m)v_n$ for any $m, n \in \mathbb{N}$ (see Lemma 14.3.4) and letting n tend to $+\infty$. Indeed, according to Proposition 14.3.1, $T(1/m)v_n$ converges to \hat{v}_m in $C^2(K)$ and $T(1/m)\mathcal{A}v_n$ converges to $T(1/m)\mathcal{A}v$ in C(K) for any compact set $K \subset \mathbb{R}^N$. This finishes the proof.

The results of Theorem 14.4.1 can be improved if f is smoother. In such a case it can be proved that the function $u = R(\lambda, \widehat{A})f$ is a classical solution to the equation (14.4.1), in the sense that all the derivatives involved in the definition of $\mathcal{A}u$ exists in the classical sense and they are bounded and continuous in \mathbb{R}^N . Since the proof of the next corollary is similar to, and even simpler than the proof of Corollary 14.4.11, we do not go into details.

Corollary 14.4.10 Suppose that $f \in C_b^{\theta}(\mathbb{R}^N)$ for some $\theta > 1/3$. Then, the function $u = R(\lambda, A)f$ belongs to $C_b^{\theta+2/3}(\mathbb{R}^N)$ and there exists a positive constant C, independent of f, such that

$$||u||_{C_b^{\theta+2/3}(\mathbb{R}^N)} \le C||f||_{C_b^{\theta}(\mathbb{R}^N)}.$$

Now, we consider the parabolic nonhomogeneous Cauchy problem (14.4.2). We are going to show that its unique distributional solution, with the regularity properties claimed in Theorem 14.4.1, is the function u given by the variation-of-constants formula

$$u(t,x) = (T(t)f)(x) + \int_0^t (T(t-s)g(s,\cdot))(x)ds, \qquad t \in [0,T], \quad x \in \mathbb{R}^N.$$
(14.4.49)

As a first step we observe that the convolution term in (14.4.49) is well defined. Of course, it suffices to check that for any $x \in \mathbb{R}^N$, the function $w : [0,T] \times [0,T] \to \mathbb{R}$ defined by w(r,s) = (T(r)f(s))(x) for any $r,s \in [0,T]$ is continuous. For this purpose we observe that, by Theorem 14.2.7, the function $w(\cdot,s)$ is continuous for any $s \in [0,T]$, and by Proposition 14.3.1, $w(r,\cdot)$ is continuous in [0,T], uniformly with respect to $r \in [0,T]$.

We are now in a position to prove Theorem 14.4.2.

Proof of Theorem 14.4.2. (Existence). Of course, according to Theorem 14.2.7, we can restrict ourselves to proving the assertion in the case when $f \equiv 0$. For this purpose, we begin by proving that the function

$$v(t,x) = \int_0^t (T(t-s)g(s,\cdot))(x)ds, \qquad t \in [0,T], \ x \in \mathbb{R}^N$$
 (14.4.50)

is a distributional solution to (14.4.2) (with $f \equiv 0$). For this purpose, we introduce a sequence $\{g_n\} \in C_b^{1,2}([0,T] \times \mathbb{R}^N)$ converging locally uniformly in $[0,T] \times \mathbb{R}^N$ to g, and we denote by v_n the function defined by (14.4.50) with g being replaced with g_n . Using the estimate (14.2.34) with h = k = 2 we can show that v_n is a classical solution to the problem (14.4.2), where we

take $f \equiv 0$ and we replace g with g_n (see also the proof of the forthcoming Corollary 14.4.11). Moreover, v_n converges to v uniformly in $[0,T] \times K$ for any compact set $K \subset \mathbb{R}^N$. Indeed,

$$\sup_{\substack{(t,x)\in[0,T]\times K}} |v_n(t,x)-v(t,x)|$$

$$= \sup_{\substack{(t,x)\in[0,T]\times K}} \left| \int_0^t (T(s)(g_n(t-s,\cdot)-g(t-s,\cdot)))(x)ds \right|$$

$$\leq \sup_{x\in K} \int_0^T \left(\sup_{r\in[0,T]} |g_n(r,\cdot)-g(r,\cdot)|\right)(x)ds$$

(recall that $\{T(t)\}$ is an order preserving semigroup, see Theorem 14.2.7). Since $\sup_{t\in[0,T]}|g_n(r,\cdot)-g(r,\cdot)|$ converges to 0 locally uniformly in \mathbb{R}^N , as n tends to $+\infty$, Proposition 14.3.1 implies that the last side of the previous chain of inequalities vanishes as n tends to $+\infty$. Hence, v_n tends to v uniformly in $[0,T]\times K$.

Now, we observe that, for any smooth function $\varphi \in C_c^{\infty}((0,T) \times \mathbb{R}^N)$,

$$\int_{(0,T)\times\mathbb{R}^N} g_n \varphi \, dt \, dx = \int_{(0,T)\times\mathbb{R}^N} (D_t v_n - \mathcal{A}v_n) \varphi \, dt \, dx$$
$$= \int_{(0,T)\times\mathbb{R}^N} v_n (-D_t \varphi - \mathcal{A}^* \varphi) \, dt \, dx,$$

where \mathcal{A}^* is given by (14.3.9). Letting n go to $+\infty$ we deduce that v is a distributional solution of (14.4.2), with $f \equiv 0$.

The regularity properties of u and the estimate (14.4.4) can be proved arguing as in the proof of Proposition 6.2.6, taking Proposition 14.4.7 into account.

(Uniqueness). Let us now prove that the function u in (14.4.49) is the unique distributional solution to the problem (14.4.2) which is twice continuously differentiable in $[0,T]\times\mathbb{R}^N$ with respect to the space variables x_j $(j=1,\ldots,r)$. We adapt to this situation the proof of Theorem 14.4.1. For this purpose, let w be a solution to the Cauchy problem (14.4.2) with $f\equiv 0$ and $g\equiv 0$, such that $w, D_i w, D_{ij} w \in C_b([0,T]\times\mathbb{R}^N)$ for any $i,j=1,\ldots,r$. We extend w by continuity to $\mathbb{R}\times\mathbb{R}^N$ by setting $\overline{w}(t,\cdot)=0$ for any t<0 and $\overline{w}(t,\cdot)=w(T,\cdot)$ for any t>T. The so extended function is as smooth as w is, i.e., $D_i\overline{w}$ and $D_{ij}\overline{w}$ are continuous in $\mathbb{R}\times\mathbb{R}^N$ for any $i,j\leq r$. Then, we regularize \overline{w} by convolution by setting

$$w_n(t,x) = (\overline{w} \star (\varrho_n \varphi_n))(t,x) := \int_{\mathbb{R} \times \mathbb{R}^N} \overline{w}(t-s,x-y)\varrho_n(s)\varphi_n(y) \, ds \, dy,$$
(14.4.51)

with $\varrho_n(s) = n\varrho(ns)$, $\varphi_n(x) = n^N \varphi(nx)$, where $\varrho \in C_c^{\infty}(\mathbb{R})$ and $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ are such that $\chi_{(-1/2,1/2)} \leq \varrho \leq \chi_{(-1,1)}$, $\chi_{B(1/2)} \leq \varphi \leq \chi_{B(1)}$, and $\|\varrho\|_{L^1(\mathbb{R})} = \|\varphi\|_{L^1(\mathbb{R}^N)} = 1$.

Let us show that $D_t w_n - \mathcal{A}w_n$ converges to 0 locally uniformly in $(0,T) \times \mathbb{R}^N$. For this purpose, we observe that the smoothness of w implies that $\mathcal{A}w \in C([0,T] \times \mathbb{R}^N)$ and that $\mathcal{A}w_n$ converges to $\mathcal{A}w$ locally uniformly in $(0,T) \times \mathbb{R}^N$. Here, $D_t w - \mathcal{B}w$ is meant in the sense of distributions. Therefore, we can restrict ourselves to showing that $(D_t - \mathcal{B})w_n$ converges to $(D_t - \mathcal{B})w \in C([0,T] \times \mathbb{R}^N)$ locally uniformly in $(0,T) \times \mathbb{R}^N$. To show such a property, we first prove that

$$(D_t - \mathcal{B})w_n = \overline{(D_t w - \mathcal{B}w)} \star \varrho_n \varphi_n + \text{Tr}(B)w_n + \overline{w} \star (\varrho_n \mathcal{B}(\varphi_n)), \quad (14.4.52)$$

in $(1/n, T - 1/n) \times \mathbb{R}^N$, where $\overline{D_t w - \mathcal{B}w}$ denotes any continuous extension of $D_t w - \mathcal{B}w$ to the whole of $\mathbb{R} \times \mathbb{R}^N$.

An integration by parts shows that (14.4.52) holds true in the case when $\overline{w} \in C_b^{1,2}([0,T] \times \mathbb{R}^N)$. In the general case, we approximate \overline{w} by a sequence of smooth functions $\{w_m\} \subset C_b^{1,2}(\mathbb{R} \times \mathbb{R}^N)$ converging to \overline{w} locally uniformly in $\mathbb{R} \times \mathbb{R}^N$, and we define the function w_n^m $(m,n \in \mathbb{N})$ accordingly to (14.4.51), with \overline{w} being replaced with w_m . As it is immediately seen,

$$(D_t - \mathcal{B})w_n^m = (D_t w^m - \mathcal{B}w^m) \star \varrho_n \varphi_n + \text{Tr}(B)w_n^m + w^m \star (\varrho_n \mathcal{B}\varphi_n),$$

in $[0,T] \times \mathbb{R}^N$. Since w^m converges to \overline{w} locally uniformly, then the function $\operatorname{Tr}(B)w_n^m + w^m \star (\varrho_n \mathcal{B}\varphi_n)$ converges to $\operatorname{Tr}(B)w_n + \overline{w} \star (\varrho_n \mathcal{B}\varphi_n)$, locally uniformly in $[0,T] \times \mathbb{R}^N$. As far as the term $(D_t w^m - \mathcal{B}w^m) \star \varrho_n \varphi_n$ is concerned, we observe that

$$((D_t w^m - \mathcal{B}w^m) \star \varrho_n \varphi_n)(t, x)$$

$$= \int_{[t-1/n, t+1/n] \times (x+B(1/n))} (D_t w^m - \mathcal{B}w^m)(s, y) \varrho_n(t-s) \varphi_n(x-y) \, ds \, dy,$$
(14.4.53)

for any t>0 and any $x\in\mathbb{R}^N$. Suppose that $t\in(1/n,T-1/n)$. Then the function $\varrho_n(t-\cdot)\varphi_n(x-\cdot)$ is compactly supported in $(0,T)\times\mathbb{R}^N$. Therefore, since $D_tw^m-\mathcal{B}w^m$ converges to $D_tw-\mathcal{B}w\in C([0,T]\times\mathbb{R}^N)$ in the sense of distributions, letting m go to $+\infty$ in (14.4.53), we deduce that

$$\lim_{m \to +\infty} ((D_t w^m - \mathcal{B} w^m) \star \varrho_n \varphi_n)(t, x)$$

$$= \int_{[t-1/n, t+1/n] \times (x+B(1/n))} (D_t w - \mathcal{B} w)(s, y) \varrho_n(t-s) \varphi_n(x-y) \, ds \, dy$$

$$= \int_{\mathbb{R} \times \mathbb{R}^N} (\overline{D_t w - \mathcal{B} w})(t-s, x-y) \varrho_n(s) \varphi_n(y) \, ds \, dy,$$

for any $t \in (1/n, T-1/n)$ and any $x \in \mathbb{R}^N$. The formula (14.4.52) follows. Now, arguing as in the proof of Lemma 14.4.9, from (14.4.52) we can conclude that $D_t w_n - \mathcal{B} w_n$ converges locally uniformly in $(0, T) \times \mathbb{R}^N$ to $D_t w - \mathcal{B} w$. Therefore, $D_t w_n - \mathcal{A} w_n$ converges to 0 locally uniformly in $(0,T) \times \mathbb{R}^N$, as n tends to $+\infty$.

To conclude the proof, we observe that, for any $n \in \mathbb{N}$ and any $t_0 \in (0, T)$, the function $z_n = w_n(\cdot + t_0, \cdot)$ is a classical solution to the Cauchy problem

$$\begin{cases}
D_t z_n(t, x) = \mathcal{A} z_n(t, x) + f_n(t + t_0, x), & t \in [0, T - t_0], x \in \mathbb{R}^N, \\
z_n(0, x) = w_n(t_0, x), & x \in \mathbb{R}^N,
\end{cases} (14.4.54)$$

where $f_n := D_t z_n - \mathcal{A} z_n$. Since $f_n \in C_b^{1,2}([0,T] \times \mathbb{R}^N)$, the function $h_n : [0,T-t_0] \times \mathbb{R}^N \to \mathbb{R}$ defined by

$$h_n(t,x) = (T(t)w_n(t_0,\cdot))(x) + \int_0^t (T(t-s)f_n(t+t_0-s,\cdot))(x)ds,$$

for any $t \in [0, T - t_0]$ and any $x \in \mathbb{R}^N$, is a classical solution to the Cauchy problem (14.4.54) (see also the proof of Corollary 14.4.11). The maximum principle (see Theorem 14.2.6) implies that $h_n = z_n$. Now, since z_n , $f_n(\cdot + t_0, \cdot)$ converge, locally uniformly in $[0, T - t_0] \times \mathbb{R}^N$ to $w(\cdot + t_0, \cdot)$ and 0, respectively, taking Proposition 14.3.1 into account, we get

$$w(t+t_0,x) = (T(t)w(t_0,\cdot))(x), t \in [0,T-t_0], x \in \mathbb{R}^N,$$

for any $t_0 \in (0, T)$. Letting t_0 go to 0, and recalling that $u(0, \cdot) = 0$, we deduce that $w \equiv 0$. This finishes the proof.

To conclude this section we show that the results in Theorem 14.4.2 can be improved if f and g are smoother. In such a case the function u in (14.4.49) is a classical solution to the Cauchy problem (14.4.2), in the sense that the time derivative of u as well as all the space derivatives involved in the definition of Au exist in the classical sense and they are bounded and continuous in $[0,T] \times \mathbb{R}^N$.

Corollary 14.4.11 Suppose that the assumptions of Theorem 14.4.2 are satisfied. Further, suppose that $f \in C_b^{\theta+2/3}(\mathbb{R}^N)$, $g(t,\cdot) \in C_b^{\theta}(\mathbb{R}^N)$ for some $\theta > 1/3$ and any $t \in [0,T]$, with

$$\sup_{t\in[0,T]}\|g(t,\cdot)\|_{C^\theta_b(\mathbb{R}^N)}<+\infty.$$

Then, the function u in (14.4.49) is once continuously differentiable in $[0,T] \times \mathbb{R}^N$ with respect to the time and space variables. Moreover, for any $t \in [0,T]$, $u(t,\cdot)$ belongs to $C_b^{\theta+2/3}(\mathbb{R}^N)$ and there exists a positive constant C, independent of f and g, such that

$$\sup_{t \in [0,T]} \|u(t,\cdot)\|_{C_b^{\theta+2/3}(\mathbb{R}^N)} \le C \left(\|f\|_{C_b^{\theta+2/3}(\mathbb{R}^N)} + \sup_{t \in [0,T]} \|g(t,\cdot)\|_{C_b^{\theta}(\mathbb{R}^N)} \right). \tag{14.4.55}$$

Proof. As in the proof of Theorem 14.4.2, we can restrict ourselves to dealing with the function v defined in (14.4.50). Using an interpolation argument similar to that used in the proof of Proposition 14.4.7, from (14.2.1)-(14.2.3) and (14.2.34) (with h=k) we deduce that, for any $0 \le \alpha \le \beta \le 3$ and any $\omega > 0$, there exist two positive constants C and ω such that

$$||T(t)||_{L(C_h^{\alpha}(\mathbb{R}^N), C_h^{\beta}(\mathbb{R}^N))} \le Ce^{\omega t}t^{-\frac{3(\beta-\alpha)}{2}}, \qquad t > 0.$$
 (14.4.56)

Now, arguing as in the proof of Proposition 6.2.6, we can prove that $v(t,\cdot)\in C_b^{\theta+2/3}(\mathbb{R}^N)$ for any t>0 and

$$\sup_{t \in [0,T]} \|v(t,\cdot)\|_{C_b^{\theta+2/3}(\mathbb{R}^N)} \le C_1 \sup_{t \in [0,T]} \|g(t,\cdot)\|_{C_b^{\theta}(\mathbb{R}^N)}, \tag{14.4.57}$$

for some positive constant C_1 , independent of g. Therefore, (14.4.55) follows from (14.2.34) and (14.4.57).

To conclude the proof, let us show that v is continuously differentiable with respect to time in $(0,T] \times \mathbb{R}^N$ and $D_t v = \mathcal{A}v + g$. This can be done arguing as in the proof of Proposition 6.2.5, with minor changes. For this reason we limit ourselves to sketching the proof, pointing out the main differences.

Using (14.4.5) with $\alpha = \theta/3$ and $\beta = 2/3 + \theta/6$, we deduce that

$$||D_{ij}T(t-s+r)g(s,\cdot)||_{\infty} \le c(t-s) \sup_{t \in [0,T]} ||g(t,\cdot)||_{C_b^{\theta}(\mathbb{R}^N)}, \quad (14.4.58)$$

for any $i, j \leq r$ and some positive function $c \in L^1(0,T)$. Now, (14.4.56) (with $(\alpha,\beta)=(\theta,1)$) and (14.4.58) imply that, for any compact set $K \subset \mathbb{R}^N$, there exists a function $\hat{c}_K \in L^1(0,T)$ such that

$$\|\mathcal{A}T(t-s+r)g(s,\cdot)\|_{C(K)} \le \hat{c}_K(t-s) \sup_{t \in [0,T]} \|g(t,\cdot)\|_{C_b^{\theta}(\mathbb{R}^N)}.$$
 (14.4.59)

Hence, the same arguments as in the proof of Proposition 6.2.5 show that v is differentiable from the right, with respect to the time variable, in $(0,T) \times \mathbb{R}^N$ and $D_t^+ v = \mathcal{A}v + g$ in $(0,T) \times \mathbb{R}^N$.

To conclude that v is continuously differentiable with respect to the time variable in $[0,T]\times\mathbb{R}^N$, we just need to show that $\mathcal{A}v$ is continuous in $[0,T]\times\mathbb{R}^N$. Here, the proof differs a bit from that of Proposition 6.2.5. Hence, we go into the details. Fix $t_0,t\in[0,T]$. From Proposition A.4.4 and the estimate (14.4.4) we get

$$||v(t,\cdot,y) - v(t_0,\cdot,y)||_{C^2(\overline{B}(R))}$$

$$\leq C_R ||v(t,\cdot,y) - v(t_0,\cdot,y)||_{C(\overline{B}(R))}^{\frac{\theta}{2+\theta}} ||v(t,\cdot,y) - v(t_0,\cdot,y)||_{C_b^{2+\theta}(\overline{B}(R))}^{\frac{2}{(2+\theta)}}$$

$$\leq 2C_R' ||v(t,\cdot,y) - v(t_0,\cdot,y)||_{C(\overline{B}(R))}^{\frac{\theta}{2+\theta}},$$

for any R > 0 and any $B(R) \subset \mathbb{R}^r$. Since $v \in C_b([0,T] \times \mathbb{R}^N)$, we deduce that the functions $t \mapsto D_i v(t,x,y)$ and $t \mapsto D_{ij} v(t,x,y)$ $(i,j=1,\ldots,r)$ are

continuous in [0,T], uniformly with respect to (x,y) in bounded subsets of \mathbb{R}^N . Therefore, $D_i v, D_{ij} v \in C_b([0,T] \times \mathbb{R}^N)$. Similarly, taking the estimate (14.4.57) into account, one can show that $D_i v \in C_b([0,T] \times \mathbb{R}^N)$ for any j > r. It follows that $Av \in C([0,T] \times \mathbb{R}^N)$ and we are done.

$\begin{array}{c} {\rm Part~IV} \\ {\bf Appendices} \end{array}$

Appendix A

Basic notions of functional analysis in Banach spaces

In this appendix we collect a few basic results on linear operators, and on elementary spectral theory that we use in this book. For more details and for the proofs of the results that we present, we refer the reader mainly to [81, 147].

A.1 Bounded, compact and closed linear operators

Let X and Y be two Banach spaces. We denote by L(X,Y) the vector space of linear and bounded operators $T: X \to Y$. We endow it with the norm

$$||T||_{L(X,Y)} = \sup_{\|x \in X \atop \|x\| = 1} ||Tx||_{Y} = \sup_{x \in X \setminus \{0\}} \frac{||Tx||_{Y}}{\|x\|_{X}}.$$
 (A.1.1)

If Y = X, we write L(X) instead of L(X, X).

The norm in (A.1.1) makes L(X,Y) a Banach space.

An operator $T \in L(X,Y)$ is called *compact* if it maps the closed ball B(1) of X into a relatively compact set in Y.

Proposition A.1.1 ([147], Chapter 10, Section 2) Let X, Y, Z be three Banach spaces. Then, the following properties are met:

- (i) the set of all the compact operators from X to Y is a closed subspace of L(X,Y);
- (ii) for any compact operator $L: X \to Y$ and any bounded operator $M \in B(Y,Z)$ (resp. $M \in B(Z,X)$), the operator $M \circ L$ (resp. $L \circ M$) is a compact operator.

A very useful criterion for the convergence of a given sequence of bounded linear operators is given in the following proposition.

Proposition A.1.2 Let $\{M_n\} \in L(X)$ be a sequence of bounded operators with $\sup_{n \in \mathbb{N}} \|M_n\|_{L(X)} < +\infty$ and such that $M_n x$ converges in X, as n tends

468

to $+\infty$, for any x in a dense subspace of X. Then, M_n converges pointwise in X to a bounded linear operator M which satisfies $||M||_{L(X)} \leq C$.

We now introduce another class of linear operators which we use in this book. If D(A) is a vector subspace of X and $A:D(A)\subset X\to Y$ is a linear operator, we say that A is *closed* if its graph

$$\mathcal{G}_A = \{(x, y) \in X \times Y : x \in D(A), \ y = Ax\}$$

is a closed subset of $X \times Y$. As it is easily seen, A is closed if and only if, for any sequence $\{x_n\} \subset D(A)$ such that x_n and Ax_n converge, respectively, to some elements $x \in X$ and $y \in Y$, as n tends to $+\infty$, then $x \in D(A)$ and y = Ax.

In general, a closed operator is not continuous in $(X, \|\cdot\|)$. It turns out to be bounded if we endow D(A) with the graph norm

$$||x||_{D(A)} = ||x||_X + ||Ax||_Y, \qquad x \in D(A).$$
 (A.1.2)

Note that D(A) is a Banach space when it is endowed with the graph norm.

An operator $A: D(A) \subset X \to Y$ is said to be *closable* if there exists a (closed) operator \overline{A} whose graph coincides with the closure of \mathcal{G}_A . The operator \overline{A} is called the *closure* of A. In such a case any $x \in D(\overline{A})$ is the limit of a sequence $\{x_n\} \subset D(A)$ which is a Cauchy sequence with respect to the graph norm (A.1.2). Moreover, $\overline{A}x := \lim_{n \to +\infty} Ax_n$.

Equivalently, A is closable if, for any sequence $\{x_n\} \in D(A)$ converging to 0 in X and such that Ax_n converges to some $y \in Y$ as n tends to $+\infty$, then y = 0.

Now, let X be a Hilbert space with inner product $\langle \cdot, \cdot \rangle_X$.

For any linear operator $A:D(A)\subset X\to X$ with dense domain, the *adjoint* A^* of A is the operator $A^*:D(A^*)\subset X\to X$ defined as follows:

$$D(A^*) = \{ x \in X : \exists y \in X \text{ s.t. } \langle Az, x \rangle_X = \langle z, y \rangle_X, \quad \forall z \in D(A) \}, \quad A^*x = y.$$

The operator $A:D(A)\subset X\to X$ is said to be self-adjoint if $D(A)=D(A^*)$ and $A=A^*$.

Finally, $A:D(A)\subset X\to X$ is nonpositive if $\langle Ax,x\rangle_X\leq 0$ for any $x\in D(A)$.

A.2 Vector valued Riemann integral

In this section we define the Riemann integral for vector-valued functions. For more details and for the proof of the results that we present here, we refer the reader to [46, Chapter 2], [48, Chapter 3] and [72, Chapter 3].

Definition A.2.1 A bounded function $f:[a,b] \mapsto X$ $(-\infty < a < b < +\infty)$ is said to be integrable on [a,b] if there exists $x \in X$ with the following property: for any $\varepsilon > 0$ there exists $\delta > 0$ such that, for any partition $\mathcal{P} = \{a = t_0 < t_1 < \ldots < t_n = b\}$ of [a,b], with $\max_{i=1,\ldots,n}(t_i - t_{i-1}) < \delta$, and for any choice of the points $\xi_i \in [t_{i-1},t_i]$, we have

$$\left\|x - \sum_{i=1}^{n} f(\xi_i)(t_i - t_{i-1})\right\| < \varepsilon.$$

In this case we define

$$\int_{a}^{b} f(t)dt = x.$$

Arguing as in the real-valued case, one can easily check that the set of functions that are integrable in [a,b] is a vector space and that the integral over [a,b] is a linear operator in this vector space. In particular, if $f:[a,b]\to X$ is continuous, then it is integrable. Moreover, if f is integrable in [a,b], then the map $t\mapsto \|f(t)\|_X$ is integrable in [a,b] as well. Finally, if f is integrable in [a,b], then it is integrable in any $[c,d]\subset [a,b]$ and

$$\int_{a}^{b} f(t)dt = \int_{a}^{c} f(t)dt + \int_{c}^{d} f(t)dt,$$

for any $c \in (a, b)$.

As in the real-valued case, the definition of Riemann integral can be easily extended to the case of unbounded intervals or unbounded functions.

Definition A.2.2 Let $I \subset \mathbb{R}$ be an interval with endpoints a and b $(-\infty \le a < b \le +\infty)$ with a and b not necessarily in I. Moreover, let $f: I \to X$ be Riemann integrable in [c,d] for any a < c < d < b. We say that f admits an improper integral in I if, for any $t_0 \in I$, the limits

$$\lim_{c \to a^+} \int_c^{t_0} f(t)dt, \qquad \lim_{d \to b^-} \int_{t_0}^d f(t)dt$$

exist in X. In this case we set

$$\int_{I} f(x)dx = \lim_{c \to a^{+}} \int_{c}^{t_{0}} f(t)dt + \lim_{d \to b^{-}} \int_{t_{0}}^{d} f(t)dt.$$

Note that the previous definition is independent of t_0 .

We now recall the definition of the integral of vector-valued functions of a complex variable, along a smooth curve γ .

Definition A.2.3 Let Ω be an open subset of \mathbb{C} , $f:\Omega \to X$ be a continuous function and $\gamma:[a,b]\to\Omega$ be a piecewise C^1 -curve. The integral of f along γ is defined as follows:

$$\int_{\gamma} f(z)dz = \int_{a}^{b} f(\gamma(t))\gamma'(t)dt.$$

As in the case of vector-valued functions defined in a real interval, we can define the *improper complex integrals* in an obvious way.

Definition A.2.4 Let $\Omega \subset \mathbb{C}$ be a (possibly) unbounded open set. Moreover, let I = (a, b) be a (possibly unbounded) interval and $\gamma : I \to \mathbb{C}$ be a (piecewise) C^1 curve in Ω . We say that f admits an improper integral along γ if for any $t_0 \in (a, b)$ the limits

$$\lim_{s \to a^+} \int_s^{t_0} f(\gamma(\tau)) \gamma'(\tau) d\tau \qquad \text{and} \qquad \lim_{s \to b^-} \int_{t_0}^b f(\gamma(\tau)) \gamma'(\tau) d\tau$$

exist in X. In such a case, we set

$$\int_{\gamma} f(z)dz = \lim_{s \to a^+} \int_{s}^{t_0} f(\gamma(\tau))\gamma'(\tau)d\tau + \lim_{s \to b^-} \int_{t_0}^{s} f(\gamma(\tau))\gamma'(\tau)d\tau.$$

Note that the definition of the improper integral is independent of the choice of t_0 . Moreover, when I is bounded and the integral of f along γ exists, then f admits an improper integral along γ and the two integrals coincide.

A.3 Spectrum and resolvent

In this section, we recall the notions of resolvent and spectrum of a linear operator.

Definition A.3.1 Let $A: D(A) \subset X \to X$ be a linear operator. The resolvent set $\rho(A)$ and the spectrum $\sigma(A)$ of A are defined by

$$\rho(A) = \{ \lambda \in \mathbb{C} : \exists (\lambda I - A)^{-1} \in L(X) \}, \qquad \sigma(A) = \mathbb{C} \setminus \rho(A). \quad (A.3.1)$$

The complex numbers $\lambda \in \sigma(A)$ such that $\lambda I - A$ is not injective are called the eigenvalues of A, and the elements $x \in D(A)$ such that $x \neq 0$ and $Ax = \lambda x$ are called the eigenvectors (or eigenfunctions, when X is a function space) of A relative to the eigenvalue λ . The set $\sigma_p(A)$ whose elements are the eigenvalues of A is called the point spectrum of A.

If $\lambda \in \rho(A)$, the operator $R(\lambda, A) := (\lambda I - A)^{-1}$ is called the resolvent operator.

Let us recall some simple properties of resolvent and spectrum. First of all, since the inverse of a bounded operator is a closed operator, it is clear that, if $\rho(A) \neq \emptyset$, then A is closed.

The family of operators $\{R(\lambda, A) : \lambda \in \rho(A)\}$ satisfies the so-called *resolvent identity*

$$R(\lambda, A) - R(\mu, A) = (\mu - \lambda)R(\lambda, A)R(\mu, A), \qquad \lambda, \mu \in \rho(A).$$
 (A.3.2)

Such a formula characterizes the resolvent operators, as specified in the following proposition.

Proposition A.3.2 ([147], Section 8.4) Let $\Omega \subset \mathbb{C}$ be an open set, and let $\{F(\lambda) : \lambda \in \Omega\} \subset L(X)$ be a family of linear operators verifying the resolvent identity

$$F(\lambda) - F(\mu) = (\mu - \lambda)F(\lambda)F(\mu), \quad \lambda, \mu \in \Omega.$$

If the operator $F(\lambda_0)$ is invertible, for some $\lambda_0 \in \Omega$, then there exists a closed linear operator $A: D(A) \subset X \to X$ such that $\rho(A)$ contains Ω , and $R(\lambda, A) = F(\lambda)$ for any $\lambda \in \Omega$.

As the following proposition shows, the spectrum of an operator A is always closed in \mathbb{C} .

Proposition A.3.3 Let λ_0 be in $\rho(A)$. Then, $|\lambda - \lambda_0| < ||R(\lambda_0, A)||_{L(X)}^{-1}$ implies that λ belongs to $\rho(A)$ and the equality

$$R(\lambda, A) = R(\lambda_0, A)(I + (\lambda - \lambda_0)R(\lambda_0, A))^{-1}$$
(A.3.3)

holds. As a consequence, $\rho(A)$ is open and $\sigma(A)$ is closed.

Further properties of the resolvent operator are listed in the next proposition.

Proposition A.3.4 The function $R(\cdot, A)$ is holomorphic in $\rho(A)$. Moreover, the domain of analyticity of the function $\lambda \mapsto R(\lambda, A)$ is $\rho(A)$ and the estimate

$$||R(\lambda, A)||_{L(X)} \ge \frac{1}{\operatorname{dist}(\lambda, \sigma(A))}, \qquad \lambda \in \rho(A)$$
 (A.3.4)

holds.

As a consequence of Proposition A.3.4 we obtain the following Laurent expansion of $R(\cdot, A)$ in a neighborhood of an isolated point of $\sigma(A)$.

Proposition A.3.5 ([81], Chapter 3, Section 6.5) Suppose that $\lambda \in \mathbb{C}$ is an isolated point in $\sigma(A)$. Then, there exists r > 0 such that

$$R(\mu, A) = \sum_{n=0}^{+\infty} (\mu - \lambda)^n S^{n+1} + P + \sum_{n=1}^{+\infty} \frac{D^n}{(\mu - \lambda)^{n+1}},$$
 (A.3.5)

for any $0 < |\mu - \lambda| < r$. Here, P is the projection defined by

$$P = \frac{1}{2\pi i} \int_{\gamma} R(\xi, A) d\xi, \tag{A.3.6}$$

where γ is the boundary of the ball $\lambda + B(r)$ (oriented counterclockwise), such that $\lambda + B(r) \cap \sigma(A) = {\lambda}$. The operators D and S are defined, respectively, by

$$D = (\lambda I - A)P, \qquad S = \lim_{\xi \to \lambda} R(\xi, A)(I - P).$$

In particular, λ is the unique element of the spectrum of the restriction of A to P(X), whereas the spectrum of the restriction of A to (I-P)(X) is $\sigma(A) \setminus \{\lambda\}$. Finally, $X = P(X) \oplus (I - P)(X)$.

The next theorem gives an estimate from above of the norm of the resolvent set of a densely defined operator $A:D(A)\subset X\to X$, in terms of its numerical range

$$r(A) = \{ \langle x', Ax \rangle : x \in D(A), \|x\| = 1, x' \in X', \|x'\| = 1, \langle x', x \rangle = 1 \}.$$

Theorem A.3.6 ([123], Chapter 1, Theorem 3.9) Let $A:D(A)\subset X$ $\to X$ be a densely defined closed operator in X. If $\lambda \in \mathbb{C} \setminus r(A)$, then the operator $\lambda I - A$ is one to one and has closed range. Moreover, if Σ is a component of $\mathbb{C} \setminus r(A)$ which intersects $\rho(A)$, then the spectrum of A is contained $in \mathbb{C} \setminus \Sigma \ and$

$$||R(\lambda, A)||_{L(X)} \le \frac{1}{\operatorname{dist}(\lambda, \overline{r(A)})}, \qquad \lambda \in \mathbb{C} \setminus \Sigma.$$

In the case when $R(\lambda, A)$ is a compact operator for some (and, hence, for any) $\lambda \in \rho(A)$, then $\sigma(A)$ has a very easy structure, as the following proposition shows.

Proposition A.3.7 Let $A: X \to X$ be a closed operator with $\sigma(A) \neq \emptyset$ and compact resolvent operator. Then, the following properties are met:

- (i) $\sigma(A)$ consists of an at most countable set of points. Any $\lambda \in \sigma(A)$ is an eigenvalue of A and the corresponding eigenspace has finite dimension;
- (ii) for any $\lambda \in \sigma(A)$, the vector space of the generalized eigenvectors of A (i.e., the vector space of all $x \in X$ such that $(\lambda I - A)^k x = 0$ for some $k \in \mathbb{N}$) has finite dimension;
- (iii) any $\lambda \in \sigma(A)$, is a pole of the function $R(\cdot, A)$ whose order coincides with the dimension of the vector space of all the generalized eigenvectors associated with the eigenvalue λ . Finally, the order of the pole coincides also with the index $i_{\lambda}(A)$ of the eigenvalue λ (i.e., with the smallest integer k such that $Ker((\lambda I - A)^k) = Ker((\lambda I - A)^{k+1})$.

Proof. The proof can be obtained combining the results in [81, Chapter 3, Sections 6.4, 6.5 and 6.8]. For the reader's convenience we give some details of how this can be done.

Since, for any $\xi \in \rho(A)$, $R(\xi, A)$ is a compact operator then, according to Proposition A.1.1, the projection P in (A.3.6) is a compact operator as well. Hence, P(X) is a finite dimensional subspace of X (see [81, Chapter 3, Section 4]).

Now, the arguments in [81, Chapter 3, Section 6.5] show that any $\lambda \in \sigma(A)$ is a pole of the resolvent operator, with finite multiplicity that coincides with the dimension m of P(X). According to the decomposition theorem (see [81, Chapter 3, Theorem 6.17]), λ is the unique element in the spectrum of the restriction of A to P(X), and it belongs to the resolvent set of the restriction of A to P(X). It follows that P(X) is the space of generalized eigenvectors of A corresponding to λ . Now, the equality $i_{\lambda}(A) = m$ follows observing that $i_{\lambda}(A) = i_{\lambda}(\tilde{A})$, since, according to a well-known result of linear algebra, $i_{\lambda}(\tilde{A}) = m$.

A.4 Some results from interpolation theory

In this section we recall some results from the interpolation theory. For the proof of the results that we present here, if not otherwise specified, we refer the reader to [104] and [141].

Let us introduce the interpolation spaces $(X,Y)_{\theta,\infty}$.

Definition A.4.1 Let X, Y be two Banach spaces with Y continuously embedded in X. Moreover, let $K:(0,+\infty)\times X\to\mathbb{R}$ be the function defined by

$$K(\xi, x) = \inf_{x=a+b} (\|a\|_X + \xi \|b\|_Y), \qquad \xi > 0, \quad x \in X.$$

For any $\theta \in [0,1]$, the interpolation space $(X,Y)_{\theta,\infty}$ is defined by

$$(X,Y)_{\theta,\infty} = \{x \in X : \sup_{0 < \xi < 1} \xi^{-\theta} K(\xi,x) < +\infty\},$$

and it is normed by

$$||x||_{\theta,\infty} := \sup_{0 < \xi < 1} \xi^{-\theta} K(\xi, x), \quad x \in (X, Y)_{\theta,\infty}.$$
 (A.4.1)

For any $\theta \in [0,1]$, $(X,Y)_{\theta,\infty}$ is a Banach space when it is endowed with the norm in (A.4.1) and

$$Y \subset (X,Y)_{\theta,\infty} \subset X$$
 (A.4.2)

with continuous embeddings.

The topological inclusions in (A.4.2) and the following proposition show the reason why $(X,Y)_{\theta,\infty}$ are called interpolation spaces.

Proposition A.4.2 Let X_1 , X_2 , Y_1 , Y_2 be four Banach spaces with Y_j continuously embedded in X_j for j = 1, 2. Moreover, let $T \in L(X_1, X_2) \cap L(Y_1, Y_2)$. Then, for any $\theta \in (0, 1)$, $T \in L((X_1, Y_1)_{\theta, \infty}, (X_2, Y_2)_{\theta, \infty})$. Moreover,

$$||T||_{L((X_1,Y_1)_{\theta,\infty},(X_2,Y_2)_{\theta,\infty})} \le ||T||_{L(X_1,X_2)}^{1-\theta} ||T||_{L(Y_1,Y_2)}^{\theta}.$$
(A.4.3)

We now introduce the definition of spaces of class J_{θ} and K_{θ} ($\theta \in [0,1]$) which play a central role in the interpolation theory.

Definition A.4.3 Let X, Y, Z be a triplet of Banach spaces such that $Y \subset Z \subset X$ with continuous embeddings. Moreover, let $\theta \in [0,1]$. We say that Z belongs to the class J_{θ} between X and Y (in short $Z \in J_{\theta}(X,Y)$) if there exists a positive constant $C = C(\theta)$ such that

$$\|x\|_Z \leq C \|x\|_X^{1-\theta} \|x\|_Y^\theta, \qquad x \in Z.$$

Similarly, we say that Z belongs to the class K_{θ} between X and Y (in short $Z \in K_{\theta}(X,Y)$) if $Z \subset (X,Y)_{\theta,\infty}$ with a continuous embedding.

Abstract and concrete examples of spaces of class J_{θ} are given in the following two propositions.

Proposition A.4.4 Let $0 < \theta < \alpha$ and let $\Omega \subset \mathbb{R}^N$ $(N \ge 1)$ be an open set with boundary uniformly of class C^{α} (possibly $\Omega = \mathbb{R}^N$). Then, $C_b^{\theta}(\overline{\Omega})$ belongs to the class $J_{\theta/\alpha}$ between $C_b(\overline{\Omega})$ and $C_b^{\alpha}(\overline{\Omega})$.

Proposition A.4.5 For any $\theta \in [0,1]$ and any pair of Banach spaces X and Y, with Y continuously embedded in X, $(X,Y)_{\theta,\infty} \in J_{\theta}(X,Y) \cap K_{\theta}(X,Y)$.

The next proposition show some interesting properties of the spaces of class J_{α} .

Proposition A.4.6 Let $Y \subset Z \subset X$ be a triplet of Banach spaces such that $Z \in J_{\alpha}(X,Y)$ for some $\alpha \in (0,1)$. Further, let $I \subset \mathbb{R}$ be an interval. Then, the following properties are met:

(i) if $u \in B(I;Y) \cap C_b^{\theta}(I;X)$ for some $\theta \in [0,1)$, then $u \in C_b^{\theta(1-\alpha)}(I;Z)$. Moreover, there exists a positive constant C such that

$$\|u\|_{C_b^{\theta(1-\alpha)}(I;Z)} \leq C \|u\|_{C_b^{\theta}(I;X)}^{1-\alpha} \|u\|_{B(I;Y)}^{\alpha};$$

(ii) if $u \in B(I;Y) \cap \text{Lip}(I;X)$, then $u \in C_b^{1-\alpha}(I;Z)$. Moreover, there exists a positive constant C such that

$$||u||_{C_b^{1-\alpha}(I;Z)} \le C||u||_{\mathrm{Lip}(I;X)}^{1-\alpha}||u||_{B(I;Y)}^{\alpha}.$$

We now state one of the most useful tools from interpolation theory: the so-called *reiteration theorem*.

Theorem A.4.7 Let $0 \le \theta_1 < \theta_2 \le 1$. Suppose that $Z_j \in K_{\theta_j}(X,Y)$ (j = 1, 2). Then, for any $\theta \in (0, 1)$

$$(Z_1, Z_2)_{\theta,\infty} \subset (X, Y)_{(1-\theta)\theta_1 + \theta\theta_2,\infty},$$

with a continuous embedding. Similarly, if $Z_j \in J_{\theta_j}(X,Y)$ (j=1,2), then, for any $\theta \in (0,1)$

$$(X,Y)_{(1-\theta)\theta_1+\theta\theta_2,\infty}\subset (Z_1,Z_2)_{\theta,\infty},$$

with a continuous embedding. In particular, if $Z_j \in J_{\theta_j}(X,Y) \cap K_{\theta_j}(X,Y)$ (j=1,2), then

$$(X,Y)_{(1-\theta)\theta_1+\theta\theta_2,\infty}=(Z_1,Z_2)_{\theta,\infty},$$

with equivalence of the corresponding norms.

Using the interpolation theorem some spaces of Hölder continuous functions can be characterized as interpolation spaces.

Theorem A.4.8 For any $0 \le \alpha < \beta$ and any $\theta \in (0,1)$ it holds that

$$(C_b^{\alpha}(\mathbb{R}^N), C_b^{\beta}(\mathbb{R}^N))_{\theta, \infty} = C_b^{\alpha + \theta(\beta - \alpha)}(\mathbb{R}^N), \tag{A.4.4}$$

with equivalence of the corresponding norms. Moreover,

$$(BUC(\mathbb{R}^N), C_b^{\beta}(\mathbb{R}^N))_{\theta,\infty} = C_b^{\theta\beta}(\mathbb{R}^N). \tag{A.4.5}$$

Proof. For the proof of (A.4.4) see [104, Theorem 1.2.17 & Corollary 1.2.18]. The last part of the theorem follows from the previous one through the reiteration theorem, recalling that $BUC(\mathbb{R}^N)$ belongs to $J_0(C_b(\mathbb{R}^N), C_b^{\beta}(\mathbb{R}^N)) \cap K_0(C_b(\mathbb{R}^N), C_b^{\beta}(\mathbb{R}^N))$.

To conclude this section we state the Riesz-Thorin interpolation theorem. For the proof we refer the reader to [133, Theorem IX.17].

Theorem A.4.9 (Riesz-Thorin interpolation theorem) Let (Ω_1, μ_1) and (Ω_2, μ_2) be two σ -finite spaces. Moreover, let T be a linear operator such that

$$T \in L(L^{p_0}(\Omega_1, \mu_1), L^{q_0}(\Omega_2, \mu_2)) \cap L(L^{p_1}(\Omega_1, \mu_1), L^{q_1}(\Omega_2, \mu_2)),$$

476 Appendix A. Basic notions of Functional Analysis in Banach spaces

for some $p_0, p_1, q_0, q_1 \in [1, +\infty]$. Then,

$$T \in L(L^{p_{\theta}}(\Omega_1, \mu_1), L^{q_{\theta}}(\Omega_2, \mu_2)), \qquad \theta \in (0, 1),$$

where

$$\frac{1}{p_{\theta}} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1}, \qquad \frac{1}{q_{\theta}} = \frac{1-\theta}{q_0} + \frac{\theta}{q_1}.$$

Moreover,

$$\begin{split} & \|T\|_{L(L^{p_{\theta}}(\Omega_{1},\mu_{1}),L^{q_{\theta}}(\Omega_{2},\mu_{2}))} \\ & \leq \|T\|_{L(L^{p_{0}}(\Omega_{1},\mu_{1}),L^{q_{0}}(\Omega_{2},\mu_{2}))}^{1-\theta} \|T\|_{L(L^{p_{1}}(\Omega_{1},\mu_{1}),L^{q_{1}}(\Omega_{2},\mu_{2}))}^{\theta}. \end{split}$$

$Appendix \,\, B$

An overview on strongly continuous and analytic semigroups

In this appendix we recall the definitions and the main properties of strongly continuous and analytic semigroups that we use in this book. If not otherwise specified, we refer the reader to [51, 72, 104, 123] for the proofs of the results of this appendix.

To begin with we give the definition of semigroup of bounded linear operators. Throughout this appendix, if not otherwise specified, X will denote a Banach space.

Definition B.0.1 A one-parameter family $\{T(t)\}$ of linear bounded operators in X is said to be a semigroup of bounded linear operators if

$$T(t+s) = T(t) \circ T(s), \qquad s, t \ge 0.$$
 (B.0.1)

We simply write T(t)T(s) for $T(t)\circ T(s)$ when there is no damage of confusion.

B.1 Strongly continuous semigroups

We begin this section with the definition of the strongly continuous semigroups.

Definition B.1.1 A semigroup $\{T(t)\}$ of bounded linear operators, defined in X, is strongly continuous if, for any $x \in X$, T(t)x tends to x as t tends to 0 from the right.

Using the semigroup rule (B.0.1), one can see that, if $\{T(t)\}$ is a strongly continuous semigroup, then the function $t \mapsto T(t)x$ is continuous in $[0, +\infty)$ for any $x \in X$. Moreover, it is also possible to show that there exist two constants $M \geq 1$ and $\omega \in \mathbb{R}$ such that

$$||T(t)||_{L(X)} \le Me^{\omega t}, \qquad t > 0.$$
 (B.1.1)

When $\omega = 0$ and $M_{\omega} = 1$, we say that $\{T(t)\}$ is a strongly continuous semigroup of contractions in X.

For any strongly continuous semigroup we define its exponential growth bound as the infimum of the set of all the $\omega \in \mathbb{R}$ such that (B.1.1) holds for some $M \geq 1$. It is possible to show that the exponential growth bound of $\{T(t)\}$ coincides with the spectral bound s(A) of its infinitesimal generator, where

$$s(A) = \sup \{ \text{Re } \lambda : \lambda \in \sigma(A) \}.$$

See, e.g., [8, Theorem 5.3.1]).

Remark B.1.2 If $\{T(t)\}$ is a strongly continuous semigroup, then the mapping $(t,x)\mapsto T(t)x$ is locally uniformly continuous in $[0,+\infty)\times X$. See, e.g., [51, Chapter 1, Lemma 5.2]. As a consequence, for any compact set $K\subset X$ and any T>0, the mapping $t\mapsto T(t)x$ is continuous in [0,T], uniformly with respect to $x\in K$.

We can now give the definition of the infinitesimal generator of a strongly continuous semigroup.

Definition B.1.3 Let $\{T(t)\}$ be a strongly continuous semigroup in X. The infinitesimal generator of $\{T(t)\}$ is the operator $A: D(A) \subset X \to X$ defined by

$$\begin{cases} D(A) = \left\{ x \in X : \exists g \in X \text{ s.t. } g = \lim_{t \to 0^+} \frac{T(t)x - x}{t} \right\}, \\ Ax = \lim_{t \to 0^+} \frac{T(t)x - x}{t}. \end{cases}$$
(B.1.2)

It is easy to see that the function $t \mapsto T(t)x$ is continuously Fréchet differentiable in $[0, +\infty)$ with values in X if and only if $x \in D(A)$. In this case $D_t T(t)x = AT(t)x = T(t)Ax$ for any $t \ge 0$.

As the following proposition shows, the infinitesimal generator characterizes uniquely the strongly continuous semigroups.

Proposition B.1.4 Let $\{S(t)\}$ and $\{T(t)\}$ be two strongly continuous semi-groups in X and denote by A and B their infinitesimal generators. If A=B, then $S(t) \equiv T(t)$ for any t>0.

The Hille-Yosida theorem is a keystone in the theory of strongly continuous semigroups since it allows us to give a complete characterization of the infinitesimal generators of strongly continuous semigroups.

Theorem B.1.5 (Hille-Yosida) A linear operator $A: D(A) \subset X \to X$ is the infinitesimal generator of a strongly continuous semigroup $\{T(t)\}$ satisfying (B.1.1) for some $M \geq 1$ and some $\omega \in \mathbb{R}$, if and only if

(i) A is closed and its domain is dense in X;

(ii) the resolvent set $\rho(A)$ contains the half line $(\omega, +\infty)$ and, for any $\lambda > \omega$ and any $n \in \mathbb{N}$,

$$\|R(\lambda,A)\|_{L(X)} \leq \frac{M}{(\lambda-\omega)^n}.$$

The proof of the Hille-Yosida theorem shows a connection between the resolvent operator and the semigroup. In fact, the resolvent operator is the Laplace transform of the semigroup, namely

$$R(\lambda, A)x = \int_0^{+\infty} e^{-\lambda t} T(t)x \, dt, \qquad \text{Re } \lambda > \omega, \quad x \in X.$$
 (B.1.3)

Hence, many properties of $R(\lambda, A)$ can be deduced from the corresponding properties of the semigroup.

Another important result in the theory of strongly continuous semigroups is the Lumer-Phillips theorem. It provides an alternative characterization of the infinitesimal generators of strongly continuous semigroups of contractions.

Definition B.1.6 Let $A:D(A)\subset X\to X$ be a linear operator. A is dissipative if

$$\|\lambda x - Ax\|_X \ge \lambda \|x\|_X, \qquad \lambda > 0, \quad x \in D(A).$$

Theorem B.1.7 (Lumer-Phillips) Let $A:D(A)\subset X\to X$ be a linear operator with dense domain. Then, the following properties are met:

- (i) if A is dissipative and there exists $\lambda_0 > 0$ such that the range of the operator $\lambda_0 I A$ is X, then A is the infinitesimal generator of a strongly continuous semigroup of contractions;
- (ii) if A is the infinitesimal generator of a strongly continuous semigroup of contractions in X, then the range of the operator $\lambda I A$ is X for any $\lambda > 0$. Moreover, A is dissipative.

A straightforward consequence of the Lumer-Phillips theorem is the following proposition.

Proposition B.1.8 *Let* $A : D(A) \subset X \to X$ *be a dissipative operator. Then, the following properties are met:*

- (i) if A is closable, then the closure \overline{A} is also dissipative;
- (ii) if D(A) is dense in X and $(\lambda I A)(D(A))$ is dense in X for some (and hence all $\lambda > 0$), then the closure \overline{A} is the infinitesimal generator of a strongly continuous semigroup of contractions.

To conclude this section, we recall the definition of a core of a linear operator A and we provide a useful sufficient condition guaranteeing that a set E is a core of A, when A is the infinitesimal generator of a strongly continuous semigroup. Finally, we state the famous Trotter-Kato theorem.

Definition B.1.9 Let A be a linear operator in a Banach space X with domain D(A). A subspace E of D(A) is a core of A if E is dense in D(A) in the graph norm (see (A.1.2)).

Proposition B.1.10 Let $A: D(A) \subset X \to X$ be the generator of a strongly continuous semigroup of contractions $\{S(t)\}$ in X. Moreover, let E be a subspace of D(A) dense in X such that $S(t)(E) \subset E$ for any t > 0. Then, E is a core of A.

Theorem B.1.11 (Trotter-Kato) Let $\{T(t)\}$, $\{T_n(t)\}$ $(n \in \mathbb{N})$ be strongly continuous semigroups of contractions in X, with infinitesimal generators A: $D(A) \subset X \to X$ and $A_n : D(A_n) \subset X \to X$, respectively. Further, let D be a core of the operator A such that $D \subset D(A_n)$ for any $n \in \mathbb{N}$. If, for any $x \in D$, A_nx tends to Ax in X, as n tends to $+\infty$, then $R(\lambda, A_n)x$ tends to $R(\lambda, A)x$, as n tends to $+\infty$, for any $k \in \mathbb{C}$ with positive real part.

B.1.1 On the closure of the sum of generators of strongly continuous semigroups

Here we briefly recall some results on the sum of closed operators that we use in Chapter 9. Throughout the subsection we make the following hypotheses.

Hypotheses B.1.12 (i) $A: D(A) \subset X \to X$ is a closed operator such that -A is the infinitesimal generator of a strongly continuous semigroup of contractions in X;

(ii)
$$0 \in \rho(A)$$
.

The analyticity of the function $\lambda R(\lambda, A)$ in $\rho(A)$ implies that $\rho(A)$ contains a sector $\Sigma_{\varphi} := \{\lambda \in \mathbb{C} \setminus \{0\} : |\arg \lambda| < \varphi\}$ for some $\varphi \in [0, \pi)$ and that there exists a positive constant C_{φ} such that $\|\lambda R(\lambda, -A)\|_{L(X)} \leq C_{\varphi}$ for any $\lambda \in \Sigma_{\varphi}$.

Therefore, one can introduce the spectral angle φ_A of A, defined by

$$\varphi_A = \inf \{ \varphi > 0 : \Sigma_{\pi - \varphi} \subset \rho(-A) \text{ and } C_{\varphi} < +\infty \}.$$

Moreover, for $s \in \mathbb{R} \setminus \{0\}$, one can define the operator A^{is} through the formula

$$A^{is}x = -\frac{\sin(i\pi s)}{\pi} \left\{ i\frac{x}{s} + \frac{1}{1+is}A^{-1}x + \int_0^1 \lambda^{is+1}R(-\lambda, A)A^{-1}x \, d\lambda + \int_1^{+\infty} \lambda^{is-1}R(-\lambda, A)Ax \, d\lambda \right\}, \qquad x \in D(A)$$

(see [86, 87]). A^{is} is a closed and densely defined operator for any $s \neq 0$ but, in general, it is not bounded in \mathbb{R} .

If the operators A^{is} are bounded for any $s \neq 0$ and the map $s \mapsto A^{is}$ $(A^0 := I)$, is locally bounded in \mathbb{R} , we say that A admits bounded imaginary powers or simply that $A \in \mathcal{BIP}(X)$.

If $A \in \mathcal{BIP}(X)$, then the limit

$$\theta_A := \lim_{|s| \to +\infty} |s|^{-1} \log (\|A^{is}\|_{L(X)})$$
(B.1.4)

is well defined and it is called the *power angle* of A.

We summarize in the following proposition a few properties of the class $\mathcal{BIP}(X)$. For a proof, we refer the reader to [34, Theorem 5.8], [35], [131] and [132, Theorem 3].

Proposition B.1.13 The following properties are met:

- (i) if $A \in \mathcal{BIP}(X)$, then $\omega I + A \in \mathcal{BIP}(X)$ for any $\omega > 0$ and $\theta_{\omega I + A} = \theta_A$;
- (ii) if $A \in \mathcal{BIP}(X)$, then $\theta_A \geq \varphi_A$;
- (iii) if $A: D(A) \subset L^p(\mathbb{R}^N, \mu) \to L^p(\mathbb{R}^N, \mu)$ (μ being a σ -finite measure, and $p \in (1, +\infty)$) is the generator of a strongly continuous semigroup of contractions $\{T(t)\}$ such that $T(t)f \geq 0$ for any $f \geq 0$, then, for any $\omega > 0$, $\omega I A \in \mathcal{BIP}(L^p(\mathbb{R}^N, \mu))$ and $\theta_{\omega I A} \leq \pi/2$;
- (iv) if A is the realization in $L^p(\mathbb{R}^N, dx)$ $(p \in (1, +\infty))$ of the uniformly elliptic operator $\mathcal{A}u = \sum_{i,j=1}^N q_{ij}D_{ij}u$ with constant coefficients, then $\theta_{\omega I-A} = 0$ for any $\omega > 0$;
- (v) if X is a Hilbert space and $A: D(A) \subset X \to X$ is a self-adjoint operator such that -A generates a strongly continuous semigroup of contractions in X, then $A \in \mathcal{BIP}(X)$ and $\theta_A = 0$.

The property (iii) in Proposition B.1.13 is known as the *transference principle*. The property (iv) holds for a more general class of elliptic operators in divergence form than those we considered here. We refer the reader to [132, Theorem C] for more details.

We now introduce a class of Banach spaces, the so-called ξ -convex spaces.

Definition B.1.14 A Banach space X is said to be ξ -convex if there exists a function $\xi: X \times X \to \mathbb{R}$ such that

- (i) ξ is convex with respect to both the variables;
- (ii) $\xi(0,0) > 0$;
- (iii) $\xi(x+y) \le |x+y|$ for any $x, y \in \partial B(1)$.

Examples of ξ -convex Banach spaces are the L^p -spaces $(p \in [1, +\infty))$ relevant to σ -finite measures in \mathbb{R}^N . In particular $L^p(\mathbb{R}^N)$ are all ξ -convex Banach spaces.

For further results on the operators which admit bounded imaginary powers and on ξ -convex Banach spaces, we refer the reader to [141, Chapter 1, Section 15], [47, Appendix], the survey paper [23] and the monograph [130].

To conclude this subsection we state two theorems which guarantee the closedness of the sum of n sectorial operators which admits bounded imaginary powers. The first one deals with the sum of n operators A_i commuting in the resolvent sense (i.e., $R(\lambda, A_i)R(\mu, A_j) = R(\mu, A_j)R(\lambda, A_i)$ for any $\lambda \in \rho(A_i)$, any $\mu \in \rho(A_j)$ and any i, j = 1, ..., N). It extends a well known result by Dore and Venni (see [47]).

Theorem B.1.15 ([132], Corollary 4) Let X be a ξ -convex space and let $A_i: D(A_i) \subset X \to X$ $(i=1,\ldots,N)$ belong to $\mathcal{BIP}(X)$ and commute in the resolvent sense. Further, suppose that the power angles θ_i satisfy $\theta_i + \theta_j < \pi$ for any $i, j = 1, \ldots, n$, $i \neq j$, and there is only one index j such that $\theta_j = \max_{i=1,\ldots,n} \theta_i$. Then, the operator $A = \sum_{i=1}^n A_i$ with domain $D(A) = \bigcap_{i=1}^n D(A_i)$ is closed, it belongs to $\mathcal{BIP}(X)$ and there exists a positive constant C such that

$$\sum_{i=1}^{n} ||A_i x||_X \le C ||Ax||_X, \qquad x \in D(A).$$
 (B.1.5)

The second theorem deals with the case of two operators which do not commute in resolvent sense. In such a case the noncommutativity of the resolvent operators of A and B is balanced by a suitable assumption on their commutators.

Theorem B.1.16 ([120], Corollary 2) Let X be a ξ -convex Banach space and let $A, B \in \mathcal{BIP}(X)$. Further suppose that the power angles θ_A and θ_B satisfy $\theta_A + \theta_B < \pi$ and that there exist C > 0, α, β satisfying $0 \le \alpha < \beta \le 1$ and φ_A, φ_B with $\varphi_A > \theta_A, \varphi_B > \theta_B$ and $\varphi_A + \varphi_B < \pi$ such that

$$||AR(\lambda, -A)[A^{-1}, R(\mu, -B)]||_{L(X)} \le \frac{C}{(1+|\lambda|^{1-\alpha})|\mu|^{1+\beta}},$$
 (B.1.6)

for any $\lambda, \mu \in \mathbb{C} \setminus \{0\}$ such that $|\arg \lambda| < \pi - \varphi_A$ and $|\arg \mu| < \pi - \varphi_B$. Then, the operator A + B with domain $D(A + B) = D(A) \cap D(B)$ is closed in X.

B.2 Analytic semigroups

In this section we introduce the analytic semigroups and we describe their main properties.

Definition B.2.1 A linear operator $A: D(A) \subset X \to X$ is said to be sectorial if there exist three constants $\omega \in \mathbb{R}$, $\theta \in (\pi/2, \pi)$, M > 0 such that

$$\begin{cases}
(i) & \rho(A) \supset S_{\theta,\omega} = \{\lambda \in \mathbb{C} : \lambda \neq \omega, |\arg(\lambda - \omega)| < \theta\}, \\
(ii) & \|R(\lambda, A)\|_{L(X)} \le \frac{M}{|\lambda - \omega|}, \qquad \lambda \in S_{\theta,\omega}.
\end{cases}$$
(B.2.1)

For any t > 0, the conditions (B.2.1) allow us to define a bounded linear operator T(t) on X as follows. For any r > 0 and any $\eta \in (\pi/2, \theta]$, let $\gamma_{r,\eta}$ be the curve

$$\{\lambda \in \mathbb{C} : |\arg \lambda| = \eta, \ |\lambda| \ge r\} \cup \{\lambda \in \mathbb{C} : |\arg \lambda| \le \eta, \ |\lambda| = r\},$$

oriented counterclockwise. We set

$$T(t) = \frac{1}{2\pi i} \int_{\gamma_{r,\eta} + \omega} e^{t\lambda} R(\lambda, A) d\lambda, \qquad t > 0, \qquad T(0) = I. \quad (B.2.2)$$

It is easy to check that the integral in (B.2.2) is well defined and it is independent of r > 0 and $\eta \in (\pi/2, \theta)$.

In the following theorem we summarize the main properties of T(t) for t>0.

Theorem B.2.2 Let A be a sectorial operator and let T(t) be given by (B.2.2), for any t > 0. Then, the following properties are met:

- (i) the family of bounded operators $\{T(t)\}$ is a semigroup;
- (ii) $T(t)x \in D(A^k)$ for any t > 0, any $x \in X$ and any $k \in \mathbb{N}$. Moreover, if $x \in D(A^k)$, then

$$A^k T(t)x = T(t)A^k x, \qquad t \ge 0;$$

(iii) there exists a constant M > 0 such that

$$||T(t)||_{L(X)} \le Me^{\omega t}, \qquad t > 0,$$
 (B.2.3)

where ω is the number in (B.2.1). Moreover, for any $\varepsilon > 0$ and any $k \in \mathbb{N}$, there exists a positive constant $C_{k,\varepsilon}$ such that

$$||t^k A^k T(t)||_{L(X)} \le C_{k,\varepsilon} e^{(\omega + \varepsilon)t}, \qquad t > 0;$$
 (B.2.4)

(iv) the function $t \mapsto T(t)$ is analytic in $(0+\infty)$ with values in L(X) and

$$\frac{d^k}{dt^k}T(t) = A^kT(t), \qquad t > 0, \ k \in \mathbb{N}.$$
 (B.2.5)

We can now give the following definition.

Definition B.2.3 Let A be a sectorial operator. The semigroup $\{T(t)\}$ defined by (B.2.2) is called the analytic semigroup generated by A (in X).

The property (iv) in Theorem B.2.2 implies that $\{T(t)\}$ is norm continuous in $(0, +\infty)$ according to the following definition.

Definition B.2.4 A semigroup of bounded linear operators $\{T(t)\}$ in X is said to be norm continuous in the interval $I \subset [0, +\infty)$ if the map $t \mapsto T(t)$ is continuous in I with respect to the topology of L(X).

The norm continuity of the semigroup allows us to write

$$R(\lambda, A) = \int_0^{+\infty} e^{-\lambda t} T(t) dt,$$
 (B.2.6)

for any $\lambda \in \mathbb{C}$, with $\text{Re}\lambda > \omega$, where the integral converges in the topology of L(X).

In the following proposition we exploit the behaviour of the function $t \mapsto T(t)x$ at t = 0.

Proposition B.2.5 The following properties are met:

- (i) if $x \in \overline{D(A)}$, then $\lim_{t \to 0^+} T(t)x = x$. Conversely, if $y = \lim_{t \to 0^+} T(t)x$ exists, then $x \in \overline{D(A)}$ and y = x;
- (ii) if $x \in D(A)$ and $Ax \in \overline{D(A)}$, then $\lim_{t\to 0^+} (T(t)x x)/t = Ax$. Conversely, if $z := \lim_{t\to 0^+} (T(t)x x)/t$ exists, then $x \in D(A)$ and $Ax = z \in \overline{D(A)}$;
- (iii) if $x \in D(A)$ and $Ax \in \overline{D(A)}$, then $\lim_{t\to 0^+} AT(t)x = Ax$.

Remark B.2.6 Let $x \in X$ and denote by $u: [0, +\infty) \to X$ the function defined by u(t) = T(t)x for any t > 0 and u(0) = x. Combining the results in Theorem B.2.2 and Proposition B.2.5, we easily deduce that $u \in C([0, +\infty); X)$ if and only if $x \in \overline{D(A)}$. Similarly, $u \in C([0, +\infty); D(A))$ (i.e., both u and Au belong to $C([0, +\infty); X)$) if and only if $x \in D(A)$ and $Ax \in \overline{D(A)}$. Since, according to Theorem B.2.2, u is differentiable in $(0, +\infty)$ and u'(t) = Au(t) for any t > 0, then $u \in C^1([0, +\infty); X)$ if and only if $x \in D(A)$ and $Ax \in \overline{D(A)}$. Moreover, $u \in C^1([0, +\infty); D(A))$ if and only if $x \in D(A^2)$ and $A^2x \in \overline{D(A)}$.

As a straightforward consequence of Proposition B.2.5(i), we deduce that an analytic semigroup $\{T(t)\}$ fails to be strongly continuous in X unless D(A) is dense in X.

Useful criteria to establish if an operator $A:D(A)\subset X\to X$ is sectorial are given by the following three theorems.

Theorem B.2.7 Let $A: D(A) \subset X \to X$ be a linear operator such that $\rho(A)$ contains the halfplane $\{\lambda \in \mathbb{C} : \text{Re } \lambda \geq K\}$ for some $K \geq 0$, and

$$\|\lambda R(\lambda, A)\|_{L(X)} \le M, \qquad \text{Re } \lambda \ge K,$$
 (B.2.7)

for some $M \geq 1$. Then, A is sectorial.

Theorem B.2.8 Let $\{T(t)\}$ be a strongly continuous semigroup in X such that the function $t \mapsto T(t)x$ is differentiable in $(0, +\infty)$ with values in X for any $x \in X$ and there exists a positive constant M such that $t \|T'(t)\|_{L(X)} \leq M$ for any $t \in (0, 1)$. Then, $\{T(t)\}$ is analytic.

Theorem B.2.9 Let X be a Hilbert space and $A: D(A) \subset X \to X$ be a self-adjoint dissipative operator. Then, A is sectorial in X and it satisfies the conditions (B.2.1) with $\omega = 0$ and arbitrary $\theta < \pi$.

Next, we recall the following classical perturbation result for analytic semigroups.

Theorem B.2.10 If $A:D(A)\subset X\to X$ is sectorial and $B:D(B)\subset X\to X$ is a linear operator such that

- (i) $D(A) \subset D(B)$;
- (ii) there exist $\theta \in (0,1)$ and C > 0 such that

$$\|Bx\| \le C \|x\|_{D(A)}^{\theta} \|x\|_X^{1-\theta}, \qquad x \in D(A).$$

Then, $A + B : D(A + B) := D(A) \to X$ is sectorial. Moreover, if A satisfies (B.2.1) with $\theta > 3\pi/4$, then the operator $-A^2$ is sectorial as well.

We now introduce the interpolation spaces $D_A(\alpha, \infty)$ ($\alpha \in (0,1)$), which play a crucial role in the theory of analytic semigroups.

Definition B.2.11 Let $\alpha \in (0,1)$. We set $D_A(\alpha,\infty) = (X,D(A))_{\alpha,\infty}$ (see Section A.4) and we call it the interpolation space of order α .

According to Proposition A.4.2, and Theorem B.2.2, one can easily see that, if $x \in D_A(\alpha, \infty)$, then the function $t \mapsto g_x(t) := t^{-\alpha}(T(t)x - x)$ is bounded in a right neighborhood of t = 0. Actually, as the following proposition shows, the function g_x is bounded near t = 0 if and only if $x \in D_A(\alpha, \infty)$.

Proposition B.2.12 The interpolation space $D_A(\alpha, \infty)$ is the set of all the $x \in X$ such that

$$[[x]]_{D_A(\alpha,\infty)} := \sup_{0 < t \le 1} t^{-\alpha} ||T(t)x - x||_X < +\infty.$$

Moreover, the norm

$$x \mapsto ||x||_X + [[x]]_{D_A(\alpha,\infty)}$$

is equivalent to the norm $\|\cdot\|_{(X,D(A))_{\alpha,\infty}}$ defined in (A.4.1). Equivalently, $D_A(\alpha,\infty)$ can be defined as the set of all the $x\in X$ such that

$$[[[x]]]_{D_A(\alpha,\infty)}:=\sup_{0<\xi<1}\|\xi^{1-\theta}AT(\xi)x\|_X<+\infty.$$

Moreover, the norm

$$x \mapsto ||x||_X + [[[x]]]_{D_A(\alpha,\infty)}$$

is equivalent to the norm $\|\cdot\|_{(X,D(A))_{\alpha,\infty}}$.

As a consequence of Theorem B.2.2 and Proposition B.2.12, it follows that the function $u = T(\cdot)x$ belongs to $C^{\alpha}([0,T];X)$ for some (and hence any) T>0 if and only if $x\in D_A(\alpha,\infty)$. Similarly, $u\in C^{\alpha}([0,T];D(A))$ (or, equivalently, $u \in C^{1+\alpha}([0,T];X)$ if and only if $x \in D_A(1+\alpha,\infty)$, where

$$D_A(1+\alpha,\infty) := \{x \in D(A) : Ax \in D_A(\alpha,\infty)\}.$$

To conclude this section, let us consider the abstract Cauchy problem

$$\begin{cases} u'(t) = Au(t) + f(t), & t \in (0, T], \\ u(0) = x, \end{cases}$$
 (B.2.8)

where $f:[0,T]\to X$ is a given continuous function and $x\in X$.

Definition B.2.13 We say that $u:[0,T] \to X$ is a strict solution to (B.2.8) if $u \in C^1([0,T];X) \cap C([0,T];D(A))$ and it satisfies (B.2.8). Similarly, a function $u:[0,T]\to X$ is a classical solution to (B.2.8) if $u\in$ $C^1((0,T];X) \cap C((0,T];D(A)) \cap C([0,T];X)$ and it satisfies (B.2.8).

We have seen that, in the case when f = 0 and $x \in \overline{D(A)}$, the function $u = T(\cdot)x$ is a classical solution to (B.2.8) and it is strict provided that $x \in D(A)$ and $Ax \in D(A)$. Actually, one can show that the previous one is the unique classical (resp. strict) solution to (B.2.8) with f = 0. Indeed, if $f \in C((0,T];X)$ and the function $t \mapsto ||f(t)||_X$ is integrable in (0,T), then any classical (resp. strict) solution u to (B.2.8) is given by the variation-ofconstants formula

$$u(t) = T(t)x + \int_0^t T(t-s)f(s)ds, \qquad t \in [0,T].$$
 (B.2.9)

Whenever the integral in (B.2.9) does make sense, the function u defined by (B.2.9) is said to be a mild solution of (B.2.8).

Appendix C

PDE's and analytic semigroups

In this appendix we recall some classical results on PDE's of elliptic and parabolic problems that we use in this book. Namely, we state some well-known (interior) Schauder estimates, some maximum principle and we see the relations between the theory of PDE's and the theory of analytic semigroups.

Throughout this appendix, by \mathcal{A} we denote the elliptic operator defined on smooth functions by

$$Au(x) = \sum_{i,j=1}^{N} q_{ij}(x)D_{ij}u(x) + \sum_{i=1}^{N} b_i(x)D_iu(x) + c(x)u(x), \qquad (C.0.1)$$

with real coefficients q_{ij} , b_i (i, j = 1, ..., N) and c defined in an open set Ω . We do not assume any smoothness assumption on the domain Ω unless it is explicitly mentioned.

The following condition is always assumed in this appendix.

Hypothesis C.0.1 $q_{ij} \equiv q_{ji}$ for any i, j = 1, ..., N and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \mu_0|\xi|^2, \qquad \xi \in \Omega, \ x \in \overline{\Omega},$$

for some positive constant μ_0 .

C.1 A priori estimates

In this section we recall some classical L^p - and Schauder interior estimates for solutions to elliptic and parabolic problems in bounded domains Ω or in the whole of \mathbb{R}^N .

We begin with the elliptic case and recall well-known interior L^p -estimates and L^p -estimates up to a part of the boundary of Ω , where homogeneous Neumann boundary conditions are prescribed.

Theorem C.1.1 ([66], Theorems 9.11 & 9.19) Let Ω be an open set and p any real number in the interval $(1, +\infty)$. Then, the following properties are met:

(i) if the coefficients of the operator A are bounded and continuous in Ω , then, for any open set $\Omega' \subset\subset \Omega$, there exists a positive constant C, depending on p, Ω , Ω' , μ_0 and the moduli of continuity of the coefficients q_{ij} in Ω' (i, j = 1, ..., N), such that

$$||u||_{W^{2,p}(\Omega')} \le C(||u||_{L^p(\Omega)} + ||\mathcal{A}u||_{L^p(\Omega)}),$$

for any $u \in L^p(\Omega) \cap W^{2,p}_{loc}(\Omega)$ such that $Au \in L^p(\Omega)$;

(ii) if the coefficients q_{ij} , b_i (i, j = 1, ..., N) and c are locally Lipschitz continuous in Ω , $u \in W^{2,p}_{loc}(\Omega)$ and $Au \in W^{1,p}_{loc}(\Omega)$, then $u \in W^{3,p}_{loc}(\Omega)$ and, for any $\Omega' \subset \subset \Omega'' \subset \subset \Omega$, there exists a positive constant \widetilde{C} , depending on p, Ω' , Ω'' and the coefficients of the operator A, such that

$$||u||_{W^{3,p}(\Omega')} \le \widetilde{C}(||u||_{L^p(\Omega'')} + ||\mathcal{A}u||_{W^{1,p}(\Omega'')}).$$

We also recall the following L^p -estimates up to the boundary of Ω .

Theorem C.1.2 Let $p \in (1, +\infty)$ and let Ω be an open set of class C^2 . Moreover, let Ω_0 and Ω_1 be two bounded subsets of Ω , with Ω_1 of class C^2 , such that their boundaries intersect the boundary of Ω and $\operatorname{dist}(\Omega_0, \Omega \setminus \Omega_1) > 0$. Further, assume that the coefficients of the operator A are continuous in $\overline{\Omega}$. Then, there exists a positive constant C, depending on $p, N, \mu, \Omega_0, \Omega_1$, the supnorm of the coefficients of the operator A and the modulus of continuity of the leading coefficients in Ω_1 , such that

$$||u||_{W^{2,p}(\Omega_0)} \le C (||\mathcal{A}u||_{L^p(\Omega_1)} + ||u||_{L^p(\Omega_1)}),$$

for any function $u \in W^{2,p}(\Omega)$ such that $\frac{\partial u}{\partial \nu} = 0$ on $\partial \Omega \cap \partial \Omega_1$.

To conclude the analysis of the elliptic case, we consider the following result concerning the regularity of distributional solutions to elliptic equations. Such results are essentially due to S. Agmon (see [3] and [4], in the case when p=2), which deals with the case when \mathbb{R}^N is replaced with a bounded set Ω . Since as far as we know they seem not to be written, at least in the most used textbooks, in the form that we need, for the reader's convenience we provide a proof of them. Such a proof is a variant of a proof due to G. Metafune, D. Pallara and A. Rhandi.

Theorem C.1.3 Assume that Hypothesis C.0.1 is satisfied. Moreover, let $p \in (1, +\infty)$, denote by p' the conjugate exponent to p (i.e., 1/p' = 1 - 1/p) and set

$$A = \sum_{i,j=1}^{N} q_{ij} D_{ij} + \sum_{i=1}^{N} b_i D_i.$$

Then, the following properties are met:

(i) if $q_{ij} \in C_b^1(\mathbb{R}^N)$, $b_i \in C_b(\mathbb{R}^N)$ (i, j = 1, ..., N) and $u \in L^p(\mathbb{R}^N)$ satisfies the estimate

$$\left| \int_{\mathbb{R}^N} u \mathcal{A} \varphi dx \right| \le C \|\varphi\|_{W^{1,p'}(\mathbb{R}^N)}, \tag{C.1.1}$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$, then u belongs to $W^{1,p}(\mathbb{R}^N)$;

- (ii) if $q_{ij} \in C^1(\mathbb{R}^N)$, $b_i \in C(\mathbb{R}^N)$ (i, j = 1, ..., N) (not necessarily bounded) and $u \in L^p_{loc}(\mathbb{R}^N)$ satisfies (C.1.1) for any $\varphi \in C^\infty_c(\mathbb{R}^N)$, then $u \in W^{1,p}_{loc}(\mathbb{R}^N)$;
- (iii) if $q_{ij} \in C_b^2(\mathbb{R}^N)$, $b_i \in C_b^1(\mathbb{R}^N)$ (i, j = 1, ..., N), $\operatorname{div} b \in L^p(\mathbb{R}^N)$ and $f, u \in L^p(\mathbb{R}^N)$ are such that

$$\int_{\mathbb{R}^N} u \mathcal{A} \varphi dx = \int_{\mathbb{R}^N} f \varphi dx, \qquad (C.1.2)$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$, then $u \in W^{2,p}(\mathbb{R}^N)$. In the case when $\mathcal{A} = \sum_{i,j=1}^N D_i(q_{ij}D_{ij}) + \sum_{i=1}^N b_iD_i$ the same result holds under the weaker assumptions $q_{ij}, b_i \in C_b^1(\mathbb{R}^N)$ (i, j = 1, ..., N);

(iv) if $q_{ij} \in C^2(\mathbb{R}^N)$, $b_i \in C^1(\mathbb{R}^N)$ (i, j = 1, ..., N) (not necessarily bounded) and $f, u \in L^p_{loc}(\mathbb{R}^N)$ satisfy (C.1.2) for any $\varphi \in C^\infty_c(\mathbb{R}^N)$, then $u \in W^{2,p}_{loc}(\mathbb{R}^N)$. In the case when $\mathcal{A} = \sum_{i,j=1}^N D_i(q_{ij}D_{ij}) + \sum_{i=1}^N b_iD_i$ the same result holds under the weaker assumptions $q_{ij}, b_i \in C^1(\mathbb{R}^N)$ (i, j = 1, ..., N).

Proof. We begin by proving the property (i). For this purpose, let $A_1:W^{2,p'}(\mathbb{R}^N)\to L^{p'}(\mathbb{R}^N)$ be the operator defined by

$$A_1 v = \sum_{i,j=1}^{N} q_{ij} D_{ij} v,$$
 (C.1.3)

for any $v \in W^{2,p'}(\mathbb{R}^N)$. From (C.1.1) we deduce that, for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$,

$$\left| \int_{\mathbb{R}^N} u A_1 \varphi dx \right| \le C \|\varphi\|_{W^{1,p'}(\mathbb{R}^N)}. \tag{C.1.4}$$

For any $h \in \mathbb{R}^N$, set $\tau_h u = |h|^{-1}(u(\cdot + h) - u)$. An explicit computation shows that

$$\int_{\mathbb{R}^N} \tau_h u \, \mathcal{A}_1 \varphi \, dx = \int_{\mathbb{R}^N} u \mathcal{A}_1(\tau_{-h}\varphi) \, dx + \int_{\mathbb{R}^N} u(\tau_{-h}q_{ij}) D_{ij}\varphi(\cdot -h) \, dx,$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$. Using the boundedness of the diffusion coefficients q_{ij} in \mathbb{R}^N (i, j = 1, ..., N) and (C.1.4), we obtain

$$\left| \int_{\mathbb{R}^N} \tau_h u(\lambda \varphi - A_1 \varphi) dx \right| \le C_1 \|\varphi\|_{W^{2,p'}(\mathbb{R}^N)}, \tag{C.1.5}$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$, any $\lambda > 0$ and some positive constant $C_1 = C_1(\lambda)$, independent of φ and h. Since $u \in L^p(\mathbb{R}^N)$, by density, (C.1.5) extends to any function in $W^{2,p'}(\mathbb{R}^N)$. We fix a large λ so that the operator $\lambda - A_1$ is invertible from $W^{2,p'}(\mathbb{R}^N)$ to $L^{p'}(\mathbb{R}^N)$ (see the forthcoming Theorem C.3.6(i)), and we set $\varphi = R(\lambda, A_1)(\tau_h u | \tau_h u|^{p-2})$. The function φ satisfies

$$\|\varphi\|_{W^{2,p'}(\mathbb{R}^N)} \le C_2 \|\tau_h u\|_{L^p(\mathbb{R}^N)}^{p-1},$$
 (C.1.6)

for some positive constant C_2 , independent of h. Then, plugging this particular φ into (C.1.5) and taking (C.1.6) into account, we obtain

$$\int_{\mathbb{R}^N} |\tau_h u|^p dx \le C_3,\tag{C.1.7}$$

with C_3 being independent of h. Now, from (C.1.7), we immediately deduce that $u \in W^{1,p}(\mathbb{R}^N)$.

Proof of (ii). Let us fix r > 0 and let $\varphi, \psi_r \in C_c^{\infty}(\mathbb{R}^N)$ be two smooth functions with $\chi_{B(r)} \leq \psi_r \leq \chi_{B(2r)}$. Since

$$\psi_r \mathcal{A}\varphi = \mathcal{A}(\psi_r \varphi) - \varphi \mathcal{A}\psi_r - 2\sum_{i,j=1}^N q_{ij} D_i \varphi D_j \psi_r, \qquad (C.1.8)$$

we immediately deduce that

$$\int_{\mathbb{R}^N} u\psi_r \mathcal{A}\varphi \, dx = \int_{\mathbb{R}^N} u\mathcal{A}(\varphi\psi_r) dx - \int_{\mathbb{R}^N} u\varphi \mathcal{A}\psi_r dx$$
$$-2\int_{\mathbb{R}^N} u \sum_{i,j=1}^N q_{ij} D_i \varphi D_j \psi_r dx.$$

Hence from (C.1.1), it follows that

$$\left| \int u\psi_r \mathcal{A}\varphi dx \right| \le C \|\varphi\|_{W^{1,p'}(\mathbb{R}^N)}.$$

Now, we consider a function $\eta \in C_c^{\infty}(\mathbb{R}^N)$ such that $\chi_{B(2r)} \leq \eta \leq \chi_{B(4r)}$ and we introduce the operator $\widetilde{\mathcal{A}} = \sum_{i,j=1}^N \widetilde{q}_{ij} D_{ij} + \sum_{i=1}^N \widetilde{b}_i D_i$, where

$$\widetilde{q}_{ij} = \eta q_{ij} + (1 - \eta)\delta_{ij}, \qquad \widetilde{b}_i = \eta b_i, \qquad i, j = 1, \dots, N.$$
 (C.1.9)

Of course,

$$\int u\psi_r \mathcal{A}\varphi dx = \int u\psi_r \widetilde{\mathcal{A}}\varphi dx$$

and the operator $\widetilde{\mathcal{A}}$ is uniformly elliptic in \mathbb{R}^N and has the diffusion and the drift coefficients belonging, respectively, to $C_b^1(\mathbb{R}^N)$ and to $C_b(\mathbb{R}^N)$. Hence,

we can apply the property (i) to the function $u\psi_r$, obtaining that $u\psi_r \in W^{1,p}(\mathbb{R}^N)$ and, consequently, that $u \in W^{1,p}(B(r))$. The arbitrariness of r > 0 implies that $u \in W^{1,p}_{loc}(\mathbb{R}^N)$.

Proof of (iii). We first assume that \mathcal{A} is not in divergence form. According to the property (i), we know that $u \in W^{1,p}(\mathbb{R}^N)$. Therefore, taking (C.1.2) into account, we get

$$\int_{\mathbb{R}^N} u A_1 \varphi dx = \int_{\mathbb{R}^N} f_1 \varphi dx, \qquad (C.1.10)$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ where A_1 is given by (C.1.3), $f_1 = f + u \operatorname{div} b + \sum_{i=1}^N b_i D_i u$ belong to $L^p(\mathbb{R}^N)$. Hence,

$$\int_{\mathbb{R}^N} u(\lambda \varphi - A_1 \varphi) dx = \int_{\mathbb{R}^N} g \varphi dx$$
 (C.1.11)

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$, where $g = \lambda u - f_1$ belongs to $L^p(\mathbb{R}^N)$. By density, we can extend (C.1.11) to any function $\varphi \in W^{2,p'}(\mathbb{R}^N)$. Now, we fix λ in the resolvent sets of both the operator A_1 and its adjoint A_1^* defined by

$$A_1^* u = \sum_{i,j=1}^N q_{ij} D_{ij} u + 2 \sum_{i,j=1}^N D_j q_{ij} D_i u + \sum_{i,j=1}^N (D_{ij} q_{ij}) u,$$

for any $u \in D(A_1^*) = W^{2,p}(\mathbb{R}^N)$ (see again the forthcoming Theorem C.3.6(i)). To prove that $u \in W^{2,p}(\mathbb{R}^N)$, we show that $u = R(\lambda, A_1^*)g$. For this purpose, it suffices to observe that (C.1.11) is satisfied if we insert $R(\lambda, A_1^*)g$ in place of u. Therefore, the function $z = R(\lambda, A_1^*)g - u$ satisfies

$$\int_{\mathbb{R}^N} z(\lambda \varphi - A_1 \varphi) dx = 0, \qquad (C.1.12)$$

for any $\varphi \in W^{2,p'}(\mathbb{R}^N)$. Since $\lambda I - A_1$ is surjective from $W^{2,p'}(\mathbb{R}^N)$ into $L^{p'}(\mathbb{R}^N)$, it is now immediate to check that z = 0.

In the case when \mathcal{A} is in divergence form, repeating the same computations as above (with the obvious changes), one can easily check that the same result holds also in the case when q_{ij} $(i,j=1,\ldots,N)$ belong to $C_b^1(\mathbb{R}^N)$. Indeed, in the above proof, the second-order derivatives of the diffusion coefficients q_{ij} $(i,j=1,\ldots,N)$ appeared only in the definition of the adjoint operator A_1^* . In the case when \mathcal{A} is in divergence form the operator A_1^* is given by

$$A_1^* u = \sum_{i,j=1}^N D_i(q_{ij}D_j u), \qquad u \in D(A_1^*) = W^{2,p}(\mathbb{R}^N)$$

and, therefore, there is no need of requiring the existence of the second order derivatives of the diffusion coefficients.

Proof of (iv). We just consider the case when \mathcal{A} is not in divergence form, since, as it has been pointed out at the end of the proof of the property (iii), the other case is completely similar. According to the property (ii), $u \in W_{\text{loc}}^{1,p}(\mathbb{R}^N)$. Therefore, also in this situation, u satisfies (C.1.10) for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$, where now $f_1 \in L_{\text{loc}}^p(\mathbb{R}^N)$. Fix r > 0 and let ψ_r be as in the proof of the property (ii). Taking (C.1.8) (written with A_1 instead of \mathcal{A}) into account, one can easily show that the function $v = u\psi_r$ satisfies

$$\int_{\mathbb{R}^N} (\lambda \varphi - A_1 \varphi) v dx = \int_{\mathbb{R}^N} g \varphi dx, \tag{C.1.13}$$

for any $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ and any $\lambda > 0$, where

$$g = \lambda v - \psi_r \left(f + u \operatorname{div} b + \sum_{i=1}^{N} b_i D_i u \right) + u A_1 \psi_r - 2 \sum_{i,j=1}^{N} D_i (q_{ij} u D_j \psi_r)$$

belongs to $L^p(\mathbb{R}^N)$. As it is immediately seen the function v satisfies the formula (C.1.13) also if we replace A_1 with the operator $\widetilde{A}_1 = \sum_{i,j=1}^N \widetilde{q}_{ij} D_{ij}$ when the coefficients \widetilde{q}_{ij} $(i,j=1,\ldots,N)$ are given by (C.1.9). Now, since $\widetilde{q}_{ij} \in C_b^2(\mathbb{R}^N)$ $(i,j=1,\ldots,N)$, according to the property (iii), $v \in W^{2,p}(\mathbb{R}^N)$ so that $u \in W^{2,p}(B(r))$ and the arbitrariness of r > 0 allows us to conclude.

We now consider the parabolic problems associated with the operator \mathcal{A} . For this purpose, we denote by $d:[0,+\infty)\times\overline{\Omega}\to\mathbb{R}$ the function defined by

$$d(t,x) = d(t,x) = \operatorname{dist}(x,\partial\Omega) \wedge \sqrt{t}, \qquad t > 0, \ x \in \overline{\Omega}.$$

Theorem C.1.4 ([60], Thms. 3.5, 3.10 and [101]) Assume that the coefficients q_{ij} , b_i (i, j = 1, ..., N) and c of the operator A belong to $C_b^{\alpha}(\overline{\Omega})$ for some $\alpha \in (0,1)$. Further assume that $u \in C_{loc}^{1+\alpha/2,2+\alpha}((0,T)\times\Omega)$ is a bounded (with respect to the sup-norm) solution of the equation

$$D_t u(t,x) - \mathcal{A}u(t,x) = 0, \qquad t \in (0,T), \ x \in \Omega.$$

Then, the following properties are met:

(i) there exists a positive constant C_1 , depending only on the coefficients of A, such that

$$|d^{0}, u|_{\alpha} + \sum_{i=1}^{N} |d, D_{i}u|_{\alpha} + \sum_{i,j=1}^{N} |d^{2}, D_{i}D_{j}u|_{\alpha} + |d^{2}, D_{t}u|_{\alpha} \le C_{1} \sup_{(0,T) \times \Omega} |u|,$$
(C.1.14)

where

$$\begin{split} |d^m,u|_{\alpha} &= \sup_{(t,x)\in(0,T)\times\Omega} |(d(t,x))^m u(t,x)| \\ &+ \sup_{(t,x),(s,y)\in(0,T)\times\Omega \atop (t,x)\neq(s,y)} (d(t,x)\wedge d(s,y))^{m+\alpha} \frac{|u(t,x)-u(s,y)|}{(|x-y|^2+|t-s|)^{\alpha/2}}. \end{split}$$

In particular, for any open set $\Omega' \subset\subset \Omega$ and any $s \in (0,T)$, there exists a positive constant C_2 depending on s, the coefficients of the operator A, Ω , Ω' and T, such that

$$||u||_{C^{1+\alpha/2,2+\alpha}([s,T)\times\Omega')} \le C_2 \sup_{(0,T)\times\Omega} |u|.$$
 (C.1.15)

Moreover, if $\operatorname{dist}(\Omega, \Omega') > \sqrt{T}$, then

$$\sup_{(t,x)\in(0,T)\times\Omega'} \left(t^{\frac{1}{2}} |Du(t,x)| + t|D^2u(t,x)| \right) \le C_3 \sup_{(t,x)\in(0,T)\times\Omega} |u(t,x)|, \tag{C.1.16}$$

for some positive constant C_3 , depending on the coefficients of the operator A, Ω , Ω' and T;

(ii) let the coefficients q_{ij} , b_i and c belong to $C^{k+\alpha}(\Omega)$ for any i, j = 1, ..., N and some $k \in \mathbb{N}$. Then, u is continuously differentiable in Ω_T up to the (k+2)-th-order, with respect to the space variables. Moreover, $D^k u \in C^{1+\alpha/2,2+\alpha}_{loc}((0,T)\times\Omega)$ and

$$||D^k u||_{C^{1+\alpha/2,2+\alpha}([\varepsilon',T]\times\Omega')} \le C_4 \sup_{(t,x)\in[\varepsilon,T]\times\Omega} |u(t,x)|, \quad (C.1.17)$$

for any $\varepsilon, \varepsilon' > 0$ ($\varepsilon < \varepsilon' < T$), any open set $\Omega' \subset\subset \Omega$ and some positive constant C_4 , depending on $\varepsilon, \varepsilon'$, T, Ω, Ω' , $||q_{ij}||_{C^{1+\alpha}(\Omega)}$ and $||b_i||_{C^{1+\alpha}(\Omega)}$ (i, j = 1, ..., N) and $||c||_{C^{1+\alpha}(\Omega)}$.

Finally, if $\operatorname{dist}(\Omega', \Omega) \geq \sqrt{T}$, then

$$\sup_{(t,x)\in(0,T)\times\Omega'} t^{\frac{3}{2}} |D^3 u(t,x)| \le C \sup_{(t,x)\in(0,T)\times\Omega} |u(t,x)|.$$
 (C.1.18)

The next theorem provides us with Schauder estimates up to (a part of) the boundary of Ω .

Theorem C.1.5 ([94], Theorem 4.10.1) Let Ω be an open subset of \mathbb{R}^N with boundary of class $C^{2+\alpha}$ for some $\alpha \in (0,1)$, and let Ω' and Ω'' be two bounded subsets of Ω such that $\Omega' \subset \Omega'' \subset \Omega$ and $\operatorname{dist}(\Omega', \Omega \setminus \Omega'') > 0$. Moreover, assume that the coefficients q_{ij} , b_i (i, j = 1, ..., N) and c of the operator A belong to $C^{\alpha}_{\operatorname{loc}}(\overline{\Omega})$. Finally assume that $u \in C^{1+\alpha/2,2+\alpha}([T_1,T_2] \times \overline{\Omega}'')$ solves the differential equation $D_t u - \mathcal{A}u = 0$ in $(T_1,T_2) \times \Omega''$ for some $0 \leq T_1 < T_2$. Then, the following properties are met:

(i) if $u \equiv 0$ (resp. $\frac{\partial u}{\partial \nu} \equiv 0$)) on $(T_1, T_2) \times \partial \Omega''$, then, for any $T^* \in (T_1, T_2)$, there exists a positive constant $C = C_{T^*, T_1, T_2, \Omega', \Omega''}$ such that

$$||u||_{C^{1+\alpha/2,2+\alpha}([T^*,T_2]\times\overline{\Omega}')} \le C||u||_{L^{\infty}((T_1,T_2)\times\Omega'')};$$
 (C.1.19)

(ii) if $T_1 = T^* = 0$, $u \equiv 0$ (resp. $\frac{\partial u}{\partial \nu} \equiv 0$)) on $(0, T_2) \times \partial \Omega''$, then there exists a positive constant $C = C_{T_1, T_2, \Omega', \Omega''}$ such that

$$||u||_{C^{1+\alpha/2,2+\alpha}([0,T_2]\times\overline{\Omega}')} \le C\left(||u||_{L^{\infty}((0,T_2)\times\Omega'')} + ||u(0,\cdot)||_{C^{2+\alpha}(\Omega'')}\right).$$
(C.1.20)

C.2 Classical maximum principles

In this section we collect the classical maximum principles for continuous solutions to both the elliptic equations (C.2.1), (C.2.2) and for the parabolic Cauchy problems (C.3.2) and (C.3.3) that we use in this book. For the proofs we refer the reader to [22], [89, Chapter 8] and [129].

We make the following assumptions on Ω and on the coefficients of the operator A.

Hypotheses C.2.1 (i) Ω is either an open bounded set with boundary of class C^2 or $\Omega = \mathbb{R}^N$;

- (ii) q_{ij} , b_i (i, j = 1, ..., N) and c belong to $C_b(\overline{\Omega})$;
- (iii) Hypothesis C.0.1 is satisfied.

Moreover, we set

$$c_0 = \sup_{x \in \Omega} c(x)$$

and denote by $\nu = \nu(x)$ the outward unit normal vector to $\partial\Omega$ at $x \in \partial\Omega$.

Theorem C.2.2 Let $\lambda > c_0$ and suppose that $u \in W^{2,p}_{loc}(\Omega)$ for any $p \in (1,+\infty)$ satisfies the differential inequality $\lambda u - \mathcal{A}u \geq 0$. Then, the following properties are met:

- (i) if $u \ge 0$ on $\partial\Omega$, then $u \ge 0$ in Ω ;
- (ii) if $\frac{\partial u}{\partial \nu} \geq 0$ on $\partial \Omega$, then $u \geq 0$ on Ω .

Moreover, if $\lambda > c_0$, $f \in C_b(\overline{\Omega})$ and $u \in \bigcap_{1 \le p < +\infty} W^{2,p}_{loc}(\Omega)$ solves the problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ u(x) = 0, & x \in \partial\Omega. \end{cases}$$
 (C.2.1)

or

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ \frac{\partial u}{\partial \nu}(x) = 0, & x \in \partial\Omega, \end{cases}$$
 (C.2.2)

then

$$||u||_{\infty} \le \frac{1}{\lambda - c_0} ||f||_{\infty}. \tag{C.2.3}$$

Concerning the parabolic equation $D_t u - Au = 0$ we have the following result.

Proposition C.2.3 Fix T > 0. Then, the following properties are met:

(i) if
$$z \in C_b([0,T] \times \overline{\Omega}) \cap C^{1,2}((0,T] \times \Omega)$$
 satisfies

$$\begin{cases} D_t z(t,x) - \mathcal{A} z(t,x) \le 0, & t \in (0,T], x \in \Omega, \\ z(t,x) \le 0, & t \in (0,T], x \in \partial \Omega, \\ z(0,x) \le 0, & x \in \overline{\Omega}, \end{cases}$$

then z < 0 in $[0, T] \times \overline{\Omega}$.

(ii) if
$$z \in C_b([0,T] \times \overline{\Omega}) \cap C^{0,1}((0,T] \times \overline{\Omega}) \times \cap C^{1,2}((0,T] \times \Omega)$$
 satisfies
$$\begin{cases} D_t z(t,x) - \mathcal{A}z(t,x) \leq 0, & t \in (0,T], \ x \in \Omega, \\ \frac{\partial z}{\partial \nu}(t,x) \leq 0, & t \in (0,T], \ x \in \partial \Omega, \\ z(0,x) \leq 0, & x \in \Omega, \end{cases}$$

then z < 0 in $[0, T] \times \overline{\Omega}$.

(iii) If
$$z \in C_b([0,T] \times \overline{\Omega}) \cap C^{0,1}((0,T] \times \overline{\Omega}) \times \cap C^{1,2}((0,T] \times \Omega)$$
 satisfies
$$D_t z(t,x) - \mathcal{A}z(t,x) \leq 0, \qquad t \in (0,T], \ x \in \Omega,$$

and z attains its maximum value M in $(0,T] \times \Omega$ at some point (t_0,x_0) , then z = M in $[0,t_0] \times \Omega$.

Further, if Ω satisfies the interior sphere condition and z attains its maximum value at some point $(t,x) \in (0,T] \times \partial \Omega$, then $\partial z/\partial \nu > 0$ at (t,x).

C.3 Existence of classical solution to PDE's and analytic semigroups

Here, we make the following assumptions on Ω and on the coefficients of the operator \mathcal{A} .

Hypotheses C.3.1 (i) Ω is an open set with a boundary which is uniformly of class $C^{2+2\alpha}$ for some $\alpha \in (0,1)$ (or, possibly, $\Omega = \mathbb{R}^N$);

- (ii) q_{ij} , b_i (i, j = 1, ..., N) and c belong to $C_b^{2\alpha}(\overline{\Omega})$;
- (iii) $q_{ij} \equiv q_{ji}$ for any i, j = 1, ..., N and

$$\sum_{i,j=1}^{N} q_{ij}(x)\xi_i\xi_j \ge \mu_0|\xi|^2, \qquad \xi \in \mathbb{R}^N, \ x \in \overline{\Omega},$$

for some positive constant μ_0 .

For notational convenience, we set

$$c_0 = \sup_{x \in \overline{\Omega}} c(x), \tag{C.3.1}$$

and we denote by $\nu = \nu(x)$ the exterior unit normal vector to $\partial\Omega$ at $x \in \partial\Omega$.

$$\begin{cases} D_t u(t,x) - \mathcal{A}u(t,x) = 0, & t > 0, x \in \Omega, \\ u(t,x) = 0, & t > 0, x \in \partial\Omega, \\ u(0,x) = f(x), & x \in \Omega \end{cases}$$
 (C.3.2)

and the Cauchy-Neumann problem

$$\begin{cases}
D_t u(t,x) - \mathcal{A}u(t,x) = 0, & t > 0, x \in \Omega, \\
\frac{\partial u}{\partial \nu}(t,x) = 0, & t > 0, x \in \partial\Omega, \\
u(0,x) = f(x), & x \in \Omega.
\end{cases}$$
(C.3.3)

More precisely, the following results can be proved (see also [74], [78], [79], [94, Chapter 4, Theorem 5.2] and [140, Chapter 5, Sections 3 & 5]).

Proposition C.3.2 Let Hypotheses C.3.1 be satisfied. For any $f \in C_b(\overline{\Omega})$, problem (C.3.2) admits a unique solution $u \in C([0, +\infty) \times \overline{\Omega} \setminus (\{0\} \times \partial \Omega)) \cap C^{1,2}((0, +\infty) \times \Omega)$, which is bounded in $[0, T] \times \overline{\Omega}$ for any T > 0. Moreover, it satisfies

$$||u(t,\cdot)||_{\infty} \le e^{c_0 t} ||f||_{\infty}, \qquad t > 0;$$
 (C.3.4)

and

$$t^{\frac{k}{2}} \|D^k T(t)f\|_{\infty} \le C \|f\|_{\infty}, \qquad t \in (0, T),$$

for any T > 0 some constant C = C(T), independent of f, and any k = 1, 2, 3. Here, c_0 is given by C.3.1. Moreover, there exists a unique function $G_{\Omega} \in C((0, +\infty) \times \Omega \times \Omega)$ such that

$$u(t,x) = \int_{\Omega} G_{\Omega}(t,x,y)f(y)dy, \quad t > 0, \ x \in \Omega.$$

The function G_{Ω} is called the fundamental solution of the problem (C.3.2), it is positive in $(0, +\infty) \times \Omega \times \Omega$ and satisfies

$$G_{\Omega}(t+s,x,y) = \int_{\Omega} G_{\Omega}(s,x,z)G_{\Omega}(t,z,y)dz, \qquad (C.3.5)$$

for any s,t>0 and any $x,y\in\Omega$. Moreover, for any t>0 the function $G(t,\cdot,\cdot)$ is bounded in $\Omega\times\Omega$.

Finally, if $f \in C_0(\Omega)$, then $u \in C([0, +\infty) \times \overline{\Omega}) \cap C_{loc}^{1+\alpha, 2+2\alpha}((0, +\infty) \times \overline{\Omega})$; if $f \in C_c^{2+\alpha}(\mathbb{R}^N)$, then $u \in C^{1+\alpha/2, 2+\alpha}([0, T] \times \overline{\Omega})$ for any T > 0, and, if $f \in C_c^{2+k+\alpha}(\Omega)$ and Ω is of class $C^{2+k+\alpha}$ for some $k \in \mathbb{N}$, then all the space derivatives of u up to the k-th order belong to $C^{1+\alpha/2, 2+\alpha}([0, T] \times \overline{\Omega})$ for any T > 0.

Similar results hold for the Cauchy-Neumann problem (C.3.3). Here, we limit ourselves to stating the results that we actually need in this book.

Proposition C.3.3 For any $f \in C_b(\overline{\Omega})$, there exists a unique solution $u \in C([0,+\infty) \times \overline{\Omega}) \cap C_{loc}^{1+\alpha/2,2+\alpha}((0,+\infty) \times \overline{\Omega})$ to the Cauchy-Neumann problem (C.3.3), which is bounded in $[0,T] \times \Omega$ for any T > 0. Moreover,

$$||u(t,\cdot)||_{\infty} \le e^{c_0 t} ||f||_{\infty}, \qquad t > 0;$$
 (C.3.6)

and

$$t^{\frac{k}{2}} \|D^k u(t, \cdot)\|_{\infty} \le C \|f\|_{\infty}, \qquad t \in (0, T),$$

for any T > 0, any k = 1, 2, 3 and some positive constant C = C(T). Here, c_0 is given by (C.3.1).

Concerning the elliptic equation $\lambda u - Au = f$, we have the following results.

Proposition C.3.4 Under Hypotheses C.3.1, for any $f \in C_b(\overline{\Omega})$ and any $\lambda > c_0$, there exists a unique solution $u \in \bigcap_{1 \le p < +\infty} W^{2,p}(\Omega)$ to the Dirichlet problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ u(x) = 0, & x \in \partial\Omega. \end{cases}$$
 (C.3.7)

The function u can be represented by

$$u(x) = \int_{\Omega} K_{\lambda}^{\Omega}(x, y) f(y) dy, \qquad x \in \Omega,$$
 (C.3.8)

where K_{λ}^{Ω} , the so-called Green's function, is given by

$$K_{\lambda}^{\Omega}(x,y) = \int_{0}^{+\infty} e^{-\lambda t} G_{\Omega}(t,x,y) dt, \qquad x,y \in \Omega.$$
 (C.3.9)

Proposition C.3.5 Under Hypotheses C.3.1, for any $f \in C_b(\overline{\Omega})$ and any $\lambda > c_0$, there exists a unique solution $u \in \bigcap_{1 to the Neumann$ problem

$$\begin{cases} \lambda u(x) - \mathcal{A}u(x) = f(x), & x \in \Omega, \\ \frac{\partial u}{\partial \nu}(x) = 0, & x \in \partial \Omega. \end{cases}$$
 (C.3.10)

Most of the results of this section may be rephrased in terms of analytic semigroups. Indeed, the following theorem holds.

Theorem C.3.6 Under Hypotheses C.3.1, the following properties are met:

- (i) for any $p \in (1, +\infty)$ the operator $A_p : D(A_p) \to L^p(\mathbb{R}^N)$, defined by $A_p u = \mathcal{A}u$ for any $u \in D(A_p) := W^{2,p}(\mathbb{R}^N)$, is sectorial in $L^p(\mathbb{R}^N)$ and $D(A_n)$ is dense in $L^p(\mathbb{R}^N)$:
- (ii) let A_p be the operator defined by

$$\begin{cases} D(A_p) = W^{2,p}(\Omega) \cap W_0^{1,p}(\Omega), \\ A_p u = \mathcal{A}u, & u \in D(A_p). \end{cases}$$

Then, A_p is sectorial in $L^p(\Omega)$, and $D(A_p)$ is dense in $L^p(\Omega)$;

(iii) for any $p \in (1, +\infty)$ let A_p be the operator defined by

$$\begin{cases} D(A_p) = \left\{ u \in W^{2,p}(\Omega) : \frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega \right\}, \\ A_p u = \mathcal{A}u, \ u \in D(A_p). \end{cases}$$

Then, the operator A_p is sectorial in $L^p(\Omega)$ and $D(A_p)$ is dense in $L^p(\Omega)$. Moreover, $\rho(A) \supset (c_0, +\infty)$;

(iv) let $A: D(A) \to C_b(\overline{\Omega})$ be defined by

$$\begin{cases}
D(A) = \left\{ u \in \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2,p}(\Omega) : u_{|\partial\Omega} = 0, \ Au \in C_b(\overline{\Omega}) \right\}, \\
Au = Au, \ u \in D(A).
\end{cases}$$
(C.3.11)

Then, A is sectorial in $C_b(\overline{\Omega})$ and $\overline{D(A)} = C_0(\overline{\Omega}) = \{u \in C_b(\overline{\Omega}) : u = 0\}$ 0 on $\partial\Omega$. Moreover, $\rho(A) \supset (c_0, +\infty)$ and

$$D_A(\alpha,\infty) = \left\{ u \in C_b^{2\alpha}(\overline{\Omega}) : u_{\left|\partial\Omega\right.} = 0 \right\},$$

$$D_A(1+\alpha,\infty) = \left\{ u \in C_b^{2+2\alpha}(\overline{\Omega}) : u_{\left|\partial\Omega\right.} = (\mathcal{A}u)_{\left|\partial\Omega\right.} = 0 \right\}.$$

Finally, $\{u \in C_b^1(\overline{\Omega}) : u_{|\partial\Omega} = 0\}$ belongs to the class $J_{1/2}$ between $C_b(\overline{\Omega})$ and D(A).

All the previous results hold also in the case when Ω is replaced with \mathbb{R}^N , provided we drop out any boundary condition.

(v) Let $A:D(A)\to C_b(\overline{\Omega})$ be defined by

$$\begin{cases} D(A) = \left\{ u \in \bigcap_{1 \le p < +\infty} W_{\text{loc}}^{2,p}(\Omega) : \frac{\partial u}{\partial \nu} = 0, \text{ on } \partial \Omega, u, Au \in C_b(\overline{\Omega}) \right\}, \\ Au = Au, u \in D(A). \end{cases}$$

 $-^{(C.3.12)}$

Then, A is sectorial in $C_b(\overline{\Omega})$ and its domain is dense in $C(\overline{\Omega})$. Moreover,

$$D_A(\alpha, \infty) = \begin{cases} C_b^{2\alpha}(\overline{\Omega}), & \alpha \in (0, 1/2), \\ \left\{ u \in C_b^{2\alpha}(\overline{\Omega}) : \frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega \right\}, & \alpha \in (1/2, 1); \end{cases}$$
(C.3.13)

$$D_A(1+\alpha,\infty)$$

$$= \left\{ \begin{cases} \left\{ u \in C_b^{2+2\alpha}(\overline{\Omega}) : \frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega \right\}, & \alpha \in (0, 1/2), \\ \left\{ u \in C_b^{2+2\alpha}(\overline{\Omega}) : \frac{\partial u}{\partial \nu} = \frac{\partial (\mathcal{A}u)}{\partial \nu} = 0 \text{ on } \partial \Omega \right\}, & \alpha \in (1/2, 1). \end{cases}$$
(C.3.14)

Finally, $\{u \in C_b^1(\overline{\Omega}) : \partial u/\partial \nu = 0 \text{ on } \partial \Omega\}$ belongs to the class $J_{1/2}$ between $C_b(\overline{\Omega})$ and D(A).

For the proof we refer the reader to [1, 4, 5, 6, 24, 25, 137, 138].

Remark C.3.7 The properties (ii) and (iii) in Theorem C.3.6, combined with the results in Section B.2, show that the classical solution to problem (C.3.2) (resp. (C.3.3)) in Proposition C.3.2 (resp. in Proposition C.3.3) is given by $u = T(\cdot)f$, where $\{T(t)\}$ is the analytic semigroup generated in $C_b(\overline{\Omega})$ by the realization of the operator \mathcal{A} with homogeneous Dirichlet boundary conditions (resp. with homogeneous Neumann boundary conditions).

Similar results hold for the solutions to the elliptic equations considered in Propositions C.3.4 and C.3.5. Namely, the solution u to (C.3.7) (resp. to

(C.3.10)) is given by $u = R(\lambda, A)f$, where $R(\lambda, A)$ is the resolvent operator associated with the operator A in Theorem C.3.6(iv) (resp. Theorem C.3.6(v)).

Appendix D

Some properties of the distance function

In this appendix we collect some regularity results of the distance function $d(x) = \operatorname{dist}(x, \partial\Omega)$, when $\partial\Omega$ is the boundary of a smooth open subset Ω of \mathbb{R}^N . We limit ourselves to state them in the case when Ω is bounded since they are classical results (see, e.g., [66, Section 14.6]). Moreover, we show that most of them may be extended also to the case where Ω is unbounded. The results we present here are taken from [59].

Definition D.0.1 We say that an open subset Ω of \mathbb{R}^N is uniformly of class $C^{2+\alpha}$, for some $\alpha \in [0,1)$, if there exist a countable open covering $\{U_n\}$ of $\partial\Omega$ and a family of diffeomorphisms $g_n: \overline{U}_n \to \overline{B}(1)$ of class $C^{2+\alpha}$, such that $g_n(U_n \cap \Omega) = \{x \in B(1) : x_N > 0\}$, $g_n(U_n \cap \partial\Omega) = \{x \in B(1) : x_N = 0\}$ (where x_N denotes the N-th component of the vector x) and, moreover,

- (i) there exists $k \in \mathbb{N}$ such that $\bigcap_{n \in J} U_n = \emptyset$ if $\operatorname{card}(J) > k$;
- (ii) there exists $\varepsilon > 0$ such that $\{x \in \Omega : d(x, \partial\Omega) < \varepsilon\} \subset \bigcup_{n \in \mathbb{N}} V_n$, where $V_n = g_n^{-1}(B(1/2))$;
- (iii) there exists C > 0 such that $||g_n||_{C^{2+\alpha}} + ||g_n^{-1}||_{C^{2+\alpha}} \le C$, for any $n \in \mathbb{N}$.

Remark D.0.2 It is possible to show that if Ω is uniformly of class $C^{2+\alpha}$, then the diffeomorphism g_n $(n \in \mathbb{N})$ can be chosen so that

$$(\operatorname{Jac} g_n)(x)\nu(x) = -\alpha_n e_N, \qquad x \in U_n \cap \partial\Omega, \quad n \in \mathbb{N},$$

for some positive function α_n , where ν denotes the unit outward normal and $e_N = (0, \ldots, 0, 1)$. Roughly speaking, the diffeomorphisms g_n leave the normal direction unchanged. We refer the reader to [146, Chapter 1, Section 2.4] for the proof of this property.

Remark D.0.3 As a consequence of the condition (iii) in Definition D.0.1, the principal curvatures $\lambda_1, \ldots, \lambda_{N-1}$ of $\partial\Omega$ are bounded both from above and below with respect to x, hence

$$M_0 := \inf_{x \in \partial \Omega} \left\{ \frac{\partial \nu}{\partial \tau}(x) \cdot \tau, \ |\tau| = 1, \ \tau \cdot \nu(x) = 0 \right\} > -\infty,$$

since, locally, $\partial \nu / \partial \tau \cdot \tau = \lambda_1 \tau_1^2 + \dots + \lambda_{N-1} \tau_{N-1}^2$.

Definition D.0.4 The open set Ω is said to satisfy the uniform interior sphere condition if, for any point $z_0 \in \partial \Omega$, there exists a ball $B(z_0, r_{z_0}) \subset \Omega$ such that $\overline{B}(z_0, r_{z_0}) \cap \partial \Omega = \{z_0\}$, and $\inf_{z_0 \in \Omega} r_{z_0} > 0$.

As the following proposition shows, if $\partial\Omega$ is uniformly of class C^2 , then Ω satisfies the interior sphere condition.

Proposition D.0.5 ([59], Prop. B.2) If $\partial\Omega$ is uniformly of class C^2 , then it satisfies a uniform interior sphere condition.

Proof. Let $y_0 \in \partial\Omega$ and let $j \in \mathbb{N}$ be such that $y_0 \in V_j$. Further, denote by $\nu(y_0)$ and T_{y_0} , respectively, the unit inward normal vector and the tangent space to $\partial\Omega$ at y_0 . By the implicit function theorem and by a rotation of coordinates, we may assume that the x_N axis lies in the direction $\nu(y_0)$ and that $\partial\Omega\cap V_j$ is given by the equation $x_N=\psi_j(x')$, where $x'=(x_1,\ldots,x_{N-1})$ and ψ_j is of class $C^{2+\alpha}$. Moreover $\Omega\cap V_j\subseteq \{x_N<\psi_j(x')\}$. Now, let B be a tangent ball to T_{y_0} at y_0 lying in the set $\{x_N<\psi_j(x')\}$. Taking a smaller radius r_0 if necessary, we find that $B\subseteq\Omega$ and $\overline{B}\cap\partial\Omega=\{y_0\}$. Moreover, comparing the curvatures of $T(y_0)$, B, and $\partial\Omega$ at y_0 we have $0<1/r_0\le k_i(y_0)$, where k_i for $i=1,\ldots,N-1$ denote the so-called principal curvatures of $\partial\Omega$. Since k_i are related to the second order derivatives of ψ_j (see [66, (14.93)]) which are uniformly bounded, we obtain that $k_i(y_0)\le C$, hence $r_0\ge C^{-1}$.

Now, we are ready to prove the main result of this appendix.

Proposition D.0.6 ([59], Proposition B.3) Let $\partial\Omega$ be uniformly of class C^2 , Further, fix $\delta = 2/C$ and set $\Omega_{\delta} = \{y \in \overline{\Omega} \mid d(y) < \delta\}$. Then,

- (i) for any $x \in \Omega_{\delta}$ there exists a unique $y = y(x) \in \partial \Omega$ such that |x y| = d(x);
- (ii) $d \in C_b^2(\Omega_\delta)$;
- (iii) $Dd(x) = \nu(y(x))$, for any $x \in \Omega_{\delta}$.

Proof. We limit ourselves to proving the property (i). Indeed, the proof of the properties (ii) and (iii) relies on the property (i) and the implicit function theorem, and it is completely similar to that given in [66, Section 14.6] in the case when Ω is bounded.

Of course, the existence part in (i) is obvious. To prove the uniqueness of the point y=y(x) at any $x\in\Omega_{\delta}$, let $y\in\partial\Omega$ be such that d(x)=|x-y|. According to Proposition D.0.5, there exists a ball $B=\xi+B(\rho)$ such that $B\subset\Omega$ and $\overline{B}\cap\partial\Omega=\{y\}$. Moreover from the definition of δ it follows that $x\in B$. It is easy to see that x and ξ lie on the normal direction $\nu(y)$ and that the balls $x+B(d(x)), \, \xi+B(\rho)$ are tangent at y. Then, x+B(d(x)) still verifies the interior sphere condition at y. It follows that for any $z\in\partial\Omega\setminus\{y\}$,

one has $z \notin x + B(d(x))$, so that y is actually the unique point such that |x-y| = d(x).

Appendix E

Function spaces: definitions and main properties

In this appendix we collect all the function spaces that we consider in this book.

E.1 Spaces of functions that are continuous or Hölder continuous in domains $\Omega \subset \mathbb{R}^N$

E.1.1 Isotropic spaces

Given an open set $\Omega \subset \mathbb{R}^N$ (even $\Omega = \mathbb{R}^N$), we denote by $C_b(\Omega)$ (resp. $C_b(\overline{\Omega})$) the space of bounded and continuous functions $f: \Omega \to \mathbb{R}$ (resp. $f: \overline{\Omega} \to \mathbb{R}$) and we endow it with the sup-norm, i.e.,

$$||f||_{\infty} = \sup_{x \in \Omega} |f(x)|, \qquad f \in C_b(\Omega) \quad \text{(resp. } f \in C_b(\overline{\Omega})\text{)}.$$

Let Ω be bounded, $C_0(\Omega)$ denotes the subspace of $C_b(\overline{\Omega})$ of functions $f: \overline{\Omega} \to \mathbb{R}$ such that f = 0 on $\partial \Omega$. When Ω is unbounded, $C_0(\overline{\Omega})$ denotes the subset of $C_b(\overline{\Omega})$ of functions $f: \overline{\Omega} \to \mathbb{R}$ such that f = 0 on $\partial \Omega$ and

$$\lim_{\substack{|x| \to +\infty \\ x \in \Omega}} f(x) = 0.$$

Next, for any k > 0, we denote by $C_b^k(\overline{\Omega})$ the space of functions $f: \overline{\Omega} \to \mathbb{R}$ which are continuously differentiable in Ω up to the [k]-order and such that $D^{\alpha}f \in C_b(\overline{\Omega})$ for any $|\alpha| \leq [k]$ ([k] denoting the integer part of k) and $D^{\alpha}f$ is (k-[k])-Hölder continuous for any $|\alpha| = [k]$. We endow $C_b^k(\overline{\Omega})$ with the norm

$$||f||_{C_b^k(\overline{\Omega})} = \sum_{|\alpha| \le [k]} ||D^{\alpha}f||_{\infty} + \sum_{|\alpha| = [k]} [D^{\alpha}f]_{C^{k-[k]}(\overline{\Omega})}, \qquad f \in C_b^k(\overline{\Omega}), \quad (E.1.1)$$

where

$$[f]_{C^{k-[k]}(\overline{\Omega})} = \sup_{\substack{x,y \in \Omega \\ x \neq y}} \frac{|f(x) - f(y)|}{|x - y|^{k-[k]}}.$$

Similarly, $C^k_{\text{loc}}(\mathbb{R}^N)$, $(k \in \mathbb{R}_+)$, is the set of functions $f: \mathbb{R}^N \to \mathbb{R}$ which are continuously differentiable up to the [k]-th-order in \mathbb{R}^N and have k-th-order

derivatives which are (k-[k])-Hölder continuous in K for any compact set $K \subset \mathbb{R}^N$.

We say that $f \in BUC^k(\mathbb{R}^N)$ $(k \in \mathbb{N})$ if $f \in C_b^k(\mathbb{R}^N)$ and $D^{\alpha}f$ is uniformly continuous in \mathbb{R}^N for any $|\alpha| = k$. We endow it with the norm in (E.1.1).

Moreover, for any open set $\Omega \subset \mathbb{R}^N$ and any $k \geq 0$, we denote by $C_c^k(\Omega)$ (resp. $C_c^k(\overline{\Omega})$) the subset of $C_b^k(\overline{\Omega})$ of functions $f: \overline{\Omega} \to \mathbb{R}$ compactly supported in Ω (resp. in $\overline{\Omega}$). We endow both $C_c^k(\Omega)$ and $C_c^k(\overline{\Omega})$ with the norm of $C_b^k(\overline{\Omega})$. We say that $f \in C_c^\infty(\Omega)$ if $f \in C_c^k(\Omega)$ for any $k \in \mathbb{N}$.

Finally, for any smooth open set Ω , we denote by $C^1_{\nu}(\overline{\Omega})$ the set of functions $f \in C^1_{h}(\overline{\Omega})$ such that the normal derivative $\partial f/\partial \nu$ identically vanishes on $\partial \Omega$.

Sometimes we also consider spaces of vector-valued continuous functions. For any $k \in [0, +\infty)$, and any $m \in \mathbb{N}$, we denote by $C^k(\Omega, \mathbb{R}^m)$ (resp. $C^k_{\text{loc}}(\Omega, \mathbb{R}^m)$) the set of functions $f = (f_1, \ldots, f_m) : \Omega \to \mathbb{R}^m$ such that $f_j \in C^k(\Omega)$ (resp. $f_j \in C^k_{\text{loc}}(\Omega)$) for any $j = 1, \ldots, m$.

E.1.2 Anisotropic Hölder spaces in \mathbb{R}^N

We now define the anisotropic Hölder spaces that we use in this book. For this purpose, for any $r \in \{1, \ldots, N-1\}$, we split $\mathbb{R}^N := \mathbb{R}^r \times \mathbb{R}^{N-r}$.

To define the spaces $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$, let us introduce the Zygmund spaces $C_b^{\alpha}(\mathbb{R}^m)$ $(m \in \mathbb{N})$. For any $\gamma \in (0,1]$, the Zygmund space $C_b^{\gamma}(\mathbb{R}^m)$ is the set of functions $f \in C_b(\mathbb{R}^m)$ such that

$$[f]_{\mathcal{C}^{\gamma}_b(\mathbb{R}^m)} = \sup_{\stackrel{x,y \in \mathbb{R}^m}{x \neq x}} \frac{|f(x) - 2f\left(\frac{x+y}{2}\right) + f(y)|}{|x-y|^{\gamma}} < +\infty.$$

 $\mathcal{C}_b^{\gamma}(\mathbb{R}^m)$ is a Banach space when it is endowed with the norm

$$||f||_{\mathcal{C}_{b}^{\gamma}(\mathbb{R}^{m})} = ||f||_{\infty} + [f]_{\mathcal{C}_{b}^{\gamma}(\mathbb{R}^{m})}, \qquad f \in \mathcal{C}_{b}^{\gamma}(\mathbb{R}^{m}).$$

For $\gamma > 1$ such that $\gamma \notin \mathbb{N}$,

$$C_b^{\gamma}(\mathbb{R}^m) = \left\{ f \in C_b^{[\gamma]}(\mathbb{R}^m) : D^{\alpha} f \in C_b^{\gamma - [\gamma]}(\mathbb{R}^m) \text{ for any } |\alpha| = [\gamma] \right\},\,$$

and it is normed by

$$\|f\|_{\mathcal{C}^{\gamma}_b(\mathbb{R}^m)} = \|f\|_{C^{[\gamma]}_b(\mathbb{R}^m)} + \sum_{|\alpha|=[\gamma]} [D^{\alpha}f]_{\mathcal{C}^{\gamma-[\gamma]}_b(\mathbb{R}^m)}.$$

Finally, if $\gamma \in \mathbb{N}$, $\gamma > 1$,

$$\mathcal{C}_b^{\gamma}(\mathbb{R}^m) = \left\{ f \in C_b^{\gamma-1}(\mathbb{R}^m) : D^{\alpha}f \in \mathcal{C}_b^1(\mathbb{R}^m), \quad \text{for any } |\alpha| = \gamma - 1 \right\}$$

and it is normed by

$$||f||_{\mathcal{C}_b^{\gamma}(\mathbb{R}^m)} = ||f||_{C_b^{\gamma-1}(\mathbb{R}^m)} + \sum_{|\alpha|=\gamma-1} [D^{\alpha}f]_{\mathcal{C}_b^1(\mathbb{R}^m)}.$$

We can now define the space $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$ for any $\theta\in(0,1]$, by setting

$$\begin{split} \mathcal{C}^{3\theta,\theta}_{b,r}(\mathbb{R}^N) &= \left\{ f: \mathbb{R}^N \to \mathbb{R}: f(\cdot,y) \in \mathcal{C}^{3\theta}_b(\mathbb{R}^r) \quad \forall y \in \mathbb{R}^{N-r}, \\ \sup_{y \in \mathbb{R}^{N-r}} \|f(\cdot,y)\|_{\mathcal{C}^{3\theta}_b(\mathbb{R}^r)} < +\infty, \\ f(x,\cdot) &\in \mathcal{C}^{\theta}_b(\mathbb{R}^{N-r}) \quad \forall x \in \mathbb{R}^r, \quad \sup_{x \in \mathbb{R}^r} \|f(x,\cdot)\|_{\mathcal{C}^{\theta}_b(\mathbb{R}^{N-r})} < +\infty \right\} \end{split}$$

and we norm it by

$$\|f\|_{\mathcal{C}^{3\theta,\theta}_{b,r}(\mathbb{R}^{N})} = \sup_{y \in \mathbb{R}^{N-r}} \|f(\cdot,y)\|_{\mathcal{C}^{3\theta}_{b}(\mathbb{R}^{r})} + \sup_{x \in \mathbb{R}^{r}} [f(x,\cdot)]_{\mathcal{C}^{\theta}_{b}(\mathbb{R}^{N-r})}.$$

Similarly, we define and norm the spaces $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$, replacing everywhere the Zygmund spaces \mathcal{C}_b^{α} with the usual Hölder spaces C_b^{α} .

When no confusion may arise, we simply write $C_b^{3\theta,\theta}(\mathbb{R}^N)$ and $C_b^{3\theta,\theta}(\mathbb{R}^N)$.

Remark E.1.1 It is well known (see, e.g., [104, Proposition 0.2.2]) that the spaces $C_b^{\gamma}(\mathbb{R}^m)$ and $C_b^{\gamma}(\mathbb{R}^m)$ coincide when $\gamma \notin \mathbb{N}$ and the corresponding norms are equivalent, whereas, if $\gamma \in \mathbb{N}$, $C_b^{\gamma}(\mathbb{R}^m)$ is properly continuously embedded in $C_b^{\gamma}(\mathbb{R}^m)$. It follows that $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N) = C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$ with equivalence of the corresponding norms, if $\theta \in \mathbb{R}_+ \setminus (\frac{1}{3}\mathbb{N})$, while $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$ is properly and continuously embedded in $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$ otherwise.

Remark E.1.2 By interpolation, it is easy to check that, if $f \in C_{b,r}^{3\theta,\theta}(\mathbb{R}^N)$ and $3\theta > 1$, then all the derivatives of f, with respect to the first r variables and up to the $[3\theta]$ -order are (bounded and) continuous in \mathbb{R}^N . Indeed, for any $x \in \mathbb{R}^N$ split x = (y, z) where $y = (x_1, \ldots, x_r)$ and $z = (x_{r+1}, \ldots, x_N)$, and let $k \in \mathbb{N}$ be such that $k < 3\theta$. Since there exists a positive constant C such that

$$\|g\|_{C^k_{b,r}(\mathbb{R}^r)} \leq C\|g\|_{C_b(\mathbb{R}^r)}^{1-\frac{k}{3\theta}}\|g\|_{C_b^{3\theta}(\mathbb{R}^r)}^{\frac{k}{3\theta}}, \qquad g \in C^{3\theta}_{b,r}(\mathbb{R}^r)$$

(see Proposition A.4.4), we easily deduce that

$$||f(\cdot, y_{2}) - f(\cdot, y_{1})||_{C_{b}^{k}(\mathbb{R}^{r})}$$

$$\leq C||f(\cdot, y_{2}) - f(\cdot, y_{1})||_{C_{b}(\mathbb{R}^{r})}^{1 - \frac{k}{3\theta}}||f(\cdot, y_{2}) - f(\cdot, y_{1})||_{C_{b}^{3\theta}(\mathbb{R}^{r})}^{\frac{k}{3\theta}}$$

$$\leq 2C||f||_{C_{b,r}^{3\theta,\theta}(\mathbb{R}^{N})}||y_{2} - y_{1}||^{\theta - \frac{k}{3}}, \tag{E.1.2}$$

for any $z_1, z_2 \in \mathbb{R}^{N-r}$. From (E.1.2) we deduce that the k-th derivatives of f with respect to the first r variables belong to $C_{b,r}^{3\theta-k,\theta-k/3}(\mathbb{R}^N)$, so, in particular, they are continuous in \mathbb{R}^N .

E.2 Parabolic Hölder spaces

Given any interval $I \subset (0, +\infty)$, any set $D \subset \mathbb{R}^N$ with nonempty interior part, and two integers $h, k \in \mathbb{N} \cup \{0\}$, $h \leq 1$, we denote by $C^{h,k}(I \times D)$, the set of functions $f: I \times D \to \mathbb{R}$ which are continuous in $I \times D$, continuously differentiable in the interior of $I \times D$ up to the h-th order with respect to the time variable and up to the k-th order with respect to the space variables and such with all the derivatives which can be extended by continuity to $I \times D$. When $I \times D$ is bounded, we norm $C^{h,k}(I \times D)$ by setting

$$||f||_{C_b^{h,k}(I \times D)} = \sum_{j=0}^h ||D_t^{(j)} f||_{\infty} + \sum_{0 < |\alpha| < k} ||D^{\alpha} f||_{\infty}.$$
 (E.2.1)

Moreover, when $I \times D$ is unbounded we write $C_b^{h,k}(I \times D)$ to denote the set of functions belonging to $C_b^{h,k}(I \times D)$ which are bounded together with their derivatives. We endow $C_b^{h,k}(I \times D)$ with the norm in (E.2.1).

We denote by $C_c^{\infty}(I \times D)$ the set of functions $f: I \times D \to \mathbb{R}$ compactly supported in $I \times D$, which admit classical derivatives $D_t^j D^{\alpha} f$ of any order in $I \times D$.

Next, for any $\alpha \in (0,1)$ and any $k \in \mathbb{N} \cup \{0\}$, we denote by $C^{1+\alpha/2,2+\alpha}(I \times D)$ the subspace of $C^{1,2}(I \times \Omega)$ of functions $f: I \times D \to \mathbb{R}$ such that the functions $D_t f, D_x^{\beta} f$ ($|\beta| \leq 2$) are bounded and satisfy

$$\|f\|_{C^{1+\alpha/2,2+\alpha}(I\times D)}:=\sup_{x\in D}\|f(\cdot,x)\|_{C_b^{1+\alpha/2}(I)}+\sup_{t\in I}\|f(t,\cdot)\|_{C_b^{2+\alpha}(D)}<+\infty.$$
 (E.2.2)

It is a Banach space when endowed with the norm in (E.2.2).

Similarly, we say that $f \in C^{1+\alpha/2,2+\alpha}_{loc}(I \times D)$ if $f \in C^{1+\alpha/2,2+\alpha}(F)$ for any open set $F := (a,b) \times \Omega'$ with compact closure in $I \times D$.

Let Y be a subspace of $C_b(\overline{\Omega})$. By B(I;Y) and $C_b^k(I;Y)$ $(k \geq 0)$ we denote the set of functions $f:I \to Y$ which are, respectively, bounded in I and differentiable in I (with values in Y) up to the [k]-th order, with bounded derivatives, $f^{([k])}$ being Hölder continuous in I with exponent k-[k]. We norm these spaces by setting

$$||f||_{B(I;Y)} = \sup_{t \in I} ||f(t,\cdot)||_Y,$$

and

$$||f||_{C^k(I;Y)} = \sum_{j=0}^{[k]} ||f^{(j)}||_{B(I;Y)} + \sup_{\substack{s,t \in I \\ s \neq t}} \frac{||f(t) - f(s)||_Y}{|t - s|^{k - [k]}}.$$

Sometimes, we identify a bounded and continuous function $f: I \to \Omega \to \mathbb{R}$ with an element of the space $C(I; C_b(\Omega))$ in the natural way. Modulo this

identification, we can say that $f \in C^{1+\alpha/2,2+k+\alpha}(I \times \Omega)$ $(k \in \mathbb{N} \cup \{0\})$ if and only if $f \in C^{1+\alpha/2}(I; C_b(\overline{\Omega})) \cap C(I; C_b^{2+k+\alpha}(\Omega))$.

E.3 L^p and Sobolev spaces

Given an open set $\Omega \subset \mathbb{R}^N$ (possibly $\Omega = \mathbb{R}^N$), we denote by $B_b(\Omega)$ the space of bounded Borel measurable functions defined in Ω .

We now define the usual Sobolev spaces. Let μ be a positive measure defined on the σ -algebra of all the Borel sets of Ω . We denote by $L^p(\Omega, \mu)$ $(p \in [1, +\infty))$ the space of Borel measurable functions $f: \Omega \to \mathbb{R}$ such that

$$||f||_{L^p(\Omega,\mu)} := \left(\int_{\Omega} |f|^p d\mu\right)^{\frac{1}{p}} < +\infty,$$
 (E.3.1)

if $p < +\infty$ and

$$||f||_{\infty} = \min \{\Lambda > 0 \text{ s.t. } |f(x)| \le \Lambda, \ \mu - a.e. \text{ in } \Omega \}.$$
 (E.3.2)

 $L^p(\Omega,\mu)$ $(p\in[1,+\infty])$ is a Banach space when endowed with the norm in (E.3.1) and (E.3.2).

By $L^p(\Omega)$ $(p \in [1, +\infty])$ we denote the usual L^p spaces related to the Lebesgue measure defined on the σ -algebra of the Lebesgue measurable sets.

By $L^p_{loc}(\Omega)$ $(p \in [1, +\infty])$ we denote the space of functions f which belong to $L^p(F)$ for any open set $F \subset \mathbb{R}^N$ with compact closure.

The Sobolev space $W^{k,p}(\Omega,\mu)$ $(k \in \mathbb{N}, p \in [1,+\infty])$ is the subspace of $L^p(\Omega,\mu)$ of functions f which admits distributional derivatives up to the k-th-order in $L^p(\Omega,\mu)$. It is a Banach space when endowed with the norm

$$||f||_{W^{k,p}(\Omega,\mu)} = \sum_{|\alpha| \le k} ||D^{\alpha}f||_{L^p(\Omega,\mu)}, \quad f \in W^{k,p}(\Omega,\mu).$$

As above, we simply write $W^{k,p}(\Omega)$ when the underlying measure is the Lebesgue measure.

We say that $u \in W^{k,p}_{loc}(\Omega)$ $(k \in \mathbb{N}, p \in [1, +\infty])$ if $u \in W^{k,p}(F)$ for any open set $F \subset \mathbb{R}^N$ with compact closure.

For more details on L^p and Sobolev spaces we refer the reader to [2].

References

- [1] P. Acquistapace and B. Terreni. Hölder classes with boundary conditions as interpolation spaces. *Math. Z.*, 195(4):451–471, 1987.
- [2] R.A. Adams. Sobolev spaces. Academic Press Inc., New York, 1975.
- [3] S. Agmon. The L^p approach to the Dirichlet problem. I. Regularity theorems. Ann. Scuola Norm. Sup. Pisa (3), 13:405–448, 1959.
- [4] S. Agmon. On the eigenfunctions and the eigenvalues of general elliptic boundary value problems. *Comm. Pure Appl. Math*, 15:119–147, 1962.
- [5] S. Agmon, A. Douglis, and L. Nirenberg. Estimates near the boundary for solutions of elliptic partial differential equations satisfying general boundary conditions. I. Comm. Pure Appl. Math., 12:623–727, 1959.
- [6] S. Agmon, A. Douglis, and L. Nirenberg. Estimates near the boundary for solutions of elliptic partial differential equations satisfying general boundary conditions. II. Comm. Pure Appl. Math., 17:35–92, 1964.
- [7] W. Arendt. Gaussian estimates and interpolation of the spectrum in L^p . Differential Integral Equations, 7(5-6):1153–1168, 1994.
- [8] W. Arendt, C. Batty, M. Hieber, and F. Neubrander. Vector-valued Laplace transforms and Cauchy problems, volume 96 of Monographs in Mathematics. Birkhäuser, Basel, 2001.
- [9] D.G. Aronson and P. Besala. Parabolic equations with unbounded coefficients. *J. Differential Equations*, 3:1–14, 1967.
- [10] R. Azencott. Behaviour of diffusion semigroups at infinity. Bull. Soc. Math. France, 102:193–240, 1974.
- [11] D. Bakry. Transformations de Riesz pour les semigroupes symétriques. Seconde partie: Etude sous la condition $\Gamma_2 \geq 0$. In Séminaire de Probabilités XIX, volume 1123 of Lecture Notes in Math., pages 145–174, Berlin, 1985. Springer-Verlag. (French).
- [12] D. Bakry and M. Émery. Diffusions hypercontractives. In Séminaire de

- Probabilités XIX, volume 1123 of Lecture Note in Math., pages 177–206, Berlin, 1985. Springer-Verlag. (French).
- [13] D. Bakry and M. Ledoux. Lévy-Gromov's isoperimetric inequality for an infinite dimensional diffusion generator. *Invent. Math.*, 123(2):259– 281, 1996.
- [14] S. Bernstein. Sur la généralisation du problème de Dirichlet, I. *Math. Ann.*, 62(2):253–271, 1906. (French).
- [15] M. Bertoldi and S. Fornaro. Gradient estimates in parabolic problems with unbounded coefficients. *Studia Math.*, 165(3):221–254, 2004.
- [16] M. Bertoldi, S. Fornaro, and L. Lorenzi. Gradient estimates for parabolic problems with unbounded coefficients in non convex unbounded domains. *Forum Math.* (to appear).
- [17] M. Bertoldi, S. Fornaro, and L. Lorenzi. Pointwise gradient estimates in exterior domains. *Arch. Math.* (*Basel*) (to appear).
- [18] M. Bertoldi and L. Lorenzi. Estimates of the derivatives for parabolic problems with unbounded coefficients. *Trans. Amer. Math. Soc.*, 357(7):2627–2664, 2005.
- [19] W. Bodanko. Sur le problème de Cauchy et les problèmes de Fourier pour les équations paraboliques dans un domaine non bornè. *Ann. Polon. Math.*, 18:79–94, 1966. (French).
- [20] V.I. Bogachev, N.V. Krylov, and M. Röckner. Differentiability of invariant measures and of transition probabilities of singular diffusions. *Dokl. Akad. Nauk*, 376(2):151–154, 2001. (Russian).
- [21] V.I. Bogachev, N.V. Krylov, and M. Röckner. On regularity of transition probabilities and invariant measures of singular diffusions under minimal conditions. *Comm. Partial Differential Equations*, 26(11-12):2037–2080, 2001.
- [22] J.-M. Bony. Principe du maximum dans le espaces de Sobolev. C.R. Acad. Sci. Paris, Série A, 265:333–336, 1967. (French).
- [23] D.L. Burkholder. Martingales and Fourier analysis in Banach Spaces. In Probability and analysis, volume 1206 of Lecture Notes in Math., pages 61–108. Springer-Verlag, Berlin, 1986.
- [24] S. Campanato. Generation of analytic semigroups, in the Hölder topology, by elliptic operators of second order with neumann boundary conditions. *Le Matematiche*, 35(1-2):61–72, 1980.

- [25] S. Campanato. Generation of analytic semigroups by elliptic operators of second order in Hölder spaces. *Ann. Scuola Norm. Sup. Pisa Cl. Sci.* (4), 8(3):495–512, 1981.
- [26] P. Cannarsa and V. Vespri. Generation of analytic semigroups by elliptic operators with unbounded coefficients. SIAM J. Math. Anal., 18(3):857–872, 1987.
- [27] P. Cannarsa and V. Vespri. Generation of analytic semigroups in the L^p -topology by elliptic operators in \mathbb{R}^N . Israel J. Math., 61(3):235–255, 1988.
- [28] S. Cerrai. A Hille-Yosida theorem for weakly continuous semigroups. Semigroup Forum, 49:349–367, 1994.
- [29] S. Cerrai. Some results for second order elliptic operators having unbounded coefficients. *Differential Integral Equations*, 11(4):561–588, 1998.
- [30] S. Cerrai. Second order PDE's in finite and infinite dimension. A probabilistic approach, volume 1762 of Lecture Notes in Math. Springer-Verlag, Berlin, 2001.
- [31] R. Chill, E. Fasangova, G. Metafune, and D. Pallara. The sector of analyticity of the Ornstein-Uhlenbeck semigroup on L^p spaces with respect to invariant measure. J. London Math. Soc. (2), 71(3):703–722, 2005.
- [32] A. Chojnowska-Michalik and B. Goldys. Nonsymmetric Ornstein-Uhlenbeck semigroups as second quantized operator. *J. Math. Kyoto Univ.*, 36(3):481–498, 1996.
- [33] A. Chojnowska-Michalik and B. Goldys. Symmetric Ornstein-Uhlenbeck semigroups and their generators. *Probab. Theory Related Fields*, 124(4):459–486, 2002.
- [34] P. Clemént and J. Prüss. Completely positive measures and Feller semi-groups. *Math. Ann.*, 287(1):73–105, 1990.
- [35] R.R. Coifman and G. Weiss. Transference methods in analysis. volume 31 of *Regional Conference Series in Mathematics*. American Mathematical Society, Providence, R.I., 1976.
- [36] G. Da Prato. Perturbation of the Ornstein-Uhlenbeck semigroups. Rend. Istit. Mat. Univ. Trieste, 28(suppl.):101–126, 1997.
- [37] G. Da Prato. Regularity results for some degenerate parabolic equations. Riv. Mat. Univ. Parma, $6(2^*):245-257$, 1999.

- [38] G. Da Prato and P. Grisvard. Sommes d'opérateurs linéaires et équations différentelles opérationelles. *J. Math. Pures Appl.* (9), 54(3):305–387, 1975. (French).
- [39] G. Da Prato and A. Lunardi. On the Ornstein-Uhlenbeck operator in spaces of continuous functions. *J. Funct. Anal.*, 131(1):94–114, 1995.
- [40] G. Da Prato and A. Lunardi. Elliptic operators with unbounded drift coefficients and Neumann boundary condition. J. Differential Equations, 198(1):35–52, 2004.
- [41] G. Da Prato and V. Vespri. Maximal L^p regularity for elliptic equations with unbounded coefficients. *Nonlin. Anal.*, 49(6):747–755, 2002.
- [42] G. Da Prato and J. Zabczyk. Ergodicity for infinite-dimensional systems, volume 229 of London Mathematical Society Lecture Note Series. Cambridge University Press.
- [43] G. Da Prato and J. Zabczyk. Stochastic Equations in Infinite Dimensions, volume 44 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 1992.
- [44] E.B. Davies. L^1 properties of second order elliptic operators. Bull. London Math. Soc., 17(5):417–436, 1985.
- [45] E.B. Davies. Heat kernels and spectral theory, volume 92 of Cambridge Tracts in Mathematics. Cambridge University Press, Cambridge, 1990.
- [46] J. Diestel and J.J. Uhl. *Vector measures*, volume 15 of *Amer. Math. Soc. Math. Surveys.* Amer. Math. Soc., Providence, R.I., 1977.
- [47] G. Dore and A. Venni. On the closedness of the sum of two closed operators. *Math. Z.*, 196(2):189–201, 1987.
- [48] N. Dunford and J.T. Schwartz. *Linear Operators*. Interscience Publishers, Ltd., London, 1958.
- [49] E.B. Dynkin. Markov Processes, volume 121 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, Berlin-Göttingen-Heidelberg, 1965.
- [50] S. Eidelman, S. Kamin, and F. Porper. Uniqueness of solutions of the Cauchy problem for parabolic equations degenerating at infinity. *Asymptot. Anal.*, 22(3-4):349–358, 2000.
- [51] K.J. Engel and R. Nagel. One-parameter semigroups for linear evolution equations, volume 194 of Graduate Texts in Mathematics. Springer-Verlag, New York, 2000.

- [52] J.B. Epperson. The hypercontractive approach to exactly bounding an operator with complex Gaussian kernel. *J. Funct. Anal.*, 87(1):1–30, 1989.
- [53] S.N. Ethier and T.G. Kurtz. Markov Processes. John Wiley & Sons, Inc., New York, 1986.
- [54] L.C. Evans and R.F. Gariepy. Measure theory and fine properties of functions. CRC Press, Boca Raton, FL, 1992.
- [55] G.F. Feissner. A Gaussian measure analogue to Sobolev's inequality. PhD thesis, Cornell University, 1972.
- [56] G.F. Feissner. Hypercontractive semigroups and Sobolev's inequality. Trans. Amer. Math. Soc., 210:51–62, 1975.
- [57] W. Feller. Diffusion processes in one dimension. *Trans. Amer. Math. Soc.*, 77:1–31, 1954.
- [58] S. Fornaro and V. Manco. On the domain of some ordinary differential operators in spaces of continuous functions. *Arch. Math.* (*Basel*), 82(4):335–343, 2004.
- [59] S. Fornaro, G. Metafune, and E. Priola. Gradient estimates for parabolic Dirichlet problems in unbounded domains. *J. Differential Equations*, 205(2):329–353, 2004.
- [60] A. Friedman. Partial differential equations of parabolic type. Prentice-Hall, Inc., Englewood Cliffs, N.J., 1964.
- [61] M. Fuhrman. Analyticity of transition semigroups and closability of bilinear forms in Hilbert spaces. *Studia Math.*, 115(1):53–71, 1995.
- [62] M. Fukushima, Y. Oshima, and M. Takeda. Dirichlet forms and symmetric Markov processes, volume 19 of de Gruyter Studies in Mathematics. Walter de Gruyter, Berlin, New York, 1994.
- [63] N. Garofalo and E. Lanconelli. Level sets of the fundamental solution and Harnack inequality for degenerate equations of kolmogorov type. Trans. Amer. Math. Soc., 321(2):775–792, 1990.
- [64] M. Giaquinta. Introduction to regularity theory for nonlinear elliptic systems. Birkhäuser Verlag, Basel, 1993.
- [65] I.I. Gihman and A.V. Skorokhod. *The theory of stochastic processes*. Springer-Verlag, New York-Heidelberg, 1975.
- [66] D. Gilbarg and N.S. Trudinger. Elliptic partial differential equations of

- second order, 2nd Edition. Springer-Verlag, New York, 2001.
- [67] B. Goldys. On analyticity of Ornstein-Uhlenbeck semigroups. Atti Accad. Naz. Lincei Cl. Sci. Fis. Mat. Natur. Rend. Lincei (9) Mat. Appl., 10(3):131–140, 1999.
- [68] B. Goldys and M. Kocan. Diffusion semigroups in spaces of continuous functions with mixed topology. *J. Differential Equations*, 173(1):17–39, 2001.
- [69] L. Gross. Logarithmic Sobolev inequalities. Amer. J. Math., 97(4):1061– 1083, 1975.
- [70] L. Gross. Logarithmic Sobolev inequalities and contractivity properties of semigroups. In *Dirichlet forms* (*Varenna*, 1992), volume 1563 of *Lecture Notes in Math.*, pages 54–88. Springer-Verlag, Berlin, 1993.
- [71] M. Hieber, L. Lorenzi, and A. Rhandi. Uniform and L^p estimates of the derivatives for parabolic operators with unbounded coefficients in exterior domains and applications. Technical Report 425, Dipart. Mat. Univ. Parma, 2005.
- [72] E. Hille and R.S. Phillips. Functional analysis and semigroups, volume 31 of Amer. Math. Soc. Coll. Publ. Amer. Math. Soc., Providence, R.I., 1957.
- [73] L. Hörmander. Hypoelliptic second order differential equations. Acta Math., 119:147–171, 1967.
- [74] A.M. Il'in, A.S. Kalashnikov, and O.A. Oleinik. Second order linear equations of parabolic type. *Usphehi Mat. Nauk.*, 17(3):3–146, 1962. (Russian).
- [75] K. Itô and H.P. McKean Jr. Diffusion processes and their sample paths, volume 125 of Grundlehren der mathematischen Wissenschaften. Academic Press, New York, 1964.
- [76] S. Itô. Fundamental solutions of parabolic differential equations and boundary value problems. *Jap. J. Math.*, 27:55–102, 1957.
- [77] B. Jefferies. Weakly integrable semigroups on locally convex spaces. J. Funct. Anal., 66(3):347-364, 1986.
- [78] F. John. General properties of linear elliptic partial differential equations. In *Proc. Symp. Spectral Theory and Differential Problems*, pages 113–175, Oklahoma Agricultural and Mechanical College, Stillwater, Oklahoma, 1951.

- [79] F. John. Plane waves and spherical means applied to partial differential equations. Interscience Publishers, Inc., New York, London, 1955.
- [80] C.B. Morrey Jr. Multiple integrals in the calculus of variations, volume 130 of Grundlehren der mathematischen Wissenschaften. Springer-Verlag, New York, 1966.
- [81] T. Kato. Perturbation theory for linear operators. Second edition, volume 132 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, New York, 1976.
- [82] T. Kato. L^p-theory of Schrödinger operators with a singular potential. In Aspects of Positivity in Functional Analysis (Tübingen (1985)), volume 122 of North-Holland Math. Stud., pages 63–78, Amsterdam, 1986. North-Holland.
- [83] R.Z. Khas'minskii. Stochastic stability of differential equations. Sijthoff and Noordhoff, Alphen aan den Rijn, 1980.
- [84] A.E. Kogoj and E. Lanconelli. An invariant Harnack inequality for a class of hypoelliptic ultraparabolic equations. *Mediterr. J. Math*, 1(1):51–80, 2004.
- [85] A. Kolmogorov. Zufällige bewegungen (zur theorie der brownschen bewegung). Ann. of Math. (2), 35(1):116–117, 1934. (German).
- [86] H. Komatsu. Fractional powers of operators. Pacific J. Math., 19:285–346, 1966.
- [87] S.G. Krein. Linear differential equations in Banach space, volume 29 of Translations of Mathematical Monographs. American Mathematical Society, Providence, R.I., 1972.
- [88] N.V. Krylov. Introduction to the theory of diffusion processes, volume 142 of Translations of Mathematical Monographs. American Mathematical Society, Providence, R.I., 1995.
- [89] N.V. Krylov. Lectures on elliptic and parabolic equations in Hölder spaces, volume 12 of Graduate Studies in Mathematics. American Mathematical Society, Providence, R.I., 1996.
- [90] M. Krzyżański. Sur la solution fondamentale de l'équation linéaire normale du type parabolique dont le dernier coefficient est non borné. I. Atti Accad. Naz. Lincei Rend. Cl. Sci. Fis. Mat. Nat. (8), 32:326–330, 1962. (French).
- [91] M. Krzyżański. Sur la solution fondamentale de l'équation linéaire normale du type parabolique dont le dernier coefficient est non borné. II.

- Atti Accad. Naz. Lincei Rend. Cl. Sci. Fis. Mat. Nat. (8), 32:471–476, 1962. (French).
- [92] M. Krzyżański and A. Szybiak. Construction et étude de la solution fondamentale de l'équation linéaire du type parabolique dont le dernier coefficient est non borné. I. Atti Accad. Naz. Lincei Rend. Cl. Sci. Fis. Mat. Nat. (8), 27:26–30, 1959. (French).
- [93] M. Krzyżański and A. Szybiak. Construction et étude de la solution fondamentale de l'équation linéaire du type parabolique dont le dernier coefficient est non borné. II. Atti Accad. Naz. Lincei Rend. Cl. Sci. Fis. Mat. Nat. (8), 27:113–117, 1959. (French).
- [94] O.A. Ladyzhenskaja, V.A. Solonnikov, and N.N. Uralceva. Linear and quasilinear equations of parabolic type, volume 23 of Translations of Mathematical Monographs. American Mathematical Society, Providence, R.I., 1967.
- [95] E. Lanconelli. On a class of Kolmogorov-Fokker-Planck operators. In *Progress in elliptic and parabolic partial differential equations* (Capri, 1994), volume 350 of Pitman Res. Notes Math. Ser., pages 173–183, Harlow, 1996. Longman.
- [96] E. Lanconelli and F. Lascialfari. A boundary value problem for a class of quasilinear operators of Fokker-Planck type. *Ann. Univ. Ferrara Sez. VII*, 41(suppl.):65–84, 1997.
- [97] E. Lanconelli, A. Pascucci, and S. Polidoro. Linear and nonlinear ultraparabolic equations of Kolmogorov type arising in diffusion theory and in finance. In *Nonlinear problems in mathematical physics and related* topics, II, volume 2 of *Int. Math. Ser.* (N.Y.), pages 243–265, New York, 2002. Kluwer/Plenum.
- [98] E. Lanconelli and S. Polidoro. On a class of hypoelliptic evolution operators. *Rend. Sem. Mat. Univ. Politec. Torino*, 52(1):29–63, 1994.
- [99] E.M. Landis. Second order equations of elliptic and parabolic type, volume 171 of Translations of Mathematical Monographs. American Mathematical Society, Providence, R.I., 1998.
- [100] S. Lang. Linear algebra. Third edition. Undergraduate Texts in Mathematics. Springer-Verlag, New York, 1987.
- [101] G. Lieberman. Second order parabolic differential equations. World Scientific Publishing Co. Inc., River Edge, N.J., 1996.
- $\left[102\right]\;$ L. Lorenzi. Estimates of the derivatives of a class of parabolic degenerate

- operators with unbounded coefficients in \mathbb{R}^N . Ann. Scuola Norm. Sup. Pisa Cl. Sci. (5), 4(2):255–293, 2005.
- [103] L. Lorenzi. Schauder estimates for elliptic and parabolic degenerate problems with unbounded coefficients in \mathbb{R}^N . Differential Integral Equations, 18(5):531–566, 2005.
- [104] A. Lunardi. Analytic Semigroups and Optimal Regularity in Parabolic Problems, volume 16 of Progress in Nonlinear Differential Equations and their Applications. Birkhäuser, Basel, 1995.
- [105] A. Lunardi. An interpolation method to characterize domains of generators of semigroup. Semigroup Forum, 53(3):321–329, 1996.
- [106] A. Lunardi. On the Ornstein-Uhlenbeck Operator in L^2 spaces with respect to invariant measures. Trans. Amer. Math. Soc., 349(1):155–169, 1997.
- [107] A. Lunardi. Schauder estimates for a class of degenerate elliptic and parabolic operators with unbounded coefficients in \mathbb{R}^N . Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 24(1):133–164, 1997.
- [108] A. Lunardi. Schauder theorems for linear elliptic and parabolic problems with unbounded coefficients in \mathbb{R}^n . Studia Math., 128(2):171–198, 1998.
- [109] P. Malliavin. Stochastic analysis, volume 313 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, Berlin, 1997.
- [110] M. Manfredini. The Dirichlet problem for a class of ultraparabolic equations. Adv. Differential Equations, 2(5):831–866, 1997.
- [111] M. Manfredini and A. Pascucci. A priori estimates for quasilinear degenerate parabolic equations. Proc. Amer. Math. Soc., 131:1115–1120, 2003.
- [112] G. Metafune. L^p -spectrum of Ornstein-Uhlenbeck operators. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 30(1):97–124, 2001.
- [113] G. Metafune, D. Pallara, and E. Priola. Spectrum of the Ornstein-Uhlenbeck operators in L^p spaces with respect to invariant measures. J. Funct. Anal., 196(1):40–60, 2002.
- [114] G. Metafune, D. Pallara, and A. Rhandi. Global properties of invariant measures. J. Funct. Anal., 223(2):396–424, 2005.
- [115] G. Metafune, D. Pallara, and M. Wacker. Compactness properties of Feller semigroups. *Studia Math.*, 153(2):179–206, 2002.

- [116] G. Metafune, D. Pallara, and M. Wacker. Feller semigroups on \mathbb{R}^N . Semigroup Forum, 65(2):159–205, 2002.
- [117] G. Metafune and E. Priola. Some classes of non-analytic Markov semi-groups. J. Math. Anal. Appl., 294(2):596–613, 2004.
- [118] G. Metafune, J. Prüss, A. Rhandi, and R. Schnaubelt. The domain of the Ornstein-Uhlenbeck operator on an L^p-Space with invariant measure. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (5), 1(2):471–485, 2002.
- [119] G. Metafune, J. Prüss, A. Rhandi, and R. Schnaubelt. L^p -regularity for elliptic operators with unbounded coefficients. Adv. Differential Equations, 10(10):1131-1164, 2005.
- [120] S. Monniaux and J. Prüss. A theorem of Dore-Venni type for noncommuting operators. *Trans. Amer. Math. Soc.*, 349(12):4787–4814, 1997.
- [121] E. Nelson. The free Markoff field. J. Funct. Anal., 12:211–227, 1973.
- [122] A. Pascucci. Hölder regularity for a Kolmogorov equation. *Trans. Amer. Math. Soc.*, 355(3):901–924, 2003.
- [123] A. Pazy. Semigroups of linear operators and applications to partial differential equations, volume 44 of Applied Mathematical Sciences. Springer-Verlag, Berlin, 1983.
- [124] K. Petersen. Ergodic theory, volume 2 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 1983.
- [125] E. Priola. On a class of Markov type semigroups in spaces of uniformly continuous and bounded functions. *Studia Math.*, 136(3):271–295, 1999.
- [126] E. Priola. The Cauchy problem for a class of Markov-type semigroups. Commun. Appl. Anal., 5(1):49–75, 2001.
- [127] E. Priola. On a Dirichlet problem involving an Ornstein-Uhlenbeck operator. *Potential Anal.*, 18(3):251–287, 2003.
- [128] E. Priola and J. Zabczyk. Liouville theorems for non-local operators. J. Funct. Anal., 216(2):455–490, 2004.
- [129] M.H. Protter and H.F. Weinberger. *Maximum principles in differential equations*. Prentice-Hall, Inc., Englewood Cliffs, N.J., 1967.
- [130] J. Prüss. Evolutionary integral equations and applications, volume 87 of Monographs in Mathematics. Birkhäuser Verlag, Basel, 1993.
- $\left[131\right]\,$ J. Prüss and H. Sohr. On operators with bounded imaginary powers in

- Banach spaces. Math. Z., 203(3):429-452, 1990.
- [132] J. Prüss and H. Sohr. Imaginary powers of elliptic second order differential operators in L^p -space. *Hiroshima Math. J.*, 23(1):161–192, 1993.
- [133] M. Reed and B. Simon. *Methods of modern mathematical physics II, Fourier Analysis, self adjontness*. Academic Press, New York, San Francisco, London, 1975.
- [134] O.S. Rothaus. Logarithmic Sobolev inequalities and the spectrum of Schrödinger operators. *J. Funct. Anal.*, 42(1):110–120, 1981.
- [135] W. Rudin. *Real and complex analysis*. McGraw Hill Book Co., New York-Toronto, 1966.
- [136] W. Stannat. (Nonsymmetric) Dirichlet operators on L^1 : existence, uniqueness and associated Markov processes. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 28(1):99–140, 1999.
- [137] H.B. Stewart. Generation of analytic semigroups by strongly elliptic operators. *Trans. Amer. Math. Soc.*, 199:141–162, 1974.
- [138] H.B. Stewart. Generation of analytic semigroups by strongly elliptic operators under general boundary conditions. *Trans. Amer. Math. Soc.*, 259(1):299–310, 1980.
- [139] D.W. Stroock and S.R.S. Varadhan. Multidimensional diffusion processes, volume 233 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, New York, 1979.
- [140] H. Tanabe. Functional analytic methods in partial differential equations, volume 204 of Monographs and Textbooks in Pure and Applied Mathematics. Marcel Dekker, Inc., New York, 1997.
- [141] H. Triebel. Interpolation theory, function spaces, differential operators, volume 18 of North-Holland Mathematical Library. North-Holland Publishing Co., Amsterdam-New York, 1978.
- [142] N.S. Trudinger. Linear elliptic operators with measurable coefficients. *Ann. Scuola Norm. Sup. Pisa* (3), 27:265–308, 1973.
- [143] F.-Y. Wang. On estimation of the logarithmic Sobolev constant and gradient estimates of heat semigroups. *Probab. Theory Related Fields*, 108(1):87–101, 1997.
- [144] F.-Y. Wang. Gradient estimates of Dirichlet heat semigroups and application to isoperimetric inequalities. *Ann. Probab.*, 32(1A):424–440, 2004.

522

- [145] A. Wiweger. A topologisation of Saks spaces. Bull. Acad. Polon. Sci. Cl. III, 5:773–777, 1957.
- [146] J. Wloka. *Partial differential equations*. Cambridge University Press, 1987.
- [147] K. Yosida. Functional Analysis, volume 123 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, Berlin, 1965.
- [148] J. Zabczyk. Parabolic equations on Hilbert spaces. In Stochastic PDE's and Kolmogorov equations in infinite dimensions (Cetraro, 1998), Lecture Notes in Math.
- [149] J. Zabczyk. *Mathematical Control Theory: an introduction*. Birkhäuser Boston, Inc., Boston, 1992.

B(I;Y), 508	$L_{\mathrm{loc}}^{p}(\Omega), 509$
$BUC^k(\mathbb{R}^N), 506$	$R(\lambda)$, 4, 8, 17, 21, 60, 62, 63, 115,
$B_b(\Omega), 509$	162, 346, 441
$C^1_{\nu}(\overline{\Omega}), 506$	$R(\lambda, L_p)$, 157, 162, 176, 180, 184,
$C^{k}(I;Y), 508$	212, 265
$C^k(\Omega,\mathbb{R}^m), 506$	$R(\lambda, \widehat{A}), 85-87, 91, 92, 284, 286,$
$C_b^k(\overline{\Omega}), 505$	291, 346–348, 441–443, 456,
$C_c^k(\Omega), 506$	458, 459
$C^{k}_{\mathrm{loc}}(\Omega,\mathbb{R}^{m}), 506$	$S_{\theta,\omega}, 483$
$C^k_{loc}(\mathbb{R}^N), 505$	$W^{k,p}(\Omega), 509$
$C^{1+\alpha/2,2+\alpha}(I \times D), 508$	$W^{k,p}(\Omega,\mu), 509$
$C_{\text{loc}}^{1+\alpha/2,2+\alpha}(I \times D), 508$ $C_{b,r}^{3\theta,\theta}(\mathbb{R}^N), 507$	$W_{\mathrm{loc}}^{k,p}(\Omega), 509$
$C_{\cdot}^{100}(\mathbb{R}^{N})$ 507	$D_{\max}(\mathcal{A}), 3, 4, 7, 8, 26, 27, 47, 51,$
$C_c^{b,r}(\Omega)$, 506	53-55, 60, 63, 65, 73, 86,
$C^{h,k}(I \times D), 508$	110, 111, 115, 150, 174,
$C_b^{h,k}(I \times D)$, 508	178, 180, 209, 223
$C_b (1 \wedge D), 505$ $C_0(\Omega), 505$	p(t, x; dy), 13-16, 27, 28, 31, 55, 64,
$C_b(\Omega)$, 505 $C_b(\Omega)$, 505	67, 69, 70, 72-76, 80, 84,
$C_b(\mathbb{R}^N), 505$	130, 162, 171, 175, 176,
$C_b(\overline{\Omega}), 505$ $C_b(\overline{\Omega}), 505$	341, 343, 354, 386, 440
$D(L_p)$, 162, 207, 209, 210, 223, 227,	$C_b^{3\theta,\theta}(\mathbb{R}^N), 507$
(28, 234, 292, 360, 404)	$\mathcal{C}^{3\theta,\theta}_{b,r}(\mathbb{R}^N), 507$
$D(\widehat{A}), 26, 27, 47, 55, 56, 73, 86,$	5,7
159, 162, 163, 209, 210,	\widehat{A} (weak generator), 5, 6, 25, 68,
222, 223, 225, 228–230, 291,	115, 159, 209, 210, 283,
346, 348, 441–443, 455, 458	284, 286–291, 346, 348, 386,
$D_A(1+\alpha,\infty),\ 486$	441, 442
$D_A(\alpha+1,\infty)$, 383, 486, 499	abstract Cauchy problem, 486
$D_A(\alpha,\infty)$, 485, 499	accessible point
G(t, x, y), 14, 136, 137	exit, 39, 40, 47, 90
$G_{\Omega}(t,x,y), 497, 498$	regular, 39, 40, 160
K_{λ} , 8	adjoint operator, 443, 468
L(X), 467	analytic semigroup, 18, 250, 256,
L(X,Y), 467	259, 300, 310, 311, 315,
$L^p(\Omega)$, 509	320, 321, 323, 324, 342,
$L^p(\Omega,\mu)$, 509	378–380, 383, 484
	•

of contractions, 387 equivalent processes, 28, 30, 31 Ascoli-Arzelà theorem, 8, 19, 71, ergodic measure, 163, 167, 358 74, 83, 440 exit time, 29, 31, 173 exponential growth bound, 478 Bernstein method, 95, 300, 313, 367, 368, 371, 451 filtration, 30 bounded imaginary powers (\mathcal{BIP}) , Fréchet differentiable function, 168, 259, 263, 281, 481 478 bounded pointwise convergence, 22 fundamental solution, 6, 497 Chacon-Ornstein theorem, 167, 358 generalized eigenfunction, 265, 270, classical solution, 486 271to the Cauchy problem in \mathbb{R}^N , generalized eigenvector, 472, 473 generator of $\{T(t)\}\$ in L^p_μ , 403 5 to the Cauchy-Dirichlet probgenerator of $\{T(t)\}$ in $L^p_\mu(L_p)$, 155, 156, 162, 176, 207, 209, lem in Ω , 305 to the Cauchy-Neumann prob-210, 227, 251, 254, 255, lem in Ω , 329 258, 259, 264, 267, 272, to the degenerate Cauchy prob-273, 360, 404 lem in \mathbb{R}^N , 408 graph of a linear operator, 468 closable operator, 468 Green's function, 4, 8, 9, 14, 161, closed operator, 468 498 closure of a linear operator, 183, 468 Harnack inequality, 15 Hermite polynomials, 276, 278 compact operator, 156, 176, 179, 217, 238, 467, 473 Hille-Yosida theorem, 92, 478 hypercontractivity, 158, 228 compact resolvent operator, 265, 472 compactness of T(t)in $C_b(\mathbb{R}^N)$, 67, 68, 76, 77, 80, improper 83-85 complex integrals, 470 integral, 469 in L_{μ}^{p} , 216, 217, 265 index of an eigenvalue, 269, 472 conservation of probability, 62, 64 infinitesimal generator of a strongly conservative semigroup, 62–64, 69, continuous semigroup, 181, 72, 75, 174, 238 478, 480 core, 162, 180, 181, 184, 185, 209, infinitesimally invariant measure, 160, 211, 228, 254–256, 261, 262, 274, 280, 291, 360, 403, 191 interior L^p estimates, 66, 487 480interior Schauder estimates, 19, 21, dissipative operator, 181, 212, 364, 84, 341, 384, 393 interpolation theorem, 475 479, 485 invariance of $C_0(\mathbb{R}^N)$, 83–85, 91, distance function, 302, 369, 501 eigenvalues, 246, 470 irreducible semigroup, 7, 20, 166, eigenvectors, 470 244, 344, 409

Kalman rank condition, 240 Khas'minskii theorem, 168, 172, 205, 207, 359, 403 Krylov-Bogoliubov theorem, 171 Laplace transform, 54, 92, 479 Liouville theorem, 124, 149, 155,

Laplace transform, 54, 92, 479
Liouville theorem, 124, 149, 155, 168, 265
logarithmic Sobolev inequality (LSI), 157, 158, 221–223, 227–230
and Poincaré inequality, 230
Lumer-Phillips theorem, 294, 320, 321, 380, 479
Lyapunov function, 52, 59, 61, 156, 299, 390

Markov process, 27–30, 173 Markov time, 28, 29 maximum principle, 55, 60, 299–302, 306, 307, 310, 313, 326, 327, 494 mild solution, 117, 459, 486 minimal solution, 5, 8, 13

mixed topology, 22, 25

 $\begin{array}{c} \text{nonhomogeneous Cauchy problem,} \\ 95,\,117,\,409,\,459 \\ \text{norm-continuous semigroup,}\,67,71,\\ 72,\,85,\,177,\,477 \\ \text{normal transition function,}\,13,\,28 \\ \text{numerical range,}\,323,\,380,\,472 \end{array}$

optimal Schauder estimates the elliptic case, 113, 115 the parabolic case, 113, 120 order preserving semigroup, 432, 434, 460

Poincaré inequality, 157, 218–220, 230, 364, 365 and spectral gap, 220, 365 point spectrum, 470 power angle, 259, 263, 281, 481 Prokhorov theorem, 168, 171, 176

reiteration theorem, 451, 475

resolvent identity, 456, 471
resolvent operator, 269, 470
resolvent set, 470
Riemann integral, 468
Riesz-Thorin interpolation theorem,
177, 259, 475

sector of analyticity, 274 sectorial operator, 483 self-adjoint operator, 212, 251, 253, 259, 263, 364, 485 semigroup

of bounded linear operators, 477 of contractions, 113, 284, 432, 437, 440

semigroup of contractions, 155 space of class

 J_{θ} , 330, 451, 474, 475 K_{θ} , 451, 474, 475 space of class J_{α} , 345 spectral angle, 480 spectral bound, 238, 265, 478 spectral gap, 178 in L_{μ}^{2} , 157, 220, 365

 $\begin{array}{c} \text{spectrum, } 470 \\ \text{of } \widehat{A}, \, 243, \, 284, \, 286, \, 289 \\ \text{of } A_p, \, 235, \, 267, \, 281 \\ \text{of } L_p, \, 176, \, 234, \, 264, \, 265, \, 272- \\ 274 \end{array}$

stochastic differential equation, 30 stochastically continuous transition function, 13, 14, 16 strict solution, 486

strong Feller property, 7, 20, 71, 113, 166, 244, 344, 409, 441

strong Markov process, 29 strongly continuous analytic semigroup, 212, 281, 320, 333, 340, 364, 375, 380

strongly continuous semigroup, 18, 68, 91, 155, 179, 183, 184, 234, 235, 246, 262, 271, 273, 274, 279, 280, 283,

320, 364, 380, 398, 477, 480 of contractions, 162, 179, 183, 184, 285, 292, 293, 310, 320, 353, 403, 477, 481 strongly continuous semigroup of contractions, 360 subinvariant measure, 183

tight, 67, 70, 80, 168, 171–176 totally bounded, 83, 216, 217 transference principle, 481 transition family, 67, 69 transition function, 13, 28, 31, 64 transition probabilities, 28 Trotter-Kato theorem, 285, 480

unaccessible point, 48 entrance, 39, 40, 46, 47, 77, 78, 90 natural, 39–41, 46

variation-of-constants formula, 12, 117, 340, 345, 436, 459, 486

Wiener process, 30